

## Studies on Disease Prevention and Control, Decontamination and Sterilization, Microbial Adaptive Responses and Survival, Alternative Therapies, and Sustainability

Submitted by Neil J. Rowan (M.Sc. Ph.D.)

[Volume 2]

September 2024

A Thesis Submitted in Fulfilment of the Requirements for the Higher Degree of Doctor of Science (D.Sc) in the Faculty of Science

### Declaration

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copy right of this thesis belongs to the author under of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgment must always be made to the use of any material contained in, or derived from, this thesis.

Nene Rowen

Signed:

Date: 27<sup>th</sup> September, 2024

#### Abstract

This thesis describes my independent research studies starting in 1996 on five related areas that have advanced disease prevention and control including sustainable technologies to meet significant societal challenges. There are ever increasing demands for specialist foods and sophisticated devices to meet complexities of modern society including serving vulnerable groups. Whilst there is an expanding volume of published literature on developing food production, and to a much lesser degree medical devices, there is a need to understand why traditional and emerging decontamination and sterilization modalities work and what conditions or circumstances operating at the interface between microbial destruction and maintaining a desired product functionality could support microbial survivors and potentially foodborne or iatrogenic-mediated infection. There is also a dearth in knowledge surrounding the real-time detection of viable fastidious pathogenic microorganisms (such as complex parasites or drug-resistant fungi) post selection of appropriate technologies to safely treat foods and to decontaminate complex reusable medical devices. There is also a dearth of published information on appropriate cellular and molecular indicators to inform critical mechanistic information underpinning testing, verification and validation of new decontamination technologies. Elucidating holistically, the key parameters governing reliable and effective decontamination, provides evidence-based data to inform next-generation products from design thinking to automation in order to meet emerging societal needs.

The **first section** provides critical new insights and knowledge on conditions promoting the potential survival of microbial pathogens in sensitive foods such as reconstituted foods destined for vulnerable populations. It describes preparation and storage-abuse conditions promoting adaptive microbial survival and toxin production leading. It describes occurrence of such abuse conditions in hospital-prepared feeds in a HIV ward along with implemented of my recommended solutions that informed new guidelines of practice and helped to mitigate against future food-borne illnesses. This section characterizes processing conditions promoting the occurrence of atypical pathogens in sensitive foods, such as thermal-stresses leading to atypical cellular appearance and virulence factor expression *Listeria monocytogenes* that also enabled survival in human polymorphonuclear leukocytes.

The **second section** elucidates the first reporting on reliable and repeatable operational conditions underpinning non-thermal processing technologies (pulsed UV light, pulsed-plasma gas-discharge, pulsed electric fields), which also encompasses key mechanistic knowledge on critical cellular and molecular determinants governing irreversible microbial cell death. Commensurate studies report on development of alternative biomarkers to monitor and evaluate real-time disinfection performance including first report of a combined cell culture-qPCR assay for complex entero-parasites. New methodologies for toxicological end-point determinations in processing technologies are described.

3

First reporting on microbial kinetic inactivation and modelling for non-thermal technologies. Studies also elucidate appropriate treatment dose for effective killing of biological indicators including development of vaporized hydrogen peroxide as a new thermal sterilization modality for medtech.

The third section elucidates and develops non-thermal decontamination and sterilization technologies at commercial scale for established and new applications including for medical devices, food/feed, and for pollination industry (such as for decontaminating heat-sensitive pollen of complex pathogens fed to bees). Depending on the application, these studies include technologies encompassing x-ray, electron-beam, gamma-irradiation, pulsed UV and the co-development of real-8578/-+parametric release for treated products. Understanding the holistic interplay of all applied and inimical stresses governing effective microbial lethality defines critical knowledge including desirable end-to-end sterility assurance conditions ranging from elucidation to verification and validation of technological applications that meets safety. This section describes first classification system for effective cleaning of complex features in reusable medical devices and revisits efficacy of Spaulding's classification for device sterilisation and patient safety using this combinational new cleaning method. A holistic subject-matter knowledge of decontamination helps society meet unforeseen threats such as my elucidation and the first published recommendation of appropriate sterilization technologies and conditions for the safe reuse of PPE arising from critical supplying chain shortages during COVID-19 pandemic along with for sustainable waste management. Studies advance shellfish depuration and decontamination for recalcitrant fastidious norovirus pathogens attached to bivalve tissue.

The **fourth** section elucidates and develops novel alternative therapies and approaches for combatting antimicrobial-drug resistant (AMR) pathogens including bacteria and fungi, such as for lung delivery and for animal feed applications. Studies also address diagnostics for AMR pathogens that are at crisis point for society linked to decontamination. Studies also address elucidation of alternative antimicrobial and biofilm-disrupting bioactives used synergistically and in combination, yet tolerant of and suitable for medical device production-processes for smart coating applications.

The **last section** describes development of sustainable innovation including use of appropriate disinfection technologies. This includes first studies on development of an integrated multi-trophic aquaculture system in the peatlands aligned with zero-waste, zero pollution and climate action principles. It addresses digital transformation including new 'in-field' real-time monitoring and combined use of bioinformatics and next-generation sequencing to advance sustainable food innovation. It develops and applies new models including life cycle assessment and ecological tools.

4

#### Acknowledgements

Life is a journey that we are told is taking one step after another, that is lonely road. When we have someone to walk with us and support us along the way, that is living. I dedicate this thesis to my parents Ruth and Breffni, to my wife Michelle, to my daughter Chloe and to my sons Kevin and Liam for their endless love, support and encouragement! This thesis has been a psychological journey that captured precious time that was not shared enough with them. I apologise to them for when I have been distant and not focusing on their needs. I discussed this higher DSc submission with my father Breffni sadly before he passed way in 2014; it's a real relief (catharsis) that have I eventually managed to address it.

I was the first of many generations to have attended University. I feel this background has positively influenced my steadfast commitment to pursuit of research excellence and has helped with my engagement with students who attend university from all walks of life. I am grateful to the University of Galway for awarding me a professional soccer scholarship during my undergraduate years as a microbiology student. I am grateful to the late Professor John E Smith offering me an AFRC educational scholarship to pursue postgraduate studies at Strathclyde University. Thank you to Prof John G. Anderson and Prof Scott MacGregor for their friendship and encouragement at Strathclyde University as I forged an independent research pathway. I was pleased that my grading of 5\* during the UK Research Assessment Exercise of 2023 reflected my commitment to this pursuit of academic excellence and to growing research and innovation at the University of Strathclyde.

To my friends at Dublin Institute of Technology, thank you for keeping me sane given that I commuted ca. 6 hours every day for 3 years to lecture there; thus, affording our children the opportunity of attend school locally in Athlone. Thank you to the School of Medicine at the University of Galway for appointing me as an Adjunct Professor. Also, to the University of KwaZulu Natal, and the University of Malaysia for Honorary Professorship appointments in Biotechnology. To the University of Stanford for recognising my contributions to these fields of study as reflected in my appearance on 2023 list of top 2% scientists globally. To the Technological University of Shannon (TUS), for bestowing the inaugural 'Outstanding Contribution to Research' award earlier this year. I have spent the past 20 years working at TUS here in Ireland (formerly Athlone Institute of Technology (AIT)) that was predominantly teaching-led. I thank TUS for my appointment as the first Director of the Bioscience Research Institute since 2012 and for its trust in me to inform process changes to grow research scale and capacity that contributed to it becoming a new university in October 2020.

#### **Table of Content**

#### Section 4: Complementary and Alternative Therapies and Approaches

Masterson, C.H., Murphy, E.J., Gonzalez, H., Major, I., McCarthy, S.D., O'Toole, D., 11 Laffey, J.393G. and Rowan, N.J (2020). Purified  $\beta$ -glucans from the Shiitake mushroom ameliorates a405ntibiotic-resistant *Klebsiella pneumoniae*-induced pulmonary sepsis. *Letters in Applie424d Microbiology* **71** (4), 405-412.

Murphy, E.J., Masterons, C., Rezoagli, E., O'Toole, D., Major, I., Stack, D., Lynch, M., Laffey, J.G. and Rowan, N.J (2020). β-Glucan extracts from the same edible shiitake mushroom *Lentinus edodes* produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects — Implications for coronavirus disease (COVID-19) immunotherapies. *Science of the Total Environment*, **732**, https://doi.org/10.1016/j.scitotenv.2020.139330

Murphy, E.J., Rezoagli, E, Pogue, R., Simonassi-Paiva, B., Aldin, I., Fehrenbach, G.W., O'Niell, E., Major, I., Laffey, J.G. and Rowan, N. (2022). Immunomodulatory activity of β-glucan polysaccharides isolated from different species of mushroom – A potential treatment for inflammatory lung conditions. *Science of the Total Environment*, **809**, https://doi.org/10.1016/j.scitotenv.2021.152177

Murphy, E.J., Rezoagli, E., Major, I., Rowan, N.J., Laffey, J.G (2020).  $\beta$ -Glucan 42 Metabolic and Immunomodulatory Properties and Potential for Clinical Application. *Journal of Fungi*, **6** (4), <u>https://doi.org/10.3390/jof6040356</u>

Galanakis, C.M., Aldawoud, T.M., Rizou, M., Rowan, N.J. and Ibrahim, S.A. (2020).59Food Ingredients and Active Compounds against the Coronavirus Disease (COVID-19)9Pandemic:AComprehensiveReview.Foods,9(11),https://doi.org/10.3390/foods91117011

Wan Mohtar, W.A.A.Q.I, Ilham, Z., Jamaludin, A.A and Rowan, N.J (2021). Use of Zebrafish Embryo Assay to Evaluate Toxicity and Safety of Bioreactor-Grown Exopolysaccharides and Endopolysaccharides from European *Ganoderma applanatum* Mycelium for Future Aquaculture Applications. *International Journal of Molecular Sciences*, **22** (4), <u>https://doi.org/10.3390/ijms22041675</u>

Wan Mohtar, W.A.A.Q.I., Fafek, N.A., Thiran, J.P., Rahman, J.F.P., Yerima, G., Subramaniam, K. and Rowan, N.J. (2021). Investigations on the use of exopolysaccharide derived from mycelial extract of Ganoderma lucidum as functional feed ingredient for aquaculture-farmed red hybrid Tilapia (Oreochromis sp.). *Future Foods*, **3**, https://doi.org/10.1016/j.fufo.2021.100018

Usuldin, S.R.A., Wan-Mohtar, W.A.A.Q.I., IIham, Z., Abdullah, N.R. and Rowan, N. 98 (2021). *In vivo* toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications. *Nature Scientific Reports*, **11**, <u>https://doi.org/10.1038/s41598-021-02486-7</u>

Masterson, K., Meade, E., Garvey, M., Lynch, M, Major, I and Rowan, N.J (2021). 111 Development of a low-temperature extrusion process for production of GRAS bioactivepolymer loaded compounds for targeting antimicrobial-resistant (AMR) bacteria. *Science of the Total Environment*, **80**0, https://doi.org/10.1016/j.scitotenv.2021.149545

10

Masterson, K., Major, I., Lynch, M. and Rowan, N.J. (2023). Synergy Assessment of Four 120 Antimicrobial Bioactive Compounds for the Combinational Treatment of Bacterial Pathogens. *Biomedicines*, **1**1 (8), https://doi.org/10.3390/biomedicines11082216

Rowan, N.J., Meade, E., Garvey, M. (2021). Efficacy of frontline chemical biocides and disinfection approaches for inactivating SARS-CoV-2 variants of concern that cause coronavirus disease with the emergence of opportunities for green eco-solutions. *Current Opinion in Environmental Science and Health*, **23**, https://doi.org/10.1016/j.coesh.2021.100290

Meade, E., Hehir, S., Rowan, N.J. and Garvey, M. (2022). Mycotherapy: Potential of Fungal Bioactives for the Treatment of Mental Health Disorders and Morbidities of Chronic Pain. *Journal of Fungi*, **8** (3), https://doi.org/10.3390/jof8030290

#### Section 5: Elucidating and Developing New Sustainable Technologies and 179 Frameworks

Naughton, S., Kavanagh, S., Lynch, M. and Rowan, N.J. (2020). Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: A case study from the Republic of Ireland. *Aquaculture*, **526**, https://doi.org/10.1016/j.aquaculture.2020.735377

Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E. and Rowan, N.J. (2018) 194 Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. *PeerJ 6 Aquatic Biology*, **e5281**, https://doi.org/10.7717/peerj.5281

Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E. and Rowan, N.J. (2018). Full water quality monitoring of a traditional flow-through rainbow trout farm. *Fishes* **3** (3),https://doi.org/10.3390/fishes3030028

Rowan, N.J., Fort, A., O'Neill, E.A., Clifford, E., Jansen, M., Helfert, M., Toner, D., Maguire, J. and Tiwari, B. (2024). Development of a novel recirculatory multitrophic peatland system for the production of high-value bio-based products at scale embracing zero waste and pollution principles to unlock sustainable development goals. *Case Studies in* <u>Chemical</u> and <u>Environmental</u> <u>Engineering</u>, **9**, https://doi.org/10.1016/j.cscee.2024.100763

O'Neill, E.A. and Rowan, N.J. (2022). Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance – A comparative case study from the Republic of Ireland. *Science of the Total Environment*, **802**, https://doi.org/10.1016/j.scitotenv.2021.149800

O'Neill, E.A., Morse, A.P. and Rowan, N.J. (2022). Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: Implications and opportunities for advancing research and disruptive innovation post COVID-19 era. *Science of the Total Environment*, **819**, https://doi.org/10.1016/j.scitotenv.2022.153073

O'Neill, E.A., Rowan, N.J. and Fogarty, A.M. (2019). Novel use of the alga 264 *Pseudokirchneriella subcapitata*, as an early-warning indicator to identify climate change

ambiguity in aquatic environments using freshwater finfish as a case study. *Science of the Total Environment*, 692, 209-218.

O'Neill, E.A., Fehrenbach, G., Murphy, E., Alencar, S.A., Pogue, R. and Rowan, N.J. (2022). Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: Case study from the Republic of Ireland. Science of the Total Environment, 851 (2), <u>https://doi.org/10.1016/j.scitotenv.2022.158392</u>

O'Neill, E.A. and Rowan, N.J. (2023). Potential disruptive effects of zoosporic parasites 284 on peatland-based organic freshwater aquaculture: Case study from the Republic of Ireland. Science of the Total Environment, 868, https://doi.org/10.1016/j.scitotenv.2023.161495

Rowan, N.J., Murray, N., Qiao, Y., O'Neill, E., Clifford, E., Barcelo, D. and Power, D.M. (2022). Digital transformation of peatland eco-innovations ('Paludiculture'): Enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products and services. *Science of the Total Environment*, **838** (3), https://doi.org/10.1016/j.scitotenv.2022.156328

Galanakis, C.M., Rizou, M., Aldawoud, T.M.S., Ucak. I. and Rowan, N.J. (2021). 313 Innovations and technology disruptions in the food sector within the COVID-19 pandemic and post-lockdown era. Trends in Food Science and Technology, 110, 193-200

Fehrenbach, G.W., Murphy, E., Pogue, R., Carter, F., Clifford, E., Major, I. and Rowan, N. (2023). Pulsed ultraviolet disinfection of artificially contaminated seawater seeded with high levels of pathogen disease indicators as an alternative for the shellfish industry depuration systems. *Environmental Science and Pollution Research*, **30**, 70771-70782.

Fehrenbach, G.W., Pogue, R., Carter, F., Clifford, E. and Rowan, N.J. (2022). Implications 333 for the seafood industry, consumers and the environment arising from contamination of shellfish with pharmaceuticals, plastics and potentially toxic elements: A case study from Irish waters with a global orientation. *Science of the Total Environment*, **844**, https://doi.org/10.1016/j.scitotenv.2022.157067

Finnegan, W., Clifford, E., Goggins, J., O'Leary N., Dobson, A., Rowan, N., Xiao, L., 347 Miao, S., Fitzhenry, K., Leonard, P., Tarpey, E., Gil-Pulido, B., Gao, F. and Zhan, X. (2018). Dairywater – striving for sustainability within the diary processing industry in the Republic of Ireland. *Journal of Diary Research*, **85** (3), 366-374.

Tiedeken, E.J., Tahar, A., McHugh, B. and Rowan, N.J. (2017). Monitoring, sources, 356 receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – A 20 year systematic review. *Science of the Total Environment*, **574**, 1140-1163.

Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E. and Rowan N.J. (2017). Development380of a semi-quantitative risk assessment model for evaluating environmental threat posed by380the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: An380Irish case study. Science of The Total Environment, 603-604-, 627-638.380

Tahar, A., Tiedeken, E.J. and Rowan, N.J. (2018). Occurrence and geodatabase mapping 392 of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – A first overview of the situation in the Republic of Ireland. *Science of the Total Environment*, **616-617**-, 187-197.

Rowan, N.J. (2023). Current decontamination challenges and potentially complementary solutions to safeguard the vulnerable seafood industry from recalcitrant human norovirus in live shellfish: Quo Vadis? *Science of the Total Environment*, **874**, https://doi.org/10.1016/j.scitotenv.2023.162380

Ruiz-Salmon, I., Laso, J., Margallo, M., Villanaueva-Rey, P., Rodriguez, E., Quinteiro, 423 P., Claudia Dias, A., Almeida, C., Clifford, E., Reguiero, L., Mendez, D., Anglada, C., Noirot, C., Rowan, N.J., Vazquez-Rowe, I. and Aldaco, R. (2021). Life cycle assessment of fish and seafood processed products – A review of methodologies and new challenges. *Science of the Total Environment*, **761**, https://doi.org/10.1016/j.scitotenv.2020.144094

Almeida, C., Laso, J., Baptista de Sousa, Cooney, R., Quinterio, P., Rowan, N.J., Dias, A.C., Clifford E., Reboredo, R.G., Margallo, M., Nunes, M.L. and Marques, A. (2023). Seafood consumers engagement in reducing environmental impacts from packaging. *Science of the Total Environment*, **863**, <u>https://doi.org/10.1016/j.scitotenv.2022.160846</u>

Rowan, N.,J. and Casey, O. (2021). 'Empower Eco' multi-actor HUB: A triple helix 458 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. *Current Opinion in Environmental Science and Health*, **21**, https://doi.org/10.1016/j.coesh.2021.100254

Cooney, R., Baptista de Sousa, D., Fernandez Rios, Mellett, S., Rowan, N., Morse, A.P., 470 Hayes, M., Laso, J., Regueiro, L., Wan, A., Clifford, E. (2023). A circular economy framework for seafood waste valorisation to meet challenges and opportunities for intensive production and sustainability. *Journal of Cleaner Production*, **392**, https://doi.org/10.1016/j.jclepro.2023.136283

Rowan, N.J. and Galanakis, C.M. (2020). Unlocking challenges and opportunities 500 presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? *Science of the Total Environment*. **748**, https://doi.org/10.1016/j.scitotenv.2020.141362

Rowan, N.J. (2023). Challenges and future opportunities to unlock the critical supply 500 chain of personal and protective equipment (PPE) encompassing decontamination and reuse under emergency use authorization (EUA) conditions during the COVID-19 pandemic: Through a reflective circularity and sustainability lens. *Science of the Total Environment*, **866**, https://doi.org/10.1016/j.scitotenv.2023.16145

#### Appendix 1. Full list of publications including uncited material in this thesis512

#### Appendix 2. Supporting letters confirming lead authorship on cited papers 523

#### Section Four - Complementary and Alternative Therapies and Approaches'

This explored in part a collaboration with Prof John Laffey in Clinical Anaesthesia and Intensive Care Medicine at the University College Galway. Studies describe the first isolation, purification and extraction of sterile beta glucan from shiitake mushroom that was used to combat antibiotic resistant Klebsiella pneumoniae using a lung model. An understanding of appropriate novel bioactives that could prime immune system to combat antimicrobial resistant bacteria (AMR) that could potentially survive novel device processing would be deemed important from a preventive disease perspective. Studies based on isolation of beta-glucans from other medicinal fungi also revealed positive immunological markers for lung health application. There is a growing volume of literature on this beta-glucan topic where our independent studies revealed shortcomings with bioactive extraction, purification and dose internationally that has stalled production of a safe product for clinical use. Studies using shiitake mushroom-derived beta-glucans provided some insights for therapies to combat complex lung health conditions and had relevance for SARS-COV-2 virus. The use of medical fungi and extracts as immune-stimulants did not produce undesirable toxicological end-points when studied in fisheries including zebra-fish. Bio-based products (such as Nisin, zince oxide, silver nitrate and chitosan) were studied using a new mild-temperature extrusion process for inhibiting or inactivating AMR bacterial pathogens and biofilm development (with and without non-processing) such as for medical device coating research and applications. This can inform future use of appropriate Green bioactive alternatives and linked smart polymer research that is described in this thesis. The role of alternative biocides for addressing complex pathogens such as emergence of drug-resistant fungi is also described.



#### ORIGINAL ARTICLE

## Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant *Klebsiella pneumoniae*-induced pulmonary sepsis

C.H. Masterson<sup>1</sup> (b), E.J. Murphy<sup>2</sup>, H. Gonzalez<sup>1</sup>, I. Major<sup>3</sup>, S.D. McCarthy<sup>1</sup>, D. O'Toole<sup>1</sup>, J.G. Laffey<sup>1,4</sup> and N.J. Rowan<sup>5</sup> (b)

- 1 Lung Biology Group, National University of Ireland, Galway, Ireland
- 2 Bioscience Research Institute, Athlone Institute of Technology, Athlone, Ireland
- 3 Materials Research Institute, Athlone Institute of Technology, Athlone, Ireland
- 4 Anesthesia and Intensive Care Medicine, University Hospital Galway, Galway, Ireland
- 5 Centre for Disinfection, Sterilisation and Biosecurity, Athlone Institute of Technology, Athlone, Ireland

Significance and Impact of the Study: Bacterial infection remains the main cause of acute respiratory distress syndrome and the need for an effective therapeutic intervention is evident. This study has demonstrated for the first time that purified  $\beta$ -glucans from the Shiitake mushroom *Lentinus edodes* can be used to attenuate the injury resulting from antibiotic-resistant *Klebsiella pneumoniae* pulmonary infection. Intravenous administration of  $\beta$ -glucan shows potential for treating sepsis-induced lung injury as it effectively reduces bacterial load, inflammatory white cell influx, protein leakage to the lungs and improves lung physiological parameters. Use of lentinan shows promise as potential therapeutic intervention to combat bacterial pulmonary sepsis.

#### Keywords

beta glucans, *Klebsiella pneumoniae*, lung injury, medicinal mushrooms, sepsis.

#### Correspondence

Claire Masterson, Lung Biology Group, Biosciences Research Building, National University of Ireland, Galway, Ireland. E-mail: Claire.mast@gmail.com

2020/0583: received 1 April 2020, revised 26 June 2020 and accepted 16 July 2020

doi:10.1111/lam.13358

#### Abstract

Bacterial infection remains the main cause of acute respiratory distress syndrome and is a leading cause of death and disability in critically ill patients. Here we report on the use of purified  $\beta$ -glucan (lentinan) extracts from Lentinus edodes (Shiitake) mushroom that can reduce infection by a multidrugresistant clinical isolate of Klebsiella pneumoniae in a rodent pneumonia model, likely through immunomodulation. Adult male Sprague-Dawley rats were subjected to intra-tracheal administration of K. pneumoniae to induce pulmonary sepsis and randomized to three groups; vehicle control (Vehicle, n = 12), commercial lentinan (CL, n = 8) or in-house extracted lentinan (IHL, n = 8) were administered intravenously 1 h postinfection. Physiological parameters and blood gas analysis were measured, bacterial counts from bronchoalveolar-lavage (BAL) were determined, along with differential staining of white cells and measurement of protein concentration in BAL 48 h after pneumonia induction. Use of IHL extract significantly decreased BAL CFU counts. Both CL and IHL extractions reduced protein concentration in BAL. Use of IHL resulted in an improvement in physiological parameters compared to controls and CL. In conclusion, administration of lentinan to treat sepsisinduced lung injury appears safe and effective and may exert its effects in an immunomodulatory manner.

#### Introduction

Acute respiratory distress syndrome (ARDS) is a leading cause of death and disability in critically ill patients (Rubenfeld *et al.* 2005) resulting from a loss of respiratory

system compliance and hypoxia due to alveolar–capillary barrier disruption (Ferguson *et al.* 2005) which can lead to multiple organ failure and death (Ware and Matthay 2000). Although ARDS can arise from many causes, sepsisassociated ARDS is the most common type (Bellani *et al.*  2016) with approximately 40% of patients with severe sepsis, and 23% of mechanically ventilated patients developing ARDS (Bellani et al. 2016, McNicholas et al. 2018). The inflammatory response associated with the presence of a microbial infection involves the recruitment of inflammatory white cells and an over-production of inflammatory cytokines leading to alveolar-capillary barrier damage and pulmonary oedema instigating further infiltration, inflammation and widespread damage (Grommes and Soehnlein 2011). Interventions to combat the progression of ARDS to date have been mostly unsuccessful (Laffey and Kavanagh 2018; Lewis et al. 2019). Thus far the most promising therapy for the treatment of ARDS has been the implementation of supportive therapies such as fluid management, ventilatory strategies and prone positioning to improve oxygenation (Kon et al. 2015). The collective failure of pharmacologic therapies shows a clear need for the introduction of novel treatments for sepsis and associated ARDS. Targeting the immune response as a method of controlling the inflammatory response would serve as an attractive option in combating the initial infection and detrimental outcome of an overzealous inflammatory milieu in sepsis-induced ARDS.

β-glucans are polysaccharides naturally found in the cell walls of plants, bacteria, fungi, yeast and algae (Sullivan et al. 2006). They are comprised of chains of D-glucose rings connected commonly through a 1-3 glycosidic bond. Depending on source, β-glucans vary in chain length and level of side branching. Mushroom β-glucans, including lentinan extracts from Lentinus edodes (Shiitake), consist of a 1,3 linked back bone with 1,6 linked side branches (Wan-Mohtar et al. 2016). The molecular structure of different β-glucans confers different properties including solubility and receptor binding which is of interest in relation to leukocyte activation. For example, β-glucans with poor solubility confer direct leukocyte activation due to clustering of the receptor binding site (Sahasrabudhe et al. 2016), whereas modified soluble forms of β-glucan enhanced leukocyte activity without complete activation (Poutsiaka et al. 1993). Dectin-1 is a β-glucan-specific receptor (Herre et al. 2004) found predominantly on macrophages and neutrophils as well as dendritic cells and T-Lymphocytes (Brown et al. 2003; Taylor et al. 2002) and functions as a receptor specific for  $\beta$ -glucans with 1,3 and/or 1,6 linkages (Willment *et al.* 2003). It has been known for quite some time that  $\beta$ -glucans are the effective components in fungal products, and mushroom extracts have been used as medicine for thousands of years (Chang 2002; Sullivan et al. 2006).

Recently, there has been an increasing interest in the use of  $\beta$ -glucans from natural sources for treating infection in animals and humans (Carballo *et al.* 2019), including use of Shiitake mushrooms (Zhang *et al.* 2019).

Carballo et al. (2019) exploited use of  $\beta$ -glucans from yeast in feed to modulate immune responses to control Vibrio genus-related infection in farmed fish. McCarty et al. recently described the potential role of  $\beta$ -glucan as a natural nutraceutical for boosting type 1 interferon response to RNA viruses such as influenza and coronavirus (Mccarty and Dinicolantonio 2020). Sepsis, influenza and coronavirus cause an inflammatory storm in the lungs and it is this inflammatory storm that leads to respiratory distress, organ failure and death (Shi et al. 2020). There is emerging evidence to suggest that certain nutraceuticals, such as β-glucans, may help reduce inflammation in the lungs with RNA viruses and other problematic bacterial pathogens (Vetvicka and Vetvickova 2015). β-glucans may also help boost type 1 interferon response to these pathogens, which is the body's primary means of help to create antimicrobial antibodies to fight off these infections (Mccarty and Dinicolantonio 2020). Here we aimed to investigate the potential for our inhouse purified lentinan β-glucan preparation (compared to commercially produced lentinan β-glucan) in combating antibiotic-resistant Klebsiella pneumoniae pulmonary infection.

#### **Results and discussion**

# The characteristics of $\beta$ -glucan compounds varies depending on preparation

Immunomodulatory polysaccharides, such as β-glucans, are nontoxic and do not result in side effects commonly seen with the use of bacterial or synthetic compounds (Rice et al. 2004) making them an attractive adjunct therapeutic strategy in critically ill or high-risk patients. Megazyme analysis of commercial lentinan (CL) and IHL samples revealed differences in both  $\beta$  and  $\alpha$ -glucan % w/ w content. CL was shown to have a significantly higher  $\alpha$ glucan content at  $16 \pm 1.5\%$  compared with only  $2 \pm 0.5\%$  for IHL extract. However, IHL had a significantly higher  $\beta$ -glucan content at 76  $\pm$  3.5% compared with  $48 \pm 3.0$  for CL samples (Fig. S1). Further studies by our group using the same samples revealed differences between specimens including variances in particle size and particle dispersity and elemental content, but also that each sample contained the same  $\beta$ -glucan compound (Murphy et al. 2020). This corroborates findings of Zhang et al. where similar medicinal fungi may exhibit significant differences in the molecular mass range profile and chain conformation of the beta-glucan samples. A study conducted by Vetvicka and Vetvickova (2018) suggested that the use of a highly purified yeast-derived β-glucan was superior to crudely isolated, less active  $\beta$ -glucans from various sources. In rodent models of septic shock and as we have shown here, the IHL which contains a purer composition of β-glucan out-performs the commercially sourced CL. β-glucans at concentrations used in this study (20 mg kg<sup>-1</sup>) were based on the maximum dose of CL that could be administered due to the viscosity of the solution, and did not affect viability or proliferation of K. pneumoniae bacteria (data not shown). This suggests an indirect effect of the compounds as described previously (Vetvicka and Vetvickova 2015) whereby β-glucans act upon the immune system to instigate clearance of infection and are not directly antibacterial themselves. Recent linked in vitro studies using same β-glucan extracts revealed that both IHL and CL demonstrated low to minimal toxicity in pulmonary cell lines. We observed that IHL and CL had both inflammatory and anti-inflammatory properties, with IHL showing a greater immunemodulating profile compared to CL supporting the immunomodulation potential of this novel B-glucan extract (Murphy et al. 2020).

## B-glucan can effectively attenuate bacterial counts and white cell infiltrates to the lungs during pneumonia

Whilst direct antibacterial properties of CL and IHL could not be demonstrated, here we have shown that the numbers of viable bacteria present in the BAL fluid of treated animals was significantly reduced. The administration of 20 mg kg<sup>-1</sup>  $\beta$ -glucan was significantly effective at

lowering BAL WCC compared to vehicle (Fig. 1, panel b, P < 0.05), however, only the IHL was shown to significantly reduce the numbers of viable K. pneumoniae in the BAL fluid (Fig. 1, panel a, P < 0.05). Differential counts of the total white cells in the BAL fluid revealed that both CL and IHL could reduce the numbers of inflammatory PMNs infiltrating to the lung (Fig. 1, panel c, P < 0.01) but only IHL increased the numbers of monocytes/macrophages in the BAL fluid (Fig. 1, panel d, P < 0.05). It has been shown in several studies that β-glucans exert a potent effect on cells of the innate immune system which would explain the reduction in bacterial numbers and the differences in innate immune cell populations as shown here. B-glucans are not produced by humans or animals and are therefore classed as 'non-self' upon introduction to the body activating innate and adaptive immune responses (Brown and Gordon 2005). B-glucans can bind to and activate circulating monocytes increasing their cytotoxic activity, phagocytic activity, reactive oxygen species and nitric oxide production and cytokine secretion (Chan et al. 2009). In mouse models of influenza yeast  $\beta$ glucan has been shown to increase dendritic cell activation conferring protection from infection (Camilli et al. 2018, Vetvicka and Vetvickova). In humans, administration of  $\beta$ -glucan has been previously shown to increase blood monocyte proliferation and maturation of dendritic cells (Chan et al. 2007), whereas we have demonstrated an increase in monocyte numbers in the BAL fluid of



**Figure 1** The effects of IHL and CL on the bacterial counts and white cell infiltrates to the lungs during pneumonia. Rodent models of *Klebsiella pneumoniae* infection were treated intravenously with 20 mg kg<sup>-1</sup> of IHL or CL β-glucan. Bacterial and white cell counts were performed on the BAL fluid (panels a & b). Differential staining of the BAL white cells allowed numeration of the PMN (panel c) and Mo fractions (panel d). Data are expressed as mean  $\pm$  SD (n = 6-10/group). \*P < 0.05, \*\*P < 0.01 (one-way ANOVA with Newman–Keuls multiple comparison test) with respect to Vehicle group. IHL = In-house lentinan; CL = Commercial lentinan; CFU = colony forming unit; BAL = bronchoalveolar lavage; PMN = Polymorphonuclear neutrophils; Mo = Monocytes.

Letters in Applied Microbiology 71, 405-412 © 2020 The Society for Applied Microbiology

treated animals whilst also demonstrating a decrease in the inflammatory PMN fraction.

## B-glucans improve lung function and levels of circulating metabolites

Administration of IHL significantly improved arterial oxygenation (P < 0.05) during pneumonia compared to vehicle controls. CL also improved arterial oxygen levels, albeit not significantly (Fig. 2, panel a). Static lung compliance measurements revealed that lung structural indices were also improved. Blood gas analyses of circulating metabolites revealed a significant decrease in plasma lactate levels after treatment with IHL  $\beta$ -glucans (Fig. 2, panel b, P < 0.05). The administration of IHL  $\beta$ -glucan also returned blood glucose levels back to normal compared to vehicle and sham (Fig. 2, panel c).

While both CL and IHL increased compliance towards normal values, only IHL reached significance (Fig. 3, panel a, P < 0.05). The wet:dry ratio, an indicator of lung leak and tissue oedema, was positively decreased by IHL compared to vehicle control (Fig. 3, panel b, P < 0.001) and arterial–alveolar gradient values trended towards a decrease in the IHL-treated group (Fig. 3, panel c).

## B-glucans enhance the inflammatory response to infection

Cytokine measurements in the BAL of animal models revealed that the treatment of *K. pneumonia* using  $\beta$ -glucans seemed to enhance the inflammatory response to infection by significantly decreasing the levels of anti-inflammatory IL-10 (P < 0.01) and modestly increasing levels of TNF- $\alpha$  (P = 0.108) compared to vehicle control (Fig. 4, panels a & b). Measurement of BAL protein levels indicated that treatment with  $\beta$ -glucan significantly decreased the levels of protein representing a decrease in

alveolar–capillary leakage (Fig. 4, panel c, P < 0.05) and supports the finding that IHL could improve lung structural indices as shown by measurements in static lung compliance and wet:dry ratio (Fig. 3, panels a,b). A study has demonstrated that the prophylactic administration of  $\beta$ -glucan can confer a protective effect, lowering TNF- $\alpha$ production and enhancing endotoxin clearance (Vereschagin *et al.* 1998). Here however, we have demonstrated a very modest increase in inflammatory TNF- $\alpha$  accompanied by a significant decrease in anti-inflammatory IL-10. This may be explained by the prophylactic administration used in previous studies which may have resulted in a state of 'trained immunity' (Quintin *et al.* 2012). In contrast,  $\beta$ -glucan in the present study was administered 1h postbacterial inoculation.

There is strong potential to produce commercially scalable production of  $\beta$ -glucans from Shiitake mushroom mycelium using controlled bioreactor studies for global research and innovation (Wan-Mohtar *et al.* 2016). With more research such as the present study and others (Vetvicka and Vetvickova 2018) showing that the production process can significantly impact the activity and efficacy of the extracted  $\beta$ -glucan it is imperative that it is given due attention. Here we have shown that the isolation of a pure  $\beta$ -glucan product from the Shiitake mushroom is effective in attenuating the injury parameters associated with bacterial pneumonia and propose the use of naturally derived immunomodulators to be a novel strategy in combating ARDS.

#### Materials and methods

#### Klebsiella pneumoniae-Induced Lung Injury

All work was approved by the Animal Care Research Ethics Committee of the National University of Ireland, Galway and conducted under license from the Health



**Figure 2** The effects of IHL and CL on blood gas indices during pneumonia. Rodent models of *Klebsiella pneumoniae* infection were treated intravenously with 20 mg kg<sup>-1</sup> of IHL or CL  $\beta$ -glucan. Samples of arterial blood were analysed using a blood gas analyser during mechanical ventilation. Arterial oxygenation (panel a), and plasma lactate and glucose readings were recorded (panels b & c). Data are expressed as mean  $\pm$  SD (n = 6-10/group). \*P < 0.05, NS = not significant (one-way ANOVA with Newman–Keuls multiple comparison test) with respect to Vehicle group and/or Sham group. IHL = In-house lentinan; CL = Commercial lentinan; pO<sub>2</sub> = partial pressure of oxygen.

Letters in Applied Microbiology 71, 405-412 © 2020 The Society for Applied Microbiology



**Figure 3** The effects of IHL and CL on physiological parameters during pneumonia. Rodent models of *Klebsiella pneumoniae* infection were treated intravenously with 20 mg kg<sup>-1</sup> of IHL or CL  $\beta$ -glucan. Static compliance was measured under anaesthesia (panel a), and samples of lung tissue were dried postmortem for analysis of wet : dry ratios (panel b). Samples of arterial blood were analysed using a blood gas analyser during mechanical ventilation at FiO<sub>2</sub> 1·0 and AaDO<sub>2</sub> was calculated (panel c). Data are expressed as mean  $\pm$  SD (n = 6-10/group). \*P < 0.05, \*\*\*P < 0.001, (one-way ANOVA with Newman–Keuls multiple comparison test) with respect to Vehicle group. IHL = In-house lentinan; CL = Commercial lentinan; AaDO<sub>2</sub> = arterial alveolar oxygen gradient.



**Figure 4** The effects of IHL and CL on cytokine profiles during pneumonia. Rodent models of *Klebsiella pneumoniae* infection were treated intravenously with 20 mg kg<sup>-1</sup> of IHL or CL  $\beta$ -glucan. Postmortem BAL samples were analysed for anti-inflammatory IL-10 (panel a) and inflammatory TNF- $\alpha$  (panel b) and protein concentration was assessed using a protein assay of the BAL fluid (panel c). Data are expressed as mean  $\pm$  SD (n = 6-10/group). \*P < 0.05, \*\*P < 0.01, (one-way ANOVA with Newman–Keuls multiple comparison test) with respect to Vehicle group. IHL = Inhouse lentinan; CL = commercial lentinan; BAL = bronchoalveolar lavage; IL-10 = interleukin 10; TNF- $\alpha$  = tumor necrosis factor alpha.

Products Regulatory Authority, Ireland (Licence number AE19125/P053, Number of animals approved on licence —246; Animals enrolled in this series—28). Specific-pathogen-free adult male Sprague–Dawley (CD) rats (Charles River Laboratories, Kent, UK) weighing between 300 and 450 g were used in all experiments. These animals were chosen based on the successful establishment of bacterial pneumonia using intratracheal *Escherichia coli* by our research group which was used in several studies (O'croinin *et al.* 2005; Chonghaile *et al.* 2008; Devaney *et al.* 2015; Masterson *et al.* 2018). The use of rats allows for adequate sample sizes of arterial blood, bronchoalveolar lavage (BAL) fluid and tissues for analysis.

In all groups, animals were anaesthetised using vaporized isoflurane (Iso-Vet, Chanelle, Co. Galway, Ireland). The animals were orotracheally intubated under direct vision using a guide wire and a 14G catheter (BD Insyte®; BD Biosciences, Berkshire, UK). A bolus of  $1 \times 10^9$  CFU of *K. pneumoniae* in a 300-µl PBS suspension was instilled followed by a bolus of air. The animals were allowed to recover from anaesthesia as previously described (Curley *et al.* 2017; Devaney *et al.* 2015; Devaney *et al.* 2013; O'Croinin *et al.* 2005; O'Croinin *et al.* 2008) before proceeding to treatment. Piloting results for the *K. pneumoniae* pneumonia model are included in the supplementary data (Fig. S2).

#### Clinical isolate

*Klebsiella pneumoniae* clinical isolate was provided by Tullamore General Hospital (Health Service Executive, Co. Offaly, Ireland) as a stab culture. Using a Clinical and Laboratory Standards Institute (CLSI)-compliant Vitek 2 automated analyser to obtain antibiotic sensitivities, this organism was determined to be resistant to gentamicin and aztreonam and therefore categorized as a multidrugresistant extended-spectrum beta-lactamase as per European Antimicrobial Resistance Surveillance Network guidelines 2017 (Giske *et al.* 2017). Of note, the patient from whom the isolate was derived was successfully treated using meropenem following the failure of broadspectrum antibiotics. The sample was isolated from blood but originated from the urine via a urinary tract infection complication.

*Klebsiella pneumoniae* cultures were propagated using tryptone soya broth (Fisher Scientific Ireland) and colonies were identified using UTI Brilliance Clarity Agar (LIP Fannin, Galway, Ireland). Colony forming units were quantified using a combination of serial dilution cultures and optical density readings at 600 nm. Cultures were prepared for animal models of pneumonia at  $1 \times 10^9$  CFU per 300 µl in sterile PBS (Sigma Aldrich, Wicklow, Ireland).

#### Beta glucan isolation and preparation

Commercial lentinan was sourced from Carbosynth (FL33321, Berkshire, UK) and resuspended in sterile H<sub>2</sub>O (Sigma Aldrich). In-house produced lentinan (IHL) was extracted from the fruiting bodies of L. edodes which were purchased from Fancy Fungi Ltd (Co. Wexford, Ireland). Lentinan extracts were analysed for (1,3)-(1,6)-beta-glucan content using the Megazyme yeast and mushroom kit (K-YBGL; Megazyme Ltd Co. Wicklow, Ireland). Assays were carried out according to the manufacturer's instructions. Briefly, samples were milled, and placed in 12 mol  $l^{-1}$  H<sub>2</sub>SO<sub>4</sub> at -4°C for 2 h to solubilize the glucans. The samples were then hydrolyzed in 2 mol  $l^{-1}$ H<sub>2</sub>SO<sub>4</sub> at 100°C for 2 h. After incubation any remaining glucan fragments were quantitatively hydrolyzed to glucose using a mixture of exo-1,3-β-glucanase and β-glucosidase which gives a measurement of total glucan. The alpha glucan and sucrose content of the sample is determined by hydrolyzing specifically to D-glucose and D-fructose. Glucose was measured with amyloglucosidase and invertase using Glucose Determination Reagent. β-glucan was determined by the difference in both measurements.

## In-house vs commercial lentinan beta glucans in bacterial pneumoniae

One hour following intra-tracheal instillation of *K. pneumoniae* bacteria, animals were randomized to receive: (i) vehicle (PBS, 300 µl, n = 12); (ii) 20 mg kg<sup>-1</sup> IHL β-glucan (n = 8); (iii) 20mg kg<sup>-1</sup> CL β-glucan (n = 8) and the degree of injury assessed at 48 h postpneumonia induction.

#### In vivo assessment

Animals were anaesthetized with subcutaneous ketamine 75 mg kg<sup>-1</sup> (Ketalar; Pfizer, Cork, Ireland) and medetomidine 0.5 mg kg<sup>-1</sup> (Domitor; Vétoquinol, Dublin, Ireland), intravenous and intra-arterial access was secured, and a tracheostomy tube was inserted. Anaesthesia was maintained with 2 mg kg<sup>-1</sup> Alfaxan® (Jurox PLC, UK) and paralysis with 0.5 mg kg<sup>-1</sup> atracurium besylate (Tracrium; GlaxoSmithKline, Dublin, Ireland), and mechanical ventilation commenced using 7 ml kg<sup>-1</sup> tidal volume. Arterial blood pressure, airway pressure, lung static compliance and arterial blood gas analyses were performed as previously described (Curley *et al.* 2012; Devaney *et al.* 2015).

#### Ex vivo analyses

Following exsanguination under anaesthesia, bronchoalveolar lavage (BAL) was performed, and BAL fluid differential leukocyte counts and lung bacterial colony counts were completed. BAL concentrations of TNF- $\alpha$  and IL-10 cytokines were determined using ELISA (R&D Systems, Abingdon, UK) and total BAL protein was measured (Micro BCA; Pierce, Rockford, IL). All *ex vivo* analyses were performed by blinded investigators.

#### Statistical analysis

A sample size of 8–12 animals per group was determined for a three-group design based on previously published experimental data from these models by our group (Chonghaile *et al.* 2008; O'Croinin *et al.* 2005; Devaney *et al.* 2015). Data are reported as means ( $\pm$ SD) or as medians (interquartile range). Data were analysed using Sigma Stat (SYSTAT® software, Richmond, CA). The distribution of all data was tested for normality using Kolmogorov–Smirnov tests. Data were analysed by Student's unpaired *t*-tests or one-way ANOVA as appropriate, with post hoc testing using Student–Newman–Keuls between group comparisons as appropriate. Underlying model assumptions were deemed appropriate based on suitable residual plots. A two-tailed *P* value of <0.05 was considered significant.

#### Acknowledgements

This work was supported by the Irish Health Research Board (HRB-2015-POR-644).

#### **Conflict of Interest**

No conflict of interest declared.

#### References

Bellani, G., Laffey, J.G., Pham, T., Fan, E., Brochard, L.,
Esteban, A., Gattinoni, L., Van Haren, F. *et al.* (2016)
Epidemiology, patterns of care, and mortality for patients with acute respiratory distress syndrome in intensive care units in 50 countries. *JAMA* 315, 788–800.

Downloaded from https://academic.oup.com/lambio/article/71/4/405/6698073 by Athlone Institute of Technology user on 31 January 2023

Brown, G.D. and Gordon, S. (2005) Immune recognition of fungal beta-glucans. *Cell Microbiol* **7**, 471–479.

Brown, G.D., Herre, J., Williams, D.L., Willment, J.A., Marshall, A.S. and Gordon, S. (2003) Dectin-1 mediates the biological effects of beta-glucans. *J Exp Med* 197, 1119–1124.

Camilli, G., Tabouret, G. and Quintin, J. (2018) The Complexity of Fungal beta-Glucan in Health and Disease: Effects on the Mononuclear Phagocyte System. *Front Immunol* 9, 673.

Carballo, C., Pinto, P.I.S., Mateus, A.P., Berbel, C., Guerreiro, C.C., Martinez-Blanch, J.F., Codoner, F.M., Mantecon, L. *et al.* (2019) Yeast beta-glucans and microalgal extracts modulate the immune response and gut microbiome in Senegalese sole (*Solea senegalensis*). *Fish Shellfish Immunol* **92**, 31–39.

Chan, G.C., Chan, W.K. and Sze, D.M. (2009) The effects of beta-glucan on human immune and cancer cells. *J Hematol Oncol* **2**, 25.

Chan, W.K., Law, H.K.W., Lin, Z.B., Lau, Y.L. and Chan, G.C. (2007) Response of human dendritic cells to different immunomodulatory polysaccharides derived from mushroom and barley. *Int Immunol* 19, 891–899.

Chang, R. (2002) Bioactive polysaccharides from traditional Chinese medicine herbs as anticancer adjuvants. *J Altern Complement Med* **8**, 559–65.

Chonghaile, M.N., Higgins, B.D., Costello, J. and Laffey, J.G. (2008) Hypercapnic acidosis attenuates lung injury induced by established bacterial pneumonia. *Anesthesiology* 109, 837–48.

Curley, G.F., Hayes, M., Ansari, B., Shaw, G., Ryan, A., Barry, F., O'brien, T., O'toole, D. *et al.* (2012) Mesenchymal stem cells enhance recovery and repair following ventilatorinduced lung injury in the rat. *Thorax* 67, 496–501.

Curley, G.F., Jerkic, M., Dixon, S., Hogan, G., Masterson, C., O'Toole, D., Devaney, J. and Laffey, J.G. (2017) Cryopreserved, xeno-free human umbilical cord mesenchymal stromal cells reduce lung injury severity and bacterial burden in rodent *Escherichia coli*-induced acute respiratory distress syndrome. *Crit Care Med* **45**, e202–e212.

Devaney, J., Curley, G.F., Hayes, M., Masterson, C., Ansari, B., O'brien, T., O'toole, D. & Laffey, J.G. (2013) Inhibition of pulmonary nuclear factor kappa-B decreases the severity of acute *Escherichia coli* pneumonia but worsens prolonged pneumonia. *Crit Care* 17, R82.

Devaney, J., Horie, S., Masterson, C., Elliman, S., Barry, F., O'brien, T., Curley, G.F., O'toole, D. *et al.* (2015) Human mesenchymal stromal cells decrease the severity of acute lung injury induced by *E. coli* in the rat. *Thorax* 70, 625–635.

Ferguson, N.D., Frutos-Vivar, F., Esteban, A., Fernandez-Segoviano, P., Aramburu, J.A., Najera, L. and Stewart, T.E. (2005) Acute respiratory distress syndrome: underrecognition by clinicians and diagnostic accuracy of three clinical definitions. *Crit Care Med* 33, 2228–2234. Giske, C.G., Martinez-Martinez, L., Cantón, R., Stefani, S., Skov, R., Glupczynski, Y., Nordmann, P., Wootton, M. et al. (2017) EUCAST Guidelines for Detection of Resistance Mechanisms and Specific Resistances of Clinical and/or Epidemiological Importance, 2nd edn. Testing, E. C. O. A. S. Sweden: European Union.

Grommes, J. and Soehnlein, O. (2011) Contribution of neutrophils to acute lung injury. *Mol Med* **17**, 293–307.

Herre, J., Marshall, A.S., Caron, E., Edwards, A.D., Williams, D.L., Schweighoffer, E., Tybulewicz, V., Sousa, R.E. *et al.* (2004) Dectin-1 uses novel mechanisms for yeast phagocytosis in macrophages. *Blood* **104**, 4038–4045.

Kon, Z.N., Dahi, S., Evans, C.F., Byrnes, K.A., Bittle, G.J., Wehman, B., Rector, R.P., Mccormick, B.M. *et al.* (2015) Long-term venovenous extracorporeal membrane oxygenation support for acute respiratory distress syndrome. *Ann Thorac Surg* **100**, 2059–2063.

Laffey, J.G. and Kavanagh, B.P. (2018) Negative trials in critical care: why most research is probably wrong. *Lancet Respir Med* **6**, 659–660.

Lewis, S.R., Pritchard, M.W., Thomas, C.M. and Smith, A.F. (2019) Pharmacological agents for adults with acute respiratory distress syndrome. *Cochrane Database Syst Rev* 7, CD004477.

Masterson, C., Devaney, J., Horie, S., O'Flynn, L., Deedigan, L., Elliman, S., Barry, F., O'Brien, T. *et al.* (2018) Syndecan-2-positive, bone marrow-derived human mesenchymal stromal cells attenuate bacterial-induced acute lung injury and enhance resolution of ventilatorinduced lung injury in rats. *Anesthesiology* **129**, 502–516.

Mccarty, M.F. and Dinicolantonio, J.J. (2020) Nutraceuticals have potential for boosting the type 1 interferon response to RNA viruses including influenza and coronavirus. *Prog Cardiovasc Dis* https://doi.org/10.1016/j.pcad.2020.02.007.

Mcnicholas, B.A., Rooney, G.M. and Laffey, J.G. (2018) Lessons to learn from epidemiologic studies in ARDS. *Curr Opin Crit Care* 24, 41–48.

Murphy, E.J., Masterson, C., Rezoagli, E., O'Toole, D., Major, I., Stack, G.D., Lynch, M., Laffey, J.G. *et al.* (2020) beta-Glucan extracts from the same edible shiitake mushroom *Lentinus edodes* produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects - implications for coronavirus disease (COVID-19) immunotherapies. *Sci Total Environ* **732**, 139330.

O'croinin, D.F., Hopkins, N.O., Moore, M.M., Boylan, J.F., Mcloughlin, P. & Laffey, J.G. (2005) Hypercapnic acidosis does not modulate the severity of bacterial pneumoniainduced lung injury. *Crit Care Med* 33, 2606–2612.

O'croinin, D.F., Nichol, A.D., Hopkins, N., Boylan, J., O'brien, S., O'connor, C., Laffey, J.G. & Mcloughlin, P. (2008) Sustained hypercapnic acidosis during pulmonary infection increases bacterial load and worsens lung injury. *Crit Care Med* 36, 2128–2135.

Poutsiaka, D.D., Mengozzi, M., Vannier, E., Sinha, B. and Dinarello, C.A. (1993) Cross-linking of the beta-glucan

Letters in Applied Microbiology 71, 405-412 © 2020 The Society for Applied Microbiology

receptor on human monocytes results in interleukin-1 receptor antagonist but not interleukin-1 production. *Blood* **82**, 3695–3700.

- Quintin, J., Saeed, S., Martens, J.H., Giamarellos-Bourboulis,
  E.J., Ifrim, D.C., Logie, C., Jacobs, L. Jansen, T. *et al.* (2012) Candida albicans infection affords protection against reinfection via functional reprogramming of monocytes. *Cell Host Microbe* 12, 223–232.
- Rice, P.J., Lockhart, B.E., Barker, L.A., Adams, E.L., Ensley,
  H.E. and Williams, D.L. (2004) Pharmacokinetics of fungal (1-3)-beta-D-glucans following intravenous administration in rats. *Int Immunopharmacol* 4, 1209–1215.
- Rubenfeld, G.D., Caldwell, E., Peabody, E., Weaver, J., Martin, D.P., Neff, M., Stern, E.J. and Hudson, L.D. (2005)
  Incidence and outcomes of acute lung injury. *N Engl J Med* 353, 1685–1693.
- Sahasrabudhe, N.M., Dokter-Fokkens, J. and De Vos, P. (2016) Particulate beta-glucans synergistically activate TLR4 and Dectin-1 in human dendritic cells. *Mol Nutr Food Res* **60**, 2514–2522.
- Shi, Y., Wang, Y., Shao, C., Huang, J., Gan, J., Huang, X., Bucci, E., Piacentini, M. *et al.* (2020) COVID-19 infection: the perspectives on immune responses. *Cell Death Differ* 27, 1451–1454.
- Sullivan, R., Smith, J.E. and Rowan, N.J. (2006) Medicinal mushrooms and cancer therapy: translating a traditional practice into Western medicine. *Perspect Biol Med* 49, 159–170.
- Taylor, P.R., Brown, G.D., Reid, D.M., Willment, J.A., Martinez-Pomares, L., Gordon, S. and Wong, S.Y. (2002) The beta-glucan receptor, dectin-1, is predominantly expressed on the surface of cells of the monocyte/ macrophage and neutrophil lineages. *J Immunol* 169, 3876–3682.
- Vereschagin, E.I., van Lambalgen, A.A., Dushkin, M.I., Schwartz, Y.S., Polyakov, L., Heemskerk, A., Huisman, E., Thijs, L.G. *et al.* (1998) Soluble glucan protects against endotoxin shock in the rat: the role of the scavenger receptor. *Shock* 9, 193–198.

- Vetvicka, V. and Vetvickova, J. (2015) Glucan supplementation enhances the immune response against an influenza challenge in mice. *Ann Transl Med* **3**, 22.
- Vetvicka, V. and Vetvickova, J. (2018) Glucans and Cancer: Comparison of Commercially Available beta-glucans - Part IV. Anticancer Res 38, 1327–1333.
- Wan-Mohtar, W.A., Young, L., Abbott, G.M., Clements, C., Harvey, L.M. and Mcneil, B. (2016) Antimicrobial properties and cytotoxicity of sulfated (1,3)-beta-D-glucan from the mycelium of the mushroom *Ganoderma lucidum*. *J Microbiol Biotechnol* 26, 999–1010.
- Ware, L.B. and Matthay, M.A. (2000) The acute respiratory distress syndrome. *N Engl J Med* **342**, 1334–1349.
- Willment, J.A., Lin, H.H., Reid, D.M., Taylor, P.R., Williams, D.L., Wong, S.Y., Gordon, S. and Brown, G.D. (2003)
  Dectin-1 expression and function are enhanced on alternatively activated and GM-CSF-treated macrophages and are negatively regulated by IL-10, dexamethasone, and lipopolysaccharide. *J Immunol* 171, 4569–4573.
- Zhang, J., Wen, C., Gu, J., Ji, C., Duan, Y. and Zhang, H. (2019) Effects of subcritical water extraction microenvironment on the structure and biological activities of polysaccharides from Lentinus edodes. *Int J Biol Macromol* **123**, 1002–1011.

#### **Supporting Information**

Additional Supporting Information may be found in the online version of this article:

**Figure S1.**  $\beta$ -Glucan content was analysed using the Megazyme analysis kit and  $\alpha$  (white bars) and  $\beta$ -glucan (grey bars) content expressed as % (w/w) of total glucan in each sample (adapted from Murphy *et al.* 2020).

**Figure** S2. *Klebsiella pneumoniae* models were established using a dose escalation of viable bacteria  $(2 \times 10^7 - 1 \times 10^9 \text{ CFU} \text{ per animal})$  given intratracheally.

Contents lists available at ScienceDirect







journal homepage: www.elsevier.com/locate/scitotenv

## β-Glucan extracts from the same edible shiitake mushroom *Lentinus* edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects – Implications for coronavirus disease (COVID-19) immunotherapies



Emma J. Murphy <sup>a,\*</sup>, Claire Masterson <sup>b</sup>, Emanuele Rezoagli <sup>b,c,d</sup>, Daniel O'Toole <sup>b</sup>, Ian Major <sup>e</sup>, Gary D. Stack <sup>a</sup>, Mark Lynch <sup>a</sup>, John G. Laffey <sup>b,c</sup>, Neil J. Rowan <sup>a,f</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Athlone, Ireland

<sup>b</sup> Lung Biology Group, Regenerative Medicine Institute at CURAM Centre for Medical Devices, National University of Ireland Galway, Galway, Ireland

<sup>c</sup> Anaesthesia and Intensive Care Medicine, University Hospital Galway, Galway, Ireland

<sup>d</sup> Department of Medicine and Surgery, University of Milano-Bicocca, Monza (MB), Italy

<sup>e</sup> Materials Research Institute, Athlone Institute of Technology, Athlone, Ireland

<sup>f</sup> Centre for Disinfection, Sterilization and Biosecurity, Athlone Institute of Technology, Athlone, Ireland

#### HIGHLIGHTS

- β-Glucans from shiitake mushroom reduces IL-1β, IL-6 in in vitro lung injury model.
- β-Glucans from same source can differ in immunomodulatory and pulmonary cytoprotective effects.
- β-Glucans can reduce oxidative stress and activate macrophages.
- β-Glucans may ameliorate cytokine storm that causes ARDS as seen with COVID-19.

#### ARTICLE INFO

Article history: Received 13 April 2020 Received in revised form 7 May 2020 Accepted 8 May 2020 Available online 11 May 2020

Editor: Damia Barcelo

Keywords: β-Glucans Medicinal mushroom COVID-19 Immunomodulation

#### GRAPHICAL ABSTRACT



#### ABSTRACT

Coronavirus pneumonia is accompanied by rapid virus replication, where a large number of inflammatory cell infiltration and cytokine storm may lead to acute lung injury, acute respiratory distress syndrome (ARDS) and death. The uncontrolled release of pro-inflammatory cytokines, including interleukin (IL)-1 $\beta$  and IL-6, is associated with ARDS. This constituted the first study to report on the variability in physicochemical properties of  $\beta$ glucans extracts from the same edible mushroom *Lentinus edodes* on the reduction of these pro-inflammatory cytokines and oxidative stress. Specifically, the impact on the immunomodulatory and cytoprotective properties of our novel in 'house' (IH-Lentinan, IHL) and a commercial (Carbosynth-Lentinan, CL) Lentinan extract were investigated using in vitro models of lung injury and macrophage phagocytosis. CL comprised higher amounts of  $\alpha$ glucans and correspondingly less  $\beta$ -glucans. The two lentinan extracts demonstrated varying immunomodulatory activities. Both Lentinan extracts reduced cytokine-induced NF- $\kappa$ B activation in human alveolar epithelial A549 cells, with the IHL extract proving more effective at lower doses. In contrast, in activated THP-1 derived

\* Corresponding author at: Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Ireland. *E-mail address*: emurphy@ait.ie (E.J. Murphy).

Cytokine storm Disease mitigation macrophages, the CL extract more effectively attenuated pro-inflammatory cytokine production (TNF- $\alpha$ , IL-8, IL-2, IL-6, IL-22) as well as TGF- $\beta$  and IL-10. The CL extract attenuated oxidative stress-induced early apoptosis, while the IHL extract attenuated late apoptosis. Our findings demonstrate significant physicochemical differences between Lentinan extracts, which produce differential in vitro immunomodulatory and pulmonary cytoprotective effects that may also have positive relevance to candidate COVID-19 therapeutics targeting cyto-kine storm.

© 2020 Published by Elsevier B.V.

#### 1. Introduction

The pandemic outbreak of coronavirus disease 2019 (COVID-19) is rapidly spreading globally (Zhang et al., 2020; Rowan and Laffey, 2020). Reports from China showed that about 20% of COVID-19 patients developed severe disease, resulting in a fatality of 4% (Zhang et al., 2020). A large part of COVID-19 patients in China experienced severe complications including acute respiratory distress syndrome (ARDS) requiring admission to intensive care unit (ICU) (Zhang et al., 2020). ARDS is a devastating condition of severe respiratory failure with 40% mortality and for which novel therapies are urgently needed (Rezoagli et al., 2017). It is characterized by widespread inflammation of the lungs, where extrapulmonary infections are a key aetiology of ARDS onset (Laffey and Matthay, 2017). The inflammatory insult results in lung parenchyma injury and activation of the immune system with an upregulation of pro-inflammatory cytokines (Laffey and Matthay, 2017). The current international standard of intervention includes ventilatory management and organ support (Bellani et al., 2016). The innate immune system plays a pivotal role in the pathophysiology of ARDS (Chousterman et al., 2017).

COVID-19 causes an inflammatory or cytokine storm (CS) in the lungs with the excessive and uncontrolled release of proinflammatory cytokines, including interleukin (IL)-1 $\beta$  and IL-6 (Conti et al., 2020). The binding of COVID-19 to the Toll-Like Receptor (TLR) causes the release of pro-IL-1 $\beta$  which is cleaved by caspase-1, followed by the production of IL-1 $\beta$  that is a mediator of lung inflammation, fever and fibrosis. Researchers have confirmed level of inflammatory factors in patients with COVID-19 including elevation of IL-6 in non-survival groups (Huang et al., 2020), as compared with that of the survivals. Therefore, how to block the CS and when to initiate anti-inflammation therapy is critical for reducing the death rate of COVID-19 (Channappanavar and Perlman, 2017; Chousterman et al., 2017). Suppression of pro-inflammatory IL-1 family members and IL-6 has been shown to have a therapeutic effect in many inflammatory diseases, including viral infections (Zhang et al., 2020).

There is an increasing interest in the medicinal use of mushrooms nutraceuticals that have been previously reported to exhibit wideranging activities including anti-inflammatory, anti-tumor and immune-modulating capabilities (Pelizon et al., 2005; Zheng et al., 2005; Akramiene et al., 2007; Kumar, 2015). β-Glucans are one of the main active components derived from mushrooms (Smith et al., 2002; Zhu et al., 2015). These are glucose polymers that are linked together through 1,3 linear  $\beta$ -glycosidic chains. Complexity and variation in the compound derive from side branching structures (Stone, 2009). βglucans isolated from fungi commonly possess side branching at the 1,4 or 1,6 position (Kaur et al., 2020). Variance will also occur with chain length and many these variances are species-specific and dictate biological activity (Sullivan et al., 2006; Chaichian et al., 2020). Lentinans are a specific class of  $\beta$ -glucans extracted from the edible mushroom *Lentinus edodes*, and are composed of a  $\beta$ -(1–3)-glucose backbone with two (1-6)- $\beta$ -glucose branches of each five glucose units (Sullivan et al., 2006; Kaur et al., 2020). There has been an increasing interest in their use for treating disease in animals and humans (Carballo et al., 2019; McCarty and DiNicolantonio, 2020). McCarty and DiNicolantonio (2020) recently described the potential role of  $\beta$ - glucan as a natural nutraceutical for boosting type 1 interferon response to RNA viruses such as influenza and coronavirus.

Putative use of  $\beta$ -glucans in mitigating lung infections correlates with findings from our recent in vivo studies to address ARDS (Masterson et al., 2019). However, there were significant challenges in identifying a reliable and repeatable source of  $\beta$ -glucans suitable for lung delivery as findings from screening of over 20 natural and commercial products screened revealed that they were unsuitable for lung delivery due to microbial contamination or exhibiting very low levels of  $\beta$ -glucan. We recently reported that purified  $\beta$ -glucans (Lentinan) from the Shiitake mushroom Lentinus edodes, obtained using our inhouse novel extraction method can be used to reduce populations of clinical isolate Klebsiella pneumoniae harbouring multiple antibiotic resistances in an in vivo lung infection model (Masterson et al., 2019). We reported that administration of Lentinan shows potential for treating sepsis-induced lung injury as it effectively reduces bacterial load in arterial blood and BAL, reduces white cell count protein inflammation to the lungs, and improves lung physiological parameters. Evidence also showed that in-house Lentinan extracted supported vital pO2 along with promoting lung cellular repair.

This constitutes the first study to compare a commercially sourced Lentinan extract from the edible mushroom *Lentinus edodes* (referred to as Carbosynth-Lentinan) to that of an in-house hot-water extract (IHL) of the same mushroom in order to evaluate immunomodulatory properties. These were characterized using an in vitro lung injury model with a focus on profiling components associated with cytokine storm. It is hypothesized that  $\beta$ -glucans derived from exotic mushrooms have the potential to alleviate the immune cascade in pathological conditions, such as ARDS that is experienced by COVID-19 patients.

#### 2. Materials and methods

#### 2.1. Materials

Commercial Lentinan (CL) was sourced from Carbosynth (FL33321, Compton, Berkshire, UK). Fruiting bodies of *Lentinus edodes* were purchased from Fancy Fungi (Ringaheen, Co. Wexford Ireland). IHL was extracted from the fruiting bodies using a novel process. A549 cells (used at passage 90) and THP-1 cells (used at passage 10) were obtained from the American Type Culture Collection, (ATCC, Rockville, MD, USA). Cells were cultured in RPMI-1640 (Sigma-Aldrich, St. Louis, MO, USA), supplemented with 10% fetal calf serum (Sigma-Aldrich), 1% penicillin G (100 U/mL) and streptomycin (100  $\mu$ g/mL) solution (Sigma-Aldrich) at 37 °C in 95% air/5% CO<sub>2</sub> environment. For differentiation into macrophages, THP-1 cells were treated with phorbol 12-myristate 13-acetate (PMA) (Peprotech EC, London, UK) at a concentration of 100 ng/mL for 48 h.

#### 2.2. Physicochemical characterization of $\beta$ -glucan samples

#### 2.2.1. Megazyme analysis

Extracts were analyzed for (1,3)-(1,6)- $\beta$ -glucan content using the Megazyme yeast and mushroom kit (K-YBGL) (Megazyme Ltd., Bray, Co. Wicklow, Ireland). Assays were carried out according to manufacturer's instructions. Briefly, samples were milled, and placed in 12 M

H<sub>2</sub>SO<sub>4</sub> at -4 °C for 2 h to solubilize the glucans. The samples were then hydrolyzed in 2 M H<sub>2</sub>SO<sub>4</sub> at 100 °C for 2 h. After incubation, any remaining glucan fragments were quantitatively hydrolyzed to glucose using a mixture of *exo*-1,3- $\beta$ -glucanase and  $\beta$ -glucosidase which gives a measurement of total glucan. The alpha glucan and sucrose content of the sample is determined by hydrolyzing specifically to D-glucose and D-fructose. Glucose was measured with amyloglucosidase and invertase using a glucose oxidase peroxidase GOPOD reagent.  $\beta$ -Glucan was determined by the difference in both measurements.

Attenuated total reflectance Fourier transform infrared spectroscopy (ATR-FTIR).

ATR-FTIR was carried out on a Perkin Elmer Spectrum fitted with a universal ATR sampling accessory. All data were recorded at ambient temperature, in the spectral range of 4000–650 cm<sup>-1</sup>, utilizing a 16 scan per sample cycle and a fixed universal compression force of 80 N. Subsequent analysis was carried out using Spectrum software.

#### 2.2.2. Scanning electron microscopy (SEM)

SEM was performed on a Mira SEM (Tescan Oxford Instruments, UK) using a range of magnifications to evaluate the surface morphology of the extracts using the function of secondary electrons. Samples were placed on an aluminum stub and were gold-coated using Baltec SCD 005 sputter coater (BAL-TEC GmbH, Chemnitz Germany) for 110 s at 0.1 mbar vacuum before observation.

#### 2.2.3. Nuclear magnetic resonance (NMR)

<sup>1</sup>H NMR spectra were obtained using an Agilent Technologies Ultra High Field (UHF) 800 MHz NMR system. Spectra were analyzed using ACD NMR software. Samples were prepared in deuterated water (D<sub>2</sub>O).

2.3. Immunomodulatory properties of  $\beta$ -glucan samples from Lentinus edodes

#### 2.3.1. Cell viability assays

Cells were seeded at a density of  $4 \times 10^5$  cells/well in a 96 well plate. After 24 h, cells were treated with varying concentrations of both CL and IHL. Cells were incubated with samples for a further 24 h. For MTT assay, media was aspirated and cells were treated with 10% 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide (MTT) in RPMI for 3.5 h at 37 °C in a humidified incubator with 5% CO<sub>2</sub>. MTT and media were aspirated and formazan product was solubilized through the addition of 100 µL of DMSO per well. Solubilized product was quantitatively measured at 540 nm using a Synergy<sup>TM</sup> HT Multi-Mode Microplate Reader (BioTek, Winooski, USA). Results were expressed as percentage viability with respect to vehicle control.

#### 2.3.2. Enzyme linked immunosorbent assay (ELISA)

Human Duoset sandwich ELISA kit (RnD Systems MN, USA) was used to measure cytokine levels in the medium after  $\beta$ -glucan exposure. All ELISAs were performed according to manufacturer's instructions. Results were expressed either in pg/mL or in ng/mL.

#### 2.3.3. Cell injury

Pulmonary alveolar type II A549 cells were seeded at a density of  $4 \times 10^5$  cells/well in 96 well plates. After 24 h cells were injured with 1 ng/mL of IL-1 $\beta$  (Peprotech, Rocky Hill, NJ) in RPMI supplemented with 1% penicillin/streptomycin. THP-1 cells were seeded at a density of  $4 \times 10^5$  cells/well in 96 well plates and 24 h later injured with 100 ng/mL LPS (Sigma) in RPMI supplemented with 1% penicillin/streptomycin. Cells were then treated with  $\beta$ -glucan samples for a further 24 h.

#### 2.3.4. Luciferase assays

A549 cells stably transfected with NF- $\kappa$ B-luc reporter gene were obtained from the National University of Ireland Galway. These A549 cells were seeded at a density of 4  $\times$  10<sup>5</sup> cells/well in 96 well plates. After

24 h, IL-1 $\beta$  and  $\beta$ -glucan samples were added to each well for 3.5 h at 37 °C in a humidified incubator with 5% CO<sub>2</sub>. 100 µL of luciferase substrate (SolarGlow Molecutools, Dublin, Ireland) which includes lysis buffer, was added to each well. After brief agitation on an orbital shaker, luminescence was assessed in a Microplate Reader (BioTek, USA). Results were expressed as fold induction with respect to vehicle treated controls.

#### 2.3.5. Phagocytosis assay

The Vybrant<sup>TM</sup> phagocytosis assay kit (Invitrogen, Carlsbad, CA) was used to measure the phagocytosis capacity of THP-1 cells after exposure to  $\beta$ -glucan samples. Samples were analyzed according to manufacturer's instructions. Phagocytic index was calculated and results were graphed as percentage phagocytic index with respect to vehicle treated controls.

#### 2.3.6. Hydrogen peroxide (H<sub>2</sub>O<sub>2</sub>) injury assay

To determine the effect of  $H_2O_2$  on cell viability, cell apoptosis and necrosis were assessed using propidium iodide (PI) and Annexin-VFITC conjugated antibodies (Miltenyl Biotech). Cells were initially treated with a dose range of  $H_2O_2$  from 0.001 mM to 20 mM. As 10 mM of  $H_2O_2$ , reduced cell viability by 50%, it was selected as the injuring concentration for further experiments. Cells were simultaneously treated with  $\beta$ -glucan samples. After 24 h of incubation, cells were complexed with Annexin-VFITC conjugated antibody at 1:1000 dilution for a further 10 min in the dark. Cells were washed in PBS three times and resuspended in flow buffer (Miltenyl Biotech). Cells were then automatically incubated with PI using the MACS Quant analyzer. Cells from each sample were then analyzed by MACSQUANT Analyzer 10 (Miltenyl Biotech). Data was analyzed using Flowlogic software(Miltenyl Biotech) with a guide to interpretation of findings represented in Table 1.

#### 2.4. Statistical data analysis

Continuous data were described as mean  $\pm$  standard deviation (SD). Differences in glucan content between CL and IHL were tested using unpaired Student's t-test. Differences of kb induction in A459 cells over increasing doses of CL and IHL were explored using a two-way analysis of variance. The viability of A549 cells was tested using increasing doses of β-glucans (i.e. IHL or CL) using a one-way analysis of variance. Post-hoc multiple comparison of increasing doses of CL and IHL versus 0 mg/mL was performed by controlling the false discovery rate using the twostage step-up method of Benjamini, Krieger and Yekuteli. Differences in cytokine levels and phagocytosis were tested between different groups (i.e. LPS, CL and IHL) or between different doses (i.e. 1, 5 and 10 mg/mL) of  $\beta$ -glucans using a one-way analysis of variance. Posthoc multiple comparison versus PBS was performed by controlling the false discovery rate using the two-stage step-up methods of Benjamini, Krieger and Yekuteli. Statistical significance was reached with a pvalue < .05 (2-tailed). Statistical analyses were performed using GraphPad Prism 7a (GraphPad Software, San Diego, CA, USA) and Microsoft Excel for Mac 2017, Version 15.32 (Microsoft Corporation, Redmond, WA).

#### 3. Results & discussion

#### 3.1. Physicochemical characterization of $\beta$ -glucan samples

Megazyme analysis of Carbosynth-lentinan (CL) and In-house Lentinan (IHL) samples are displayed in Fig. **1**. Data shows both  $\beta$  and alpha glucan % w/w content. CL was shown to have a significantly higher alpha glucan content. IHL was shown to have a significantly higher  $\beta$ glucan content and were purer and cleaner in composition. Scanning electron microscopy paired with **energy dispersive X-ray analysis was carried out to determine particle size and elemental content, with results displayed in Figs. 2 and 3 respectively. Fig. 2, a** 

#### Table 1

Interpretation of H<sub>2</sub>O<sub>2</sub> assay derived from using flow-cytometry data.

Quadrant location:	Location representation:	Annexin V:	Propidium iodide:	Cell condition:
Lower left		Negative	Negative	Living cells
Lower right	***	Positive	Negative	Early apoptotic
Upper right	-460 -1600	Negative	Positive	Late apoptotic cells
Upper left		Positive	Positive	Necrotic cells
	***			

dimensional representation of particle size, shows that CL has a uniform particle dispersity. Conversely, IHL shows a more heterogeneous particle mix. Fig. 3 shows that IHL contains fewer elements compared to CL, as sulfur, silicon and chlorine were not present in IHL. These results, paired with Megazyme results, reaffirm that IHL is a cleaner preparation with fewer constituents.

FT-IR spectra of the CL and IHL are shown in Fig. 4. The IR analysis reveals a strong absorption peak in the fingerprint region at approximately 1015 cm<sup>-1</sup> in both the commercial and in-house lentinan samples. The presence of equivalent absorption peaks, in the absorption range characteristic of polysaccharides, strongly suggests that both samples contain the same  $\beta$ -glucan compound. **Preliminary NMR analysis of the commercial and in-house Lentinan suggests significant** 





differences between the products. The <sup>1</sup>H spectra of the compounds are shown in Fig. 5.

Previous researchers have reported that variability in β-glucan chain length, branching and composition can vary based principally on on source (Kaur et al., 2020). Extraction procedures will have an effect on the structure and purity of the  $\beta$ -glucan product, which may have an effect of its bioactivity. Thus, there has been an obstacle in understanding biological activities and this has hindered potential therapeutic development (Zhang et al., 2011). Scanning electron microscopy paired with energy dispersive X-ray analysis showed that the IHL had a diverse heterogenous particle composition in comparison to CL, which had a more uniform composition. IHL was also found to have less chemical elements (sulfur, silicon and chlorine) compared to CL. These results suggest that IHL is a purer  $\beta$ -glucan product, compared to the commercially available sample. FTIR characterization of the samples shows that both samples contain the same β-glucan compound (Fig. 4). NMR analysis contradicted this and showed that there is a significant difference between both samples (refer to supplementary information). Bak and colleagues carried out a study to measure the glucan contents in the fruiting bodies of L. edodes mushroom from various cultivars which was found to be variable based on cultivar (Bak et al., 2014). Gil-Ramirez and coauthors found a variance between mushroom samples based on growth conditions, degree of fruiting body maturing body as well as a difference between fresh and fruiting bodies (Gil-Ramirez et al., 2011). Studies have also shown variance in β-glucan content dependent on the country of origin (Bak et al., 2014). L. edodes from Japan (49.5% β-glucan w/ w) having a higher content than that of mushrooms purchased in Iran  $(38\% \beta$ -glucan w/w).

Immunomodulatory properties of  $\beta$ -glucan samples from Shiitake mushroom *L. edodes*.

To investigate the anti-inflammatory effects of the  $\beta$ -glucan samples, transformed human airway epithelial cells (A549) were used. Initial experiments showed that a dose of 10 ng/mL of IL-1 $\beta$  gave an IL-8 release response, indicative of NF- $\kappa$ B activation. To further examine



Fig. 2. Dimensional representation of commercial (A) and in-house (B) samples analyzed by scanning electron microscopy.



Fig. 3. Quantitative analysis of elemental contents in commercial (A) and in-house (B) samples determined by energy dispersive X-ray analyzer.



Fig. 4. FTIR spectra of the commercial (A) and in-house lentinan (B). \$23\$



Fig. 5. A549 cells transfected with NF-KB reporter gene were stimulated with IL-1β for 1 h before administration of β-glucan samples. \*p < .05; \*\*p < .01; \*\*\*p < .001 versus 0 mg/mL.

the effects of the  $\beta$ -glucan samples on the NF- $\kappa$ B pathway, A549 cells stably transfected with the NF-KB luciferase reporter gene were used. IL-1B stimulation caused a 6-fold increase in luciferase expression. Both  $\beta$ -glucan samples reduced this expression, with IHL inhibiting this increase at a dose of 1 mg/mL (Fig. 5B). CL was able to reduce IL- $1\beta$  induced NF- $\kappa$ B pathway activation at the higher concentrations (Fig. 5A). This IL-1B pulmonary model finding is directly relevant to COVID-19 as IL-1 $\beta$  is a prominent part of the 'cytokine storm' response (Zhang et al., 2020). Hypercytokinaemia is considered a prominent mechanism of injury in COVID-19 patients. Furthermore, there are reports of the IL-1 Ra blocker Anakinra being effective for COVID-19, where Anakinra is being tested as a potential therapy for clinical trials. There are currently no treatments directed at halting the cytokine storm and acute lung injury to stop the progression from manageable hypoxia to frank respiratory failure and ARDS in patients with COVID-19 infection. Preventing progression from early acute hypoxia and cytokine release syndrome to frank hypoxic respiratory failure and ARDS could have a huge impact on the foreseeable overflow of the ICU units (https://clinicaltrials.gov/ct2/show/NCT04330638). The aforementioned reported that in ventilated patients, preventing the onset of ARDS, or shortening ICU stay could also be crucial in this regard. Furthermore, the clinical status after 15 days treatment was evaluated to measure the effectiveness of tocilizumab, tocilizumab and anakinra, siltuximab, siltuximab and anakinra and anakinra on restoring lung homeostasis, using single IV injection (siltuximab or tocilizumab) combined or not with daily subcutaneous injections of anakinra until 28 days or hospital discharge, whichever is first.

IHL was able to achieve these effects at the lower concentrations analyzed. MTT assays were preforming to ensure that cell viability was not contributing to the anti-inflammatory effects observed. Results show that both extracts did not elicit any cytotoxic effects at all concentrations, despite a trend toward a decrease in viability that was seen at the highest dose tested (10 mg/mL), but was only evident for the CL group (p = .096) (Fig. 6). No statistical differences were observed for multiple comparisons of singe *p*-values for MTT findings, which is attributed to the broad data variability represented by large standard deviation observed (Fig. 6). Post-hoc multiple comparisons of increasing doses of CL and IHL versus test control (0 mg/mL) were used to ensure that MTT findings were not over-stated in the assessment of preclinical data to reduce risk of false discovery. Statistical analysis data of multiple comparisons for single *p*-values for MTT findings are also provided in supporting information.

The capability of the  $\beta$ -glucan samples to induce or inhibit phagocytosis was also investigated. In the absence of injury, both CL (Fig. 7A+C) and IHL (Fig. 7B+C) appeared to reduce phagocytic index. After injury, both samples reduced phagocytic index (Fig. 7).

Macrophages and monocytes recognize  $\beta$ -glucans by various receptors present on their membrane (Vaclav et al., 2013). This recognition will result in the secretion of cytokines (Netea et al., 2008). Therefore, the immunomodulatory activity of the  $\beta$ -glucan samples was analyzed, in relation to their ability to influence the secretion of cytokines from macrophages. Results demonstrated that the samples induced/suppressed cytokine release at different concentrations. For example, IHL induced the secretion of IL-6 – conversely CL suppressed it. Lentinan analyzed by previous researchers was found to increase the release of TNF- $\alpha$  and IL-6 (Morales et al., 2019). This immunomodulatory effect was also analyzed by their ability to influence the phagocytic activity of macrophages. Both samples suppressed phagocytosis after LPS insult, but CL exclusively suppressed phagocytosis in the absence of injury. Dectin-1 receptor is assumed to be one of the main receptors responsible for the recognition of  $\beta$ -glucans. Activation of dectin-1 with 1,3–1,6 β-glucans can trigger cytokine release and phagocytosis (Brown and Gordon, 2001, 2005; Willment et al., 2003; Herre et al., 2004).



Fig. 6. MTT analysis of A549 cells treated with  $\beta$ -glucan samples to ensure antiinflammatory properties' are not related to a reduction in toxicity.

7



Fig. 7. The potential of β-glucan samples to alleviate phagocytic index of THP-1 cells was determined using Vybrant Phagocytosis assay kit \*p < .05; \*\*p < .01; \*\*\*p < .001 versus PBS.

To determine the potential alleviating effects of the  $\beta$ -glucans on oxidative stress injury, THP-1 cells were treated with 10 mM of H<sub>2</sub>O<sub>2</sub> to induce oxidative stress subsequent to  $\beta$ -glucan treatment. In terms of viability (Panel a), all doses of  $\beta$ -glucans increased viability after injury, CL however significantly increased viability at all doses administered (1, 2.5 and 5 mg/mL). CL significantly reduced early apoptosis after injury at all doses tested (Fig. 8B). IHL appeared to reduce both early and late apoptosis (Fig. 8A+B), however this was not significant. IHL at 1 mg/mL and 5 mg/mL, or CL at 1 mg/mL and 2.5 mg/mL appear to increase necrosis although not significantly (Fig. 8C).

Reactive oxygen species (ROS), which are generated as intermediates in metabolic pathways, are prevalent in pathological conditions. Oxidative stress can occur at a pulmonary level (Rahman and MacNee, 2000). Cellular-derived ROS are produced enzymatically by inflammatory and epithelial cells (Marwick et al., 2007). It has been suggested that oxidants play a contributing role to cell injury as well as leakage of fluid into the lung interstitial space (Liu, 2008). The potential of βglucan to alleviate oxidative stress is relatively unknown.  $\beta$ -Glucans were used to investigate the ability to alleviate oxidative stress in H<sub>2</sub>O<sub>2</sub>-treated THP-1 cells where apoptosis and necrosis were measured using flow cytometry. Propidium iodide (PI) in conjunction with Annexin V was used to determine if THP-1 cells were viable, apoptotic or necrotic. Cell status was assessed based on differences in plasma membrane integrity and permeability (Vermes et al., 2000; Farrell et al., 2011). The results from this experiment were variable and effects were source dependent. Both CL and IHL sources at both conditions tested had the ability to increase viability after injury. CL significantly reduced early apoptosis, and IHL showed a trend to reduce early apoptosis although not significantly. Our results are in line with the findings reported by Zi and colleagues who observed that Lentinan had the ability to alleviate oxidative stress (Zi et al., 2019). Reduction in oxidative stress by use of Lentinan is relevant to COVID-19 intervention. The renin-angiotensin (RAS) signaling pathway, oxidative stress and cell death, cytokines storm and endothelial dysfunction are four major pathways involved in the pathogenesis of COVID-19 (Kouhpayeh et al., 2020). Therapeutic candidates that inhibit RAS and quench oxidative stress would be relevant for COVID-19.

An important assay to determine glucan effect is measurement of cytokine production. In this study, we investigated the effect  $\beta$ -glucan samples had on pro and anti-inflammatory cytokine expression. Inflammatory chemokines and cytokines measured included IL-8, IL-2 and TNF- $\alpha$ . Pleiotropic cytokines included TGF- $\beta$ 1 and IL-6. Antiinflammatory cytokines included IL-10 and IL-22 (Fig. 9). LPS (100 ng/mL) was also tested as it is a known inducer of inflammatory cytokines in immune cells. IHL (1 mg/mL) significantly induced the secretion of inflammatory mediators IL-8 and TNF- $\alpha$ . CL significantly induced the secretion of IL-8. IHL significantly increased the secretion of IL-10 and significantly decreased the secretion IL-22, as did CL. Other cytokines measured included IL-6 and TGF-B1. CL reduced the secretion of IL-6 and TGF-B1. Conversely, IHL significantly increased the secretion of IL-6 and CL decreased secretion. No differences in TGF- $\beta$ 1 concentration were observed when IHL was administered, however CL significantly reduced secretion.

In conclusion, findings of this in-vitro investigation showed that  $\beta$ glucan from Lentinus edodes demonstrated potential for the treatment of lung injury. When compared to a commercial source of the same mushroom, the in-house Lentinan extract contained higher levels of  $\beta$ -glucan and lower levels of  $\alpha$ -glucan. Both Lentinan products reduced inflammation in a lung epithelial model, and IHL achieved this effect at lower doses. Physicochemical characterization studies found important differences in the composition of the Lentinan extracts, as determined by SEM, ATR-FTIR, and NMR, with the CL exhibiting higher amounts of alpha-glucans and correspondingly less B-glucans. The two Lentinan extracts demonstrated varying immunomodulatory activities. Both Lentinan extracts reduced cytokine induced NF-KB activation in human alveolar epithelial A549 cells, with the IHL extract proving more effective at lower doses. In contrast, in activated THP-1 derived macrophages, the CL extract more effectively attenuated proinflammatory cytokine production (TNF- $\alpha$ , IL-8, IL-2, IL-6, IL-22). The CL extract attenuated oxidative stress-induced early apoptosis, while



**Fig. 8.** Flow cytometric analysis of apoptosis (early and late) and necrosis of THP-1 cells after oxidative stress induction and subsequent β-glucan treatment \**p* < .05; \*\**p* < .01; \*\*\**p* < .001 versus 0 mg/mL

the IHL extract attenuated late apoptosis. The implications of these findings infers that defining a reliable and repeatable source of  $\beta$ -glucan, where processes can be tailored to control chain lengths may potentially reduce key cytokines involved in cytokine storm experienced in severe cases of COVID-19. In the future,  $\beta$ -glucans may be delivered as a tailored cocktail matched against critical time-points in the form of future nutraceutical-based intervention for tackling COVID-19. Such  $\beta$ -glucans immunomodulatory cocktails may also have adjacent applications for

8

addressing ARDS that is an important pathophysiological event seen with sepsis. These purified  $\beta$ -glucan combinational cocktails may be produced on a large commercial scale using bioreactors for global deployment (Tafuek et al., 2020).

To maintain functional bioactivity and to increase  $\beta$ -glucan yield, less harmful extraction processes are required that includes cessation of enzyme and harsh chemical usage as adopted using this IHL approach. Findings from this timely study highlight the putative potential

E.J. Murphy et al. / Science of the Total Environment 732 (2020) 139330



Fig. 9. The effects of  $\beta$ -glucan samples on cytokine expression of THP-1 cells preactivated with PMA. \*p < .05; \*\*p < .01; \*\*\*p < .01;

for use  $\beta$ -glucan extractions from the edible mushroom Future research is also required to study putative relationship (if any) of variation in the extraction methods producing different  $\beta$ -glucan preparations and cytokine storm associated with COVID-19. *L. edodes* may also have future potential by way of influencing immunotherapies for addressing COVID-19 that rely on reducing cytokine storm. Further clinical studies are merited to refine  $\beta$ -glucan as a countermeasure for tackling cytokine storm that causes ARDS, as evident with COVID-19.

#### Funding

The authors acknowledge funding support from the Health Research Board Project [grant number HRB-HRA-POR-2014-664], Republic of Ireland, and Athlone Institute of Technology Doctoral Scholarship [grant number AIT0614], Republic of Ireland.

#### **Contributions to publication**

All author contributed equally to this research paper.

#### **Declaration of competing interest**

The authors declare no conflict of interest.

#### Acknowledgements

The authors are grateful to all the researchers whom we cited in this study for their significant and valuable research.

9

#### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi. org/10.1016/j.scitotenv.2020.139330.

#### References

- Akramiene, D., et al., 2007. Effects of beta-glucans on the immune system. Medicina (Kaunas, Lithuania) https://doi.org/10.3390/medicina43080076.
- Bak, W.C., et al., 2014. Determination of glucan contents in the fruiting bodies and mycelia of Lentinula edodes cultivars. Mycobiology https://doi.org/10.5941/ MYCO.2014.42.3.301.
- Bellani, G., et al., 2016. The LUNG SAFE study: a presentation of the prevalence of ARDS according to the Berlin definition! Crit. Care https://doi.org/10.1186/s13054-016-1443-x.
- Brown, G.D., Gordon, S., 2001. Immune recognition. A new receptor for beta-glucans. Nature https://doi.org/10.1038/35092620.
- Brown, G.D., Gordon, S., 2005. Immune recognition of fungal β-glucans. Cell. Microbiol. https://doi.org/10.1111/j.1462-5822.2005.00505.x.
- Carballo, C., Pinto, P.I.S., Mateus, A.P., Berbei, C., Guerreiro, C.C., Martinez-Blanch, J.F., Condoner, F.M., Mantecon, L., Power, D.M., Manchado, M., 2019. Yeast beta-glucans and microalgal extracts modulate the immune response and gut microbiome in Senegalese sole (Solea senegalensis). Fish Shellfish Immunol. 92, 31–39.
- Channappanavar, R., Perlman, S., 2017. Pathogenic human coronovirus infections: causes and consequences of cytokine storm and immunopathology. Semin. Immunopathol. 39, 529–539.
- Chichian, S., Moazzami, B., Sadoughi, F., Haddad, H., Marsa Zaroudi, K., Azemi, Z., 2020. Functional activities of beta-glucans in the prevention or treatment of cervical cancer. J. Ovarian Res. 13, 24. https://doi.org/10.1186/s13048-020.
- J. Ovarian Res. 13, 24. https://doi.org/10.1186/s13048-020. Chousterman, B.G., Swirski, F.K., Weber, G.F., 2017. Cytokine storm and sepsis disease pathogenesis. Semin. Immunopathol. 39, 517–528.
- Conti, P., Ronconi, G., Caraffa, A., Gallenga, C.E., Ross, R., Frydas, I., Kritas, S.K., 2020. Induction of pro-inflammatory cytokines (IL-1 and IL-6) and lung inflammation by Coronavirus-19 (COVI-19 or SARS-CoV-2): anti-inflammatory strategies. J. Biol. Regul. Homeost. Agents 34 (2). https://doi.org/10.23812/CONTI-E Mar 14. pii: 1. (Epub ahead of print).
- Farrell, H., Hayes, J., Laffey, J.G., Rowan, N.J., 2011. Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant *Candida albicans*. J. Microbiol. Methods 84 (2), 317–326.
- Gil-Ramirez, A., et al., 2011. Edible mushrooms as potential sources of new hypocholesterolemic compounds. Proceedings of the 7th International Conference on Mushroom Biology and Mushroom Products.
- Herre, J., et al., 2004. Dectin-1 uses novel mechanisms for yeast phagocytosis in macrophages. Blood https://doi.org/10.1182/blood-2004-03-1140.
- Huang, C., Wang, Y., Li, X., Ren, L., Zhao, J., Hu, Y., 2020. Clinical features of patients infected with 2019 novel coronavirus in Wuhan, China. Lancet 395, 497–506.
- Kaur, R., Sharma, M., Ji D, Xu M, Agyei, D. 2020. Structural features, modification, and functionalities of beta-glucan. Fibers. 8(1):1. https://doi.org/https://doi.org/10.3390/ fib8010001.
- Kouhpayeh, S., Shariati, L., Boshtam, M., Rahimmanesh, I., Mirian, M., Zeinalian, M., Salarijazi, A., Khanahmad, N., Damavandi, M.S., Sadeghi, P., Khanahmad, H., 2020. The Molecular Story of COVID-19; NAD+ Depletion Addresses all Questions in this Infection. Preprints. https://doi.org/10.20944/preprints202003.0346.v1 2020030346.
- Kumar, P., 2015. Role of edible mushrooms as functional foods a review. South Asian J. Food Technol. Environ. 1 (3), 211–218.
- Laffey, J.G., Matthay, M.A., 2017. Fifty years of research in ARDS. Cell based therapy for ARDS: biology and potential therapeutic value. Am. J. Respir. Crit. Care Med. https:// doi.org/10.1164/rccm.201701-0107CP.
- Liu, R.M., 2008. Oxidative stress, plasminogen activator inhibitor 1, and lung fibrosis. Antioxid. Redox Signal. https://doi.org/10.1089/ars.2007.1903.
- Marwick, J.A., et al., 2007. Oxidative stress and steroid resistance in asthma and COPD: pharmacological manipulation of HDAC-2 as a therapeutic strategy. Expert Opin. Ther. Targets https://doi.org/10.1517/14728222.11.6.745.
- Masterson, C.H., Murphy, E., Major, I., Gonzalez, H., O'Toole, D., McCarthy, S., Laffey, J.G., Rowan, N., 2019. Purified beta-glucan from the *Lentinus edodes* mushroom attenuates antibiotic resistant *Klebsiella pneumonia*-induced pulmonary sepsis. Am. J. Respir. Crit.

Care Med. 199, A1222 2019. https://www.atsjournals.org/doi/abs/10.1164/ajrccm-conference.2019.199.1\_MeetingAbstracts.A1222.

- McCarty, M.F., DiNicolantonio, J.J., 2020. Nutraceuticals have potential for boosting the type 1 interferon response to RNA viruses including influenza and coronavirus. Prog. Cardiovasc. Dis. https://doi.org/10.1016/j.pcad.202.02.007.
- Morales, D., et al., 2019. Isolation and comparison of α- and β-D-glucans from shiitake mushrooms (*Lentinula edodes*) with different biological activities. Carbohydrate Polymers. Elsevier Ltd, p. 115521. https://doi.org/10.1016/i.carbpol.2019.115521.
- Netea, M.G., et al., 2008. An integrated model of the recognition of *Candida albicans* by the innate immune system. Nat. Rev. Microbiol. https://doi.org/10.1038/nrmicro1815.
- Pelizon, A.C., et al., 2005. Immunomodulatory activities associated with  $\beta$ -glucan derived from Saccharomyces cerevisiae. Physiol. Res. 54, 557–564.
- Rahman, I., MacNee, W., 2000. Oxidative stress and regulation of glutathione in lung inflammation. Eur. Respir. J. https://doi.org/10.1034/j.1399-3003.2000.016003534.x.
- Rezoagli, E., Fumagalli, R., Bellani, G., 2017. Definition and epidemiology of acute respiratory distress syndrome. Ann. Transl. Med. https://doi.org/10.21037/atm.2017.06.62.
- Rowan, N.J., Laffey, J.G., 2020. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from coronavirus disease (COVID-19) pandemic – case study from the Republic of Ireland. Sci. Total Environ. 725, 138532 (Online pre-print 5th April, 2020). https://www.sciencedirect. com/science/article/pii/S0048969720320453?via%3Dihub.
- Smith, J.E., Rowan, N.J., Sullivan, R., 2002. Medicinal mushrooms: a rapidly developing area of biotechnology for cancer therapy and other bioactives. Biotechnol. Lett. 24 (22), 1839–1845.
- Stone, B.A., 2009. Chemistry of β-glucans. Chemistry, Biochemistry, and Biology of 1–3 Beta Glucans and Related Polysaccharides. https://doi.org/10.1016/B978-0-12-373971-1.00002-9.
- Sullivan, R., Smith, J.E., Rowan, N.J., 2006. Medicinal mushrooms and cancer therapy: translating a traditional practice into Western medicine. Perspect. Biol. Med. 49 (2), 159–170.
- Taufek, N.M., Harith, H.H., Hafiz, M., Zulllhamd, A. R., Rowan, N., Wan-Mohtar W.A.A.Q.I., 2020. Performance of Mycelial Biomass and Exopolysaccharide from Malaysian *Ganoderma lucidum* for the Fungivore Red Hybrid Tilapia (*Oreochromis* Sp.) in Zebrafish Embryo. vol 17 https://doi.org/10.1016/j.aqrep.2020.100322.
- Vaclav, V., et al., 2013. Placebo-driven clinical trials of transfer point glucan #300 in children with chronic respiratory problems: III. Clinical findings. Am. J. Immunol. https:// doi.org/10.3844/ajisp.2013.88.93.
- Vermes, I., Haanen, C., Reutelingsperger, C., 2000. Flow cytometry of apoptotic cell death. J. Immunol. Methods https://doi.org/10.1016/S0022-1759(00)00233-7.
- Willment, J.A., et al., 2003. Dectin-1 expression and function are enhanced on alternatively activated and GM-CSF-treated macrophages and are negatively regulated by IL-10, dexamethasone, and lipopolysaccharide. J. Immunol. https://doi.org/10.4049/ jimmunol.171.9.4569.
- Zhang, Y., et al., 2011. Advances in lentinan: isolation, structure, chain conformation and bioactivities. Food Hydrocoll. https://doi.org/10.1016/j.foodhyd.2010.02.001.
- Zhang, W., Zhao, Y., Zhang, F., Wang, Q., Li, T., Liu, Z., Wang, J., Qin, Y., Zhang, X., Yan, X., Zeng, Z., Zhang, S. 2020. The use of anti-inflammatory drugs in the treatment of people with severe coronavirus disease 2019 (COVID-19): the perspectives of clinical immunologists from China, Clin. Immunol. (published online 25 March, 2025) https:// doi.org/https://doi.org/10.1016/j.clim.2020.108393.
- Zheng, R., et al., 2005. Characterization and immunomodulating activities of polysaccharide from *Lentinus edodes*. Int. Immunopharmacol. https://doi.org/10.1016/j. intimp.2004.11.011.
- Zhu, F., et al., 2015. β-Glucans from edible and medicinal mushrooms: characteristics, physicochemical and biological activities. J. Food Compos. Anal. https://doi.org/ 10.1016/j.jfca.2015.01.019.
- Zi, Y., et al., 2019. Lentinan inhibits oxidative stress and inflammatory cytokine production induced by benzo(a)pyrene in human keratinocytes. J. Cosmet. Dermatol. https://doi.org/10.1111/jocd.13005.

Contents lists available at ScienceDirect





# Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

# Immunomodulatory activity of $\beta$ -glucan polysaccharides isolated from different species of mushroom – A potential treatment for inflammatory lung conditions

Emma J. Murphy <sup>a,b,1</sup>, Emanuele Rezoagli <sup>c,d,e,\*,1</sup>, Robert Pogue <sup>a,f</sup>, Bianca Simonassi-Paiva <sup>a</sup>, Ismin Izwani Zainol Abidin <sup>g</sup>, Gustavo Waltzer Fehrenbach <sup>a</sup>, Emer O'Neil <sup>a</sup>, Ian Major <sup>g</sup>, John G. Laffey <sup>c,d</sup>, Neil Rowan <sup>a</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Athlone, Ireland

<sup>c</sup> Lung Biology Group, Regenerative Medicine Institute at CURAM Centre for Medical Devices, School of Medicine, National University of Ireland Galway, Galway, Ireland

<sup>d</sup> Anaesthesia and Intensive Care Medicine, University Hospital Galway, Galway, Ireland

<sup>e</sup> Department of Medicine and Surgery, University of Milan-Bicocca, Monza, Italy

<sup>f</sup> Post-Graduate Program in Genomic Sciences and Biotechnology, Catholic University of Brasilia, Brazil

<sup>8</sup> Materials Research Institute, Athlone Institute of Technology, Athlone, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- There is biological variance in glucan content between different mushroom species.
- β-glucans extracts reduce inflammatory biomarkers in macrophages after LPS and Cytomix injury.
- β-glucans extracts reduce inflammatory cytokine secretion in in-vitro lung cells after injury with inflammatory biomarkers.

#### ARTICLE INFO

Article history: Received 6 August 2021 Received in revised form 29 November 2021 Accepted 30 November 2021 Available online 4 December 2021

Editor: Lotfi Aleya

*Keywords*: β-Glucans 

 Weith of the set of the

#### ABSTRACT

Acute respiratory distress syndrome (ARDS) is the most common form of acute severe hypoxemic respiratory failure in the critically ill with a hospital mortality of 40%. Alveolar inflammation is one of the hallmarks for this disease.  $\beta$ -Glucans are polysaccharides isolated from a variety of natural sources including mushrooms, with documented immune modulating properties. To investigate the immunomodulatory activity of  $\beta$ -glucans and their potential as a treatment for ARDS, we isolated and measured glucan-rich polysaccharides from seven species of mushrooms. We used three models of in-vitro injury in THP-1 macrophages, Peripheral blood mononuclear cells (CD14 + ) (PMBCs) isolated from healthy volunteers and lung epithelial cell lines. We observed variance between  $\beta$ -glucan content in extracts isolated from seven mushroom species. The extracts with the highest  $\beta$ -glucan content found was *Lentinus edodes* which contained 70% w/w and *Hypsizygus tessellatus* which contained 80% w/w with low levels of  $\alpha$ -glucan. The

E-mail address: emanuele.rezoagli@unimib.it (E. Rezoagli).



<sup>&</sup>lt;sup>b</sup> Department of Graduate Studies, Limerick Institute of Technology, Limerick, Ireland

<sup>\*</sup> Corresponding author at: Department of Medicine and Surgery, University of Milan-Bicocca, Monza, Italy.

<sup>&</sup>lt;sup>1</sup> E.J. Murphy and E. Rezoagli equally contributed.

THP-1 macrophages Lung injury ARDS Medicinal mushrooms One-health extracts had the ability to induce secretion of up to 4000 pg/mL of the inflammatory cytokine IL-6, and up to 5000 pg/mL and 500 pg/mL of the anti-inflammatory cytokines IL-22 and IL-10, respectively, at a concentration of 1 mg/mL in THP-1 macrophages. In the presence of cytokine injury, IL-8 was reduced from 15,000 pg/mL to as low as 10,000 pg/mL in THP-1 macrophages. After insult with LPS, phagocytosis dropped from 70–90% to as low 10% in CD14 + PBMCs. After LPS insult CCL8 relative gene expression was reduced, and IL-10 relative gene expression increased from 50 to 250-fold in THP-1 macrophages. In lung epithelial cells, both A549 and BEAS-2B after IL-1 $\beta$  insult, IL-8 levels dropped from 10,000 pg/mL to as low as 6000 pg/mL. TNF- $\alpha$  levels dropped 10-fold from 100 pg/mL to just below 10 pg/mL. These results demonstrate the therapeutic potential of  $\beta$ -glucans in inflammatory lung conditions. Findings also advance bio-based research that connects green innovation with One Health applications for the betterment of society.

#### 1. Introduction

Acute respiratory distress syndrome (ARDS), is the most common form of acute severe hypoxemic respiratory failure in the critically ill (Rezoagli et al., 2017). The syndrome is defined by: acute onset of hypoxemia (PaO2:FiO2 ratio <300) and bilateral pulmonary opacities not explained by cardiac failure or fluid overload (Bellani et al., 2016). ARDS is a diffuse inflammatory reaction and can be characterised by an explosive acute inflammatory response in lung parenchyma (Crimi and Slutsky, 2004), impairing the principal function of gas exchange, which can lead to hypoxaemia. Treatment is mainly focused on clinical management as there remains no effective direct pharmacological therapy for this condition (Rezoagli et al., 2019). There is an urgent need for treatment as mortality and morbidity are unacceptably high at 40% (Horie et al., 2020). Furthermore, infection by severe acute respiratory syndrome coronavirus-2 (SARS-CoV-2) has led to further incidences of COVID-19-related ARDS, which is associated with 70% of fatal cases (Rezoagli et al., 2021; Li et al., 2020; Zhou et al., 2020). In both ARDS and in COVID-19-induced ARDS there is a marked increase in serum levels of inflammatory cytokines and chemokines, which is a major contributor to disease severity and ultimately death (G. Chen et al., 2020; X. Chen et al., 2020; Huang et al., 2020; Mehta et al., 2020; Qin et al., 2020). The pathophysiology of ARDS is associated with numerous target immune cells including macrophages. The lung microenvironment during injury determines the functional phenotype of macrophages, which can promote either wound healing or inflammation (Nanchal and Truwit, 2018). Thus, in developing potential therapeutics it is important to understand the potential effects of these therapeutics on lung tissue and on macrophages. An ideal treatment for this condition would aim at reducing the effects of the proinflammatory cascade and would seek to maximize the anti-inflammatory immunomodulatory response (Zambelli et al., 2021).

β-Glucans are defined as complex polysaccharides that are found in an abundance of sources including fungi, yeast, grain, bacteria, and algae (Murphy et al., 2021). β-Glucans can be classified structurally as either 1,3 1,4-linked or 1,3 1,6-linked, which is dependent on their source (Cui et al., 2011; Pogue et al., 2021; Murphy et al., 2020). These molecules are widely marketed as biologically active molecules (bioactives) (Wang et al., 2017a). There are over 200 clinical trials registered for their use for a range of applications. There are also licenced drugs containing β-glucans on the market since 1980 in Japan, for the treatment of cancer (Novak and Vetvicka, 2008; Takeshita et al., 1991; Yang et al., 2019). β-Glucans as pharmaceutical agents have also been authorised in several countries, including the United States of America, Canada, Finland, Sweden, China and Korea (van Steenwijk et al., 2021). The diverse functional effects of these molecules include alteration of lipid and glucose metabolism, cholesterol reduction, obesity regulation and reduction of cardiovascular and diabetic risk, modulating the gut microbiome, altering lipid and glucose metabolism and beneficial effects on gastrointestinal conditions such as irritable bowel syndrome (Drozdowski et al., 2010; Maki et al., 2003; McRorie and McKeown, 2017; Sima et al., 2018; Tiwari and Cummins, 2011). β-Glucans, specifically from non-cereal sources, are widely documented for their immunomodulatory properties, with the ability to stimulate the immune response and initiate inflammatory properties, and to promote resistance to infections (Ooi and Liu, 2012). (Bohn and BeMiller, 1995).

Mushroom-derived  $\beta$ -glucans are the most potent immune modulators (Borchers et al., 1999; Ooi and Liu, 2012; Lorenzen and Anke, 1998; Ooi and Liu, 1999; Tzianabos, 2000; Wasser and Weis, 1999). Moreover, they have demonstrated therapeutic effects in alleviating infective respiratory conditions (Fuller et al., 2012; Jesenak et al., 2013; Yamauchi et al., 2008). They have also been documented to reduce pro-inflammatory cyto-kines, increase anti-inflammatory cytokines, increased formation of antiox-idants as well reduction of inflammatory cells in preclinical lung injury models (Bedirli et al., 2007; Jedinak et al., 2011; Johnson et al., 2009; Kofuji et al., 2012; Soltys and Quinn, 1999; Yamada et al., 2007). These beneficial effects can also be seen in clinical trials. When patients were administered  $\beta$ -glucans for the prevention of nosocomial pneumonia and sepsis, the treatment group compared to the control group had lower incidences of pneumonia as well as a lower mortality rate (De Felippe et al., 1993).

We have previously investigated the effects of a commercial  $\beta$ -glucan and an in-house extract of  $\beta$ -glucans from the mushroom Lentinus edodes (Masterson et al., 2020; Murphy et al., 2020a, 2020b, 2020c). Specifically, Murphy et al. (2019) showed that  $\beta$ -glucans from the same mushroom, one isolated by hot water extraction and one sourced commercially had different effects, namely reduction in inflammatory cytokines, reduction in phagocytic activity of macrophages after LPS insult and reduction of inflammatory response in in-vitro lung cells. Thus, to continue this work and understand the potential immunomodulatory properties of other mushroom β-glucans as a potential treatment for inflammatory lung conditions like ARDS we decided to replicate the assays and include additional test parameters. In the current study, we first extracted and measured β-glucans from seven species of mushroom to determine BRM variance among species by applying them to a monocytic cell line and an in-vitro lung injury model. Second, we isolated CD14+ monocytes from healthy volunteers and exposed the cells to the extracts, then measured phagocytic activity. Third, we simulated an injurious environment on two types of alveolar cell lines using IL-1 $\beta$  and measured cytokine expression. Finally, we extended this assay to a monocytic cell line, which was inflamed with different insults (LPS and cytomix). It has recently emerged that macrophages are reduced and equally as inflamed as lung cells during COVID-19 infection. Therefore, after injury we measured cytokine release, gene expression, and phagocytosis of these cells to determine immune-modulatory potential in an inflammatory micro-environment.

#### 2. Materials & methods

Commercial Lentinan (CLE) was sourced from Carbosynth (FL33321, Compton, Berkshire, UK). Fruiting bodies of mushrooms were kindly gifted by Garryhinch Wood Exotics Ltd. Garryhinch, Portarlington, Co Offaly, Ireland. The fruiting body of *Agaricus blazeii* was kindly gifted by Professor Leo van Griensven, Wageningen University, The Netherlands. Other species of mushroom included; *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), Pleurotus eryngii (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).

#### 2.1. β-Glucan extraction

To extract  $\beta$ -glucans from the fruiting bodies of the mushrooms, the method used previously by Murphy et al. (2019) was employed. Briefly –

the fruiting bodies were washed and dried. The samples were blended into a fine powder. Roughly, 100 g of dried blended biomass was placed in 1 Litre of water and autoclaved. After autoclaving the polysaccharides were precipitated from supernatant using 100% Ethanol. Precipitates were dried and solubilised in PBS for analysis.

#### 2.2. β-Glucan quantification

Extracts were analysed for 1-3 1-6  $\beta$ -glucan content using the Megazyme yeast and mushroom kit (K-YBGL) (Megazyme Ltd., Bray, Co. Wicklow, Ireland). Assays were carried out according to manufacturer's instructions. After milling, samples were placed in 12 M H<sub>2</sub>SO<sub>4</sub> at -4 °C for 2 h to solubilize the  $\beta$ -glucans. Samples were then hydrolysed in 2 M H<sub>2</sub>SO<sub>4</sub> at 100 °C for a further 2 h. Any remaining  $\beta$ -glucan fragments were quantitatively hydrolysed to glucose using a mixture of exo-1,3- $\beta$ -glucanase and  $\beta$ -glucosidase which gives a measurement of total  $\beta$ -glucan content after substrate addition. The  $\alpha$ -glucan content of the sample was determined by hydrolysing specifically to D-glucose and D-fructose. Glucose was measured with amyloglucosidase and invertase using a glucose oxidase peroxidase GOPOD reagent.  $\beta$ -Glucan content was determined by the difference between the two measurements.

#### 2.3. Blood donor cell collection

Blood sample collection was approved by the Athlone Institute of Technology Ethics Committee. Blood samples were obtained from healthy volunteers for isolation of immune cells. A total of 15 mL was collected from each donor. Individual cells were isolated from 5 mL aliquots of collected blood. Samples were magnetically labelled with whole blood microbeads (Miltenyi Biotec, Germany) to isolate cells based on specific surface molecules according to the manufacturer's instructions, using the autoMACS separator (Miltenyi Biotec).

#### 2.4. Cell culture

A549 cells (used at passage 90), BEAS-2B cells (used at passage 10), and THP-1 monocyte cells (used at passage 20), were obtained from the American Type Culture Collection (ATCC, Rockville, MD, USA). Cells were cultured in RPMI-1640 (Sigma-Aldrich, St. Louis, MO, USA), supplemented with 10% fetal calf serum (Sigma-Aldrich), 1% penicillin G (100 U/mL) and streptomycin (100  $\mu$ g/mL) solution (Sigma-Aldrich), at 37 °C a 5% CO<sub>2</sub> environment. For differentiation into macrophages, THP-1 monocyte cells were treated with phorbol 12-myristate 13-acetate (PMA) for differentiation into THP-1 macrophages. (Peprotech EC, London, UK), at a concentration of 100 ng/mL, for 48 h.

#### 2.5. CD14+ PBMCs

CD14+ cells were positively isolated based on the surface molecule CD14, which is primarily found on monocytes (Shin et al., 2019). Isolated cells were cultured in RPMI-1640 (Sigma-Aldrich, St. Louis, MO, USA), supplemented with 10% fetal calf serum (Sigma-Aldrich), 1% penicillin G (100 U/mL)/streptomycin (100  $\mu$ g/mL) solution (Sigma-Aldrich) and 50 ng/mL of macrophage colony stimulating growth factor (MCSGF) (RnD Systems MN, USA) at 37 °C in a 5% CO<sub>2</sub> environment.

#### 2.6. Cell injury and $\beta$ -glucan treatment

All cell types were treated with 1 mg/mL of  $\beta$ -glucan in PBS based on other published work (Jung et al., 2007; Murphy et al., 2020a, 2020b, 2020c; Sari et al., 2020; Sivinski et al., 2020). For injury assays THP-1 PMA differentiated macrophage cells were seeded at a density of  $4 \times 10^5$  cells/well in 96-well plates, and 24 h later were injured with two different types of insult: either LPS (100 ng/mL) (Sigma) or cytomix (TNF- $\alpha$ , IFN- $\Upsilon$ , IL-1 $\beta$ ), at 25 ng/mL (Immunotools), in RPMI supplemented with 1% penicillin/streptomycin. After 24 h cells were washed three times in PBS and

treated with 1 mg/mL of extracts for a further 24 h before analysis. Pulmonary alveolar type II A549 cells were seeded at a density of 4  $\times$  10<sup>5</sup> cells/ well in 96 well plates. After 24 h cells were injured with 1 ng/mL of IL-1 $\beta$  (Peprotech, Rocky Hill, NJ) in RPMI supplemented with 1% penicillin/ streptomycin.

#### 2.7. Enzyme linked immunosorbent assay (ELISA)

A human Duoset sandwich ELISA kit (RnD Systems MN, USA) was used to measure cytokine levels in the medium after  $\beta$ -glucan exposure. All ELISA assays were performed according to the manufacturer's instructions. Results were expressed either in pg/mL or in ng/mL.

#### 2.8. Phagocytosis assays

To determine Phagocytic activity, THP-1 macrophages (PMA differentiated) and CD14+ cells were seeded into 96-well plates at 4 × 105 cells/ well. After 24 h, cells were either injured or treated with PBS. After a further 2 h, cells were treated with  $\beta$ -glucan extracts. After a further 24 h cells were washed with PBS and incubated with Alexa Fluor 488conjugated E.coli (K-12 strain) Bioparticles (E13231; Life Technologies) for 2 h, after which cells were washed three times with PBS to remove residual particles before resuspension in FACS flow buffer and measured for fluorescent particles by flow cytometry (Miltenyi Biotec, Germany).

#### 2.9. RNA extraction

For RNA extraction from THP-1 macrophage cells, Media was removed, the cells were washed  $3 \times$  with PBS, and RNA was extracted using the Purelink RNA Mini kit (Thermo-Fisher), according to the manufacturer's instructions. RNA was analysed using a NanoDrop 1000 spectrophotometer (Thermo Fisher Scientific, DE, USA) to determine RNA concentrations and A260/A280 ratios.

#### 2.10. cDNA synthesis and real-time PCR

cDNA was prepared for the replicate samples using the SensiFAST cDNA Synthesis kit (Bioline), according to the manufacturer's instructions. RNA input for all samples was normalized to the sample, so that 325.5 ng of total RNA were used in each reaction. Real-time quantitative PCR was performed using pre-designed TaqMan Gene Expression Assays for the respective genes, together with the TaqMan Gene Expression Master Mix (Thermo-Fisher). The transcripts examined were: TLR2, IL-10, CCL8, CLEC-7a and MCSGF. Reactions were carried out on the LightCycler 96 equipment (Roche), using the GAPDH transcript as endogenous control. Relative gene expression was calculated using the 2^-ddCq method.

#### 2.11. Statistical analysis

Continuous data were expressed as mean and standard error of the mean (SEM). Differences of continuous variables between species of mushrooms and PBS and injury (i.e. LPS or Cytomix or IL-1  $\beta$ ) were assessed by one-way analysis of variance for independent measures. Post-hoc comparisons were investigated by controlling the False Discovery Rate using the two-stage step-up method of Benjamini, Krieger and Yekutieli test. Statistical significance was considered with a p-value < 0.05 (two-sided). Statistical analyses were performed using STATA/MP 16.0 for Windows (StataCorp LLC, College Station, TX 77845, USA) and GraphPad Prism 8 for Windows (Version 8.0.2, FraphPad Software, Inc.).

#### 3. Results

#### 3.1. β-Glucans quantification

 $\beta$ -glucans were extracted from seven species of mushrooms as previously described (Murphy et al., 2020a, 2020b, 2020c). After extraction

and isolation, the Megazyme 1,3 1,6 kit was used to determine the concentration of  $\alpha$ - and  $\beta$ -glucans.

There was variance between species as can be seen in the relative concentrations of  $\beta$ -glucan and  $\alpha$ -glucan displayed in Fig. 1. P.E appeared to have the purest concentration of  $\beta$ -glucan compared to other extracts. A.B and P.O appeared to have high levels of contaminating  $\alpha$ -glucan present. H.T yielded the highest concentration of  $\beta$ -glucan with little contaminating  $\alpha$ -glucan. H.T yielded the highest concentration of  $\beta$ -glucan with little contaminating  $\alpha$ -glucan.

#### 3.2. Effect of $\beta$ -glucans on macrophages

#### 3.2.1. The direct effect of $\beta$ -glucans on THP-1 macrophages

To understand the effects of  $\beta$ -glucan extracts on macrophages, THP-1 macrophages were treated with each of the extracts and cytokine secretion levels were measured by an ELISA. Results show (Fig. 2) that each extract had a different effect on the cytokine release profile from THP-1 macrophages. Extracts had the potential to increase secretion of both inflammatory cytokines (IL-6, IL-8, TNF- $\alpha$ ) and anti-inflammatory cytokines (IL-10 and IL-22). CLE is a commercial source of  $\beta$ -glucan, and its extraction method is unknown; all other extracts were processed as described in the methods Section 2.1. Although the extracts have different effects on the cytokine secretion profile, the pattern was generally similar except for CLE. CLE induced lower secretion levels of IL-6 (Panel A), IL-22 (Panel C), and IL-10 (Panel E), compared to the other extracts and lower levels of IL-2 (Panel F) compared to the PBS control. The remaining extracts increased IL-6, TNF-α (Panel B) and IL-10 secretion and maintained IL-22 and IL-2 compared to PBS control. CLE and some of the extracts (L.E, C.L.E and P. C) appeared to increase the secretion of the chemokine IL-8 compared to control (Panel D).

## 3.2.2. Effect of $\beta$ -glucans on phagocytosis by THP-1 and CD14 + PBMC macrophages

To compare the effects of  $\beta$ -glucan extracts on a macrophage cell line (THP-1) and on fresh PBMCs (CD14+), cells were treated with extracts, and after 24 h phagocytic activity was measured and displayed in Fig. 3. Results varied: Panel A shows the effects on THP-1 macrophages; LPS significantly increased phagocytosis relative to untreated cells. Four extracts (P. M, P.C, P. E and H.T) were able to significantly reduce phagocytosis, with PM showing by far the greatest reduction. CD14+ PBMCs (Fig. 3, Panel B) showed varying responses to the extracts in terms of phagocytosis, as expected due to donor variability. The extract A.B significantly reduced the



Fig. 1. The percentage w/w  $\alpha$ -glucan and  $\beta$ -glucan content in mushroom extracts using Megazyme. Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota* microspora (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus* eryngii (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).

phagocytic activity, although the overall percentage phagocytosis was low in these cells. The extracts showed a tendency toward reduction in PBMCs, however, the results do not show significance, potentially due to donor variability.

To gain a further understanding of the mechanism, expression levels of four genes were measured. Two genes were related to cytokine/chemokine response: IL-10 (Fig. 4; Panel B), and CCL8 (IL-8; Fig. 4, Panel D). Two genes corresponded to cell surface ligands associated with  $\beta$ -glucan recognition: toll-like receptor 2 TLR-2 (Fig. 4, Panel A), and dectin-1 (CLEC7a, Fig. 4, Panel C).

Expression of TLR2 in THP-1 macrophages showed no significant difference in induction between LPS and the  $\beta$ -glucans. However, compared to PBS control, P.O induced expression of TLR2. All the extracts significantly inhibited the relative gene expression levels of CLEC7a compared to both PBS control and LPS in THP-1 macrophages. IL-10 expression showed no significant increase in gene expression compared to controls except for with P.M. The extracts did not significantly induce the expression of CCL8 compared to PBS controls and induction was significantly lower compared to LPS.

To determine the effect of  $\beta$ -glucans on macrophages after injury, THP-1 macrophages were injured with Cytomix (IL-1 $\beta$ , TNF- $\alpha$  & IFN- $\gamma$ ), and then treated with  $\beta$ -glucan extracts (Fig. 5). ELISA assay results show the  $\beta$ -glucan extracts from L.E, P.E, H.T and A.B significantly increased the secretion of IL-6 after insult (Panel A). P.O and P.C did not have the same induction profile as when directly treated with injury (Fig. 2 Panel A), which induced ~2000 pg/mL secretory levels of IL-6. After insult with cytomix P. O and P.C induced ~1000 pg/mL secretory levels of IL-6. IL-8 secretion was increased after cytomix treatment alone (Fig. 5, Panel B). However, the strongest inducers of IL-8 secretion with direct treatment were L.E, C.L.E and P.C (Fig. 2 Panel D), all of which significantly reduced the secretion of this inflammatory chemokine after injury except for LE which was not significant. There was no significant effect of  $\beta$ -glucan treatments on TNF- $\alpha$  secretion after cytomix insult (Fig. 5, Panel C).

Both THP-1 macrophages and PBMCs were analysed for phagocytic activity (Fig. 6) after injury with LPS. The THP-1 macrophages after injury (Fig. 6, Panel A) had a very similar response to those with  $\beta$ -glucan extracts alone (Fig. 3 Panel A) compared with treatment after injury. When CD14 + cells were treated with  $\beta$ -glucan extracts, only A.B significantly reduced phagocytosis (Fig. 3 Panel B). However, when administered after LPS, all  $\beta$ -glucan extracts reduced percentage phagocytosis (Fig. 6, panel B). Panel C displays the phagocytosis percentage of THP-1 macrophages after cytomix insult; C.L.E, P.M, P.O, P.C and P.E reduced the phagocytic activity after insult and treatment.

Inflammatory gene expression and anti-inflammatory gene expression markers were analysed after both types of injury, as displayed in Fig. 7. The inflammatory marker CCL8 was measured after LPS injury (Panel A) and cytomix (Panel B). In the presence of LPS, CCL8 relative gene expression was significantly reduced after treatment with all extracts. In the presence of cytomix insult, L.E, P.O, P.C and H.T all increased the relative gene expression of CCL8. The anti-inflammatory marker IL-10 was measured after LPS injury (Panel C) and cytomix (Panel D). After LPS insult L.E, H.T and A.B all significantly increased the expression of IL-10 gene compared to injury alone. All the extracts increased the expression of IL-10 compared to injury alone.

#### 3.3. Effect of $\beta$ -glucans in an in-vitro lung injury model

In previous work carried out by this group, we established that  $\beta$ -glucan extracts (L.E) have the potential to reduce inflammation in alveolar A549 cell lines (Murphy et al., 2019). To further expand on this work, the same assays were repeated with another alveolar cell line BEAS-2B, using six new extracts.

#### 3.3.1. A549 cells

To determine the direct effects of the  $\beta$ -glucan extracts on lung cells, extracts were incubated with A549 cells for 24 h and supernatant was



Fig. 2. The effect of the β-glucan extracts on cytokine expression in THP-1 macrophages (PMA differentiated) measured using ELISA. Panel A; IL-6, Panel B; TNF-α, Panel C; IL-22, Panel D; IL-8, Panel E; IL-10 Panel F; IL-2. p < 0.05 versus PBS. Cells were treated with 1 mg/mL of extracts for 24 h before cytokine analysis. Phosphate buffer saline (PBS), Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).

measured for cytokines. Results are displayed in Fig. 8. Panel A shows the release of IL-6 after treatment; P.M, P.A, P.E, H.T and A.B all significantly induce the secretion of IL-6 compared to PBS control. Panel B shows the secretion of IL-8. All the extracts induced the secretion of IL-8 from A549 cells with respect to PBS control except for P.E. When A549 cells were treated with the extracts, TNF- $\alpha$  secretion was increased with L.E, P.O, P.E, H.T and A.B with respect to PBS control (Panel C).

A549 cells were then treated with IL-1 $\beta$  to induce a cytokine injury. Subsequently the cells were treated with the  $\beta$ -glucans extracts and the cytokine analysis was repeated, as displayed in Fig. 8. Panel D shows that C.LE., P.M and P.A slightly reduced IL-6 secretion (though not significantly), after insult except for L.E which increases secretion with respect to injury alone. Panel E shows that after insult L.E, C.L.E, P.O, P.E and A. B reduce the section of IL-8, which is an opposite response to when cells are treated in the absence of injury (Fig. 8; Panel B). The extracts P.E, H.T and A.B significantly reduced TNF- $\alpha$  secretion after IL-1 $\beta$  insult (Panel F) which is again an opposite response to when the cells are treated alone with extracts (Panel C).

#### 3.3.2. BEAS-2B

To understand if the  $\beta$ -glucan extracts would have a similar effect in another lung epithelial cell line, BEAS-2B cells were treated with the  $\beta$ glucans extracts. The supernatant was then measured for cytokine secretion. Results are displayed in Fig. 9. All extracts induced the secretion of IL-6 (Panel A), and IL-8 (Panel B) with respect to PBS control. Panel C shows the secretion of TNF- $\alpha$  after treatment, C.L.E increased secretion but the other extracts had no effect.

Like the A549 cells, BEAS-2B cells were then treated with IL-1 $\beta$  to induce a cytokine injury; cells were then treated with the  $\beta$ -glucans extracts and the cytokine analysis was repeated, as displayed in Fig. 9. There was no effect on IL-6 secretion (Panel D). Panel E shows that after insult C.L.E, P.A, P.E, H.T and A.B reduced the section of IL-8 which is an opposite response to when cells are treated in the absence of injury (Fig. 9; Panel B). The extracts P.E, H.T and A.B significantly reduced TNF- $\alpha$  secretion after IL-1 $\beta$  insult (Fig. 9; Panel F) however these levels were the same in the absence of injury (Fig. 9; Panel C) suggesting expression levels are maintained in the presence of injury.



Fig. 3. The effects of the β-glucan extracts on percentage phagocytosis measured using flow cytometry analysis of uptake of Alexa Fluor 488-conjugated E.coli (K-12 strain) Bioparticles. THP-1 macrophages were treated with 1 mg/mL of extracts for 24 h before phagocytic analysis. Panel A; THP-1 macrophages; Panel B; CD14+ primary macrophages. Phosphate buffer saline (PBS), Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).



Fig. 4. The effect of the  $\beta$ -glucan extracts on gene expression levels of THP-1 macrophages (PMA differentiated), relative to PBS-treated cells (expression level = 1.0). Panel A; TLR2, Panel B; IL-10, Panel C; CLEC7a, Panel D; CCL8. Differences in relative gene expression \*p < 0.05 versus PBS;  $^{\#}p < 0.05$  versus LPS. Cells were treated with 1 mg/mL of extracts for 24 h before analysis of gene expression levels. Phosphate buffer saline (PBS), Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C.), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).





Science of the Total Environment 809 (2022) 152177



**Fig. 5.** The effects of the β-glucan extracts on THP-1 macrophages (PMA differentiated) after cytokine insult measured using ELISA. Panel A; IL-6, Panel B; IL-8, Panel C; TNF-α. \*p < 0.05 versus cytomix. Cells were treated with cytomix (TNF-α, IFN-Υ, IL-1β), at 25 ng/mL for 24 h, after which they were washed with PBS and treated with extracts (1 mg/mL) for 24 h before cytokine analysis. Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).

#### 4. Discussion

ARDS-associated lung injury develops a state which is marked by an increase in serum levels of inflammatory chemokines and cytokines; this is a

**Fig. 6.** The effects of the β-glucan extracts on THP-1 macrophages (PMA differentiated) after cytokine insult or LPS insult. Percentage phagocytosis measured using flow cytometry analysis of uptake of Alexa Fluor 488-conjugated *E. coli* (K-12 strain) Bioparticles. Panel A; THP-1 macrophages (PMA differentiated) treated with LPS, Panel B; CD14 + PBMCs treated with LPS. Panel C; THP-1 macrophages (PMA differentiated) treated with cytomix. \*p < 0.05 versus Cytomix or LPS. Cells were treated with cytomix (TNF-α, IFN- Y, IL-1β), at 25 ng/mL or LPS for 24 h after which they were washed with PBS and treated with extracts (1 mg/mL) for 24 h before phagocytic activity analysis. Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazeii* (A.B).



Fig. 7. The effects of the  $\beta$ -glucan extracts on THP-1 macrophages (PMA differentiated) after LPS (Panels A, C) or Cytomix (Panels B, D) injury relative to PBS-treated cells (expression level = 1.0). Panels A and B show IL-8 expression after  $\beta$ -glucan treatments, relative to insult; Panels C and D show IL-10 expression after  $\beta$ -glucan treatments, relative to insult. \*p < 0.05 versus LPS /Cytomix. THP-1 macrophages (PMA differentiated) were treated with cytomix (TNF- $\alpha$ , IFN-  $\gamma$ , IL-1 $\beta$ ), at 25 ng/mL or LPS for 24 h after which they were washed with PBS and treated with extracts (1 mg/mL) for 24 h before gene expression analysis. Commercial Lentinan (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazei* (A.B).

major contributor to disease severity and ultimately death (G. Chen et al., 2020; X. Chen et al., 2020; Huang et al., 2020; Mehta et al., 2020; Qin et al., 2020). There is currently a vast literature relating to the immunomodulatory effects of β-glucan (Rao et al., 2020), and there is a huge level of enthusiasm regarding their therapeutic potential. Thus with this in mind, there are three aims to this work. Firstly, to understand if β-glucans extracted in the same way from different species of mushroom contained the same levels of  $\beta$ -glucan content and if these extracts had the same effect on a key player in cellular immunity and response - macrophages. The second aim was to determine if the samples could elicit a response or prime immune cells, and whether there would be a difference in priming effects. Our final aim was to determine whether there may be the potential for the extracts to be used in hyperinflammatory conditions such as ARDS. To understand this, two types of in-vitro models were used - injured macrophages and lung injury models. This approach will potentially facilitate a greater understanding into the biological variance of these compounds and realising their therapeutic potential.

It is recognised in the literature that  $\beta$ -glucans from different sources can exert different biological effects, and that different extraction methods may help to optimize performance. Our experiments were performed using the same extraction method on all seven mushroom species. As such, we can compare the immunomodulatory effects of the  $\beta$ -glucan extracts by the same extraction procedure. Further studies may examine the effects of altering extraction parameters on  $\beta$ -glucan activity. Furthermore, future structural analyses would allow us to deepen the structure-function relationship. Further studies may examine the effects of altering extraction parameters on  $\beta$ -glucan activity. Furthermore, future structural analyses would allow us to deepen the structure-function relationship. The diverse mechanisms of action of  $\beta$ -glucans is unknown. There are differences in the effects of  $\beta$ -glucans that can be observed between similar preparations from the same species or source. The cellular pathways that are activated after recognition are also not fully understood. β-Glucans appear to be recognised as pathogen associated molecular patterns (PAMPs) and modulate immune function via this pathway (Brown and Gordon, 2005; Borchers et al., 1999). However, the exact mechanism by which  $\beta$ -glucans suppress inflammatory cytokines and induce anti-inflammatory cytokines are complex, and incompletely understood. With this in mind, previous work by this group, Murphy et al., 2019 investigated the differential effects of two  $\beta$ -glucan extracts in an in-vitro lung injury model and in an in-vivo model of pulmonary sepsis (Masterson et al., 2020). Once, determined fungal βglucans had immune-modulatory effects in lung injury pre-clinical models, the next advancement is to highlight other potential fungal derived  $\boldsymbol{\beta}$ glucans with immune-modulatory activity. Once identified, future studies will investigate the structure-activity relationship to gain an understanding of how these molecules elicit their effects and the pathways associated with these effects.

The Key findings of this work include - There is a variance in the levels of  $\beta$ -glucan between mushroom species extracted in the same way. This study found that *Lentinus edodes* and *Hypsizygus tessellatus* had the highest levels of  $\beta$ -glucan content when measured using the Megazyme assay.


**Fig. 8.** The effect of the β-glucan extracts on cytokine expression in A549 cells measured using ELISA. Panel A; IL-6, Panel B; IL-8, Panel C; TNF-α. The effect of the β-glucan extracts on cytokine expression in BEAS-2B cells after IL-1β insult, measured using ELISA. Panel D; IL-6, Panel E; IL-8, Panel F; TNF-α. p < 0.05 versus PBS or IL-1β. Uninjured cells were treated with extracts (1 mg/mL) for 24 h before cytokine analysis. Cells were treated with IL-1β at 1 ng/mL for 24 h after which they were washed with PBS and treated with extracts (1 mg/mL) for 24 h before cytokine analysis. Phosphate buffer saline (PBS), *Commercial Lentinan* (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazei* (A.B).

Most extracts had the ability to induce both pro and anti-inflammatory cytokines individually at a concentration of 1 mg/mL in THP-1 macrophages. In the presence of a paracrine insult of a cocktail of cytokines; IL-8 was reduced in THP-1 macrophages. Also observed was a reduction in phagocytosis in THP-1 macrophages and CD14 + macrophages in the presence and absence of injury. After LPS insult, CCL8 relative gene expression was reduced, and IL-10 gene expression was increased in THP-1 macrophages. In lung epithelial cells, the extracts had the ability to reduce two cytokines (IL-8 and TNF- $\alpha$ ) which are heavily correlated to pathogenesis of inflammation in the presence of IL-1 $\beta$ .

#### 4.1. β-Glucans quantification in mushroom species

Hot water extracts were prepared from seven species of mushroom, and  $\beta$ -glucan content was determined. Results show that although extracts were isolated by the same method, each species yielded different levels of  $\alpha$ - and

β-glucans. Although there is some evidence to suggest that α-glucans can have immune-modulating properties (Masuda et al., 2017; Okamoto et al., 2007). There is substantially more evidence to suggest that the βglucan molecule is the immune-stimulating compound found in mushrooms. These results highlight the variability between β-glucan contents in the different mushroom species. Two other studies using the same analysis procedure found variance among mushroom species (McCleary and Draga, 2016; Sari et al., 2017). Other studies have found that α- and starch glucans are usually of low abundance in cultivated mushrooms (Bak et al., 2014; Sari et al., 2017; Synytsya et al., 2008).

#### 4.2. Effects of $\beta$ -glucans on macrophages

Macrophages have the potential to intensify inflammation or exhibit regulatory repair activity during injury (Wynn and Barron, 2010).



**Fig. 9.** The effect of the β-glucan extracts on cytokine expression in BEAS-2B cells measured using ELISA. Panel A; IL-6, Panel B; IL-8, Panel C; TNF-α. The effect of the β-glucan extracts on cytokine expression in BEAS-2B cells after IL-1β insult, measured using ELISA. Panel D; IL-6, Panel E; IL-8, Panel F; TNF-α. The effect of the β-glucan extracts on cytokine expression in BEAS-2B cells after IL-1β insult, measured using ELISA. Panel D; IL-6, Panel E; IL-8, Panel F; TNF-α. p < 0.05 versus PBS or IL-1β. Uninjured cells were treated with extracts (1 mg/mL) for 24 h before cytokine analysis. Cells were treated with IL-1β at 1 ng/mL for 24 h after which they were washed with PBS and treated with extracts (1 mg/mL) for 24 h before cytokine analysis. Phosphate buffer saline (PBS), *Commercial Lentinan* (C.L.E.), *Lentinus edodes* (L.E), *Pholiota microspora* (P.M), *Pleurotus ostreatus* (P.O), *Pleurotus citrinopileatus* (P.C), *Pleurotus eryngii* (P.E), *Hypsizygus tessellatus* (H.T) and *Agaricus blazei* (A.B).

As well as variance in content there is also evidential variance in response, which is most evident in Fig. 2 Panel A, measurement of IL-6. P.M and P.O have similar levels of  $\beta$ -glucan content (Fig. 1), yet P.M induced THP-1 macrophages to produce nearly double the amount of IL-6 in comparison to P.O, according to the ELISA assay. This could be correlated to the higher levels of  $\alpha$ -glucan, in the P.O sample or to structural variances between  $\beta$ -glucans from different species. However, the high amount of  $\alpha$ -glucan present in A.B sample does not hinder its activity in stimulating IL-6 secretion.

Variance can also be seen in phagocytic activity (Fig. 3, Panel A), where some samples (P.M, P.C and H.T) reduced phagocytosis in THP-1 macrophages. A.B reduced phagocytic activity in the donor PBMCs. Other extracts had no effect on phagocytic index. The THP-1 macrophages were differentiated using PMA, and the PBMCs were differentiated using MCSGF. Thus, as they should have a high phagocytic potential in this assay, it is interesting that some of the samples appeared to reduce this. This ability is potentially useful especially in conditions where macrophages are hypersensitive, and phagocytosis is uncontrolled.

Previous research has also shown that varied sources and structures lead to a varied biological response (Bohn and BeMiller, 1995; Bose et al., 2014; Demleitner et al., 1992; Driscoll et al., 2009; Goodridge et al., 2009; Volman et al., 2008; Wang et al., 2017b). As such, our results are in agreement with the literature in that  $\beta$ -glucan from different mushroom sources can induce varied responses. Further investigation into these correlations may identify optimized  $\beta$ -glucan sources for treatment of different pathological conditions.

Dectin-1 is a type II membrane receptor, which is documented as one of the principal receptors for  $\beta$ -glucans (Baert et al., 2015). TLR 2, 4 and 6 cobind to dectin-1 after  $\beta$ -glucan recognition (Guo et al., 2015), modulating and contributing to cell responses including the release of pro and antiinflammatory cytokines and phagocytic activity (Kanjan et al., 2017). The results of the present study showed a low- to absent expression for the gene dectin-1 receptor (CLEC7a). This could be for two reasons; a limitation of this study was that the samples were taken at 24 h when the gene could be (temporarily) switched off. Secondly, dectin-1 does not recognise all  $\beta$ glucans equally; studies have shown that dectin-1 reacts differently based on structural determinants such as side-branching and size of the molecule (Adams et al., 2008). No gene expression could also be correlated to inhibition of CLEC7a, which could be correlated to a negative feedback effect. This result warrants a further timeline study to understand this mechanism.

Nonetheless these results demonstrate that the  $\beta$ -glucan samples are recognised by macrophages of a cell line lineage and from fresh PBMCs. This recognition can induce the secretion of both pro- and anti-inflammatory cytokines, reduce phagocytic activity, and alter gene expression levels reducing pro-inflammatory chemokines and increasing the secretion of the anti-inflammatory marker IL-10. Extracts increased secretion of both inflammatory cytokines (IL-6, IL-8, TNFα) and anti-inflammatory cytokines (IL- 10 and IL-22). M1 macrophage polarization is associated with the secretion of pro-inflammatory cytokines: IL-1β, IL-6, and TNF-α (Bouhlel et al., 2007). M2 macrophage polarization is associated with the secretion of anti-inflammatory cytokines IL-10 (Arora et al., 2018; Wang et al., 2014). As the  $\beta$ -glucan extracts induce the secretion of both, it is possible that they stimulate the cells into a mixed population of M1/M2 macrophages. The commercial sample C.L.E had a different effect on the cells. L.E and C.L.E are isolated from the same mushroom species, again showing the great variances between  $\beta$ -glucan samples which can be dependent on cultivation, seasonal variation as well as extraction procedure. Taken together these results show the potential of β-glucans from mushrooms to behave as biological response modifiers.

To understand the immunomodulatory effects of  $\beta$ -glucan in an inflammatory M1 phenotype-inducing environment two types of insult were used. Firstly, LPS which stimulates macrophages toward an M1 phenotype (Zheng et al., 2013) and secondly a cocktail of cytokines (cytomix) was used to stimulate an inflammatory environment (Farley et al., 2009). After insult  $\beta$ -glucan samples were added to determine if the effects of the insult could be tempered.

After treatment with cytomix, some of the  $\beta$ -glucan extracts increased the secretion of IL-6. However, after insult, some of the extracts (P.O and P.C) induced less secretion of IL-6, compared with  $\beta$ -glucan alone, thus suggesting that the immune response is reduced in the presence of an injuring agent (cytomix) (Fig. 5 Panel a). One interesting finding in this study is the reduction of phagocytosis of PBMCs after LPS insult. All  $\beta$ -glucan extracts reduced the phagocytic index in the presence of LPS to just under half of the activity of positive controls. This result demonstrates the potential of  $\beta$ -glucans to modulate macrophage activity as these cells are from healthy volunteers. There is donor variation in these samples which is to be expected; future studies would investigate this effect in larger groups of healthy volunteers. The  $\beta$ -glucans also reduced phagocytosis after cytomix insult.

Impressively, the  $\beta$ -glucan samples reduced IL-8 gene expression levels after LPS injury and increased the gene expression levels of IL-10. This demonstrates an intracellular shift from an inflammatory phenotype to an anti-inflammatory phenotype in the presence of LPS. Although IL-8 was not reduced in the presence of cytomix, IL-10 was increased, again demonstrating a shift to a more anti-inflammatory response.

During SARS-CoV-2 macrophages communicate with target cells through chemokines and phagocytic signaling (Qi et al., 2020). Macrophages respond to initial infection as a result of the inflammatory cytokines secreted by type II alveolar cells which include IL-1 $\beta$ , IL-6 and TNF- $\alpha$  (Denney and Ho, 2018). When aiming to reduce the response of macrophages in inflammatory conditions, it is also important to target the alveolar cells at the centre of the injury.

#### 4.3. Effect of $\beta$ -glucans in an in-vitro lung injury model

Cytokines and chemokines have an important role in immunity as well as in immune pathology as a dysregulated response has the potential to cause extensive tissue and organ damage, especially in the lungs (Pedersen and Ho, 2020). As SARS-Cov-2 infection is associated with the production of inflammatory cytokines we investigated the effects  $\beta$ -glucans would have in an inflammatory environment by measuring cytokine production after IL-1 $\beta$  insult on two types of alveolar cell lines; A549 and BEAS-2b. When A549 (Fig. 8) and BEAS-2b cells (Fig. 9) were treated with the  $\beta$ -glucan extracts, all inflammatory cytokines were elevated. However, in the presence of inflammatory insults, some of the inflammatory cytokines were reduced significantly. Figs. 8

and 9 Panel E and F, shows that the extracts had the ability to reduce IL-8 and TNF- $\alpha$  in both A549 cells and BEAS-2B. In reducing the cytokine expression and inflammation of lung tissue the inflammatory process can be avoided and blood gas transfer potentially unaffected or minimally affected. After injury, when the invading pathogen is eliminated, large numbers of inflammatory monocytes and macrophages can be recruited to the distal alveolar space because of chemokine gradients, this can also exceed the total number of resident macrophages (Davies et al., 2013; Galli et al., 2011).

As epithelial cells are the main source of anti-viral responses in the first 24–48 h window after infection, this is an important result. Important signals are transmitted to innate immune cells which are translated to adaptive immune responses (Geller and Yan, 2020). By firstly priming these cells with bioactives such as  $\beta$ -glucans to respond to infection, innate cells are recruited, and a memory is created for prevention of a secondary infection. More importantly, if the cells are modulated by the  $\beta$ -glucans.

As the infection is more lung-centred than multi-organ-centred (McGonagle et al., 2020).

In-vitro lung epithelial cells represent a good model to determine potential targets. This study has shown that  $\beta$ -glucan extracts from mushrooms can reduce inflammatory responses in models of in-vitro lung injury.

#### 5. Conclusions

There is a growing awareness of therapies directed to modulate the immune response in many pathological contexts. Medicinal mushrooms, which contain the complex  $\beta$ -glucans sugars have been used to treat an array of conditions for centuries including inflammatory conditions. Previously, we have demonstrated that  $\beta$ -glucans from the same mushroom isolated by different methods have differential immune-modulation abilities in an in-vitro model as well as in an invivo preclinical model. Following on from this, the current work has demonstrated the potential of  $\beta$ -glucans as immunomodulators with dual functions, firstly as immune priming agents that may bolster the capacity of the body to maintain homeostasis in the face of infectious and other challenges, and secondly to temper the immune response following infection, thus helping to avoid the serious sequelae associated with immune hyper-inflammatory response in immune and epithelial cells in inflammatory lung conditions such as ARDS. Future work will investigate relationship between the structure of β-glucans and mechanistic effects at cell and molecular levels. The main findings of this research also strongly align with emergence of green innovation for OneHealth applications (Rowan and Galanakis, 2020).

#### CRediT authorship contribution statement

Emma Murphy: conceptualization, methodology, data curation, visualization, formal analysis, writing original draft, review and editing. Emanuele Rezoagli: conceptualization, methodology, data curation, visualization, formal analysis, writing original draft, review and editing. Robert Pogue: methodology, data curation, formal analysis, review and editing. Bianca Simonassi-Paiva: methodology, formal analysis, review and editing. Ismin Izwani Zainol Abidin: methodology, formal analysis, review and editing. Gustavo Waltzer Fehrenbach: methodology, formal analysis, review and editing. Emer O'Neil: methodology. John Laffey: supervision, funding acquisition, conceptualization, review and editing. Ian Major: methodology, supervision, review and editing. Neil Rowan: funding acquisition, project administration, conceptualization, supervision, methodology, review & editing.

#### Declaration of competing interest

The authors declare no conflict of interest.

#### E.J. Murphy et al.

#### References

- Adams, E.L., et al., 2008. Differential high-affinity interaction of Dectin-1 with natural or synthetic glucans is dependent upon primary structure and is influenced by polymer chain length and side-chain branching. J. Pharmacol. Exp. Ther. 325 (1), 115–123. https:// doi.org/10.1124/jpet.107.133124.
- Arora, S., et al., 2018. Macrophages: their role, activation and polarization in pulmonary diseases. Immunobiology. Elsevier GmbH, pp. 383–396. https://doi.org/10.1016/j.imbio. 2017.11.001.
- Baert, K., Sonck, E., et al., 2015. Cell type-specific differences in β-glucan recognition and signalling in porcine innate immune cells. Dev. Comp. Immunol. 48 (1), 192–203. https:// doi.org/10.1016/j.dci.2014.10.005.
- Bak, W.C., et al., 2014. Determination of glucan contents in the fruiting bodies and mycelia of lentinula edodes cultivars. Mycobiology, 301–304 https://doi.org/10.5941/MYCO.2014. 42.3.301.
- Bedirli, A., et al., 2007. Beta-glucan attenuates inflammatory cytokine release and prevents acute lung injury in an experimental model of sepsis. Shock 27 (4), 397–401. https:// doi.org/10.1097/01.shk.0000245030.24235.fl.
- Bellani, G., et al., 2016. The LUNG SAFE study: a presentation of the prevalence of ARDS ac-
- cording to the Berlin Definition! Crit. Care https://doi.org/10.1186/s13054-016-1443-x. Bohn, J.A., BeMiller, J.N., 1995. (1→3)-β-d-glucans as biological response modifiers: a review of structure-functional activity relationships. Carbohydr. Polym. 28 (1), 3–14. https:// doi.org/10.1016/0144-8617(95)00076-3.
- Borchers, A.T., et al., 1999. Mushrooms, tumors, and immunity. Exp. Biol. Med, pp. 281–293. https://doi.org/10.3181/00379727-221-44412.
- Bose, N., et al., 2014. Differential regulation of oxidative burst by distinct β-glucan- binding receptors and signaling pathways in human peripheral blood mononuclear cells. Glycobiology 24 (4), 379–391. https://doi.org/10.1093/glycob/cwu005.
- Bouhlel, M.A., et al., 2007. PPARγ activation primes human monocytes into alternative M2 macrophages with anti-inflammatory properties. Cell Metabolism 6 (2), 137–143. https://doi.org/10.1016/j.cmet.2007.06.010 Cell Press.
- Brown, G.D., Gordon, S., 2005. Immune recognition of fungal  $\beta$ -glucans. Cell. Microbiol., 471–479. https://doi.org/10.1111/j.1462-5822.2005.00505.x. John Wiley & Sons, Ltd.
- Chen, G., et al., 2020. Clinical and immunological features of severe and moderate coronavirus disease 2019. J. Clin. Investig. 130 (5), 2620–2629. https://doi.org/10.1172/ JCI137244.
- Chen, X., et al., 2020. Macrophage polarization and its role in the pathogenesis of acute lung injury/acute respiratory distress syndrome. Inflammation Research. Springer Science and Business Media Deutschland GmbH, pp. 883–895. https://doi.org/10.1007/s00011-020-01378-2.
- Crimi, E., Slutsky, A.S., 2004. Inflammation and the acute respiratory distress syndrome. Best Pract. Res. Clin. Anaesthesiol. 18 (3), 477–492. https://doi.org/10.1016/j.bpa.2003.12. 007.
- Cui, S.W., Wang, Q., Zhang, M., 2011. β-Glucans. RSC Polymer Chemistry Series1(3). Multidisciplinary Digital Publishing Institute, pp. 319–345. https://doi.org/10.3390/encyclopedia1030064.
- Davies, L.C., et al., 2013. Tissue-resident macrophages. Nature Immunology, 986–995 https://doi.org/10.1038/ni.2705.
- Demleitner, S., Kraus, J., Franz, G., 1992. Synthesis and antitumour activity of sulfoalkyl derivatives of curdlan and lichenan. Carbohydr. Res. 226 (2), 247–252. https://doi.org/10. 1016/0008-6215(92)84072-Z.
- De Felippe, J., et al., 1993. Infection prevention in patients with severe multiple trauma with the immunomodulator beta 1-3 polyglucose (glucan). Surg. Gynecol. Obstet. 177 (4), 383–388.
- Denney, L., Ho, L.P., 2018. The role of respiratory epithelium in host defence against influenza virus infection. Biomed. J., 218–233. https://doi.org/10.1016/j.bj.2018.08.004.
- Driscoll, M., et al., 2009. Therapeutic potential of various  $\beta$ -glucan sources in conjunction with anti-tumor monoclonal antibody in cancer therapy. Cancer Biol. Ther. 8 (3), 218–225. https://doi.org/10.4161/cbt.8.3.7337.
- Drozdowski, L.A., et al., 2010. β-Glucan extracts inhibit the in vitro intestinal uptake of longchain fatty acids and cholesterol and down-regulate genes involved in lipogenesis and lipid transport in rats. J. Nutr. Biochem. 21 (8), 695–701. https://doi.org/10.1016/j. jnutbio.2009.04.003.
- Farley, K.S., Wang, L., Mehta, S., 2009. Septic pulmonary microvascular endothelial cell injury: role of alveolar macrophage NADPH oxidase. 296 (3), 480–488. https://doi.org/ 10.1152/AJPLUNG.90201.2008 https://doi.org/10.1152/ajplung.90201.2008. American Physiological Society.
- Fuller, R., et al., 2012. Influence of yeast-derived 1,3/1,6 glucopolysaccharide on circulating cytokines and chemokines with respect to upper respiratory tract infections. Nutrition 28 (6), 665–669. https://doi.org/10.1016/j.nut.2011.11.012.
- Galli, S.J., Borregaard, N., Wynn, T.A., 2011. Phenotypic and functional plasticity of cells of innate immunity: macrophages, mast cells and neutrophils. Nat. Immunol., 1035–1044 https://doi.org/10.1038/ni.2109.
- Geller, A., Yan, J., 2020. Could the induction of trained immunity by β-glucan serve as a defense against COVID-19? Front. Immunol. 11. https://doi.org/10.3389/fimmu.2020. 01782.
- Goodridge, H.S., Wolf, A.J., Underhill, D.M., 2009. B-glucan recognition by the innate immune system. Immunol. Rev. 230 (1), 38–50. https://doi.org/10.1111/j.1600-065X. 2009.00793.x.
- Guo, Y., et al., 2015. Leptospiral lipopolysaccharide stimulates the expression of toll-like receptor 2 and cytokines in pig fibroblasts. Anim. Sci. J. 86 (2), 238–244. https://doi. org/10.1111/asj.12254.
- Horie, S., et al., 2020. Emerging pharmacological therapies for ARDS: COVID-19 and beyond. Intensive Care Med. 46 (12), 2265–2283. https://doi.org/10.1007/s00134-020-06141-z. Springer Science and Business Media Deutschland GmbH.

- Huang, C., et al., 2020. Clinical features of patients infected with 2019 novel coronavirus in Wuhan, China. Lancet 395 (10223), 497–506. https://doi.org/10.1016/S0140-6736 (20)30183-5.
- Jedinak, A., et al., 2011. Anti-inflammatory activity of edible oyster mushroom is mediated through the inhibition of NF-kB and AP-1 signaling. Nutr. J. 10 (1). https://doi.org/10. 1186/1475-2891-10-52.
- Jesenak, M., et al., 2013. Immunomodulatory effect of pleuran (β-glucan from *Pleurotus* ostreatus) in children with recurrent respiratory tract infections. Int. Immunopharmacol. 15 (2), 395–399. https://doi.org/10.1016/j.intimp.2012.11.020.
- Johnson, E., et al., 2009. Effect of an extract based on the medicinal mushroom agaricus blazei murill on release of cytokines, chemokines and leukocyte growth factors in human blood ex vivo and in vivo. Scand. J. Immunol. 69 (3), 242–250. https://doi.org/10.1111/j. 1365-3083.2008.02218.x.
- Jung, H.K., et al., 2007. Production and physicochemical characterization of β-glucan produced by *Paenibacillus polymyxa* JB115. Biotechnol. Bioprocess Eng. 12 (6), 713–719.
- Kanjan, P., et al., 2017. Immune effects of β-glucan are determined by combined effects on Dectin-1, TLR2, 4 and 5. J. Funct. Foods 37, 433–440. https://doi.org/10.1016/j.jff. 2017.07.061.
- Kofuji, K., et al., 2012. Antioxidant activity of β -glucan. ISRN Pharmaceutics 2012, 1–5. https://doi.org/10.5402/2012/125864.
- Li, G., et al., 2020. Coronavirus infections and immune responses. J. Med. Virol. 424–432. https://doi.org/10.1002/jmv.25685.
- Lorenzen, K., Anke, T., 1998. Basidiomycetes as a source for new bioactive natural products. Curr. Org. Chem. 2, 329–364.
- Maki, K.C., et al., 2003. Food products containing free tall oil-based phytosterols and oat  $\beta$  glucan lower serum total and LDL cholesterol in hypercholesterolemic adults. J. Nutr. 133 (3), 808–813. https://doi.org/10.1093/jn/133.3.808.
- Masterson, C.H., et al., 2020. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant *Klebsiella pneumoniae*-induced pulmonary sepsis. Lett. Appl. Microbiol. 71 (4), 405–412. https://doi.org/10.1111/lam.13358. Blackwell Publishing Ltd.
- Masuda, Y., et al., 2017. Antitumor activity of orally administered maitake α-glucan by stimulating antitumor immune response in murine tumor. PLoS ONE 12 (3). https://doi.org/ 10.1371/journal.pone.0173621.
- McCleary, B.V., Draga, A., 2016. Measurement of  $\beta$ -glucan in mushrooms and mycelial products. J. AOAC Int. 99 (2), 364–373. https://doi.org/10.5740/jaoacint.15-0289.
- McGonagle, D., et al., 2020. The role of cytokines including Interleukin-6 in COVID-19 induced pneumonia and macrophage activation syndrome-like disease. Autoimmun. Rev. https://doi.org/10.1016/j.autrev.2020.102537.
- McRorie, J.W., McKeown, N.M., 2017. Understanding the physics of functional fibers in the gastrointestinal tract: an evidence-based approach to resolving enduring misconceptions about insoluble and soluble fiber. J. Acad. Nutr. Diet. 117 (2), 251–264. https://doi.org/ 10.1016/j.jand.2016.09.021.
- Mehta, P., et al., 2020. COVID-19: consider cytokine storm syndromes and immunosuppression. Lancet, 1033–1034 https://doi.org/10.1016/S0140-6736(20)30628-0.
- Murphy, E.J., et al., 2019. Immunomodulation properties of a novel β-glucan extract from the mushroom lentinus edodes in an in-vitro lung injury model. https://doi.org/10.1164/ ajrccm-conference.2019.199.1\_meetingabstracts.a2114 pp. A2114–A2114.
- Murphy, Emma J., Masterson, C., et al., 2020. β-glucan extracts from the same edible shiitake mushroom lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects — implications for coronavirus disease (COVID-19) immunotherapies. Sci. Total Environ. 732. https://doi.org/10.1016/j.scitotenv.2020.139330.
- Murphy, Emma J., et al., 2020. β-glucan metabolic and immunomodulatory properties and potential for clinical application. Journal of fungi (Basel, Switzerland). MDPI AG 6 (4), 1–33. https://doi.org/10.3390/jof6040356.
- Murphy, Emma J., Rezoagli, E., et al., 2020. B-glucan metabolic and immunomodulatory properties and potential for clinical application. J. Fungi 6 (4), 1–33. https://doi.org/ 10.3390/jof6040356.
- Murphy, Emma J., et al., 2021. β-Glucans. Encyclopedia 1 (3), 831–847. https://doi.org/10. 3390/encyclopedia1030064 2021.
- Nanchal, R.S., Truwit, J.D., 2018. Recent advances in understanding and treating acute respiratory distress syndrome [version 1; referees: 2 approved]. F1000Research https://doi. org/10.12688/f1000research.15493.1 F1000 Research Ltd.
- Novak, M., Vetvicka, V., 2008. β-glucans, history, and the present: immunomodulatory aspects and mechanisms of action. J. Immunotoxicol., 47–57 https://doi.org/10.1080/ 15476910802019045.
- Okamoto, S., et al., 2007. Inflammatory immune responses by water-insoluble α-glucans. J. Dent. Res. 86 (3), 242–248. https://doi.org/10.1177/154405910708600309.
- Ooi, V.E.C., Liu, F., 1999. A review of pharmacological activities of mushroom polysaccharides. Int. J. Med. Mushrooms 1 (3), 195–206. https://doi.org/10.1615/ intjmedmushrooms.v1.i3.10.
- Ooi, V.E.C., Liu, F., 2012. Immunomodulation and anti-cancer activity of polysaccharideprotein complexes. Curr. Med. Chem. 7 (7), 715–729. https://doi.org/10.2174/ 0929867003374705.
- Pedersen, S.F., Ho, Y.C., 2020. SARS-CoV-2: a storm is raging. J. Clin. Investig., 2202–2205 https://doi.org/10.1172/JCI137647.
- Pogue, R., et al., 2021. Exploiting immunomodulatory properties of β-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms: concise review of existing knowledge, innovation and future opportunities. Curr. Opin. Environ. Sci. Health 21. Elsevier BV, p. 100248. https://doi.org/10.1016/j.coesh. 2021.100248.
- Qi, F., et al., 2020. Single cell RNA sequencing of 13 human tissues identify cell types and receptors of human coronaviruses. Biochem. Biophys. Res. Commun. 526 (1), 135–140. https://doi.org/10.1016/j.bbrc.2020.03.044.
- Qin, C., et al., 2020. Dysregulation of immune response in patients with coronavirus 2019 (COVID-19) in Wuhan, China. Clin. Infect. Dis. 71 (15), 762–768. https://doi.org/10. 1093/cid/ciaa248.

#### E.J. Murphy et al.

- Rezoagli, E., Fumagalli, R., Bellani, G., 2017 Jul. Ann. Transl. Med. 5 (14), 282. https://doi. org/10.21037/atm.2017.06.62.
- Rezoagli, E., et al., 2019. The safety and efficiency of addressing ards using stem cell therapies in clinical trials. Stem Cell-Based Therapy for Lung Disease, 219–238. https://doi.org/10. 1007/978-3-030-29403-8\_12.
- Rezoagli, E., et al., 2021. Development of a critical care response experiences from Italy during the coronavirus disease 2019 pandemic. Anesthesiol. Clin. 39 (2), 265–284. https:// doi.org/10.1016/j.anclin.2021.02.003.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? Sci. Total Environ. 748. https://doi.org/10.1016/j.scitotenv.2020.141362.
- Sari, M., et al., 2017. Screening of beta-glucan contents in commercially cultivated and wild growing mushrooms. Food Chem. 216, 45–51. https://doi.org/10.1016/j.foodchem. 2016.08.010.
- Sari, M., et al., 2020. The birch bracket medicinal mushroom, *Fomitopsis betulina* (Agaricomycetes)-bioactive source for beta-glucan fraction with tumor cell migration blocking ability. Int. J. Med. Mushrooms 22 (1), 1–13. https://doi.org/10.1615/ IntJMedMushrooms.2019033291.
- Shin, J. Il, et al., 2019. Inflammasomes and autoimmune and rheumatic diseases: a comprehensive review. J. Autoimmun.
- Sima, P., Vannucci, L., Vetvicka, V., 2018. β-glucans and cholesterol (review). Int. J. Mol. Med., 1799–1808. https://doi.org/10.3892/ijmm.2018.3411. Spandidos Publications.
- Sivinski, S.E., et al., 2020. Development of an in vitro macrophage screening system on the immunomodulating effects of feed components. J. Anim. Sci. Biotechnol. 11 (1). https://doi.org/10.1186/s40104-020-00497-4.
- Soltys, J., Quinn, M.T., 1999. Modulation of endotoxin- and enterotoxin-induced cytokine release by in vivo treatment with  $\beta$ -(1,6)-branched  $\beta$ -(1,3)-glucan. Infect. Immun. 67 (1), 244–252. https://doi.org/10.1128/iai.67.1.244-252.1999.
- Synytsya, A., et al., 2008. Mushrooms of genus pleurotus as a source of dietary fibres and glucans for food supplements. Czech J. Food Sci. 26 (6), 441–446. https://doi.org/10. 17221/1361-cjfs.
- Takeshita, K., et al., 1991. Diversity of complement activation by lentinan, an antitumor polysaccharide, in gastric cancer patients. Nippon Geka Gakkai zasshi 92 (1), 5–11.
- Tiwari, U., Cummins, E., 2011. Meta-analysis of the effect of  $\beta$ -glucan intake on blood cholesterol and glucose levels. Nutrition 27 (10), 1008–1016. https://doi.org/10.1016/j.nut. 2010.11.006.

- Tzianabos, A.O., 2000. Polysaccharide immunomodulators as therapeutic agents: structural aspects and biologic function. Clin. Microbiol. Rev., 523–533. https://doi.org/10.1128/ CMR.13.4.523-533.2000.
- van Steenwijk, H.P., Bast, A., de Boer, A., 2021. Immunomodulating effects of fungal betaglucans: from traditional use to medicine. Nutrients 13 (4). https://doi.org/10.3390/ nu13041333.
- Volman, J.J., Ramakers, J.D., Plat, J., 2008. Dietary modulation of immune function by βglucans. Physiol. Behav. 94 (2), 276–284. https://doi.org/10.1016/j.physbeh.2007.11. 045.
- Wang, N., Liang, H., Zen, K., 2014. Molecular mechanisms that influence the macrophage M1-M2 polarization balance. Front. Immunol. 614. https://doi.org/10.3389/fimmu.2014. 00614. Frontiers Media S.A.
- Wang, Q., Wang, F., Xu, Z., Ding, Z., 2017a. Bioactive mushroom polysaccharides: a review on monosaccharide composition, biosynthesis and regulation. Molecules 22 (6), 955.
- Wang, Q., et al., 2017b. β-Glucans: relationships between modification, conformation and functional activities. Molecules 22 (2). https://doi.org/10.3390/molecules22020257.
- Wasser, S.P., Weis, A.L., 1999. Medicinal properties of substances occurring in higher basidiomycetes mushrooms: current perspectives (review). Int. J. Med. Mushrooms 1 (1), 31–62. https://doi.org/10.1615/intjmedmushrooms.v1.i1.30.
- Wynn, T.A., Barron, L., 2010. Macrophages: master regulators of inflammation and fibrosis. Seminars in Liver Disease, pp. 245–257 https://doi.org/10.1055/s-0030-1255354.
- Yamada, J., et al., 2007. Alleviation of seasonal allergic symptoms with superfine β-1,3-glucan: a randomized study. J. Allergy Clin. Immunol. 119 (5), 1119–1126. https://doi. org/10.1016/i.iaci.2007.02.005.
- Yamauchi, E., et al., 2008. Contribution of lung fibroblast migration in the fibrotic process of airway remodeling in asthma. Allergol. Int. 57 (1), 73–78. https://doi.org/10.2332/ allergolint.O-06-481.
- Yang, D., Zhou, Z., Zhang, L., 2019. An overview of fungal glycan-based therapeutics. Prog. Mol. Biol. Transl. Sci., pp. 135–163. https://doi.org/10.1016/bs.pmbts.2019.02.001
- Zheng, X.F., et al., 2013. Lipopolysaccharide-induced M2 to M1 macrophage transformation for IL-12p70 production is blocked by *Candida albicans* mediated up-regulation of EBI3 expression. PloS one 8 (5). https://doi.org/10.1371/JOURNAL.PONE.0063967.
- Zambelli, V, et al., 2021. Hexarelin modulates lung mechanics, inflammation, and fibrosis in acute lung injury. Drug Target Insights. https://doi.org/10.33393/dti.2021.2347.
- Zhou, F., et al., 2020. Clinical course and risk factors for mortality of adult inpatients with COVID-19 in Wuhan, China: a retrospective cohort study. Lancet 395 (10229), 1054–1062. https://doi.org/10.1016/S0140-6736(20)30566-3.





# Entry β-Glucans

# Emma J. Murphy <sup>1,2</sup>, Emanuele Rezoagli <sup>3,4,5</sup>, Ian Major <sup>2</sup>, Neil Rowan <sup>1</sup>, and John G. Laffey <sup>3,4,\*</sup>

- <sup>1</sup> Bioscience Research Institute, Athlone Institute of Technology, N37 HD68 Athlone, Ireland; emurphy@ait.ie (E.J.M.); nrowan@ait.ie (N.R.)
- <sup>2</sup> Materials Research Institute, Athlone Institute of Technology, N37 HD68 Athlone, Ireland; imajor@ait.ie
- <sup>3</sup> Lung Biology Group, Regenerative Medicine Institute at CURAM Centre for Medical Devices, School of Medicine, National University of Ireland Galway, H91 TK33 Galway, Ireland; emanuele.rezoagli@unimib.it
  - Anaesthesia and Intensive Care Medicine, University Hospital Galway, H91 TK33 Galway, Ireland
- <sup>5</sup> Department of Medicine and Surgery, University of Milan-Bicocca, 20900 Monza, Italy
- Correspondence: john.laffey@nuigalway.ie; Tel.: +353-91-49349

**Definition:** Definition $\beta$ -glucans are complex polysaccharides that are found in several plants and foods, including mushrooms.  $\beta$ -glucans display an array of potentially therapeutic properties.

Keywords: β-glucan; clinical trials; biomedicine; immunomodulation; metabolism

# 1. Introduction

4

 $\beta$ -glucans/Beta-glucans are a large class of complex polysaccharides that can be found in an abundance of sources. Depending on origin,  $\beta$ -glucans can be classified as cereal or non-cereal derived. Cereal sources of  $\beta$ -glucans include oat and barley and non-cereal sources can include mushroom, algae, bacteria and seaweed [1].  $\beta$ -glucans are biologically active compounds that have been widely reported to improve health [2]. Specific to this group of polysaccharides is a 1,3 beta-glycosidic linked backbone; separate to this, the polysaccharide can take many forms, dictated by origin.

There is a growing interest in foods that have the potential to lower the risk or incidences of chronic diseases or promote lifespan as well as have anti-aging properties. This has led to an increase in awareness of the effect of diet on health [3,4].

In 1979, Stephen DeFelice devised the term nutraceutical, which may be isolated nutrients, dietary supplements, genetically engineered foods and herbal products [5]. Nutraceuticals are defined as a food or food component that provides medical or health benefits, including prevention and/or treatment of disease [5]. Similarly, bioactive compounds are defined as "essential and non-essential compounds that occur in nature, are part of the food chain and are shown to have an effect on human health" [6].

Bioactive substances in food provide health benefits beyond the nutritional benefits of the product [7].  $\beta$ -glucans are reported to be both a bioactive and a nutraceutical. Their therapeutic effects can also be largely classified into two categories, metabolic/GI effects or immune-modulatory effects, which is largely based on structure, determined by source [1,8].

Metabolic effects are usually observed with cereal derived  $\beta$ -glucans. Effects include modulation of the gut microbiome, cholesterol reduction and decreased cardiovascular and diabetic risk. Non-cereal  $\beta$ -glucans are associated with immune-modulatory effects, anti-tumor effects, wound healing and alleviation of immune-related conditions, as demonstrated in Figure 1 [1].  $\beta$ -glucans are also administered as an animal and fish feed additive to increase [9,10]. These molecules also have applications in the food industry for thickening and for gelation purposes [11].

In this encyclopedia entry on  $\beta$ -glucans, we start from their initial discovery, then examine  $\beta$ -glucan sources, characterize their complex and diverse structures, and examine the implications of their structural variations on their activity profile. The therapeutic



Citation: Murphy, E.J.; Rezoagli, E.; Major, I.; Rowan, N.; Laffey, J.G. β-Glucans. *Encyclopedia* **2021**, *1*, 831–847. https://doi.org/10.3390/ encyclopedia1030064

Academic Editor: Fabrizio Araniti

Received: 9 June 2021 Accepted: 11 August 2021 Published: 19 August 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). potential in different disease conditions is then discussed, and the barriers to fully realizing this potential is dissected in some detail. Finally, we examine other uses of  $\beta$ -glucan in animal health and their application in the food industry.



**Figure 1.** Mechanisms and activity of  $\beta$ -glucan which are dependent on source.  $\beta$ -glucan can be classified as cereal derived (upper panel), or non-cereal derived (lower panel). Picture originally published in [1]. Modified with permission.

# 2. History of β-Glucans

As  $\beta$ -glucans are a structural component of mushrooms, their use as an unknown therapeutic in medicinal mushrooms dates back hundreds of years [12].  $\beta$ -glucans, as an active compound, followed two different pathways of discovery. In Western medicine, they were initially discovered by Pilliner et al., in 1954 [13].  $\beta$ -glucans were discovered unintentionally through the discovery of Properdin in the compliment system. Zymosan was the stimulatory agent used in these experiments. It is a crude mixture of yeast cell

wall particles, which is now established to contain high levels of  $\beta$ -glucans. In 1961, Riggi and Dilugio identified the active compound in zymosan to be a polysaccharide. They concluded that the polysaccharide required a 1,3 beta type linkage for activity [13].

Independently, concurrently in Japan,  $\beta$ -glucans were being investigated for antitumor properties, with the first study in 1969. Chikara et al. isolated  $\beta$ -glucans from the mushroom *Lentinus edodes* and demonstrated it to have an ability to inhibit sarcoma in mice [14].  $\beta$ -glucans were previously used as a traditional medicine for cancer therapy [15–17]. To date,  $\beta$ -glucans isolated from *Lentinus edodes* named Lentinan and Polysaccharide K are two licensed drugs in Japan [18].

# Developments

In the wake of these pioneering studies,  $\beta$ -glucan research has gained considerable attention and advancements, with these bioactives being registered in clinical trials for an array of conditions including inflammatory conditions, cardiometabolic diseases, obesity and cancer, and more recently, as a nutritional supplement for the treatment of COVID-19 in conjunction with vaccination (Efficacy and Tolerability of ABBC1 in Volunteers Receiving the Influenza or Covid-19 Vaccine—Full Text View—ClinicalTrials.gov) [19]. The US Food and Drug Administration has also recognized the benefits of consuming  $\beta$ -glucans, recommending a 3 g/day intake of oat bran, a source of  $\beta$ -glucan and registered as a cholesterol-reducing food [20].

#### **3.** Sources and Structure of β-Glucans

As mentioned,  $\beta$ -glucans can be classified by source: cereal or non-cereal. This classification is largely based on structure, as  $\beta$ -glucans from cereal have a different structure than those originating from non-cereal sources. Secondly,  $\beta$ -glucans from cereal sources can differ; for example,  $\beta$ -glucans from barley can be structurally different to those of oat. The same diversity can occur between  $\beta$ -glucans from non-cereal sources. All  $\beta$ -glucans are homo-polysaccharides and essentially composed of glucose units linked together and thus have a characteristic 1,3 linked backbone [21,22], which is fundamental to activity [23]. The structural difference occurs at branching off this backbone, which is dictated by source.  $\beta$ -glucans can be unbranched or branched [24]. Branching can usually occur at either the 1,4 or 1,6 position [24]. These molecular and structural characteristics will determine functional activity as  $\beta$ -glucans have a defined structure–activity relationship [25].

Cereal or grain derived  $\beta$ -glucans usually have 1,3 1,4 glycosidic linkages without any 1,6 bonds or branching [26,27]. Non-cereal sources usually have 1,6 linked branches off the main side chain. Figure 2 demonstrates the structural differences between  $\beta$ glucans. Other glucans have no branching, such as Curdlan, a glucan isolated from *Agrobacterium* that contains no side branching, just a beta-D glucan backbone [28]. There are, however, exceptions—*Sorghum arundinaceum*, an ancient cereal grain, was found to contain  $\beta$ -glucans with alpha 1,4 linked D-glucopyranose residues with 1,3, 1,6 branching points [29]. Moreover, different species of Sorghum have different structures; *Sorghum bicolor* contains 1,3 with 1,4 linkages [30].

 $\beta$ -glucans from the same species source may have variances in structure. Variances can include chain length, degrees of branching, polymerization and 3D conformational structure. Three-dimensional conformational structure can be random coil, single helix or triple helix [2]. Furthermore, other factors that can influence the structure include growth conditions, extraction procedure and analysis [20,31].

It has been shown that  $\beta$ -glucans specifically of mushroom origin have anti-tumor properties. Lentinan is a  $\beta$ -glucan isolated from mushroom which has a  $\beta$ -helix conformation [32,33]. If the helix is destroyed or removed, the anti-tumor activity decreases. Other studies demonstrated that molecular weight and chain conformation are all dependent on anti-tumor activity [34]. The ability of  $\beta$ -glucans to interact and modulate immunity is correlated to polymer length, degree of branching and tertiary structure [23,35]. Larger  $\beta$ -glucans have been shown to activate leucocytes directly. Activation initiates phagocytosis,



antimicrobial activities and production of cytokines. Medium to smaller-sized  $\beta$ -glucans induce NF- $\kappa$ B and mediate inflammation [36].



**Figure 2.** All  $\beta$ -glucans have a 1,3 linked backbone. The variances in structure occur with glycosidic side branches. Variances can also occur in chain length or degrees of polymerization, branching interval and branch length. Source and extraction technique will influence final structure. Various analytical methods can be used for the characterization of  $\beta$ -glucans, as outlined in Table 1.

 $\beta$ -glucans can also be isolated from macro and microalgae sources. Microalgae are unicellular photosynthetic organisms, whereas macroalgae are larger multicellular organisms. Laminarian is a water-soluble  $\beta$ -glucan extracted from the cell wall storage of *Eisenia bicyclis*, a brown macroalgae. Similarly, to other non-cereal sources, the structure consists of a 1,3 linked backbone with 1,6 side branching. The glycosidic bonds have different ratios depending on the habit or season of extraction [37,38].

Laminarian possesses anti-inflammatory, anti-apoptotic, anti-tumor, antioxidant and anticoagulant properties. More recently, anti-cancer effects via enhanced apoptotic cellular death and angiogenic potential have been reported [37]. This polysaccharide is also being exploited as a carrier for gene delivery [39]. Topical laminarian-based creams improve wound healing via collagen deposition and promotion of skin regeneration [40]. Oral supplementation of laminarian has also been shown to suppress the progressive development of precancerous lesions in mice [38].

Soluble  $\beta$ -glucan from macro algae has been applied in the aquaculture industry to fish in the early stages of development as well as cultures of crustaceans as a bath treatment to enhance immunity in early life [41]. Species of brown algae (*Phaeophyceae*) and red algae (*Gracilariaceae*) contain a 1,3 1,6 linked  $\beta$ -glucan with immunomodulatory properties—via activation of spleen lymphocytes [41].

The microalga *Euglena gracilis* is a well reported source of  $\beta$ -glucans. The microalga is of the division Euglenoophyceae and family Euglenales [42,43].  $\beta$ -glucans isolated from this source are called paramylon, which are large, linear, unbranched 1,3  $\beta$ -glucans with the potential to support immune function [44]. A clinical trial that investigated the effects of supplementation with *E. gracilis* that contained a 50%  $\beta$ -1,3-glucan on immune function reduced the severity of upper respiratory tract infections (URTI), decreasing sick days and symptoms compared to placebo [43].

#### Extraction, Purification and Characterisation Methods

There are four types of extraction procedures used for isolation of  $\beta$ -glucans: alkaline extraction, acidic extraction, enzymatic extraction and water extraction. More advanced methods can include ultrasound-assisted extraction, microwave-assisted extraction and superheated water extraction.

Cereal derived  $\beta$ -glucans are generally located in the aleurone (proteins stored as granules), in the sub-aleurone or in the cell wall of endospores, which are all located in oat, barley, wheat and rice [45]. Extractions from cereal sources can be difficult and therefore,  $\beta$ -glucans from these sources are commercially more expensive [46]. There is no standard method for the extraction of  $\beta$ -glucans. Most researchers modify commonly published methods. The method selected is usually based on solubility of the  $\beta$ -glucan in hot water or solvents and is followed by precipitation using 2-propanol or ethanol [47]. Other published methods include accelerated solvent extraction (ASE) [48], sodium hydroxide as initial extraction solvent [47]. Table 1 lists the various extraction methods for  $\beta$ -glucans as well as analytical methods for identification.

For  $\beta$ -glucans from non-cereal sources, hot water extraction is the most common method [49]. Samples are usually milled, extracted with water and then precipitated with solvents such as ethanol [50]. Other methods have used ethanol precipitation and chromatography methods using Sephadex or Di-Ethyl-Amino-Ethyl (DEAE) [47,51].

Simple extraction techniques will isolate the  $\beta$ -glucans to some extent. Usually, there are other contaminating materials present, e.g., proteins. Enzymatic techniques can be used for the removal of proteins. The isolated  $\beta$ -glucans, by simple extraction techniques, are usually fit for purpose. Purification and extensive extraction techniques can be expensive and tedious. However, different extraction and purification techniques will yield different products and cause inconsistences between results. A way to control this is to fully characterize  $\beta$ -glucans after extraction, as it is important that  $\beta$ -glucans retain their structural characteristics after extraction [52]. Methods applied for characterization include gel permeation chromatography (GPC) [53] for molecular weight estimation, Fourier-transform

infrared spectroscopy (FTIR) and nuclear magnetic resonance spectroscopy (NMR) for structural characterization [54,55]. For the detection of  $\beta$ -glucans in products, enzymatic reactions are also available [56].

Table 1. Extraction and characterization methods for $\beta$ -glucan.
---

Source	Extraction Method	Extraction Method Analytical Method	
Hull-less Barley Bran	Ultrasonic extraction, Hot water extraction, Microwave extraction, Microwave assisted ultrasonic extraction	Megazyme commercial quantification kit, Structure determined using FTIR, Molecular weights determined using gel permeation chromatography (GPC)	[57]
Hull-less Barley	Alkali extraction and ethanol precipitation.	High pressure size exclusion chromatography (HPSEC), Methylation, Gas Chromatography Mass Spectroscopy (GC-MS)	[21]
Barley	Ethanol Extraction and enzyme treatment with amylase.	Megazyme commercial quantification kit, Molecular weight was determined using size-exclusion ultra-high-performance liquid chromatography	[58]
Barley Bran	Enzyme extraction using α-amylase, protease, glucoamylase, pullunlanase and xylanase	Megazyme commercial quantification kit, HPSEC, Rheology, SEM	[59]
Barley	Pressurized aqueous ethanol	Megazyme commercial quantification kit, GPC	[60]
Barley	Ethanol and water extraction	N/A	[61]
Oat and Barley	Multistage approach; Solvent (acetone) and enzyme extraction α-amylase and hot water extraction	Megazyme commercial quantification kit, Asymmetric flow field-flow fractionation (AF4) coupled to multiangle light scattering (MALS), differential refractive index (dRI) and fluorescence (FL) detection, High performance anion exchange chromatography	[62]
Oat	Multistage enzymatic and solvent extraction—enzymes α-amylase, amyloglucosidase, and papain	Megazyme commercial quantification kit	[63]
Oat	Subcritical-water extraction	Megazyme commercial quantification kit HPLC	[64]
Oat	Enzymatic extraction- $\alpha$ -amylase	Megazyme commercial quantification kit	[65]
Corn pericarp	Anion exchange chromatography and affinity chromatography	HPLC, NMR, Methylation	[66]
Castanea mollissima	Water extraction followed by ethanol extraction purified using anion exchange chromatography	Phenol- sulfuric acid method, High performance gel permeation chromatography (HPGPC) HPLC, FTIR, Methylation analysis	[67]
Pueraria lobata	Ethanol extraction followed by cold water extraction followed by further ethanol extraction. Fractions of extracts collected using DEAE-Sepharose chromatography column	Carbohydrate content determined by phenol-sulfuric acid method, HPGPC, Congo red method used to determine triple helix structure.	[68]
Ziziphus jujuba Mill	3 Phase extraction, Aqueous alkaline extraction; acidic precipitation of proteins at their isoelectric point, precipitation of glucans with absolute ethanol	Megazyme commercial kit, FTIR, Surface structural differences determined using scanning electron microscope.	

Source	Extraction Method	Analytical Method	
Tuber melanosporum	Extracted pressurized liquids. Water and ethanol used as extraction solvents.	Analyzed by NMR and Gas chromatography mass spectroscopy GC_MS	[70]
Phaseolus vulgaris	Sonication of cell wall residue	Gel filtration chromatography, Methylation analysis, NMR	[71]
Punica granatum	Alkaline treatment, isoelectric precipitation, alcohol precipitation	FTIR	[72]
Euglena cantabrica	Pressurized liquid extraction (PLE) using different temperature 40–180 °C using green solvents (ethanol-water) mixtures.	Extracts analyzed by high-pressure size-exclusion chromatography coupled to an evaporative light-scattering detector.	[73]
Laminaria hyperborea	Hydrothermal assisted extraction—acid extraction, temperature and pressure treatment, filtration and freeze drying.	Concentration calculated using Megazyme commercial kit.	[74]
Saccharomyces cerevisiae	Hot water Extraction & Enzymatic Treatment; enzymes-protease and lipase	Concentration calculated using Megazyme commercial kit.	[75]
Saccharomyces cerevisiae	n/a	Congo red assay—colorimetric	[76]
Saccharomyces cerevisiae	Yeast was cultured in yeast extract-peptone-glucose (YBG) broth to produce ß-glucans. Cells were sonicated. Alkaline-acid extraction used as extraction method.	n/a	[77]
Saccharomyces cerevisiae	Acid-base extraction method	FTIR analysis of structure, HPLC	[78]
Saccharomyces cerevisiae	Cell exposure to hot water (autoclaving), thermally induced autolysis, homogenization in a bead mill, sonication	FTIR, Megazyme commercial kit	[79]
Saccharomyces cerevisiae	Alkaline and acidic extraction	FTIR, NMR	[80]
Pleurotus eryngii	Water extraction at different temperatures and pressures	High pressure size exclusion chromatography (HPSEC), Gel permeation chromatography (GPC)	[81]
Cantharellus tubaeformis	Pressurized hot water extraction (80–240 °C)	Megazyme commercial kit	[82]
Pleurotus sajor caju	Hot aqueous extraction	NMR spectroscopy, HPSEC, Methylation analysis	[83]
Agaricus bisporus, Lentinula edodes and Pleurotus ostreatus	Pressurized water extraction (PWE)	Megazyme commercial kit	[84]
Mushroom by-products	Mechanical agitation and ultrasound assistance in ethanol/water solutions	and ultrasound Megazyme commercial kit	
Ganoderma lucidum	rma lucidum    Hot water extraction, Soxhlet extraction, ultrasound assisted extraction.    Content and ration of branching determined by enzymatic-HPAEC-PAD detection, FTIR, SEM		[86]
Pholiota nameko	Defatting process with cold water, Hot aqueous extraction Enzyme treatment with amylase	NMR, Methylation	[87]

# Table 1. Cont.

Source	Extraction Method	Analytical Method	Ref
Pleurotus ostreatus	Methanol extraction.Megazyme commercial kit,Hot water extraction;GPC,Acid hydrolysisHydrophilic interaction chromatography		[88]
Lentinus edodes-	Orthogonal alkaline extraction with sodium hydroxide—further purification ethanol precipitation and anion exchange chromatography	<sup>1</sup> H-NMR High-performance gel permeation chromatography–refractive index–multi-angle laser light scattering (HPGPC-RI-MALLS) Methylation analysis	[89]

Table 1. Cont.

# 4. Activity

### 4.1. Absorption

Although other routes of administration of  $\beta$ -glucans have been explored, oral administration is the most common, as  $\beta$ -glucans are a food-derived compound. When orally administered, studies have shown that small  $\beta$ -glucan concentrations in other tissues are low. Studies have also demonstrated that  $\beta$ -glucans reach the small intestine intact. The primary reaction occurs between the molecule and the gut epithelia [90]. Peyer's patches are groupings of lymphoid follicles in the ileum region of the small intestine. It is in this location that immune stimulatory components are taken up to lymphatic circulation. In the Peyer's patches, there are circulating antigen presenting cells (APCs) [91]. When  $\beta$ -glucans reach the Peyer's patches, dendritic cells or microfold (M cells) recognize  $\beta$ -glucans and present them to other immune cells located here [92,93].

Pathogen-associated molecular patterns (PAMPs) are essentially the signature markings of pathogens. It is how they are recognized by immune counterparts. PAMPs that contain a sugar component such LPS from Gram-negative bacteria are classified as sugar complexed PAMPs (SCPs). SCPs are functional and structural components of the pathogen [94]. Receptors recognize these unfamiliar microbial structures and respond.  $\beta$ -glucans are a structural component of fungi and therefore are recognized as a PAMP. This is beneficial as  $\beta$ -glucans initiate an immune response without introducing an infectious agent such as a fungus or bacteria. As PAMPs,  $\beta$ -glucans can stimulate immune cells in isolated form—i.e., they are not required to be attached to the infectious agent—they have been developed as adjuvants and immunotherapeutics [95].

There is huge diversification of carbohydrate structure and branching, creating a range of complex monomers and polymers, and there are specific receptors to recognize these conformations which elicit the necessary response [96,97]. The PAMP recognition ability of the immune cells requires receptors, namely, pathogen recognition receptors (PRRs) [98].

#### 4.2. Cellular Activation

It is well established that cellular biological activities—namely, immune modulatory activity—are dependent on recognition and binding of receptors. C-type lectin receptors (CLRs) are the main pathogen recognition receptors involved in fungal recognition [99]. Initial or broad-spectrum recognition of  $\beta$ -glucans is based on the 1,3 backbones usually by a receptor called Dectin-1 or the  $\beta$ -glucan receptor [92]. It is reported that  $\beta$ -glucans activate intestinal epithelial cells to secrete pro-inflammatory cytokines in a Dectin-1 and Syk-dependent manner [100].

Dectin-1, a CLR-type receptor, is present on macrophages, dendritic cells, neutrophils and on pathways where pathogens invade [101–103]. After activation, Dectin-1 will trigger the NF-κB pathway, which will induce cytokine synthesis by Syk-dependent pathways. Pathways can include NF-κB inducting kinase (SYK-NIK) pathways or Syk independent pathways, including the Ras associated factor-l (RAF-1) pathway [104].

Complement receptor 3 (CR3) is mainly expressed on NK cells, dendritic cells, macrophages and neutrophils [105]. This specific receptor usually responds to iC3B opsonized cells and pathogens [106].

CR3 is involved in the recognition of endogenous ligands, but also acts as an opsonic receptor for the complement component and as a non-opsonic receptor for  $\beta$ -glucan. The anti-tumorigenic properties of  $\beta$ -glucans are correlated to recognition of this receptor and deficiency is correlated to a loss in activity [107].

Other receptors that recognize  $\beta$ -glucan include lactosylceramide receptor, scavenger receptors and Toll like receptors (TLR), namely TLR2 [108,109]. TLR 2,4 and 6 co-bind to dectin-1 after glucan recognition [110]. Lactosylceramide receptors are found in plasma membrane of many different cells [95]. Binding and recognition of these receptors induces ingestion, respiratory burst, microbial killing, inflammatory processes and the release of chemical mediators that activate and recruit other cells [111].

Dectin-1 can interact with  $\beta$ -glucans over a wide range of affinities; although 1,6 branching does not influence binding, branched  $\beta$ -glucans appear to have a stronger affinity in comparison to linear  $\beta$ -glucans [112]. Other influences over recognition include chain conformation and polymerization. Extraction procedure can have a huge impact on  $\beta$ -glucan conformation [113]. Thus, the isolation and purification procedures must be carefully selected, as it will ultimately influence the activity of  $\beta$ -glucans.

#### 5. Therapeutic Potential and Challenges

# 5.1. Therapeutic Potential

In May 2006, the US Food and Drug Administration acknowledged that  $\beta$ -glucans reduce the risk of coronary heart disease. Health claims relating to  $\beta$ -glucans specifically for maintenance of blood cholesterol were approved in Europe in 2009 [114].

From a metabolic viewpoint,  $\beta$ -glucans have the ability to lower cholesterol and improve glycemic control [115–117]. Viscous  $\beta$ -glucans have been shown to modulate host bile acid metabolism [117]. In lowering cholesterol and triglycerides, the risk of cardiovascular diseases is also reduced [118]. Yeast  $\beta$ -glucans serve as prebiotics, improving gastrointestinal function by enhancing intestinal microbiota [1,119].

Immune modulatory properties include stimulation of the immune response and initiation of inflammatory processes as well as improving resistance to infections [120]. Non cereal  $\beta$ -glucans alleviate allergic conditions [121,122].  $\beta$ -glucans have also been shown to enhance bacterial clearance from blood and reduce mortality in a rat model of intra-abdominal sepsis [123].

Immuno-oncology is the study of treatments that utilize the immune system to target cancer [124].  $\beta$ -glucans have shown promise in this area as they stimulate immune cells to target cancerous cells.  $\beta$ -glucans have been demonstrated to induce macrophages and NK cells to attack and destroy cancerous cells [125]. Mushroom  $\beta$ -glucans promoted T-cell immunomodulation and neutrophil infiltration into tumors in mice, which lead to tumor growth inhibition [126]. Oral administration of  $\beta$ -glucans has similar effects to other routes of administration when used as an anti-tumor immunotherapy [107].

Promising results in clinical trials using  $\beta$ -glucans as an adjuvant in conjunction with monoclonal antibodies for cell mediated tumor killing has established them as a potential intervention in the field of immune-oncology [127].

 $\beta$ -glucans have also gained attention as vaccine adjuvants. When combined with vaccines, they have the ability to enhance sensitivity by stimulating immune cells [128]. Stimulants that are used for immune modulatory functions are referred to as immune adjuvants [129]. As  $\beta$ -glucans are recognized by immune counterparts as PAMPs, they are widely investigated for their potential as immune adjuvants [130,131]. They have therefore been administered as adjuvants in vaccines or counterparts in vaccine delivery systems [132,133].

#### 5.2. Therapeutic Challenges

The therapeutic promise of  $\beta$ -glucans for an array of clinical conditions is considerable [127]. The translation to clinical application is not as successful for several reasons. Firstly, like all naturally derived products,  $\beta$ -glucans can represent a complex mixture of ingredients which can all contribute to or prevent activity. Extraction methods will greatly affect the final compound. Intense extraction procedures can alter the natural configuration of the  $\beta$ -glucan compound. Softer extraction procedures can leave other contaminating residues present, which can dilute, contaminate or even contribute to activity [27].

Secondly, there is huge diversity among  $\beta$ -glucan structures. Diversity occurs between  $\beta$ -glucans from different species, but there are also variances in extracted samples from the same species. These differences include degree of branching, monosaccharide composition, linkage ratio and linkage type [134]. The molecular weight of the  $\beta$ -glucan greatly influences activity, which can be dictated by growth conditions and the extraction procedure. For example,  $\beta$ -glucans isolated from barley in one experiment was shown to have a molecular weight of  $3 \times 10^4$  g/mol and in a separate experiment found to have a molecular weight of  $270 \times 10^4$  g/mol [31]. The distribution of polysaccharides from aleurone to endosperm in cereal sources can also vary depending on origin. In one study, aleurone was found to contain 26% w/w  $\beta$ -glucan in comparison to endosperm, which contained 70%. A study investigating the individual cytolytic composition of different barley varieties found that cytolytic malt modifications have a lower  $\beta$ -glucan content. This work confirms that breeding progress reduces  $\beta$ -glucan content in barley [135]. These variances can cause huge inconsistencies in reported therapeutic effects. These two issues can potentially be resolved by producing standardized methods for the extraction of  $\beta$ glucans categorized by source, as the location of  $\beta$ -glucans is different in each. At present, there is no standardized extraction method.

Furthermore, very few reports state the way in which  $\beta$ -glucans are extracted and the final concentration yield. A potential solution to this is that all  $\beta$ -glucans should be fully characterized in terms of structure, molecular weight and degree of purity. Therefore, activity can be compared, receptor pathways defined and mechanisms of action elucidated.

Finally, disease targets are also very broad. In categorizing and defining structure and disease targets, clinical effects can be better understood. Optimal routes of administration and general applications such as length of treatment and dose also vary in both preclinical and clinical trials, leading to conflicting findings [136]. These too need to be defined for comparison to overcome inconsistences, as there is huge enthusiasm for the clinical use of  $\beta$ -glucans.

#### 5.3. Animal Health

 $\beta$ -glucans administered as adjuvants in vaccines have also shown promise in animal models. In vivo models have shown upregulated macrophage and dendritic cell recruitment and maturation. Antigen-specific CD8+ T-cell responses were also increased. Delivery by various routes of administration showed similar activity [137,138].

There are constant influxes of studies demonstrating the immune modulating properties effects of  $\beta$ -glucans. There is also a ban on the use of antibiotics administered to farmed animals for growth and immune promoting purposes. The use of  $\beta$ -glucans as an alternative is increasing [139]. In fisheries, it is tedious and costly to vaccinate fish to improve immunity and immune strength against pathogens; one of the major benefits of food component immune modulators is that they can be administered as food and still achieve the same effects. Studies have shown that feeding  $\beta$ -glucans to healthy fish has increased non-specific and specific immunity levels and increased protection against bacterial infections [140].

Algae  $\beta$ -glucans have demonstrated improved immunity in dogs [141]. In mice, the bioactives had significant effects as a prophylactic treatment to reduce anthrax infection and inhibition of cancer cells through stimulation of cytokines [142,143]. Sheep oral supplementation to ewes had positive effects on reproductive performance, growth rate and body composition [144]. Lambs, when administered  $\beta$ -glucans, had a significant increase in phagocytic and respiratory burst activities [145,146]. In pigs, there was an improvement in health status [139] and when administered to piglets, there was a reduction in the susceptibility of newborn piglets to enterotoxigenic bacterial infection [147]. When

chicks are administered dietary supplementation with yeast derived  $\beta$ -glucans, Salmonella colonization in the cecum was reduced [148]; this effect was also observed in turkeys [149]. With these studies and many more,  $\beta$ -glucans are proving to be an attractive safe alternative to antibiotic administration.

### 5.4. Applications of $\beta$ -Glucans in the Food Industry

β-glucans are added to food products for a variety of reasons, most often as texture enhancers [150]. β-glucans originating from barley have been added as a thickener to liquid products as well as a source of dietary fiber [21]. β-glucan have also been demonstrated to increase water absorption [151], be a fat substitute [152] and act as a stabilizer in the formation of foam and emulsion. [153]. When used as a fat substitute in meat, it improved texture and retained moisture [152]. When added to food, β-glucans can enhance texture and replace fats but also have beneficial therapeutic effects against metabolic conditions. β-glucans are not digestible by enzymes in the human gastrointestinal tract, and therefore, when ingested in food, they behave as soluble dietary fibers [154]. Highly viscous βglucans have the ability to lower cholesterol and improve glycemic control as well as satiety control [117]. There is also evidence to suggest that when eaten, β-glucans interact and positively influence the gut microbiome [155].

The circular economy repurposes residues or by-products into raw materials for other processes, with the aim of reducing waste. Brewers spent grain (BSG) is an abundant by-product from the brewing industry. It contains many bioactives and is a source of 1,3 1,4  $\beta$ -glucans. Previously, BSG was mainly used in animal nutrition. However,  $\beta$ -glucans can be extracted from BSG and used for human nutrition or included in food with beneficial effects, including reduction in the postprandial glycemic response by up to 50% [156].

# 6. Conclusions and Perspectives

 $\beta$ -glucans are a diverse class of complex polysaccharides that can be found in an abundance of sources and are classified by source as cereal or non-cereal, based on differences in their branching patterns. These structural differences, caused in part by their source, and the extraction and purification methods used, confer different effect profiles on members of the  $\beta$ -glucans family. There is a significant body of literature attesting to their therapeutic potential, which vary from anti-tumor and immunomodulatory effects described predominantly with mushroom derived  $\beta$ -glucans, to metabolic effects such as improved glycemic control and cholesterol-lowering effects with cereal  $\beta$ -glucans, to probiotic properties.

While over 200 registered clinical trials of  $\beta$ -glucans clearly attest to the interest in these compounds, significant barriers exist to realizing their therapeutic effects. A key issue is the fact that the structure versus function profile remains incompletely understood. Addressing these barriers will require optimization of isolation and purification procedures, careful characterization of the relationship between specific variations in  $\beta$ -glucan structure and their effect profile. This approach would facilitate a greater understanding  $\beta$ -glucan, enabling identification of the most promising compounds for clinical testing, and should realize the promise of this intriguing class of compounds.

**Author Contributions:** Conceptualization, J.G.L. and E.J.M.; original draft preparation, E.J.M., E.R. and J.G.L.; writing—review and editing, E.J.M., E.R., I.M., N.R. and J.G.L.; visualization, E.J.M.; supervision, J.G.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Conflicts of Interest: The authors declare no conflict of interest.

Entry Link on the Encyclopedia Platform: Https://encyclopedia.pub/14401.

# References

- Murphy, E.J.; Rezoagli, E.; Major, I.; Rowan, N.J.; Laffey, J.G. β-Glucan Metabolic and Immunomodulatory Properties and Potential for Clinical Application. J. Fungi 2020, 6, 356. [CrossRef]
- Wang, Q.; Sheng, X.; Shi, A.; Hu, H.; Yang, Y.; Liu, L.; Fei, L.; Liu, H. β-Glucans: Relationships between modification, conformation and functional activities. *Molecules* 2017, 22, 257. [CrossRef]
- 3. Lordan, S.; Ross, R.P.; Stanton, C. Marine bioactives as functional food ingredients: Potential to reduce the incidence of chronic diseases. *Mar. Drugs* **2011**, *9*, 1056–1100. [CrossRef]
- 4. Ahnen, R.T.; Jonnalagadda, S.S.; Slavin, J.L. Role of plant protein in nutrition, wellness, and health. *Nutr. Rev.* 2019, 77, 735–747. [CrossRef]
- 5. Kalra, E.K. Nutraceutical-Definition and introduction. AAPS PharmSci 2003, 5, 27. [CrossRef] [PubMed]
- Frank, J.; Fukagawa, N.K.; Bilia, A.R.; Johnson, E.J.; Kwon, O.; Prakash, V.; Miyazawa, T.; Clifford, M.N.; Kay, C.D.; Crozier, A.; et al. Terms and nomenclature used for plant-derived components in nutrition and related research: Efforts toward harmonization. *Nutr. Rev.* 2020, *78*, 451–458. [CrossRef] [PubMed]
- Santos, D.I.; Saraiva, J.M.A.; Vicente, A.A.; Moldão-Martins, M. Methods for determining bioavailability and bioaccessibility of bioactive compounds and nutrients. In *Innovative Thermal and Non-Thermal Processing, Bioaccessibility and Bioavailability of Nutrients and Bioactive Compounds*; Elsevier: Amsterdam, The Netherlands, 2019; pp. 23–54.
- Ulmius, M.; Önning, G.; Nilsson, L. Solution behavior of barley β-glucan as studied with asymmetrical flow field-flow fractionation. *Food Hydrocoll.* 2012, 26, 175–180. [CrossRef]
- Rodrigues, M.V.; Zanuzzo, F.S.; Koch, J.F.A.; de Oliveira, C.A.F.; Sima, P.; Vetvicka, V. Development of Fish Immunity and the Role of β-Glucan in Immune Responses. *Molecules* 2020, 25, 5378. [CrossRef]
- 10. Byrne, K.A.; Loving, C.L.; McGill, J.L. Innate Immunomodulation in Food Animals: Evidence for Trained Immunity? *Front. Immunol.* 2020, 11. [CrossRef]
- 11. Ahmad, A.; Anjum, F.M.; Zahoor, T.; Nawaz, H.; Dilshad, S.M.R. Beta glucan: A valuable functional ingredient in foods. *Crit. Rev. Food Sci. Nutr.* **2012**, *52*, 201–212. [CrossRef]
- 12. Vetvicka, V.; Teplyakova, T.V.; Shintyapina, A.B.; Korolenko, T.A. Effects of medicinal fungi-derived β-glucan on tumor progression. *J. Fungi* **2021**, *7*, 250. [CrossRef] [PubMed]
- Pillemer, L.; Blum, L.; Lepow, I.H.; Ross, O.A.; Todd, E.W.; Wardlaw, A.C. The properdin system and immunity: I. Demonstration and isolation of a new serum protein, properdin, and its role in immune phenomena. *Science* 1954, 120, 279–285. [CrossRef] [PubMed]
- 14. Chihara, G.; Maeda, Y.; Hamuro, J.; Sasaki, T.; Fukuoka, F. Inhibition of mouse sarcoma 180 by polysaccharides from Lentinus edodes (Berk.) sing. *Nature* 1969, 222, 687–688. [CrossRef] [PubMed]
- 15. Takeshita, K.; Saito, N.; Sato, Y.; Maruyama, M.; Sunagawa, M.; Habu, H.; Endo, M. Diversity of complement activation by lentinan, an antitumor polysaccharide, in gastric cancer patients. *Nippon. Geka Gakkai Zasshi* **1991**, *92*, 5–11.
- 16. Kimura, Y.; Tojima, H.; Fukase, S.; Takeda, K. Clinical evaluation of sizofilan as assistant immunotherapy in treatment of head and neck cancer. *Acta Otolaryngol.* **1994**, *114*, 192–195. [CrossRef] [PubMed]
- Matsuoka, H.; Seo, Y.; Wakasugp, H.; Saito, T.; Tomoda, H. Lentinan potentiates immunity and prolongs the survival time of some patients. *Anticancer Res.* 1997, 17, 2751–2755. Available online: https://europepmc.org/article/med/9252710 (accessed on 18 May 2021).
- Ina, K.; Kataoka, T.; Ando, T. The Use of Lentinan for Treating Gastric Cancer. Anticancer Agents Med. Chem. 2013, 13, 681–688.
  [CrossRef]
- 19. Efficacy and Tolerability of ABBC1 in Volunteers Receiving the Influenza or COVID-19 Vaccine-Full Text View-ClinicalTrials.gov. Available online: https://clinicaltrials.gov/ct2/show/NCT04798677?term=bETA-GLUCAN&cond=Covid19&draw=2&rank=1 (accessed on 21 May 2021).
- 20. Henrion, M.; Francey, C.; Lê, K.A.; Lamothe, L. Cereal B-glucans: The impact of processing and how it affects physiological responses. *Nutrients* **2019**, *11*, 1729. [CrossRef]
- Zhang, H.; Zhang, N.; Xiong, Z.; Wang, G.; Xia, Y.; Lai, P.; Ai, L. Structural characterization and rheological properties of β-D-glucan from hull-less barley (Hordeum vulgare L. var. nudum Hook. f.). *Phytochemistry* 2018, 155, 155–163. [CrossRef]
- 22. Friedman, M. Mushroom polysaccharides: Chemistry and antiobesity, antidiabetes, anticancer, and antibiotic properties in cells, rodents, and humans. *Foods* **2016**, *5*, 80. [CrossRef]
- 23. Han, B.; Baruah, K.; Cox, E.; Vanrompay, D.; Bossier, P. Structure-Functional Activity Relationship of β-Glucans From the Perspective of Immunomodulation: A Mini-Review. *Front. Immunol.* **2020**, *11*, 658. [CrossRef]
- 24. Kaur, R.; Sharma, M.; Ji, D.; Xu, M.; Agyei, D. Structural features, modification, and functionalities of beta-glucan. *Fibers* **2020**, *8*, 1. [CrossRef]
- Rieder, A.; Grimmer, S.; Kolset, S.O.; Michaelsen, T.E.; Knutsen, S.H. Cereal β-glucan preparations of different weight average molecular weights induce variable cytokine secretion in human intestinal epithelial cell lines. *Food Chem.* 2011, 128, 1037–1043. [CrossRef]
- 26. Jin, Y.; Li, P.; Wang, F. β-glucans as potential immunoadjuvants: A review on the adjuvanticity, structure-activity relationship and receptor recognition properties. *Vaccine* **2018**, *36*, 5235–5244. [CrossRef] [PubMed]

- Du, B.; Meenu, M.; Liu, H.; Xu, B. A concise review on the molecular structure and function relationship of β-glucan. *Int. J. Mol. Sci.* 2019, 20, 4032. [CrossRef] [PubMed]
- 28. Zhan, X.B.; Lin, C.C.; Zhang, H.T. Recent advances in curdlan biosynthesis, biotechnological production, and applications. *Appl. Microbiol. Biotechnol.* **2012**, *93*, 525–531. [CrossRef]
- 29. Da Silva, B.P.; Silva, G.M.; Mendes, T.P.; Parente, J.P. Structural characteristics of a bioactive polysaccharide from Sorghum arundinaceum. Z. Naturforsch.-Sect. C J. Biosci. 2003, 58, 342–346. [CrossRef] [PubMed]
- Ermawar, R.A.; Collins, H.M.; Byrt, C.S.; Betts, N.S.; Henderson, M.; Shirley, N.J.; Schwerdt, J.; Lahnstein, J.; Fincher, G.B.; Burton, R.A. Distribution, structure and biosynthetic gene families of (1,3;1,4)-β-glucan in Sorghum bicolor. *J. Integr. Plant Biol.* 2015, 57, 429–445. [CrossRef]
- Lazaridou, A.; Biliaderis, C.G. Molecular aspects of cereal β-glucan functionality: Physical properties, technological applications and physiological effects. J. Cereal Sci. 2007, 46, 101–118. [CrossRef]
- Jin, Y.; Cai, L.; Yang, Q.; Luo, Z.; Liang, L.; Liang, Y.; Wu, B.; Ding, L.; Zhang, D.; Xu, X.; et al. Anti-leukemia activities of selenium nanoparticles embedded in nanotube consisted of triple-helix β-D-glucan. *Carbohydr. Polym.* 2020, 240, 116329. [CrossRef] [PubMed]
- 33. Lyu, F.; Xu, X.; Zhang, L. Natural polysaccharides with different conformations: Extraction, structure and anti-tumor activity. *J. Mater. Chem. B* 2020, *8*, 9652–9667. [CrossRef] [PubMed]
- Zheng, X.; Lu, F.; Xu, X.; Zhang, L. Extended chain conformation of β-glucan and its effect on antitumor activity. *J. Mater. Chem. B* 2017, *5*, 5623–5631. [CrossRef]
- 35. Stier, H.; Ebbeskotte, V.; Gruenwald, J. Immune-modulatory effects of dietary Yeast Beta-1,3/1,6-D-glucan. *Nutr. J.* **2014**, *13*, 38. [CrossRef]
- Kim, H.S.; Hong, J.T.; Kim, Y.; Han, S.-B. Stimulatory Effect of β-glucans on Immune Cells. *Immune Netw.* 2011, 11, 191. [CrossRef] [PubMed]
- Zargarzadeh, M.; Amaral, A.J.R.; Custódio, C.A.; Mano, J.F. Biomedical applications of laminarin. *Carbohydr. Polym.* 2020, 232. [CrossRef]
- 38. Desamero, M.J.; Kakuta, S.; Chambers, J.K.; Uchida, K.; Hachimura, S.; Takamoto, M.; Nakayama, J.; Nakayama, H.; Kyuwa, S. Orally administered brown seaweed-derived β-glucan effectively restrained development of gastric dysplasia in A4gnt KO mice that spontaneously develop gastric adenocarcinoma. *Int. Immunopharmacol.* 2018, 60, 211–220. [CrossRef]
- Hong, S.J.; Ahn, M.H.; Sangshetti, J.; Choung, P.H.; Arote, R.B. Sugar-based gene delivery systems: Current knowledge and new perspectives. *Carbohydr. Polym.* 2018, 181, 1180–1193. [CrossRef]
- Sellimi, S.; Maalej, H.; Rekik, D.M.; Benslima, A.; Ksouda, G.; Hamdi, M.; Sahnoun, Z.; Li, S.; Nasri, M.; Hajji, M. Antioxidant, antibacterial and in vivo wound healing properties of laminaran purified from Cystoseira barbata seaweed. *Int. J. Biol. Macromol.* 2018, 119, 633–644. [CrossRef]
- 41. Bobadilla, F.; Rodriguez-Tirado, C.; Imarai, M.; Galotto, M.J.; Andersson, R. Soluble β-1,3/1,6-glucan in seaweed from the southern hemisphere and its immunomodulatory effect. *Carbohydr. Polym.* **2013**, *92*, 241–248. [CrossRef]
- 42. Krajčovič, J.; Vesteg, M.; Schwartzbach, S.D. Euglenoid flagellates: A multifaceted biotechnology platform. *J. Biotechnol.* 2015, 202, 135–145. [CrossRef] [PubMed]
- 43. Evans, M.; Falcone, P.H.; Crowley, D.C.; Sulley, A.M.; Campbell, M.; Zakaria, N.; Lasrado, J.A.; Fritz, E.P.; Herrlinger, K.A. Effect of a Euglena gracilis fermentate on immune function in healthy, active adults: A randomized, double-blind, placebo-controlled trial. *Nutrients* **2019**, *11*, 2926. [CrossRef] [PubMed]
- 44. Russo, R.; Barsanti, L.; Evangelista, V.; Frassanito, A.M.; Longo, V.; Pucci, L.; Penno, G.; Gualtieri, P. Euglena gracilis paramylon activates human lymphocytes by upregulating pro-inflammatory factors. *Food Sci. Nutr.* **2017**, *5*, 205–214. [CrossRef] [PubMed]
- 45. Sikora, P.; Tosh, S.M.; Brummer, Y.; Olsson, O. Identification of high β-glucan oat lines and localization and chemical characterization of their seed kernel β-glucans. *Food Chem.* **2013**, *137*, 83–91. [CrossRef] [PubMed]
- Daou, C.; Zhang, H. Oat Beta-Glucan: Its Role in Health Promotion and Prevention of Diseases. *Compr. Rev. Food Sci. Food Saf.* 2012, 11, 355–365. [CrossRef]
- 47. Zhu, F.; Du, B.; Xu, B. A critical review on production and industrial applications of beta-glucans. *Food Hydrocoll.* **2016**, *52*, 275–288. [CrossRef]
- 48. Du, B.; Zhu, F.; Xu, B. β-Glucan extraction from bran of hull-less barley by accelerated solvent extraction combined with response surface methodology. *J. Cereal Sci.* 2014, 59, 95–100. [CrossRef]
- 49. Yan, J.K.; Wang, W.Q.; Wu, J.Y. Recent advances in Cordyceps sinensis polysaccharides: Mycelial fermentation, isolation, structure, and bioactivities: A review. J. Funct. Foods 2014, 6, 33–47. [CrossRef]
- 50. Bhanja, S.K.; Rout, D.; Patra, P.; Sen, I.K.; Nandan, C.K.; Islam, S.S. Water-insoluble glucans from the edible fungus Ramaria botrytis. *Bioact. Carbohydr. Diet. Fibre* **2014**, *3*, 52–58. [CrossRef]
- Kim, Y.W.; Kim, K.H.; Choi, H.J.; Lee, D.S. Anti-diabetic activity of β-glucans and their enzymatically hydrolyzed oligosaccharides from Agaricus blazei. *Biotechnol. Lett.* 2005, 27, 483–487. [CrossRef]
- 52. Brennan, C.S.; Cleary, L.J. The potential use of cereal (1→3,1→4)-β-d-glucans as functional food ingredients. *J. Cereal Sci.* 2005, 42, 1–13. [CrossRef]
- 53. Zhu, F.; Du, B.; Bian, Z.; Xu, B. β-Glucans from edible and medicinal mushrooms: Characteristics, physicochemical and biological activities. *J. Food Compos. Anal.* **2015**, *41*, 165–173. [CrossRef]

- 54. Kono, H.; Kondo, N.; Hirabayashi, K.; Ogata, M.; Totani, K.; Ikematsu, S.; Osada, M. NMR spectroscopic structural characterization of a water-soluble  $\beta$ -(1  $\rightarrow$  3, 1  $\rightarrow$  6)-glucan from Aureobasidium pullulans. *Carbohydr. Polym.* **2017**, 174, 876–886. [CrossRef]
- 55. Murphy, E.J.; Masterson, C.; Rezoagli, E.; O'Toole, D.; Major, I.; Stack, G.D.; Lynch, M.; Laffey, J.G.; Rowan, N.J. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—Implications for coronavirus disease (COVID-19) immunotherapies. *Sci. Total Environ.* **2020**, *732*, 139330. [CrossRef]
- 56. McCleary, B.V.; Draga, A. Measurement of β-Glucan in mushrooms and mycelial products. *J. AOAC Int.* **2016**, *99*, 364–373. [CrossRef]
- Liu, H.; Li, Y.; You, M.; Liu, X. Comparison of physicochemical properties of β-glucans extracted from hull-less barley bran by different methods. *Int. J. Biol. Macromol.* 2021, *182*, 1192–1199. [CrossRef]
- 58. Kim, H.J.; Kim, H.J. Physicochemical characteristics and in vitro bile acid binding and starch digestion of β-glucans extracted from different varieties of Jeju barley. *Food Sci. Biotechnol.* **2017**, *26*, 1501–1510. [CrossRef] [PubMed]
- Karimi, R.; Azizi, M.H.; Xu, Q. Effect of different enzymatic extractions on molecular weight distribution, rheological and microstructural properties of barley bran β-glucan. *Int. J. Biol. Macromol.* 2019, *126*, 298–309. [CrossRef]
- Benito-Román, Ó.; Alvarez, V.H.; Alonso, E.; Cocero, M.J.; Saldaña, M.D.A. Pressurized aqueous ethanol extraction of β-glucans and phenolic compounds from waxy barley. *Food Res. Int.* 2015, *75*, 252–259. [CrossRef] [PubMed]
- 61. Chen, H.; Nie, Q.; Xie, M.; Yao, H.; Zhang, K.; Yin, J.; Nie, S. Protective effects of β-glucan isolated from highland barley on ethanol-induced gastric damage in rats and its benefits to mice gut conditions. *Food Res. Int.* **2019**, *122*, 157–166. [CrossRef]
- 62. Zielke, C.; Kosik, O.; Ainalem, M.L.; Lovegrove, A.; Stradner, A.; Nilsson, L. Characterization of cereal β-glucan extracts from oat and barley and quantification of proteinaceous matter. *PLoS ONE* **2017**, *12*, e0172034. [CrossRef]
- 63. Harasym, J.; Zyła, E.; Dziendzikowska, K.; Gromadzka-Ostrowska, J. Proteinaceous residue removal from oat β-glucan extracts obtained by alkalinewater extraction. *Molecules* **2019**, *24*, 1729. [CrossRef]
- 64. Yoo, H.U.; Ko, M.J.; Chung, M.S. Hydrolysis of beta-glucan in oat flour during subcritical-water extraction. *Food Chem.* **2020**, *308*. [CrossRef] [PubMed]
- 65. Karp, S.; Wyrwisz, J.; Kurek, M.A. The impact of different levels of oat β-glucan and water on gluten-free cake rheology and physicochemical characterisation. *J. Food Sci. Technol.* **2020**, *57*, 3628–3638. [CrossRef] [PubMed]
- 66. Yoshida, T.; Honda, Y.; Tsujimoto, T.; Uyama, H.; Azuma, J.I. Selective isolation of β-glucan from corn pericarp hemicelluloses by affinity chromatography on cellulose column. *Carbohydr. Polym.* **2014**, *111*, 538–542. [CrossRef] [PubMed]
- 67. Li, H.; Wang, Y.; Wang, C.; Zhang, S.; Li, S.; Zhou, G.; Wang, S.; Zhang, J. Extraction, selenylation modification and antitumor activity of the glucan from Castanea mollissima Blume. *Glycoconj. J.* **2017**, *34*, 207–217. [CrossRef]
- 68. Dong, Z.; Zhang, M.; Li, H.; Zhan, Q.; Lai, F.; Wu, H. Structural characterization and immunomodulatory activity of a novel polysaccharide from Pueraria lobata (Willd.) Ohwi root. *Int. J. Biol. Macromol.* **2020**, *154*, 1556–1564. [CrossRef] [PubMed]
- Fazio, A.; la Torre, C.; Caroleo, M.C.; Caputo, P.; Plastina, P.; Cione, E. Isolation and purification of glucans from an Italian cultivar of Ziziphus jujuba Mill. And in vitro effect on skin repair. *Molecules* 2020, 25, 968. [CrossRef] [PubMed]
- 70. Tejedor-Calvo, E.; Morales, D.; Marco, P.; Sánchez, S.; Garcia-Barreda, S.; Smiderle, F.R.; Iacomini, M.; Villalva, M.; Santoyo, S.; Soler-Rivas, C. Screening of bioactive compounds in truffles and evaluation of pressurized liquid extractions (PLE) to obtain fractions with biological activities. *Food Res. Int.* 2020, 132, 109054. [CrossRef]
- Alonso-Simón, A.; Encina, A.E.; Seyama, T.; Kondo, T.; García-Angulo, P.; Álvarez, J.M.; Acebes, J.L.; Hayashi, T. Purification and characterization of a soluble β-1,4-glucan from bean (Phaseolus vulgaris L.)-cultured cells dehabituated to dichlobenil. *Planta* 2013, 237, 1475–1482. [CrossRef]
- Fazio, A.; Iacopetta, D.; la Torre, C.; Ceramella, J.; Muià, N.; Catalano, A.; Carocci, A.; Sinicropi, M.S. Finding solutions for agricultural wastes: Antioxidant and antitumor properties of pomegranate Akko peel extracts and β-glucan recovery†. *Food Funct.* 2018, *9*, 6619–6632. [CrossRef]
- Muñoz-Almagro, N.; Gilbert-López, B.; Pozuelo-Rollón, M.C.; García-Fernandez, Y.; Almeida, C.; Villamiel, M.; Mendiola, J.A.; Ibáñez, E. Exploring the microalga euglena cantabrica by pressurized liquid extraction to obtain bioactive compounds. *Mar. Drugs* 2020, 18, 308. [CrossRef] [PubMed]
- Rajauria, G.; Ravindran, R.; Garcia-Vaquero, M.; Rai, D.K.; Sweeney, T.; O'Doherty, J. Molecular characteristics and antioxidant activity of laminarin extracted from the seaweed species Laminaria hyperborea, using hydrothermal-assisted extraction and a multi-step purification procedure. *Food Hydrocoll.* 2021, 112, 106332. [CrossRef]
- 75. Borchani, C.; Fonteyn, F.; Jamin, G.; Paquot, M.; Blecker, C.; Thonart, P. Enzymatic process for the fractionation of baker's yeast cell wall (Saccharomyces cerevisiae). *Food Chem.* **2014**, *163*, 108–113. [CrossRef]
- Kupetz, M.; Procopio, S.; Sacher, B.; Becker, T. Critical review of the methods of β-glucan analysis and its significance in the beer filtration process. *Eur. Food Res. Technol.* 2015, 241, 725–736. [CrossRef]
- 77. Dhewantara, F.X.R. Cholesterol-lowering effect of beta glucan extracted from saccharomyces cerevisiae in rats. *Sci. Pharm.* **2016**, *84*, 153–165. [CrossRef]
- 78. Amer, E.M.; Saber, S.H.; Markeb, A.A.; Elkhawaga, A.A.; Mekhemer, I.M.A.; Zohri, A.N.A.; Abujamel, T.S.; Harakeh, S.; Abd-Allah, E.A. Enhancement of β-glucan biological activity using a modified acid-base extraction method from saccharomyces cerevisiae. *Molecules* 2021, 26, 2113. [CrossRef]

- 79. Bzducha-Wróbel, A.; Błłazejak, S.; Kawarska, A.; Stasiak-Rózańska, L.; Gientka, I.; Majewska, E. Evaluation of the efficiency of different disruption methods on yeast cell wall preparation for β-glucan isolation. *Molecules* 2014, 19, 20941–20961. [CrossRef]
- Upadhyay, T.K.; Fatima, N.; Sharma, D.; Saravanakumar, V.; Sharma, R. Preparation and characterization of beta-glucan particles containing a payload of nanoembedded rifabutin for enhanced targeted delivery to macrophages. *EXCLI J.* 2017, 16, 210–228. [CrossRef]
- 81. Abreu, H.; Zavadinack, M.; Smiderle, F.R.; Cipriani, T.R.; Cordeiro, L.M.C.; Iacomini, M. Polysaccharides from Pleurotus eryngii: Selective extraction methodologies and their modulatory effects on THP-1 macrophages. *Carbohydr. Polym.* **2021**, 252. [CrossRef]
- 82. Rodríguez-Seoane, P.; González-Muñoz, M.J.; Falqué, E.; Domínguez, H. Pressurized hot water extraction of β-glucans from Cantharellus tubaeformis. *Electrophoresis* **2018**, *39*, 1892–1898. [CrossRef] [PubMed]
- Carbonero, E.R.; Ruthes, A.C.; Freitas, C.S.; Utrilla, P.; Gálvez, J.; da Silva, E.V.; Sassaki, G.L.; Gorin, P.A.J.; Iacomini, M. Chemical and biological properties of a highly branched β-glucan from edible mushroom Pleurotus sajor-caju. *Carbohydr. Polym.* 2012, 90, 814–819. [CrossRef]
- Palanisamy, M.; Aldars-García, L.; Gil-Ramírez, A.; Ruiz-Rodríguez, A.; Marín, F.R.; Reglero, G.; Soler-Rivas, C. Pressurized water extraction of β-glucan enriched fractions with bile acids-binding capacities obtained from edible mushrooms. *Biotechnol. Prog.* 2014, 30, 391–400. [CrossRef] [PubMed]
- 85. Umaña, M.; Eim, V.; Garau, C.; Rosselló, C.; Simal, S. Ultrasound-assisted extraction of ergosterol and antioxidant components from mushroom by-products and the attainment of a β-glucan rich residue. *Food Chem.* **2020**, 332. [CrossRef]
- Alzorqi, I.; Sudheer, S.; Lu, T.J.; Manickam, S. Ultrasonically extracted β-D-glucan from artificially cultivated mushroom, characteristic properties and antioxidant activity. *Ultrason. Sonochem.* 2017, *35*, 531–540. [CrossRef] [PubMed]
- Abreu, H.; Simas, F.F.; Smiderle, F.R.; Sovrani, V.; Dallazen, J.L.; Maria-Ferreira, D.; Werner, M.F.; Cordeiro, L.M.C.; Iacomini, M. Gelling functional property, anti-inflammatory and antinociceptive bioactivities of β-D-glucan from the edible mushroom Pholiota nameko. *Int. J. Biol. Macromol.* 2019, *122*, 1128–1135. [CrossRef]
- 88. Szwengiel, A.; Stachowiak, B. Deproteinization of water-soluble ß-glucan during acid extraction from fruiting bodies of Pleurotus ostreatus mushrooms. *Carbohydr. Polym.* **2016**, *146*, 310–319. [CrossRef]
- 89. Li, J.; Cai, C.; Zheng, M.; Hao, J.; Wang, Y.; Hu, M.; Fan, L.; Yu, G. Alkaline extraction, structural characterization, and bioactivities of (1→6)-β-d-glucan from lentinus edodes. *Molecules* **2019**, 24, 1610. [CrossRef]
- 90. Rieder, A.; Knutsen, S.H.; Fernandez, A.S.; Ballance, S. At a high dose even partially degraded beta-glucan with decreased solubility significantly reduced the glycaemic response to bread. *Food Funct.* **2019**, *10*, 1529–1539. [CrossRef]
- 91. Panneerselvam, D.; Budh, D.P. Peyer Patches. In Encyclopedia of Cancer; Springer: Berlin/Heidelberg, Germany, 2011; p. 2831.
- Batbayar, S.; Lee, D.H.; Kim, H.W. Immunomodulation of fungal β-glucan in host defense signaling by dectin-1. *Biomol. Ther.* 2012, 20, 433–445. [CrossRef]
- 93. Reboldi, A.; Cyster, J.G. Peyer's patches: Organizing B-cell responses at the intestinal frontier. *Immunol. Rev.* 2016, 271, 230–245. [CrossRef] [PubMed]
- 94. Mahla, R.S.; Reddy, M.C.; Prasad, D.V.R.; Kumar, H. Sweeten PAMPs: Role of sugar complexed PAMPs in innate immunity and vaccine biology. *Front. Immunol.* **2013**, *4*. [CrossRef] [PubMed]
- Camilli, G.; Tabouret, G.; Quintin, J. The Complexity of Fungal β-Glucan in Health and Disease: Effects on the Mononuclear Phagocyte System. *Front. Immunol.* 2018, 9. [CrossRef] [PubMed]
- 96. Kolarich, D.; Lepenies, B.; Seeberger, P.H. Glycomics, glycoproteomics and the immune system. *Curr. Opin. Chem. Biol.* **2012**, 16, 214–220. [CrossRef] [PubMed]
- 97. Rabinovich, G.A.; van Kooyk, Y.; Cobb, B.A. Glycobiology of immune responses. Ann. N. Y. Acad. Sci. 2012, 1253, 1–15. [CrossRef]
- 98. Patin, E.C.; Thompson, A.; Orr, S.J. Pattern recognition receptors in fungal immunity. *Semin. Cell Dev. Biol.* 2019, *89*, 24–33. [CrossRef] [PubMed]
- 99. Tang, J.; Lin, G.; Langdon, W.Y.; Tao, L.; Zhang, J. Regulation of C-type lectin receptor-mediated antifungal immunity. *Front. Immunol.* **2018**, *9*, 123. [CrossRef] [PubMed]
- 100. Cohen-Kedar, S.; Baram, L.; Elad, H.; Brazowski, E.; Guzner-Gur, H.; Dotan, I. Human intestinal epithelial cells respond to β-glucans via Dectin-1 and Syk. *Eur. J. Immunol.* 2014, 44, 3729–3740. [CrossRef] [PubMed]
- Branzk, N.; Lubojemska, A.; Hardison, S.E.; Wang, Q.; Gutierrez, M.G.; Brown, G.D.; Papayannopoulos, V. Neutrophils sense microbe size and selectively release neutrophil extracellular traps in response to large pathogens. *Nat. Immunol.* 2014, 15, 1017–1025. [CrossRef]
- 102. Li, P.; Wang, F. Polysaccharides: Candidates of promising vaccine adjuvants. Drug Discov. Ther. 2015, 9, 88–93. [CrossRef]
- Kardani, K.; Bolhassani, A.; Shahbazi, S. Prime-boost vaccine strategy against viral infections: Mechanisms and benefits. *Vaccine* 2016, 34, 413–423. [CrossRef]
- 104. Haas, T.; Heidegger, S.; Wintges, A.; Bscheider, M.; Bek, S.; Fischer, J.C.; Eisenkolb, G.; Schmickl, M.; Spoerl, S.; Peschel, C.; et al. Card9 controls Dectin-1-induced T-cell cytotoxicity and tumor growth in mice. *Eur. J. Immunol.* **2017**, *47*, 872–879. [CrossRef]
- 105. Vorup-Jensen, T.; Jensen, R.K. Structural immunology of complement receptors 3 and 4. Front. Immunol. 2018, 9, 2716. [CrossRef]
- 106. Bajic, G.; Yatime, L.; Sim, R.B.; Vorup-Jensen, T.; Andersen, G.R. Structural insight on the recognition of surface-bound opsonins by the integrin i domain of complement receptor 3. *Proc. Natl. Acad. Sci. USA* **2013**, *110*, 16426–16431. [CrossRef]
- 107. Geller, A.; Shrestha, R.; Yan, J. Yeast-derived β-glucan in cancer: Novel uses of a traditional therapeutic. *Int. J. Mol. Sci.* **2019**, 20, 3618. [CrossRef]

- 108. Vera, J.; Fenutria, R.; Cañadas, O.; Figueras, M.; Mota, R.; Sarrias, M.R.; Williams, D.L.; Casals, C.; Yelamos, J.; Lozano, F. The CD5 ectodomain interacts with conserved fungal cell wall components and protects from zymosan-induced septic shock-like syndrome. *Proc. Natl. Acad. Sci. USA* 2009, 106, 1506–1511. [CrossRef]
- 109. Yang, C.; Gao, J.; Dong, H.; Zhu, P.F.; Wang, Z.G.; Jiang, J.X. Expressions of scavenger receptor, CD14 and protective mechanisms of carboxymethyl-β2-1, 3-glucan in posttraumatic endotoxemia in mice. *J. Trauma-Inj. Infect. Crit. Care* 2008, 65, 1471–1477. [CrossRef] [PubMed]
- 110. Guo, Y.; Fukuda, T.; Donai, K.; Kuroda, K.; Masuda, M.; Nakamura, S.; Yoneyama, H.; Isogai, E. Leptospiral lipopolysaccharide stimulates the expression of toll-like receptor 2 and cytokines in pig fibroblasts. *Anim. Sci. J.* **2015**, *86*, 238–244. [CrossRef]
- 111. Medzhitov, R. Recognition of microorganisms and activation of the immune response. *Nature* **2007**, *449*, 819–826. [CrossRef] [PubMed]
- 112. Adams, E.L.; Rice, P.J.; Graves, B.; Ensley, H.E.; Yu, H.; Brown, G.D.; Gordon, S.; Monteiro, M.A.; Papp-Szabo, E.; Lowman, D.W.; et al. Differential high-affinity interaction of Dectin-1 with natural or synthetic glucans is dependent upon primary structure and is influenced by polymer chain length and side-chain branching. *J. Pharmacol. Exp. Ther.* 2008, 325, 115–123. [CrossRef] [PubMed]
- Maheshwari, G.; Sowrirajan, S.; Joseph, B. Extraction and Isolation of β-Glucan from Grain Sources—A Review. J. Food Sci. 2017, 82, 1535–1545. [CrossRef] [PubMed]
- 114. Suzuki, T.; Kusano, K.; Kondo, N.; Nishikawa, K.; Kuge, T.; Ohno, N. Biological Activity of High-Purity β-1,3-1,6-Glucan Derived from the Black Yeast Aureobasidium pullulans: A Literature Review. *Nutrients* **2021**, *13*, 242. [CrossRef]
- 115. Drozdowski, L.A.; Reimer, R.A.; Temelli, F.; Bell, R.C.; Vasanthan, T.; Thomson, A.B.R. β-Glucan extracts inhibit the in vitro intestinal uptake of long-chain fatty acids and cholesterol and down-regulate genes involved in lipogenesis and lipid transport in rats. *J. Nutr. Biochem.* 2010, 21, 695–701. [CrossRef]
- 116. Tiwari, U.; Cummins, E. Meta-analysis of the effect of β-glucan intake on blood cholesterol and glucose levels. *Nutrition* 2011, 27, 1008–1016. [CrossRef] [PubMed]
- 117. McRorie, J.W.; McKeown, N.M. Understanding the Physics of Functional Fibers in the Gastrointestinal Tract: An Evidence-Based Approach to Resolving Enduring Misconceptions about Insoluble and Soluble Fiber. J. Acad. Nutr. Diet. 2017, 117, 251–264. [CrossRef] [PubMed]
- 118. Sima, P.; Vannucci, L.; Vetvicka, V. β-glucans and cholesterol (Review). Int. J. Mol. Med. 2018, 41, 1799–1808. [CrossRef] [PubMed]
- Wang, H.; Chen, G.; Li, X.; Zheng, F.; Zeng, X. Yeast β-glucan, a potential prebiotic, showed a similar probiotic activity to inulin. *Food Funct.* 2020, *11*, 10386–10396. [CrossRef]
- 120. Ooi, V.E.C.; Liu, F. Immunomodulation and Anti-Cancer Activity of Polysaccharide-Protein Complexes. *Curr. Med. Chem.* **2012**, 7, 715–729. [CrossRef] [PubMed]
- 121. Kofuji, K.; Aoki, A.; Tsubaki, K.; Konishi, M.; Isobe, T.; Murata, Y. Antioxidant Activity of β-Glucan. ISRN Pharm. 2012, 2012, 125864. [CrossRef]
- 122. Yamada, J.; Hamuro, J.; Hatanaka, H.; Hamabata, K.; Kinoshita, S. Alleviation of seasonal allergic symptoms with superfine β-1,3-glucan: A randomized study. *J. Allergy Clin. Immunol.* **2007**, *119*, 1119–1126. [CrossRef]
- 123. Quintin, J. Fungal mediated innate immune memory, what have we learned? Semin. Cell Dev. Biol. 2019, 89, 71–77. [CrossRef]
- Kamta, J.; Chaar, M.; Ande, A.; Altomare, D.A.; Ait-Oudhia, S. Advancing cancer therapy with present and emerging immunooncology approaches. *Front. Oncol.* 2017, 7, 64. [CrossRef]
- 125. Ayeka, P.A. Potential of Mushroom Compounds as Immunomodulators in Cancer Immunotherapy: A Review. *Evid.-Based Complementary Altern. Med.* 2018, 2018, 7271509. [CrossRef]
- 126. Zou, S.; Duan, B.; Xu, X. Inhibition of tumor growth by β-glucans through promoting CD4 + T cell immunomodulation and neutrophil-killing in mice. *Carbohydr. Polym.* 2019, 213, 370–381. [CrossRef]
- 127. Vetvicka, V.; Vannucci, L.; Sima, P.; Richter, J. Beta glucan: Supplement or drug? From laboratory to clinical trials. *Molecules* **2019**, 24, 1251. [CrossRef]
- 128. Vetvicka, V.; Vannucci, L.; Sima, P. B-Glucan As a New Tool in Vaccine Development. Scand. J. Immunol. 2020, 91. [CrossRef]
- 129. Temizoz, B.; Kuroda, E.; Ishii, K.J. Vaccine adjuvants as potential cancer immunotherapeutics. *Int. Immunol.* **2016**, *28*, 329–338. [CrossRef]
- Levitz, S.M.; Huang, H.; Ostroff, G.R.; Specht, C.A. Exploiting fungal cell wall components in vaccines. *Semin. Immunopathol.* 2015, 37, 199–207. [CrossRef]
- Fesel, P.H.; Zuccaro, A. β-glucan: Crucial component of the fungal cell wall and elusive MAMP in plants. *Fungal Genet. Biol.* 2016, 90, 53–60. [CrossRef]
- 132. Bazan, S.B.; Breinig, T.; Schmitt, M.J.; Breinig, F. Heat treatment improves antigen-specific T cell activation after protein delivery by several but not all yeast genera. *Vaccine* **2014**, *32*, 2591–2598. [CrossRef]
- Miyamoto, N.; Mochizuki, S.; Sakurai, K. Designing an immunocyte-targeting delivery system by use of beta-glucan. *Vaccine* 2018, 36, 186–189. [CrossRef]
- 134. Yuan, H.; Lan, P.; He, Y.; Li, C.; Ma, X. Effect of the modifications on the physicochemical and biological properties of β-glucan-a critical review. *Molecules* **2020**, *25*, 57. [CrossRef] [PubMed]
- 135. Gastl, M.; Kupetz, M.; Becker, T. Determination of Cytolytic Malt Modification–Part I: Influence of Variety Characteristics. J. Am. Soc. Brew. Chem. 2020, 79, 53–65. [CrossRef]

- 136. Castro, E.d.; Calder, P.C.; Roche, H.M. β-1,3/1,6-Glucans and Immunity: State of the Art and Future Directions. *Mol. Nutr. Food Res.* **2021**, *65*, 1901071. [CrossRef]
- Miyamoto, N.; Mochizuki, S.; Sakurai, K. Enhanced immunostimulation with crosslinked CpG-DNA/β-1,3-glucan nanoparticle through hybridization. *Chem. Lett.* 2014, 43, 991–993. [CrossRef]
- 138. Kobiyama, K.; Temizoz, B.; Kanuma, T.; Ozasa, K.; Momota, M.; Yamamoto, T.; Aoshi, T.; Kuroda, E.; Ishii, K.J. Species-dependent role of type I IFNs and IL-12 in the CTL response induced by humanized CpG complexed with β-glucan. *Eur. J. Immunol.* 2016, 46, 1142–1151. [CrossRef]
- Vetvicka, V.; Vannucci, L.; Sima, P. The Effects of β-Glucan on Pig Growth and Immunity. *Open Biochem. J.* 2014, *1*, 89–93.
  [CrossRef]
- 140. Pogue, R.; Murphy, E.J.; Fehrenbach, G.W.; Rezoagli, E.; Rowan, N.J. Exploiting immunomodulatory properties of β-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms: Concise review of existing knowledge, innovation and future opportunities. *Curr. Opin. Environ. Sci. Health* **2021**, *21*, 100248. [CrossRef]
- 141. Paris, S.; Chapat, L.; Pasin, M.; Lambiel, M.; Sharrock, T.E.; Shukla, R.; Sigoillot-Claude, C.; Bonnet, J.M.; Poulet, H.; Freyburger, L.; et al. β-Glucan-Induced Trained Immunity in Dogs. *Front. Immunol.* **2020**, *11*. [CrossRef]
- Roudi, R.; Mohammadi, S.R.; Roudbary, M.; Mohsenzadegan, M. Lung cancer and β-glucans: Review of potential therapeutic applications. *Investig. New Drugs* 2017, 35, 509–517. [CrossRef]
- 143. Vetvicka, V.; Vetvickova, J. Comparison of Immunological Effects of Commercially Available β-Glucans. 2014. Available online: http://www.hoajonline.com/journals/pdf/2054-9903-1-1.pdf (accessed on 18 March 2021).
- 144. Zaleska, B.; Milewski, S.; Zabek, K. Impact of Saccharomyces cerevisiae supplementation on reproductive performance, milk yield in ewes and offspring growth. *Arch. Anim. Breed.* **2015**, *58*, 79–83. [CrossRef]
- 145. Zabek, K.; Milewski, S.; Wójcik, R.; Siwicki, A.K. Effect of β-1,3/1,6-D-glucan in diet on productivity and humoral and cellular defense mechanisms in sheep. Acta Vet. Brno 2013, 82, 141–146. [CrossRef]
- 146. Khalkhane, A.S.; Habibian, R. Effect of dietary B-glucan supplementation on humoral and cellular immunologic factors in lambs. *Glob. Vet.* **2013**, *11*, 38–43. [CrossRef]
- 147. Stuyven, E.; Cox, E.; Vancaeneghem, S.; Arnouts, S.; Deprez, P.; Goddeeris, B.M. Effect of β-glucans on an ETEC infection in piglets. *Vet. Immunol. Immunopathol.* **2009**, *128*, 60–66. [CrossRef]
- 148. Shao, Y.; Guo, Y.; Wang, Z. β-1,3/1,6-Glucan alleviated intestinal mucosal barrier impairment of broiler chickens challenged with Salmonella enterica serovar Typhimurium. *Poult. Sci.* **2013**, *92*, 1764–1773. [CrossRef]
- Huff, G.R.; Huff, W.E.; Farnell, M.B.; Rath, N.C.; de los Santos, F.S.; Donoghue, A.M. Bacterial clearance, heterophil function, and hematological parameters of transport-stressed turkey poults supplemented with dietary yeast extract. *Poult. Sci.* 2010, 89, 447–456. [CrossRef] [PubMed]
- Mejía, S.M.V.; de Francisco, A.; Bohrer, B.M. A comprehensive review on cereal β-glucan: Extraction, characterization, causes of degradation, and food application. *Crit. Rev. Food Sci. Nutr.* 2020, *60*, 3693–3704. [CrossRef] [PubMed]
- 151. Izydorczyk, M.S.; Dexter, J.E. Barley β-glucans and arabinoxylans: Molecular structure, physicochemical properties, and uses in food products-a Review. *Food Res. Int.* **2008**, *41*, 850–868. [CrossRef]
- 152. Piñero, M.P.; Parra, K.; Huerta-Leidenz, N.; de Moreno, L.A.; Ferrer, M.; Araujo, S.; Barboza, Y. Effect of oat's soluble fibre (β-glucan) as a fat replacer on physical, chemical, microbiological and sensory properties of low-fat beef patties. *Meat Sci.* 2008, 80, 675–680. [CrossRef]
- 153. Liu, R.; Wang, N.; Li, Q.; Zhang, M. Comparative studies on physicochemical properties of raw and hydrolyzed oat β-glucan and their application in low-fat meatballs. *Food Hydrocoll.* **2015**, *51*, 424–431. [CrossRef]
- 154. El Khoury, D.; Cuda, C.; Luhovyy, B.L.; Anderson, G.H. Beta glucan: Health benefits in obesity and metabolic syndrome. *J. Nutr. Metab.* **2012**, 2012, 851362. [CrossRef]
- 155. Ryan, P.M.; London, L.E.E.; Bjorndahl, T.C.; Mandal, R.; Murphy, K.; Fitzgerald, G.F.; Shanahan, F.; Ross, R.P.; Wishart, D.S.; Caplice, N.M.; et al. Microbiome and metabolome modifying effects of several cardiovascular disease interventions in apo-E-/mice. *Microbiome* 2017, 5, 30. [CrossRef] [PubMed]
- 156. Steiner, J.; Procopio, S.; Becker, T. Brewer's spent grain: Source of value-added polysaccharides for the food industry in reference to the health claims. *Eur. Food Res. Technol.* **2015**, *241*, 303–315. [CrossRef]



Review

# Food Ingredients and Active Compounds against the Coronavirus Disease (COVID-19) Pandemic: A Comprehensive Review

Charis M. Galanakis <sup>1,2,3,\*</sup>, Turki M.S. Aldawoud <sup>2</sup>, Myrto Rizou <sup>1</sup>, Neil J. Rowan <sup>4,5</sup>, and Salam A. Ibrahim <sup>6,\*</sup>

- <sup>1</sup> Research & Innovation Department, Galanakis Laboratories, 73131 Chania, Greece; myrtorizou@chemlab.gr
- <sup>2</sup> College of Science, King Saud University, Riyadh 12372, Saudi Arabia; tdawoud@ksu.edu.sa
- <sup>3</sup> Food Waste Recovery Group, ISEKI Food Association, 1190 Vienna, Austria
- <sup>4</sup> Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, N37 F6D7 Athlone, Ireland; nrowan@ait.ie
- <sup>5</sup> Empower Eco Sustainability Hub, Lough Boora, R35 DA50 Co. Offaly, Ireland
- <sup>6</sup> Food and Nutritional Sciences Program, North Carolina Agricultural and Technical State University, Greensboro, NC 27401, USA
- \* Correspondence: cgalanakis@chemlab.gr (C.M.G.); ibrah001@ncat.edu (S.A.I.)

Received: 30 September 2020; Accepted: 16 November 2020; Published: 20 November 2020



MDF

Abstract: As media reports have noted, the COVID-19 pandemic has accelerated market mainstreaming of immune-boosting food bioactives, supplements, and nutraceuticals. However, most studies reporting on the potential of bioactives against COVID-19 transmission have been uploaded as preprints with little opportunity to revise content for benefit and impact. The current review discusses current best evidence and information underpinning the role of food ingredients and bioactive compounds in supporting immune functions in humans and animals, specifically in the prevention and treatment of COVID-19 disease. Up to now, some evidence from randomized population and clinical trials has suggested that vitamin D levels may be linked to COVID-19 transmission and severity. Numerous theoretical studies have pointed to polyphenols and particularly flavonoids as potential inhibitors of SARS-CoV-2 infection. There is also inconclusive evidence to support the future use of  $\beta$ -glucan to address COVID-19 due in part to variability in immune response arising from heterogeneity in polysaccharide branch and chain length for different sources and the absence of a standardized extraction method. To confirm the promising outcomes and hypotheses for bioactive compounds, more randomized and controlled clinical studies are needed. The results of such studies would have a profound effect on the prospects of food supplements and nutraceuticals as potential prophylaxis against COVID-19 and serve to help consumers to protect themselves during the post-lockdown recovery era.

Keywords: SARS-CoV-2; COVID-19; vitamin D; immune system; functional foods; nutraceuticals

# 1. Introduction

Over the past 20 years, much scientific focus has been trained on the identification of active ingredients in food products (e.g., vitamins) that promote human health. The terms "functional foods" and "superfoods" have become very popular concerning foods and food products that claim health benefits. To follow this trend, numerous food ingredients and active compounds have been investigated as health-promoting agents with possible antimicrobial functions, anti-inflammatory activities, and potential antiviral actions [1]. Except for the vitamins, the compounds under investigation include bioactive peptides, polysaccharides, bioactive lipids, and natural polyphenols [2]. Nutraceuticals' demand increases despite the false efficacy claims, the concerns about quality, the lack of clinical

evidence, and the self-medication implications for severe illness [3]. There is still uncertainty about the "nutraceutical" term, as these products have been reported as neither nutritious nor pharmaceutical. In many countries, there is no legal definition of "functional food" and "nutraceuticals" terms that are used interchangeably. The term "nutraceutical" (originally coined by Stephen DeFelice in 1989) is broadly used to include various naturally occurring products, such as fortified foods, functional foods, and food supplements. However, while the definition of a "food supplement" is clear, the definition of a "nutraceutical" (syncretic neologism of the words nutrient and pharmaceutical) fits in the grey area between food, food supplements, and pharmaceuticals [4]. In addition to their essential nutritional value, nutraceuticals are intended to promote health benefits, where these products typically comprise ingredients that are referred to as generally recognized as safe (GRAS) internationally.

As the world entered the COVID-19 pandemic, the market of bioactive ingredients supporting the immune system was among the four aspects of food systems (including also food safety, food security, and sustainability) directly affected by this crisis [5,6]. In particular, consumers around the world stocked up on vitamin C and other botanical ingredients, whereas panic buying and shortages ensued [7,8]. Subsequently, the efforts of researchers and the scientific community to investigate bioactive compounds that could support the immune system, protect against viruses of the lower respiratory tract, and restrict the transmission of SARS-CoV-2 (the novel coronavirus causing COVID-19 disease) accelerated, resulting in some recent reviews summarizing the potential of natural compounds against SARS-Cov-2 and coronaviruses in general [9–11].

Since there is still relatively limited knowledge about SARS-CoV-2, the urgent need for fast-track prevention and treatment strategies has led researchers to publish articles in "torrents" [12]. The disadvantage of this expeditious approach to glean new knowledge is that the associated surge in recently published studies has as yet to be subjected to any review for benefit or impact. It is envisaged that incremental innovations sustaining agri-food and new products will be influenced by pressing need for new health interventions to combat COVID-19, along with responding to a surge in interest in accelerating green deal innovations and climate change [13].

Therefore, this current timely study provides a comprehensive review on the role of food and plant ingredients and active compounds against COVID-19 disease, as claimed in the torrents of early and fast-released studies. In addition, it commensurately explores the important contribution of food supplements and nutraceuticals in supporting a healthy society including considering "long COVID-19" for survivors [14]. For example, peripheral inflammation caused by COVID-19 may have long-term consequences in those that recover, leading to chronic medical conditions such as dementia and neurodegenerative disease, likely through neuroinflammatory mechanisms that can be compounded by an unhealthy diet [14]. Thus, there is a pressing need for wider access to healthy foods, and people should be made aware that healthy eating habits may reduce susceptibility to and long-term complications from COVID-19.

### 2. The Role of Food Ingredients and Active Compounds in Supporting the Human Immune System

As it is well known from mechanistic and clinical data, vitamins and folate, polysaccharides and dietary fiber, lipids, peptides, and natural polyphenols are important for the body's immune system against viruses [1,15,16]. Table 1 presents a summary of the health benefits and possible action mode against SARS-CoV-2 virus of food ingredients and bioactive compounds. These bioactive compounds can be found in fruits and vegetables, the consumption of which is highly recommended [17,18]. For example, citrus, kiwi, and vegetables such as broccoli contain high amounts of vitamin C (ascorbic acid), which is necessary for the repair of body tissues and immune's function [19]. Likewise, vitamin C can restrict infection of the lower respiratory tract under certain conditions [20] and prevent the common cold [21]. Vegetables such as sweet potato, spinach, and carrots contain high amounts of vitamin A (comprised of oil-soluble compounds such as  $\beta$ -carotene, retinol, and retinoic acid). Vitamin A is known to support immune function and protect against infections when administered

at pharmacological or high dosage (10 nM or higher) due to its regulatory roles in humoral immune processes and cellular immune responses [22].

Bioactive peptides are composed of several of amino acids arranged in different configurations, and their molecular weights are <6000 kDa. Currently, more than 1500 different bioactive peptides have been reported. Although the correlation between structure and functional properties is not well established, many bioactive peptides share some structural features that include a peptide residue length between 2–20 amino acids, and the presence of hydrophobic amino acids in addition to proline, lysine, or arginine groups [23]. Bioactives are classified by their action mode (e.g., antimicrobial, antioxidant, immunomodulatory, anti-inflammatory, antithrombotic) and their binding capacity to micronutrients such as minerals. Dairy products are the primary sources of peptides, e.g., < 3 and 3–10 kDa peptide fractions obtained from fermented milks with specific Lactobacillus plantarum strains [24]. However, peptides can be obtained from other animal sources such as bovine blood, gelatin, meat, egg, and fish. Some plant sources of bioactive peptides are soy peanut, sorghum, pumpkins, mushroom, and rice. Peptides can also be released during the fermentation process of cheese, yogurt, and other fermented products. Peptides can be produced during cheese ripening, whereas lactic acid bacteria in general can produce a wide range of them during fermentation. For instance, some anti-inflammatory activities of peptide fractions can be obtained from fermented milk with specific *Lactobacillus plantarum* strains [24]. The probiotic strain of Lactobacillus gasseri SBT2055 (LG2055) has also been shown to reduce virus respiratory infection by suppressing the replication of the virus. Results demonstrated that following the respiratory syncytial infection, LG2055 enhanced the expression of IFN- $\beta$  and IFN- $\gamma$  at the gene level in the mice's lungs. Another study demonstrated that several lactic acid bacteria-induced small protein molecules known as immune interferon (IFN- $\gamma$  or IFN- $\beta$ ) in the mice's lungs, contributing to clearance of respiratory virus [23]. Finally, another study has shown that L. paracasei strains cultured in the presence of artichokes' phenolic extracts (4 mg/mL) have anti-inflammatory effects on dendritic cells, suggesting potential synergistic effects between different bioactive compounds [25].

Polysaccharides are complex carbohydrate polymers that are naturally found in various food sources including seaweeds and cereal grains and are also produced by microorganisms such as lactic acid bacteria. Polysaccharides have many bioactivities such as antioxidant, immunomodulatory, antidiabetic, anticancer, as well as liver and renal protective functions. A large number of studies have been conducted on the isolation and structural identification of polysaccharides from edible and medicinal mushrooms, and a total of more than forty polysaccharides were reported. In addition, these studies have shown that polysaccharides from mushrooms have antivirus activity against a wide range of viruses, including hepatitis B, influenza, enterovirus, herpes simplex virus, porcine circovirus, rotavirus, and others. For instance, Coriolus versicolor's polysaccharide peptide has been demonstrated to possess immunomodulatory properties with the ability to triggers a Toll-like receptor 4 showing insignificant toxicity [26]. One explanation for this antiviral activity could be attributed to the restriction of adsorption and penetration of the virus. The polysaccharides from mushrooms exhibited potential anti-HIV activity by downregulating replication of the virus and upregulating certain antiviral chemokines (Stromal Cell Derived Factor-1 alpha, SDF-1 $\alpha$ , and Macrophage Inflammatory Protein, MIP-1 $\alpha/\beta$ ). These chemokines block the coreceptors of HIV-1 in THP1 cells from leukemia patients and blood mononuclear cells [26].

Compounds	Type of Study	Health Benefits	Mode of Action against SARS-CoV-2
Bioactive peptides (e.g., <3 and 3–10 kDa peptide fractions obtained from fermented milks with specific <i>Lactobacillus plantarum</i> strains)	In vitro	Control hormone release, anti-inflammatory, anti-hemolytic, anti-mutagenic, antioxidant and antimicrobial activities [23–25]	Disruption of viral spike protein [26]
Polysaccharides	In vitro In vivo	Antiviral activity stimulate ROS <sup>1</sup> , reduce risk factors for chronic diseases, improve metabolism and digestibility [27]	Reduction in inflammatory responses Prevention of ARDS <sup>2</sup> [28-32]
Vitamins (A, C, E, and D)	In vitro Clinical	Support immune function Protect against infections Prevent common cold [19–22]	Restriction of ACE2 <sup>3</sup> activity Promotion of innate immunity [33–38]
Medicinal Herbs	Clinical In vitro	Prevention of influenza viruses [33,34,39,40]	Improve COVID-19 patients recovery [41,42]
Bioactive lipids (fatty acids, phytosterols, carotenoids)	In vivo	Enhance immune response Anti-inflammatory activities Reduce risk of cardiovascular diseases [43]	Inhibition of ACE <sup>4</sup> and restriction of virus ability to enter the cells [44]
Natural polyphenols (flavonoids, phenolic acids, stilbenes, lignans)	In vitro In silico	Anti-inflammatory, antimicrobial and antioxidant activities, antiviral capacity, prevent digestion issues, reduce the risk of chronic diseases [45–53]	Inhibition of viral replication Disruption of viral spike protein Inhibition of SARS-CoV-2 protease [26,54–69]

Table 1. Health benefits and possible action mode against SARS-CoV-2 virus of food ingredients and bioactive compounds.

<sup>1</sup> ROS: reactive oxygen species; <sup>2</sup> ARDS: acute respiratory distress syndrome; <sup>3</sup> ACE2: angiotensin-converting enzyme 2; <sup>4</sup> ACE: angiotensin-converting enzyme.

Bioactive lipids comprise several endogenous molecules affecting a wide array of biological processes. Typical examples of lipids include omega-3 fatty acids and their metabolic products, carotenoids, phytosterols, and fat-soluble vitamins, phenolic lipids, and acylglycerol derivatives. Numerous studies have found an association between the consumption of certain bioactive lipids and the prevention, delay, or treatment of chronic and acute diseases such as cardiovascular disease, cancer, osteoporosis, and immune disorders. One possible protective mechanism for the bioactive lipids against virus infection is the prevention of inflammation. Bioactive lipids such as oleic acid have exhibited antiviral protection by inducing leakage and even lysis of cell membranes including the virus lipid membrane. It is also known that these bioactive lipids show phagocytic capacity of macrophages and can facilitate the removal of any damage by viral infection [27].

Polyphenolic compounds from plant foods, extracts, and food processing by-products have well known antioxidant activities [28–32]. Polyphenolic compounds can be subdivided into flavonoids, phenolic acid, polyphenolic amides, resveratrol, and other polyphenols. These compounds are antioxidants and also have antimicrobial and antiviral activities. Several recent studies explored the in vitro antiviral capacity of different polyphenols [33–35]. In addition, Vázquez-Calvo et al. [36] conducted a study related to the effect of polyphenols (e.g., epigallocatechin, epigallocatechin gallate, epicatechin, catechin, cyanidin, and delphinidin) on Dengue, Zika, and Nile viruses. These enveloped plus-strand RNA viruses are transmitted by mosquitoes, posing a serious threat to health. The results of this study showed that the aforementioned polyphenols influenced the attachment and entry step of the RNA into the host cells were dose-dependent in a 1–10  $\mu$ M range and thereby reduced the infectivity of the viruses.

#### 3. Food Ingredients and Active Compounds against COVID-19 Disease

The oral supplementation or intravenous administration with food bioactives and nutraceuticals has been proposed as an alternative approach against COVID-19 disease based mainly on their anti-inflammatory properties, but also to reflect their ability to inhibit viruses' (e.g., SARS-CoV, MERS-CoV and SARS-CoV-2) activity by disrupting their protein envelopes [37]. For instance, these compounds can enhance the response of type 1 interferon to RNA viruses such as influenza and coronavirus [38]. The possible mode of action against SARS-CoV-2 of several bioactive compounds and food ingredients is presented in Table 1.

Plant secondary metabolites (e.g.,  $\beta$ -carboline, quinoline alkaloids like cinchonine, skimmianin, dictamine, and quinine, as well as isoquinoline alkaloids such as emetine, berberine, and sanguinarine) can act as DNA intercalators (similar to chloroquine's proposed in vitro mechanisms of action against SARS-CoV-2) inhibiting the replication of the virus [39]. Another anti-replication mechanism is through binding of angiotensin-converting enzyme 2 (ACE-2). This enzyme is a type I transmembrane metallocarboxypeptidase that mainly expressed in the renal tubular epithelium, and the vascular endothelial cells, but also in the lung, and the kidney [40–42]. Different studies have referred to ACE-2 as a cellular entry for SARS-CoV-2 in the body together with other factors [43,44]. In particular, the spike proteins of the virus surface binds on ACE-2 and diffuses the virus into the target cells [45]. Subsequently, compounds that exhibit a binding affinity for the core amino acid of ACE-2 have been proposed as potential preventive or therapeutic agents (these compounds mainly attack free viral particles and to a lesser extent viruses that have already entered host cells), since they are able to interfere with or avoid the host-viral interaction [39,46]. For example, omega-3 fatty acids and their metabolites inhibit angiotensin-converting enzymes (ACE, which is a precursor of ACE-2) and thus suppress ACE2-expression, reducing the availability of receptors to SARS-CoV-2 and subsequently restricting its ability to enter the target cell [37,47].

Using simulations of binding free energy and molecular dynamics, stilbene-based compounds such as resveratrol has been shown to be potential anti-COVID-19 candidates that can theoretically disrupt the virus's spike protein [48]. Relevant action mechanisms have also been proposed for other bioactive compounds such as folic acid. Folic acid is a water-soluble form of vitamin B that can inhibit furin activity that facilitates the cleavage between ACE-2 and spike proteins of SARS-CoV-2 [49]. Those proteins have also been studied as possible targets of bioactive peptides in mitigation of SARS-CoV-2 pathology. In particular, bioactive peptides with unique amino acid sequences can mitigate such targets including, type II transmembrane serine proteases (TMPRSS2) inhibition, furin cleavage, and the renin–angiotensin–aldosterone system (RAAS) members. Based on current evidence and structure-function analysis, multiple bioactive peptides present potency to neutralize the virus [50].

Adem et al. [51] conducted a molecular docking study to identify the ability of 80 flavonoid compounds to bind 3-chymotrypsin-like protease (3CLpro), which is known to be an important enzyme for the replication of SARS-CoV. Other polyphenols and flavonoids such as protocatechuic acid, punicalagin, theaflavin gallate, kaempferol, theaflavin digallate [52], pedunculagin, tercatain, punicalin, [53], epigallocatechin gallate [54], riboflavin, daidzein, genistein, phycocyanobilin, cyanidin [55] hispidin, lepidine E [56], and hesperidin [57] have been proposed as potential inhibitors of the main SARS-CoV-2 protease in similar studies. Besides, hesperidin and other flavonoids have been referred to possess better binding poses than common drugs against COVID-19, e.g., nelfinavir, chloroquine, and hydroxychloroquine sulfate [51,58]. Another phenolic compound, quercetin, is known to display a broad range of antiviral properties that can interfere at multiple steps of pathogen virulence (virus entry, replication, and protein assembly). These therapeutic effects can be augmented by the co-administration of vitamin C. Furthermore, quercetin has recently been identified as an inhibitor of SARS-CoV-2 3CLpro [59], and also, it has been shown to act synergistically with vitamin D and have anti-viral action against SARS-CoV-2 with the proposed dosage for both substances being up to 500 mg taken two times a day for prophylaxis and mild cases [60]. Considering the different flavonoids, a thorough bioinformatic analysis of small molecules interacting with ACE-2 using the SUMMIT supercomputer indicated the structural analogue eriodictyol (5,7,30,40-tetrahydroxyflavanone) of luteolin as the most potential inhibitor of SARS-CoV-2 [61]. Finally, both quercetin and other flavonoids from different plants (e.g., litchi seeds) have shown similar activity when applied in traditional Chinese medicine [62]. In general, supplementation with dietary flavonoids up to 1-2 g/day is considered to be safe, but higher cumulative dosages should be avoided, as they can negatively affect the metabolism of the liver. Flavonoids can be obtained from different plant sources, but it is peanut shells that may affect persons allergic to peanuts, or fava beans, consumption of which could cause hemolytic anemia to Mediterranean extraction persons who lack the enzyme glucose-6-phosphate dehydrogenase (G6PD). Such patients should also not be administered the antimalarial drugs chloroquine and hydroxychloroquine, which have been advocated based on anecdotal reports for the treatment of COVID-19 [63,64].

Nevertheless, although these theoretical approaches and hypotheses about the potential role of plant secondary metabolites against COVID-19 seem to be reasonable, most of them have not been justified yet either with in vitro studies or with clinical trials. In one of the first relevant studies, Murphy et al. [65] used an in vitro lung injury model to study the activity of  $\beta$ -glucans from the *Shittake* mushroom. The authors using 1, 5–10 mg/mL of  $\beta$ -glucans reported that they reduce inflammatory responses associated with acute respiratory distress syndrome (ARDS) in vitro that includes reduced cytokine production, oxidative stress, necrosis, and apoptosis, suggesting potential for amelioration of SARS-COV-2 through a role in preventing cytokine storm. The latest occurs when white blood cells accumulate by infections such as SARS-CoV-2, releasing inflammatory cytokines. Differential in vitro immunomodulatory and pulmonary cytoprotective effects were attributed in part to variance in  $\beta$ -glucan structure that will have cross-cutting relevance to COVID-19 interventions. ARDS was evident in severe complications experienced by COVID-19 patients in China admitted to intensive care units [66], where the specific involvement in reducing cytokines IL-1 $\beta$  and IL-6 production is a strategy for COVID-19 intervention [67]. Although not studied yet,  $\beta$ -glucan from other sources could result in similar responses, e.g.,  $\beta$ -glucan derived from algae (in a diet of 108 mg/kg) has been shown to enhance immune responses of weaned pigs experimentally infected with a pathogenic E. coli. In particular, the supplementation of  $\beta$ -glucan also reduced white blood cells, neutrophils, serum tumor necrosis factor (TNF)- $\alpha$ , cortisol, and haptoglobin and down-regulated (p < 0.05) the mRNA expression of several immune genes (IL1B, IL6, and TNFA) in ileal mucosa of E. coli challenged pigs, compared with the control diet [68].

On the other hand, Hetland et al. [69] recently highlighted the potential for medicinal mushrooms, such as *Basidiomyceta Agaricus blazei Murill*, *Hericium erinaceus*, and *Grifola frondosa*, for prophylactic and therapeutic potential against severe lung inflammation that often follows COVID-19 infection. An AbM-based mushroom extract (Andosan<sup>™</sup>), also containing *Hericium erinaceus* and *Grifola frondosa*, has been shown to significantly reduce bacteremia and increase survival in mice with pneumococcal sepsis and to improve symptoms and quality of life of patients with inflammatory bowel disease via an anti-inflammatory effect. Hence, such mushroom extracts could have prophylactic or therapeutic effect against the pneumonic superinfection and severe lung inflammation that often complicates COVID-19 infection.

Besides, several authors highlighted the potential benefits of boosting immune response and wellbeing of individuals undergoing self-quarantine for prevention and treatment of COVID-19; such as, the consumption of foods rich in vitamins A and D, zinc, and selenium that are found in mushroom, oily fish (salmon, sardines), egg yolk, milk, cheese, whole grain and wheat bran cereals, and dark green leafy vegetables [70–72].

Novel therapies are urgently required to address ARDS that is associated with severe respiratory failure with approximately 40% mortality [73]. Up-regulation of pro-inflammatory cytokines associated with lung parenchyma injury is an important causation element of ARDS [74,75].  $\beta$ -Glucans from Shiitake also reduced populations of a multiple-antibiotic resistant isolate *Klebsiella pneumoniae* in an in vivo lung infection model [76,77]. Findings showed that  $\beta$ -glucan improved lung physiological parameters, reduced white cell count protein inflammation to the lung, reduced bacterial levels in the bronchoalveolar lavage and arterial blood parameter, and supported vital partial pressure of oxygen (pO2) along with promoting lung cellular repair. The same research group recently reported on a novel means of delivering nebulized bioactives in aerosols for effective lung delivery that has potential implications for use of smart nutraceuticals in future treatment and recovery of COVID-19 patients [78]. The aforementioned also supports the findings of Bediril et al. [79], whereas in an experimental sepsis model,  $\beta$ -glucan attenuated inflammatory cytokine release (tumor necrosis factor- $\alpha$ , interleukin-1 $\beta$  and interleukin- $\beta$ ) and prevented acute lung injury. While there is emergence of smart infection models, such as for lung [65,76,77] or gut delivery [80], there is a pressing need to pursue controlled clinical studies on use of  $\beta$ -glucan to prove efficacy for human health.

Moreover, Luo et al. [81] reported on the use of bilberry (Vaccinium myrtillus Linnaeus), which investigated the anti-inflammatory effects of bilberry extract (containing 42% anthocyanin) on alleviating liver injury and croton oil-induced ear edema using a rodent infection model. In particular, bilberry extract inhibited effectively liver inflammation and croton oil-induced ear edema caused by P. acnes. The administration of bilberry extract suppressed the protein levels of nuclear factor-kB, tumor necrosis factor- $\alpha$ , and inducible nitric oxide synthase, and the increase in liver mRNA levels of interleukin-6, tumor necrosis factor- $\alpha$  (TNF- $\alpha$ ), interleukin-1 $\beta$ , and inducible nitric oxide synthase. Bilberry extract treatment also reduced liver malondialdehyde and nitric oxide contents. These results suggest anti-inflammatory potential for future use in natural products and healthy food, such as for COVID-19 needs. Garden blue blueberry (Vaccinium ashei Reade) anthocyanin extracts were also recently reported for their capacity and underlying mechanisms in protecting from lipopolysaccharide (LPS)-stimulated inflammation in vitro [82]. Enzyme-linked immunosorbent assay (ELISA) studies indicated that blueberry extract restricted importantly the production of interleukin-1 $\beta$ , prostaglandin E2, interferon- $\gamma$ , and nitric oxide. Further analysis with real-time PCR showed that in LPS-stimulated RAW 264.7 cells, the mRNA expression levels of cyclooxygenase 2, tumor necrosis factor- $\alpha$ , monocyte chemoattractant protein-1, interleukin-6, and interleukin-1 $\beta$  were suppressed. However, detailed

Other reviews and studies discuss the role of vitamins (particularly A, E, and D) against COVID-19 transmission. For example, isotretinoin, which is a derivative of vitamin A, was recently suggested to restrict the activity of angiotensin-converting enzyme 2 (ACE2) [83]. In contrast, it is known that the decrease in levels of vitamin D and E in cattle could increase the infection possibilities by bovine coronavirus [84]. Besides, the increased risk of acute respiratory distress and lung injury syndrome is attributed to vitamin D deficiency. In particular, it is known that vitamin D promotes innate immunity processes, e.g., it suppresses adaptive immunity by decreasing the maturation of dendritic cells, diminishing the ability of dendritic cells to present antigen to CD4 cells, suppressing the proliferation of CD4 cells, and their differentiation into Th1 and Th17, and via promotion of Th2 and Treg [85]. Taking into account that vitamin D possesses immunomodulatory and anti-inflammatory properties as well as antiviral effects indicated by numerous studies with well-established data [86], the respective supplementation of patients may support their immune system against COVID-19 [87]. Ilie et al. [88] confirmed this using randomized population studies. More specifically, the mean levels of vitamin D for 20 European countries and morbidity and mortality caused by COVID-19 were acquired. Negative correlations between mean levels of vitamin D (average 56 mmol/L  $\pm$  10.61) in each country and the number of COVID-19 cases/1 M population (mean  $295.95 \pm 298.7$ , and mortality/1 M population (mean  $5.96 \pm 15.13$ ) were observed. Vitamin D levels are severely low in the aging population especially in Spain, Italy, and Switzerland. In another effort, Daneshkhah et al. [89] combined data from two clinical studies to suggest that vitamin D may reduce COVID-19 severity by suppressing the cytokine storm in patients. A possible link between high C-reactive protein (CRP) and vitamin D deficiency and calculated an odds ratio of 1.8 among the elderly (age greater than or equal to 60 years) in low-income families and an odds ratio of 1.9 among the elderly (age greater than or equal to 60 years) in high-income families. COVID-19 patient-level data show a notable odds ratio of 3.4 for high CRP in severe COVID-19 patients. Lau et al. [90] conducted a retrospective observational study, reviewing the medical records of COVID-19 and suggesting that vitamin D insufficiency may play a role in the progress of COVID-19 disease. However, in order to validate this kind of hypotheses, more randomized controlled studies are needed, since the latest investigation did not adequately address risk-stratify subjects.

Another prevention strategy against the transmission of SARS-CoV-2 may be the consumption of herbs. Herbal medicines (e.g., traditional Chinese or Unani) are rich in dietary antioxidants such as polyphenols and vitamins and have shown some promising results for the treatment of relevant diseases. For example, Astragulus membranaceus extracts have shown in vitro anti-influenza virus activity [91,92]. Moreover, the consumption of ginseng root has been proposed for the prevention of influenza [93]. Hui et al. [94] revised the outcomes of population, cohort, and clinical studies that used Chinese herbal formulas for the prevention of SARS and H1N1 influenza transmission in older people and high-risk populations, suggesting the potential protective role of such formulas against COVID-19 disease. The most frequently used herbs in these studies included Fructus forsythia (Lianqiao), Radix glycyrrhizae (Gancao), Radix saposhnikoviae (Fangfeng), Radix astragali (Huangqi), Lonicerae Japonicae Flos (Jinyinhua), and Rhizoma Atractylodis Macrocephalae (Baizhu). Luo et al. [95] conducted an empirical study based on the treatment of 54 COVID-19 patients from Wuhan (China), noting that traditional Chinese medicine improved patients' recovery. Ang et al. [96] analyzed 28 (26 Chinese and two Korean government-issued) guidelines and numerous herbal formulas, identifying different patterns for mild, moderate, severe, and recovery stages of COVID-19 disease. Glycyrrhizae Radix et Rhizoma was the most used herb in the Chinese guidelines. Silveira et al. [97] studied the possible positive effects of 39 herbal medicines on COVID-19 patients and concluded that five of them (Althaea officinalis, Commiphora molmol, Glycyrrhiza glabra, Hedera helix and Sambucus nigra) can be used as immune system boosters in early and mild cases, whereas the benefits/risks assessment of 12 (Allium sativum, Andrographis paniculata, Echinacea angustifolia, Echinacea purpurea, Eucalyptus globules essential oil, Justicia pectoralis, Magnolia officinalis, Mikania glomerata, Pelargonium sidoides, Pimpinella anisum, Salix sp, Zingiber

66

*officinale*) others were promising. Similarly, ibuprofen showed promising results, although there was not enough evidence to endorse the use of paracetamol and/or codeine. In any case, it is important to confirm the potential protective role of the above formulas with prospective and rigorous clinical studies [94,96].

# 4. The Prospects of Food Supplements and Nutraceuticals in the Era of the COVID-19 Pandemic

The pandemic has disrupted citizen purchase behavior in a way that is referred to as the "ripple effect"; an upstream propagation of the disruptions to all actors involved in the supply chains. Subsequently, a considerable effort is required by authorities to support the primary sector in terms of economy planification, digitalization, and products' eco-labeling [98]. To this line, forthcoming technologies for the agri-food sector will be influenced by the growing demand to produce safer and nutritious foods to meet growing populations that reflects dynamic changes in eating habits such as personalized nutrition, alternative protein sources, and attitudes towards climate change and digitization [99]. The interplay of food insecurity, malnutrition, and obesity on dietary behaviors amid the COVID-19 pandemic indicate opportunities for stakeholders to address social and structural determinants of healthy eating as a treatment strategy to improve the health span of individuals [100].

Thereby, the COVID-19 pandemic has widely affected the food sector with consumers increasingly seeking sustainable, organic, and functional foods [101,102]. Immunity was among consumers' highest priorities, but the current pandemic has forced them to re-evaluate their eating patterns and lifestyles. According to a survey of 23,000 individual consumers by FMCG Gurus [103], 72% of European shoppers are eager to change their eating behavior by turning to healthier choices following the era of COVID-19 pandemic. Consequently, products fortified with bioactive ingredients and nutraceuticals are more popular than ever [104]. These products may be developed from food processing by-products [30,105], plants, yeast, seaweeds, algae [106], fungi, or mushrooms that reduce inflammatory responses that are typically associated with cytokine storm in severe COVID-19 patients. At the same time, there is a significant interest from existing businesses and various start-ups in animal protein alternatives, bio-based fibers, and other bio-based products (e.g., biofuels to bioplastics) that fall in the concept of the circular bioeconomy. In order to be socially accepted, the latest needs to rely on residual bio-based feedstock and waste, hence reducing its dependency on crops that compete with food markets [99].

Following this trend, several companies are releasing relevant products such as chocolate balls containing  $\beta$ -glucan (recovered from mushrooms) that are targeting improving children's immunity during the post-lockdown period [107]. Subsequently, the food industry claims more recognition of immune-boosting ingredients [108], as well as highlighting the emerging need for more systematic research and collaboration with academic and governmental institutions in this field [109,110]. Indeed, despite many studies suggesting the protective role of food ingredients, the legal framework for relevant health and nutritional claims is strict, especially in Europe [111]. In this regard, organizations such as the Food and Drug Administration (FDA) and the Global Organization for Eicosapentaenoic Acid and Docosahexaenoic Acid (GOED) have released warnings and informative letters for the related food industries to avoid product claims of general immunity. It is feared that such premature or untested claims could lead to further confusion with regard to food and nutrition-related prevention or treatment of COVID-19 disease [112]. Chinese authorities have gone so far as to call out the names of products officially (e.g., probiotic pulp, vegetables, and tea) and respective companies that have unproven health claims against COVID-19 [112].

COVID-19 will present both challenges and opportunities for the development of new innovative nutraceuticals and business models that may lead to techno-socioeconomic disruption in the food ecosystem throughout the supply chain and marketplace [9]. This disruption will lead to innovative products that converge disciplines such as innovation in manufacturing, services, business processes, and Internet and communication technologies (ICT). Virtual accelerator hubs for connecting micro with small and medium enterprises (SMEs) that exploit advances in ICT through immersive technologies will become more popular for informing disruptive innovation in situ and for remote end-user

applications. This tendency will enable hurdling restrictions that may come with networking and training innovators or employees in meeting rooms that may persist as a barrier to innovate for post-COVID-19 disruptors [99]. Priority is to achieve international consensus on datasets to harmonize methods for reliable and repeatable processing and to inform clinical trials [10] that can be facilitated by promoting open access to findings. COVID-19 pandemic has provided an almost instantaneous void or dearth in critical information to inform consumer market preferences, beliefs, perceptions, attitudes, and barriers towards change that will meet this particular need during and post this pandemic [113]. There is a pressing need to exploit existing or to create new multi-agency enterprise hubs related to academia that will support and accelerate innovators and businesses (such as the ones in agri-food), and full-span commercialization of products and services using the nine-stage technology readiness assessment. This fact should include off-site pilot-data generation where there is an increasing trajectory towards sustainable innovation, the green agenda, and digitization [98]. Therefore, establishing convincing and compelling evidenced-based information as to the real benefits to human health from existing and new nutraceuticals will be important moving forward. Besides, there is a trend towards online purchases of nutraceuticals as well as greater focus on security and adulteration [114].

#### 5. Conclusions

The promising outcomes of the above studies along with the well-documented role of food bioactive ingredients in supporting the immune system and the supplementation of consumers' diets with vitamins, tannins, polyphenols, flavonoids, bioactive lipids, and herbs are driving current market growth trends in the food and nutraceutical sector. Moreover, these trends will most likely continue to drive the market in the post-lockdown era [4]. However, as of 10 November 2020, at the time of this writing, there has not been adequate published evidence correlating the consumption of food bioactives with direct prevention or recovery from COVID-19 disease. Some evidence (including randomized population and clinical trials) does exist regarding the role of vitamin D against COVID-19 disease. In addition, numerous theoretical studies have suggested polyphenolic compounds (mostly flavonoids) as potential inhibitors of SARS-CoV-2 transmission. The potential for using  $\beta$ -glucan to address COVID-19 has also been suggested by taking into account the variability in immune response arising from heterogeneity in polysaccharide branch and chain lengths of different sources. Researchers have recommended drinking plenty of water, along with consuming foods rich in minerals such as magnesium and zinc and vitamins C, D, and E, in addition to a better life style that can boost immunity to help fight infection [115]. Because these supplements can support or improve the function of the immune system, the market prospects for nutraceuticals and functional foods within the post-lockdown period remain high, as a result of the increased interest of health-conscious consumers. Businesses will thus be seeking to fill the demand for new knowledge surrounding consumer preferences, needs, and attitudes toward nutraceuticals and functional foods in order to address the challenges and opportunities created by COVID-19 disease. The aforementioned changes that are coming will also tend to inform or even accelerate innovation across the food industry ecosystem including advances in ICT and manufacturing.

Author Contributions: Conceptualization: C.M.G., T.M.S.A., N.J.R., and S.A.I.; writing—original draft: C.M.G., T.M.S.A., N.J.R., and S.A.I.; writing—review and editing: C.M.G., T.M.S.A., M.R., N.J.R., and S.A.I. All authors have read and agreed to the published version of the manuscript.

**Funding:** Author (S.A.I.) would also like to acknowledge the support of the Agricultural Research Station at North Carolina Agricultural and Technical State University (Greensboro, NC 27411, USA). This research was funded, in part, by Grants (project Number NC.X337-5-21-170-1 and NC.X341-5-21-170-1) from the National Institute of Food and Agriculture (NIFA). Its contents are solely the responsibility of the authors and do not necessarily represent the official views of NIFA.

Conflicts of Interest: The authors declare that they have no competing or conflict of interests.

# References

- 1. Watson, R.; Preedy, V. (Eds.) *Bioactive Foods in Promoting Health*, 2nd ed.; Fruits, Vegetables, and Herbs; Academic Press: Cambridge, MA, USA, 2016.
- 2. González, S. Dietary Bioactive Compounds and Human Health and Disease. *Nutrients* **2020**, *12*, 348. [CrossRef] [PubMed]
- 3. Williamson, E.M.; Liu, X.; Izzo, A.A. Trends in use, pharmacology, and clinical applications of emerging herbal nutraceuticals. *Br. J. Pharmacol.* **2020**, *177*, 1227–1240. [CrossRef] [PubMed]
- 4. Santini, A.; Cammarata, S.M.; Capone, G.; Ianaro, A.; Tenore, G.C.; Pani, L.; Novellino, E. Nutraceuticals: Opening the debate for a regulatory framework. *Br. J. Clin. Pharmacol.* **2018**, *84*, 659–672. [CrossRef]
- 5. Galanakis, C.M. The Food Systems in the Era of the Coronavirus (COVID-19) Pandemic Crisis. *Foods* **2020**, *9*, 523. [CrossRef] [PubMed]
- 6. Rizou, M.; Galanakis, I.M.; Aldawoud, T.M.; Galanakis, C.M. Safety of foods, food supply chain and environment within the COVID-19 pandemic. *Trends Food Sci. Technol.* **2020**, *102*, 293–299. [CrossRef]
- Schyltz, H. Panic Buying of İmmune Products Causes Shortfalls, Greater Supply Gaps Loom in Months Ahead. 2020. Available online: https://www.nutraingredients.com/Article/2020/03/16/Panic-buying-of-immuneproducts-causes-shortfalls-greater-supply-gaps-loom-in-months-ahead (accessed on 16 March 2020).
- 8. Masterson, D. Industry Offers Insight into Vitamin C Supply Chain. 2020. Available online: https://www. nutraingredients.com/Article/2020/04/02/Industry-offers-insight-into-vitamin-C-supply-chain (accessed on 2 April 2020).
- 9. Xian, Y.; Zhang, J.; Bian, Z.; Zhou, H.; Zhang, Z.; Lin, Z.-X.; Xu, H.-X. Bioactive natural compounds against human coronaviruses: A review and perspective. *Acta Pharm. Sin. B* **2020**, *10*, 1163–1174. [CrossRef]
- Hensel, A.; Bauer, R.; Heinrich, M.; Spiegler, V.; Kayser, O.; Hempel, G.; Kraft, K. Challenges at the Time of COVID-19: Opportunities and Innovations in Antivirals from Nature. *Planta Med.* 2020, *86*, 659–664. [CrossRef]
- 11. Antonio, A.D.S.; Wiedemann, L.S.M.; Junior, V.F.V. Natural products' role against COVID-19. *RSC Adv.* 2020, 10, 23379–23393. [CrossRef]
- 12. The Economist. Scientific Research on the Coronavirus is Being Released in a Torrent. 2020. Available online: https://www.economist.com/science-and-technology/2020/05/07/scientific-research-on-the-coronavirus-is-being-released-in-a-torrent (accessed on 7 May 2020).
- 13. Rowan, N.J. Pulsed light as an emerging technology to cause disruption for food and adjacent industries–Quo vadis? *Trends Food Sci. Technol.* **2019**, *88*, 316–332. [CrossRef]
- 14. Butler, M.J.; Barrientos, R.M. The impact of nutrition on COVID-19 susceptibility and long-term consequences. *Brain Behav. Immun.* **2020**, *87*, 53–54. [CrossRef]
- Calder, P.C.; Carr, A.C.; Gombart, A.F.; Eggersdorfer, M. Optimal Nutritional Status for a Well-Functioning Immune System Is an Important Factor to Protect against Viral Infections. *Nutrients* 2020, *12*, 1181. [CrossRef] [PubMed]
- Castro, E.D.M.; Calder, P.C.; Roche, H.M. β-1,3/1,6-glucans and Immunity: State of the Art and Future Directions. *Mol. Nutr. Food Res. J.* 2020, e1901071. [CrossRef] [PubMed]
- 17. Naik, S.R.; Thakare, V.N.; Joshi, F.P. Functional Foods and Herbs as Potential Immunoadjuvants and Medicines in Maintaining Healthy Immune System: A Commentary. *J. Complement. Integr. Med.* **2010**, *7*, 1. [CrossRef]
- Gibson, A.; Edgar, J.D.; E Neville, C.; Gilchrist, S.E.C.M.; McKinley, M.C.; Patterson, C.C.; Young, I.S.; Woodside, J.V. Effect of fruit and vegetable consumption on immune function in older people: A randomized controlled trial. *Am. J. Clin. Nutr.* 2012, *96*, 1429–1436. [CrossRef] [PubMed]
- 19. Carr, A.C.; Maggini, S. Vitamin C and Immune Function. *Nutrients* 2017, 9, 1211. [CrossRef]
- 20. Hemilä, H. Vitamin C intake and susceptibility to pneumonia. *Pediatr. Infect. Dis. J.* **1997**, *16*, 836–837. [CrossRef]
- 21. Gómez, E.; Quidel, S.; Soto, G.B.; Ortigoza, Á. Does vitamin C prevent the common cold? *Medwave* 2018, 18, e7236. [CrossRef]
- 22. Huang, Z.; Liu, Y.; Qi, G.; Brand, D.D.; Zheng, S.G. Role of Vitamin A in the Immune System. *J. Clin. Med.* 2018, 7, 258. [CrossRef]
- 23. Sanchez, A.; Vazquez, A. Bioactive Peptides: A Review. Food Qual. Saf. 2017, 1, 29–46. [CrossRef]

- 24. Aguilar-Toalá, J.; Santiago-López, L.; Peres, C.; Garcia, H.; Vallejo-Cordoba, B.; González-Córdova, A.; Hernández-Mendoza, A. Assessment of multifunctional activity of bioactive peptides derived from fermented milk by specific Lactobacillus plantarum strains. *J. Dairy Sci.* **2017**, *100*, 65–75. [CrossRef]
- 25. Sisto, A.; Luongo, D.; Treppiccione, L.; De Bellis, P.; Di Venere, D.; Lavermicocca, P.; Rossi, B.M. Effect of Lactobacillus paracasei Culture Filtrates and Artichoke Polyphenols on Cytokine Production by Dendritic Cells. *Nutrients* **2016**, *8*, 635. [CrossRef] [PubMed]
- Rodríguez-Valentín, M.; López, S.; Rivera, M.; Ríos-Olivares, E.; Cubano, L.; Boukli, N.M. Naturally derived anti-HIV polysaccharide peptide (PSP) triggers a toll-like receptor 4-dependent antiviral immune response. *J. Immunol. Res.* 2018, 2018, 1–15. [CrossRef] [PubMed]
- 27. Laskin, D.L.; Sunil, V.R.; Gardner, C.R.; Laskin, J.D. Macrophages and Tissue Injury: Agents of Defense or Destruction? *Annu. Rev. Pharmacol. Toxicol.* **2011**, *51*, 267–288. [CrossRef] [PubMed]
- 28. Galanakis, C.M. Recovery of high added-value components from food wastes: Conventional, emerging technologies and commercialized applications. *Trends Food Sci. Technol.* **2012**, *26*, 68–87. [CrossRef]
- 29. Galanakis, C.M. Emerging technologies for the production of nutraceuticals from agricultural by-products: A viewpoint of opportunities and challenges. *Food Bioprod. Process.* **2013**, *91*, 575–579. [CrossRef]
- 30. Galanakis, C.M. Separation of functional macromolecules and micromolecules: From ultrafiltration to the border of nanofiltration. *Trends Food Sci. Technol.* **2015**, *42*, 44–63. [CrossRef]
- 31. Kovacevic, D.B.; Barba, F.J.; Granato, D.; Galanakis, C.M.; Herceg, Z.; Dragovic-Uzelac, V.; Prutnik, P. Pressurized hot water extraction (PHWE) for the green recovery of bioactive compounds from steviol glycosides from Stevis rebaudiana Bertoni Leaves. *Food Chem.* **2018**, 254, 150–157. [CrossRef]
- 32. Galanakis, C.M.; Tsatalas, P.; Galanakis, I.M. Implementation of phenols recovered from olive mill wastewater as UV booster in cosmetics. *Ind. Crop. Prod.* **2018**, *111*, 30–37. [CrossRef]
- 33. Hsu, S. Compounds Derived from Epigallocatechin-3-Gallate (EGCG) as a Novel Approach to the Prevention of Viral Infections. *Inflamm. Allergy Drug Targets* **2015**, *14*, 13–18. [CrossRef]
- Parker, S.; May, B.; Zhang, C.; Zhang, A.L.; Lu, C.; Xue, C.C. A Pharmacological Review of Bioactive Constituents of Paeonia lactiflora Pallas and Paeonia veitchii Lynch. *Phytother. Res.* 2016, 30, 1445–1473. [CrossRef]
- 35. Yang, C.; Zhi, X.; Xu, H.; Lv, M. Advances on Semisynthesis, Total Synthesis, and Structure-Activity Relationships of Honokiol and Magnolol Derivatives. *Mini-Rev. Med. Chem.* **2016**, *16*, 1. [CrossRef] [PubMed]
- Vázquez-Calvo, Á.; De Oya, N.J.; Martín-Acebes, M.A.; Garcia-Moruno, E.; Saiz, J.-C. Antiviral Properties of the Natural Polyphenols Delphinidin and Epigallocatechin Gallate against the Flaviviruses West Nile Virus, Zika Virus, and Dengue Virus. *Front. Microbiol.* 2017, *8*, 1314. [CrossRef] [PubMed]
- 37. Das, U.N. Can Bioactive Lipids Inactivate Coronavirus (COVID-19)? *Arch. Med. Res.* 2020, *51*, 282–286. [CrossRef] [PubMed]
- 38. Mccarty, M.F.; DiNicolantonio, J.J. Nutraceuticals have potential for boosting the type 1 interferon response to RNA viruses including influenza and coronavirus. *Prog. Cardiovasc. Dis.* **2020**, *63*, 383–385. [CrossRef]
- 39. Wink, P. Potential of DNA Intercalating Alkaloids and Other Plant Secondary Metabolites against SARS-CoV-2 Causing COVID-19. *Diversity* **2020**, in press. [CrossRef]
- 40. Harmer, D.; Gilbert, M.; Borman, R.; Clark, K.L. Quantitative mRNA expression profiling of ACE 2, a novel homologue of angiotensin converting enzyme. *FEBS Lett.* **2002**, *532*, 107–110. [CrossRef]
- Kuba, K.; Imai, Y.; Ohto-Nakanishi, T.; Penninger, J.M. Trilogy of ACE2: A peptidase in the renin–angiotensin system, a SARS receptor, and a partner for amino acid transporters. *Pharmacol. Ther.* 2010, 128, 119–128. [CrossRef]
- Jiang, F.; Yang, J.; Zhang, Y.; Dong, M.; Wang, S.; Zhang, Q.; Liu, F.F.; Zhang, K.; Zhang, C. Angiotensin-converting enzyme 2 and angiotensin 1–7: novel therapeutic targets. *Nat. Rev. Cardiol.* 2014, 11, 413–426. [CrossRef]
- Zhou, P.; Yang, X.-L.; Wang, X.-G.; Hu, B.; Zhang, L.; Zhang, W.; Si, H.-R.; Zhu, Y.; Li, B.; Huang, C.-L.; et al. A pneumonia outbreak associated with a new coronavirus of probable bat origin. *Nature* 2020, 579, 270–273. [CrossRef]
- 44. Hoffmann, M.; Kleine-Weber, H.; Schroeder, S.; Krüger, N.; Herrler, T.; Erichsen, S.; Schiergens, T.S.; Herrler, G.; Wu, N.-H.; Nitsche, A.; et al. SARS-CoV-2 Cell Entry Depends on ACE2 and TMPRSS2 and Is Blocked by a Clinically Proven Protease Inhibitor. *Cell* **2020**, *181*, 271–280.e8. [CrossRef]

- Bosch, B.J.; Van Der Zee, R.; De Haan, C.A.M.; Rottier, P.J.M. The Coronavirus Spike Protein Is a Class I Virus Fusion Protein: Structural and Functional Characterization of the Fusion Core Complex. *J. Virol.* 2003, 77, 8801–8811. [CrossRef] [PubMed]
- 46. Sivaraman, D.; Pradeep, P.S. Scope of phytotherapeutics in targeting ACE2 mediated Host-Viral Interface of SARS-CoV2 that causes COVID-19. *ChemRxiv Preprint* **2020**. [CrossRef]
- Das, U.N. Response to: Bioactive Lipids and Coronavirus (COVID-19)-further Discussion. *Arch. Med. Res.* 2020, 51, 445–449. [CrossRef] [PubMed]
- 48. Wahedi, H.M.; Ahmad, S.; Abbasi, S.W. Stilbene-based natural compounds as promising drug candidates against COVID-19. *J. Biomol. Struct. Dyn.* **2020**, 1–10. [CrossRef] [PubMed]
- 49. Sheybani, Z.; Dokoohaki, M.H.; Negahdaripour, M.; Dehdashti, M.; Zolghadr, H.; Moghadami, M.; et al. The Role of Folic Acid in the Management of Respiratory Disease Caused by COVID-19. *ChemRxiv Preprint* **2020**. [CrossRef]
- Bhullar, K.S.; Drews, S.J.; Wu, J. Translating bioactive peptides for COVID-19 therapy. *Eur. J. Pharmacol.* 2020, 173661. [CrossRef]
- Adem, S.; Eyupoglu, V.; Sarfraz, I.; Rasul, A.; Ali, M. Identification of Potent COVID-19 Main Protease (Mpro) Inhibitors from Natural Polyphenols: An in Silico Strategy Unveils a Hope against CORONA. *Preprints* 2020. [CrossRef]
- 52. Bhatia, S.; Giri, S.; Lal, A.F.; Singh, S. Battle Against Coronavirus: Repurposing Old Friends (Food Borne Polyphenols) for New Enemy (COVID-19). *ChemRxiv Preprint* **2020**. [CrossRef]
- 53. Khalifa, I.; Zhu, W.; Nafie, M.S.; Dutta, K.; Li, C. Anti-COVID-19 Effects of Ten Structurally Different Hydrolysable Tannins through Binding with the Catalytic-Closed Sites of COVID-19 Main Protease: An In-Silico Approach. *Preprints* **2020**, 2020030277. [CrossRef]
- 54. Maiti, S.; Banerjee, A. Epigallocatechin gallate and theaflavin gallate interaction in SARS-CoV -2 spike-protein central channel with reference to the hydroxychloroquine interaction: Bioinformatics and molecular docking study. *Drug Dev. Res.* **2020**. [CrossRef]
- Pendyala, B.; Patras, A. In silico Screening of Food Bioactive Compounds to Predict Potential Inhibitors of COVID-19 Main protease (Mpro) and RNA-dependent RNA polymerase (RdRp). *ChemRxiv Preprint* 2020. [CrossRef]
- 56. Benarous, K.; Yousfi, M. Hispidin and Lepidine E: two Natural Compounds and Folic acid as Potential Inhibitors of 2019-novel coronavirus Main Protease (2019-nCoVMpro), molecular docking and SAR study. *Curr. Comput. Drug Des.* **2020**, *16*, 1–14. [CrossRef]
- 57. Utomo, R.Y.; Ikawati, M.; Meiyanto, E. Revealing the Potency of Citrus and Galangal Constituents to Halt SARS-CoV-2 Infection. *Preprints* **2020**, 2020030214. [CrossRef]
- Tallei, T.E.; Tumilaar, S.G.; Niode, N.J.; Fatimawali, F.; Kepel, B.J.; Idroes, R.; Effendi, Y. Potential of Plant Bioactive Compounds as SARS-CoV-2 Main Protease (Mpro) and Spike (S) Glycoprotein Inhibitors: A Molecular Docking Study. *Preprints* 2020. [CrossRef]
- 59. DeRosa, G.; Maffioli, P.; D'Angelo, A.; Di Pierro, F. A role for quercetin in coronavirus disease 2019 (COVID-19). *Phytother. Res.* **2020**. [CrossRef]
- 60. Biancatelli, R.M.L.C.; Berrill, M.; Catravas, J.D.; Marik, P.E. Quercetin and Vitamin C: An Experimental, Synergistic Therapy for the Prevention and Treatment of SARS-CoV-2 Related Disease (COVID-19). *Front. Immunol.* **2020**, *11*, 1451. [CrossRef]
- 61. Smith, M.; Smith, J.C. Repurposing therapeutics for COVID-19: Supercomputer-based docking to the SARS-CoV-2 viral spike protein and viral spike protein-human ACE2 interface. *Chem-Rxiv Preprint* **2020**. [CrossRef]
- 62. Yang, Y.; Islam, S.; Wang, J.; Li, Y.; Chen, X. Traditional Chinese Medicine in the Treatment of Patients Infected with 2019-New Coronavirus (SARS-CoV-2): A Review and Perspective. *Int. J. Biol. Sci.* **2020**, *16*, 1708–1717. [CrossRef]
- Ross, J.A.; Kasum, C.M. Dietary flavonoids: Bioavailability, metabolic effects, and safety. *Annu. Rev. Nutr.* 2002, 22, 19–34. [CrossRef]
- 64. Theoharides, T.C. COVID-19, pulmonary mast cells, cytokine storms, and beneficial actions of luteolin. *BioFactors* **2020**, *46*, 306–308. [CrossRef]

- 65. Murphy, E.J.; Masterson, C.; Rezoagli, E.; O'Toole, D.; Major, I.; Stack, G.D.; Lynch, M.; Laffey, J.G.; Rowan, N.J. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—Implications for coronavirus disease (COVID-19) immunotherapies. *Sci. Total Environ.* **2020**, *732*, 139330. [CrossRef] [PubMed]
- Zhang, W.; Zhao, Y.; Zhang, F.; Wang, Q.; Li, T.; Liu, Z.; Wang, J.; Qin, Y.; Zhang, X.; Yan, X.; et al. The use of anti-inflammatory drugs in the treatment of people with severe coronavirus disease 2019 (COVID-19): The Perspectives of clinical immunologists from China. *Clin. Immunol.* 2020, 214, 108393. [CrossRef] [PubMed]
- 67. Conti, P.; Ronconi, G.; Caraffa, A.; Gallenga, C.E.; Ross, R.; Frydas, I.; Kritas, S.K. Induction of pro-inflammatory cytokines (IL-1 and IL-6) and lung inflammation by Coronavirus-19 (COVI-19 or SARS-CoV-2): anti-inflammatory strategies. *J. Biol. Regul. Homeost. Agents* **2020**, in press.
- 68. Kim, K.; Ehrlich, A.; Perng, V.; Chase, J.A.; Raybould, H.; Li, X.; Atwill, E.R.; Whelan, R.; Sokale, A.; Liu, Y. Algae-derived β-glucan enhanced gut health and immune responses of weaned pigs experimentally infected with a pathogenic E. coli. *Anim. Feed. Sci. Technol.* **2019**, *248*, 114–125. [CrossRef]
- 69. Hetland, G.; Johnson, E.; Bernardshaw, S.V.; Grinde, B. Can medicinal mushrooms have prophylactic or therapeutic effect against COVID-19 and its pneumonic superinfection and complicating inflammation. *Scandanavian J. Immunol.* **2020**. [CrossRef]
- Jayawardena, R.; Sooriyaarachchi, P.; Chourdakis, M.; Jeewandara, C.; Ranasinghe, P. Enhancing immunity in viral infections, with special emphasis on COVID-19: A review. *Diabetes Metab. Syndr. Clin. Res. Rev.* 2020, 14, 367–382. [CrossRef]
- 71. Chowdhury, M.A.; Hossain, N.; Kashem, M.A.; Shalid, M.A.; Alam, A. Immune-response in COVID-19—A review. *J. Infect. Public Health* **2020**, *13*, 1619–1629. [CrossRef]
- 72. Muscogiuri, G.; Barrea, L.; Savastano, S.; Colao, A. Nutritional recommendations for CoVID-19 quarantine. *Eur. J. Clin. Nutr.* **2020**, *74*, 850–851. [CrossRef]
- 73. Rezoagli, E.; Fumagalli, R.; Bellani, G. Definition and epidemiology of acute respiratory distress syndrome. *Ann. Transl. Med.* **2017**, *5*, 282. [CrossRef]
- 74. Laffey, J.G.; Matthay, M.A. Fifty Years of Research in ARDS. Cell Based Therapy for ARDS: Biology and Potential Therapeutic Value. *Am. J. Respir. Crit. Care Med.* **2017**, *196*, 266–273. [CrossRef]
- 75. Chousterman, B.G.; Swirski, F.K.; Weber, G.F. Cytokine storm and sepsis disease pathogenesis. *Semin. Immunopathol.* **2017**, *39*, 517–528. [CrossRef] [PubMed]
- Masterson, C.H.; Murphy, E.; Major, I.; Gonzalez, H.; O'Toole, D.; McCarthy, S.; Laffey, J.G.; Rowan, N. Purified beta-glucan from the Lentinus edodes mushroom attenuates antibiotic resistant *Klebsiella pneumoniae*-induced pulmonary sepsis. *Am. J. Respir. Crit. Care Med.* 2019, 199, A122.
- 77. Masterson, C.; Murphy, E.J.; Gonzalez, H.; Major, I.; McCarthy, S.D.; O'Toole, D.; Laffey, J.G.; Rowan, N.J. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumoniae-induced pulmonary sepsis. *Lett. Appl. Microbiol.* **2020**, *7*, 405–412. [CrossRef]
- 78. McCarty, S.D.; Horgan, E.; Ali, A.; Masterson, C.; Laffey, J.G.; MacLoughlin, R.; O'Toole, D. Nebulized mesenchemyal stem cells derived conditioned medium retains antibacterial properties against clinical pathogen isolates. *J. Aerosol Med. Pulm. Drug Deliv.* **2019**, *33*, 140–152. [CrossRef] [PubMed]
- 79. Bedirli, A.; Kerem, M.; Pasaoglu, H.; Akyurek, N.; Tezcaner, T.; Elbeg, S.; Memis, L.; Sakrak, O. Beta-glucan attenuates inflammatory cytokine release and prevents acute lung injury in an experimental model of sepsis. *Shock* **2007**, *27*, 397–401. [CrossRef]
- Carballo, C.; Pinto, P.I.; Mateus, A.P.; Berbel, C.; Guerreiro, C.C.; Martinez-Blanch, J.F.; Codoñer, F.M.; Mantecon, L.; Power, D.M.; Manchado, M. Yeast β-glucans and microalgal extracts modulate the immune response and gut microbiome in Senegalese sole (Solea senegalensis). *Fish Shellfish Immunol.* 2019, *92*, 31–39. [CrossRef]
- Luo, H.; Lv, X.D.; Wang, G.E.; Li, Y.F.; Kurihara, H.; He, R.R. Anti-inflammatory effects of anthocyanins-rich extract from bilberry on crotron-oil-induced ear edema and Propionobacteracterium acnes plus LPS-induced liver damage in mice. *Int. J. Food Sci. Nutr.* 2014, *65*, 594–601. [CrossRef]
- 82. Xu, W.; Zhou, Q.; Yao, Y.; Li, X.; Zhang, J.-I.; Su, G.H.; Deng, A.P. Inhibitory effect of Gardanblue bluberry anthocyanin extracts on liposaccharide-stimulated inflammation response in RAW 264. 7 cell. *J. Zheinjiang Uni-Sci. B (Biomed. Biotech.)* **2016**, *17*, 425–436. [CrossRef]
- Sinha, S.; Cheng, K.; Aldape, K.; Schiff, E.; Ruppin, E. Systematic Cell Line-Based Identification of Drugs Modifying ACE2 Expression. *Preprints* 2020, *97*, 84. [CrossRef]
- 84. Nonnecke, B.J.; McGill, J.L.; Ridpath, J.F.; Sacco, R.E.; Lippolis, J.D.; Reinhardt, T.A. Acute phase response elicited by experimental bovine diarrhea virus (BVDV) infection is associated with decreased vitamin D and E status of vitamin-replete preruminant calves. *J. Dairy Sci.* **2014**, *97*, 5566–5579. [CrossRef]
- 85. Bikle, D. Nonclassic Actions of Vitamin D. J. Clin. Endocrinol. Metab. 2009, 94, 26–34. [CrossRef] [PubMed]
- 86. Jakovac, H. COVID-19 and vitamin D—Is there a link and an opportunity for intervention? *Am. J. Physiol. Metab.* **2020**, *318*, E589. [CrossRef] [PubMed]
- 87. Honardoost, M.; Ghavideldarestani, M.; Khamseh, M.E. Role of vitamin D in pathogenesis and severity of COVID-19 infection. *Arch. Physiol. Biochem.* **2020**, 1–7. [CrossRef] [PubMed]
- 88. Ilie, P.C.; Stefanescu, S.; Smith, L. The role of vitamin D in the prevention of coronavirus disease 2019 infection and mortality. *Aging Clin. Exp. Res.* **2020**, *32*, 1195–1198. [CrossRef] [PubMed]
- 89. Daneshkhah, A.; Agrawal, V.; Eshein, A.; Subramanian, H.; Roy, H.K.; Backman, V. The Possible Role of Vitamin D in Suppressing Cytokine Storm and Associated Mortality in COVID-19. *Patients* **2020**. [CrossRef]
- 90. Lau, F.H.; Majumder, R.; Torabi, R.; Saeg, F.; Hoffman, R.; Cirillo, J.D.; Greiffenstein, P. Vitamin D Insufficiency is Prevalent in Severe COVID-19. **2020**. [CrossRef]
- 91. Nikhat, S.; Fazil, M. Overview of Covid-19, its prevention and management in the light of Unani medicine. *Sci. Total Environ.* **2020**, *728*, 138859. [CrossRef]
- Liang, Y.; Zhang, Q.; Zhang, L.; Wang, R.; Xu, X.; Hu, X. Astragalus Membranaceus Treatment Protects Raw264.7 Cells from Influenza Virus by Regulating G1 Phase and the TLR3-Mediated Signaling Pathway. *Evid. Based Complement. Altern. Med.* 2019, 2019, 1–10. [CrossRef]
- Luo, H.; Tang, Q.-L.; Shang, Y.-X.; Liang, S.-B.; Yang, M.; Robinson, N.; Liu, J. Can Chinese Medicine Be Used for Prevention of Corona Virus Disease 2019 (COVID-19)? A Review of Historical Classics, Research Evidence and Current Prevention Programs. *Chin. J. Integr. Med.* 2020, *26*, 243–250. [CrossRef]
- 94. Im, K.; Kim, J.; Min, H. Ginseng, the natural effectual antiviral: Protective effects of Korean Red Ginseng against viral infection. *J. Ginseng Res.* **2016**, *40*, 309–314. [CrossRef]
- 95. Luo, E.; Zhang, D.; Luo, H.; Liu, B.; Zhao, K.; Zhao, Y.-H.; Bian, Y.; Wang, Y. Treatment efficacy analysis of traditional Chinese medicine for novel coronavirus pneumonia (COVID-19): An empirical study from Wuhan, Hubei Province, China. *Chin. Med.* **2020**, *15*, 1–13. [CrossRef] [PubMed]
- 96. Ang, L.; Lee, H.W.; Choi, J.Y.; Zhang, J.; Lee, M.S. Herbal medicine and pattern identification for treating COVID-19: a rapid review of guidelines. *Integr. Med. Res.* **2020**, *9*, 100407. [CrossRef] [PubMed]
- 97. Silveira, D.; Prieto-Garcia, J.M.; Boylan, F.; Estrada, O.; Fonseca-Bazzo, Y.M.; Jamal, C.M.; Magalhães, P.O.; Pereira, E.O.; Tomczyk, M.; Heinrich, M. COVID-19: Is There Evidence for the Use of Herbal Medicines as Adjuvant Symptomatic Therapy? *Front. Pharmacol.* 2020, *11*, 581840. [CrossRef] [PubMed]
- 98. Aldaco, R.; Hoehn, D.; Laso, J.; Margallo, M.; Ruiz-Salmón, J.; Cristobal, J.; Kahhat, R.; Villanueva-Rey, P.; Bala, A.; Batlle-Bayer, L.; et al. Food waste management during the COVID-19 outbreak: A holistic climate, economic and nutritional approach. *Sci. Total Environ.* 2020, 742, 140524. [CrossRef]
- Rowan, N.J.; Galanakis, C.M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? *Sci. Total. Environ.* 2020, 748, 141362. [CrossRef]
- 100. Huizar, M.I.; Arena, R.; Laddu, D.R. The global food syndemic: The impact of food insecurity, Malnutrition and obesity on the healthspan amid the COVID-19 pandemic. *Prog. Cardiovasc. Dis.* **2020**. [CrossRef]
- Askew, K. COVID-19 Crisis Accelerates Mainstreaming of Immune Boosting Beta-Glucans 2020-20 April 2020. Available online: https://www.foodnavigator.com/Article/2020/04/20/COVID-19-crisis-acceleratesmainstreaming-of-immune-boosting-beta-glucans (accessed on 20 April 2020).
- 102. Askew, K. Organic Food's Coronavirus Boost: 'Health Crises Have a Long-Term Impact on Consumer Demand'-20 April 2020. Available online: https://www.foodnavigator.com/Article/2020/05/06/Organic-food-gets-coronavirus-boost (accessed on 20 April 2020).
- FMCC Gurus. The Impact of COVID-19 in 2020 & Beyond. Available online: https://fmcggurus.com/covid-19/ (accessed on 11 May 2020).
- 104. Koe, T. Chocolate Balls for Immune Health: Singapor Firm's New Product for Kids amid COVID-19. 2020. Available online: https://www.nutraingredients-asia.com/Article/2020/04/30/Chocolate-balls-for-immunehealth-Singapore-firm-s-new-product-for-kids-amid-COVID-19 (accessed on 11 May 2020).

- 105. Galanakis, C.M. Phenols recovered from olive mill wastewater as additives in meat products. *Trends Food Sci. Technol.* **2018**, *79*, 98–105. [CrossRef]
- 106. Roselló-Soto, E.; Galanakis, C.M.; Brnčić, M.; Orlien, V.; Trujillo, F.J.; Mawson, R.; Knoerzer, K.; Tiwari, B.K.; Barba, F.J. Clean recovery of antioxidant compounds from plant foods, by-products and algae assisted by ultrasounds processing. Modeling approaches to optimize processing conditions. *Trends Food Sci. Technol.* 2015, 42, 134–149. [CrossRef]
- 107. Daniells, S. NPA's Fabricant: 'It's a Global Pandemic. We Need to Speak a Global Language When It Comes to the Science around Vitamin D and Zinc.' Available online: https://www.nutraingredients.com/Article/2020/05/06/NPA-s-Fabricant-It-s-a-global-pandemic.-We-need-to-speak-a-global-language-when-it-comes-to-the-science-around-vitamin-D-and-zinc (accessed on 11 May 2020).
- 108. Koe, T. Immunity İnnovation: Gencor to Study PEA's İmmune Boosting Effects via Human Trial. 2020. Available online: https://www.nutraingredients-asia.com/Article/2020/03/19/Immunity-innovation-Gencorto-study-PEA-s-immune-boosting-effects-via-human-trial# (accessed on 19 March 2020).
- 109. Koe, T. COVID-19 and Nutra Research: Where Does the İndustry Need to Focus İts SCİENTİFİC Endeavours? 2020. Available online: https://www.nutraingredients.com/Article/2020/04/01/COVID-19-and-nutra-research-Where-does-the-industry-need-to-focus-its-scientific-endeavours (accessed on 1 April 2020).
- Galanakis, C.M.; Cvejic, J.; Verardo, V.; Segura-Carretero, A. Food use for social innovation by optimizing food waste recovery strategies. In *Innovation Strategies for the Food Industry: Tools for Implementation*; Galanakis, C.M., Ed.; Elsevier Inc.: Waltham, MA, USA, 2016.
- 111. GOED. Industry Advisory: Omega-3 Immunity Claims. Available online: https://goedomega3.com/storage/ app/media/scientific-reports/GOED%20Advisory%20-%20Immunity%20Claim%20-%202020%2003.pdf (accessed on 24 March 2020).
- 112. Koe, T. Unproven COVID-19 Health Claims: China's Crackdown on Ads for Oral Sprays, Probiotics and Anti-Hangover Tea 2020. Available online: https://www.nutraingredients.com/Article/2020/04/07/Unproven-COVID-19-health-claims-China-s-crackdown-on-ads-for-oral-sprays-probiotics-and-anti-hangover-tea (accessed on 11 May 2020).
- 113. Rowan, N.J.; Laffey, J.G. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic–Case study from the Republic of Ireland. *Sci. Total Environ.* **2020**, *725*, 138532. [CrossRef]
- 114. Bullimore, S. COVID-19: Nutraceutical Platers Respond. NBR Nutraceutical Business Reviews. Available online: https://www.nutraceuticalbusinessreview.com/news/article\_page/COVID-19\_Nutraceutical\_players\_ respond/163975 (accessed on 16 May 2020).
- 115. Arshad, M.S.; Khan, U.; Sadiq, A.; Khalid, W.; Hussain, M.; Yasmeen, A.; Asghar, Z.; Rehana, H. Coronavirus disease (COVID-19) and immunity booster green foods: A mini review. *Food Sci. Nutr.* 2020, *8*, 3971–3976. [CrossRef]

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).





## Article Use of Zebrafish Embryo Assay to Evaluate Toxicity and Safety of Bioreactor-Grown Exopolysaccharides and Endopolysaccharides from European *Ganoderma applanatum* Mycelium for Future Aquaculture Applications

Wan Abd Al Qadr Imad Wan-Mohtar <sup>1,2,3</sup>, Zul Ilham <sup>2,4</sup>, Adi Ainurzaman Jamaludin <sup>4</sup> and Neil Rowan <sup>3,5,\*</sup>

- <sup>1</sup> Functional Omics and Bioprocess Development Laboratory, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, Kuala Lumpur 50603, Malaysia; qadyr@um.edu.my
- <sup>2</sup> Bioresources and Bioprocessing Research Group, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, Kuala Lumpur 50603, Malaysia; ilham@um.edu.my
- <sup>3</sup> Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, N37 WO89 Athlone, Westmeath, Ireland
- <sup>4</sup> Environmental Science and Management Program, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, Kuala Lumpur 50603, Malaysia; adiainurzaman@um.edu.my
- <sup>5</sup> Empower Eco Innovation Hub, Lough Boora, Co., R35 DA50 Tullamore, Offaly, Ireland
- Correspondence: nrowan@ait.ie; Tel.: +353-(0)90-64671838



Citation: Wan-Mohtar, W.A.A.Q.I.; Ilham, Z.; Jamaludin, A.A.; Rowan, N. Use of Zebrafish Embryo Assay to Evaluate Toxicity and Safety of Bioreactor-Grown Exopolysaccharides and Endopolysaccharides from European *Ganoderma applanatum* Mycelium for Future Aquaculture Applications. *Int. J. Mol. Sci.* 2021, 22, 1675. https://doi.org/10.3390/ ijms22041675

Academic Editor: Toshio Ohshima Received: 21 January 2021 Accepted: 4 February 2021 Published: 7 February 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Abstract: Natural mycelial exopolysaccharide (EPS) and endopolysaccharide (ENS) extracted from bioreactor-cultivated European *Ganoderma applanatum* mushrooms are of potential high commercial value for both food and adjacent biopharmaceutical industries. In order to evaluate their potential toxicity for aquaculture application, both EPS (0.01–10 mg/mL) and ENS (0.01–10 mg/mL) extracts were tested for Zebrafish Embryo Toxicity (ZFET); early development effects on Zebrafish Embryos (ZE) were also analyzed between 24 and 120 h post-fertilization (HPF). Both EPS and ENS are considered non-toxic with LC<sub>50</sub> of 1.41 mg/mL and 0.87 mg/mL respectively. Both EPS and ENS did not delay hatching and teratogenic defect towards ZE with <1.0 mg/mL, respectively. No significant changes in the ZE heart rate were detected following treatment with the two compounds tested (EPS: 0.01–10 mg/mL: 176.44  $\pm$  0.77 beats/min and ENS: 0.01–10 mg/mL: 148.44  $\pm$  17.75 beats/min) compared to normal ZE (120–180 beats/min). These initial findings support future pre-clinical trials in adult fish models with view to safely using EPS and ENS as potential feed supplements for supplements for development of the aquaculture industry.

Keywords: Ganoderma applanatum; exopolysaccharide; endopolysaccharide; zebrafish; toxicology

#### 1. Introduction

Over the past two decades, medicinal mushroom exopolysaccharides (EPS) and endopolysaccharide (ENS) have been increasingly exploited for human benefits including clinical trials [1], neurological protection [2], potent in vitro biological activities [3], chicken patty [4], anti-oral cancer [5], and breast cancer models [6]; however, there is still limited development as yet on their commercial use by the aquaculture industry. Development and exploitation of medicinal mushroom products has significant potential to improve farmed fish health, where appropriate innovation in fish feed and formulation potentially offers safe, affordable and future intensive sustainability needs [7–9]. Recent evidence suggests the use of bioreactor-grown medicinal mushroom *Ganoderma* sp. extracts in aquaculturelike areas including fish-feed [10], fish toxicity [11], and have passed potential clinical application [12]. These potential applications are largely attributed to its bioactive EPS-ENS components with immuno-stimulatory properties, which can be scaled efficiently and repeatedly for commercial use by deploying a specifically-tailored mushroom bioreactor system [13,14].

2 of 15

Currently, there is an innovation gap in terms of specific use and development of fungal biotechnology for meeting emerging needs in the aquaculture industry globally. However, there are some examples of specific mushroom usage including Shiitake fruiting body for fingerlings of Carps [15] and Rainbow trout feed [16]; thus, use of Ganodermaderived EPS-ENS that comprise beta-glucans warrants further investigation and testing. Mushroom scientists have previously reported that *G. applanatum* EPS-ENS demonstrated promise as novel disease mitigation therapeutic [17–19], suggesting potential for use as disease prevention counter-measures for adjacent aquaculture industry in the form of prebiotics [20].

The creative idea of using extracts from naturally-occurring algae to address key challenges and opportunities in fish health has been exemplified by Marino et al., [21] who studied use of algal extracts as potential alternatives to that of antibiotics for farming sea bass and gilthead sea bream. Marino et al., [21] highlighted increased interest in the use of natural-alternatives materials to that of frontline antibiotics and biocides, particularly where there is a trend towards producing farmed high-value fish with organic status. Cascio et al., [22] reported on the potential "game-changer" use of a microalgae (Spirulina) targeting beneficial effects in the gastrointestinal tract of Zebrafish; this has implications for potentially stimulating the breeding of animals. Such studies show the importance of Zebrafish (Danio rerio) assay due to the ability of Zebrafish stomach system to recognize small extracts such as mRNA and immune-priming genes. The use of beta-glucans-derived from medicinal mushrooms, yeast, algae, seaweed and so forth is of increasing interest as these bio-based therapeutics have shown potential for immune-priming of fish for disease mitigation and for added food security. However, the vast majority of studies to date have focused on extraction of bioactives from raw materials, but not from bioreactor grown systems that are important for larger pilot scale and commercial deployment. It is therefore necessary to test and verify the toxicity status of EPS-ENS polysaccharide extract prior to pilot and commercial use; such as, the use of a zebrafish larvae model to evaluate these bioactive substances as part of a fish feed formulation study [23,24]. For example, use of antibacterial polysaccharides from Undaria pinnatifida (macroalgae) was reported by Rizzo et al., [25] for the treatment of prominent aquaculture disease Vibrio harveyi causes death of marine fish [26] due to vasculitis, gastro-enteritis and eye lesions.

Our study used the ZFET strategy as it was previously reported to be a fast, affordable and sensitive approach [11]; it also offers greater flexibility in terms of using multiple sample insertions to evaluate toxicological effects due to early stages of embryonic development [11] and for continuity in terms of relevance to the gastrointestinal anatomy in small intestine mammals [22]. This study reports on the use of ZFET assay for testing European *G. applanatum* extracts using seven different EPS-ENS concentrations that specifically addresses LC<sub>50</sub>, embryonic hatching delays, teratogenic defect, and heart rate response with clear microscopic images.

#### 2. Results

#### 2.1. Effects of EPS and ENS on the Survival Rate of Zebrafish Embryos

In the ZFET assay, the survival rate of the embryos of the zebrafish in the range of 0 to 120 h old at the cell culture concentration of 0.01 to 10 mg/mL was analyzed. Based on standard, the normal hatching for zebrafish embryos is median of 48 to 72 h of post-fertilization (HPF), thus the survival rate of the embryos (prior hatch) and larva (post hatch) treated with EPS-ENS extract was determined for maximum of five days (120 h). In Figure 1, we observe that 100% of untreated embryos survived for up to 120 HPF. However, among pretreating cells with EPS-ENS doses of 0.01 to 1 mg/mL, the cell survival rate was 88% at 72 HPF. The survival rate dropped from 88% at 72 HPF to 0% for all cells pretreated at the highest EPS-ENS concentrations (5–10 mg/mL). No embryos at >72 HPF managed to survive and developed at EPS concentration >5 mg/mL (Figure 1a). In the meantime, Figure 1b, on low concentrations (0.5–1 mg/mL) of ENS extract, an average drop in survival rate was relatively small (85%), while at higher concentrations (5–10 mg/mL), a lower



survival rate (<30%) was observed at just 4 days of age (96 HPF). No embryo survived after 120 HPF for ENS at concentrations >5 mg/mL. Collectively, it shows that EPS-ENS delay hatching optimum concentrations of less than <1 mg/mL.

**Figure 1.** The performance of *Ganoderma applanatum* (**a**) EPS and (**b**) ENS extract at concentrations of 0.01–10 mg/mL on the survival rate of zebrafish embryos at 0–120 h. No embryos survived for both samples at concentration tested > 5.0 mg/mL after 96 h post fertilization (HPF). Symbols: \* p < 0.05, \*\* p < 0.01 and \*\*\* p < 0.001.

#### 2.2. Performance of EPS-ENS Doses on the Zebrafish Embryos Mortality

Overall, extracts of European *G. applanatum* EPS-ENS have lethal effects that vary with dose and time. In Figure 2, at low concentrations, EPS (<3 mg/mL) and ENS (<1 mg/mL) were both able to prevent 90% of zebrafish embryos from dying. However, at high concentrations, 4 mg/mL EPS and 2 mg/mL ENS did not perform better (with low survival rate) throughout the embryo development stage, showing dead embryos post 72 HPF. Consequently, the LC<sub>50</sub> value of zebrafish embryos exposed to EPS extract tested was



1.41 mg/mL, whereas the LC<sub>50</sub> value for ENS extract tested was much lower than this at 0.87 mg/mL.

**Figure 2.** Effect of *Ganoderma applanatum* (**a**) EPS extract at concentrations of 0.01–10 mg/mL and (**b**) ENS at concentrations of 0.01–10 mg/mL on zebrafish embryos mortality rate after 120 HPF. The LC<sub>50</sub> value for EPS extract was 1.41 mg/mL while LC<sub>50</sub> value for EPS extract was 0.87 mg/mL. Symbols: \* p < 0.05, \*\* p < 0.01 and \*\*\* p < 0.001.

#### 2.3. Performance of EPS and ENS Doses on the Zebrafish Embryos Hatching Capability

Subsequently, different EPS-ENS concentrations of extract may affect the hatchability percentages. Figure 3 shows a trend of decreasing hatching rates with increased concentrations of the extract (>1 mg/mL) once treated into zebrafish embryos. It also depicts the hatching rate during (a) EPS treatments at 0.01–10 mg/mL and (b) ENS treatments at 0.01 to 10 mg/mL during 0 to 120 HPF period. For EPS, there was no visible changes in terms of hatching capability at concentrations less than 1 mg/mL, but the rate dropped to 35% at 48 HPF once exposed to 5 mg/mL. Zebrafish did not hatch or survive when treated with a concentration of 10 mg/mL EPS, indicating that high amounts of EPS causes strong suppression of zebrafish embryo at 24 HPF. On the other hand, less than 81% of the embryos hatched on 48 h treatment with ENS at concentrations >0.5 mg/mL; although, zebrafish larvae treated with ENS concentrations at 5 to 10 mg/mL showed the lower hatching rate (<65%) due to high mortality rate after 72 HPF.



**Figure 3.** Hatching rate of zebrafish embryos at 0 to 120 h of post exposure with *Ganoderma applanatum* EPS and ENS extract at concentrations of 0.01–10 mg/mL. (a) For EPS, low hatching rate (<40%) was observed at concentration 5.0 mg/mL due to high mortality rate. Meanwhile, (b) for ENS, low hatching rate (<30%) was observed at concentrations 10 mg/mL due to high mortality rate. High hatching rate was observed at concentrations >1.0 mg/mL (>80%). Symbols: \* p < 0.05, \*\* p < 0.01 and \*\*\* p < 0.001.

#### 2.4. The Effects of EPS and ENS on the Heart Rate of Zebrafish Embryos

In the development of many model organisms, zebrafish heart represents the core of functional organ [27]. Based on Figure 4, both extracts of EPS and ENS at low concentrations <1 mg/mL, showed no significant difference towards the heart rate of untreated zebrafish embryo at 96 HPF. The heart rate at day 4 (96 HPF) for EPS (Figure 4a) treatment was healthily recorded at 176.44  $\pm$  0.77 beats min<sup>-1</sup> at concentrations <1 mg/mL compared to normal average zebrafish embryonic heart rate (120–180 beats min<sup>-1</sup>). Meanwhile, ENS (Figure 4b) treatment also gave normal heart rate (148.44  $\pm$  17.75 beats min<sup>-1</sup>) at

concentrations <1 mg/mL. Since EPS extract at 5 and 10 mg/mL killed the zebrafish at 96 HPF, the embryonic heart rate at these concentrations were omitted. However, the embryo was barely alive when treated with 5 mg/mL of ENS (90 beats min<sup>-1</sup>) and unrecorded at 10 mg/mL due to death. Both EPS and ENS treatments verified the previous report in which the normal heart rate of zebrafish embryo nearly mimics human condition at 120 to 180 beats min<sup>-1</sup> [28].



**Figure 4.** Effect of *Ganoderma applanatum* EPS and ENS extract at concentrations of 0.01–1.0 mg/mL on heart rate of zebrafish embryos at 96 HPF. For (**a**) EPS, no data at concentrations >5.0 mg/mL due to embryo death. Meanwhile, (**b**) for ENS, no data at concentrations 10.0 mg/mL recorded due to embryo death. \*\*\* p < 0.05 significantly different from the untreated group (zebrafish embryos in medium only).

#### 2.5. Performance of EPS and ENS Doses on the Morphology of Larvae and Zebrafish Embryos Development

The performance of EPS and ENS doses were observed from 0 to 120 HPF period particularly on larvae and zebrafish embryos morphological defects. Figure 5 depicts no visible teratogenic effect of either EPS or ENS at concentrations <1 mg/mL on the embryogenesis stages at 120 h after exposing the embryos to them. Based on these data, it clearly suggested that either EPS or ENS has affected larvae and zebrafish embryos development before and after hatching.



**Figure 5.** Effect of EPS-ENS extracts (0.01–1 mg/mL) of European *Ganoderma applanatum* showing normal zebrafish embryogenesis at different HPF development. The inverted microscope was used to produce the images and edited using paint 3D [11]. There were 4 periods depicted: Blastula (4 HPF), Segmentation (24 HPF), Pharyngula (48 HPF), and Hatching (72 HPF). Y—yolk sac; S—somites; P—pericardium; O—ear bud; M—melanophores; G—gut; F—fin; Ch—chorion; C—chorda; Bc—blood cells; An—anus; A—eye anlage. Scale bar = 0.5 mm.

As both zebrafish embryo and larvae development were unaffected when treated with 1 mg/mL EPS and 1 mg/mL ENS, various abnormalities were observed as the concentration significantly spiked to 5 mg/mL EPS (Figure 6) and 1.31 mg/mL ENS (Figure 7). One of the most prone abnormalities was observed for ENS showing the period of coagulated embryos from 24 HPF (segmentation) to 48 HPF (pharyngula), and loss of yolk sac, resulting in unhatched embryos even after 120 HPF. Meanwhile, ENS-treated zebrafish managed to hatch at 72 HPF, however with tail malformation and broken blood cells after 120 HPF. Most of normal features were also absent such as fin, gut, and melanophores.



**Figure 6.** Images of zebrafish embryo and larvae development at 0 HPF, 24 HPF, 48 HPF, 72 HPF, 96 HPF and 120 HPF after treated with European *Ganoderma applanatum* at high EPS concentration of 5.0 mg/mL. Images were captured using inverted microscope at  $100 \times (0 \text{ and } 24 \text{ HPF})$  and  $40 \times \text{magnification } (48-20 \text{ HPF})$ .



(b) ENS at 1.31 mg/mL



**Figure 7.** Images of zebrafish embryo and larvae development at 0 HPF, 24 HPF, 48 HPF, 72 HPF, 96 HPF and 120 HPF after treated with European *Ganoderma applanatum* at high ENS concentration of 1.31 mg/mL. Images were captured using inverted microscope at  $100 \times (0 \text{ and } 24 \text{ HPF})$  and  $40 \times \text{magnification } (48-20 \text{ HPF})$ .

#### 3. Discussion

In this study, both EPS and ENS extracted from bioreactor-grown European *G. applanatum* were analyzed for acute toxic effects in treated zebrafish embryo. EPS-ENS originates from edible mushroom species and has potential to be used as high-value product in the form of dietary fish feed supplement [29]. Recently, use of EPS-ENS (or beta-glucan) extracts has increased in interest for various applications, such as for preventing biofilms [30] and for fish-feed [10]. Beta-glucans derived from mushrooms and other sources, have also been reported as potential new therapeutic to help fight Coronavirus disease (COVID-19) in the form of potential immunotherapies [31]; to alleviate use of frontline antibiotics for combatting resilient bacteria [32]; as an alternative to the use of silver nanoparticles [33]; or for other potential applications such as cosmetics [34], immune priming or boosting [12], and prevention of cell death [2]. However, study on the toxicity of these specific compounds are lacking, including use of Zebrafish testing, which would help inform product development and applications.

Using this ZFET technique, healthy zebrafish embryos, or larvae, have been used as surrogate animal model, which offers several advantages such as efficient, fast, reliable and comparable to human model [23,35]. This suits the "Zebrafish 3.0" toxicity model as the strategy share many physiological and cellular characteristics with higher vertebrates. In addition, the current ZFET assay uses 96-well plate that have a direct contact with EPS-ENS, mimicking the close contact with the demersal characteristic of zebrafish embryo. Secondly, extra-uterine and transparency can be examined, representing clear phenotypic embryonic development changes. In this study, teratogenic and embryotoxic effects evaluations are essential for EPS-ENS to determine the concentration that is safe for consumption. With booming global health industry and products from plants and fungi, claimed to have strong pharmacological responses, most of them are still dubious on the toxicological profiles.

In this study, fertilized zebrafish embryos were exposed to concentrations of *G. ap*planatum extract, EPS (0.01–10 mg/mL) and ENS (0.01–10 mg/mL) that were shown to be non-toxic using this ZFET technique. Overall, EPS at concentrations <1 mg/mL and ENS at <1 mg/mL did not cause delay hatching towards the embryo and with 88% survival rate at 24 to 120 HPF. Additionally, there were no significant changes in both EPS and ENS at concentration <1 mg/mL on the embryo heart rate compared to the normal ones. Moreover, teratogenic effect and zebrafish embryo abnormalities can only be observed at concentrations >5 mg/mL and >1 mg/mL in EPS and ENS, respectively. The test showed that EPS has a higher  $LC_{50}$  value of 1.41 mg/mL, meaning that it is better than ENS with lower LC<sub>50</sub> value (0.87 mg/mL). Even though both EPS and ENS extracts originated from G. applanatum mycelium, they may differ in terms of compound composition which originated from fruiting body and different mycelial extraction procedures [36–38]. Former research verified that similar EPS from the sister *G. lucidum* mycelium exhibit a broad range of bioactivities, including immunostimulant and antitumorigenic effects [39], which are higher than those of the fruiting bodies [34]. Meanwhile, ENS has given a lower  $LC_{50}$  value than EPS due to its different mycelial extraction methodology; EPS is directly extracted from the surface of fungal mycelium while ENS has undergone series of physico-chemical extractions from the internal part of dried fungal mycelium [5].

Morphological deformities, including tail malformation, could restrict the embryo's ability to break the chorion (Ch in Figure 6) and hatch out post 5 days. Likewise, the absence of heartbeat and coagulated embryo is considered as lethal effects. The work of Dulay et al. [40] has confirmed that tail malformation in zebrafish embryo does occur when exposed to 10 mg/mL of *Ganoderma* sp. fruiting body extract. In addition, it depicted that the safety threshold concentration is <10 mg/mL for any Ganoderma sp. fruiting body extract [40], which is in-line with the current data described for *G. applanatum* EPS and ENS possessing less toxic responses.

Some medicinal mushrooms have also been studied on the effects of toxicity towards zebrafish embryos in comparison to *Ganoderma* species. Exposure of termite-mound mushroom *Termitomyces clypeatus* extract at concentrations of >0.1% reduces the zebrafish embryos hatchability and <50% extract gave visible teratogenic effects post 48 HPF [41,42]. On the other hand, exposure of grey oyster mushroom *Pleurotus ostreatus* ethanol extract killed the zebrafish embryos using 2.5 and 5% concentrations post 12 HPF, while obstructed growth and tail malformation can be observe at 1% dose [43].

Recent studies have reported that *G. applanatum* polysaccharides are heavily used as food supplements for human species [19,44,45] to improve growth and immunity but none was tested as aquatic feed. The closest comparison were only by its popular sister *G. lucidum* at acceptable concentrations ranging from 1.0 to 1.5 g/kg on giant freshwater prawn (*Macrobrachium rosenbergii*) [46] and grass carp (*Ctenopharyngodon Idella*) [47], which enhance innate immune response and development. Therefore, to gain safety concentration for both aquaculture feed and human drug, both EPS and ENS extracts from *G. applanatum* toxicity reports are essential. Henceforth, the current novel ZFET data may provide useful information for assessing the potential health risks of the EPS-ENS consortia. Still, further tests are merited to evaluate the  $LC_{50}$  value of EPS-ENS extract for informing large-scale human trials and for larger animals (e.g., pig, rabbit, and adult trout), before this innovation can be deployed for commercial use.

This constitutes the first toxicology study that addresses bioreactor-cultured European *G. applanatum* (temperature climate mushroom) extracts where findings are compared to that of extracts derived from its sister *G. lucidum* (Table 1). Three studies have previously reported on the evaluation of EPS from *G. lucidum* using ZFET assay (2648  $\mu$ g/mL) [11], normal prostate cell line (500  $\mu$ g/mL) [48], and normal human lung cell (1000  $\mu$ g/mL) [49]. The findings from this present study reports on testing and evaluation of both non-toxic mycelial EPS (1410  $\mu$ g/mL) and ENS (870  $\mu$ g/mL) of stirred-tank bioreactor grown *G. applanatum* samples using the zebrafish 3.0 toxicity model. Moreover, this study provided evidence to support the safe use of European *G. applanatum* bioactive polysaccharides via Zebrafish model.

		Non-Toxic Conce		
Source	Toxicological Model	Exopolysaccharide (EPS)	Endopolysaccharide (ENS)	References
<i>G. applanatum</i> BGS6Ap	In vivo—Zebrafish embryos and larvae	1410	870	Current study
G. lucidum QRS 5120	In vivo—Zebrafish embryos and larvae	2648	NA	[11]
G. lucidum BCCM 31549	In vitro—normal human prostate cell line (PN2TA)	500	NA	[48]
G. lucidum	In vitro—normal human lung cell (WRL68)	1000	NA	[49]

**Table 1.** Comparison with published non-toxicity assessment of exopolysaccharide (EPS) and endopolysaccharide (ENS) from the mushroom *Ganoderma* sp.

#### 4. Materials and Methods

#### 4.1. European Ganoderma Applanatum Bioreactor Fermentation

European *G. applanatum* strain BGS6Ap was isolated by Prof. Dr. Anita Klaus from the Republic of Serbia in the wild region of Mount Kosmaj at temperate environment (18 °C–24 °C), coordinate of 44°27′57″ N, 20°33′52″ E, and maintained at 4 °C on Malt Extract Agar (MEA) slants prior to fermentation setup. The bioreactor fermentation of *G. applanatum* consists of double-seed culture stages. The seed culture medium condition was previously optimized for Ganoderma sp. liquid batch fermentation at fixed metrics (g/L): Yeast extract 1, Glucose 30, KH<sub>2</sub>PO<sub>4</sub> 0.5, MgSO<sub>4</sub> 0.5, and NH<sub>4</sub>Cl<sub>4</sub> 4 [50]. The fungus was cultivated for ten days at pre-optimized condition of 25 °C, pH 6, 10 g/L of glucose and 150 rpm in a 5-L stirred-tank (STR- Labfors, Infors H-T, Basel, Switzerland) bioreactor [51] to produce fungal biomass.

#### 4.2. G. applanatum EPS-ENS Preparations

The fungal biomass from the harvested *G. applanatum* culture was subjected to Buchner funnel filtration, followed by triple washing with distilled water. To obtain EPS, the filtrate was precipitated by the addition of four volumes of 99% (v/v) cold-ethanol-shock treatment and left overnight at 4 °C. After 24 h, crude EPS was formed naturally by agglomeration, and centrifuged at 10,000 rpm for 20 min. Meanwhile, to obtain ENS, the initial washed-fungal biomass was dried in a dryer until it powdered. The dried biomass was mixed with 1 g:20 mL distilled water, then subjected to heat-extraction (24 °C for 30 min) [52]. The resulting crude ENS was precipitated using the similar procedure of crude EPS. Finally, both EPS and ENS crude extracts were slow-dried to constant weight prior to toxicity studies. Stock powder of dried EPS-ENS was diluted in embryo medium (Danio-SprintM solution) producing working solutions (10 mg/mL). The mixture was diluted in 2-fold serial dilutions obtaining six concentrations ranging from 0.01–5 mg/mL in a 96-well microplate (200 µL). Embryos as in embryo medium were assigned as standard control (untreated).

#### 4.3. Zebrafish Maintenance and Breeding

The use of zebrafish was approved under the by the Institutional Animal Care and Use Committee (IACUC), Faculty of Biotechnology and Biomolecular Sciences, Universiti Putra Malaysia. Briefly, zebrafish of both sexes were placed into a breeding tank on the day the breeding tank was prepared. Following the first day, embryos were collected, washed, and then incubated in Danio-SprintM solution. Dead and coagulated embryos were discarded in an attempt to find healthy embryos [11].

#### 4.4. Zebrafish Embryo Toxicity (ZFET) Assay

The standard of Organization for Economic Cooperation and Development (OECD) standard was used in the ZFET assay [53]. Initially, one zebrafish embryo per well system at 0 HPF were subjected to diluted EPS-ENS samples (200  $\mu$ L) in 96-well microplates (Corning<sup>®</sup> 96 Well Clear Polystyrene Microplate, Corning, NY, USA) at seven different concentrations ranging from 0.01–10 mg/mL. The EPS-ENS and untreated samples were tested with a total of 24 replicate of embryos per exposure group slight above minimum standard [54,55]. Successful treated embryos were incubated at room temperature (25–28 °C) for 5 days. The cumulative mortality and developmental malformations of embryos and larvae were observed and determined every 24 h from 0–120 HPF as previously suggested by Parenti et al., [56]. The results indicate that mortality, hatching rate, heart rate, morphological malformation or teratogenic defects were observed using THUNDER Imager 3D Live Cell & 3D Cell Culture & 3D Assay, Leica Microsystems GmbH, Wetzlar, Germany. Heartbeat (1 min) and lethal endpoints (no heartbeat and clear coagulation) were assessed as according to Taufek et al., [11]. Anomalous development includes yolk sac oedema, curved body, non-hatched, bent tail and pericardial oedema. The principle of toxicity (LC<sub>50</sub>) values for EPS-ENS derivative >1 mg/mL are considered relatively harmless, 0.1–1 mg/mL are considered practically non-toxic, 0.01–0.1 mg/mL are considered slightly toxic, 0.001–0.01 mg/mL are considered moderately toxic, 0.0001–0.001 mg/mL are considered highly toxic and <0.0001 mg/mL are considered super toxic. Video S1: EPS-ENS Untreated vs. 0.01, 0.5 and 1 mg/mL treatments was supplied as supplementary data.

#### 4.5. Statistical Analysis

ZFET response graphs and the lethal concentration at 50% (LC<sub>50</sub>) of treated samples toward zebrafish embryos were produced by using GraphPad Prism version 9.0 (GraphPad Software, Inc., 2365 Northside Dr. Suite 560, San Diego, CA, USA). Heart rate response was presented as mean  $\pm$  standard error of mean (S.E.M) from three different animals. One-way analysis of variance (ANOVA) was used to carry out the significant differences with a post hoc test via Dunnett's Multiple Comparison. The significant difference was considered at \* p < 0.05, \*\* p < 0.01 and \*\*\* p < 0.001 between the means of treated group as compared to embryos in embryo medium.

#### 5. Conclusions

In conclusion, this represents the first study to report on the use of zebrafish embryo toxicity (ZFET) assay on bioreactor-grown European medicinal mushroom *G. applanatum* EPS and ENS extracts. EPS (LC<sub>50</sub>: 1.41 mg/mL) was harmless while ENS (LC<sub>50</sub>: 0.87 mg/mL) are practically non-toxic. The ZFET assay offers a fast, affordable, robust, and efficient early development approach to evaluating extracts from medicinal fungi for future use in aquaculture [57]. Findings will inform and guide future "foodomics" research spanning molecular toxicology and linked proteomics and metablomics as it relates to targeting specific transcripts for biosensor development that will support end-to-end monitoring of value chain, including using of machine learning for mitigating food waste and improving "circularlity" for the food sector. Specifically, this study provides new knowledge that will inform efficacy for future immune-priming and disease mitigation innovation for global food security.

**Supplementary Materials:** The following are available online at https://www.mdpi.com/1422-006 7/22/4/1675/s1, Video S1: EPS-ENS Untreated vs. 0.01, 0.5 and 1 mg/mL treatments.

Author Contributions: Conceptualization, W.A.A.Q.I.W.-M. and N.R.; methodology, W.A.A.Q.I.W.-M. and N.R.; software, W.A.A.Q.I.W.-M. and N.R.; validation, Z.I. and A.A.J.; formal analysis, W.A.A.Q.I.W.-M.; investigation, W.A.A.Q.I.W.-M.; resources, N.R.; data curation, W.A.A.Q.I.W.-M. and N.R.; writing—original draft preparation, W.A.A.Q.I.W.-M. and N.R.; writing—review and editing, Z.I. and A.A.J.; visualization, W.A.A.Q.I.W.-M.; supervision, W.A.A.Q.I.W.-M.; project administration, W.A.A.Q.I.W.-M. and N.R.; funding acquisition, W.A.A.Q.I.W.-M. and N.R. All authors have read and agreed to the published version of the manuscript.

**Funding:** We want to thank Universiti Malaya under the Research University Grant-Faculty Program GPF084A-2020 and MOHE- Top 100 (IIRG)-IISS: IIRG003A-2020IISS awarded to Wan-Mohtar. The work was also funded by H2020 MSCA RISE ICHTHYS project [number 872217/19] awarded to Neil Rowan.

**Institutional Review Board Statement:** The study was conducted according to the guidelines of the Organization for Economic Cooperation and Development (OECD) No. 236; Fish Embryo Acute Toxicity (FET) Test (OECD, 2013) under compliance of IACUC UPM.

Informed Consent Statement: Not applicable.

**Data Availability Statement:** The data presented in this study are available on request from the corresponding author. The data are not publicly available due to privacy.

**Acknowledgments:** We would like to thank Anita Klaus for European *G. applanatum* strain BGS6Ap isolated by from the Republic of Serbia in the wild region of Mount Kosmaj. Sincere gratitude to Danio Assay Team, Department of Biochemistry, Faculty of Biotechnology and Biomolecular Sciences, Universiti Putra Malaysia, Serdang, Selangor for technical support.

**Conflicts of Interest:** The authors declare no conflict of interest.

#### References

- Venturella, G.; Ferraro, V.; Cirlincione, F.; Gargano, M.L. Medicinal Mushrooms: Bioactive Compounds, Use, and Clinical Trials. *Int. J. Mol. Sci.* 2021, 22, 634. [CrossRef]
- Liang, Z.; Yuan, Z.; Guo, J.; Wu, J.; Yi, J.; Deng, J.; Shan, Y. Ganoderma lucidum polysaccharides prevent palmitic acid-evoked apoptosis and autophagy in intestinal porcine epithelial cell line via restoration of mitochondrial function and regulation of MAPK and AMPK/Akt/mTOR signaling pathway. *Int. J. Mol. Sci.* 2019, 20, 478. [CrossRef]
- Klaus, A.; Wan-Mohtar, W.A.A.Q.I.; Nikolić, B.; Cvetković, S.; Vunduk, J. Pink oyster mushroom Pleurotus flabellatus mycelium produced by an airlift bioreactor—the evidence of potent in vitro biological activities. World J. Microbiol. Biotechnol. 2021, 37, 1–14.
- Wan-Mohtar, W.A.A.Q.I.; Halim-Lim, S.A.; Kamarudin, N.Z.; Rukayadi, Y.; Abd Rahim, M.H.; Jamaludin, A.A.; Ilham, Z. Fruiting-body-base flour from an Oyster mushroom waste in the development of antioxidative chicken patty. *J. Food Sci.* 2020, *85*, 3124–3133. [CrossRef]

- Abdullah, N.R.; Sharif, F.; Azizan, N.H.; Hafidz, I.F.M.; Supramani, S.; Usuldin, S.R.A.; Ahmad, R.; Wan-Mohtar, W.A.A.Q.I. Pellet diameter of Ganoderma lucidum in a repeated-batch fermentation for the trio total production of biomass-exopolysaccharideendopolysaccharide and its anti-oral cancer beta-glucan response. *AIMS Microbiol.* 2020, *6*, 379. [CrossRef] [PubMed]
- Roda, E.; De Luca, F.; Di Iorio, C.; Ratto, D.; Siciliani, S.; Ferrari, B.; Cobelli, F.; Borsci, G.; Priori, E.C.; Chinosi, S. Novel Medicinal Mushroom Blend as a Promising Supplement in Integrative Oncology: A Multi-Tiered Study using 4T1 Triple-Negative Mouse Breast Cancer Model. *Int. J. Mol. Sci.* 2020, 21, 3479. [CrossRef] [PubMed]
- O'Neill, E.A.; Rowan, N.J.; Fogarty, A.M. Novel use of the alga Pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. *Sci. Total Environ.* 2019, 692, 209–218. [CrossRef]
- 8. Tahar, A.; Kennedy, A.; Fitzgerald, R.D.; Clifford, E.; Rowan, N. Full water quality monitoring of a traditional flow-through rainbow trout farm. *Fishes* **2018**, *3*, 28. [CrossRef]
- 9. O'Neill, E.A.; Stejskal, V.; Clifford, E.; Rowan, N.J. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production—A case study from the Republic of Ireland. *Sci. Total Environ.* **2020**, *706*, 136044. [CrossRef]
- Wan-Mohtar, W.A.A.Q.I.; Taufek, N.M.; Yerima, G.; Rahman, J.; Thiran, J.P.; Subramaniam, K.; Sabaratnam, V. Effect of bioreactorgrown biomass from Ganoderma lucidum mycelium on growth performance and physiological response of red hybrid tilapia (*Oreochromis* sp.) for sustainable aquaculture. *Org. Agric.* 2020. [CrossRef]
- Taufek, N.M.; Harith, H.H.; Abd Rahim, M.H.; Ilham, Z.; Rowan, N.; Wan-Mohtar, W.A.A.Q.I. Performance of mycelial biomass and exopolysaccharide from Malaysian Ganoderma lucidum for the fungivore red hybrid Tilapia (*Oreochromis* sp.) in Zebrafish embryo. *Aquac. Rep.* 2020, 17, 100322. [CrossRef]
- 12. Murphy, E.J.; Rezoagli, E.; Major, I.; Rowan, N.J.; Laffey, J.G. β-glucan metabolic and immunomodulatory properties and potential for clinical application. *J. Fungi* **2020**, *6*, 356. [CrossRef]
- Supramani, S.; Jailani, N.; Ramarao, K.; Zain, N.A.M.; Klaus, A.; Ahmad, R.; Wan, W.A.A.Q.I. Pellet diameter and morphology of European Ganoderma pfeifferi in a repeated-batch fermentation for exopolysaccharide production. *Biocatal. Agric. Biotechnol.* 2019, 19, 101118. [CrossRef]
- 14. Hassan, N.A.; Supramani, S.; Sohedein, M.N.A.; Usuldin, S.R.A.; Klaus, A.; Ilham, Z.; Chen, W.-H.; Wan-Mohtar, W.A.A.Q.I. Efficient biomass-exopolysaccharide production from an identified wild-Serbian Ganoderma lucidum strain BGF4A1 mycelium in a controlled submerged fermentation. *Biocatal. Agric. Biotechnol.* **2019**, *21*, 101305. [CrossRef]
- 15. Deborah Paripuranam, T.; Divya, V.; Ulaganathan, P.; Balamurugan, V.; Umamaheswari, S. Replacing fish meal with earthworm and mushroom meals in practical diets of Labeo rohita and Hemigrammus caudovittatus fingerlings. *Indian J. Anim. Res.* **2011**, 45, 115–119.
- Baba, E.; Uluköy, G.; Öntaş, C. Effects of feed supplemented with *Lentinula edodes* mushroom extract on the immune response of rainbow trout, Oncorhynchus mykiss, and disease resistance against *Lactococcus garvieae*. *Aquaculture* 2015, 448, 476–482. [CrossRef]
- Liang, D.; Yong, T.; Chen, S.; Xie, Y.; Chen, D.; Zhou, X.; Li, D.; Li, M.; Su, L.; Zuo, D. Hypouricemic effect of 2, 5dihydroxyacetophenone, a computational screened bioactive compound from *Ganoderma applanatum*, on hyperuricemic mice. *Int. J. Mol. Sci.* 2018, *19*, 1394. [CrossRef] [PubMed]
- Tang, L.; Zhu, Z.-F.; Cao, L.-P.; Shen, M.; Gao, Y.; Tu, C.-J.; Zhang, Z.-H.; Shan, W.-G. Thermosensitive gel of polysaccharide from *Ganoderma applanatum* combined with paclitaxel for mice with 4T1 breast cancer. *Zhongguo Zhong Yao Za Zhi Zhongguo Zhongyao Zazhi China J. Chin. Mater. Med.* 2020, 45, 2533–2539.
- 19. Hanyu, X.; Lanyue, L.; Miao, D.; Wentao, F.; Cangran, C.; Hui, S. Effect of Ganoderma applanatum polysaccharides on MAPK/ERK pathway affecting autophagy in breast cancer MCF-7 cells. *Int. J. Biol. Macromol.* **2020**, *146*, 353–362. [CrossRef]
- 20. Rahmann, G.; Grimm, D.; Kuenz, A.; Hessel, E. Combining land-based organic and landless food production: A concept for a circular and sustainable food chain for Africa in 2100. *Org. Agric.* **2020**, *10*, 9–21. [CrossRef]
- Marino, F.; Di Caro, G.; Gugliandolo, C.; Spano, A.; Faggio, C.; Genovese, G.; Morabito, M.; Russo, A.; Barreca, D.; Fazio, F. Preliminary study on the in vitro and in vivo effects of *Asparagopsis taxiformis* bioactive phycoderivates on teleosts. *Front. Physiol.* 2016, 7, 459. [CrossRef] [PubMed]
- Lo Cascio, P.; Calabrò, C.; Bertuccio, C.; Iaria, C.; Marino, F.; Denaro, M.G. Immunohistochemical characterization of PepT1 and ghrelin in gastrointestinal tract of zebrafish: Effects of Spirulina vegetarian diet on the neuroendocrine system cells after alimentary stress. *Front. Physiol.* 2018, *9*, 614. [CrossRef] [PubMed]
- 23. Ghilardi, A.; Diana, A.; Prosperi, L.; Del Giacco, L. Expression pattern of the small muscle protein, X-linked (smpx) gene during zebrafish embryonic and larval developmental stages. *Gene Expr. Patterns* **2020**, *36*, 119110. [CrossRef]
- Parolini, M.; Ghilardi, A.; De Felice, B.; Del Giacco, L. Environmental concentration of fluoxetine disturbs larvae behavior and increases the defense response at molecular level in zebrafish (*Danio rerio*). *Environ. Sci. Pollut. Res.* 2019, 26, 34943–34952. [CrossRef]
- Rizzo, C.; Genovese, G.; Morabito, M.; Faggio, C.; Pagano, M.; Spanò, A.; Zammuto, V.; Minicante, S.A.; Manghisi, A.; Cigala, R.M. Potential antibacterial activity of marine macroalgae against pathogens relevant for aquaculture and human health. *J. Pure Appl. Microbiol.* 2017, 11, 1695–1706. [CrossRef]
- Zhang, X.-H.; He, X.; Austin, B. Vibrio harveyi: A serious pathogen of fish and invertebrates in mariculture. *Mar. Life Sci. Technol.* 2020, 2, 1–15. [CrossRef] [PubMed]

- 27. Bakkers, J. Zebrafish as a model to study cardiac development and human cardiac disease. *Cardiovasc. Res.* 2011, 91, 279–288. [CrossRef]
- Baker, K.; Warren, K.S.; Yellen, G.; Fishman, M.C. Defective "pacemaker" current (Ih) in a zebrafish mutant with a slow heart rate. Proc. Natl. Acad. Sci. USA 1997, 94, 4554–4559. [CrossRef]
- Kozarski, M.S.; Klaus, A.S.; Vunduk, J.Đ.; Jakovljević, D.M.; Jadranin, M.B.; Nikšić, M.P. Health impact of commercially cultivated mushroom Agaricus bisporus and wild-growing mushroom *Ganoderma resinaceum*—a comparative overview. *J. Serb. Chem. Soc.* 2020, *85*, 721–735. [CrossRef]
- Vunduk, J.; Wan-Mohtar, W.A.A.Q.I.; Mohamad, S.A.; Abd Halim, N.H.; Dzomir, A.Z.M.; Žižak, Ž.; Klaus, A. Polysaccharides of *Pleurotus flabellatus* strain Mynuk produced by submerged fermentation as a promising novel tool against adhesion and biofilm formation of foodborne pathogens. *LWT* 2019, *112*, 108221. [CrossRef]
- Murphy, E.J.; Masterson, C.; Rezoagli, E.; O'Toole, D.; Major, I.; Stack, G.D.; Lynch, M.; Laffey, J.G.; Rowan, N.J. β-Glucan extracts from the same edible shiitake mushroom *Lentinus edodes* produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—Implications for coronavirus disease (COVID-19) immunotherapies. *Sci. Total Environ.* 2020, 732, 139330. [CrossRef]
- Masterson, C.H.; Murphy, E.J.; Gonzalez, H.; Major, I.; McCarthy, S.; O'Toole, D.; Laffey, J.G.; Rowan, N.J. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumoniae-induced pulmonary sepsis. *Lett. Appl. Microbiol.* 2020, 71, 405–412. [CrossRef] [PubMed]
- 33. Klaus, A.; Petrovic, P.; Vunduk, J.; Pavlovic, V.; Van Griensven, L.J.L.D. The antimicrobial activities of silver nanoparticles synthesized from medicinal mushrooms. *Int. J. Med. Mushrooms* **2020**, *22*, 869–883. [CrossRef] [PubMed]
- Kozarski, M.; Klaus, A.; Jakovljević, D.; Todorović, N.; Wan-Mohtar, W.A.A.Q.I.; Nikšić, M. Ganoderma lucidum as a cosmeceutical: Antiradical potential and inhibitory effect on hyperpigmentation and skin extracellular matrix degradation enzymes. *Arch. Biol. Sci.* 2019, *71*, 253–264. [CrossRef]
- 35. Della Torre, C.; Maggioni, D.; Ghilardi, A.; Parolini, M.; Santo, N.; Landi, C.; Madaschi, L.; Magni, S.; Tasselli, S.; Ascagni, M. The interactions of fullerene C60 and Benzo (α) pyrene influence their bioavailability and toxicity to zebrafish embryos. *Environ. Pollut.* **2018**, 241, 999–1008. [CrossRef] [PubMed]
- 36. Zhao, L.Y.; Dong, Y.H.; Chen, G.T.; Hu, Q.H. Extraction, purification, characterization and antitumor activity of polysaccharides from Ganoderma lucidum. *Carbohydr. Polym.* **2010**, *80*, 783–789. [CrossRef]
- 37. Ma, Y.; He, H.; Wu, J.; Wang, C.; Chao, K.; Huang, Q. Assessment of Polysaccharides from Mycelia of genus Ganoderma by Mid-Infrared and Near-Infrared Spectroscopy. *Sci. Rep.* **2018**, *8*, 10. [CrossRef] [PubMed]
- 38. Huang, Y.; Li, N.; Wan, J.B.; Zhang, D.; Yan, C. Structural characterization and antioxidant activity of a novel heteropolysaccharide from the submerged fermentation mycelia of *Ganoderma capense*. *Carbohydr. Polym.* **2015**, *134*, 752–760. [CrossRef]
- Ahmadifar, E.; Dawood, M.A.; Moghadam, M.S.; Sheikhzadeh, N.; Hoseinifar, S.H.; Musthafa, M.S. Modulation of immune parameters and antioxidant defense in zebrafish (*Danio rerio*) using dietary apple cider vinegar. *Aquaculture* 2019, 513, 734412. [CrossRef]
- Dulay, R.M.R.; Kalaw, S.P.; Reyes, R.G.; Alfonso, N.F.; Eguchi, F. Teratogenic and Toxic Effects of Lingzhi or Reishi Medicinal Mushroom, Ganoderma lucidum (W.Curt.:Fr.) P. Karst. (Higher Basidiomycetes), on Zebrafish Embryo as Model. *Int. J. Med. Mushrooms* 2012, 14, 507–512. [CrossRef] [PubMed]
- De Castro, M.E.G.; Dulay, R.M.R.; Enriquez, M. Toxic and Teratogenic Effects of Medicinal and Culinary Mushroom, *Termitomyces clypeatus*, Collected from the Termite Mound in Mt. Makiling Forest Reserve, Los Baños, Laguna, Philippines on Developing Embryos Of Zebrafish (*Danio rerio*). *Der Pharm. Lett.* 2016, *8*, 237–242.
- 42. Wu, X.; Hao, Q.; Teame, T.; Ding, Q.; Liu, H.; Ran, C.; Yang, Y.; Xia, L.; Wei, S.; Zhou, Z. Gut microbiota induced by dietary GWF<sup>®</sup> contributes to growth promotion, immune regulation and disease resistance in hybrid sturgeon (Acipenserbaerii x Acipenserschrenckii): Insights from a germ-free zebrafish model. *Aquaculture* 2020, 520, 734966. [CrossRef]
- 43. De Castro, M.; Dulay, R.M.R. Toxic and teratogenic effects of *Lentinus sajor-caju* and *Pleurotus* ostreatus ethanolic extracts in Danio rerio embryo model. *Int. J. Biol. Pharm. Allied Sci.* 2015, 4, 2261–2269.
- 44. Gao, Z.; Yuan, F.; Li, H.; Feng, Y.; Zhang, Y.; Zhang, C.; Zhang, J.; Song, Z.; Jia, L. The ameliorations of *Ganoderma applanatum* residue polysaccharides against CCl4 induced liver injury. *Int. J. Biol. Macromol.* **2019**, *137*, 1130–1140. [CrossRef]
- 45. Zhen, D.; Su, L.; Miao, Y.; Zhao, F.; Ren, G.; Mahfuz, S.; Song, H. Purification, partial characterization and inducing tumor cell apoptosis activity of a polysaccharide from *Ganoderma applanatum*. *Int. J. Biol. Macromol.* **2018**, *115*, 10–17. [CrossRef]
- 46. Mohan, K.; Muralisankar, T.; Uthayakumar, V.; Chandirasekar, R.; Rajan, D.K. Dietary *Ganoderma lucidum* polysaccharides to enhance the growth, immune response and disease resistance of freshwater prawn *Macrobrachium rosenbergii*. *Aquac. Rep.* **2019**, 14, 100203. [CrossRef]
- 47. Chithra, E.; Padmanaban, A.; Mohan, K. Potential use of *Ganoderma lucidum* polysaccharides as a feed supplement in diets on survival and growth performance of the grass carp, *Ctenopharyngodon idella*. *Int. J. Fish. Aquac. Stud.* **2016**, *4*, 328–333.
- Wan-Mohtar, W.A.Q.I.; Young, L.; Abbott, G.M.; Clements, C.; Harvey, L.M.; McNeil, B. Antimicrobial properties and cytotoxicity of sulfated (1,3)-beta-D-glucan from the mycelium of the mushroom *Ganoderma lucidum*. J Microbiol. Biotechnol. 2016, 26, 999–1010. [CrossRef] [PubMed]
- 49. Chung, W.T.; Lee, S.H.; Dai Kim, J.; Park, Y.S.; Hwang, B.; Lee, S.Y.; Lee, H.Y. Effect of mycelial culture broth of Ganoderma lucidum on the growth characteristics of human cell lines. *J. Biosci. Bioeng.* **2001**, *92*, 550–555. [CrossRef]

- 50. Supramani, S.; Rahayu Ahmad, Z.I.; Annuar, M.S.M.; Klaus, A.; Wan-Mohtar, W.A.A.Q.I. Optimisation of biomass, exopolysaccharide and intracellular polysaccharide production from the mycelium of an identified *Ganoderma lucidum* strain QRS 5120 using response surface methodology. *AIMS Microbiol.* **2019**, *5*, 19. [CrossRef] [PubMed]
- 51. Usuldin, S.R.A.; Mahmud, N.; Ilham, Z.; Ikram, N.K.K.; Ahmad, R.; Wan-Mohtar, W.A.A.Q.I. In-depth spectral characterization of antioxidative (1, 3)-β-D-glucan from the mycelium of an identified tiger milk mushroom *Lignosus rhinocerus* strain ABI in a stirred-tank bioreactor. *Biocatal. Agric. Biotechnol.* **2020**, *23*, 101455. [CrossRef]
- Nik Ubaidillah, N.H.; Abdullah, N.; Sabaratnam, V. Isolation of the intracellular and extracellular polysaccharides of *Ganoderma* neojaponicum (Imazeki) and characterization of their immunomodulatory properties. *Electron. J. Biotechnol.* 2015, 18, 188–195. [CrossRef]
- 53. OECD. Test No. 236: Fish Embryo Acute Toxicity (FET) Test; OECD Publishing: Paris, France, 2013.
- 54. Aspatwar, A.; Hammaren, M.M.; Parikka, M.; Parkkila, S. Rapid Evaluation of Toxicity of Chemical Compounds Using Zebrafish Embryos. J. Vis. Exp. 2019. [CrossRef] [PubMed]
- 55. Gourmelon, A.; Delrue, N. Validation in support of internationally harmonised OECD test guidelines for assessing the safety of chemicals. *Valid. Altern. Methods Toxic. Test.* **2016**, *856*, 9–32.
- 56. Parenti, C.C.; Ghilardi, A.; Della Torre, C.; Magni, S.; Del Giacco, L.; Binelli, A. Evaluation of the infiltration of polystyrene nanobeads in zebrafish embryo tissues after short-term exposure and the related biochemical and behavioural effects. *Environ. Pollut.* **2019**, 254, 112947. [CrossRef]
- 57. Rowan, N.J.; Galanakis, C.M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? *Sci. Total Environ.* **2020**, *748*, 141362. [CrossRef]



Contents lists available at ScienceDirect

### **Future Foods**



journal homepage: www.elsevier.com/locate/fufo

# Investigations on the use of exopolysaccharide derived from mycelial extract of *Ganoderma lucidum* as functional feed ingredient for aquaculture-farmed red hybrid Tilapia (*Oreochromis* sp.)



Wan Abd Al Qadr Imad Wan-Mohtar<sup>a,c</sup>, Norhidayah Mohd Taufek<sup>b</sup>, Jaganath Puvanen Thiran<sup>b</sup>, Janathu Ferdaus Pajaru Rahman<sup>b</sup>, Grema Yerima<sup>b</sup>, Kumeera Subramaniam<sup>b</sup>, Neil Rowan<sup>c,d,\*</sup>

<sup>a</sup> Functional Omics and Bioprocess Development Laboratory, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, 50603 Kuala Lumpur, Malaysia

<sup>b</sup> AquaNutri Biotech Research Laboratory, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, 50603 Kuala Lumpur, Malaysia

<sup>c</sup> Bioscience Research Institute, Athlone Institute of Technology, Ireland

<sup>d</sup> Empower Eco Sustainablity HUB, Lough Boora, Co. Offaly, Ireland

#### ARTICLE INFO

Keywords: Ganoderma lucidum Exopolysaccharides Red hybrid Tilapia Growth performance Sustainable aquaculture Functional Food

#### ABSTRACT

There is growing interest in medicinal fungi for promoting immunomodulation benefits in humans and animals. In this study, exopolysaccharide (EPS) from *Ganoderma lucidum* was used in feed (EFF) on production of farmed-red hybrid Tilapia performance for sustainable aquaculture. The feeding trial was conducted for a period of 42 days. Tilapia weight gain (WG) was significantly the highest  $(18.75 \pm 0.03 \text{ g/fish})$  using 3 g/kg of EFF compared to control (0 g/kg EFF; 12.4 g/fish). Fish fed with 3 g/kg of EFF were shown to be healthiest as attested by measured organosomatic indices that achieved an hepatosomatic index (HSI) of 2.22% and condition factor (CF) of 1.46% compared to control (1.97% HSI; 1.60 CF). Antioxidant activity of Glutathione S-transferase (GST) exhibited significantly lower activities in 1 and 2 g/kg of EFF while increasing catalase (CAT) activity was detected in higher EFF (2-3 g/kg of EPS). Muscle compositions were not affected by the EFF feeding regimes. In contrast, EFF feeding produced significantly higher haemoglobin (7.43 g/dl), haematocrit (37.5%), red blood cell (2.69  $10^6/\text{mm}^3$ ) compared to control group. In summary, EFF was beneficial in promoting growth, stimulating antioxidant enzymes, and health status of red hybrid Tilapia, which highlights the potential for future use in intensive sustainable aquaculture practices.

#### 1. Introduction

Aquaculture is a rapidly growing food sector that has been characterised by considerable investment in many parts of the world including Malaysia (O'Neill et al., 2019; Tahar et al., 2018a; Tahar et al., 2018b; Ruiz-Salmón et al., 2020). It has been identified as a sustainable solution for addressing problems of depleting natural wild fish stock globally (O'Neill et al., 2020). Rowan and Galanaksi (2020) have also described the important role of developing intensive aquaculture practices for transitioning beyond the global economic recession, which has been caused by ongoing COVID-19 pandemic. Over the past four decades, a significant increase in fish supply has been achieved through the intensity of aquaculture production that has led to a stressful environment for the fish cultured in captivity (FAO, 2018; Naughton et al., 2020). Operational flux, such as the imposition of stressful environmental growth conditions in aquaculture, may lead to elevated susceptibility to diseases where administration of immunomodulatory therapeutics in feed has been reported previously to alleviate onset or severity including positive impact on fish gut microbiome (Carballo et al., 2019).

The utilisation of antibiotics in managing diseases is a concern in the aquaculture industry due to the increased risk of developing antimicrobial resistance in animal production (Naughton et al., 2020). A growing interest in organic farming has warranted the use of natural alternatives, such as medicinal plants and fungi as a feed supplement, as a way to reduce the adverse side effect of chemical residue from antibiotics (Baba et al., 2015; Wan-Mohtar, 2021). These prebiotic and probiotic ingredients have been reported to improve immunity as well as a growth promoter in fish and shrimp without compromising their health and also sustainable for the environment (Mohan et al., 2019a; Rufchaei et al., 2019). Probiotics are important feed supplements in aquaculture due to their capacity to enhance health and to prevent disease (Carballo et al., 2019). Carballo and co-workers (2019) recently reported that such ingredients are typically non-digestable molecules such as yeast-derived beta-glucans, which selectively modulate the intestinal microbiome by

\* Corresponding author at: Bioscience Research Institute, Athlone Institute of Technology, Ireland. *E-mail address:* nrowan@ait.ie (N. Rowan).

https://doi.org/10.1016/j.fufo.2021.100018

Received 13 October 2020; Received in revised form 2 February 2021; Accepted 4 February 2021

2666-8335/© 2021 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)

promoting indigenous microbial populations and commensurately prevention of unwanted pathogen proliferation. In spite of demonstrating the potential benefits of using prebiotics to improve animal health, their use in aquaculture remains limited and their mode of mechanistic action is far from understood (Dimitrogou et al., 2011; Akhter et al., 2015; Morrison et al., 2016; Ruiz-Salmón et al., 2020; Ruiz-Salmón et al., 2021)

Use of medicinal mushrooms has been previously reported to provide health benefits in both animals and humans through immunomodulatory function (Smith et al., 2002; Rowan et al., 2003; Sullivan et al., 2006; Masterson et al., 2020; Murphy et al. 2020a; Murphy et al., 2020b; Galanakis et al., 2020). Ganoderma lucidum is one such medicinal mushroom known for its valuable properties to treat many diseases (Cao et al., 2018). Regarded as a 'marvellous herb', G. lucidum is widely used in many countries, particularly in East Asia due to its therapeutic effects (Kozarski et al., 2019; Wan-Mohtar et al., 2020). It contains functional metabolites that possess antifungal, antibacterial and anti-inflammatory properties (Rowan et al., 2003; Cao et al., 2018; Wan-Mohtar et al., 2017). The bioactive metabolites, particularly polysaccharide extracted from G. lucidum, have demonstrated immunomodulatory and anticancer properties (Meng et al., 2014). However, the low yield of fungal polysaccharides extracted from fruiting bodies and mycelium of G. lucidum is appreciated as a technological limitation: consequently, it has prompted advance research in the area of fungal exopolysaccharide (EPS) through submerged liquid fermentation (Supramani et al., 2019b). Technically, the production of EPS is viewed to be less-harsh during the extraction process, hence reduce product degradation and resulted in costeffectiveness (Yuan et al., 2012).

Tilapia production is projected to increase from 58% in 2016 to 62% in 2030 (FAO, 2018) In Malaysia, red hybrid tilapia (Oreochromis sp.) is considered as the top commercial species besides catfish (DOF, 2018). However, there are many constraints in developing sustainable, yet intensive culture of tilapia. One of the top challenges is the disease threat that could results in economic loss. Several bacteria disease such as Streptococcosis, Columnaris and Haemorrhagic septicaemia are amongst common infections of tilapia culture in Malaysia. Viral infection such as Tilapia Lake Virus (TiLV) is currently a significant threat to global tilapia production and major outbreak has also been reported in Malaysia. The interaction of pathogenic bacteria and TiLV was reported to be synergistic, which resulted in strengthening of both pathogens and increasing severity of the disease (Amal et al., 2018; Elumalai et al., 2019; Jansen et al., 2019). According to USDA (2019) there is no vaccine available for the treatment of TiLV and appropriate alternative mitigation and biosecurity measures are needed to address this challenge. Hence, to reduce the challenges in the stressful event during intensive tilapia culture, introducing feed additive containing immunostimulant properties would be an advantage to boost their immunity. Therefore, this constitutes the first study to examine the potential of EPS fish-feed (EFF) extracted from the mycelium of G. lucidum on growth, antioxidant responses, body composition and haematological indices of Malaysian red hybrid tilapia.

#### 2. Materials and method

# 2.1. Production of fungal exopolysaccharides from the mycelium of Ganoderma lucidum

Fungal biomass was pre-grown in a 5-L stirred-tank bioreactor system (STR, Fermetec, Malaysia) under controlled submerged fermentation strictly according to Usuldin et al. (2019) using *Ganoderma lucidum* strain QRS 5120 (Supramani et al., 2019a) obtained from Functional Omics and Bioprocess Development Laboratory, Institute of Biological Sciences, Universiti Malaya. The fermented *G. lucidum* mycelial biomass was subjected to cold 95% (v/v) ethanolic extraction at 1:4 ratio and left for 48 hours at 4 °C for macromolecules precipitation. The method of extraction was conducted according to Hassan et al. (2019) whereas the

#### Table 1

Formulation and nutritional composition of exopolysaccharide fish feed (EFF) originated from the mycelium of *Ganoderma lucidum*.

Ingredients (g/kg)	Control <sup>5</sup>	1 g/kg	2 g/kg	3 g/kg
Fish meal (g)	300.0	300.0	300.0	300.0
Corn meal (g)	193.9	193.5	193.1	192.7
Rice bran meal (g)	199.3	198.9	198.6	198.2
Soybean meal (g)	236.8	236.6	236.3	236.1
Exopolysaccharide (g)	0.0	1.0	2.0	3.0
Lysine (g)	10.0	10.0	10.0	10.0
Methionine (g)	5.0	5.0	5.0	5.0
Vitamin premix(g) <sup>1</sup>	1.0	1.0	1.0	1.0
Mineral premix(g) <sup>2</sup>	3.0	3.0	3.0	3.0
Dicalcium phosphate (g)	10.0	10.0	10.0	10.0
Fish Oil (ml)	40.0	40.0	40.0	40.0
Total (g)	1000	1000	1000	1000
Nutritional composition				
Protein (%)	37.86	36.30	37.02	35.19
Lipid (%)	6.29	6.29	4.84	6.67
Fiber (%)	1.63	1.70	1.92	1.76
Dry matter (%)	85.65	85.69	85.73	85.77
Ash (%)	12.00	12.86	12.40	12.45
Carbohydrate (%) <sup>3</sup>	42.22	42.85	43.82	43.93
Gross energy (kJ/g) <sup>4</sup>	18.98	18.72	18.49	18.80

<sup>1</sup> The vitamin premix supplied the following per 100g diet: vitamin A, 500IU; vitamin D<sub>3</sub>, 100IU; vitamin E, 0.75 mg; vitamin K, 0.02 mg; vitamin B<sub>1</sub>, 1.0 mg; vitamin B<sub>2</sub>, 0.5 mg; vitamin B<sub>3</sub>, 0.3 mg; vitamin B<sub>6</sub>, 0.02 mg; folic acid, 0.1 mg; biotin, 0.235 mg; pantothenic acid, 1.0 mg; inositol, 2.5 mg.

<sup>2</sup> The mineral premix supplied the following per kg diet: selenium, 0.2 mg; iron, 8.0 mg; manganase, 1.0 mg; zinc, 8.0 mg; copper, 0.15 mg; potassium chloride, 0.4 mg; magnesium oxide, 0.6 mg; sodium bicarbonate, 1.5 mg; iodine, 1.0 mg; cobalt, 0.25 mg. <sup>3</sup>Carbohydrate = 100 - (% protein + % lipid + % fiber + % ash).

 $^4$  Gross energy was calculated as 23.9, 39.8, and 17.6 kJ/g for protein, lipid, and NFE, respectively.

<sup>5</sup> Feed: Control = 0.0 g/kg of EPS,

centrifuged (9000 rpm, 15 min) macromolecule precipitates was freezedried to produce crude EPS.

#### 2.2. Preparation of experimental diets

Four experimental EFF were formulated to be isoenergetic composition (~19g/kJ) containing 0 (control), supplementation of 1(1E), 2(2E) and 3(3E) g EPS per kg diet (Table 1). The supplements (in dried powder form) were uniformly mixed with dry ingredients and pelleted with extruder machine (KCM, Y132M-A). The pelleted feed was then ovendried at  $60^{\circ}$ C and later stored in  $4^{\circ}$ C until further use.

#### 2.3. Experimental animal and setup

Red hybrid Tilapia fingerlings were purchased from a local fish farm in Sungai Buloh, Malaysia and acclimated for two weeks prior to the feeding trial. Fish were hand-fed with commercial diet, *ad libitum*, twice per day during the acclimatisation period. One hundred twenty fish  $(16.19 \pm 0.24 \text{ g})$  were stocked into each of the eight tanks (100 L) with a density of 15 fish per treatment in duplicate cultures as justified by Wan-Mohtar et al., (2020). Feeding was performed twice daily at a 3% body weight ratio throughout the 42-days of feeding trial and feeding intake was recorded.

Each tank was equipped with filtration and constant aeration with dechlorinated water throughout the study. Water temperature were maintained at 28–29°C, pH at 6.0–6.8, DO above 4.0, ammonia <0.80 mg/L, and nitrate <1.9 mg/L. Approximately 25-30% of water was changed every two days, and uneaten feed and faeces were collected every day by siphoning at the bottom of the tanks to maintain water quality.

All procedures of fish rearing and handling were performed according to the Universiti Malaya Animal Care and Use Policy (UM ACUP) (IACUC-002), which was established according to the Malaysian Animal Act 1953 (Act 647).

#### 2.4. Growth performance and organosomatic indices

After termination of the feeding trial at day 42, fish were starved for 24 hours before weighing. All the fish were anaesthetised with clove oil and weigh individually to determine the growth performance including body weight gain (BWG), specific growth rate (SGR), feed conversion ratio (FCR), survival rate (SR) and feed intake (FI). Organosomatic indices including visceromatic index (VSI), hepatosomatic index (HSI) and condition factor (CF) were also measured post-feeding trial.

BWG(g/fish) = final weight(g) - initial weight(g)

FI(g/fish) = total feed for the feeding period(g)/number of fish alive

 $SGR(\%) = [(Ln \ final \ weight) - (Ln \ initial \ weight)/days \ of \ trial] \times 100$ 

FCR = feed intake(g)/weight gain(g)

 $SR(\%) = final number of fish alive/initial number of fish alive \times 100$ 

 $CF = [bod y weight (g) / (bod y length (cm)] \times 100$ 

 $VSI(\%) = [viscera \ weight(g)/whole \ body \ weight(g)] \times 100$ 

 $HSI(\%) = [wet weight of liver(g)/final weight of fish(g)] \times 100$ 

#### 2.5. Sample collection

#### 2.5.1. Liver

Five fishes were randomly selected in each tank and sacrificed for its liver sample collection. Data for the body weight and liver weight were recorded individually according to each sacrificed fish. A total of 1.0 g of each liver was homogenized in 10ml buffer containing 25mM sodium phosphate buffer (pH 7.4), 0.1 mM protease inhibitor, 1.0 mM ethylenediaminetetraacetic acid (EDTA), 0.1 mM dithiothreitol (DTT) and 0.1 phenylthiourea (PTU). The sample was then homogenized using a laboratory homogeniser at 150 rpm for 2 minutes until a distributed uniform liquid was achieved. Later, the homogenates were centrifuged at 100,000 rpm (Beckman Coulter Optima L-100K Ultracentrifuge), at 4°C for 30 minutes and the supernatants were separated and later stored at -80°C for further analysis.

#### 2.5.2. Blood and serum

Blood samples were collected randomly from five fishes in each tank through the caudal puncture. Approximately 1 ml of the blood sample was collected by using a 1-ml syringe with a needle size of 22G 1  $\frac{1}{2}$  inch from each fish.. For each individual fish, 0.5 ml of the blood was transferred to a heparinised vacutainer tube for haematological analysis (complete blood counts) and the remaining 0.5 ml was transferred into a non-heparinised serum vacutainer tube and later centrifuged at 5000 x g at 4°C for 10 min, according to the method described by Zhang et al. (2019). The five blood samples from each replicate were pooled together. The isolated serum was then stored at -20°C for further analysis.

#### 2.6. Analytical methods

#### 2.6.1. Proximate composition

Proximate composition of the feed ingredients (Table 2), experimental diets (Table 1) and fish muscle were determined according to the Standard Method of AOAC (2005). Crude protein was analysed by using the semi-auto Kjeldahl system (N  $\times$  6.25) (Kjeltec semi auto-analyzer). Crude lipid was determined by Soxhlet extraction method using petroleum ether as the solvent. Moisture was measured by drying the sample in an oven at 105 °C overnight and later continue in a muffle furnace for ash content at 600 °C for 6 h. The crude fibre was analyzed after acid and alkali digestion. Carbohydrate or NFE data were obtained through calculation [100 - (% crude protein + % crude lipid + % crude fiber + % crude ash)] and gross energy was determined by using the physiological values (crude protein x 23.9) + (crude lipid x 39.8) + (carbohydrate x 17.6) according to Schulz et al. (2005).

#### 2.6.2. Antioxidant enzyme assay

Glutathione S-transferase (GST) was analysed spectrophotometrically by measuring the activity towards 1-Chloro-2,4-dinitrobenzene (CDNB) as a substrate at 340 nm as described by the method from Habig et al. (1974). Catalase activity (CAT) was measured according to the decline in  $H_2O_2$  at 240 nm according to Claiborne (1985). Both antioxidant enzyme activities were expressed in nmol/min/mL.

#### 2.6.3. Haematology and serum protein analysis

Complete blood counts (CBC) which include red blood cells (RBC), white blood cells (WBC), haemoglobin (HGB), haematocrit (HCT) and RBC indices namely: mean corpuscular volume (MCV), mean corpuscular haemoglobin (MCH), and mean corpuscular haemoglobin concentration (MCHC) were measured using automatic haematology analyser (Sysmex XN, Germany). Serum protein analysis was measured by using Advia 2400 Chemistry System (Siemens Healthineers, Germany)

#### 2.7. Statistical analysis

The data were analysed by one-way analysis of variance (ANOVA) followed by Duncan's multiple range test to compare the means between individual treatments using SPSS software version 23 (SPSS Inc., Chicago IL. USA). Data were expressed as mean  $\pm$  standard deviation (SD) values and differences were considered statistically significant at *P* < 0.05.

#### 3. Results

#### 3.1. Growth performance, feed efficiency and organosomatic indices

At the end of the 6-week trial, the effect of different EFF on growth performance and feed utilisation in red hybrid Tilapia is shown in Table 2. Precise observation of fish morphological measured bi-weekly was characterized in Figure 1. The groups treated with 2E (2 g/kg EFF) and 3E (3 g/kg EFF) had significantly higher WG and SGR compared with those fed with control although no significant difference was observed in 1E (1 g/kg EFF) group. In addition, FCR, FI and SR did not affect all the experimental diets (P > 0.05) throughout the feeding trial. Based on the organosomatic indices results, HSI and VSI were significantly affected by the supplementation of EFF. Groups with either high EFF (2 and 3 g/kg) and control shows significantly lower VSI value was recorded in 3E (3 g/kg EFF). In contrast, significantly lower VSI value was recorded in 3E (3 g/kg EFF) fed fish compared to other diets (P < 0.05). However, the supplementation of EFF did not influence the CF of red hybrid Tilapia during the experimental period.

#### 3.2. Oxidative stress of experimental fish

The GST activity in liver of red hybrid Tilapia was found to be significantly higher (P < 0.05) in 3 g/kg of EFF (38.17 nmol/min/ml)

0.05

0.14

0.12

0.25

0.06

0.04

0.18

### Table 2 Growth performance of the red hybrid Tilapia treated with EPS fish feed (EFF) originated from the mycelium of Ganoderma lucidum.

SGR (%/day)

FI (g/fish)

FCR

Performance <sup>1</sup>	Experimental EPS f					
Control <sup>5</sup>	1 g/kg	2 g/kg	3 g/kg	F	P-value	
Initial Weight (g/fish)	17.45 ± 1.48	16.20 ± 0.14	15.40 ± 1.41	15.65 ± 1.91	0.85	0.54
Final weight (g/fish)	$28.36 \pm 0.22^{a}$	$32.75 \pm 0.35^{ab}$	33.75 ± 2.21 <sup>ab</sup>	34.40 ±1.88 <sup>b</sup>	3.58	0.13
BWG(g/fish) <sup>4</sup>	$11.87 \pm 0.64^{a}$	$16.55 \pm 0.21^{ab}$	$18.37 \pm 0.80^{b}$	$18.75 \pm 0.03^{b}$	12.00	0.02

 $1.88 \pm 0.06^{b}$ 

 $1.32 \pm 0.06$ 

 $24.24 \pm 2.13$ 

 $1.89 \pm 0.16^{b}$ 

 $1.31~\pm~0.14$ 

 $24.61 \pm 2.60$ 

7.10

3.19

1.25

3.27

5.67

3.23

1.80

SR (%)  $100\,\pm\,0.00$ 93.33 ± 6.70  $100\,\pm\,0.00$  $96.67 \pm 3.33$ Organosomatic indices<sup>2</sup>  $1.97 \pm 0.14^{b}$  $2.27 \pm 0.40^{\circ}$ HSI (%)  $1.42 + 0.51^{a}$  $2.22 + 0.45^{\circ}$  $10.71 \pm 1.18^{ab}$ VSI (%)  $11.02 \pm 0.99^{b}$  $9.75 \pm 1.06^{ab}$  $9.42 \pm 0.87^{a}$ CF (%)  $1.60 \pm 0.19$  $1.51 \pm 0.08$  $1.48\,\pm\,0.04$  $1.46\,\pm\,0.05$ 

 $1.60 \pm 0.11^{ab}$ 

 $1.72\,\pm\,0.23$ 

 $26.25 \pm 0.98$ 

<sup>3</sup>Mean values in the same row with different superscripts are significantly different (P < 0.05)

<sup>1</sup> The results represent mean± SD of 15 fishes per tank (\*14 fishes for control) on growth performance

 $^2\,$  The results represent mean± SD of 3 fishes per tank on organosomatic indices

 $1.16 \pm 0.22^{a}$ 

 $2.00 \pm 0.05$ 

 $23.78 \pm 0.60$ 

<sup>4</sup> BWG: Body Weight gain;, SGR: Specific growth rate, FCR: Feed conversion ratio, SR: Survival rate, FI: feed intake, HSI: Hepatosomatic index, VSI: Viscerosomatic index, CF: Condition factor

<sup>5</sup> Feed: Control = 0.0 g/kg of EPS



Fig. 1. Bi-weekly weight gain of red hybrid Tilapia (Oreochromis sp.) treated with different concentration of EPS fish feed (EFF) originated from the mycelium of Ganoderma lucidum in 6-week feeding trial.

(Table 3). On the other hand, no significant difference was detected in the activity of dietary 1E (1 g/kg EFF) and 2E (2 g/kg EFF) compared to control (P > 0.05). Similar trend was observed in 3E (3 g/kg EFF) with significantly higher CAT activity (230.51 nmol/min/ml) compared to other diets (P < 0.05).

diets while control was recorded as the lowest group. Other nutrient components including crude protein, crude lipid, dry matter, ash and gross energy did not have a significant effect amongst the treatment (P > 0.05). In contrast, 1E (1 g/kg EFF) diet had a significantly lower level (P < 0.05) of carbohydrate compared to other EFF. Control EFF did not differ significantly (P > 0.05) compared to 2E (2 g/kg EFF) and 3E (3 g/kg EFF) in terms of carbohydrate muscle content.

#### 3.3. Fish muscle nutritional composition

The fish muscle composition treated with EFF after 42-day of feeding trial is presented in Table 4. The data revealed that only fibre was significantly affected by EFF treatments. The fibre content of fish treated with 2 g/kg EFF is comparatively higher (P < 0.05) compared to other

#### 3.4. Haematological indices

Haemoglobin (Hb) and RBC of red hybrid Tilapia were affected by experimental EFF and elevated significantly (P < 0.05) in a linear fashion

#### Table 3

Antioxidant responses of red hybrid Tilapia (Oreochromis sp.) treated with different concentration of EPS fish feed (EFF) originated from the mycelium of Ganoderma lucidum after 42 days.

Antioxidant enzymes (nmol/min/mg protein) <sup>4</sup>	Experimental EPS feed					
Control <sup>3</sup>	1 g/kg	2 g/kg	3 g/kg	F	P-value	
Glutathione-S-transferase Catalase	$\begin{array}{l} 20.77  \pm  0.99^{a} \\ 117.81  \pm  7.24^{a} \end{array}$	$11.18 \pm 1.14^{a}$ 176.73 ± 7.99 <sup>b</sup>	$16.91 \pm 1.03^{a}$ 225.90 ± 2.17 <sup>c</sup>	$38.17 \pm 6.85^{b}$ $230.51 \pm 13.04^{c}$	21.52 75.99	0.006 0.001

<sup>1</sup>The results represent mean $\pm$  SD of 5 fishes per tank

<sup>2</sup>Mean values in the same row with different superscripts are significantly different (P < 0.05)

<sup>3</sup> Feed: Control = 0.0 g/kg of EPS

<sup>4</sup> Specific activity of enzyme measured in nmol/min/mg protein.

#### Table 4

Fish muscle composition of red hybrid Tilapia (Oreochromis sp.) treated with different concentration of EPS fish feed (EFF) originated from the mycelium of Ganoderma lucidum after 42 days.

Components (%)	Experimental fish fe	xperimental fish feed <sup>3,4</sup>				
Control <sup>5</sup>	1 g/kg	2 g/kg	3 g/kg	F	P-value	
Crude protein	81.87 ± 1.73	88.70 ± 2.01	83.80 ± 3.13	83.21 ± 1.47	3.75	0.12
Crude lipid	$1.50 \pm 0.09$	$1.68 \pm 0.11$	1.37 ± 0.25	$1.45 \pm 0.33$	0.72	0.59
Crude fiber	$0.03 \pm 0.00^{a}$	$0.20 \pm 0.00^{bc}$	$0.22 \pm 0.08^{\circ}$	$0.06 \pm 0.06^{ab}$	7.18	0.04
Dry matter (wet weight basis)	$24.20 \pm 0.14$	$24.40 \pm 0.11$	$25.40 \pm 0.20$	$25.70 \pm 0.12$	0.42	0.53
Ash	$6.60 \pm 0.41$	$6.93 \pm 0.63$	$5.90 \pm 0.89$	6.55 ± 1.06	0.79	0.56
Carbohydrate <sup>1</sup>	$10.00 \pm 1.32$	2.50 ± 2.38	8.70 ± 3.66	8.73 ± 2.89	3.14	0.15
Gross energy (kJ/g) <sup>2</sup>	$21.93 \pm 0.22$	$22.52 \pm 0.26$	$22.11 \pm 0.20$	$22.00~\pm~0.03$	3.73	0.12

 $^1$  Carbohydrate % = 100 – (crude protein % + crude lipid % + ash % + crude fiber %).

<sup>2</sup> Gross energy was calculated as 23.9 kJ/g, 39.8 kJ/g, and 17.6 kJ/g for protein, lipid, and carbohydrate, respectively

 $^3$  The result represents the mean  $\pm$  SD of 3 fishes per tank

<sup>4</sup> Mean values in the same row with different superscripts are significantly different (P < 0.05).

<sup>5</sup> Feed: Control = 0.0 g/kg of EPS

#### Table 5

Haematological parameters of red hybrid Tilapia (*Oreochromis* sp.) treated with different concentration of EPS fish feed (EFF) originated from the mycelium of *Ganoderma lucidum* after 42 days.

$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Parameters <sup>3</sup>	Experimental diets1,2	2				
HGB (g/dl) $5.75 \pm 0.70^3$ $6.98 \pm 0.10^b$ $7.08 \pm 0.51^b$ $7.43 \pm 0.22^b$ $10.18$ $0.001$ HCT (%) $30.00 \pm 3.46^a$ $36.00 \pm 1.41^b$ $39.00 \pm 2.31^b$ $37.50 \pm 0.58^b$ $12.66$ $0.001$ RBC ( $10^6/mm^3$ ) $2.04 \pm 0.29^a$ $2.47 \pm 0.14^b$ $2.57 \pm 0.24^b$ $2.69 \pm 0.02^b$ $7.92$ $0.004$ WBC ( $10^3/mm^3$ ) $133.88 \pm 27.84$ $155.20 \pm 6.64$ $161.63 \pm 36.34$ $176.65 \pm 21.99$ $1.92$ $0.18$ MCV (fl) $148.00 \pm 3.46^{bc}$ $145.75 \pm 4.35^b$ $152.50 \pm 4.04^c$ $139.50 \pm 1.29^a$ $9.57$ $0.002$ MCH (pg) $28.20 \pm 0.68$ $28.28 \pm 0.94$ $27.83 \pm 4.52$ $27.65 \pm 0.62$ $0.07$ $0.98$	Control <sup>4</sup>	1 g/kg	2 g/kg	3 g/kg	F	P-value	
MCHC (g/L)         190./5 $\pm 2.22$ 193./5 $\pm 4.35$ 182.25 $\pm 24.84$ 198.50 $\pm 2.89$ 1.15         0.37           Serum total protein (g/dl) $3.20 \pm 0.12$ $2.75 \pm 0.40$ $3.23 \pm 0.10$ $3.03 \pm 0.26$ $3.01$ $0.07$	HGB (g/dl) HCT (%) RBC (10 <sup>6</sup> /mm <sup>3</sup> ) WBC (10 <sup>3</sup> /mm <sup>3</sup> ) MCV (fl) MCH (pg) MCHC (g/L) Serum total protein (g/dl)	$\begin{array}{c} 5.75 \pm 0.70^{a} \\ 30.00 \pm 3.46^{a} \\ 2.04 \pm 0.29^{a} \\ 133.88 \pm 27.84 \\ 148.00 \pm 3.46^{bc} \\ 28.20 \pm 0.68 \\ 190.75 \pm 2.22 \\ 3.20 \pm 0.12 \end{array}$	$\begin{array}{c} 6.98 \pm 0.10^{b} \\ 36.00 \pm 1.41^{b} \\ 2.47 \pm 0.14^{b} \\ 155.20 \pm 6.64 \\ 145.75 \pm 4.35^{b} \\ 28.28 \pm 0.94 \\ 193.75 \pm 4.35 \\ 2.75 \pm 0.40 \end{array}$	$\begin{array}{c} 7.08 \pm 0.51^{\rm b} \\ 39.00 \pm 2.31^{\rm b} \\ 2.57 \pm 0.24^{\rm b} \\ 161.63 \pm 36.34 \\ 152.50 \pm 4.04^{\rm c} \\ 27.83 \pm 4.52 \\ 182.25 \pm 24.84 \\ 3.23 \pm 0.10 \end{array}$	$\begin{array}{c} 7.43 \pm 0.22^{b} \\ 37.50 \pm 0.58^{b} \\ 2.69 \pm 0.02^{b} \\ 176.65 \pm 21.99 \\ 139.50 \pm 1.29^{a} \\ 27.65 \pm 0.62 \\ 198.50 \pm 2.89 \\ 3.03 \pm 0.26 \end{array}$	10.18 12.66 7.92 1.92 9.57 0.07 1.15 3.01	0.001 0.001 0.004 0.18 0.002 0.98 0.37 0.07

<sup>1</sup> The result represents mean  $\pm$  SD of 5 fishes per tank

<sup>2</sup> Mean values in the same row with different superscripts are significantly different (p< 0.05).

<sup>3</sup> HGB = Haemoglobin; HCT = hematocrit; RBC = red blood cell; WBC = white blood cell; MCV = mean corpuscular volume; MCH = mean corpuscular haemoglobin; MCHC = mean corpuscular haemoglobin concentration.

<sup>4</sup> Feed: Control = 0.0 g/kg of EPS

as the supplementation level EFF increased from 0 (control) to 3 g/kg (Table 5). Similarly, a significant higher level of HCT was observed in experimental diets compared to control. White blood cell (WBC) count, MCH and MCHC values of fish did not vary significantly (P > 0.05) among different treatments, albeit discernible improvement was seen in WBC as the EFF increased. Nevertheless, the MCV was significantly lower in high EFF (3E, 3 g/kg) than other diets (P < 0.05). For biochemical serum total protein, no significant differences were observed between all treatments.

#### 4. Discussion

The emergence of bacterial antibiotic-resistance due to their excessive use has stimulated research and development to elucidate alternative eco-friendly compounds from fungal products. In addition to been considered inexpensive, safe and biodegradable, use of fungal polysaccharides (such as  $\beta$ -glucans) are unlikely to influence bacterial resistance to front-line antibiotics that is reached a crisis-point, according to the World Health Organisation (Reverter et al., 2014; WHO, 2019). One of the most promising medicinal mushroom, *G. lucidum*, has demonstrated potential for human and animal use by way of improved immune function: however, these bioactives have not been developed or tested as potentially essential aquaculture-based protein for farmed fish. Moreover, recent adjacent findings have shown that exopolysacharides, extracted from the same *Ganoderma lucidum*, is considered as non-toxic, as demonstrated by use zebrafish embryo toxicity (ZFET) assay: thus, inferring it's safe use in aquaculture feed (Taufek et al., 2020).

Aligned with previous findings, the results of this preliminary study revealed that administration of EPS from the mycelium of *G. lucidum* improved growth performance and feed efficiency, antioxidant activities and haematological indices in red hybrid Tilapia. To the best of our knowledge, this constitutes the first study to examine the performance of EPS from the fermented mycelium of *G. lucidum* as a feed supplement for red hybrid Tilapia. Previous published studies were limited to using mature fruiting bodies of *G. lucidum* for animal feed trials including fish (*Ctenopharyngodon Idella*) (Chithra et al., 2016), prawn (Mohan et al., 2016) and poultry (Bederska-Łojewska et al., 2017; Khan et al., 2019). Previous study by Chitra et al, (2016) indicated that administration of 1.0g/kg of *Ganoderma lucidum* polysaccharides regulate better performance of grass carp. Hence in the current study, the concentration of EPS was observed at 1, 2 and 3 g/kg diet. This study demonstrated that the supplementation of 2 to 3 g/kg EPS in red hybrid Tilapia diet provide better growth when observed in terms of WG, SGR and organosomatic indices of HSI. Moreover, improved FCR values were also observed in these supplemented diets when compared to control, which demonstrates the ability of EPS to promote fish growth.

Antioxidant activity has been regarded as an essential measure to detect fish well-being. Glutathione S-transferase (GST) plays a critical role in protecting cells in terms of toxicity of xenobiotics electrophiles and from a foreign contaminants perspective (Dasari et al., 2018). In aquatic animals, significant alteration of GST activities will reflect the metabolic disturbance and cell damage in specific organs (Mohan et al., 2015; Raji et al., 2018). The increasing level of GST in any material implies that it may contain a compound that could trigger the biotransformation of xenobiotics. Our findings suggest that incorporation of EPS did not differ significantly in between control, 1E and 2E although 1E diet exhibited numerically lower activities than other treatments.

As the first line of defence mechanism, catalase (CAT) is correlated with an increasing concentration of H<sub>2</sub>O<sub>2</sub> (Wilhelm Filho et al., 2005). In contrast to GST, high CAT activity was observed in EPS supplemented groups when compared to control, whereby group fed with 3E shows significantly higher activity than the others. This might corresponds to the previous study by Wu et al. (2016) which reported the presence of ß-1,3/1,6 glucan polysaccharides in cell wall of G.lucidum. The CAT activity trend is in accordance to the growth performance analysis where high SGR and WG, as well as numerically low FCR value, was observed in the group fed with 3E, which might be attributed to high metabolic rate (Ahmed et al., 2017; Taufek et al., 2016). Ahmed et al. (2017) investigated the effect of hot water extract (HWE) of the waste mushroomstalk (Pleurotus sp.) on CAT activity of tilapia. They found a significant increase in CAT activity in 0.5% and 1% of HWE when compared to control, which is contributed by the presence of EPS. Oxidative stress response findings reported in this current study will inform new innovation in terms of fish supplement. Consequently, there are pressing opportunities that include potential immediate issues in fish farming from farm to fork, that focues on reduced environmental stress conditions.

Fish fillet is the edible part of the fish that is expected to contain high nutritional value. Consequently, the nutritional composition of fish fillet was addressed in this study with a particular role on investigating the influence of EPS supplementation on key determinants. Specifically, high protein content was observed in all EPS-supplemented fish. This contrasts to previous studies that focused on red tilapia body composition where fish were fed a diet supplemented by 10% with mushroom stalk meal (*Pleurotus sajor-caju*) (Abdul Rahman Jabir et al., 2012). Other researchers have also reported that the lipid content of fish fed the experimental diets (1.37 - 1.68 %) were higher than Nile tilapia that were fed supplemented diets containing red algae (*Gracilaria arcuata*) (Younis et al., 2018).

Haematological parameters are essential tools in understanding the physiological status of experimental fish. Overall, the HGB, HCT, and RBC had significant effects on all EPS group as compared to control, whereby the highest level was detected in 3E group. It was observed that increasing RBC is correlated with the increasing trend of HGB and HCT due to their synergistic linkage (Audu et al., 2014). Also, EPS-treated groups produced higher WBC despite no significant differences were found amongst treatments, which corresponds to the trend of RBC as immunity indicator. A similar trend was observed in the study made by Chitsaz et al. (2018) during supplementation of a medic-

inal mushrooms extract (*Lentinula edodes*,) in great sturgeon juvenile (*Huso huso*). Increasing in RBC and WBC values were also reported by Mohan et al. (2019b) in shrimp fed with exopolysaccharides. Red blood cell indices are mean corpuscular haemoglobin (MCH) and mean corpuscular volume (MCV) which used to diagnose the level of anaemia (Şahan, Duman, 2010). The maturation of red blood cells determined the value of MCV and MCH. The unstable trend was observed in MCV of fish in all treatments suggesting the presence of immature red blood cells which have a larger cell volume and lower haemoglobin (Carvalho, Fernandes, 2006).

The findings of this present study correlate well with the trend towards the use of prebiotics and probiotics in feed supplements for aquaculture production (Akhter et al., 2015; Carballo et al., 2019; Dimitroglou et al., 2011; Galankis et al., 2020). This also reflects a trend towards the use of natural processes for disease mitigation in aquaculture and adjacent food production processes for intensive sustainability (Rowan, 2019). Use of probiotics and prebiotics are normally non-digestable molecules that selectively modulate the intestinal micobiome by promoting indigenous microbial populations and by preventing pathogen proliferation (Carballo et al., 2019). Recent findings have demonstrated that certain carbohydrates (exopoly and oligosaccharides) as promising prebiotics for animal and human health including betaglucans that are fermented in the gut resulting in enhanced production of short chain fatty acids (SCFA) such as formate, acetate, proponate and butyrate (Carballo et al., 2019; Mahdhi et al., 2020; Morrison, Preston, 2016). These secondary metabolites also modulate production and release of cytokines, chemokines and leukocyte recruitment for health benefits (Morrison, Preston, 2016). Use of beta-glucans derived from medicinal fungi, as described in this present study, shows potential as an immune-priming feed supplement for aquaculture production. However, future studies focusing on interaction of beta-glucans derived from medicinal fungi with gut microbiota in fish is merited.

#### Conclusion

EPS fish-feed formulation, based on the mycelium of Ganoderma lucidum, exerted potent benefits for red hybrid Tilapia growth performance and improved antioxidant activity of GST and CAT enzymes as well as haematological indices of RBC, HCT and HGB. Nutritional muscle composition was minorly affected by EFF supplementation. Findings from this preliminary study highlight the potential benefits of using EFF in fish feed supplementation to enhance the health and immunity of Tilapia and other farmed fish along with adjacent scale-up studies for commercial aquaculture deployment. This timely study has significant implications for the smart intensification of aquaculture through innovative steps in feed formulation, which will provide vital future food for meeting the pressing dietary needs of our growing global populations. Developing sustaining and disruptive technologies in the the food sector will also enable pivoting of society beyond recession created by ongoing COVID-19 (Rowan and Galanakis, 2020) through provision of safe and nutritious supply chain that includes health benefits (Galanakis et al., 2021). Transnational modelling of key socio-economic drivers informing sustainable development of marine industry, enabled by life cycle assessment of products and services (Ruiz-Salmón et al., 2020), will inform policy that includes Europe's Green New Deal (Rowan and Galanakis, 2020). This transdisciplinary study has significant implications for the smart sustainable intensification of aquaculture through innovative steps in feed formulation, which will provide future safe nutritious food to meet pressing needs for our growing global populations.

#### **Declaration of Competing Interest**

The authors declare that they have no competing conflict of interest and the authors alone are responsible for the content and writing of the paper.

#### Acknowledgement

We want to thank the Ministry of Higher Education Malaysia on RU Geran-Fakulti Program Universiti Malaya (GPF084A-2020) to Dr Wan-Mohtar and for the Fundamental Research Grant Scheme (FRGS FP066-2018) awarded to Dr Norhidayah. The work was also sponsored in part by collaborative H2020 MSCA RISE ICHTHYS project [number 872217/19] Prof Neil Rowan.

#### References

- Abdul Rahman Jabir, M.D., Razak, S.A., Sabaratnam, V., 2012. Effect of mushroom supplementation as a prebiotic compound in super worm based diet on growth performance of red tilapia fingerlings. Sains Malaysiana 41, 1197–1203.
- Ahmed, M., Abdullah, N., Shuib, A.S., Abdul Razak, S., 2017. Influence of raw polysaccharide extract from mushroom stalk waste on growth and pH perturbation induced-stress in Nile tilapia. Oreochromis niloticus. Aquaculture. 468, 60–70.
- Akhter, N., Wu, B., Memon, A.M., Mohsin, M., 2015. Probiotics and prebiotics associated with aquaculture: a review. Fish & Shellfish Immunol. 45, 733–741.
- Amal, M.N.A., Koh, C.B., Nurliyana, M., Suhaiba, M., Nor-Amalina, Z., Santha, S., Diyana-Nadhirah, K.P., Yusof, M.T., Ina-Salwany, M.Y., Zamri-Saad, M., 2018. A case of natural co-infection of Tilapia Lake Virus and Aeromonas veronii in a Malaysian red hybrid tilapia (Oreochromis niloticus×O. mossambicus) farm experiencing high mortality. Aquaculture 485, 12–16.
- Audu, B.S., Adamu, K.M., Nonyelu, O.N., 2014. Changes in haematological parameters of Clarias gariepinus exposed to century plant (Agave Americana) Leaf Dust.
- Baba, E., Uluköy, G., Öntaş, C., 2015. Effects of feed supplemented with Lentinula edodes mushroom extract on the immune response of rainbow trout, Oncorhynchus mykiss, and disease resistance against Lactococcus garvieae. Aquaculture 448, 476–482.
- Bederska-Lojewska, D., Świątkiewicz, S., Muszyńska, B., 2017. The use of Basidiomycota mushrooms in poultry nutrition—A review. Anim. Feed Sci. Technol. 230, 59–69.
   Cao, Y., Xu, X., Liu, S., Huang, L., Gu, J., 2018. Ganoderma: a cancer immunotherapy
- review. Front. Pharmacol. 9. Carballo, C., Pinto, P.I.S., Mateus, A.P., Berbel, C., Guerreiro, C.C., Martinez-Blanch, J.F., Codoñer, F.M., Mantecon, L., Power, D.M., Manchado, M., 2019. Yeast β-glucans and
- microalgal extracts modulate the immune response and gut microbiome in Senegalese sole (Solea senegalensis). Fish & Shellfish Immunol. 92, 31–39. Carvalho, C.S., Fernandes, M.N., 2006. Effect of temperature on copper toxicity and hema-
- tological responses in the neuropical fish Prochilodus scrofa at low and high pH. Aquaculture 251, 109–117.
- Chithra, E., Padmanaban, A., Mohan, K., 2016. Potential use of Ganoderma lucidum polysaccharides as a feed supplement in diets on survival and growth performance of the grass carp, Ctenopharyngodon idella. Int. J. Fish. Aquac. Stud. 4, 328–333.
- Chitsaz, H., Akrami, R., Ahmadi, Z., 2018. Effects of mushroom (Lentinula edodes) extract on growth performance, immune response and hemato-biochemical parameters of great sturgeon juvenile (Huso huso Linnaeus, 1754). Iran. J. Aquat. Anim. Health 4, 29–48.
- Claiborne, A., 1985. Catalase activity. In: CRC Handbook of Methods for Oxygen Radical Research, 1, pp. 283–284.
- Dasari, S., Ganjayi, M.S., Meriga, B., 2018. Glutathione S-transferase is a good biomarker in acrylamide induced neurotoxicity and genotoxicity. Interdiscip. Toxicol. 11, 115.

Dimitroglou, A., Merrifield, D.L., Carnevali, O., Picchietti, S., Avella, M., Daniels, C., Güroy, D., Davies, S.J., 2011. Microbial manipulations to improve fish health and production–a Mediterranean perspective. Fish Shellfish Immunol. 30, 1–16.

- DOF, 2018. Perangkaan Perikanan Tahun 2018. Department of Fisheries Malaysia, https://www.dof.gov.my/dof2/resources/user\_29/Documents/Perangkaan% 20Perikanan/2018%20Jilid%201/5.jadual\_akuakultur\_2018.pdf.
- Elumalai, P., Prakash, P., Musthafa, M.S., Faggio, C., 2010. Effect of alkoxy glycerol on growth performance, immune response and disease resistance in Nile Tilapia (Oreochromis niloticus). Res. Vet. Sci. 123, 298–304.
- FAO, 2018. The State of World Fisheries and Aquaculture Meeting the sustainable development goals. The State of World Fisheries and Aquaculture, FAO, Rome, Italy, p. 227.
- Galanakis, C.M., Aldawood, T.M.S., Rizou, M., Rowan, N.J., Ibrahim, S.A, 2020. Food ingredients and active compounds against the coronavirus disease (COVID-19) pandemic: a comprehensive review. Foods. 9 (11), 1701 doi:10.3390/foods9111701.
- Gananakis, C.M.Rizou, M., Aldawoud, T.M.S., Ucak, I., Rowan, N.J., 2021. Innovations and technology disruptions in the food sector within the COVID-19 pandemic and post-lockdown era.. Journal of Applied Microbiology 110, 193–200 doi:10.1016/j.tifs.2021.02.002.
- Habig, W.H., Pabst, M.J., Jakoby, W.B., 1974. Glutathione S-transferases. The first enzymatic step in mercapturic acid formation. J. Biol. Chem. 249, 7130–7139.
- Hassan, N.A., Supramani, S., Sohedein, M.N.A., Usuldin, S.R.A., Klaus, A., Ilham, Z., Chen, W.-H., Wan, W.A.A.Q.I., 2019. Efficient biomass-exopolysaccharide production from an identified wild-Serbian Ganoderma lucidum strain BGF4A1 mycelium in a controlled submerged fermentation. Biocatal. Agric. Biotechnol. 21, 101305.
- Jansen, M.D., Dong, H.T., Mohan, C.V., 2019. Tilapia lake virus: a threat to the global tilapia industry? Rev. Aquacult. 11, 725–739.
- Khan, S.H., Mukhtar, N., Iqbal, J., 2019. Role of mushroom as dietary supplement on performance of poultry. J. Dietary Suppl. 16, 611–624.
- Kozarski, M., Klaus, A., Jakovljevic, D., Todorovic, N., Wan-Mohtar, W.A.A.Q., Niksic, M., 2019. Ganoderma lucidum as a cosmeceutical: Antiradical potential and inhibitory

effect on hyperpigmentation and skin extracellular matrix degradation enzymes. Arch Biol. Sci. 71, 253–264.

- Mahdhi, A., Chakroun, I., Espinosa-Ruiz, C., Messina, C.M., Arena, R., Majdoub, H., Santulli, A., Mzoughi, R., Esteban, M.A., 2020. Dietary administration effects of exopolysaccharide from potential probiotic strains on immune and antioxidant status and nutritional value of European sea bass (Dicentrarchus labrax L.). Res. Vet. Sci. 131, 51–58.
- Masterson, C.H., Murphy, E.J., Gonzalez, E., Major, I., McCarthy, S.D., O'Toole, D., Laffey, J.G., Rowan, N.J., 2020. Purified *β*-glucans from the Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumoniae-induced pulmonary sepsis. Lett Applied Microbiol 71 (4), 405–412.
- Meng, L.-Z., Xie, J., Lv, G.-P., Hu, D.-J., Zhao, J., Duan, J.-A., Li, S.-P., 2014. A Comparative study on immunomodulatory activity of polysaccharides from two official species of ganoderma (Lingzhi). Nutr. Cancer 66, 1124–1131.
- Mohan, K., Padmanaban, M., Uthayakumar, V., 2015. Ganoderma lucidum (Higher Basidiomycetes. Am. J. Biol. Life Sci. 3, 168–175.
- Mohan, K., Muralisankar, T., Uthayakumar, V., Chandirasekar, R., Karthick Rajan, D., 2019a. Dietary Ganoderma lucidum polysaccharides to enhance the growth, immune response and disease resistance of freshwater prawn Macrobrachium rosenbergii. Aquacult. Rep. 14, 100203.
- Mohan, K., Padmanaban, A.M., Uthayakumar, V., Chandirasekar, R., Muralisankar, T., Santhanam, P., 2016. Effect of dietary Ganoderma lucidum polysaccharides on biological and physiological responses of the giant freshwater prawn Macrobrachium rosenbergii. Aquaculture 464, 42–49.
- Mohan, K., Ravichandran, S., Muralisankar, T., Uthayakumar, V., Chandirasekar, R., Seedevi, P., Rajan, D.K., 2019b. Potential uses of fungal polysaccharides as immunostimulants in fish and shrimp aquaculture: a review. Aquaculture 500, 250–263.
- Morrison, D.J., Preston, T., 2016. Formation of short chain fatty acids by the gut microbiota and their impact on human metabolism. Gut Microbes 7, 189–200.
- Murphy, E., Masteron, C., Rezoagli, E., O'Toole, D., Major, I., Stack, G., Lynch, M., Laffey, J.G., Rowan, N.J. 2020. β-Glucan extracts from the same edible shiitake mushroom *Lentinus edodes* produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects — Implications for coronavirus disease (COVID-19) immunotherapies. Vol 732 https://doi.org/10.1016/j.scitotenv.2020.139330
- Murphy, E.J., Rezoagli, E., Major, I., Rowan, N.J., Laffey, J.G., 2020b. β-glucan metabolic and immunomodulatory properties and potential for clinical application. J. Fungi 6 (4), 356. doi:10.3390/jof6040356.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N. 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: A case study from the Republic of Ireland. https://doi.org/10.1016/j.aquaculture.2020.735377
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga *Pseudokirchner-iella subcapitata*, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218.
- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production–A case study from the Republic of Ireland. Sci. Total Environ. 706, 136044.
- Raji, A.A., Alaba, P.A., Yusuf, H., Abu Bakar, N.H., Taufek, N.M., Muin, H., Alias, Z., Milow, P., Abdul Razak, S., 2018. Fishmeal replacement with Spirulina Platensis and Chlorella vulgaris in African catfish (Clarias gariepinus) diet: Effect on antioxidant enzyme activities and haematological parameters. Res. Vet. Sci. 119, 67–75.
- Reverter, M., Bontemps, N., Lecchini, D., Banaigs, B., Sasal, P., 2014. Use of plant extracts in fish aquaculture as an alternative to chemotherapy: current status and future perspectives. Aquaculture 433, 50–61.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? Sci. Total Environ. doi:10.1016/j.scitotenv.2020.141362.
- Rowan, N.J., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries–Quo vadis?. Trends in food science & technology.
- Rowan, N.J., Smith, J.E., Sullivan, R., 2003. Immunomodulatory activities of mushroom glucans and polysaccharide–protein complexes in animals and humans (a review). Int. J. Med. Mushrooms 5.
- Rufchaei, R., Mirvaghefi, A., Hoseinifar, S.H., Valipour, A., Nedaei, S., 2019. Effects of dietary administration of water hyacinth (Eichhornia crassipes) leaves extracts on innate immune parameters, antioxidant defence and disease resistance in rainbow trout (Oncorhynchus mykiss). Aquaculture, 734533.
- Ruiz-Salmón, I., Margallo, M., Lazo, J., Villaneuva-Rey, P., Marino, D., Quinteiro, P., Dias, A.C., Nunes, M.L., Marques, A., Feijoo, G., Moreira, M.T., Loubet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N., Mendez-Paz, D., Iglesias-Parga, Z., Anglda, C., Martin, J-C., Irabien, A., Aldaco, R., 2020. Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Current Opin. Environ. Sci. Health 13, 101–106.
- Ruiz-Salmón, I., Lazo, J., Margallo, M., Villaneuva-Rey, P., Rodriguez, E., Quinteiro, P., Dias, A.C., Almeida, C., Nunes, M.L., Marques, A., Cortes, A., Moreira, A.T., Feijoo, G., Loubet, P., Sonnemann, G., Morse, A.P., Cooney, R., Clifford, E., Regueiro, L., Mendez, D., Anglada, C., Noiret, C., Rowan, N., Vazquez-Rowe, I., Aldcaco, R., 2021. Life cylce assessment of fish and seafood processed products- a review of methodologies and new challenges. Sci. Total Environ. 761 doi:10.116/j.scitotenv.2020.144094.
- Şahan, A., Duman, S., 2010. Influence of β-glucan applications on some non-specific cellular immune response and haematologic parameters of healthy Nile tilapia (Oreochromis niloticus L., 1758). Turk. J. Vet. Anim. Sci. 34, 75–81.
- Schulz, C., Knaus, U., Wirth, M., Rennert, B., 2005. Effects of varying dietary fatty acid profile on growth performance, fatty acid, body and tissue composition of juvenile pike perch (Sander lucioperca). Aquacult. Nutr. 11, 403–413.

- Smith, J.E., Rowan, N.J., Sullivan, R., 2002. Medicinal mushrooms: a rapidly developing area of biotechnology for cancer therapy and other bioactivities. Biotechnol. Lett. 24, 1839–1845.
- Sullivan, R., Smith, J.E., Rowan, N.J., 2006. Medicinal mushrooms and cancer therapy: translating a traditional practice into Western medicine. Perspect. Biol. Med. 49, 159–170.
- Supramani, S., Ahmad, R., Ilham, Z., Annuar, M.S.M., Klaus, A., Wan-Mohtar, W.A.A.Q.I., 2019a. Optimisation of biomass, exopolysaccharide and intracellular polysaccharide production from the mycelium of an identified <em>Ganoderma lucidum</em>strain QRS 5120 using response surface methodology. AIMS Microbiol. 5, 19–38.
- Supramani, S., Jailani, N., Ramarao, K., Mohd Zain, N.A., Klaus, A., Ahmad, R., Wan-Mohtar, W.A.A.Q.I., 2019b. Pellet diameter and morphology of European Ganoderma pfeifferi in a repeated-batch fermentation for exopolysaccharide production. Biocatal. Agric. Biotechnol. 19, 101118.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018a. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, e5281.
- Tahar, A., Kennedy, A., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018b. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3, 28.
- Taufek, N.M., Aspani, F., Muin, H., Raji, A.A., Razak, S.A., Alias, Z., 2016. The effect of dietary cricket meal (Gryllus bimaculatus) on growth performance, antioxidant enzyme activities, and haematological response of African catfish (Clarias gariepinus). Fish Physiol Biochem 42, 1143–1155.
- Taufek, N.M., Harith, H.H., Abd Rahim, M.H., Ilham, Z., Rowan, N., Wan-Mohtar, W.A.A.Q.I., 2020. Performance of mycelial biomass and exopolysaccharide from Malaysian Ganoderma lucidum for the fungivore red hybrid Tilapia (Oreochromis sp.) in Zebrafish embryo. Aquacult. Rep. 17, 100322.
- USDA, 2019. Emerging Risk to Animal Health Notice Tilapia Lake Virus (TiLV). United States Department of Agriculture, https://www.aphis.usda.gov/animal\_health/ downloads/import/tilv-ed-notice.pdf.
- Usuldin, S.R.A., Mahmud, N., Ilham, Z., Ikram, N.K.K., Ahmad, R., Wan-Mohtar, W.A.A.Q.I., 2019. In-depth spectral characterization of antioxidative (1,3)-β-D-glucan from the mycelium of an identified tiger milk mushroom Lig-

nosus rhinocerus strain ABI in a stirred-tank bioreactor. Biocatal. Agric. Biotechnol., 101455.

- Wan-Mohtar, W.A.A.Q.I.IIham, Z., Jamaludin, A.A. Rowan, N., 2021. Use of zebrafish embryo assay to evaluate toxicity and safety of bioreactor-grown exopolysaccharides and endopolysaccharides from European Ganoderma applanatum mycelium for future aquaculture applications. Int J Mol Sci 22 (4). doi:10.3390/ijms22041675.
- Wan-Mohtar, W.A.A.Q.I., Taufek, N.M., Yerima, G., Rahman, J., Thiran, J.P., Subramaniam, K., Sabaratnam, 2020. Effect of bioreactor-grown biomass from *Ganoderma lucidum* mycelium on growth performance and physiological response of red hybrid tilapia (*Oreochromis* sp.) for sustainable aquaculture. Org. Agr. doi:10.1007/s13165-020-00303-5.
- Wan-Mohtar, W.A.A.Q.I., Viegelmann, C., Klaus, A., Lim, S.A.H., 2017. Antifungal-demelanizing properties and RAW264.7 macrophages stimulation of glucan sulfate from the mycelium of the mushroom Ganoderma lucidum. Food Sci. Biotechnol. 26, 159–165.
- WHO, 2019. New report calling for urgent action to avert antimicrobial resistance crisis. https://www.who.int/news/item/29-04-2019-new-report-calls-for-urgent-action-toavert-antimicrobial-resistance-crisis (accessed 27 November, 2020).
- Wilhelm Filho, D., Torres, M.A., Zaniboni-Filho, E., Pedrosa, R.C., 2005. Effect of different oxygen tensions on weight gain, feed conversion, and antioxidant status in piapara, Leporinus elongatus (Valenciennes, 1847). Aquaculture 244, 349–357.
- Wu, Y.-S., Ho, S.-Y., Nan, F.-H., Chen, S.-N., 2016. Ganoderma lucidum beta 1, 3/1, 6 glucan as an immunomodulator in inflammation induced by a high-cholesterol diet. BMC Compl. Altern. Med. 16, 500.
- Younis, E.-S.M., Al-Quffail, A.S., Al-Asgah, N.A., Abdel-Warith, A.-W.A., Al-Hafedh, Y.S., 2018. Effect of dietary fish meal replacement by red algae, Gracilaria arcuata, on growth performance and body composition of Nile tilapia Oreochromis niloticus. Saudi J. Biol. Sci. 25, 198–203.
- Yuan, B., Chi, X., Zhang, R., 2012. Optimization of exopolysaccharides production from a novel strain of Ganoderma lucidum CAU5501 in submerged culture. Braz. J. Microbiol. 43, 490–497.
- Zhang, J., Zhong, L., Peng, M., Chu, W., Liu, Z., Dai, Z., Hu, Y., 2019. Replacement of fish meal with soy protein concentrate in diet of juvenile rice field eel Monopterus albus. Aquacult. Rep. 15, 100235.

# scientific reports



# **OPEN** In vivo toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications

Siti Rokhiyah Ahmad Usuldin<sup>1,2</sup>, Wan Abd Al Qadr Imad Wan-Mohtar<sup>2,3,6,8</sup>, Zul Ilham<sup>3,4</sup>, Adi Ainurzaman Jamaludin<sup>4</sup>, Nur Raihan Abdullah<sup>5</sup> & Neil Rowan<sup>6,7,8</sup>

Natural mycelial biomass (MB) and exopolysaccharides (EPS) of Malaysian tiger milk mushroom Lignosus rhinocerus are considered high-end components due to their high commercial potential value in drug discovery. This study aims to evaluate the toxicity of the mushroom extracts' generated in a bioreactor using the zebrafish embryo toxicity (ZFET) model assay as a new therapy for treating asthma. Both MB and EPS extracts, at concentrations 0.16–10 mg/mL, were tested for ZFET and early development effects on Zebrafish Embryos (ZE) during 24–120 h post-fertilisation (HPF). Findings revealed that MB was deemed safe with an  $LC_{50}$  of 0.77 mg/mL; the EPS were non-toxic ( $LC_{50}$  of 0.41 mg/mL). Neither MB nor EPS delayed hatching nor teratogenic defects in the treated ZE at a 2.5 mg/mL dose. There were no significant changes in the ZE heart rate after treatments with MB (130 beats/min) and EPS (140 beats/min), compared to that of normal ZE (120–180 beats/min). Mixing both natural compounds MB and EPS did not affect toxicity using ZFET testing; thus, intimating their safe future use as therapeutic interventions. This represents the first study to have used the ZFET assay on MB and EPS extracts of L. rhinocerus for future health applications.

Mushrooms are filamentous fungi that have been used worldwide since prehistoric times<sup>1,2</sup>, and medically since at least 3000 BCE<sup>3,4</sup>. Mushrooms are heterotrophic organisms and have become increasingly important in research and industry<sup>5</sup>; food, medicines, cosmetics, detergents, and biofuels are examples of high-value products manufactured from fungi<sup>6,7</sup>. Furthermore, mushroom-derived extracts are becoming increasingly popular due to their potential usage in a wide range of vital health applications<sup>8,9</sup>. A focused interest in biorefining these products, such as via Green New Deal innovations from food waste streams, reflects a stronger emphasis on expanding 'circularity' and bioeconomy<sup>10-14</sup>. Mushrooms are classified in the kingdom of Fungi and have many active constituents, including, but possibly not limited to, polysaccharides, polysaccharide peptides, proteins, terpenoids, and nucleotides<sup>15</sup>. The most studied and used medicinally active ingredient in mushrooms is  $\beta$ -glucan. Previous research has revealed that  $\beta$ -glucans have broad metabolic and gastro-intestinal effects, including modulating the gut microbiome, altering lipid and glucose metabolism, and reducing cholesterol; thus, leading to the use of  $\beta$ -glucan as potential therapy for treating metabolic syndrome, obesity and diet regulation, gastrointestinal conditions, and to reduce the risk of cardiovascular and diabetes<sup>16,17</sup>.

The bioactive extracts derived from mushrooms can modulate the immune response affecting hematopoietic stem cells, lymphocytes, macrophages, T cells, dendritic cells, and natural killer cells. Murphy et al.<sup>18</sup> reviewed

<sup>1</sup>Agro-Biotechnology Institute, Malaysia (ABI), National Institutes of Biotechnology Malaysia (NIMB), c/o HQ MARDI, 43400 Serdang, Selangor, Malaysia. <sup>2</sup>Functional Omics and Bioprocess Development Laboratory, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, 50603 Kuala Lumpur, Malaysia. <sup>3</sup>Bioresources and Bioprocessing Research Group, Institute of Biological Sciences, Faculty of Sciences, Universiti Malaya, 50603 Kuala Lumpur, Malaysia. <sup>4</sup>Environmental Science and Management Program, Institute of Biological Sciences, Faculty of Science, Universiti Malaya, 50603 Kuala Lumpur, Malaysia. <sup>5</sup>Department of Biotechnology, Kulliyyah of Science, International Islamic University Malaysia, 25200 Kuantan, Pahang, Malaysia. <sup>6</sup>Bioscience Research Institute, Athlone Institute of Technology, Athlone, Ireland. <sup>7</sup>Empower Eco Innovation Hub, Boora, Co. Offaly, Ireland. <sup>8</sup>These authors contributed equally: Wan Abd Al Qadr Imad Wan-Mohtar and Neil Rowan. Zemail: nrowan@ait.ie

Scientific Reports | (2021) 11:23079

over 200 patents that highlighted the therapeutic potential of  $\beta$ -glucans; this is evidenced by the fact that two glucans were licensed in Japan as immune-adjuvant therapy for treating cancer. Moreover, the pronounced immune-modulatory effects of  $\beta$ -glucans<sup>18,19</sup> promoted their usage as adjuvant agents for treating cancers, immune-mediated conditions, rhinitis, respiratory infections, and to enhance wound healing<sup>16,20</sup>. However, further clinical testing and translation of  $\beta$ -glucans face significant challenges due to differences in source and extraction procedures<sup>16</sup>. We recently identified the active ingredients of *Lignosus rhinocerus* using 2D NMR analyses and reported on the antioxidative potential of (1,3)- $\beta$ -D-glucan as an essential constituent<sup>21</sup>. However, many other compounds extracted from medicinal mushrooms have yet to be named, which are often referred to by gel chromatography fraction<sup>15</sup>; thus, highlighting the need to conduct and report on their safety.

The tiger milk mushroom, *Lignosus rhinocerus*, belongs to the Basidiomycota section of the Polyporaceae family and is classified as a filamentous fungus<sup>22,23</sup>. *L. rhinocerus* was grown in submerged-liquid fermentation (SLF) using a laboratory-scale stirred-tank bioreactor to achieve bulk cultivation and commensurate production of polysaccharides<sup>21</sup>. When compared to solid-state fermentation (SSF), SLF has several advantages, including limited space requirements, ease of scale-up, reliable and reproducible processing, ease of monitoring, and versatility<sup>24</sup>. Artificially cultivated *L. rhinocerus* is also an excellent replacement in developing therapeutic items. For example, exopolysaccharides (EPS) isolated from mushroom mycelial biomass (MB) have pharmacological properties as immunomodulatory, anti-inflammatory, antibacterial, antiviral, and antioxidant activities<sup>25</sup>. Chen et al.<sup>26</sup> discovered that *L. rhinocerotis* mycelium grown in SLF does not cause mutagenicity or genotoxicity. The US Food and Drug Administration (FDA) standards, on the other hand, demand substantial proof of no hazard for commercial usage<sup>26</sup>.

Asthma affects 300 million individuals worldwide and is caused by a complex combination of inherited and environmental variables<sup>23</sup>. Allergic asthma is a long-term condition characterised by wheezing, shortness of breath, chest tightness, and coughing. In Malaysia, indigenous peoples have long used *L. rhinocerus* to treat asthma, while the majority of today's asthma medications are made up of steroids and other anti-inflammatory drugs<sup>23</sup>. Recent studies have reported on the efficacy of using EPS from medicinal mushrooms to ameliorate pro- and anti-inflammatory responses using ex vivo and in vivo infection models with therapeutic potential<sup>16,18,27</sup>. Aqueous extracts of *L. rhinocerotis* were reported to help reduce asthma-related variables in an asthma model<sup>28</sup>. In addition, a previous toxicity study indicated that feeding 1000 mg/kg of *L. rhinocerus* extract to rats had no detrimental consequences, hence it was considered safe<sup>29</sup>. As a result, more effective asthma treatment is required using *L. rhinocerus* as a helpful adjuvant or alternative to currently available asthma medications.

Zebrafish embryos have been extensively studied and documented as a reliable and popular model for developmental biology, toxicity, and, more recently, drug discovery<sup>30</sup>. Zebrafish may be readily bred, reared, and maintained in the laboratory<sup>31</sup>. Zebrafish embryos develop quickly, where they are fully developed five days after conception. Light microscopy can straightforwardly examine morphological structures and internal organs, such as the brain, eyes, heart, liver, and kidney due to the embryo's transparency. Dyes can be used to measure organ-specific and overall developmental toxicity visually or quantitatively. Due to its small size, a single Zebrafish embryo can be maintained in low fluid volumes for the first six days of development, including microtiter plates. The permeability of zebrafish embryos is prominent; for example, small chemicals added to fish water permeate the embryos, simplifying drug administration and assay processing<sup>32</sup>. Chemical screening can be completed after a few days due to the embryo's rapid development. The zebrafish is therefore a unique vertebrate model for high-throughput chemical screening, beneficial for pre-clinical drug discovery and toxicity assessment<sup>33,34</sup>.

A recent publication evaluating the toxicity of biomass-EPS comparable medicinal mushroom mycelial extracts revealed that the zebrafish embryo toxicity (ZFET) assay could be deployed as a safety screening approach before pre-clinical testing according to national and international standards<sup>35</sup>. Compared to human cell lines, research on the ZFET model is quick, resilient, efficient, and cost-effective for early development investigations; it also represents relevant genetic structure and equivalent critical organs and tissues<sup>36,37</sup>. Thus, this study aims to determine the toxicity of mushroom extracts using ZFET before they are developed and potentially commercialised as a new therapeutic intervention. To the best of our knowledge, there have been no toxicity studies using the ZFET model describing the use of MB and EPS of *L. rhinocerus* generated in the bioreactor.

As a result, this constitutes the second study to determine the toxicity levels of extracted MB and EPS from *L. rhinocerus* using a ZFET model to ensure product safety throughout the pre-commercialisation phase. Notably, the present rare *L. rhinocerus* strain ABI (Agro-Biotechnology Institute Malaysia) was successfully isolated and identified from a tropical forest near Lata Iskandar, Pahang, Malaysia<sup>21</sup>; however, limited information has been published on its therapeutic potential. This study reports on the use of ZFET assay on bioreactor-grown Malaysian medicinal mushroom *L. rhinocerus* MB and EPS extracts. This study explicitly addresses LC<sub>50</sub>, embryonic hatching delays, teratogenic defect, and heart rate response with clear microscopic images. Furthermore, these findings also support the possibility of future pre-clinical trials involving the safe use of MB and EPS for prospective health applications in respiratory diseases.

#### Results

**Zebrafish embryo survival rate after MB and EPS exposure.** The survival rate of zebrafish embryos following MB and EPS exposure was studied between 0 and 120 h at MB and EPS extract concentrations of 0.16–10 mg/mL. The study period included larvae as the zebrafish embryos hatch typically 48 to 72 h post-fertilisation (HPF). The survival rate of untreated embryos, between 0 and 120 HPF, was 100% (Fig. 1a). At 48 HPF, the survival percentage for embryos treated with MB fell to 85% and 60% at >5 mg/mL and 10 mg/mL, respectively. At 72 HPF, the survival rate declined to 80%, 65%, and 10% at <2.5 mg/mL, 5 mg/mL, and >10 mg/mL, respectively. At concentrations > 1.25 mg/mL, the survival rate at 96 HPF was 20%, and after 120 HPF; it was observed that no embryos survived at concentrations > 1.25 mg/mL (Fig. 1a). The survival rate of embryos



**Figure 1.** The performance of Tiger milk mushroom, *Lignosus rhinocerus* strain ABI (**a**) MB and (**b**) EPS extract at concentrations of 0.16–10 mg/mL on the survival rate of zebrafish embryos at 0–120 h. Symbols: \*\*\*p < 0.001, \*\*p < 0.01 and \*p < 0.05. No embryos survived for both samples at concentration tested > 5.0 mg/mL after 96 h-post-fertilization (HPF).

(prior to hatching) and larvae (post-hatching) treated with EPS (0.110 mg/mL) during the five days is shown in Fig. 1b. Between 0 and 120 h of HPF, untreated embryos (control) exhibited a 100% survival rate. After 72 h of HPF exposure, the survival rate declined to 90%, 85%, and 50% at a concentration of 0.63 mg/mL, 1.25 mg/mL, and 5 mg/mL, respectively. At 96 HPF, the survival rate declined to 75% at concentrations < 0.63 mg/mL and 30% at concentrations > 1.25 mg/mL. At 120 HPF, survival rates at concentrations 0.63 mg/mL declined to 30%, while survival rates at concentrations > 1.25 mg/mL were 0%, with no surviving embryos. Overall, the results suggest that MB and EPS extracts delay hatching at doses < 1.25 mg/mL.

**Zebrafish embryos mortality after MB and EPS exposure.** Overall, MB and EPS extracts had doseand time-dependent fatal effects. Figure 2 shows a high survival rate (90%) of zebrafish embryos at concentrations of MB and EPS extracts < 1.25 mg/mL. Both MB and EPS extracts had a low survival rate at high concentrations (> 1.25 mg/mL), and none survived after 96 HPF. As a result, the fatal concentration for 50% (LC<sub>50</sub> value) of zebrafish embryos exposed to MB was 0.77 mg/mL, while the LC<sub>50</sub> value of the EPS extract was 0.41 mg/mL.

**Zebrafish embryos hatching after MB and EPS exposure.** Based on the embryo observations, increasing the mushroom extract concentrations can decrease the percentage hatchability. Figure 3a illustrates the hatching rate of zebrafish embryos treated with MB and EPS (both at 0.1610 mg/mL) at 0–120 HPF. No significant changes in the hatching rate were found when the zebrafish embryos were treated with MB extract at a 0.63 mg/mL concentration. However, at 48 HPF, the rate declined to 80% at concentrations > 1.25 mg/mL. At 72 HPF, the hatching rate was lowered to 65% at 5 mg/mL. Further reduction was observed (25% hatching rate) when treated with 10 mg/mL MB, implying a high death rate after 72 HPF. The hatching rate of EPS did not alter significantly after the treatment with 0.63 mg/mL. However, due to a significant mortality rate at 72 HPF, zebrafish larvae treated with EPS at 10 mg/mL doses had the lowest hatching rate (30%).



**Figure 2.** Effect of Tiger milk mushroom, *Lignosus rhinocerus* strain ABI (**a**) MB extract at concentrations of 0.16–10 mg/mL and (**b**) EPS at concentrations of 0.01–10 mg/mL on zebrafish embryos mortality rate after 120 HPF. Symbols: \*\*\*p < 0.001, \*\*p < 0.01 and \*p < 0.05. The LC<sub>50</sub> value for MB extract was 0.77 mg/mL while LC<sub>50</sub> value for EPS extract was 0.41 mg/mL.



**Figure 3.** Hatching rate of zebrafish embryos at 0 to 120 HPF with Tiger milk mushroom *Lignosus rhinocerus* strain ABI, with MB and EPS extract at concentrations of 0.16–10 mg/mL. Symbols: \*\*\*p < 0.001, \*\*p < 0.01 and \*p < 0.05. (**a**) For MB, a low hatching rate (<25%) was observed at concentration 10.0 mg/mL due to a high mortality rate. Meanwhile, (**b**) for EPS, a low hatching rate (<30%) was observed at concentrations of 10 mg/mL due to a high mortality rate. High hatching rate was observed at concentrations > 1.25 mg/mL (>80%).



**Figure 4.** Effect of Tiger milk mushroom *Lignosus rhinocerus* strain ABI, with MB and EPS extract at concentrations of 0.16–10 mg/mL on the heart rate of zebrafish embryos at 96 HPF. \*\*\*p<0.05 significantly different from the untreated group (zebrafish embryos in medium only). \*P<0.05 significantly different from the untreated group (zebrafish embryos in medium only). For (**a**) MB, no data at concentrations>2.5 mg/mL due to embryo death. Meanwhile, (**b**) for EPS, no data at concentrations>2.5 mg/mL was recorded due to embryo death.

**Zebrafish embryos heart rate after MB and EPS exposure.** During the development of many model species, including zebrafish, the heart is the major functioning organ<sup>38</sup>. Previous research has shown that the average heart rate of zebrafish embryos is 120–180 bpm, which is much closer to that of humans<sup>39</sup>. As shown in Fig. 4, the heart rates of zebrafish larvae at 96 HPF (4 days) for both the MB (Fig. 4a) and EPS (Fig. 4b) treatments were 130 and 140 bpm, respectively. Both extracts exhibited no significant difference in the heart rate of zebrafish larvae at 96 HPF at lower concentrations (relative to higher doses in Fig. 3), ranging between 0.161.25 mg/mL for MB and 0.161.25 mg/mL for EPS. The heart rate of zebrafish larvae at these concentrations was not determined because both MB and EPS extracts at 2.5, 5, and 10 mg/mL demonstrated very little to no survival at 96 HPF.

**Morphology of the larvae and zebrafish embryos after MB and EPS exposure.** Potential morphological abnormalities in embryos and larvae were measured from 0 to 120 HPF. There was no apparent teratogenic effect on embryos and larvae after 120 h of exposure to MB and EPS at 0.63 mg/mL and 1.25 mg/mL, respectively (Fig. 5). These findings infer that MB and EPS have no teratogenic effects on zebrafish embryo development prior to- and post-hatching. The unaffected development of zebrafish embryos and larvae after exposure to 0.63 mg/mL MB and 1.25 mg/mL EPS are shown in Fig. 6 and Fig. 7; however, numerous defects were observed when the concentration of MB and EPS increased to 10 mg/mL (Fig. 8 and Fig. 9). Coagulated embryos observed between 24 HPF (segmentation) and 48 HPF (pharyngula), along with the loss of yolk sac preventing hatching, were the most common abnormalities reported using MB treatments. Moreover, EPS-treated zebrafish hatched at 72 HPF, where tail deformity and damaged blood cells were observed after 120 HPF, with various defects included missing fins, guts, and melanophores.

#### Discussion

*Lignosus rhinocerus* is well-known for its therapeutic values, particularly as potential treatment of respiratory diseases. Previous reports have highlighted that the sclerotia, mycelium, and exopolysaccharides of *L. rhinocerus* contain similar bioactive compounds to  $\beta$ -glucans<sup>21</sup>. Nowadays,  $\beta$ -glucans have gained appeal for several emerging applications, including biopolymers<sup>40</sup> and biomedicines<sup>41</sup>. Notable potential therapeutic properties recently uncovered of mushroom-derived  $\beta$ -glucans include: (a) new or complementary immunotherapies

Scientific Reports | (2021) 11:23079 |



**Figure 5.** Effect of MB-EPS extracts (0.16–10 mg/mL) of Tiger Milk mushroom *Lignosus rhinocerus* strain ABI showing normal zebrafish embryogenesis at different HPF development. There were four periods depicted as according to Taufek et al.<sup>16</sup>: (**a**) Blastula (4 HPF), (**b**) Segmentation (24 HPF), (**c**) Pharyngula (48 HPF), and (**d**) Hatching (72 HPF). A—eye anlage; An—anus; Bc—blood cells; C—chorda; Ch—chorion; F—fin; G—gut; M—melanophores; O—ear bud; P—pericardium; S—somites; Y—yolk sac. Scale bar=0.5 mm. The inverted microscope procedure was used to produce the images.

against Coronavirus disease (SARS-CoV-2)<sup>18</sup>; (b) new therapeutic agent for mitigating diseases associated with gastrointestinal mucosal damage, such as peptic ulcers and inflammatory bowel disease<sup>42</sup>; (c) anticancer drugs for lung and breast cancer<sup>43</sup>; and (d) asthmatic treatment<sup>28,44</sup>. However, there is a substantial gap in knowledge surrounding the toxicity (if any) of these mushroom-derived bioactive compounds, particularly on MB-EPS extracts. Therefore, the use of Zebrafish trials could aid product development and implementation. Hence, this work investigated and reported on the acute toxicity of zebrafish embryos post exposure to MB and EPS derived from a rare Malaysian-origin Tiger Milk mushroom *L. rhinocerus* grown in a bioreactor.

The ZFET approach was used to expose fertilised zebrafish embryos to quantities of *L. rhinocerus* extract, MB (0.16–10 mg/mL), and EPS (0.16–10 mg/mL) shown to be non-toxic. Overall, both MB and EPS at 2.5 mg/mL concentrations did not delay embryo hatching and had a > 80% survival rate between 24 and 120 HPF. In addition, there were no significant differences in the embryo heart rate between the MB and EPS concentrations of 1.25 mg/mL. At MB and EPS doses of > 0.63 mg/mL and > 1.25 mg/mL, respectively, teratogenic effects were observed with evident zebrafish embryo defects. The test revealed that MB has a larger LC<sub>50</sub> value of 0.77 mg/mL than EPS, with a lower LC<sub>50</sub> value of 0.41 mg/mL. Although both MB and EPS extracts were obtained from *L. rhinocerus* mycelium, the compound composition may differ owing to the fruiting body and mycelial extraction procedures<sup>45,46</sup>. *L. rhinocerus* mycelium and culture broth demonstrated similar or increased bioactivities, including antioxidant capacities, compared to the use of fruiting bodies<sup>45</sup>. Moreover, EPS exhibited a lower LC<sub>50</sub> value than MB did due to its different mycelial extraction methodology. This is possibly related to MB being directly obtained from dried fungal mycelia<sup>21,47</sup>. The embryo's ability to burst through the chorion (Fig. 8) and hatch after five days may be limited by morphological defects such as tail deformity. A coagulated embryo and the absence of a heartbeat are both considered deadly.

Certain medicinal mushrooms have also been tested for their toxicity on zebrafish embryos in comparison to *Lignosus* species. Recent research on *Ganoderma lucidum* exposure found that MB did not affect ZE hatching at concentrations ranging from 250 to 5000 g/mL and EPS at 3000 g/mL. Notably, neither MB nor EPS were teratogenic at concentrations < 3000 g/mL<sup>35</sup>. Neither EPS or endopolysaccharide (ENS) concentrations of 1 mg/mL in *G. applanatum* cause embryo hatching delays. They were shown to have an 88% survival rate when tested from 24 to 120 HPF<sup>48</sup>. Consequently, this new ZFET data could be helpful in the identification of potential health risks associated with the MB-EPS consortia. However, more testing is merited to identify the LC<sub>50</sub> value of MB-EPS extract for large-scale human trials and larger animals before this innovation may be used commercially

#### Mycelial Biomass (MB) at 0.63 mg/mL



**Figure 6.** Illustrations of zebrafish embryo and larvae development after treated with Tiger Milk mushroom *Lignosus rhinocerus* strain ABI at EPS concentration of 0.63 mg/mL. Descriptions were captured using an inverted microscope at 100X (0 and 24 HPF) and 40X magnification (48–20 HPF).





**Figure 7.** Illustrations of zebrafish embryo and larvae development after treated with Tiger Milk mushroom *Lignosus rhinocerus* strain ABI at EPS concentration of 1.25 mg/mL. Descriptions were captured using an inverted microscope at 100X (0 and 24 HPF) and 40X magnification (48–20 HPF).

#### (a) Mycelial Biomass (MB) at 10.0 mg/mL



**Figure 8.** Illustrations of zebrafish embryo and larvae development after treated with Tiger Milk mushroom *Lignosus rhinocerus* strain ABI at high EPS concentration of 10.0 mg/mL. Descriptions were captured using inverted microscope at 100X (0 and 24 HPF) and 40X magnification (48–20 HPF).



**Figure 9.** Illustrations of zebrafish embryo and larvae development after treated with Tiger Milk *Lignosus rhinocerus* strain ABI at high EPS concentration of 10.0 mg/mL. Descriptions were captured using an inverted microscope at 100X (0 and 24 HPF) and 40X magnification (48–20 HPF).

(b) Exopolysaccharide (EPS) at 10.0 mg/ml

			Non-toxic concentrations (mg/mL)		
Fungal source	Toxicity model	Image	Mycelial Biomass (MB)	Exopolysaccharide (EPS)	References
L. rhinocerus	In vivo—Zebrafish embryos and larvae	0	0.77	0.41	Current study
L. rhinocerotis	In vitro—Cervical cancer cells (Ca Ski, HPV-16)	NA	25	-	50
L. rhinocerotis	In vitro—Differentiating mouse neuroblastoma (N2a) cells	с —	1.75-5.93	-	51
L. rhinocerotis	In vitro-MTT assay	NA	0.2	-	45
L. rhinocerotis	In vivo—Developmental toxicity in pregnant Sprague–Dawley (SD) rats	NA	3.4	-	52

**Table 1.** Similarity with literature for non-toxicity evaluation of mycelium biomass (MB) and exopolysaccharide (EPS) from the rare Tiger milk mushroom *Lignosus* sp. [NA: Not Available].

(e.g. pigs, rabbits, and adult trout). A similar biosafety approach using zebrafish was used to test EPS from the wild-Serbian mushroom *G. applanatum*<sup>49</sup>, which exhibited a higher yet safe  $LC_{50}$  value (1.41 mg/mL) than that of the current wild-Malaysian *L. rhinocerus* study (0.41 mg/mL). The MB of *L. rhinocerus* demonstrated a harmless biosafety status of bioreactor cultivated *L. rhinocerus* mycelia and EPS products; thus, supporting further pre-commercialisation trials. Usuldin et al.<sup>21</sup> found that MB production (~6 g/L: 30 g dry form) from a 5-L bioreactor culture supports high EPS yield, which can be produced in large quantities. When compared to dried polysaccharides, powdered MB is more applicable in the pharmaceutical industry. The latter is notable as 300 mg of dry tuber biomass from the Malaysian *L. rhinocerus* has been reported to potentially improve respiratory health in both in vivo and in vitro models<sup>34</sup>.

This study therefore constitutes the first toxicity investigation of *L. rhinocerus* grown in a bioreactor, with the results compared with that of extracts from other *Lignosus* species. Table 1 shows details of four studies assessing the effect of *L. rhinocerus* MB on cervical cancer cells  $(24 \text{ mg/mL})^{50}$ , neurite bearing cells  $(1.75-5.93 \text{ mg/mL})^{51}$ , MTT assay for normal human cells  $(200 \mu \text{g/mL})^{45}$ , and developmental toxicity in pregnant Sprague–Dawley (SD) rats  $(3.4 \text{ mg/mL})^{52}$ . Notwithstanding this, there is no published research on the toxicity of EPS. The study results are significant where the Zebrafish 3.0 toxicity model was used to evaluate and assess what was to be non-toxic mycelial biomass (0.77 mg/mL) and EPS (0.41 mg/mL) in *L. rhinocerus* bioreactor samples. This Zebrafish model offered evidence that the use of Malaysian bioactive mycelial biomass and polysaccharides *L. rhinocerus* may be safe as a new therapeutic intervention.

Furthermore, the findings from this research highlight the increasing trend towards the intensive yet sustainable exploitation of bio-based resources from food and marine ecosystems, from the emergence of the bioeconomy<sup>11</sup>. These bio-inspired materials may be refined and scaled up for commercial use through advances in biotechnology, as described here<sup>49</sup>. Notably, this emerging area will be future-proofed through accelerating digitalisation, where metadata outputs will potentially inform food for therapeutics, cosmetics, personal care products, and smart packaging, along with offering putative interventions to help mitigate the Covid-19 disease<sup>10,51,53</sup>.

#### Conclusion

In conclusion, this is the first study on the use of ZFET assay on bioreactor-grown Malaysian medicinal tiger milk mushroom *L. rhinocerus* MB and EPS extracts. MB ( $LC_{50}$ : 0.77 mg/mL) was harmless, whereas EPS ( $LC_{50}$ : 0.41 mg/mL) are practically non-toxic. The ZFET assay offers a fast, affordable, robust, and efficient early development approach to evaluating extracts from medicinal fungi for future use as asthmatic medication. Specifically, this study provides evidence of the potential of *L. rhinocerus* as an alternative or adjuvant to the current drugs used for the management of respiratory diseases. Additionally, for the early medication development process, zebrafish can be utilised to quickly discover potentially dangerous chemicals and prioritise compounds for additional pre-clinical and clinical testing. The adaptation of conventional instruments in conjunction with new nanotechnology discoveries will help to further increase the use of zebrafish for drug screening.

#### Methods

**Tiger milk mushroom.** Wild Malaysian tiger milk mushroom, *L. rhinocerus* strain ABI was isolated from Lata Iskandar, Pahang, Malaysia, from the tropical rainforest (23 °C to 28 °C; 4.1949° N, 101.1923° E)<sup>21</sup>. The sclerotium was cultured on a potato dextrose agar (PDA) plate (Sigma-Aldrich, Dorset, UK) and incubated at 30 °C under dark conditions. The strain was stored and maintained on PDA slants at 4 °C<sup>54</sup>.

**Culture conditions.** The fungal inoculum was prepared according to Wan Mohtar et al.<sup>55</sup> blueprints fungal production plan, including two seed culture stages. The mycelium was cultivated for ten days under dark conditions at an initial pH of 5, 150 rpm, and 30 °C with slight adjustments for the first seed culture. Four mycelial agar squares (1 cm x 1 cm each) were cut from a ten-day-old plate culture and inoculated in a 250 mL Erlenmeyer flask using sterile scalpels (100 mL of medium). The first seed culture was then homogenised for 10 sec with a sterile Waring hand mixer to produce more hyphal tips with uniform mycelium diameters. The homogenised mycelial culture was transferred to a 500 mL shake flask (200 mL medium) as the inoculum for the second seed culture and incubated for 11 d under dark circumstances on an orbital shaker at initial pH 5, 150 rpm, and 30 °C. Unless otherwise stated, the liquid culture medium of seed cultures contained glucose (3% (w/ v), yeast extract (0.1% (w/ v), peptone (0.2% (w/ v), potassium dihydrogen phosphate (KH<sub>2</sub>PO<sub>4</sub>) (0.046% (w/ v), dipotassium hydrogen phosphate (K<sub>2</sub>HPO<sub>4</sub>) (0.1% (w/ v), and magnesium sulphate heptahydrate (MgSO4.7H).

**High-scale bioreactor fermentation.** A stirred-tank (STR) bioreactor was used with a total volume of 5 L (3.5-L working volume) (Sartorius Stedim, Biostat B-plus, Germany). Blueprint of Usuldin et al.<sup>21</sup> was followed; 10% (v/v) of the seed culture was used to inoculate the STR using parameters as follows: temperature (30 °C); pH 5.0; dissolved oxygen (DO) (20–40%); air flow rate 3 L/min; agitation speed (200 rpm). The mycelium was cultured in the bioreactor for 11 d and the resulting mycelial pellets were isolated. The media formulation for the bioreactor used was the same as that for the shake flask, unless otherwise stated.

**Mycelial biomass and exopolysaccharide production.** The bioreactor's harvested mycelial biomass (MB) was filtered three times with distilled water using a vacuum Buchner funnel filter. The filtered MB was dried at 35 °C in a food dehydrator until it reached a consistent weight<sup>47</sup>. The filtrate was precipitated by adding 95% (v/ v) ethanol at a ratio of 1:4 to the filtrate and left overnight at 4 °C to obtain the EPS. After that, the sample was centrifuged for 15 min at 10,000 rpm. The supernatant was discarded, and the pellet was dried at 35 °C in a food dehydrator until it reached a constant weight.

**Sample preparation for the toxicity test.** Dried MB and EPS were prepared at room temperature for toxicity testing. A 10 mg/mL of stock solution was prepared by dissolving dried MB and EPS in embryo media (Danio-SprintM media), which was then diluted two-fold and further in a 96-well microplate (200  $\mu$ L/well) using serial dilutions to obtain seven different concentrations in the 0.16–10 mg/mL range. For a standard control, zebrafish embryos in embryo media solution were used as an untreated control sample (0 mg/mL).

**Upkeep and breeding of zebrafish system.** A couple of adult zebrafish were placed in a breeding tank the day before the breeding occurred to set up the system. The following day, embryos were cleansed and incubated in the embryo medium (Danio-SprintM media) for two hours. Only healthy fertilised embryos were selected for the ZFET testing; meanwhile, the dead and coagulated embryos were discarded<sup>35</sup>.

**Zebrafish embryo toxicity (ZFET) test.** Firstly, at 0 HPF, zebrafish embryos were exposed to samples (200  $\mu$ L) in 96-well microplates (embryo/well) at seven different concentrations ranging from 0.16 to 10 mg/mL. The experiments were designed with an exposure group, both treated and untreated, containing 12 embryos each. The successfully treated embryos were cultured at ambient temperature (25 °C to 28 °C) for five days. The cumulative mortality and development abnormalities of zebrafish embryos and larvae were observed and examined for every 24 HPF from 0 to 120 HPF. Data of the survival rate, hatching rate, heart rate, morphological malformations, and teratogenic defects were captured and recorded using an inverted microscope coupled with a digital camera. The heartbeats were counted using a stopwatch (three embryos/min). Lethal endpoints were defined based on coagulation and the nonappearance of a heartbeat. Developmental defects such as pericardial oedema, yolk sac oedema, non-hatched, twisted body, and twisted tail were observed and recorded. The LC<sub>50</sub> values were considered based on the principle of toxicity, in which > 1 mg/mL are considered relatively harmless, 0.1–1 mg/mL non-toxic, 0.01–0.1 mg/mL slightly toxic, 0.001–0.01 mg/mL moderately toxic, 0.0001–0.001 mg/mL highly toxic, and > 0.0001 mg/mL are super toxic.

**Ethics declaration.** The breeding of Zebrafish (Danio rerio F. Hamilton, 1822) broodstocks and the in vivo methodology was approved by the Institutional Animal Care and Use Committee (IACUC) of Universiti Putra Malaysia (UPM), Malaysia and a licensed Danio Assay Laboratories Sdn. Bhd. (1,075,617-T), Director, Animal Biochemistry & Biotechnology Laboratory (ABBTech), Department of Biochemistry, Faculty of Biotechnology & Biomolecular Sciences, UPM, Selangor. The research was carried out in accordance with the Organization for Economic Cooperation and Development (OECD) No. 236: Fish Embryo Acute Toxicity (FET) Test (OECD, 2013)<sup>56</sup>, under compliance of IACUC UPM using triplicates of all samples and ARRIVE guidelines.

**Statistical evaluation.** All of the graphs and figures were produced using GraphPad Prism v.8.0. (GraphPad Soft-ware, Inc.). The lethal concentration at 50% ( $LC_{50}$ ) of treated samples toward zebrafish embryos was

107

evaluated using the same methods. The heart rates of three different animals were presented as a mean standard error of mean (SEM). A one-way analysis of variance (ANOVA) was used to determine significant differences, followed by a Dunnett's Multiple Comparison post-hoc test. differences between the means of the treated group and embryos in embryo media were set at  $p 0.001^{**}$ ,  $p 0.01^{**}$ ,  $p 0.05^{*}$ .

Received: 3 August 2021; Accepted: 12 November 2021 Published online: 29 November 2021

#### References

- Waktola, G. & Temesgen, T. Application of mushroom as food and medicine. Adv. Biotechnol. Microbiol. 113, 1–4. https://doi.org/ 10.19080/AIBM.2018.11.555817 (2018).
- Vunduk, J. *et al.* Polysaccharides of Pleurotus flabellatus strain Mynuk produced by submerged fermentation as a promising novel tool against adhesion and biofilm formation of foodborne pathogens. *Lwt* 112, 108221. https://doi.org/10.1016/j.lwt.2019.05.119 (2019).
- Sullivan, R., Smith, J. E. & Rowan, N. J. Medicinal mushrooms and cancer therapy: Translating a traditional practice into Western medicine. *Perspect. Biol. Med.* 49, 159–170. https://doi.org/10.1353/pbm.2006.0034 (2006).
- Smith, J. E., Rowan, N. J. & Sullivan, R. Medicinal mushrooms: A rapidly developing area of biotechnology for cancer therapy and other bioactivities. *Biotech. Lett.* 24, 1839–1845. https://doi.org/10.1023/A:1020994628109 (2002).
- 5. Sujarit, K., Suwannarach, N., Kumla, J. & Lomthong, T. Mushrooms: Splendid gifts for the cosmetic industry. *Chiang Mai J. Sci.* **48**, 699–725 (2021).
- Cerimi, K., Akkaya, K. C., Pohl, C., Schmidt, B. & Neubauer, P. Fungi as source for new bio-based materials: A patent review. Fungal Biol. Biotechnol. 6, 1–10. https://doi.org/10.1186/s40694-019-0080-y (2019).
- 7. Meyer, V. *et al.* Current challenges of research on filamentous fungi in relation to human welfare and a sustainable bio-economy: A white paper. *Fungal Biol. Biotechnol.* **3**, 1–17. https://doi.org/10.1186/s40694-016-0024-8 (2016).
- Wan-Mohtar, W. et al. Fruiting-body-base flour from an oyster mushroom-a waste source of antioxidative flour for developing potential functional cookies and steamed-bun. AIMS Agric. Food https://doi.org/10.3934/agrfood.2018.4.481 (2018).
- 9. Wan-Mohtar, W. A. A. I. *et al.* Fruiting-body-base flour from an Oyster mushroom waste in the development of antioxidative chicken patty. *J. Food Sci.* **85**, 3124–3133. https://doi.org/10.1111/1750-3841.15402 (2020).
- Rowan, N. J. & Galanakis, C. M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis?. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.141362 (2020).
- Rowan, N. J. & Casey, O. Empower Eco Multi-Actor HUB: A triple helix "academia-industry-authority" approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations' Sustainable Development Goals framework. *Current Opin. Environ. Sci. Health* https://doi.org/10.1016/j.coesh.2021.100254 (2021).
- Tran, T. N. T. et al. Modelling drying kinetic of oyster mushroom dehydration–The optimization of drying conditions for dehydratation of Pleurotus species. Mater. Sci. Energy Technol. 3, 840–845. https://doi.org/10.1016/j.mset.2020.10.005 (2020).
- 13. Águila-Almanza, E. *et al.* Facile and green approach in managing sand crab carapace biowaste for obtention of high deacetylation percentage chitosan. *J. Environ. Chem. Eng.* **9**, 105229. https://doi.org/10.1021/cm501852w (2021).
- 14. Zhou, Y. *et al.* Characterization of whey protein isolate and pectin composite film catalyzed by small laccase from Streptomyces coelicolor. *Environ. Technol. Innov.* **19**, 100999. https://doi.org/10.1016/j.eti.2020.100999 (2020).
- 15. Guggenheim, A. G., Wright, K. M. & Zwickey, H. L. Immune modulation from five major mushrooms: Application to integrative oncology. Integrat. Med. Clin. J. 13, 32 (2014).
- Murphy, E. J., Rezoagli, E., Major, I., Rowan, N. J. & Laffey, J. G. β-glucan metabolic and immunomodulatory properties and potential for clinical application. J. Fungi 6, 356. https://doi.org/10.3390/jof6040356 (2020).
- Mahari, W. A. W. et al. A review on valorization of oyster mushroom and waste generated in the mushroom cultivation industry. J. Hazard. Mater. 400, 123156. https://doi.org/10.1016/j.jhazmat.2020.123156 (2020).
- Murphy, E. J. et al. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—Implications for coronavirus disease (COVID-19) immunotherapies. Sci. Total Environ. 732, 139330. https://doi.org/10.1016/j.scitotenv.2020.139330 (2020).
- Pogue, R., Murphy, E. J., Fehrenbach, G. W., Rezoagli, E. & Rowan, N. J. Exploiting immunomodulatory properties of beta-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms: concise review of existing knowledge, innovation and future opportunities. *Current Opin. Environ. Sci. Health* https://doi.org/10.1016/j.coesh.2021.100248 (2021).
- Pushparajah, V. et al. Characterisation of a new fungal immunomodulatory protein from tiger milk mushroom Lignosus rhinocerotis. Sci. Rep. 6, 30010. https://doi.org/10.1038/srep30010 (2016).
- Usuldin, S. R. A. *et al.* In-depth spectral characterization of antioxidative (1, 3)-β-D-glucan from the mycelium of an identified tiger milk mushroom Lignosus rhinocerus strain ABI in a stirred-tank bioreactor. *Biocatal. Agric. Biotechnol.* 23, 101455. https:// doi.org/10.1016/j.bcab.2019.101455 (2020).
- Abdullah, N., Haimi, M. Z. D., Lau, B. F. & Annuar, M. S. M. Domestication of a wild medicinal sclerotial mushroom, *Lignosus rhinocerotis* (Cooke) Ryvarden. *Ind. Crops Prod.* 47, 256–261. https://doi.org/10.1016/j.indcrop.2013.03.012 (2013).
- Johnathan, M., Gan, S. H., Ezumi, M. F. W., Faezahtul, A. H. & Nurul, A. A. Phytochemical profiles and inhibitory effects of Tiger Milk mushroom (Lignosus rhinocerus) extract on ovalbumin-induced airway inflammation in a rodent model of asthma. *BMC Complement. Altern. Med.* 16, 167. https://doi.org/10.1186/s12906-016-1141-x (2016).
- Supramani, S.A.R.I.Z.A.M.S.M.K.A.W.-M.W.A.A.Q.I. Optimisation of biomass, exopolysaccharide and intracellular polysaccharide production from the mycelium of an identified Ganoderma lucidum strain QRS 5120 using response surface methodology. *AIMS Microbiol.* 5, 19–38. https://doi.org/10.3934/microbiol.2019.1.19 (2019).
- Phan, C.-W., David, P. & Sabaratnam, V. Edible and medicinal mushrooms: Emerging brain food for the mitigation of neurodegenerative diseases. J. Med. Food 20, 1–10. https://doi.org/10.1089/jmf.2016.3740 (2017).
- Chen, T.-I., Zhuang, H.-W., Chiao, Y.-C. & Chen, C.-C. Mutagenicity and genotoxicity effects of Lignosus rhinocerotis mushroom mycelium. J. Ethnopharmacol. 149, 70–74. https://doi.org/10.1016/j.jep.2013.06.001 (2013).
- Masterson, C. H. *et al.* Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumoniaeinduced pulmonary sepsis. *Lett. Appl. Microbiol.* **71**, 405–412. https://doi.org/10.1111/lam.13358 (2020).
- Johnathan, M. *et al.* Lignosus rhinocerotis Cooke Ryvarden ameliorates airway inflammation, mucus hypersecretion and airway hyperresponsiveness in a murine model of asthma. *PLoS ONE* 16, e0249091. https://doi.org/10.1371/journal.pone.0247639 (2021).
- Lee, S. S., Tan, N. H., Fung, S. Y., Pailoor, J. & Sim, S. M. Evaluation of the sub-acute toxicity of the sclerotium of Lignosus rhinocerus (Cooke), the Tiger Milk mushroom. J. Ethnopharmacol. 138, 192–200. https://doi.org/10.1016/j.jep.2011.09.004 (2011).
- Zhang, C., Willett, C. & Fremgen, T. Zebrafish: an animal model for toxicological studies. *Current Protocols Toxicol.* 17, 171–1718. https://doi.org/10.1002/0471140856.tx0107s17 (2003).
- 31. Reed, B. & Jennings, M. Guidance on the housing and care of zebrafish danio rerio. *Guidance on the housing and care of zebrafish Danio rerio.* (2011).
- Su, T. et al. The feasibility of the zebrafish embryo as a promising alternative for acute toxicity test using various fish species: A critical review. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2021.147705 (2021).
- Abramenko, N. et al. Acute toxicity of Cu-MOF nanoparticles (nanoHKUST-1) towards embryos and adult Zebrafish. Int. J. Mol. Sci. 22, 5568. https://doi.org/10.3390/ijms22115568 (2021).
- Dubińska-Magiera, M., Migocka-Patrzałek, M., Lewandowski, D., Daczewska, M. & Jagla, K. Zebrafish as a model for the study of lipid-lowering drug-induced myopathies. Int. J. Mol. Sci. 22, 5654. https://doi.org/10.3390/ijms22115654 (2021).
- Taufek, N. M. et al. Performance of mycelial biomass and exopolysaccharide from Malaysian Ganoderma lucidum for the fungivore red hybrid Tilapia (Oreochromis sp.) in Zebrafish embryo. Aquaculture Rep. https://doi.org/10.1016/j.aqrep.2020.100322 (2020).
- 36. Paatero, I. *et al.* Analyses in zebrafish embryos reveal that nanotoxicity profiles are dependent on surface-functionalization controlled penetrance of biological membranes. *Sci. Rep.* 7, 8423. https://doi.org/10.1038/s41598-017-09312-z (2017).
- Saleem, S. & Kannan, R. R. Zebrafish: an emerging real-time model system to study Alzheimer's disease and neurospecific drug discovery. *Cell Death Discov.* 4, 45. https://doi.org/10.1038/s41420-018-0109-7 (2018).
- Bakkers, J. Zebrafish as a model to study cardiac development and human cardiac disease. Cardiovasc. Res. 91, 279–288. https:// doi.org/10.1093/cvr/cvr098 (2011).
- Baker, K., Warren, K. S., Yellen, G. & Fishman, M. C. Defective, "pacemaker" current (Ih) in a zebrafish mutant with a slow heart rate. Proc. Natl. Acad. Sci. 94, 4554–4559. https://doi.org/10.1073/pnas.94.9.4554 (1997).
- 40. Choudhary, S. in New and Future Developments in Microbial Biotechnology and Bioengineering 171-181 (2020).
- 41. van Steenwijk, H. P., Bast, A. & de Boer, A. Immunomodulating effects of fungal beta-glucans: From traditional use to medicine. *Nutrients* 1, 3. https://doi.org/10.3390/nu13041333 (2021).
- 42. Veeraperumal, S. *et al.* Restitution of epithelial cells during intestinal mucosal wound healing: The effect of a polysaccharide from the sclerotium of Lignosus rhinocerotis (Cooke) Ryvarden. *J. Ethnopharmacol.* **274**, 114024. https://doi.org/10.1016/j.jep.2021. 114024 (2021).
- Vetvicka, V., Teplyakova, T. V., Shintyapina, A. B. & Korolenko, T. A. Effects of medicinal fungi-derived β-glucan on tumor progression. J. Fungi 7, 250. https://doi.org/10.3390/jof7040250 (2021).
- 44. Tan, E. S. S., Leo, T. K. & Tan, C. K. Effect of tiger milk mushroom (Lignosus rhinocerus) supplementation on respiratory health, immunity and antioxidant status: An open-label prospective study. Sci. Rep. 11, 11781. https://doi.org/10.1038/s41598-021-91256-6 (2021).
- Lau, B. F. et al. The potential of mycelium and culture broth of *Lignosus rhinocerotis* as substitutes for the naturally occurring sclerotium with regard to antioxidant capacity, cytotoxic effect, and low-molecular-weight chemical constituents. PLoS ONE 9, e102509. https://doi.org/10.1371/journal.pone.0102509 (2014).
- Lau, B. F., Abdullah, N. & Aminudin, N. Chemical composition of the tiger's milk mushroom, *Lignosus rhinocerotis* (Cooke) Ryvarden, from different developmental stages. J. Agric. Food Chem. 61, 4890–4897. https://doi.org/10.1021/jf4002507 (2013).
- Abdullah, N. R. *et al.* Pellet diameter of Ganoderma lucidum in a repeated-batch fermentation for the trio total production of biomass-exopolysaccharide-endopolysaccharide and its anti-oral cancer beta-glucan response. *AIMS Microbiol.* 6, 379–400. https:// doi.org/10.3934/microbiol.2020023 (2020).
- Wan-Mohtar, W. A. A. Q. I. et al. Investigations on the use of exopolysaccharide derived from mycelial extract of Ganoderma lucidum as functional feed ingredient for aquaculture-farmed red hybrid Tilapia (Oreochromis sp.). Future Foods https://doi.org/ 10.1016/j.fufo.2021.100018 (2021).
- Wan-Mohtar, W. A. A. Q. I., Ilham, Z., Jamaludin, A. A. & Rowan, N. Use of zebrafish embryo assay to evaluate toxicity and safety of bioreactor-grown exopolysaccharides and endopolysaccharides from European ganoderma applanatum mycelium for future aquaculture applications. *Int. J. Mol. Sci.* 22, 1675 (2021).
- Lau, B. F., Abdullah, N., Aminudin, N., Lee, H. B. & Tan, P. J. Ethnomedicinal uses, pharmacological activities, and cultivation of Lignosus spp. (tigers milk mushrooms) in Malaysia—A review. J. Ethnopharmacol. 169, 441–458. https://doi.org/10.1016/j.jep. 2015.04.042 (2015).
- Phan, C. W., David, P., Naidu, M., Wong, K. H. & Sabaratnam, V. Neurite outgrowth stimulatory effects of culinary-medicinal mushrooms and their toxicity assessment using differentiating Neuro-2a and embryonic fibroblast BALB/3T3. BMC Complement Altern. Med. 13, 261. https://doi.org/10.1186/1472-6882-13-261 (2013).
- 52. Jhou, B. Y., Liu, H. H., Yeh, S. H. & Chen, C. C. Oral reproductive and developmental toxicity of Lignosus rhinocerotis mycelium in rat. *J. Ethnopharmacol.* 208, 66–71. https://doi.org/10.1016/j.jep.2017.06.029 (2017).
- Galanakis, C. M., Aldawoud, T. M. S., Rizou, M., Rowan, N. J. & Ibrahim, S. A. Food ingredients and active compounds against the coronavirus disease (COVID-19) pandemic: A comprehensive review. *Foods* 9, 1701. https://doi.org/10.3390/foods9111701 (2020).
- Wan-Mohtar, W. A. *et al.* Antimicrobial properties and cytotoxicity of sulfated (1,3)-beta-D-glucan from the mycelium of the mushroom *Ganoderma lucidum. J. Microbiol. Biotechnol.* 26, 999–1010. https://doi.org/10.4014/jmb.1510.10018 (2016).
- Wan-Mohtar, W. A. A. Q. I., Abd, M. R., Harvey, L. M. & McNeil, B. Exopolysaccharide production by Ganoderma lucidum immobilised on polyurethane foam in a repeated-batch fermentation. *Biocatal Agric Biotechnol* 8, 24–31. https://doi.org/10.1016/j. bcab.2016.08.002 (2016).
- 56. OECD. Test No. 236: Fish Embryo Acute Toxicity (FET) Test. (2013).

# Acknowledgements

We want to thank Universiti Malaya under the GPF084A-2020 and IIRG003A-2020IISS awarded to Dr Wan-Mohtar, ST007-2021 (Dr Ilham) and Agro-Biotechnology Institute (ABI), Malaysia, under the Bio-Analytical Industry Development Programme (BIDP) awarded to Ms Siti Usuldin. The work was also funded by H2020 MSCA RISE project (number 872217/19) and Interreg Atlantic Area Neptunus Project (EAPA\_576/2018) awarded to Prof Neil Rowan.

#### **Author contributions**

Conceptualization, A.U.; methodology, W.M.; software, W.M.; validation, A.J; formal analysis, N.R.; investigation, Z.I.; resources, N.R.; data curation, N.R.; writing—original draft preparation, A.U.; writing—review and editing, A.J.; visualization, R.A.; supervision, Z.I.; project administration, W.M.; funding acquisition and writing—review and editing, N.R. All authors reviewed the manuscript. A.U: Ahmad Usuldin; R.A: Raihan Abdullah; Z.I: Zul Ilham; A.J: Ainurzaman Jamaludin; W.M.: Wan-Mohtar; N.R: Neil Rowan.

# Funding

Universiti Malaysia Scholarship, GPF084A-2020, GPF084A-2020, and ST007-2021. Horizon 2020 Framework Programme, 872217/10, Just Transition, MRTT 112. Interreg Atlantic Area Neptunus Project (EAPA\_576/2018).

# **Competing interests**

The authors declare no competing interests.

# Additional information

Correspondence and requests for materials should be addressed to N.R.

Reprints and permissions information is available at www.nature.com/reprints.

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

© The Author(s) 2021

Contents lists available at ScienceDirect







# Development of a low-temperature extrusion process for production of GRAS bioactive-polymer loaded compounds for targeting antimicrobial-resistant (AMR) bacteria



Kevin Masterson<sup>a,\*</sup>, Elaine Meade<sup>d</sup>, Mary Garvey<sup>d</sup>, Mark Lynch<sup>a</sup>, Ian Major<sup>b</sup>, Neil J. Rowan<sup>a,c</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Ireland

<sup>b</sup> Materials Research Institute, Athlone Institute of Technology, Ireland

<sup>c</sup> Centre for Disinfection, Sterilization and Biosecurity, Athlone Institute of Technology, Ireland

<sup>d</sup> Department of Life Science, Sligo Institute of Technology, Ash Lane, Sligo, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- New copolymer process for delivery of heat-sensitive antimicrobial bioactives
- Efficacy against a range of veterinary antimicrobial resistant bacterial isolates
- Process appropriate for delivery of nextgeneration bioactives
- A new 'One-health' approach for potentially addressing priority AMR challenges



#### ARTICLE INFO

Article history: Received 4 June 2021 Received in revised form 5 August 2021 Accepted 5 August 2021 Available online 10 August 2021

Editor: Damia Barcelo

Keywords: Disease mitigation Novel materials GRAS Thermally-sensitive Antimicrobial-resistance Polymer processing

# ABSTRACT

Antimicrobial resistance (AMR) is recognised globally as one of the greatest threats to human and animal health; thus, discovery of alternative antibacterial agents to address AMR is a priority challenge. This study constitutes the first report of a low-melting temperature, polymer- extrusion process for the smart delivery of thermally-sensitive antimicrobial bioactives, including generally-regarded-as-safe (GRAS) bioactives derived from various sources. Bioactives were assessed before and after extrusion by determining their respective minimum inhibitory concentrations (MIC). WHO-priority AMR-bacterial isolates causing zoonotic infections were evaluated along with use of standard ATCC strains. Findings revealed that this copolymer method was capable of delivering thermally-sensitive bioactives with varying degrees of growth inhibition against the AMR-bacterial strains. The extrusion process was found to increase the effect of nisin against MRSA (4-fold increase) and *L. monocytogenes* (6.4-fold increase), silver nitrate (AgNO<sub>3</sub>) against *E. coli* (3.6-fold increase) and *S. epidermidis* (1.25-fold increase), and chitosan against *S. aureus* (1.25-fold). Findings show the potential applicability of this polymer extrusion process for developing future bioactive-loaded polymer compounds; thus, highlighting the potential of converging bio-based industry with novel materials for enabling 'One-Health' solutions.

© 2021 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

## 1. Introduction

\* Corresponding author. *E-mail address:* k.masterson@research.ait.ie (K. Masterson). Antimicrobial resistance (AMR) is recognised globally as one of the greatest threats to human health (Interagency Coordination Group on

https://doi.org/10.1016/j.scitotenv.2021.149545

0048-9697/© 2021 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Antimicrobial Resistance, 2019). The World Health Organisation (WHO) has called for immediate action to counteract what has become an "antibiotic resistance crisis" (World Health Organistaion, 2020). The emergence and spread of AMR is directly related to misuse and overuse of antibiotics across various sectors (Talebi Bezmin Abadi et al., 2019). WHO have developed the antibiotic classification framework, Access, Watch, Reserve (AWaRe), for aiding antibiotic stewardship (WHO report on Surveillance of Antibiotic Consumption, 2018). Access antibiotics are classified as having the lowest potential for resistance, Watch antibiotics are those that are prone to AMR, while Reserve antibiotics are classed for use as "last resort" only. It was reported that between 2000 and 2015, global consumption of Watch antibiotics increased by 90.9% per-capita, while consumption of Access antibiotics increased by 26.2% (Klein et al., 2021). On any given day, about one in every three hospital patients in Ireland are being treated with antibiotics (Plachouras et al., 2018). This is further complicated by antibiotic use in the agricultural sector, which has also been identified as a major contributor to AMR on a global scale (Economou and Gousia, 2015; Manyi-Loh et al., 2018).

In collaboration with other major World Organisations, the WHO has designed an approach, known as "One-Health", which aims to combine resources across multiple sectors particularly converging public health and animal health (FAO et al., 2008). The concept of One-Health is not new; however, in recent times it has become increasingly important such as in the area of AMR (Evans and Leighton, 2014). Studies of plasmid-encoded colistin resistance (mcr-1), initially found in pigs in China, have shown its presence in humans and animals worldwide; thus highlighting the one health link between humans, animals and the environment in the context of AMR (Caniaux et al., 2017). Additionally, there are various types of antimicrobial resistant organisms (ARO), some of which are resistant to "last resort" antibiotics (Tacconelli et al., 2018). As such, there is a pressing need to develop new or alternative antimicrobial compounds to combat the global threat of AMR for human and animal intervention (Aminov, 2010; Interagency Coordination Group on Antimicrobial Resistance, 2019; Ventola, 2015).

There are a number of potential alternatives currently under investigation, such as phages and various organic and inorganic compounds; however, such alternatives hold limitations in treating bacterial infections (Fenton et al., 2013; Gill et al., 2006; Kwiatek et al., 2012; O'Flaherty et al., 2005; Wilson and González, 2003; Wittebole et al., 2014). Bacteriophages have a narrow spectrum of effect and are generally tailored per bacterial species, leading to increased costs and higher probability of inefficacy versus infections with more complex aetiology (Brevne et al., 2017; Gomes et al., 2016; Porter et al., 2016). Biofilms also limit the efficacy of therapeutics by reducing or preventing interaction with their targets (Margues et al., 2017). For example, certain biomedical devices are complex in design with narrow lumen diameters and are prone to unwanted biofilm formations as are challenging to clean (Bhattacharya et al., 2015; Wagner et al., 2011). However, alternative interventions include using polymer-based materials in order to prevent a build-up of bacterial cells in such devices; thus, stopping initial attachment and preventing or disrupting the formation of biofilm (Costerton, 2005; Konai et al., 2018). Development of alternative biobased therapeutics and smart materials also has implications for efficacy of reprocessing and sterilization (Chen et al., 2019).

Polymer-based therapeutic delivery solutions are plausible through the use of smart manufacturing approaches, such as hot-melt extrusion (HME), which necessitates the capacity of the active compound to withstand thermal processing to maintain bioactivity (Simões et al., 2019). However, traditional antibiotics have low thermal stability, excluding them as possible candidates for HME applications (Wylie et al., 2021). There is growing interest in exploiting bio-based ingredients as potential antimicrobial therapies; however, there is a gap in knowledge as to effective non-thermal processing and delivery approaches (Rowan and Galanakis, 2020; Rowan and Casey, 2021). HME has seen growing use in the area of pharmaceuticals as it allows the production of biologically active polymers with various formulations, dosage forms, and can enhance the physical properties of the bioactive component (such as increased water solubility, improved stability and shelf-life) (Patil et al., 2016).

This constitutes the first study to report on use of low-temperature polymer extrusion process using four bioactive compounds for one health applications. Silver nitrate (AgNO<sub>3</sub>), Zinc Oxide (ZnO), nisin and chitosan were assessed similarly to that of antibiotics, by determining their minimum inhibitory concentration (MIC) against three standard, ATCC bacteria strains. These bioactives were assessed for their processing potential by incorporating them into a polymer by HME and then re-assessing their antibacterial capabilities. By assessing their potential as antibiotic alternatives, further studies can be conducted to determine suitable areas they can be utilised, replacing traditional antibiotics; thus, potentially lowering their overall use and mitigating against occurrence of AMR development as a consequence of antibiotic misuse.

AgNO<sub>3</sub> is the most commonly used and documented silver salt derivative, and has raised great interest recently for its revival as an antimicrobial (Atiyeh et al., 2007; Balazs et al., 2004; Gao et al., 2018; Prabhu and Poulose, 2012; Russell and Hugo, 1994). While previous studies have shown that AgNO<sub>3</sub> incorporation into hydrogels and liquid-based polymers can increase it's antimicrobial efficacy (dos Santos et al., 2012; Wu et al., 2018), this present study represents the first to incorporate the compound into a solid-form polymer by HME and assess its antimicrobial abilities.

ZnO is a commonly-used ingredient in cosmetics and other various areas, categorised generally recognised as safe (GRAS) by the FDA; it has recently elicited interest due to its antimicrobial abilities, particularly when used as a nano-particle (Beyth et al., 2015; Padmavathy and Vijayaraghavan, 2008; Pasquet et al., 2014; U.S. Food and Drug Administration, 2019). ZnO and ZnO nanoparticles (ZnO-NPs) have shown numerous successes in their ability as antimicrobials against bacteria and fungi. A number of studies in the food packaging area have shown ZnO-NPs suitability for polymer incorporation, demonstrating their resilience to processing, holding their antimicrobial abilities while in polymer form (Espitia et al., 2012; Silvestre et al., 2016).

Nisin, a bacteriocin, has been studied for use in food packaging to reduce spoilage. While recent studies have focused primarily on its use as a general antibacterial agent, previous works have shown its active properties within polymer preparations (Kawada-Matsuo et al., 2019; Lewies et al., 2018; Tong et al., 2014). Chitosan has been also studied in food packaging, drug delivery and as a synergistic carrier of antimicrobials (Kim et al., 2017; Kravanja et al., 2019). Both nisin and chitosan are recognised GRAS compounds, where they were previously investigated for use in active food packaging due to their known antimicrobial abilities and established non-toxicity in mammalian studies (Asli et al., 2017; Bastarrachea et al., 2015; Cardozo et al., 2014; Moon et al., 2007; J. Wu et al., 2007; Yang et al., 2014).

While the antimicrobial efficacy of these bioactives have been previously reported, this constitutes the first study to report on use alone and combined by way of incorporation into a solid polymer by HME for antimicrobial use. While their efficacy toward bacteria is important, the ability of these bioactives to be utilised in polymer forms potential creates opportunities for their adaptability across a number of important sectors such as biomedical devices (implants, catheters etc.) and in veterinary products. The HME process was chosen as it allows for complex formulations of the active ingredients, and incorporation into various polymer carriers allowing for refinement of the final product's physical characteristics. In addition, this supports and potentially enables incorporation of smart bioactive products that have been biorefined from difference sources for antimicrobial use; thus, providing a process for current markets and practises. The HME process will also enable combinations of individual bioactives that will support broad-spectrum antimicrobial properties. Specifically, this study assesses the efficacy of these bioactive-polymer combinations against wild-type and AMR bacteria, including pathogenic veterinary strains.

#### 2. Materials & methods

#### 2.1. Isolation and characterization of veterinary zoonotic strains

Bacterial isolates including *Escherichia coli*, Methicillin-resistant *Staphylococcus aureus* (MRSA), Vancomycin Resistant Enterococci (VRE), *Listeria monocytogenes* and *Acinetobacter baumannii* were obtained from diagnostic testing of canine, equine and farm animals manifesting with conditions such as bacteraemia, renal infection, open wound infections, and mastitis. Collected samples of infection or disease in the form of urine, blood material, milk and swabs were provided by registered veterinary personnel in sterile containers (Cruinn Diagnostics, Dublin, Ireland). Liquid samples were immediately inoculated onto nutrient agar and incubated at 37 °C for 24 h. Swabs were inoculated in nutrient broth and incubated at 37 °C for up to 24 h under rotary conditions (125 rpm) before streaking onto nutrient agar plates.

Individual colonies were re-streaked for isolation and pure isolated colonies inoculated into nutrient broth for further biochemical characterization. Colonies were identified based on their morphological characteristics, biochemical profile, and growth on selective agars, specifically CHROMagar<sup>™</sup> Acinetobacter (CHROMagar<sup>™</sup>, Paris, France), Harlequin<sup>™</sup> E. coli/Coliform Medium, Harlequin™ Listeria Chromogenic Agar, Baird Parker agar (LabM, Cruinn Diagnostics Ltd., Dublin, Ireland) and BBL™ Enterococcosel<sup>™</sup> Agar (Becton, Dickinson and Company, Dublin, Ireland). Identity was confirmed via polymerase chain reaction (PCR). Specifically, a single colony of each bacterial test isolate was subcultured in nutrient broth and incubated overnight at 37 °C. Genomic DNA was directly extracted using the GenElute™ Bacterial Genomic kit (Sigma Aldrich, Dublin, Ireland) according to the manufacturer's instructions. The bacterial primers ITS\_8F (5'- AGAGTTTGATCCTGGCTCAG -3') and ITS\_U1492R (5'- GGTTACCTTGTTACGACTT -3') (Sigma Aldrich, Dublin, Ireland) were used for amplification of 16s rRNA gene. PCR was performed in a total reaction volume of 20 µL, containing 17 µL red Taq 1.1× master mix (VWR, Dublin, Ireland) 1 µL ITS\_8F, 1 µL ITS\_U1492R and 1 µL of pure genomic DNA eluate. DNA amplification was performed in a thermo cycler (VWR, Dublin, Ireland) using the recommended parameters. Following DNA amplification, the PCR products were examined by electrophoresis on a 1% w/v agarose gel run at 120 V for 50 min. Successful reactions were sent to Source Bioscience (Waterford, Ireland) for clean-up and gene sequencing of products. Strains were stored long term in 20% glycerol at -20 °C and short term in nutrient broth at 5 °C. Identity of strains was confirmed via Gram stain and selective agars prior to each experimental set up.

#### 2.2. Antibiotic resistance profile of veterinary isolates

Antibiotic resistance profiles were established using CHROMagar™ agars selective for Extended Spectrum Beta-Lactamase (ESBL), vancomycin-resistant enterococci (VRE) and Methicillin-resistant Staphylococcus aureus (MRSA) (CHROMagar™, Paris, France) and a range of antibiotic susceptibility disks (ThermoFisher Scientific, Ireland) as per European Committee for Antibiotic Susceptibility Testing (EUCAST) guidelines (EUCAST, 2020). Specifically, colonies of an overnight bacterial culture suspended in sterile saline at a density of 0.5 McFarland (ca.  $1 \times 10^8$  cfu/mL) were overlaid on to Mueller-Hinton agar (MHA) (4 mm) in 90 mm circular petri plate as per the EUCAST disk diffusion method (EUCAST, 2020). An antibiotic inoculated disk was placed in the centre of the plate and incubated inverted for 18 h at 37 °C. Antibiotics used during profiling include Streptomycin, Vancomycin, Chloramphenicol, Erythromycin, Ampicillin, Amoxicillin/ Clavulanic acid, Cefpodoxime, Cefotaxime, Aztreonam, Doripenem, Meropenem, Ciprofloxacin, Levofloxacin, Colistin, Doxycycline.

Zones of inhibition were measured and used to determine the bacterial species resistance profile. The absence of a zone of inhibition denotes complete resistance (R) of the species against the tested antibiotic. Susceptible species were graded as being completely susceptible (S), or as having intermediate susceptibility (I), based on the ability of the test drug to produce a zone diameter according to EUCAST zone diameter guidelines. MRSA and VRE, which are listed as high importance, were assessed for resistance to vancomycin and guinolones amongst other therapeutics. Isolates that tested positive on CHROMagar™ ESBL and displayed resistance to the extended-spectrum cephalosporin group of antibiotics were selected for phenotype confirmation of ESBL production. ESBL detection and characterization are recommended for public health and infection control purposes (EUCAST, 2020). This was carried out by placing cefpodoxime (10  $\mu$ g) and cefpodoxime/clavulanate (10  $\mu$ g/1  $\mu$ g) discs on an inoculated MHA plate, 30 mm apart. Plates were then incubated overnight at 37 °C. A zone diameter of ≥5 mm is considered positive for ESBL production.

#### 2.3. Hot-melt extrusion (HME) co-polymer process

The bioactive compounds, nisin (2.5% w/w, SKU: N5764, CAS: 1414-45-5), Chitosan (low molecular weight, SKU: 448869, CAS: 9012-76-4), Zinc Oxide (ZnO, nanopowder: <100 nm particle size, SKU: 544906, CAS: 1314-13-2) and Silver Nitrate (AgNO<sub>3</sub>, SKU: S8157, CAS: 7761-88-8) were purchased from Sigma Aldrich (Sigma-Aldrich Ireland Limited, Wicklow, Ireland), and incorporated into a polymer by hot-melt extrusion (HME) process using a PRISM Twin Screw Extruder-16-TC (Twin bore diameter: 16 mm, screw diameter: 15.6 mm, channel depth: 3.3 mm, barrel length: 384 mm). Polymers were processed using an upper barrel temperatures 140 °C and screw speed of 100 RPM. The HME process was selected to match material stability of bioactives such that it enabled processing without affecting material functionality or activity post-processing. Poly-vinyl-pyrrolidone/vinyl acetate (PVPVA) was purchased from BASF (BASF SE Headquarters, Ludwigshafen, Germany) and chosen as the copolymer carrier due to its low melting temperature, which would ensure minimal thermal damage to the test bioactives. PVPVA is a hydrophilic polymer matrix that dissolves readily in water; thus, forming bioactive polymer solution when loaded with drugs. Processed PVPVA also has a very solid yet brittle composition allowing it to be ground to a fine powder, allowing for easier solution preparation.

#### 2.4. Bioactive solution preparation

Chitosan was dissolved in 1% (v/v) acetic acid and then adjusted to pH 5.5 with 0.4 M sodium hydroxide (NaOH). ZnO was suspended in dH<sub>2</sub>O. Nisin was dissolved in a solution of 400 mM sodium chloride (NaCl), pH 3.25. These solutions were then sterilised by autoclaving. Nisin concentrations are reported in terms of active nisin content, where 1 g of commercial nisin powder contains 25 mg of active nisin. AgNO<sub>3</sub> was put to a solution of 28% (v/v) Poly (ethylene glycol), average molecular weight 400 (PEG-400) and 26% (w/v) d-sorbitol This solution was then filter sterilised by use of a 0.2  $\mu$ m syringe filter tip. The bioactive polymers were ground into a fine powder using a mortar and pestle and prepared as per their respective bioactive, apart from AgNO<sub>3</sub>-ploymer, which was dissolved in dH<sub>2</sub>O and sterilised by autoclaving.

#### 2.5. Antibacterial assay

The minimum inhibitory concentration (MIC) was determined for each standard bacterial and veterinary isolate strain by use of the broth microdilution method, adapted from previous literature (Wiegand et al., 2008). The standard strains, *Escherichia coli* (ATCC 25922, NCTC 12241) and *Staphylococcus aureus* (ATCC 29213, NCTC 12973) were purchased from Public Health England (Culture Collections, Public Health England, Salisbury, UK). Staphylococcus epidermidis (ATCC 35984) was purchased from ATCC (LGC Standards, Middlesex, UK). Briefly, microdilution assays were carried out in untreated 96-well plates using Mueller-Hinton broth (MHB). Bacterial strains were subcultured twice from frozen stocks before use. Working stock dilutions of the bioactives and bioactive polymers were prepared in MHB, aliquoted into the first column of a 96-well plate and serial diluted along the plate. Treatment vehicles  $(T_v)$  were included to assess any effects that the solutions may have (excluding the bioactive itself). Two columns were left untreated, with one of these left uninoculated to act as a sterility and positive control (SC) and the other as a growth and negative control (GC). Colonies of the test bacteria were taken from an overnight streak plate used to prepare a 0.5 MacFarland bacterial suspension ( $1 \times 10^8$  cfu/mL). This suspension was adjusted and aliquoted into the wells of the plate (excluding the SC) giving a final in-well bacterial concentration of  $5 \times 10^5$  cfu/mL. Plates were incubated in a shaker incubator (120 rpm) for 18 h at 37 °C. Wells were observed for growth as determined by visible turbidity and absorbance readings (625 nm). The MIC was determined as the lowest concentration of a treatment to prevent cell growth (i.e. broth turbidity).

# 2.6. Statistical analysis

Experiments were carried out in replicates of three or four. Significance between MIC values of the bioactives before and after polymer processing was determined by use of either a paired *t*-test (one-tailed), or a two-way ANOVA model with Bonferroni post-hoc test. A P value <0.05 was considered significant.

### 3. Results

#### 3.1. Resistance profile of veterinary isolates

Resistance profiles were established in accordance with the WHO priority pathogen list for veterinary isolates MRSA, VRE, *L. monocytogenes*, with critically important *E. coli*, and *A. baumannii* also being assessed for resistance to 3rd generation cephalosporins and carbapenems, amongst other drug classes (Table 1). *A. baumannii* and *E. coli* both exhibited resistance to vancomycin and ampicillin, with ESBL activity, which was confirmed, with the bacteria exhibiting zones of 15 mm and 20 mm vs Cefpodoxime discs, and 23 mm and 28 mm vs Cefpodoxime + clavulanic acid discs, respectively. *E. coli* also demonstrated resistance to amoxicillin/ clavulanic acid. *E. coli* and *A. baumannii* displayed intermediate susceptibility to an array of antibiotics including streptomycin (10 mm, 15 mm),

#### Table 1

Resistance profile of veterinary isolates to a range of antibiotics as determined by zones of inhibition with reference to EUCAST cut off points.

			Zone diameter (mm) of bacterial species				
Drug class	Antibiotic	Conc. (µg/disc)	A. buamannii	E. coli (93)	MRSA <sup>a</sup>	VRE <sup>a</sup>	L. monocytogenes
Aminoglycoside	Streptomycin	10	15	10	R	R	11
Glycopeptide	Vancomycin	30	R	R	16	R(12)	13
Chloramphenicol	Chloramphenicol	30	9	21[24]	22(18)[24]	20	27
Macrolide	Erythromycin	15	9	9	15(18)[26]	7	15(25)
Penicillin	Ampicillin	10	R	R	R	15	10(16)
Penicillin-like	Ampicillin/clav	20:10	25	14(19)	11	24	25
	Cefpodoxime	10	10	20(21)	R	15	R
Cephalosporins	Cefotaxime	5	16	16(17)	R	R	R
Monobactam	Aztreonam	30	32	25(21)	-	-	-
	Doripenem	10	29	22[9]	16	25	38
Carbapenenis	Meropenem	10	25(15)	23(16)	30	12	30(26)
Quinolones	Ciprofloxacin	5	30(21)	36(22)	28(24)	R(15)	10
	Levofloxacin	5	32(20)	31[33](19)	29(22)	R(15)	26
Polymyxin	Colistin	10	13	12[9]		-	-
Tetracycline	Doxycycline	30	10	12	11	30	40

R donates complete resistance to antibiotic.

() EUCAST 2020 cut-off zone diameter (mm) for antibiotic resistance for certain species and antibiotic. Zones below this are deemed resistant.

[] EUCAST 2019 cut-off zone diameter (mm) for antibiotic resistance for certain species and antibiotic. Zones below this are deemed resistant.

<sup>a</sup> WHO high priority pathogens.

#### Table 2

Antibiotic Minimum Inhibitory Concentrations for Veterinary Isolate strains. Table shows the minimum inhibitory concentration (MIC) of various traditional antibiotics versus veterinary bacterial isolates.

		Bacterial species			
		A. baumannii	E. coli (93)	MRSA	L. monocytogenes
Antibiotic (µg/mL)	Streptomycin Vancomycin Erythromycin Azithromycin Amoxicillin Ceftazidime (3rd) Cefotaxime (3rd) Ceftriaxone (3rd) Ceftriaxone (3rd) Aztreonam Meronenem	A. baumannii 8 R 32 4 256 64 128 64 64 64 256 0 5	E coli (93) 4 256 R 8 128 16 16 16 4 2 8 8 0 125	MRSA R 1 1 8 256 32 32 32 32 32 32 - 32	L. monocytogenes 16 0.5 16 32 4 R R R 16 - 16 - 16
	Doxycycline Tetracycline Ciprofloxacin Levofloxacin	16 128 0.125 0.25	0.123 64 128 0.25 0.25	128 128 0.25 0.25	8 16 2 2

erythromycin (9 mm, 9 mm), and chloramphenicol (21 mm, 9 mm) respectively.

MRSA give clear indication of resistance to streptomycin and 3rd generation cephalosporins, with intermediate susceptibility to vancomycin (16 mm), tetracycline (11 mm), and erythromycin (22 mm). The high priority pathogen VRE demonstrates clear resistance to streptomycin, cephalosporins and quinolones, with low susceptibility to the macrolide erythromycin (7 mm) and full susceptibility to tetracyclines (doxycycline, 30 mm). *L. monocytogenes* displays resistance to ampicillin, erythromycin, and cephalosporins.

#### 3.2. Minimum inhibitory concentration of antibiotics

The MIC of an array of antibiotics was determined against each veterinary isolate of MRSA, VRE, *E. coli* and *A. baumannii* (Table 2).

#### 3.3. Minimum inhibitory concentration of bioactives

The MIC values of the bioactive compounds were determined for each ATCC bacterial strain and veterinary isolate. Results of the MIC assays gave positive indication to the bacterial inhibitory properties of AgNO<sub>3</sub>, chitosan, ZnO and nisin versus the three ATCC bacterial strains, *E. coli, S. aureus* and *S. epidermidis*. The AgNO<sub>3</sub> preparation exhibited

#### Table 3

Bioactive Minimum Inhibitory Concentrations versus ATCC Bacterial Strains. Table shows the mean minimum inhibitory concentration (MIC) of silver nitrate (AgNO<sub>3</sub>), chitosan, zinc oxide (ZnO) and nisin against the ATCC bacterial species, *E. coli, S. aureus* and *S. epidermidis*, as determined by use of broth microdilution assays. Bioactives were assessed before and after extrusion with the co-polymer PVPVA via hot-melt extrusion (HME). Significance changes of MIC were determined by use of a 2-way ANOVA and shown in terms of a P value, N = 4.

		Bacterial species			
		E. coli (ATCC 25922)	S. aureus (ATCC 25913)	S. epidermidis (ATCC 35984)	
Bioactive MIC (µg/mL)	AgNO <sub>3</sub> AgNO <sub>3</sub> -PVPVA Chitosan Chitosan-PVPVA ZnO ZnO-PVPVA	31.25 8.789 156.3 175.8 203.125* 562.5*	42.97*** 109.4*** 390.6 312.5 156.25 312.5	19.53 15.63 156.3 208.3 97.6625 140.625	
	Nisin Nisin-PVPVA	No MIC <sup>a</sup> No MIC <sup>a</sup>	6.833** 19.53**	4.885 8.3	

<sup>a</sup> Up to 0.125 mg/mL tested.

\* P < 0.05.

\*\* P < 0.01.

\*\*\* P < 0.001.

effective growth inhibitory effects, as determined by the absence of broth turbidity, against *all tested bacterial strains* at concentrations between 15 and 62.5  $\mu$ g/mL (Tables 3 & 4). The AgNO<sub>3</sub> T<sub>v</sub>, consisting of PEG-400 and d-sorbitol, exhibited no effects upon bacterial growth.

Chitosan showed efficacy versus all test strains at concentrations between 78 and 625  $\mu$ g/mL except for VRE which required up to 1250  $\mu$ g/mL (Tables 3 & 4). The Chitosan T<sub>v</sub> only held noticeable effect at the higher concentrations of 0.125–0.25% acetic acid (AcOH), which is equivalent to a chitosan concentration of 1250–2500  $\mu$ g/mL. The AcOH concentration in the MIC range would be between 0.0137 and 0.0235% AcOH. This indicates that the T<sub>v</sub> has no effect on the acquired MIC values as they are much too low.

The ZnO suspension held varying degrees of efficacy, with a mean concentration range of 78.125–312.5 µg/mL versus ATCC while the MRSA vet isolate required up to 7500 µg/mL and VRE vet isolate showed no inhibition, even with testing up to 30 mg/mL (Tables 3 & 4). The ZnO suspension was effective in inhibiting growth of the three ATCC strains, being least effective against *E. coli* and most effective against *S. epidermidis*. This can be accounted for by the bacteria's intrinsic resistance and pathological strengths. Gram-negative bacterium, such as *E. coli*, have multiple, thin layers of membrane combined with an inner peptidoglycan cell wall-layer, which present formidable barrier for therapeutics. Gram-positive bacteria, such as *S. aureus* and *S. epidermidis*, comprise of a single, outer cell membrane under a thick peptidoglycan layer that has actually been shown to enable therapeutics by aiding their absorption into the cell. Furthermore, *S. epidermidis* is a well-known opportunistic, biofilm forming bacteria recognised as an

etiological agent in complex device-mediated infection in healthcare and in veterinary practice. It generally exhibits low resistance to antimicrobials while in planktonic forms, requiring biofilm formation to become more resistant, thus justifying its lower MIC compared to *E. coli* and even *S. aureus*. The resistance of VRE to ZnO may be accounted for by its alternative peptidoglycan synthesis, which alters the bonding potential of its outer peptidoglycan layer, preventing ZnO from interacting and penetrating the bacteria cell (Ahmed and Baptiste, 2018).

Nisin was effective in inhibiting the growth of the ATCC strains of *S. aureus* and *S. epidermidis*, with MIC values between 3.9 and 7.81 µg/mL. Effective inhibition was also seen against the veterinary isolates MRSA, VRE and L. *monocytogenes*, with MICs ranging between 12.5 and 15.6 µg/mL. Nisin held no inhibitory effect against *E. coli* (both ATCC and veterinary isolates) or *A. baumannii*, even with concentrations up to 125 µg/mL. This finding is appreciated given that nisin cannot carry out its mechanisms of growth inhibition versus Gram negative bacteria due to outer cell membrane preventing this bioactive from interacting with its target, the intramembrane molecule lipid II.

# 3.4. Minimum inhibitory concentration of bioactive loaded polymer compounds

The bioactives were extruded into a polymer and assessed using the broth microdilution assay to determine their MIC versus both ATCC and veterinary isolates. Samples of the stock polymer, PVPVA, were also included at concentrations up to 60 mg/mL to assess their effects upon the bacteria. The MIC values of the bioactives before and after polymer incorporation versus the ATCC bacterial strains are presented together for comparison (Tables 3 & 4).

AgNO<sub>3</sub> held notable inhibitory effects against tested strains, before and after polymer processing (Tables 3 & 4). MIC values against *E. coli* and *S. epidermidis* obtained after extrusion into the PVPVA polymer were lower than those obtained from AgNO<sub>3</sub> pre-hot melt extrusion, noting an increase in its efficacy. AgNO<sub>3</sub> proved the most consistently effective of the four compounds, displaying consistently low MIC values against each test bacterial strain while also exhibiting lowered MIC values following polymer incorporation. This showing that not only does AgNO<sub>3</sub> still hold function following the HME process but appears to have its mechanism of actions enhanced. Additional observations have shown the bioactive to have greater stability when incorporated into the polymer, denoted by a lack of colouration after long-term air exposure, an occurrence normally seen with unmodified AgNO<sub>3</sub>.

Chitosan was quite effective against tested bacterial strains, with very little interference seen from its  $T_v$ . Following HME, chitosan exhibited no notable change in efficacy (Tables 3 & 4). Although the MICs versus *S. aureus, L. monocytogenes* and *E. coli* (*isolate*) were lowered 1.25-fold (390.6  $\rightarrow$  312.5 µg/mL), 1.5-fold (234.38  $\rightarrow$  156.25 µg/mL) and 1.5-fold (234.38  $\rightarrow$  156.25 µg/mL) respectively, showing an increase in efficacy, these were not considered significant (P > 0.05).

#### Table 4

*Mean Minimum Inhibitory Concentrations* versus *Veterinary Isolates*. Table shows the minimum inhibitory concentration (MIC) of silver nitrate (AgNO<sub>3</sub>), nisin, zinc oxide (ZnO) and chitosan, before and after incorporation with the polymer PVPVA against the veterinary isolates, *MRSA*, *VRE*, *L. monocytogenes*, *E. coli* and *A. baumannii*. Significant changes in MIC were determined by use of either a paired *t*-test (nisin) or a two-way ANOVA with Bonferroni post-test, and shown in terms of a P value. N = 3.

		Veterinary Isolate				
		MRSA	VRE	L. monocytogenes	E. coli (isolate)	A. baumannii
Bioactive (µg/mL)	AgNO <sub>3</sub>	20.83	20.83	15.63	15.63**	13.02
	AgNO <sub>3</sub> -PVPVA	31.25	31.25	15.63	31.25**	13.02
	Nisin	15.6**	15.6**	12.5**	No MIC	No MIC
	Nisin-PVPVA	3.9**	1.95**	1.95**	No MIC	No MIC
	Chitosan	208.33	1250	234.38	234.38	260.42
	Chitosan-PVPVA	260.42	1666.67	156.25	156.25	416.67
	ZnO	4583.33**	No MIC <sup>a</sup>	937.5	390.63	364.58
	ZnO-PVPVA	25000**	No MIC <sup>a</sup>	937.5	1562.5	208.17

<sup>a</sup> Up to 30 mg/mL tested.

\*\* P < 0.01.

ZnO appeared to have its antimicrobial abilities greatly hindered by the polymer processing, as all MIC values were observed to increased. *E. coli* and *S. aureus* MIC values saw a 2.8-fold (203.13  $\rightarrow$  562.5 µg/mL) and a 2-fold (156.25  $\rightarrow$  312.5 µg/mL) increase, respectively (Table 3). MRSA and *E. coli* (*isolate*) MICs were seen to increase 5.5-fold (P < 0.01) and 4-fold respectively (Table 4). The MIC against *S. epidermidis* demonstrated the lowest effect from the polymer process, with a 1.4fold increase (97.66  $\rightarrow$  140.63 µg/mL). While it is unlikely that the heat from the polymer process caused this loss of potency (as ZnO has a melting temperature of 1975 °C), it is possible that the shearing effect of the twin-screw extruder may have denatured it. Although it is probable that the polymer itself bound too strongly to the ZnO molecules; thus preventing it from freely interacting with bacterial cells.

Incorporation of nisin into a PVPVA polymer through HME exhibited a significant decrease in efficiency, as denoted by an increase in MIC values against the ATCC strains S. aureus (6.83  $\rightarrow$  19.53 µg/mL) and S. epidermidis  $(4.89 \rightarrow 8.3 \,\mu\text{g/mL})$ , which represents a 2.9-fold (P < 0.01) and 1.7-fold increase in MIC values respectively (Table 3). However, the opposite was observed in assays against the veterinary isolates, which resulted in nisin-PVPVA achieving lower MIC values versus MRSA (15.6  $\rightarrow$  3.9  $\mu$ g/mL), VRE (15.6  $\rightarrow$  1.95  $\mu$ g/mL) and *L. monocytogenes* (12.5  $\rightarrow$  1.95 µg/mL); thus, representing 4-fold, 8-fold and 6-fold decreases, respectively (Table 4). This increase in potency is noteworthy as was found to be statistically significant (P < 0.01). Another noted characteristic of the nisin-PVPVA polymer was its increased shelf life in solution. Many of the additional components of the commercial nisin powder are intentionally left within its composition in-order to increase the compounds shelf life. However once in solution, it loses potency after 7-10 days. The nisin-PVPVA were observed to hold same level of potency for up to 30 days following solution preparation, which could be accredited to the polymer supporting the nisin's polycyclic structure stability.

Overall, the four bioactives exhibited satisfactory bacterial growth inhibition against *E. coli, S. aureus* and *S. epidermidis*. In terms of effective treatment dose, AgNO<sub>3</sub> and nisin held the greatest efficiency to lowest concentrations. Chitosan had equally consistent results across bacterial strains and, along with AgNO<sub>3</sub>, was the least negatively affected following HME with PVPVA. All bioactives have demonstrated their suitability for HME as they have retained sufficient activity post-processing, with enhanced solubility and potency lifetime.

#### 4. Discussion

There has been a pressing need for alternative therapies in the treatment of bacterial infections and diseases. With the current antibiotic resistance pandemic, stringent restrictions and management are being implemented to hinder its impact nationally and globally. The use of antibiotics in medicine has taken a hesitant approach in recent years due to their misuse being observed in various sectors worldwide, including agriculture, which having a huge influence on the increasing levels of AMR bacteria. The monitoring and documenting of AMR is a vital phase in controlling its emergence and instigating future actions. Simply differentiating which drugs bacteria susceptible to, and which they are resistant to, is essential to clinicians, where susceptibility results guide treatment options. The emergence of S. aureus strains displaying intermediate susceptibility (VISA) or full resistance (VRSA) to vancomycin, is currently a significant threat to public health and safety where they are associated with hard-to-treat nosocomial infections globally. The evidence of varying levels of resistance in these veterinary isolates contributes to the association between the veterinary use of antibiotics and emergence/proliferation of antibiotic resistance. With the increasing demand on livestock production, there will undoubtedly be an increase in the veterinary use of antibiotics with proliferation of AMR and environmental pollution with resistant genes, which can impact upon the human health sector. In conjunction with monitoring resistance profiles, research on the sensitivity of treated bacteria to antibiotic alternatives is important for monitoring and controlling the threat of AMR.

Here, four bioactive compounds, silver nitrate (AgNO<sub>3</sub>), nisin, chitosan and zinc oxide (ZnO), were presented and assessed for their antibacterial abilities under conditions equal to that of antibiotic testing. Studies have been conducted to evaluate and compare AgNO<sub>3</sub> and silver nanoparticles (AgNPs), with reported AgNO<sub>3</sub> MIC values ranging between 31 and 140 µg/mL against *E. coli* and 31–80 µg/mL against *S. aureus*, which closely resemble MIC values determined the in the present study (Lima et al., 2019; Salman, 2017). Another study investigated the antibacterial and cytotoxic properties of AgNO<sub>3</sub>, reporting MIC values of 6 µg/mL versus *E. coli* and *S. aureus*, which are slightly lower than those described in the present study (31.25 and 42.97 µg/mL respectively) (Mulley et al., 2014). The present study has found that AgNO<sub>3</sub> not only holds antibacterial ability following HME, but exhibited an increased efficacy against ATCC *E. coli* and *S. epidermidis* strains.

While nisin has been extensively studied for its antibacterial abilities, there appears to be large variation between studies in terms of reported MIC values (Kawada-Matsuo et al., 2019; Lewies et al., 2018; Tong et al., 2014). Kawada-Matsuo et al. (2019) reported on a similar solution prepartion method where determined MIC values (3.2-6.4 µg/mL) against S. aureus, were similar to what were observed in this present study (6.8 µg/mL). Tong et al. (2014) reported MIC values of 1000 µg/mL against Enterococcus species, which is much greater than those reported here (15.6 µg/mL versus VRE). The variation may be a result of the preparations used in its testing, where the current study utilises a set pH and salt content for preparing the nisin solution; where some literature reports using a dilute acid in-order to dissolve the nisin after which the solution is sterilised by autoclaving or filtration. Other studies have reported the need of a specific salt content inorder to stabilise nisin in solution, without which would increase its susceptibility to the damaging effects of the autoclaving process (Rollema et al., 1995; Yamazaki et al., 2000). Results from this study show nisin to inhibit bacterial species in a similar capacity as before in relation to S. aureus, or, in some cases, to even higher efficacy in terms of Enterococcus species tested.

The majority of studies that focus on use of nisin are concerned with its incorporation into food packaging materials, either as an active component or as a coating on processed polymer films. There are numerous polymer materials examined as a carrier for nisin, such as polylactic acid (PLA), nitrocellulose (NC), methylcellulose (MC), polyethylene oxide (PEO), and even natural polymers such as chitosan (Cutter et al., 2001; Cha et al., 2003; Jin and Zhang, 2008; Jin et al., 2009; Imran et al., 2010; Imran et al., 2014; Han et al., 2017). Many of these studies have reported increased antimicrobial activity following nisin incorporation. However, many of these methods involve numerous and timeconsuming steps for the preparation of the nisin/polymer, requiring mixing of the components in a liquid form and then allowing them to dry. The incorporation of nisin into a polymer by extrusion has also been documented, although is reported that the heat and shear forces of the process have negative effects upon the antimicrobial abilities of nisin (Gharsallaoui et al., 2016). The present study represents the first known to extrude nisin using the PVPVA co-polymer. Results show successful incorporation of the nisin into the polymer, retaining activity after HME and even showing increased efficacy against the veterinary isolate strains, MRSA, VRE, L. monocytogenes. Additionally, several advantages over previous studies can be noted in that the preparation was combined using dry stocks, the product was processed in a much shorter time-span with a prolonged shelf.

The reported antimicrobial abilities of chitosan can vary greatly, with studies by Aliasghari et al. (2016) and Zaghloul (2015) showing MIC values much higher than those presented here (ranging between 625 and 1250  $\mu$ g/mL). While a separate study by Shanmugam et al. (2016) reported lower MIC values of 100  $\mu$ g/mL versus *E. coli* and *S. aureus*. The lower MIC reported by Shanmugam et al. (2016) is quite significant in comparison to the values presented here (156.3–390.6  $\mu$ g/mL), however it should be noted that the study extracted and produced its own form of chitosan, whereas the present study utilised

commercial chitosan, similarly to the two other mentioned studies. Modifications of chitosan to increase its antimicrobial potential have also shown success by reporting lowered MIC values (Hassan et al., 2018); however, these were not superior to the present study (Hassan et al., 2018). Chitosan itself being a polymer, commonly used in HME and other processes, owes for its resilience. Notably, the HME process alone without bioactives, as reported in this study, had no significant effect upon antibacterial effect. This present study also reported that ZnO can be effectively incorporated into a polymer for antimicrobial action. Pantani et al. (2013) noted that ZnO extrusion with the polymer polylactic acid (PLA) hindered diffusion of ZnO within the polymer, which would interfere with its antimicrobial abilities and may explain the reduced antimicrobial efficacy exhibited during this study.

#### 5. Conclusion

The present study has demonstrated the potential application of four GRAS bioactives, namely AgNO<sub>3</sub>, chitosan, ZnO and nisin, as alternative antibiotic compounds. The bioactives were able to inhibit bacterial growth of various standard ATCC strains and antibiotic resistant veterinary strains under similar conditions to that used in traditional antibiotic testing. Furthermore, the bioactives gave a clear indication to their suitability for polymer processing and incorporation, opening the way for their application as potential alternatives in many areas that include antibiotic use. In addition, the bioactive-polymer incorporation supports their use where traditional antibiotics were not previously appropriate (due to over/under-exposure, leading to AMR development) such as general disinfectants, antimicrobial-infused wound treatments, antimicrobial-embedded polymer devices and biofilm-based infection treatments. This constitutes the first study to report on the novel use of GRAS bioactives that have been embedded in a polymer carrier with view to use as potential, alterative antimicrobial therapeutic for challenging zoonotic and human infections. Future studies merited include determining biofilm disruptive capabilities of these bioactivepolymer combinations along with commensurate biocompatibility. The relationship between use of nonthermal reprocessing and sterilization modalities, such as electron beam and vaporized hydrogen peroxide, and maintenance of bioactive-polymer functionaty post treatment are also merited (McEvoy and Rowan, 2019; McEvoy et al., 2021). Also, studes revealing molecular and cellular mechanisms underpinning the effectiveness of this novel GRAS bioactive-polymer combintion on targeted AMR bacteria will help advance this platform technology (Farrell et al., 2011). Initial studies surrounding use of this novel copolymer delivery process shows promise for use of heat-sensitive bioactives, such as for bio-based bioactives mined from food and from the bioeconomy.

#### Funding acknowledgement

The authors acknowledge funding support from Athlone Institute of Technology Presidential Doctorate project (AIT-2017-008) for this work.

#### **CRediT** authorship contribution statement

**Kevin Masterson:** Conceptualization, Data curation, Formal analysis, Methodology, Writing – original draft. **Elaine Meade:** Data curation, Formal analysis, Methodology, Writing – original draft. **Mary Garvey:** Formal analysis, Investigation, Methodology, Writing – original draft. **Mark Lynch:** Formal analysis, Methodology, Supervision, Writing – original draft. **Ian Major:** Conceptualization, Formal analysis, Supervision, Writing – original draft, Writing – review & editing. **Neil J. Rowan:** Conceptualization, Formal analysis, Methodology, Project administration, Resources, Supervision, Writing – original draft, Writing – review & editing.

# **Declaration of competing interest**

There is no conflict of interest for authors.

#### References

- Ahmed, M.O., Baptiste, K.E., 2018. Vancomycin-resistant enterococci: a review of antimicrobial resistance mechanisms and perspectives of human and animal health. Microb. Drug Resist. 24 (5), 590–606. https://doi.org/10.1089/mdr.2017.0147.
- Aliasghari, A., Khorasgani, M.R., Vaezifar, S., Rahimi, F., Younesi, H., Khoroushi, M., 2016. Evaluation of antibacterial efficiency of chitosan and chitosan nanoparticles on cariogenic streptococci: an in vitro study. Iranian J. Microbiol. 8 (2), 93–100.
- Aminov, R.I., 2010. A brief history of the antibiotic era: lessons learned and challenges for the future. Front. Microbiol. 1 (DEC), 1–7. https://doi.org/10.3389/fmicb.2010.00134.
- Asli, A., Brouillette, E., Ster, C., Ghinet, M.G., Brzezinski, R., Lacasse, P., Jacques, M., Malouin, F., 2017. Antibiofilm and antibacterial effects of specific chitosan molecules on Staphylococcus aureus isolates associated with bovine mastitis. PLoS ONE https://doi.org/ 10.1371/journal.pone.0176988.
- Atiyeh, B.S., Costagliola, M., Hayek, S.N., Dibo, S.A., 2007. Effect of silver on burn wound infection control and healing: review of the literature. Burns 33 (2), 139–148. https://doi.org/10.1016/j.burns.2006.06.010.
- Balazs, D.J., Triandafillu, K., Wood, P., Chevolot, Y., Van Delden, C., Harms, H., Hollenstein, C., Mathieu, H.J., 2004. Inhibition of bacterial adhesion on PVC endotracheal tubes by RF-oxygen glow discharge, sodium hydroxide and silver nitrate treatments. Biomaterials 25 (11), 2139–2151. https://doi.org/10.1016/j.biomaterials.2003.08.053.
- Bastarrachea, L., Wong, D., Roman, M., Lin, Z., Goddard, J., 2015. Active Packaging Coatings. Coatings 5 (4), 771–791. https://doi.org/10.3390/coatings5040771.
- Beyth, N., Houri-haddad, Y., Domb, A., Khan, W., Hazan, R., 2015. <book-title>Alternative Antimicrobial Approach: Nano-Antimicrobial Materials</book-title>. Alternative Antimicrobial Approach : Nano-Antimicrobial Materials https://doi.org/10.1155/ 2015/246012.
- Bhattacharya, M., Wozniak, D.J., Stoodley, P., Hall-Stoodly, L., 2015. Prevention and treatment of Staphylococcus aureus biofilms. Expert Rev. Anti-Infect. Ther. 13 (12), 1499–1516. https://doi.org/10.1586/14787210.2015.1100533.Prevention.
- Breyne, K., Honaker, R.W., Hobbs, Z., Richter, M., Zaczek, M., Spangler, T., Steenbrugge, J., Lu, R., Kinkhabwala, A., Marchon, B., Meyer, E., Mokres, L., 2017. Efficacy and safety of a bovine-associated Staphylococcus aureus phage cocktail in a murine model of mastitis. Front. Microbiol. 8 (NOV), 1–11. https://doi.org/10.3389/fmicb.2017.02348.
- Caniaux, I., van Belkum, A., Zambardi, G., Poirel, L., Gros, M.F., 2017. MCR: modern colistin resistance. Eur. J. Clin. Microbiol. Infect. Dis. 36 (3), 415–420. https://doi.org/10.1007/ s10096-016-2846-y.
- Cardozo, V.F., Lancheros, C.A.C., Narciso, A.M., Valereto, E.C.S., Kobayashi, R.K.T., Seabra, A.B., Nakazato, G., 2014. Evaluation of antibacterial activity of nitric oxide-releasing polymeric particles against Staphylococcus aureus and Escherichia coli from bovine mastitis. Int. J. Pharm. https://doi.org/10.1016/j.ijpharm.2014.06.051.
- Cha, D.S., Cooksey, K., Chinnan, M.S., Park, H.J., 2003. Release of nisin from various heatpressed and cast films. LWT Food Sci. Technol. 36 (2), 209–213. https://doi.org/10. 1016/S0023-6438(02)00209-8.
- Chen, Y., Neff, M., McEvoy, B., Cao, Z., Pezzoli, R., Murphy, A., Gately, N., Hopkins, M., Rowan, N., Devine, D.M., 2019. 3D printed polymers are less stable than injection moulded counterparts when exposed to terminal sterilisation processes using novel vaporized hydrogen perioxide and electron beam processes. Polymer 183. https://doi.org/10.1016/j.polymer.2019.121870.
- Costerton, J.W., 2005. Biofilm theory can guide the treatment of device-related orthopaedic infections. Clin. Orthop. Relat. Res. 437, 7–11. https://doi.org/10.1097/00003086-200508000-00003.
- Cutter, C.N., Willett, J.L., GRSiragusa., 2001. Improved antimicrobial activity of nisinincorporated polymer films by formulation change and addition of food grade chelator. Lett. Appl. Microbiol. 33 (4), 325–328. https://doi.org/10.1046/j.1472-765X.2001. 01005.x.
- dos Santos, C.A., Jozala, A.F., Pessoa, A., Seckler, M.M., 2012. Antimicrobial effectiveness of silver nanoparticles co-stabilized by the bioactive copolymer pluronic F68. Journal of Nanobiotechnology 10 (1), 1. https://doi.org/10.1186/1477-3155-10-43.
- Economou, V., Gousia, P., 2015. Agriculture and food animals as a source of antimicrobialresistant bacteria. Infection and Drug Resistance. 8, pp. 49–61. https://doi.org/10. 2147/IDR.S55778.
- Espitia, P.J.P., de Andrade, N.J., Cruz, R.S., Medeiros, E.A.A., Soares, N.de F.F., Coimbra, J.S. dos R., 2012. Zinc oxide nanoparticles: synthesis, antimicrobial activity and food packaging applications. Food Bioprocess Technol. (5), 1447–1464 https://doi.org/10. 1007/s11947-012-0797-6.
- EUCAST, 2020. European Comminttee on Antimicrobil Testing: Antimicrobial Susceptibility Testing. https://www.eucast.org/ast\_of\_bacteria/. (Accessed 10 August 2021).
- Evans, B.R., Leighton, F.A., 2014. A history of one health. Rev. Sci. Tech. 33 (2), 413–420. https://doi.org/10.20506/rst.33.2.2298.
- FAO, OIE, WHO, UNICEF, UN System Influenza Coordination, The World Bank, 2008. A strategic framework for reducing risks of infectious diseases at the animal-humanecosystems interface. Contributing to One World, One Health, October, p. 68.
- Farrell, H., Hayes, J., Laffey, J., Rowan, N., 2011. Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant Candida albicans. J. Microbiol. Methods 84 (2), 317–326. https://doi.org/10.1016/j.mimet.2010.12.021.
- Fenton, M., Keary, R., McAuliffe, O., Ross, R.P., O'Mahony, J., Coffey, A., 2013. Bacteriophage-derived peptidase CHAPKeliminates and prevents staphylococcal biofilms. Int. J. Microbiol. https://doi.org/10.1155/2013/625341.

- Gao, S.S., Zhao, I.S., Duffin, S., Duangthip, D., Lo, E.C.M., Chu, C.H., 2018. Revitalising silver nitrate for caries management. Int. J. Environ. Res. Public Health 15 (1). https://doi. org/10.3390/ijerph15010080.
- Gharsallaoui, A., Joly, C., Oulahal, N., Degraeve, P., 2016. Nisin as a food preservative: part 2: antimicrobial polymer materials containing nisin. Crit. Rev. Food Sci. Nutr. 56 (8), 1275–1289. https://doi.org/10.1080/10408398.2013.763766.
- Gill, J.J., Pacan, J.C., Carson, M.E., Leslie, K.E., Griffiths, M.W., Sabour, P.M., 2006. Efficacy and pharmacokinetics of bacteriophage therapy in treatment of subclinical Staphylococcus aureus mastitis in lactating dairy cattle. Antimicrob. Agents Chemother. 50 (9), 2912–2918. https://doi.org/10.1128/AAC.01630-05.
- Gomes, F., Saavedra, M.J., Henriques, M., 2016. Bovine mastitis disease/pathogenicity: evidence of the potential role of microbial biofilms. Pathogens and Disease 74 (3), 1–7. https://doi.org/10.1093/femspd/ftw006.
- Han, D., Sherman, S., Filocamo, S., Steckl, A.J., 2017. Long-term antimicrobial effect of nisin released from electrospun triaxial fiber membranes. Acta Biomater. 53, 242–249. https://doi.org/10.1016/j.actbio.2017.02.029.
- Hassan, M.A., Omer, A.M., Abbas, E., Baset, W.M.A., Tamer, T.M., 2018. Preparation, physicochemical characterization and antimicrobial activities of novel two phenolic chitosan schiff base derivatives. Sci. Rep. 8 (1), 1–14. https://doi.org/10.1038/s41598-018-29650-w.
- Imran, M., El-Fahmy, S., Revol-Junelles, A.M., Desobry, S., 2010. Cellulose derivative based active coatings: effects of nisin and plasticizer on physico-chemical and antimicrobial properties of hydroxypropyl methylcellulose films. Carbohydr. Polym. 81 (2), 219–225. https://doi.org/10.1016/j.carbpol.2010.02.021.
- Imran, M., Klouj, A., Revol-junelles, A., Desobry, S., 2014. Controlled release of nisin from HPMC, sodium caseinate, poly-lactic acid and chitosan for active packaging applications. J. Food Eng. 143, 178–185. https://doi.org/10.1016/j.jfoodeng.2014.06.040.
- Interagency Coordination Group on Antimicrobial Resistance, 2019. No time to wait: securing the future from drug-resistant infections report to the secretary-general of the United Nations. World Health Organisation 1, 1–24. https://www.who.int/antimicrobial-resistance/interagency-coordination-group/IACG\_final\_report\_EN.pdf? ua=1.
- Jin, T., Zhang, H., 2008. Biodegradable polylactic acid polymer with nisin for use in antimicrobial food packaging. J. Food Sci. 73 (3). https://doi.org/10.1111/j.1750-3841.2008. 00681.x.
- Jin, Tony, Liu, L., Zhang, H., Hicks, K., 2009. Antimicrobial activity of nisin incorporated in pectin and polylactic acid composite films against listeria monocytogenes. Int. J. Food Sci. Technol. 44 (2), 322–329. https://doi.org/10.1111/j.1365-2621.2008.01719.x.
- Kawada-Matsuo, M., Watanabe, A., Arii, K., Oogai, Y., Noguchi, K., Miyawaki, S., Hayashi, T., Komatsuzawa, H., 2019. An alternative nisin a resistance mechanism affects virulence in Staphylococcus aureus. Appl. Environ. Microbiol. https://doi.org/10.1101/716191.
- Kim, J.H., Yu, D., Eom, S.H., Kim, S.H., Oh, J., Jung, W.K., Kim, Y.M., 2017. Synergistic antibacterial effects of chitosan-caffeic acid conjugate against antibiotic-resistant acnerelated bacteria. Marine Drugs 15 (6), 1–10. https://doi.org/10.3390/md15060167.
- Klein, E.Y., Milkowska-Shibata, M., Tseng, K.K., Sharland, M., Gandra, S., Pulcini, C., Laxminarayan, R., 2021. Assessment of WHO antibiotic consumption and access targets in 76 countries, 2000–15: an analysis of pharmaceutical sales data. Lancet Infect. Dis. 21 (1), 107–115. https://doi.org/10.1016/S1473-3099(20)30332-7.
- Konai, M.M., Bhattacharjee, B., Ghosh, S., Haldar, J., 2018. Recent Progress in polymer research to tackle infections and antimicrobial resistance. Biomacromolecules 19 (6), 1888–1917. https://doi.org/10.1021/acs.biomac.8b00458.
- Kravanja, G., Primožic, M., Knez, Ž., Leitgeb, M., 2019. Chitosan-based (Nano)materials for novel biomedical applications. Molecules 24 (10). https://doi.org/10.3390/ molecules24101960.
- Kwiatek, M., Parasion, S., Mizak, L., Gryko, R., Bartoszcze, M., Kocik, J., 2012. Characterization of a bacteriophage, isolated from a cow with mastitis, that is lytic against Staphylococcus aureus strains. Arch. Virol. https://doi.org/10.1007/s00705-011-1160-3.
- Lewies, A., Du Plessis, L.H., Wentzel, J.F., 2018. The cytotoxic, antimicrobial and anticancer properties of the antimicrobial peptide nisin Z alone and in combination with conventional treatments. Cytotoxicity, July. https://doi.org/10.5772/intechopen.71927.
- Lima, K.O., Vasconcelos, A.A., de Sousa Júnior, J.J.V., Escher, S.K.S., Nakazato, G., Taube Júnior, P.S., 2019. Green synthesis of silver nanoparticles using Amazon fruits. International Journal of Nanoscience and Nanotechnology 15 (3), 179–188.
- Manyi-Loh, C., Mamphweli, S., Meyer, E., Okoh, A., 2018. Antibiotic use in agriculture and its consequential resistance in environmental sources: potential public health implications. Molecules Vol. 23, Issue 4. https://doi.org/10.3390/molecules23040795.
- Marques, V.F., da Motta, C.C., de Melo, D.A., Soares, B.da S., Barbosa, H.S., de Souza, M.M.S., Coelho, S.de M.de O., Coelho, I.da S., 2017. Biofilm production and beta-lactamic resistance in Brazilian Staphylococcus aureus isolates from bovine mastitis. Braz. J. Microbiol. 48 (1), 118–124. https://doi.org/10.1016/j.bjm.2016.10.001.
- McEvoy, B., Rowan, N.J., 2019. Terminal sterilization of medical devices using vaporized hydrogen peroxide: a review of current methods and emerging opportunities. Journal of Applied Microbiologyg 127 (5), 1403–1420.
- McEvoy, B., Lynch, M., Rowan, N.J., 2021. Opportunities for the application of real-time bacterial analysis using flow cytometry for the advancement of sterilization microbiology. J. Appl. Microbiol. 130 (6), 1794–1812.
- Moon, J.S., Kim, H.K., Koo, H.C., Joo, Y.S., Nam, H.M., Park, Y.H., Kang, M.II., 2007. The antibacterial and immunostimulative effect of chitosan-oligosaccharides against infection by Staphylococcus aureus isolated from bovine mastitis. Appl. Microbiol. Biotechnol. 75 (5), 989–998. https://doi.org/10.1007/s00253-007-0898-8.
- Mulley, G., Jenkins, A.T.A., Waterfield, N.R., 2014. Inactivation of the antibacterial and cytotoxic properties of silver ions by biologically relevant compounds. PLoS ONE 9 (4), 2–10. https://doi.org/10.1371/journal.pone.0094409.
- O'Flaherty, S., Coffey, A., Meaney, W.J., Fitzgerald, G.F., Ross, R.P., 2005. Inhibition of bacteriophage K proliferation on Staphylococcus aureus in raw bovine milk. Lett. Appl. Microbiol. https://doi.org/10.1111/j.1472-765X.2005.01762.x.

- Padmavathy, N., Vijayaraghavan, R., 2008. Enhanced bioactivity of ZnO nanoparticles an antimicrobial study. Sci. Technol. Adv. Mater. 9 (3). https://doi.org/10.1088/1468-6996/9/3/035004.
- Pantani, R., Gorrasi, G., Vigliotta, G., Murariu, M., Dubois, P., 2013. PLA-ZnO nanocomposite films: water vapor barrier properties and specific end-use characteristics. Eur. Polym. J. 49 (11), 3471–3482. https://doi.org/10.1016/j.eurpolymj.2013.08.005.
- Pasquet, J., Chevalier, Y., Pelletier, J., Couval, E., Bouvier, D., Bolzinger, M.A., 2014. The contribution of zinc ions to the antimicrobial activity of zinc oxide. Colloids Surf. A Physicochem. Eng. Asp. 457 (1), 263–274. https://doi.org/10.1016/j.colsurfa.2014. 05.057.
- Patil, H., Tiwari, R.V., Repka, M.A., 2016. Hot-melt extrusion: from theory to application in pharmaceutical formulation. AAPS PharmSciTech 17 (1), 20–42. https://doi.org/10. 1208/s12249-015-0360-7.
- Plachouras, D., Kärki, T., Hansen, S., Hopkins, S., Lyytikäinen, O., Moro, M.L., Reilly, J., Zarb, P., Zingg, W., Kinross, P., Weist, K., Monnet, D.L., Suetens, C., Strauss, R., Presterl, E., Latour, K., Vandael, E., Dobreva, E., Ivanov, I.N., Florentin, D., 2018. Antimicrobial use in european acute care hospitals: Results from the second point prevalence survey (PPS) of healthcare-associated infections and antimicrobial use, 2016 to 2017. Eurosurveillance 23 (46). https://doi.org/10.2807/1560-7917.ES.23.46.1800393.
- Porter, J., Anderson, J., Carter, L., Donjacour, E., Paros, M., 2016. In vitro evaluation of a novel bacteriophage cocktail as a preventative for bovine coliform mastitis. J. Dairy Sci. 99 (3), 2053–2062. https://doi.org/10.3168/jds.2015-9748.
- Prabhu, S., Poulose, E.K., 2012. Silver nanoparticles : mechanism of antimicrobial action, synthesis, medical applications, and toxicity effects. International Nano Letters 2012 (2), 1–10. https://doi.org/10.1186/2228-5326-2-32.
- Rollema, H.A., Kuipers, O.P., Both, P., de Vos, W.M., Siezen, R.J., 1995. Improvement of solubility and stability of the antimicrobial peptide nisin by protein engineering. Appl. Environ. Microbiol. 61 (8), 2873–2878.
- Rowan, N.J., Casey, O., 2021. Empower eco multiactor hub; a triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new green Deal opportunities under a United Nations sustaining development goals framework. Curr. Opin. Environ. Sci. Health https:// doi.org/10.1016/j.coesh.2021.100254.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? Sci. Total Environ. 748, 141362. https://doi.org/10.1016/j.scitotenv. 2020.141362.
- Russell, A.D., Hugo, W.B., 1994. Antimicrobial activity and action of silver. Prog. Med. Chem. 31 (C), 351–370. https://doi.org/10.1016/S0079-6468(08)70024-9.
- Salman, H.D., 2017. Journal of global pharma technology evaluation and comparison the antibacterial activity of silver Nano particles (AgNPs) and silver nitrate (AgNO3) on some pathogenic bacteria. Journal of Global Pharma Technology 10 (9), 238–248.
- Shanmugam, A., Kathiresan, K., Nayak, L., 2016. Preparation, characterization and antibacterial activity of chitosan and phosphorylated chitosan from cuttlebone of Sepia kobiensis (Hoyle, 1885). Biotechnol. Rep. 9, 25–30. https://doi.org/10.1016/j.btre. 2015.10.007.
- Silvestre, C., Duraccio, D., Marra, A., Strongone, V., Cimmino, S., 2016. Development of antibacterial composite films based on isotactic polypropylene and coated ZnO particles for active food packaging. Coatings 6 (1), 4. https://doi.org/10.3390/coatings6010004.
- Simões, M.F., Pinto, R.M.A., Simões, S., 2019. Hot-melt extrusion in the pharmaceutical industry: toward filing a new drug application. Drug Discov. Today 24 (9), 1749–1768. https://doi.org/10.1016/j.drudis.2019.05.013.
- Tacconelli, E., Carrara, E., Savoldi, A., Harbarth, S., Mendelson, M., Monnet, D.L., Pulcini, C., Kahlmeter, G., Kluytmans, J., Carmeli, Y., Ouellette, M., Outterson, K., Patel, J., Cavaleri, M., Cox, E.M., Houchens, C.R., Grayson, M.L., Hansen, P., Singh, N., Zorzet, A., 2018. Discovery, research, and development of new antibiotics: the WHO priority list of antibiotic-resistant bacteria and tuberculosis. Lancet Infect. Dis. 18 (3), 318–327. https://doi.org/10.1016/S1473-3099(17)30753-3.
- Talebi Bezmin Abadi, A., Rizvanov, A.A., Haertlé, T., Blatt, N.L., 2019. World Health Organization report: current crisis of antibiotic resistance. BioNanoScience 9 (4), 778–788. https://doi.org/10.1007/s12668-019-00658-4.
- Tong, Z., Zhang, Y., Ling, J., Ma, J., Huang, L., Zhang, L., 2014. An in vitro study on the effects of nisin on the antibacterial activities of 18 antibiotics against enterococcus faecalis. PLoS ONE 9 (2). https://doi.org/10.1371/journal.pone.0089209.
- U.S. Food & Drug Administration, 2019. CFR Code of Federal Regulations Title 21. https:// www.accessdata.fda.gov/scripts/cdrh/cfdocs/cfcfr/CFRSearch.cfm?fr=182.8991.
- Ventola, C.L., 2015. The antibiotic resistance crisis: part 2: management strategies and new agents. P & T : A Peer-Reviewed Journal for Formulary Management 40 (5), 344–352. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=4422635&tool= pmcentrez&rendertype=abstract.
- Wagner, C., Aytac, S., Maria Hänsch, G., 2011. Biofilm growth on implants: bacteria prefer plasma coats. Int. J. Artif. Organs 34 (9), 811–817. https://doi.org/10.5301/ijao. 5000061.
- WHO report on Surveillance of Antibiotic Consumption, E. implementation, 2018. WHO Report on Surveillance of Antibiotic Consumption. In Who. http://apps.who.int/iris/ 0Ahttps://apps.who.int/iris/bitstream/handle/10665/277359/9789241514880-eng. pdf.
- Wiegand, I., Hilpert, K., Hancock, R.E.W., 2008. Agar and broth dilution methods to determine the minimal inhibitory concentration (MIC) of antimicrobial substances. Nat. Protoc. 3 (2), 163–175. https://doi.org/10.1038/nprot.2007.521.
- Wilson, D.J., González, R.N., 2003. Vaccination strategies for reducing clinical severity of coliform mastitis. Vet. Clin. N. Am. Food Anim. Pract. 19 (1), 187–197. https://doi. org/10.1016/S0749-0720(02)00070-1.
- Wittebole, X., De Roock, S., Opal, S.M., 2014. A historical overview of bacteriophage therapy as an alternative to antibiotics for the treatment of bacterial pathogens. Virulence 5 (1), 209–218. https://doi.org/10.4161/viru.25991.

- World Health Organistaion, 2020. WHO | UN interagency coordination group (IACG) on antimicrobial resistance. WHO 1, 1–25. https://www.who.int/antimicrobial-resistance/interagency-coordination-group/en/.
- Wu, J., Hu, S., Cao, L., 2007. Therapeutic effect of nisin Z on subclinical mastitis in lactating cows. Antimicrob. Agents Chemother. 51 (9), 3131–3135. https://doi.org/10.1128/ AAC.00629-07.
- AAC.00025-07.
  Wu, F., He, D., Chen, L., Liu, F., Huang, H., Dai, J., Zhang, S., You, J., 2018. Antibacterial coordination polymer hydrogels composed of silver(i)-PEGylated bisimidazolylbenzyl alcohol. RSC Adv. 8 (37), 20829–20835. https://doi.org/10.1039/c8ra00682b.
  Wylie, M.P., Irwin, N.J., Howard, D., Heydon, K., McCoy, C.P., 2021. Hot-melt extrusion of
- Wylie, M.P., Irwin, N.J., Howard, D., Heydon, K., McCoy, C.P., 2021. Hot-melt extrusion of photodynamic antimicrobial polymers for prevention of microbial contamination. J. Photochem. Photobiol. B 214 (November 2020), 112098. https://doi.org/10.1016/ j.jphotobiol.2020.112098.
- Yamazaki, K., Murakami, M., Kawai, Y., Inoue, N., Matsuda, T., 2000. Use of nisin for inhibition of alicyclobacillus acidoterrestris in acidic drinks. Food Microbiol. 17 (3), 315–320. https://doi.org/10.1006/fmic.1999.0309.
- Yang, S.C., Lin, C.H., Sung, C.T., Fang, J.Y., 2014. Antibacterial activities of bacteriocins: application in foods and pharmaceuticals. Front. Microbiol. 5 (MAY), 1–10. https://doi.org/10.3389/fmicb.2014.00241.
- Zaghloul, R., 2015. Comparison of antibacterial activity of fungal chitosan and some preservatives against some foodborne pathogenic bacteria. Egypt. J. Microbiol. 50 (1), 31–42. https://doi.org/10.21608/ejm.2015.233.





# Article Synergy Assessment of Four Antimicrobial Bioactive Compounds for the Combinational Treatment of Bacterial Pathogens

Kevin Masterson <sup>1,\*</sup>, Ian Major <sup>2</sup>, Mark Lynch <sup>1</sup> and Neil Rowan <sup>1</sup>

- <sup>1</sup> Bioscience Research Institute, Technological University of the Shannon, N37 HD68 Athlone, Ireland; marklynch@tus.ie (M.L.); nrowan@tus.ie (N.R.)
- <sup>2</sup> PRISM Research Institute, Technological University of the Shannon, N37 HD68 Athlone, Ireland; imajor@tus.ie
- \* Correspondence: k.masterson@research.ait.ie

Abstract: Antimicrobial resistance (AMR) has become a topic of great concern in recent years, with much effort being committed to developing alternative treatments for resistant bacterial pathogens. Drug combinational therapies have been a major area of research for several years, with modern iterations using combining well-established antibiotics and other antimicrobials with the aim of discovering complementary mechanisms. Previously, we characterised four GRAS antimicrobials that can withstand thermal polymer extrusion processes for novel medical device-based and therapeutic applications. In the present study, four antimicrobial bioactive-silver nitrate, nisin, chitosan and zinc oxide-were assessed for their potential combined use as an alternative synergistic treatment for AMR bacteria via a broth microdilution assay based on a checkerboard format. The bioactives were tested in arrangements of two-, three- and four-drug combinations, and their interactions were determined and expressed in terms of a synergy score. Results have revealed interesting interactions based on treatments against recognised test bacterial strains that cause human and animal infections, namely E. coli, S. aureus and S. epidermidis. Silver nitrate was seen to greatly enhance the efficacy of its paired treatment. Combinations with nisin, which is a lantibiotic, exhibited the most interesting results, as nisin has no effect against Gram-negative bacteria when used alone; however, it demonstrated antimicrobial effects when combined with silver nitrate or chitosan. This study constitutes the first study to both report on practical three- and four-drug combinational assays and utilise these methods for the assessment of established and emerging antimicrobials. The novel methods and results presented in this study show the potential to explore previously unknown drug combination compatibility measures in an ease-of-use- and high-throughput-based format, which can greatly help future research that aims to identify appropriate alternative treatments for AMR, including the screening of potential new bioactives biorefined from various sources.

**Keywords:** antimicrobial resistance; bioactives; synergy analysis; drug combinations; two-drug combination; three-drug combination; four-drug combination; checkerboard assay; broth microdilution

# 1. Introduction

Antimicrobial resistance (AMR) has become a topic of academic interest, as it has reached a crisis point that has driven scientists to consider novel appropriate solutions to overcome it [1–3]. An ever-increasing number of antibiotic resistant bacterial species have emerged that pose serious threats to modern medicine, causing a loss in efficacy of critical front-line therapeutics [4]. Antibiotics remain our primary means of eliminating pathological bacterial infections, and while there has been a recent resurgence in the development of novel antibiotic compounds, additional ways of tackling AMR bacteria are urgently needed [5,6]. Research that aims to discover alternative antimicrobials has



Citation: Masterson, K.; Major, I.; Lynch, M.; Rowan, N. Synergy Assessment of Four Antimicrobial Bioactive Compounds for the Combinational Treatment of Bacterial Pathogens. *Biomedicines* **2023**, *11*, 2216. https://doi.org/10.3390/ biomedicines11082216

Academic Editor: Maria A. Bonifacio

Received: 29 June 2023 Revised: 22 July 2023 Accepted: 31 July 2023 Published: 7 August 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). been a major topic of interest, as it particularly hopes to circumvent the emerging resistance to mainstay antibiotics [7–9]. Additionally, the co-development of methods that can assess the efficacy of appropriate combinations of already established antimicrobial compounds is important to reduce reliance on a single treatment [10-12]. While a vast number of antimicrobial compounds are in use today, many have specific modes of action and, thus, have a narrow effective spectrum in terms of the bacterial species that they can target [13]. This issue can reduce their suitability in medical settings, which ideally require a more broad-spectrum treatment, given the frequent occurrence of co-infections [14]. Furthermore, selecting a narrow-effect spectrum that relies on a singular mechanism of microbial inactivation or inhibition can also make it easier for exposed bacteria to develop unwanted resistance. Thus, the use of two or more treatments in combination to treat bacterial infections represents a highly promising avenue of research. Checkerboard assays are well-documented methods used to assess the effects of different treatments when used in combination, whereby serially diluted concentrations of treatments are combined across a 96-well microtiter plate [15-18]. The resulting effects of combination therapy can be described as synergistic, additive, or antagonistic [16,19,20]. Synergy describes a total effect greater than the sum of the individual effects. An additive effect shows that the combined drugs exhibit a total effect equal to the sum of the individual effects, being no lesser or greater. An antagonistic effect describes combinations in which the total effect is lessened compared to the sum of the individual effects [21]. Combination therapies that result in an overall synergistic effect can allow a much greater impact to result from treatments that would normally hold less or, perhaps no, effect when used alone, such as in the case of AMR bacteria. While co-treatment therapies have been widely used in the treatment of diseases such as cancer, there is a rising interest in the synergistic abilities of previously established antimicrobial compounds [10,12,16].

In a previous study reported by these authors, the individual antibacterial capabilities of four GRAS bioactive compounds—silver nitrate ( $AgNO_3$ ), nisin, chitosan and zinc oxide (ZnO)—were assessed against a number of type-strain bacterial species, as well as AMR wild-type strains [22]. These GRAS bioactives were unusual in the sense that they withstood temperatures used to extrude and process polymers used in the manufacturing of medical devices; thus, these bioactives offer interesting options for new therapeutic research. In the present study, these four bioactives will be assessed regarding their antimicrobial capabilities in combination with one another, using arrangements of two-, three- and four-drug combinations. For this initial combinational study, three standard type strains will be used, namely Escherichia coli, Staphylococcus aureus and Staphylococcus epidermidis. E. coli and S. aureus were chosen as they represent Gram-negative and Gram-positive bacteria, respectively. S. epidermidis was included as it represents opportunistic Grampositive bacterial pathogens and was observed to hold atypical behaviour against these four compounds in the previous study relative to *S. aureus*. The antimicrobial abilities of the two-drug combinations will be determined via use of a standard broth microdilution protocol in a checkerboard assay format, through which growth will be measured using turbidity absorbance readings. Three- and four-drug combinations will be assessed via use of novel versions of the checkerboard assay developed in the present study. The readings will be used to calculate the % growth of each treatment relative to the 100% growth control. While the checkerboard assay is a method commonly utilised to assess combination effects, there are various methods and programs developed for analysis of results [15-18]. The end results identified in the present study will analysed via use of the recently developed "synergy" python package, which can analyse large amounts of combinations and report their synergy scores [23].

# 2. Materials and Methods

# 2.1. Bioactive Solution Preparation

Silver nitrate (AgNO<sub>3</sub>) (SKU: S8157, CAS: 7761-88-8), nisin, 2.5% (SKU: N5764, CAS: 1414-45-5), chitosan of low molecular weight (SKU: 448869, CAS: 9012-76-4), zinc oxide

(ZnO) and nanopowder of <100 nm in particle size (SKU: 544906, CAS: 1314-13-2) were purchased from Sigma-Aldrich/Merck (Merck Life Science Limited, Arklow, Co. Wicklow, Ireland). Chitosan was dissolved in 1% (v/v) acetic acid and adjusted to pH 5.5 using 0.4 M sodium hydroxide (NaOH). ZnO was suspended in dH<sub>2</sub>O. Nisin was dissolved in a solution of 400 mM sodium chloride (NaCl), which had a pH of 3.25. These solutions were then sterilised through autoclaving. Nisin concentrations were reported in terms of active nisin content, with 1 g of commercial nisin powder containing 25 mg of active nisin. AgNO<sub>3</sub> was placed into a solution of 28% (v/v) Poly (ethylene glycol), which had an average molecular weight of 400 (PEG-400) and 26% (w/v) d-sorbitol [24]. This solution was then filter sterilised through use of a 0.2-micrometer syringe filter tip.

# 2.2. Bacterial Cell Culture

The standard strains, namely *E. coli* (ATCC 25922, NCTC 12241) and *S. aureus* (ATCC 29213, NCTC 12973), were purchased from Public Health England (Culture Collections, Public Health England, Salisbury, UK). *S. epidermidis* (ATCC 35984) was purchased from ATCC (LGC Standards, Middlesex, UK). Cultures were prepared via overnight incubation using tryptone soy agar (TSA). Colonies were then suspended in Mueller–Hinton broth (MHB) to 0.5 MacFarland absorbance for use as inoculum [25,26].

#### 2.3. Two-Drug Combinational Broth Microdilution Assay

All steps were conducted under aseptic conditions or in closed systems. The antimicrobial properties of each bioactive solution in combinations of two were assessed in terms of their growth inhibitory capabilities, as determined via use of the broth microdilution method adapted from a previously published protocol [26]. Broth microdilution assays were carried out in flat bottom 96-well plates (untreated) against three chosen bacterial strains, namely E. coli, S. aureus and S. epidermidis. Before use, microplate lids were treated using a hydrophilic coating (20% (v/v) of isopropyl alcohol (IPA), 0.5% (v/v) of Triton-X100) [27]. Bacterial inoculums were prepared to give a final in-well concentration of  $5 \times 10^{5}$  cfu/mL, as determined via absorbance readings. Two-drug combination assays were prepared in an  $8 \times 8$  checkerboard layout, allowing a total of 64 combinations. Dilutions of drug A and B were prepared in Mueller–Hinton broth (MHB) at a concentration four times higher  $(4\times)$  than the highest desired final in-well concentration. Serial dilutions (1:2) of drug A and drug B were prepared in separate 96-well plates and combined in the final test plate (1:2 dilution) (See Figures S1 and S2). Each well was then inoculated with the prepared bacterial inoculum (1:2 dilution). Absorbance of the plate was measured using a BioTek<sup>®</sup> Synergy HT microplate reader and Gen5 Microplate Reader Software (Version 2.01.14) (BioTek<sup>®</sup> Instruments GmbH, Bad Friedrichshall, Germany). The plate was read using an endpoint absorbance read at 625 nM, and results were recorded as time-point 0 (t = 0) before incubation. This process allowed measurement of any turbidity caused by treatments and was be used as a blank. The plate was placed in a container to help prevent loss of well volume due to evaporation. The container was placed on a rotary incubator at 120 RPM, 37 °C for 18 h. Following incubation, plate absorbance was read (variable shake, and the 1-min endpoint absorbance was read at 625 nm). Results were recorded as timepoint 18 (t = 18). The absorbance values were used to calculate the %inhibition for each treatment well.

#### 2.4. Three-Drug Combinational Broth Microdilution Assay

Three-drug combination assays were carried out as per the two-drug combination assay, albeit using a  $6 \times 6$  checkerboard layout. Six such checkerboards were prepared by combining drug A and drug B, as per a two-drug combination assay, and different concentrations of drug C was added to each individual checkerboard. This setup allowed 6 concentrations of drug A, drug B and drug C to be assessed in combination ( $6 \times 6 \times 6$ ), with a total of 216 combinations (See Figure S3 for example layout). The experiment was split across three 96-well plates, allowing two  $6 \times 6$  checkerboards per plate. A separate

broth microdilution assay of single treatments was also carried out as a control to ensure that the treatments and bacteria tested performed in a nominal manner. Incubations and

# 2.5. Four-Drug Combinational Broth Microdilution Assay

Four-drug combination assays were carried out in a  $4 \times 4$  checkerboard layout, which built on the three-drug layout design. The layout was designed in such a way that four  $4 \times 4$  checkerboards (CBs) were set up within four 96-well plates. Each CB combined drug A with drug B. Each of these four CBs then had a different concentration of drug C added to it. To all CBs within each plate, a difference concentration of drug D will be added. The resulting system will yield a  $4 \times 4 \times 4 \times 4$  combination (totalling in 256 combinations) (see Figure S4 for example layout).

absorbance readings were carried out as per two-drug combinational assay.

A separate broth microdilution assay was also carried out using drugs A, B, C and D in tandem with the four-drug combination assay, which was used as a control to ensure that the individual treatments and bacteria being tested performed in a nominal manner. Plate and inoculum preparation, incubations and absorbance readings were all carried out as per the two-drug combinational assay.

# 2.6. Analysis of Results for the Determination of Synergy/Antagonism

Results from drug combination assays were analysed to determine drug interactions in terms of synergy or antagonism via the "synergy" python package [28]. Input data for synergy were prepared in an excel document using the concentration of each drug ( $\mu$ g/mL) and the % growth. Input data contained an individual column for the concentration of each drug ("drug1.conc", "drug2.conc", "drug3.conc" or "drug4.conc"). The Bliss model was chosen due to its simplicity and ability to analyse four-drug combinations. The reported response was expressed in terms of % growth. The response was input under the column "effect" and expressed as a decimal fraction of 1 (i.e., 100% growth = 1.0, 50% = 0.5, 0% = 0.0). Data were then exported as a .csv file. The synergy package was opened and run using PyCharm (version 2020.2) (JetBrains s.r.o, Prague, Czech Republic), which is a pythonintegrated development environment (IDE). Following the synergy documentation, input data were imported and analysed using the Bliss model. Results were expressed in terms of a synergy score, with a positive score indicating synergy, a score of 0 representing no effect and negative scores representing antagonism.

# 3. Results

Due to the number of combinations analysed during this study, only the three highestscoring interactions of each combination and their average values will be reported and discussed. Synergy scores represent the magnitude of the combination interactions, where a positive score indicates synergy, scores close to 0 indicate an additive effect, a score of 0 represents no effect and negative scores represent antagonism. 2D heat-maps of all 64 combinations of each two-drug combination against each test bacterial species are presented in Figures 1–6 showing each combination's synergy score based on a colour scale, which is shown in the legend. Bar graphs have been prepared and presented in Figures 7–9 for each two-drug, three-drug and four-drug combination respectively, showing the three highest-scoring drug combinations (Combo 1–3 on the *x*-axis) at the drug concentrations ( $\mu$ g/mL) shown on the left *y*-axis, with the calculated Bliss synergy score shown on the right *y*-axis (see Tables S1–S5 for the graphed data).



**Figure 1.** AgNO<sub>3</sub>–chitosan synergy heat map: Graphs show heat map of synergy between silver nitrate (AgNO<sub>3</sub>) and Chitosan in inhibiting *E. coli, S. aureus* and *S. epidermidis* growth as determined via broth microdilution and absorbance readings. Inhibition results were analysed using the synergy python package using the Bliss synergy model. The synergy python package produced the heatmap graphs of each combination result, giving visual presentations of combinations of synergy (green) or antagonism (purple). n = 3.



**Figure 2.** AgNO<sub>3</sub>–nisin synergy heat map: Graphs show heat map of synergy between silver nitrate (AgNO<sub>3</sub>) and Nisin in inhibiting *E. coli*, *S. aureus* and *S. epidermidis* growth as determined via broth

microdilution and absorbance readings. Inhibition results were analysed with the synergy python package using the Bliss synergy model. The synergy python package produced the heatmap graphs of each combination result, giving visual presentations of combinations of high (green) and low (purple) synergy. n = 3.



**Figure 3.** AgNO<sub>3</sub>–ZnO synergy heat map: Graphs show heat map of synergy between silver nitrate (AgNO<sub>3</sub>) and zinc oxide (ZnO) in inhibiting *E. coli*, *S. aureus* and *S. epidermidis* growth, as determined via broth microdilution and absorbance readings. Inhibition results were analysed via the synergy python package using the Bliss synergy model. The synergy python package produced the heatmap graphs of each combination result, giving visual presentations of combinations of high (green) and low (purple) synergy. n = 3.

- 3.1. Two-Drug Combinations
- AgNO<sub>3</sub>-Chitosan

AgNO<sub>3</sub> and Chitosan reported good synergistic interactions against each bacterial strain. The combination reported the highest average synergy scores against *E. coli* (average 0.4) and *S. aureus* (average 0.32). While the average concentration of chitosan was similar to that of the MIC versus *E. coli*, AgNO<sub>3</sub> was reported to be present in lower concentrations. The most effective combination versus *S. aureus* reported concentrations that were 1/2 the MIC, with inhibition being approximately 69%. Results versus *S. epidermidis* reported good overall synergy, as much lower concentrations of each treatment exhibited more effective inhibition, and the second reported combination exhibited 99% inhibition, with 1/3 of the MIC of AgNO<sub>3</sub> and less than 1/2 of the MIC of chitosan being used.

• AgNO<sub>3</sub>-Nisin

AgNO<sub>3</sub> and Nisin demonstrated a number of highly synergistic combinations (an average 0.32 versus *E. coli* and an average 0.24 versus *S. aureus*), as well as reporting the highest two-drug score from this study (average 0.68 versus *S. epidermidis*). While the highest-scoring combinations versus *E. coli* did not report inhibition exceeding 70%, there was moderate synergy observed compared to AgNO<sub>3</sub> used alone at the same concentrations.

The third highest-scoring combination versus *S. aureus* reported 99% inhibition, using less than 1/4 MIC of AgNO<sub>3</sub> and 1/10 MIC of nisin. The three highest-scoring combinations versus *S. epidermidis* indicated that a concentration of  $10 \mu g/mL$  AgNO<sub>3</sub> was most effective in enabling nisin, which was reported to be present in relatively low concentrations, while still having a notable effect upon bacterial growth.



**Figure 4.** Chitosan–nisin synergy heat map: Graphs show heat map of synergy between Chitosan and Nisin in inhibiting *E. coli, S. aureus* and *S. epidermidis* growth, as determined via broth microdilution and absorbance readings. Inhibition results were analysed via the synergy python package using the Bliss synergy model. The synergy python package produced the heatmap graphs of each combination result, giving visual presentations of combinations of high (green) and low (purple) synergy. n = 3.

AgNO<sub>3</sub>-ZnO

AgNO<sub>3</sub> and ZnO reported moderate synergy against *E. coli* (average 0.22) and *S. aureus* (average 0.26), as well as relatively high synergy versus *S. epidermidis* (average 0.44). The highest reported *E. coli* combination exhibited 98.5% growth inhibition at an AgNO<sub>3</sub> concentration 1/4MIC and a ZnO concentration 1/2.5MIC, demonstrating a noticeable increase in the efficacy in both treatments. *S. aureus* results reported that lower concentrations of both AgNO<sub>3</sub> and ZnO exhibited greater effect when combined. One reported combination exhibited 95.5% growth inhibition using 1/1.8MIC AgNO<sub>3</sub> and 1/2.5MIC ZnO. AgNO<sub>3</sub> and ZnO demonstrated the second highest-scoring average of all two-drug combinations (average 0.44) versus *S. epidermidis*. Reported combinations exhibited effective growth inhibition at much lower concentrations, even reaching 95.4% growth inhibition with 1/1.6MIC AgNO<sub>3</sub> and 1/3.33MIC ZnO.

Nisin–Chitosan

Nisin and chitosan reported mixed results in combination. The highest-scoring combinations were identified versus *S. aureus* (average 0.24); however, the highest inhibition of these combinations reached only 50%, with no major reductions being seen in the concentrations of nisin or chitosan. Results versus *E. coli* show that greater concentrations of chitosan were needed to enable nisin; however, these concentrations exceeded the MIC of chitosan, making the combination ineffective. Results versus *S. epidermidis* demonstrated no major interactions, being close a synergy score of 0 in all combinations. Only one combination reported effective synergy, which exhibited 87.3% inhibition with a score of 0.11; however, the concentration of nisin used in this combination exceeded that of its MIC when tested alone.



**Figure 5.** Chitosan–ZnO synergy heat map: Graphs show heat map of synergy between Chitosan and zinc oxide (ZnO) in inhibiting *E. coli*, *S. aureus* and *S. epidermidis* growth, as determined via broth microdilution and absorbance readings. Inhibition results were analysed via the synergy python package using the Bliss synergy model. The synergy python package produced the heatmap graphs of each combination result, giving visual presentations of combinations of high (green) and low (purple) synergy. n = 3.

Chitosan–ZnO

Chitosan and ZnO reported very few synergistic interactions versus *E. coli* (average 0.11), *S. aureus* (average 0.09) and *S. epidermidis* (average 0.27). Analysis of interactions versus *E. coli* show that a high concentration of chitosan was required for synergy to be identified; however, the amount of chitosan was  $2 \times$  MIC, and the synergy score was relatively low. While synergy was seen versus *S. aureus* at quite low concentrations of the two highest-scoring combinations, the exhibited growth inhibition was not noteworthy (3.1%, 2.6%, respectively). Combinations versus *S. epidermidis* reported moderate synergy at quite low concentrations of each combination; however, the inhibition did not exceed 31%.

• Nisin–ZnO

Nisin and ZnO reported low synergy versus *E. coli* (average 0.08), *S. aureus* (average 0.06) and *S. epidermidis* (average 0.14). The highest-scoring combination versus *E. coli* (0.09) did not yield noteworthy inhibition, while the next highest-scoring combinations reported concentrations of ZnO that exceed the MIC in order to enable nisin. Highest-scoring combinations versus *S. aureus* reported low concentrations of each treatment; however, they had no noteworthy growth inhibitory effect (3.7–14%). Combinations versus *S. epidermidis* exhibited moderate inhibitory effects; however, the concentrations of ZnO exceeded that of the average MIC, and concentrations of nisin were not much lower that the previously reported MIC average.



**Figure 6.** Nisin–ZnO synergy heat map: Graphs that show heat map of synergy between nisin and zinc oxide (ZnO) regarding the inhibition of *E. coli*, *S. aureus* and *S. epidermidis* growth, as determined via broth microdilution and absorbance readings. Inhibition results were analysed via the synergy python package using the Bliss synergy model. The synergy python package produced the heatmap graphs of each combination results, giving visual presentations of combinations of high (green) and low (purple) synergy. n = 3.

# 3.2. Three-Drug Combinations

Chitosan–AgNO<sub>3</sub>–Nisin

Chitosan, AgNO<sub>3</sub> and nisin reported moderate-to-high synergy in growth inhibition versus E. coli (average 0.38), S. aureus (average 0.56) and S. epidermidis (average 0.43). While the higher scoring combinations versus *E. coli* included high concentrations of chitosan (80–160  $\mu$ g/mL), reported concentrations of AgNO<sub>3</sub> were low (2–4  $\mu$ g/mL) and had 99% inhibition. Concentrations of nisin were rather high, relative to those of the other test species (3.91–7.81 µg/mL). The highest-scoring combination versus S. aureus reported relatively low concentrations of each compound (78.13 µg/mL chitosan, 8 µg/mL AgNO<sub>3</sub> and 10.63 µg/mL nisin) and expressed 99% inhibition. The second highest-scoring combination showed similar concentrations and levels of inhibition; however, when used twice, the amount of AgNO<sub>3</sub> (16  $\mu$ g/mL) was still less than the previously reported MIC. The third highest-scoring combination reported lower chitosan (39.06  $\mu$ g/mL), though it did not fully inhibit S. aureus growth (71.45%). These combinations versus S. aureus reported the second highest average score of all three-drug test combinations. Combinations versus S. epidermidis reported near full inhibition (92–96%), as well as good synergy and low concentrations of chitosan (39.06–78.13 µg/mL) and nisin (0.63–1.25 µg/mL); however, concentrations of AgNO<sub>3</sub> were near to the MIC ( $8-16 \mu g/mL$ ).

• Chitosan–AgNO<sub>3</sub>–ZnO

Chitosan, AgNO<sub>3</sub> and ZnO held moderately low synergy versus *E. coli* (average 0.14), *S. aureus* (average 0.28) and *S. epidermidis* (average 0.35), with combinations versus *E. coli* having the lowest scores of all three-drug combinations. Scores for *E. coli* were quite low,

and each of the reported combinations exhibited full inhibition (97.15–100%) at low concentrations of chitosan (40–80  $\mu$ g/mL), AgNO<sub>3</sub> (0.5–4  $\mu$ g/mL) and ZnO (20  $\mu$ g/mL). The highest scores versus *S. aureus* reported consistently low concentrations of ZnO (62.5  $\mu$ g/mL), as well as low concentrations of chitosan (39.06–78.13  $\mu$ g/mL) and AgNO<sub>3</sub> (8–16  $\mu$ g/mL), while growth inhibition was high (97.55–98.32%). Combinations versus *S. epidermidis* reported good growth inhibition (87.95–98.4%) at relatively low concentrations of chitosan (20–40  $\mu$ g/mL), AgNO<sub>3</sub> (2–4  $\mu$ g/mL) and ZnO (10  $\mu$ g/mL).

Nisin–AgNO<sub>3</sub>–ZnO

Nisin, AgNO<sub>3</sub> and ZnO reported strong synergy versus *E. coli* (average 0.36), *S. aureus* (average 0.38) and *S. epidermidis* (average 0.53). Results versus *E. coli* show that very high concentrations of nisin (31.25  $\mu$ g/mL) yielded high synergy at low concentrations of AgNO<sub>3</sub> (0.5–1  $\mu$ g/mL) and ZnO (31.25–62.5  $\mu$ g/mL); however, growth inhibition did not exceed 77%. Combinations versus *S. aureus* reported moderate synergy at low concentrations of nisin (0.63–1.25  $\mu$ g/mL) and moderate concentrations of AgNO<sub>3</sub> (4–16  $\mu$ g/mL) and ZnO (39.06  $\mu$ g/mL). Combinations versus *S. epidermidis* reported the third highest average score synergy at low concentrations of nisin (1.25–2.5  $\mu$ g/mL), AgNO<sub>3</sub> (2–4  $\mu$ g/mL) and ZnO (5  $\mu$ g/mL); however, these combinations exhibited low-to-moderate growth inhibition (33.6–72.3%).



Figure 7. Cont.



**(B)** 

Figure 7. Cont.



Figure 7. Cont.

1.5

1.0

0.5

0.0

Combo 3

Synergy score



(D)

Figure 7. Cont.





(E)

Figure 7. Cont.





**Figure 7.** Synergy score results of the top three two-drug combinations: Bar graphs that present the concentrations of each drug and their respective synergy scores from the top three two-drug combinations (**A**–**F**) against *E. coli, S. aureus* and *S. epidermidis*. (**A**) AgNO<sub>3</sub>/Chitosan, (**B**) Nisin/AgNO<sub>3</sub>, (**C**) AgNO<sub>3</sub>/ZnO, (**D**) Nisin/Chitosan, (**E**) Chitosan/ZnO and (**F**) Nisin/ZnO. Bars show drug concentrations, as indicated on the left *y*-axis, and the line/symbols show each combination's (combo) synergy score, as indicated on the right *y*-axis.

Nisin–Chitosan–ZnO

Nisin, chitosan and ZnO reported low inhibition synergy versus *E. coli* (average 0.23) and *S. epidermidis* (average 0.21); however, combinations versus *S. aureus* reported the highest synergy score across all three-drug combinations (average 0.83). Concentrations of the reported combinations versus *E. coli* indicated poor synergy between treatments, as high concentrations of chitosan (9.77–312.5  $\mu$ g/mL) and ZnO (31.25–125  $\mu$ g/mL) were utilised. Combinations that used low concentrations of each drug exhibited very low inhibition (26.6%).

The highest-scoring combination (0.93) versus *S. aureus* reported low concentrations of nisin (3.91  $\mu$ g/mL), chitosan (39.06  $\mu$ g/mL) and ZnO (62.5  $\mu$ g/mL) with high inhibition (98.7%). The second highest-scoring combination (0.84) also reported low concentrations of nisin (0.977  $\mu$ g/mL), chitosan (156.25  $\mu$ g/mL) and ZnO (62.5  $\mu$ g/mL) with high inhibition (99.4%). While the third highest-scoring combination reported low concentrations of nisin (0.977  $\mu$ g/mL), chitosan (78.13  $\mu$ g/mL) and ZnO (62.5  $\mu$ g/mL), the reported growth

inhibition was moderate (71.5%). Moreover, while results versus *S. epidermidis* reported low concentrations of nisin (0.98–1.95  $\mu$ g/mL), chitosan (39.06  $\mu$ g/mL) and ZnO (31.25  $\mu$ g/mL), along with high growth inhibition (99%), there was little synergy observed, as denoted by the two highest scores. The third highest-scoring combination reported a very high concentration of chitosan (625  $\mu$ g/mL).



Chitosan/AgNO3/Nisin vs S. epidermidis



Chitosan
 AgNO<sub>3</sub>
 Nisin
 Synergy Score

(A)

Figure 8. Cont.







Chitosan/AgNO3/ZnO vs. E. coli 100 1.5 80 60 Concetration (µg/mL) 1.0 Synergy score 40 20 5 4 3. 2 1 0 0.0 Combo 1 Combo 2 Combo 3

Chitosan/AgNO<sub>3</sub>/ZnO vs. S. epidermidis



Figure 8. Cont.







80-

60

40·





Nisin/AgNO<sub>3</sub>/ZnO vs S. epidermidis



Figure 8. Cont.

Nisin/AgNO<sub>3</sub>/ZnO vs S. aureus





(**C**)





**Figure 8.** Synergy score results of top three three-drug combinations: Bar graphs presenting the concentrations of each drug and their respective synergy scores from the top three three-drug combinations (**A**–**D**) against *E. coli, S. aureus* and *S. epidermidis*. (**A**) Chitosan/AgNO<sub>3</sub>/Nisin, (**B**) Chitosan/AgNO<sub>3</sub>/ZnO, (**C**) Nisin/AgNO<sub>3</sub>/ZnO and (**D**) Nisin/Chitosan/ZnO. Bars show drug concentrations as indicated on the left *y*-axis and the line/symbols show each combination (combo) synergy score as indicated on the right *y*-axis.

# 3.3. Four-Drug Combinations

AgNO<sub>3</sub>–Nisin–Chitosan–ZnO

Chitosan, nisin, AgNO<sub>3</sub> and ZnO exhibited moderately high synergy in combination versus *E. coli* (average 0.36) and very high synergy versus *S. aureus* (average 0.91) and *S. epidermidis* (average 1.11). While the compound's average synergy score versus *E. coli* is lower than that versus the other two test species, results indicate the positive contributions of each treatment at low concentrations of chitosan (80  $\mu$ g/mL), nisin (1.95–31.25  $\mu$ g/mL), AgNO<sub>3</sub> (8  $\mu$ g/mL) and ZnO (10–40  $\mu$ g/mL), which showed effective inhibition (69–98.9%). The highest-scoring combination, which exhibited 98.9% inhibition, reported a very high concentration of nisin (31.25  $\mu$ g/mL), indicating that nisin had a strong influence within the combination.

0

Combo 1

Reported synergy scores versus S. aureus are quite high, with the top scoring combinations exhibiting effective inhibition (97.5–99.5%) at low concentrations of chitosan  $(19.53-78.13 \ \mu g/mL)$ , nisin  $(0.39-1.56 \ \mu g/mL)$  and AgNO<sub>3</sub> (4  $\mu g/mL)$  and moderate concentrations of ZnO (62.50  $\mu$ g/mL).

Combinations versus S. epidermidis reported the highest synergy scores identified within the present study at low concentrations of chitosan (20–80  $\mu$ g/mL), nisin (1.25  $\mu$ g/mL), AgNO<sub>3</sub> (8  $\mu$ g/mL) and ZnO (10–40  $\mu$ g/mL). While the top combination reported a very high synergy score (1.3), the reported inhibition greatly deviated (stdev 69.18), having an average value of 24.96%. The second highest (1.13) and third highest (0.9) scoring combinations exhibited stable inhibition (98.7–99.1%) at similarly low concentrations of each treatment.



Chitosan/Nisin/AgNO<sub>3</sub>/ZnO vs E. coli

Chitosan/Nisin/AgNO<sub>3</sub>/ZnO vs S. aureus



Figure 9. Synergy score results of the top three four-drug combinations: Bar graphs that present the concentrations of each drug and their respective synergy scores from the top three four-drug combinations (Chitosan/Nisin/AgNO<sub>3</sub>/ZnO) against E. coli, S. aureus and S. epidermidis. Bars show drug concentrations, as indicated on the left *y*-axis, and the line/symbols show each combination (combo) synergy score, as indicated on the right y-axis.

# 4. Discussion

Combo 2

Antimicrobial synergy holds great promise as a solution for use in meeting the AMR crisis for several reasons. While a bacterial species may hold or even develop resistance to a single therapeutic agent, co-treatment with an alternative compound that exhibits

0.0

Combo 3

alternative modes of antimicrobial action could help to alleviate this issue. Additionally, certain groups of bacteria hold intrinsic metabolic or physical characteristics that can prevent certain classes of antimicrobials from exhibiting their effect. Co-treatment using a compound that can disrupt these characteristics would allow the primary treatment to carry out its effect unimpeded. Gram-negative bacteria are an example of one such group, as they have an additional outer membrane that can act to prevent compounds from reaching their target ligands. Following this example, nisin is a poly-cyclic lantibiotic that targets the inner-membrane-bound lipid II molecule. Due to the presence of an outer membrane, nisin is prevented from reaching its target, thus rendering it ineffective [29,30]. However, in theory, it would be possible to enable nisin by combining it via treatment with an additional compound that targets the outer membrane. By removing the outer membrane or compromising its integrity, nisin could freely interact with its lipid-II target. While this interaction can be clearly deemed to be synergistic, it is not enough on its own to observe a positive end result from the combination. While it would stand to reason that combining two or more already well-known and effective treatments would produce a greater gross effect than that of each individual treatment, previous studies of drug combinations have shown this predicted outcome to be incorrect, as have the results presented in this study [16,20,31]. To determine the synergistic abilities of two or more compounds, it is necessary to assess an array of various concentrations in different combinations. It is not important to determine the highest effect of combined treatments; the concept is to instead determine combinations that express a higher effect than that of the individual drugs at an identical concentration. The aim is to more easily discern the ratio of each drug required to enable another's mechanism of action, thus giving the most efficient synergy.

# 4.1. Inhibition and Synergy

Previously,  $AgNO_3$  was shown to be the most effective bacterial growth inhibitor of the tested bacterial species [22]. Nisin was shown to have very efficient inhibitory effects on test Gram-positive bacterial species, while having no effect on Gram-positive bacteria. Both compounds differ majorly in their modes of action, with AgNO<sub>3</sub> permeating bacterial membranes through reactive silver ions  $(Ag^{2+})$ , while nisin has specific binding affinity to the lipid-II molecules bound in the inner bacterial membrane. Nisin's inability to affect Gram-negative bacteria is based on its inability to breach its outer membrane and interact with the lipid-II ligand. By combining both AgNO<sub>3</sub> and nisin, it was hypothesised that the reactive Ag<sup>2+</sup> ions of AgNO<sub>3</sub> breach the Gram-negative outer bacterial membrane, allowing nisin to reach its target ligand [32–35]. Similar hypotheses were devised regarding ZnO and nisin, as ZnO had efficacy against Gram-negative and Gram-positive bacteria, as well as a similar mode of action wherein it destabilises membranes through release of  $Zn^{2+}$  ions and reactive oxygen species (ROS) [36–39]. Chitosan also had a noteworthy effect on all test strains, though it also had an alternate mechanism via which it targeted the bacterial cell wall [40,41]. The varying mechanisms had great significance for combinational studies and allowed us to observe whether effects unlock one other treatments' drawbacks (AgNO<sub>3</sub>-nisin, ZnO-nisin), stack upon against them (AgNO<sub>3</sub>-ZnO) or complement them (chitosan–nisin, chitosan–AgNO<sub>3</sub>, chitosan–ZnO).

In this study, the combinatory compatibility of four chosen bioactives was successfully established, as were the magnitude of their interactions with one another. The checkerboard assay was utilised to screen the inhibitory effect of bioactive combinations against each test bacterial strain. The checkerboard assay was a well-established method used to screen drug combinations in various areas of clinical research [12,15–18]. Through the use of the synergy python package and the Bliss model, the synergy score of each test combination was successfully determined. The results of this study presented interesting interactions between the bioactives, many of which were predictable, though others were unanticipated.

# 4.2. Two-Drug Combination Synergy

The results of two-drug combination studies carried out against the Gram-negative bacteria E. coli have yielded varying results. Nisin, which is a lantibiotic that targets the inner-membrane-bound lipid II molecule, is hindered by the outer membrane found in Gram-negative bacteria, which prevents nisin from carrying out its mechanism of action. It was hypothesised that combining nisin with a compound capable of penetrating the outer membrane, such as AgNO<sub>3</sub> or ZnO, would enable nisin, with the resulting interaction being marked as synergetic. Combinations of nisin–AgNO<sub>3</sub> exhibited moderate-to-high synergy (average 0.32), showing a consistent concentration of AgNO<sub>3</sub> (8.49  $\mu$ g/mL) to be the most accommodating compound for varying concentrations of nisin. While the inhibition ranged between 64 and 68% for this combination, it shows that nisin was able to have an effect upon a previously unaffected target. In contrast, ZnO was not found to enable nisin; rather, it appeared that nisin was antagonizing ZnO, as the concentrations of ZnO in the most synergistic combinations were higher than that of its previously determined MIC. Combinations of nisin-chitosan also exhibited undesired results, with higher concentrations of chitosan being utilised to observe an inhibitory effect. While such results are unfavourable, they still present a promising observation, showing that nisin had an effect on Gram-negative bacteria. Combinations of AgNO<sub>3</sub>-chitosan exhibited a strong synergistic interaction, with effects being evident at lower concentrations of AgNO<sub>3</sub>, which would indicate chitosan's ability to enable it. Chitosan has also shown to enable ZnO, which also exhibited lower concentrations; however, these combinations scored quite low, which reflects the fact that the concentration of chitosan was quite high.

Two-drug combinations used to inhibit *S. aureus* and *S. epidermidis* growth presented some moderate-to-strong synergistic combinations; however, there was a pattern of ZnO not effectively combining with nisin or chitosan. AgNO<sub>3</sub> demonstrates itself to be the most effect bioactive, enabling all other bioactives with which it is combined, reporting lower concentrations with higher inhibition responses. The highest overall scoring two-drug combination involved nisin–AgNO<sub>3</sub> versus *S. epidermidis*. Chitosan also demonstrated notable synergy with most bioactives, though it only effectively combined with ZnO against *S. epidermidis*, as much lower concentrations of both gave a greater response; however, the inhibition response was weak.

# 4.3. Three-Drug Combinations against Gram-Negative Bacteria

Increasing the combination number can further alter the effect exhibited by treatments, as is evident from three-drug combinations. Combinations that included nisin were shown to demonstrate high synergy versus *E. coli* with near full inhibition. Following the twodrug analysis, it was predictable that chitosan–AgNO<sub>3</sub>–nisin would effectively synergise, presenting the highest-scoring combination versus *E. coli*. Furthermore, the relatively high concentrations of nisin in this combination showed that it had an active effect on *E. coli*, as it can be presumed to be heavily involved in enabling resistance (i.e., a concentration close to 0 would indicate little-to-no input). A more unpredictable result was seen regarding combinations that involved ZnO, as two-drug combinations demonstrated ZnO to be a poor component in combination, though three-drug combinations showed opposing results. Nisin–ZnO was the lowest scoring combination versus *E. coli*; however, with the inclusion of AgNO<sub>3</sub> or chitosan, these combinations were the second and third highest-scoring threedrug combinations against *E. coli*, respectively. Most interestingly, the nisin–AgNO<sub>3</sub>–ZnO combination highlights how little AgNO<sub>3</sub> was reported in the higher scoring combinations, while relatively high concentrations of nisin were reported. This result again indicates nisin's active role in the combination, whereas  $AgNO_3$  is at too low a concentration to have an inhibitory effect. This result could also demonstrate the ability of  $AgNO_3$  to enable the mechanism of nisin. While chitosan–AgNO<sub>3</sub>–ZnO reported low scoring combinations, the results seemed to be promising, with low concentrations of all three bioactives and nearly full inhibition reported. This result, once again, does not follow the patterns observed in two-drug combinations of the same bioactives.

## 4.4. Three-Drug Combinations against Gram-Positive Bacteria

Three-drug combinations versus *S. aureus* and *S. epidermidis* offered interesting points of comparison. The chitosan–AgNO<sub>3</sub>–nisin combination scored highly versus both bacterial strains; however, concentrations of AgNO<sub>3</sub> were quite high at low concentrations of nisin. Scores from combinations that involved ZnO also proved to be quite unpredictable versus Gram-positive bacteria. Combinations of nisin–ZnO and chitosan–ZnO against *S. aureus* scored quite poorly; however, nisin–chitosan–ZnO reported the highest score of all three-drug combinations. In contrast, this combination had the second lowest score against *S. epidermidis*. The individual concentrations were quite low, and the reported synergy scores were also quite low. An interesting observation of this combination is that it was also predictable based on the two-drug combinations of nisin–chitosan, nisin–ZnO and chitosan–ZnO, which produced synergy scores that very closely averaged that of the nisin–chitosan–ZnO synergy score. Concentrations versus *S. aureus* indicate that ZnO enabled the effects of chitosan and AgNO<sub>3</sub>; however, concentrations versus *S. epidermidis* did not indicate that any single bioactive enabled another bioactive, highlighting the even distribution of activity between the three bioactives.

Nisin–AgNO<sub>3</sub>–ZnO demonstrated strong synergy versus both Gram-positive bacteria. While the reported synergy was particularly high against *S. epidermidis*, the reported inhibitory effects were quite low. Two-drug reports show that nisin–ZnO combinations interacted very poorly, which implies that the influence of AgNO<sub>3</sub> caused the three-drug combinations to more favourably interact, which was also predictable given the synergy scores of nisin–AgNO<sub>3</sub> and AgNO<sub>3</sub>–ZnO.

# 4.5. Four-Drug Combinations

While a four-drug combination gave valuable insights into interactions between all four treatments at once, the  $4 \times 4$  sized checkerboard had some disadvantages relative to larger  $6 \times 6$  or  $8 \times 8$  checkerboards, primarily the fact that it could not accommodate enough combinations to generate a full model of the possible combinational interactions. However, if determined via a two- or three-drug assay, key concentrations can be selected and utilised within the four-drug assay for further investigation. As such, the present four-drug assay layout should be used as a follow-on study, rather than as an initial combinational study, due to such limitations. Likewise, with the chosen three-drug assay layout, the four-drug assay layout allowed reduced experimental size and a faster setup.

Four-drug combinations reported a marked increase in the efficacy of all four bioactives relative to their individual capabilities against each bacterial strain. The combination of chitosan–nisin–AgNO<sub>3</sub>–ZnO against *E. coli* exhibited some predictable results, with synergy scores comparable to scores derived from two- and three-drug combinations. While concentrations of each bioactive in the highest-scoring combinations were lower than their individual MICs, concentrations of chitosan and AgNO<sub>3</sub> were still quite moderate. Furthermore, only the highest-scoring combination reported complete inhibition while using a high concentration of nisin, which was expected due to nisin's inability to target Gram-negative species.

The reported four-drug synergy scores against *S. aureus* and *S. epidermidis* were very high relative to other scores determined during this study. Concentrations versus *S. aureus* were notably lower than their individual MICs, thus having strong inhibitory effects. While *S. epidermidis* reported the highest synergy score of this study, its highest-scoring combination reported low inhibition. Compared to the other two reported combinations, a slight increase in either chitosan or ZnO was sufficient to push the effects toward complete inhibition, albeit remaining well below their individual MICs. From analysis of the most effective combinations against each individual bacterial strain, the most effective concentrations of each drug in combination were determined, which gave a four-drug combination that could cause complete inhibition against all tested bacterial strains, while the amount of each drug used was limited (See Table 1).

**Table 1.** Most effective concentrations of the four bioactive compounds in combination against *E. coli, S. aureus* and *S. epidermidis*. These concentrations were established by evaluating the highest-scoring combinations against each bacterial species and determining the lowest concentrations of each compound that would cause complete inhibition of all three bacterial species.

Bioactive	Most Effective Concentration (µg/mL)
Chitosan	80
Nisin	2
AgNO <sub>3</sub>	8
ZnO	60

# 5. Conclusions

Combinational antimicrobial bioactive studies hold great potential in many areas of clinical, pharmaceutical and medical research. The discovery of cross-treatment synergism could potentially unlock many new avenues of therapy for various pathogens and conditions. While drug synergy is a key target of combinational studies, drug antagonism is also a well-documented occurrence in pharmaceuticals, and while there are several models under development for its prediction, in many cases, it is difficult to determine which treatments may interact negatively without performing pre-clinical or clinical studies. Though it is important to find compatible combinations of drugs, another key goal is finding combinations in which the individual drugs are more effective within a combination than they are when acting alone. Determining synergy scores is an efficient method of screening many combinations of treatments and deducing the most effective option. It is evident from results presented here that treatment interactions cannot be accurately predicted and can differ greatly between bacterial strains. Furthermore, increasing the combination number has also been shown to have an unpredictable effect, as two-drug combinations cannot predict the effects of three-drug combinations of the same components, and likewise, twoand three-drug combinations cannot predict the effects of four-drug combinations. Current findings show that models previously used to predict drug combinations cannot be wholly trusted, as there are aberrant results presented in this study that contradict predictive models. While such results may not be considered advantageous, they provide knowledge critical to the development of combinational treatments. Within the scope of investigating new or previously unsuitable compounds as alternative antimicrobial treatments, the possibility of using as little of each compound possible while still holding an antimicrobial effect holds great potential. Identification of antimicrobial compounds based on complementary modes of action also provides an additional avenue for the discovery of novel treatments that combat AMR bacteria, wherein the activity of one treatment may enable the activity of another treatment that is naturally ineffective against the bacteria, as was the case of nisin against Gram-negative species in the present study.

There is also a pressing need to use effective screening methods to evaluate alternative bioactives that are biorefined via various environmental and food waste streams to help address the shortage in appropriate antimicrobials and the development of AMR [42,43]. Interestingly, there is increased interest in exploring new sources of antimicrobials, such as marine, peatland and food waste streams, that may present stressful environments that favor the production of unique antimicrobial bioactives [44]. This simple mass-throughput screening-based approach to evaluating combinational bioactives will also help address the surge in resistance to anti-fungal drugs among problematical fungi that cause significant human and animal infections [45,46]. There is also a proportionate interest in progressing interdisciplinary research through Quadruple Helix Hub frameworks (combining academia, industry, society and regulators), which use shared access to specialist equipment and subject-matter experts across disciplines to overcome these challenges [47].

The four chosen bioactives, namely AgNO<sub>3</sub>, ZnO, nisin and chitosan, which have previously been characterised in terms of their individual antimicrobial abilities, have now been characterised for their combinational interactions in two-, three-, and four-drug

arrangements. Using this data, we accurately determined the most effective concentrations of each compound required for most effective microbial growth inhibition, limiting the amount of each compound needed to inhibit bacterial growth while having a broader spectrum of effect. Using this data, it is clear that the use of these compounds in combination produces a much more effective inhibitor of microbial growth, as the concentration of each individual bioactive is much lower than that of their MIC alone. As the current study relied on combinations of serial dilutions, more extensive analyses must be carried out regarding the most synergistic combinations, with further testing used to determine the most ideal concentration combinations and verify their efficacy. Additionally, the methods and analysis procedures presented have been shown to produce detailed and high-throughput assessments of drug combinations, and as such, they should be carried forward to evaluate additional treatments and combinations. The methodology could also be adjusted accordingly to allow studies of pharmaceuticals in other fields of research.

**Supplementary Materials:** The following supporting information can be downloaded via the following link: https://www.mdpi.com/article/10.3390/biomedicines11082216/s1, Figure S1. Drug combination setup, Figure S2. Two-drug combination assay final layout, Figure S3. Three-drug combination assay final layout, Figure S4. Drug combination assay final layout; Table S1. Three highest Bliss scoring two-drug combinations against *E. coli*, Table S2. Three highest Bliss scoring two-drug combinations against *S. aureus*. Table S3. Three highest Bliss scores of each two-drug combination versus *S. epidermidis*. Table S4. Three highest Bliss synergy scores of each three-drug combination versus *E. coli*, *S. aureus* and *S. epidermidis*. Table S5. Three highest Bliss synergy scores and the average of each four-drug combination versus *E. coli*, *S. aureus* and *S. epidermidis*.

**Author Contributions:** Conceptualisation, K.M., I.M. and N.R.; methodology, K.M.; investigation, K.M.; writing—original draft preparation, K.M.; writing—review and editing, all authors; supervision, N.R., I.M. and M.L.; funding acquisition, N.R. All authors have read and agreed to the published version of the manuscript.

**Funding:** The authors acknowledge funding support provided by DAFM-2022-PSS125 and Athlone Institute of Technology (grant No. AIT-2017-008) for this work.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

**Conflicts of Interest:** The authors declare no conflict of interest.

# References

- Hsia, Y.; Lee, B.R.; Versporten, A.; Yang, Y.; Bielicki, J.; Jackson, C.; Newland, J.; Goossens, H.; Magrini, N.; Sharland, M. Use of the WHO Access, Watch, and Reserve classification to define patterns of hospital antibiotic use (AWaRe): An analysis of paediatric survey data from 56 countries. *Lancet Glob. Health* 2019, 7, e861–e871. [CrossRef] [PubMed]
- 2. Dadgostar, P. Antimicrobial resistance: Implications and costs. Infect. Drug Resist. 2019, 12, 3903–3910. [CrossRef] [PubMed]
- 3. Interagency Coordination Group on Antimicrobial Resistance. No Time to Wait: Securing the Future from Drug-Resistant Infections Report to the Secretary-General of the United Nations. 2019. Available online: https://www.who.int/antimicrobial-resistance/interagency-coordination-group/IACG\_final\_report\_EN.pdf?ua=1 (accessed on 29 May 2020).
- Talebi Bezmin Abadi, A.; Rizvanov, A.A.; Haertlé, T.; Blatt, N.L. World Health Organization Report: Current Crisis of Antibiotic Resistance. *Bionanoscience* 2019, 9, 778–788. [CrossRef]
- Koulenti, D.; Xu, E.; Yin Sum Mok, I.; Song, A.; Karageorgopoulos, D.E.; Armaganidis, A.; Lipman, J.; Tsiodras, S. Novel Antibiotics for Multidrug-Resistant Gram-Positive Microorganisms. *Microorganisms* 2019, 7, 270. [CrossRef]
- Tacconelli, E.; Carrara, E.; Savoldi, A.; Harbarth, S.; Mendelson, M.; Monnet, D.L.; Pulcini, C.; Kahlmeter, G.; Kluytmans, J.; Carmeli, Y.; et al. Discovery, research, and development of new antibiotics: The WHO priority list of antibiotic-resistant bacteria and tuberculosis. *Lancet Infect. Dis.* 2018, 18, 318–327. [CrossRef] [PubMed]
- Rima, M.; Rima, M.; Fajloun, Z.; Sabatier, J.M.; Bechinger, B.; Naas, T. Antimicrobial peptides: A potent alternative to antibiotics. *Antibiotics* 2021, 10, 1095. [CrossRef]
- 8. Girard, M.; Bee, G. Invited review: Tannins as a potential alternative to antibiotics to prevent coliform diarrhea in weaned pigs. *Animal* **2020**, *14*, 95–107. [CrossRef]
- Alekish, M.; Ismail, Z.B.; Albiss, B.; Nawasrah, S. In vitro antibacterial effects of zinc oxide nanoparticles on multiple drugresistant strains of *Staphylococcus aureus* and *Escherichia coli*: An alternative approach for antibacterial therapy of mastitis in sheep. *Vet. World* 2018, 11, 1428–1432. [CrossRef]
- Duss, F.R.; Garcia De La Mària, C.; Croxatto, A.; Giulieri, S.; Lamoth, F.; Manuel, O.; Miró, J.M. Successful treatment with daptomycin and ceftaroline of MDR *Staphylococcus aureus* native valve endocarditis: A case report. *J. Antimicrob. Chemother.* 2019, 74, 2626–2630. [CrossRef]
- Gonzales, P.R.; Pesesky, M.W.; Bouley, R.; Ballard, A.; Biddy, B.A.; Suckow, M.A.; Wolter, W.R.; Schroeder, V.A.; Burnham, C.A.D.; Mobashery, S.; et al. Synergistic, collaterally sensitive β-lactam combinations suppress resistance in MRSA. *Nat. Chem. Biol.* 2015, 11, 855–861. [CrossRef]
- 12. Fatsis-Kavalopoulos, N.; Roemhild, R.; Tang, P.C.; Kreuger, J.; Andersson, D.I. CombiANT: Antibiotic interaction testing made easy. *PLoS Biol.* **2020**, *18*, e3000856. [CrossRef]
- Acuña, L.; Morero, R.D.; Bellomio, A. Development of Wide-Spectrum Hybrid Bacteriocins for Food Biopreservation. Food Bioprocess Technol. 2011, 4, 1029–1049. [CrossRef]
- 14. Singh, S.B.; Young, K.; Silver, L.L. What is an "ideal" antibiotic? Discovery challenges and path forward. *Biochem. Pharmacol.* **2017**, 133, 63–73. [CrossRef]
- 15. Mataraci, E.; Dosler, S. In vitro activities of antibiotics and antimicrobial cationic peptides alone and in combination against methicillin-resistant *Staphylococcus aureus* biofilms. *Antimicrob. Agents Chemother.* **2012**, *56*, 6366–6371. [CrossRef]
- Torres, N.S.; Montelongo-Jauregui, D.; Abercrombie, J.J.; Srinivasan, A.; Lopez-Ribot, J.L.; Ramasubramanian, A.K.; Leung, K.P. Antimicrobial and antibiofilm activity of synergistic combinations of a commercially available small compound library with colistin against *Pseudomonas aeruginosa*. *Front. Microbiol.* 2018, *9*, 2541. [CrossRef]
- 17. Cokol-Cakmak, M.; Cetiner, S.; Erdem, N.; Bakan, F.; Cokolid, M. Guided screen for synergistic three-drug Combinations. *PLoS ONE* **2020**, *15*, e0235929. [CrossRef]
- Meletiadis, J.; Pournaras, S.; Roilides, E.; Walsh, T.J. Defining fractional inhibitory concentration index cutoffs for additive interactions based on self-drug additive combinations, Monte Carlo simulation analysis, and in vitro-in vivo correlation data for antifungal drug combinations against *Aspergillus fumi*. *Antimicrob. Agents Chemother.* 2010, 54, 602–609. [CrossRef]
- Bollenbach, T. Antimicrobial interactions: Mechanisms and implications for drug discovery and resistance evolution. *Curr. Opin. Microbiol.* 2015, 27, 1–9. [CrossRef]
- Yadav, B.; Wennerberg, K.; Aittokallio, T.; Tang, J. Searching for Drug Synergy in Complex Dose-Response Landscapes Using an Interaction Potency Model. *Comput. Struct. Biotechnol. J.* 2015, 13, 504–513. [CrossRef]
- 21. Tang, J.; Wennerberg, K.; Aittokallio, T. What is synergy? The Saariselkä agreement revisited. *Front. Pharmacol.* 2015, *6*, 181. [CrossRef]
- Masterson, K.; Meade, E.; Garvey, M.; Lynch, M.; Major, I.; Rowan, N.J. Development of a low-temperature extrusion process for production of GRAS bioactive-polymer loaded compounds for targeting antimicrobial-resistant (AMR) bacteria. *Sci. Total Environ.* 2021, 800, 149545. [CrossRef] [PubMed]
- Wooten, D.J.; Meyer, C.T.; Lubbock, A.L.R.; Quaranta, V.; Lopez, C.F. MuSyC is a consensus framework that unifies multi-drug synergy metrics for combinatorial drug discovery. *Nat. Commun.* 2021, 12, 4607. [CrossRef] [PubMed]
- Boekema, B. How to Prevent Silver Nitrate (AgNO<sub>3</sub>) Precipitating in Growth Broth?. ResearchGate. 2018. Available online: https://www.researchgate.net/post/How\_to\_prevent\_Silver\_Nitrate\_AgNO3\_precipitating\_in\_growth\_broth (accessed on 25 January 2018).
- Harrison, J.J.; Stremick, C.A.; Turner, R.J.; Allan, N.D.; Olson, M.E.; Ceri, H. Microtiter susceptibility testing of microbes growing on peg lids: A miniaturized biofilm model for high-throughput screening. *Nat. Protoc.* 2010, *5*, 1236–1254. [CrossRef] [PubMed]
- 26. Wiegand, I.; Hilpert, K.; Hancock, R.E.W. Agar and broth dilution methods to determine the minimal inhibitory concentration (MIC) of antimicrobial substances. *Nat. Protoc.* **2008**, *3*, 163–175. [CrossRef]
- 27. Brewster, J.D. A simple micro-growth assay for enumerating bacteria. J. Microbiol. Methods 2003, 53, 77–86. [CrossRef]
- 28. Wooten, D.J.; Albert, R. Synergy: A Python library for calculating, analyzing and visualizing drug combination synergy. *Bioinformatics* **2021**, *37*, 1473–1474. [CrossRef]
- 29. He, L.; Zou, L.; Yang, Q.; Xia, J.; Zhou, K.; Zhu, Y.; Han, X.; Pu, B.; Hu, B.; Deng, W. Antimicrobial Activities of Nisin, Tea Polyphenols, and Chitosan and their Combinations in Chilled Mutton. *J. Food Sci.* **2016**, *81*, 1466–1471. [CrossRef]
- 30. Ruhr, E.; Sahl, H.G. Mode of action of the peptide antibiotic nisin and influence on the membrane potential of whole cells and on cytoplasmic and artificial membrane vesicles. *Antimicrob. Agents Chemother.* **1985**, *27*, 841–845. [CrossRef]
- Chen, D.; Liu, X.; Yang, Y.; Yang, H.; Lu, P. Systematic synergy modeling: Understanding drug synergy from a systems biology perspective. BMC Syst. Biol. 2015, 9, 56. [CrossRef]
- Gut, I.M.; Blanke, S.R.; Van Der Donk, W.A. Mechanism of inhibition of bacillus anthracis spore outgrowth by the Lantibiotic Nisin. ACS Chem. Biol. 2011, 6, 744–752. [CrossRef]
- Pandian, S.R.K.; Deepak, V.; Kalishwaralal, K.; Viswanathan, P.; Sangiliyandi, G. Mechanism of bactericidal activity of silver nitrate—A concentration dependent bi-functional molecule. *Braz. J. Microbiol.* 2010, 41, 805–809. [CrossRef]
- 34. Prabhu, S.; Poulose, E.K. Silver nanoparticles: Mechanism of antimicrobial action, synthesis, medical applications, and toxicity effects. *Int. Nano Lett.* **2012**, 2012, 32. [CrossRef]

- 35. Van Heusden, H.E.; De Kruijff, B.; Breukink, E. Lipid II induces a transmembrane orientation of the pore-forming peptide lantibiotic nisin. *Biochemistry* **2002**, *41*, 12171–12178. [CrossRef]
- 36. Espitia, P.J.P.; Soares, N.d.F.F.; Coimbra, J.S.d.R.; de Andrade, N.J.; Cruz, R.S.; Medeiros, E.A.A. Zinc Oxide Nanoparticles: Synthesis, Antimicrobial Activity and Food Packaging Applications. *Food Bioprocess Technol.* **2012**, *5*, 1447–1464. [CrossRef]
- 37. Fiedot-Toboła, M.; Ciesielska, M.; Maliszewska, I.; Rac-Rumijowska, O.; Suchorska-Woźniak, P.; Teterycz, H.; Bryjak, M. Deposition of zinc oxide on different polymer textiles and their antibacterial properties. *Materials* **2018**, *11*, 707. [CrossRef]
- Pasquet, J.; Chevalier, Y.; Couval, E.; Bouvier, D.; Noizet, G.; Morlière, C.; Bolzinger, M.A. Antimicrobial activity of zinc oxide particles on five micro-organisms of the Challenge Tests related to their physicochemical properties. *Int. J. Pharm.* 2014, 460, 92–100. [CrossRef]
- 39. Pasquet, J.; Chevalier, Y.; Pelletier, J.; Couval, E.; Bouvier, D.; Bolzinger, M.A. The contribution of zinc ions to the antimicrobial activity of zinc oxide. *Colloids Surfaces A Physicochem. Eng. Asp.* **2014**, 457, 263–274. [CrossRef]
- 40. Kunjachan, S.; Jose, S. Understanding the mechanism of ionic gelation for synthesis of chitosan nanoparticles using qualitative techniques. *Asian J. Pharm.* **2010**, *4*, 148. [CrossRef]
- Qin, C.; Li, H.; Xiao, Q.; Liu, Y.; Zhu, J.; Du, Y. Water-solubility of chitosan and its antimicrobial activity. *Carbohydr. Polym.* 2006, 63, 367–374. [CrossRef]
- 42. Rowan, N.J.; Galanakis, C.M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? *Sci. Total Environ.* **2020**, *748*, 141362. [CrossRef]
- 43. Rowan, N.J.; Pogue, R. Editorial overview: Green new deal era—Current challenges and emerging opportunities for developing sustaining and disruptive innovation. *Curr. Opin. Environ. Sci. Health* **2021**, 22, 100294. [CrossRef]
- O'Neill, E.A.; McKeon Bennett, M.; Rowan, N.J. Peatland-based innovation can potentially support and enable the sustainable development goals of the United Nations: Case study from the Republic of Ireland. *Case Stud. Chem. Environ. Eng.* 2022, 6, 100251. [CrossRef]
- 45. Garvey, M.; Meade, E.; Rowan, N.J. Effectiveness of front line and emerging fungal disease prevention and control interventions and opportunities to address appropriate eco-sustainable solutions. *Sci. Total Environ.* **2022**, *851*, 158284. [CrossRef] [PubMed]
- Garvey, M.; Rowan, N.J. Pathogenic Drug Resistant Fungi: A Review of Mitigation Strategies. Int. J. Mol. Sci. 2023, 24, 1584. [CrossRef]
- Rowan, N.J.; Casey, O. Empower Eco multiactor HUB: A triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. *Curr. Opin. Environ. Sci. Health* 2021, 21, 100254. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



**ScienceDirect** 

### Efficacy of frontline chemical biocides and disinfection approaches for inactivating SARS-CoV-2 variants of concern that cause coronavirus disease with the emergence of opportunities for green eco-solutions

Neil J. Rowan<sup>1,2,a</sup>, Elaine Meade<sup>3</sup> and Mary Garvey<sup>3</sup>

#### Abstract

The emergence of severe acute respiratory disease (SARS-CoV-2) variants that cause coronavirus disease is of global concern. Severe acute respiratory disease variants of concern (VOC) exhibiting greater transmissibility, and potentially increased risk of hospitalization, severity and mortality, are attributed to molecular mutations in outer viral surface spike proteins. Thus, there is a reliance on using appropriate counter-disease measures, including non-pharmaceutical interventions and vaccination. The best evidence suggests that the use of frontline biocides effectively inactivate coronavirus similarly, including VOC, such as 202012/01, 501Y.V2 and P.1 that have rapidly replaced the wild-type variant in the United Kingdom, South Africa and Brazil, respectively. However, this review highlights that efficacy of VOC-disinfection will depend on the type of biocide and the parameters governing the activity. VOC are likely to be similar in size to the wild-type strain, thus implying that existing guidelines for use and re-use of face masks post disinfection remain relevant. Monitoring to avoid injudicious use of biocides during the coronavirus disease era is required as prolonged and excessive biocide usage may negatively impact our receiving environments; thus, highlighting the potential for alternative more environmentalfriendly sustainable biocide solutions. Traditional biocides may promote cross-antimicrobial resistance to antibiotics in problematical bacteria. The existing filtration efficacy of face masks is likely to perform similarly for VOC due to similar viral size; however, advances in face mask manufacturing by way incorporating new anti-viral materials will potentially enhance their design and functionality for existing and potential future pandemics.

#### Addresses

<sup>1</sup> Centre for Disinfection and Sterilisation, Athlone Institute of Technology, Dublin Road, Athlone, Ireland

<sup>2</sup> Department of Nursing and Healthcare, Athlone Institute of Technology, Dublin Road, Athlone, Ireland

<sup>3</sup> Department of Life Science, Institute of Technology, Sligo, Ash Lane, Sligo, Ireland

Corresponding author: Rowan, Neil J (nrowan@ait.ie)

<sup>a</sup> Given his role as Guest Editor Neil J. Rowan had no involvement in the peer-review of this article and has no access to information

regarding its peer-review. Full responsibility for the editorial process for this article was delegated to Damia Barceló Culleres.

Current Opinion in Environmental Science & Health 2021, 23:100290

This review comes from a themed issue on Environmental Toxicology 2021: Disruptive Green Deal Innovations

Edited by Neil J. Rowan and Robert Pogue

For a complete overview see the Issue and the Editorial

https://doi.org/10.1016/j.coesh.2021.100290

2468-5844/© 2021 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

#### Keywords

COVID-19, Disinfection, PPE, Biocides, SAR-Cov-2 variants, Healthcare, Environment.

# Introduction — coronavirus and its implications for maintaining healthcare provision

The coronavirus disease (COVID-19) pandemic, caused by the severe acute respiratory coronavirus 2 (SARS-CoV-2), has imposed tremendous challenges on healthcare systems globally [1-3]. At the time of writing (30th March, 2021), there have been 127,628,928 cases of COVID-19 worldwide, including 2,791,055 deaths [4]. COVID-19 elicits a broad infection spectrum ranging from very mild, non-respiratory symptoms to severe acute respiratory illness, sepsis with organ dysfunction and death; however, some infected people can be asymptomatic [1]. Evidence highlighting the contributions of super-spreaders of infectious airborne viral particles, including the more transmissible SARS-CoV-2 variants of concern (VOC), has also contributed to the occurrence of third and fourth waves of COVID-19 infections [5-7]. Addressing the ongoing COVID-19 pandemic has created unprecedented logistical challenges to maintain critical supplies of single-use personal and protective equipment (PPE) [8,9], where reuse and

disinfection occurred under emergency use authorization. At the outset of COVID-19, there was a void in knowledge to effectively counter this disease; however, there is an increasing understanding of the potential role of different strategies to address COVID-19, including adopting non-pharmaceutical interventions (NPIs, such as correct wearing of face masks, hand hygiene, use PPE, maintaining social distancing, detection testing, contact tracing), along with delivering new vaccines [8,9]. Data generated from predictive mathematical modelling of multiple-contributing factors influencing the occurrence of COVID-19, and commensurate efficacy of disease counter-measures, is increasing, such information is translated to calculating risk probability to monitor and manage the basic reproduction number  $R_0$  in the following [9]. It is challenging to appreciate the actual efficacy of specific COVID-19 disease interventions in real-time given the swiftly moving pace of this pandemic. This current opinion focuses on understanding the efficacy of frontline biocides and disinfection approaches against SARS-CoV-2 variants of concern.

#### Coronaviruses and implications for meeting personal and protective equipment supply chain shortage and disinfection reprocessing

SARS-CoV-2 is a large positive-stranded RNA virus with an outer lipid envelope containing glycoprotein spikes (Figure 1) [10]. In general, enveloped viruses, such as coronaviruses, are more sensitive to environmental deleterious stresses, such as chemical biocides, than

Figure	1
rigure	

similarly-treated naked viruses, due to the presence of a lipid membrane [9,11,12]. Coronaviruses range from 60 to 140 nm in size, which is below the 300 nm pore diameter used in multiple layers of material used to make face masks. However, the use of multiple layers in single-use plastic face mask reduces the probable risk of penetration and transmission by acting as a barrier to respiratory droplets [13,14], which is likely to apply similarly to mitigating against VOC transmission. The effectiveness of single-use plastic filtering-face piece respirators face masks varies based on type and certification that is defined across three levels of protection depending upon leakage of particles into the interior of the mask that are 22% (FFP1, such as medical and procedural masks), and 8% (FFP2, such as N95-type respirators), and 2% for non-disposable FFP3-type respirators [2]. Use of non-thermal biocidal and disinfection approaches, such as vaporized hydrogen peroxide (30-35% VH2O2) and moist heat (60-65 °C for 30 min), and ultraviolet light at 254 nm (or fluence at 2000 mJ/cm<sup>2</sup>), has been applied for reprocessing FFP1 and FFP2 type respirators, such as under emergency use authorization [2,25–27]. Non-thermal disinfection approaches of FFPs have been selected to enable retention of filtration performance, material compatibility, comfort fit, and pressure drop. In addition, there has been increased usage of alternative, cost-effective, home-made, cloth or fabric face coverings by the general public, where particle penetration efficacy was improved by using more than one layer of cotton-polypropylene and by introducing pleats when compared to testing using a fitted



Structural components of SARS-COV-2 (left) and effective biocidal agents known to deactivate the virus (right).

N95 FFP2-type control [15–17]. Combining similar mild heat conditions, along with the use of detergent, for reuse of face coverings is theoretically plausible for coronavirus-disinfection; however, there remains a lack of information on disinfection and efficacy for informing the frequency of reuse that maintains filtration functionality [2,9]. Post COVID-19, it is likely that the changes in medical practice will drive sustaining or increasing high demand for PPE.

# Understanding coronaviruses and the role biocides in breaking cycle of COVID-19 infection

Coronaviruses are typically inactivated on different surfaces within 4–5 days at ambient room temperatures on different surfaces, such as tissue, wood, glass, plastic, stainless steel, surgical masks, and paper, that can be influenced by humidity, viral load, presence of organic matter [9,18]; for example, colder conditions, such as refrigeration (4 °C), may extend SARS-COV-2 viability on surfaces beyond 14 days [19,20]. It took 14 days at 20 °C to reduce SARS-CoV-2 on nitrile gloves by 5 log orders using simulated typical infectious body fluids from infective patients; however, viral persistence was evident up to 21 days on plastic face shields, N100 FFPs, and polyethylene overalls [21]. These observations imply that the colder conditions associated with winter may support longer survival of SARS-CoV-2 on contact surfaces and when suspended in aerosols. There is a pressing need to understand the role of different interventions in breaking the cycle of SARS-CoV-2 infection (Figure 2) to protect frontline healthcare workers and patients [9]. This includes generating data over a longitudinal period to evaluate and harmonize deployment of these interventions (singly or combined) to prevent or reduce the risk of COVID-19 with a focus on

Figure 2



Use of chemical biocides and other disinfectants to break cycle of COVID-19 disease.

infections agent (SARS-CoV-2 and co-infections), reservoir, portal of exit, mode of transmission, portal of entry, and susceptible host (Figure 2). The efficacy of such data will be informed by mathematical modelling and randomized control studies [9]. The use of disinfectants or chemical biocides and disinfection approaches will feature strongly as key disease countermeasures in breaking the cycle of infection (Figure 2). Factors that influence the efficacy and performance of different types of biocides are varied and complex [22,23]; however, the parameters governing selection and performance of different biocide types are generally well-established, where the degree of application depends upon categories of risk to patients aligned with a commensurate level of treatment required to be achieved on contact surfaces and in the environment (Table 1). 'COVID-19 fatigue' of citizens is likely to play a contributing role in meeting compliance with deploying disease counter-measures to effectively manage new cases to protect frontline healthcare workers [24].

#### SARS-CoV-2 VOC

The World Health Organization monitors public health events associated with SARS-CoV-2 VOC [3]. Key features for these VOC are presented in Table 2. VOC 202012/01, 501Y.V2, and P.1 have commonly demonstrated an increase in transmissibility compared to wildtype (non-VOC) variants and have demonstrated a propensity for rapidly replacing other circulating SARS-CoV-2 strains. Variants 202012/01, 501Y.V2, and P.1 rapidly replaced the wild-type variant in the United Kingdom, South Africa, and Brazil, respectively. However, it is highly likely that more transmissible and pathogenic VOCs will emerge during this pandemic as the virus has ample opportunity to mutate given the high numbers of infected hosts globally. However, in terms of existing VOC, the WHO deploys 'a logistic model of competitive growth that highlighted additive increases in the effective reproduction number (Rt) relative to the wild-type variant that was estimated at 41% (95% CI: 41-42%) for 202012/01, 36% (95% CI: 32-40%) for 501Y.V2, and 11% (95% CI: 7-16%) for P.1' [3]. The transmissibility of P.1 is such that it is rapidly replacing the wild-type variant at a local level. Recent studies have shown VOC 202012/01 may be associated with an increased risk of hospitalization, severity, and mortality. There is a growing body of evidence on vaccine-induced neutralizing antibody activity against VOCs (Table 2). The findings support that neutralizing activity is largely sustained against this variant. However, these findings highlight the importance of using combinational approaches, including the use of biocides for surface disinfection, as important to limit transmission of VOCs. Key mutations affect viral nonstructural proteins that are unlikely to affect the efficacy of frontline biocides described in Tables 3 and 4. There is an increasing interest in the future proof design

of face masks by also incorporating potentially new antiviral materials with the provision for more environmentally friendly non-metal nanomaterials [66].

## Indication of biocide efficacy against coronavirus

Biocides encompass chemicals with antiseptic, disinfectant, and/or preservative activity (Table 3). Biocides are used for a broad range of purposes, 'usually with inanimate objectives (hard surface disinfectants), externally on the skin (antiseptics and topical antimicrobials), to prevent or limit microbial infection for preoperative skin infection or incorporated (preservatives) into pharmaceutical, cosmetic, or other types of products to prevent microbial contamination' [28]. Desirable properties of biocides include virucidal within the time that can allow for it to be in contact with materials to be treated: effectiveness not diminished under conditions of disinfection; does not damage material treated, has a suitable spectrum of activity; low toxicity and resistance to it has not emerged; inexpensive. Biocidal efficacy is influenced by several, and sometimes, inter-related factors - notably concentration, period of contact, pH, temperature, presence of organic or other interfering or enhancing materials or compounds, nature, numbers (dose), location (planktonic, biofilm), and condition of microorganisms (recalcitrant endospores vs sensitive enveloped viruses) (Table 3). For example, concentration exponent  $(\eta)$  is particularly important as it measures the effect of concentration of dilution based on the activity of the biocide [22,23]. Biocides with high  $\eta$ -values (such as alcohols, phenols) rapidly lose efficacy when diluted, whereas those with low n-values (such as QACs, chlorhexidine, orthopthalaldehyde) retain considerable activity on dilution. This difference is highly relevant when considering both lethal disinfection activity and potential implications on receiving environment, where the potentially deleterious impact of biocide residues must be considered [28]. In addition, many frontline biocides have optimal pH activity, such as hypochlorite and phenolics are most effective at acid pH, whereas glutaraldehyde and cationic biocides (e.g. QACS) are most potent at alkaline pH. Several researchers have reported that biocide activity can be influenced by interaction with organic matter (e.g. dirt, blood, serum, vomit, the presence of biofilm), and non-ionic surfaces, and adsorption on containers and other contact surfaces (Table 3).

Coronaviruses are incapable of supporting independent life; thus, biocide disinfection is determined by using *in vitro* bioassays, where reduction of cytopathic effects in tissue culture monolayers are observed that is attributed to a reduction in viral infectivity compared with untreated controls. Surviving viral fractions are typically expressed through  $Log_{10}$  reductions enumerated either by determining the 50% titration reduction endpoint for infectivity (known as tissue culture infectious dose 50%,

#### Table 1

#### Factors governing anti-viral efficacy of biocides.

Factors <sup>a</sup>	Comments	Relevance and implication for usage in practice
Factors characteristic of biocide		
Concentration	Understand the effect of dilution upon activity — concentration must be 'cidal' to viruses	Appropriate staff training
Contact time	Length of exposure can often enhance biocidal efficacy	Appropriate staff training
Organic load	Diminish the activity of biocide and protect other contaminating bacteria of concern	Understand physicochemical factors governing biocidal action
Formulation	Influences inactivation performance against coronaviruses and intended surface or application for treatment	Understand nation of active agent and impact on intended contact material
Temperature	Increased activity against viruses can be achieved with higher temperatures and relevant for some devices (e.g. endoscopes)	Appropriate staff training
pH	Affects biocide (stability and ionization) and affects growth of co-infective microorganisms	Less relevant for healthcare environment
Biological and environmental fact	ors	
Presence of biofilm	Provides protective menstrua or environment that can be found on equipment and in certain surfaces	Combine physical cleaning along with chemical action
Viral load	The greater the population number of viruses present the more difficult it can be to disinfect	Biocides often used in excess at high level concentration — SARS-CoV appear sensitive to low and moderate levels

Categories of risk as defined for patients and treatment of surfaces, equipment, environm	nent de la constant de la constant de la constant de la constant de la constant de la constant de la constant d
High risk	Sterilization such as use of VH2O2
Intermediate risk	Disinfection
Low risk	Cleaning and drying usually sufficient; disinfection
Requirements of chemical biocides or disinfectants	
Spectrum of activity	'Cidal' as opposed to 'static' activity as latter is not appropriate
Efficacy	Rapid action, particularly on surfaces
Incompatibility	Should not be neutralized or diminished easily
Toxicity	Should be minimal
Damages to surfaces, or materials	Corrosiveness should be minimal, especially at dilution of use. Should not damage contact surface to be disinfected
Costs	Should be affordable, particularly to ensure supply chain

<sup>a</sup> Factors listed in order of importance – adapted from the study by Michie, West, and Harvey [24].

#### Table 2

Synopsis of key information on SARS-CoV-2 variants of concern, as reported by World Health Organization on 23rd March, 2021.				
Emerging information <sup>a</sup>		Variant of concern (VOC)		
Next strain clade	20I/501Y.V1	20H/501Y.V2	20J/501Y.V3	
PANGO lineage	B.1.1.7	B.1.351	B.1.1.28.1 (alias P.10	
Alternate name	VOC 20201/01	VOC 202012/02		
First detected	United Kingdom	South Africa	Brazil	
First appearance	20 September, 2020	Early August, 2020	December 2020	
Key spike mutations	H69/V70 deletion; Y144 deletion; N501Y, A5700, P681H	L242/A243/L244 deletion; K417N E484K, N501Y	N417T, E484K; N501Y	
Key mutation in common		5106/G107/F108 deletion in Non-Structural Protein 6 (NSP6)		
Transmissibility	Increased (36–75%), increased secondary attack rate (10–13%)	Increased (1.50 (95% CI: 1.20-2.13) times more transmissible than previously circulating variant	Increased, more transmissible than previous circulating variants	
Severity	Possible increased risk of hospitalization, severity and mortality	Possible increased risk in hospital mortality by 20%	Under investigation, limited impact	
Neutralization capacity	Slight reduction, but overall neutralising titres still remained	Decreased, suggesting potential increased risk of infection.	Decreased reinfections reported	
Potential Impacts on vaccines	<ul> <li>No significant impact on post-vaccine neutralization by Moderna, Pfizer-BioNTech, Oxford-AstraZeneca, Novavax</li> <li>No significant change in prevention of disease by Oxford-AstraZeneca, Novavax, and Pfizer-BioNTech</li> <li>Evidence for prevention of infection evidence limited — reduced effect reported for Oxford-AstraZeneca</li> <li>125 (7)</li> </ul>	<ul> <li>Post-vaccine neutralization reductions range from minimal to moderate for Moderna and Pfizer; however, there is also some evidence of more substantial reductions</li> <li>Substantial reductions found for Oxford-AstraZeneca products</li> <li>There is no evidence to inform vaccine impact on asymptomatic infection by 501Y.V2</li> <li>73 (11)</li> </ul>	<ul> <li>Limited to modest reduction in post-vaccine neutralization by Oxford-Astrazeneca, Moderna, and Pfizer-BioNTech vaccines.</li> <li>Preliminary suggestion of loss of neutralization following vaccination with Sinovac</li> </ul>	
(newly reported in last week)	125 (7)	73 (11)	41 (0)	

<sup>a</sup> Adapted from WHO []; note, consult this reference report for more detailed information on emerging information on key VOC.

#### Table 3

General properties, strengths	and limitations of frontline chemical	biocides against coronaviruses.		
Biocide type and active igredient	Mechanism of virucidal Action	General usage	Limitations	Strengths
Alcohols Isopropyl alcohol (isopropanol) Ethyl Alcohol (Ethanol)	Disrupts cell envelope, coagulates and denatures proteins. Isopropyl alcohol is lipophilic disrupting lipid membrane.	Skin antisepsis (ca 70% v/v) Small equipment disinfection, for example, thermometers, critical tools, non-invasive probes	Not sporicidal Prolonged and repeat usage affects integrity of materials such as plastics. Flammable	No-staining, low toxicity, mild pleasant odour
Cationic surfactants — QAC such as BZK, MBAT, DDA	Mostly disrupt by solvating or disrupting cell envelope — cationic ammonium groups with hydrophilic heads	Fomites (200 ppm),	Require warmer temperature and longer periods for achieving MEC Low affinity against non-enveloped viruses	Nontoxic, colourless and odourless — retain activity in hard water, high tolerance to organic matter
Oxidising agent — Sodium hypochlorite	Oxidation of cell envelope	Household bleach — dissolves in water to form hypochlorous acid — used in clinical area for fomites, non-critical surfaces where there is blood spillage or vomit	Sensitive to presence of organic matter and porous material — can range from <1000 pm to 10,000 ppm depending on organic material — cleaning step and ventilation needed	Fast acting at low concentrations — inactivates envelope and non-envelope viruses
Oxidising agent Hydrogen peroxide	Hydroxyl free radicals cleave or crosslink biomolecules including proteins, nucleic acids, an lipids	Skin antisepsis (0.125% v/v); contact surfaces (35% v/v)	Limited information. Concentration of 0.5% effective against enveloped and non-envelop viruses.	Decomposes to form water and oxygen — effective against SARS-CoV-2 and surrogates — can be used on stainless steel
Halogenated compounds — Povidone iodine and Povidone Iodone (PVP-1)	Possibly blocking receptor for viral binding. lodine can inhibit viral enzymes (neuraminidase) essential for viral release from host	<ul> <li>PVP-1 (0.23%) used for rapid skin, oral cavity, nasal disinfection.</li> <li>Povidone iodine used at 7.5–10% pre-operative skin disinfection, antiseptic hand washes, scrubs, ointments</li> </ul>	Can be cytotoxic and cause skin irritancy — Is an iodophor is mixture of iodine and carrier polymerpolyvinl pyrrolidone — not suitable for use with silicone products	PVP-1 water soluble, stains can be removed by washing. Substitute or used in combination with for alcohol-based disinfection products.
Aldehydes Glutaraldehyde. Formaldehyde and OPA	Chemically alkylating the amino and sulfhydryl groups of proteins and amino groups of nucleic acid bases	High level broad spectrum virucidal disinfection — vaccine production — decontaminates of surgical equipment, endoscopes, dialysers.	High reactivity, hazardous to health — irritant. Apart from OPE, more reactive at alkaline conditions. Pungent odour <1 ppm, monitoring.	Rubber, plastics, lensed instruments are tolerant. OPA chemically stable over pH 3–9, non-irritant, stains skin wear PPE.

QAC – Quaternary ammonium compounds; BZK - benzalkonium chloride; mon; MBAT - biz(tri-methyl ammonium methylene chloride)-alkaly (C<sub>9-15</sub>) toluene; DDA – didecyldimethyl ammonium chloride; OPA – Ortho-phthalaldehyde or 1,2-dicarboxybenzaldehyde.

MEC -lowest concentration of biocide that reduces virus titre by 99.9% or greater compared to control reactions. Adapted from Lin [], Dev Kumar [].

Table 4						
Use of differen	Use of different disinfection approaches for inactivating SARS-CoV-2 and its' surrogate indicators.					
Disinfectant		Parameters	SARS-CoV-2 & Surrogate species	Reduction Assay used		
Chemical	Ethanol	60-70%, 1 min, hard surfaces, ceramic and porcelain tiles -carrier test.	hCoV (HCoV-229e)	3 - 4 log, TCID <sub>50</sub> assay [53]		
	$H_2O_2$	0.5%, 15 min, surface carrier test 1–6%, 30 s, suspension testing of oral mouth wash	SARS-CoV-2 SARS-CoV-2, USA–WA1/2020 strain	6 log plaque assay using Vero E6 cells [52] 1–1.8 log TCID assay using Vero 76 cells [54]		
	QAC – BAC QAC – DDAC	0.04% w/v, 1 min, steel surface. quantitative carrier test 0.025%, 3 days, suspension test	Parainfluenzavirus type 3 (HPIV-3) and human coronavirus 229-E (HOV-229E) Canine Coronavirus	3 log Plaque assay using MA-104 line of rhesus monkey kidney cells 4 log TCID assay using A72 fibroma cell line [55]		
	Sodium hypochlorite	0.1%, 1 min, suspension test. 6%, 30 s surface carrier test.	SARS-CoV-2	4 log [39] TCID assav [40]		
	IPA	70–90%, 30 s 70–80%, 15 s, ceramic and porcelain tiles -carrier test.	SARS-CoV-1 hCoV (HCoV-229e)	4 log, TCID assay [53]		
	Acetic acid	6%, 5 min, aqueous suspension test.	SARS-CoV-2 (Hu/DP/Kng/19-020 strain)	4 log TCID <sub>50</sub> assay using VeroE6-TMPRSS2 cells [43]		
	Glutaraldehyde Formaldehyde	0.5%, 2 min, suspension test 0.7–1%, 2 min, suspension test	SARS-CoV isolate FFM-1	3 log TCID using human embryonic lung fibroblasts		
	Povidone iodine	1–2.5%, 15 s, suspension testing of oral mouth wash	SARS-CoV-2, USA-WA1/2020 strain	4 log TCID assay using Vero 76 cells [54]		
Technologies	Steam sterilisation	121 °C, 5 min, medical masks, N95 respirators	Avian coronavirus (H120)	2 log inoculation of embryonated eggs, real-time TaoMan RT-PCR assay [56]		
	Heat	56 °C, 30 min, 65 °C, 15 min, 98 °C 2 min, suspension test.	SARS-CoV-2	5 log TCID <sub>50</sub> assay using Vero E6 cells [57].		
	Deep UV LED	265, 280, and 300 nm, 20 s, hard surfaces, carrier test	SARS-CoV-2	3.3 log Plaque assay using Vero E6 cells [58]		
	Simulated sunlight	60 min on hard surfaces, carrier test on surface dried droplets.	SARS-CoV-2 USA-WA1/2020	4 log. TCID <sub>50</sub> assay using Vero cells (ATCC CCL- 81) [60]		
	UVC	254 nm, 4-9 s, wet and dried droplets	SARS-CoV-2	3 log Plaque assav [61]		
	Ozone	30 ppmv, 40 min 100 ppmv, 30 min, 1000 ppmv, 20 min on surfaces, carrier test.	hCoV 229E (HuCoV-229E)	95% reduction (1 log) HEK-293 cells and imaging using IncuCyte ZOOM system [36]		
	Vapourised H <sub>2</sub> O <sub>2</sub>	0.5%, 60 s, surface of stainless steel disks, carrier test.	feline calicivirus, human adenovirus type 1, avian and swine influenza virus	4 log TCID50 assay using A-549 cells, MDCK cells [59]		
	Chlorine dioxide gas (ClO2)	30–300 ppm, 25 °C to 30 °C, 1.5–3 h, in vivo.	avian infectious bronchitis coronavirus	infected chick embryos as models [62]		
	Gamma radiation (cobalt-60)	1-5 MRads, suspension test.	arenavirus, bunyavirus, coronavirus, filovirus, flavivirus, orthomyxovirus, paramyxovirus	6 log TCID <sub>50</sub> assay Vero cells [63]		

TCID<sub>50</sub> assay - Tissue Culture Infectious Dose assay, QAC quaternary ammonia compound, BAC benzalkonium chloride, DDAC didecyl dimethyl ammonium chloride.

or TCID<sub>50</sub> assay) or by performing a viral plague assay: however, reverse transcriptase - polymerase chain reaction (RT-PCR) using threshold (Ct value) is also used to determine viral load through detection of specific genes. Determining factors influencing biocides efficacy has traditionally been conducted to evaluate minimum inhibitory concentrations or lethal effects such as European suspension test, rate-of-kill test, and in-use test that are more suitable for anti-bacterial agents [29]. The Sterilization industry relies upon 12 log<sub>10</sub> reductions of recalcitrant bacterial spores as biological indicators or surrogates, such as Geobacillus stearothermophilus, Bacillus atrophaeus, for determining sterility assurance levels for different sterilants where there is significant overkill to ensure validation of processes [25,30,31,32]. Thus, existing disinfection processes are ultimately based upon the probability of viral reduction where there is a pressing need to elucidate robust real-time inactivation enumeration methods [such as 31], which is likely to be informed by predictive modelling and may create opportunities for machine learning and artificial intelligence.

#### **Disinfection of SARS-CoV-2**

As an enveloped virus, SARS-CoV-2 is susceptible to commercial disinfectant chemicals, technologies, and physical disinfection methods [33,34] (Table 4). Recent studies have detected SARS-CoV-2 RNA on surfaces in isolation wards 28-days following exposure via RT-PCR methods [35], where the infectivity ability of viral RNA is unknown. However, it is unlikely that undamaged viral RNA realized on treated surfaces would remain a significant risk because of its inability to enter human lung cells as RNA only. Determination of biocidal efficacy against SARS-CoV-2 is not always feasible as the virus requires biosafety level 3 or higher; therefore, fewer pathogenic viruses as surrogate indicators of infectivity are frequently used [36]. Virucidal efficacy is determined by quantitative suspension tests, namely EN 14476 requiring 4 log reduction using surrogate enveloped species such as polio, adenovirus and murine norovirus, where efficacy against SARS-CoV-2 has yet to be elucidated experimentally. The framework of the European Committee for Standardization outlines surrogate species suitable for disinfection studies for many microorganisms. For virucidal activity against enveloped viruses, including SARS-CoV-2, the vaccinia virus has been specified as the relevant test organism according to this framework [37]. In clinical settings, SARS-CoV-2 has been detected on surfaces in intensive care units  $(4.4-5.2 \log_{10})$ , in isolation rooms, and on general wards  $(2.8-4.0 \log_{10})$  [38]. While the viral load of SARS-CoV-2 on fomites directly following contact with infected persons is currently unknown, it is known that the virus remains infectious on surfaces for up to 9 days [39,40] depending on the surface material, pH, temperature, and humidity [40]. The suitability of suspension tests to determine efficacy on surfaces is unknown, particularly where the organic matter may be present such as in healthcare settings. ISO 18184 is a surface carrier test for virucidal activity; at present, no studies have demonstrated biocidal efficacy against SARS-CoV-2 using this method. The disinfection of surfaces and hand sanitation is of paramount importance in controlling viral transmission as recommended by the WHO. Disinfectant efficacy is affected by viral type, organic matter, viral titre, pH, viral clumping, biocide contact time and concentration. As such, cleaning is an essential prerequisite for disinfection to remove contaminating organic matter. In healthcare settings, disinfection agents in use include high-pressure steam sterilization, dry heat, UV-light, ethylene oxide (EtO) gas, hydrogen peroxide gas plasma, and biocidal chemicals [41] (Table 4).

The environmental protection agency (EPA) has approved various chemical biocide for use domestically and clinically to reduce coronavirus transmission, including quaternary ammonium (QACs), hydrogen peroxide  $(H_2O_2)$ , peroxyacetic acid, isopropanol (IPA), ethanol, sodium hypochlorite, octanoic acid, phenolic, among others [41]. Viral inactivation is resultant from disruption of the cell structure, destruction of the lipid envelope, protein coagulation, nucleic acid, and protein denaturation [23] (Tables 3 and 4). Studies assessing disinfection efficacy are difficult to compare because of innate experimental variations and lack of standardized procedures, including test material, varied exposure times, viral load, test chemical or combinations and the organic inhibitor used [40] (Table 4). Studies report efficacy of 70-90% IPA with 30 s exposure against SARS-CoV with 1-3% H<sub>2</sub>O<sub>2</sub> demonstrating efficacy after 1 min exposure [42], while 0.1% sodium hypochlorite was effective in 1 min [39] (Table 4). A concentration of 6% acetic acid reduced human coronavirus (hCo-V) viability by 3.5 log<sub>10</sub> after 1 min contact time [43] on surfaces. 60-70% ethanol reduced surface viral load by >3  $\log_{10}$  after 1 min exposure in healthcare settings [44]. For the inactivation of SARS-CoV-2, the most common disinfectants used are 62-70% ethanol, 0.5% H<sub>2</sub>O<sub>2</sub>, and 0.1% sodium hypochlorite, which are effective with 1 min exposure via oxidative reactions [42]. Pulsed plasma gas discharge has also produced biocidal water comprising short-lived oxygenated free radicals that has potential contact surface disinfection leaving no unwanted chemical residues [45]. For hand disinfection, the WHO recommends the use of 75% IPA or 80% ethanol hand rubs for 30 s to inactivate SARS-CoV-2 [46]. However, there is increasing opportunities to exploit advances in digitization and modelling to inform the future efficacy of disinfectants, including combining biocidal approaches and to hurdle limitations for broad applications [9,47]. Material compatability and functionality are important factors to consider when using new eco-alternatives to conventional biocides that includes combinational treatments [64].

#### Conclusion

Variant strains of SARS-CoV-2 are constantly emerging due to innate mutagenic changes in the viral genome, primarily altering structural components such as spike proteins. Although the efficacy of vaccinations program against these variants is unknown, such structural changes are unlikely to confer disinfection resistance as non-specific destruction of proteins and lipids of the viral capsid occur. Frontline biocides appear to be affective against VOC, but factors governing usage needs careful consideration. Vigilance is needed to protect our environment during the COVID-19 era, particularly by avoiding injudicious use of biocide that may negatively impact our agroecosystems [48]. Prolonged and excessive biocide use may give rise to situations that potentially promote cross-antimicrobial resistance in problematical bacteria to frontline antibiotics [49]. Adaptive resistance to frontline biocides has been reported since the early 1990s such as against bisphenol, triclosan, glutaraldehyde, and oxidising agents [22]. Paul et al. [48] highlighted that extensive application of biocides affects microbial flora that may lead to a decrease in the number and diversity of beneficial microbes that may directly affect the functioning of nutrient cycles; thus, careful considerations should be given to biocide neutralization, environmental management and sustainability [50,51]. Understanding these factors is important for the training of end-users, (e.g. healthcare, industry and community), to ensure the efficacy of biocidal product is maintained and effectively neutralized, along with monitoring policy for effective and responsible deployment of biocides. This current opinion supports Article 18 of the European Union's biocidal products regulation that directs the European Commission to issue a report on how the biocidal products regulation contributes to the sustainable use of biocidal products to reduce the risks posed to human health, animal health, and the environment by biocidal products. The aforementioned also recommends a series of actions to be completed by 2024, including investing additional resources in enforcement activities; developing the best available techniques reference documents that can be relevant for biocidal products used in industrial processes, and encouraging the development and implementation of standards that could contribute to the sustainable use of biocidal products and alternatives to biocidal products.

#### Funding

No funding.

#### **Declaration of competing interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### References

Papers of particular interest, published within the period of review, have been highlighted as:

- \* of special interest
- \*\* of outstanding interest
- Rowan NJ, Laffey JG: Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic-Case study from the Republic of Ireland. *Sci Total Environ* 2020, **725**:138532, https://doi.org/ 10.1016/j.scitoteny.2020.138532.

Describes strategy, rationale and first use of non-thermal disinfection for meeting PPE including VH202, UV, mild heating and low level sodium hypochlorite biocidal washes. Highlights the need for strategic management of PPE supply chain and careful selection of disinfection that meets purpose and regulatory alignment, such as under Emergency Use Authorization.

 Rowan NJ, Laffey JG: Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from coronavirus disease (COVID-19) pandemic – implications for efficacy, re-use and sustainable waste management. *Sci Total Environ* 2021, **72**:1422259, https:// doi.org/10.1016/j.scitotenv.2020.142259.

Describes innovation for PPE reprocessing including alignment with disinfection with a focus on meeting ongoing pandemic needs and future sustainable waste management.

 World Health Organization: COVID-19 weekly epidemiological *update*. 2021. file:///C:/Users/Neil/Downloads/20210323\_Weekly\_ Epi\_Update\_32.pdf. Accessed 23 March 2021.

Monitors variants of concern and interest for SARS-CoV-2 by way of analysis of key information published internationally. Specifically addresses genomic variance, transmissibility, vaccine and diagnostic efficacy as it relates to viral mutational changes and potential impact on hospitalization, severity and mortality.

- European Centre for Disease Prevention and Control: COVID-19 situation week 12. updated 1<sup>st</sup> April, 20201. 2021. https://www. ecde.europa.eu/en/covid-19. Accessed 1 April 2021.
- World Health Organization: SARS-CoV-2 variants –emergencies preparedness response. 2020. https://www.who.int/csr/don/31december-2020-sars-cov2-variants/en/. Accessed 21 March 2021.
- World Health Organization: *Transmission of SARS-CoV2: implications for infection control prevention*. 2020. https://www.who.int/ news-room/commentaries/detail/transmission-of-sars-cov-2implications-for-infection-prevention-precautions. Accessed 25 March 2021.
- Nature communications: COVID research updates: new coronavirus variants spur multi-talented antibody response. 2021. https:// www.nature.com/articles/d41586-020-00502-w. Accessed 30 March 2021.
- Chu DK, Akl EA, Duda S, Solo K, Yaacoub S, Schunemann HJ: Physical distancing, face masks, and eye protection to prevent person-to-person transmission of SARS-CoV-2 and COVID-19: a systematic review and meta-analysis. Lancet 2020, 395:1973–1987.
- Rowan NJ, Moral RA: Disposable face masks and reusable face coverings as non-pharmaceutical interventions (NPIs) to prevent transmission of SARS-CoV-2 variants that cause Coronavirus disease (COVID-19): role of new sustainable NPI design innovations and predictive mathematical modelling. *Sci Total Environ* 2021:145530, https://doi.org/10.1016/ i.scitoteny.2021.145530.

First description of use of mathematical modelling to evaluate efficacy of non-pharmaceutical interventions for curtailing COVD-19 pandemic. Highlights challenges of relying upon limited data in the short term for risk mitigation where there is need to conduct randomised control trials with greater sampling data over a longitudinal period, which is difficult given the swiftly moving pace of a pandemic.

 Dev Kumar G, Mishra A, Dunn L, Townsend A, Oguaginma IC, Bright KR, Gerba CP: Biocides and novel antimicrobial agents for the mitigation of coronavirus. *Front Microbiol* 2020, https:// doi.org/10.3389/fmicb.2020.01351.

- Pinon A, Vialette M: Survival of viruses in water. Intervirology 2018, 61:214–222.
- Figueroa A, Hauck R, Saldias-Rodriguez J, Gallardo RA: Combination of quaternary ammonium and glutaraldehyde as a disinfectant against enveloped and non-enveloped viruses. J Appl Poultry Res 2017, 26:491–497.
- Chua MH, Cheng W, Goh SS, Kong J, Li B, Lim JYC, Mao J, Wang S, Xue K, Yang L, Ye E, Zhang K, Cheong WC, Tan BH, Li Z, Tan BH, Loh XJ: Face masks in the new COVID-19 normal: materials, testing and perspectives. AAAS Res 2020, https://doi.org/10.34133/2020/7286735.
- 14. Wibisono Y, Fadila CR, Saiful S, Bilad MR: Facile approaches of \* polymeric face masks reuse and reinforcements for microaerosol droplets and virus filtration: a review. *Polymers* 2020, 12:2516, https://doi.org/10.3390/polym12112516.

Reviews materials for sustainable face masks that will inform ongoing and future pandemics.

 Fischer EP, Fishcer MC, Grass D, Henrion I, Warren SW,
 \* Westmand E: Low cost measurement of facemask efficacy for following expelled droplets. *Sci Adv* 2020, https://doi.org/ 10.1126/sciadv.abd3083.

First to review different materials and design for face coverings used in COVID-19 pandemic. Highlighted the importance of using several layers and pleats to improve filtration performance where N95 FFP2 was used as test positive control for comparison.

- Sickbert-Bennett EE, Samet M, Clapp PW: Filtration efficiency of hospital face mask alternatives available for use during the COVID-19 pandemic. JAMA Intern Med 2020, https://doi.org/ 10.1001/jamaintermed.2020.4221.
- Asadi S, Cappak CD, Barreda S, Wexler A, Bouvier NM, Ristenpart WD: Efficacy of masks and face coverings in controlling outward aerosol particle emission from expiratory activities. *Sci Rep Nat Res* 2020, https://doi.org/10.1038/ s414598-020-72798-7.
- Derraik YGB, Anderson WA, Connelly EA, Anderson YC: Rapid review of SARS-CoV-1 and SARS-CoV-2 viability, susceptibility to treatment, and the disinfection and reuse of PPE, particularly filtering facepiece respirators. Int J Environ Res Public Health 2020, 17:6117, https://doi.org/10.3390/ ijerph17176117.

Comprehensive review of Coronavirus (SARS-CoV-1 and SARS-CoV-2) as it relates to viral survival on different contact surfaces and viral persistence on similar surfaces post disinfection treatments, which included UV irradiation and mild heating.

- Chin AWH, Chu JTS, Perera MRA, Hui KPY, Yen HL, Chan MCW, Peris M, Poon LLM: Stability of SARS-CoV-2 in different environmental conditions. *Lancet Microbe* 2020, 1:e10.
- Chan KH, Sridhar S, Zhang RR, Chu H, Fung AYF, Chan G, Chan JFW, To KKW, Hung IFM, Cheng VCC, Yeun KY: Factors affecting stability and infectivity of SARS-CoV-2. J Hosp Infect 2020, 106:226–231, https://doi.org/10.1016/j.jhin.2020.07.009.
- Klasoff SG, Strong JE, Funk D, Cutts TA: Stability of SARS-Cov-2 on critical personal protective equipment. *MedRxiv* 2020, https://doi.org/10.1101/2020.06.20128884.
- Maillard JY: Antimicrobial biocides in the healthcare envi ronment: efficacy, usage, policies and perceived problems. Therapeut Clin Risk Manag 2005, 1:307–320.

Describes putative involvement of biocides in conferring cross protection to frontline antibiotics.

- Lin Q, Lim JYC, Xue K, Yew PYM, Owh C, Chee PL, Loh XJ: Sanitizing agents for virus inactivation and disinfection. Views; 2020, https://doi.org/10.1002/viw2.16.
- Michie S, West R, Harvey N: The concept of "fatigue" in tackling covid-19. BMJ 2020, https://doi.org/10.1136/bmj.m4171.
- 25. Russo R, Levine C, Grady C, Pexioto B, McCormick-Ell J,
  \* Block T, Cresko A, Delmas G, Chitale P, Frees A, Riuz A, Alland D: Decontaminating N95 respirators during the COVID-19 pandemic: simple and practical approaches to increase decontamination capacity, speed, safety, and ease of use. J Hosp Infect 2021, 109:52–57.

Description of V2HO2 process for non-thermal processing of PPE under Emergency Use Authorization. PPE are designed for single use;

thus, there is careful consideration to maintaining filtration efficacy, material compatibility, pressure drop and comfort/fit post reprocessing. This is a very highly regulated industry/domain.

- de Man P, van Straten B, van den Dobbelsteen J, van der Eijk A, Horeman T, Koeleman H: Sterilization of disposable face masks by means of standardized dry and steam sterilization processes; an alternative in the fight against mask shortages due to COVID-19. *J Hosp Infect* 2020, https://doi.org/10.1016/ j.jhin.2020.04.001. S0195670120301766.
- Rubio-Romero JC, del Carmen M, Ferreira P, Torecilla JA, Santiago G, Castro C: Disposable masks, disinfection and sterilisation for reuse, and non-certified manufacturing. *Sci Total Environ* 2020, https://doi.org/10.1016/j.ssci.2020.104830.
- Russel AD: Introduction on biocides into clinical practice and the impact on antibiotic resistance bacteria. J Appl Microbiol 2003, 92:S121S-S135S.
- Environmental Protection Agency: Virucidal effectiveness testing:
   using feline calcivirus as surrogate for norovirus. 2017. https://www. epa.gov/pesticide-registration/virucidal-effectiveness-testing-usingfeline-calcilvirus-surrogate-norovirus. Accessed 30 March 2021. Important guidance information on use of viral surrogates to inform virucidal effectiveness of biocides.
- McEvoy B, Rowan NJ: Terminal sterilization of medical de- vices using vaporized hydrogen peroxide: a review of current methods and emerging opportunities. J Appl Microbiol 2019, 127:1403-1420.

First review to comprehensive compare and contrast development of nonthemal sterilisation modalities (VH2O2, electron beam, x-ray) for treatment of materials and devices. It is likely that this information informed adjacent needs for COVID-19 in terms of reprocessing of PPE where there is a gap in knowledge.

- **31.** Rowan NJ: **Pulsed light as an emerging technology to cause disruption for food and adjacent industries–Quo vadis?** *Trends Food Sci Technol* 2019, **88**:316–332.
- McEvoy B, Lynch M, Rowan NJ: Opportunities for the application of real-time bacterial cell analysis using flow cytometry for the advancement of sterilization microbiology. J Appl Microbiol 2020, https://doi.org/10.1111/jam.14876.

First to review potential use of flow cytometry to inform sterilisation modality including key challenges and emerging opportunities.

 Xiling G, Yin C, Ling W, et al.: In vitro inactivation of SARS CoV-2 by commonly used disinfection products and methods. Sci Rep 2021, 11:2418, https://doi.org/10.1038/ s41598-021-82148-w.

Reports on the investigation of disinfection products on SARS-CoV-2 inactivation using standard quantitative suspension tests. Optimal exposure times and concentrations for varying disinfection agents are elucidated providing key information in preventing disease transmission. The use of SARS-CoV-2 instead of a surrogate species makes this experimental data highly relevant and topical. Infectivity was determined via TCID50 assay using Vero-E6 cells which are derived from the kidney of an African green monkey and are used for the culture of viral species of medical importance.

- Al-Gheethi Adel, Al-Sahari, Mohammed, Abdul Malek Marlinda, Noman Efaq, Al-Maqtari Qais, Mohamed Radin, Talip Balkis A, Alkhadher Sadeq, Hossain Md Sohrab: Disinfection methods and survival of SARS-CoV-2 in the environment and contaminated materials: a bibliometric analysis. Sustainability 2020, 12:7378, https://doi.org/10.3390/su12187378.
- Zhou Yunyun, Zeng Yuyang, Chen Changzheng: Presence of SARS-CoV-2 RNA in isolation ward environment 28 Days after exposure. Int J Infect Dis 2020, https://doi.org/10.1016/ j.ijid.2020.06.015. S1201971220304513.
- Zucker I, Lester Y, Alter J, et al.: Pseudoviruses for the assessment of coronavirus disinfection by ozone. Environ Chem Lett 2021, https://doi.org/10.1007/s10311-020-01160.
- Steinhauer K, Meister TL, Todt D, Krawczyk A, Pasvogel L,
   Becker B, Paulma D, Bischoff B, Eggers M, Pfaender S, Brill FHH, Steinmann E: Virucidal efficacy of different formulations for hand and surface disinfection targeting SARS CoV-2. J Hosp Infect 2021, https://doi.org/10.1016/j.jhin.2021.03.015.

Infect 2021, https://doi.org/10.1016/j.jhin.2021.03.015. Investigates the virucidal efficacy of disinfectants which claim activity against enveloped viruses as specified in the European Standard EN 14476 and German national DVV/RKI guidelines. Surface biocides and hand sanitizers are described in terms of their anti-viral activity where studies conclude SARAS-CoV-2 has similar sensitivity to the standard test viral surrogate vaccinia as used in EN 14476 or DVV/RKI testing guidelines. The study is of significance as it elucidates the suitability of the recommended surrogate vaccinia to represent SARS-CoV-2 in disinfection studies.

 Marquès M, Domingo JL: Contamination of inert surfaces by \* SARS-CoV-2: persistence, stability and infectivity. A review. Environ Res 2021, 193:110559, https://doi.org/10.1016/ j.envres.2020.110559.

Outlines the importance of surfaces in viral transmission by reviewing relevant scientific literature. Studies describe the survival of SARS-CoV-2 on surfaces and inactivation using common disinfection solutions. Findings conclude that the virus can remain infective on surfaces for days depending on material type with copper surfaces the least hospitable material. Importantly studies also conclude that washing hands and regular disinfection should reduce coronavirus transmission.

 Kampf G, Todt D, Pfaender S, Steinmann E: Persistence of coronaviruses on inanimate surfaces and its inactivation with biocidal agents. J Hosp Infect 2020, https://doi.org/10.1016/ j.jhin.2020.01.022. S0195670120300463-.

Review of efficacy of frontline biocides for SARS-CoV-2 at outset of COVD-19 pandemic. This is possibly one of the most cited references in this domain that provides initial reference data on potential biocides to deploy to mitigation transmission of this virus in the environment.

 Pedreira A, Taskon Y, García MR: A critical review of disinfection processes to control SARS-CoV-2 transmission in the food industry. *Foods* 2021, 10:283, https://doi.org/10.3390/ foods10020283.

Summaries the best available literature to propose a holistic view for the disinfection process outlining optimal variable such as exposure time and concentration for the successful inactivation of SARS-CoV-2 and other pathogens. Pedreira et al., 2021 also considers the environmental impact of disinfection techniques where negative effect on the environmental is often likely following excessive disinfection protocols

- Rai Nagendra Kumar, Ashok Anushruti, Akondi Butchi Raju: Consequences of chemical impact of disinfectants: safe preventive measures against COVID-19. Crit Rev Toxicol 2020, https://doi.org/10.1080/10408444.2020.1790499.
- Al-Sayah MH: Chemical disinfectants of COVID-19: an overview. J Water Health 2020, 18:843–848, https://doi.org/10.2166/ wh.2020.108.
- Bedrosian N, Mitchell E, Rohm E, Rothe M, Kelly C, String G, Lantagne D: A systematic review of surface contamination, stability, and disinfection data on SARS-CoV-2 (through July 10, 2020). Environ Sci Technol 2020, https://doi.org/10.1021/ acs.est.0c05651.
- Lauritano Dorina, Moreo Giulia, Limongelli Luisa, Nardone Michele, Carinci Francesco: Environmental disinfection strategies to prevent indirect transmission of SARS-CoV2 in healthcare settings. *Appl Sci* 2020, 10:6291, https:// doi.org/10.3390/app10186291.
- 45. Hayes J, Kirf D, Garvey M, Rowan N: Disinfection and toxicological assessments of pulsed UV and pulsed-plasma gasdischarge treated-water containing the waterborne protozoan enteroparasite Cryptosporidium parvum. *J Microbiol Methods* 2013, 94:325–337.
- Kampf G, Brüggemann Y, Kaba HEJ, Steinmann J, Pfaender S, Scheithauer S, Steinmann E: Potential sources, modes of transmission and effectiveness of prevention measures against SARS-CoV-2. *J Hosp Infect* 2020, 106:678–697, https:// doi.org/10.1016/j.jhin.2020.09.022.
- Schrank CL, Minbiole KPC, Wuest WM: Are quaternary ammonium compounds, the workhorse disinfectants, effective against severe acute respiratory syndrome coronavirus-2? ACS Infect Dis 2020, 6:1553–1557.
- Paul D, Mondal SK, Mandal SM: Biologia Futura: use of biocides during COVID-19-global reshuffling of the microbiota. *Biol Futura* 2021, https://doi.org/10.1007/s42977-021-00069-1.

Reviews potential impact of chemical biocides on agrosystems including genotoxicity.

- Usman M, Farooq M, Hanna K: Environmental side effects of the injudicious use of antimicrobials in the era of COVID-19. *Sci Total Environ* 2020, 745, https://doi.org/10.1016/ j.scitotenv.2020.141053.
- Rowan NJ, Galanakis CM: Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: quo Vadis? Sci Total Environ 2020:141362, https://doi.org/10.1016/ j.scitotenv.2020.141362.
- 51. Rowan NJ, Casey O: Empower Eco Multi-Actor HUB: a triple \* helix "academia-industry-authority" approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations' Sustainable Development Goals framework. *Curr Opin Environ Sci Health* 2021, https://doi.org/10.1016/ j.coesh.2021.100254.

First to describe contribution of multi-actor hubs that are triple helix focused (academia-industry-authority) for testing new green innovation including disinfection technologies that are aligned with vision and needs of the Green Deal era. This included social, public and technological readiness levels and how this relates to UN Sustainable Development Goals.

- Mileto D, Mancon A, Staurenghi F, Rizzo S, Gismondo MR, Guidotti M: Inactivation of SARS-CoV-2 in the liquid phase: are aqueous hydrogen peroxide and sodium percarbonate efficient decontamination agents? ACS Chem Health Saf 2021, https://doi.org/10.1021/acs.chas.0c00095.
- Meyers C, Kass R, Goldenberg D, Milici J, Alam S, Robison R: Ethanol and isopropanol inactivation of human coronavirus on hard surfaces. J Hosp Infect 2021, 107:45–49, https://doi.org/ 10.1016/j.jhin.2020.09.026.
- Bidra Avinash S, Pelletier Jesse S, Westover Jonna B, Frank Samantha, Brown Seth M, Tessema Belachew: Comparison of in vitro inactivation of SARS CoV-2 with hydrogen peroxide and povidoneâ. Iodine oral antiseptic rinses. *J Prosthodont* 2020, 13220, https://doi.org/10.1111/jopr.13220.
- Schrank Cassandra L, Minbiole Kevin PC, Wuest William M: Are quaternary ammonium compounds, the workhorse disinfectants, effective against severe acute respiratory syndromecoronavirus-2? ACS Infect Dis 2020, https://doi.org/10.1021/ acsinfecdis.0c00265. acsinfecdis.0c00265-.
- Van Loon J, Veelaert L, Van Goethem S, Watts R, Verwulgen S, Verlinden JC, Du Bois E: Reuse of filtering facepiece respirators in the COVID-19 era. Sustainability 2021, 13:797, https:// doi.org/10.3390/su13020797.
- Auerswald H, Yan S, Dul S, In S, Dussart P, Martin NF, Karlsson EA, Garcia-Rivera JA: Assessment of inactivation procedures for SARS-CoV-2. J Gen Virol 2021, https://doi.org/ 10.1099/jgv.0.001539.

Describes the importance of viral handling and biosafety measures to ensure safety during disinfection studies. Inactivation data relating to the efficacy of several chemical and physical methods are investigated against SARS-CoV-2 isolates from Cambodia. Studies conclude that chemical (AVL, inactivating sample buffer and formaldehyde) and heattreatment (56 and 98 °C) methods completely inactivated viral loads of up to 5 log10 ensuring operator safety during experimental procedures and outline effective control measures for disease transmission.

- Minamikawa T, Koma T, Suzuki A, et al.: Quantitative evaluation of SARS-CoV-2 inactivation using a deep ultraviolet lightemitting diode. Sci Rep 2021, 11:5070, https://doi.org/10.1038/ s41598-021-84592-0.
- Goyal SM, Chander Y, Yezli S, Otter JA: Evaluating the virucidal efficacy of hydrogen peroxide vapour. J Hosp Infect 2014, 86:255–259, https://doi.org/10.1016/j.jhin.2014.02.003.
- Ratnesar-Shumate Shanna, Williams Gregory, Green Brian, Krause Melissa, Holland Brian, Wood Stewart, Bohannon Jordan, Boydston Jeremy, Freeburger Denise, Hooper Idris, Beck Katie, Yeager John, Altamura Louis A, Biryukov Jennifer, Yolitz Jason, Schuit Michael, Wahl Victoria, Hevey Michael, Dabisch Paul: Simulated sunlight rapidly inactivates SARS-CoV-2 on surfaces. J Infect Dis 2020, https://doi.org/10.1093/infdis/jiaa274. jiaa274.

- Storm N, McKay LGA, Downs SN, *et al.*: Rapid and complete inactivation of SARS-CoV-2 by ultraviolet-C irradiation. *Sci Rep* 2020, 10:22421, https://doi.org/10.1038/s41598-020-79600-8.
- Zambrano-Estrada X, Dominquez-Sanchez C, Banuet-Martinez M, Guerrero de la Rosa F, Garcia-Gasca T, Prieto-Valiente L, Acevedo-Whitehouse K: Evaluation of the antiviral effect of chlorine dioxide (CIO2) using a vertebrate model inoculated with avian coronavirus. *BioRxiv* 2020, https:// doi.org/10.1101/2020.10.13.336768. 2020.10.13.336768.
- Feldmann F, Shupert WL, Haddock E, Twardoski B, Feldmann H: Gamma irradiation as an effective method for inactivation of emerging viral pathogens. *Am J Trop Med Hyg* 2019, 100:1275–1277, https://doi.org/10.4269/ajtmh.18-0937.
- Zhou J, Hu Z, Sabihi F, Chen Z, Zhu M: Progress and perspectives on antiviral protective materials. Adv Fibr Mater 2020, 2:123–129.





Elaine Meade <sup>1</sup>, Sarah Hehir <sup>1,2</sup>, Neil Rowan <sup>3</sup> and Mary Garvey <sup>1,2,\*</sup>

- <sup>1</sup> Department of Life Science, Sligo Institute of Technology, F91 YW50 Sligo, Ireland; elaine.meade@mail.itsligo.ie (E.M.); hehir.sarah@itsligo.ie (S.H.)
- <sup>2</sup> Centre for Precision Engineering, Materials and Manufacturing Research (PEM), Institute of Technology, F91 YW50 Sligo, Ireland
- <sup>3</sup> Bioscience Research Institute, Technical University Shannon Midlands Midwest, N37 HD68 Athlone, Ireland; neil.rowan@tus.ie
- \* Correspondence: garvey.mary@itsligo.ie; Tel.: +353-071-9305529

Abstract: Mushrooms have been used as traditional medicine for millennia, fungi are the main natural source of psychedelic compounds. There is now increasing interest in using fungal active compounds such as psychedelics for alleviating symptoms of mental health disorders including major depressive disorder, anxiety, and addiction. The anxiolytic, antidepressant and anti-addictive effect of these compounds has raised awareness stimulating neuropharmacological investigations. Micro-dosing or acute dosing with psychedelics including Lysergic acid diethylamide (LSD) and psilocybin may offer patients treatment options which are unmet by current therapeutic options. Studies suggest that either dosing regimen produces a rapid and long-lasting effect on the patient post administration with a good safety profile. Psychedelics can also modulate immune systems including pro-inflammatory cytokines suggesting a potential in the treatment of auto-immune and other chronic pain conditions. This literature review aims to explore recent evidence relating to the application of fungal bioactives in treating chronic mental health and chronic pain morbidities.



#### 1. Introduction

There are many recognised mental health disorders or mental illnesses including (but not limited to) anxiety disorders, mood disorders (depression, bipolar disorder, and cyclothymic disorder) psychotic disorders (schizophrenia), eating disorders (anorexia nervosa, bulimia nervosa), impulse control and addiction disorders, obsessive-compulsive disorder (OCD), post-traumatic stress disorder (PTSD) and personality disorders [1]. Each mental illness manifests with a variable array of symptoms that differ depending on the illness present; it is accepted, however, that a person's mood, thinking, perceptions, anhedonia (pleasure sensation) and behaviours are affected. According to the 2013 Diagnostic and Statistical Manual of Mental Disorders 5 (DSM-5), "A mental disorder is a syndrome characterized by clinically significant disturbance in an individual's cognition, emotion regulation, or behaviour that reflects a dysfunction in the psychological, biological, or developmental processes underlying mental functioning" [1]. These issues are undoubtedly proliferated by the stigma toward mental disorders such as anxiety, clinical depression (major depressive disorder), and OCD where a lack of understanding in society prohibits afflicted persons from seeking help or speaking out [2]. Stigma where mentally ill persons are perceived as dangerous, unpredictable, or as having a weakness of character has excluded them from society to an extent, further exasperating the issue [2]. The use of the DSM-5 in the diagnosis of mental illness, however, remains under scrutiny as understanding and expansion of mental illness progresses [3]. The World Health Organisation (WHO, Geneva, Switzerland) created the WHO International Consortium in Psychiatric Epidemiology



**Citation:** Meade, E.; Hehir, S.; Rowan, N.; Garvey, M. Mycotherapy: Potential of Fungal Bioactives for the Treatment of Mental Health Disorders and Morbidities of Chronic Pain. *J. Fungi* **2022**, *8*, 290. https:// doi.org/10.3390/jof8030290

Academic Editor: Laurent Dufossé

Received: 24 February 2022 Accepted: 10 March 2022 Published: 11 March 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). (ICPE) in 1998 with the aim of gathering data on mental illness via a diagnostic instrument, i.e., the WHO Composite International Diagnostic Interview (CIDI) surveys [4]. The WHO states that mental health illness is increasing yearly across the globe with approximately 20% of children and adolescents being diagnosed with a mental illness where suicide is the second leading cause of death in 15- to 29-year-olds [5]. In Europe, mental health disorders affect approximately 165 million people yearly with anxiety, mood and addictive disorders being most common [6]. Suffering a mental health disorder is devasting for the individual afflicted but also impacts the family unit as substance abuse and suicidal idealisation often manifests. Considering the wider burden of mental health disease and its associated functional impairment, the socio-economic impact must also be considered due to their increasing prevalence and high disability rate where treatment options often remain unmet [7].

The direct economic burden mental illness relates to diagnosis, treatment, or hospitalisation costs with indirect costs associated with the impact on economic growth including loss of income due to disability, absence from work and cost of additional care [6].

The impact of chronic pain co-morbidities associated with mental health must also be considered including functional disorders such as fibromyalgia and other functional somatic syndromes (FSSs). The treatment of mental health disorders is typically of combination of drug therapy based on psychoactive drugs (Table 1), mood stabilisers or antipsychotic drugs, and psychotherapeutic treatments. In many cases, however, mentally ill persons are non-responsive to therapy, termed treatment resistant [8]. Currently, 30% of clinically depressed are treatment resistant with this cohort having increased issues with social interactions, occupational difficulties, declining physical health and often suicidal thoughts [9]. Additionally, therapeutics currently in use are prone to numerous side effects where treatment is ultimately discontinued in patients (Table 1). To provide therapeutic options for such persons, and to reduce the burden of disease, it is imperative that we seek effective alternatives. The use of psychedelics, such as psilocybin (magic mushrooms), Lysergic acid diethylamide (LSD), and ayahuasca have drawn attention due to their noticeable ability to alter consciousness in personally meaningful, therapeutic, and spiritual ways [10]. Indeed, clinical trials on psilocybin demonstrated its efficacy in reducing cancer-related anxiety and depression, treatment-resistant depression, major depressive disorder (MDD), and substance misuse [11]. This review aims to highlight and discuss the potential of fungal biologics including psilocybin, LSD and others as therapeutic alternatives (mycotherapy) for alleviating the symptoms of mental health disorders. The current knowledge underlining the potential of fungal biologics is highlighted and discussed in the context of common mental health disorders including addiction, MDD and anxiety where current treatment is ineffective or causes significant undesirable side effects. Additionally, this review will review the potential of said biologics to also alleviate the symptoms of chronic pain conditions commonly associated with mental illness including fibromyalgia and irritable bowel syndrome.

Disorder	Treatment	Mode of Action	Efficacy	Side Effects
	SSRIs e.g., sertraline, escitalopram SNRIs e.g., venlafaxine, duloxetine	Inhibit the reuptake of 5-HT Inhibit the reuptake of NE and 5-HT	First-line treatments for PD, GAD, SAD, and PTSD [12]	GI problems (nausea, diarrhoea, dyspepsia, bleeding), dry mouth, headaches, dizziness, anxiety, insomnia, and sexual
	TCAs e.g., amitriptyline, imipramine	(and/or DA) Inhibit the reuptake of NE and 5-HT	Equivocal efficacy with SSRIs; however, cause more adverse side effects due to their anticholinergic activity [14]	dysfunction [13] Nausea, dry mouth, constipation, weight gain, blurred vision, light-headedness, confusion, sedation, urine retention and tachycardia [15]
Anxiety disorders *	MAOIs e.g., moclobemide, phenelzine	Inhibit the mitochondrial enzyme monoamine oxidase	Third-line treatment for refractory SAD and PD, i.e., considered for patients who are non-responsive to other treatments [16]	Dry mouth, nausea, diarrhoea, constipation, drowsiness, insomnia, dizziness/or light-headedness, fatigue, urinary problems, sexual dysfunction, hypertensive crisis reaction, and serotonin syndrome [17]
	Benzodiazepines e.g., alprazolam, diazepam, clonazepam	Positive allosteric modulators of GABA-A, resulting in increased frequency of chloride channel opening	Effective and fast acting in the treatment of GAD. Recommended as second-line therapy and for short duration use due potential risks of tolerance, dependence, abuse, or misuse [13]	Drowsiness, lethargy, fatigue, and potential for dependence. Higher doses can cause impaired motor coordination, dizziness, vertigo, slurred speech, blurry vision, mood swings, euphoria and hostile or erratic behaviour
	Pregabalin	Calcium channel modulator [18]	Effective as a monotherapy for GAD, or as an adjunct to SSRIs/SNRIs in	Sedation, dizziness, somnolence, dry mouth, amblyopia, diarrhoea, weight gain and potential for dependence [12]
	Buspirone	High affinity for 5-HT <sub>1A</sub> receptors [21]	ueaunent-resistant GAD [19,20]	Nausea, headaches, dizziness, and fatigue [13]

Table 1. Treatment of mental health disorders.

Disorder	Treatment	Mode of Action	Efficacy	Side Effects
Mood disorders *	Lithium	Multiple mechanisms including modulation of (GABA)-ergic and glutamatergic neurotransmission, and alteration of voltage-gated ion	First-line treatment for prevention of manic and depressive episodes of bipolar disorder (BD) [24]	Cardiac problems, cognitive problems, acne, psoriasis, thyroid problems, nausea, vomiting, weight gain, hyponatremia, sedation, decreased libido, and teratogenic [25]
	valproic acid	channels or intracellular signalling pathways [22,23]	First-line treatment for acute mania and maintenance of BD [26]	problems, hair loss, hypothyroidism, aplastic anaemia, Leukopenia, increased transaminases, hepatitis, SLE-like syndrome, hyponatremia, tremor, decreased libido, infertility and teratogenic [25]
	Carbamazepine		Effective as a monotherapy to treat manic symptoms of bipolar or as adjunct to lithium or valproic acid [27]	Cardiac problems, cognitive problems, acne, hair loss, hypothyroidism, PCOS. diarrhoea, nausea, vomiting, pancreatitis, increased transaminases, metabolic syndrome, weight gain, sedation, tremor, decreased sexual function, infertility and teratogenic [25]
Psychotic disorders	First-generation antipsychotics (FGA) e.g., Chlorpromazine, haloperidol	D2 antagonists: work by inhibiting dopaminergic neurotransmission [28]	Effective in the treatment and maintenance of schizophrenia, acute mania with psychotic symptoms, major depressive order with psychotic features, and delusional disorder [28]	Adverse effects are drug specific and include anticholinergic effects (dry mouth, blurry vision, tachycardia, constipation), sedation, distonias, weight gain, increased lipids, parkinsonism (tremor, rigidity, bradykinesia), akathisia tardive
	Second-generation antipsychotics (SGA) e.g., quetiapine, aripiprazole	Serotonin-dopamine antagonists: work by blocking D2 dopamine receptors as well as serotonin receptor antagonist action [28]	Same clinical efficacy as FGA, with the exception of clozapine, which has unique efficacy against treatment resistant schizophrenia [30]	dyskinesia, sialorrhea, orthostatic hypotension, neuroleptic malignant syndrome, sexual disfunction, neutropenia/agranulocytosis, and myocarditis [29]

Disorder	Treatment	Mode of Action	Efficacy	Side Effects
	Olanzapine (SGA)	Block dopaminergic (D1-4 antagonism) and serotonergic (5-HT <sub>2A/2C</sub> antagonism) receptors [31]	Effective as an adjunctive therapy in treatment of AN, increasing appetite and decreasing anxiety and ruminating thoughts involving body image and food [32]	Dizziness, orthostatic hypotension, hypercholesterolemia, hyperglycaemia, weight gain, extra-pyramidal symptoms, dry mouth, hyperprolactinemia, and insomnia [32]
Eating disorder	Antidepressants(SSRIs, SNRIs, TCAs, MAOIs)	Defined above	Effective as an adjunctive therapy in treatment of BN and BED, reducing the crisis of binge eating, purging phenomena and improving mood and anxiety [33]	Listed above
	Mood stabilizers e.g., topiramate	Blocks voltage gated sodium channels, enhances GABA-A receptor activity, reduces membrane depolarization by AMPA/Kainate receptors and is a weak inhibitor of carbonic anhydrase [34]	Shown efficacy in treatment of BN and BED, reducing the crisis of binge eating, purging phenomena and promoting weight loss (in overweight or obese patients) [33]	Paraesthesia, fatigue, cognitive problems, dizziness, somnolence, psychomotor slowing, memory/concentration difficulties, nervousness, confusion, weight loss [34]
Impulse control, addiction, and obsessive-compulsive disorders	Antidepressants e.g., SSRIs and clomipramine (TCA) Mood stabilisers e.g., olanzapine,	Potently inhibit the reuptake of 5-HT	Effective as a monotherapy or as an augmentation agent in the treatment	Listed above
	carbamazepine	Defined above Non-specific competitive opioid antagonist with highest affinity for the mu-opioid receptors in the CNS [39]	IED and pyromania), addiction and compulsive disorders [35–39]	Nausea, vomiting, abdominal pain, decreased appetite, dizziness, lethargy, headaches and sleep disorders [40]
Personality disorders	Antidepressants (SSRIs, SNRIs) Quetiapine Naltrexone	Defined above	Shown efficacy in the treatment of BPD [41,42]	Listed above

Abbreviations: SSRI = selective serotonin reuptake inhibitors, SNRI = serotonin and norepinephrine reuptake inhibitor, TCA = Tricyclic antidepressants, MAOIs = Monoamine oxidase inhibitors, FGA = first generation antipsychotics, SGA = second generation antipsychotics, 5-HT = 5-hydroxytryptamine, NE = norepinephrine, DA = dopamine, GABA = γ-AMPA = aminobutyric acid, α-Amino-3-hydroxy-5-methyl-4-isoxazolepropionic acid, PD = panic disorder, GAD = generalised anxiety disorder, SAD = social anxiety disorder, PTSD = post-traumatic stress disorder, BD = bipolar disorder, AN = anorexia nervosa, BN = bulimia nervosa, BED = binge eating disorder, PG = pathological gambling, KM = kleptomania, TTM = trichotillomania, IED = intermittent explosive disorder, BPD = Borderline Personality Disorder, SLE = systemic lupus erythematosus, PCOS = Polycystic ovary syndrome, CNS = central nervous system. \* Many drugs listed for the treatment of anxiety are also employed for the treatment of mood disorders including SSRIs, NSRIs and TCAs.

Table 1. Cont.

#### 2. Mental Health Disorders

The incidence of mental health disorders is increasing globally, yet a fundamental understanding of the aetiology of mental health disease remains elusive. Assumptions relate severe mental illness to a small number of genes with a specific relationship between genotype and mental illness where biological pathways and environmental factors impact on illness [43]. Studies have demonstrated a strong relationship between severe trauma such as childhood abuse and/or abandonment with adult mental health disorders and a diminished quality of life [44]. The most widely discussed theories of depression are based on the regulation of the monoamines, norepinephrine (NE), dopamine and serotonin. The catecholamine hypothesis of depression developed in the 1960s remains the basis of treatment of mental health disorders such as MDD and anxiety disorder. Based on the theory that such mental illness is a result of a NE or dopamine deficiency at central nervous system (CNS) synapses, treatment aims to increase dopamine or NE levels to alleviate symptoms [45]. The serotonin hypothesis of depression also from the 1960s relates mood disorders to a deficiency in serotonin (5-hydroxytryptamine, 5-HT) a neurotransmitter involved in regulating emotion and mood [46]. There are currently 14 5-HT receptor subtypes identified having varying affinities for serotonin, present presynaptically, postsynaptically in the CNS and throughout the brain being predominately G protein coupled receptors (except 5-HT3 receptor) [47]. Agonism of the 5-HT<sub>2A</sub> receptors is involved in increasing cortical glutamate levels [48]. The neuroendocrine hypothalamic-pituitary-adrenal (HPA) axis a natural pathway activated in times of stress also becomes hyperactive in states of depression. The HPA axis is involved in maintaining blood glucocorticoid levels, where elevated levels can disrupt HPA function [49]. Studies have shown that cortisol elevation long term results in cognitive and other medical issues in patients and exacerbate depressive symptoms as shown in mice [50]. Thus, it is generally accepted that neuropsychiatric diseases such as PTSD, MDD, anxiety etc are a result of a chemical imbalance in neural circuits distributed across the brain and CNS related to a loss of neuro plasticity [51]. Consequently, treatment of mental health disorders is based on prescribing anti-depressants namely selective serotonin reuptake inhibitors (SSRIs), serotonin-norepinephrine reuptake inhibitors (SNRIs), dopamine reuptake inhibitors (DRIs) and tricyclic antidepressants (TCAs) which aim to increase neurotransmitter levels at the synapses and increase neuroplasticity. While the efficacy of this approach remains under question the side effects associated with the varying therapeutics in these categories are undeniable. Serotonin syndrome for example which is a potentially life-threatening condition can be caused by using serotonergic drugs and overactivation of 5-HT<sub>1A</sub> and 5-HT<sub>2A</sub> receptors [52]. Additionally, antidepressants such as TCAs and SSRIs are known to increase the concentration of neurotransmitters at synapses within minutes, yet it takes weeks to months for mental illness symptoms to regress in patients [53]. Additional theories of depression include the neurogenic hypothesis of depression and inflammatory hypothesis based on an increase in proinflammatory cytokines in the nervous system [7]. The 5-HT receptors are known to regulate the release of cytokines interleukin (IL) and tumour necrosis factor (TNF) and modulate immune macrophage and dendritic cell function and aid in maintaining an anti-inflammatory state in vivo [54]. With increasing rates of mental illness and suicide ideation and tendencies, it is imperative that a more rapid acting treatment option is established.

#### Mental Health and Co-Morbidities of Chronic Pain

Functional somatic syndrome is the term used for conditions that do not appear to have an underlying biological pathology [55]. These debilitating conditions are becoming increasingly frequent imposing a significant economic burden. Chronic fatigue syndrome (CFS), irritable bowel syndrome (IBS) and fibromyalgia (FM) amongst others are FSSs where pain, fatigue, muscle, and joint aches often present without an identifiable cause. Theories have arisen (the 'Lumpers' vs. 'Splitters' debate) suggesting that certain FSSs have one underlying aetiology and are associated with psychiatric disorder's including anxiety and depression. While the pathophysiology of FM remains unknown it has been postulated that

thinly myelinated and unmyelinated C-nerve fibres have been reported in FMS patients [56] suggesting a link between loss of nerve myelin and chronic pain. Additionally, studies suggest that a dysfunction of dopamine is present in FM patients where dopamine agonists appear effective in treating symptoms [57]. Studies have demonstrated decreased cortical dopamine receptor binding in FM patients [57]. FM characterised by extreme hyperalgesia is highly prevalent (3 times more prevalent) in persons with major depressive disorder than in patients without, studies also show high rates of MDD amongst family members of FM patients suggesting a genetic aspect [58]. Additionally, sleep abnormalities in FM are similar to those present with a 5-HT dysfunction in depressed individuals [58]. CFS appears less linked to mental health disorders where psychiatric status is not considered an important causal contributor to CFS [59]. IBS is a painful syndrome characterized by chronic abdominal pain, altered bowel habits, correlated to significant psychological distress and psychiatric comorbidities anxiety, panic disorder, MDD and suicidal ideation [60]. Polymorphisms in the serotonin transporter gene (SERT) has been associated with higher risk of depression in IBS patients with a dysfunction of serotonin transporters impacting the emotion regulating regions of the brain [61,62]. At present, treatment of FM and IBS is also reliant on the use of antidepressant drugs TCAs and SNRIs where successful therapy has supported the belief that a dysfunction of serotonin and NE is present in FSSs similarly to MDD [58].

The relationship between mental health disorders and autoimmune conditions (including multiple sclerosis, rheumatoid arthritis, colitis, and lupus erythematosus) has been well established where there is increased risk of MDD, anxiety and schizophrenia in autoimmune patients. Theories suggest the excessive inflammation characteristic of autoimmune diseases may be a casual factor impacting the CNS or via cytokine interactions between nerve and immune cells of the patient [54]. The relationship between increased levels of cytokines including tumour necrosis factor- $\alpha$  (TNF- $\alpha$ ) and IL-6 and major depression disorder is now recognised [47]. This raises the question if fungal extracts showing promise in treating mental health disorders may also alleviate the symptoms of certain FSSs namely FM and IBS and auto immune conditions.

#### 3. Fungal Biologics: Unlocking the Potential of Eastern Practice into Western Medicine

There has been an increasing interest in translating potential medicinal cures derived from medicinal mushroom used for centuries in Eastern practice into Western medicine [63,64]. Modern medical practice relies heavily on the use of highly purified pharmaceutical compounds whose purity can be easily assessed and whose pharmaceutical activity and toxicity show clear structure-function relationships. In contrast, for decades, many mushroom-derived medicines contain mixtures of natural compounds had not undergone detailed chemical analyses and whose mechanism of action had not been elucidated [65]. With the decline in the number of new therapeutics produced from the pharmaceutical industry, novel biologic agents are being sought from traditional medicine. Translating traditional Eastern practices into acceptable evidence-based Western therapies is challenging given different manufacturing standards, criteria of purity, and under-powered clinical trials making assessment of efficacy and toxicity by Western standards of clinical evidence difficult. Purified bioactive compounds derived from medicinal mushrooms using appropriate drug discovery programs are a potentially important new source of psychoactive agents. For example, the therapeutic potential of medicinal mushrooms is evidenced by the fact that two glucan isolates were licensed as drugs in Japan as immune-adjuvant therapy for cancer in 1980. Moreover, Murphy et al. (2021) noted that approximately 200 registered clinical trials have appeared in the literature focusing on use of beta-glucans extracted from a diversity of medicinal mushrooms with a focus on cancer treatments using mainly oral administration [65]. These authors observed significant challenges exist to further clinical testing and translation of mushroom-derived biologics. The diverse range of conditions for which biologics from medicinal fungi are in clinical testing underlines the incomplete understanding of the diverse mechanisms of action that is a key knowledge gap. By far the greatest mechanistic information relates to elucidating immunomodulatory potential of medicinal mushrooms [66,67]. Furthermore, important differences appear to exist in the effects of apparently similar mushroom-derived preparations, which may be due to differences in sources and extraction procedures, another poorly understood issue [65]. In general, there is a dearth in evidence-based knowledge from registered clinical trials on potential use of biologics derived from medicinal mushrooms for alleviating mental health conditions. Therefore, there is growing interest in adopting a Quadruple Helix approach (academia–industry–policy–society) to support emerging industries, including medicinal fungi companies, in the 'bioprospecting' and development of new therapeutic products informed by compliance with appropriate regulatory framework and clinical trials; moreover, this multi-actor framework also enables greater societal awareness of the potential for higher fungi to meet a diversity of emerging needs [68–70].

Outside of the use of extracts and along with the beta-glucans, a plethora of fungal bioactives have been isolated and fully purified including further polysaccharides, peptides and polyphenols, medium-sized macromolecules such as triterpenoids and the small molecule type structures of the indole alkaloids including tryptamines. Yildiz et al. (2015), for example, reports on the potent antioxidant action of phenol compounds in many medicinal mushrooms [71]. The focus here will be on the bioactives with low molecular weights which can cross the BBB and have structural features and effects similar to the common neurotransmitters. The neurotransmitters, serotonin, dopamine and NE are monoamines derived from either tyrosine or tryptophan. Some of the most interesting and potent compounds discussed herein are derived from the same precursors. Indeed, serotonin itself and psilocybin are in fact both substituted tryptamines (Figure 1).



Figure 1. Monoamine neurotransmitters and fungal bioactives.

Psychedelics or serotonergic hallucinogens are natural substances which have agonism for the 5-HT receptors and include psilocybin, dimethyltryptamine (DMT) from the chacruna (ayahuasca brew) and jurema plants and mescaline from the peyote and San Pedro cacti [72]. Psychedelic compounds produce hallucinogenic experiences via activity in brain regions regulating cognition, emotions, self-awareness and perception of pain [73]. Psychedelics are included in the psychoplastogens class of compounds which have effect on neural plasticity in key circuits involved in brain health [53]. By definition psychoplastogens produce a quantifiable change in plasticity within a short time frame (>72 h) post administration [74]. Agonism of the 5-HT receptor group is associated with neuroplastic changes potentially reducing depression and anxiety symptoms [75]. Studies show that psychoplastogens have a range of therapeutic potential in treating neuropsychiatric diseases including anxiety disorder, mood disorders, PTSD, and addiction [74]. The 5-HT receptor group also has an important role in the gut-brain axis and is linked to the pathophysiology of IBS via dysregulated serotonin signalling [76]. Mushrooms containing psychoplastogens have been in use as traditional medicine in numerous cultures as healing rituals for millennia. Fungal biologics of therapeutic potential include lysergic acid diethylamide (LSD) a synthetic derivative of the precursor LSA (d-lysergic acid) or ergine from the rye ergot fungus *Claviceps purpurea*, muscimol and a DMT derivative from *Amanita muscaria* mushroom (amongst others) and psilocybin, a hallucinogenic pro-drug of psilocin found in Psilocybe mushrooms (magic mushrooms) (Table 2). Psilocybe alkaloids are more recently obtained from magic truffles (fungis sclerotia) as this species is currently overlooked in the prohibition laws banning sale [48]. Non hallucinogenic fungal biologics showing potential for mental health treatment include compounds from the edible mushroom *Hericium erinaceus* and ergothioneine from *Pleurotus cornucopiae*.

Table 2. Fungal biologics of therapeutic potential.

	Mescaline	Psilocybin/Psilocin	LSD
mics	Naturally occurring substituted phenethylamine extracted from the peyote cactus	Naturally occurring indole-alkylamine (tryptamine) extracted from Psilocybe mushrooms	Semisynthetic indole-alkylamine (ergoline) derived from lysergic acid found in <i>Claviceps purpurea</i>
dyna	5-HT releasing agent, catecholamine-like structure [77]	Close structural analogue of 5-HT	Close structural analogue of 5-HT
rmaco	Primarily interacts at 5-HT <sub>2A/2C</sub> and $\alpha_{2-}$ adrenergic receptors [78]	Primarily interacts at 5-HT <sub>2A/2C</sub> and 5-HT <sub>1A</sub> , 5-HT <sub>2C</sub> [79]	Mixed 5-HT <sub>2</sub> /5-HT <sub>1</sub> receptor partial agonist [80]
Pha	Low binding affinity at dopaminergic and histaminergic receptors [78]	Indirectly increases DA concentration but has no affinity for D2 receptors [81]	High affinity dopaminergic, adrenergic, and histaminergic receptors [82,83]
	Can be ingested orally, smoked, or insufflated	Can be ingested orally or intravenously	Can be ingested orally, smoked, injected, or snorted
	Relatively low-potency: active doses in the 200–400 mg range) [84]	$20 \times$ more potent than mescaline: active doses in 10–30 mg range [81]	2000× more potent mescaline: active doses in 25–200 μg range [85]
cinetics	Rapidly absorbed in GI and distributed to the kidneys and liver	Rapidly absorbed and dephosphorylated to psilocin (bioactive form) [81]	Rapidly absorbed in GI and distributed to tissues and organs
macol	Low lipid solubility, weak ability to penetrate BBB [77]	Lipid soluble, can easily cross BBB [86]	Can easily cross BBB [87]
Phan	Detoxification via oxidative deamination Long-lasting, half-life of 6 hrs	Detoxification via demethylation and oxidative deamination Half-life of 3 hrs	Detoxification via N-dealkylation and/or oxidation processes Half-life of 3.6 hrs
	Eliminated in urine mainly in the unchanged form (81.4%) and the remaining as the metabolite TMPA [88]	Eliminated in urine mainly as glucuronidated metabolites (80%) as well as unaltered psilocybin (3–10%) [89]	Eliminated in urine mainly as metabolites, only 1% of the dose is excreted unchanged) [87]
Efficacy	Acute experiences of psychological insight during mescaline use are associated with self-reported improvements in anxiety disorders, depression, and substance abuse [78,84]	Therapeutic efficacy in treating mood and anxiety disorders, depression, cluster headaches, chronic pain, intractable phantom pain, obsessive-compulsive disorder, and substance abuse [79,81,90]	Therapeutic efficacy in treating anxiety disorders, depression, cluster headaches, obsessive-compulsive disorder, substance abuse, psychosomatic illnesses, and anxiety in relation to life-threatening diseases [91,92]

#### 3.1. Psilocybe Mushrooms

Undoubtedly psychedelics from the Psilocybe species are currently the more commonly investigated fungal compounds for their biological and therapeutic potential. Psychedelic compounds are known to exert their effects via activation of serotonergic 5-HT receptors dispersed throughout the central and peripheral nervous system impacting on cognition, mood and emotion [93]. Psilocybin is a potent agonist of 5-HT<sub>2A</sub> receptors and moderate agonist of 5-HT<sub>1A</sub> and 5-HT<sub>2C3</sub> receptors present in the thalamus and cortex of the brain [79]. These receptors are also associated with peripheral and central nervous system pain perception [73] justifying the use of SSRIs in the treatment of chronic pain conditions such as FM. Psilocybin has also demonstrated affinity for dopamine receptors and serotonin transporter protein [94]. Both psilocybin and its metabolite psilocin, pass through the blood–brain barrier (BBB) where psilocin is 1.5 times more potent. Indeed, between the 1950s and 1970s, researchers actively investigated the potential of psilocybin in the treatment of OCD, addiction, and neurotic behaviours [72] but studies were inhibited by prohibition until the early 1990s. Psychedelics are now believed to have additional beneficial biological activities including promoting neural plasticity and immune modulation [95]. Additionally, psilocybin positively impacted symptoms of MDD, treatment resistant depression, substance addiction and anxiety (anxiolytic) in afflicted persons and terminal cancer patients [96]. Comparative studies where one dose of ca. 30 mg/70 kgpsilocybin was administered to terminally ill patients reduced symptoms of depression and anxiety and improved optimism in 51 cancer patients for up to 6 months post administration [97]. In one trial study, psilocybin reduced symptoms of MDD, anxiety and anhedonia as early as 1 week post administration of 10 mg and 25 mg (seven days apart) with no observed side effects to the patient [75]. Studies assessing the impact of psilocybin on alcohol addiction showed patients had significantly fewer drinking days in the treatment period of 8 weeks at 300 µg/kg 4 weeks apart [79]. Promising results were also observed with the use of psilocybin in conjunction with cognitive behavioural therapy (CBT) to aid in smoking cessation [98]. A long term follow up study reported 60% abstinence rates after more than 12 months compared with a 31% anstinence rate at 12 months using conventional therapies. Small scale clinical studies also demonstrate the efficacy of psilocybin at reducing the symptoms of OCD where SSRI treatment failed [99]. Similar to the action of SSRIs on 5-HT receptors alleviating chronic pain in patients with neuropathic, musculoskeletal pain and fibromyalgia, psilocybin may reduce chronic pain symptoms [73]. Clinical trials are warranted on the administration of psilocybin to chronic pain patients including FSSs comparative to current treatment options such as SSRIs and opioids.

#### 3.2. Claviceps purpurea

The fungus Claviceps purpurea is extensively known for its production of ergot alkaloids having activity on the nervous system and smooth muscles where therapeutic application includes treating Parkinson's disease, cluster headaches and migraine [100]. LSD is a semi-synthetic derivative of lysergic acid found in fungus *Claviceps purpurea* [73]. LSD was also a psychedelic compound highly researched in the 1950s until the Controlled Substances Act was introduced in 1970 [72]. Like psilocybin, LSD has activity on brain processes involved in cognition, emotion, and perception via affinity for serotonin receptors [101]. LSD is an agonist of 5-HT with high affinity for 5-HT2A and 5-HT2C and is small enough to pass the BBB being approximately 100 times more potent than psilocybin [73]. Interestingly, LSD is also an agonist of the dopaminergic receptors with relatively high affinity for dopamine 2 receptors [102] and an agonist of the trace amine associate receptor 1 (TAAR1) [103]. Currently, in the treatment of psychotic illness the therapeutic efficacy of active pharmaceutical ingredients (APIs) is required to be greater than 70% occupancy of D2 receptors [103]. TAAR1 is essential in the regulation of monoamines (dopamine, NE, serotonin) in neurons of the CNS and neuro immune modulation [104] via agonism of trace amines ( $\beta$ -phenylethylamine, p-tyramine, and tryptamine). Dysregulation of trace amines and TAAR1 receptor dysfunction has been identified in psychotic disorders including MDD, bipolar disorder and schizophrenia [103]. Additionally, the TAAR1 receptors are associated with the pathophysiology of IBS [76]. Studies have demonstrated the positive effects of LSD on anxiety relating to terminal illness [93] with positive affects also observed for PTSD and addiction. The analgesic activity of LSD has also been described in terminally ill patients lasting up to 12 h post administration [73]. While reports on pain management of headaches are self-reported, patients state that sub hallucinogenic concentrations (micro dosing) of LSD prophylactically and metaphylactically alleviated migraine and cluster

headaches [105]. One clinical study determined that 20 µg LSD increased pain tolerance and reduced pain perception in patients compared to placebo with similar outcomes to oxycodone and morphine [106]. Such analgesic effects may be attributed to the dopaminergic activity of LSD suggesting positive effects on neurological disease such as FM. LSD's mechanism of action is pleiotropic affecting serotonin (5-HT), dopamine and TAAR1 receptor pathways potentially alleviating symptoms of mental illness and chronic pain conditions FM and IBS. It is important to note that LSD is a complex molecule, effecting many receptor pathways, where long-term administration may result in psychotic-like symptoms or Hallucinogen Persisting Perception Disorder (HPPD) or flashbacks in the user [103]. Such effects, however, are deemed low risk, uncommon and mostly associated with recreational use [107]. Additionally, secalonic acid A and its isomers are also found in Claviceps purpurea which has anti-cancer activity via topoisomerase I and II inhibition [108].

#### 3.3. Amanita muscaria

A. muscaria contains many biologically active compounds including the psychoactive alkaloids: muscarine, ibotenic acid and muscimol [109] where ibotenic and muscimol are structurally similar to gamma-aminobutyric acid (GABA) having effect on glutamate receptors in the CNS [110] and can cross the BBB. Ibotenic acid is a potent neurotoxin via activation of N-methyl-d-aspartate (NMDA) receptors where muscimol has a strong psychoactivating action [111]. The alkaloid muscarine is an acetylcholine agonist of the parasympathetic nervous system typically negatively impacting the functioning of numerous organs but cannot pass the BBB [109]. Over stimulation of muscarinic cholinergic system and acetylcholine is associated with the aetiology of depression suggesting that the agonist muscarine will elevate depressive symptoms in patients. Studies by Corbett 1991, demonstrate that muscimol improved social behaviour and had anxiolytic-like effects post systemic administration to test rats, similar to diazepam [112]. Additionally, muscimol when used as a combination therapy with endomorphin-1 reduced the symptoms of neuropathic pain due to spinal cord injury when administered for 7 consecutive days [113]. Bufotenine (5-HO-DMT) is an alkaloid and a substituted tryptamine similar to serotonin and psilocin, also present in the Amanita mushroom. As such, it has the ability to cross the BBB having agonistic effects on 5-HT1A and 5-HT1B receptors [114]. Another substituted tryptamine,5-MeO-DMT, requires enzymatic activity of a monoamine oxidase (MAO) inhibitor to induce its psychedelic effects orally [48]. Recently,  $\beta$ -carbolines, which are well-known MAO inhibitors have been isolated directly from psilocybe mushrooms.  $\beta$ -carbolines, as well as psilocybin itself, are derivatives of tryptophan. This represents an interesting natural product pathway where two different types of products from the same precursor contribute to the same pharmacological effect [115]. Tryptamines have been used to treat symptoms of PTSD, MDD, addiction and anxiety [116] and show potential as treatment with low risks for addiction.

#### 3.4. Hericium erinaceus (H. erinaceus)

The medical benefits of *H. erinaceus* have been well documented with anticancer, antioxidative, anti-inflammatory and antimicrobial activity amongst others [7] where neurotropic compounds (Figure 1) present in the mushroom are more recently being considered for neuropsychiatric disorders. Hericenones and erinacines (cyathane derivatives) are biological neurotropic compounds found in the fruiting body and mycelium of *H. erinaceus*. These compounds can pass the BBB due to their low molecular weight [117] which may beneficially impact on Alzheimer's and Parkinson's disease due to their impact on nerve growth factor (NGF) [118]. Indeed, many biological active compounds are present in the fruiting bodies of *H. erinaceus* including aromatic compounds, diterpenoids, steroids, and polysaccharides [119]. NGF is essential for protecting nerve tissue and maintaining neuron functionality where studies have shown NGF levels in MDD patients is also significantly reduced [120]. Interestingly, NGF itself is unable to pass the BBB highlighting the benefits of these compounds which stimulate NGF production within the brain and

encourage nerve myelination throughout the CNS [121]. Perhaps such myelination effects may also aid in alleviating the symptoms of FM in patients where improper myelination of nerves is a possible issue [56]. Initial studies suggest that the NGF enhancing activity of these extracts may reduce MDD and anxiety [122]. Elevated levels of NGF are associated with neuroplasticity and neurogenesis which may improve mood in MDD patients based on the neurogenic hypothesis of depression [7]. Additionally, in vivo studies on depressed animals demonstrated that when fed with *H. erinaceus* levels of the monoamine's serotonin, dopamine and NE were elevated [123]. In vivo studies report that erinacine can increase monamine expression and modulate anti-inflammatory brain-derived neurotrophic factor (BDNF) signalling in depressed animals [124]. Amycenone, a proprietary mixture of compounds which are present in the fruiting body of *H. erinaceus*, demonstrated antiinflammatory activity against cytokines TNF and IL and reduced inflammatory induced depression in test animals [119] a similar activity observed with SSRIs and SNRIs. Such anti-inflammatory activity may also prove beneficial in alleviating symptoms of autoimmune co-morbidities of mental illness. Studies show when consuming H. erinaceus adults with mild cognitive impairment and menopausal women were less depressed, anxious with improved cognitive abilities compared to placebo control groups [121]. Studies are warranted however, confirming the efficacy of biological extracts from *H. erinaceus* mushrooms comparative to consuming the entire fruiting body of the mushroom at a therapeutic level. Additionally, while these compounds have been shown to affect the levels of monoamines, the mechanism of modulation requires further investigation.

#### 3.5. Pleurotus cornucopiae

One of the fungal bioactives with the most diverse effects and potential uses currently under widespread investigation is Ergothionine. Found in large amounts in oyster mushrooms (*Pleurotus cornucopiae* var. *citrinopileatus*), Ergothione is a metabolite of the aromatic amino acid histidine. It has been shown to have neuroprotective and anti-inflammatory effects as well as being potentially involved in neurogenesis, all of which are characteristics of compounds which may be useful in the treatment of the aforementioned disorders [125]. Dietary ergothionine was shown to significantly reduce immobility time in forced swim test (FST) and tail spin test (TST) which are used as models of depression in mice [126]. Additionally, oral administration of ergothioneine prior to and during another mouse model of depression (Social Defeat Stress (SDS) paradigm) had a preventative effect on depressive behaviours such as social avoidance and sleep abnormalities [127]. Ergothioneine is a substrate of carnitine/organic cation transporter OCTN1 and this has led to investigation of the potential targeting of such atypical monoamine transporters as a strategy for the development of new anti-depressants [128].

#### 4. Pharmacological Consideration of Mycotherapy

Bioactive compounds which promote neuro plasticity and induce long term changes in mood, emotion and cognition may offer therapeutic options to chronically mentally ill patients [95]. When considering the administration of psychedelic compounds as therapeutics the efficacy of single dosing versus micro-dosing must be considered as current studies vary between self-medicating/micro dosing patients and single dose clinical trials. The time frame required for onset of action is an important consideration as current drug therapy (SSRI, NSRIs) for mental illness requires weeks to months for therapeutic effect. Studies indicate the effects of the fungal bioactives described occur rapidly and have a prolonged lasting effect on the symptoms of the diseases investigated [93]. Additionally, there appears to be no, or limited side effects or withdrawal symptoms associated with acute therapy or micro-dosing. Tolerance can develop however, relating to down regulation or de-sensitization of the 5-HT receptors and a cross tolerance with other psychedelic compounds may occur [47]. Psilocybin has low toxicity with chronic exposure and moderate toxicity in cases of acute exposure and has low addiction and dependence issues [75]. LSD which is more potent than psilocybin induced some dose dependent physical and psychological symptoms including derealization, depersonalization and dissociation [106]. Psilocybin has pharmacodynamic effect lasting between 4–6 h with LSD lasting up to 12 h, an elimination half-life of 3.5 h and 3 h for LSD [129] and psilocybin [130] respectively, has been established. Route of administration is an important formulation consideration for psychotherapy as tryptamine and its derivates are prone to first pass metabolism. The addition of a monoamine oxidase inhibitor may be needed for oral delivery formulations [48]. Such an inhibitor would increase bioavailability and pharmacological efficacy of bioactive amines including psilocybin and LSD. The administration of such compounds is not straight forward as each has specific receptor affinities, duration of action and potency. The therapeutic index for each compound must be established to fully determine the safety profile as long-term treatment may be required for chronic mental health illness or chronic pain conditions. Currently, mood and behavioural studies are conducted in animals where analysis of symptoms and extrapolation to humans is difficult. Consuming such compounds may affect the patient's consciousness and provide insights and existential and spiritual questions [99]. For example, at 12 month follow up in the smoking cessation work by Johnson et al., 86.7% of the participants rated their psilocybin experience among the 5 most personally meaningful and spiritually significant of their lives. Administration of these compounds, therefore, must be in a controlled manner to avoid the risk to persons prone to psychosis and to determine the patient's mindset prior to administration. To this end, many of the studies involving the psychedelics and compounds which alter consciousness have been carried out in the presence of psychiatrists and counselling professionals specifically trained in psychedelic psychotherapy. Some alternative methods of determining activity have been utilized based on gene ontology, predictive analysis and Kyoto Encyclopedia of Genes and Genomes (KEGG) pathway analysis to investigate fungal biologics [124].

There is a pressing need to reach international consensus and agreement on the defined production methods for bioactives produced from medicinal fungi for this potential purpose as batch-to-batch variation in production methods and types of mushrooms may generated different structural-bioactives that may lead to vary functional activities [67,131,132]. This situation will be addressed by promoting international collaborations between academia and industry partners in the areas of biotechnology, mycology, toxicology, bioinformatics and biopharma.

#### 5. Conclusions

Bioactives found in higher fungi such as mushrooms have chemical structures similar to neurotransmitters and can act as agonists of receptor pathways involved in psychiatric conditions. Harnessing this activity as therapy for chronic mental health and pain diseases may offer benefits where therapeutic needs are currently unmet. Considering the high morbidity and mortality rates associated with these disorders and the economic burden placed on health systems, it is imperative competent treatment options are investigated. Bioactives such as psychedelics effect a person's cognition, emotion and mood often having a long-lasting effect on symptoms of mental illness. Interest has arisen in tryptamine derivatives LSD and psilocybin and actives including hericenones and erinacines for the treatment of mood disorders, anxiety disorders, PTSD and addiction. Additionally, compounds acting as agonists of serotonin receptors may aid in alleviating symptoms of inflammatory conditions by regulating pro-inflammatory cytokines TNF and IL. While the potential benefits of such mycotherapy emerge, studies are warranted to determine their full pharmacokinetic and pharmacodynamic profiles before implementation as alternative drug therapy can be considered. While addiction and dependence do not appear to be an issue with such compounds, tolerance and cumulative effects must be considered. The complete aetiology of mental health disorders remains undetermined with the relationship between the mind, the brain and CNS a multifaceted interaction. As knowledge emerges giving a better understanding of mental illness, advances in neuropharmacology will undoubtedly follow. Perhaps such advances will include mycotherapy. Currently, the biggest challenge associated with psychedelic pharmacology studies is prohibition

where psychedelics are classed as a Schedule 1 drug and the social stigma associated with their use. Chong et al., 2021 implied a predictive analysis method to investigate the molecular mechanisms of *H. erinaceus* [124], perhaps this offers a novel means of determining efficacy prior to clinical trials in line with prohibition. Other methods including ontology and Kyoto Encyclopedia of Genes and Genomes pathway analysis to investigate fungal biologics may also be implemented. Using animal species to model depression system in humans is currently standard protocol. Such assays allow researchers to investigate neural circuitry and molecular and cellular pathways in acute and chronic states. While such models off advantages, perhaps going forward inclusive studies of depression may employ several strains and multiple testing models. Where studies must compare genetic and epigenetic expressions, and activity to better extrapolate to human patients. Until a more comprehensive understanding of the aetiology of depression is established critical analysis of current drug therapy and fungal bioactives is warranted on existing animal and clinical studies to accurately determine the exact mode of action of fungal biologicals. Adopting 'Quadruple Helix' framework that unites academia, industry, policy, and society will further enable and support bioprospecting, development, testing and validation of bioactives of interest extracted from higher fungi (mushrooms) such as to advance the field of chronic pain and mental health disorders and conditions. Currently, international drug policies are slowly evolving as the potential benefits of these compounds are being revealed; however, care must be taken at clinical therapy stage.

**Author Contributions:** Conceptualization of this article was by M.G. and S.H. Design, research, and writing of this article was conducted by E.M., S.H., N.R. and M.G. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Acknowledgments: M.G., S.H. and E.M. would like to acknowledge the PEM centre and IT Sligo.

**Conflicts of Interest:** The author declares no conflict of interest.

#### References

- 1. American Psychiatric Association. *DSM 5 Diagnostic and Statistical Manual of Mental Disorders;* American Psychiatric Association: Washington, DC, USA, 2013; 947p.
- Ebneter, D.S.; Latner, J.D. Stigmatizing attitudes differ across mental health disorders: A comparison of stigma across eating disorders, obesity, and major depressive disorder. J. Nerv. Ment. Dis. 2013, 201, 281–285. [CrossRef]
- 3. Adam, D. Mental health: On the spectrum. *Nature* 2013, 496, 416–418. [CrossRef]
- Kessler, R.C.; Angermeyer, M.; Anthony, J.C.; De Graaf, R.; Demyttenaere, K.; Gasquet, I.; De Girolamo, G.; Gluzman, S.; Gureje, O.; Haro, J.M.; et al. Lifetime prevalence and age-of-onset distributions of mental disorders in the World Health Organization's World Mental Health Survey Initiative. *World Psychiatry* 2007, *6*, 168–176. [PubMed]
- 5. World Health Organisation. Mental Health. Available online: https://www.who.int/health-topics/mental-health#tab=tab\_12021 (accessed on 23 February 2022).
- 6. Trautmann, S.; Rehm, J.; Wittchen, H.-U. The economic costs of mental disorders: Do our societies react appropriately to the burden of mental disorders? *EMBO Rep.* 2016, *17*, 1245–1249. [CrossRef] [PubMed]
- Chong, P.S.; Fung, M.-L.; Wong, K.H.; Lim, L.W. Therapeutic Potential of Hericium erinaceus for Depressive Disorder. *Int. J. Mol. Sci.* 2019, 21, 163. [CrossRef] [PubMed]
- 8. Keller, M.B. Issues in treatment-resistant depression. J. Clin. Psychiatry 2005, 66 (Suppl. 8), 5–12.
- 9. Al-Harbi, K.S. Treatment-resistant depression: Therapeutic trends, challenges, and future directions. *Patient Prefer. Adherence* **2012**, *6*, 369–388. [CrossRef]
- 10. Johansen, P.; Krebs, T.S. Psychedelics not linked to mental health problems or suicidal behavior: A population study. *J. Psychopharmacol.* **2015**, *29*, 270–279. [CrossRef]
- 11. Davis, A.K.; Agin-Liebes, G.; España, M.; Pilecki, B.; Luoma, J. Attitudes and Beliefs about the Therapeutic Use of Psychedelic Drugs among Psychologists in the United States. *J. Psychoact. Drugs* **2021**, *53*, 1–10. [CrossRef]
- 12. Strawn, J.R.; Geracioti, L.; Rajdev, N.; Clemenza, K.; Levine, A. Pharmacotherapy for generalized anxiety disorder in adult and pediatric patients: An evidence-based treatment review. *Expert Opin. Pharmacother.* **2018**, *19*, 1057–1070. [CrossRef]

- Garakani, A.; Murrough, J.W.; Freire, R.C.; Thom, R.P.; Larkin, K.; Buono, F.D.; Iosifescu, D.V. Pharmacotherapy of Anxiety Disorders: Current and Emerging Treatment Options. Review. *Front. Psychiatry* 2020, *11*, 1412. [CrossRef]
- 14. Moraczewski, J.; Aedma, K.K. Tricyclic Antidepressants; StatPearls Publishing: Treasure Island, FL, USA, 2022.
- 15. Schneider, J.; Patterson, M.; Jimenez, X.F. Beyond depression: Other uses for tricyclic antidepressants. *Clevel. Clin. J. Med.* **2019**, *86*, 807–814. [CrossRef]
- 16. Chamberlain, S.R.; Baldwin, D.S. Monoamine oxidase inhibitors (MAOIs) in psychiatric practice: How to use them safely and effectively. *CNS Drugs* **2021**, *35*, 703–716. [CrossRef]
- 17. Sub Laban, T.; Saadabadi, A. Monoamine Oxidase Inhibitors (MAOI); StatPearls Publishing: Treasure Island, FL, USA, 2022.
- 18. Ansara, E.D. Management of treatment-resistant generalized anxiety disorder. Ment. Health Clin. 2020, 10, 326–334. [CrossRef]
- 19. Sartori, S.B.; Singewald, N. Novel pharmacological targets in drug development for the treatment of anxiety and anxiety-related disorders. *Pharmacol. Ther.* **2019**, 204, 107402. [CrossRef]
- 20. Shmuts, R.; Kay, A.; Beck, M. Buspirone: A forgotten friend. Curr. Psychiatry 2020, 19, 20.
- 21. Wilson, T.K.; Tripp, J. Buspirone; StatPearls Publishing: Treasure Island, FL, USA, 2022.
- 22. Schloesser, R.J.; Martinowich, K.; Manji, H.K. Mood-stabilizing drugs: Mechanisms of action. *Trends Neurosci.* **2012**, *35*, 36–46. [CrossRef]
- Iannaccone, T.; Sellitto, C.; Manzo, V.; Colucci, F.; Giudice, V.; Stefanelli, B.; Iuliano, A.; Corrivetti, G.; Filippelli, A. Pharmacogenetics of carbamazepine and valproate: Focus on polymorphisms of drug metabolizing enzymes and transporters. *Pharmaceuticals* 2021, 14, 204. [CrossRef]
- 24. Pérez de Mendiola, X.; Hidalgo-Mazzei, D.; Vieta, E.; González-Pinto, A. Overview of lithium's use: A nationwide survey. *Int. J. Bipolar Disord.* **2021**, *9*, 10. [CrossRef]
- 25. Murru, A.; Popovic, D.; Pacchiarotti, I.; Hidalgo, D.; León-Caballero, J.; Vieta, E. Management of Adverse Effects of Mood Stabilizers. *Curr. Psychiatry Rep.* **2015**, *17*, 66. [CrossRef]
- 26. Bourin, M. Mechanism of action of valproic acid and its derivatives. SOJ Pharm. Sci. 2020, 7, 1-4.
- Grunze, A.; Amann, B.L.; Grunze, H. Efficacy of Carbamazepine and Its Derivatives in the Treatment of Bipolar Disorder. *Medicina* 2021, 57, 433. [CrossRef]
- 28. Chokhawala, K.; Stevens, L. Antipsychotic Medications; StatPearls Publishing: Treasure Island, FL, USA, 2022.
- 29. Stroup, T.S.; Gray, N. Management of common adverse effects of antipsychotic medications. *World Psychiatry Off. J. World Psychiatr. Assoc.* 2018, 17, 341–356. [CrossRef]
- 30. Lally, J.; MacCabe, J.H. Antipsychotic medication in schizophrenia: A review. Br. Med. Bull. 2015, 114, 169–179. [CrossRef]
- 31. Norris, M.L.; Spettigue, W.; Buchholz, A.; Henderson, K.A.; Gomez, R.; Maras, D.; Gaboury, I.; Ni, A. Olanzapine use for the adjunctive treatment of adolescents with anorexia nervosa. *J. Child Adolesc. Psychopharmacol.* **2011**, *21*, 213–220. [CrossRef]
- Çöpür, S.; Çöpür, M. Olanzapine in the treatment of anorexia nervosa: A systematic review. *Egypt. J. Neurol. Psychiatry Neurosurg.* 2020, 56, 60. [CrossRef]
- Milano, W.; De Rosa, M.; Milano, L.; Riccio, A.; Sanseverino, B.; Capasso, A. The Pharmacological Options in the Treatment of Eating Disorders. *ISRN Pharmacol.* 2013, 2013, 352865. [CrossRef]
- 34. Fariba, K.; Saadabadi, A. Topiramate; StatPearls Publishing: Treasure Island, FL, USA, 2020.
- Grant, J.E.; Schreiber, L.; Odlaug, B.L. Impulse control disorders: Updated review of clinical characteristics and pharmacological management. *Front. Psychiatry* 2011, 2, 1.
- 36. Douaihy, A.B.; Kelly, T.M.; Sullivan, C. Medications for substance use disorders. *Soc. Work Public Health* **2013**, *28*, 264–278. [CrossRef]
- 37. Fontenelle, L.F.; Oostermeijer, S.; Harrison, B.J.; Pantelis, C.; Yücel, M. Obsessive-compulsive disorder, impulse control disorders and drug addiction. *Drugs* **2011**, *71*, 827–840. [CrossRef]
- Kayser, R.R. Pharmacotherapy for treatment-resistant obsessive-compulsive disorder. J. Clin. Psychiatry 2020, 81, 14428. [CrossRef] [PubMed]
- Mouaffak, F.; Leite, C.; Hamzaoui, S.; Benyamina, A.; Laqueille, X.; Kebir, O. Naltrexone in the treatment of broadly defined behavioral addictions: A review and meta-analysis of randomized controlled trials. *Eur. Addict. Res.* 2017, 23, 204–210. [CrossRef] [PubMed]
- Bolton, M.; Hodkinson, A.; Boda, S.; Mould, A.; Panagioti, M.; Rhodes, S.; Riste, L.; van Marwijk, H. Serious adverse events reported in placebo randomised controlled trials of oral naltrexone: A systematic review and meta-analysis. *BMC Med.* 2019, 17, 10. [CrossRef] [PubMed]
- Timäus, C.; Meiser, M.; Bandelow, B.; Engel, K.R.; Paschke, A.M.; Wiltfang, J.; Wedekind, D. Pharmacotherapy of borderline personality disorder: What has changed over two decades? A retrospective evaluation of clinical practice. *BMC Psychiatry* 2019, 19, 393. [CrossRef]
- 42. Bozzatello, P.; Ghirardini, C.; Uscinska, M.; Rocca, P.; Bellino, S. Pharmacotherapy of personality disorders: What we know and what we have to search for. *Future Neurol.* **2017**, *12*, 199–222. [CrossRef]
- Uher, R.; Zwicker, A. Etiology in psychiatry: Embracing the reality of poly-gene-environmental causation of mental illness. World Psychiatry Off. J. World Psychiatr. Assoc. 2017, 16, 121–129. [CrossRef]
- 44. Wu, N.S.; Schairer, L.C.; Dellor, E.; Grella, C. Childhood trauma and health outcomes in adults with comorbid substance abuse and mental health disorders. *Addict. Behav.* 2010, *35*, 68–71. [CrossRef]

- 45. Gold, P.W.; Wong, M.-L. Re-assessing the catecholamine hypothesis of depression: The case of melancholic depression. *Mol. Psychiatry* **2021**, *26*, 6121–6124. [CrossRef]
- 46. Albert, P.R.; Benkelfat, C.; Descarries, L. The neurobiology of depression—Revisiting the serotonin hypothesis. I. Cellular and molecular mechanisms. *Philos. Trans. R. Soc. Lond. Ser. B Biol. Sci.* **2012**, *367*, 2378–2381. [CrossRef]
- 47. Baumeister, D.; Barnes, G.; Giaroli, G.; Tracy, D. Classical hallucinogens as antidepressants? A review of pharmacodynamics and putative clinical roles. *Ther. Adv. Psychopharmacol.* **2014**, *4*, 156–169. [CrossRef]
- 48. Malaca, S.; Lo Faro, A.F.; Tamborra, A.; Pichini, S.; Busardò, F.P.; Huestis, M.A. Toxicology and Analysis of Psychoactive Tryptamines. *Int. J. Mol. Sci.* 2020, 21, 9279. [CrossRef]
- 49. Lew, S.Y.; Lim, S.H.; Lim, L.W.; Wong, K.H. Neuroprotective effects of *Hericium erinaceus* (Bull.: Fr.) Pers. against high-dose corticosterone-induced oxidative stress in PC-12 cells. *BMC Complement. Med. Ther.* **2020**, *20*, 340. [CrossRef]
- Conrad, C.D. Chronic stress-induced hippocampal vulnerability: The glucocorticoid vulnerability hypothesis. *Rev. Neurosci.* 2008, 19, 395–412. [CrossRef]
- Crocker, L.; Heller, W.; Warren, S.; O'Hare, A.; Infantolino, Z.; Miller, G. Relationships among cognition, emotion, and motivation: Implications for intervention and neuroplasticity in psychopathology. Review. *Front. Hum. Neurosci.* 2013, 7, 261. [CrossRef]
- 52. Volpi-Abadie, J.; Kaye, A.M.; Kaye, A.D. Serotonin syndrome. Ochsner J. 2013, 13, 533–540.
- 53. Vargas, M.V.; Meyer, R.; Avanes, A.A.; Rus, M.; Olson, D.E. Psychedelics and Other Psychoplastogens for Treating Mental Illness. Review. *Front. Psychiatry* **2021**, *12*, 1691. [CrossRef]
- 54. Thompson, C.; Szabo, A. Psychedelics as a novel approach to treating autoimmune conditions. *Immunol. Lett.* **2020**, *228*, 45–54. [CrossRef]
- Janssens, K.A.; Zijlema, W.L.; Joustra, M.L.; Rosmalen, J.G. Mood and Anxiety Disorders in Chronic Fatigue Syndrome, Fibromyalgia, and Irritable Bowel Syndrome: Results from the LifeLines Cohort Study. *Psychosom. Med.* 2015, 77, 449–457. [CrossRef]
- 56. Üçeyler, N.; Sommer, C. Fibromyalgiesyndrom. Z. Rheumatol. 2015, 74, 490–495. [CrossRef]
- 57. Albrecht, D.S.; MacKie, P.J.; Kareken, D.A.; Hutchins, G.D.; Chumin, E.J.; Christian, B.T.; Yoder, K.K. Differential dopamine function in fibromyalgia. *Brain Imaging Behav.* 2016, 10, 829–839. [CrossRef]
- 58. Moret, C.; Briley, M. Antidepressants in the treatment of fibromyalgia. *Neuropsychiatr. Dis. Treat.* **2006**, *2*, 537–548. [CrossRef] [PubMed]
- 59. Natelson, B.H.; Lin, J.S.; Lange, G.; Khan, S.; Stegner, A.; Unger, E.R. The effect of comorbid medical and psychiatric diagnoses on chronic fatigue syndrome. *Ann. Med.* **2019**, *51*, 371–378. [CrossRef] [PubMed]
- 60. Fadgyas-Stanculete, M.; Buga, A.-M.; Popa-Wagner, A.; Dumitrascu, D.L. The relationship between irritable bowel syndrome and psychiatric disorders: From molecular changes to clinical manifestations. *J. Mol. Psychiatry* **2014**, *2*, 4. [CrossRef] [PubMed]
- 61. Jarrett, M.E.; Kohen, R.; Cain, K.C.; Burr, R.L.; Poppe, A.; Navaja, G.P.; Heitkemper, M.M. Relationship of SERT polymorphisms to depressive and anxiety symptoms in irritable bowel syndrome. *Biol. Res. Nurs.* **2007**, *9*, 161–169. [CrossRef]
- Fukudo, S.; Kanazawa, M.; Mizuno, T.; Hamaguchi, T.; Kano, M.; Watanabe, S.; Sagami, Y.; Shoji, T.; Endo, Y.; Hongo, M.; et al. Impact of serotonin transporter gene polymorphism on brain activation by colorectal distention. *Neuroimage* 2009, 47, 946–951. [CrossRef]
- 63. Smith, J.E.; Rowan, N.J.; Sullivan, R. Medicinal mushrooms: A rapidly developing area of biotechnology for cancer therapy and other bioactivities. *Biotechnol. Lett.* **2002**, *24*, 1839–1845. [CrossRef]
- 64. Sullivan, R.; Smith, J.E.; Rowan, N.J. Medicinal mushrooms and cancer therapy: Translating a traditional practice into Western medicine. *Perspect. Biol. Med.* 2006, 49, 159–170. [CrossRef]
- Murphy, E.J.; Rezoagli, E.; Major, I.; Rowan, N.J.; Laffey, J.G. β-glucan metabolic and immunomodulatory properties and potential for clinical application. J. Fungi 2020, 6, 356. [CrossRef]
- 66. Murphy, E.J.; Masterson, C.; Rezoagli, E.; O'Toole, D.; Major, I.; Stack, G.D.; Lynch, M.; Laffey, J.G.; Rowan, N.J. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—Implications for coronavirus disease (COVID-19) immunotherapies. *Sci. Total Environ.* **2020**, *732*, 139330. [CrossRef]
- 67. Murphy, E.J.; Rezoagli, E.; Pogue, R.; Simonassi-Paiva, B.; Izwani, I.; Abidin, Z.; Waltzer Fehrenbach, G.; O'Neil, E.; Major, I.; Laffey, J.G.; et al. Immunomodulatory activity of β-glucan polysaccharides isolated from different species of mushroom–A potential treatment for inflammatory lung conditions. *Sci. Total Environ.* 2022, *809*, 152177. [CrossRef]
- 68. Rowan, N.J.; Galanakis, C.M. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? *Sci. Total Environ.* **2020**, *748*, 141362. [CrossRef] [PubMed]
- Rowan, N.J.; Casey, O. Empower Eco multiactor HUB: A triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. *Curr. Opin. Environ. Sci. Health* 2021, 21, 100254. [CrossRef]
- Allam, Z.; Sharifi, A.; Giurco, D.; Sharpe, S.A. On the theoretical conceptualisations, knowledge structures and trends of green new deals. *Sustainability* 2021, 13, 12529. [CrossRef]
- Yildiz, O.; Can, Z.; Laghari, A.Q.; Şahin, H.; Malkoç, M. Wild edible mushrooms as a natural source of phenolics and antioxidants. J. Food Biochem. 2015, 39, 148–154. [CrossRef]

- 72. Dos Santos, R.G.; Bouso, J.C.; Rocha, J.M.; Rossi, G.N.; Hallak, J.E. The use of classic hallucinogens/psychedelics in a therapeutic context: Healthcare policy opportunities and challenges. *Risk Manag. Healthc. Policy* **2021**, *14*, 901. [CrossRef]
- 73. Whelan, A.; Johnson, M.I. Lysergic acid diethylamide and psilocybin for the management of patients with persistent pain: A potential role? *Pain Manag.* **2018**, *8*, 217–229. [CrossRef]
- Olson, D.E. Psychoplastogens: A Promising Class of Plasticity-Promoting Neurotherapeutics. J. Exp. Neurosci. 2018, 12, 1–4. [CrossRef]
- 75. Gill, H.; Gill, B.; Chen-Li, D.; El-Halabi, S.; Rodrigues, N.B.; Cha, D.S.; Lipsitz, O.; Lee, Y.; Rosenblat, J.D.; Majeed, A.; et al. The emerging role of psilocybin and MDMA in the treatment of mental illness. *Expert Rev. Neurother.* 2020, 20, 1263–1273. [CrossRef]
- Pretorius, L.; Smith, C. The trace aminergic system: A gender-sensitive therapeutic target for IBS? J. Biomed. Sci. 2020, 27, 95. [CrossRef]
- 77. Mayet, S. A review of common psychedelic drugs. S. Afr. J. Anaesth. Analg. 2020, 26, S113–S117. [CrossRef]
- Uthaug, M.V.; Davis, A.K.; Haas, T.F.; Dawis, D.; Dolan, S.B.; Lancelotta, R.; Timmermann, C.; Ramaekers, J.G. The epidemiology of mescaline use: Pattern of use, motivations for consumption, and perceived consequences, benefits, and acute and enduring subjective effects. J. Psychopharmacol. 2021, 36, 309–320. [CrossRef]
- 79. Daniel, J.; Haberman, M. Clinical potential of psilocybin as a treatment for mental health conditions. *Ment. Health Clin.* **2017**, *7*, 24–28. [CrossRef]
- 80. Passie, T.; Halpern, J.H.; Stichtenoth, D.O.; Emrich, H.M.; Hintzen, A. The pharmacology of lysergic acid diethylamide: A review. *CNS Neurosci. Ther.* **2008**, *14*, 295–314. [CrossRef]
- Lowe, H.; Toyang, N.; Steele, B.; Valentine, H.; Grant, J.; Ali, A.; Ngwa, W.; Gordon, L. The Therapeutic Potential of Psilocybin. *Molecules* 2021, 26, 2948. [CrossRef]
- Halberstadt, A.L.; Geyer, M.A. Multiple receptors contribute to the behavioral effects of indoleamine hallucinogens. *Neuropharma-cology* 2011, 61, 364–381. [CrossRef]
- Cumming, P.; Scheidegger, M.; Dornbierer, D.; Palner, M.; Quednow, B.B.; Martin-Soelch, C. Molecular and functional imaging studies of psychedelic drug action in animals and humans. *Molecules* 2021, 26, 2451. [CrossRef]
- 84. Agin-Liebes, G.; Haas, T.F.; Lancelotta, R.; Uthaug, M.V.; Ramaekers, J.G.; Davis, A.K. Naturalistic use of mescaline is associated with self-reported psychiatric improvements and enduring positive life changes. *ACS Pharmacol. Transl. Sci.* **2021**, *4*, 543–552. [CrossRef]
- Holze, F.; Vizeli, P.; Ley, L.; Müller, F.; Dolder, P.; Stocker, M.; Duthaler, U.; Varghese, N.; Eckert, A.; Borgwardt, S.; et al. Acute dose-dependent effects of lysergic acid diethylamide in a double-blind placebo-controlled study in healthy subjects. *Neuropsychopharmacology* 2021, 46, 537–544. [CrossRef]
- Dinis-Oliveira, R.J. Metabolism of psilocybin and psilocin: Clinical and forensic toxicological relevance. *Drug Metab. Rev.* 2017, 49, 84–91. [CrossRef]
- Libânio Osório Marta, R.F. Metabolism of lysergic acid diethylamide (LSD): An update. Drug Metab. Rev. 2019, 51, 378–387.
   [CrossRef]
- 88. Dinis-Oliveira, R.J.; Pereira, C.L.; da Silva, D.D. Pharmacokinetic and Pharmacodynamic Aspects of Peyote and Mescaline: Clinical and Forensic Repercussions. *Curr. Mol. Pharmacol.* **2019**, *12*, 184–194. [CrossRef]
- 89. Passie, T.; Seifert, J.; Schneider, U.; Emrich, H.M. The pharmacology of psilocybin. Addict. Biol. 2002, 7, 357–364. [CrossRef]
- Carhart-Harris, R.L.; Bolstridge, M.; Rucker, J.; Day, C.M.J.; Erritzoe, D.; Kaelen, M.; Bloomfield, M.; Rickard, J.A.; Forbs, B.; Feilding, A.; et al. Psilocybin with psychological support for treatment-resistant depression: An open-label feasibility study. *Lancet Psychiatry* 2016, *3*, 619–627. [CrossRef]
- Fuentes, J.J.; Fonseca, F.; Elices, M.; Farré, M.; Torrens, M. Therapeutic use of LSD in psychiatry: A systematic review of randomized-controlled clinical trials. *Front. Psychiatry* 2020, 10, 943. [CrossRef]
- 92. Nichols, D.E. Psychedelics. Pharmacol. Rev. 2016, 68, 264–355. [CrossRef]
- 93. Sherwood, A.M.; Prisinzano, T.E. Novel psychotherapeutics—A cautiously optimistic focus on hallucinogens. *Expert Rev. Clin. Pharmacol.* **2018**, *11*, 1–3. [CrossRef]
- 94. Carlini, E.A.; Maia, L.O. Plant and fungal hallucinogens as toxic and therapeutic agents. In *Plant Toxins Toxinology*; Gopalakrishnakone, P., Carlini, C., Ligabue-Braun, R., Eds.; Springer: Berlin, Germany, 2017; pp. 37–80.
- 95. Olson, D.E. The Promise of Psychedelic Science. ACS Pharmacol. Transl. Sci. 2021, 4, 413–415. [CrossRef]
- 96. Ross, S.; Bossis, A.; Guss, J.; Agin-Liebes, G.; Malone, T.; Cohen, B.; Mennenga, S.E.; Belser, A.; Kalliontzi, K.; Babb, J.; et al. Rapid and sustained symptom reduction following psilocybin treatment for anxiety and depression in patients with life-threatening cancer: A randomized controlled trial. *J. Psychopharmacol.* **2016**, *30*, 1165–1180. [CrossRef]
- 97. Griffiths, R.R.; Johnson, M.W.; Carducci, M.A.; Umbrich, A.; Richards, W.A.; Richards, B.D.; Cosimano, M.P.; Klinedinst, M.A. Psilocybin produces substantial and sustained decreases in depression and anxiety in patients with life-threatening cancer: A randomized double-blind trial. *J. Psychopharmacol.* 2016, 30, 1181–1197. [CrossRef]
- 98. Johnson, M.W.; Garcia-Romeu, A.; Griffiths, R.R. Long-term follow-up of psilocybin-facilitated smoking cessation. *Am. J. Drug Alcohol Abus.* 2017, 43, 55–60. [CrossRef]
- 99. Moreno, F.A.; Wiegand, C.B.; Taitano, E.K.; Delgado, P.L. Safety, tolerability, and efficacy of psilocybin in 9 patients with obsessive-compulsive disorder. *J. Clin. Psychiatry* 2006, *67*, 1735–1740. [CrossRef] [PubMed]

- Negård, M.; Uhlig, S.; Kauserud, H.; Andersen, T.; Høiland, K.; Vrålstad, T. Links between genetic groups, indole alkaloid profiles and ecology within the grass-parasitic *Claviceps purpurea* species complex. *Toxins* 2015, 7, 1431–1456. [CrossRef] [PubMed]
- Rodríguez Arce, J.M.; Winkelman, M.J. Psychedelics, sociality, and human evolution. *Front. Psychol.* 2021, 12, 729425. [CrossRef] [PubMed]
- Halberstadt, A.L. Hallucinogenic drugs: A new study answers old questions about LSD. Curr. Biol. 2017, 27, R156–R158.
   [CrossRef]
- 103. De Gregorio, D.; Comai, S.; Posa, L.; Gobbi, G. d-Lysergic acid diethylamide (LSD) as a model of psychosis: Mechanism of action and pharmacology. *Int. J. Mol. Sci.* **2016**, *17*, 1953. [CrossRef]
- 104. Rogers, T.J. The molecular basis for neuroimmune receptor signaling. J. Neuroimmune Pharmacol. 2012, 7, 722–724. [CrossRef]
- 105. Hutten, N.R.P.W.; Mason, N.L.; Dolder, P.C.; Kuypers, K.P.C. Self-rated effectiveness of microdosing with psychedelics for mental and physical health problems among microdosers. *Front. Psychiatry* **2019**, *10*, 672. [CrossRef]
- 106. Ramaekers, J.G.; Hutten, N.; Mason, N.L.; Dolder, P.; Theunissen, E.L.; Holze, F.; Liechti, M.E.; Feilding, A.; Kuypers, K.P.C. A low dose of lysergic acid diethylamide decreases pain perception in healthy volunteers. J. Psychopharmacol. 2021, 35, 398–405. [CrossRef]
- 107. Tupper, K.W.; Wood, E.; Yensen, R.; Johnson, M.W. Psychedelic medicine: A re-emerging therapeutic paradigm. *Can. Med. Assoc. J.* **2015**, *187*, 1054–1059. [CrossRef]
- 108. Lünne, F.; Köhler, J.; Stroh, C.; Müller, L.; Daniliuc, C.G.; Mück-Lichtenfeld, C.; Würthwein, E.-U.; Esselen, M.; Humpf, H.-U.; Kalinina, S.A. Insights into Ergochromes of the Plant Pathogen *Claviceps purpurea*. J. Nat. Prod. **2021**, 84, 2630–2643. [CrossRef]
- 109. Carboué, Q.; Lopez, M. Amanita muscaria: Ecology, Chemistry, Myths. Encyclopedia 2021, 1, 905–914. [CrossRef]
- Michelot, D.; Melendez-Howell, L.M. Amanita muscaria: Chemistry, biology, toxicology, and ethnomycology. Mycol. Res. 2003, 107, 131–146. [CrossRef]
- Mikaszewska-Sokolewicz, M.A.; Pankowska, S.; Janiak, M.; Pruszczyk, P.; Łazowski, T.; Jankowski, K. Coma in the course of severe poisoning after consumption of red fly agaric (*Amanita muscaria*). Acta Biochim. Pol. 2016, 63, 181–182. [CrossRef]
- 112. Corbett, R.; Fielding, S.; Cornfeldt, M.; Dunn, R.W. GABAmimetic agents display anxiolytic-like effects in the social interaction and elevated plus maze procedures. *Psychopharmacology* **1991**, *104*, 312–316. [CrossRef]
- Hosseini, M.; Karami, Z.; Yousefifard, M.; Janzadeh, A.; Zamani, E.; Nasirinezhad, F. Simultaneous intrathecal injection of muscimol and endomorphin-1 alleviates neuropathic pain in rat model of spinal cord injury. *Brain Behav.* 2020, 10, e01576. [CrossRef]
- 114. Mishraki-Berkowitz, T.; Kochelski, E.; Kavanagh, P.; O'Brien, J.; Dunne, C.; Talbot, B.; Ennis, P.; Wolf, U. The Psilocin (4-hydroxy-N, N-dimethyltryptamine) and Bufotenine (5-hydroxy-N, N-dimethyltryptamine) Case: Ensuring the Correct Isomer has Been Identified. J. Forensic Sci. 2020, 65, 1450–1457. [CrossRef]
- 115. Blei, F.; Dörner, S.; Fricke, J.; Baldeweg, F.; Trottmann, F.; Komor, A.; Meyer, F.; Hertweck, C.; Hoffmeister, D. Simultaneous production of psilocybin and a cocktail of β–carboline monoamine oxidase inhibitors in "magic" mushrooms. *Chem.–A Eur. J.* 2020, *26*, 729–734. [CrossRef]
- Davis, A.K.; Barsuglia, J.P.; Lancelotta, R.; Grant, R.M.; Renn, E. The epidemiology of 5-methoxy- N, N-dimethyltryptamine (5-MeO-DMT) use: Benefits, consequences, patterns of use, subjective effects, and reasons for consumption. *J. Psychopharmacol.* 2018, 32, 779–792. [CrossRef]
- 117. Ma, B.-J.; Shen, J.-W.; Yu, H.-Y.; Ruan, Y.; Wu, T.-T.; Zhao, X. Hericenones and erinacines: Stimulators of nerve growth factor (NGF) biosynthesis in *Hericium erinaceus*. *Mycology* **2010**, *1*, 92–98. [CrossRef]
- 118. Rai, S.N.; Mishra, D.; Singh, P.; Vamanu, E.; Singh, M.P. Therapeutic applications of mushrooms and their biomolecules along with a glimpse of in silico approach in neurodegenerative diseases. *Biomed. Pharmacother.* **2021**, 137, 111377. [CrossRef]
- 119. Yao, W.; Zhang, J.-c.; Dong, C.; Zhuang, C.; Hirota, S.; Inanaga, K.; Hashimoto, K. Effects of amycenone on serum levels of tumor necrosis factor-α, interleukin-10, and depression-like behavior in mice after lipopolysaccharide administration. *Pharmacol. Biochem. Behav.* **2015**, *136*, 7–12. [CrossRef] [PubMed]
- 120. Li, I.; Lee, L.-Y.; Tzeng, T.-T.; Chen, W.-P.; Chen, Y.-P.; Shiao, Y.-J.; Chen, C.-C. Neurohealth properties of *Hericium erinaceus* mycelia enriched with erinacines. *Behav. Neurol.* **2018**, 2018, 5802634. [CrossRef] [PubMed]
- 121. Kim, Y.O.; Lee, S.W.; Kim, J.S. A comprehensive review of the therapeutic effects of *Hericium erinaceus* in neurodegenerative disease. *J. Mushroom* **2014**, *12*, 77–81. [CrossRef]
- 122. Nagano, M.; Shimizu, K.; Kondo, R.; Hayashi, C.; Sato, D.; Kitagawa, K.; Ohnuki, K. Reduction of depression and anxiety by 4 weeks *Hericium erinaceus* intake. *Biomed. Res.* **2010**, *31*, 231–237. [CrossRef]
- 123. Chiu, C.-H.; Chyau, C.-C.; Chen, C.-C.; Lee, L.-Y.; Chen, W.-P.; Liu, Y.-L.; Lin, W.-H.; Mong, M.-C. Erinacine A-enriched *Hericium erinaceus* mycelium produces antidepressant-like effects through modulating BDNF/PI3K/Akt/GSK-3β signaling in mice. *Int. J. Mol. Sci.* 2018, *19*, 341. [CrossRef]
- 124. Chong, P.S.; Poon, C.H.; Roy, J.; Tsui, K.C.; Lew, S.Y.; Lok Phang, M.W.; Yuenyinn Tan, R.J.; Cheng, P.G.; Fung, M.-L.; Wong, K.H.; et al. Neurogenesis-dependent antidepressant-like activity of *Hericium erinaceus* in an animal model of depression. *Chin. Med.* 2021, 16, 132. [CrossRef]
- 125. Cheah, I.K.; Halliwell, B. Ergothioneine, recent developments. Redox Biol. 2021, 42, 101868. [CrossRef]

- 126. Nakamichi, N.; Nakayama, K.; Ishimoto, T.; Masuo, Y.; Wakayama, T.; Sekiguchi, H.; Sutoh, K.; Usumi, K.; Iseki, S.; Kato, Y. Food-derived hydrophilic antioxidant ergothioneine is distributed to the brain and exerts antidepressant effect in mice. *Brain Behav.* 2016, 6, e00477. [CrossRef]
- 127. Matsuda, Y.; Ozawa, N.; Shinozaki, T.; Wakabayashi, K.-i.; Suzuki, K.; Kawano, Y.; Ohtsu, I.; Tatebayashi, Y. Ergothioneine, a metabolite of the gut bacterium *Lactobacillus reuteri*, protects against stress-induced sleep disturbances. *Transl. Psychiatry* 2020, 10, 170. [CrossRef]
- 128. Orrico-Sanchez, A.; Chausset-Boissarie, L.; Alves de Sousa, R.; Coutens, B.; Rezai Amin, S.; Vialou, V.; Louis, F.; Hessani, A.; Dansette, P.M.; Zornoza, T.; et al. Antidepressant efficacy of a selective organic cation transporter blocker in a mouse model of depression. *Mol. Psychiatry* 2020, 25, 1245–1259. [CrossRef]
- 129. Dolder, P.C.; Schmid, Y.; Steuer, A.E.; Kraemer, T.; Rentsch, K.M.; Hammann, F.; Liechti, M.E. Pharmacokinetics and pharmacodynamics of lysergic acid diethylamide in healthy subjects. *Clin. Pharmacokinet.* **2017**, *56*, 1219–1230. [CrossRef]
- Brown, R.T.; Nicholas, C.R.; Cozzi, N.V.; Gassman, M.C.; Cooper, K.M.; Muller, D.; Thomas, C.D.; Hetzel, S.J.; Henriquez, K.M.; Ribaudo, A.S.; et al. Pharmacokinetics of escalating doses of oral psilocybin in healthy adults. *Clin. Pharmacokinet.* 2017, 56, 1543–1554. [CrossRef]
- 131. Masterson, C.H.; Murphy, E.J.; Gonzalez, H.; Major, I.; McCarthy, S.D.; O'Toole, D.; Laffey, J.G.; Rowan, N.J. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant *Klebsiella pneumoniae*-induced pulmonary sepsis. *Lett. Appl. Microbiol.* 2020, 71, 405–412. [CrossRef]
- 132. Usuldin, S.R.A.; Wan-Mohtar, W.A.A.Q.I.; Ilham, Z.; Jamaludin, A.A.; Abdullah, N.R.; Rowan, N. In vivo toxicity of bioreactorgrown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications. *Sci. Rep.* **2021**, *11*, 23079. [CrossRef]

#### Section Five – Elucidating and Developing New Sustainable Technologies and Frameworks

The final section describes some of my latter years research as Director of the Bioscience Research Institute in Ireland at TUS on the holistic development of sustainable innovation at demonstration sites. These studies with and without novel processing or disease-mitigating technologies where emphasis is on developing a new integrated, multitrophic aquaculture system (IMTA) that recirculates clean water based on use of fish waste stream by microalgae and duckweed present in the ponds. Studies report on the first use IMTA in the peatlands as alternative means of sustainable land use aligned to zero waste, zero pollution and climate action principles. Studies showed that an understanding of key disease preventative approaches was critical to informing reliable and repeatable operation of the IMTA site where pulsed UV was removed as an end-of-pipe solution (as no discharge of aquaculture effluent to receiving water occurs negating PUV or other technology use). Studies report on the simultaneous use of hand-held diagnostic equipment with 'in field' sensors for monitoring fish culture ponds, fish health and disease mitigation. Bioinformatics with next-generation sequencing are also used to support and inform traditional enumerating techniques for entire ecosystem in IMTA peatland site. Studies revealed that climate change produced three successive storm events that flooded this IMTA system causing significant fish mortality where there was a requirement to introduce disease mitigation approaches (non-chemical solutions), as site is organic status. Studies describe first use of multi-actor hubs including triple to quintuple helix frameworks for integrated development of innovation. Studies address modelling and geo-mapping of chemicals of emerging concerns on the European Watch List in water and the importance of understanding and identifying sensitive detection methods and having appropriate disinfection assets at waste water treatment facilities as critical control points. Studies describe life cycle assessment and ecological tools to confirm efficacy and appropriateness of innovation for food and environmental deployment. Studies also compare and address large scale-disinfection technologies for effective treatment and enumeration of food and water-borne parasites in their different life-cycle development/growth phases as these are particularly complex, recalcitrant and fastidious pathogens.

Contents lists available at ScienceDirect

### Aquaculture



journal homepage: www.elsevier.com/locate/aquaculture

Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: A case study from the Republic of Ireland

Check for updates

Sarah Naughton<sup>a,\*</sup>, Siobhán Kavanagh<sup>a,b</sup>, Mark Lynch<sup>a</sup>, Neil J. Rowan<sup>a,c</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Ireland

<sup>b</sup> Department of Life and Physical Science, Athlone Institute of Technology, Ireland

<sup>c</sup> Centre for Disinfection, Sterilisation and Biosecurity, Athlone Institute of Technology, Ireland

#### ARTICLE INFO

Keywords: Microalgae Freshwater aquaculture Waste treatment Resource utilization Sustainability

#### ABSTRACT

There has been growing interest in exploiting microalgae as a natural process for low cost wastewater treatment and for water quality control and remediation in aquaculture. This constitutes the first study to report on a strong relationship between use of sophisticated wet-laboratory flow cytometry equipment and in-field AlgaeTorch<sup>®</sup> technologies for determining microalgae and bacteria population dynamics in a freshwater pillpond aquaculture farm over a 10-month monitoring period producing Eurasian Perch, *Perca fluviatilis*, in the Republic of Ireland. Nitrate levels and temperature were the most significant factors influencing microalgae numbers in rearing and treatment ponds as determined by Principle Component Analysis. Variance in climate, namely drought conditions that occurred during monitoring period, did not affect microalgae or microbial numbers. *Chlorophyta, Bacillariophyta* and *Cryptophyta* were the most dominant algal divisions observed in this recirculating aquaculture system, many of these are recognized as a natural source of beneficial prebiotics for fish. Determining baseline microalgal profiles in rearing water, followed by elucidating physicochemical parameters governing wastewater treatment performance, can inform future intensification and diversification of freshwater aquaculture by exploiting and replicating knowledge of favourable algal-microbial ecosystems. Furthermore, holistic datasets can be utilised for smart agriculture by way of informing management tools for future remote monitoring and decision-making by producers.

#### 1. Introduction

Aquaculture has become the fastest-growing food sector in the world (FAO, 2018; Ruis-Salmon et al., 2020). During the period 2010 to 2030, global aquaculture production needs to increase threefold in order to meet the demands for fish and food (DAFM, 2015; O'Neill et al., 2019; O'Neill et al., 2020). In the aquaculture industry, water quality needs to be closely monitored in order to maintain, as closely as possible, the optimal growth conditions for a given cultured fish, and consequently to ensure optimal production (European Food Safety Authority (EFSA), 2008). Commensurately, water pollution has become a concern, posing threats to environmental protection that will retard intensive sustainability of aquaculture globally (Tahar et al., 2018;

Tahar et al., 2019). To overcome these challenges, significant effort has been devoted to control wastewater pollution and to improve water quality control in aquaculture (Han et al., 2019). However, traditional environmental remediation approaches, such as aeration, filtration and other physical technologies, require high energy consumption or substantial add to the investment that increases total cost and financial burden of the industry (Longo et al., 2016; Tahar et al., 2019). These traditional technologies are often unable to fully utilise and recycle resources such as nutrients (including nitrogen, phosphorous and carbon) along with producing large amounts of  $CO_2$  and sludge that cause secondary environmental pollution (Lu et al., 2019a). Moreover, antibiotics and medicines are frequently used in aquaculture to mitigate against disease and to reduce risk of outbreaks, which adds to growing

\* Corresponding author.

https://doi.org/10.1016/j.aquaculture.2020.735377

Received 21 November 2019; Received in revised form 4 March 2020; Accepted 14 April 2020 Available online 16 April 2020

0044-8486/ © 2020 Elsevier B.V. All rights reserved.

E-mail address: s.naughton@research.ait.ie (S. Naughton).
concerns over the antibiotic-resistance crisis globally (Muziasari et al., 2016). Consequently, there has been growing interest in the development of alternative or complementary environmental-friendly and economically feasible solutions to advance aquaculture (O'Neill et al., 2020).

In aquaculture, fish are reared at high densities for increased productivity, which can lead to the build-up of in-organic nutrients, excreted waste and feed remnants that can lead to unwanted eutrophication in the receiving aquatic environment (Bentzon-Tilia et al., 2016). When conditions are optimal, namely high nutrient loads, high temperature and sunlight, algae can grow to an excessive number forming either harmful or beneficial blooms in the rearing water that can effect fish health (Drikas et al., 2001). With high density rearing practices, harmful pathogens also have a greater chance of rapid circulation and persistence resulting in the potential to cause a disease outbreak. There is pressing need to monitor and manage microalgae, bacteria and key parameters in rearing water on fish farms that both detects and mitigates for the emergence of pathogens in order to make remedial management actions in real time (O'Neill et al., 2020). However, biomass analysis tends to be an insufficient method for speciation of microalgae as they lack distinctive features (Li et al., 2012). This highlights the importance of traditional time-consuming microscopic analysis in aquaculture.

There are many approaches to enumerating microalgae that have been traditionally limited to using conventional slide methods, such as the Sedgewick-Rafter slide, the Palmer-Maloney slide, the Petroff-Hausser slide and the standard haemocytometer method (Guillard, 1978; Han et al., 2019). The most common method for phytoplankton enumeration, especially for multispecies communities, is the Utermöhl method that requires an inverted microscopic with sophisticated optics in order to ensure reliable results (Vuorio et al., 2007). There is also considerable inter-variation between operators of the method, and variation between microscopes used that affects reliability and robustness (Vuorio et al., 2007). The use of real-time enumeration methods in marine aquaculture, such as flow cytometry (FCM), enables rapidcounting and differentiation of more than 10,000 cells or targets per second that can add to accuracy of algal determinations in a non-destructive manner (Endo et al., 2010). Recently, researchers have reported on the use of handheld monitoring devices, such as AlgaeTorch®, for taxonomic classification of algae through measuring fluorescence (Szymański et al., 2017). However, comparative use of conventional and sophisticated wet-laboratory equipment (such as flow cytometry) with on-farm monitoring devices (such as AlgaeTorch®), has yet to be reported for characterization and monitoring of freshwater aquaculture that relies on using natural biological processes for controlling water quality.

Thus, the aim of this timely study was to gain a comprehensive understanding of the role and relationship of microalga species with key bacterial and physicochemical parameters in freshwater aquaculture over a 10-month monitoring period using both wet-laboratory and on-farm measurements in order to define and enhance freshwater aquaculture.

#### 2. Materials and methods

#### 2.1. Study site and sampling

Samples were collected from a freshwater fish farm located in Boyle, Co. Sligo, Republic of Ireland that produces Eurasian perch (*Perca fluviatilis*). It contains broodstock tanks, a hatchery for eggs and larvae and a nursery for juvenile growth. All of these indoor systems are based in tanks and operate under recirculating aquaculture systems (RAS). There are three grow-out ponds for the larger fish. These are earthen pill-pond systems based on low surface flow water sourced from a local stream. This study focused on one of the grow out ponds, designated Pond 1 that is denoted by the red perimeter shown in Fig. 1. Pond 1 is divided into two connecting ponds; the fish pond (FP) that is stocked with adult perch, and the treatment pond (TP) devoid of fish. Flow is circulated in and out of the FP region using paddle wheels guided by walls to aid aeration. The TP supports the growth of microalgae that also provides oxygenation and wastewater remediation. A schematic is also inserted into Fig. 1 to help contextualize the operational concept. Sampling occurred in triplicate from March 2018 to November 2018 in order to capture seasonality as briefly outlined in Table 1. Samples were transported in a cooler box to the laboratory for analysis. Preserved samples were stored at 4 °C until further analysis was carried out.

#### 2.2. Physicochemical parameter measurement

Physicochemical parameters that were measured in the laboratory included nitrates, nitrites, ammonium and phosphorus concentrations, pH and carbonate hardness. Whilst nitrates, nitrites, ammonia, phosphate, pH, oxygen and turbidity were measured *in situ* on the farm by standard methods. All parameters measured in the lab were carried out using individual test kits for each specific parameter as outlined in Table 2. After initial preservation with sulphuric acid, the pH was increased to between 6 and 7, which was required for all tests. Each test was carried out as per the individual kit instruction manual in duplicate. A Jenway UV–Vis Spectrophotometer was used for the spectrophotometric tests.

#### 2.3. Use of flow cytometry to enumerate microalgae

The MACSQuant<sup>®</sup> Analyzer 10 (Miltenyi Biotec) flow cytometry (FCM) system was used for this study. Samples were centrifuged at 3500 xg for 20 mins, thereafter the cell pellet was resuspended in running buffer (PBS with 1 mM EDTA, 0.2% Tween, 0.1% sodium azide, 0.2  $\mu$ m filtered). Samples were stained with nucleic acid dye SYBR Green (1:10,000×), to separate the DNA-containing cells from cellular debris and sedimentation present in the pond. SYBR Green fluorescence was detected on the B1 channel. An unstained algal control was required to eliminate natural auto-fluorescence in the B1 channel caused by excitation by the blue laser. Relative Cell Size and granularity was determined by forward scatter (FSC) and side scatter (SSC) channels. The blue laser stimulates chlorophyll (Chl) fluorescence on the B3 channel, and red laser stimulates phycocyanin (PC) fluorescence (R1 channel). Each sample was analysed in triplicate.

The gating method to obtain and enumerate the desired population, *i.e.* algae excluding cyanobacteria, was adapted from Moorhouse et al. (2018), Haynes et al. (2015) and Read et al. (2014) for phytoplankton and plankton analysis. This process is explained in Fig. 2 and involved running an unstained representative for each sample for the elimination of auto-fluorescence. This gate was used to determine DNA-containing cells in the SYBR Green-stained samples, without natural autofluorescence interference. Fig. 2(B) illustrates the distribution of cells present with all auto-fluorescent cells or unwanted material falling outside of the area/gate of interest. The Chl<sup>+</sup> population was isolated by acquiring the cells that fluoresced in the B3 channel. Fig. 2(C) illustrates the Chl<sup>+</sup> population with FSC on the x-axis relating to cell size and SSC on the y-axis relating to cell granularity and complexity. Cyanobacteria were present in this population as they also possess chlorophyll. This group was eliminated by graphing B3 against R1, and gating around the  $Chl^+/PC^-$  population. This is illustrated in Fig. 2(D) The final population depicted in Fig. 2(E) is the population of interest viewed under FSC and SSC. The isolated cyanobacterial populations from each sample were also analysed and enumerated to determine the trends over the duration of the study. All flow cytometry data gating was carried out using FlowJo software package.

#### 2.4. Microalga profile development and in-field AlgaeTorch monitoring

Identification using a standard inverted microscope for



Fig. 1. Perch farm flow through ponds - aerial view, with insert schematic of Pond 1 layout.

#### Table 1

Sampling regime for obtaining rearing water profile at the perch farm.

Sample type	Volume	Application	Frequency	Preservative
Algal	100 ml	Enumeration, Identification & Profile Development	Biweekly	1% Lugol's Iodine
Bacterial	500 ml	Enumeration	Biweekly	1% Formaldehyde
Physicochemical	500 ml	Parameter Measurement	Biweekly	Sulphuric Acid

#### Table 2

Test kits and devices for physicochemical parameter analysis at the perch farm including acceptable limits.

Parameter	Measurement	R <sup>2</sup> Coefficient	Result Ranges (m	g/l)	MAC (mg/l)		
			Fish pond (FP)	Treatment pond (TP)	SI No. 77 (2019)	Boyd (1998)	
Nitrates	Photometric [Spectroquant® Nitrate Test - 1 00713]	0.998	< 0.01–3.27	< 0.01–3.25	50	0.2–10.0	
Nitrites	[Spectroquant Nitrate Test = 1.05/13] Photometric [Spectroquant® Nitrite Test = 1.14776]	1	< 0.01–0.12	< 0.01–0.10	0.03	< 0.3	
Ammonium	Photometric	0.998	0.12–1.89	0.08–1.69	-	0.2–2.0	
Ammonia	[Spectroquant <sup>®</sup> Ammonium Test – 1.14/52] Calculation	-	< 0.01	< 0.01	< 0.03	< 0.1	
Phosphates	Photometric	0.999	0.17–1.66	0.10-2.20	0.025	0.005-0.2	
Carbonate Hardness	Titration [MColortest <sup>™</sup> Carbonate Hardness Test – 1.08048	-	70.06–650.57	50.04-630.55	-	50–200	

morphological analysis and photographic identification keys were used in conjunction with FCM for enumeration to maximise the information obtained. In addition to FCM analysis of microalgal populations, the chlorophyll *a* and Cyanobacteria populations were also measured *in situ* using the AlgaeTorch<sup>®</sup> (bbe Moldaenke). The AlgaeTorch<sup>®</sup> is based on real-time *in vivo* fluorescent measurement upon excitation of the microalgal cells in response to six LEDs of three different wavelengths, 470, 525 and 610 nm. The measurement analysis carried out is in accordance with ISO 10260:1992 Water quality – Measurement of biochemical parameters – Spectrometric determination of the chlorophyll*a* concentration and DIN 38412/16:1985 German standard methods for the examination of water, waste water and sludge; Test methods using water organisms (Group L); Determination of chlorophyll *a* in surface water (L 16). Both chlorophyll a and cyanobacterial content were measured in µg chl-a/l (bbe Moldaenke, 2017).

#### 2.5. Bacterial enumeration using epifluorescent microscopy

Sample preparation for bacterial enumeration was carried out as per Bloem and Vos (2004). SYBR Gold was used to stain the microbial cells. Bacteria were filtered using Sartorius filtration apparatus onto 0.22 µm isopore membrane filters and a support Whatman™ filter was used to enhance vacuum distribution filtered onto the membrane. Residual background staining was removed using distilled deionised water. The filter was placed on a glass slide and counted immediately to avoid fading of the stain (Kumaravel et al., 2009).



(caption on next page)

**Fig. 2.** Flow cytometry algal gating process involving (A) Gating for autofluorescence in the unstained sample, (B) Gating for relevant living organisms in the SYBR Green stained sample, (C) Chlorophyll positive population of the living cells viewed under FSC vs SSC, (D) Gating to eliminate cyanobacterial population and (E) Algal population of interest for enumeration viewed under FSC vs. SSC. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)



Fig. 3. AlgaeTorch diagram outlining the main components for operation (Source: bbe Moldaenke).

For each filter, a selection of random fields were counted until a total of at least 300 cells were counted over a minimum five fields. Counting was facilitated *via* the use of an epifluorescent microscope under oil immersion ( $100 \times$  objective lens). The samples were observed under a blue optical filter, which has an excitation from 465 to 505 nm, 510 nm cut-off; emission from 515 to 565 nm as adopted from Shibata et al. (2006). Bacterial cells were visualised as green dots or lines. Once a count was obtained, the following formula was used for enumeration of bacterial cells per millilitre:

Count x 176 Dilution factor x No. of fields counts x 0.186

where 176 was the area of the field of view on the microscope, and 0.186 was the filter area. Each sample was counted in triplicate and where numbers exceeded 100 cells per field of view, a minimum of 4 counts were obtained or a further dilution was made.

#### 2.6. Statistical analysis

All statistical analysis for physicochemical measurements, algal enumeration and bacterial enumeration was carried out using GraphPad Prism 8. A D'Agostino-Pearson normality test was carried out to test the normality of all data. Results from this informed the use of a parametric unpaired t-test or a nonparametric Mann-Whitney test for comparison of results between the FP and the TP. A value of  $p \le 0.05$ indicated a significant difference at the 95% level of confidence. FlowJo software package was used for analysing flow cytometry data and for the generation of cytograms in order to establish an algal cell count. XLSTAT was used to generate the Principle Component Analysis (PCA) using Pearson correlation matrix scores to compare all parameters analysed. The closer the score to 1 or (-) 1 the greater the positive or negative correlation between two parameters, respectively. In the case where correlations existed between parameters, yellow scores denote a moderately strong correlation, red scores denote a strong correlation and blue scores denote a very strong correlation.

#### 3. Results

#### 3.1. Physicochemical parameter analysis

All physicochemical measurements analysed in the lab are displayed

in Table 2. Fig. 4 displays the physicochemical parameter trends for the FP with similar results achieved for the TP. Parameter levels were assessed according to Boyd (1998) and SI No. 77 of 2019 - European Union Environmental Objectives (Surface Waters) (Amendment) Regulations 2019. Nitrates, nitrites, ammonium and phosphates results were expressed in mg/l, with water hardness expressed in the most common form as calcium carbonate hardness (CaCO<sub>3</sub>) as per Wurts and Durborow (1992). In all cases an  $R^2$  value of >0.99 was achieved for each standard curve, indicating the reliability of the test method with very little variation. The nitrate concentration peaked in October reaching levels of 3.27 mg/l and 3.25 mg/l for the FP and TP, respectively. The levels were lowest at the end of May with levels below the limit of detection recorded for both the FP and TP (Fig. 4). Nitrite levels ranged from below the level of detection in March to 0.118 mg/l and 0.103 mg/l in June, for the FP and TP, respectively (Fig. 4). The ammonium trends were lowest at levels of 0.12 and 0.08 mg/l for the FP and TP in April, respectively. The highest ammonium concentration levels were in June for both ponds, with a concentration of 1.89 mg/l in the FP and 1.69 mg/l in the TP (Fig. 4). The highest phosphate concentrations were measured at 1.66 mg/l in the FP and 2.20 mg/l in the TP (Fig. 4). Water hardness levels ranged from 70.06 to 650.57 mg  $CaCO_3/I$  in the FP and 50.04 to 630.55 mg  $CaCO_3/I$  in the TP (Fig. 4). The water hardness was highest in September, with average hardness levels reaching 157.95 mg CaCO<sub>3</sub>/l in FP and 129.53 mg CaCO<sub>3</sub>/l in TP (Fig. 4), when the two high outliers reached in September were excluded.

There were no statistically significant differences between the data obtained for both sections of the pond regardless of the test used for analysis, as reported for nitrate (p = .9455), nitrite (p = .9347), ammonium (p = .7567), phosphate (p = .9215) and carbonate hardness (p = .0619).

#### 3.2. Microalgal enumeration

The algal numbers from March until November of 2018 in the perch farm are displayed in Fig. 5 expressed as algal cells per ml, which was measured using FCM. As stated in the gating methodology for algal enumeration, the desired population included the chlorophyll population excluding cells positive for PC, therefore the graphs below do not include the majority of the Cyanobacteria numbers. The algal population in terms of enumeration remained steady for March and April and



Fig. 4. Trends for nitrogen, phosphate and carbonate hardness for the FP at the perch farm, 2018.

then increased in May when the light levels increased and temperatures increased to highs of 19 °C and 21 °C in May and July. Algal numbers peaked in late June, with counts of  $1.54 \times 10^5$  cells/ml and  $1.57 \times 10^5$  cells/ml observed for the FP and TP, respectively (Fig. 5). Lowest numbers were detected in the winter months *via* flow cytometry analysis of less than 100 cells/ml. There was a slight decrease in the algal numbers in August in the TP compared with the FP, a trend that was reversed in September, when a higher algal count was observed for the TP (Fig. 5). A *p*-value of 0.821, following a Mann-Whitney test indicated that there was no significant difference between the values obtained for both the FP and TP in terms of enumeration.

#### 3.3. Cyanobacterial population enumeration

The total number of chlorophyll-containing cells was determined *via* flow cytometry. The PC<sup>+</sup> population was assumed to closely represent the cyanobacterial population in this study. This data was used to establish the cyanobacterial population from the total number of chlorophyll positive cells. Figs. 6 and 7 illustrate the total chlorophyll positive population and the number of cells in this population that represent algal cells compared to cyanobacterial cells for the FP and TP, respectively. On average, Cyanobacteria accounted for 80% of the chlorophyll-containing cells present in the rearing water, whereas the algal cells only accounted for 20% of the chlorophyll-containing cells.

#### 3.4. Microalgal community profiling

Through use of the inverted light microscope, the microalgal profile and species diversity for each month was determined. The dominant species for each month are presented in Table 3 for the FP and Table 4 for the TP. The most numerous species are listed with the dominating phylum for each month shaded in pink. Over 40 different algal species were observed, with the majority identified to species level using photographic identification keys. Chlorophyta was the most dominant phylum for the FP and the TP overall, with the most commonly occuring species including Chlamydomonas sp., Chlorella sp., Dictyosphaerium sp., Monoraphidium sp., Pandorina sp., Scenedesmus sp., Selenastrum sp., Tetraspora sp. and Westella sp. In the FP the most dominant phylum for June and September was Bacillariophyta (diatoms). In June this was mainly Stephanodiscus sp., and Cyclotella sp. In September when the algal number decreased in the FP compared to the TP, the dominant species in the FP were Aulodiscus sp., Hyalodiscus sp. and Cyclotella sp., whereas, the TP was completely dominated by Cryptomonas sp., of the Cryptophyta phylum.

#### 3.5. Bacterial enumeration

Bacterial counts were conducted in order to provide supplementary data to algae enumeration and profiling for correlation purposes. The overall average count of total bacteria for the pond was  $6.33 \times 10^6$ 



Fig. 5. Algae cells/ml in from March to November 2018 in Pond 1 at the perch farm.



Fig. 6. Enumeration of total chlorophyll-containing cells consisting of cyanobacterial and algal cells in the FP at the perch farm, 2018.

cell/ml. The trend for this analysis from March to November 2018 is displayed in Fig. 8 expressed as  $log_{10}$  cells/ml.

#### 3.6. Principle component analysis

In order to analyse the large volume of data, PCA analysis was conducted in Excel using the XLSTAT software to observe any correlations and or variability between parameters, which can be difficult to ascertain from individual and even overlaid parameters. This type of analysis provides tables and graphs through which observations on the relationships between parameters were made. The parameters analysed included a combination of variables measured in the laboratory and measured on site. The lab parameters analysed were nitrate concentration, nitrite concentration, ammonium concentration, phosphate concentration, water hardness (mg  $CaCO_3/1$ ), bacterial numbers, algal

numbers, chlorophyll containing cells and cyanobacterial cells, with the latter four parameters measured in cells/ml. The parameters analysed that were recorded on site included pH, oxygen (mg/l), turbidity (FTU), feeding rate and total chlorophyll-*a* and Cyanobacteria measured using an AlgaeTorch<sup>®</sup> that is in good agreement with FCM determinations, with results in  $\mu$ g/l. Tables 5 and 6 display the correlation matrix scores obtained for the parameters analysed in this study.

#### 3.7. Weather conditions

The Republic of Ireland experienced one of its hottest summers on record in 2018 that coincided with the sampling period in this study (Met Éireann, 2019). Drought conditions and national hosepipe bans were put in place for most of the country up to the end of August 2018 (O'Neill et al., 2019). As a result of these unusual weather conditions,



Fig. 7. Enumeration of total chlorophyll-containing cells consisting of cyanobacterial and algal cells in the TP at the perch farm, 2018.

#### Table 3

Dominant algal species from March to November 2018 in the FP of Pond 1 at the perch farm.

		* Shading indicates the	
	Most to least dominant in each group	dominant group	
Month	Species		
March	Chlorella, Monoraphidium - Chlorophyta		
April	Chlamydomonas, Chlorella, Scenedesmus, Monoraphidium - Chlorophyta	Chroomonas, Cryptomonas - Cryptophyta	Trachelomas - Euglenophyta
May	Dictyosphaerium, Pandorina, Chlamydomonas, Tetraspora, Westella, - Chlorophyta		
June	Dictyospaherium, Westella, Chlamydomonas, Chlorella, Tetraspora - Chlorophyta	Stephanodiscus, Cyclotella - Bacillariophyta	
July	Dictyosphaerium, Westella, Chlamydomonas, Chlorella, Sphaerocystis - Chlorophyta		
August	Westella, Dictyosphaerium, Chlamydomonas, Scenedesmus - Chlorophyta	Cryptomonas - Cryptophyta	
September	Dictyosphaerium, Selenastrum - Chlorophyta	Aulodiscus, Hyalodiscus, Cyclotella - Bacillariophyta	
October	Selenastrum, Dictyosphaerium, Chlorella - Chlorophyta		
November	Chlorella, Dictyosphaerium, Chlamydomonas - Chlorophyta	Cryptomonas - Cryptophyta	Chroococcus - Cyanophyta

#### Table 4

Dominant algal species from March to November 2018 in the TP of Pond 1 of the perch farm.

		* Shading indicates the	
	Most to least dominant in each group	dominant group	
Month	Species		
March	Chlorella, Monoraphidium - Chlorophyta		
April	Chlorella, Monoraphidium, Dictyosphaerium, Scenedesmus, Chlamydomonas - Chlorophyta	Trachelomas - Euglenophyta	Aphanocapsa - Cyanophyta
May	Chlamydomonas, Chlorella, Pandorina, Westella, Tetraspora - Chlorophyta	Stephanodiscus, Cyclotella - Bacillariophyta	Cryptomonas - Cryptophyta
June	Dictyosphaerium, Westella, Chlorella, Chlamydomonas, Tetraspora - Chlorophyta	Cyclotella - Bacillariophyta	
July	Dictyosphaerium, Westella, Chlamydomonas - Chlorophyta	Snowella - Cyanophyta	
August	Westella, Chlamydomonas, Tetraspora - Chlorophyta		
September	Chlorella - Chlorophyta	Cryptomonas - Cryptophyta	
October	Dictyosphaerium, Selenastrum, Chlorella - Chlorophyta	Aphanocapsa - Cyanophyta	
November	Dictyosphaerium, Westella, Chlamydomonas - Chlorophyta		

mean rainfall and temperature data collected at three Met Eireann weather stations surrounding and closest to the fish farm in the Republic of Ireland, were observed. The stations were located in Knock, Markree and Mount Dillion. Decreases in the average monthly rainfall (Fig. 9(A)) and increases in temperature (Fig. 9(B)) were observed across the weather stations.

#### 4. Discussion

#### 4.1. Physicochemical parameter analysis

The highest nitrate concentrations of 3.27 mg/l and 3.25 mg/l observed for the FP and TP, respectively are below the 50 mg/l maximum acceptable limit of SI No. 77 of 2019. When the levels were lowest at the end of May there was a water change in the pond, therefore the fish were removed. This process may have reduced the level of nitrates, with one of the main sources being fish waste, as well as decaying



Fig. 8. Bacterial enumeration (log10 cells/ml) from March to November 2018 in Pond 1 at the perch farm.

organic matter in the water (Jiménez-Montealegre et al., 2002; Thajuddin and Subramanian, 2005), which would have also been partially removed in the process. Nitrite levels were below the limit of detection in March and highest in June with concentrations of 0.118 mg/l and 0.103 mg/l for the FP and TP, respectively. The temperature was highest at this point ranging between 19 °C and 21 °C and the bacterial counts were also highest in June overall. As the initial steps in this process involves the oxidation of ammonia to nitrite (Hargreaves, 1998; Helfrich and Libey, 1990), Fig. 4 illustrates the decrease of ammonium levels at the start of June, followed by an increase in the nitrite levels, respectively, which may in part account for this increase in nitrite concentration. The partial change of water in the pond at the end of May could have contributed to the rise in ammonium. A sudden decrease in bacterial and algal numbers evident from Figs. 5 and 8 may have impacted the nutrient recycling process. The ammonium ion tends to be harmless to fish unless extremely high concentrations are reached (Boyd and Lichtkoppler, 1979). The highest phosphate concentrations were detected in May for both ponds. Lund (1965) stated that phosphorus levels can decrease in the summer, and as the summer months progressed, the phosphate concentrations measured during this study did in fact decrease. This may in part be due to the high levels of bacteria in the summer which are major competitors of algae for the

#### Table 5

Correlation coefficient scores for PCA analysis carried out on TP parameters at the perch farm using XLSTAT (Chl = chlorophyll; Ctn. = containing; Cyano = cyanobacterial).

Variables	fëtrate	Jänne	Ammonium	Phosphate	Temperature	pH	Oxygen	Turbidity	Hardness	Feeding Rate	Bacteria	Algae	Chi-Stn Ceils	Cyano Cells	Chi (Torch)	Cyano (Torch)
Nitrate	1												0.	No Relat	ionship	
Nitrite	0.200	1											>0	0.2 = Very	Weak	
Ammonium	0.118	0.342	1										0.4	1-0.6 = N	Aderately	Strong
Phosphate	0.261	0.550	0.426	1									0.1	8 1.0 = V 0 = Perfec	ery Strong t Linear Rel	ationship
Temperature	-0.173	0.249	0.026	0.268	1											
pH	-0,433	-0.203	-0.331	-0.251	1516	1										
Oxygen	-0.483	-0.234	-0.120	-0.292	-0.338	0.195	1									
Turbidity	-0.167	0.258	0.421	0.493	-0.001	-0.267	0.161	1								
Hardness	0.240	0.512	0.099	0.171	0.108	0.005	-0.220	-0.145	1	ľ.						
Feeding Rate	-0.093	-0.091	-0.076	0.068	0.420	0.367	-0.292	-0.060	0.030	1						
Bacteria	-0.030	0.216	-0.080	0.257	0.402	0.304	-0.002	-0.121	0.100	0.235	1					
Algae	-0.369	0.230	0.043	0.190	0.329	0.264	0.336	0.319	-0.160	0.108	0.555	1				
Ol-Cin Cells	-0.463	0.366	0.065	0.284	0.069	-0.043	0.327	0.484	-0.151	-0.115	0.004	0.441	1			
Cyano Cells	-0,435	0.350	0.063	0.270	0.023	-0.084	0.295	0.463	-0.137	-0.137	-0.082	0.314	0.990	1		
Chil (Torch)	-0.539	0.211	-0.073	0.286	273	0.490	0.057	0.357	-0.131	0.308	0.339	0.511	0.555	0.508	1	
Cyano(Torch)	-0.568	0.175	-0.206	0.060	0.445	0.580	0.377	0.100	-0.103	0.151	0.489	180	0.527	0.466		1

#### Table 6

Correlation coefficient scores for PCA analysis carried out on FP parameters at the perch farm using XLSTAT (Chl = chlorophyll; Ctn. = containing; Cyano = cyanobacterial).

Variables	Nitrate	Ninite	Ammonium	Phosphate	Temperature	PH	Oxygen	Turbidity	Hardness	Feeding Rate	Bacteria	Algae	Chi Ctn. Cells	Cvano. Cella	(Tarch)	(Forch)
Nitrate	1													= No Rela	tionship	
Mitrine	0.282	1											3	0.2 = Ver	y Weak Weak	
Ammonium	0.150	0.295	1										(	.4 - 0.6 =	Moderately	Strong
Phosphate	0.398	1624	0.438	1									1	1.0 = Perfe	very Strong ct Linear Rei	ationship
Temperature	-0.261	0.234	0.029	0.276	1											
pH	-0.500	-0.215	-0.338	-0.263	2515	1										
Owygen	-0.494	-0.211	-0.150	-0.354	-0.338	0.195	1									
Turbidity	-0.155	0.264	0.391	0.399	-0.001	-0.267	0.161	1								
Hardness	0.245	0.424	0.325	0.209	-0.065	-0.138	-0.079	0.087	1							
Feeding Rate	-0.061	-0.105	-0.081	0.048	0.420	0.367	-0.292	-0.060	-0.052	1						
Bactoria	-0.110	0.173	-0.068	0.233	0.384	0.282	0.019	-0.100	-0.099	0.276	1					
Algae	-0.357	0.281	0.252	0.229	0.370	0.187	0.256	0.377	-0.030	0.140	0.539	1				
Chi Ctr. Cells	-0.481	0.321	0.119	0.101	0.131	-0.029	0.270	0.496	0.014	0.019	0.067	0.607	1			
Cvano Cells	-0.459	0.303	0.085	0.069	0.069	-0.075	0.247	0.477	0.023	-0.010	-0.032	0,475	0 988	1		
Chi (Tarch)	-0.599	0.199	-0.084	0.174	0.731	0.490	0.057	0.357	-0.169	0.308	0.352	0.502	0.567	0.526	1	
Cyang (Torth)	-0.573	0.179	-0.257	0.024	0.445	0.580	0.377	0.100	-0.159	0.151	0.480	0.548	0.481	0.421		1





Fig. 9. (A) Mean total rainfall (mm) for three met offices nearby to sampled fish farm, and (B) mean temperature in degrees Celsius recorded at same three met offices, over periods 2017 to 2019.

uptake and utilization of inorganic phosphorus (Lund, 1965).

#### 4.2. Algal enumeration

The algal population remained steady across the sampling months but peaked when the temperatures were at the highest (19 °C and 21 °C) between May and July. The decrease in the algal numbers in the TP compared to the FP in August may have been due to the removal of weeds and duckweed from the outer area of the TP which may have led to a dilution of the algal population. It is a common trend that algal cells reach highest concentrations during the summer months due to increasing irradiance, longer daylight hours and higher temperatures. Li et al. (2012) found that algae numbers were higher in summer than winter as is the general trend of algal growth in lakes or ponds. However, the decrease from summer to winter was only marginal, from  $3.45 \times 10^3$  to  $1.46 \times 10^3$  cells/1, whereas in this project cell numbers declined to much lower levels in winter, decreasing to 9 cells/ml in the first week of November.

PCA analysis indicated a strong correlation between temperature and the chlorophyll content measured using the AlgaeTorch<sup>®</sup>, with a coefficient score of 0.791. This would be expected as temperature and daylight are two of the most important factors in terms of phytoplankton growth, so an increase in temperature will lead to an increase in phytoplankton. This correlation was not, however, present for the chlorophyll analysis carried out in the lab, even though this parameter correlated to the on-site chlorophyll measurements. This result may imply that although certain parameters have been proven to share relationships, this may not always be reflected in the data as other variables may interfere or an increased number of data points may be required.

There was a moderately strong positive correlation between algae and bacterial enumeration data in both the FP and TP, with coefficient scores of 0.555 and 0.539, respectively. This correlation is commonly observed due to the symbiotic relationship shared between these two biological entities. Algae require nitrogen as an essential element for building structural and functional proteins (Hu, 2004). It is available in the soil organic matter (SOM); however, nitrogen is not in a bioavailable form for algae to utilise. Nitrogen-fixing bacteria convert the nitrogen (nitrogen fixation) into a form that can then be utilised by algae (Neospark, 2014; Thajuddin and Subramanian, 2005). The rate of nitrogen fixation largely depends on the bacterial species present in the water and the concentration of ammonia (Hargreaves, 1998). This nitrogen fixation process highlights the important interdependent relationship that exists between algae and bacteria. Another aspect of this dynamic relationship involves organic matter, on which bacteria thrive (Amon and Benner, 1996; Baines and Pace, 1991; Blancheton et al., 2013). One of the principal sources of organic matter in the rearing water is primary production by microalgae, followed by excreta and feed pellets (Baines and Pace, 1991; Moriarty, 1997). Aerobic bacteria present in the water body break down this organic matter into CO<sub>2</sub> and ammonia (Phang, 1991). Algae then utilise the CO<sub>2</sub> for photosynthesis and release oxygen during the process, which in turn oxygenates the water for the fish (Neospark, 2014). Algae also uptake the ammonia as well as heavy metals, reducing the availability of toxic substances for fish to consume (Neori et al., 2004).

Phosphorus was also a determining factor for plankton richness in the study carried out by Li et al. (2012) on an artificial lake, whereas transparency negatively correlated with plankton communities. This negative correlation was related to the negative correlation achieved between transparency and the presence of algae, indicating that algae have a major impact on the turbidity of water (Li et al., 2012). This finding agrees with the moderately strong positive correlation achieved between turbidity and chlorophyll-containing cells in both the FP and TP in this study.

Microalgae are capable of assimilating nutrients in eutrophic water bodies (Leng et al., 2018) along with wastewater remediation from

many sources including food industry, agriculture and municipal effluents (Han et al., 2019). Chlorella sp. and Scenedesmus sp. have been previously reported to positively contribute to the natural treatment of wastewater due to their efficiency at nutrient, antibiotic and heavy metal removal (Chen et al., 2012; Choi and Lee, 2012; Delgadillo-Mirquez et al., 2016; Delrue et al., 2016; de Godos et al., 2012; Min et al., 2011; Nurdogan and Oswald, 1995). In addition to treating wastewater, microalgae can also synthesize value-added components such as proteins, lipids and natural pigments (Han et al., 2019). Previous researchers have reported on the value-added biomass derived from microalgae activities that could contribute to aquaculture feed along with augmenting immunity of farmed fish (Sirakov et al., 2015; Ansari et al., 2017). Microalgae-assisted aquaculture generates oxygen as a natural aeration process that can also influence and adjust microbial communities, which, if effectively controlled, could be applied to negate oxygen depletion and unwanted algal blooms (Han et al., 2019; Lu et al., 2019b).

A biotic balance should be achieved to ensure that the algal and bacterial numbers and populations are having this positive influence on productivity, rather than negatively influencing rearing water quality and consequently the production and efficiency of an aquaculture farm. Findings from this present study provides knowledge regarding microbial interactions, and ecology of these systems, that prevents the utilization of microbial communities in the assessment, improvement and control of aquaculture farms. There is increasing evidence to suggest that a co-dependent relationship exists between phytoplankton and nutrients in rearing water, as phytoplankton abundance depends on nutrient availability and nutrient cycling depends majorly on the presence of phytoplankton (Li et al., 2012; Bentzon-Tilia et al., 2016). This highlights how important the microalga balance is in rearing water, as not only do the algae produce oxygen during daylight (Moriarty, 1997), but also recycle metabolites that would otherwise build up in the water. Moriarty (1997) also highlights the importance of nitrogen and phosphorus on microalgae productivity.

#### 4.3. Cyanobacterial population enumeration

The Cyanobacteria population accounted for the majority of the chlorophyll-containing cells detected/recorded throughout the sampling period in both ponds. This observation is highlighted further with the use of the moving average trend line in Figs. 6 and 7. While certain species of Cyanobacteria, for example *Microcystis* sp., can release toxins into the water and exhert detrimental effects on other organisms, Cyanobacteria are also beneficial for processes such as nitrification. Throughout this process, the nutrients are taken up by the cells and are therefore removed from the water, increasing the water quality in terms of nutrient pollution. In a study carried out by Liu et al. (2018) for the treatment of aquaculture using Chlorophyta, prior to inoculation with the green microalgae, Cyanobacteria were responsible for the partial removal of the pollutants from the aquaculture water, highlighting the importance of Cyanobacteria in the rearing water.

Following the PCA analysis, the highest correlation for the FP and the TP was the positive one observed between chlorophyll-containing cells and the cyanobacterial cells with coefficient scores of 0.990 and 0.988 for the FP and TP, respectively. This clearly corresponds to the data displayed in Figs. 5 and 6, reiterating the observation that the Cyanobacteria phylum represented most of the chlorophyll source present in pond 1 at the perch farm compared to algae. There was a strong positive correlation between the same two parameters when measured on site using an algal torch with a score of 0.652, which indicates that both on site and in lab measurements of phytoplankton produced similar trends. There was a moderately strong positive correlation between algal counts and chlorophyll-containing cells for the FP with a score of 0.441. Whereas a strong positive correlation was established for the same parameters in the TP, with a score of 0.607. This would suggest that the trends for algal numbers in the TP were more in line with the overall trends for chlorophyll-containing cells compared to the algal numbers in the FP. This may be due to the presence of fish in the FP, which would have impacted the algal numbers to a higher extent, by uptake into the diet, for example.

In the case of nitrates, the data for nitrate concentration in both the FP and the TP was negatively correlated with chlorophyll and cyanobacterial parameters measured. This would indicate that the presence of chlorophyll-containing cells/pigment, the majority of which corresponded to Cyanobacteria, had a negative impact on nitrate levels. Phytoplankton are known for the uptake and removal of certain nutrients from the water and the coefficient scores reflect this fact. This finding is comparable to results determined in a study carried out by Choi and Lee (2010), where the growth of Cyanobacteria and algae inhibited the maximum nitrification rate by a factor of 4 in an autotrophic bioreactor. Hu et al. (2000) also established similar results in an assessment of the removal of nitrate from groundwater by Cyanobacteria, with *Synechococcus* sp. displaying the highest rate of nitrate removal.

#### 4.4. Algal community profiling

Chlorella and Monoraphidium were the most common algae present in both the FP and TP in March and both species remained quite dominant in April. Chlorella has been reported as one of the most effective algal species at nutrient uptake, particularly nitrogen and phosphorus (Wang et al., 2010). In this study, the nitrate, nitrite and ammonium concentrations all decreased in early/mid-June when the Chlorella population remained prevalent. The effectiveness of Chlorella sp., in particular Chlorella vulgaris, at nutrient removal is also evident from its common use in the treatment of wastewater (Abdel-Raouf et al., 2012; Choi and Lee, 2012; Delrue et al., 2016; de Godos et al., 2012; Min et al., 2011). Chlorella kessleri, synonymous with Parachlorellakessleri, has also shown great potential for pollutant removal from aquaculture wastewater. Liu et al. (2018) inoculated aquaculture wastewater with five Chlorophyta species with P. kessleri exhibiting the greatest rate of nutrient uptake in terms of COD, nitrogen and total phosphorus. Monoraphidium sp. have also been reported for successful nutrient uptake. In a biodiesel production study carried out by Holbrook et al. (2014) Monoraphidium reduced concentrations of nitrates and phosphates to < 5 mg/l and < 1 mg/l, respectively. Therefore, Monoraphidium sp. are potentially useful organisms for phytoremediation of aquaculture water if the cell densities are increased. Sanchis-Perucho et al. (2018) discovered that the nutrient removal efficiency of a consortium of Monoraphidium and Scenedesmus sp. was more effective than the removal of nitrogen and phosphorus compared to Chlorella.

There was an increase in algal diversity for the month of April, with Chlorophyta dominating in both the FP and TP. The most numerous Chlorophyta observed, other than Chlorella and Monoraphidium, included Pandorina sp., Chlamydomonas sp., Dictyosphaerium sp., Kirchneriella sp. and Scenedesmus sp., with S. obtusus, S. quadricauda, S. obliquus, S. opaliensis and S. acuminatus all identified. The presence of different clonal populations for some algae species, such as a four- and eight-colony formation of Scenedesmus sp., compared to single-celled organisms may be attributed to the selective pressures in aquatic environments. For example, a study carried out by Zhu et al. (2015) demonstrated that upon exposure to Daphnia filtrate, acting as a predator, the Scenedesmus sp. increased the rate of the formation of four- and eight-celled populations. The presence of both the four- and eight-celled Scenedesmus sp. at the collaborating farm in this study may be indicative of the selective pressure that was present in the rearing water for the duration of the study, due to the abundant diversity that was evident from all the samples analysed. There may also have been selective pressures due to the dramatic changes in meteorological and environmental conditions, ranging from snow in March to drought in the summer months. Crytophytes and Euglenophyta were also observed

in April in the form of *Chroomonas* sp. and *Cryptomonas* sp., and *Tra-chelomonas* sp., respectively. *Pediastrum* sp. was observed in the sample for May, which was not present prior to then. This species remained present until October, after which it was not observed.

Diatoms, mainly *Cyclotella* and *Stephanodiscus* sp., were the most frequent species observed in the month of May for the TP and June for the FP and the TP. According to Stoermer and Julius (2003) diatoms tend to be specific to certain habitats which allows for their use as indicators of water quality, with *Stephanodiscus* considered to be one of the most common and ubiquitous freshwater diatoms. In July, *Merismopedia* sp. was present which had not been present prior to then. *Synura* sp. were quite dominant in July for both the FP and TP. There was an increase in the presence of *Scenedesmus* sp. in August compared to any other month.

The species diversity for both the FP and the TP was very similar every month, however with one major exception. In the month of September, the TP was completely dominated by *Cryptomonas* sp. Contrastingly, very large diatoms, which resembled *Aulacodiscus*, *Hyalodiscus* and *Cyclotella* sp. dominated the FP. The presence of these diatoms seemed to cause a decrease in other species present, possibly due to feeding or out-competing with other species for nutrients. In fact, the algal counts for mid-September were lower for the FP compared to the TP.

According to Vuorio et al. (2007), when analysing multispecies communities of phytoplankton, enumeration procedures can be complicated and more information regarding water quality can be determined by phytoplankton community analysis compared to basic nutrient or chlorophyll a measurements. Therefore, it was important to perform a two-step algal analysis procedure in the form of flow cytometry and microscopy. Other factors can also be problematic for the algal identification process. Stoermer and Julius (2003) state that the average size of diatomic cells decreases after each vegetative life cycle, which can lead to variability in cell morphology of the same species. Environmental conditions such as salt levels can also alter diatom morphology (Stoermer and Julius, 2003). As well as that Small et al. (2016) stated that in terms of the capacity of photoautotrophic systems, such as algae, to remove nutrient waste from the water depends largely on energy uptake from sunlight, which is very unpredictable in a climate such as the one in Ireland. With variations in climate change and increasing temperatures worldwide, certain microalgae species may not be able to grow, and as they are a source of oxygen in the ponds, the use of a natural means of producing oxygen for biological processes may no longer be an option.

Resident bacteria and other microbes, which can limit or influence microalga growth due to availability of nutrients, contaminated microalgae populations in the pill pond. Compared with previously described closed photo-bioreactors (Han et al., 2019), a pill pond system has a much lower investment and operational cost but higher volume, making it more suitable for treatment of aquaculture wastewater. Han et al. (2019) indicated that two critical factors should be considered for advancing this type of aquaculture process. Firstly, location and pond depth need to be properly evaluated so as to improve light transmittance and photosynthesis rate. Secondly, the relationship between microalgae and bacteria should be thoroughly investigated to be fully elucidated. Van den Hende et al. (2014) reported that microalgaebacterial flocs contributed to the removal of 28% COD, 53% BOD<sub>5</sub>, 31% Total Nitrogen, and 64% Total Phosphorus in aquaculture wastewater (12 m<sup>3</sup> raceway pond), suggesting that the threat of bacteria to microalgae is putatively low when a beneficial cooperation is established.

#### 5. Conclusion

The results of this study conducted at an aquaculture perch farm in the Republic of Ireland provide a baseline for the rearing water microalgae and physicochemical ecosystem interactions. In pond aquaculture that often relies upon natural wastewater treatment processes,

there is much less control over environmental conditions compared to closed tanks, which is why the ecosystem dynamic needs to be understood and possibly manipulated for successful and sustainable fish production. Identification of the most influential biological species in more depth would provide the opportunity of transplantation of specific microbial assemblages when required for certain processes, i.e. the addition of a specific bacteria for nitrification or the fertilisation of a specific algal species for nutrient removal or oxygen supplementation during the daytime. However, in order for this to be possible and beneficial, the function of each species needs to be determined. In addition, more sustainable and effective disease control measures need to be implemented for successful management and eradication of unwanted pathogens and possibly for control of the algal population, once identified. Without the baseline information, the required knowledge to inform prevention measures rather than undergoing treatment processes would be unfeasible. This constitutes the first study that reported good agreement between use of real-time laboratory-based techniques and in-field monitoring technologies for enumerating microalgae and bacterial communities, where it is envisaged that use of these combinational approaches will aid the future development of aquaculture processes. Limitations associated with findings relate to the fact that flow cytometry is highly specialised and not broadly available to support the aquaculture industry, but could be provided as a specialist contract service. Having an in-depth knowledge of this characterization would also be the basis for future diagnostic applications such as the design and development of diagnostic molecular kits.

#### Credit author statement

CRediT roles:

Sarah Naughton: Data curation, formal analysis, investigation, methodology, software, validation, writing original paper and revision. Siobhan Kavanagh: funding acquisition, investigation, methodology, supervision, writing original manuscript and editing revision.

Mark Lynch: investigation, methodology, supervision, writing original paper and making edits to revision.

Neil J Rowan: Conceptualization, funding acquisition, investigation, methodology, project administration, resources, supervision, visualization, writing orginal draft and revision with editing.

#### **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgements

The authors wish to thank the collaborating aquaculture farm, Bord Iascaigh Mhara (BIM) and European Monetary Fund (EMF) for cofunding this research under the Knowledge Gateway Scheme (17KGS004).

#### References

- Abdel-Raouf, N., Al-Homaidan, A.A., Ibraheem, I.B.M., 2012. Microalgae and wastewater treatment. Saudi J. Biol. Sci. 19, 257–275. https://doi.org/10.1016/j.sjbs.2012.04. 005.
- Amon, R.M.W., Benner, R., 1996. Bacterial utilization of different size classes of dissolved organic matter. Limnol. Oceanogr. 41, 41–51.
- Ansari, F.A., Singh, P., Guldhe, A., Bux.F., 2017. Microalgal cultivation using aquaculture wastewater: integrated biomass generation and nutrient remediation. Algal Res. 21, 169–177.
- Baines, S.B., Pace, M.L., 1991. The production of dissolved organic matter by phytoplankton and its importance to bacteria: patterns across marine and freshwater systems. Limnol. Oceanogr. 36, 1078–1090.
- Bentzon-Tilia, M., Sonnenschien, E.C., Gram, L., 2016. Monitoring and managing microbes in aquaculture - Towards a sustainable industry. Microb. Biotechnol. 9 (5),

Aquaculture 526 (2020) 735377

567-584

- Blancheton, J.P., Attramadal, K.J.K., Michaud, L., d'Orbcastel, E.R., Vadstein, O., 2013. Insight into bacterial population in aquaculture systems and its implication. Aquac. Eng. 53, 30–39. https://doi.org/10.1016/j.aquaeng.2012.11.009.
- Bloem, J., Vos, A., 2004. Detection, identification and classification of microbes using other methods. In: Kowalchuk, G.A., de Bruijn, F., Head, I.M., Van der Zijpp, A.J., van Elsas, J.D. (Eds.), Molecular Microbial Ecology Manual, pp. 861–873.
- Boyd, C.E., 1998. 'Water quality for pond aquaculture', Research and development series. International Centre for Aquaculture and Aquatic Environments, Alabama Agricultural Experiment Station, Auburn University, Alabama No. 43.
- Boyd, C.E., Lichtkoppler, F., 1979. Water Quality Management in Pond Fish Culture. Research and Development Series.
- Chen, C., Chang, H., Kao, P., Pan, J., Chang, J., 2012. Biosorption of cadmium by CO2fixing microalga Scenedesmus obliquus CNW-N. Bioresour. Technol. 105, 74–80. https://doi.org/10.1016/j.biortech.2011.11.124.
- Choi, H., Lee, S., 2012. Effects of microalgae on the removal of nutrients from wastewater: various concentrations of Chlorella vulgaris. Environ. Eng. Res. 17, 3–8.
- DAFM, 2015. National Strategic Plan for Sustainable Aquaculture Development.
- de Godos, I., Muñoz, Raúl, Guieysse, B., 2012. Tetracycline removal during wastewater treatment in high-rate algal ponds. J. Hazard. Mater. 229–230, 446–449. https://doi. org/10.1016/j.jhazmat.2012.05.106.
- Delgadillo-Mirquez, L., Lopes, F., Taidi, B., Pareau, D., 2016. Nitrogen and phosphate removal from wastewater with a mixed microalgae and bacteria culture. Biotechnol. Rep. 11, 18–26. https://doi.org/10.1016/j.btre.2016.04.003.
- Delrue, F., Álvarez-Díaz, P.D., Fon-Sing, S., Fleury, G., Sassi, J.-F., 2016. The environmental biorefinery : using microalgae to remediate wastewater, a win-win paradigm. Energies 9, 1–19. https://doi.org/10.3390/en9030132.
- Drikas, M., Chow, C.W.K., House, J., Burch, M.D., 2001. Using coagulation, flocculation, and settling to remove toxic cyanobacteria. Am. Water Works Assoc. 100–111.
- Endo, H., Nakayama, J., Hayashi, T., 2010. Application of flow cytometry to environmental control in marine aquaculture. Mater. Sci. Eng. C 12 (1–2), 83–88.
- EFSA, 2008. Scientific Opinion of the Panel on Animal Health and Animal Welfare on a request from the European Commission on the Animal welfare aspects of husbandry systems for farmed Trout. The EFSA Journal 796 (September), 1–22.
- FAO, 2018. The State of World Fisheries and Aquaculture. Food and Agriculture Organization of the United Nations.
- Guillard, R.R.L., 1978. Counting slides. In: Sournia, A. (Ed.), Phytoplankton Manual. United Nations Educational, Scientific and Cultural Organization, Paris, pp. 182–188. https://doi.org/10.2216/i0031-8884-19-4-341.1.
- Han, P., Fan, L., Zhou, W., 2019. A review on the use of microalgae for sustainable aquaculture. Appl. Sci. 2019 (9), 2377. https://doi.org/10.3390/app9112377.
- Hargreaves, J.a., 1998. Nitrogen biogeochemistry of aquaculture ponds. Aquaculture 166, 181–212. https://doi.org/10.1016/S0044-8486(98)00298-1.
- Haynes, M., Seegers, B., Saluk, A., 2015. Advanced analysis of aquatic plankton using flow cytometry. ACEA Biosci. https://doi.org/10.1007/978-3-319-48671-0\_31.
- Helfrich, L.a., Libey, G., 1990. Fish Farming in Recirculating Aquaculture Systems (Ras). Dep. Fish. Wildl. Sci.
- Holbrook, G.P., Davidson, Z., Tatara, R.A., Ziemer, N.L., Rosentrater, K.A., Grayburn, W.S., 2014. Use of the microalga Monoraphidium sp. grown in wastewater as a feedstock for biodiesel: cultivation and fuel characteristics. Appl. Energy 131, 386–393. https://doi.org/10.1016/j.apenergy.2014.06.043.
- Hu, Q., 2004. Environmental effects on cell composition. In: Richmond, A. (Ed.), Handbook of Microalgal Culture. Blackwell Publishing Ltd., pp. 83–94. https://doi. org/10.1002/9780470995280.ch5.
- Hu, Q., Westerhoff, P., Vermaas, W., 2002. Removal of nitrate from groundwater by cyanobacteria: Quantitative assessment of factors influencing nitrate uptake. Appl. Environ. Microbiol. 66 (1), 133–139.
- Jiménez-Montealegre, R., Verdegem, M.C.J., van Dam, A., Verreth, J.A.J., 2002. Conceptualization and validation of a dynamic model for the simulation of nitrogen transformations and fluxes in fish ponds. Ecol. Model. 147, 123–152. https://doi.org/ 10.1016/S0304-3800(01)00403-3.
- Kumaravel, T.S., Vilhar, B., Faux, S.P., Jha, A.N., 2009. Comet assay measurements: a perspective. Cell Biol. Toxicol. 25, 53–64. https://doi.org/10.1007/s10565-007-9043-9.
- Leng, L., Li, J., Wen, Z., Zhou, W., 2018. Use of microalgae to recycle nutrients in aqueous phase derived from hydrothermal liquefaction process. Biosresour. Technol. 256, 529–542.
- Li, X., Yu, Y., Zhang, T., Feng, W., Ao, H., Yan, Q., 2012. Seasonal variation of plankton communities influenced by environmental factors in an artificial lake. Chin. J. Oceanol. Limnol. 30 (3), 397–403.
- Liu, Y., Lv, J., Feng, J., Liu, Q., Nan, F., Xie, S., 2018. Treatment of real aquaculture wastewater from a fishery utilizing phytoremediation with microalgae. J. Chem. Technol. Biotechnol. 1–12. https://doi.org/10.1002/jctb.5837.
- Longo, S., d'Antoni, B.M., Bongards, M., Chaparro, A., Cronrath, A., Fatone, F., Lema, J.M., Maauricio-Iglesias, M., Soares, A., Hospido, A., 2016. Monitoring and diagnosis of energy consumption in wastewater treatment plants. A state of the art and proposal for improvement. Appl. Energy 179, 1251–1268.
- Lu, Q., Han, P., Xiao, Y., Liu, T., Chen, F., Leng, I., Liu, H., Zhou, J., 2019a. The novel approach of using microbial system for sustainable development of aquaponics. J. Chem. Prod. 217, 573–575.
- Lu, Q., Ji, C., Yan, Y., Ziao, Y., Li, J., Leng, L., Zhou, W., 2019b. Application of novel micoralge-film based air purifier to improve air quality through oxygen production and fine particulates removal. J. Chem. Technol. Biotechnol. 94, 1057–1063.
- Lund, J.W.G., 1965. The ecology of the freshwater phytoplankton. Biol. Rev. 40, 231–290. https://doi.org/10.1111/j.1469-185X.1965.tb00803.x.
- Met Éireann, 2019. Monthly Data [WWW Document]. Irish Meteorol. Serv URL. https://

www.met.ie/climate/available-data/monthly-data (accessed 20.10.2019).

- Min, M., Wang, L., Li, Y., Mohr, M.J., Hu, B., Zhou, W., Chen, P., Ruan, R., 2011. Cultivating Chlorella sp . in a pilot-scale photobioreactor using centrate wastewater for microalgae biomass production and wastewater nutrient removal. Appl. Biochem. Biotechnol. 165, 123–137. https://doi.org/10.1007/s12010-011-9238-7.
- Moldaenke, Bbe, 2017. AlgaeTorch the Handy Measurement Instrument for Rapid Deployment.
- Moorhouse, H.L., Read, D.S., McGowan, S., Wagner, M., Roberts, C., Armstrong, L.K., Nicholls, D.J.E., Wickham, H.D., Hutchins, M.G., Bowes, M.J., 2018. Characterisation of a major phytoplankton bloom in the River Thames (UK) using flow cytometry and high performance liquid chromatography. Sci. Total Environ. 624, 366–376. https:// doi.org/10.1016/j.scitotenv.2017.12.128.
- Moriarty, D.J.W., 1997. The role of microorganisms in aquaculture ponds. Aquaculture 151, 333–349.
- Muziasari, W.I., Parnanen, K., Johnson, T.A., Lyra, C., Karkman, A., Stedfeld, R.D., Tammien, M., Tiedje, J.M., Virta, M., 2016. Aquaculture changes the profile of antibiotic resistance and mobile genetic element associated genes in Baltic Sea sediments. FEMS Microbiol. Ecol. 92 fiw052.
- Neori, A., Chopin, T., Troell, M., Buschmann, A.H., Kraemer, G.P., Halling, C., Shpigel, M., Yarish, C., 2004. Integrated aquaculture: rationale, evolution and state of the art emphasizing seaweed biofiltration in modern mariculture. Aquaculture 231, 361–391. https://doi.org/10.1016/j.aquaculture.2003.11.015.
- Neospark, 2014. Importance of Plankton in Aquaculture and The Benefits of EcoPlankt-Aqua [WWW Document]. Neospark Aquac. Tech Files, Hyperbad (India) URL. http://neospark.com/aqua-techfiles.html (accessed 3.13.17).
- Nurdogan, Y., Oswald, W.J., 1995. Enhanced nutrient removal in high-rate ponds. Water Sci. Technol. 31, 33–43.
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga Pseudokirknierella subcapitata as an early-warning indicator to identify climate change amgibuity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218.
- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of pealtands as future locations for the sustainable intensification of freshwater aquaculture production - a case study from the Republic of Ireland. Sci. Total Environ. 706 (1 March). https:// doi.org/10.1016/j.scitotenv.2019.136044.
- Phang, S.-M., 1991. Role of Algae in Livestock-Fish Integrated Systems. pp. 1-11.
- Read, D.S., Bowes, M.J., Newbold, L.K., Whiteley, A.S., 2014. Weekly flow cytometric analysis of riverine phytoplankton to determine seasonal bloom dynamics. Environ Sci Process Impacts 16, 594–603. https://doi.org/10.1039/c3em00657c.
- Ruis-Salmon, I., Margallo, M., Laso, J., Villaneuva-Rey, P., Marino, D., Quinteiro, P., Dias, A.C., Nunes, M.L., Marques, A., Felijoo, G., Moreira, M.T., Loubet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N., Mendez-Paz, D., Iglesias-Parga X., Anglada, C., Martin, J.C., Irabien, A., Aldaco, R., 2020. Addressing challengis and opportunities fo the European seafood sector under a circular economy. Curr. Opin.

Environ. Sci. Health 13, 101-106.

- Sanchis-Perucho, P., Duran, F., Barat, R., Pachés, M., Aguado, D., 2018. Microalgae population dynamics growth with AnMBR effluent: effect of light and phosphorus on concentration. Water Sci. Technol. 77, 2566–2577. https://doi.org/10.2166/wst. 2018.207.
- Shibata, A., Goto, Y., Saito, H., Kikuchi, T., Toda, T., Taguchi, S., 2006. Comparison of SYBR Green I and SYBR Gold stains for enumerating bacteria and viruses by epifluorescence microscopy. Aquat. Microb. Ecol. 43, 223–231. https://doi.org/10. 3354/ame043223.
- Sirakov, I., Velichkova, K., Stoyanova, S., Staykov, Y., 2015. The importance of microalge for aquaculture industry. Review. Int. K. Fish Aquat. Stud. 2, 81–84.
- Small, B.C., Hardy, R.W., Tucker, C.S., 2016. Enhancing fish performance in aquaculture. Anim. Front. 6, 42–49. https://doi.org/10.2527/af.2016-0043.
- Stoermer, E.F., Julius, M.L., 2003. Centric diatoms. In: Wehr, J.D., Sheath, R.G. (Eds.), Freshwater Algae of North America: Ecology and Classification. Elsevier Inc., USA, pp. 559–594. https://doi.org/10.1016/B978-0-12-741550-5.50016-7.
- Szymański, N., Dąbrowski, P., Zabochnicka-Świątek, M., Panchal, B., Lohse, D., Kalaji, H.M., 2017. Taxonomic classification of algae by the use of chlorophyll a fluorescence. Sci. Rev. Eng. Environ. Sci. 26 (4), 470–480. https://doi.org/10.22630/PNIKS. 2017.26.4.45.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, 5281. https://doi.org/10.7717/peerj.5281.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2019. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishers 3, 28. https://doi.org/10.3390/fishes3030028.
- Thajuddin, N., Subramanian, G., 2005. Cyanobacterial biodiversity and potential applications in biotechnology. Curr. Sci. 89, 47–57.
- Van Den Hende, S., Claessens, L., Muylder, E., Boon, N., Vervaeren, H., 2014. Microalgal bacterial flocs originating from aquaculture wastewater treatment as diet ingredient for Litopenaeus vannamei (Boone). Aquacult. Res. 47. https://doi.org/10.1111/are. 12564.
- Vuorio, K., Lepistö, L., Holopainen, A.L., 2007. Intercalibrations of freshwater phytoplankton analyses. Boreal Environ. Res. 12, 561–569.
- Wang, L., Min, M., Li, Y., Chen, P., Chen, Y., Liu, Y., Wang, Y., Ruan, R., 2010. Cultivation of green algae Chlorella sp. in different wastewaters from municipal wastewater treatment plant. Appl. Biochem. Biotechnol. 162 (4), 1174–1186.
- Wurts, W.A., Durborow, R.M., 1992. Interactions of pH, carbon dioxide, alkalinity and hardness in fish ponds. South. Reg. Aquac. Cent. 1–4. https://doi.org/10.1007/ s004380051177.
- Zhu, X., Wang, J., Lu, Y., Chen, Q., Yang, Z., 2015. Grazer-induced morphological defense in Scenedesmus obliquus is affected by competition against Microcystis aeruginosa. Scientific Reports 5 (12743), 1–10.

# Peer

## Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland

Alexandre Tahar<sup>1</sup>, Alan M. Kennedy<sup>2</sup>, Richard D. Fitzgerald<sup>3,†</sup>, Eoghan Clifford<sup>2</sup> and Neil Rowan<sup>1</sup>

<sup>1</sup> Bioscience Research Institute, Athlone Institute of Technology, Athlone, Ireland

<sup>2</sup> Civil Engineering, National University of Ireland, Galway, Ireland

<sup>3</sup> Ryan Institute, National University of Ireland, Galway, Ireland

<sup>†</sup>Deceased.

## ABSTRACT

In the context of future aquaculture intensification, a longitudinal ten-year evaluation of the current traditional rainbow trout production in Ireland was performed. Publically available and independent data obtained from local authorities were gathered and analysed. Inlet and outlet concentrations of parameters such as BOD<sub>5</sub>, ammonium, nitrite, dissolved oxygen and pH for four consecutive flow-through fish farms covering the four seasons over a ten-year period (2005–2015) were analysed. The objectives of the study were (i) to characterize the impact of each fish farm on water quality in function of their respective production and identify any seasonal variability, (ii) to quantify the cumulative impact of the four farms on the river quality and to check if the self-purification capacity of the river was enough to allow the river to reach back its background levels for the analysed parameters, (iii) to build a baseline study for Ireland in order to extrapolate as a dataset for expected climate change and production intensification. For most of the parameter analysed, no significant impact of the fish farming activity on water quality/river quality was observed. These results, the first ones generated in Ireland so far, will have to be completed by a survey on biodiversity and ecotoxicology and compared after production intensification and the likely future introduction of water treatment systems on the different sites.

Subjects Aquaculture, Fisheries and Fish Science, Ecology, Aquatic and Marine Chemistry, Environmental Contamination and Remediation, Environmental Impacts Keywords Pond-based, Freshwater aquaculture, Environmental impact, Rainbow trout, Receiving water, Ireland

## **INTRODUCTION**

With the wild fish catching capacities almost reached and with the overall human population increase on the planet, fish from aquaculture is and will increasingly become a more important food source for human consumption in the near future (e.g., *Donnely, 2011*; *Guilpart et al., 2012*). Aquaculture production increased more than five-fold from 1990 to 2012, while the world capture fisheries increased with only 8% at the same time (*Krause et al., 2015*). Furthermore, it is planned that this increase in production will be sharper in the next decades in Ireland, notably with the objective of increasing food export (including

Submitted 8 February 2018 Accepted 14 June 2018 Published 24 July 2018

Corresponding author Alexandre Tahar, atahar@ait.ie

Academic editor Marcia Marques

Additional Information and Declarations can be found on page 19

DOI 10.7717/peerj.5281

Copyright 2018 Tahar et al.

Distributed under Creative Commons CC-BY 4.0

OPEN ACCESS

aquaculture fish products) by 85% by 2025 through the application of the Food Wise 2025, the strategic plan for the development of agri-food sector (*DAFM*, 2015). The future and current intensification of this activity is associated with meeting more stringent environmental regulations to ensure a sustainable and environmentally friendly production. At the European scale, freshwater fish farming is essentially regulated through the Water Framework Directive (WFD) that aims at providing the good chemical and ecological status of the rivers (*Aubin, Tocqueville & Kaushik, 2011; European Commission, 2000; Guilpart et al., 2012*). Currently, there is a dearth in scientific knowledge on the potential impact and environmental risk of aquaculture farming practices on receiving water quality as it relates to both WFD compliance and related EU Rivers Basin Management Plan 2016–2021.

Various anthropogenic activities can cause the deterioration of river quality including point source pollution such as wastewater treatment plants (WWTPs) with the discharge of partially treated effluents or non-point sources such as agricultural activities (e.g., *Papatryphon et al.*, 2005). The traditional freshwater fish farming industry generally operates using flow-through systems (FT) without any water treatment, with the oxygen levels being maintained by relatively high volume water abstraction & flow through the farm. Fish farming activity generates wastes from fish excreta and uneaten feed which if discharged untreated could potentially impact the water quality of the discharge receiving water (Caramel et al., 2014; Garcia et al., 2014; Lalonde, Ernst & Garron, 2015; Lazzari & Baldisserotto, 2008; Sindilariu, Brinker & Reiter, 2009; Verdegem, 2013). The potential impacts of such intensive FT fish farming on receiving river quality can include increased concentrations of five-day biochemical oxygen demand (BOD<sub>5</sub>), a drop in dissolved oxygen concentrations (DO), an enrichment of total suspended solids (TSS) content in the receiving water, in nutrients such as nitrogen generally characterised by total ammonia nitrogen (TAN), and phosphorus generally characterised by orthophosphate (PO<sub>4</sub>-P) that could both potentially lead to surface water eutrophication (e.g., *Boaventura et al.*, 1997; Caramel et al., 2014; Garcia et al., 2014; Lazzari & Baldisserotto, 2008; Teodorowicz, 2013). This impact will depend on the production system employed, the production intensity and on the type of feed but also on the assimilative capacity of the receiving water (*Boaventura* et al., 1997; Caramel et al., 2014).

A number of studies have focused on the characterization of freshwater fish farm effluents across the world (e.g., *Boaventura et al., 1997; Guilpart et al., 2012; Lalonde, Ernst & Garron, 2015; Neto, Nocko & Ostrensky, 2015; True, Johnson & Chen, 2004*). Most of these studies have inherent drawbacks due to their respective framework; some studies focused only on a short period of time of one year or less (e.g., *Caramel et al., 2014; Všetičková et al., 2012; Živić et al., 2009*) or relied on individual water samples (e.g., *Caramel et al., 2013; Všetičková et al., 2012; Živić et al., 1996; Lalonde, Ernst & Garron, 2015; Noroozrajabi et al., 2013; Všetičková et al., 2012*), while the impact of aquaculture should be assessed using longer term evaluation process encompassing production, fish life stages and river characteristics variations (*Aubin, Tocqueville & Kaushik, 2011; Hennessy et al., 1996*). Other studies (e.g., *Koçer et al., 2013; Sindilariu, Brinker & Reiter, 2009; Všetičková et al., 2012; Yalcuk, Pakdil & Kantürer, 2014*) have focused on fish farm gates (comparison of farm inlet and outlet water quality) and did not aim at evaluating the impact they might have on receiving water. Some limited studies

(*Aubin, Tocqueville & Kaushik, 2011; Boaventura et al., 1997; Pulatsu et al., 2004*) analysed the impact of freshwater fish farming on the receiving water quality during a long period of time (i.e., at least two years duration), allowing one to fully understand the impact of specific farms on specific rivers and in specific places. However, data from these studies are difficult to extrapolate and apply into different contexts in light of the parameters and conditions such as species, temperature, aquaculture practices, chemicals used, receiving water features (e.g., flow, hydromorphology) and local environmental conditions that may be very specific to individual farms and locations (*Lalonde, Ernst & Garron, 2015*).

With the increasing demand for fish and fish products worldwide (e.g., Guilpart et al., 2012), the freshwater aquaculture industry is facing the challenge of finding the way to produce more without any associated environmental degradation (Martins et al., 2010; Sturrock et al., 2008; Teodorowicz, 2013). In some countries such as Denmark, freshwater aquaculture practices have recently evolved to model trout farms, systems that were able to fulfil both production and environmental objectives (Jokumsen & Svendsen, 2010; Lalonde, Ernst & Garron, 2015; Teodorowicz, 2013). However, some other countries such as Poland (Teodorowicz, 2013), France (Papatryphon et al., 2005) and Ireland still require a drastic evolution of the aquaculture practices through more advanced systems in order to fulfil those ambitious objectives. To the best of our knowledge, there is no existing data about the impact of freshwater fish farming on downstream river quality in Ireland. There is, therefore, a pressing need to evaluate the impact of aquaculture activity on a longer-term basis using evidence-based data that are publically available, and to ascertain if there is any accumulation of pollution in receiving water. Thus, the main aim of the present study was to evaluate such data in terms of reported aquaculture influent and effluent parameters over a ten year period in order to establish relationship (if any) with receiving water quality in Ireland. In that aim, the impact of four consecutive FT traditional freshwater rainbow trout (Oncorhynchus mykiss) farms on the receiving water quality was assessed through publically available historical data analysis (2005–2015). The objectives of the study were (i) to characterize the impact of each fish farm on water quality in function of their respective production and identify any seasonal variability, (ii) to quantify the cumulative impact of the four farms on the river quality and to check if the self-purification capacity of the river was enough to allow the river to reach back its background levels for the analysed parameters, (iii) to build a baseline study for Ireland in order to extrapolate as a dataset for expected climate change and production intensification.

## **MATERIALS AND METHODS**

## **Fish farms description**

The studied facilities are all Irish traditional rainbow trout fish farms, all operating in a FT system during the studied period. The schematic representation of fish farms in Fig. 1 shows the very specific configuration of the area with four different farms, all abstracting the water needed and having their discharge into the same river, all in a relative small area (about 4 km between the first and the last farm). This specific configuration allowed the study of the potential cumulative impact the fish farms might have on river quality.



Full-size DOI: 10.7717/peerj.5281/fig-1

Farm 1 (F1) was a hatchery and juvenile (i.e., <100 g) production site. It was operating in FT in a concrete channel separated in different consecutive sections and had a partial recirculation of the water associated to an oxygenation step (no water treatment). As a hatchery site, this farm was the most sensitive to water quality, and which is why it was historically located at the most upstream location where the water quality was not theoretically affected by the fish farm activity yet. Farm 2 (F2) was about 2 km downstream from F1 and was a medium size (i.e., 100–500 g) on-growing site that was operating in FT in an earthen pond. Farm 3 (F3) was about 1.5 km downstream from F2 and was an on-growing site (i.e., 500 g to harvest size) that was operating in FT in earthen ponds with a partial recirculation of the water associated to oxygenation (no water treatment). Farm 4 (F4) was about 500 m downstream from F3 and was an on-growing site (i.e., 500 g to harvest size) that was operating in FT in an earthen pond associated with a sedimentation pond aiming at removing settleable solids from the discharge water.

On average, the four farms produced 75, 50, 165 and 100 tons a year respectively during the investigated period (2005–2015).

Considering that there was no or limited form of water treatment on these farms (Fig. 1), the objective was to evaluate the impact of the whole pool of farms, and of successive individual farms, on the water quality in order to check if there was any accumulation of pollution (e.g., TSS, BOD<sub>5</sub>, TAN, PO<sub>4</sub>-P) within their receiving water. According to Ireland's EPA (http://www.epa.ie/hydronet/#Water%20Levels), the river 95% percentile flow was calculated as being 1.1 m<sup>3</sup>/s (average over 30 years of 5.5 m<sup>3</sup>/s). It is noteworthy

that there was no agricultural activity or WWTPs within the area (i.e., upstream and in between the different farms) and that therefore, in the present study, any significant modification of the river quality downstream from the farms will be attributed to the fish farming activity and will represent its impact on the river quality.

The farms water intake flow was measured at regular intervals by the fish farmer and were about  $0.4 \text{ m}^3$ /s on average on each farm (range  $0.1-0.6 \text{ m}^3$ /s depending on the farm and the flow conditions in the river).

## The discharge licence

Each Irish County Council (Local Authority) require two licences to operate—an aquaculture licence issued by the Department of Agriculture, Food & the Marine and a Trade Effluent Discharge Licence issued by the relevant Local Authority.

For these four farms the discharge licences states both (i) a regulation on the maximum water abstraction rate that the farm cannot exceed depending on the river flow, and (ii) a maximum differential concentration (or value) between fish farms influent and effluent waters for a range of parameters (i.e.,  $1 \text{ mgO}_2/\text{L}$  for BOD<sub>5</sub>, 10 mg/L for TSS, 5 NTU for turbidity, 0.4 mgN/L for TAN, 0.002 mgN/L for nitrite (NO<sub>2</sub>-N) and 0.2 mgP/L for PO<sub>4</sub>-P). For three other parameters, the regulation defines absolute limit values for the farm effluent (i.e., 60% saturation for DO, range 6–9 for pH and ambient temperature). The discharge licence specified that each fish farm had to be sampled by the regulatory agency for both inlet and outlet water at a frequency of four times a year (i.e., generally one sampling per season).

## The parameters monitored

A full ten-year record of historical data (2005–2015) generated by an independent and accredited water analysis lab was gathered for this study. The present study focused on the parameters that were monitored and analysed during the studied period. Hence regulated parameters (presented in the previous section) and total oxidised nitrogen (TON), representing the sum of NO<sub>2</sub>-N and nitrate (NO<sub>3</sub>-N), were considered for the present study.

Surprisingly, according to the publicly available documents that were consulted for this study, ammonium (NH<sub>4</sub>-N) was monitored instead of TAN (regulated parameter) that represents the sum of NH<sub>4</sub>-N and ammonia (NH<sub>3</sub>-N), the latter being much more toxic for fish (*Tomasso*, 1994). The predominance of one ammonia form or another depends on pH and temperature with NH<sub>3</sub>-N predominating at high temperature and pH values (*Emerson et al.*, 1975). Therefore, for the present study, pH and temperature values were used to enable an estimation of TAN concentrations from NH<sub>4</sub>-N measurements.

Validated standard methods (*APHA*, 2012; *APHA*, 2005) were employed by the accredited labs for the analysis of each monitored parameter.

## Sampling

All samples were performed as spot samples and no composite samples were taken. According to the Local Authority responsible for the water monitoring, all the inlet sampling locations ( $i_{1-4}$ , see Fig. 1) were located in the river itself immediately upstream from the inlet channel of each farm (taken every time at the same section at the centre of the river). The outlet sampling spots ( $o_{1-4}$ , see Fig. 1) were directly located in the outlet channels of each farm (farm effluents) and therefore do not include any dilution by the downstream river.

The discharge licence did not specify a sampling method and thus it is likely that spot samples were employed as they were considered the most efficient and cheap sampling method. Samples (i.e., 1L in PET bottles) were stabilized in acidic conditions (to reach a pH value below 2 in order to avoid any transformation of the nitrogenous compounds) and brought to the labs where they were kept refrigerated before analysis. Dissolved oxygen saturation levels, pH and temperature values were obtained *in-situ* before sample stabilization by a multi-parameter sensor (YSI 51B oxygenmeter and WTW pH 330 pH meter). Separate samples were taken for the analysis of BOD<sub>5</sub> parameter (not acidified).

## Approach employed for the quantification of the fish farms impacts

The following range of assessments were considered when evaluating the impacts of the farms on the receiving river water—individual farm impact on water quality, and cumulative impact of the four fish farms on river water quality. Additionally, in order to check a potential higher impact during summer condition (as observed in *Lalonde, Ernst & Garron, 2015*), the seasonal variability of the impacts was also studied by considering the evolution of the effluents, quality and the impacts for the different seasons during the year. For this purpose, data obtained from a monitoring in December, January and February were classified as "winter data"; the ones from the months of March, April and May as "spring data"; the ones from the months of June, July and August as "summer data" and the ones from the months of September, October and November as "autumn data".

As discussed above, the four fish farms were abstracting the water from and discharging into the same river (Fig. 1). Therefore, the fish farms inlet analysis results (i.e.,  $I_1$ ,  $I_2$ ,  $I_3$ and  $I_4$ ) were employed to study the evolution of the water quality along the river across the 4 farms. We noticed that this method was used to study the impact of the three first farms (i.e., farms 1–3). The evaluation of the cumulative impact of the three first farms (F1, F2 and F3) on the river quality was achieved by the comparison of the results obtained in  $i_1$  and  $i_4$  sampling locations. This allowed us to take account of the dilution of the effluent of each farm by the river and therefore to take account of the pollution load emitted by the farms. The impact of F4 was not possible to assess this way because river quality downstream from this farm was not monitored (i.e., the only location of these farm effluent monitoring points that does not take account of the dilution by the river, see Fig. 1).

Unfortunately, those results were not correlated with the fish parameters because we could not have access to accurate records of these biological data but only of average yearly production. Therefore, this study only focused on water quality parameters.

### Data treatment

The following equation was used to calculate the differential concentrations  $(D_i)$  for each monitored farm (i) and parameter introduced above.

$$D_i = O_i - I_i.$$

With  $O_i$  and  $I_i$  the outlet and inlet concentrations at the farm *i* for a given parameter on a given sampling event.

Only concentration values were considered and treated although load values (taking farm outlet flow values in consideration) would have been a better way to characterize the impact of the different farms. However, flow values were not measured continuously on the different sites and this did not allow for an accurate calculation of the loads for the different parameters at different times. Also, river concentrations were employed to characterize the impact of the different farms on river quality; those river concentrations are taking river dilution into account and were therefore considered as a good approach to consider the load of each of parameters considered.

Statistical tests were performed using MATLAB software (MathWorks, Natick, MA, USA) in order to assess if datasets means were significantly different (e.g., concentrations measured upstream and downstream from each farm) and if the fish farms had an impact on the water quality. Box plot representations of the data were chosen in order to show the dispersion of the data and to avoid any misinterpretation due to the occurrence of extreme values. Two samples Student's *t*-tests were employed to compare datasets after checking the normal distribution of each dataset (validity domain of this statistical test). A confidence interval of 95% was systematically employed.

Many results from the datasets obtained from the regulation body were below the limit of quantification (LOQ) associated to the analytical method employed. In this case, for the calculations performed in the present study and because the sampling and analysis were not performed by our team, the data was considered as equal to the LOQ (substitution approach). The real value of the data could be lower than LOQ but this "conservative" approach of "below LOQ = LOQ" was chosen here, although other approach could have been used such as "below LOQ = LOQ/2" or "below LOQ = LOQ/ $\sqrt{2}$ " (*Hornund & Reed, 1990*; *US Environmental Protection Agency, 2003*), to avoid any misinterpretations.

## **RESULTS AND DISCUSSION**

# General overview of the results—outcome of inlet and outlet values for each parameters

More than 1,000 inlet/outlet couples of data were gathered overall for all the parameters and the four fish farms investigated. The full dataset as well as the representation of the data for each parameter and each farm and the monthly distribution of the data are presented in supplementary materials sections (SP1, Tables S1 and S2).

Figure 2 gives an overview of the gathered data for all farms and presents a global comparison of inlet and outlet water quality observed globally and for all the parameters monitored.

Globally, by comparing inlet and outlet values for each parameter, Fig. 2 shows the mean increase or decrease of the monitored parameters values caused by the four farms without distinction. No significant differences between farms inlets and outlets were observed for TON, temperature, BOD<sub>5</sub>, turbidity, TSS and NO<sub>2</sub>-N (P value > 0.05) demonstrating that the investigated fish farms might have no impact on downstream river quality for these



Figure 2 Box-plot global representation of inlet and outlet values compiled and gathered for the four fish farms and for all the monitored parameters (2005–2015).

Full-size DOI: 10.7717/peerj.5281/fig-2

parameters. However, a significant increase from fish farms inlets to outlets was observed for NH<sub>4</sub>-N (P value < 0.001), PO<sub>4</sub>-P (P value < 0.001) and BOD<sub>5</sub> (P value < 0.01) demonstrating that the investigated fish farms might have an impact on the downstream river quality through the release of those compounds. However, the increase in NH<sub>4</sub>-N was only about 0.08 mgN/L on average ( $\pm 0.08$  mgN/L, n = 130, see Fig. 2F), which can be considered as a relatively low value compared to values found in other studies such as a 1.46 mgN/L increase (*Boaventura et al.*, 1997). The same observation can be done for  $PO_4$ -P with an average increase of about 0.015 mgP/L ( $\pm 0.03$  mgP/L, n = 130, see Fig. 2G) found in the present study and a range of 0.06-0.58 mgP/L increase found in another study where three different rainbow trout farms were investigated during one year (Boaventura et al., 1997). Some other studies observed some increase from rainbow trout farms inlets to outlets for NH<sub>4</sub>-N (Caramel et al., 2014; Guilpart et al., 2012; Kırkağaç, Pulatsu & Topcu, 2009; Lalonde, Ernst & Garron, 2015), BOD<sub>5</sub> (Teodorowicz, 2013), total nitrogen (Caramel et al., 2014; Lalonde, Ernst & Garron, 2015), total phosphorus (Caramel et al., 2014; Kırkağaç, Pulatsu & Topcu, 2009; Lalonde, Ernst & Garron, 2015) and PO<sub>4</sub>-P (Caramel et al., 2014; Guilpart et al., 2012). Furthermore, a significant decrease was observed in the present study at fish farms outlet compared to inlets for both DO (P value < 0.001) and pH (P value = 0.001) showing, as expected, (i) a global oxygen consumption due to fish metabolism (Boyd & Tucker, 1998), and (ii) a production of carbon dioxide by fish with the consequence of lower pH value at the outlets (Boyd & Tucker, 1998). An average decrease in DO of about  $1.42 \text{ mgO}_2/\text{L}$  ( $\pm 1.28 \text{ mgO}_2/\text{L}$ , n = 101, see Fig. 21) from farms inlets to farms outlets was observed, that is in agreement with previously published results such as a study where a DO decrease of  $0.7-2.4 \text{ mgO}_2/\text{L}$  depending on the fish farm investigated was found (Boaventura et al., 1997). A deeper focus on these data, with an emphasis on the inlet and outlet of each farm, will be presented in the following sections of this study as to whether or not they confirm these global trends. An evaluation of potentially significant impacts of those individual fish farms on the receiving water quality for the monitored parameters will be provided.

## Impact on water quality—individual farms level

The objective of this section are (i) to quantify the impact of each farm on water quality and (ii) to address their compliance to their discharge licences criteria.

## Parameters regulated in term of differential concentrations

Here we present an outcome of the differential concentrations for five regulated parameters (i.e., BOD<sub>5</sub>, NO<sub>2</sub>-N, PO<sub>4</sub>-P, TSS and turbidity) and for ammonium (NH<sub>4</sub>-N) for each farm (Fig. 3).

A first general observation of the results is that for most of the regulated parameters (i.e., turbidity, TSS, PO<sub>4</sub>-P, NH<sub>4</sub>-N, BOD<sub>5</sub>) neither the average nor median values are higher than their associated differential limit value. The different farms were therefore globally in compliance with their discharge licence for those parameters. An exception is NO<sub>2</sub>-N, with an average differential concentration higher than the limit value (i.e., 0.006  $\pm$  0.025 mgN/L, n = 40 vs 0.002 mgN/L, see Fig. 3C) for F4. However, the median value was equal



Figure 3 Box-plot representation of the differential concentrations ( $D_i$ ) between inlet and outlet of each of the four fish farms and for each regulated parameter (2005–2015). Numbers 1, 2, 3 and 4 represent the consecutive fish farms in the order of the river flow direction. In brackets, the discharge licence limit value for each regulated parameter as mentioned in the discharge licence. Median, average, Q1 and Q3 are presented. For NH<sub>4</sub>-N, the limit value associated is for total ammonia (TAN).

Full-size DOI: 10.7717/peerj.5281/fig-3

to the limit value (i.e., 0.002 mgN/L) underpinning the influence of extreme values on the average calculation. For this parameter, a large number of data were <LOQ and as stated above were considered as equal to the LOQ; for F1 a very large proportion of the data (i.e., about 95%) were <LOQ at both inlet and outlet giving resulting differential concentrations of 0 mgN/L according to the equation presented in the materials and method section (it is noticed that the same result would have been obtained with a different choice of values for analysis below the LOQ). Surprisingly, the LOQ for NO<sub>2</sub>-N was generally 0.06 mgN/L which is higher than the differential concentration limit of 0.002 mgN/L. This shows the limit of the approach set by the Local Authority to set some very low differential limits

that cannot actually be addressed by the chemical analysis procedure typically applied in accredited chemical analysis labs.

For BOD<sub>5</sub>, a global compliance with the discharge licence was observed with both average and median values below the differential limit value (i.e.,  $1 \text{ mgO}_2/\text{L}$ , see Fig. 3B) for the four farms. However, for F2, F3 and F4 a substantial number of individual differentials values was above the limit value (i.e., about a quarter of the values for the three farms) demonstrating that even if a general compliance was observed the compliance was not achieved all the time. The low differential limit value for this parameter (i.e.,  $1 \text{ mgO}_2/\text{L}$ ) compared to other fish farms in the country (i.e., generally 2 mg/L) might explain this relative high number of "non-compliance" data. Furthermore, according to the statistical analysis performed, no significant difference were observed between the different farms for BOD<sub>5</sub> differential values (i.e., *P* values > 0.05).

For TAN (i.e.,  $NH_4-N + NH_3-N$ ), the compliance cannot be directly assessed by the available  $NH_4-N$  monitoring; however, considering that pH values were always below 8 (i.e., range 6.5–7.5 pH units, see Supplementary Material SP1) and the temperature range of 5–15 °C during all the monitoring period (see Supplementary Material SP1),  $NH_4-N$  was highly predominating over  $NH_3-N$  and therefore TAN concentrations were similar to  $NH_4-N$  (about 0.1 mgN/L depending on the farm, see Fig. 3F). Therefore, the conclusion is that all farms were in compliance with the differential concentration limit for TAN of 0.4 mgN/L. Furthermore, according to the statistical analysis performed, some significant differences were observed between the different farms in terms of  $NH_4-N$  differential values; mean  $NH_4-N$  differential value for F1 was revealed to be significantly lower than for F2 (*P* value > 0.05), F3 (*P* value < 0.001) and F4 (*P* value < 0.01). No significant difference was observed between F2, F3 and F4 for this parameter.

Farm 2 and F4 showed a substantial proportion of negative differential values for turbidity and TSS (i.e., nearly half of the total number of data for these farms, see Fig. 3E) demonstrating that those farms were polishing the water regarding solids. The earthen pond configuration and low water velocity for F2 and the presence of a sedimentation pond in F4 could explain this observation (Fig. 1). However, the presence of such solid removal processes did not enhance the removal of other compounds such as BOD<sub>5</sub> and PO<sub>4</sub>-P that could have been reduced by the presence of a sedimentation pond (*Teodorowicz, 2013*). The statistical analysis of the data showed that the TSS differential mean value was significantly lower for F2 than for F3 (*P* value < 0.01). No other significant difference between farms was observed for both TSS and turbidity.

Farm 3 was the farm with the highest average yearly production of 165 tons (Fig. 1) and was also the one associated to the highest average differential values for PO<sub>4</sub>-P and NH<sub>4</sub>-N (i.e.,  $0.02 \pm 0.042$  mgP/L, n = 42 and  $0.1 \pm 0.08$  mgN/L, n = 42, respectively, see Figs. 3D and 3F), even if the difference with the other farms were not significant according to the statistical analysis. In the present study, a higher fish production led to a higher release of phosphorus and nitrogenous wastes as was observed in other studies such as *Guilpart et al. (2012)* where a proportionality relationship was observed between nutrient release and farm production. For the same set of parameters, F1 was revealed to be less impacting than the other farms with on average lower differential values than any other farm. This might

be due to a relative low yearly production in this farm (i.e., 75 tons, see Fig. 1) and the fact that this was a hatchery and juvenile production farm as it was observed in another study (*Guilpart et al., 2012; Teodorowicz, 2013*). The statistical analysis revealed that mean PO<sub>4</sub>-P differential value for F1 was significantly lower than for F3 (*P* value = 0.01) and for F4 (*P* value < 0.001). No other significant difference between the different farms was observed for PO<sub>4</sub>-P.

Globally, a general compliance with the discharge licence for all the regulated parameters was observed for the four farms and therefore the studied individual fish farms were not substantially impacting water quality. However, it was also observed that the F3 and F4 had on average higher differential values than the other farms. This fact could be due to higher yearly production for F3 and F4 compared to the other farms (Fig. 1) (*Boaventura et al., 1997; Guilpart et al., 2012; Teodorowicz, 2013*).

### Parameters regulated in term of absolute values

Three parameters were regulated in term of absolute effluent values (i.e., DO, pH, temperature). The limit value for DO was 60% saturation in water and it appeared in the present study that no value was found to be below this threshold (average value for all farm outlets of 80% saturation, see Supplementary Materials SP1). The accepted range for pH was 6–9 pH units and no value was found to be out of this range, with an actual range of 6.5–7.5 pH units for all farms (see Supplementary Materials SP1). For temperature, the discharge licence stated that the effluent had to be "ambient" and no significant differences were observed between farms inlets and outlets (see Supplementary Materials SP1) meaning that the outlet water was actually at ambient temperature. Therefore, the four different farms were in compliance with the discharge licence for these three parameters and the farms were not substantially impacting water quality for these parameters.

The statistical analysis revealed that for temperature, the mean differential value for F3 was significantly higher than for F4 (P value < 0.05). This could be due to the different configuration of those farms; F3 being a multi-pond based farm with an assumed (not measured) associated relatively high hydraulic retention time (HRT), allowing a higher water heating potential than F4 which was assumed to be associated to a lower HRT due to its different configuration (i.e., only two ponds) and therefore a lower heating potential than F3. The statistical analysis for the other parameters (i.e., pH and DO) mean differential values did not reveal any significant differences between the four farms.

## Cumulative impact of the fish farms on the river quality

The evolution of the quality of the inlet water through the four consecutive fish farms is presented in Fig. 4.

## Upstream water quality

Background river quality (i.e.,  $I_1$ , see Fig. 4) was on average very suitable for rainbow trout production with temperature range of 5–15 °C (see Supplementary Materials SP1), high level of DO (i.e., about 10.3 ± 1.29 mgO<sub>2</sub>/L, n = 17 on average, see Fig. 4G), low levels of BOD<sub>5</sub> (i.e., about 2.06 ± 0.95 mgO<sub>2</sub>/L, n = 17 on average, see Fig. 4B), NH<sub>4</sub>-N (i.e., 0.08 ± 0.004 mgN/L, n = 17 on average, see Fig. 4D), NO<sub>2</sub>-N (i.e., 0.06 ± 0.012 mgN/L, n = 17



**Figure 4** Box-plot representation of the water quality evolution through the four different fish farms inlets showing the cumulative impact of the fish farms on river quality for the monitored parameters. Numbers 1, 2, 3 and 4 represent the consecutive farms in the order of the river flow. Median, average, Q1 and Q3 are presented.

Full-size 🖾 DOI: 10.7717/peerj.5281/fig-4

on average, see Fig. 4H) and TON (i.e.,  $1.36 \pm 0.32$  mgN/L, n = 16 on average, see Fig. 4A). Overall, these levels are all suitable for fish growth and all below the proven chronic toxicity levels for salmonids (*EEC*, 1978). Furthermore, these values confirmed that there was no potentially polluting activity (e.g., agriculture, WWTP) upstream from the fish farms that would have degraded the river quality for the parameters monitored.

## BOD<sub>5</sub>/dissolved oxygen (DO) patterns through the river flow

Overall there was no negative impact observed on river water quality due to the farms in terms of BOD<sub>5</sub> and DO concentrations which remained at about  $2 \text{ mgO}_2/\text{L}$  and  $10 \text{ mgO}_2/\text{L}$  respectively across the four different farms inlets and therefore along the river.

For DO, these results demonstrated that even if a significant decrease was observed at farm scale (as mentioned before), the river had the potential to reoxygenate between the different farms to reach its background level. In a similar study dealing with the cumulative impact of five different rainbow trout farms in Turkey, *Pulatsu et al. (2004)* observed an impact of fish farming that significantly decreased the DO levels in the river at 100 m downstream from the last farm. This relative short distance compared to distances between the different farms in the present study might explain why an impact was observed in this study.

For BOD<sub>5</sub>, no significant difference in the inlet mean concentrations for the four farms was observed in the present study according to the statistical test applied (P value > 0.05) (Fig. 4B). This demonstrated that even if some  $BOD_5$  differential concentrations were higher than the limit values for F2, F3 and F4, this had globally no impact on the river quality. A possible explanation for this observed trend could be a combination of (i) a relative high river flow compared to average fish farms water uptake flow providing a high dilution capacity to the river, (ii) a relative low differential concentrations for BOD<sub>5</sub> associated to river self-cleaning potential for this compound as it was observed in a study dealing with nine different rainbow trout and carp farms in Poland (Teodorowicz, 2013). On the contrary, *Pulatsu et al.* (2004) observed a significant impact of trout farming on the BOD<sub>5</sub> concentration in the river. However, in this study the downstream monitoring station was only at 100 m from the last farm investigated that might not be enough distance for the river to clean itself or to reoxygenate as it was observed in other studies (e.g., Boaventura et al., 1997; Teodorowicz, 2013). Depending on river/farm flow and river characteristics, a distance of 2-3 km downstream the fish farm is considered as being necessary to allow a self-purification of the river regarding BOD<sub>5</sub> (*Boaventura et al.*, 1997).

## Ammonium pattern through the river flow

Slight variations were observed for NH<sub>4</sub>-N concentrations across the four fish farms inlets and therefore along the river. However, upstream and downstream farms inlets (i.e., F1 and F4 respectively) were associated to similar NH<sub>4</sub>-N mean concentrations of 0.08 mg/L (no significant differences between those two farms, *P* value > 0.05) (Fig. 4D). Thus, as for BOD<sub>5</sub> parameter, we can assume that the distance between the different farms in the present study was long enough to allow the river to purify itself regarding NH<sub>4</sub>-N and to get back to its initial background concentration. Therefore, there was not any impact of the fish farms on the river quality in term of NH<sub>4</sub>-N concentrations. However a significant difference was observed between F3 and F4 for this parameters (*P* value < 0.01) with a higher value at F4 inlet meaning that F3 might have had an impact on river quality for NH<sub>4</sub>-N. The relative high average yearly production in F3 (i.e., 165 tons, see Fig. 1) associated to the relative short distance between those two farms (i.e., 500 m, see Fig. 1) might explain this observation due to a limited self-cleaning potential by the river within this relative short distance (*Boaventura et al., 1997; Lalonde, Ernst & Garron, 2015*). Depending on farm production and dry weather flow for the receiving water, *Boaventura et al. (1997)* estimated at 3–12 km the distance necessary for the river to get back to its initial organic chemical compound (including NH<sub>4</sub>-N) concentration downstream from a given trout farm. With such low differential concentrations in F4 (0.08  $\pm$  0.08 mgN/L, n = 41, on average, see Fig. 3F) and the self-cleaning potential of the river for this compound (*Boaventura et al., 1997; Lalonde, Ernst & Garron, 2015*), it appears very unlikely that this fish farm would have any impact on the downstream river quality in term of NH<sub>4</sub>-N concentration.

## Nitrite pattern through the river flow

A decrease in NO<sub>2</sub>-N concentrations was observed across the fish farm inlets and therefore along the river (Fig. 4H). Median NO<sub>2</sub>-N concentration was revealed to be 0.06 and 0.03 mgN/L at the upstream (i.e., F1 inlet) and downstream (i.e., F4 inlet) monitoring locations respectively. Therefore, considering this observation, the fish farm activity might have had a positive impact on the river quality regarding the NO<sub>2</sub>-N concentration. To our knowledge, this potential "purification" of a river by trout farms regarding NO<sub>2</sub>-N was never observed in fish farms where no denitrification step is applied, as was the case for the investigated fish farms in the present study where no water treatment processes were applied.

However, this result might be an artefact because as observed before most of the results obtained for F1 inlet location were below the LOQ of generally 0.06 mgN/L. As previously mentioned, in these cases, a value equal to the LOQ was considered for the present study but the real values could have been lower. The analytical LOQ was generally lower for F4 than for F1; this could explain why this "apparent dilution" of the river by the farms effluent was obtained for this parameter. The relative small dataset for F1 (n = 17) compared to the other farms (n = 30–41) might also explain the differences observed between upstream and downstream locations (i.e., F1 and F4 respectively). Other studies demonstrated the potential impact of trout farming on NO<sub>2</sub>-N levels in river. *Pulatsu et al.* (2004) observed a significant increase from 0.019 to 0.581 mgN/L from upstream to 100 m downstream from five different trout farms. This could confirm the occurrence of an artefact with the results of the present study.

## Total oxidized nitrogen pattern through the river flow

A slight increase was observed across the fish farms inlets and therefore along the river for TON with median concentrations of 1.4 and 1.8 mgN/L for the upstream (i.e., F1) and downstream (i.e., F4) locations respectively (Fig. 4A). Therefore, fish farming might have a slight impact on the river quality considering this parameter and might be responsible for a release of NO<sub>3</sub>-N (not directly monitored). However, according to statistical analysis, no significant difference was observed between F1 and F4 inlets for this parameter average

values (*P* value > 0.05). This result is in agreement with the literature on this topic; in one study, NO<sub>3</sub>-N mean concentrations were observed to increase from 0.13 mgN/L to 0.43 mgN/L from upstream to downstream of freshwater salmonids farms in Canada but no significant difference was observed (*Lalonde, Ernst & Garron, 2015*). In another study, a significant impact of trout farming on river quality for both NO<sub>3</sub>-N and NO<sub>2</sub>-N was demonstrated (*Pulatsu et al., 2004*). Another study dealing with the potential impact of eight different trout farms in France demonstrated that there was no trend for NO<sub>3</sub>-N and that fish farms could be either responsible of an increase or a decrease of NO<sub>3</sub>-N concentrations in the receiving water (i.e., 100, 1,000 m downstream from the farms) (*Guilpart et al., 2012*).

### Orthophosphate pattern through the river flow

Stable PO<sub>4</sub>-P concentrations were observed from F1 inlet to F3 inlet. Then, an increase was observed on PO<sub>4</sub>-P concentration from 0.01 mgP/L to almost 0.02 mgP/L between F3 and F4 inlets respectively (i.e., median values, see Fig. 4E). Therefore, and considering that there was not any other potentially polluting activity in the area (i.e., WWTP, agriculture), F3 had an impact on the river quality in term of PO<sub>4</sub>-P concentration. This was confirmed by the statistical analysis revealing that mean PO<sub>4</sub>-P inlet concentrations for F3 and F4 were significantly different (*P* value < 0.001) and that PO<sub>4</sub>-P concentrations were significantly higher at the F4 inlet than at F3 inlet. As for NH<sub>4</sub>-N this trend might be due to both a relative high production in F3 and short distance between F3 and F4. Considering similar PO<sub>4</sub>-P differential concentrations for F3 and F4 of about 0.02 mgP/L (see Fig. 3D) and a similar water uptake flow both farms, it is highly possible that an impact of F4 would have been observed if the river was monitored directly downstream from this fish farm. However, considering the self-cleaning potential of the river, the river might have got back to its PO<sub>4</sub>-P background concentration of about 0.01 mgP/L (see Fig. 4E) a few kilometres downstream from F4 (Boaventura et al., 1997). In another study, an impact of trout farming on total phosphorus (TP) on the river quality was observed with an increase from 0.069 mgP/L upstream to 0.117 mgP/L downstream from the fish farms (*Pulatsu et al., 2004*). It is assumed that this increase in TP is likely to be due to an increase in the reactive form, PO<sub>4</sub>-P, which was the monitored parameter in the present study.

## Total suspended solids/turbidity patterns through the river flow

A slight TSS (and turbidity) increase were observed all along the river from 2 mg/L (2 NTU) to 3 mg/L (2.5 NTU) for upstream (i.e., F1 inlet) and downstream (i.e., F4 inlet) locations respectively (Figs. 4F and 4C). This demonstrated that the four fish farms had a very limited impact on the river quality in term of TSS (and turbidity). This limited impact was confirmed by the statistical analysis that demonstrated that there was no significant differences between TSS and turbidity average values between F1 and F4 inlet values (*P* value > 0.05). Considering relative low TSS and turbidity differential values observed for F4 (see Fig. 3E), it is unlikely that this fish farm would impact the downstream river quality for these parameters. This result is in agreement with the literature (*Lalonde, Ernst & Garron, 2015; Pulatsu et al., 2004*) where a non-significant impact of trout farms on receiving water quality was observed for TSS.

## Temperature/pH patterns through the river flow

No significant impact of fish farms on both river temperature and pH was observed in this study. This result is in agreement with *Pulatsu et al. (2004)*, who did not observe any impact of trout farms on the receiving water quality for these parameters.

## Seasonal variability (influent/effluent quality, impact)

Differential values were gathered by season with no distinction between the different farms. The results are presented in Fig. 5.

The results were compared statistically season by season in order to show any significant difference from one season to another for each parameter. Overall, except for NH<sub>4</sub>-N (Fig. 5C), no significant difference was observed between the average results obtained for each season for all the parameters monitored. Even if differential values were on average higher in summer for TON, NH<sub>4</sub>-N, PO<sub>4</sub>-P and NO<sub>2</sub>-N, the statistical analysis revealed that no significant differences were observed with mean values calculated for the other seasons. The analysis of the results obtained for NH<sub>4</sub>-N revealed that mean values obtained for the spring season were significantly lower (P value < 0.01) than the results obtained for any other season. No significant difference was observed when the other seasons were compared. These results demonstrate that the season change had an overall limited impact on the polluting potential of the fish farms investigated and on the parameters monitored and that, despite apparent higher differential values for nutrient during summer, fish farming was not significantly more impacting during this season than during any other seasons in terms of discharge. This observation is in agreement with another study where a negligible impact of season on 15 trout farms effluent quality was observed for all the parameters monitored except for TP (Lalonde, Ernst & Garron, 2015). However, other studies demonstrated that the impact of trout farms on receiving water quality was higher in summer conditions when stocks are high, and both river flow and DO levels are low (Boyd & Tucker, 1998). In the future, climate change might increase the gap between winter and summer conditions with potential higher temperature/lower flows in rivers during the summer; therefore, the impact on aquaculture and on effluent water quality might be more and more significant.

## CONCLUSION

A ten-year longitudinal survey (2005–2015) of the impact of four consecutive FT trout farms was performed for the first time in Ireland based only on publically available and independently generated data analysis. First, it was demonstrated that publically available data, although not always of very good quality, can be used to reliably assess the impact of fish farming on receiving water quality. The impact of each farm on water quality was assessed, and it was demonstrated that the impact was significant for NH<sub>4</sub>-N and DO and more important for the farms which produced the most during the investigated period. Those results were expected in regard to other studies performed on the impact of traditional trout production across the world (*Boaventura et al., 1997; Caramel et al., 2014; Garcia et al., 2014; Teodorowicz, 2013*). The cumulative impact of the fish farms on receiving water quality was also assessed and it was demonstrated that the distance between



**Figure 5** Box-plot representation of the seasonal variation of the differential values compiled and gathered for the four fish farms for the monitored parameters. Median, average, Q1 and Q3 are presented.

Full-size DOI: 10.7717/peerj.5281/fig-5

the different fish farms was globally sufficient to allow the river for self-purification regarding the parameters analysed and no overall cumulative impact was observed for the parameters considered. However, considering the relative high water volumes extracted by the studied farm, an increase production would not be possible without the addition of water treatment technologies to apply water reuse and therefore to reduce the demand on water without impacting receiving water quality. The present study represents the first benchmarking of the freshwater fish farming industry in Ireland and will be used as a baseline study, along with a study of the potential impacts on river's hydromorphology, for comparison before the evolution through more advanced practices and the expected implementation of water treatment processes in a near future due to the more and more stringent legislative framework. However, only the relevance of using the water quality parameters can be discussed. Indeed, notions such as the impact on biodiversity and ecotoxicology were not assessed in the present study and will represent a challenge for future studies in order to fully take account of the impact of the aquaculture industry in regard to the implication of the WFD and the expected future intensification of the production. The present study will provide a basis for policy and commensurate decisions on future fish farm licencing and for meeting good quality status under WFD as it pertains to the operation of fish farming facilities.

## **ADDITIONAL INFORMATION AND DECLARATIONS**

## Funding

This work was supported by Ireland's Department of Agriculture, Food and the Marine through the Morefish project and by Bord Iascaigh Mhara though the EcoAqua project. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

## **Grant Disclosures**

The following grant information was disclosed by the authors: Ireland's Department of Agriculture, Food and the Marine. Bord Iascaigh Mhara.

## **Competing Interests**

The authors declare there are no competing interests.

## **Author Contributions**

- Alexandre Tahar conceived and designed the experiments, performed the experiments, analyzed the data, contributed reagents/materials/analysis tools, prepared figures and/or tables, authored or reviewed drafts of the paper, approved the final draft.
- Alan M. Kennedy and Richard D. Fitzgerald conceived and designed the experiments, performed the experiments, contributed reagents/materials/analysis tools, authored or reviewed drafts of the paper, approved the final draft.

• Eoghan Clifford and Neil Rowan conceived and designed the experiments, contributed reagents/materials/analysis tools, authored or reviewed drafts of the paper, approved the final draft.

## **Data Availability**

The following information was supplied regarding data availability: The raw data are provided as Supplemental Files.

## **Supplemental Information**

Supplemental information for this article can be found online at http://dx.doi.org/10.7717/ peerj.5281#supplemental-information.

## REFERENCES

- American Public Health Association (APHA). 2005. Standard methods for the examination of water and wastewater. Washington D.C.: Association APH.
- American Public Health Association (APHA). 2012. In: Rice EW, Bridgewater L, eds. *Standard methods for the examination of water and wastewater*. 22nd edition. Washington, D.C.: American Public Health Association, American Water Works Association, Water Environment Federation.
- Aubin J, Tocqueville A, Kaushik S. 2011. Characterisation of waste output from flowthrough trout farms in France: comparison of nutrient mass-balance modelling and hydrological methods. *Aquatic Living Resources* 24:63–70 DOI 10.1051/alr/2011008.
- Boaventura R, Pedro AM, Coimbra J, Lencastre E. 1997. Trout farm effluents: characterization and impact on the receiving streams. *Environmental Pollution* 95:379–387 DOI 10.1016/S0269-7491(96)00117-0.
- **Boyd C, Tucker C. 1998.** *Pond aquaculture water quality management.* Boston: Kluwer Academic Publishers, 700.
- Caramel B, Moraes M, Carmo C, Vaz-dos Santos A, Tabata Y, Osti J, Ishikawa C, Cerqueira M, Mercante C. 2014. Water quality assessment of a trout farming effluent, Bocaina, Brazil. *Journal of Water Resource and Protection* 6:909–915 DOI 10.4236/jwarp.2014.610086.
- **Donnely N. 2011.** Potential diversification opportunities for the land-based aquaculture sector in Northern Ireland European Union. *Available at http://www. aquacultureinitiative.eu/diversification%20report.pdf*.
- **EEC. 1978.** Council Directive 78/659/EEC of 18 July 1978 on the quality of fresh waters needing protection or improvement in order to support fish life—Repealed by Directive 2006/44/EC. *Available at https://publications.europa.eu/en/publication-detail/-/publication/4208a321-9981-4bee-93ae-3341c28735e2/language-en.*
- **Emerson K, Russo RC, Lund RE, Thurston RV. 1975.** Aqueous ammonia equilibrium calculations: effect of ph and temperature. *Journal of the Fisheries Research Board of Canada* **32**:2379–2383 DOI 10.1139/f75-274.

- **European Commission. 2000.** Directive 2000/60/EC of the European parliament and of the council establishing a framework for the community action in the field of water policy. *Available at https://www.eea.europa.eu/policy-documents/directive-2000-60-ec-of*.
- Garcia F, Kimpara JM, Valenti WC, Ambrosio LA. 2014. Emergy assessment of tilapia cage farming in a hydroelectric reservoir. *Ecological Engineering* 68:72–79 DOI 10.1016/j.ecoleng.2014.03.076.
- Guilpart A, Roussel J-M, Aubin J, Caquet T, Marle M, Le Bris H. 2012. The use of benthic invertebrate community and water quality analyses to assess ecological consequences of fish farm effluents in rivers. *Ecological Indicators* 23:356–365 DOI 10.1016/j.ecolind.2012.04.019.
- Hennessy MM, Wilson L, Struthers W, Kelly LA. 1996. Waste loadings from two freshwater Atlantic Salmon Juvenile farms in Scotland. *Water, Air, and Soil Pollution* 86:235–249 DOI 10.1007/BF00279159.
- **Hornund RW, Reed LD. 1990.** Estimation of average concentration in the presence of nondetectable values. *Applied Occupational and Environmental Hygiene* **5**(1):46–51.
- **Irish Department of Agriculture, Food and the Marine (DAFM). 2015.** *National strategic plan for sustainable aquaculture.* Dublin: Irish Department of Agriculture, Food and the Marine.
- Jokumsen A, Svendsen LM. 2010. Farming of Freshwater rainbow trout in Denmark. Lyngby: Technical University of Denmark, National Institute of Aquatic Resources. *Available at http://orbit.dtu.dk/files/6581106/219-10\_Farming-of-freshwaterrainbow-trout-in-denmark-v2.pdf*.
- Kırkağaç MU, Pulatsu S, Topcu A. 2009. Trout farm effluent effects on water sediment quality and benthos. *Clean Soil Air Water* **37**(**4**–**5**):386–391.
- Koçer MAT, Kanyılmaz M, Yılayaz A, Sevgili H. 2013. Waste loading into a regulated stream from land-based trout farms. *Aquaculture Environment Interactions* 3:187–195 DOI 10.3354/aei00059.
- Krause G, Brugere C, Diedrich A, Ebeling MW, Ferse SCA, Mikkelsen E, Pérez Agúndez JA, Stead SM, Stybel N, Troell M. 2015. A revolution without people? Closing the people-policy gap in aquaculture development. *Aquaculture* 447:44–55 DOI 10.1016/j.aquaculture.2015.02.009.
- Lalonde BA, Ernst W, Garron C. 2015. Chemical and physical characterisation of effluents from land-based fish farms in Atlantic Canada. *Aquaculture International* 23:535–546 DOI 10.1007/s10499-014-9834-y.
- Lazzari R, Baldisserotto B. 2008. Nitrogen and phosphorus waste in fish farming. *Boletim do Instituto de Pesca* 34:591–600.
- Martins CIM, Eding EH, Verdegem MCJ, Heinsbroek LTN, Schneider O, Blancheton JP, d'Orbcastel ER, Verreth JAJ. 2010. New developments in recirculating aquaculture systems in Europe: a perspective on environmental sustainability. *Aquacultural Engineering* **43**:83–93 DOI 10.1016/j.aquaeng.2010.09.002.

- **Neto RM, Nocko HR, Ostrensky A. 2015.** Environmental characterization and impacts of fish farming in the cascade reservoirs of the Paranapanema River, Brazil. *Aquaculture Environment Interactions* **6**:255–272 DOI 10.3354/aei00130.
- Noroozrajabi A, Ghorbani R, Abdi O, Nabavi E. 2013. The impact of rainbow trout farm effluents on water physicochemical properties of daryasar stream. *World Journal of Fish and Marine Sciences* 5:342–346.
- Papatryphon E, Petit J, Van Der Werf H, Kaushik S, Claver K. 2005. Nutrientbalance modeling as a tool for environmental management in aquaculture: the case of trout farming in france. *Environmental Management* 35:161–174 DOI 10.1007/s00267-004-4020-z.
- **Pulatsu S, Rad F, Köksal G, Aydın F, Benli AÇK, Topçu A. 2004.** The impact of rainbow trout farm effluents on water quality of karasu stream, Turkey. *Turkish Journal of Fisheries and Aquatic Sciences* **4**:9–15.
- Sindilariu PD, Brinker A, Reiter R. 2009. Factors influencing the efficiency of constructed wetlands used for the treatment of intensive trout farm effluent. *Ecological Engineering* 35:711–722 DOI 10.1016/j.ecoleng.2008.11.007.
- Sturrock H, Newton R, Paffrath S, Bostock J, Muir J, Young J, Immink A, Dickson M. 2008. Prospective analysis of the aquaculture sector in the EU—Part 2: characterisation of emerging aquaculture systems. In: (IPTS) IP, ed. JRC Scientific and Technical Reports—EUR 23409 EN/2. Luxembourg: Office for Official Publications of the European Communities.
- **Teodorowicz M. 2013.** Surface water quality and intensive fish culture. *Archives of Polish Fisheries* **21**:65–111 DOI 10.2478/aopf-2013-0007.
- **Tomasso JR. 1994.** Toxicity of nitrogenous wastes to aquaculture animals. *Reviews in Fisheries Science* **2**:291–314 DOI 10.1080/10641269409388560.
- True B, Johnson W, Chen S. 2004. Reducing phosphorus discharge from flow-through aquaculture I: facility and effluent characterization. *Aquacultural Engineering* 32:129–144 DOI 10.1016/j.aquaeng.2004.08.001.
- **US Environmental Protection Agency. 2003.** In: Croghan CW, Egeghy PP, eds. *Methods of dealing with values below the limit of detection using SAS.* Washington, D.C.: EPA.
- **Verdegem MCJ. 2013.** Nutrient discharge from aquaculture operations in function of system design and production environment. *Reviews in Aquaculture* **5**:158–171 DOI 10.1111/raq.12011.
- Všetičková L, Adamek Z, Rozkošný M, Sedláček P. 2012. Effects of semi-intensive carp pond farming on discharged water quality. *Acta Ichthyologica Et Piscatoria* 42:223–231.
- Yalcuk A, Pakdil N, Kantürer O. 2014. Investigation of the effects of fish farms in bolu (turkey) on aquatic pollution. *International Journal of Agricultural and Food Research* 3(1):1–13.
- Živić I, Marković Z, Filipović-Rojka Z, Živić M. 2009. Influence of a trout farm on water quality and macrozoobenthos communities of the receiving stream (Trešnjica River, Serbia). *International Review of Hydrobiology* 94:673–687 DOI 10.1002/iroh.200811137.





## Full Water Quality Monitoring of a Traditional Flow-Through Rainbow Trout Farm

Alexandre Tahar <sup>1,\*</sup>, Alan Kennedy <sup>2</sup>, Richard D. Fitzgerald <sup>3</sup>, Eoghan Clifford <sup>2,3</sup> and Neil Rowan <sup>1</sup>

- <sup>1</sup> Athlone Institute of Technology, Dublin Road, Athlone, County Westmeath, Ireland; nrowan@ait.ie
- <sup>2</sup> Civil Engineering, College of Engineering and Informatics, National University of Ireland, Galway, Ireland; michaelalan.kennedy@nuigalway.ie (A.K.); eoghan.clifford@nuigalway.ie (E.C.)
- <sup>3</sup> Ryan Institute, National University of Ireland, Galway, Ireland; richard.fitzgerald@nuigalway.ie
- \* Correspondence: atahar@ait.ie

Received: 10 May 2018; Accepted: 11 July 2018; Published: 17 July 2018



**Abstract:** Traditional freshwater rainbow trout farms are still popular in some European countries such as Poland, France and Ireland. These systems generally operate in flow-through configuration. The impact such production systems might have on water quality remains mostly unknown. The present study was set up to fulfil this objective of monitoring water quality on different fish farm locations in order to identify the impacts of the whole farm (comparison of farm inlet and outlet) and at pond scale in order to understand the water quality dynamics and to better understand the impact of multiple water reuse (water passes) in a given pond on water quality. In the absence of any sort of water treatment, an increase in the number of water passes was shown to create an increase in ammonium concentration along the farm. Finally, this traditional flow-through rainbow trout production system was revealed to almost be at its full carrying capacity with respect to internal water quality parameters and fish welfare. To increase fish production, some water treatment techniques (solid/liquid separation, nitrification) would have to be introduced in order to optimize the rearing water quality for fish growth and to minimize the release of pollutants in the receiving water to limit the impact on the environment.

**Keywords:** rainbow trout; traditional flow-through; pond-based; rearing water quality; environmental impact

#### 1. Introduction

With the wild fish catching capacities reached and the overall human population increase, fish produced through aquaculture is and will be more and more an important food source for human consumption in the near future [1,2]. Aquaculture production increased more than five-fold from 1990 to 2012, while the world capture fisheries increased by only 8% at the same time [3]. The future and current intensification of this activity is associated with more and more stringent environmental regulations to ensure a sustainable and environmentally friendly production process/model [4–6]. At European scale, freshwater fish farming is essentially regulated through the water framework directive (WFD), which aims at achieving good chemical and ecological status of the rivers [1,7,8].

The traditional freshwater fish farming industry generally operates using flow-through systems (FT) without any water treatment, the dissolved oxygen (DO) levels in the culture water being maintained by relatively high volumes of water abstracted and flowing through the fish farm. The oxygen loss due to both fish activity and sediments is generally compensated by mechanical aerators such as paddle-wheels. In traditional FT systems, this aeration allows the fish farmers to use the water several times in different ponds before release (multi-pass); however, the impact of
this water reuse on water quality and the potential accumulation of waste in the rearing water has not been clearly identified so far. When the amount of DO is too low in the fish farm inlet water, fish farmers generally use supplementary oxygenation systems such as oxygenation cones in order to maintain optimum DO levels. Furthermore, fish farming activity generates dissolved and solid wastes such as nutrients (nitrogen and phosphorus) and organic matter from fish excreta and uneaten feed. Therefore, if the effluent is discharged untreated, it could potentially impact the discharge water quality compared to make-up water, and as a consequence impact the receiving water quality [9–14].

In Ireland, a large proportion of the freshwater production is still carried out using traditional FT systems without, or with very limited, water treatment before discharge. To the best of our knowledge, there is no existing published data on the effluent quality and flow, allowing estimation of pollutants fluxes and the potential impact the fish farming activity might have on water quality in Ireland. Models could be employed for the evaluation of organic matter (OM), nitrogen and phosphorus fluxes from feed composition and fish data (stocking density, average fish mass, overall fish mass) by the nutritional method [15]. However, missing data on fish/feed (especially on feed composition, ingested feed quantity, digestibility coefficients and the proportion of uneaten feed must be available according to [15]) forced us to evaluate the different fluxes by the hydrobiological method (involving flow measurements and water quality analysis).

Traditional flow-through systems generally involve a multi-use (or multi-pass) of water, with water passing through different ponds with no water treatment before being released to the receiving water, although some reuse might involve a supplementary oxygenation step. The objective of this study was to assess the water quality at different farm levels (pond, 1-pass, 2-pass and 3 or more pass ponds, whole farm) of a traditional rainbow trout farm in Ireland. To that aim, the whole farm (inlet/outlet and strategic fish farm locations) was monitored during two consecutive dry days. The monitoring program included water physico-chemical quality (dissolved oxygen, turbidity, nitrate, ammonium, and pH) and flow measurement in order to determine the flow dynamic of the whole farm and its different ponds. Each parameter was monitored on an hourly basis at the different selected farm locations in order to study fluxes and to identify any diurnal variation.

#### 2. Results and Discussion

#### 2.1. Fish and Flow Parameters

The table below summarizes the data related to (i) the biomass for each pond (i.e., total biomass, number of fish, average fish weight and stocking densities) that were obtained from the fish farmers, and (ii) flow values that were measured for each monitored pond (Tables 1 and 2).

FISH											
	Volume	No. of Fish	Average Weight	Biomass	Stocking Density						
Pond	m <sup>3</sup>		g/Fish	kg	kg/m <sup>3</sup>						
11	121	3000	667	2000	17						
12	144	3069	558	1713	12						
13	147	12,000	170	2040	14						
$1_4$	153	7696	610	4695	31						
1 <sub>5</sub>	140	9736	800	7789	56						
16	159	12,000	170	2040	13						
1 <sub>7</sub>	122	5409	980	5301	43						
18	86	15,750	200	3150	37						
19	68	0	/	0	0						
110	56	10,500	260	2730	49						
1 <sub>11</sub>	44	10,500	260	2730	62						
1 <sub>12</sub>	41	0	/	0	0						
113		10,080		2400	73						
after harvest	33	nd	240	600	19						
114	25	10,080	220	2218	89						
1 <sub>15</sub>	32	8000	280	2240	70						
1 <sub>16</sub>	39	8000	220	1760	45						
1 <sub>17</sub>	36	8000	220	1760	49						
$1_{18}$	16	0	/	0	0						
21	592	6000	1600	9600	16						
22	207	6500	1300	8450	41						
23	256	11,041	1330	14,685	57						
32	301	1800	1100	1980	7						
3 <sub>3</sub>	435	1283	1400	1796	4						

**Table 1.** Outcome of fish data for the whole fish farm at the moment of the experiment. The figures 0, 1, 2 and 3 for ponds number are theoretical numbers of water uses (see Figure 7).

Table 2.	Outcome o	of flow o	data me	easured	on the	selected	fish	farm	locations	at the	e moment	of the	9
experime	ent. Standar	d devia	tion (St	. Dev.);	Minim	um (Min	.); Ma	iximu	m (Max.).				

			FLOW			
	Mean Value	St. Dev.	Min.	Max.	Number	
Pond	L/s	L/s	L/s	L/s		Comment
15	19.9		17.5	24.5	35	
1 <sub>13</sub>	7.3	0.1	7.3	7.4	10	
1 <sub>15</sub>	6.8	0.7	5.9	7.5	39	
	-1.8	0.2	-1.3	-2.1	11	$2_1$ outlet 1
$2_{1}$	35.2	2.3	29.8	37.2	42	$2_1$ outlet 2
	66.1	4.3	0.0	96.3	370	recirculation
2	6.0	0.8	4.4	7.7	23	$2_2$ outlet 1
42	34.5	0.8	33.2	35.7	21	$2_2$ outlet 2
23	76.1	7.8	57.6	148.5	473	Farm outlet 1
32	50.3	7.7	34.1	63.9	20	
33	58.1	21.5	0.0	112.7	565	Farm outlet 2

Table 1 shows the range of fish sizes being reared at the time of the experiment (170–1600 g/fish). Different stocking densities were also observed in the different ponds (4–89 kg/m<sup>3</sup>).

The flow measurements showed a flow dynamic slightly different than what was expected initially in terms of number of water passes. For example, a negative flow value was obtained between  $2_1$ and  $2_2$  ponds (Table 2). This might be a consequence of the recirculation pump in operation in  $2_1$  that would have applied a suction of the water from  $2_2$  (average recirculation flow of 66 L/s, see Table 2). This demonstrated that  $2_2$  was completely isolated from  $2_1$  and therefore isolated from any recirculated water. This is the reason why, contrary to initial expectations where it would have been a "3-pass pond", 2<sub>2</sub> was considered as a "2-pass pond", as it only received water from  $1_{8-12}$  ponds (i.e., "1-pass ponds"). Furthermore, we observed that the average flow from 2<sub>2</sub> to 2<sub>3</sub> pond was very limited (i.e., 6 L/s, see Table 2). This is why 2<sub>3</sub> pond was considered a "mostly 2-pass pond", as it received water from six "1-pass ponds" (i.e.,  $1_{13-18}$  ponds at 7 L/s each, see Table 2) and only one "2-pass pond" (i.e.,  $2_2$  pond at 6 L/s, see Table 2).

Temperature was monitored in different ponds during the experiment and water temperature was 8-9.5 °C in all ponds during the whole experiment.

#### 2.2. Impact of the Whole Fish Farm on Water Quality

The objective was to assess the overall impact of the farm on water quality (i.e., farm influent vs. effluent water quality). Farm inlet (before oxygenation, feeding channel, i.e.,  $0_0$ ) and farm outlet water quality were compared. Differential concentrations ( $\Delta_C(t)$ , see Equation (1)) were calculated between inlet and mean outlet concentrations between the two outlets (considering the two farm outlets from  $2_3$  and  $3_3$  ponds had similar flows, see Table 2) monitored during the two days of experiment for all the studied parameters (see Section 3.2).

The figure below presents an outcome of the differential values obtained between the fish farm outlets and inlet for five parameters (i.e., DO, turbidity, nitrate (NO<sub>3</sub>–N), ammonium (NH<sub>4</sub>–N) and pH) during the two days of monitoring (Figure 1).



**Figure 1.** Whole farm differential values between outlets and inlet of dissolved oxygen (DO), turbidity (in NTU (Nephelometric Turbidity Unit)), nitrate, ammonium and pH during the monitoring program. Vertical dash lines represent feeding times.

Globally, negative  $\Delta_C$  were observed for DO, pH and NO<sub>3</sub>–N for the duration of the monitoring. Such results were expected for DO and pH due to O<sub>2</sub> consumption by fish and sediments and the CO<sub>2</sub> production leading to a pH drop, as observed on other studies [16,17]. However, positive  $\Delta_C$  were observed for turbidity and NH<sub>4</sub>–N, meaning that the farm activity "created" solids and NH<sub>4</sub>–N. Except for NO<sub>3</sub>–N and DO,  $\Delta_C(t)$  appeared relatively stable all along the experiment and no diurnal effect was observed at this overall farm scale. For DO, higher absolute  $\Delta_C(t)$  (i.e., down to almost –40%) were observed during the days compared to night times (i.e., about –20%); this might be due to the higher O<sub>2</sub> consumption by fish during the day due to the occurrences of feedings and higher fish activity [18].

The table below summarizes these data by providing average and associated standard deviation of the overall farm  $\Delta_C(t)$  values (Table 3). This presentation of the overall farm mean  $\Delta_C$  shows a global picture of the influence of the fish farm on water quality. Associated fluxes calculated as presented in Section 3.4 (Equations (2) and (3)) are also presented.

	D	0	Turb.	NO <sub>3</sub> -N	NH <sub>4</sub> -N	pН
	%	mg/L	NTU	mgN/L	mgN/L	
mean difference	-27.39	-3.09	1.19	-0.36	0.30	-0.39
st. dev.	6.05	0.67	1.44	0.72	0.04	0.07
min.	-38.10	-4.33	-1.10	-1.83	0.24	-0.50
max.	-19.40	-2.25	6.01	0.61	0.38	-0.30
count	24		16	15	21	11
flux (mg/s) *	NA	-413.9	159.3	-48.6	40.4	NA
flux $(g/h)$ *		-1489.9	573.5	-174.8	145.4	
flux (g/h/kg <sub>fish</sub> ) **		-0.018	0.007	-0.002	0.002	
flux (mg/h/kg <sub>fish</sub> ) **		-18.37	7.07	-2.16	1.79	

**Table 3.** Summary of the overall farm differential data between outlets and inlet. Turbidity (Turb.); nitrate (NO<sub>3</sub>–N); ammonium (NH<sub>4</sub>–N); not applicable (NA).

\* fluxes calculated at farm scale using the average cumulated outlet flow rate of 134 L/s (see Table 2); \*\* fluxes calculated considering a total biomass reared on farm of 81,094 kg (see Table 1).

One of the parameters most affected by the farm was the DO saturation rate, with an average global "loss" of about 27% between farm inlet and outlet despite supplementary aeration/oxygenation steps on the farm (see Figure 7). Considering the total biomass and the outlet flows measured, mean DO F value was  $-18.4 \text{ mgO}_2/\text{h/kg}_{fish}$ .

An increase in turbidity values from farm inlet to outlet was observed with an average positive  $\Delta_{\rm C}$  of about 1.2 Nephelometric Turbidity Unit (NTU). This value appears relatively low compared to values measured at the outlets of some ponds (up to 20 NTU), and this may be due to the sedimentation of suspended solids in ponds, which would have lowered the release of suspended solids at the farm outlet. A peak in turbidity increase (i.e.,  $\Delta_{\rm C}(t)$  of 6 NTU) was observed at the beginning of the first night of the monitoring (Figure 1) which could be due to fish metabolism and the release of faeces a few hours after the afternoon feeding [16–18]. In terms of flux, average F value was 7.1 NTU/h/kg<sub>fish</sub>.

A global average drop in NO<sub>3</sub>–N concentrations of about 0.36 mgN/L between farm inlet and outlet was observed (Table 3). As shown on Figure 1, and regarding the relatively high standard deviation associated to the average differential value for this parameter (i.e., 0.72 mgN/L), a high variability of NO<sub>3</sub>–N  $\Delta_C$  during the experiment (i.e., range –1.83–+0.61) was observed. A negative value of average flux of –2.2 mgN/h/kg<sub>fish</sub> was obtained.

As expected, NH<sub>4</sub>–N appeared to be "produced" by the farm, with a positive average  $\Delta_C$  of 0.30 mgN/L (see Table 3). This represents a significant increase regarding the average farm inlet concentration of about 0.30 mgN/L (see Supplementary Materials). This  $\Delta_C$  remained stable all along the experiment, with a relatively narrow range observed (0.24–0.38 mgN/L). Therefore, the release of NH<sub>4</sub>–N by the farm seemed not be affected by light and feeding regimes. It is noticed that this apparent NH<sub>4</sub>–N "production" might be an artefact due to the overall pH drop of 0.39 units (Table 3) that would have changed the ammonia equilibrium through the ionized form, NH<sub>4</sub>–N. An average F value of 1.8 mgN/h/kg<sub>fish</sub> was obtained.

An average pH decrease of about 0.39 pH units between farm inlet and outlet was observed (Table 3) that might a consequence of a CO<sub>2</sub> production by fish metabolism [16–18]. This  $\Delta_C$  remained stable all along the experiment with a relatively narrow range observed (-0.50–-0.30 pH units, see Table 3).

#### 2.3. Comparison of the Water Quality from Inlet/1-Pass/2-Pass/3 Pass Ponds

The objective of this part of the study was to identify trends between the water quality of the farm inlet, 1-pass, 2-pass and 3-pass waters in order (i) to check if fish farm inlet water impacted water quality flowing through the farm and (ii) to identify potential trends in water quality parameters throughout the farm.

According to the flow measurements performed, only  $2_2$  and  $2_3$  ponds were considered as "2-pass ponds" (see Table 2 and Section 2.1). Ponds  $3_2$  and  $3_3$  were considered as "at least 3-pass ponds"

5 of 19

because, as explained before, they received some recirculated water from  $2_1$  (see Table 2 and Section 2.1). It is noticed that  $2_1$  pond was kept alone for this analysis because it had the specificity to receive directly recirculated water from  $1_{1-7}$ . This made the number of water passes of this pond difficult to evaluate from the flow measurements performed.

Average values for inlets (after supplementary oxygenation, i.e.,  $1_{13}$  and  $1_{15}$  inlets), 1-pass ( $1_{13}$  and  $1_{15}$  outlets), 2-pass (or "mostly 2-pass ponds", i.e.,  $2_2$  and  $2_3$  outlets) and "at least 3-pass ponds" (i.e.,  $3_2$  and  $3_3$  outlets),  $2_1$  outlet and associated standard deviation were calculated. The two "1-pass ponds" monitored (i.e.,  $1_{13}$  and  $1_{15}$ ) were considered separately because of the  $1_{13}$  pond harvesting that occurred after 24 h into the monitoring period. For more simplicity, in the following paragraphs the terms "2-pass ponds" and "3-pass ponds" will be employed instead of "mostly 2-pass ponds" and "at least 3-pass ponds" and "overall results for all the different pond categories (number of water passes) and for all the monitored parameters are presented on the figure below (Figure 2).



**Figure 2.** Average evolution of water quality (i.e., dissolved oxygen saturation rate, turbidity, nitrate, ammonium and pH) at the farm inlets and in "1-pass", "2-pass", "3-pass" and 2<sub>1</sub> ponds outlets. Vertical dash lines represent feeding times.

#### 2.3.1. Dissolved Oxygen

As expected, over the whole farm, the most oxygenated waters were the fish farm inlet waters, with saturation values in the range 85–100%. A higher DO saturation rate was measured during day time compared to night time (average DO values of 96% and 84%, respectively). These higher DO values during the day might be a consequence of the plants and phytoplankton photosynthesis occurring with sunlight and producing O<sub>2</sub> [17]. The same trend was observed at the inlets of  $1_{13}/1_{15}$  ponds after supplementary oxygenation by the two oxygenation cones located in the inlet channels  $0_2$  and  $0_3$  (see Figure 7) and that were in operation continuously during the whole monitoring program. However, the impact of the oxygenation cones on the level of DO saturation of the water appeared to be negligible during day time (i.e., 2–3% higher in the oxygenated water (i.e.,  $1_{13}/1_{15}$  ponds inlets) than in the inlet water from the inlet channel  $0_0$ , see Supplementary Materials). This impact was revealed to be higher at night time with a difference between oxygenated and non-oxygenated water up to 10% (i.e., 84% and 94% for non-oxygenated and oxygenated water respectively, see Supplementary Materials).

The monitoring of the three "1-pass ponds" showed that a systematic drop in the DO level was observed after every feeding event. This drop appeared to occur immediately after feeding. This would coincide with an increase in fish activity and in the metabolic demands associated with digestion [18].

Interestingly, the "3-pass ponds" did not show the least oxygenated water, with values in the range 65–85% of DO saturation at their outlets. According to the flow measurement performed (see Table 1), it was demonstrated that those ponds were fed more or less directly by recirculated water (i.e., water from  $2_1$  pond). This demonstrated the efficiency of applying recirculation/supplementary oxygenation and the associated impact on DO saturation rate observed all along the farm from  $2_1$  pond to the  $3_3$  farm outlet (see Figure 7).

The least oxygenated ponds were the "2-pass ponds", with values in the range 45–65%. These values were in the same range than the ones observed for "1-pass ponds" water from  $1_{15}$  (i.e., 50–70%), demonstrating the ability of paddle-wheel aerators to compensate for the loss of oxygen that occurred in those "2-pass ponds" due to fish/sediments consumption. Furthermore, the two "2-pass ponds" were not fed in any way by recirculated water (aeration of these ponds relied only on paddle-wheel aerators, see Table 2 and Section 3.1), and that could explain the relative low oxygen saturation levels observed in those ponds compared to the "3-pass ponds". Among the two "2-pass ponds",  $2_3$  pond had a higher DO saturation at its outlet than  $2_2$  pond during the whole monitoring period (see Supplementary Materials); this might be due to the differences in aeration systems from one pond to another (i.e., 2<sub>2</sub> being aerated with two paddle-wheel aerators and 2<sub>3</sub> with only one paddle-wheel, see Figure 7). This observation would confirm the ability of paddle-wheel aerators in compensating for the DO loss due to fish and sediments. Furthermore,  $2_1$  and  $3_2$  ponds outlets had similar DO profiles during the experiment (see Supplementary Materials). Thus, the aeration systems in operation in  $3_2$  (i.e., two paddle-wheels see Figure 7) were efficient enough to maintain the DO saturation levels higher than 70%. However, it was observed that DO levels were generally lower in  $3_3$  than in  $2_1/3_2$ ponds showing that the aeration system (i.e., one paddle-wheel, see Figure 7) was not efficient enough to maintain the same DO level in this pond.

#### 2.3.2. Turbidity

As expected, the lowest turbidity levels and variability were observed for the farm inlet waters. However, the comparison of the different pond classes is not straightforward. The potential sedimentation of solids along the farm might be a potential explanation of the difficulty to analyze the results for this parameter [16,18]. The relatively low turbidity values measured (a few NTU) and the potentially high uncertainty associated with turbidity measurements at these low levels might be another potential explanation. However, the same peak of turbidity was observed for the two "1-pass ponds" ( $1_{13}$  and  $1_{15}$ ) on the first day of monitoring (at about 8 h monitoring time) and the same general turbidity profiles were observed for those two ponds (Figure 2). The fact that the turbidity measurements were repeatable from those two ponds with similar fish characteristics (i.e., same total

biomass and average fish weight, see Table 1) indicates that the turbidity measurements performed were accurate and reliable. The greater variability in turbidity was observed for the 1-pass ponds outlets (i.e.,  $1_{13}$  and  $1_{15}$ ). This may be due to the sedimentation of suspended solids all along the farm that would have smoothed the associated turbidity profiles [16,18].

#### 2.3.3. Nitrate

The same profile was observed for NO<sub>3</sub>–N concentrations at any farm level for the duration of the monitoring. The general increase of concentration observed at farm inlet was also observed at each level of water passes. However, the ponds associated to the highest levels of passes (i.e., "3-pass ponds" and  $2_1$ ) were associated to the lowest NO<sub>3</sub>–N concentrations (i.e., in the range 25–29 mgN/L) while the farm inlet concentrations were in the range 26–32 mg/L. It is noteworthy that the "3-pass ponds" and  $2_1$  pond (all fed with recirculated water) showed similar NO<sub>3</sub>–N profile. This demonstrated that, as for the DO profile,  $2_1$  pond could not be considered a "2-pass pond" for the NO<sub>3</sub>–N profile. Additionally, the "2-pass ponds" were not affected in terms of the NO<sub>3</sub>–N concentration profile by the fish farming activity. This was highlighted by the comparison with the farm inlet NO<sub>3</sub>–N concentrations that were observed to be in similar range than 2-pass ponds all along the experiment (range of 2-pass ponds 27–32 mgN/L).

#### 2.3.4. Ammonium

A clear distinction between  $NH_4$ –N levels at the farm inlets, "1-pass", "2-pass" and "3-pass ponds" outlets was observed. As expected, the higher the number of passes is, the higher the  $NH_4$ –Nconcentration. This demonstrates the progressive accumulation of  $NH_4$ –N across the whole farm due to fish excretion and/or to progressive dissolution of sediments [16,18].

Farm inlet NH<sub>4</sub>–N concentrations were in the range 0.20–0.34 mgN/L.

The "1-pass pond" outlets' NH<sub>4</sub>–N concentrations were in the range 0.30–0.45 mgN/L (before  $1_{13}$  harvest). It is noteworthy that almost the exact same NH<sub>4</sub>–N profile was obtained at the outlet of both "1-pass ponds" (i.e.,  $1_{13}$  before harvest and  $1_{15}$ ); then, lower values were observed after  $1_{13}$  harvest during the second day of monitoring. This highlighted both (i) the good quality and reproducibility of the data generated regarding the use of the multisensor probes in terms of NH<sub>4</sub>–N monitoring and (ii) the impact of fish in NH<sub>4</sub>–N production in a pond.

The "2-pass ponds" average NH<sub>4</sub>–N outlet concentrations were in the range 0.47–0.62 mgN/L, which is a higher level than the ones observed for "1-pass ponds". A great variability between the values measured for the two "2-pass ponds" considered (i.e.,  $2_2$  and  $2_3$ ) was observed; the NH<sub>4</sub>–N concentrations were indeed systematically higher for  $2_2$ , by about 0.1 mgN/L. This could be due to the different receiving waters for those two ponds ( $2_2$  received water from the five  $1_{8-12}$  "1-pass ponds" (not monitored during this study) and  $2_3$  pond received water from the six  $1_{13-18}$  "1-pass ponds"). This could also be due to the different biomass reared in both  $2_2$  and  $2_3$  ponds (see Table 1).

The "3-pass" and 2<sub>1</sub> ponds NH<sub>4</sub>–N outlet concentrations were in the range 0.49–0.81 mgN/L, which was the highest range recorded over the whole farm. This observation is in agreement with both (i) the expected NH<sub>4</sub>–N accumulation across the whole farm and (ii) the water recirculation without any nitrification step to oxidize NH<sub>4</sub>–N (into NO<sub>2</sub>–N and NO<sub>3</sub>–N) leading to an NH<sub>4</sub>–N accumulation within the recirculation loop. This underpinned the impact of recirculation on the 2<sub>1</sub> pond water quality. It is noticed that the 2<sub>2</sub> pond outlet NH<sub>4</sub>–N concentrations were in the same range that those measured at "3-pass pond" outlets.

#### 2.3.5. pH

As expected, an impact of the farm on pH values was observed with a general decrease of pH across the whole farm, likely due to the CO<sub>2</sub> production by fish metabolism [17].

The inlet mean values were in the range 8.2–8.4 pH units with lower values generally observed at night. The "1-pass pond" outlets' pH values were measured in the range 7.9–8.0 (before harvest for  $1_{13}$ 

pond). It is interesting to observe values up to 8.2 pH units after harvest at the  $1_{13}$  pond outlet (similar to the values observed at the inlet). This highlights the impact of fish on pH variation. The "2-pass ponds", "3-pass ponds" and  $2_1$  pond outlets pH values were in the range 7.8–7.95 and 7.8–7.9 and 7.7–7.8 pH units, respectively. The lowest values were therefore observed for the recirculated water from the  $2_1$  pond. This again demonstrated the impact of recirculation on water quality. This could be due to the recirculation loop that could have brought back water with low pH into this pond. As mentioned before, this could explain the high NH<sub>4</sub>–N concentration measured in this pond.

Of note is that the ponds with the highest number of water passes had the lowest pH values and the highest  $NH_4$ –N concentrations. At low pH (i.e., <8), and at the temperatures measured during this monitoring program, the ionized form of ammonia ( $NH_4$ –N, relatively non-toxic to fish) is highly predominating compared to the toxic unionized form ( $NH_3$ –N). Therefore, the highest  $NH_4$ –N concentrations observed in the high number of passes ponds could be just an artefact due to pH variation and not due to an accumulation of fish wastes.

#### 2.4. Impact of 1-Pass Ponds on Water Quality

Three different "1-pass ponds" were intensively monitored during the experiment (i.e.,  $1_5$ ,  $1_{13}$  and  $1_{15}$ , see Figure 7). As for the whole farm,  $\Delta_C(t)$ , calculated as explained in Section 3.4 and using Equation (1), are presented. Fluxes, calculated as explained in Section 3.4 (Equations (2) and (3)), are also presented. The flux values will be specifically discussed in Section 2.5 below.

#### 2.4.1. 1<sub>15</sub> Pond

The figure below presents the pattern of  $\Delta_C$  between the outlet and inlet of  $1_{15}$  pond during the monitoring program (Figure 3).



**Figure 3.** Pond 1<sub>15</sub> differential values between outlet and inlet for dissolved oxygen and pH (**A**) and for turbidity, nitrate and ammonium (**B**) during the monitoring program. Vertical dash lines represent feeding times.

As for the whole farm scale, different trends were observed, depending on the parameter considered at this pond scale. Negative  $\Delta_C$  were obtained for pH, DO and NO<sub>3</sub>–N. Positive values were obtained for turbidity and NH<sub>4</sub>–N. For NO<sub>3</sub>–N, a general increase in  $\Delta_C$  from the beginning to the end of the experiment (from –1.5 to about 0 mgN/L) was observed. The  $\Delta_C$  for NH<sub>4</sub>–N was higher during night times. Dissolved oxygen showed diurnal variation also, with higher consumption (or lower  $\Delta_C$ ) during day time with drops likely to be due to feedings [19]. No clear trends or diurnal variation were observed for the other parameters.

The  $\Delta_C$  data are summarized in the table below, which provides average values, associated standard deviation and the calculated fluxes for each parameter (Table 4). Fluxes data, as described in Section 3.4 are also presented.

	Ι	00	Turb.	NO <sub>3</sub> -N	NH <sub>4</sub> –N	
	%	mg/L	NTU	mgN/L	mgN/L	- рн
mean difference	-33.50	-3.80	1.85	-0.50	0.13	-0.38
st. dev.	4.88	0.53	3.15	0.50	0.03	0.04
min.	-45.50	-5.04	0.18	-1.43	0.08	-0.40
max.	-27.30	-2.92	13.32	0.08	0.18	-0.30
count	24		16	15	21	10
flux (mg/s) *	NA	-26.6	12.9	-3.5	0.9	NA
flux $(g/h)$ *		-95.7	46.5	-12.7	3.2	
flux $(g/h/kg_{fish})$ **		-0.043	0.021	-0.006	0.001	
flux (mg/h/kg <sub>fish</sub> ) **		-42.74	20.76	-5.65	1.44	

**Table 4.** Summary of the 1<sub>15</sub> pond differential data and fluxes between outlets and inlet.

\* fluxes calculated at  $1_{15}$  pond scale (including the impact of fish and sediments) with the average flow rate of 7L/s (see Table 2); \*\* fluxes calculated considering a total biomass reared in  $1_{15}$  pond of 2240 kg (see Table 1).

#### 2.4.2. 1<sub>13</sub> Pond

As for  $1_{15}$  pond, the figure below presents the evolution of  $\Delta_C$  between outlet and inlet of  $1_{13}$  pond during all the monitoring program (Figure 4). It is noticed that this pond was partially harvested on day two of the experiment (after about 24 h of monitoring) (see Table 1).



**Figure 4.** Pond  $1_{13}$  differential values between outlet and inlet for dissolved oxygen and pH (**A**) and for turbidity, nitrate and ammonium (**B**) during the monitoring program. Vertical light dash lines represent feeding times.

As for the whole farm and the  $1_{15}$  pond, negative  $\Delta_C$  values were observed for NO<sub>3</sub>–N, pH and DO and positive ones for turbidity and NH<sub>4</sub>–N. Before harvest,  $\Delta_C$  for NH<sub>4</sub>–N were higher at night, confirming the observations made for  $1_{15}$  pond (see Section 2.4.1). After harvest, all  $\Delta_C$  profiles changed for all parameters and all the absolute values were lower. As expected, the effect of the fish activity in this pond was therefore reduced after harvesting, as the total biomass was reduced (see Table 1).

The  $\Delta_C$  data are summarized in the table below, which provides average values, associated standard deviation and the calculated fluxes for each parameter before and after harvest (Table 5). Fluxes data, as described in Section 2.4 and according to Equations (2) and (3) are also presented.

			Before	Harvest			After Harvest					
		DO	Turb.	NO <sub>3</sub> –N	NH <sub>4</sub> –N	T	Γ	00	Turb.	NO <sub>3</sub> –N	NH <sub>4</sub> -N	
	%	mg/L	NTU	mgN/L	mgN/L	рн	%	mg/L	NTU	mgN/L	mgN/L	рн
mean difference	-40	-4.55124	2.295	-0.37125	0.136364	-0.34286	-23.01	-2.607	0.561667	-0.06714	0.07	-0.225
st. dev.	6.685	0.708759	4.367466	0.590507	0.036952	0.053452	5.641208	0.630768	0.579738	0.568997	0.02357	0.05
min.	-51.8	-5.73	-0.06	-1.06	0.08	-0.4	-33.7	-3.83	-0.08	-0.85	0.02	-0.3
max.	-24.1	-2.87	14.52	0.86	0.19	-0.3	-16.8	-1.92	1.45	0.67	0.1	-0.2
count	14		10	8	11	7	10		6	7	10	4
flux (mg/s) *	NA	-31.8587	16.065	-2.59875	0.954545	NA	NA	-18.249	3.931667	-0.47	0.49	NA
flux $(g/h)$ *		-114.691	57.834	-9.3555	3.436364			-65.6964	14.154	-1.692	1.764	
$flux (g/h/kg_{fish}) **$		-0.04779	0.024098	-0.0039	0.001432			-0.10949	0.02359	-0.00282	0.00294	
flux (mg/h/kg <sub>fish</sub> ) **		-47.788	24.0975	-3.89813	1.431818			-109.494	23.59	-2.82	2.94	

**Table 5.** Summary of the 1<sub>13</sub> pond differential data and fluxes between outlets and inlet before and after harvest.

\* fluxes calculated at 1<sub>13</sub> pond scale (including the impact of fishes and sediments) with the average flow rate of 7L/s (see Table 1); \*\* fluxes calculated considering a total biomass reared in 1<sub>13</sub> pond of 2400 kg and 600 kg before and after harvest respectively (see Table 1).

The comparison of  $1_{15}$  and  $1_{13}$  ponds before harvest (similar total and average fish weight in those two ponds as presented in Table 1) showed very similar values, demonstrating the reproducibility of our approach. The effect of harvesting on the differential values appears very clear when results before and after harvest are compared. Mean DO consumption dropped from 40 to 23%, the mean pH difference dropped from 0.34 to 0.23 pH units, mean turbidity production dropped from 2.30 to 0.56 NTU and the mean NH<sub>4</sub>–N production dropped from 0.14 to 0.07 mgN/L.

It is noteworthy that despite the fact that the  $1_{13}$  pond was harvested and that only 600 kg of fish remained after harvest, the pond fluxes in terms of DO and NH<sub>4</sub>–N are not negligible. This underpins the potential impact of the sediments accumulated at the bottom of the tank might have on DO consumption or NH<sub>4</sub>–N production. The role of sediments on the residual  $\Delta_C$  values obtained after harvest will be discussed in Section 2.5.

#### 2.4.3. 15 Pond

The figure below presents the evolution of  $\Delta_C$  between outlet and inlet of  $1_{13}$  pond during the entire monitoring program (Figure 5).



**Figure 5.** Pond 1<sub>5</sub> differential values between outlet and inlet for dissolved oxygen and pH (**A**) and for turbidity, nitrate and ammonium (**B**) during the monitoring program. The inlet monitoring location was changed after 8 h of monitoring. Vertical dash lines represent feeding times.

This pond had three main particularities compared to  $1_{13}/1_{15}$  ponds regarding DO:

- (1) It was supplied with super-saturated water from the recirculation loop (average 140% saturation rate, see Supplementary Materials),
- (2) It had a paddle-wheel aerator running during the experiment (see Figure 7),
- (3) Bigger fish than in  $1_{13}/1_{15}$  ponds were reared (average weight of 800 g and 220–240 g respectively, see Table 1)

The table below presents a summary of the  $\Delta_C$  values observed for all parameters during the whole experiment (Table 6). The amount of data is lower for this pond because the initial inlet monitoring location had to be changed for another during the experiment as the initial one was observed to be unsuitable for monitoring.

	D	0	Turb.	NO <sub>3</sub> –N	NH <sub>4</sub> -N	рH
	%	mg/L	NTU	mgN/L	mgN/L	P
mean difference	-56.82	-6.41	1.49	0.08	0.18	0.01
st. dev.	4.42	0.49	1.87	0.22	0.05	0.04
min.	-65.50	-7.39	-2.57	-0.12	0.10	0.00
max.	-49.70	-5.48	4.58	0.49	0.27	0.10
count	17	17	10	11	16	8
flux (mg/s) *	NA	-128.1	29.8	1.6	3.7	NA
flux $(g/h)$ *		-461.2	107.1	5.8	13.2	
$flux (g/h/kg_{fish}) **$		-0.059	0.014	0.001	0.002	
flux (mg/h/kg <sub>fish</sub> ) **		-59.21	13.75	0.75	1.69	

Table 6.	Summary	of the $1_5$	pond	differential	data ar	nd fluxes	between	outlets	and i	inlet.
----------	---------	--------------	------	--------------	---------	-----------	---------	---------	-------	--------

\* fluxes calculated at  $1_5$  pond scale (including the impact of fishes and sediments) with the average flow rate of 20 L/s (see Table 2); \*\* fluxes calculated considering a total biomass reared in  $1_5$  pond of 7789 kg (see Table 1).

A relatively high DO consumption rate of  $59 \text{ mgO}_2/h/kg_{\text{fish}}$  compared to the ones found for  $1_{13}/1_{15}$  ponds (43 and 48 mgO<sub>2</sub>/h/kg<sub>fish</sub>, respectively; see Tables 3 and 4) was observed. The NH<sub>4</sub>–N production rate was found to be similar to what was found for the other "1-pass ponds" and for the whole farm (Table 3).

Contrary to the other "1-pass ponds" monitored (i.e.,  $1_{13}$  and  $1_{15}$ ), a paddle-wheel aerator was employed in this pond for supplementary aeration (Figure 7) and thus, had to be considered in the DO mass balance for this  $1_5$  pond. However, the impact of the paddle-wheel aerator in increasing DO levels in the pond could not be assessed.

These results regarding the higher DO consumption rate in  $1_5$  compared to  $1_{13}/1_{15}$  ponds was unexpected for at least two reasons:

- (1) The presence of a paddle-wheel aerator should have increased the DO amount received by the pond, and therefore decreased the overall consumption rate (because the calculation did not into account the O<sub>2</sub> input from the paddle-wheel aerator).
- (2) From the literature [16], bigger fish are supposed to consume less  $O_2$  by mass unit than smaller ones, and the comparisons of results obtained for  $1_5$  and  $1_{13}/1_{15}$  ponds appeared not to confirm this statement.

The different hypotheses that might explain these results could be:

- In relation to the very high initial DO saturation rate at the inlet of this 1<sub>5</sub> pond (i.e., 140% on average) that would have modified the O<sub>2</sub> liquid/air equilibrium and therefore could have caused a faster decrease in DO concentration within the pond (which would have caused the O<sub>2</sub> loss by liquid/air transfer higher and give a higher apparent consumption rate).
- In relation to sediment amount at the bottom of this pond that could be higher per square meter than in 1<sub>13</sub>/1<sub>15</sub> ponds that could have caused a higher sediment DO consumption rate for this 1<sub>5</sub> pond.

#### 2.5. Outcome on Fluxes

The figure below presents the mean F(t) values calculated from Equation (3) for (i) the three "1-pass ponds" monitored (i.e., 1<sub>5</sub>, 1<sub>13</sub> before and after harvest and 1<sub>15</sub> ponds) and (ii) the whole farm (Figure 6).



**Figure 6.** Mean fluxes calculated for DO, turbidity (**A**), NO<sub>3</sub>–N and NH<sub>4</sub>–N (**B**) at different farm levels (1-pass ponds and whole farm).

The differences in DO fluxes between  $1_3$  and the other "1-pass ponds" were already discussed in a previous Section (Section 2.4.3) and it was assumed that the higher fluxes observed in  $1_3$  pond were due to a higher initial DO saturation rate. It is confirmed here that  $1_{13}$  pond before harvest and  $1_{15}$  pond had similar DO fluxes, with F values being -48 and  $-43 \text{ mgO}_2/\text{h/kg}_{fish}$  respectively. The lower F value observed for the whole farm (i.e.,  $F = -18 \text{ mgO}_2/\text{h/kg}_{fish}$ ) can be explained by the supply of oxygen given by both aeration and oxygenation devices across the farm that would have lowered the apparent loss of  $O_2$  at farm scale. The highest F value observed for  $1_{13}$  pond after harvest (i.e.,  $-109 \text{ mgO}_2/\text{h/kg}_{fish}$ ) could be an artefact due the impact of sediments (fish faeces and uneaten feed) on the overall DO consumption. The same observation can be done for NH<sub>4</sub>–N flux with a very high F value observed for  $1_{13}$  after harvest compared to other ponds. It is noteworthy that no data were generated in order to quantify the potential impact of other compartments (plants, algae, phytoplankton, soil etc.) on the DO/NH<sub>4</sub>–N levels in ponds. However, no diurnal variation was observed in DO consumption after harvest in pond  $1_{13}$  (Figure 5); this observation would confirm that the impact of phytoplankton and plants was negligible compared to the impact of sediments and fish.

For turbidity, F values for  $1_{13}$  before and after harvest and for  $1_{15}$  were similar (i.e., 24.1, 23.6 and 20.8 NTU/h/kg<sub>fish</sub> respectively). The F value for  $1_5$  pond was slightly lower (i.e., 13.8 NTU/h/kg<sub>fish</sub>), which could be caused by the higher size of this pond and lower water velocity, which would enhance the settling capacities of this pond, increasing the amount of solids removed from water and thus the F value for turbidity. At farm scale, F value was even lower (i.e., 7.1 NTU/h/kg<sub>fish</sub>); this observation could be due to the settling capacity of the whole farm [16,18].

For NO<sub>3</sub>–N, a negative flux was observed for all ponds except 1<sub>3</sub> pond. This pond was the only one to receive some recirculated water which could have created this flux difference and this apparent NO<sub>3</sub>–N production by the pond by bringing back some low NO<sub>3</sub>–N content water into the loop.

#### 3. Materials and Methods

#### 3.1. Farm Description

The studied fish farm was an earth pond-based rainbow trout farm operating mostly in FT configuration with an additional recirculation loop involving a supplementary oxygenation step. The general layout of the fish farm and the different monitoring locations are presented on the figure below (Figure 7). Fish of different growth stages were distributed into the different ponds according to their weight. Fish were fed manually twice a day until apparent satiation with manufactured pellets (Efico Alfa from Biomar, Brande, Denmark, 41-44% protein, 25-28% lipid, 16-19% carbohydrate, 0.8-2.4% fiber, 4-7% ash, 0.9% total phosphorus, according to the supplier) of different size (pellets diameter) according to the size/weight of fish to be fed in each of the different ponds (see Table 1).



**Figure 7.** Fish farm general layout (A: paddle-wheel aerator, C: oxygen cone; O: high pressure oxygenator; M: physico-chemical monitoring location; F: water flow monitoring location). Some ponds (i.e.,  $1_9$ ,  $1_{12}$  and  $1_{18}$ ) were not stocked at the moment of monitoring, they are represented in grey. The figures 0, 1, 2 and 3 are theoretical numbers of water uses.

Supplementary oxygenation of the inlet water was achieved using oxygenation cones with supplementary aeration provided within the ponds by mechanical aerators (paddle-wheels) (Figure 7). Additionally, there was a partial recirculation of the water from  $2_1$  pond to  $0_1$  pond also involving a supplementary oxygenation step. It is noteworthy that this recirculation was applied without any water treatment system (i.e., neither solid removal nor nitrification step applied), hence this farm would still fall within the classification of "flow-through farm".

According to the farm layout and the flow dynamic we initially defined the expected number of water passes for each pond. For one given pond, the number of water passes was defined as the number of ponds the water flowed through. As an example, all the ponds labelled  $1_{1-18}$  were considered as 1-pass ponds.

#### 3.2. Water Quality and Flow Monitoring

Water quality parameters (pH, DO, NO<sub>3</sub>–N, NH<sub>4</sub>–N, temperature) were analyzed in situ using a multi-sensor probe (YSI, Xylem Inc., Tunbridge Wells, United Kingdom).

Flows were monitored in different farm locations (Figure 7) using ultrasonic flow meters (Nivus PCM4 portable ultrasonic flow measurement devices, Eppingen, Germany). Both  $2_3$  and  $3_3$  ponds outlets were monitored continuously and some other locations ( $1_5$ ,  $1_{13}$ ,  $1_{15}$ ,  $2_1$ ,  $2_2$  and  $3_2$  ponds outlets) were monitored at set intervals during the monitoring program in order to understand the whole farm dynamic and to be able calculate/estimate fluxes at (i) farm outlet, and (ii) within the farm. It is noteworthy that the assumption was made that farm inlet flow (not measured) was equal to the farm outlet flow and that no water loss occurred within the farm.

#### 3.3. Experimental Set-Up

The objective was to assess flows and water quality dynamic over the whole fish farm. As all the fish farm locations could not be monitored continuously due to lack of both time and human resources, choices of the farm locations to monitor were made in order to cover most of the farm operation and to fulfil the objectives presented in Section 1. To that aim, the farm locations monitored were the following:

- (1) The general inlet channel  $0_0$  as the fish farm inlet (i.e., before supplementary oxygenation by the cones, see Figure 7).
- (2) The inlets and outlets of three "1-pass ponds" (i.e., 1<sub>5</sub>, 1<sub>13</sub> and 1<sub>15</sub>, see Figure 7) that were chosen both to represent the diversity of the farm (i.e., different average fish weights, total biomass in

 $1_5$  and  $1_{13}/1_{15}$  ponds with larger fish in  $1_5$  as presented in Table 1) and to generate results from similar ponds in order to obtain a duplication of the results and confirm that the protocol set-up was accurate (i.e.,  $1_{13}$  and  $1_{15}$  with similar average fish weight and total biomass, see Table 1).

- (3) The outlets of all the other ponds (i.e., 2-pass, 3-pass and more) in order to identify the dynamic of the farm and to assess the influence of the number of water passes on water quality.
- (4) The recirculated water (from 2<sub>1</sub> to inlet channel 0<sub>1</sub>, see Figure 7) in order to assess the impact of the recirculation loop on water quality.

Due to limited personnel resources and time operational constraints, only the ponds outlets' water quality was monitored. For simplification purposes, outlet concentrations were considered as equal to those encountered within the pond themselves.

A period of two consecutive dry-weather days of monitoring was considered in order to avoid any effect on flows that would have been created by rain. It was checked that no rain event was observed upstream during the whole experiment and the week before.

#### 3.4. Data Treatment

Differential concentrations ( $\Delta_C(t)$ ) were calculated as being the difference between outlet and inlet concentrations for a given parameter at a given time. These concentrations were calculated using the following equation (Equation (1)):

$$\Delta_{C}(t) = C_{outlet}(t) - C_{inlet}(t) \tag{1}$$

where  $\Delta_C(t)$ , generally expressed in mg/L (except for turbidity and pH), represents the differential concentration of a given parameter at the given time *t*.

These  $\Delta_C(t)$  were calculated at (i) farm scale ( $C_{outlet}(t)$  being the mean value calculated for the two farm outlets (i.e.,  $2_3$  and  $3_3$ ) for a given time and  $C_{inlet}(t)$  the value measured at farm inlet (i.e.,  $0_0$ ) at the same time for a given parameter) in order to study the impact of the whole farm on water quality, and (ii) at pond scale in order to study the impact of each monitored pond on water quality and to assess the influence of water passes on water quality. The  $\Delta_C(t)$  were calculated for all monitored times and also averaged for the duration of the monitoring. Hence, a positive value for  $\Delta_C(t)$  would mean an increase in the concentration of a given parameter from inlet to outlet, and therefore a "production" by the farm (or by the pond considered); similarly, a negative  $\Delta_C(t)$  value would mean a decrease in the concentration of a given parameter from inlet to outlet and therefore a "consumption" by the farm (or by the pond considered).

From the obtained  $\Delta_C(t)$  values, flow values were used to calculate fluxes (f(t), generally expressed in mg/h, except for pH and turbidity) both at farm and pond scales for all the monitored parameters. The f(t) were calculated as follow (Equation (2)):

$$f(t) = \Delta_C(t) \times Q_{outlet}.$$
 (2)

where f(t) (mg/h) is the flux of a given parameter from the whole farm or from a given pond and  $Q_{outlet}$  (L/h) the flow at the outlet of the farm or of a given pond.

Those fluxes were converted using the total biomass in presence in each pond (see Table 1) to obtain fluxes in  $mg/h/kg_{fish}$ . according to the following equation (Equation (3)):

$$F(T) = \frac{f(t)}{m_{T,pond}}$$
(3)

With  $m_{T,vond}$  (kg<sub>fish</sub>), the total biomass in the pond considered.

All F(t) values were averaged for the farm and for the monitored ponds for all the monitored parameters to obtain comparable mean flux values between the different ponds monitored.

#### 4. Conclusions

A traditional flow-through rainbow trout farm was monitored extensively for the first time in Ireland. It is recalled that traditional flow-through systems still represent 90% of the trout production at the country scale today, and even if practices are evolving through recirculated aquaculture systems (RAS), this type of practice remains very common at the EU scale (France, Poland, Czech Republic, etc.). Therefore, the results and observations from this study are very relevant to the industry.

A combination of flow and water quality monitoring made it possible to understand the dynamics of water and its nutrient and dissolved oxygen content. To allow for a better understanding and analysis of the results, the concept of "multi water passes" was defined as the number of ponds the water flows through before release at the farm outlet. The flow measurements performed illustrated the difficulty in assigning a number of water passes to each pond, and the results obtained were not in agreement with the expected number of water passes (negligible or even negative flows observed at some farm locations). Thus, this study gives the impact and limits of multi water reuse on water quality and of the implementation of supplementary aeration/oxygenation, still practiced broadly in Ireland and in Europe. Furthermore, this study made it possible to attain proper and independent monitoring of the effluent water quality of the farm effluent and of the impact it might have on the receiving water quality.

The impact of water passes on water quality was identified, with a notable increase in NH<sub>4</sub>–N concentrations when the number of water passes increases. However, the loss in DO was globally compensated by the oxygenation/aeration technologies (oxygenation cones, recirculation loop and paddlewheel aerators) employed onsite. The role of sediments in DO consumption and NH<sub>4</sub>–N production was clearly identified as being substantial, and sediments could be responsible for a significant DO consumption, which has to be compensated by the operation of some oxygenation/aeration devices with their associated costs (energy, investment). It is, therefore, recommended for fish farmers still employing traditional flow-through systems to "de-sludge" their ponds more frequently in order to limit DO loss and NH<sub>4</sub>–N increase in the rearing water.

The industry has to produce more to be economically viable in the future [20]. However, it was shown during the present study that ammonium levels increased with the number of water passes, and that it could reach concentrations up to almost 1 mgN/L in ponds. This shows that increasing production in the same farm configuration would increase nutrients concentration even more, which could hamper production due to an expected impact on fish health from this concentration [21]. Thus, with regards to fish welfare, the outcomes of this study make it possible to identify areas and practices within the FT systems that show water quality levels close to threshold levels for fish welfare. From this, appropriate actions and practices can be implemented. Therefore, if this fish farm were to produce more, some water treatment processes (solid/liquid separation, nitrification) would have to be set up in order to both limit the build-up of nutrients in the rearing water and to reduce the nutrient release in the effluent to fulfil the WFD objectives. It also identifies performances of the industry with respect to the WFD, criteria thereby facilitating gap analysis for the implementation of water treatment technologies. Finally, this study also acts as a benchmark for the industry to define what improvements are required and develop a roadmap for the sustainable development of the freshwater aquaculture sector in Ireland and beyond (specifically for trout production).

**Supplementary Materials:** The following are available online at http://www.mdpi.com/2410-3888/3/3/28/s1, raw data excel files.

Author Contributions: Conceptualization, A.T. and R.D.F.; Data curation, A.T. and A.K.; Formal analysis, A.T., R.D.F. and E.C.; Funding acquisition, A.K. and R.D.F.; Investigation, E.C. and N.R.; Methodology, A.T. and A.K.; Project administration, A.K. and E.C.; Supervision, E.C. and N.R.; Validation, A.K., E.C. and N.R.; Visualization, R.D.F.; Writing—original draft, A.T.; Writing—review & editing, A.T.

**Funding:** The authors express gratitude to Ireland's Department of Agriculture, Food and the Marine for supporting this study through the MOREFISH project and Bord Iascaigh Mhara for supporting the EcoAqua project.

**Acknowledgments:** The authors would like to acknowledge Ronan Cooney (MSc. NUIG) and Conor Behan (MSc. NUIG) for their active participation to the monitoring campaigns.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- 1. Guilpart, A.; Roussel, J.-M.; Aubin, J.; Caquet, T.; Marle, M.; Le Bris, H. The use of benthic invertebrate community and water quality analyses to assess ecological consequences of fish farm effluents in rivers. *Ecol. Indic.* **2012**, *23*, 356–365. [CrossRef]
- 2. Donnely, N. *Potential Diversification Opportunities for the Land-Based Aquaculture Sector in Northern Ireland;* European Union: Brussels, Belgium, 2011; p. 144.
- 3. Krause, G.; Brugere, C.; Diedrich, A.; Ebeling, M.W.; Ferse, S.C.A.; Mikkelsen, E.; Pérez Agúndez, J.A.; Stead, S.M.; Stybel, N.; Troell, M. A revolution without people? Closing the people-policy gap in aquaculture development. *Aquaculture* **2015**, *447*, 44–55. [CrossRef]
- 4. Jokumsen, A.; Svendsen, L.M. *Farming of Freshwater Rainbow Trout in Denmark*; Technical University of Denmark, National Institute of Aquatic Resources, DTU Aqua: Charlottenlund, Denmark, 2010; p. 50.
- 5. Sturrock, H.; Newton, R.; Paffrath, S.; Bostock, J.; Muir, J.; Young, J.; Immink, A.; Dickson, M. *Prospective Analysis of the Aquaculture Sector in the EU—Part 2: Characterisation of Emerging Aquaculture Systems;* JRC—European Commission: Ispra, Italy, 2008.
- 6. Martins, C.I.M.; Eding, E.H.; Verdegem, M.C.J.; Heinsbroek, L.T.N.; Schneider, O.; Blancheton, J.P.; d'Orbcastel, E.R.; Verreth, J.A.J. New developments in recirculating aquaculture systems in Europe: A perspective on environmental sustainability. *Aquac. Eng.* **2010**, *43*, 83–93. [CrossRef]
- 7. European Commission (EC). *Directive 2000/60/EC of the European Parliament and of the Council Establishing a Framework for the Community Action in the Field of Water Policy;* European Commission, Ed.; European Commission: Ispra, Italy, 2000.
- Aubin, J.; Tocqueville, A.; Kaushik, S. Characterization of waste output from flow-through trout farms in France: Comparison of nutrient mass-balance modelling and hydrological methods. *Aquat. Living Resour.* 2011, 24, 63–70. [CrossRef]
- 9. Garcia, F.; Kimpara, J.M.; Valenti, W.C.; Ambrosio, L.A. Emergy assessment of tilapia cage farming in a hydroelectric reservoir. *Ecol. Eng.* **2014**, *68*, 72–79. [CrossRef]
- 10. Lalonde, B.A.; Ernst, W.; Garron, C. Chemical and physical characterization of effluents from land-based fish farms in Atlantic Canada. *Aquac. Int.* **2015**, *23*, 535–546. [CrossRef]
- 11. Lazzari, R.; Baldisserotto, B. Nitrogen and phosphorus waste in fish farming. *Bol. Inst. Pesca* 2008, 34, 591–600.
- 12. Sindilariu, P.D.; Brinker, A.; Reiter, R. Factors influencing the efficiency of constructed wetlands used for the treatment of intensive trout farm effluent. *Ecol. Eng.* **2009**, *35*, 711–722. [CrossRef]
- 13. Verdegem, M.C.J. Nutrient discharge from aquaculture operations in function of system design and production environment. *Rev. Aquac.* 2013, *5*, 158–171. [CrossRef]
- 14. Caramel, B.; Moraes, M.; Carmo, C.; Vaz-dos-Santos, A.; Tabata, Y.; Osti, J.; Ishikawa, C.; Cerqueira, M.; Mercante, C. Water quality assessment of a trout farming effluent, Bocaina, Brazil. *J. Water Resour. Prot.* **2014**, *6*, 909–915. [CrossRef]
- 15. Roque d'Orbcastel, E.; Person-Le Ruyet, J.; Le Bayon, N.; Blancheton, J.P. Comparative growth and welfare in rainbow trout reared in recirculating and flow through rearing systems. *Aquac. Eng.* **2009**, *40*, 79–86. [CrossRef]
- 16. Boyd, C.; Tucker, C. *Pond Aquaculture Water Quality Management*; Kluwer Academic Publishers: Boston, MA, USA, 1998; p. 700.
- 17. Lawson, T.B. *Fundamentals of Aquacultural Engineering*; Kluwer Academic Publishers: Boston, MA, USA, 1995; p. 362.
- 18. Hargreaves, J.A.; Tucker, C.S. *Managing Ammonia in Fish Ponds*; Southern Regional Aquaculture Center: Stoneville, MS, USA, 2004.
- Alsop, D.; Wood, C. The interactive effects of feeding and exercise on oxygen consumption, swimming performance and protein usage in juvenile rainbow trout (*Oncorhynchus mykiss*). J. Exp. Biol. 1997, 200, 2337. [PubMed]

- 20. Irish Department of Agriculture, Food and the Marine (DAFM). *National Strategic Plan for Sustainable Aquaculture*; Irish Department of Agriculture, Food and the Marine: Dublin, Ireland, 2015.
- 21. Haywood, G.P. Ammonia toxicity in teleost fishes: A review. In *Canadian Technical Report of Fisheries and Aquatic Sciences;* Department of Fisheries and Oceans–Fisheries Research Branch: Nanaimo, British Colombia, Canada, 1983; Volume 1177.



© 2018 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).

Contents lists available at ScienceDirect



Case Studies in Chemical and Environmental Engineering

journal homepage: www.sciencedirect.com/journal/case-studies-in-chemicaland-environmental-engineering

Case Report



Neil J. Rowan<sup>a,b,\*</sup>, Antoine Fort<sup>a,b</sup>, Emer A. O'Neill<sup>a,b</sup>, Eoghan Clifford<sup>c</sup>, Marcel Jansen<sup>d</sup>, Markus Helfert<sup>e</sup>, Damien Toner<sup>f</sup>, Julie Maguire<sup>g</sup>, Brijesh Tiwari<sup>h</sup>

<sup>a</sup> Technological University of the Shannon, Midlands Campus, University Road, Athlone, Co Westmeath, N37 HD68, Ireland

<sup>b</sup> Empower Eco Sustainability Hub, Technological University of the Shannon, East Campus, University Road, Athlone, Co Westmeath, N37 F6D7, Ireland

<sup>d</sup> School of Biological, Earth and Environmental Sciences & Environmental Research Institute, University College Cork, Distillery Fields, North Mall, Cork, T23 N73K,

<sup>f</sup> Bord Iascaigh Mhara, Crofton Road, Dún Laoghaire, Co Dublin, A96 E5A0, Ireland

<sup>g</sup> Indigo Rock Marine Research Centre, Gearhies, Bantry, Co Cork, P75 AX07, Ireland

h Teagasc Food Research Centre, Moorepark, Moorepark West, Fermoy, Co Cork, P61 C996, Ireland

#### ARTICLE INFO

Keywords: Bioeconomy Peatland Commercial demonstrator Aquaculture Change of land use Sustainability

#### ABSTRACT

This case study describes the novel development and demonstration of commercial, green, bio-based products using a peatland based recirculating integrated multi-trophic aquaculture (IMTA) system in the Irish Midlands. This site enables the transition from traditional peat harvesting for energy generation towards alternative sustainable employment. The system effectively addresses sustainable in-land freshwater aquaculture development. It also demonstrates value-chain products at scale for new feeds arising from the fish waste-stream by exploiting cascades from the fish culture waste-stream (bio-fertiliser) for cultivating duckweed and macroalgae. These plants can then be bio-refined and valorised to produce new products. The system also provides a circular demonstrator site that will facilitate industry and entrepreneurs to develop and test new innovations and ideas. By providing this open-access site to support companies in testing, financial constraints such as access to specialist equipment and technical expertise will be off-set thus enabling entrepreneurs and industry to develop new commercial products at scale. Additionally, the outputs from this system will help address and inform several United Nations sustainable development goals.

#### 1. Introduction

The steady growth in global population coupled with the commensurate demand for edible protein has contributed to an agricultural intensification that presents challenges for the sustainability of livestock farming [1]. Specifically, there is a need to reduce the production and release of agriculture waste products to protect our fragile environment [2]. Conversely, new opportunities are arising to valorise agricultural waste, for example; through the co-production of different bio-based products from raw materials (biomass) that can be used for the production of food, feed and biofuels [3]. Such a circular bioeconomy is aligned with a renewed focus on rural resilience and regional development across Europe [1] that seeks to promote zero-carbon changes in land use, including rewetting peatland [4]. The bioeconomy offers new sustainable and climate-resilient pathways for economic development aiding a fair and just transition (JT) from a fossil fuel driven economy [5]. However, there remains pressing technical and financial challenges that includes scalability and delivering clear messaging where solutions can be met through pilot commercial demonstration activities. Consequently, this case study describes the use of a peatland-based, recirculating integrated multi-trophic aquaculture (IMTA) system in the Irish JT Territory to overcome some of these hurdles through organic fish-production and culturing duckweed as a protein crop. This does not just include technical advances, but also

\* Corresponding author. Technological University of the Shannon, Midlands Campus, University Road, Athlone, Co Westmeath, N37 HD68, Ireland. *E-mail address:* neil.rowan@tus.ie (N.J. Rowan).

https://doi.org/10.1016/j.cscee.2024.100763

Received 17 March 2024; Received in revised form 13 May 2024; Accepted 16 May 2024 Available online 16 May 2024 2666-0164/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).





<sup>&</sup>lt;sup>c</sup> Civil Engineering, School of Engineering, University of Galway, University Road, Galway, H91 TK33, Ireland

Ireland

e School of Business Maynooth, Innovation Value Institute, Maynooth University, Maynooth, Co Kildare, W23 F2H6, Ireland

embraces, for example, multi-actor stakeholder engagement using a quadruple helix approach [6,7] that integrates industry, policy-makers, society and academics.

#### 2. IMTA process at Mount Lucas wind farm in Irish peatlands

The IMTA site at Mount Lucas in the Irish midlands is a former commercial peat-harvesting site that has operated several years as a successful organic fish farm using the principle of integrated multitropic aquaculture i.e., where two or more organisms are cultured or farmed together [8]. Building on that core principle, this case study describes the development and future trajectory of the site. Ziegler et al. [9] reported that key challenges affecting the development of peatland-based innovation include the lack of shared systems of data generation for meta-analysis, corporate strategy, risk-mitigation and business disruption. In Ireland, there is nearly 100,000 ha. degraded peatland used for commercial peat cutting thus presenting scope to develop and demonstrate an IMTA approach to deliver solutions at scale for the bioeconomy. The IMTA farm is located in the Irish Midlands on a wind farm  $(53^{\circ}17'3'' \text{ N} - 7^{\circ}11'45'' \text{ W})$  and is a cut-away rewetted glacial till site, from which peat was previously extracted [9,10]. Four split (pill) aquaculture ponds were dug-out on the infertile cut-away substrate and used for culturing of rainbow trout (Oncorhynchus mykiss) and European perch (Perca fluviatilis) [10]. The system is circulatory and fishponds are connected by means of channels to an algae and duckweed lagoon that serves as a natural fish effluent waste treatment system (see Fig. 1). Energy is locally generated by wind turbines. Water is rarely taken into (only done so to compensate for evaporation) or released from (only during times of excessive rainfall, the system to a nearby bog river.

# 3. Use of freshwater macroalgae for nutrient recycling and biomass production

Early studies have revealed the presence of a large variety of algae in

the IMTA system at Mount Lucas [11]. As part of the IMTA process, the fish cultivation waste stream will be used to cultivate macroalgae in land-based tanks. Freshwater macroalgae, unlike their marine counterparts, are a relatively overlooked group of plants that, to date, have not been fully harnessed at scale for their nutrient removal potential or their biomass composition in aquaculture systems. Several species have been shown to be indicators for eutrophication through their sustained growth in nutrient-rich freshwater bodies [12]. Such characteristics will be leveraged in Mount Lucas to efficiently recycle the fish effluent (which mostly contains ammonium and phosphates) into useable biomass. Macroalgae have distinct advantages over freshwater microalgae, namely their containment within the site, ease of harvest, and higher resistance to bacterial contamination in the environment [13].

Determining the diversity and performance of local freshwater macroalgae species under high nutrient load and/or a seasonal pattern has yet to be characterized. Screening for growth and nutrient removal potential of Irish freshwater macroalgal species, particularly Chlorophyta such as Cladophora, Oedogonium, Monostroma, Pithophora and freshwater *Ulva* genera will advance system performance [8]. Some species of these genera have been shown to grow in freshwater at very low to no salinity conditions and sustain high growth in high-nutrient environments [14,15]. Growth and nutrient uptake kinetics will be interpreted in the context of continuous monitoring of nutrient levels using sensors. An on-site nursery informing production of fresh macroalgae biomass in land-based tanks will be deployed with focus on determining seasonality of species to ensure an all year-round process. Harvested macroalgae will be bio-refined for bioactives such as proteins, carbohydrates, lipids and antioxidants. For example; it has already been shown that Oedogonium represents a high-quality source of proteins for animal feed [16], while Cladophora biomass has been shown to possess a wide range of possible applications from feed, nutraceuticals and composites to bio-stimulants [17].



**Fig. 1.** Aerial view of the full IMTA system. Culture ponds (blue squares), reservoir (orange square), treatment lagoon (green square) and neighbouring wetland (yellow triangle) are visible. Airlift (white squares) and paddlewheel (orange circles) locations have been indicated. Peatland/bog river (blue dashed lines) has been included along with the intake point (grey circle) and overflow point (grey rectangle). Red arrows indicate water flow in and around the system. Wind turbine (WT) 19 which provides all electrical needs for the farm can be seen in the bottom left-hand corner. (For interpretation of the references to colour in this figure legend, the reader is referred to the Web version of this article.)

# 4. Duckweed for sustainable valorisation of aquaculture wastewater

Duckweed (Lemnaceae) species have gained considerable attention as a sustainable source of high-quality nutrition, biofuels, and pharmaceuticals, as well as effective organisms for the phytoremediation of wastewaters [18]. A protein content of up to 45 % makes duckweed biomass nutritionally interesting as an ingredient for animal feeds, or for human use [18]. Previous work conducted on the pilot IMTA site at Mount Lucas has demonstrated the potential to generate high biomass yields from duckweed, with yields in excess of 30 tonnes [19] containing some ten tonnes of protein. Such yields are substantial in comparison with those of traditional crops such as soybean that rarely exceed five tonnes per year. The duckweed protein content was found to range between 20 % and 25 % [19], but there is scope to increase the protein content to about 40 % [20], through an increase in nitrogen in the water column. Thus, integrated management involving fish-husbandry, nutrient monitoring and duckweed manipulation can increase the economic case for the cultivation of duckweed as a novel protein crop. The strength of the IMTA is that it comprises of an integrated system in which fish, duckweed and algae species engage in mutually beneficial, as well as antagonistic interactions which need to be aligned with stakeholder interests (i.e., protein extraction technology) in accordance with priorities for a JT.

Moreover, outdoor duckweed cultivation has gained in popularity in recent decades, but it can be challenging to optimise, to control operationally and to integrate into a more comprehensive system such as an IMTA. Substantial challenges remain with respect to scaling up smallscale systems (up to several m<sup>2</sup>) to semi-commercial, large-scale systems. This includes understanding the relevant importance of the diffusion and mixing processes in facilitating plant nutrient supply to the thin surface layer of the floating duckweed plant. Optimising recirculating flow through technology, as described in this IMTA peatland model, coupled with sensor supports systems with artificial intelligence (AI) and machine-learning growth models, and monitoring of growth using an UAV, can maximise biomass yields. The duckweed performance data will be used to infer performance modelling of the farm and identify positive and negative drivers of duckweed growth in-situ (e.g., nutrient levels, flow rates, weather conditions), but also to calculate the capacity of the duckweed system to manage water nutrient levels that are safe for diverse fish species. Furthermore, such monitoring will facilitate accurate timing of autonomous harvesting, optimised to maximise yield. To avoid harvesting becoming a bottleneck to commercial exploitation, a SWOT analysis of key activities, including biomass cultivation, harvesting, biomass pre-treatment and protein extraction, will be undertaken.

#### 5. Conversion systems, bio-refinery and valorisation

Bio-refinery concepts comprise of an integration of processes to make high-value-added products such as food or feed ingredients (e.g., proteins, fibres), biomaterials, nutraceuticals (e.g., bioactives), and lowervalue-market products (e.g., biogas). The Mount Lucas IMTA system addresses production of macroalgal and duckweed with a biorefinery approach that will yield marketable products without producing unused waste or side streams (Fig. 2). The key challenges associated with the development of sustainable zero-waste value chains are; (1) post-harvest interventions for biomass; (2) on-site preservation; (3) pre-treatment technologies; and (4) robust, energy-efficient conversion systems (Fig. 3). For industry, two aspects of biomass processing methods can be considered. One is on-site preservation using ensiling. The more preferred but energy-intensive method currently employed is hot air drying. Currently, most large-scale biomass drying is carried out using conventional hot air dying (10–15 h. at 50°C) without pre-treatments. Macroalgal and duckweed biomass will need to be intensively dried post-harvest in order to reduce their moisture content from 70-80 % to



Fig. 2. Zero waste approach for Biorefineries'.

8–10 % ahead of further fractionation. From a cost perspective, it is preferable to avoid processing such as intensive drying, which is expensive to run, especially when drying large volumes of biomass that has a very high moisture content (>70 %), such as macroalgal and duckweed biomass. However, the application of alternative novel drying techniques, such as microwave-assisted and ultrasound-assisted drying, can improve drying efficiencies and product quality. Alternatively, establishing bio-refineries from fresh biomass will limit the degradation of valuable unstable bioactive compounds such as polyphenols; however, this puts a strain on the producer, as transporting fresh biomass is logistically challenging, expensive and inefficient. In an ideal situation, the bio-refinery processing plant would be located close to the biomass production site, as is the vision for the IMTA system.

One of the key bottlenecks of bio-refinery has been the separation of some fractions without wasting other fractions through simple, costeffective and scalable processes with low-energy requirements [21]. Hence, key steps in establishing a zero-waste bio-refinery for the Mount Lucas site can be adapted from IEA Bioenergy (https://www.ieabioene rgyreview.org/) (Table 1). Novel extraction and processing technologies play an important role in establishing energy-efficient bio-refineries by reducing dependencies on solvents and improving extraction efficiencies while preserving desired functional properties. The application of a range of new technologies for valorisation of biomass has been reviewed extensively [22]. Table 2 outlines the key features of novel technologies for the extraction of compounds of interest. One of the best strategies for improving extraction efficiencies is to develop a combination of sequential extraction technologies to maximise the synergies [23]. Such sequential extraction of ingredients (proteins, fibres and bioactives) has been established for macroalgae [24]. The zero-waste potential of macroalgal bio-refinery has been realised in the production of several products for food and pharmaceutical applications [25], and the potential to integrate advances in bio-refinery technologies with a commercially viable business has been highlighted [26].

#### 6. From innovation to sustainable business ideas

The pilot IMTA demonstrator site will facilitate industry and entrepreneurs to develop and test innovations and ideas matching business canvas model. Environmental sustainability (including risks, threats and bottlenecks) of new products and value chains, demonstrated on-site, will be determined using life cycle assessment (LCA). Appropriate LCA



Fig. 3. Valorisation of biomass.

#### Table 1

Key steps for developing energy efficient zero waste biorefineries.

- 1 Determining chemical composition and physicochemical properties of target biomass (e.g., macroalgae, duckweed)
- 2 Define the target bio-products and physicochemical properties required for potential applications (e.g., agri-food, nutraceuticals, energy)
- 3 Develop an inventory of technologies or any interventions required to produce targeted bio-products (e.g., pre-treatments, processing technologies, solvents)
- 4 Inventory of input materials (e.g., solvents, energy, biomass)
- 5 Data collection for each unit operation (e.g., conversion, fractionation, processing
- 6 Determine the chemical compounds to be transformed via pre-treatment/processes (e.g., side streams generated)
- 7 Carry out cost-benefit analysis (LCA/LCCA) to meet targets (e.g., economic, environmental and social indicators)

#### Table 2

Comparison of selected extraction techniques.

	Hydrodynamic Cavitation	Ultrasound cavitation	Microwave	Pulsed electric field	High pressure processing	Enzymatic processes	Chemical processes
Scalability Operating cost	Medium Medium High	Low Medium Medium	Medium Medium Medium	Medium Medium Medium	High High High	Low High Medium	Medium Low Medium
Energy requirement	Low	Low	Medium	Medium	High	Low	Low
Residue					Particulates	Enzymatic residue	Chemical residue
Harsh condition			High temperature		High pressure		
Selectivity	Low	Low	Low	High	Low	High	High

and life-cycle costing analysis (LLCA) will be conducted for value chains across the IMTA, supporting optimal engagement with stakeholders/ beneficiaries with a commercial market orientation [27]. Ecological and ecotoxicological assessments that also embrace the impact of weather variance and climate change will be conducted in accordance with the methods of O'Neill et al. [28]. A go-to-market strategy will assist the transition from pilot ideas to market-ready innovations using appropriate business analysis and modelling tools (e.g., SWOT, PESTELE, PM-CANVAS). Innovation studies thus far at this IMTA site (such as aquaculture and protein harvesting from duckweed) [8,18,28,29] have implemented proven methodologies such as the 'innovation sweet-spot' i.e., the optimal intersection at which feasibility (what is technically possible), viability (what can be sustainable for a business) and desirability (what users or customers need and want) converge, leading to successful and impactful innovations [30].

#### 7. Summary

The development of a fully monitored and characterised IMTA site in the peatland at Mount Lucas presents a timely opportunity to demonstrate high-value bio-based products at scale. This peatland recirculatory site will be informed by an integrated multi-actor stakeholder approach that will help identify and overcome technical and economic challenges for viable new products and services that embrace appropriate change of land use. The value stream generated from the fish culture waste stream presents many exciting business opportunities that will be balanced by conducting an appropriate environmental LCA. Fish

production (freshwater aquaculture) in the rewetted peatland is an important activity for food security. However, an on-site hatchery and mesocosm are needed to increase nutrients in the waste stream so that sufficient amounts of biomass are produced for duckweed and macroalgae bio-refinery activities, leading to high-value products and services. This IMTA model provides an open-access site for supporting companies in the testing and development of new green products. Additionally, financial constraints in terms of access to specialist equipment and technical expertise are off-set. This IMTA site will operate at the critical interface between top-down government policies (and strategies) for informing effective bioeconomy activities to meet end-user/beneficiary needs and addressing a fair and JT for communities pivoting to lowcarbon economies. Current and future outputs from this IMTA site will help address and inform several Unite Nations Sustainable Development Goals (UNSDGs) including; No Poverty (SDG 1), Zero Hunger (SDG 2), Good Health and Wellbeing (SDG 3), Responsible Consumption and Production (SDG 12), and Life Below Water (SDG 14).

#### **CRediT** authorship contribution statement

Neil J. Rowan: Writing – review & editing, Writing – original draft, Visualization, Methodology, Funding acquisition, Conceptualization. Antoine Fort: Writing – review & editing, Writing – original draft, Methodology, Investigation. Emer A. O'Neill: Writing – review & editing, Writing – original draft, Methodology, Data curation. Eoghan Clifford: Writing – review & editing, Writing – original draft, Methodology. Marcel Jansen: Writing – review & editing, Writing – original draft, Methodology, Funding acquisition, Conceptualization. **Markus Helfert:** Writing – review & editing, Writing – original draft, Methodology, Funding acquisition. **Damien Toner:** Writing – review & editing, Writing – original draft, Methodology. **Julie Maguire:** Writing – review & editing, Writing – original draft, Methodology. **Brijesh Tiwari:** Writing – review & editing, Writing – original draft, Funding acquisition, Conceptualization.

#### Declaration of competing interest

The authors declare no conflict of interest.

#### Data availability

Data will be made available on request.

#### Acknowledgements

The authors acknowledge funding support from the Department of Agriculture, Food and the Marine (2022 DAFM PSS125 'BioDMJT' project).

#### References

- R. Ravindran, S. Koopmans, J.P.M. Sanders, H. McMahon, J. Gaffey, Production of green biorefinery protein concentrate derived from perennial ryegrass as an alternative feed for pigs, Cleanroom Technol. 3 (2021) 656–669, https://doi.org/ 10.3390/cleantechnol3030039.
- [2] Department of Housing, Local Government and Heritage, The Nitrates Directive, 2023. https://www.gov.ie/en/publication/b87ad-nitrates-directive. accessed 11.28.23.
- [3] M.L. Colgrave, S. Dominik, A.B. Tobin, R. Stockmann, C. Simon, C.A. Howitt, D. P. Belobrajdic, C. Paull, T. Vanhercke, Perspectives on future protein production, J. Agric. Food Chem. 69 (2021) 15076–15083, https://doi.org/10.1021/acs.jafc.1c05989.
- [4] N. Sakuntaladewi, D. Rachmanadi, D. Mendham, T.W. Yuwati, B. Winarno, B. T. Premono, S. Lestari, A.Ramawati Ardhana, K. Budiningsih, D.C. Hidayat, M. Iqbal, Can we simultaneously restore peatlands and improve livelihoods? Exploring community home yard innovations in utilizing degraded peatland, Land 11 (2022) 150, https://doi.org/10.3390/land11020150.
- [5] N.J. Rowan, R. Pogue, Editorial overview: green new deal era current challenges and emerging opportunities for developing sustaining and disruptive innovation, Curr. Opin. Environ. Sci. Health 22 (2021) 100294, https://doi.org/10.1016/J. COESH.2021.100294.
- [6] N.J. Rowan, O. Casey, Empower Eco multiactor HUB: a triple helix 'academiaindustry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework, Curr. Opin. Environ. Sci. Health 21 (2021) 100254. https://doi.org/10.1016/J.COESH.2021.100254.
- [7] N.J. Rowan, C.M. Galanakis, Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: quo Vadis? Sci. Total Environ. 748 (2020) 141362 https://doi.org/ 10.1016/J.SCITOTENV.2020.141362.
- [8] E.A. O'Neill, G. Fehrenbach, E. Murphy, S.A. Alencar, R. Pogue, N.J. Rowan, Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: case study from the Republic of Ireland, Sci. Total Environ. 851 (2022) 158392, https://doi.org/10.1016/j.scitotenv.2022.158392.
- [9] V. Stejskal, S. Paolacci, D. Toner, M.A.K. Jansen, A novel multitrophic concept for the cultivation of fish and duckweed: a technical note, J. Clean. Prod. 366 (2022) 132881, https://doi.org/10.1016/J.JCLEPRO.2022.132881.
- [10] R. Ziegler, W. Wichtmann, S. Abel, R. Kemp, M. Simard, H. Joosten, Wet peatland utilisation for climate protection – an international survey of paludiculture innovation, Clean Eng. Technol. 5 (2021) 2666–7908, https://doi.org/10.1016/j. clet.2021.100305.
- [11] E.A. O'Neill, V. Stejskal, E. Clifford, N.J. Rowan, Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production –

a case study from the Republic of Ireland, Sci. Total Environ. 706 (2020) 136044, https://doi.org/10.1016/j.scitotenv.2019.136044.

- [12] E.A. O'Neill, A.P. Morse, N.J. Rowan, Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: implications and opportunities for advancing research and disruptive innovation post COVID-19 era, Sci. Total Environ. 819 (2022) 153073, https://doi.org/10.1016/J.SCITOTENV.2022.153073.
- [13] Y.I. Gubelit, Opportunistic macroalgae as a component in assessment of eutrophication, Diversity 14 (2022) 1112, https://doi.org/10.3390/d14121112.
- [14] L.N. Nguyen, L. Aditya, H.P. Vu, A. Hasan Johir, L. Bennar, P. Ralph, N.B. Hoang, J. Zdarta, L.D. Nghiem, Nutrient removal by algae-based wastewater treatment, Curr. Pollut. Rep. 8 (2022) 369–383, https://doi.org/10.1007/s40726-022-00230x.
- [15] R.J. Lawton, A.J. Cole, D.A. Roberts, N.A. Paul, R. De Nys, The industrial ecology of freshwater macroalgae for biomass applications, Algal Res. 24 (2017) 486–491, https://doi.org/10.1016/j.algal.2016.08.019.
- [16] R.C. Rajak, S. Jacob, S. Kim, A holistic zero waste biorefinery approach for macroalgal biomass utilization: a review, Sci. Total Environ. 716 (2020) 137067, https://doi.org/10.1016/j.scitotenv.2020.137067.
- [17] M.J. Vucko, A.J. Cole, J.A. Moorhead, J. Pit, R. De Nys, The freshwater macroalga Oedogonium intermedium can meet the nutritional requirements of the herbivorous fish Ancistrus cirrhosus, Algal Res. 27 (2017) 21–31, https://doi.org/ 10.1016/j.algal.2017.08.020.
- [18] M. Munir, R. Qureshi, M. Bibi, A. Mahmood Khan, Pharmaceutical aptitude of Cladophora: a comprehensive review, Algal Res. 39 (2019) 101476, https://doi. org/10.1016/j.algal.2019.101476.
- [19] N.E. Coughlan, E. Walsh, P. Bolger, G. Burnell, N. O'Leary, M. O'Mahoney, S. Paolacci, D. Wall, M.A.K. Jansen, Duckweed bioreactors: challenges and opportunities for large-scale indoor cultivation of Lemnaceae, J. Clean. Prod. 336 (2022) 130285, https://doi.org/10.1016/j.jclepro.2021.130285.
- [20] S. Paolacci, V. Stejskal, D. Toner, M.A.K. Jansen, Wastewater valorisation in an integrated multitrophic aquaculture system; assessing nutrient removal and biomass production by duckweed species, Environ. Pollut. 302 (2022) 119059, https://doi.org/10.1016/J.ENVPOL.2022.119059.
- [21] R. Chakrabarti, W.D. Clark, J.G. Sharma, R.K. Goswami, A.K. Shrivastav, D. R. Tocher, Mass production of Lemna minor and its amino acid and fatty acid profiles, Front. Chem. 6 (2018) 479, https://doi.org/10.3389/fchem.2018.00479.
- [22] K.W. Chew, J.Y. Yap, P. Loke Show, N.H. Suan, J.C. Juan, C. Ling, D.-J. Lee, J.-S. Chang, Microalgae biorefinery: high value products perspectives, Bioresour. Technol. 229 (2017) 53–62, https://doi.org/10.1016/j.biortech.2017.01.006.
- [23] V. Ummat, S.P. Sivagnanam, G. Rajauria, C. O'Donnell, B. Kumar Tiwari, Advances in pre-treatment techniques and green extraction technologies for bioactives from seaweeds, Trends Food Sci. Technol. 110 (2021) 90–106, https://doi.org/10.1016/ j.tifs.2021.01.018.
- [24] A.E. Nilsson, K. Bergman, L.P. Gomez Barrio, E.M. Cabral, K. Tiwari, Life cycle assessment of a seaweed-based biorefinery concept for production of food, materials, and energy, Algal Res. 65 (2022) 102725, https://doi.org/10.1016/j. algal.2022.102725.
- [25] J. Sadhukhan, S. Gadkari, E. Martinez-Hernandez, K.S. Ng, M. Shemfe, E. Torres-Garcia, J. Lynch, Novel macroalgae (seaweed) biorefinery systems for integrated chemical, protein, salt, nutrient and mineral extractions and environmental protection by green synthesis and life cycle sustainability assessments, Green Chem. 21 (2019) 2635–2655, https://doi.org/10.1039/C9GC00607A.
- [26] K. Balina, F. Romagnoli, D. Blumberga, Seaweed biorefinery concept for sustainable use of marine resources, Energy Proc. 128 (2017) 504–511, https:// doi.org/10.1016/j.egypro.2017.09.067.
- [27] N.J. Rowan, N. Murray, Y. Qiao, E. O'Neill, E. Clifford, D. Barceló, D.M. Power, Digital transformation of peatland eco-innovations ('Paludiculture'): enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products and services, Sci. Total Environ. 838 (2022) 156328, https://doi.org/ 10.1016/J.SCITOTENV.2022.156328.
- [28] E.A. O'Neill, M. McKeon Bennett, N.J. Rowan, Peatland-based innovation can potentially support and enable the sustainable development goals of the United Nations: case study from the Republic of Ireland, Case Stud. Chem. Environ. Eng. 6 (2022) 100251, https://doi.org/10.1016/J.CSCEE.2022.100251.
- [29] N.J. Rowan, The role of digital technologies in supporting and improving fishery and aquaculture across the supply chain – quo Vadis? Aquac. Fish 8 (2023) 365–374, https://doi.org/10.1016/J.AAF.2022.06.003.
- [30] E. O'Neill, V. Stejskal, S. Paulacci, Quo vadis Development of a novel peatlandbased recirculating aquaculture multi-trophic pond system (RAMPS) in the Irish midlands with a global orientation, Case Studies Chem. Environ. Eng. 9 (2024) 100748. https://doi.org/10.1016/j.cscee.2024.100748.

Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/scitotenv



# Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance – A comparative case study from the Republic of Ireland



### Emer A. O'Neill<sup>a,b,\*</sup>, Neil J. Rowan<sup>a,b</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co. Westmeath, Ireland
<sup>b</sup> Empower Eco Sustainability Hub, Boora, Co. Offaly, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- Traditional monitoring of aquaculture outputs using physicochemical parameters has limited efficacy.
- Complementary use of algae supports environmental monitoring of aquaculture.
- Duckweed supports and improves efficacy of aquaculture wastewater treatment.
- Algae are a potentially rapid and sensitive bioindicator of aquaculture water quality.
- Algae are a potential early warning tool for assessing impacts of climate change in aquaculture.

#### ARTICLE INFO

Article history: Received 2 July 2021 Received in revised form 16 August 2021 Accepted 17 August 2021 Available online xxxx

Editor: Damia Barcelo

Keywords: Algae Bioindicator Ecotoxicology Climate change Freshwater aquaculture Sustainability



### ABSTRACT

Aquaculture is one of the fastest growing food producing industries globally, providing ~50% of fish for human consumption. However, the rapid growth of aquaculture presents a range of challenges including balancing environmental impact that can be influenced by variations in climatic conditions. Monitoring of physicochemical parameters is traditionally used to evaluate aquaculture output quality; however, this approach does not indicate the cumulative ecotoxicological effects on receiving waters. Specifically, this case study investigated the relationship between measuring traditional physicochemical parameters and the health of the alga Pseudokirchneriella subcapitata in order to evaluate the potential ecotoxicological effects of freshwater aquaculture on the receiving aquatic ecosystem in the Irish midlands. This constituted the first 2-year longitudinal study conducted in 2018 and 2019 that reports on the efficacy of using algae as a natural bioindicator to monitor and assess freshwater aquaculture wastewater from a traditional flow-through fish farm producing Eurasian Perch (Perca fluviatilis); monitoring was compared over a same six-month period in the same location each year. Findings demonstrated significant differences between the two monitoring periods when using P. subcapitata for assessing the quality of aquaculture intake (P = 0.030) and output (P = 0.039). No stimulatory effects were observed during 2019 unlike >50% rates experienced the previous year. These observations coincided with changes in climatic conditions whereby the 2018 period experienced extended levels of drought; whereas non-drought conditions were observed during 2019. Findings suggest that reliance upon traditional monitoring techniques may not provide

\* Corresponding author at: Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co. Westmeath, Ireland. *E-mail address:* e.oneill@research.ait.ie (E.A. O'Neill).

sufficient robustness or versatility to address emerging issues, such as extremes in climate variance, which may influence the future intensive sustainability of freshwater aquaculture. This research supports the complementary use of *P. subcapitata* as a rapid and simple early-warning bioindicator for measuring aquaculture output quality on receiving aquatic ecosystems.

© 2021 Elsevier B.V. All rights reserved.

#### 1. Introduction

The depletion of wild capture fishery practices has resulted in the rapid development of aquaculture (Han et al., 2019) making it the fastest growing food producing industry worldwide (Ottinger et al., 2016; O'Neill et al., 2019, 2020). According to the FAO (2018), aquaculture now accounts for ~50% of fish produced for human consumption: this figure is expected to rise to ~62% by 2030 (Fredricks et al., 2015: Liu et al., 2017). The dramatic increase in aquaculture production is attributed to over exploitation of wild fisheries that are now at their maximum sustainable yields, along with increased consumer demand for fish (Tahar et al., 2018a, 2018b). Farmed fish is rich in protein and is also a more efficient protein utilisation and feed conversion source than other animals destined for protein production (Tschirner and Kloas, 2017). However, despite its numerous advantages, the rapid increase in aquaculture production has resulted in the emergence of several issues within the industry that include limitations in water and space, increased incidences of disease and increased environmental concerns (Ngo et al., 2016; Troell et al., 2017; Han et al., 2019; O'Neill et al., 2019). Stenevik and Sundby (2007) have also indicated that variations in climatic conditions have demonstrated substantial effects on increases as well as decreases in stocking densities; therefore, the success of fish stock assessment depends to a large extent on the ability to predict impacts climate change has on the dynamics of aquatic ecosystems. These treats have hindered the sustainable development and expansion of the industry (Han et al., 2019).

The ecological importance of algae have received consideration in studies focusing on natural approaches to wastewater remediation in freshwater aquaculture (Naughton et al., 2020) including potential influence of climate variance on process performance (O'Neill et al., 2019). Previous researchers have also noted the potential of algal communities to exhibit many attributes as biological indicators of spatial and temporal environmental change (Omar, 2010); additionally, microalgae have been reported as potentially useful monitoring quality of water bodies (Zaghloul et al., 2020; Parus and Karbowska, 2020; Kadam et al., 2020; Tsarenko et al., 2021). Parus and Karbowska (2020) recently reported on the possibility of using the algae *Ulva* and Cystoseira as natural indicators of environmental cleanliness given that these species where shown to accumulate metals. Parmar and Rawtani (2016) described several potential advantages for use of bioindicators, namely (1) biological impacts can be determined; (2) potential synergetic and antagonistic impacts of various combined pollutants on ecosystems can be exhibited; (3) early stage diagnosis of putative harmful effects of toxins on human and animal health can be monitored; and (4) can be considered as a potentially viable economic alternative to use of conventional sophisticated methods.

According to Rindi (2014), terrestrial algae (green algae and diatoms) are more directly affected by climate change and can therefore respond in a more immediate way. This is attributed in part to the fact that algae have short generations, fast turnovers and respond quickly to changes in environmental conditions. Sarmaja-Korjonen et al. (2006) demonstrated that algae appeared to be comparatively good indicators of environmental conditions by representing productivity disparities during changing climatic conditions. Hallegraeff (2010) has also indicated that changes in algal communities can putatively provide a sensitive early warning for climate-driven uncertainties in aquatic ecosystems. There has been increased interest in alternative uses for microalgae within aquaculture to assist with sustainability, in addition to enabling ecotoxicological

assessment and water quality control (Han et al., 2019; O'Neill et al., 2019). According to Han et al. (2019), microalgae can also be utilised in aquaculture for wastewater assimilation, oxygen production and partial feed replacement. The microalgae *Pseudokirchneriella subcapitata* (*P. subcapitata*) has previously been suggested as a potential early warning indicator for altering issues associated with in aquaculture processing due to environmental variances, including climate change (O'Neill et al., 2019).

Fish farm wastewater is traditionally high in nutrient rich products (Ngo et al., 2016; Sikder et al., 2016). Nitrogen, phosphorus and organic matter are characteristic of this nutrient rich waste which is normally as a result of metabolic waste products and left over food (Jegatheesan et al., 2011; O'Neill et al., 2019). If this is released into a water body untreated, water pollution will develop leading to issues that may include eutrophication in that aquatic system (Martinez-Porchas et al., 2014). Eutrophication occurs when a water body is put under pressure with large levels of organic matter and nutrient waste that is taken in and biologically processed which in turn leads to algal blooms (Jegatheesan et al., 2011; Martinez-Porchas et al., 2014; Sikder et al., 2016). Algal blooms in turn can lead to decreases in light and oxygen production, which can suffocate aquatic life (Jegatheesan et al., 2011; Chislock et al., 2013; O'Neill et al., 2019). Organic matter and nutrient waste is typically as a result of the application of artificial feed supplementation which is necessary in order to increase and maintain yields to meet the increased demands (Kolarevic et al., 2014; Feucht and Zander, 2015; O'Neill et al., 2019).

Water quality is typically assessed to determine the potential effects it may have in its receiving system; this is traditionally conducted by means of physicochemical analysis (da Silva et al., 2017). The use of these parameters alone will only provide a limited window in time of the water quality for a system (O'Neill et al., 2019; O'Neill et al., 2020). Inclusion of bioassays to assess the potential effects on aquatic ecosystems and the organisms therein will provide a broader scope on the quality of water. Microalgae are primary producers and are keystones in aquatic food chains. They represent an imperative group of highly sensitive photosynthetic organisms frequently used to assess aquatic systems (Rodgher et al., 2012). *Pseudokirchneriella subcapitata* (*P. subcapitata*) is unicellular green algae most commonly used and recommended for ecotoxicological assessment due to its being inexpensive, and both highly reliable and reproducible (ISO, 2012).

The hypothesis of this study is that algal traditionally used in ecotoxicology bioassays can be further utilised for the real-time sustainable enhancement of aquaculture as it provides a potential means of monitoring the influence of adverse environmental effects caused by extreme weather events attributed to variances in climate. Thus, the aim of this research is to determine the robustness of *P. subcapitata* as a putative early warning bio-indicator for monitoring impact of climate variance using an Irish freshwater aquaculture farm as a case study.

#### 2. Materials & methods

#### 2.1. Sampling

Intake and output water samples were collected from a freshwater fish farm located in Boyle, Co. Sligo (Fig. 1). The farm cultures European perch (*Perca fluviatilis*) and consists of three culture ponds that use a flow through system, a hatchery and nursery that use a recirculating aquaculture system (RAS) and a constructed wetland that is used for culture water treatment. Grab samples were collected



**Fig. 1.** Map of Ireland indication the approximate location of the freshwater fish farm (53°58′16″ N, 08°24′44″ W) indicated with yellow, and the three closest Met Eireann weather stations (Markree – 54°10′30″ N, 08°27′20″ W; Mount Dillon – 53°43′37″ N, 07°58′51″ W; Knock – 53°54′22″ N, 08°49′4″ W) surrounding the farm, indicated by orange. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

in 5 L octagonal carboy HDPE bottles (Lennox) and transported directly to the lab via car approximately 70 km away. Samples were collected directly from the intake and output sources once a month from March 2019 to August 2019 as this was the time period analysed during the previous study at same freshwater aquaculture farm reported in O'Neill et al. (2019). Samples from the settlement pond were also collected for analysis in order to determine the treatment efficacy of the constructed wetland, which was not fully operational until June 2019 reflecting period of maintenance. Wastewater collection occurred on the same day, at approximately the same time, during each month of monitoring where sampling points are displayed in Fig. 2. Triplicate samples were analysed from the same 5 L grab sample.

#### 2.2. Physicochemical analysis

The Statutory Instrument (S.I.) 77/2019, S.I. 272/2009, and the Irish Environmental Protection Agency's (EPA) water quality parameters (Environmental Protection Agency, 2001; Irish Statutory Office, 2009; Irish Statutory Office, 2019) were followed to measure water quality parameters. Discharge licensing in Ireland is currently based on an individual basis. Grab samples collected represented 30 min of the 24 h period; composite sampling was not possible. To compensate for the latter, results complied in this study were also compared to previous research studies conducted on a range of aquaculture facilities (Table 7 in the Supplementary data).



**Fig. 2.** Schematic of the Irish freshwater fish farm layout indicating the locations of the collection points for the intake (red), output (green) and settlement pond (yellow) water samples. 1) hatchery, 2) nursery, 3-4) mesocosms, 5-7) culture ponds, 8) settlement pond, 9) constructed wetland, 10) holding tank, 11) river. Black arrows indicate flow of water. Note: Schematic is not to scale. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

#### Table 1

Summary of the methods used to assess the physicochemical parameters investigated on the Irish freshwater aquaculture intake, output and settlement pond water samples. The method employed, detection limit for all kits used and standard water and wastewater analysis methods numbers have been included.

Physicochemical parameter	Method	Detection limit (mg L <sup>-1</sup> )	Standard method number
Alkalinity	Titrimetric	-	2320-В
BOD	Membrane electrode	-	5210-B
COD	Photometric	0-150 15-300	5220-D
DO	Membrane electrode	-	4500-0 G
Hardness	Titrimetric	-	2340-С
$NH_4^+$	Photometric	0.013-3.86 2.6-193.0	4500-NH <sub>3</sub> -F
NO <sub>2</sub>	Photometric	0.007-3.28	345-1
NO <sub>3</sub>	Photometric	0.4-110.7	4500-NO <sub>3</sub>
рН	Membrane electrode	-	2310-В
PO <sub>4</sub> <sup>3-</sup>	Photometric	0.007-15.3 1.5-92.0	4500-P-C
Suspended solids	Gravimetric	-	2540-D
Temperature	Thermometer	-	2550-В

 $NH_4^+$  = ammonium,  $NO_2^-$  = nitrite,  $NO_3^-$  = nitrate,  $PO_4^{3^-}$  = orthophosphate, DO = dissolved oxygen, BOD = biochemical oxygen demand, COD = chemical oxygen demand.

Physicochemical parameters – temperature, pH, NH<sub>4</sub><sup>+</sup>, NO<sub>2</sub>, NO<sub>3</sub>, PO<sub>4</sub><sup>-</sup>, DO, BOD, COD, suspended solids, hardness and alkalinity were analysed within 24 h of collection to remove the need for preservation. Spectroquant® kits (Sigma Aldrich) were used as per the manufacturer's instructions to assess NH<sub>4</sub><sup>+</sup>, NO<sub>2</sub>, NO<sub>3</sub>, PO<sub>4</sub><sup>2-</sup> and COD levels. Temperature and pH were analysed using the VWR pHenomenal<sup>TM</sup> MU 6100 L meter and VWR 111662-1157 pH probe. DO and BOD<sub>5day</sub> were assessed using the Jenway 9500 DO<sub>2</sub> meter and probe. The suspended solids were analysed by filtration using a Buchner flask and funnel. Alkalinity was assessed via titration using phenolphthalein indicator, methyl orange indicator and hydrochloric acid. Hardness was analysed via titration using pH 10 buffer, erichrome black and EDTA. A summary of all physicochemical methods employed in this study, including each standard method number, are shown in Table 1.

#### 2.3. Ecotoxicity analysis

The unicellular freshwater green algae *P. subcapitata* was used to determine the quality of the water. A culture was obtained from The Culture Collection of Algae and Protozoa (CCAP 278/4: SAMS Limited,

Scottish Marine Institute, Oban, Argyll, Scotland, U.K.) and grown in standard Jarworski's culture medium at 23  $^{\circ}C \pm 2 ^{\circ}C$  exposed to continuous illumination (lux 6000-10,000). Additionally, starter cultures of Asterionella formosa (CCAP 1005/9) and Monoraphidium contortum (CCAP 245/2) were obtained from The Culture Collection of Algae and Protozoa (SAMS Limited, Scottish Marine Institute, Oban, Argyll, Scotland). P. subcapitata was compared with A. formosa and M. contortum to ensure that P. subcapitata was representative of Irish aquatic algae (Table 2). Algae were sub-cultured every three days to ensure the growth rate remained in the exponential phase. Analysis was conducted as per the Water quality - Fresh water algal growth inhibition test with unicellular green algae ISO (8692:2012) guidelines. The P. subcapitata was exposed to the intake and output samples for 72 h at 23  $^{\circ}C \pm 2 ^{\circ}C$  exposed to continuous illumination under static conditions. The percent of algal growth rate inhibition (E<sub>r</sub>C<sub>50</sub>) was calculated by comparing samples to a negative control containing just the Jarworski's culture medium. The  $E_r C_{50}$  is the concentration at which the has been a 50% reduction in the growth rate relative to the control within 72 h (ISO, 2012). Eqs. (1), (2) and (3) were taken directly from the ISO (8692:2012) guidelines (ISO, 2012) and calculations were conducted as follows;

Algae cells mL<sup>-1</sup> = 
$$\frac{n}{0.02} \times 10^3$$
 (1)

where

n = the number of cells counted using a haemocytometer.

Average specific growth rate (
$$\mu$$
) =  $\frac{\ln X_n - \ln X_0}{T_n - T_0}$  (2)

where

In = natural log of  $X_n$  = Algae cells mL<sup>-1</sup> at 72 h  $X_0$  = Algae cells mL<sup>-1</sup> at 0 h  $T_n$  = Duration of test  $T_0$  = Time zero.

Percent growth rate inhibition = 
$$\frac{C\mu - T\mu}{C\mu} \times 100$$
 (3)

where

 $C\mu$  = Average specific growth rate for control  $T\mu$  = Average specific growth rate for treatment.

#### Table 2

Mean concentrations calculated for each parameter investigated in this study conducted in 2019 (intake and output water 2019) and the previous study conducted in 2018 on the same fish farm (intake and output water 2018) by O'Neill et al. (2019). All data is based on the average across six months. S.D. has been indicated.

Parameter	Intake water		Output water	
	2018	2019	2018	2019
$NH_{4}^{+}$ (mg L <sup>-1</sup> )	$0.16\pm0.18$	$0.06\pm0.09$	$1.16 \pm 0.64$	$0.53 \pm 0.53$
$NO_{2}^{-}$ (mg L <sup>-1</sup> )	$0.02 \pm 0.01$	$0.01 \pm 0.01$	$0.32 \pm 0.38$	$0.10\pm0.07$
$NO_{3} (mg L^{-1})$	$3.62 \pm 1.60$	$1.81 \pm 1.27$	$5.29 \pm 5.56$	$1.74 \pm 1.10$
$PO_4^{3-}$ (mg L <sup>-1</sup> )	$1.76 \pm 0.84$	$0.63 \pm 1.14$	$3.78 \pm 2.00$	$0.77 \pm 0.51$
DO (mg $O_2 L^{-1}$ )	$10.31 \pm 0.87$	$10.76 \pm 2.75$	$5.10 \pm 2.85$	$7.66 \pm 3.06$
BOD (mg $O_2 L^{-1}$ )	$2.27 \pm 1.47$	$2.68 \pm 0.70$	$3.24 \pm 1.95$	$2.80\pm0.96$
$COD (mg O_2 L^{-1})$	45.91 ± 40.81	$25.97 \pm 9.98$	$76.44 \pm 59.06$	$19.24 \pm 11.68$
Temperature (°C)	$14.76 \pm 2.53$	$13.85 \pm 1.35$	$15.53 \pm 2.66$	$14.23 \pm 1.48$
рН	$7.76 \pm 0.19$	$7.70 \pm 0.14$	$7.11 \pm 0.18$	$7.14\pm0.06$
Suspended solids (mg L <sup>-1</sup> )	40.17 ± 79.08	$20.50 \pm 8.00$	83.67 ± 144.33	19.22 ± 9.23
Hardness (mg CaCO <sub>3</sub> L <sup>-1</sup> )	$100.49 \pm 9.22$	$106.24 \pm 12.18$	$116.03 \pm 16.80$	111.58 ± 22.45
Alkalinity (mg CaCO <sub>3</sub> L <sup>-1</sup> )	$122.55 \pm 17.71$	$135.03 \pm 20.49$	$128.91 \pm 18.19$	129.47 ± 17.98
P. subcapitata (% growth rate inhibition)	43.14 ± 18.47	$13.66 \pm 1.44$	$-2.70 \pm 20.41$	$9.73  \pm  2.03$

 $NH_4^+$  = ammonium,  $NO_2^-$  = nitrite,  $NO_3^-$  = nitrate,  $PO_4^{3^-}$  = orthophosphate, DO = dissolved oxygen, BOD = biochemical oxygen demand, COD = chemical oxygen demand.

#### 2.4. Statistical analysis

Statistical analyses were conducted using MINITAB 18 and GRAPHPAD PRISM 8. The generated data were grouped and subjected to normality testing (Anderson-Darling) to ensure all samples were normally distributed. Unpaired *t*-tests and ANOVA were used to identify any significant differences in the variables. P < 0.05 indicated a statistically significant difference. Pearson's correlation (*r*) was used to assess if any correlations between the algae and/or the physicochemical parameters existed (Fig. 3).

#### 3. Results

#### 3.1. Physicochemical analysis

Results determined for the physicochemical parameters investigated in this study on Irish freshwater fish farm intake, output and settlement pond water samples are displayed in Fig. 4. Increases in NH<sub>4</sub><sup>+</sup>, NO<sub>2</sub>, PO<sub>3</sub><sup>2-</sup>, BOD and temperature, along with decreases in DO, COD, pH and alkalinity occurred when comparing the intake and output water from the fish farm. Fluctuations from month-to-month in NO<sub>3</sub> and



**Fig. 3.** Breakdown for the physicochemical parameters investigated on Irish freshwater aquaculture intake (green), output (blue) and settlement pond (yellow) water samples from March 2019 to August 2019. Parameters investigated were A)  $NH_4^+$ , B)  $NO_2^-$ , C)  $NO_3^-$ , D)  $PO_4^{3-}$ , E) DO, F) BOD, G) COD, H) temperature, I) pH, J) suspended solids, K) hardness and L) alkalinity. Red lines indicate levels set out by S.I. 272 of 2009 and 77 of 2019. Black lines indicate levels set out by the Irish EPAs parameters for water quality. NOTE: Dilution factor of the receiving water body has not been included. Lines do not appear on temperature and CaCO3 as no limits were indicated. S.D. indicated, n = 9. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)



**Fig. 4.** Percentage growth rate inhibition observed in *P. subcapitata* proceeding exposure to Irish freshwater aquaculture intake (green), output (blue) and settlement pond (yellow) water samples for 72 h at 23 °C  $\pm$  2 °C under continuous illumination from March 2019 to august 2019. S.D. indicated, *n* = 9. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

hardness were also observed. With the exception of NO<sub>2</sub> (P = 0.011) and pH (P = 0.025), no statistically significant (one-way ANOVA) differences were indicated. With the exception of NH<sub>4</sub><sup>+</sup> and NO<sub>2</sub> levels in May and June, decreases were observed in the physicochemical parameters between the settlement pond and output water. With the exception of suspended solids (P = 0.044), no statistically significant (one-way ANOVA) differences were observed; one-way ANOVA was conducted across all three sampling points and no statistically significant differences were observed. Two-way ANOVA was conducted in order to take the sampling month into consideration; with the exception of BOD (P = 0.083) and suspended solids (P = 0.150), a statistically significant difference was observed in all parameters.

#### 3.2. Algal bioassay analysis

The percentage growth rate inhibition observed in the intake, output and settlement pond water are displayed in Fig. 5. With the exception of samples for May and July, a decrease in growth rate inhibition between the intake and output samples was demonstrated. A decrease was also observed in all samples between the settlement pond and output water. No statistically significant (one-way ANOVA) differences were observed for either set of samples. One-way ANOVA was conducted across the three sampling points and no statistically significant differences were indicated. Two-way ANOVA was conducted to determine whether any statistically significant differences were observed when the sampling month was taken into consideration. A statistically significant difference (P = 0.001) was indicated when the sampling month was included.

#### 3.3. Comparative study

Table 3 summarises an average of all results obtained during a previous study conducted at a similar time of year on the same fish farm during times of extreme weather conditions (heat wave and drought) by O'Neill et al. (2019) and those determined in this study which were conducted under normal weather conditions for the Republic of Ireland. With the exception of the dissolved oxygen, pH, alkalinity and hardness, all physicochemical concentrations decreased during the similar time periods of 2018 and 2019 in both the intake and output water samples. The pH, alkalinity and hardness remained similar whilst dissolved oxygen levels increased. For the *P. subcapitata*, considerable decreases in inhibition toxicity were observed in the intake water and no stimulation was observed in the output water from this study compared to 2018. Both of which demonstrated a statistically significant (*t*-tests) difference (P = 0.030 for the intake water and P = 0.039 for the output water).

# 3.4. Correlation studies for monitoring periods of freshwater aquaculture farm

Correlation studies were conducted between all parameters investigated at the three sampling points. A positive correlation between two parameters indicates that as one parameter increases or decreases, so too does the other parameter. A negative correlation between two parameters indicates that as one parameter increases or decreases, the opposite occurs with the other parameter i.e., an inverse relationship. All results for the intake, output and settlement pond water samples are displayed in Tables 3, 4, 5, and 6 respectively. In the intake samples a negative correlation was observed between P. subcapitata and temperature as well as pH. A negative correlation was indicated between  $NH_4^+$ and NO<sub>2</sub>. A positive correlation was identified between NO<sub>2</sub> and alkalinity. The NO<sub>3</sub> demonstrated a negative correlation with DO and a positive correlation with suspended solids. A positive correlation was observed between PO<sub>4</sub><sup>3-</sup> and alkalinity. A negative correlation was identified between DO and suspended solids. In the output samples a positive correlation between temperature and NO<sub>3</sub> was indicated. A negative correlation was identified between DO and NH<sub>4</sub><sup>+</sup> as well as  $NO_2^-$ . Hardness displayed a positive correlation between  $PO_4^{3-}$  and alkalinity. In the settlement pond a positive correlation between *P. subcapitata* and PO<sub>4</sub><sup>3-</sup> was observed. Temperature demonstrated a positive correlation with pH, NH<sub>4</sub><sup>+</sup>, hardness and alkalinity. The pH indicated a positive correlation with NH<sub>4</sub><sup>+</sup>, BOD hardness and alkalinity.



**Fig. 5.** Average A) rainfall and B) temperature recorded for 2018 (blue) and 2019 (yellow) at three Met Eireann weather stations surrounding the freshwater fish farm during the sampling period of March 2019 to August 2019. Stations were located at 1) Markree, Co. Sligo, 2) Knock, Co. Mayo and 3) Mount Dillon, Co. Roscommon. Stations were located north-west, south-west and south-east of the fish farm, respectively. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

#### Table 3

Correlation matrix for *P. subcapitata* and all physicochemical parameters investigated on the Irish freshwater aquaculture intake water samples. Bold figures indicate where statistically significant differences were observed. Breakdown of correlation ranges are also indicated.

	P. sub	Т	рН	$\mathrm{NH}_4^+$	$NO_2^-$	$NO_3^-$	PO43-	DO	BOD	COD	SS	Н	А
P. sub	1.000								0 = Nore	elationship			
Т	-0.830	1.000							>0-0.3 =	Weak relation	onship		
pН	-0.951	0.794	1.000						0.3-0.5 =	Moderately	weak relation	nship	
$NH_4^+$	-0.183	0.360	0.451	1.000					0.5-0.7 =	Moderately	strong relation	onship	
NO <sub>2</sub>	-0.260	0.037	-0.020	-0.845	1.000				0.7-<1 =	Strong relat	ionship		
NO <sub>3</sub>	-0.727	0.687	0.557	-0.173	0.656	1.000			1 = Perfe	ect linear rela	tionship		
PO4 <sup>3-</sup>	-0.286	0.024	0.133	-0.444	0.785	0.728	1.000						
DO	0.726	-0.809	-0.553	0.057	-0.468	-0.876	-0.400	1.000					
BOD	0.307	-0.627	-0.351	-0.582	0.232	-0.465	-0.073	0.625	1.000				
COD	-0.623	0.448	0.372	-0.523	0.712	0.623	0.353	-0.749	0.007	1.000			
SS	-0.702	0.594	0.466	-0.380	0.730	0.855	0.559	-0.915	-0.295	0.924	1.000		
Н	-0.061	0.326	-0.138	-0.426	0.588	0.638	0.471	-0.502	-0.264	0.254	0.438	1.000	
А	-0.394	0.201	0.218	-0.497	0.831	0.800	0.916	-0.440	0.017	0.400	0.577	0.659	1.000

 $P. sub = P. subcapitata, T = temperature, NH_{+}^{4} = ammonium, NO_{2}^{2} = nitrite, NO_{3}^{3} = nitrate, PO_{4}^{3^{2}} = orthophosphate, DO = dissolved oxygen, BOD = biochemical oxygen demand, COD = chemical oxygen demand, SS = suspended solids, H = hardness, A = alkalinity.$ 

A positive correlation was identified between  $NH_4^+$  and DO as well as BOD. A positive correlation was observed between  $NO_2^-$  and  $NO_3^-$ . DO demonstrated positive correlations with  $PO_4^{3-}$  and BOD. Finally, a positive correlation was identified between hardness and alkalinity.

3.5. Weather conditions influencing water quality on monitored aquaculture farm

Due to observations determined in the previous study, conducted by same authors (O'Neill et al., 2019), as a result of dramatic weather conditions experienced during 2018 in the Republic of Ireland that coincided with occurrence of the hottest summer recorded to date by Irish Meteorological Office (Met Eireann) resulting in a nationwide hosepipe ban due to limit negative impacts of drought; weather conditions during the sampling period have been included. For continuity, mean temperature and rainfall data collected by Met Eireann at the same three weather stations surrounding the fish farm as the previous study have been included for this sample period (Met Éireann, 2019). Stations were situated at Markree Co. Sligo, Knock Co. Mayo and Mount Dillon Co. Roscommon, as shown in Fig. 1. Increases in the mean rainfall (Fig. 5A) and decreases in the mean temperature (Fig. 5B) were observed for 2019 versus the same time period in 2018 across the three stations. Maxima temperatures had also decreased (Fig. 6 - Supplementary). Statistical analysis found that the relationships between the algae and the rainfall and temperature switched. A moderately strong inverse relationship (r = -0.559) between the algae and temperature, and a weak inverse relationship (r = -0.209) between the algae and rainfall now existed.

#### 4. Discussion

#### 4.1. Physicochemical evaluation

In order to ascertain whether the processes conducted in the fish farm altered the quality of the water, the physicochemical results determined in the intake and output water were compared firstly to one another and then to the previous study conducted. It should be appreciated that the dilution factor of the receiving river on potential impact of aquaculture effluent has not been considered in this research. The presence of  $NH_4^+$ ,  $NO_2^-$  and  $NO_3^-$  in the output water suggested that the nitrification process (enzymatic oxidation of  $NH_4^+$  to  $NO_3^-$  by way of  $NO_2^-$ ) was occurring. Increases in these parameters between the intake and output samples suggested that, for the most part, their production was due to practices within the farm. These were most likely due to the presence of fish waste and uneaten artificial pelleted feed used in the cultural process. Increases above guidance levels (1 mg NH<sub>4</sub><sup>+</sup> L<sup>-1</sup> and 0.03 mg  $NO_{2}^{-1}L^{-1}$  in the parameters were only observed between May and June, which suggested a potential cause for concern. However, this was most likely attributed to the constructed wetland, which was not functioning to its optimal capacity due to undergoing maintenance works: no discharge of aquaculture effluent was released during these times. Levels of monitored physicochemical parameters dropped below

Table 4

Correlation matrix for *P. subcapitata* and all physicochemical parameters investigated on the Irish freshwater aquaculture output water samples. Bold figures indicate where statistically significant differences were observed. Breakdown of correlation ranges are also indicated.

	P. sub	Т	рН	$\mathrm{NH}_4^+$	NO <sub>2</sub>	NO <sub>3</sub>	PO43-	DO	BOD	COD	SS	Н	А
P. sub	1.000								0 = No relationship				
Т	-0.537	1.000							>0-0.3 = Weak relationship				
pН	-0.332	0.419	1.000						0.3-0.5 = Moderately weak relationship				
NH <sub>4</sub> <sup>+</sup>	-0.210	0.496	-0.262	1.000					0.5-0.7 = Moderately strong relationship				
$NO_2^-$	0.131	0.213	-0.645	0.733	1.000				0.7-<1 =	= Strong relat	ionship	-	
NO <sub>3</sub>	-0.627	0.929	0.234	0.439	0.161	1.000			1 = Perfect linear relationship				
PO43-	-0.488	0.139	-0.500	0.433	0.656	0.213	1.000				•		
DO	0.296	-0.670	0.370	-0.816	-0.811	-0.700	-0.612	1.000					
BOD	-0.102	-0.518	-0.702	0.128	0.431	-0.405	0.766	-0.112	1.000				
COD	0.007	-0.362	-0.505	-0.529	-0.092	-0.070	0.252	0.095	0.385	1.000			
SS	-0.442	-0.194	-0.004	0.304	0.124	-0.267	0.525	0.033	0.634	-0.324	1.000		
Н	-0.574	0.344	-0.514	0.494	0.603	0.522	0.902	-0.769	0.558	0.364	0.246	1.000	
А	-0.639	0.479	-0.327	0.494	0.299	0.736	0.544	-0.701	0.197	0.296	0.019	0.839	1.000

 $P. sub = P. subcapitata, T = temperature, NH_{+}^{+} = ammonium, NO_{2}^{2} = nitrite, NO_{3}^{3} = nitrate, PO_{4}^{3-} = orthophosphate, DO = dissolved oxygen, BOD = biochemical oxygen demand, COD = chemical oxygen demand, SS = suspended solids, H = hardness, A = alkalinity.$ 

#### Table 5

Correlation matrix for *P. subcapitata* and all physicochemical parameters investigated on the Irish freshwater aquaculture settlement pond water samples. Bold figures indicate where statistically significant differences were observed. Breakdown of correlation ranges are also indicated.

	P. sub	Т	рН	$\mathrm{NH}_4^+$	$NO_2^-$	NO <sub>3</sub>	PO4 <sup>3-</sup>	DO	BOD	COD	SS	Н	А
P. sub	1.000								0 = No 1	elationship			
Т	0.386	1.000							>0-0.3 =	= Weak relatio	onship		
pН	0.566	0.938	1.000						0.3-0.5 = Moderately weak relationship				
$NH_4^+$	0.714	0.816	0.958	1.000					0.5-0.7 = Moderately strong relationship				
NO <sub>2</sub>	0.133	0.464	0.229	0.107	1.000				0.7-<1 =	= Strong relati	onship		
NO <sub>3</sub>	0.168	0.417	0.182	0.079	0.992	1.000			1 = Perf	ect linear relat	tionship		
PO43-	0.976	0.393	0.580	0.715	-0.004	0.032	1.000						
DO	0.784	0.525	0.765	0.883	-0.288	-0.287	0.847	1.000					
BOD	0.679	0.786	0.945	0.986	-0.034	-0.060	0.714	0.927	1.000				
COD	0.593	0.362	0.523	0.532	-0.322	-0.336	0.730	0.744	0.616	1.000			
SS	-0.255	0.468	0.514	0.438	-0.085	-0.183	-0.271	0.201	0.441	-0.077	1.000		
Н	0.251	0.898	0.843	0.677	0.205	0.147	0.341	0.512	0.709	0.609	0.412	1.000	
А	0.356	0.942	0.920	0.781	0.279	0.212	0.407	0.571	0.785	0.582	0.488	0.974	1.000

 $P. sub = P. subcapitata, T = temperature, NH_{+}^{4} = ammonium, NO_{2}^{2} = nitrite, NO_{3}^{3} = nitrate, PO_{4}^{3} = orthophosphate, DO = dissolved oxygen, BOD = biochemical oxygen demand, COD = chemical oxygen demand, SS = suspended solids, H = hardness, A = alkalinity.$ 

guidance values, once the wetland was fully functional where low levels were also observed in the intake water. It is likely that agricultural processes (cattle and sheep farming) and forestry processes (tree felling) occurring upstream of the fish farm contributed to these measured physicochemical parameters.

The  $PO_4^{3-}$  levels in the output water was greater than that of the input as a result of the processes within the aquaculture farm. However, levels observed in the intake suggested that agricultural and forestry processes upstream of the farm could have also contributed to levels. Concentrations of  $PO_4^{3-}$  were greater than guidance levels (0.35 mg  $PO_4^{3-}$  L<sup>-1</sup>) suggesting a potential cause for concern as excess levels can result in the promotion of algal blooms leading to potential hypoxic conditions in the water body (O'Neill et al., 2019). However, once maintenance was completed on the constructed wetland, levels detected in the output water were reduced to guidance levels indicating no foreseen issues.

A decrease in oxygen levels was observed between the intake and output water due to the aquaculture process. This decrease may also have been due to changes in seasonality. According to Alam et al. (2007), da Silva et al. (2017) and O'Neill et al. (2019) oxygen levels  $\geq$ 4 mg O<sub>2</sub> L<sup>-1</sup> are sufficient to maintain aquatic life. Although levels in the output water dropped just below the guidance value (7 mg  $O_2 L^{-1}$ for cyprinid waters), levels remained above the critical 4 mg  $O_2 L^{-1}$ level and as a result had indicated no cause for concern. BOD levels between the intake and output water fluctuated i.e., BOD increased between the intake and output water during March, June and July, whilst decreases were observed in April, May and August. Despite fluctuations, BOD levels remained below the guidance value of 5 mg O<sub>2</sub> L<sup>-1</sup> for cyprinid waters suggested by the Irish EPA. Additionally, the dilution factor of the receiving water system has not been included therefore BOD levels would further decrease upon release. With the exception of March, COD levels decreased between the intake and output water. This suggested that the COD levels were not due to processes within the farm and were more likely due to works being conducted upstream. Despite this, COD levels were well below the guidance value of 40 mg  $O_2$  L<sup>-1</sup>.

Suspended solids levels were greater between the intake and output water during the months of March, April and May which were most likely due to increases in production processes in the farm. However, this trend reversed for the months of June, July and August. This was most likely due to high levels of tree felling being conducted in the forestry upstream of the farm during this time. Levels were greater than the guidance level of 25 mg L<sup>-1</sup> during May and June but this was most likely due to the constructed wetland maintenance work as once the wetland became fully functional again after the June maintenance, concentrations dropped well below this level. Once again, it should be noted that water did not leave the farm during this time.

Temperature between the intake and output water samples remained consistent with increases observed during the summer months, as would be expected. Fish farms must not release water that is greater than 20 °C. At no point during the study did temperatures rise to this level. Aquaculture waters are recommended to have a pH of between 6 and 9 (EPA, 2001). All samples remained within this range. The intake samples were slightly more alkaline than the output samples. Output samples had greater CaCO<sub>3</sub> levels and therefore a greater buffering capacity which may account for pH levels of just about neutral (pH 7) in the output water. CaCO<sub>3</sub> levels were measured for hardness. Results suggested that the water was slightly to moderately hard. This correlates with water hardness demonstrated around Boyle, Co. Roscommon (O'Neill et al., 2019).

All parameters were then compared to the previous year's study. In 2018, Ireland experienced its hottest summer on record whereby the country experienced long periods of drought. The physicochemical parameters were greater in 2018 than that of this study (2019) for the similar time period (O'Neill et al., 2019). This was most likely due to increased flow rates as a result of increased rainfall resulting in no drought conditions being observed in 2019. As this research only focused on one type of fish farm culturing one specific species of fish (European Perch) results from this study were also compared to previous aquaculture studies. These studies were located worldwide and encompassed a range of different aquaculture systems culturing several different species of fish, as shown in Table 6. The studies reviewed demonstrated similar or higher levels than the concentrations observed in this study.

#### 4.2. Algal bioassay evaluation

Inhibition of the growth rate of the *P. subcapitata* was observed in both the intake and output water samples. The presence of growth rate inhibition suggested that algal blooms downstream of the fish farm would be unlikely. However, growth rate inhibition is still demonstration of a toxic effect. This inhibition may result in loss of biodiversity in the receiving water body (Rabalais, 2002; O'Neill et al., 2019). Exclusive of the months of May and July, the percentage of growth rate inhibition was found to decrease between the intake and output water samples. The inhibition toxicity detected throughout the study was at sub-lethal levels. Additionally, toxicity was reduced once the water had passed through the fish farm's constructed wetland. This suggested that the farm itself was successfully improving the water quality.

When results were compared to the previous study of 2018, a statistically significant difference was observed in both the intake

(P = 0.030) and the output (P = 0.039) water samples. Unlike the previous study, no growth rate stimulation was observed in the output water. Equally, considerably lower levels (sub-lethal) of toxicity were observed in the intake water, e.g. levels of up to 75% growth rate inhibition were observed during the drought conditions of 2018 (O'Neill et al., 2019). This reduction was most likely due to the reduced temperatures and resulting increased flow rates.

Results were than compared to previous studies that utilised *P. subcapitata* to assess fish farm output water. Miashiro et al. (2012) demonstrated similar results in a Brazilian study (with a traditionally much warmer climate than Ireland) to the previous study conducted on the fish farm by O'Neill et al. (2019) during the heat wave and drought conditions, where by similar levels of growth rate stimulation were observed. The current study however, conformed to most of the available research on the effects of fish farm output water on *P. subcapitata*. Guéguen et al. (2004), Ivanova and Groudeva (2006) and Ma et al. (2006) all observed similar growth inhibition levels to those demonstrated. These studies were also conducted in countries (Poland and Bulgaria) with similar temperate weather conditions to those normally experienced in Ireland.

#### 4.3. Constructed wetland evaluation

The previous study conducted by O'Neill et al. (2019) indicated that there may have been issues with the constructed wetland due to increased concentration of nitrogenous and phosphorus waste in the output water samples. However, it was unclear whether this issue may have been due to the extreme weather conditions experienced during 2018 in Ireland. As a result, samples were included at the exit point of the settlement pond to ascertain the efficacy of the wetland. This was the point at which the wastewater entered the constructed wetland. Evaluation of the settlement pond demonstrated that, when fully functional after the June maintenance, the constructed wetland was effective in the removal of waste products from the water. This efficacy may also be due in part to the re-introduction of duckweed (Lemna minor). The previous study found spikes in nitrogenous waste concentrations when the duckweed was removed from the farm. Duckweed has the ability to use  $NO_3^-$  as a nutrient source (O'Neill et al., 2019) and research is ongoing in this area.

#### 4.4. Climate change

According to the Intergovernmental Panel on Climate Change (2020), the momentum of climate change had greatly increased in 2019. Climate change is the most troubling scientific issues of our time (Bulkeley and Newell, 2015). The Bulletin of the Atomic Scientists (2020) has now moved the hypothetical Doomsday Clock to 100 s to midnight which is the closest it's ever been to the "point of no return" represented by midnight. Originally introduced in 1947 due to the threat of nuclear weapons, climate change is now considered an equal threat to that (Weisberger, 2020). This research has further indicated that climate change has a direct impact on fish farming, as suggested by the lack of algal growth stimulation or high levels of growth inhibition due normal weather conditions reported in this study. Algal growth and temperature still demonstrate a strong correlationship (r = -0.830) in the intake samples. This research has further demonstrated the ability of P. subcapitata to be utilised as an early warning indicator for climate change ambiguity in freshwater aquaculture.

#### 5. Conclusion

The findings of this timely study responds to the main tenets of the recent intergovernmental report on global climate change (IPCC, 2021) that seeks urgent viable and resilient technological solutions to help future proof for a climate-smart environmentally friendly agri-food sector, including fisheries. Moreover, this 'code red for

humanity' IPCC report on climate change clearly highlights that human or anthropogenic activity has contributed greatly to greenhouse gas levels in the atmosphere where there is a pressing need to reduce carbon dioxide and methane emissions, and to stall rising global temperatures that leads to extreme weather events.

- Regarding the latter, there is pressing need for countries to use innovative approaches to support and to develop sustainable food systems delivering benefits for the sector, for society, and for the environment. The findings of this present study will support and enable viable and resilient primary producers to provide food that are safe, nutritious and appealing; thus, using eco-technologies and talent to inform innovative, competitive and resilient agri-food sector regionally, and internationally (Rowan and Pogue, 2021).
- Specifically, this study revealed that the freshwater microalga *P. subcapitata* can be used for the real-time prediction of potential adverse environmental issues associated with freshwater aquaculture wastewater, which can be seen as complementary to relying upon using traditional physicochemical parametric measurements.
- As this research focused exclusively on one type of fish farm in the Republic of Ireland, use of this algal bioindicator technique should be also applied to evaluate different types of aquaculture farms including pond-based, flow-through, and recirculation in order to ensure harmonised results across a range of culture systems and fish species.
- Inclusion of additional ecotoxicological bioassays such as a full test battery, encompassing different trophic levels (e.g., Daphnia magna – primary consumer, Vibrio fischeri – decomposer) (Garvey et al., 2013) within the aquatic ecosystems should also be considered for future studies in order to develop a better understanding of the potential environmental effects' aquaculture processes could have on water bodies.
- There is merit in conducting molecular profiling of naturally occurring microalgae in order to incorporate these as a cocktail of native species representative of local natural aquatic ecosystems, which will support and inform biodiversity, conservation management along with enhanced bioindicator performance. For example, Kadam et al. (2020) identified 33 Taxa belonging to 27 genera of microalgae when they considered development of a putative 'Algal Genus Pollution Index' for potentially assessing water bodies in the Doon valley, India.
- The constructed wetland servicing this aquaculture farm needs to be increased in size in order to be effective in treating volume of the wastewater effluent where efficacy of treatment can be also influenced by extreme weather events that influence flow rates.
- The lack of growth rate stimulation and decrease in growth rate inhibition when compared to the previous study (O'Neill et al., 2019) supports future use of *P. subcapitata* as an early warning indicator to potential issues in fish farms associated with climate change where unpredictable and more erratic weather conditions may become more frequent. It is appreciated that there a dearth in evidence-based literature on the use of microalgae as a bioindicator for monitoring impact of climate change and its potential effects in aquaculture.
- Whilst this present research has demonstrated interested findings, there is a need to pursue catchment based-studies that incorporates an extended number of locations and inter-laboratory evaluations for to improve technological rigor and stakeholder acceptance including policy-makers.
- Increasing NH<sub>3</sub> levels in the monitored fish ponds can be potentially toxic to fish that require further investigation.
- There are emerging opportunities for use of natural microalgae in the development of predictive environmental risk models that will help inform the quality status of water catchments, along with evaluating commensurate efficacy of intervention strategies, such as municipal wastewater treatment plants (Tahar et al., 2017).

Supplementary data to this article can be found online at https://doi. org/10.1016/j.scitotenv.2021.149800.

#### **CRediT authorship contribution statement**

**Emer A. O'Neill:** Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Project administration, Resources, Software, Validation, Visualization, Writing – original draft, Writing – review & editing. **Neil J. Rowan:** Conceptualization, Funding acquisition, Resources, Supervision, Writing – review & editing.

#### **Declaration of competing interest**

The authors declare that there are no competing interests or conflict of interest with respect to the publication of this article.

#### Acknowledgements

The authors would like to thank Athlone Institute of Technology (President Seed Fund), for providing financial support for this research. The authors would also like to acknowledge and thank Paul Kearney for access to his freshwater aquaculture facility in order to conduct this research.

#### References

- Alam, M.J.B., Islam, M.R., Muyen, Z., Mamun, M., Islam, S., 2007. Water quality parameters along rivers. Int. J. Environ. Sci. Technol. 4, 159–167.
- Bulkeley, H., Newell, P., 2015. Governing Climate Change. 2nd ed. Routledge Taylor & Francis Group, London.
- Bulletin of the Atomic Scientists, 2020. 2020 doomsday clock statement [WWW document]. URL. https://thebulletin.org/doomsday-clock/current-time/# (accessed 1.28.20).
- Chislock, M.F., Doster, E., Zitomer, R.A., Wilson, A.E., 2013. Eutrophication: causes, consequences, and controls in aquatic ecosystems | learn science at scitable. Nat. Educ. Knowl. 4, 10.
- da Silva, A.Q., Nilin, J., Santaella, S.T., Oriel, Bonilla, H., 2017. Ecotoxicological and physicochemical evaluation of an effluent of a shrimp farm located in Northeastern Brazil. Panam. I. Aquat. Sci. 12, 263–272.
- EPA, 2001. Parameters of Water Quality Interpretation and Standards. Johnstown Castle.
- FAO, 2018. The State of World Fisheries and Aquaculture Meeting the Sustainable Development Goals.
- Feucht, Y., Zander, K., 2015. Of earth ponds, flow-through and closed recirculation systems – German consumers' understanding of sustainable aquaculture and its communication. Aquaculture 438, 151–158.
- Fredricks, B.K.T., Jewell, S., Survey, U.S.G. 2015. Literature Review of the Potential Effects of Formalin on Nitrogen Oxidation Efficiency of the Biofilters of Recirculating Aquaculture Systems (RAS) for Freshwater Finfish.
- Garvey, M., Hayes, J., Clifford, E., Rowan, N., 2013. Ecotoxicological assessment of pulsed ultraviolet light-treated water containing species and Cryptosporidium parvum using a microbiotest test battery. Water Environ. J. 29 (1), 27–35.
- Guéguen, C., Gilbin, R., Pardos, M., Dominik, J., 2004. Water toxicity and metal contamination assessment of a polluted river: the upper Vistula River (Poland). Appl. Geochem. 19, 153–162.
- Hallegraeff, G.M., 2010. Ocean climate change, phytoplankton community responses, and harmful algal blooms: a formidable predictive challenge. J. Phycol. 46, 220–235.
- Han, P., Lu, Q., Fan, L., Zhou, W., 2019. A review on the use of microalgae for sustainable aquaculture. Appl. Sci. 9, 1–20.
- Intergovernmental Panel on Climate Change, 2020. Global Warming of 1.5 oC.
- Intergovernmental Panel on Climate Change (IPCC), 2021. AR6 climate change 2021. The physical science basis. https://www.ipcc.ch/report/ar6/wg1/ (accessed 16th August, 2021).
- Irish Statutory Office, 2009. S.I. No. 272/2009 European Communities Environmental Objectives (surface waters) regulations 2009 [WWW document]. URL. Electron. Irish Statut. B (accessed 4.5.19). http://www.irishstatutebook.ie/eli/2009/si/272/ made/en/print.
- Irish Statutory Office, 2019. S.I. No. 77/2019 European Union Environmental Objectives (surface waters) (amendment) regulations 2019 [WWW document]. 2019. URL. Electron. Irish Statut. B (accessed 1.14.20). http://www.irishstatutebook.ie/eli/2019/ si/77/made/en/pdf.
- ISO, 2012. ISO (8692:2012) Water Quality Fresh Water Algal Growth Inhibition Test With Unicellular Green Algae (No. 8692), ISO 8692:2012(en) Water Quality — Fresh Water Algal Growth Inhibition Test With Unicellular Green Algae, p. 3 Geneva, Switzerland.
- Ivanova, I., Groudeva, V., 2006. Use of selenastrum capricornutum growth inhibition test for testing toxicity of metal ions in soil and water. Biotechnol. Biotechnol. Equip. 20, 179–183.
- Jegatheesan, V., Shu, L., Visvanathan, C., 2011. Aquaculture Effluent: Impacts and Remedies for Protecting the Environment and Human Health. Encyclopedia of Environmental Health. Elsevier Science, pp. 123–135.

- Kadam, A.D., Kishore, G., Mishra, D.K., Arunachalam, K., 2020. Microalgal diversity as an indicator of the state of the environment of water bodies of Doon valley in Western Mimalaya, India. Ecol. Indic. 112. https://doi.org/10.1016/i.ecolind.2020.106077.
- Kolarevic, J., Baeverfjord, G., Takle, H., Ytteborg, E., Reiten, B.K.M., Nergård, S., Terjesen, B.F., 2014. Performance and welfare of Atlantic salmon smolt reared in recirculating or flow through aquaculture systems. Aquaculture 432, 15–25.
- Liu, X., Steele, J.C., Meng, X.-Z.Z., 2017. Usage, residue, and human health risk of antibiotics in Chinese aquaculture: a review. Environ. Pollut. 223, 161–169.
- Ma, J., Wang, S., Wang, P., Ma, L., Chen, X., Xu, R., 2006. Toxicity assessment of 40 herbicides to the green alga raphidocelis subcapitata. Ecotoxicol. Environ. Saf. 63, 456–462.
- Martinez-Porchas, M., Martinez-Cordova, L.R., Lopez-Elias, J.A., Porchas-Cornejo, M.A., 2014. Bioremediation of aquaculture effluents. Microbial Biodegradation and Bioremediation. Elsevier, London, pp. 542–555.
- Met Éireann, 2019. Monthly data [WWW document]. Irish Meteorol. Serv. https://www. met.ie/climate/available-data/monthly-data Miashiro, L., Lombardi, J.V., Mercante, C.T.J., 2012. Ecotoxicity assessment in aquaculture
- Miashiro, L., Lombardi, J.V., Mercante, C.T.J., 2012. Ecotoxicity assessment in aquaculture system using the test organism pseudokirchneriella subcapitata (Chlorophyceae). Acta Sci. Biol. Sci. 34, 373–379.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system. Aquaculture 526, 735377.
- Ngo, H.H., Guo, W., Tram Vo, T.P., Nghiem, L.D., Hai, F.L., 2016. Aerobic treatment of effluents from the aquaculture industry. Current Developments in Biotechnology and Bioengineering. Elsevier, Amsterdam, Netherlands, pp. 35–77.
- Omar, W.M.W., 2010. Perspectives on the use of algae as biological indicators for monitoring and predicting aquatic environments, with special reference to Malaysian freshwater ecosystems. Trop. Life Sci. Res. 21 (2), 51–67.
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218.
- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. Sci. Total Environ. 706, 136044.
- Ottinger, M., Clauss, K., Kuenzer, C., 2016. Aquaculture: relevance, distribution, impacts and spatial assessments - a review. Ocean Coast. Manag. 119, 244–266.
- Parmar, T.K., Rawtani, D., 2016. Bioindicators: the nature indicator of environmental pollution. Front. Life Sci. 9 (2), 110–118.
- Parus, A., Karbowska, B., 2020. Marine algae as natural indictor of environmental cleanliness. Water Air Soil Pollut. 231 (97). https://doi.org/10.1007/s11270-020-4434-0.
- Rabalais, N.N., 2002. Nitrogen in aquatic ecosystems. Ambio 31, 102-112.
- Rindi, F., 2014. Terrestrial green algae: systematics, biogeography and expected responses to climate change. In: Hodkinson, T.R., Jones, M.B., Waldren, S., Parnell, J.A.N. (Eds.), Climate Change, Ecology and Systematics. Cambridge University Press, New York, pp. 201–227.
- Rodgher, S., Espíndola, E.L.G., Simões, F.C.F., Tonietto, A.E., 2012. Cadmium and chromium toxicity to pseudokirchneriella subcapitata and Microcystis aeruginosa. Braz. Arch. Biol. Technol. 55, 161–169.
- Rowan, N.J., Pogue, R., 2021. Editorial overview: green new deal era current challenges and emerging opportunities for developing sustaining and disruptive innovation. Curr. Opin. Environ. Sci. Health https://doi.org/10.1016/j.coesh.2021.100294.
- Sarmaja-Korjonen, K., Seppänen, A., Bennike, O., 2006. Pediastrum algae from the classic late glacial Bølling Sø site, Denmark: response of aquatic biota to climate change. Rev. Palaeobot. Palynol. 138, 95–107.

Sikder, M.N.A., Min, W.W., Ziyad, A.O., Prem Kumar, P., Dinesh Kumar, R., 2016. Sustainable treatment of aquaculture effluents in future-a review. Int. Res. J. Adv. Eng. Sci. 1, 190–193.

- Stenevik, E.K., Sundby, S., 2007. Impacts of climate change on commercial fish stocks in norwegian waters. Mar. Policy 31, 19–31.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638.
- Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E., Rowan, N., 2018a. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3, 28.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018b. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, 1–22.
- Troell, M., Kautsky, N., Beveridge, M., Henriksson, P., Primavera, J., Rönnbäck, P., Folke, C., Jonell, M., 2017. Aquaculture. Reference Module in Life Sciences.
- Tsarenko, P.M., Bilous, O.P., Kryvosheia-Zakharova, O.M., Lilitska, H.H., Barinova, S., 2021. Diversity of algae and cyanobacteria and bioindication characteristics of the alpine Lake nesamovyte (Eastern Carpathians, Ukraine) from 100 years ago to the present. Diversity 13, 256. https://doi.org/10.3390/d13060256.
- Tschirner, M., Kloas, W., 2017. Increasing the sustainability of aquaculture systems insects as alternative protein source for fish diets. GAIA - Ecol. Perspect. Sci. Soc. 26, 332–340.
- Weisberger, M., 2020. Doomsday clock is now 100 seconds from midnight | live science. [WWW Document]. URL. https://www.livescience.com/doomsday-clock-update-2020.html (accessed 1.28.20).
- Zaghloul, A., Saber, M., Gadow, S., Awad, F., 2020. Biological indicators for pollution detection in terrestrial and aquatic ecosystems. Bull. Natl. Res. Cent. 44 (127). https://doi. org/10.1186/s42269-020-00385-x.

Contents lists available at ScienceDirect

Science of the Total Environment



journal homepage: www.elsevier.com/locate/scitotenv

# Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: Implications and opportunities for advancing research and disruptive innovation post COVID-19 era



SUSTAINABILIT OOD PRODU

### E.A. O'Neill<sup>a,\*</sup>, A.P. Morse<sup>b</sup>, N.J. Rowan<sup>a</sup>

Bioscience Research Institute, Technological University of the Shannon - Midlands and Midwest, University Road, Athlone, Ireland Department of Geography and Planning, School of Environmental Sciences, University of Liverpool, UK

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- · Integrated multi-trophic (IMTA) aquaculture offers a sustainable food process.
- · IMTA can be affected by extremes in weather/climate events.
- IMTA balance can be disrupted by emergence of toxigenic algae.
- Future need for in-farm real-time monitoring of IMTA systems

#### ARTICLE INFO

Article history Received 23 November 2021 Received in revised form 7 January 2022 Accepted 8 January 2022 Available online 14 January 2022

Editor: Damià Barceló

Keywords: Sustainable aquaculture climate change food security environmental indicators disruptive innovation paludiculture



Advancing wet peatland 'paludiculture' innovation present enormous potential to sustain carbon-cycles, reduce greenhouse-gas (GHG) gas emissions and to transition communities to low-carbon economies; however, there is limited scientific-evidence to support and enable direct commercial viability of eco-friendly products and services. This timely study reports on a novel, paludiculture-based, integrated-multi-trophic-aquaculture (IMTA) system for sustainable food production in the Irish midlands. This freshwater IMTA process relies on a naturally occurring ecosystem of microalgae, bacteria and duckweed in ponds for managing waste and water quality that is powered by wind turbines; however, as it is recirculating, it does not rely upon end-of-pipe solutions and does not discharge effluent to receiving waters. This constitutes the first report on the effects of extreme weather events on the performance of this IMTA system that produces European perch (Perca fluviatilis), rainbow trout (Oncorhynchus mykiis) during Spring 2020. Sampling coincided with lockdown periods of worker mobility restriction due to COVID-19 pandemic. Observations revealed that the frequency and intensity of storms generated high levels of rainfall that disrupted the algal and bacterial ecosystem in the IMTA leading to the emergence and predominance of toxic cyanobacteria that caused fish mortality. There is a pressing need for international agreement on standardized set of environmental indicators to advance paludiculture innovation that addresses climate-change and sustainability. This study describes important technical parameters for advancing freshwater aquaculture (IMTA), which can be future refined using real-time monitoringtools at farm level to inform management decision-making based on evaluating environmental indicators and weather data. The relevance of these findings to informing global sustaining and disruptive research and innovation in

Corresponding author. E-mail address: e.oneill@research.ait.ie (E.A. O'Neill).

http://dx.doi.org/10.1016/j.scitotenv.2022.153073

0048-9697/© 2022 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

paludiculture is presented, along with alignment with UN Sustainable Development goals. This study also addresses global challenges and opportunities highlighting a commensurate need for international agreement on resilient indicators encompassing linked ecological, societal, cultural, economic and cultural domains.

#### 1. Introduction

Unlocking the challenges of climate-orientated peatland management will enable conservation of their important carbon stocks (Wichmaan et al., 2016; Ziegler et al., 2021). However, some 15% of peatlands globally are degraded due to drainage-base agriculture and forestry that are key contributors to greenhouse gas emissions (GHG) (Uràk et al., 2017; Ziegler et al., 2021). The controlled rewetting of peatlands offers solutions to preventing GHG emissions and biodiversity loss (Ziegler et al., 2021), and presents opportunities for farming and innovative use of these rewetted peatlands that is termed 'paludiculture' (O'Neill et al., 2019; Ziegler et al., 2021). There is strong interest in developing commercial paludiculture activities including fuel, horticulture, aquaculture, and construction material; however, there is a need to address knowledge gaps that will influence the economic viability of paludiculture-derived products and services (Ziegler et al., 2021; O'Neill and Rowan, 2022). Factors affecting effective deployment of paludiculture are complex as fostering sustainable green innovation requires a holistic strategic response that can be effectively facilitated through the 'quadruple helix' approach of industry, academia, government and society (Rowan and Casey, 2021; Tan et al., 2021).

Key challenges affecting the development of paludiculture-based innovation include the lack of shared real-time systems of data generation for met-analysis, corporate strategy, risk-mitigation, and business disruption (Ziegler et al., 2021). There is a lack of consensus on international standardization of paludiculture innovation in terms of outcome sets, and agreement on sustaining measured variables that must encompass protecting biodiversity (Ziegler et al., 2021). Unlocking the disruptive potential of paludiculture innovation can be met by collaborate use of in-field environmental test-bed facilities (Rowan and Casey, 2021).

Aquaculture is the fastest growing food producing industry in the world that accounts for half of the fish produced globally for human consumption (Nielsen et al., 2016; Liu et al., 2017; O'Neill et al., 2019, O'Neill et al., 2020; O'Neill and Rowan, 2022). Farmed fish is considered to be a more efficient protein utilisation and feed conversion source than other animals destined for protein production (Tschirner and Kloas, 2017). According to the United Nations (UN), aquaculture now provides fish availability to countries and regions that would have previously been limited or nonexistent, often at affordable prices; thus, providing improved nutrition and food security (Fish Farming Expert, 2020; Rowan and Casey, 2021). The increasing interest in exploiting low-cost environmentally-friendly 'natural' processes in aquaculture has led to the timely development of integrated multi-trophic aquaculture systems or IMTA, along with accelerating efforts to implement eco-innovation and to improve the monitoring of traditional processes (Granada et al., 2016; Tahar et al., 2018a, 2018b; Naughton et al., 2020). However, advances in aquaculture must also be balanced with the commensurate need to meet commitments as set out by many European directives for environmental protection as well as the Water Framework Directive (WFD), where the latter aims to achieve good water status in all waters across all EU member countries (Voulvoulis et al., 2017; WFD Ireland, 2018).

In addition to these environmental concerns, extreme weather events, such as drought have displayed substantial effects on variances in stocking densities within aquaculture facilities therefore, successful stock assessments will be influenced by our ability to understand, monitor and to predict variances in weather and climate change on aquaculture ecosystem dynamics (Rowan and Pogue, 2021; O'Neill and Rowan, 2022). An increasing number of studies have intimated that climate vulnerability and climate change can have adverse impacts on global food production and food security (Iizumi and Ramankutty, 2015). These authors reported that ongoing climate change, and associated variances in intensity, frequency, and

duration of weather/climate extremes, in conjunction with growing populations and dietary requirements, further complicate this drive to improve food sustainability and security. Therefore, stakeholders including policymakers, urgently require better estimates of the likely incidence of extreme weather events, their impact on food production and security, and the commensurate consequential impact in terms of socio-economic losses (Chavez et al., 2015). There is a pressing need to address the uncertainties in climate model predictions not only over years, but also at regional and local scale to inform efficacy of food production systems, and interventions. Mowi Ireland, a coastal aquaculture farmer, recently lost approximately 80,000 salmon due to toxigenic plankton bloom where the rearing water was 13 °C compared to typical 11.5 °C, which was potentially attributed to climate change (Moore, 2021).

This constitutes the first study to report on the potential effects of extreme weather events on the technical performance of a novel peatlands-based IMTA process located in the Republic of Ireland that lead to fish mortality, which coincided with occurrence of COVID-19 pandemic. It describes the worldwide relevance of data generated by using a Quadruple Helix (academia-industry-government-society) approach to advancing paludiculture innovation along with challenges and opportunities. It describes measurable technical and environmental variables for the development of this IMTA system that embraces extreme weather events; it describes smart tools to inform technical, policy and societal readiness level of paludiculture innovation aligned with U.N.s' sustainable development goals.

#### 2. Materials and methods

#### 2.1. Sampling

Oasis fish farm is an innovative peatland cut-away integrated multitrophic aquaculture (IMTA) system process set in the middle of Mount Lucas Wind Farm, Co. Offaly (53°17'3" N – 7°11'45" W). This IMTA holds European perch (Perca fluviatilis), rainbow trout (Oncorhynchus mykiis), common duckweed (Lemna minor) and gibbous duckweed (Lemna gibba) and exploits use of microalgae for waste removal. The aquaculture system consists of four split (pill) ponds connected to an algae and duckweed lagoon with 16 channels serving as a treatment system. Fish are kept at a density that does not exceed the organic farming standard (e.g., <20 kg/m<sup>-3</sup> for perch), using screens at the D-ends of each split pond. The space between two D-end fish culture areas is also used to treat waste with free living algae in suspension. Flow in each split pond is generated and water is circulated using an airlift. Each D-end fish culture area is equipped with oxygen and temperature probes connected to paddlewheels to provide extra oxygen when necessary. The farm is designed to hold a maximum of 32,000 kg of fish.

Water samples were collected from the Oasis farm in five liter octagonal carboy HDPE bottles (Lennox) and transported in insulated cool boxes directly to the lab, 62 km away, via car. Samples were collected every two weeks from December 2019 to February 2020 and then once a week until October 2020. Samples were taken from each on the culture ponds, the entry and exit points of the duckweed lagoons and the reservoir during this study. Samples were also taken from the overflow tank during times when there may have been a potential for discharge. Refer to Fig. 1 for the locations of all sample points within the farm.

#### 2.2. Physicochemical analysis

Good water quality is critical for the cultivation of fish as well as for the receiving ecosystems attached to aquaculture facilities. Analysing the physicochemical parameters is the most common method of determining the



Fig. 1. Aerial view of Oasis Fish Farm, located in Ballycon, Co. Offaly. The culture ponds, water reservoir, algae & duckweed wastewater treatment channels, overflow tank and bog river are all visible. Blue lines indicate the direction of the flow of water. The green squares indicate all sampling points within the farm to monitor the IMTA process.

current water quality (Shukla et al., 2013). No definite set of physicochemical parameters specific for Irish aquaculture water and wastewater could be found. Therefore, the physicochemical parameters routinely monitored in Irish fish farms to assess water quality were applied. Additionally, a range of previous studies conducted on aquaculture facilities across the world were researched. The range of parameters investigated in these studies were also applied to this research. As no values for the individual physicochemical parameter levels could be found in relation to Irish aquaculture, the standard Irish EPA water quality parameters based on the Freshwater Fish Directive [78/659/EEC], Surface Water Regulations [1989] and surface water regulations [SI 272 of 2009] and amendments [SI 77 of 2019] were used as guidance (EPA, 2001; Irish Statutory Office, 2009; Irish Statutory OfficeOffice, 2019).

Water parameters; temperature, pH, ammonium  $(NH_4^+)$ , nitrite  $(NO_2^-)$ , nitrate  $(NO_3^-)$ , orthophosphate  $(PO_4^{3^-})$ , dissolved oxygen (DO), biochemical oxygen demand (BOD), suspended solids (SS), dissolved solids (DS), hardness, and alkalinity – were investigated in the laboratory within 24 h of collection to prevent the need for preservation. The Standard Methods for the Analysis of Water and Wastewater were used for all of the parameters employed. See Table 1. Spectroquant® photometric kits were used to assess the  $NH_4^+$ ,  $NO_2^-$ ,  $NO3^-$  and  $PO_4^{3^-}$ . Analysis was conducted as per the manufacturer's instructions. Absorbance was analysed using a Shimadzu UV-2250 spectrophotometer. Temperature, pH, dissolved solids and conductivity were analysed using a VWR pHenomenal<sup>TM</sup> MU 6100 L meter, VWR 111662–1157 pH probe and VWR CO11 conductivity probe. DO and BOD<sub>5day</sub> were analysed using a Jenway 9500 DO<sub>2</sub> meter and probe. Suspended solids were analysed via filtration using a Buchner flask,

Buchner funnel and Whatman 0.45  $\mu$ m pore membrane filter. Hardness was assessed via titration using pH 10 buffer, Erichrome black and EDTA. Alkalinity was analysed by titration using phenolphthalein indicator, methyl orange indicator and hydrochloric acid.

#### Table 1

Summary of all physicochemical methods applied to this research. All parameters/ variables, their respective methods, standard analysis of water and wastewater method numbers and the detection limits for parameters where photometric test kits were employed have been included.

Parameter/variable	Analytical method	Standard analysis method number	Detection limit (mg $L^{-1}$ )
Alkalinity	Titrimetric	2320-В	-
Ammonium (NH <sub>4</sub> <sup>+</sup> )	Photometric	4500-NH <sub>3</sub> -F	0.013-3.86
			2.6-193.0
Biochemical oxygen	Membrane electrode	5210-В	-
Disastand array (DO)	No 1	4500.0.0	
Dissolved oxygen (DO)	Membrane electrode	4500-0 G	-
Dissolved solids (DS)	Electrode	2540-C	-
Hardness	Titrimetric	2340-C	-
Nitrate $(NO_3^-)$	Photometric	4500-NO <sub>3</sub>	0.4-110.7
Nitrite $(NO_2^-)$	Photometric	345–1	0.007-3.28
Orthophosphate (PO <sub>4</sub> <sup>3-</sup> )	Photometric	4500-P-C	0.007-15.3
			1.5-92.0
pН	Membrane electrode	2310-В	-
Suspended solids (SS)	Gravimetric	2540-D	-
Temperature	Thermometer	2550-В	-
#### 2.3. Algae enumeration

Algae cells were both manually and automatically counted. All manual enumeration was conducted using a Superior Marienfeld Neubauer Improved Haemocytometer (0.1 mm, 0.0025 mm<sup>2</sup>, Tiefe Depth Profondeur No: 717810) and a Nikon YS100 light microscope. The Miltenyi Biotec MACSQuant® Analyser 10 Flow Cytometer (FCM) was used for the automated enumeration of the algae. Preparation of phytoplankton samples for flow cytometry was adapted from Naughton et al. (2020). A ten millimeter aliquot of each sample preserved with Lugols Iodine were centrifuged at  $3500 \times g$  for 20 min. The supernatant was removed and the algae pellet was re-suspended in flow buffer. The flow buffer was prepared by adding 1 mM EDTA, 0.2% Tween and 0.1% NaN<sub>3</sub> to 1 L phosphate saline buffer or PBS (Merck). The buffer was filtered using a 0.20 µm filter (Sigma-Aldrich) to remove impurities which may interfere with the flow cytometer. The resuspended sample was divided into two aliquots (one three milliliter aliquot and one seven milliliter aliquot). The three milliliter aliquot was

used for the unstained negative control samples. The seven milliliter aliquot was used for the stained samples. Two hundred microliters of 10X SYBR Green was added to the seven milliliter aliquot and incubated for 15 min in the dark at room temperature. Using two milliliter Eppendorf's (Merck), 1.5 mL from the unstained aliquot and three 1.5 mL's of the stained aliquot were centrifuged at  $3500 \times g$  for 15 min. The supernatant was removed and the pellets were re-suspended in 1.5 mL of fresh flow buffer. Samples were then loaded onto a round-bottomed 96 well plate. Two hundred microliters of each aliquot was loaded onto the plate i.e., four wells containing one unstained and three (triplicate) stained aliquots were loaded for each sample.

The instrument was set to uptake 100  $\mu$ L of each sample for analysis. The trigger point for the FSC laser was set at 1.0 to eliminate the detection of as much debris as possible in the samples. The FlowJo<sup>TM</sup> v10.7 software program was used for the analysis of the data generated from the FCM. Gating was used to enumerate the algal and cyanobacterial populations. The gating method was adapted from Haynes et al. (2016), Moorhouse et al.



**Fig. 2.** Flow cytometry dot diagrams for the enumeration of algae and cyanobacteria. A) Unstained samples to eliminate autofluorescence interference, B) cells stained with SYBR Green for the enumerations of algal and cyanobacterial populations, C) chlorophyll and phycocyanin levels used to distinguish between algae and cyanobacteria, D) enumeration of both the algae and the cyanobacteria populations.

(2018), Naughton et al. (2020) and Read et al. (2014). The unstained sample (negative control) was first gated (Fig. 2A) to eliminate as much autofluorescence interference as possible. This gate was then applied to the stained samples (Fig. 2B) in order to identify and enumerate the cells present in each sample. As per Moorhouse et al. (2018) and Naughton et al. (2020), the blue B3 channel, which was used to identify chlorophyll positive cells, was plotted against the red R1 channel, which was used to identify phycocyanin positive cells to distinguish between algae and cyanobacteria. (Fig. 2C). Enumeration of the algal and cyanobacterial populations were then established (Fig. 2D).

#### 2.4. Statistical analysis

All statistical analysis and construction of dose response curves, standard curves, etc. were performed on GRAPHPAD PRISM 7, 8 and 9, and MINITAB 18 and 19. The data generated were grouped and subject to normality tests (Anderson-Darling), to determine if samples were from a normal distribution (p > 0.05 = normal distribution). This in turn would establish whether parametric or non-parametric testing was to be conducted on results. As there was normal distribution, parametric testing was applied. *t*-tests and ANOVA were used to determine if any significant



**Fig. 3.** Breakdown of all physicochemical parameters investigated on the novel trial fish farm between December 2019 and October 2020. Parameters investigated were A) ammonium and nitrite, B) nitrate and orthophosphate, C) dissolved oxygen and biochemical oxygen demand, D) pH and Temperature, E) suspended and dissolved solids, and F) hardness and alkalinity. S.D. indicated, n = 8. Samples missing from April and May 2020 due to COVID-19 lockdown and restrictions in the Republic of Ireland.

differences were observed in the variables (p < 0.05 = significant difference). Unpaired tests were used as different sets of samples were analysed to assess the quality of the aquaculture water samples. For the correlation studies, the Pearson's correlation coefficient (r) was used to determine whether any relationships existed between any of the parameters investigated.

#### 3. Results & discussion

# 3.1. Effects of extreme weather variance on the technical performance of a new recirculating IMTA process in the Irish peatland

This timely study describes the effects of extreme weather events experienced in Spring period of 2020 on the performance of a fully-integrated multi-trophic aquaculture process developed in the Irish peatlands. This IMTA process uses a balanced ecosystem of naturally-occurring microalgae, bacteria and duckweed to regulate waste and maintain water quality; in addition, it uses wind turbines as a renewable source of energy to operate aeration systems in the circulatory freshwater aquaculture ponds (O'Neill et al. 2020). This study characterises this IMTA process by way of physicochemical and algae monitoring over a year-long case study. As no statistically significant differences were observed between each of the sampling points results were averaged for ease of reporting (refer to Fig. 3.).

#### 3.1.1. Physicochemical analysis

Prior to the first COVID-19 national lockdown in the Republic of Ireland in March 2020, very little to no NH<sup>+</sup><sub>4</sub> levels were detected within the IMTA farm. This was believed to be due to a combination of issues associated with cyanobacteria levels and increased levels of rainfall experienced in February 2020. NH<sup>+</sup><sub>4</sub> levels increased across all sampling points from the end of July 2020 with spikes of up to 0.90 mg NH<sup>+</sup><sub>4</sub> L<sup>-1</sup> observed at the beginning of October 2020. However, levels did not rise to greater than the guidance values of 1 mg NH<sup>+</sup><sub>4</sub> L<sup>-1</sup> (EPA, 2001). Additionally, no NH<sup>+</sup><sub>4</sub> was detected at the discharge point during times of possible overflow and release indicating no potential issues associated with NH<sup>+</sup><sub>4</sub> for the receiving peatlands. Levels of NH<sup>+</sup><sub>4</sub> also decreased between the culture ponds and the duckweed lagoon (treatment lagoon) suggesting that the treatment process was effective at reducing NH<sup>+</sup><sub>4</sub> levels in wastewater.

Before the lockdown period, NO<sub>2</sub><sup>-</sup> levels fluctuated between 0 mg NO<sub>2</sub><sup>--</sup> L<sup>-1</sup> and 0.03 mg NO<sub>2</sub><sup>-</sup> L<sup>-1</sup>. However, levels increased greatly after this period, with concentrations spiking to between 0.25 mg NO<sub>2</sub><sup>-</sup> L<sup>-1</sup> and 0.30 mg NO<sub>2</sub><sup>-</sup> L<sup>-1</sup> in mid-July and mid-September, respectively. This was a tenfold increase on previous levels as well as being tenfold greater than the guidance value of 0.03 mg L<sup>-1</sup> (EPA, 2001). NO<sub>2</sub><sup>-</sup> is highly toxic to aquatic life (O'Neill et al., 2019; Pollice et al., 2002) but is extremely unstable and would not remain in this form for long as it would be quickly transformed to NO<sub>3</sub><sup>-</sup> (Durborow et al., 1997; O'Neill et al., 2019; O'Neill et al., 2020). As no overflow and release occurred during the times of high levels, the NO<sub>2</sub><sup>-</sup> would not cause issues within the bog.

 $NO_3^-$  levels dropped considerably the month prior to lockdown (February 2020) going from >8 mg  $NO_3^-$  L<sup>-1</sup> to 0 mg  $NO_3^-$  L<sup>-1</sup>. This coincided with changes in weather conditions and excessive rainfall experienced throughout the month. Once analysis recommenced after the lockdown period,  $NO_3^-$  levels slowly increased reaching levels >8 mg  $NO_3^-$  L<sup>-1</sup> in September before dropping back to between 2 mg  $NO_3^-$  L<sup>-1</sup> and 4 mg  $NO_3^-$  L<sup>-1</sup> in October. Levels were well below the guidance value of 50 mg L<sup>-1</sup> (EPA, 2001). The increased levels in  $NO_3^-$  observed in this study may be due to the increased levels of  $NO_2^-$  also observed.

 $PO_4^{3-}$  levels were observed across all sampling points within the farm. Statistical analysis demonstrated no significant difference between the different sampling points (p = 0.2160).  $PO_4^{3-}$  levels detected were above the guidance value of <0.035 mg L<sup>-1</sup> (Irish Statutory OfficeOffice, 2019, 2009) indicating there may be potential issues within the farm and additional treatment process to reduce  $PO_4^{3-}$  levels may need to be considered as the algae and duckweed lagoon is not effectively removing it.

Variations in DO levels and BOD levels were observed across the eight sampling points. No statistically significant differences were observed across both parameters (DO p = 0.1421, BOD p = 0.5464). DO levels fluctuated between 4 mg  $O_2 L^{-1}$  and 10 mg  $O_2 L^{-1}$ . The recommended DO concentration present in salmonid waters is  $\geq 9$  mg  $O_2 L^{-1}$  and in cyprinid waters is  $\geq 7$  mg  $O_2 L^{-1}$  (EPA, 2001). Levels continually increased and decreased above and below these recommended levels. However, they did not drop below the threshold of 4 mg  $O_2 L^{-1}$  required for sufficient maintenance of aquatic life (Alam et al., 2007; da Silva et al., 2017; O'Neill et al., 2020).

The SI 272/2009 and SI 77/2019 recommended a mean BOD concentration of 1.30 mg  $O_2 L^{-1}$  for high water status and 1.50 mg  $O_2 L^{-1}$  for good water status (Irish Statutory Office, 2009; Irish Statutory OfficeOffice, 2019). However, the EPA suggested  $\leq 3 \text{ mg O}_2 \text{ L}^{-1}$  and  $\leq 6 \text{ mg } O_2 \text{ L}^{-1}$  for salmonid and cyprinid waters, respectively (EPA, 2001). Issues were indicated with the BOD levels observed across all sampling points. The BOD is caused by microorganisms using O2 when consuming organic matter therefore organic matter needs to be reduced in order to decrease BOD levels (EPA, 2001; Gupta et al., 2017; Kasuya et al., 1998; Lee and Nikraz, 2015; Mcintosh and Fitzsimmons, 2003; Sultana et al., 2017). Increasing O<sub>2</sub> levels and the addition of filtration to remove some of the organic matter have been found to decrease BOD levels (Gupta et al., 2017; Lee and Nikraz, 2015). Work is ongoing to reduce BOD levels. However, as no water was released from the farm during times of increased BOD levels, no concerns associated with this issue affecting the surrounding peatland habitat were foreseen.

Fluxes in the levels of suspended solids were observed across all sampling points, while dissolved solid concentrations remained more consistent throughout the study. The levels of suspended solids observed throughout the summer months (June, July, August) were well above the guidance value of 25 mg  $L^{-1}$  (EPA, 2001) reaching highs of >120 mg  $L^{-1}$ . Given that suspended solids can cause gill irritation, signs of which were observed in some of the fish, this was considered to be a major issue. It was also believed that this issue was linked to the issues with BOD previously mentioned. After filtration methods were applied to different areas of the farm, suspended solid levels dropped back to below the MAC level by September and remained so until the end of the study. Dissolved solid concentrations observed in the study were well below the suggested concentration of <300 mg  $L^{-1}$  for excellent water status (WHO, 2003).

Fluctuations were indicated in the temperature range and the pH range observed across all of the sampling points. The elevations in temperature were observed between June and September as would be expected given the season (summer). Although no specific guidance value for temperature could be established as all species of fish have a slightly different optimum temperature, any water released into an aquatic system must be <20 °C (EPA, 2001). Temperatures were >20 °C only once in mid-July. However, as water was not released from the system, this was not deemed to be an issue. Recommended pH levels of between pH 6 and pH 8 were suggested (EPA, 2001; Irish Statutory Office, 2009; Irish Statutory OfficeOffice, 2019). The pH levels remained within this range throughout the study. The pH levels were just below pH 8 however, levels dropped to just above pH 7 from August 2020 to September 2020. Although levels remained within the recommended range, the alteration in pH levels may have had an effect on the BOD issues observed in the farm. Alterations in pH can decrease the rate of organic removal rates thus affecting BOD measurements (Mukherjee et al., 1968).

CaCO<sub>3</sub> levels were measured in the eight sampling points in order to determine hardness and alkalinity levels. Statistical analysis was conducted for each parameter and no significant differences were observed (hardness p = 0.5237, alkalinity p = 0.4806). Hardness levels observed suggested that the water was slight to moderately hard. This correlated with water hardness maps of Ireland which demonstrated water in the midlands around Co. Offaly were also slight to moderately hard. It has been suggested that fish prefer a minimum of 20 mg CaCO<sub>3</sub> L<sup>-1</sup> alkalinity levels. Levels recorded within the farm remained above this optimum threshold throughout

the study (Boyd and Tucker, 2015; EPA, 2001). Alkalinity levels observed in this study demonstrated similar results to those reported by Stephens and Farris (2004b).

All physicochemical findings were then compared to previous research. Results were similar to those observed in other studies with the exception of NO<sub>2</sub>, suspended solids and BOD (Boaventura et al., 1997; Camargo, 1994; Cao et al., 2007; Caramel et al., 2014; Costanzo et al., 2004; da Silva et al., 2017; Fadaeifard et al., 2011; Guilpart et al., 2012; Lalonde et al., 2014; Moreira et al., 2010; Namin et al., 2013; Noroozrajabi et al., 2013; O'Neill et al., 2019; Pulatsü et al., 2004; Stephens and Farris, 2004a, 2004b; Ziemann et al., 1992; Živić et al., 2009). A ten-fold increase was detected in the NO2-levels and suspended solids levels were higher than the majority of the previous studies, as too were BOD levels. Correlations were indicated between BOD levels and a range of parameters including pH, temperature, dissolved oxygen, alkalinity and suspended solids. This demonstrated the importance of maintaining high DO levels as oxygen is vital for the BOD process. Abnormal or irregular pH levels, which were observed for a time in the farm, can decrease the rate of removal of organic compounds which affect BOD levels. By proxy, changes in alkalinity will also have an impact (Chinedu et al., 2015). Small amounts of all suspended solids are considered volatile suspended solids and exert greater pressures on the oxygen demand thus increasing BOD levels (Gerardi and Lytle, 2015). Finally, as temperatures increase so too does BOD removal rates as higher temperatures enhance microbes respiration rates (Lim et al., 2001). The range of correlations with BOD has demonstrated how complex the process is and may be why issues were encountered in controlling the BOD levels within the farm.

#### 3.1.2. Algae & cyanobacteria analysis

Lower levels of algae were observed during the winter months which is to be expected as temperatures are lower and less sunlight is experienced. The spring month displayed a rise in levels which corresponded with the increase in light and temperature. However, a drop in algal numbers were observed just prior to the first COVID-19 national lockdown that occurred in March 2020. This drop was most likely due to excessive levels of rain fall experienced during the month of February. Algae numbers consistently remained between  $1 \times 10^5$  and  $5 \times 10^5$  cells mL<sup>-1</sup> after the lockdown period until the end of the study. This suggested the stabilisation had occurred. Moderately strong correlationships were observed with most of the nitrogen nutrients indicating that  $NH_4^+$ ,  $NO_2^-$  and  $NO_3^-$  play a vital role in maintaining optimum algae levels in the novel IMTA process. Results also found that the higher and more stable the levels of  $NO_3^-$  present, the more stable the algae numbers. Given that  $NO_3^-$  is algae's preferred form of nutrient, and NH<sub>4</sub><sup>+</sup> and NO<sub>2</sub><sup>-</sup> are necessary for the natural production of NO<sub>3</sub><sup>-</sup> via the nitrification process, this result was expected.

Cyanobacteria levels were also monitored in parallel to the algal numbers, as shown in Fig. 4. Although many species of cyanobacteria can provide beneficial elements (e.g., Spirulina) the presence of increased levels cyanobacteria was found to have a negative impact on the novel IMTA system. Increased incidences of mortality were observed as cyanobacterial levels rose. Much like with freshwater bodies, cyanobacteria numbers were always present in the system. They remained below the level of algae being reported highlighting algae's ability to control cyanobacteria levels as both are competing for the nitrogen nutrient source. However, cyanobacteria levels were found to increase just before the lockdown period, demonstrating an inverse relationship with the algae. This suggested that the cyanobacteria were out competing the algae for nutrients. Again, this coincided with extreme weather conditions. Once levels stabilised after the lockdown period, they were once again consistently below the level of algae, remaining between  $1 \times 10^4$  and  $1 \times 10^5$  cells mL<sup>-1</sup>. Reducing in mortality levels also coincided with the stabilisation of cyanobacteria levels.

Correlation studies were conducted between the algae and cyanobacterial levels and all of the physicochemical parameters. In addition to a correlation observed between the algae and the cyanobacteria themselves, correlations were observed between both counts and a range



**Fig. 4.** Cell counts in cells  $mL^{-1}$  of algae (bar) and cyanobacteria (line) observed throughout the novel trial fish farm between December 2019 and October 2020. Cell counts conducted via flow cytometry. S.D. indicated, n = 8. Samples missing from April 2020 due to COVID-19 lockdown in the Republic of Ireland. Red line indicates when excessive levels of rainfall were observed.

of physicochemical parameters. A correlation was observed with the N parameters ( $NH_4^+$ ,  $NO_2^-$  and  $NO_3^-$ ) as well as with the P ( $PO_4^{3-}$ ). As N and P are both necessary for algal growth this was expected. It also highlighted the need to ensure these nutrients were present in the system to ensure the continued presence of algae that was necessary for the novel IMTA process to be effective. The correlation between the pH and algae was also expected as it is well known that although algae can tolerate small fluctuations in pH, increased and more frequent fluctuations can slow down growth rates (Dubinsky and Rotem, 1974).

#### 3.1.3. Climate variance

During this research process, algae demonstrated the potential to be used as an early warning indicator for highlighting issues associated with climate change having experiences a range of different weather conditions including flooding conditions in 2020 where record levels of rain fell due to the development of a number of storm systems in close proximity to one another. February 2020 was one of the wettest on record for the Republic of Ireland. This high level of rainfall was as a result of two extratropical cyclone storms hitting Ireland in that month and in close proximity to one another. Storm Ciara (formed 7th February 2020, dissipated 16th February 2020) and Storm Dennis (formed 11th February 2020, dissipated 18th February) affected Ireland less than a week apart. Just after this weather event cyanobacteria levels began to rise within the farm, as shown in Fig. 4, as well as fish mortalities (up to 44%) which had up until that point remained consistently low (  $\Box$ 3%). Veterinary post-mortems found signs of hepatotoxicity (liver necrosis) and high instances of gill irritation. Unfortunately, the first lockdown period began in March 2020 and as a result no samples could be analysed after this point until May 2020. Literature searches were conducted remotely to find an appropriate action of reducing or removing the cyanobacterial levels without having additional consequences on the fish health. Both Iredale et al. (2012) and Rajabi et al. (2010) demonstrated successful cyanobacterial control and removal from freshwater bodies with the application of barley straw. This method was suggested and then subsequently applied to the culture ponds at which point the farm reported a reduction in mortalities and cyanobacterial levels. Findings from the overarching Bord Iascaigh Mhara (BIM, 2020) project subsequently revealed that 3402 species of bacteria and microalgae were occurring in the IMTA ecosystem that were identified using next generation sequencing and bioinformatics, of which 1864 are algal. This demonstrated the complexity of the study given the high level of algal diversity within the farm. Of these 1864; 1551 species or sub-species of algae were identified across 210 genera, 60 were identified on family as opposed to genus or species, 42 were classified as uncultured.

In order to determine whether the excessive levels of rainfall indirectly caused issues in February and the lack there of caused issues in May, Met



Fig. 5. Map of Ireland indication the approximate location of Oasis fish farm (53°17′03″ N, 07°11′45″ W) indicated with orange, and the three closest Met Eireann weather stations (Gurteen – 53°02′24″ N, 08°00′36″ W; Oak Park – 52°51′36″ N, 06°55′36″ W; Mullingar – 53°33′36″ N, 07°20′24″ W) surrounding the farm, indicated by yellow.

Éireann metadata was once again investigated. The three closest weather stations surround Oasis fish farm at Mount Lucas were used (Mullingar, Co. Westmeath; Oak Park, Co. Carlow; and Gurteen, Co Tipperary), as indicated in Fig. 5. The Irish midlands traditionally get an average of 70.3 mm of rainfall for the month of February. However, according to Met Éireann, 197.7 mm of rainfall fell for that month (Met Éireann, 2021), as shown in Fig. 6. This subsequently would have diluted down all nutrient levels and reduced algal / cyanobacterial numbers within the farm. As nutrients build back up, the ammonium, which is cyanobacteria's preferred source of nutrients, is used by the cyanobacteria before it has a chance to be converted to nitrate via the nitrification process (Herrero et al., 2001) which is the algae's preferred form. This in turn allows the cyanobacteria to grow and out compete the algae. Normally the higher levels of algae and increased levels of nitrate control the levels of cyanobacteria. Leachate or runoff from the bog itself many also be contributing to the issues however no

studies on this have been conduct to date. Cyanobacteria levels rose again in May with similar levels or mortality begin observed again. However, there were no instances of rainfall during this instance (Fig. 6.). In fact, May 2020 was considered one of the driest in recent year's (Met Éireann, 2021). No physicochemical analysis could be conducted at this time due to COVID-19 restrictions still being in place. Again, Veterinary postmortems found signs of hepatotoxicity and gill irritation. The Barley straw had been removed from the ponds prior to this event and was therefore though to be the main cause for the problem. It may also be as a seasonal event as cyanobacteria have been known to "bloom" during the spring and early summer months as temperatures increase. However, as this was the first study ever conducted on a peatland IMTA system additional research and analysis would need to be conducted.

Overland flow of water from rainfall events can impact on infrastructure and cause flooding, which is different to riverine flooding. Overland flow



Fig. 6. Average A) rainfall and B) temperature recorded for 2019 (blue) and 2020 (yellow) at three Met Eireann weather stations surrounding the Oasis fish farm during the sampling period of December 2019 to October 2020. Stations were located at 1) Mullingar, Co. Westmeath, 2) Oak Park, Co. Carlow and 3) Gurteen, Co Tipperary. Stations were located north, south-west and south-east of the fish farm, respectively.

#### Table 2

Relevance of this 'paludiculture-based' IMTA system to enabling the advancement of adjacent and emerging worldwide cross-cutting research and innovation including UN SDGs.

Core topics and enabling innovation, products and services	UN SDG number
<ul> <li>Paludiculture 'wet peatland' innovation</li> <li>*Collaborative advancement and standardization of innovation for direct economic viability including horticulture, food production and bio-based industry (Tan et al., 2021; Ziegler et al., 2021)</li> <li>*Ecotoxicology, biodiversity and safety evaluation of emerging innovation and services (Rowan and Galanakis, 2020; Wan Mohtar et al., 2021; Usuldin et al., 2021)</li> <li>*Sustainable Carbon Cycles - supporting the European Commission in reaching climate neutrality through specific focus on quality, credibility and certification of carbon removals in land sector using IMTA system (European Commission, 2021).</li> <li>*Development of novel multimodal sensing technologies, IoT devices, Microsoft Azure Cloud and AI models to inform effective paludiculture innovation and management including carbon reduction (Science Foundation Ireland (SFI), 2021)</li> <li>*Ecosystem Service Management and Pollination, including social enterprises (Goblirsch et al., 2021)</li> <li>*Biorefinery of bio-based products from IMTA-generated microalgae for one-health applications (Wichmaan et al., 2020; Rowan and Pogue, 2021; McKeon-Bennett and Hodkinson, 2021) and bioplastics (Silva, 2021)</li> <li>*Real-time monitoring, testing and development of in-field technologies with sophisticated laboratory equipment, such as flow cytometry with in-field algaTorch™ (Naughton et al., 2020)</li> <li>*Nexus between water – food – energy for fisheries development including development of LCA, MFA and PCA tools for informing sustainability of products and services (Ruiz-Salmón et al., 2020a,b)</li> <li>*Social marketing, awareness and stakeholder behavioural change towards low-carbon food security including circular bioeconomy (Domegan, 2021)</li> <li>*Effective communication strategies for seafood to promote health and sustainability (Sacchetttini et al., 2021)</li> <li>*Sustainable recirculating alternatives to traditional 'end-of-pipe' techno</li></ul>	1. 2. 3. 6. 7. 8. 9. 11. 13. 14. 15.
Quadruple helix hub concept and approach to sustainable research and innovation *Trans-regional development of Quadruple Helix Hubs (industry-government-academia-society) for advancing low carbon innovation and supporting communities and stakeholders (Rowan and Casey, 2021) *Development of living lab linked to environmental test beds for advancing research, enterprise, entrepreneurship and innovation with nexus to education, training and job creation (Li et al., 2018); Blue Hatch Aquaculture Accelerator Program, 2020: Rowan and Casey, 2021; Rowan and Galanakis, 2020) *Investing and enabling eco-innovation including green finance, design thinking and high potential start up accelerator investment linked to stage gate needs (Rowan and Pogue, 2021) *Early-technical validation (test-the-tech, experimentation/validation in pre-pilot); scaling to real-life setting *Transnational modelling and cluster development including forging European Innovation HUBs such as (Sharebiotech, 2021) and Regional University Network-European Universities (RUN-EU, 2020).	2. 4. 5. 9. 11. 12. 13.
Climate strategy, risk management, awareness and transparency *Publically-backed climate risk insurance for farmers that considers weather indexing for protection against extreme weather events (Doherty et al., 2021). *Risk modelling and assessment of key variables affecting asset and innovation performance and sustainability including use of EPA's SPR model (Tiedeken et al., 2017; Tahar et al., 2017; Tahar et al., 2018) *Supporting sustaining and disruptive innovation and services that also embraces technological, political and societal readiness levels (Geels, 2018; Rowan and Casey, 2021; Schuelke-Leech, 2021) *Current limited short-term focus on risk management, transparency post review of 1168 companies. Also, a lack of understanding of climate	4. 9. 11. 12. 13.

#### Table 2 (continued)

Core topics and enabling innovation, products and services	UN SDG
	number
risk with few companies having climate strategies (Coppola et al	
2019)	
*International consensus on best indicators reflecting climate	
resilience, economy, society and culture (Barry and Hoyne, 2021).	
COVID-19 pandemic	1.
*Food and supply chain disruption including research provision (Guan	2.
et al., 2020; Herreor et al., 2020; Galanakis et al., 2021; Rowan and	3.
Galanakis, 2020)	9.
*Biorefining alternative sustainable materials for multiple-use PPE	
(Rowan and Laffey, 2020a,b; Rowan and Moral, 2021)	
*Food security and safety (Galanakis et al., 2021)	
International policy and cohesiveness	1.
*Informing EU Green New Deal; UN Sustainable Development Goals;	2.
Paris Climate Agreement (Galanakis et al., 2022)	4.
	9. 11
	16
	10.
Climate change and action plans	3.
*Weather events and disease mitigating simulation and predictive	4.
models (Jenkings and Kane, 2019)	13.
*Development of in-field biosensors (and digital-sensors) to monitor	14.
performance of agri-food security linked to extreme weather events	15.
(Naughton et al., 2020; O'Neill and Rowan, 2022).	
*Informing multi-disciplinary and cross-functional international	
projects to seek agreement on climate change indicators that addresses	
ecosystem resilience with nexus to economic and sociocultural	
indicators (Barry and Hoyne, 2021).	
Divited transformation	1
*Development of dropes and satellites. IoT devices. AI and	1.
cloud-enabled technologies to monitor carbon sequestration of	3
peatlands and other land types to improve understanding on human	4.
activity on land use and how it relates to climate change (Science	5.
Foundation Ireland (SFI), 2021)	6.
*ArcGIS mapping to inform restoration and rehabilitation of peatlands	7.
linked to ecology/biodiversity to inform carbon cycles (Rowan and	8.
Casey, 2021) *Development of digital twin that includes ARVR	9.
(Quality of Experience) for remote specialist virtual training (Braga	10.
Rodriguez et al., 2020; Rowan and Galanakis, 2020)	11.
*Precision agriculture including blockchain, AI, robotics, along with	13.
combining machine learning with bioinformatics to future proof	14.
environmental test beds and validate new products and services	
Including ecological/blodiversity forecasting (Rowan and Pogue, 2021;	
*Painfall radar data and rainfall information collected from	
commercial microwaya mobile phone backhone transmission networks	
(Chwala and Kunstmann, 2019): worldwide there is an interest in	
changes to sub-daily rainfall regimes for example through the INTENSE	
project (Blenkinsop et al., 2018)	
*Agriculture IoT, construction of agriculture IoT infrastructures, data	
security and data sharing, sustainable energy solutions, economic	
analysis and operation management in agriculture IoT, and IoT-based	
agriculture financing and e-business modes. IoT-based precision	
agriculture (Ruan et al., 2019).	
United Nations Sustainable Development Goals (UN SDGs); Augmented	Reality Virtual

Reality (ARVR); Life Cycle Analysis (LCA); Material Flow Analysis (MFA); Principle Component Analysis (PCA); Artificial Intelligence (AI); Internet of Things (IoT)

occurs when either the ground is already saturated, due to previous rainfall, and the new rain falling can only run off; or when the rainfall intensity exceeds the infiltration rate of the ground and again the rain will run off, these situations are often called a flash flood (Dimitriou, 2011). Detecting trends in long-term rainfall records is difficult due to the highly variable nature of rainfall and scarcity, often, of long records. Information on rainfall intensity, explicitly sub-daily and sub-hourly records, is much more sparse and short term in its record; although, data generated from growing hobby weather station networks can be potentially used to help fill this gap. Other sources could include processed rainfall radar data and rainfall information can be gathered from commercial microwave mobile phone

backbone transmission networks (Chwala and Kunstmann, 2019). Worldwide, there is an interest in changes to sub-daily rainfall regimes; for example, through the INTENSE project (Blenkinsop et al., 2018).

Extratropical cyclones, produce more than 70% of the winter rainfall in north-west Europe. The islands of Ireland and Great Britain had their stormiest winter on record during 2012/2013 with more than two intense cyclones per week (Priestley et al., 2017). It is plausible that impacts due to global climate change in the sea surface temperatures in the Western tropical West Pacific along with the reduction in Arctic sea-ice may have allowed this stormy period of weather to occur. Noone et al. (2016) working on a recently homogenised Irish rainfall data set from 1850 to 2010 found there was a positive trend for the winter months and a negative trend for the summer months. In Ireland there are interests in the impact of unusual rainfall events on infrastructure. The impact of rainfall events and the destabilisation of railway embankments in Ireland was studies leading to an intensity and duration curve for significant failures (Martinović et al., 2018). A similar approach may be able to be taken in safeguarding other assets including aquaculture. There is also interest in blanket peat failures on slopes in Ireland, due to excess rainfall (Jennings and Kane, 2019). Flood impacts on aquaculture have been studied in the Czech Republic where during flooding in 2002, 2006, 2009, and 2013 some 54% of fish in pond-based system were lost (Rutkayová et al., 2018). Many of these losses seem to be from flash flood events attributed to riverine and overland flooding. Rutkayová et al., 2018 found the impacts to be different between juvenile and adult fish, and by species. These researchers also noted the importance in term of impacts of what they term the 'train effect', which is a repeated series of large rainfall events over the same area in a short space of time. This is a classic high impact situation where a series of non-record breaking, but high events when combined over a short space of time can have extreme impacts. Commensurate use of tools to inform management and decision-making at the farm level informed by life cycle assessment, material flow analysis and risk modelling will support and enable solutions to these complex challenges (Tahar et al., 2017; Ruiz-Salmón et al., 2020a; Ruiz-Salmón et al., 2020b).

#### 4. Implications and opportunities of IMTA findings for informing emerging paludiculture and adjacent research and innovation

Findings from this fully-recirculating IMTA study have broader implications for informing worldwide collaborative research and innovation in many multidisciplinary and cross-functional domains beyond the initial focus of paludiculture (Table 2), and alignment with United Nations' Sustainable Development Goals (Table 3). Traditionally, there remains reliance on flow-through aquaculture processes that relies upon bespoke understanding of process performance and interventions including end-of-pipe solutions to safeguard receiving waters from polluted effluents (Rowan, 2011; Tahar et al., 2017). The reliable and repeatable operation of this freshwater IMTA process will enable generation of data that can inform the international development and standardization of paludiculture innovation (Tan et al., 2021; Ziegler et al., 2021). This wet-peatland aquculture innovation can be used for informing efficacy of carbon sequestration, GHG emission reductions and carbon cycles (Science Foundation Ireland (SFI), 2021), ecotoxicology and biodiversity (Rowan and Galanakis, 2020), food production, safety and security (Galanakis et al., 2021; Sacchetttini et al., 2021), ecosystem service management and pollination (Goblirsch et al., 2021), real-time monitoring of in-field technologies linked to living labs (Naughton et al., 2020); suitability of new environmental-friendly aquafeeds (Cooney et al., 2021), digital transformation including novel multimodal sensing technologies, IoT devices, AI and cloud-based innovation to inform effective management of carbon reduction (Science Foundation Ireland SFI, 2022); along with use of biorefinery concept to extract high value bio-based materials from microalgae and duckweed for one health applications (Rowan and Pogue, 2021) (Table 2).

#### Table 3

Indicative examples of IMTA activities and tools to support, enable and accelerate potential green innovation disruption under United Nations' Sustainable Development Goals (SDGS\*).

UN sustainable development		Sustaining or potentially disruptive activity
goa	1*	
1	No poverty	Food production and value chain. Food security and managing climate events to avoid supply chain disruption. Production of high protein sustainable foods via IMTA system. Researcher mobility creating opportunities in education.
2	Zero hunger	Development and future climate-proofing sustainable agri-food processes and crops along with alternative sources for protein that includes training.
3	Good health and wellbeing	Adopting One Health approach to informing eco-green innovation community transition and social enterprise for health including COVID-19 that considers bio-refinery concept (Rowan and Galanakis, 2020; Galanakis et al., 2021)
4	Quality education	Openly sharing knowledge, discoveries and growing collaborations in academia, that cross-cuts STEM with Social Science and humanities, to inform behavioural change. Use of IMTA for specialist training linked to new eco innovation education that encompass green finance, food disruption, climate change.
5	Gender equality	IMTA framework is aligns with gender equality focused with equal representation in research, innovation, entrepreneurship, social enterprise, outreach, education.
6	Clean water and sanitation	Innovative green research and enterprise to promote natural resources for water quality and mitigate waste that moves beyond end-of-pipe solutions
7	Affordable and clean energy	IMTA powered by wind turbines, but also enables development of LCA tools for investing nexus between energy, water and food systems. This also enables commensurate research on sustainable carbon cycles (European Commission, 2021) and green energy using paludiculture platform.
8	Decent work and economic growth	IMTA adopts Quadruple Helix Hub approach that connects industry, government, academic and society – this will lead to informing new job creation and new eco-focused Start Ups along with risk management, transparency for climate action governance in business beyond short-term needs (Rowan and Casey, 2021). This will inform future economic indicators and growth via international collaboration.
9	Industry, innovation, infrastructure	IMTA is trigger new eco or green innovation and research that considers future infrastructure to provide standardization of outcome sets for climate change and sustainability using living labs and environmental demonstrator facilities – this will pump-prime new bio-based research, food production and circularity.
10	Reduced inequalities	Adopting a Quadruple Helix approach will enable broad stakeholder engagement to ensured equalities are met.
11	Sustainable cities and communities	IMTA system is a green innovation aligned with supporting needs of low-carbon communities for regional development and regeneration. This is aligned with European Just Transition and European Green Deal initiatives. This will be informed by social marketing and appropriate communication strategies (Domegan, 2021; Sacchetttini et al., 2021).
12	Responsible consumption and production	IMTA system will support digital transformation of what is a defined sustainable process to support responsible consumption, and production that will inform global needs. Future disruptive innovation likely to emerge from digital domain.
13	Climate action	IMTA system can inform efficacy of future food production processes as it pertains across addressing climate events and inform environmental, ecological, social, political and cultural indicators as state-of-the-art collaborative international demonstrator facility.
14	Life below water	IMTA supports freshwater aquaculture and studies on biodiversity related to natural aquatic ecosystems in peatlands
15	Life on land	IMTA supports biodiversity, pollination and ecosystem service management
14	Peace, justice, strong inst.	IMTA has core tenets that blends academia, industry, authority with communities.
17	Partnerships for the goals	IMTA supports and enable national and international partnerships aligned with UN SDGs that includes mobility and training

This IMTA process will also support and enable development of low-carbon community and social enterprises promoted through the European Just Transition and Green Deal Initiatives (Rowan and Galanakis, 2020), which can be effectively managed through the use of 'Quadruple Helix Hubs' that converge academia, industry, government and society (Rowan and Casey, 2021). This Quadruple Helix Hub framework will also support and enable the commensurate delivery of paludiculture-based and adjacent research, innovation, entrepreneurship linked to education and specialist training of stakeholders and communities, which are essential converging activities underpinning trans-regional European Innovation Hubs. The development and deployment of a validated IMTA process in the peatlands will also support future sustaining and disruptive innovation, where the design, applicability and maturity of the innovation, product or service may be informed by use of various smart evaluation tools including technical, political and societal readiness levels (Fig. 7), life cycle assessment, material flow analysis, principle component analysis, ecotoxicology and so forth. For example, Ruiz-Salmón and co-workers (2020b) used LCA to highlight the important nexus between water, food and energy for development of fisheries including aquaculture across the European Atlantic Area. The open knowledge sharing of IMTA data will also support research into new green business development and transnational modelling of research and innovation (Rowan and Casey, 2021), which will include climate-related awareness, risk management and transparency (Tahar et al., 2019; Coppola et al., 2019) (Table 2). IMTA process and environmental data that can inform future risk mitigation for innovation will be particularly relevant. For example, Coppola et al. (2019) reported that Deloitte asked 1168 Chief Financial Officers (CFOs) what their companies are doing to help address climate change where a thorough understanding of climate risks was rare, few companies had governance mechanisms in place to implement comprehensive climate strategies, and targets for carbon emission reductions were usually not aligned with the Paris Agreement. Coppola et al. (2019) also noted that more than US\$30 trillion in funds were held in sustainable or green investments in the five major markets tracked by the Global Sustainable Investment Alliance. The measured effects of extreme weather events, using this technically-defined IMTA process, will also inform development of predictive and simulated disease models influenced by climate change (Interreg Neptunas Project, 2019).

Barry and Hoyne (2021) reported that changes in weather systems, such as increased precipitation, snow and ice events, heatwaves and storms, have led the European Commission to develop new policies and strategies to deal with extreme events. Findings from, and future use of this IMTA system, will help inform a robust set of appropriate indicators to assess the impacts of climate change on adaptive food production and security at local and national levels. Barry and Hoyne (2021) noted that these indicators must encompass a multidisciplinary scope that also embraces ecological, social, cultural and economic changes, with greater awareness within all areas of society (Table 2). Process technical parameters used in this IMTA study will provide a useful environmental case study to help evaluate impact of climate change on paludiculture and other food systems beyond baseline observations of rainfall, temperature, GHG emissions, sea level rise/flooding and soil degradation (Aguiar et al., 2018). There is a pressing need for international agreement of relevant indicators existing outcome set used for scientific audiences that also address public interest and effects of climate change; these should allow public consumption and education through clear and concise communication standards (Williams and Eggleston, 2017). These IMTA findings contribute to climate resilience that relates to the capacity for ecosystems to respond to impacts, events or disturbances that are associated with climate change (Baho et al., 2017).

Moreover, Barry and Hoyne (2021) suggested seeking international agreement on indicators to inform ecological resilience, along with economic (such as number of new SME creation, innovation, investment in training, and specialist upskilling), social enterprise and cultural indicators (such as diversity of youth initiatives to increase civil action, solidarity and engagement). This IMTA system will also support need to inform natural capital and improvements in biodiversity, such as by promoting

Technology Readiness Levels* (TRL)*		Society Readiness Levels** (SRL)		Policy Readiness Levels*** (PFL)	
TRL 1 – Basic Research - Principles postulated and observed – no experimental proof (Discovery)         TRL 2 – Technology Concept Formulated – concept and application defined (Concept Definition)         TRL 3 – Experimental Applied Research Concept – first laboratory tests completed (Proof of Concept)         TRL 4 – Technology Validated in Lab - Small scale prototype – built and tested in lab (lab validation)	nowledge development cademia	SRL 1 - Basic Research - identifying problem and identifying societal readiness (Discovery)         SRL 2 - Formulation of problem, proposed solution(s) and potential impact, expected societal readiness; identifying intended stakeholders for project (Concept Definition)         SRL 3 - Applied Research - initial testing of proposed solution(s) with intended stakeholders (Proof of Concept)         SRL 4 - Pilot-Test Scale - concept validated through pilot testing in relevant environment to substantiate proposed impact and societal readiness (Concept	nowledge development cademia	PRL 1 -Basic Research - identifying issue/problem and identifying policy readiness (Discovery)         PRL 2 - Formulation of issue/problem, proposed solutions and potential impact; expected policy readiness; concept identification relevant to stakeholders (Concept Definition)         PRL 3 - First testing of proposed solutions, feedback, development complete (Proof of Concept).         PRL 4 - Problem validated "in lab" through pilot testing in intended environment to substantiate proposed impact, policy readiness, feedback development (lab validation)	.nowledge development .cademia
TRL 5 Large-Scale Prototype tested in intended environment (test facility validation TRL 6 – Technology demonstrated in intended environment – close to expected performance ("open water" validation) TRL 7 – System prototype demonstration in operational environment – at pre-commercial scale (system demo)	Technology Development K Collaboration	Validation) SRL 5 – Large Scale Test/system - proposed solution(s) validated; with intended stakeholders ("open water" validation) SRL 6 – Demonstrated system - solution(s) demonstrated in relevant environ and with intended stakeholders for feedback on policy SRL 7- System Refinement - refinement of product, and/or solution(s), and if needed, retesting in intended environment with stakeholders (refinement)	Societal Development Collaboration	PRL 5 – Proposed solution(s) validated; now by intended stakeholders in the area for application ("open water" validation) PRL 6 – Demonstration system in intended environ & with intended stakeholders at pre-role out scale for feedback on impact (system demo) PRL 7 – System refinement of scheme and/or solution(s), and possibly, retesting in intended environment with intended stakeholders to gain feedback (refinement)	Policy Development K Collaboration
TRL 8 – First system complete, qualified, verified –         First commercial system – manufactured issues solved (verification)         TRL 9 – Actual Full commercial system proven in operational environment – technology available for beneficiaries (deployment)	Business Development Industry	SRL 8 – First System – issues solved, proposed solution(s), as well as plan for societal adaption complete, and qualified (verification)         SRL 9 – Full Social System - actual project solution(s) in intended or relevant environment (deployment)	Stakeholder Development Governing	PTL 8 – First System - proposed solution(s), as well as plan for policy adaptation complete, and qualified (verification)         PRL 9 – Policy Implementation - actual project solution(s) proven in relevant environment. Issues solved, continued monitoring, evaluation, and review of scheme/solution (deployment)	Scheme development Government
*TRL – are indicators of status or maturity level of p	articular t	echnology been researched and commonly used for Europe	ean Comn	nission in context of Horizon Europe.	
***PRL – Used to assess the level of societal adaptat	ation to pro	oject, technology, product, process or management practice	e or inno	vation to be integrated into society	

Fig. 7. Applying IMTA ecosystem to develop and track new eco-innovation to align and commensurate with technology, societal and policy readiness levels (adapted from Rowan and Casey, 2021).

agroecological farming, re-establishing organic carbon and microbiota in the soil and land, the potential use of biochar (Galanakis et al., 2022). Funding instruments, such as the European Just Transition Fund with an overall budget of €17.5 billion will help with a fair and equitable transformation to low-carbon communities where challenges will create opportunities through digitalisation that will boost employment and growth (Table 2). George et al. (2021) reported that digital sustainability and entrepreneurship can help tackle climate change and sustainable development (Table 2). Findings of this study will also support research informing sustaining and disruptive innovation (Schuelke-Leech, 2021), where greater awareness will be enabled through social marketing (Domegan, 2021). Bio-based products harvested and refined from microalgae occurring in this IMTA ecosystem can contribute to one-health solutions, including potentially contributing to alleviating COVID-19 (Murphy et al., 2020, 2022; Pogue et al., 2021).

There is a commensurate need to develop tailored communication strategies for promoting greater awareness of health and sustainability in seafood consumption given the diversity of attitudes and perceptions reported among Italian consumers by Sacchetttini et al. (2021) (Table 2). Moreover, Domegan (2021) highlighted the pivotal role of social marketing in critically examining the interface of human and natural systems and their interconnected dynamic forces as a powerful means of influencing behaviours for the accorded transformation and betterment of individuals, communities, society and the planet. In pursuit of Green Deal Innovations, such as embodied in this IMTA process, critical emerging trends in social marketing embrace important systems science, stakeholder engagement and digital technologies.

#### 5. Conclusion

- Peatlands-based IMTA constitutes a potentially important sustainable and resilient system for advancing aquaculture, which will also support and enable development and validation of other sustaining and disruptive innovation, products and services.
  - While this IMTA model successfully produces commercial fish, and typically does not discharge to receiving water, this present study highlighted the challenges of operating a recirculating system with increased, and unexpected rainfall due to storms.
  - This IMTA system can help with improved understanding of environmental, social, cultural and economic indicators for broader appreciation of climate impacts based upon the people who are directly affected by the changes at local and national levels.
  - There is a pressing need to develop real-time approaches to monitor algae species that are potentially toxigenic, and to thoroughly investigate the impact of extreme weather events on balanced algal and microbial population in freshwater aquaculture ecosystem.
  - Given that there are potentially 1864 species of algae occurring in this
    peatland IMTA system that were identified using next generation sequencing and bioinformatics, there are pressing opportunities to exploit
    advances in flow-cytometry combined with machine learning to help unlock the challenge of purposeful monitoring.
  - Accessing and using weather data will present using opportunities to inform in field monitoring tools; for example, Ireland has a long-term data set of daily rainfall across 25 stations.
  - COP26 highlighted the pressing importance of enlisting a global response to curb carbon emissions so at to limit temperature rises to 1.5 °C; therefore, there is a commensurate need to investigate and test new sustainable food production systems (such as IMTA) that considers and hurdles extreme weather/climate events.
  - Biorefining products from this IMTA system using circular concept will help advance adjacent innovation, such as developing and testing fish feed improved for immune-stimulation to boost fish immunity to disease may help with priming against unforeseen events.
  - Further development and use of this IMTA process will help communities transition to low-carbon economies; however, there is a commensurate need to improve awareness in sustainable eco-innovation that can be

improved by increasing social enterprises and by supporting the creating of new businesses such as through the Quadruple Helix Hub concepts that aligns with collaborative European Innovation Hubs (EIH), European Green deal and European Just Transition progamme initiatives.

- A future climate-proofed IMTA system will also support and help meet the United Nations Development Goals, particularly alleviating against poverty, hunger along with fostering quality education, economic growth and innovation.
- There are pressing opportunities to advance paludiculture through digital transformation, which may lead to business model or green-disruption.
- Use of IMTA-system data to inform IoT-based precision paludiculture and agriculture.

#### CRediT authorship contribution statement

E. A. O'Neill, N. J. Rowan: Conceptualization. E. A. O'Neill: Data Curation. E. A. O'Neill, A. P. Morse, N. J. Rowan: Formal Analysis. N. J. Rowan: Funding Acquisition. E. A. O'Neill: Investigation. E. A. O'Neill: Methodology. E. A. O'Neill: Project Administration. E. A. O'Neill, N. J. Rowan: Resources. E. A. O'Neill: Software. N. J. Rowan: Supervision. E. A. O'Neill: Validation. E. A. O'Neill: Visualization. E. A. O'Neill, A. P. Morse, N. J. Rowan: Roles/Writing – Original Draft. E. A. O'Neill, A. P. Morse, N. J. Rowan: Writing – Review & Editing.

#### Declaration of competing interest

The authors declare that there are no competing interests or conflicts of interest with respect to the publication of this article.

#### Acknowledgements

The authors would like to thank Bord Iascaigh Mhara (Project BIM 19-KGS-008) and Interreg Atlantic Area Neptunus (Project EAPA\_576/2018) for funding support. We thank Gustavo W. Fehrenbach for assistance with microscopic analysis of samples during COVID-19 lockdown period.

#### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi. org/10.1016/j.scitotenv.2022.153073.

#### References

- Aguiar, F., Bentz, J., Silva, J., Fonseca, A., Swart, R., Santos, F., Penha-Lopes, G., 2018. Adaptation to climate change at local level in Europe: an overview. Environ. Sci. Pol. 86, 38–63.
- Alam, M.J.B., Islam, M.R., Muyen, Z., Mamun, M., Islam, S., 2007. Water quality parameters along rivers. Int. J. Environ. Sci. Tech. 4, 159–167.
- Baho, D., Allen, C., Garmestani, A., Fried-Petersen, H., Renes, S., Gunderson, L., Angeler, D., 2017. A quantitative framework for assessing ecological resilience. Ecol. Soc. 22.
- Barry, D., Hoyne, S., 2021. Sustainable measurement indicators to assess impact of climate change. Implications for the new green Deal era. Curr. Opin. Environ. Sci. Health 22. https://doi.org/10.1016/j.coesh.2021.100259.
- BIM, 2020. Development of IMTA Process with Focus on Contribution of Freshwater Algae, Technical Report 2019-KGS-008. https://www.ait.ie/news-and-events/news/aitbreathes-new-life-into-peatlands-bog-with-aquaculture-research. (Accessed 14 January 2022).
- Blenkinsop, S., Fowler, H.J., Barbero, R., Chan, S.C., Guerreiro, S.B., Kendon, E., Lenderink, G., Lewis, E., Li, X.-F., Westra, S., Alexander, L., Allan, R.P., Berg, P., Dunn, R.J.H., Ekström, M., Evans, J.P., Holland, G., Jones, R., Kjellström, E., Klein-Tank, A., Lettenmaier, D., Mishra, V., Prein, A.F., Sheffield, J., Tye, M.R., 2018. The INTENSE project: using observations and models to understand the past, present and future of subdaily rainfall extremes. Adv. Sci. Res. 15, 117–126. https://doi.org/10.5194/asr-15-117-2018.
- Program, Blue Hatch Aquaculture Accelerator, 2020. https://www.hatch.blue/program. (Accessed 1 July 2022).
- Boaventura, R., Pedro, A.M., Coimbra, J., Lencastre, E., 1997. Trout farm effluents: characterisation and impact on the receiving streams. Environ. Pollut. 95, 379–387.
- Boyd, C.E., Tucker, C.S., 2015. Handbook for Aquaculture Water Quality. C. S., & Craftmaster Printers, Auburn.

- Braga Rodriguez, T., O'Cathain, C., O'Connor, N.E., Murray, N., 2020. A quality of experience assessment of haptic and augmented reality feedback modalities in a gait analysis system. PLOS One 15 (3), e0230570. https://doi.org/10.1371/journal.pone.0230570.
- Camargo, J.A., 1994. The importance of biological monitoring for the ecological risk assessment of freshwater pollution; a case study. Environ. Int. 20, 229–238.
- Cao, L., Wang, W., Yang, Y., Yang, C., Yuan, Z., Xiong, S., Diana, J., 2007. Environmental impact of aquaculture and countermeasures to aquaculture pollution in China. Environ. Sci. Pollut. Res. 14, 452–462.
- Caramel, B., Moraes, B.P., Carmo, M.A.B., Vaz-Dos-Santos, C.F., Tabata, A.M., Osti, Y.A., Ishikawa, J.A.S., Cerqueira, C.M., Mercante, M.A.S., 2014. Water quality assessment of a trout farming effluent, BocainaBrazil. J. Water Resour. Prot. 6, 909–915.
- Chavez, E., Conway, G., Ghil, M., Sadler, M., 2015. Estimating the effects of extreme weather on food security. Nat. Clim. Chang. https://doi.org/10.1038/nclimate2747.
- Chinedu, I., Nwaogu, L.A., Ebe, T.E., Jjoku, J.D., 2015. Effects of meteorological events on the levels and interactions of chemical indices of a polluted freshwater system. Int. J. Environ. Science, Manag. Eng. Res. 1, 31–37.
- Chwala, C., Kunstmann, H., 2019. Commercial microwave link networks for rainfall observation: assessment of the current status and future challenges. WIREs Water. 2019 (6), e1337. https://doi.org/10.1002/wat2.1337.
- Cooney, R., Wan, A.H.L., O'Donncha, F., Clifford, E., 2021. Designing environmentally efficient aquafeeds through the use of multcriteria decision support tools. Curr. Opin. Environ. Sci. Health 23. https://doi.org/10.1016/j.coesh.2021.100276.
- Coppola, M., Krick, T., Blohmke, J., 2019. Feeling the heat? Companies are under pressure on climate change and need to do more. https://www2.deloitte.com/us/en/insights/ topics/strategy/impact-and-opportunities-of-climate-change-on-business.html. (Accessed 10 October 2021).
- Costanzo, S.D., O'Donohue, M.J., Dennison, W.C., 2004. Assessing the influence and distribution of shrimp pond effluent in a tidal mangrove creek in north-East Australia. Mar. Pollut. Bull. 48, 514–525.
- da Silva, A.Q., Nilin, J., Santaella, S.T., Oriel, Bonilla, H., 2017. Ecotoxicological and physicochemical evaluation of an effluent of a shrimp farm located in Northeastern Brazil. Panam. J. Aquat. Sci. 12, 263–272.
- Dimitriou, E., 2011. Overland flow. In: Gliński, J., Horabik, J., Lipiec, J. (Eds.), Encyclopedia of Agrophysics. Encyclopedia of Earth Sciences Series. Springer, Dordrecht https://doi. org/10.1007/978-90-481-3585-1\_104.
- Doherty, E., Mellett, S., Norton, D., McDermott, T.K.J., O'Hora, D., Ryan, M., 2021. A discrete choice experiment exploring farmer preferences for insurance against extreme weather events. J. Environ. Manag. 290. https://doi.org/10.1016/j.envman.2021.112607.
- Domegan, C., 2021. Social marketing and behavioural change in a systems setting. Curr. Opin. Environ. Sci. Health https://doi.org/10.1016/j.coesh.2021.100275.
- Dubinsky, Z., Rotem, J., 1974. Relations between algal populations and the pH of their media. Oecologia 16, 53–60.
- Durborow, R.M., Crosby, D.M., Brunson, M.W., 1997. Nitrite in Fish Ponds. Stoneville, Mississippi.

EPA, 2001. Parameters of Water Quality - Interpretation and Standards. Johnstown Castle.

European Commission, 2021. Sustainable Carbon Cycles. https://ec.europa.eu/clima/euaction/forests-and-agriculture/sustainable-carbon-cycles\_en. (Accessed 1 June 2021).

Fadaeifard, F., Raisi, M., Jalali, B., Majllesi, A., 2011. The Impact of Rainbow Trout Farm Effluents on Water Quality of Koohrang River, Iran. J. Agric. Sci. Technol. A 1207–1209.

- Expert, Fish Farming, 2020. Aquaculture has improved food security [WWW Document]. News . https://www.fishfarmingexpert.com/article/aquaculture-has-improved-foodsecurity-says-un/. (Accessed 27 May 2021).
- Galanakis, C.M., Rizou, M., Aldawoud, T.M.S., Ucak, I., Rowan, N.J., 2021. Innovations and technology disruptions in the food sector within the COVID-19 pandemic and postlockdown era. Trends Food Sci. Technol. 110 (193), 200.
- Galanakis, C.M., Brunori, G., Chiaramnti, D., Matthews, R., Panoutsou, C., Fritsche, U.R., 2022. Bioeconomy and green recovery in a post-COVID-19 era. Sci. Total Environ. 808. https://doi.org/10.1016/j.scitotenv.2021.152180.
- Geels, F.W., 2018. Disruption and low-carbon transformation, progress and new challenges in socio-technical transitions research and the multi-level perspective. Energy Res. Soc. Sci. 86, 2317–2324.
- George, G., Merrill, R.K., Schillebeeckx, S.J.D., 2021. Digital sustainability and entrepreneurship: how digital innovation are helping tackle climate change and sustainable development. Enterp. Theory Pract. 45 (5), 999–1027.
- Goblirsch, M., Eakins, J., Rowan, N.J., 2021. Disease-mitigating innovations for the pollination service industry: challenges and opportunities. Curr. Opin. Environ. Sci. Health 22. https://doi.org/10.1016/j.coesh.2021.100265.

Gerardi, M.H., Lytle, B., 2015. BOD and TSS. The Biology and Troubleshooting of Facultative Lagoons. John Wiley & Sons, Inc, pp. 195–198.

- Granada, L., Sousa, N., Lopes, S., Lemos, M.F.L., 2016. Is integrated multitrophic aquaculture the solution to the sectors' major challenges? a review. Rev. Aquac. 8, 283–300.
- Guan, D., Wang, D., Hallegatte, S., Davis, S.J., Huo, S., Li, L., Bai, Y., Lei, T., Xue, Q., Coffman, D., Cheng, D., Chen, P., Liang, X., Xu, B., Lu, X., Wang, S., Hubacek, K., Gong, P., 2020. Global-supply chain effects of COVID-19 control measures. Nat. Hum. Behav. https:// doi.org/10.1038/s41562-020-0896-8.
- Guilpart, A., Roussel, J.-M., Aubin, J., Caquet, T., Marle, M., Le Bris, H., 2012. The use of benthic invertebrate community and water quality analyses to assess ecological consequences of fish farm effluents in rivers. Ecol. Indic. 23, 356–365.

Gupta, N., Pandey, P., Hussain, J., 2017. Effect of physicochemical and biological parameters on the quality of river water of Narmada, Madhya PradeshIndia. Water Sci. 31, 11–23.

- Haynes, M., Seegers, B., Saluk, A., 2016. Advanced analysis of marine plankton using flow cytometry. Biotechniques 60, 260.
- Herreor, M., Thornton, P.K., Croz, D.M., Palmer, J., et al., 2020. Innovation can accelerate the transition towards a sustainable food system. Nat. Food 1, 266–272.
- Herrero, A., Muro-Pastor, A.M., Flores, E., 2001. Nitrogen control in cyanobacteria. J. Bacteriol. 183, 411–425.

- Iizumi, T., Ramankutty, N., 2015. How do weather and climate influence cropping area and intensity. Glob. Food Sec. 4, 46–50.
- Interreg Neptunas Project, 2019. Transnational Clustering Concept Approach to Harness Key Eco-Labelling and Key Enabling Eco-Innovations that Add-Value and Cross-Cut sea Food-Water-Energy Domains in in the Atlantic region. https://neptunus-project.eu/program/. (Accessed 14 January 2022).
- Iredale, R.S., McDonald, A.T., Adams, D.G., 2012. A series of experiments aimed at clarifying the mode of action of barley straw in cyanobacterial growth control. Water Res. 46, 6095–6103.
- Irish Statutory Office, 2009. S.I. No. 272/2009 European communities environmental objectives (surface waters) regulations 2009 [WWW Document]. Electron. Irish Statut. B.. http://www.irishstatutebook.ie/eli/2009/si/272/made/en/print. (Accessed 5 April 2019)
- Irish Statutory OfficeOffice, 2019. S.I. No. 77/2019 European union environmental objectives (surface waters) (amendment) regulations 2019 [WWW Document]. Electron. Irish Statut. B. 2019. http://www.irishstatutebook.ie/eli/2019/si/77/made/en/pdf. (Accessed 14 January 2020).
- Jennings, P., Kane, G., 2019. Assessment of peat slides and failure mechanism of blanket peat at Flughland, Ireland. Proceedings of the XVII ECSMGE-2019 - Geotechnical Engineering Foundation of the Future, pp. 1–7 https://doi.org/10.32075/ 17ECSMGE-2019-0884.
- Kasuya, K., Takagi, K., Ishiwatari, S., Yoshida, Y., Doi, Y., 1998. Biodegradabilities of various aliphatic polyesters in natural waters. Polym. Degrad. Stab. 59, 327–332.
- Lalonde, B.A., Ernst, W., Garron, C., 2014. Chemical and physical characterisation of effluents from land-based fish farms in Atlantic Canada. Aquac. Int. 23, 535–546.
- Lee, A.H., Nikraz, H., 2015. BOD: COD ratio as an indicator for river pollution. Int. Proc. Chem. Biol. Environ. Eng. 88, 89–94.
- Li, M., Porter, A.L., Souminen, A., 2018. Insights into relationship between disruptive technology/innovation and emerging technology – a bibliometric perspective. J. Food Eng. 2020, 20–27.
- Lim, B.R., Huang, X., Hu, H.Y., Goto, N., Fujie, K., 2001. Effects of temperature on biodegradation characteristics of organic pollutants and microbial community in a solid phase aerobic bioreactor treating high strength organic wastewater. Water Sci. Technol. 43, 131–137.
- Liu, X., Steele, J.C., Meng, X.-Z.Z., 2017. Usage, residue, and human health risk of antibiotics in Chinese aquaculture: a review. Environ. Pollut. 223, 161–169.
- Martinović, K., Gavin, K., Reale, C., Mangan, C., 2018. Rainfall thresholds as a landslide indicator for engineered slopes on the irish rail network. Geomorphology 306, 40–50. https://doi.org/10.1016/j.geomorph.2018.01.006.
- Mcintosh, D., Fitzsimmons, K., 2003. Characterization of effluent from an inland, low-salinity shrimp farm: what contribution could this water make if used for irrigation. Aquac. Eng. 27, 147–156.
- McKeon-Bennett, M.P., Hodkinson, T.R., 2021. Sphagnum moss as a novel growth medium in sustainable indoor agriculture systems. Curr. Opin. Environ. Sci. Health https://doi.org/ 10.1016/j.coesh.2021.100269.
- Met Éireann, 2021. Monthly data [WWW Document]. Climate. https://www.met.ie/index. php/climate/available-data/monthly-data. (Accessed 1 June 2021)
- Moore, G., 2021. https://www.fishfarmingexpert.com/article/north-wind-gives-mowi-hopeof-salvaging-crop-after-bantry-bay-die-off/. (Accessed 21 November 2021).
- Moorhouse, H.L., Read, D.S., McGowan, S., Wagner, M., Roberts, C., Armstrong, L.K., Nicholls, D.J.E., Wickham, H.D., Hutchins, M.G., Bowes, M.J., 2018. Characterisation of a major phytoplankton bloom in the river Thames (UK) using flow cytometry and high-performance liquid chromatography. Sci. Total Environ. 624, 366–376.
- Moreira, L.E.B., Lombardi, J.V., Mercante, C.T.J., Bazante-Yamaguishi, R., 2010. Ecotoxicological assessment in a pond of freshwater shrimp farming, using the cladocera Ceriodaphnia dubia as test organism. Bol. Inst. Pesca 36, 25–38.
- Mukherjee, S.K., Chatterji, A.K., Saraswat, I.P., 1968. Effect of pH on the rate of BOD of wastewater. Water Pollut. Control Fed. 40, 1934–1939.
- Murphy, E.J., Masterson, C., Reoagli, E., O'Toole, D., Major, I., Stack, D.S., Lynch, M., Laffey, J.G., Rowan, N.J., 2020. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cyto-protective effects Implications for coronavirus disease (COVID-19) immunotherapies. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.139330
- Murphy, E.J., Rezoagli, E., Pogue, R., Simonassi-Paiva, B., Abdin, I.I.Z., Fehrenbach, G.W., O'Neill, E., Major, I., Laffey, J.G., Rowan, N.J., 2022. Immunomodulatory activity of βglucan polysaccharides isolated from different species of mushroom – A potential treatment for inflammatory lung conditions. Sci. Total Environ. https://doi.org/10.1016/j. scitotenv.2021.152177
- Namin, J.I., Sharifinia, M., Makrani, A.B., 2013. Assessment of fish farm effluents on macroinvertebrates based on biological indices in Tajan River (north Iran). Casp. J. Env. Sci 11, 29–39.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system. Aquaculture 526, 735377.
- Nielsen, R., Asche, F., Nielsen, M., 2016. Restructuring european freshwater aquaculture from family-owned to large-scale firms – lessons from danish aquaculture. Aquac. Res. 47, 3852–3866.
- Noone, S., Murphy, C., Coll, J., Matthews, T., Mullan, D., Wilby, R.L., Walsh, S., 2016. Homogenization and analysis of an expanded long-term monthly rainfall network for the island of Ireland (1850–2010). Int. J. Climatol. 36, 2837–2853. https://doi.org/10.1002/joc. 4522.
- Noroozrajabi, A., Ghorbani, R., Abdi, O., Nabavi, E., 2013. The impact of rainbow trout farm effluents on water physicochemical properties of daryasar stream. World J. Fish Mar. Sci. 5, 342–346.

#### E.A. O'Neill et al.

- O'Neill, E.A., Rowan, N.J., 2022. Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance – a comparative case study from the Republic of Ireland. Sci. Total Environ. 802, 149800.
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218.
- O'Neill, E.A.E.A., Stejskal, V., Clifford, E., Rowan, N.J.N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. Sci. Total Environ. 706, 136044.
- Pogue, R., Murphy, E., Fehrenbach, G.W., Rezoagli, E., Rowan, N.J., 2021. Exploiting immunomodulatory properties of β-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms: concise review of existing knowledge, innovation and future opportunities. Curr. Opin. Environ. Sci. Health 21. https:// doi.org/10.1016/j.coesh.2021.100248.
- Pollice, A., Tandoi, V., Lestingi, C., 2002. Influence of aeration and sludge retention time on ammonium oxidation to nitrite and nitrate. Water Res. 36, 2541–2546.
- Priestley, M.D.K., Pinto, J.G., Dacre, H.F., Shaffrey, L.C., 2017. The role of cyclone clustering during the stormy winter of 2013/2014. Weather 72, 187–192. https://doi.org/10.1002/ wea.3025.
- Pulatsü, S., Rad, F., Topçu, A., 2004. The impact of rainbow trout farm effluents on water quality of Karasu streamTurkey. Turkish J. Fish. Aquat. Sci. 4, 9–15.
- Rajabi, H., Filizadeh, Y., Soltani, M., Fotokian, M.H., 2010. The use of barley straw for controlling of cyanobacteria under field application. J. Fish. Aquat. Sci. 5, 394–401.
- Read, D.S., Bowes, M.J., Newbold, L.K., Whiteley, A.S., 2014. Weekly flow cytometric analysis of riverine phytoplankton to determine seasonal bloom dynamics. Environ. Sci. Process. Impacts 16, 594–603.
- Rowan, N., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, knowledge and outlooks. Microb. Ecol. Glob. Health. https:// doi.org/10.1155/2011/462832.
- Rowan, N.J., Laffey, J.G., 2020(aa. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic - Case study from the Republic of Ireland. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.138532.
- Rowan, N.J., Laffey, J.G., 2020(bb. Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from coronavirus disease (COVID-19) pandemic – Implications for efficacy, re-use and sustainable waste management. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.142259.
- Rowan, N.J., Moral, R.A., 2021. Disposable face masks and reusable face coverings asnonpharmaceutical interventions (NPIs) to prevent transmission of SARS-CoV-2 variants that cause coronavirus disease (COVID-19): role of new sustainable NPI design innovations and predictive mathematical modelling. Sci. Total Environ. 772, 145530.
- Rowan, N.J., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo vadis? Trends food sciTechnol. 88, 316–332.
- Rowan, N.J., Casey, O., 2021. Empower Eco multiactor HUB: A triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. Curr. Opin. Environ. Sci. Heal. 21, 100254.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in Agri-food and green deal innovations: quo Vadis? Sci. Total Environ. 748. https://doi.org/10.1016/j.scitotenv.2020.141362.
- Rowan, N., O'Neill, E., Murphy, E., Pogue, R., Fehrenbach, F., Lynch, M., 2021. Aqua algae-Plus - an investigation in the role of microalgae in a novel IMTA process in the Irish midlands. Bord Iascaigh Mhara Report KGS-008-2019.
- Rowan, N.J., Pogue, R., 2021. Editorial overview: green new deal era—current challenges and emerging opportunities for developing sustaining and disruptive innovation. https://doi. org/10.1016/j.coesh.2021.100294.
- Ruan, J., Jiang, H., Zhu, C., Hu, X., Shi, Y., Liu, T., Rao, W., Chan, F.T.S., 2019. Agriculture IoT: emerging trends, cooperation networks, and outlook. IEEE Wirel. Commun. 26 (6), 56–63. https://doi.org/10.1109/MWC.001.1900096.
- Ruiz-Salmón, İ., Margallo, M., Laso, J., Villaneuva-Rey, P., Marino, D., Quinteiro, P., Dias, A.D., Nunes, A.L., Marques, A., Feijoo, G., Moreira, M.T., Loubet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N., Mendez-Paz, D., Iglesias-Parga, X., Anglada, C., Martin, J.C., Irabian, A., Aldaco, R., 2020a. Addressing challenges and opportunities for the european seafood sector under a circular economy framework. Curr. Opin. Environ. Sci. Health 13, 101–106.
- Ruiz-Salmón, I., Laso, J., Margallo, M., Villaneuva-Rey, P., Rodriguez, E., Quinteiro, P., Dias, A.C., Almeida, C., Nunez, M.L., Marquez, A., Cortes, A., Moreira, M.T., Feijoo, G., Loubet, P., Sonnemann, G., Morse, A.P., Cooney, R., Clifford, E., Regueiro, L., Mendez, D., Anglada, C., Noirot, C., Rowan, N., Vazquez-Rowe, I., Aldaco, R., 2020b. Life cycle assessment of fish and seafood processed products – a review of methodologies and new challenges. Sci. Total Environ. 761. https://doi.org/10.1016/j.scitotenv.2020.144094.
- RUN-EU, 2020. Regional University Network European University Project: Our Students will inherit the Future. https://run-eu.eu/. (Accessed 14 January 2022).
- Rutkayová, J., Vácha, F., Maršálek, M., Beneš, K., Civišová, H., Horká, P., Petrášková, E., Rost, M., Šulista, M., 2018. Fish stock losses due to extreme floods – findings from pond-based aquaculture in the Czech Republic. J. Flood Risk Manage. 11, 351–359. https://doi.org/ 10.1111/jfr3.12332.
- Sacchetttini, G., Gastellini, G., Graffigna, G., Hung, Y., Lambri, M., Marques, A., Perrella, F., Savarese, M., Verbeke, W., Capri, E., 2021. Assessing consumers' attitudes, expectations and intentions towards health and sustainability regarding seafood consumption in Italy. https://doi.org/10.1016/j.scitotenv.2021.148049.

- Science Foundation Ireland (SFI), 2021. Microsoft Ireland and SFI Launch 65m Climate Change Research Project: Terrain AI. https://www.sfi.ie/research-news/news/microsoft-sfiterrain-ai/. (Accessed 1 July 2022).
- Science Foundation Ireland (SFI), 2022. TerrainAI: Uncovering New insights to Support Effective Climate Change Decision Making. https://terrainai.com/. (Accessed 14 January 2022).
- Sharebiotech, 2021. Sharing Life Sciences Infrastructure and Skills to Benefit the Atlantic Area Biotechnology sector. https://keep.eu/projects/440/Sharing-life-science-infrastr-EN/. (Accessed 14 January 2022).
- Shukla, D., Bhadresha, K., Jain, N.K., Modi, H.A., 2013. Physicochemical analysis of water from various sources and their comparative studies. IOSR. J. Environ. Sci. Toxicol. Food Technol. 5, 89–92.
- Stephens, W.W., Farris, J.L., 2004a. A biomonitoring approach to aquaculture effluent characterization in channel catfish fingerling production. Aquaculture 241, 319–330.
- Stephens, W.W., Farris, J.L., 2004b. Instream community assessment of aquaculture effluents. Aquaculture 231, 149–162.
- Silva, A.L.P., 2021. Future proofing plastic waste management for a circular bioeconomy. Curr. Opin. Environ. Sci. Health https://doi.org/10.1016/j.coesh.2021.100263.
- Schuelke-Leech, B.A., 2021. Disruptive technologies for a green new Deal. Curr. Opin. Environ. Sci. Health https://doi.org/10.1016/j.coesh.2021.100245.
- Sultana, T., Haque, M.M., Salam, M.A., Alam, M.M., 2017. Effect of aeration on growth and production of fish in intensive aquaculture system in earthen ponds. J. Bangladesh Agril. Univ 15, 113–122.
- Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E., Rowan, N., 2018a. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3, 28.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018b. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, 1–22.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638.
- Tahar, A., Tiedeken, E.J., Rowan, N.J., 2018. Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – a first overview of the situation in the Republic of Ireland. Sci. Total Environ. 616–617, 187–197.
- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – A 20-year systematic review. Sci. Total Environ. 574, 1140–1163.
- Tan, Z.D., Lupascu, M., Mijedasa, L.S., 2021. Paludiculture as a sustainable land use alternative for tropical peatlands. Sci. Total Environ. 753, 142111.
- Tschirner, M., Kloas, W., 2017. Increasing the sustainability of aquaculture systems insects as alternative protein source for fish diets. GAIA - Ecol. Perspect. Sci. Soc. 26, 332–340.
- Uràk, I., Hartel, T., Gallé, R., Balog, A., 2017. Worldwide peatland degradation and the related carbon dioxide emissions: the importance of policy regulations. Environ. Sci. Pol. 69, 57–64.
- Usuldin, S.R.A., Wan-Mohtar, I., Ilham, Z., Jamaludin, A.A., Abdullan, N.R., Rowan, N., 2021. In-vivo toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian Tiger-Milk Mushroom mycelium for potential future health applications. Scientific Reports https://doi.org/10.21203/rs.3.rs-778200/v1.
- Voulvoulis, N., Arpon, K.D., Giakoumis, T., Barcelo, D., 2017. The EU Water Framework Directive: From Great Expectations to Problems With Implementation-NC-ND License. http://creativecommons.org/licenses/by-nc-nd/4.0/.
- Wan Mohtar, W.A.A.Q.I., Ilham, Z., Jamaludin, A.A., Rowan, N., 2021. Use of zebrafish embryo assay to evaluate toxicity and safety of bioreactor-grown exopolysaccharide and endopolysaccharides from European Ganoderma applanatum mycelium for future aquaculture applications. Int. J. Mol. Sci. 22 (4), 1675. https://doi.org/10.3390/ijms. 22041675.
- Ireland, W.F.D., 2018. The EU Water Framework Directive (WFD) [WWW Document]. Introd. to Dir. http://www.wfdireland.ie/wfd.html. (Accessed 24 July 2018).
- Who, 2003. Total dissolved solids in drinking-water background document for development of WHO guidelines for drinking-water quality. Switzerland, Geneva.
- Wichmaan, S., Krebs, M., Kumar, S., Gaudig, G., 2020. Paludiculture on former bog grassland: profitability of sphagnum farming in north West Germany. Mires Peat 20, 1–18.
- Wichmaan, W., Schroder, C., Joosten, H. (Eds.), 2016. Paludiculture Productive Use of Wet Peatlands: Climate Protection – Biodiversity – Regional Economic Benefits. Schweizbart Science Publishers, Stuttgart.
- Williams, M., Eggleston, S., 2017. Using Indicators to Explain Our Changing Climate to Policymakers and the Public. World Meteorological Organization. https://public.wmo.int/en/ resources/bulletin/using-indicators-explain-our-changing-climate-policymakers-andpublic. (Accessed 1 June 2022).
- Ziegler, R., Wichtmann, W., Abel, S., Kemp, R., Simard, M., Joosten, H., 2021. Wet peatland utilisation for climate protection – an international survey of paludiculture innovation. Clean. Energy Environ. https://doi.org/10.1016/j.clet.2021.100305.
- Ziemann, D.A., Walsh, W.A., Saphore, E.G., Fulton-Bennett, K., 1992. A survey of water quality characteristics of effluent from Hawaiian aquaculture facilities. J. World Aquac. Soc. 23, 180–191.
- Živić, I., Marković, Z., Filipović-Rojka, Z., Živić, M., 2009. Influence of a trout farm on water quality and macrozoobenthos communities of the receiving stream (Trešnjica River, Serbia). Int. Rev. Hydrobiol. 94, 673–687.

Contents lists available at ScienceDirect



Science of the Total Environment



journal homepage: www.elsevier.com/locate/scitotenv

# Novel use of the alga *Pseudokirchneriella subcapitata*, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study



### Emer A. O'Neill\*, Neil J. Rowan, Andrew M. Fogarty

Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co., Westmeath, Ireland Department of Life & Physical Science, Faculty of Science & Health, Athlone Institute of Technology, Dublin, Road, Athlone, Co. Westmeath, Ireland

#### HIGHLIGHTS

#### G R A P H I C A L A B S T R A C T

- Fish farm effluent monitored using algal bioassay and physicochemical parameters.
- Alga (*Pseudokirchneriella subcapitata*), more responsive model than physicochemical parameters alone.
- Standard water quality parameters are not applicable to fish farm wastewater.
- Duckweed and constructed wetlands were under appreciated in wastewater treatment.
- Alga demonstrated potential use as early warning system for climate change.



#### ARTICLE INFO

Article history: Received 24 May 2019 Received in revised form 15 July 2019 Accepted 15 July 2019 Available online 17 July 2019

Editor: Damia Barcelo

Keywords: Ecotoxicological Bioassay Physicochemical Fish Monitoring Sustainability

#### ABSTRACT

Aquaculture is one of the fastest growing food producing industries in the world. This dramatic increase in growth has raised many environmental concerns. Evaluation of fish farm effluent is frequently assessed by physicochemical parameters. This approach indicates potential degradation caused by the effluent and not cumulative effects on aquatic ecosystems. This study investigated relationships between physicochemical parameters (temperature, pH, conductivity, nitrogen, phosphorus, oxygen and suspended solids), typically used to assess water quality with the *Pseudokirchneriella subcapitata* algal bioassay, which evaluated the potential ecotoxicological effects that freshwater fish farm effluent has on its receiving ecosystems and organisms. Influent and effluent samples were collected from a freshwater farm facility every two weeks from April 2018 to October 2018 in the Republic of Ireland. This monitoring period coincided with one of the warmest and driest periods recorded by meteorological stations in the Republic of Ireland. Physicochemical analyses were found to be similar to those in other farm studies. After exposure of algae to the effluent, stimulation of algal growth rates increased by >50%. This stimulation was observed during monitoring. Correlation studies identified a moderately strong relationship between algal stimulation and temperature (r = -0.619). This study discovered that removal of *Lemna minor* (aquatic plant), impacted strongly on the freshwater farm pond-process to cope with nitrates.

\* Corresponding author at: Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co., Westmeath, Ireland. *E-mail address*: e.oneill@research.ait.ie (E.A. O'Neill). The constructed wetland system was unable to efficiently treat nitrates and phosphates during conditions of drought. These findings indicate that standard water quality parameters may not be applicable to inform appropriate suitability of fish farm effluent for discharge to receiving water. The research conducted in this study has suggested a potential toolbox that includes *P. subcapitata* may provide an early warning system for adverse effects as a result of climate change.

© 2019 Elsevier B.V. All rights reserved.

#### 1. Introduction

Aquaculture is one of the fastest growing food producing industries in the world (Fečkaninová et al., 2017; Liu et al., 2017), and provides one of the most sustainable forms of edible protein (Liu et al., 2017; Yue and Wang, 2017). The dramatic increase in the growth of global aquaculture production indicate its importance in modern day food supply (Jegatheesan et al., 2011), by providing a means to meet the growth in global demand (Seoane et al., 2014). However, despite this and many other advantages (Jegatheesan et al., 2011; Martinez-Porchas et al., 2014), there are many environmental concerns thought to be associated with aquaculture (Martinez-Porchas et al., 2014; Ngo et al., 2016; Troell et al., 2017), and in particular to the impacts fish farm effluent is thought to have on the receiving aquatic ecosystem (Jegatheesan et al., 2011).

In 2017, the Irish aquaculture sector produced 47,147 t of fish valued at €208.4 M which was an increase of almost €100 M since 2014 (Bord Iascaigh Mhara, 2018). The Irish aquaculture sector is primarily based in coastal areas but land-based recirculating aquaculture systems (RAS), and more traditional freshwater and land based systems are also used (Department of Agriculture Food and the Marine, 2015), and are projected to potentially grow rapidly over the next 10 years. The sector provides valuable employment on a year round basis and aids in the preservation of viable local and rural communities (Department of Agriculture Food and the Marine, 2015). The Irish aquaculture sector can be divided into shellfish and finfish culture with rope mussels, bottom grown mussels, Gigas oysters, salmon and trout the main species cultured. (Department of Agriculture Food and the Marine, 2015; Bord Iascaigh Mhara, 2018). Irish aquaculture also produces a wide range of novel species, both in marine and freshwater, such as Abalone, sea urchins, seaweed and perch (Bord Iascaigh Mhara, 2018).

Fish farm effluent typically contains nutrient rich waste products (Jegatheesan et al., 2011; Martinez-Porchas et al., 2014; Ngo et al., 2016; Sikder et al., 2016), which if released untreated into water bodies can lead to water pollution (Jegatheesan et al., 2011; Sikder et al., 2016). It may cause direct negative effects such as eutrophication (Jegatheesan et al., 2011; Martinez-Porchas et al., 2014; Ngo et al., 2016; Sikder et al., 2016; Troell et al., 2017), which is one of the greatest concerns (Ngo et al., 2016). Eutrophication is a process by which a water body receives large levels of nutrients and organic matter that can be taken in and biologically processed (Martinez-Porchas et al., 2014; Sikder et al., 2016), which in turn can result in increased levels of algal blooms and decreased levels of oxygen which can suffocate aquatic life in the water body (Jegatheesan et al., 2011; Chislock et al., 2013; Ngo et al., 2016). Fish farm effluent is commonly characterised by rich levels of nutrients, such as nitrogen and phosphorous, and organic matter (Jegatheesan et al., 2011; Ngo et al., 2016). These characteristics are as a result of uneaten or left over food and metabolic waste products such as faeces and urea (Jegatheesan et al., 2011; Martinez-Porchas et al., 2014; Ngo et al., 2016; Troell et al., 2017). In fact, the primary source of nitrogen, phosphorous and organic matter in the water is derived from the application of feed (Jegatheesan et al., 2011), which is necessary to maintain high production yields in order to meet demands (Kolarevic et al., 2014; Feucht and Zander, 2015). Another source of organic matter is the influent used to fill ponds or raceways and maintain the farms water levels (Jegatheesan et al., 2011).

The evaluation of fish farm effluent water quality is frequently assessed by the measurement of physicochemical parameters (da Silva et al., 2017), which include nitrogen, phosphorus, oxygen, suspended solids, temperature, pH, alkalinity, hardness and conductivity. Ammonium  $(NH_{4}^{+})$  is highly toxic to aquatic life (Zhang et al., 2011), and requires treatment before its release into its receiving water body (Celik et al., 2001). Nitrite (NO<sub>2</sub><sup>-</sup>), is very toxic to aquatic life (Pollice et al., 2002), but highly unstable and only remains in this form during for a short period of time during the transformation of  $NH_4^+$  to nitrate ( $NO_3^-$ ), (Durborow et al., 1997). Orthophosphate  $(PO_4^{3-})$ , a reactive form of phosphorus (Brogan et al., 2001), is one of the main causes of algal blooms and the hypoxic conditions which may occur in water bodies (Brogan et al., 2001; Barcellos et al., 2019). Biochemical oxygen demand (BOD), is the amount of oxygen used by bacteria in breaking down organic matter in the water (Lee and Nikraz, 2015). Chemical oxygen demand (COD), measures the stress a quantity of organic matter puts on a receiving water body (Lee and Nikraz, 2015). Suspended solids often consist of organic matter and elevated levels can be an indicator of eutrophic conditions (Bilotta and Brazier, 2008). Temperature is a critical environmental factor for fish farming due to its effect on growth, metabolism, survival, immune responses and oxygen consumption (Ferreira et al., 2011). Calcium carbonate (CaCO<sub>3</sub>), improves conditions for benthic animals and microbial activity and increases CO<sub>2</sub> (Ferreira et al., 2011). The alkalinity is the buffering capacity of the water body and is related to important factors in fish farming and water hardness is the amount of dissolved calcium and/or magnesium present in the water (Ferreira et al., 2011). Investigation of these parameters alone only indicate the potential degradation caused by the effluent at a given time, not their effects on aquatic ecosystems and organisms (Stephens and Farris, 2004a, 2004b; da Silva et al., 2017). Thus, ecotoxicological bioassays are used in conjunction with physicochemical analysis however, there are few studies that assess the toxic effects of fish farm effluent on aquatic ecosystems and organisms (da Silva et al., 2017).

Planktonic microalgae are primary producers and are a key component in the food chains of aquatic ecosystems (Aruoja, 2011). Sphaeropleales are one of the most dominant groups of green microalgae in the world and contain species that are considered to be very important to freshwater ecosystems (Suzuki et al., 2018). According to Rodgher et al. (2012), these microalgae constitute an important group of highly sensitive photosynthetic organisms that are frequently used to assess aquatic ecosystems. One such species in this group is Pseudokirchneriella subcapitata (P. subcapitata), CCAP 278/4, also commonly known as Raphidocelis subcapitata or Selenastrum capricornutum (Aruoja, 2011; Rodgher et al., 2012; Yamagishi et al., 2017; Suzuki et al., 2018). P. subcapitata is a unicellular algae and is consider the most widely known and used bio indicator in ecotoxicological assessments of freshwater ecosystems due to its high growth rate, high sensitivity and high reproducibility (Suzuki et al., 2018). As such, P. subcapitata has become one of the species of choice for primary producer assessment in multi-trophic testing and is the main algae recommended by the ISO (6892:2012), guidelines (International Organisation for Standardisation, 2012).

Globally, aquaculture is dominated by freshwater farming and the bulk of the production is finfish (Wang et al., 2015). The aim of this study was to conduct an ecotoxicological evaluation of Irish freshwater finfish farm effluent using the *P. subcapitata* algal bioassay ISO (8692:2012), in conjunction with traditional physicochemical analysis. Correlation studies were also conducted to determine any potential relationships between the algae and the physicochemical parameters.

#### 2. Materials & methods

#### 2.1. Sampling

Water samples were collected from a freshwater fish farm located in Boyle, Co. Roscommon (Fig. 1). The farm, which cultures perch (Perca fluviatilis), consisted of an RAS used for the hatchery and nursery, three culture ponds for the adult fish that uses a flow through system (FTS), a settlement pond and a constructed wetland for wastewater treatment. Samples were collected in 5 L octagonal carboy HDPE bottle (Lennox), and transported directly to the lab, 70 km away, via car. Samples were taken directly from the effluent source of the farm every two weeks from April 2018 to October 2018. Sampling could not be conducted in June due to unforeseen circumstances. Collection occurred on the same day (Thursdays), and at approximately the same time (10:30 a.m.). Influent samples were also collected and analysed so that any potential issues caused by works upstream of the fish farm and not as a result of works within the facility itself could be taken into consideration. Influent and effluent sampling points are shown in Fig. 2.

#### 2.2. Physicochemical analysis

Water parameters – temperature, pH,  $NH_4^+$ ,  $NO_2^-$ ,  $NO_3^-$ ,  $PO_4^{3-}$ , dissolved oxygen (DO), BOD, COD, suspended solids (SS), hardness, alkalinity and conductivity – were investigated in the laboratory within



**Fig. 1.** Map of Ireland indicating the location of the fish farm (yellow) and the closest weather stations (orange).



**Fig. 2.** Schematic of the fish farm layout. Location of the collection point for the influent (red X), and effluent (green X), are indicated. 1) Hatchery, 2) Nursery, 3–4) Mesocosms, 5–7) Culture ponds, 8) Settlement Pond, 9) Constructed Wetland, 10) Holding Tank, 11) River.

24 h of collection to prevent the need for preservation. Table 1 summarises the physicochemical methods employed in this study. Spectroquant® photometric kits were used to assess the NH<sub>4</sub><sup>+</sup>, NO<sub>2</sub><sup>-</sup>, NO<sub>3</sub><sup>-</sup>, PO<sub>4</sub><sup>3-</sup> and COD. Analysis was conducted as per the manufacturer's instructions. Temperature, pH and conductivity were analysed using a VWR pHenomenal<sup>TM</sup> MU 6100 L meter, VWR 111662–1157 pH probe and VWR CO11 conductivity probe. DO and BOD<sub>5day</sub> were analysed using a Jenway 9500 DO<sub>2</sub> meter and probe. Suspended solids were analysed via filtration using a Buchner flask and funnel. Hardness was assessed via titration using pH 10 buffer, Erichrome black and EDTA.

#### Table 1

Methods, including detection limits, used for the analyses of water quality parameters in the influent and effluent. Figure in brackets indicate the standard method number.

Parameter/variable	Method	Detection limit $(mg L^{-1})$
Temperature	Thermometer (2550-B)	-
рН	Membrane Electrode (2310-B)	-
Ammonium (NH <sub>4</sub> <sup>+</sup> )	Photometric (4500-NH <sub>3</sub> -F)	0.013-3.86
		2.6-193.0
Nitrite (NO <sub>2</sub> <sup>-</sup> )	Photometric (345–1)	0.007-3.28
Nitrate (NO <sub>3</sub> <sup>-</sup> )	Photometric (4500-NO <sub>3</sub> )	0.4-110.7
Phosphate $(PO_4^{3-})$	Photometric (4500-P-C)	0.007-15.3
		1.5-92.0
Dissolved oxygen	Membrane Electrode (4500-0 G)	-
Biochemical Oxygen demand (BOD)	Membrane Electrode (5210-B)	-
Chemical oxygen Demand	Photometric (5220-D)	0-150
(COD)		15-300
Suspended solids	Gravimetric (2540-D)	-
Hardness	Titrimetric (2340-C)	-
Alkalinity	Titrimetric (2320-B)	-
Conductivity	Electrical Conductivity	-
	(2510-A)	

Alkalinity was analysed by titration using phenolphthalein indicator, methyl orange indicator and hydrochloric acid.

#### 2.3. Toxicity testing

The freshwater unicellular green algae P. subcapitata, was used in the toxicity test. A starter culture of the P. subcapitata (Raphidocelis subcapitata), was obtained from The Culture Collection of Algae and Protozoa (CCAP 278/4; SAMS Limited, Scottish Marine Institute, Oban, Argyll, Scotland, U.K.), and grown in standard Jarworski's Medium under controlled conditions of 23  $^{\circ}C \pm 2 ^{\circ}C$  exposed to continuous illumination (lux 6000-10,000). Sub-culturing was conducted every two to three days to ensure the growth rate remained in the exponential phase. Toxicity testing was conducted as per the Water quality - Fresh water algal growth inhibition test with unicellular green algae ISO (8692:2012), guidelines. The P. subcapitata was exposed to the influent and effluent samples for 72 h under static conditions at 23  $^{\circ}C \pm 2 ^{\circ}C$  exposed to the continuous illumination. The algae growth rate inhibition and stimulation, in percent, was calculated by comparing the samples to a negative control containing just the Jarworski's medium. Calculations were conducted as follows;

Algal cells mL<sup>-1</sup> = 
$$\frac{n}{0.02} \times 10^{3}$$

where n = number of cells counted using a haemocytometer

Average specific growth rate (
$$\mu$$
), =  $\frac{\ln X_n - \ln X_o}{T_n - T_o}$ 

where  $\ln = natural \log of$ 

 $X_n = Algae cells mL^{-1}$  at 72 h

 $X_o = Algae cells mL^{-1} at 0 h$ 

 $T_n = Duration of test$ 

 $T_0 = Time zero$ 

Percent inhibition in growth rate = 
$$\frac{C\mu - T\mu}{C\mu} \times 100$$

where  $C\mu$  = Average specific growth rate for control  $T\mu$  = Average specific growth rate for treatment

#### 2.4. Statistical analysis

All statistical analyses were conducted using GRAPHPAD PRISM 8 and MINITAB 18. The data generated were grouped and subject to normality test (Anderson-Darling). *t*-tests and one-way ANOVA with Tukey were used to identify significant differences in the variables. Pearson's correlation was used to assess any correlations between the algae and/or the physicochemical parameters.

#### 3. Results

#### 3.1. Physicochemical analysis

Mean concentrations determined for the physicochemical parameters investigated on Irish freshwater fish farm influent and effluent samples over the entire testing period are summarised in Table 2 and Fig. 3 provides a monthly breakdown of the results. With the exception of the pH, increases in concentrations between the influent and effluent were observed in all parameters. When the influent and effluent were compared via statistical analysis, a significant difference was observed in the NH<sup>4</sup><sub>4</sub>, NO<sup>2</sup><sub>2</sub>, NO<sup>3</sup><sub>3</sub>, PO<sup>4</sup><sub>4</sub> and DO levels (p < 0.05).

#### Table 2

Physicochemical parameters investigated on Irish freshwater finfish aquaculture influent and effluent samples from April 2018 to October 2018. Results are presented as means  $\pm$  S.D. \*Significant differences, where p < 0.05, are indicated (n = 12).

Parameter	Influent	Effluent	P value
Temperature (°C)	$14.76\pm2.53$	$15.53 \pm 2.66$	0.854
рН	$7.76 \pm 0.19$	$7.11 \pm 0.18$	0.709
Ammonium (mg $NH_4^+ L^{-1}$ )	$0.16\pm0.18$	$1.16\pm0.64$	0.001*
Nitrite (mg $NO_2^- L^{-1}$ )	$0.02\pm0.01$	$0.32\pm0.38$	< 0.001*
Nitrate (mg $NO_3^- L^{-1}$ )	$3.62 \pm 1.60$	$5.29\pm5.56$	0.006*
Orthophosphate (mg $PO_4^{3-}L^{-1}$ )	$1.76\pm0.84$	$3.78 \pm 2.00$	0.013*
DO (mg $O_2 L^{-1}$ )	$10.31\pm0.87$	$5.10 \pm 2.85$	< 0.001*
BOD (mg $O_2 L^{-1}$ )	$2.27 \pm 1.47$	$3.24 \pm 1.95$	0.185
$COD (mg O_2 L^{-1})$	$45.91\pm40.81$	$76.44 \pm 59.06$	0.230
Suspended Solids (mg $L^{-1}$ )	$40.17\pm79.08$	$83.67 \pm 144.33$	0.073
Hardness (mg CaCO <sub>3</sub> $L^{-1}$ )	$100.49\pm9.22$	$116.03 \pm 16.80$	0.092
Alkalinity (mg CaCO <sub>3</sub> $L^{-1}$ )	$122.55 \pm 17.71$	$128.91 \pm 18.19$	0.915
Conductivity ( $\mu$ S cm <sup>-1</sup> )	$247.18 \pm 57.82$	$298.17 \pm 57.12$	0.241

DO = Dissolved Oxygen, BOD = Biochemical Oxygen Demand, COD = Chemical Oxygen Demand.

#### 3.2. Algal bioassay

The *P. subcapitata* algal bioassay ISO (8692:2012), was performed on influent and effluent samples from the Irish freshwater fish farm every two weeks to determine whether or not the rate of growth inhibition or stimulation were observed as a result of exposure to either sample. The growth rate inhibition and stimulation observed in the influent and effluent can be found in Fig. 4. Statistical analysis was conducted between the influent and effluent and a significant difference was observed ( $p \leq 0.0001$  where < 0.05 indicates a significant difference).

#### 3.3. Correlation studies

The Pearson's correlation test (Table 3), demonstrated that in the effluent, the algae (*P. subcapitata*), was negatively correlated with temperature and suspended solids and positively correlated with alkalinity. Nitrite was positively correlated with orthophosphate. Temperature was negatively correlated with dissolved oxygen and positively correlated with conductivity. The parameters hardness and alkalinity were negatively correlated with orthophosphate. Dissolved oxygen was negatively correlated with conductivity and positively correlated with nitrate. Hardness was positively correlated with alkalinity and a negative correlation with nitrite. The pH had a negative correlation with conductivity.

#### 3.4. Weather conditions

The Republic of Ireland experienced one of its hottest summers on record in 2018 that coincided with the sampling period in the study (Met Eireann, 2018b). Drought conditions and a national hose pipe ban were put in place for most of the country up to the end of August 2018 (Irish Water, 2018). As a result of these unusual weather conditions mean rainfall and temperature data collected at three Met Eireann weather stations surrounding and closest to the fish farm in the Republic of Ireland (Fig. 1), were observed. These stations were located in; Markree, Knock and Mount Dillon. Decreases in the average monthly rainfall and increases in temperature were observed across the weather stations (Fig. 5). Statistical analysis suggested that moderately strong inverse relationship existed between the algae and the rainfall (r =-0.562), and a weak relationship between the algae and the temperature (r = 0.276). Fig. 6 compares the average temperatures observed across the stations and the average maximum temperatures experienced. When the maximum temperatures were taken into account, a moderately strong relationship between the algae and temperature was observed (r = 0.505).



**Fig. 3.** Bar charts displaying monthly means for the physicochemical parameters investigated on Irish freshwater finfish aquaculture influent and effluent from April 2018 to October 2018. Parameters investigated were A) temperature, B) pH, C)  $NH_4^+$ , D)  $NO_2^-$ , E)  $NO_3^-$ , F)  $PO_4^+$ , G) dissolved oxygen, H) Biochemical Oxygen Demand, I) Chemical Oxygen Demand, J) suspended solids, K) hardness and alkalinity via CaCO<sub>3</sub> content, and L) conductivity. S.D. indicated, n = 12. Red lines indicate levels set out by S.I. 272 of 2009, which is the European Communities Environmental Objective (Surface Waters) Regulations of 2009. Blue lines indicate levels set out by the Irish EPAs Parameters for water quality, which are based on Freshwater Fish Directive [78/659/EEC], Irish Salmonid Waters Regulations [1988], and Irish Surface Water Regulations [1989]. NOTE: The dilution factor of the receiving water body has not been taken into consideration. Lines do not appear on temperature, CaCO<sub>3</sub> content and conductivity as no limits were indicated and/or the limit is well above the range of the graph.



**Fig. 4.** The percentage growth rate inhibition (positive y scale), and stimulation (negative y scale), observed in *Pseudokirchneriella subcapitata* after exposure to the freshwater finfish aquaculture influent and effluent for 72 h at 23 °C  $\pm$  2 °C under continuous illumination. Samples were collected and analysed from April 2018 to October 2018. Dates of collection appear at the bottom of each pair of results. Individual results appear in the table. Positive figures indicate inhibition and negative figures indicate stimulation. (*n* = 3, SD indicated).

#### 4. Discussion

#### 4.1. Physicochemical evaluation

As far as the authors are aware, water quality parameters specific for fish farm effluent in Irish water are not currently available and the Irish EPA are currently investigating the regulation of Irish freshwater fish farm effluent. Recommended water quality parameters set out by the Statutory Instrument (S.I.), 272/2009 (European Communities Environmental Objectives – Surface Waters – Regulations 2009), and the Irish Environmental Protection Agency's (EPA), parameters of water quality were therefore used as guidance (Environmental Protection Agency, 2001; Irish Stationery Office, 2009a, 2009b). The parameters set out in the above-mentioned Irish EPA document are based on the Freshwater Fish Directive [78/659/EEC] and/or Irish Surface Water Regulations [1989].

Presence of the highly toxic  $NH_4^+$  (Zhang et al., 2011), is an indicator of recent pollution. When comparing both samples, the concentration of

#### Table 3

Correlation matrix for the algae (*Pseudokirchneriella subcapitata*), and physicochemical parameters investigated on the Irish freshwater finfish aquaculture effluent. Bold figures indicate where significant differences (p < 0.05), have been observed. Breakdown of correlation figures are indicated in box. Breakdown of correlation values is based on Ratner (2009), and his work correlation coefficient values.

	ALGAE	PH	TEMP	$\mathrm{NH}_4^+$	$NO_2^-$	$NO_3^-$	$PO_{4}^{3-}$	DO	BOD	COD	SS	HARD.	ALK.
ALGAE	1							0 = No re	lationship				
PH	0.112	1						>0-0.3 =	Weak relation	onship			
TEMP	-0.619	-0.480	1					0.3-0.5 =	Moderately	weak relatio	nship		
$NH_{4}^{+}$	-0.285	-0.437	0.414	1				0.5-0.7 =	Moderately	strong relati	onship		
$NO_2^-$	-0.307	-0.389	0.213	0.554	1			0.7 - < 1 =	= Strong rela	tionship	•		
$NO_3^-$	0.164	-0.468	-0.475	-0.346	-0.100	1		1 = Perfe	ct linear rela	tionship			
$PO_4^{3-}$	-0.393	-0.435	0.209	0.308	0.651	0.320	1	-		-			
DO	0.356	0.408	-0.846	-0.492	-0.298	0.578	-0.028	1					
BOD	-0.197	-0.307	0.008	-0.065	0.048	0.251	0.310	0.422	1				
COD	0.270	0.310	-0.216	-0.083	-0.430	-0.003	-0.088	0.004	-0.495	1			
SS	-0.727	0.228	0.244	-0.064	-0.191	-0.121	-0.139	0.069	0.399	-0.345	1		
HARD.	0.343	0.393	-0.021	-0.095	-0.580	-0.335	-0.885	-0.156	-0.199	0.085	0.115	1	
ALK.	0.607	0.224	0.034	-0.408	-0.493	-0.228	-0.657	-0.240	-0.359	0.098	-0.368	0.701	1
COND.	-0.292	-0.547	0.841	0.394	0.251	-0.485	0.138	-0.851	-0.227	-0.083	-0.174	0.101	0.344

Temp = Temperature, DO = Dissolved Oxygen, BOD = Biochemical Oxygen Demand, COD = Chemical Oxygen Demand, SS = Suspended Solids, Hard. = Hardness, Alk. = Alkalinity, Cond. = Conductivity.



Fig. 5. Mean rainfall in mm (A), and temperature in °C (B), collected from three Met Eireann weather stations located at; 1) Markree, Co. Sligo, 2) Knock, Co. Mayo, and 3) Mount Dillon, Co Roscommon. These stations were selected as they were located to the north-west, south-west and south-east of the fish farm investigated in this study. Data from April to October in 2017 (dark grey), and 2018 (light grey), were examined.

 $NH_{4}^{+}$  present in the influent is lower than that of the effluent. This suggested that the levels of NH<sup>+</sup><sub>4</sub> detected are being generated within the farm itself. However, the small amount detected in the influent also suggests some form of pollution is being generated upstream of the farm. The concentrations observed in the effluent were more than five times greater than the 1 mg  $L^{-1}$  suggested by the Irish EPA (Environmental Protection Agency, 2001). NO<sub>2</sub><sup>-</sup> levels detected were very low mainly due to its transient nature and quick transformation from  $NH_4^+$  to  $NO_3^-$  (Durborow et al., 1997). Similar with  $NH_4^+$ , the presence of  $NO_2^$ is an indication of recent pollution. When comparing the NO<sub>2</sub><sup>-</sup> levels in both samples, levels in the effluent were much higher than that of the influent. The small amount detected in the influent suggests some form of pollution is being generated upstream of the farm. The higher levels detected in the effluent are generated within the farm itself. The concentrations detected in the effluent were one log dose greater than the 0.03 mg  $L^{-1}$  for cyprinid waters, as per the Irish EPAs suggested water quality parameters (Environmental Protection Agency, 2001).  $NO_3^-$  levels observed in the influent indicated that  $NO_3^-$  may have been entering the river upstream of the fish farm as well as its expected generation within the farm. The aquatic plant Lemna minor (duckweed), was present in the farm and can use  $NO_3^-$  as a source of nutrients (Stewart and Rhodes, 1976). A spike in the effluent at the end of September 2018 was observed and may have been due to the fact that the Lemna minor within the farm had been removed. NO<sub>3</sub><sup>-</sup> levels were well below the guidance value of 50 mg  $L^{-1}$  suggested by the Irish EPA (Environmental Protection Agency, 2001). Lemna minor's ability to use  $NO_3^-$  as a nutrient source and the spikes observed in the farm



**Fig. 6.** Mean temperatures (Bar Chart), and maximum temperatures (Line Chart), observed at three Met Eireann weather stations located closest to the fish farm, from April 2018 to October 2018. Data was based on all temperature readings collected at the three weather stations by Met Eireann (Met Eireann, 2018a).

after its removal highlighted that the duckweed holds great potential as a wastewater treatment option for fish farms and further research is currently being conducted.

 $PO_4^{3-}$  was detected in both the influent and effluent. The levels observed in the influent were less than that in the effluent. This suggested that phosphorus pollution is entering the river upstream of the fish farm, as well as being generated within the farm itself. Levels in both the influent and effluent may be cause for concern due to its ability to cause algal blooms and the hypoxic conditions which may occur thereafter in water bodies. Concentrations detected were just over two log doses greater than the recommended value of 0.035 mg  $L^{-1}$  set out by the S.I. 272/2009 for good water status (Irish Stationery Office, 2009b). The farm uses a constructed wetland pond for treatment of nitrates, phosphates and so forth, before being released. Further research is currently being conducted in the efficacy of the constructed wetland by way of expansion of treatment to address culture area. This has been the subject of previous research (Jegatheesan et al., 2011; Sharrer et al., 2016), however it still remains unclear as to the governing factors underpinning the effective ratio for culture treatment of fish farm effluent.

The recommended DO concentration in salmonid waters should be  $\geq 9 \text{ mg L}^{-1}$  and cyprinid waters (e.g. perch) should be  $\geq 7 \text{ mg O}_2 \text{ L}^{-1}$ (Environmental Protection Agency, 2001). There are no issues with the DO levels present in the influent however there may be cause for concern with levels observed in the effluent as they were well below the recommended concentration of  $\geq 7 \text{ mg } O_2 \text{ L}^{-1}$ . Levels below this concentration were only observed during the heat wave and drought conditions, and the unusual weather conditions may have played a role. Conditions began to improve once weather conditions had returned normal. Alam et al. (2007), and da Silva et al. (2017), have suggested that oxygen concentrations of  $\geq 4 \text{ mg L}^{-1}$  are sufficient for maintenance of aquatic life. S.I. 272/2009 recommend a mean BOD concentration of 1.3 mg  $L^{-1}$  for high water status and 1.5 mg  $L^{-1}$  for good water status. However, the Irish EPA has suggested  $\leq 3 \text{ mg L}^{-1}$ and  $\leq 5 \text{ mg L}^{-1}$  for salmonid and cyprinid waters, respectively (Environmental Protection Agency, 2001). The current BOD levels detected in the influent suggested no issues. The concentration of BOD detected in the effluent may be cause for concern. Although the level was below that suggested by the Irish EPA for cyprinid water, it was greater than that suggested in the S.I. 272 of 2009. COD was detected in both the influent and effluent. The levels observed in both sets of samples may be cause for concern. The mean concentration was almost double the suggested 40 mg  $L^{-1}$  set out by the Irish EPA (Environmental Protection Agency, 2001).

Two concentrations of suspended solids have been suggested by the Irish EPA (Environmental Protection Agency, 2001). Fifty mg  $L^{-1}$  as per the Surface Water Regulations [1989], and 25 mg  $L^{-1}$  as per the Freshwater Fish Directive [78/659/EEC], and Salmonid Waters Regulations

[1988]. Suspended solids can increase gill irritation and blanket the benthos (Bilotta and Brazier, 2008), therefore the lower concentration of 25 mg  $L^{-1}$  was taken as the maximum allowable concentration (MAC), in this study of an Irish freshwater finfish farm. The average levels detected in both the influent and effluent may be cause for concern as they were above the 25 mg  $L^{-1}$ .

With growing concerns associated with climate change and global warming, increases in temperatures may become more frequent. Fluctuations in temperature were observed in both sets of samples. These rises in temperatures were only observed during the elevated temperatures experienced in Ireland in 2018. The results for the pH indicated that the influent was slightly more alkaline than the effluent, which the pH just above neutral (pH 7). The slight difference in pH levels in the samples may have been due to the alkalinity levels observed in the samples. The effluent had a higher level of CaCO<sub>3</sub> and thus a better buffering capacity. The recommended pH levels should be between 6 and 9. Levels in both the influent and effluent are well within this level and therefore are present no issues.

Alkalinity has been measured as CaCO<sub>3</sub>. For hardness, CaCO<sub>3</sub> levels were also measured. Results suggested that the water is slight to moderately hard. This correlates with water hardness maps of Ireland which demonstrated water around Boyle, Co. Roscommon was slightly too moderately hard.

Some of the results observed ( $NH_4^+$ ,  $NO_2^-$ ,  $PO_4^{3-}$ , COD and suspended solids), suggested potential issues however, the dilution factor of the receiving river has not been included and needs to be taken into consideration. Equally, the condition of the influent needs to be considered. Additionally, as this research only focused on one fish farm, the results observed in this study were compared to previous studies conducted on a range of different farm effluents including; brown and rainbow trout, Atlantic salmon, catfish, shrimp and prawns. All results obtained were similar to other research studies conducted (Ziemann et al., 1992; Camargo, 1994; Boaventura et al., 1997; Mcintosh and Fitzsimmons, 2003; Biao et al., 2004; Pulatsü et al., 2004; Stephens and Farris, 2004a, 2004b; Costanzo et al., 2004; Živić et al., 2009; Moreira et al., 2010; Guilpart et al., 2012; Namin et al., 2013; Noroozrajabi et al., 2013; Herbeck et al., 2013; Caramel et al., 2014; Lalonde et al., 2014; Ferreira et al., 2015; da Silva et al., 2017), with the exception of BOD. Although not many studies included BOD, those reviewed demonstrated higher levels than the concentrations detected in this study (Boaventura et al., 1997; Mcintosh and Fitzsimmons, 2003; Ansah et al., 2012; Miashiro et al., 2012).

#### 4.2. Ecotoxicological bioassay evaluation

Growth stimulation was observed in the effluent. Stimulation occurred in mid-April and then again from July to September. This coincided with the elevated temperatures and drought conditions experienced in the Republic of Ireland in the summer of 2018. Ireland's mean summer maxima temperature is between 18 °C and 20 °C (Walsh, 2012). In 2018, temperatures exceeded 30 °C (Met Eireann, 2018b). This resulted in low rainfall levels, e.g. the three weather stations (Markree, Knock and Mount Dillion), measured an average total rainfall of only 61.9 mm for the months of May, June and July 2018 compared to 88.9 mm for the same three months in 2017 (Met Eireann, 2018a), leading to a national hose pipe ban and water restrictions (Irish Water, 2018). The ability of the effluent to cause growth stimulation suggested that the possibility of algal blooms (resulting in eutrophication), downstream of the fish farm are more likely to occur. This may result in loss of biodiversity, habitat and submerged aquatic vegetation, disruption of the ecosystems functionality, deficiencies in oxygen and modifications in food webs (Rabalais, 2002).

A higher level of growth inhibition was observed in the influent compared to the effluent. This suggested that the influent would seem unlikely to cause issues such as algal blooms. However, the high level of growth inhibition in the influent also indicated toxicity and may

result in losses to the biodiversity of the receiving water body (Rabalais, 2002). This toxicity may result in the loss of primary producers (e.g. algae), in the aquatic ecosystem. This may subsequently cause indirect adverse effects on the aquatic food chain, e.g. microcrustaceans feed on algae and fish in turn, feed on the microcrustaceans. Loss of the algae removes the food source for the microcrustaceans, resulting in their potential loss. This in turn, could result in the removal of a valuable food source for the fish. It should be noted that this toxic effect does not occur within the fish farm itself and suggests potential issues upstream of the farm. This issue may also affect the health and welfare of the fish within the farm. This is being closely monitored by the owner and no adverse effects have been observed with the fish. Further investigation into the water quality entering the fish farm has also been conducted. An old school is situated just upstream of the fish farm that has an older waste water treatment system which may have resulted in some waste entering the river. As a result of this, water is now piped from upstream of the school directly into the farm in order to assist in determining the exact cause for the poor water quality.

Most of the available research involving *P. subcapitata* focused on inhibition of growth (Guéguen et al., 2004; Ivanova and Groudeva, 2006; Ma et al., 2006). One study involving *P. subcapitata* and fish farm effluent was published by Miashiro et al. (2012), demonstrated similar results to this study, i.e. growth stimulation instead of inhibition was observed. Miashiro et al. (2012), suggested that the stimulated algal growth may have been due to the high concentration of nutrients that were observed. High levels of nutrients, were also observed in this study. The lack of available research suggested an under use of *P. subcapitata* as an early indicator of potential issues in fish farming. Stimulation of *P. subcapitata* in the effluent may be indicative of potential eutrophication downstream of the fish farm.

#### 4.3. Climate change

"There's one issue that will define the contours of this century more dramatically than any other, and that is the urgent threat of a changing climate. Climate change is no longer some far-off problem; it is happening here, it is happening now" (Obama, 2015). Climate change is considered to be one of the most troubling, challenging and unrelenting scientific issues of our time (Bulkeley and Newell, 2015). Climate change, including global warming, is a complicated and increasingly problematic challenge causing changes to rainfall and hydrology, e.g. extensive summer droughts caused by changes in rainfall (Paerl and Scott, 2010; Paerl et al., 2016).

The results observed in this study have demonstrated that climate change may have a direct impact on fish farming, as suggested by the moderately strong relationship (r = -0.619), observed between the increases in temperature and the stimulation of algal growth rates, which could lead to increased instances of eutrophication downstream of the fish farm. *P. subcapitata* should be investigated further for its ability to be utilised as a potential early warning indicator of climate change.

#### 5. Conclusion

The use of *P. subcapitata* as an early warning indicator to potential environmental issues associated with fish farms has been demonstrated as a more responsive model than physicochemical parameters alone. Evaluation of farm effluent could include ecotoxicological bioassays in order to determine any potential effects the effluent may have on the receiving aquatic ecosystem. Inclusion of the *P. subcapitata* Algal Bioassay ISO (8692:2012), in this study has demonstrated the potential eutrophication implications as a result of releasing untreated effluent from fish farms. It should be noted that this research focused on only one fish farm and therefore additional research is currently being conducted. Other bioassays that focus on different trophic levels should also be

considered in order to develop a broader picture of the potential effect's fish farm effluent poses on its receiving ecosystems.

Water quality parameters specific for fish farm effluent has not yet been established in Ireland. The Irish EPA has begun the process of regulating fish farm effluent. Results observed in this study have demonstrated that water quality parameters suggested by S.I. 272/2009 and the Irish EPA may not be applicable to fish farm effluent. The dilution factor of the receiving aquatic ecosystem is important and therefore also needs to be taken into consideration when these water quality parameters are to be determined. Results have also indicated that influent water quality is also as important when assessing fish farm effluent as it may indicate potential environmental issues as a result of activities upstream and not that which is occurring within the farm.

The study has highlighted that on review of the performance of the constructed wetland, it is envisaged that it needs to be increased in order to be effective. Also highlighted is the importance and potential use of *Lemna minor* as a waste water treatment system within fish farming facilities.

Although this study focused on the ecotoxicological effects freshwater fish farm effluent may induce on its receiving aquatic ecosystem, the research suggested that the changes in temperatures that were observed during the heat wave and drought conditions experienced in the summer months of 2018 had a direct relationship with the increased levels of algal growth stimulation detected. With irregular weather patterns becoming more frequent, especially rises in mean temperatures due to global warming, further research into the effects of climate change on aquatic ecosystems, fish farm effluent and the effects of effluents on its receiving ecosystem will need to be conducted and is currently being investigated. The research conducted in this study has suggested a potential toolbox that includes *P. subcapitata* may provide an early warning system for adverse effects as a result of climate change.

#### **CRediT authorship contribution statement**

**Emer A. O'Neill:**Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Project administration, Resources, Software, Validation, Visualization, Writing - original draft, Writing - review & editing. **Neil J. Rowan:**Conceptualization, Funding acquisition, Methodology, Project administration, Resources, Supervision, Writing - original draft, Writing - review & editing. **Andrew M. Fogarty:** Conceptualization, Funding acquisition, Methodology, Project administration, Supervision, Writing - original draft, Writing - review & editing.

#### **Declaration of Competing Interest**

The authors declare that there are no competing interests or conflicts of interest with respect to the publication of this article.

#### Acknowledgements

The authors would like to thank Athlone Institute of Technology (President Seed Fund), for providing financial support for this research. The authors would also like to thank Paul Kearney and Keywater Fisheries for access to his freshwater aquaculture facility in order to conduct this research.

#### References

- Alam, M.J.B., Islam, M.R., Muyen, Z., Mamun, M., Islam, S., 2007. Water quality parameters along rivers. Int. J. Environ. Sci. Tech. 4 (1), 159–167. https://doi.org/10.1007/ BF03325974.
- Ansah, Y.B., Frimpong, E.A., Amisah, S., 2012. Biological assessment of aquaculture effects on effluent-receiving streams in Ghana using structural and functional composition of fish and macroinvertebrate assemblages. Environ. Manag. 50 (1), 166–180. https://doi.org/10.1007/s00267-012-9858-x.

- Aruoja, V., 2011. Algae Pseudokirchneriella Subcapitata in Environmental Hazard Evaluation of Chemicals and Synthetic Nanoparticles. Estonian University of Life Sciences Available at. http://citeseerx.ist.psu.edu/viewdoc/download?doi= 10.1.1.455.17&rep=rep1&type=pdf, Accessed date: 27 May 2019.
- Barcellos, D., Queiroz, H.M., Nóbrega, G.N., de Oliveira Filho, R.L., Santaella, S.T., Otero, X.L., Ferreira, T.O., 2019. Phosphorus enriched effluents increase eutrophication risks for mangrove systems in northeastern Brazil. Marine Pollution Bulletin. Pergamon 142, 58–63. https://doi.org/10.1016/J.MARPOLBUL.2019.03.031.
- Biao, X., Zhuhong, D., Xiaorong, W., 2004. Impact of the intensive shrimp farming on the water quality of the adjacent coastal creeks from eastern China. Mar. Pollut. Bull. 48 (5–6), 543–553. https://doi.org/10.1016/j.marpolbul.2003.10.006.
- Bilotta, G.S., Brazier, R.E., 2008. Understanding the influence of suspended solids on water quality and aquatic biota. Water Res. 42 (12), 2849–2861. https://doi.org/10.1016/J. WATRES.2008.03.018 Pergamon.
- Boaventura, R., Pedro, A.M., Coimbra, J., Lencastre, E., 1997. Trout farm effluents: characterisation and impact on the receiving streams. Environ. Pollut. 95 (3), 379–387. https://doi.org/10.1177/0363546515602009.
- Brogan, J., Crowe, M., Carty, G., 2001. Developing a National Phosphorus Balance for Agriculture in Ireland. Johnstown Castle Estate Available at. www.epa.ie, Accessed date: 4 April 2019.
- Bulkeley, H., Newell, P., 2015. Governing Climate Change. 2nd edn. Routledge-Taylor & Francis Group, London.
- Camargo, J.A., 1994. The importance of biological monitoring for the ecological risk assessment of freshwater pollution: a case study. Environ. Int. 20 (2), 229–238. https://doi. org/10.1016/0160-4120(94)90140-6.
- Caramel, B., Moraes, B.P., Carmo, M.A.B., Vaz-Dos-Santos, C.F., Tabata, A.M., Osti, Y.A., Ishikawa, J.A.S., Cerqueira, C.M., Mercante, M.A.S., 2014. Water quality assessment of a trout farming effluent, Bocaina, Brazil. Journal of Water Resource and Protection 6, 909–915. https://doi.org/10.4236/jwarp.2014.610086.
- Celik, M.S., Ozdemir, B.H., Turan, M.E., Koyuncu, I., 2001. Removal if ammonia by natural clay minerals using fixed and fluidised bed column reactors. Water Sci. Technol. Water Supply 1 (1), 81–88. https://doi.org/10.2166/ws.2001.0010.
- Chislock, M.F., Doster, E., Zitomer, R.A., Wilson, A.E., 2013. Eutrophication: causes, consequences, and controls in aquatic ecosystems/learn science at Scitable. Nature Education Knowledge 4 (4), 10. https://doi.org/10.1007/978-90-481-9625-8.
- Costanzo, S.D., O'Donohue, M.J., Dennison, W.C., 2004. Assessing the influence and distribution of shrimp pond effluent in a tidal mangrove creek in north-east Australia. Mar. Pollut. Bull. 48 (5–6), 514–525. https://doi.org/10.1016/J.MARPOLBUL.2003.09.006 Pergamon.
- da Silva, A.Q., Nilin, J., Santaella, S.T., Bonilla, O.H., 2017. Ecotoxicological and physicochemical evaluation of an effluent of a shrimp farm located in northeastern Brazil. Pan-American Journal of Aquatic Sciences 12 (4), 263–272.
- Department of Agriculture Food and the Marine, 2015. National Strategic Plan for Sustainable Aquaculture Development. Dublin. Available at. https://www.agriculture.gov.ie/ media/migration/customerservice/publicconsultation/ sustainableaquaculturedevelopment/

NatStratPlanSustAquaculDevelopdraftconsult100615.pdf, Accessed date: 26 January 2018.

- Durborow, R.M., Crosby, D.M., Brunson, M.W., 1997. Nitrite in Fish Ponds. Stoneville, Mississippi. Available at:. https://appliedecology.cals.ncsu.edu/wp-content/uploads/ SRAC-0462.pdf, Accessed date: 4 April 2019.
- Environmental Protection Agency, 2001. Parameters of Water Quality-Interpretation and Standards. Johnstown Castle Available at. https://www.epa.ie/pubs/advice/water/ quality/Water\_Quality.pdf, Accessed date: 6 October 2018.
- Fečkaninová, A., Koščová, J., Mudroňová, D., Popelka, P., Toropilová, J., 2017. The use of probiotic bacteria against Aeromonas infections in salmonid aquaculture. Aquaculture 469 (1), 1–8. https://doi.org/10.1016/j.aquaculture.2016.11.042.
- Ferreira, N.C., Bonetti, C., Seiffert, W.Q., 2011. Hydrological and water quality indices as management tools in marine shrimp culture. Aquaculture 318 (3–4), 425–433. https://doi.org/10.1016/j.aquaculture.2011.05.045.
- Ferreira, J.G., Falconer, J.G., Kittiwanich, J., Ross, L., Saurel, C., Wellman, K., Zhu, C.B., Suvanachai, P., 2015. Analysis of production and environmental effects of Nile tilapia and white shrimp culture in Thailand. Aquaculture https://doi.org/10.1016/j. aquaculture.2014.08.042.
- Feucht, Y., Zander, K., 2015. Of earth ponds, flow-through and closed recirculation systems – German consumers' understanding of sustainable aquaculture and its communication. Aquaculture 438, 151–158. https://doi.org/10.1016/J. AQUACULTURE.2015.01.005 Elsevier.
- Guéguen, C., Gilbin, R., Pardos, M., Dominik, J., 2004. Water toxicity and metal contamination assessment of a polluted river: the Upper Vistula River (Poland). Appl. Geochem. 19 (1), 153–162. https://doi.org/10.1016/S0883-2927(03)00110-0.
- Guilpart, A., Roussel, J.M., Aubin, J., Caquet, T., Marle, M., Le Bris, H., 2012. The use of benthic invertebrate community and water quality analyses to assess ecological consequences of fish farm effluents in rivers. Ecol. Indic. 23, 356–365. https://doi.org/ 10.1016/j.ecolind.2012.04.019.
- Herbeck, L.S., Unger, D., Wu, Y., Jennerjahn, T.C., 2013. Effluent, nutrient and organic matter export from shrimp and fish ponds causing eutrophication in coastal and backreef waters of NE Hainan, tropical China. Cont. Shelf Res. 57, 92–104. https://doi. org/10.1016/j.csr.2012.05.006.
- International Organisation for Standardisation, 2012. ISO (8692:2012) water quality fresh water algal growth inhibition test with unicellular green algae, ISO 8692:2012 (en) water quality – fresh water algal growth inhibition test with unicellular green algae. Available at. https://www.iso.org/obp/ui/#iso:std:iso:8692:ed-3:v1:en, Accessed date: 9 April 2018.
- Irish Stationery Office, 2009a. European Communities Environmental Objectives (Surface Waters) Regulations., Statutory Instruments (Ireland).

Irish Stationery Office, 2009b. S.I. No. 272/2009-European Communities Environmental Objectives (Surface Waters) Regulations 2009, Electronic Irish Statute Book. Office of the Attorney General Available at. http://www.irishstatutebook.ie/eli/2009/si/ 272/made/en/print, Accessed date: 5 April 2019.

- Irish Water, 2018. Water Conservation Order (Hosepipe Ban) Extended in Some Areas until 30 September to Safeguard Water Supplies|Press Release|Irish Water, News. Available at. https://www.water.ie/news/water-conservation-order-/, Accessed date: 8 October 2018.
- Ivanova, I., Groudeva, V., 2006. Use of Selenastrum capricornutum growth inhibition test for testing toxicity of metal ions in soil and water. Biotechnology & Biotechnological Equipment 20 (1), 179–183. https://doi.org/10.1080/13102818.2006.10817329.
  Jegatheesan, V., Shu, L., Visvanathan, C., 2011. Aquaculture Effluent: Impacts and Reme-
- Jegatheesan, V., Shu, L., Visvanathan, C., 2011. Aquaculture Effluent: Impacts and Remedies for Protecting the Environment and Human Health. Elsevier B.V, New York, pp. 123–135 https://doi.org/10.1016/B978-0-444-52272-6.00340-8.
- Kolarevic, J., Baeverfjord, G., Takle, H., Ytteborg, E., Reiten, B.K.M., Nergård, S., Terjesen, B.F., 2014. Performance and welfare of Atlantic salmon smolt reared in recirculating or flow through aquaculture systems. Aquaculture 432 (1), 15–25. https://doi.org/ 10.1016/J.AQUACULTURE.2014.03.033 Elsevier.
- Lalonde, B.A., Ernst, W., Garron, C., 2014. Chemical and physical characterisation of effluents from land-based fish farms in Atlantic Canada. Aquac. Int. 23 (2), 535–546. https://doi.org/10.1007/s10499-014-9834-y.
- Lee, A.H., Nikraz, H., 2015. BOD: COD ratio as an Indicator for river pollution. International Proceedings of Chemical, Biological and Environmental Engineering 88 (1), 89–94. https://doi.org/10.7763/IPCBEE.
- Liu, X., Steele, J.C., Meng, X.Z., 2017. Usage, residue, and human health risk of antibiotics in Chinese aquaculture: a review. Environ. Pollut. 223 (1), 161–169. https://doi.org/ 10.1016/j.envpol.2017.01.003 Elsevier Ltd.
- Ma, J., Wang, S., Wang, P., Ma, L., Chen, X., Xu, R., 2006. Toxicity assessment of 40 herbicides to the green alga Raphidocelis subcapitata. Ecotoxicol. Environ. Saf. 63 (3), 456–462. https://doi.org/10.1016/j.ecoenv.2004.12.001.
- Martinez-Porchas, M., Martinez-Cordova, L.R., Lopez-Elias, J.A., Porchas-Cornejo, M.A., 2014. Bioremediation of aquaculture effluents. Microbial Biodegradation and Bioremediation, 1st edn Elsevier, London, pp. 542–555 https://doi.org/10.1016/B978-0-12-800021-2.00024-8.
- Mcintosh, D., Fitzsimmons, K., 2003. Characterization of effluent from an inland, lowsalinity shrimp farm: what contribution could this water make if used for irrigation. Aquac. Eng. 27 (1), 147–156. https://doi.org/10.1016/S0144-8609(02)00054-7.
- Met Eireann, 2018a. Monthly Data, Climate. Available at:. https://www.met.ie/climate/ available-data/monthly-data, Accessed date: 8 October 2018.
- Met Eireann, 2018b. Warm and Dry Weather of June and July 2018 Met Éireann-the Irish Meteorological Service, Latest News. Available at:. https://www.met.ie/warm-anddry-weather-of-june-and-july, Accessed date: 8 October 2018.
- Mhara, Bord Iascaigh, 2018. BIM Annual Aquaculture Survey 2018. Dublin. Available at. http://www.bim.ie/media/bim/content/publications/aquaculture/BIM-Annual-Aquaculture-Survey-2018.pdf, Accessed date: 31 May 2018.
- Miashiro, L, Lombardi, J.V., Mercante, C.T.J., 2012. Ecotoxicity assessment in aquaculture system using the test organism Pseudokirchneriella subcapitata (Chlorophyceae). Acta Scientiarum. Biological Sciences 34 (4), 373–379. https://doi.org/10.4025/ actascibiolsci.v34i4.9391.
- Moreira, L.E.B., Lombardi, J.V., Mercante, C.T.J., Bazante-Yamaguishi, R., 2010. Ecotoxicological assessment in a pond of freshwater shrimp farming, using the Cladocera Ceriodaphnia dubia as test organism. Bol. Inst. Pesca 36 (1), 25–38.
- Namin, J.I., Sharifinia, M., Makrani, A.B., 2013. Assessment of fish farm effluents on macroinvertebrates based on biological indices in Tajan River (North Iran). CJES Caspian Journal of Environmental Sciences Caspian J. Env. Sci. 11 (1), 29–39.
- Ngo, H.H., Guo, W., Tram Vo, T.P., Nghiem, L.D., Hai, F.I., 2016. Aerobic treatment of effluents from the aquaculture industry. Current Developments in Biotechnology and Bioengineering. Elsevier, Amsterdam, Netherlands, pp. 35–77 https://doi.org/10.1016/ B978-0-444-63665-2.00002-3.
- Noroozrajabi, A., Ghorbani, R., Abdi, O., Nabavi, E., 2013. The impact of rainbow trout farm effluents on water physicochemical properties of Daryasar stream. World Journal of Fish and Marine Sciences 5 (3), 342–346. https://doi.org/10.5829/idosi. wjfms.2013.05.03.72175.
- Obama, B., 2015. COP 21 Speech. United Nations Climate Change Conference. United Nations, Paris.
- Paerl, H.W., Scott, J.T., 2010. Throwing fuel on the fire: synergistic effects of excessive nitrogen inputs and global warming on harmful algal blooms. Environmental Science & Technology. American Chemical Society 44 (20), 7756–7758. https://doi.org/ 10.1021/es102665e.
- Paerl, H.W., Gardner, W.S., Havens, K.E., Joyner, A.R., McCarthy, M.J., Newell, S.E., Qin, B., Scott, J.T., 2016. Mitigating Cyanobacterial Harmful Algal Blooms in Aquatic

Ecosystems Impacted by Climate Change and Anthropogenic Nutrients. https://doi.org/10.1016/j.hal.2015.09.009.

- Pollice, A., Tandoi, V., Lestingi, C., 2002. Influence of aeration and sludge retention time on ammonium oxidation to nitrite and nitrate. Water Res. 36, 2541–2546. https://doi. org/10.1016/S0043-1354(01)00468-7.
- Pulatsü, S., Rad, F., Topçu, A., 2004. The impact of rainbow trout farm effluents on water quality of Karasu stream, Turkey. Turk. J. Fish. Aquat. Sci. 4 (1), 9–15 Available at:. https://www.researchgate.net/publication/284773287, Accessed date: 5 February 2019.
- Rabalais, N.N., 2002. Nitrogen in aquatic ecosystems. Ambio 31 (2), 102–112. https://doi. org/10.1639/0044-7447(2002)031[0102,NIAE]2.0.CO;2.
- Ratner, B., 2009. The correlation coefficient: its values range between +1/-1, or do they? Journal or Targeting, Measurement and Analysis for marketing 17 (2), 139–142. https://doi.org/10.1057/jt.2009.5.
- Rodgher, S., Espíndola, E.L.G., Simões, F.C.F., Tonietto, A.E., 2012. Cadmium and chromium toxicity to pseudokirchneriella subcapitata and microcystis aeruginosa. Braz. Arch. Biol. Technol. 55 (1), 161–169. https://doi.org/10.1590/S1516-89132012000100020.
- Seoane, M., Rioboo, C., Herrero, C., Cid, Á., 2014. Toxicity induced by three antibiotics commonly used in aquaculture on the marine microalga Tetraselmis suecica (Kylin) Butch. Mar. Environ. Res. 101 (1), 1–7. https://doi.org/10.1016/j. marenvres.2014.07.011.
- Sharrer, K.L., Christianson, L.E., Lepine, C., Summerfelt, S.T., 2016. Modeling and mitigation of denitrification "woodchip" bioreactor phosphorus releases during treatment of aquaculture wastewater. Ecol. Eng. 93 (1), 135–143. https://doi.org/10.1016/j. ecoleng.2016.05.019.
- Sikder, M.N.A., Min, W.W., Ziyad, A.O., Prem Kumar, P., Dinesh Kumar, R., 2016. Sustainable treatment of aquaculture effluents in future-a review. International Research Journal of Advanced Engineering and Science 1 (4), 190–193 Available at:. http:// www.irjaes.com/pdf/IRJAES-V1N4Y16/IRJAES-V1N4P209Y16.pdf, Accessed date: 4 February 2019.
- Stephens, W.W., Farris, J.L., 2004a. A biomonitoring approach to aquaculture effluent characterization in channel catfish fingerling production. Aquaculture 241 (1–4), 319–330. https://doi.org/10.1016/J.AQUACULTURE.2004.08.007 Elsevier.
- Stephens, W.W., Farris, J.L., 2004b. Instream community assessment of aquaculture effluents. Aquaculture 231 (1-4), 149–162. https://doi.org/10.1016/J.AQUACUL-TURE.2003.08.009 Elsevier.
- Stewart, G.R., Rhodes, D., 1976. Evidence for the assimilation of ammonia via the glutamine pathway in nitrate-grown Lemna minor L. FEBS Lettes 64 (2), 296–299 Available at:. https://core.ac.uk/download/pdf/82570174.pdf, Accessed date: 4 April 2019.
- Suzuki, S., Yamaguchi, H., Nakajima, N., Kawachi, M., 2018. Raphidocelis subcapitata (= Pseudokirchneriella subcapitata) provides an insight into genome evolution and environmental adaptations in the Sphaeropleales OPEN. Nature 8 (8058), 1–13. https:// doi.org/10.1038/s41598-018-26331-6.
- Troell, M., Kautsky, N., Beveridge, M., Henriksson, P., Primavera, J., Rönnbäck, P., Folke, C., Jonell, M., 2017. Aquaculture. Reference Module in Life Sciences https://doi.org/ 10.1016/B978-0-12-809633-8.02007-0.
- Walsh, S., 2012. A Summary of Climate Averages for Ireland. Dublin. Available at. www. met.ie, Accessed date: 8 October 2018.
- Wang, Q., Cheng, L., Liu, J., Li, Z., Xie, S., de Silva, S.S., 2015. Freshwater aquaculture in PR China: trends and prospects. Rev. Aquac. 5 (1), 1–20. https://doi.org/10.1111/ raq.12086.
- Yamagishi, T., Yamaguchi, H., Suzuki, S., Horie, Y., Tatarazako, N., 2017. Cell reproductive patterns in the green alga Pseudokirchneriella subcapitata (Selenastrum capricornutum) and their variations under exposure to the typical toxicants potassium dichromate and 3,5-DCP. PLoS One 12, e0171259.
- Yue, G.H., Wang, L., 2017. Current status of genome sequencing and its applications in aquaculture. Aquaculture 468, 337–347. https://doi.org/10.1016/j.aquaculture.2016.10.036 Elsevier B.V..
- Zhang, M., Wang, Z., Xu, J., Liu, Y., Ni, L., Cao, T., Xie, P., 2011. Ammonium, Microcystins, and Hypoxia of Blooms in Eutrophic Water Cause Oxidative Stress and C-N Imbalance in Submersed and Floating-Leaved Aquatic Plants in Lake Taihu, China. https://doi. org/10.1016/j.chemosphere.2010.10.038.
- Ziemann, D.A., Walsh, W.A., Saphore, E.G., Fulton-Bennett, K., 1992. A survey of water quality characteristics of effluent from Hawaiian aquaculture facilities. J. World Aquacult. Soc. 23 (3), 180–191. https://doi.org/10.1111/j.1749-7345.1992.tb00767.x.
- Živić, I., Marković, Z., Filipović-Rojka, Z., Živić, M., 2009. Influence of a trout farm on water quality and Macrozoobenthos communities of the receiving stream (Trešnjica River, Serbia). Internat. Rev. Hydrobiol. 94 (6), 673–687. https://doi.org/10.1002/ iroh.200811137.

273

Contents lists available at ScienceDirect





## Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

# Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: Case study from the Republic of Ireland



Emer A. O'Neill<sup>a,\*</sup>, Gustavo Fehrenbach<sup>a</sup>, Emma Murphy<sup>a</sup>, Sérgio A. Alencar<sup>b</sup>, Robert Pogue<sup>a,b</sup>, Neil J. Rowan<sup>a</sup>

<sup>a</sup> Bioscience Research Institute, Technological University of the Shannon: Midlands Midwest, University Road, Athlone, Co. Westmeath, Ireland <sup>b</sup> Universidade Católica de Brasilia, QS 7 LOTE 1 - Taguatinga, Brasília, DF 71966-700, Brazil

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- Integrated multi-trophic aquaculture (IMTA) will sustain food systems for peatlands.
- IMTA needs better understanding of algae for bioremediation and water quality.
- Reliance on traditional monitoring methods limits innovation and sustainability.
- Use of next generation sequencing for algae profiling will advance paludiculture.
- IMTA system delivers new 'Green' innovation balanced with environmental protection.

#### ARTICLE INFO

Editor: Damià Barceló

Keywords: Algae Aquaculture Food security Ecosystem Sustainability Next-generation sequencing



#### ABSTRACT

Development of integrated multi-trophic aquaculture (IMTA) systems constitutes a step change in the sustainable production of freshwater fish to meet emerging needs for high-protein foods globally. Recently, there has been a paradigm shift away from harvesting peat as a fuel towards the development of wettable peatland innovation (termed 'paludiculture'), such as aquaculture. Such eco-innovations support carbon sequestration and align with a balanced environmental approach to protecting biodiversity. This novel peatland-based IMTA process in the Irish midlands relies upon natural microalgae for waste treatment, recirculation and water quality where there is no use of pesticides or antibiotics. This novel IMTA system is powered with a wind turbine and the process has 'organic status'; moreover, it does not discharge aquaculture effluent to receiving water. However, there is a significant lack of understanding as to diversity of microalgae in this 'paludiculture'-based IMTA processes. This constitutes the first case study to use conventional microscopy combined with next-generation sequencing and bioinformatics to profile microalgae occurring in this novel IMTA system from pooled samples over a 12 month period in 2020. Conventional microscopy combined with classic identification revealed twenty genera of algae; with Chlorophyta and Charophyta being the most common present. However, algal DNA isolation, 16 s sequencing and bioinformatics revealed a combined total of 982 species from 341 genera across nine phyla from the same IMTA system, which emphasized a significant underestimation in the number and diversity of beneficial or potentially harmful algae in the IMTA-microbiome. These new methods also yield rich data that can be used by digital technologies to transform future monitoring and performance of the IMTA system for sustainability. The findings of this study align with many sustainability development goals of the United Nations including no poverty, zero hunger, good health and well-being, responsible consumption and production, climate change, and life below water.

\* Corresponding author.

E-mail address: emer.oneill@ait.ie (E.A. O'Neill).

http://dx.doi.org/10.1016/j.scitotenv.2022.158392

Received 19 May 2022; Received in revised form 25 August 2022; Accepted 25 August 2022 Available online 30 August 2022

0048-9697/© 2022 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### 1. Introduction

Aquaculture is the fastest growing food producing industry in the world (O'Neill et al., 2019, 2020; Naughton et al., 2020). It now dominates aquatic food production (Tacon, 2020) with >50 % of all fish produced for human consumption coming from aquaculture (FAO, 2020; O'Neill et al., 2022). This is expected to exceed 62 % by 2030 (O'Neill and Rowan, 2022). Depletion of wild fishery practices as a result of over exploitation of wild fish stocks, which are now at their maximum sustainable yields, and increased consumer demand as a result of a growing global population has attributed to the dramatic increase in aquaculture (Tahar et al., 2018a, 2018b; Han et al., 2019; O'Neill and Rowan, 2022). Aquaculture is also providing an essential and important means of food security, both directly and indirectly (O'Neill et al., 2022). Farmed fish are a highly rich source of protein and provide a more efficient protein utilisation and feed conversion source than other animals destined for protein production (Tschirner and Kloas, 2017; O'Neill et al., 2022). Previously limited or non-existent access to fish is now available in under-developed countries and regions, often at cheap prices; thus, providing improved nutrition and food security in addition to increased employment (Rowan and Casey, 2021; O'Neill et al., 2022).

At the beginning of 2020, the Irish aquaculture industry produced just over 38,238 t of fish valued at €173.5 Million for the national economy (Dennis et al., 2020). Food Wise 2025 is a strategy developed by the Department of Agriculture, Food and the Marine (DAFM) for the Irish agrifood sector. This ten-year plan set out and underlines the sectors position in the Irish economy. It also illustrated the potential for expansion within the sector (DAFM, 2015a). Food Wise 2025 predicted that the Irish agri-food sector has the potential to increase exports to €19 Billion, per annum by 2025. As part of this prediction, it proposed that the Irish aquaculture industry, or more specifically aquaculture production, could be increased to 81,700 t by 2023 in order to assist in meeting this goal (DAFM, 2015a). However, issues with the Irish aquaculture licensing process, associated with the adoption of European Union environmental protection directives resulting in space limitations have hampered the growth and the development of the industry (Moylan et al., 2017). There is now therefore, an increasing interest in exploiting low-cost environmental-friendly 'natural' processes in aquaculture (Han et al., 2019). For example, the aforementioned aquaculture issues have led to an increased research focus on developing integrated multi-trophic aquaculture systems or IMTA which many believe can help mitigate these impacts (Granada et al., 2016; Ingle et al., 2022; O'Neill et al., 2022) along with eco-innovation and monitoring of traditional processes (Tahar et al., 2018a, 2018b; Rowan, 2019). Advances in aquaculture must also be balanced with the need to meet commitments as set out by the Water Framework Directive (WFD), which aims to achieve good water status in all waters across all EU member countries (Voulvoulis et al., 2017; WFD Ireland, 2018a, 2018b; O'Neill et al., 2022). Development of new high protein fish production process using natural resources also offers potential solutions to meet many of the sustainable development goals of the United Nations (Rowan and Casey, 2021).

As part of Ireland's Strategic Plan for Sustainable Aquaculture Development, and in addition to their research into further sustainable development of traditional aquaculture processes such as flow-through systems (FTS), Bord Iascaigh Mhara (BIM), Ireland's seafood development agency, undertook a feasibility study to assess the novel use of peatlands for aquaculture diversification (DAFM, 2015b). The urgent threat of climate change, in addition to some of these peatlands now being listed as important habitats under the EU's Birds and Habitats Directives due to their scarcity, have resulted in dramatic changes in the peat industry including conversion of peatland usage to wind energy, forestry, biodiversity, amenity and waste management (Toner, 2018; Bord na Móna, 2019a: 2019b; Irish Peatland Conservation Council, 2019; O'Neill et al., 2019; Ward et al., 2019). With that, BIM further expanded the potential use of these cutaway bogs to develop Ireland's first IMTA system adhering to organic principles in order to assist in developing the Irish freshwater aquaculture industry in a sustainable manner. This trial IMTA holds European perch

(*Perca fluviatilis*), rainbow trout (*Oncorhynchus mykiis*), common duckweed (*Lemna minor*) and gibbous duckweed (*Lemna gibba*) and exploits use of microalgae for waste removal (Bord na Mona, 2019b). This IMTA process differs from traditional aquaculture practices that use water from rivers and lakes where the latter traditional systems must consider potential pollutants from agricultural runoff, industry, waste-water treatments plants and so forth. (Rowan, 2011; Tahar et al., 2017; Tiedeken et al., 2017; Tahar et al., 2018c). This IMTA process consists of four culture ponds with eight compartments for fish and a duckweed lagoon that has sixteen channels where there are airlifts for water movement and oxygen supply and paddlewheel aerators move water between fish culture ponds and the duckweed areas (O'Neill et al., 2020).

It is only in recent years that studies have been conducted, confirming the potential beneficial roles of microalgae in aquaculture (Gao et al., 2016; Ansari et al., 2017; Han et al., 2019). Microalgae could efficiently assimilate nutrients providing a good method for wastewater remediation (Wang et al., 2015a, 2015b; Leng et al., 2018; Han et al., 2019) in aquaculture, having already demonstrated promising performances in the food and agriculture industries, and in municipal wastewater treatment (De-Bashan et al., 2004; Lu et al., 2015, 2017; Han et al., 2019). Microalgae synthesise high value compounds e.g., proteins, lipids and pigments (Han et al., 2019). Studies conducted by Ansari et al. (2017), Lu et al. (2017) and Sirakov et al. (2015) have also demonstrated the application of various microalgal species for the production of biomass which could be exploited as a partial feed replacement and to enhance aquatic animal immunity. Due to the potential of these benefits, the use of microalgae in aquaculture has recently emerged into the forefront. However, the role of algae in aquaculture is still lacking (Han et al., 2019; O'Neill et al., 2019, 2020).

The overarching aim of this study was to conduct a preliminary study to determine what populations of algae were present within the novel IMTA system in order to establish what species of algae could be utilised to benefit the system and what species could potentially contribute to adverse effects so as to better inform the management and development of the process. Therefore, the objectives were (a) to establish ability of traditional approaches in determining numbers and types of microalgae in this IMTA process; and (b) to compare the efficacy of next-generation sequencing and bioinformatics to discern diversity and numbers of microalgae from pond microbiome in same system. The hypothesis is that traditional approaches used for monitoring algae in novel IMTA underestimate actual diversity and numbers of algae that limits intensive sustainability of this process.

#### 2. Methods

#### 2.1. Study site

'Oasis' fish farm is an innovative peatland cut-away integrated multitrophic aquaculture (IMTA) system process set in the middle of Mount Lucas Wind Farm, Co. Offaly (53°17'3" N - 7°11'45" W). The closed/ semi-closed aquaculture system consists of four split (pill) earthen ponds, culturing European perch (Perca fluviatilis) in one and rainbow trout (Oncorhynchus mykiis) in three, utilizing glacial till that were connected to an algae and duckweed lagoon with sixteen channels serving as a treatment system. See Fig. 1. Fish are kept at a density that does not exceed the organic farming standard (e.g.,  $< 20 \text{ kg/m}^{-3}$  for perch), using screens at the D-ends of each split pond. An average of approximately 3000 kg of feed was applied per 10,000 kg biomass per month. The space between two D-end fish culture areas is also used to treat waste with free living algae in suspension. Flow in each split pond is generated and water is circulated using an airlift. Each Dend fish culture area is equipped with oxygen and temperature probes connected to paddlewheels to provide extra oxygen when necessary. The farm is designed to hold a maximum of 32,000 Kg of fish. As this is a cutaway site, the peat was previously removed by harvesting over many years. What is left underneath is the original glacial till from which the aquaculture ponds were created; therefore, to the best of our knowledge, no oxygenation of peat occurs. The algae and aquatic plants that grow in the cultured fish ponds sink to the bottom sequester carbon and create a new sink.

E.A. O'Neill et al.



**Fig. 1.** Ariel view of trial fish farm. Sampling points have been indicated (green). Flow of water throughout the system has been included (blue). Image shows the four D-end culture ponds, the reservoir and the 16-channel duckweed and algae treatment lagoon. The turbine, which provides all electrical inputs for the farm is also visible.

#### 2.2. Sample collection, preservation and preparation

Five litre water samples were collected from the trial IMTA fish farm in 5 L octagonal carboy HDPE bottles (Lennox) and transported directly to the lab, 62 km away, *via* car in insulated boxes. Samples were collected biweekly from December 2019 to February 2020 and then once a week from March 2020 to October 2020 for the identification of algal species within the novel IMTA system. Samples were collected on the same day (Wednesday) at approximately the same time (8:30 a.m.). The sampling regime changed due to the development of unforeseen technical issues within some of the culture ponds and increased monitoring was required. Samples were taken from each of the culture ponds, the entry and exit points of the treatment lagoon. Samples from the reservoir began in June due to the commencement of culturing in it as a result of issues being observed in the culture ponds. See Fig. 1 for sampling locations. Testing began within one hour of collection.

Samples were preserved in order to minimise the loss of the biological composition and maintain as close to *in-situ* conditions as possible (Nachimuthu et al., 2020). Based on the success Naughton et al. (2020), Guillard and Sieracki (2005) and, Noble and Fuhrman (1998) observed with preservation methods, 1 % Lugol's iodine was used to preserve the samples. For each of the five L grab samples taken from the individual locations within the farm, 500 mL samples were placed into 500 mL carboy HDPE bottles (Lennox) and 1 mL of the 1 % Lugol's iodine was added. All samples were mixed well and stored at 4 °C until analysis was conducted. Analysis was conducted on both fresh and preserved samples.

#### 2.3. Physicochemical measurement

The physicochemical parameters (temperature, pH, ammonium (NH<sub>4</sub><sup>+</sup>), nitrite (NO<sub>2</sub><sup>-</sup>), nitrate (NO<sub>3</sub><sup>-</sup>), orthophosphate (PO<sub>4</sub><sup>3-</sup>), dissolved oxygen (DO), biochemical oxygen demand (BOD), suspended solids (TSS), dissolved solids (TDS), hardness, alkalinity and conductivity) were investigated in the laboratory within 24 h of collection to prevent the need for preservation. Temperature, pH, dissolved solids and conductivity were analysed using a VWR pHenomenal<sup>™</sup> MU 6100 L meter, VWR 111662–1157 pH probe and VWR CO11 conductivity probe. DO and BOD<sub>5day</sub> were analysed using a Jenway 9500 DO<sub>2</sub> meter and probe. Spectroquant® photometric kits were used to assess the NH<sub>4</sub><sup>+</sup>, NO<sub>2</sub><sup>-</sup>, NO<sub>3</sub><sup>-</sup> and PO<sub>4</sub><sup>3-</sup>. Analysis was conducted as per the manufacturer's instructions. (See Supplementary Data 1 – Method Breakdown). Absorbance was analysed using a Shimadzu UV-2250 spectrophotometer. Suspended solids were analysed *via* filtration using a Buchner flask, Buchner funnel and Whatman 0.45  $\mu$ m pore membrane filter. Hardness was assessed *via* titration using pH 10 buffer, Erichrome black and EDTA. Alkalinity was analysed by titration using phenolphthalein indicator, methyl orange indicator and HCl.

#### 2.4. Microscopic analysis

Microscopic analysis was conducted as per Naughton et al. (2020) with some modifications. Six 5 mL aliquots were taken from the sample and placed into the wells of a six well plate. This was conducted for each sampling point. Analysis was also conducted on both the fresh and preserved samples as Lugol's iodine can cause changes to cells size (Hawkins et al., 2005) and abundance (Zarauz and Irigoien, 2008). During times of high algal concentration, aliquots were diluted to 1:2, 1:5 and 1:10. The plates were left to sit for 48 h which allowed for the algae to settle out. Plates were then examined extensively using an Olympus CKX41 inverted microscope. Images were viewed at  $100 \times$  and  $400 \times$ . Twelve images per well were taken using the ISCapture software to ensure as many algae as possible could be observed and identified. These images were then analysed as per Bellinger and Sigee (2015a, 2015b, 2015c) whereby the physical features, including; size, shape, colour, morphology and motility, were recorded and then identified using identification keys and cross comparison images from the Algae Base data bank (Algae Base, 2022).

#### 2.5. Isolation of algal DNA, 18S sequencing and bioinformatics

Algal samples were centrifuged and re-suspended in 1 mL of water, before sonication at 120 W/40 kHz in a sonicator bath (Cuyson) at 50 °C for 15 min. Algal genomic DNA was isolated using the High Pure PCR Template Preparation kit (Roche). Molecular profiling was carried out by two different methods. First, 18S DNA was sequenced commercially (Macrogen, South Korea) on the HiSeq platform (Illumina), using a 300-cycle paired-end protocol, yielding an average of ~100,000 reads per sample.

Secondly, 50–100 ng of DNA was used in a PCR reaction for amplification of a 1800 bp segment of the 18S ribosomal DNA sequence, using primers described by Khaw et al. (2020), modified with adapter sequences for subsequent sequencing on the Oxford Nanopore MinIon platform (adaptor sequences underlined):

#### (F): <u>TTTCTGTTGGTGCTGATATTGC</u>GGTGATCCTGCCAGTAGTCAT ATGCTTG

#### (R): <u>ACTTGCCTGTCGCTCTATCTTC</u>GATCCTTCCGCAGGTTCACCT ACGGAAACC

Following quality assessment of the Illumina fastq files using FastQC tool (http://www.bioinformatics.babraham.ac.uk), low quality reads were trimmed with Trimmomatic (Bolger et al., 2014). Then the Qiime 2 taxonomic classification pipeline was used in order to obtain classifications at species level (Bokulich et al., 2018).

Sequencing on the MinIon platform was carried out using the PCR Barcoding Amplicons (SQK-LSK-109) protocol, according to the manufacturer's instructions. The average yield obtained was 154,374 per sample.

Quality assessment of the fastq files obtained from MinIon Nanopore sequencing data was carried out using the FastQC tool (http://www.bioinformatics.babraham.ac.uk). Adapters were removed using Porechop (v0.2.4, https://github.com/rrwick/Porechop), which trims off adapters on the ends of reads, and when a read has an adapter in its middle, it is treated as chimeric and chopped into separate reads. The Fastp tool (PMID: 30423086) was then used in order to remove low quality reads, followed by quality assessment once again with FastQC, which confirmed that poor quality bases were removed. The resulting high-quality reads

#### Table 1

Average concentrations for all physicochemical parameters monitored in Mount Lucas fish farm from December 2019 to October 2020. Measuring units and standard error of the mean (SEM) have been included, n = 10. *P* value <0.05 indicates a statistically significant difference. Water quality guidance values from Irish EPA water quality, Statutory Instrument (SI) 272/2009 and SI 77/2019 have been included.

Parameter	Average	SEM	P Value	Guidance Value
Alkalinity (mg CaCO <sub>3</sub> $L^{-1}$ )	124.33	1.41	0.4806	_
Ammonium (mg $NH_4^+ L^{-1}$ )	0.10	0.03	0.5520	< 1
BOD (mg $O_2 L^{-1}$ )	13.36	1.97	0.5464	< 6
Conductivity ( $\mu$ S cm <sup>-1</sup> )	261.44	3.69	0.3271	< 1000
Dissolved Oxygen (mg $O_2 L^{-1}$ )	7.26	0.22	0.1421	> 7
Dissolved Solids (mg $L^{-1}$ )	167.78	2.54	0.3172	< 300
Hardness (mg CaCO <sub>3</sub> $L^{-1}$ )	105.26	1.77	0.5237	-
Nitrate (mg $NO_3^- L^{-1}$ )	2.65	0.32	0.1000	< 50
Nitrite (mg $NO_2^- L^{-1}$ )	0.07	0.01	0.4867	< 0.03
Orthophosphate (mg $PO_4^{3-}L^{-1}$ )	0.05	0.03	0.2160	< 0.03
рН	7.77	0.06	0.9952	> 6 - < 9
Suspended Solids (mg $L^{-1}$ )	38.46	5.24	0.8604	< 25
Temperature (°C)	12.81	0.87	0.1671	< 20

were then clustered using the isONclust software (v0.0.4, https://github. com/ksahlin/isONclust). Each cluster, representing all reads that came from a gene, was then aligned against the National Center for Biotechnology Information (NCBI) nucleotide database (excluding fungi data in the search) using BLASTn with an e-value of 0.05.

#### 2.6. Statistical analysis

Statistical analysis was conducted in order to determine whether any statistically significant differences were observed between any of the variables monitored throughout the study. All statistical analysis were performed on GRAPHPAD PRISM 9, and MINITAB 18. The data generated were grouped and subject to normality tests (Anderson-Darling), to determine if samples were from a normal distribution (p > 0.05 = normal distribution). Parametric testing was then applied. One-way and two-way ANOVA were used to determine if any significant differences were observed in the parameters investigated (p < 0.05 = significant difference). Unpaired tests were used as different sets of samples were analysed to assess the quality of the aquaculture water samples. Grubbs test was used to determine if any outliers were indicated.

#### 3. Results

#### 3.1. Physicochemical analysis

Bi-weekly physicochemical analysis was conducted. Statistical analysis indicated no significant differences within each set of physicochemical parameters (Table 1). Therefore, results were grouped together for ease of reporting. An average of all physicochemical parameters monitored throughout the preliminary study on the trial fish farm can be found in Table 1. Established parameters were compared to guidance water quality parameters as none currently exist in Ireland based on Irish freshwater aquaculture. However, the Irish Environmental Protection Agency (EPA) are now actively reviewing this. Due to limitations, composite sampling could not be conducted. All parameters, with the exception of BOD (13.36 mg  $O_2 L^{-1}$ , P = 0.5464), nitrite (0.07 mg  $NO_2^- L^{-1}$ , P = 0.4867), orthophosphate (0.05 mg  $PO_4^{3-}L^{-1}$ , P = 0.2160) and suspended solids  $(38.46 \text{ mg L}^{-1}, \text{P} = 0.2160)$  were within the guidance values parameters (Table 1). Additional aeration and a filtration system was added to aid in reducing these parameters, after which the farm workers reported a reduction in BOD, nitrite and suspended solids. The orthophosphate remained just above the guidance value. Additional research is currently being conducted in order to reduce levels in an environmentally sustainable manner. Ideally, composite samples taken every hour across a 24 h period would provide a much clearer picture of the physicochemical composition of the farm. This should be noted when interpreting these results.

#### 3.2. Microscopic identification

Partial speciation was first conducted on all samples collected using microscopy. As similar findings were observed at all of the sampling points, results were grouped together as a whole for ease of reporting. Approximately 20 genera of algae were identified using microscopy and classic identification keys (see Table 2). The most common type of algae present was green algae (Chlorophyta and Charophyta) with a minimum of 12 genera observed (*Scenedesmus, Monoraphidium, Micractinium, Chlorella, Chlamydomonas, Pediastrum, Dictyosphaerium, Closterium, Actinastrum, Pandorina, Oocystis* and *Ankistrodesmus*). A total of 5 genera of brown algae (Ochrophyta) were observed, 4 of which were diatoms (*Cyclotella, Nitzschia, Stephanodiscus* and *Tabellaria*). Finally, *Euglena* (Euglenophyta) which are commonly referred to as naked algae, the dinoflagellate *Peridinium* (Pyrrophyta)

#### Table 2

Breakdown of the genera of algae easily identifiable under microscopic examination. Breakdown includes all sampling points collected from the novel IMTA fish farm between December 2019 and October 2020. The genus name, the individual algal phylum and types (based on pigmentation) to which they belong, the month their presence was observed and the recording of multiple species have been included.

Genus Identified	Phylum	Algae Type	Month Present	Multiple Species
Actinastrum	Chlorophyta	Green Alga	6–10	Unknown
Ankistrodesmus	Chlorophyta	Green Alga	3–9	Unknown
Chlamydomonas	Chlorophyta	Green Alga	1–4, 7–11	Unknown
Chlorella	Chlorophyta	Green Alga	1–11	Yes
Closterium	Charophyta	Green Alga	1–11	Unknown
Cyclotella	Bracillariophyta	Diatom	1–4	Unknown
Dictyosphaerium	Chlorophyta	Green Alga	1–4, 6	Unknown
Euglena	Euglenozoa	'Naked' alga	3	Unknown
Mallomonas	Ochrophyta	Brown Alga	7	Unknown
Micractinium	Chlorophyta	Green Alga	1–11	Unknown
Monoraphidium	Chlorophyta	Green Alga	1–11	Yes
Nitzschia	Bracillariophyta	Diatom	1–11	Yes
Oocystis	Chlorophyta	Green Alga	3-4, 6-8	Unknown
Pandorina	Chlorophyta	Green Alga	2-4, 6-8	Unknown
Pediastrum	Chlorophyta	Green Alga	2, 7–8, 10	Unknown
Peridinium	Miozoa	DinoflagellateYellow-Brown Alga	1-4, 6-7, 10	Unknown
Rhodomonas	Cryptophyta	Nearly Brown Alga	1-4, 6-8	Unknown
Scenedesmus	Chlorophyta	Green Alga	1–11	Yes
Stephanodiscus	Bracillariophyta	Diatom	1–10	Unknown
Tabellaria	Bracillariophyta	Diatom	6–7, 10	Unknown

1 = Dec, 2 = Jan, 3 = Feb, 4 = Mar, 5 = Apr, 6 = May, 7 = Jun, 8 = Jul, 9 = Aug, 10 = Sep & 11 = Oct.

and *Rhodomonas* (Cryptophyta) were also observed. The majority of the species were observed throughout most the entire year. *Cyclotella* and *Dictyosphaerium* were only observed between December and March / May respectively. *Pandorina, Oocystis* and *Ankistrodesmus* were found between February and August. *Tabellaria* and *Actinastrum* were observed between May and September. *Mallomonas* was only found in June and *Euglena* only in February.

Microscopic analysis also indicated that at least 4 genera (*Chlorella, Monoraphidium, Nitzschia* and *Scenedesmus*) had multiple species present. Due to the similarities and complexities of the algae when observing them under the microscope, and the high volume and variation within the individual samples, an indication of whether multiple species were present for all genera could not be immediately established. Additionally, physiological similarities within individual genera also hindered the determination of full speciation, *e.g.* many species of *Scenedesmus* appeared to be visually very similar making it very hard to differentiate between the individual species. As such, molecular identification was subsequently conducted in order to; 1) confirm the presence of the genera identified using microscopy, 2) determine whether multiple species for the respective genera were also present, and 3) identify all species present in the system, including those that were not identifiable using microscopy.

#### 3.3. Molecular identification

Crossing the Illumina and MinION data, a combined total of 982 species from 341 genera across nine phyla were identified, as shown in Fig. 2. Nine species across two genera (0.92 %) of Haptophyta, 44 species across nine genera (4.48 %) of Cryptophyta (four of which had multiple species), 38 species across nine genera (3.87 %) of Euglenophyta or Euglenozoa (with only one indicating multiple species), two species across two genera (0.20 %) of Glaucophyta and four species across three genera (0.41 %) of Rhodophyta were identified. The four remaining phyla displayed the greatest populations. Fifty species of Pyrrophyta (Miozoa), or more specifically dinoflagellates, were identified across nineteen genera (5.09 %), three genera of which were found to have multiple species. A total of 304 species of Ochrophyta were identified across 85 genera (30.96 %). Of that, 177 species across 53 genera (18.02 %) were found to be diatoms (Bracillariophyta). Six genera of Ochrophyta and nine genera of diatoms had multi species. Some 45 species of Charophyta were identified across 21 genera (4.58 %), and 486 species of Chlorophyta were identified across 191 genera (49.49 %) making it the most common phylum of algae present in the system. Only two genera of Charophyta had multi species found. However, Chlorophyta had nineteen genera with multi species.

Given the high volume of species identified, only genera where multiple species (>4) were included in the phylogenetic tree (Fig. 2) for ease of reporting. Again, for ease of reporting, only the most common genera and species, as well as those considered to potentially be the most beneficial / potential hazardous were included (Zhou et al., 2009; Lucas et al., 2019; Yarnold et al., 2019; Lee and Ryu, 2021; Al-Hussieny, 2022). A full break-down of all species can be found in the Supplementary Data 2 – Species Breakdown. Due to limitations and restrictions as a result of the COVID-19 pandemic, samples could not be analysed for each individual month as well as for each location (culture ponds, treatment lagoon and reservoir). Therefore, samples for each location were pooled together. Samples had to be sent away for analysis which severely limited how much analysis could be requested.

As no variation was observed between the four culture ponds with respect to the genera and species identified, results were grouped together. The same applied for the entry and exit points of the treatment lagoon. See Table 3 for a breakdown of the most common genera identified in their respective locations. The culture ponds demonstrated the greatest variation of algal populations with all of the most common genera, with the exception of *Frustulia, Halamphora* and *Lagerheimi*a, identified. The treatment lagoon was found to have 37 of the 46 common genera and half (23) of the most common genera identified throughout the farm were found in the reservoir.

#### 4. Discussion

Although research into the use of algae in aquaculture is still limited, many of the most abundant species identified during this preliminary IMTA peatland study hold great potential for their application to organic, sustainable aquaculture. These included Scenedesmus, Monoraphidium, Micractinium, Chlorella, Chlamydomonas, Pediastrum, Dictyosphaerium, Closterium, Actinastrum, and Ankistrodesmus. The majority of the identified genera, including those previously mentioned, have not been known to cause any adverse effects on their ecosystems. However, two species have been previously found to contribute to adverse effects. Some Pandorina species have been known to excrete toxic compounds that inhibit the growth of other algae and higher plant life (Patterson and Harris, 2007). Some species of Oocystis are well known to cause harmful algal blooms or HAB's (Pal et al., 2020). With the exception of Peridinium, which is also well known to be an instigator of HAB's (Hallegraeff et al., 2004; Ki and Han, 2007), the rest of the genus identified in the samples are not known to cause negative effects on their environment. Although Pandorina can exhibit potentially negative effects by inhibiting beneficial algae and plants, it has been reported to also be an inhibitor of Peridinium, making its presence in the system potentially advantageous (Patterson and Harris, 2007). There is a commensurate interest on the impact of extreme climate events that may affect aquaculture processes where microalgae have a key role in process performance and regulation (Naughton et al., 2020; O'Neill et al., 2022).

Microalgae have been extensively studied as a means to support waste process remediation along with production of bioactives for OneHealth applications (Naughton et al., 2020; Rowan and Pogue, 2021). Satyanarayana et al. (2011) highlighted that microalgae are a promising resource due to their high production capacity of vegetable oils. These authors highlight that microalgae possess a high growth rate, need abundantly available solar light and  $CO_2$ , and thus are more photosynthetically efficient than oil crops. Also, they tolerate high concentration of salts allowing the use of any type of water for the agriculture and the possibility of production using innovative compact photobioreactors. Tan et al. (2020) revealed the potential enormity of high-value bio-compounds derived from microalgae such as lipids, carbohydrates, proteins, and other bioactive compounds from microalgae; however, large scale commercial production and extraction processes have only recently becoming established.

There has been growing interest in exploiting microalgae as a natural process for low-cost wastewater treatment and for water quality control and remediation in aquaculture (Naughton et al., 2020). These authors evaluated an Irish based freshwater flow-through aquaculture process for production of Eurasian Perch (Perca fluviatilis) in the Republic of Ireland and revealed the predominance of microalgae and cyanobacteria where Chlorophyta, Bracillariophyta and Cryptophyta were the most dominant algal phyla present. They showed that use of flow-cytometry correlated with in-field AlgaeTorch<sup>™</sup> for analysing microalgae in aquaculture. Findings from this present study however highlighted the enormous variability in microalgae species present that may be under appreciated by use of the aforementioned conventional microscopy, flow cytometry and in field AlgaeTorch handheld sensor. However, there is also significant merit in exploring the role of Artificial Intelligence and machine learning in the real-time analysis of large next-generation sequencing data sets that will support and enable decision making on the farm. Preventive risk mitigation to combat occurrence of unwanted cyanobacteria could entail growth of beneficial helper microalgae (such as Chlorella) in bioreactors and reintroducing these into the pond as preventive or counter-measure to combat emergence of undesirable cyanobacteria that can cause fish death.

In terms of this IMTA production process supported by microalgae and nexus to pond water quality, the established physicochemical parameters were compared to water quality parameters from the Irish EPA, which were based on the Freshwater Fish Directive [78/659/EEC] and Surface Water Regulations [1989] (EPA, 2001), and the Statutory Instrument (SI) European Communities Environmental Objectives (Surface Waters) Regulations 2009 [SI 272 of 2009] and amendments [SI 77 of 2019]



Glaucophyta (2G) (2S) Rhodophyta (3G) (4S)

**Fig. 2.** Phylogenetic tree identifying genera of algae found in the novel freshwater IMTA system between December 2019 and October 2020. (n) = number of species for that specific genera, (nG) = number of genera for that specific phylum, (nS) = number of species for that specific phylum. For ease or reporting only genera with multiple species have been included. Euglenophyta = Euglenozoa, Pyrrophyta = Miozoa, Ochrophyta (Diatoms) = Bracillariophyta.

(Irish Statutory Office, 2009, 2019). With the exception of four parameters (nitrite, orthophosphate, BOD and suspended solids), all other parameters were within range of the guidance and recommended values. The orthophosphate was above the guidance value of <0.03 mg  $PO_4^{3-} L^{-1}$ , as too was the nitrite guidance value of <0.03 mg  $NO_2^{-} L^{-1}$ . An increase in these parameters should increase algal growth within the system. However, issues with both BOD and suspended solid levels may prevent this. The BOD levels were well above the <3 mg  $O_2 L^{-1}$  guidance value for salmonid waters and suspended solid levels were above the guidance value of <25 mg

 $L^{-1}$ . Excess levels of BOD in the system reduces the level of oxygen available for both the algae and the fish. Whilst higher than normal levels of suspended solids affect the level of sunlight available for algae to photosynthesise and well as induce gill irritation within the fish. Mitigation levels were applied within the farm to aid in these issues. Additional aeration for oxygen generation was added by the introduction of more paddle wheels and air lifts, and filtration systems were introduced to reduce the level of suspended solids. Reduction in levels were observed after these changes were applied.

#### Table 3

Breakdown of the location of where the most common genera with multiple species that were identified during molecular analysis. • indicates presence of genus, grey indicates absence of genus.

Ganam		Location		Conoro	Location		
Genera	Pond	Lagoon	Reservoir	Genera	Pond	Lagoon	Reservoir
Achnanthidium	•			Mallomonas	•	•	•
Amphora	•	•		Micractinium	•	•	•
Ankistrodesmus	•	•		Monoraphidium	•	•	•
Bracteacoccus	•	•	•	Nannochloropsis	•	•	
Cercomonas	٠	٠		Nitzschia	•	٠	•
Chlamydomonas	•	•	•	Ochromonas	•	•	•
Chlorella	•	•	•	Oocystis	•	•	•
Chloromonas	•	•		Paraphysomonas	•		
Chroomonas	•	•	•	Peridinium	•		•
Chrysochromulina	•	•		Prorocentrum	•		
Coelastrella	•	•	•	Prymnesium	•		
Coelastrum	•	•	•	Pseudopediastrum	•	•	•
Coleochaete	•	•		Pyramimonas	•	•	
Cryptomonas	•	•		Rhodomonas	•	•	
Cyclotella	•	•	٠	Scenedesmus	•	•	•
Desmodesmus	•	•		Scrippsiella	•		
Dinobryon	٠	٠	•	Stauroneis	•	٠	
Frustulia		•	٠	Stephanodiscus	•	•	•
Halamphora		•		Stigeoclonium	•		
Hemiselmis	•	•	٠	Synura	•	•	
Kirchneriella	•	•		Tetracystis	•		•
Klebsormidium	•	٠		Tetradesmus	•		•
Lagerheimia		•		Thalassiosira	•	٠	٠

Increasing trends in global warming already evident, the likelihood of further rise continuing, and their impacts give urgency to addressing carbon sequestration technologies more coherently and effectively (Rowan and Galanakis, 2020; Galanakis et al., 2021). Carbon dioxide (CO<sub>2</sub>) is responsible for over half the warming potential of all greenhouse gases (GHG), due to the dependence of world economies on fossil fuels. Peatlands represent an important carbon sink (Rowan and Galanakis, 2020). However, increasing the trends in global warming are already evident where scientists predict that we have a fifty-fifty chance of experiencing a 1.5 °C rise in temperature by 2026 (Frost, 2022). Thus, the unprecedented impacts of same gives urgency to addressing carbon sequestration technologies more coherently and effectively. The processes involving CO2 capture and storage (CCS) are gaining attention as an alternative for reducing CO<sub>2</sub> concentration in the ambient air (Iglina et al., 2022); however, these technologies are considered as short-term solutions, as there are still concerns about the environmental sustainability of these processes (Singh and Ahluwalia, 2012). A promising technology could be the biological capture of  $CO_2$ using microalgae due to its unmatched advantages over higher plants that can be pursued through this peatland based IMTA system. Microalgae are phototrophic microorganisms with simple nutritional requirements that have strong potential to support carbon sequestration for our planet (Singh and Ahluwalia, 2012). Iglina et al. (2022) recently reported *Chlorella* species is the best microalgae they studied at capturing CO<sub>2</sub> using vertical tubular bioreactors. These authors noted that CO<sub>2</sub> emission accounts for about 77 % of all greenhouse gases, and the calculation of greenhouse gas emissions is 56 % of all CO<sub>2</sub> imports.

Development of new sustainable food production systems as described in this case study are aligned with many of the sustainable development goals of the United Nation's (Rowan and Casey, 2021). Given recent flux in food production and security brought on by stressors such as climate change, COVID-19 and conflict in Ukraine, there is likely to be a food supply chain challenge where development and replication of such IMTA systems across the peatlands globally will help address this challenge using a green solution (Galanakis et al., 2021). Digital transformation of IMTA process will improve monitoring, performance and protection of innovation for future sustainability that includes internet of things (IoT), cloud-edge enablers for remote sensing including use of drones and robots, artificial intelligence, immersive technologies for training, and use of blockchain for cybersecurity and business model development (Rowan et al., 2022; Rowan, 2022). However, there is also significant potential for future use of artificial intelligence with machine learning for evaluating NGS and bioinformatics data for performance and trends (Liakos et al., 2018; Sharma et al., 2020; Benos et al., 2021; Meshram et al., 2021). For example, digital technologies may also inform and enable fish pumping, fish counting, automated feeding along with real-time monitoring of physicochemical parameters such as free ammonia, temperature, oxygen and pH at this IMTA site. These findings will inform the adjacent Terrain-AI (2022) that uses drone and satellite technologies to uncover new insights into supporting effective climate change decision-making across peatlands with comparisons made to grasslands, cropland and forestry.

#### 5. Conclusion

Microalgae constitute an important element of the microbiome in aquaculture ponds where there is a reliance on the efficacy of these to maintain water quality in a novel IMTA system situated in the Irish peatlands. Detailed studies revealed significant difference in the abundance of microalga based upon use of conventional microscopy compared to that of using next generation sequencing and bioinformatics for the same ponds. Given the broad diversity, there is an opportunity to exploit AI and machine learning to monitor microalgae in this IMTA system that will inform real-time decision making. There is also a pressing need to understand better the dynamic interaction between microalgae and bacteria that includes provision for modelling and predicting the impact of extreme weather events on fish biomass production and disease mitigation. This IMTA system offers potential new solution towards high protein production that aligns with many of the UN's sustainable development goals.

#### 6. Future research

The preliminary study has demonstrated how even more complex the system is than originally thought. Therefore, the next phase of research will include; Analysis of each location on a monthly basis in order to establish a better understanding of the seasonal relationship between the populations of algae present in the system versus the physicochemical parameters necessary for growth, as well as the potential impacts the changes to weather will have. Analysis into what other organisms are present within the system in order to establish their role within the system and their relationship, if any, whether it be beneficial, hazardous or unaffected, to the growth and performance of algae e.g., microscopic analysis indicated the presence of rotifers within the system. A similar month-to-month profile of all cyanobacterial species present needs to be conducted, as well as the development of mitigation procedures in order to eliminate hazardous species. Potential combined use bioinformatics with artificial intelligence including machine learning to understanding the occurrence and variance of different microalgae in this IMTA pond in real time based on operational parameters and environmental conditions including extreme weather events given that this is an open plan recirculating aquaculture system. This is aligned with precision aquaculture and will inform decision making on process adjustment and farm management. Apply digital technologies to improve real-time monitoring and performance of IMTA process that includes data-analysis and protection for improved decision making. Future research will also focus on the combined use of massive sequencing data from this IMTA system to inform the importance of phytoplankton generated in the food chains of the peat bogs and if they serve as food for fish. This approach will also be expanded to include autotrophic species comprising periphyton that can potentially represent primary productivity, such as presence of cyanoprokaryotes in freshwater environments that are possibly toxic. Finally, an investigation into the development of organic and sustainable methods to help mitigate issues that arise from unwanted species, as well as a means to limit their future presence from the system without impacting the beneficial species present.

#### CRediT authorship contribution statement

Emer A. O'Neill: Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Project administration, Resources, Software, Validation, Visualization, Writing – original draft, Writing – review & editing. Gustavo Fehrenbach: Data curation, Methodology. Emma Murphy: Data curation, Methodology. Sérgio A. Alencar: Data curation, Formal analysis, Methodology, Software, Writing – review & editing. Robert Pogue: Data curation, Formal analysis, Methodology, Software, Writing – review & editing. Neil J. Rowan: Conceptualization, Formal analysis, Funding acquisition, Resources, Supervision, Writing – original draft, Writing – review & editing.

#### Data availability

The authors are unable or have chosen not to specify which data has been used.

#### Declaration of competing interest

The authors declare that there are no competing interests of conflicts of interest with respect to the publication of this article.

#### Acknowledgements

The authors acknowledge provision of aerial image of Mount Lucas peatland site furnished by Bord Iascaigh Mhara (BIM). The authors would like to thank Interreg Atlantic Area Neptunus (Project EAPA\_576/2018), Regional University Network European University (RUN-EU Project), and Bord Iascaigh Mhara (Project 2019 BIM-KGS-008) for funding support.

#### References

- AlgaeBase, 2022. Algaebase: Listing the World's Algae [WWW Document]. URLImages (accessed 5.5.22) https://www.algaebase.org/search/images/.
- Al-Hussieny, A.A., 2022. Algae toxins and their treatment. In: Zepka, L.Q., Jacob-Lopes, E., Depra, M.C. (Eds.), Microalgae [Working Title]. IntechOpen, London, p. 102909. https://doi.org/10.5772/INTECHOPEN.102909 (accessed 11/05/2022).
- Ansari, F.A., Singh, P., Guldhe, A., Bux, F., 2017. Microalgal cultivation using aquaculture wastewater: integrated biomass generation and nutrient remediation. Algal Res. 21, 169–177. https://doi.org/10.1016/j.algal.2016.11.015.
- Bellinger, E.G., Sigee, D.C., 2015a. Freshwater Algae: Identification, Enumeration and Use as Bioindicator. 2nd ed. John Wiley & Sons Ltd, Oxford, UK.
- Bellinger, E.G., Sigee, D.C., 2015b. A key to the more frequently occurring freshwater algae. Freshwater Algae: Identification and Use as Bioindicators. Wiley-Blackwell, West Sussex, pp. 137–244.
- Bellinger, E.G., Sigee, D.C., 2015c. Introduction. Freshwater Algae: Identification and Use as Bioindicators. Wiley-Blackwell, West Sussex, pp. 1–42.
- Benos, L., Tagarakis, A.C., Dollas, G., Berruto, R., Katerias, D., 2021. Machine learning in agriculture: a comphrensive updated review. Sensors 21 (11), 3758. https://doi.org/10. 3390/s21113758.
- Bokulich, N.A., Kaehler, B.D., Rideout, J.R., Dillon, M., Bolyen, E., Knight, R., Huttley, G.A., Gregory Caporaso, J., 2018. Optimizing taxonomic classification of marker-gene amplicon sequences with QIIME 2's q2-feature-classifier plugin. Microbiome 6 (1), 90. https://doi.org/10.1186/s40168-018-0470-z.
- Bolger, A.M., Lohse, M., Usadel, B., 2014. Trimmomatic: a flexible trimmer for illumina sequence data. Bioinformatics 30 (15), 2114–2120. https://doi.org/10.1093/bioinformatics/btu170.
- Bord na Mona, 2019. Bord na Mona Story [WWW Document]. URLAbout (accessed 7.30.19) https://www.bordnamona.ie/about/bord-na-mona-story/.
- Bord na Mona, 2019. Aquaculture | Bord na Móna [WWW Document]. URLAquaculture (accessed 11.11.19) https://www.bordnamona.ie/company/our-businesses/newbusiness-and-land-development/aquaculture/.
- DAFM, 2015. Local Roots Global Reach: Food Wise 2025. A 10-Year Vision for the Irish Agrifood Industry Dublin.
- DAFM, 2015. National Strategic Plan for Sustainable Aquaculture Development Dublin.
- De-Bashan, L.E., Hernandez, J.P., Morey, T., Bashan, Y., 2004. Microalgae growth-promoting bacteria as "helpers" for microalgae: a novel approach for removing ammonium and phosphorus from municipal wastewater. Water Res. 38 (2), 466–474. https://doi.org/ 10.1016/j.watres.2003.09.022.
- Dennis, J.H., Jackson, E., Burke, B., 2020. Annual Aquaculture Report 2019-2020 Dublin.
- EPA, 2001. Parameters of Water Quality Interpretation and Standards, pp. 1–133 Johnstown Castle.
- FAO, 2020. The State of World Fisheries and Aquaculture (SOFIA). Rome, Italy. https://www.fao.org/documents/card/en/c/ca9229en.

Frost, R., 2022. Fifty-fifty chance global warming hits 1.5°C by 2022, warn scientists. https:// www.euronews.com/green/2022/05/10/we-have-a-50-50-chance-of-touching-1-5c-inthe-next-5-years-warn-scientists (accessed 11/05/2022).

- Galanakis, C.M., Rizou, M., Aldawoud, T.M.S., Ucak, I., Rowan, N.J., 2021. Innovations and technology disruptions in the food sector with the COVID-19 pandemic and postlockdown era. Trends Food Sci. Technol. 110, 193–200. https://doi.org/10.1016/j.tifs. 2021.02.002.
- Gao, F., Li, C., Yang, Z.H., Zeng, G.M., Feng, L.J., Liu, J.Zhi, Liu, M., Cai, H.Wen, 2016. Continuous microalgae cultivation in aquaculture wastewater by a membrane photobioreactor for biomass production and nutrients removal. Ecol. Eng. 92, 55–61. https://doi.org/10.1016/j.ecoleng.2016.03.046.
- Granada, L., Sousa, N., Lopes, S., Lemos, M.F.L., 2016. Is integrated multitrophic aquaculture the solution to the sectors'major challenges? - a review. Rev. Aquac. 8, 283–300. https:// doi.org/10.1111/raq.12093.
- Guillard, R.R.L., Sieracki, M.S., 2005. Counting cells in culture with light microscopy. In: Andersen, R.A. (Ed.), Algal Culturing Techniques. Elsevier Academic Press, London, UK, pp. 239–252.
- Hallegraeff, G., Anderson, D.M., Cembella, A., Wegener, A., 2004. Manual of Harmful Marine Microalgae. 2nd ed. UNESCO Publishing.
- Han, P., Lu, Q., Fan, L., Zhou, W., 2019. A review on the use of microalgae for sustainable aquaculture. Appl. Sci. 9, 1–20. https://doi.org/10.3390/app9112377.
- Hawkins, P.R., Holliday, J., Kathuria, A., Bowling, L., 2005. Change in cyanobacterial biovolume due to preservation by Lugol's iodine. Harmful Algae 4 (6), 1033–1043. https://doi.org/10.1016/J.HAL.2005.03.001.
- Iglina, T., Iglin, P., Pashchenko, D., 2022. Industrional CO2 capture by algae; a review and recent advances. Sustainability. 14 (7), 3801. https://doi.org/10.3390/su14073801.
- Ingle, K.N., Polikovsky, M., Fenta, M.C., Ingle, A.S., Golberg, A., 2022. Integration of multitrophic aquaculture approach with marine energy projects for management and restoration of coastal ecosystems of India. Ecol. Eng. 176, 106525. https://doi.org/10. 1016/j.ecoleng.2021.106525.
- Irish Peatland Conservation Council, 2019. Raised Bogs in Ireland [WWW Document]. URLA to Z Peatlands (accessed 7.29.19) http://www.ipcc.ie/a-to-z-peatlands/raised-bogs/. Irish Statutory Office, 2009. S.I. No. 272/2009 - European Communities Environmental Ob-
- Irish Statutory Office, 2009. S.I. No. 272/2009 European Communities Environmental Objectives (Surface Waters) Regulations 2009 [WWW Document]. URLElectron. Irish Statut. B (accessed 4.5.19) http://www.irishstatutebook.ie/eli/2009/si/272/made/en/print.
- Irish Statutory Office, 2019. S.I. No. 77/2019 European Union Environmental Objectives (Surface Waters) (Amendment) Regulations 2019 [WWW Document]. URLElectron. Irish Statut. B (accessed 1.14.20) http://www.irishstatutebook.ie/eli/2019/si/77/ made/en/pdf.
- Khaw, Y.S., Khong, N.M.H., Shaharuddin, N.A., Yusoff, F.M., 2020. A simple 18S rDNA approach for the identification of cultured eukaryotic microalgae with an emphasis on primers. J. Microbiol. Methods 172, 105890. https://doi.org/10.1016/j.mimet.2020. 105890.
- Ki, J.S., Han, M.S., 2007. Rapid molecular identification of the harmful freshwater dinoflagellate peridinium in various life stages using genus-specific single-cell PCR. J. Appl. Phycol. 19, 467–470. https://doi.org/10.1007/s10811-007-9157-8.
- Lee, S.M., Ryu, C.M., 2021. Algae as new kids in the beneficial plant microbiome. Front. Plant Sci. 12, 91. https://doi.org/10.3389/FPLS.2021.599742/BIBTEX.
- Leng, L., Li, J., Wen, Z., Zhou, W., 2018. Use of microalgae to recycle nutrients in aqueous phase derived from hydrothermal liquefaction process. Bioresour. Technol. 256, 529–542. https://doi.org/10.1016/j.biortech.2018.01.121.
- Liakos, K.G., Busato, P., Moshou, D., Pearson, S., Bochtis, D., 2018. Machine learning in agriculture: a review. Sensors 18 (8), 2674. https://doi.org/10.3390/s18082674.
- Lu, Q., Zhou, W., Min, M., Ma, X., Chandra, C., Doan, Y.T.T., Ma, Y., Zheng, H., Cheng, S., Griffith, R., Chen, P., Chen, C., Urriola, P.E., Shurson, G.C., Gislerød, H.R., Ruan, R., 2015. Growing chlorella sp. on meat processing wastewater for nutrient removal and biomass production. Bioresour. Technol. 198, 189–197. https://doi.org/10.1016/j.biortech. 2015.08.133.
- Lu, Q., Li, Jun, Wang, J., Li, K., Li, Jingjing, Han, P., Chen, P., Zhou, W., 2017. Exploration of a mechanism for the production of highly unsaturated fatty acids in Scenedesmus sp. at low temperature grown on oil crop residue based medium. Bioresour. Technol. 244 (1), 542–551. https://doi.org/10.1016/j.biortech.2017.08.005.
- Lucas, J.S., Southgate, P.C., Tucker, C.S., 2019. Aquaculture: Farming Aquatic Animals and Plants. John Wiley & Sons Ltd, West Sussex.
- Meshram, V., Patil, K., Meshroam, V., Hanchate, D., Ramteke, S.D., 2021. Machine learning in the agriculture domain: a state-of-the-art survey. Artif.Intell.Life Sci. 1, 100010. https:// doi.org/10.1016/j.ailsci.2021.100010.
- Moylan, M., Ó Cinnéide, L., Whelan, K., 2017. Review of the Aquaculture Licensing Process. European Martime and Fisheries Fund, Dublin.
- Nachimuthu, L.A., Watkins, M.D., Hulugalle, N., FInlay, L.A., 2020. 101012. MethodsX 7. https://doi.org/10.1016/j.mex.2020.101012.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system. Aquaculture 526, 735377. https://doi.org/10.1016/j.aquaculture.2020.735377.
- Noble, R., Fuhrman, J., 1998. Use of SYBR green I for rapid epifluorescence counts of marine viruses and bacteria. Aquat. Microb. Ecol. 14 (2), 113–118. https://doi.org/10.3354/ ame014113.
- O'Neill, E.A., Rowan, N.J., 2022. Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance a comparative case study from the Republic of Ireland. Sci. Total Environ. 802, 149800. https://doi.org/10.1016/j.scitotenv.2021.149800.
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga Pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in

aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218. https://doi.org/10.1016/j.scitotenv.2019.07.243.

- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. Sci. Total Environ. 706, 136044. https://doi. org/10.1016/j.scitotenv.2019.136044.
- O'Neill, E.A., Morse, A.P., Rowan, N.J., 2022. Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: implications and opportunities for advancing research and disruptive innovation post COVID-19 era. Sci. Total Environ. 819, 153073. https://doi.org/10. 1016/j.scitotenv.2022.153073.
- Pal, M., Yesankar, P.J., Dwivedi, A., Qureshi, A., 2020. Biotic control of harmful algal blooms (HABs): a brief review. J. Environ. Manag. 268, 110687. https://doi.org/10.1016/j. jenvman.2020.110687.
- Patterson, G.M.L., Harris, D.O., 2007. The effect of Pandorina morum (Chlorophyta) toxin on the growth of selected algae, bacteria and higher plants. Br. Phycol. J. 18 (3), 259–266. https://doi.org/10.1080/00071618300650271.
- Rowan, N.J., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, implications, and outlooks. Int. J. Microbiol. 2011, 462832. https://doi.org/10.1155/2011/462832.
- Rowan, N.J., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo vadis? Trends Food Sci.Technol. 88, 316–332. https://doi. org/10.1016/j.tifs.2019.03.027.
- Rowan, N.J., 2022. The role of digital technologies in supporting and improving fisheries and aquaculture across the supply chain – quo vadis? Aquaculture and Fisheries https://doi. org/10.1016/j.aaf.2022.06.003
- Rowan, N.J., Casey, O., 2021. Empower Eco multiactor HUB: a triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new green deal opportunities under a United Nations Sustainable Development Goals framework. Curr. Opin. Environ. Sci. Health 21, 100254. https:// doi.org/10.1016/j.coesh.2021.100254.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: quo vadis? Sci. Total Environ. 748, 141362. https://doi.org/10.1016/j.scitotenv.2020. 141362.
- Rowan, N.J., Pogue, R., 2021. Editorial overview: green new deal era current challenges and emerging opportunities for developing sustaining and disruptive innovation. Curr. Opin. Environ. Sci. Health 22, 100294. https://doi.org/10.1016/j.coesh.2021.100294.
- Rowan, N.J., Murray, N., Qiao, Y., O'Neill, E., Clifford, E., Barcelo, D., Power, D.M., 2022. Digital transformation of peatland eco-innovations ("paludiculture"): enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2022.156328.
- Satyanarayana, K.G., Mariano, A.B., Vargas, J.V.C., 2011. A review of microalgae, a versatile source for sustainable energy and materials. Int. J. Energy Res. 35 (4), 291–311. https:// doi.org/10.1002/er.1695.
- Sharma, R., Kamble, S.S., Gunasekaran, A., Kumar, V., Kumar, A., 2020. A systematic literature review on machine learning applications for sustainable agriculture supply chain performance. Comput. Oper. Res. 119, 104926. https://doi.org/10.1016/j.cor.2020. 104926.
- Singh, U.B., Ahluwalia, A.S., 2012. Microalgae: a promising tool for carbon sequestration. Mitig. Adapt. Strateg. Glob. Chang. 18, 73–95. https://doi.org/10.1007/s11027-012-9393-3.
- Sirakov, I., Velichkova, K., Stoyanova, S., Staykov, Y., 2015. The importance of micro algae for aquaculture industry - review. Int. J. Fish. Aquat. Stud. 2, 81–84.
- Tacon, A.G.J., 2020. Trends in global aquaculture and aquafeed production: 2000–2017. Rev. Fish. Sci. Aquac. 28 (1), 43–56. https://doi.org/10.1080/23308249.2019.1649634.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638. https://doi.org/10.1016/j.scitotenv. 2017.05.227.
- Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E., Rowan, N., 2018a. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3, 28. https://doi.org/10. 3390/fishes3030028.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018b. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ. 6, e5281. https://doi.org/10.7717/peerj.5281.
- Tahar, A., Tiedeken, E.J., Rowan, N.J., 2018c. Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – a first overview of the situation in the Republic of Ireland. Sci. Total Environ. 616–617, 187–197. https://doi.org/10.1016/j.scitotenv.2017.11.021.
- Tan, J.S., Lee, S.Y., Chew, K.W., Lam, M.K., Lim, J.W., Ho, S.H., Show, P.L., 2020. A review on microalgae cultivation and harvesting, and their biomass extraction processing using ionic liquids. Bioengineered 11 (1), 116–129. https://doi.org/10.1080/21655979. 2020.1711626.
- Terrain-AI, 2022. Uncovering new insights to support effective climate change decision making [WWW Document]. URLTerrain-AI (accessed 13th July, 2022) https://terrainai.com/.
- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – a 20 year systematic review. Sci. Total Environ. 574, 1140–1163. https://doi.org/10.1016/j.scitotenv.2016.09.084.
- Toner, E., 2018. Ireland slashes peat power to lower emissions. Science 362 (6420), 1222–1223. https://doi.org/10.1126/science.362.6420.1222.
- Tschirner, M., Kloas, W., 2017. Increasing the sustainability of aquaculture systems insects as alternative protein source for fish diets. GAIA - Ecol. Perspect. Sci. Soc. 26 (4), 332–340. https://doi.org/10.14512/gaia.26.4.10.

#### E.A. O'Neill et al.

- Voulvoulis, N., Arpon, K.D., Giakoumis, T., 2017. The EU water framework directive: from great expectations to problems with implementation. Sci. Total Environ. 575, 358–366. https://doi.org/10.1016/j.scitotenv.2016.09.228.
- Wang, H.M.D., Chen, C.C., Huynh, P., Chang, J.S., 2015a. Exploring the potential of using algae in cosmetics. Bioresour. Technol. 184, 355–362. https://doi.org/10.1016/j. biortech.2014.12.001.
- Wang, J., Zhou, W., Yang, H., Wang, F., Ruan, R., 2015b. Trophic mode conversion and nitrogen deprivation of microalgae for high ammonium removal from synthetic wastewater. Bioresour. Technol. 196, 668–676. https://doi.org/10.1016/j.biortech.2015.08.036.
- Ward, J., McEneaney, E., O'Dwyer, A., 2019. Boglands [WWW Document]. URLAsk About Irel. http://www.askaboutireland.ie/learning-zone/primary-students/looking-at-places/ offaly/aspects-of-offaly/boglands/.
- WFD Ireland, 2018. The EU Water Framework Directive (WFD) [WWW Document]. URLIntrod. to Dir (accessed 7.24.18) http://www.wfdireland.ie/wfd.html.

- WFD Ireland, 2018. The EU Water Framework Directive (WFD) [WWW Document]. URLIntrod. to Dir (accessed 11.11.19) http://www.wfdireland.ie/wfd.html.
- Yarnold, J., Karan, H., Oey, M., Hankamer, B., 2019. Microalgal aquafeeds as part of a circular bioeconomy. Trends Plant Sci. 24, 959–970. https://doi.org/10.1016/J.TPLANTS.2019. 06.005.
- Zarauz, L., Irigoien, X., 2008. Effects of Lugol's fixation on the size structure of natural nanomicroplankton samples, analyzed by means of an automatic counting method. J. Plankton Res. 30, 1297–1303. https://doi.org/10.1093/PLANKT/FBN084.
- Zhou, Q., Li, K., Jun, X., Bo, L., 2009. Role and functions of beneficial microorganisms in sustainable aquaculture. Bioresour. Technol. 100, 3780–3786. https://doi.org/10.1016/J. BIORTECH.2008.12.037.

Contents lists available at ScienceDirect





# Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

# Potential disruptive effects of zoosporic parasites on peatland-based organic freshwater aquaculture: Case study from the Republic of Ireland



#### Emer A. O'Neill \*, Neil J. Rowan

Centre for Sustainable Disinfection and Sustainability, Bioscience Research Institute, Technological University of the Shannon: Midlands Midwest, Athlone Campus, University Road, Athlone, Co. Westmeath, Ireland

Faculty of Science & Health, Technological University of the Shannon: Midlands Midwest, Athlone Campus, University Road, Athlone, Co. Westmeath, Ireland

#### HIGHLIGHTS

#### G R A P H I C A L A B S T R A C T

- Novel aquaculture system offers sustainable fish production.
- System depends on microalgae for waste bioremediation and water quality.
- Algal presence is important for organic processes.
- Zoosporic parasites may cause collapse of algae population.
- Climate change may contribute to zoosporic parasite issues.

#### ARTICLE INFO

Editor: Damià Barceló

Keywords: Microalgae Cyanobacteria Next generation sequencing Recirculating aquaculture multi-trophic pond system Climate Sustainability



#### ABSTRACT

Irish freshwater aquaculture holds great potential for aiding food security. However, its necessary expansion has been hampered by the adoption of important environmental EU directives. A novel peatland-based recirculating aquaculture multi-trophic pond system (RAMPS) was developed to assess its potential to assist in the sustainable development of industry whilst remaining aligned with environmental protection by adhering to organic aquaculture practices. Microalgae play a pivotal role in the farms' wastewater bioremediation. However, a collapse of the algal population within the system towards the end of the pilot study was observed. No relationship between physicochemical fluctuations and the collapse were indicated. Further investigations into the potential presence of biological agents were then conducted and fourteen species of zoosporic parasites from five different genera (Labyrinthula, Vampyrella, Amoeboaphelidium, Paraphelidium and Aphelidium) were identified after conducting next-generation sequencing (MinION). The presence of these species indicated the potential cause of algal collapse. Additionally, changes in weather conditions may have also contributed to the issue. Given the lack of data available on zoosporic parasites and their potential impact on organic aquaculture practices, additional research needs to be conducted. Developing a means to monitor and mitigate against these complex zoosporic parasites will inform food security, it will particularly help safeguard "organic" freshwater aquaculture where there is a reliance on using natural-based approaches to address disease mitigation. This information will in turn inform the replication of this RAMPs system in peatlands internationally creating local employment in green technologies, as communities' transition away from burning peat as fossil fuel. Also, zoosporic parasites may reduce important microalgae in peatland-based culture ponds that serve as exceptional sequesters of carbon. Findings of this study will inform related research that focus on the emergence of microbial pathogens in local aquatic ecosystems brought on by variances in climate.

\* Corresponding author. *E-mail address:* emer.oneill@ait.ie (E.A. O'Neill).

http://dx.doi.org/10.1016/j.scitotenv.2023.161495

Received 21 October 2022; Received in revised form 5 January 2023; Accepted 5 January 2023 Available online 10 January 2023

0048-9697/© 2023 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### 1. Introduction

Aquaculture has become the fastest growing food producing industry in the world (O'Neill et al., 2019, 2020; Houston et al., 2020) as a result of capture fisheries reaching the maximum sustainable yields, and in some instances surpassing it (Tsikliras and Froese, 2019; O'Neill and Rowan, 2022). It provides an important source for food security, both directly as a means of meeting the ever growing demand for food from an expanding global population and indirectly by providing a means of employment opportunities (Golden et al., 2021; Garlock et al., 2022). Fish produced *via* aquaculture is now available to countries that would have previously been limited or non-existent, often at cheaper prices, thus providing improved nutrition and food security (Rowan and Casey, 2021; Galanakis et al., 2021; O'Neill et al., 2022a, 2022b). However, growth of the industry has begun to slow down (Edwards et al., 2019; Mirto et al., 2022).

In Ireland, issues with licensing, environmental concerns and limited space / resources have hampered the growth of the industry (O'Neill et al., 2020, 2022b). To respond to these challenges, Bord Iascaigh Mhara (BIM), Ireland's state agency for developing the national marine fishing and aquaculture industries, developed Ireland's first trial peatland based, recirculating aquaculture multi-trophic pond systems (RAMPS) adhering to 'organic' principles, as peatlands now being listed as important habitats under the EU's Birds [79/409/EEC & 2009/147/EC] and Habitats [92/43/EC] directives. The peatland RAMPS system holds European perch (Perca fluviatilis), rainbow trout (Oncorhynchus mykiis), common duckweed (Lemna minor) and gibbous duckweed (Lemna gibba) and exploits the use of microalgae for waste removal (Bord na Móna, 2019; O'Neill et al., 2022c; Rowan et al., 2022). The process differs from traditional aquaculture practices that use water from rivers and lakes where traditional systems must consider potential pollutants from agriculture, industry, wastewater treatments plants, etc. (Rowan, 2011; Tahar et al., 2017; Tiedeken et al., 2017; Tahar et al., 2018a, 2018b, 2018c).

The potential beneficial roles of microalgae in aquaculture has only recently been confirmed (Gao et al., 2016; Ansari et al., 2017; Han et al., 2019; Nagarajan et al., 2021). For example, microalgae could efficiently assimilate nutrients providing a good method for wastewater remediation (Wang et al., 2015; Leng et al., 2018; Han et al., 2019) in aquaculture, having already demonstrated promising performances in the food and agriculture industries, and in municipal wastewater treatment (DeBashan et al., 2004; Lu et al., 2015, 2017; Han et al., 2019; Nagarajan et al., 2021; O'Neill et al., 2022b). As a result of this and other emerging benefits, the use of microalgae in aquaculture has recently emerged into the forefront. However, research into the role of microalgae in aquaculture is still lacking (Han et al., 2019; O'Neill et al., 2022a).

The development of microalgae derived products has greatly increased in recent years however, sustainable production faces biological challenges, such as the presence of zoosporic parasites (Höger et al., 2021). Zoosporic parasites are facultative parasites that produce motile spores as their infective propagules causing frequent epidemics in aquatic ecosystems (Scholz et al., 2016) as many of these parasites are known to infect vertebrates, invertebrates, vascular plants, macroalgae, phytoplankton, fungi and protists (Gleason et al., 2014a). Additionally, it is very difficult to differentiate between divergent species of zoosporic parasites as many exhibit very similar morphological characteristics. As such, it is therefore possible that the profile of zoosporic parasites present in aquatic habitats may be undescribed. Given these adversities, it is also possible that a general lack of research in this area may result in infection incidences within the susceptible algal populations of these aquatic habitats being underreported (Scholz et al., 2016). They are believed to be major drivers of phytoplankton succession, and as a consequence, infections can alter the composition of algal species in aquatic ecosystems (Scholz et al., 2016). Zoosporic parasites have received increased attention in recent years; however, research is still greatly lacking (Gleason et al., 2014b; Scholz et al., 2016).

The hypothesis behind this research evolved around the fact that dramatic losses in the microalgae populations within the peatland based trial fish farm could not be explained through use of traditional monitoring techniques and approaches. This intimated that there was other unexplained agent(s) potentially present in the RAMS system causing the observed disruptive issues. Therefore, the aim of this research was to determine whether issues associated with the loss of important algal and cyanobacterial populations within this unique peatland-based aquaculture farm may have been due to the presence and deleterious activities of zoosporic parasites and if so, what potential effects could their presence induce on this novel primary food production system and its connected aquatic environment.

#### 2. Materials & methods

#### 2.1. Study site

Oasis fish farm is an innovative trial peatland based recirculating aquaculture multi-trophic pond system (RAMPS) process set in the middle of Mount Lucas Wind Farm, Co. Offaly (53°17'3" N - 7°11'45" W). The RAMPS system combines elements from a recirculating aquaculture system (RAS) and an integrated multi-trophic aquaculture (IMTA) system). The RAMPS consists of four split (pill) ponds connected to a microalgae and duckweed lagoon with sixteen channels serving as an organic treatment system. See Fig. 1. Fish are kept at a density that does not exceed the organic farming standard (e.g.,  $<20 \text{ kg/m}^{-3}$  for perch), using screens at the D-ends of each split pond. The space between two D-end fish culture areas is also used to treat waste with free living microalgae in suspension. Flow in each split pond is generated and water is circulated using an airlift. Each D-end fish culture area is equipped with oxygen and temperature probes connected to paddlewheels to provide extra oxygen when necessary. Flow in the lagoon is generated with paddle wheels. The farm is designed to hold a maximum of 32,000 Kg of fish. European perch (Perca fluviatilis) is cultured in one pond and rainbow trout (Oncorhynchus mykiis) is cultured in the other three. No effluent is released from the farm unless excessively high levels of rainfall is experienced (O'Neill et al., 2022b; Rowan et al., 2022).

#### 2.2. Sample collection and preservation

Collection and preservation of samples were conducted as per O'Neill et al. (2022a, 2022b) with some modifications (samples were collected on different days to the previous studies conducted in parallel). Five litres (grab) water samples were collected from the trial RAMPS fish farm in octagonal carboy HDPE bottles (Lennox) and transported directly to the lab *via* car, 62 km away, in insulated boxes in order to maintain as close to *in-situ* conditions as possible. Samples were collected once a week between July 2020 and October 2020. Samples were taken from each of the culture ponds and, the entry and exit points of the treatment lagoon. See Fig. 1 for an overview of the farm sampling locations. One hundred mL of each sample was stored at -20 °C until molecular analysis was conducted. Physicochemical analysis was conducted immediately upon arrival from the fish farm.

#### 2.3. Microalgae and cyanobacteria monitoring

The microalgae and cyanobacteria levels were monitored on the farm in real time using the AlgaeTorch® (ISO 10260:1992; DIN 38412-16). The hand help device measurement is based on fluorescence which is proportional to the chlorophyll present in the microalgae and cyanobacteria (Naughton et al., 2020). Measurements were recorded in the same locations where the environmental samples were taken (Fig. 1). The device was placed into the water for approximately 15 s and the readings indicated on the device's display were recorded.

#### 2.4. Physicochemical monitoring

Physicochemical analysis was conducted as per O'Neill et al. (2022a, 2022b) with some modifications. Temperature, pH, ammonium (NH<sup>4</sup><sub>4</sub>),



**Fig. 1.** A) Location of the three meteorological weather stations (yellow box); Gurteen  $-53^{\circ}02'24''$  N,  $08^{\circ}00'36''$  W; Oak Park $-52^{\circ}51'36''$  N,  $06^{\circ}55'36''$  W; Mullingar  $-53^{\circ}33'$ 36'' N,  $07^{\circ}20'24''$  W, (red •) surrounding the trial fish farm (orange **O**). Meta data from these three stations were used to analyse any potential impacts changing weather conditions may have had on the system. B) Ariel view of the trial fish farm (Oasis) at Mount Lucas. Culture ponds (orange box), algae and duckweed lagoon (green box), reservoir (white  $\diamond$ ) and water intake location (blue  $\Delta$ ) displayed. Blue arrows indicate location of sampling points. Wind turbine which generates all electricity for the farm and the bog river which supplies water for the farm are also included. Figures adapted from O'Neill et al., 2022a. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

nitrite (NO<sub>2</sub><sup>-</sup>), nitrate (NO<sub>3</sub><sup>-</sup>), orthophosphate (PO<sub>4</sub><sup>3-</sup>), dissolved oxygen (DO), biochemical oxygen demand (BOD), total suspended solids (TSS) and total dissolved solids (TDS) were investigated in the laboratory within 24 h of collection to prevent the need for preservation. Temperature, pH and TDS were analysed using a benchtop pH and conductivity meter (VWR pHenomenal<sup>TM</sup> MU 6100 L meter, VWR 111662–1157 pH probe, VWR CO11 conductivity probe). DO and BOD<sub>5day</sub> were analysed using a benchtop DO<sub>2</sub> meter and probe (Jenway). Photometric kits were used to assess the NH<sub>4</sub><sup>+</sup>, NO<sub>2</sub><sup>-</sup>, NO<sub>3</sub><sup>-</sup> and PO<sub>4</sub><sup>3-</sup> (Merck), as per the manufacturer's

(Spectroquant®) instructions. The TSS were analysed via filtration using a Buchner flask, Buchner funnel and 0.20  $\mu m$  pore membrane filter (Whatman).

#### 2.5. Molecular analysis

Isolation of DNA, 18S sequencing and bioinformatics were conducted as per O'Neill et al. (2022b) with some modifications. The frozen water samples were first thawed, then centrifuged and the resulting pellet re-suspended in 2 mL of dH<sub>2</sub>O, before sonication at 40 kHz in a sonicator bath (Cuyson) at 45 °C for 20 min. DNA was isolated using the Roche High Pure PCR Template Preparation kit (Merck). Between 100 and 150 ng of DNA was used in a PCR reaction for the amplification of an 1800 base pair segment of the 18S ribosomal DNA sequence that was modified with adapter sequences (O'Neill et al., 2022b) for subsequent sequencing on the Oxford Nanopore MinION platform (adaptor sequences in red and underlined):

(F): 5'- <u>TTTCTGTTGGTGCTGATATTGC</u>GGTGATCCTGCCAGTAGTCAT ATGCTTG -3'.

#### (R): 3′- <u>ACTTGCCTGTCGCTCTATCTTC</u>GATCCTTCCGCAGGTTCACCT ACGGAAACC -5′.

Sequencing was conducted using the Ligation Sequencing Kit SQK-LSK114 (Oxford Nanopore Technologies) protocol, as per the manufacturer's instructions. The quality of the sequencing was assessed using the FastQC tool (v0.11.9). The adapters were removed using the Porechop adaptor trimmer tool (v0.2.4). The FastQ tool (v0.14.0) was then used to eliminate low quality reads, followed by quality assessment once again to confirm its success. The resulting high-quality reads were then grouped together using the isONclust software (v0.0.4). Each group, representing all reads that came from an individual gene, was then aligned against the National Center for Biotechnology Information (NCBI) nucleotide database using the BLASTn tool (v2.13.0) with an *E*-value of 0.05.

#### 2.6. Statistical analysis

Statistical analysis were performed on GRAPHPAD PRISM 9.3. The data generated were grouped and subject to normality tests (Anderson-Darling), to determine if samples were from a normal distribution (P > 0.05 = normal distribution). Parametric testing was then applied as a normal distribution was indicated. Grubbs test was used to determine if any outliers were indicated. *t*-tests and ANOVA (one-way with Tukey and two-way with Sidak) were used to determine if any significant differences were observed between the variables (P < 0.05 = significant difference). Unpaired tests were used as different sets of samples were analysed. Pearson's coefficient was applied to determine whether any relationship correlations (r) were indicated.

#### 3. Results

In order to obtain a more comprehensive analysis of the water quality conditions, ideally a composite sampler would be used to collect approximately 100 mL of water every hour for 24 h. However, due to technical difficulties and limitations in obtaining composite samplers, grab samples were collected. A larger volume (5 L) was collected as slowly as possible in order to aid in compensating for the lack of access to composite samples. This should be noted when interpreting these results.

#### 3.1. Microalgae and cyanobacteria

Microalgae and cyanobacteria readings across the entire system indicated no statistically significant difference (microalgae P = 0.8976, cyanobacteria P = 0.9380). Therefore, results were grouped together for ease of reporting. Microalgae levels fluctuated between 170 µg L<sup>-1</sup> and 280 µg L<sup>-1</sup> between December 2019 and August 2020, as shown in Fig. 2. However, levels dropped dramatically between August 2020 and September 2020 (Fig. 2), falling to 31.65 µg L<sup>-1</sup> before rising slightly to 95.83 µg L<sup>-1</sup> the following month (October 2020). Cyanobacteria levels remained consistently between 20 µg L<sup>-1</sup> and 40 µg L<sup>-1</sup> from January 2020 to April 2020 before spiking in May to 140 µg L<sup>-1</sup> (Fig. 2), as would be expected given the time of year. Levels dropped back to 37.45 by July 2020, as shown in Fig. 2. However, similarly with the microalgae, cyanobacteria levels dropped to 1.45 µg L<sup>-1</sup> in October 2020 (Fig. 2). Similar but not identical trends were observed in a separate study conducted in parallel



**Fig. 2.** Concentrations of algae (dark grey) and cyanobacteria (light grey) in  $\mu$ g L<sup>-1</sup> detected on the trial fish farm between December 2019 and October 2020 using the AlgaeTorch®. Red box indicates when issues were observed within the system resulting in the loss of both algal and cyanobacterial levels. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

with this research at the same site where direct algal and cyanobacterial counts were established using flow cytometry (O'Neill et al., 2022a).

#### 3.2. Weather conditions

Research conducted by O'Neill et al. (2022a) at the same location indicated changing weather conditions, most likely as a result of climate change, played a contributing factor in issues associated with fish mortalities. Therefore, Met Éireann (The Irish Meteorological Service) meta data was subsequently investigated during the time period where issues with the loss of algal and cyanobacteria populations were experienced to determine whether any relationships existed. Meta data from the three closest Met Éireann weather stations surrounding the fish farm (Mullingar, Co. Westmeath 53°33′36″ N, 07°20′24″ W; Oak Park, Co. Carlow 52°51′36″ N, 06°55′36″ W; Gurteen, Co. Tipperary 53°02′24″ N, 08°00′36″ W) were investigated. See Fig. 1 for the location of all weather stations with respect to the trial fish farm. The average monthly rainfall and temperature between July 2020 and October 2020 were included, as well as rainfall and temperature long term averages (LTA) for those months (Met Éireann, 2022), as shown in Fig. 3.

The average rainfall for the time period remained between 104 mm and 111 mm with the exception of the month of September 2020 where an average of only 59.33 mm was recorded. When data from the 2020 averages were compared to the LTA, rainfall was found to increase for the months of July, August and October but decrease for the month of September (Fig. 3). However, this was not statistically significant (P = 0.3124). The average temperature for the time period ranged between 15.43 °C and 9.57 °C. When the 2020 average data was compared to the LTA, similar trends were observed (P = 0.9830), as shown in Fig. 3. A correlation study then was conducted to determine whether the issues with algal and cyanobacterial populations (Fig. 2) were as a result of changing temperature or rainfall levels. Although moderate to strong relationships were observed for the microalgae and cyanobacteria with regards to both the rainfall and temperature data, none were statistically significant, as shown in Fig. 4 (microalgae v rainfall r = 0.647, P = 0.3527; microalgae v temperature r = 0.534, P = 0.4664; cyanobacteria v rainfall r = 0.583, P = 0.4165; cyanobacteria v temperature r = 0.517, P = 0.4827).

#### 3.3. Physicochemical analysis

The physicochemical results were compared to water quality parameters set out by the Irish Environmental Protection Agency (EPA) and the Irish Statutory Office [SI 272/2009 and SI 77/2019] for guidance



Fig. 3. A) Average rainfall (mm) and B) average temperature (°C) for 2020 (bar chart) and the long term average or LTA (line chart) records across the three meteorological weather stations (Gurteen; Oak Park; Mullingar) surrounding the Mount Lucas trial fish farm (Oasis) during the period (July 2020 – October 2020) where issues were observed within the system. SD indicated.

(EPA, 2001; Irish Statutory Office, 2009, 2019) as no physicochemical values specifically for Irish aquaculture were available (O'Neill et al., 2022a), as shown in Table 1. No issues were observed after physicochemical analysis, with the exception of  $NO_2^-$ ,  $PO_4^{3-}$ , DO and TSS. Both  $NO_2^-$  and  $PO_4^{3-}$  were consistently above the guidance value of  $0.03 \text{ mg L}^{-1}$  which may be due to the reducing levels of microalgae and cyanobacteria present in the system and the nutrients were not therefore being consumed by the organisms. The BOD levels were consistently above the guidance value of 6 mg  $O_2 L^{-1}$ . This was an on-going problem and mitigation techniques were applied (additional aeration and filtration) to aid in reducing this issue. The TSS levels were also consistently above the guidance value of 25 mg L<sup>-1</sup>. However, once additional filtration was applied TSS dropped below the guidance value. Fluctuations in oxygen were continually observed however, paddle wheels and airlifts were present in the farm to ensure oxygen levels remained above the guidance values. Oxygen levels may have also dropped during the transfer of environmental samples from the farm to the lab. Results were consistent with physicochemical analysis observed in studies conducted in parallel with this research at the same site during 2020. A correlation study was then conducted to determine whether fluctuations in the physicochemical parameters may have contributed to the loss of microalgae and cyanobacteria populations. No statistically significant correlations were observed between any of the physicochemical parameters and the microalgae or cyanobacteria results, as shown in Fig. 4.

#### 3.4. Zoosporic parasites

Samples for each month and for each location were pooled together. After analysis of the MinION data, fourteen species from five genera (*Labyrinthula, Vampyrella, Amoeboaphelidium, Paraphelidium* and *Aphelidium*) of zoosporic parasites were identified across all of the sampling locations within the system (Fig. 5). All fourteen species were identified in every month (July to October).

#### 4. Discussion

The novel RAMPS, which follows organic aquaculture principles, was initially designed to aid in alleviating issues surrounding the development of the Irish freshwater aquaculture industry as a result of the adoption of important environmental EU directives that have delayed the licensing process, as well as highlighted limitations in space and resources for the industry. Part of the organic process is the use of microalgae for wastewater bioremediation making it an important factor within the farm. This is additionally important as peatlands are now protected habitats given their increasing scarcity. Therefore, the unforeseen and unexpected loss of microalgal populations within the system, particularly around the month of September 2020 was highly problematic. Physicochemical results suggested that the fluctuations in the parameters were unlikely the cause of



**Fig. 4.** Correlation matrix used to determine whether any relationships existed between physicochemical analysis, algal levels and cyanobacterial levels observed in the trial fish farm (Oasis) and, the average rainfall and the average temperatures recorded in the surrounding meteorological weather stations between July 2020 and October 2020. Positive correlations indicated by blue, inverse correlations indicated by green. Red box indicates statistically significant results where P < 0.05. T = temperature, NH<sub>4</sub><sup>+</sup> = ammonium, NO<sub>2</sub><sup>-</sup> = nitrite, NO<sub>3</sub><sup>-</sup> = nitrate, PO<sub>4</sub><sup>3-</sup> = orthophosphate, DO = dissolved oxygen, BOD = biochemical oxygen demand, TSS = total suspended solids, TDS = total dissolved solids, A = algae, C = cyanobacteria, MER = Met Eireann Rainfall Data, MET = Met Eireann Temperature Data. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)
#### Table 1

Breakdown of the physicochemical analysis conducted on the trial fish farm (Oasis), during the period between July 2020 and October 2020, where issues within the system were observed. Values from the Irish Environmental Protection Agency and the Irish Statutory Office with regards to water quality were included for guidance as no physicochemical values specifically for Irish aquaculture were available. \* indicates when results were beyond the guidance values. S.D. indicated, n = 6.

	July	August	September	October	Guidance Values
pH Temperature (°C)	$8.11 \pm 0.08$	$7.03 \pm 0.01$	$7.48 \pm 0.06$	$7.70 \pm 0.04$	>6 - < 9
	$18.00 \pm 0.46$	17.46 ± 0.39	$13.42 \pm 0.16$	$12.16 \pm 0.10$	<25
Ammonium (mg $NH_4^+$ L <sup>-1</sup> )	$0.01 \pm 0.01$	$0.12 \pm 0.03$	$0.10 \pm 0.01$	0.33 ± 0.22	<1
Nitrite $(mg NO_2^- L^{-1})$	$0.05 \pm 0.01^*$	$0.13 \pm 0.01*$	$0.10 \pm 0.01^*$	$0.08 \pm 0.01^*$	<0.03
Nitrate (mg $NO_3^- L^{-1}$ )	$1.39 \pm 0.12$	$2.72 \pm 0.34$	4.88 ± 0.30	2.60 ± 0.77	<50
Orthophosphate $(mg PO_4^{3-} L^{-1})$	$1.11 \pm 0.07*$	$1.81 \pm 0.06^*$	$1.72 \pm 0.07^*$	$1.68 \pm 0.11^*$	<0.03
Dissolved Oxygen $(mg O_2 L^{-1})$	$5.13 \pm 0.20*$	$7.07 \pm 0.19$	5.59 ± 0.17*	6.49 ± 0.11*	>7 (C) >9 (S)
BOD $(mg O_2 L^{-1})$	$21.98 \pm 1.50^*$	$28.87 \pm 2.32^*$	22.42 ± 2.26*	26.05 ± 1.94*	<6 (C) <3 (S)
Suspended Solids $(mg L^{-1})$	72.33 ± 7.53*	$40.00 \pm 2.53^{*}$	23.68 ± 5.28	$14.00 \pm 5.51$	<25
Dissolved Solids $(mg L^{-1})$	$170.00 \pm 2.77$	169.00 ± 4.33	$172.00 \pm 4.51$	$171.00 \pm 2.47$	<300

C = Cyprinids, S = Salmonids, BOD = Biochemical Oxygen Demand.

the problem especially given the fact the fluctuations were consistently observed previously without the loss of microalgal (microalgae and cyanobacteria) numbers (O'Neill et al., 2022b). The changes in weather conditions may be potentially contributing to the issues, as moderately strong relationships were indicated, although not statistically significant. This in turn led the authors to consider that an unforeseen deleterious agent may have been the cause of the previously unexplained disruption and reduction in algal in this RAMPs system. Zoosporic parasites were subsequently investigated as they have been previously reported to cause sudden and considerable cellular death in microalgae in both natural and industrial settings (Höger et al., 2021). The presence of five known genera of zoosporic parasites genera (*Labyrinthula, Vampyrella, Amoeboaphelidium, Paraphelidium* and *Aphelidium*) was confirmed.

*Labyrinthula* are a fungal like aquatic protist found in a diverse range of habitats (both freshwater and marine) and are opportunistic zoosporic parasites known to infect diatoms (Popova et al., 2020), which were routinely observed in the system (O'Neill et al., 2022b). *Vampyrella*, which generally occur in freshwater systems, are naked amoebae that are known as 'vampire' amoebae. They puncture hole in the host's cell wall and 'suck up' the cytoplasm causing deterioration and collapse of the cell, most



Fig. 5. Phylogenetic tree indicating all zoosporic parasites identified in the trial fish farm (Oasis) between July 2020 and October 2020. See supplementary data for breakdown.

often targeting green microalgae such as Scenedesmus (Gong et al., 2015) which was one of the most common green microalgae in the system (O'Neill et al., 2022b). The three remaining genera identified (Aphelidium, Paraphelidium and Amoeboaphelidium) are all aphelids which are intracellular parasitoids of many groups of microalgae (Karpov et al., 2017). Aphelidium form cysts that penetrate the host cell and migrate the cysts contents into the cell causing deterioration and subsequent death (Karpov et al., 2020) and are commonly found to infect Scenedesmus (Gleason et al., 2014c). Parapheilidium penetrate and digest the cell wall by both mechanical and enzymatic means, targeting a range of different algal groups (Torruella et al., 2018). Amoeboaphelidium attaches itself to the host cell wall then encapsulates it and ultimately causes cell death by engulfing it (Höger et al., 2021). They have been found to infect Ankistrodesmus, Desmodesmus, Chlorella and Scenedesmus (Gleason et al., 2014c), all of which were routinely observed within the farm, with Chlorella being the most common and potentially most beneficial to the system (Ahmad et al., 2020; Arguelles, 2021; Arteaga Quico et al., 2021; O'Neill et al., 2022b). Although it cannot be inferred directly from the results, this suggested that the collapse of the microalgal populations may potentially be linked to the presence of the reported zoosporic parasites as they have the capacity to and are known to infect algal cells. Therefore, additional research is required from both an aquaculture and environmental point of view (as indicated in the future work). Not only would the presence of zoosporic parasites have a huge impact on organic aquaculture that strongly relies on microalgae, but they could also have knock on effects on surrounding aquatic ecosystems. Although this would unlikely be an issue with this system as water is very rarely released, should such parasites accidently be released from other aquaculture facilities that are directly connected to freshwater systems, the loss of microalgae (primary producers) within an aquatic ecosystem could instigate the collapse of the aquatic food chain. These findings also align with related research that considers a relations between climate variance and the emergence of infectious agents and vectors (El-Sayed and Kamel, 2020).

### 5. Conclusion

Research from this pilot study has indicated that the present of zoosporic parasites may, in part, have great adverse effects on the organic aquaculture system given their ability to demolish microalgal populations so quickly. Additionally, the changes in weather conditions, most likely due to climate change, may also be potentially contributing. The identification and real-time profiling of zoosporic parasites is significantly underappreciated for sustainable-food based systems given that this also has implications for green-industries and entrepreneurs who are focusing on developing large pond-based systems for the growth of microalgae as new biomass with a view to the bio-refining of bioactive-products for potential food, feed and health applications. Also, the microalgae in ponds serve as exceptional sequesters of carbon that could be disrupted. Ultimately, more focused research of the potential effects the presence of zoosporic parasites on organic aquaculture processes needs to be conducted, as well as an increased focus on their effects on the surrounding environments and the potential effects climate change may contribute. As such, during the next phase of the project, the following research will be conducted;

### 6. Future work

- Given that the collapse of the algal population was not consistently observed, research into the origin of the zoosporic parasites presence in the system needs determined in order to aid in the prevention of future occurrences.
- As molecular analysis could not be conducted on a week-to-week basis, additional research is needed to determine what time period throughout the year zoosporic parasites are most prevalent. This will include an extended monitoring period (12 months). Additionally, research needs to be conducted as to determine which species, and at what quantity (Real-Time PCR / qPCR), are occurring *i.e.*, are different species more prevalent throughout different periods of the year.

- Given that some species of microalgae are more beneficial than other, analysis on which species of microalgae are predominantly affected needs to be determined in order aid in preventing their entire collapse.
- Mitigation measures needs to be investigated in order to reduce and eliminate the zoosporic parasites presence without causing adverse effects on the microalgae, the fish or the surrounding system in general.
- Although the novel RAMPS is ultimately a closed system in so far as no discharge is released from the system unless excessive levels of rainfall is experienced causing the lagoon to overflow; given that the surrounding peatlands are a very important ecosystem, analysis into the potential effects zoosporic parasites on the surrounding peatland environment needs to be conducted in order to ensure all systems are protected.
- Finally, additionally research on any potential links with zoosporic emergence and changes in weather conditions needs to be considered, especially given the fact that climate change has increased the occurrence of more changing and erratic weather conditions.

### CRediT authorship contribution statement

All persons who meet authorship criteria are listed as authors, and all authors certify that they have participated sufficiently in the work to take public responsibility for the content, including participation in the concept, design, analysis, writing, or revision of the manuscript. Furthermore, each author certifies that this material or similar material has not been and will not be submitted to or published in any other publication before its appearance in the Science of the Total Environment.

CRediT roles:

Greenr releas	
Conceptualization	E. A. O'Neill, N. J. Rowan
Data Curation	E. A. O'Neill
Formal Analysis	E. A. O'Neill, N. J. Rowan
Funding Acquisition	N. J. Rowan
Investigation	E. A. O'Neill
Methodology	E. A. O'Neill
Project Administration	E. A. O'Neill
Resources	E. A. O'Neill, N. J. Rowan
Software	E. A. O'Neill
Supervision	N. J. Rowan
Validation	E. A. O'Neill
Visualization	E. A. O'Neill
Roles/Writing – Original Draft	E. A. O'Neill, N. J. Rowan
Writing – Review & Editing	E. A. O'Neill, N. J. Rowan

### Data availability

Data will be made available on request.

### Declaration of competing interest

The authors declare that there are no competing interests or conflicts of interest with respect to the publication of this article.

### Acknowledgements

The authors acknowledge provision of aerial image of Mount Lucas peatland site furnished by Bord Iascaigh Mhara (BIM). The authors would like to thank Interreg Atlantic Area Neptunus (Project EAPA\_576/2018), Regional University Network European University (RUN-EU Project), and Bord Iascaigh Mhara (Project 2019 BIM-KGS-008) for funding support. The authors would also like to thank Niamh Higgins for advice using the Nanopore MinION platform.

### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.scitotenv.2023.161495.

#### References

- Ahmad, M.T., Shariff, M., Md. Yusoff, F., Goh, Y.M., Banerjee, S., 2020. Applications of microalga Chlorella vulgaris in aquaculture. Rev. Aquac. 12 (1), 328–346. https://doi. org/10.1111/RAQ.12320.
- Ansari, F.A., Singh, P., Guldhe, A., Bux, F., 2017. Microalgal cultivation using aquaculture wastewater: integrated biomass generation and nutrient remediation. Algal Res. 21, 169–177. https://doi.org/10.1016/j.algal.2016.11.015.
- Arguelles, E.D.L.R., 2021. Biochemical composition and bioactive properties of Chlorella minutissima (Chm1) as a potential source of chemical compounds for nutritional feed supplement and disease control in aquaculture. Curr. Appl. Sci. Technol. 21 (1), 65–77.
- Arteaga Quico, C., Mariano Astocondor, M., Aquino Ortega, R., 2021. Dietary supplementation with Chlorella peruviana improve the growth and innate immune response of rainbow trout Oncorhynchus mykiss fingerlings. Aquaculture 533, 736117. https://doi.org/ 10.1016/J.AQUACULTURE.2020.736117.
- Bord na Móna, 2019. Aquaculture [WWW document]. URL https://www.bordnamona.ie/ company/our-businesses/new-business-and-land-development/aquaculture/. (Accessed 11 November 2019).
- DeBashan, L.E., Hernandez, J.P., Morey, T., Bashan, Y., 2004. Microalgae growth-promoting bacteria as "helpers" for microalgae: a novel approach for removing ammonium and phosphorus from municipal wastewater. Water Res. 38 (2), 466–474. https://doi.org/ 10.1016/j.watres.2003.09.022.
- Edwards, P., Zhang, W., Belton, B., Little, D., 2019. Misunderstandings, myths and mantras in aquaculture: its contribution to world food supplies has been systematically over reported. Mar. Policy 106, 103547. https://doi.org/10.1016/j.marpol.2019.103547.
- El-Sayed, A., Kamel, M., 2020. Climatic changes and their role in emergence and reemergence of diseases. Environ. Sci. Pollut. Res. 27, 22356–22532.
- EPA, 2001. Parameters of Water Quality Interpretation and Standards. Environmental Protection Agency, Johnstown Castle, pp. 1–133.
- Galanakis, C.M., Rizou, M., Aldawoud, T.M.S., Ucak, I., Rowan, N.J., 2021. Innovations and technology disruptions in the food sector within the COVID-19 pandemic and postlockdown era. Trends Food Sci. Technol. 110, 193–200. https://doi.org/10.1016/J. TIFS.2021.02.002.
- Gao, F., Li, C., Yang, Z.H., Zeng, G.M., Feng, L.J., Liu, J.Zhi, Liu, M., Cai, H.Wen, 2016. Continuous microalgae cultivation in aquaculture wastewater by a membrane photobioreactor for biomass production and nutrients removal. Ecol. Eng. 92, 55–61. https://doi.org/10.1016/j.ecoleng.2016.03.046.
- Garlock, T., Asche, F., Anderson, J., Ceballos-Concha, A., Love, D.C., Osmundsen, T.C., Pincinato, R.B.M., 2022. Aquaculture: the missing contributor in the food security agenda. Glob. Food Sec. 32, 100620. https://doi.org/10.1016/J.GFS.2022.100620.
- Gleason, F.H., Chambouvet, A., Sullivan, B.K., Lilje, O., Rowley, J.J.L., 2014a. Multiple zoosporic parasites pose a significant threat to amphibian populations. Fungal Ecol. 11, 181–192. https://doi.org/10.1016/J.FUNECO.2014.04.001.
- Gleason, F.H., Lilje, O., Marano, A.V., Sime-Ngando, T., Sullivan, B.K., Kirchmair, M., Neuhauser, S., 2014b. Ecological functions of zoosporic hyperparasites. Front. Microbiol. 5, 244. https://doi.org/10.3389/FMICB.2014.00244/BIBTEX.
- Gleason, F.H., Karpov, S.A., Lilje, O., Macarthur, D.J., van Otgen, F.F., Sime-Ngando, T., 2014c. Zoosporic parasites of phytoplankton. In: Jones, E.B.G., Hyde, K.D., Pang, K.L. (Eds.), Freshwater Fungi. DE GRUYTER, pp. 279–304 https://doi.org/10.1515/ 9783110333480.
- Golden, C.D., Koehn, J.Z., Shepon, A., Passarelli, S., Free, C.M., Viana, D.F., Matthey, H., Eurich, J.G., Gephart, J.A., Fluet-Chouinard, E., Nyboer, E.A., Lynch, A.J., Kjellevold, M., Bromage, S., Charlebois, P., Barange, M., Vannuccini, S., Cao, L., Kleisner, K.M., Rimm, E.B., Danaei, G., DeSisto, C., Kelahan, H., Fiorella, K.J., Little, D.C., Allison, E.H., Fanzo, J., Thilsted, S.H., 2021. Aquatic foods to nourish nations. Nature 598, 315–320. https://doi.org/10.1038/s41586-021-03917-1.
- Gong, Y., Patterson, D.J., Li, Y., Hu, Z., Sommerfeld, M., Chen, Y., Hu, Q., 2015. Vernalophrys algivore gen. nov., sp. nov. (Rhizaria: cercozoa: Vampyrellida), a new algal predator isolated from outdoor mass culture of Scenedesmus dimorphus. Appl. Environ. Microbiol. 81, 3900–3913. https://doi.org/10.1128/AEM.00160-15.
- Han, P., Lu, Q., Fan, L., Zhou, W., 2019. A review on the use of microalgae for sustainable aquaculture. Appl. Sci. 9 (11), 2377. https://doi.org/10.3390/app9112377.
- Höger, A.-L., Griehl, C., Noll, M., 2021. Infection with intracellular parasite amoeboaphelidium protococcarum induces shifts in associated bacterial communities in microalgae cultures. J. Appl. Phycol. 33, 2863–2873. https://doi.org/10.1007/s10811-021-02542-9.
- Houston, R., Bean, T., Macqueen, D., Gundappa, M., Jin, Y., Jenkins, T., Selly, S., Martin, S., Stevens, J., Santos, E., Davie, A., Robledo, D., 2020. Harnessing genomics to fast-track genetic improvement in aquaculture. Nat. Rev. Genet. 21 (7), 389–409. https://doi.org/10. 1038/s41576-020-0227-y.
- Irish Statutory Office, 2009. S.I. No. 272/2009 European Communities Environmental objectives (surface waters) regulations 2009 [WWW document]. URLElectron. Irish Statut. B. http://www.irishstatutebook.ie/eli/2009/si/272/made/en/print. (Accessed 5 April 2019).
- Irish Statutory Office, 2019. S.I. No. 77/2019 European Union Environmental objectives (surface waters) (amendment) regulations 2019 [WWW document]. 2019. URLElectron. Irish Statut. B. http://www.irishstatutebook.ie/eli/2019/si/77/made/en/pdf. (Accessed 14 January 2020).
- Karpov, S.A., Torruella, G., Moreira, D., Mamkaeva, M.A., López-García, P., 2017. Molecular phylogeny of paraphelidium letcheri sp. Nov. (Aphelida, Opisthosporidia). J. Eukaryot. Microbiol. 64, 573–578. https://doi.org/10.1111/jeu.12389.
- Karpov, S.A., Vishnyakov, A.E., López-García, P., Zorina, N.A., Ciobanu, M., Tcvetkova, V.S., Moreira, D., 2020. Morphology and molecular phylogeny of aphelidium insulamus sp. Nov. (Aphelida, Opisthosporidia). Protistology 14, 191–203. https://doi.org/10.21685/ 1680-0826-2020-14-4-3.

- Leng, L., Li, J., Wen, Z., Zhou, W., 2018. Use of microalgae to recycle nutrients in aqueous phase derived from hydrothermal liquefaction process. Bioresour. Technol. 256, 529–542. https://doi.org/10.1016/j.biortech.2018.01.121.
- Lu, Q., Zhou, W., Min, M., Ma, X., Chandra, C., Doan, Y.T.T., Ma, Y., Zheng, H., Cheng, S., Griffith, R., Chen, P., Chen, C., Urriola, P.E., Shurson, G.C., Gislerød, H.R., Ruan, R., 2015. Growing chlorella sp. On meat processing wastewater for nutrient removal and biomass production. Bioresour. Technol. 198, 189–197. https://doi.org/10.1016/j.biortech. 2015.08.133.
- Lu, Q., Li, Jun, Wang, J., Li, K., Li, Jingjing, Han, P., Chen, P., Zhou, W., 2017. Exploration of a mechanism for the production of highly unsaturated fatty acids in scenedesmus sp. at low temperature grown on oil crop residue based medium. Bioresour. Technol. 244 (1), 542–551. https://doi.org/10.1016/j.biortech.2017.08.005.
- Met Éireann, 2022. Monthly data [WWW document]. URLThe Irish Meteorological Service. https://www.met.ie/climate/available-data/monthly-data. (Accessed 25 May 2022).
- Mirto, S., Montalto, V., Mangano, M.C.M., Ape, F., Berlino, M., la Marca, C., Lucchese, M., Maricchiolo, G., Martinez, M., Rinaldi, A., Terzo, S.M.C., Celic, I., Galli, P., Sarà, G., 2022. The stakeholder's perception of socio-economic impacts generated by COVID-19 pandemic within the italian aquaculture systems. Aquaculture 553, 738127. https:// doi.org/10.1016/J.AQUACULTURE.2022.738127.
- Nagarajan, D., Varjani, S., Lee, D.J., Chang, J.S., 2021. Sustainable aquaculture and animal feed from microalgae – nutritive value and techno-functional components. Renew. Sust. Energ. Rev. 150, 111549. https://doi.org/10.1016/J.RSER.2021.111549.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system. Aquaculture 526, 735377. https:// doi.org/10.1016/j.aquaculture.2020.735377.
- O'Neill, E.A., Rowan, N.J., 2022. Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance a comparative case study from the Republic of Ireland. Sci. Total Environ. 802, 149800. https://doi.org/10.1016/j.scitotenv. 2021.149800.
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218. https://doi.org/10.1016/j.scitotenv.2019.07.243.
- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. Sci. Total Environ. 706, 136044. https://doi.org/ 10.1016/j.scitotenv.2019.136044.
- O'Neill, E.A., Morse, A.P., Rowan, N.J., 2022a. Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: implications and opportunities for advancing research and disruptive innovation post COVID-19 era. Sci. Total Environ. 819, 153073. https://doi.org/10.1016/ J.SCITOTENV.2022.153073.
- O'Neill, E.A., Fehrenbach, G., Murphy, E., Alencar, S.A., Pogue, R., Rowan, N.J., 2022b. Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: case study from the Republic of Ireland. Sci. Total Environ. 851, 158392. https://doi. org/10.1016/j.scitotenv.2022.158392.
- O'Neill, E.A., McKeon Bennett, M., Rowan, N.J., 2022c. Peatland-based innovation can potentially support and enable the sustainable development goals of the United Nations: case study from the Republic of Ireland. Case Stud. Chem. Environ. Eng. 6, 100251. https:// doi.org/10.1016/J.CSCEE.2022.100251.
- Popova, O.V., Belevich, T.A., Golyshev, S.A., Kireev, I.I., Aleoshin, V.V., 2020. Labyrinthula diatomea n. sp.—a labyrinthulid associated with marine diatoms. J. Eukaryot. Microbiol. 67, 393–402. https://doi.org/10.1111/jeu.12789.
- Rowan, N.J., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, implications, and outlooks. Int. J. Microbiol. 2011, 462832. https://doi.org/10.1155/2011/462832.
- Rowan, N.J., Casey, O., 2021. Empower eco multiactor HUB: a triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new green Deal opportunities under a United Nations sustainable development goals framework. Curr. Opin. Environ. Sci. Heal. 21, 100254. https://doi.org/ 10.1016/J.COESH.2021.100254.
- Rowan, N.J., Murray, N., Qiao, Y., O'Neill, E., Clifford, E., Barcelo, D., Power, D.M., 2022. Digital transformation of peatland eco-innovations ("paludiculture"): enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products. Sci. Total Environ. 838, 156328. https://doi.org/10.1016/j.scitotenv.2022.156328.
- Scholz, B., Guillou, L., Marano, A.V., Neuhauser, S., Sullivan, B.K., Karsten, U., Küpper, F.C., Gleason, F.H., 2016. Zoosporic parasites infecting marine diatoms-A black box that needs to be opened. Funct. Ecol. 19, 59–76. https://doi.org/10.1016/J.FUNECO.2015. 09.002.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an irish case study. Sci. Total Environ. 603–604, 627–638. https://doi.org/10.1016/j.scitotenv. 2017.05.227.
- Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E., Rowan, N., 2018a. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3 (3), 28. https://doi.org/ 10.3390/fishes3030028.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018b. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, e5281. https://doi.org/10.7717/peerj.5281.
- Tahar, A., Tiedeken, E.J., Rowan, N.J., 2018c. Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water

### E.A. O'Neill, N.J. Rowan

treatment plants – a first overview of the situation in the Republic of Ireland. Sci. Total Environ. 616–617, 187–197. https://doi.org/10.1016/j.scitotenv.2017.11.021.

- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – a 20 year systematic review. Sci. Total Environ. 574, 1140–1163. https://doi.org/10.1016/j.scitotenv.2016.09.084.
- Torruella, G., Grau-Bové, X., Moreira, D., Karpov, S.A., Burns, J.A., Sebé-Pedrós, A., Völcker, E., López-García, P., 2018. Global transcriptome analysis of the aphelid paraphelidium

tribonemae supports the phagotrophic origin of fungi. Commun. Biol. 1, 231. https://doi.org/10.1038/s42003-018-0235-z.

- Tsikliras, A., Froese, R., 2019. Maximum sustainable yields. In: Fath, B. (Ed.)Encyclopaedia of EcologyVol 1. Elsevier, Boston, pp. 108–115.
   Wang, J., Zhou, W., Yang, H., Wang, F., Ruan, R., 2015. Trophic mode conversion and nitro-
- Wang, J., Zhou, W., Yang, H., Wang, F., Ruan, R., 2015. Trophic mode conversion and nitrogen deprivation of microalgae for high ammonium removal from synthetic wastewater. Bioresour. Technol. 196, 668–676. https://doi.org/10.1016/j.biortech.2015.08.036.

Contents lists available at ScienceDirect

### Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

Review

### Digital transformation of peatland eco-innovations ('Paludiculture'): Enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products and services



Neil J. Rowan <sup>a,b,\*</sup>, Niall Murray <sup>c</sup>, Yuansong Qiao <sup>c</sup>, E. O'Neill <sup>a,b</sup>, Eoghan Clifford <sup>d</sup>, Damià Barceló <sup>e</sup>, Deborah M. Power <sup>a,f</sup>

<sup>a</sup> Bioscience Research Institute, Technological University of the Shannon Midlands Midwest (TUS), Dublin Road, Athlone, Ireland

<sup>b</sup> Empower Eco™ Sustainable Hub, Technological University of the Shannon Midlands Midwest (TUS), Dublin Road, Athlone, Ireland

<sup>c</sup> Software Research Institute, TUS, Athlone, Ireland

<sup>d</sup> School of Engineering and Informatics, NUI, Galway, Ireland

e Catalan Institute for Water Research, Faculty of Chemistry, University of Bacrelona, (ICRA), Spain

<sup>f</sup> Centro de Ciências do Mar (CCMAR), Universidade do Algarve, Campus de Gambelas 8005-139, Faro, Portugal

### HIGHLIGHTS

### GRAPHICAL ABSTRACT



 Harnessing digital transformation will enable peatland eco-innovations across the full value chain

- Mount Lucas is a paludiculture demonstrator for digital transformation
- Digital technologies developed for Agriculture 4.0 and Industry 5.0 will advance paludiculture
- Digitisation will future-proof paludiculture for climate change and sustainable eco- intensification

### ARTICLE INFO

Editor: Frederic Coulon

Keywords: Digital transformation Peatlands Climate change Disruption Sustainability innovation Natural-based solutions



### ABSTRACT

The world is heading in the wrong direction on carbon emissions where we are not on track to limit global warming to 1.5 °C; Ireland is among the countries where overall emissions have continued to rise. The development of wettable peatland products and services (termed 'Paludiculture') present significant opportunities for enabling a transition away from peat-harvesting (fossil fuels) to developing 'green' eco-innovations. However, this must be balanced with sustainable carbon sequestration and environmental protection. This complex transition from 'brown to green' must be met in real time by enabling digital technologies across the full value chain. This will potentially necessitate creation of new green-business models with the potential to support disruptive innovation. This timely paper describes digital transformation of paludiculture-based eco-innovation that will potentially lead to a paradigm shift towards using smart digital technologies to address efficiency of products and services along with future-proofing for climate change. Digital transform of paludiculture also aligns with the 'Industry 5.0 - a human-centric solution'. However, companies supporting peatland innovation may lack necessary standards, data-sharing or capabilities that can also affect viable business model propositions that can jeopardize economic, political and social sustainability. Digital solutions may reduce costs, increase productivity, improve produce develop, and achieve faster time to market for paludiculture. Digitisation also enables information systems to be open, interoperable, and user-friendly. This constitutes the first

\* Corresponding author at: Bioscience Research Institute, Technological University of the Shannon Midlands Midwest (TUS), Dublin Road, Athlone, Ireland. *E-mail address*: neil.rowan@tus.ie (N.J. Rowan).

Received 19 April 2022; Received in revised form 25 May 2022; Accepted 25 May 2022 Available online 29 May 2022 0048-9697/© 2022 The Authors. Published by Elsevier B.V. This is an open accessories under the CC BY license (http://creativecommons.org/licenses/by/4.0/). study to describe the digital transformation of paludiculture, both vertically and horizontally, in order to inform sustainability that includes process automation via AI, machine learning, IoT-Cloud informed sensors and robotics, virtual and augmented reality, and blockchain for cyber-physical systems. Thus, the aim of this paper is to describe the applicability of digital transformation to actualize the benefits and opportunities of paludiculture activities and enterprises in the Irish midlands with a global orientation.

### Contents

1.	Introduction						
2.	Development of aquaculture, aquaponics and adjacent green innovation in the Irish peatlands						
3.	Exploiting digitization to integrate and unlock vertical and horizontal value chains						
	3.1. Robotics						
	3.2. Drones and satellites						
	3.3. Sensors						
	3.4. Photonics						
	3.5. Artificial intelligence and machine learning						
	3.6. Immersive technologies, augmented reality (AR), virtual reality (VR)						
	3.7. Georeferencing and mapping						
	3.8. Edge-internet of things (IoT) systems						
4.	Pollination and ecosystem service management						
5.	Waste water recirculation – nexus between monitoring, treatment and energy using a paludiculture framework						
6.	Aquaculture and other aquatic systems						
7.	European Digital Innovation Hubs – Quo Vadis						
8.	Business modelling and sustainability for enabling accelerating Mount Lucas green innovation						
9.	Conclusions						
	9.1. Implications and recommendations						
CRe	CRediT authorship contribution statement         17						
Decl	daration of competing interest						
Ackı	nowledgments						
Refe	erences						

### 1. Introduction

The onset of COVID-19 pandemic and commensurate disruption to supply chains worldwide has created opportunities for developing new solutions to pressing societal challenges (Rowan and Laffey, 2020a; Rowan and Laffey, 2020b; Rowan and Galanakis, 2020). However, a panoply of factors impact the sustainability of green innovations, but could be ameliorated by enabling real time data management, usage and protection by engaging and implementing the vision of Industry 4.0 (a.k.a the fourth industrial revolution, digital transformation) in the agriculture and food sectors. The occurrence of COVID-19, an unpredictable and unexpected event that had severe worldwide consequences, coincided with the launch of the European 'Green New Deal', which was aimed at implementing solutions to combat climate change by supporting the sustainable intensification of eco-innovation, products and services (Rowan and Pogue, 2021). The European Just Transition initiative embraces and develops the vision of sharing knowledge openly to enhance societal inclusiveness, regenerate communities and ultimately promote a low carbon economy (Rowan and Casey, 2021). These ambitious programs and the attainment of their goals will be shaped and impacted by other mitigating factors including extreme weather events brought on by climate change (Weiskopt et al., 2020), the global energy crisis brought on by the Ukraine war and threat to the "breadbasket" of Europe and food production and security (http://www.dw.com). Despite the slowdown in carbon emissions caused by successive "lockdowns" worldwide, the current global models for business and production means the world continues to head in the wrong direction on carbon emissions in spite of the narrowing window of less than a decade to contain a catastrophic rise in global temperatures (Intergovermental Panel on Climate Change (IPCC), 2022). The world is not on track to limit warming to 1.5 °C and only a major transition in the energy sector including a substantial reduction in fossil fuel use, widespread electrification, improved energy efficiency and the use of alternative fuels can halt what is becoming an irreversible warming trend (O'Sullivan, 2022). Mondejar et al. (2021)

noted that changing climatic conditions are accomplished by high and low stress, altered rainfall patterns, elevated carbon dioxide, increased frequency of extreme weather events like flooding, droughts, cyclonic disturbances and increased saline soils. Okolo et al. (2019) reported that increased stress negatively influences agroecosystems' natural resilience; this is expected to cause dramatic environmental changes on a world scale that can affect supply chains for food, feed and clean energy. The reader is also directed to the published work of Mondejar et al. (2021) for applications of digital technologies that provide sustainable solutions to different agriculture problems.

Peatlands are a unique ecosystem that constitute an important carbon sink globally (Ziegler et al., 2021; O'Neill et al., 2022) and also represents an environment that may potentially provide a win-win scenario for new activities that could bring economic and social benefits while simultaneously contributing to reduce the harm to the climate (Rowan and Galanakis, 2020; Ziegler et al., 2021). The globalization of the current crisis (climate change, pandemics, population growth and food security) emphasise that solutions can only be found through global actions to co-create and implement solutions for these complex societal challenges and make the United Nations' Sustainability Development Goals set for 2030 a reality (Rowan and Casey, 2021).

Peatlands are a type of wetland that covers 3% of global land surface and currently approximately 15% of peatlands are degraded due to drainage for agriculture, forestry, and peat mining as a fossil fuel (Urák et al. 2017; Ziegler et al., 2021; O'Neill and Rowan, 2022), Ziegler et al. (2021) reported that the adverse consequences of intensive peatland degradation can contribute to inter alia undesirable greenhouse gas emissions, biodiversity loss and pollution of receiving waters. However, "rewilding" peatlands through controlled wetting can promote biodiversity and sustainable economies through the unique practice of paludiculture (production under permanently wet, peat-conserving and potentially peat-forming conditions, https://www.eurosite.org/). Indeed, peatlands are one of the world's most vital ecosystems, supporting a range of rare and stress-hardened plants and species, cleaning and filtering water, and mitigating flooding (Rowan and Casey, 2021). Peatlands cover just 3% of the planet, but store twice as much carbon as all the world's forests combined (Sargent, 2022). Ireland is rich in peatlands, which cover 20% of the land mass with near perfect conditions for peat soils and lock in 75% of all soil organic carbon (Rowan and Casey, 2021). Irish peatlands are estimated to actively capture around 57,000 t of carbon per year (Sargent, 2022) and constitute an important resource for carbon removal from the atmosphere and for enhancing sustainable carbon cycles (O'Neill et al., 2019).

Traditionally, use of wet peatlands (or 'paludiculture') is a potentially exciting means of alternative land use worldwide as this can lead to new sustainable employment opportunities for farmers and create opportunities for communities to fairly transition to a low carbon economy (Ziegler et al., 2021). However, promotion of paludiculture needs robust, evidence-based, collaborative research from pilot (pre-commercial) studies across a diversity of paludiculture activities to inform appropriate sustainable business models, such as for production of, fuel, fodder, food and construction materials (Wichmaan et al., 2016; Ziegler et al., 2021). Ziegler et al. (2021) and others (Tan et al., 2021; O'Neill et al., 2022) noted that support for the sustainable intensification of wet peatlands is a novel concept. O'Neill and co-workers (2019) reported for the first time the development of an innovative integrated multi-trophic aquaculture (IMTA) system for sustainable foods in peatlands of the Irish midlands. This freshwater IMTA system exploited a naturally occurring ecosystem of bacteria, algae and duckweed in ponds for managing waste and ensuring water quality and was powered by wind turbines. The described recirculating aquaculture process does not rely on complex end-of-circuit solutions for maintaining waste effluent treatments (Rowan and Galanakis, 2020; Galanakis et al., 2021).

Current freshwater aquaculture production systems are resourceconstrained and the production systems are unlikely to meet the increasing demand for aquatic foods, which is foreseen as one solution for feeding the growing world populations that is estimated to reach approximately 10 billion by 2050 (Rasmussen et al., 2018; Xia et al., 2022). Moreover, there is increasing anthropogenic-mediated destruction of natural resources caused by increasingly frequent extreme weather events that will affect the supply of safe, affordable, and nutritious food (Weiskopt et al., 2020). Aquatic ecosystems encompass about 75% of planet's surface, including that represented by the wet peatlands, which present new opportunities to intensify and diversify eco-innovation transnationally (Rowan and Casey, 2021; FAO, 2020). A knowledge gap that will have to be filled is the development and implementation of disruptive technologies in aquaculture to promote green, sustainable and profitable production models (Xia et al., 2022). Digitalization of aquaculture systems represents a new frontier and will be transformative by allowing development of precision farming solutions improve yields of food rich in high quality protein and omega-3 fatty acids (Mondejar et al., 2021; Rowan and Casey, 2021). The development and adaption of sophisticated innovations, along with open knowledge and technology exchange, is required to meet hurdles presented by lowenvironmental impact aquatic-based innovations (O'Neill et al., 2022).

There is increasing interest in adapting digital technologies such as artificial intelligence (AI) and machine learning, the Internet of Things (IoT), sensors, drones, blockchain for improving adjacent industries (Hrustek, 2020; Mondejar et al., 2021). Indeed, the disruptive approach of Industry 4.0 has started to be felt in some traditional agri-food activities and cyber-physical systems, IoT, AI and machine learning, big-data and analytics, and cloud technology have been integrated with agricultural machinery with exciting possible benefits (Tsolakis et al., 2019; Hrustek, 2020; Arvanitis and Symenonaki, 2020) (Table 1). Collectively, application of these next-generation digital technologies can support and accelerate Open knowledge exchange worldwide and can inform real time development of exciting new technologies to meet the challenges of today; although, the diversity and complexity of agri-food activities makes sector-customized solutions a challenge and a priority (Mondejar et al., 2021). Hrustek (2020) noted that digital transformation will help inform stability for the sustainable

development of the global economy including preventing or mitigating against the impact of uncertainties created by increased frequency of extreme weather events and new global crisis (financial, health, war, food and water). Hrustek (2020) also advocated that organisations must understand the key drivers of digital transformation that affect technology development and industry. Similarly, for development and future intensification of peatland innovation, a balance must be achieved between striding towards a high rate of economic progress and protecting the environment that enables delivering a climate friendly and sustainable production sector (Mondejar et al., 2021; Weizkopt et al., 2020).

The vision outlined by the authors in the present commentary is that digital transformations can inform technological, societal and political drivers of peatland, as it has for agricultural innovation. Moreover, 4IR should be deployed with the objective of enhancing the quality of life of the billions of people worldwide through creating a system that is secure and fair and establishing a framework that monitors and promotes sustainable practices (Shamin et al., 2019; Hrustek, 2020). Moving forward, peatland innovations should become an integral component of the value chain in the economy, providing products and services to society and informing food security and stability while promoting environmental, economic and social sustainability (Rowan and Galanakis, 2020). However, implementing innovative sustainable technologies typically requires far-reaching changes of the macro environment in which innovating companies operate where there is a need to strategically create an appropriate supportive external environment, such as collective system building (Planko et al., 2016; Zeigler, 2020) and quadruple helix innovation hub approach (Rowan and Casey, 2021). Commensurately, there will be a focus on achieving compliance with environmental and quality regulations as seen in agriculture sector (Hrustek, 2020). The aforementioned also aligns with the tenets of Industry 5.0 that provides a vision of the industry that looks beyond efficiency and productivity as the sole goals, but reinforces the role and the contribution of industry to society (Skobelev and Borovik, 2017a, 2017b; Nahavandi, 2019). It complements the fourth industrial revolution named Industry 4.0 by using digital transformation to place research and innovation at the service of the transition to a sustainable, human-centric and resilient industry. Industry 4.0 combines physical world of real things with their 'virtual twins' (Table 1).

Peatland innovation will also potentially mitigate against impact of reduced water supply and the environmental impact of intensive plant and livestock production to meet increased animal and human needs for high quality food production (Rowan and Casey, 2021; O'Neill et al., 2022), where the Food and Agriculture Organization (FAO, 2020) advocates adoption of digital technologies to enhance productivity and ensure food safety. Hrustek (2020) highlighted similarities to applying digital technologies to the agriculture sector where it is also envisaged that peatland innovation will require standardization and will be made available across the entire value chain where creative and adaptable approaches will be required to support sustainable development of paludiculture. It is likely that peatland innovation will also be advanced by applying smart and precise processes including process automation and robotics, peatland applications and information systems, cyber-physical systems, related tools and machines, and collection and evaluation of large amounts of data (Mondejar et al., 2021). Lessons learnt from sustainable practices in adjacent agriculture infers that sustainability across the peatlands will be centred on delivering flexibility in tandem with efficiency where future digital transformative activities will help unlock complex economic and societal challenges (Rowan and Pogue, 2021). However, the complexity and diversity of such data is likely to be vast and will require systematic approaches to make digitalization sustainable, including life cycle assessments (LCA) (Ruiz-Salmón et al., 2020; Ruiz-Salmon et al., 2021; Laso et al., 2022), material flow analysis (MFA, Abualtaher and Bar, 2019), principle component analysis (Naughton et al., 2020), bioinformatics to mitigate mislabelling in worldwide seafood market and to protect consumer health (Vindigni et al., 2021) and so forth. Cooney and co-workers

Table 1

Digital technologies - definitions and applications in peatland ecosystems

Digital technologies	Peatlands use
Information and communications technology (ICT) encompasses the capture, storage, retrieval, processing, display, representation, presentation, organization, management, security, transfer, and interchange of data and information.	Connects Peatlands ecosystem
Internet of things (IoT) - network of smart, interconnected devices and services capable of sensing or listening to requests and	Aquatic systems
perform actions using actuators. IoT enables network sensors to remote connect, track and manage products and systems.	Agroforestry
	Vertical farming
Cloud computing – use of tools and applications (such as data storage, servers, databases, software) based on a network of severs through the internet. It enables user to rent computer resources on demand to store files and applications in a virtualised servers and access all data via the internet.	Connects peatland ecosystem
Artificial Intelligence (AI) defines machines achieving human-like cognitive functions (ex. learning, reasoning, interacting) that comprises different forms of cognition and meaning understanding (such as speech recognition) and human interaction (signal sensing, smart control, simulators) rooted in algorithms and software.	Sensors, chips, robots, logistics, autonomous machines
Machine learning (ML) – a subset of AI, use and development of computer systems that learn and adapt without following explicit instructions by using algorithms and statistical models to analyse and draw inferences from patterns in data	Automated processes across peatland
Big data – continuous increase in data & technologies that needs to be collected, stored, managed and analysed. Complex and multidimensional that impacts processes, technologies. Characterised by Volume (amount of data sets), Velocity (speed of data processing). Variety (types/sources of data). Veracity (quality of data analysed).	Full peatland innovations and value chain
Blockchain is a shared digital, immutable ledger that facilitates the process of recording transitions and tracking assets in a business network using cryptographic algorithms). Blockchain protocols aggregate, validate, and relay transactions within the blockchain network. The blockchain system records the transactions in sequence. A transaction may contain a value transfer or a smart contract invocation. Almost anything of value can be tracked or transacted on a blockchain network, reducing risk and costs in business.	Traceability, security of processes across peatlands, novel business models
Photonics is a multidisciplinary field related to light including energy generation, detection, and process management. Photonics is the scientific study or application of electromagnetic energy whos basic unit is the photon, incorporating optics, laser technology, electrical engineering, material science, and information storage and processing. Photonic applications use the photon in the same what that electronic applications use the electron. Agri-photonics is a growing area of precision agriculture (Massaro et al. 2020)	Disease mitigation. Optic sensing to for protein levels in foods, water quality and fish health
Augmented reality – a technology that superimposes a computer-generated image on a user's view of the real world; thus, provide a composite view.	Training
Virtual Reality – the computer-generated simulation of a 3D image or environment that can be interacted with in a seemingly real or physical way by a person using special electronic equipment such as believe with screen inside or cloves fitted with sensors	Training
Quality of Experience (QOE) – is the degree of delight or annoyance registered by the use of an application or service.	Training
Logistics – the detailed organization and implementation of a complex operation.	Peatlands eco management
Robotics – a branch of technology that deals with the design, construction, operation and application of robots. In multi-robot or swarm robot systems, the robot collaborate to complete predefined tasks.	Aquaculture feed, monitoring
Cobot, or collaborative robot, is a robot intended for direct human robot interaction with a shared space, or where humans and robots are in proximity.	Food processing, in field logistics
Digital twin $-$ a digital win is a virtual model designed to accurately reflect a physical object.	Wind turbine fitted with various sensors for control
Edge Cloud – Edge computing is developed as complement to cloud computing, encompassing storage and compute assets located at the edge and interconnected by a scalable, application-aware network that can sense and adapt to changing needs, securely & in real time	Peatland ecosystem networking
Cybersecurity or information technology (IT) security – is the practice of protecting critical systems and sensitive information from digital attack. It is how individuals and organisations reduce the risk of a cyber attack where cyber security code function protects the devices (smartphones, laptops, tables).	Security of production, products and services
Cyber-physical systems refer to systems where software and hardware components are seamlessly integrated towards performing well defined tasks.	Security of processes.

(2021) investigated impact categories that can also be applied to evaluate emerging peatland aquaculture systems, namely using LCA and included global warming potential (GWP), acidification potential (AP), eutrophication potential (EP), freshwater and marine ecotoxicity potential (EAETP and MAEPT), cumulative energy demand (CED), net primary production use (NPPU) and water use. Rowan and Casey (2021) evaluated the maturity of paludiculture eco-innovation in terms of technology, society and policy readiness levels (O'Neill et al., 2022). The present article expands on the scope of the previous articles by:

- Addressing what digital transformation of the peatlands simply means and its' application to supporting sustainability in order to embrace wide opportunities for the development of green enterprises, job creation and to improve competitiveness of the region with global orientations.
- Describes and explores current state-of-the-art knowledge on sustainability across a diversity of peatland innovation, and adjacent activities, framed upon the use case of a novel freshwater paludiculture site in the Irish midlands and digital transformation.
- Unlocks challenges and opportunities for the development of peatland innovation as it relates to environmental, economic and societal sustainability informed by digital transformations. This includes the likely impact of climate change and the uncertainty arising from extreme weather events, habit loss or biodiversity disruption and mitigation using effective precision systems.

# 2. Development of aquaculture, aquaponics and adjacent green innovation in the Irish peatlands

Aquaculture is the fastest growing sector in agriculture and has a long history of contributing high-quality proteins to humans (O'Neill et al., 2022). Aquaculture is diversifying compared to other agriculture sectors in terms of fish species, feeds, production systems, diseases, products, business sectors and marketing (FAO, 2020; Yue and Shen, 2022). Developing live feeds, including microalgae, rotifers, and brine shrimp in hatcheries have bridged the bottleneck in aquaculture of some marine species (Yue and Shen, 2022). Selective breeding has enabled progression of quantitative genetics that has improved traits in over 60 commercial aquaculture species (Gjedrem and Robinson, 2014). This includes QTL (quantitative trait locus) mapping and maker assisted selected (MAS) that have been a basis for trait selection. Improved feed formulations to meet the nutritional requirements of each fish species have improved feed conversation rates, reduced costs and improved product quality (Tacon and Metian, 2015). Technologies and innovations for disease mitigation and management have controlled diseases in aquaculture (Kelly and Renekdas, 2020; Pogue et al., 2021). However, it will be very challenging to meet increasing demand for seafood seen as a viable option with a low environmental impact for feeding the growing world population (FAO, 2020). It is important to support and enable product diversification from aquaculture and adjacent activities that are likely to be impacted by disruptive environmental

conditions, reduced supply of fish meal and oils and extreme weather events brought on by climate change (Shen et al., 2020). These challenges, and opportunities have stimulated innovations in aquaculture (Ab Rahman et al., 2017; O'Neill et al., 2022), such as genomic selection (Houston et al., 2020; genome editing (Gratacap et al., 2019), information and digital technology (Hassan and De Filippi, 2021), recirculating aquaculture systems (RAS), development of renewable energies such as solar (Aich et al., 2020) and wind (O'Neill and Rowan, 2022; O'Neill et al., 2022), vaccines (Shefat, 2018), and novel business strategies with blockchain (Anderson et al., 2019).

The Mount Lucas peatland site covers ca. 4 ha in the Irish midlands, has also a strong trajectory to support and enable green innovation and social enterprises including vertical farming, honey production, exotic fungi/ mushroom cultivation, agro-forestry including provision for dark-sky ecotourism (Fig. 1). An advantage of peatland-based aquaculture is that it harnesses water from rivers and lakes and doesn't have the possible threats from potential pollutants in raw, untreated wastewater, such as agricultural run-off (O'Neill et al., 2020). The emphasis is on creating a balanced system in the four 1000m<sup>3</sup> fish ponds, which were created using a natural glacial till (Fig. 1). The duckweed area is 1.2 ha, linked by channels to the pond. The full volume of water is exchanged every 4 h. Among the many green boxes the project ticks is low freshwater use, with any additional water requirements normally taken care of by the Irish weather (O'Neill et al., 2019). The Mount Lucas site has a licence for 25 t of fish biomass, which works out at 13 t perch and 12 t of trout that are produced for high value markets across Europe (O'Neill et al., 2020). The underlying principle is define process for replication across peatland where the limited tonnage of fish biomass is dictated by 'organic status'. Thus, Mount Lucas hosts a blend of traditional social enterprises with emerging green technologies, products and services that requires digital transformation to effectively manage their potential along the full value chain that includes efficient wastewater recirculation and bioprospecting of bioactives. Digital transformation can also highlight the maturity of key innovations that will potentially meet several sustainable development goals of the United Nations (Galanakis et al., 2021; Rowan and Casey, 2021; O'Neill et al., 2022).

### 3. Exploiting digitization to integrate and unlock vertical and horizontal value chains

A definition of key terms and technologies that underpin digital transformation can be found in Table 1 that also connects with the envisaged peatland aquaculture innovations.

*Vertical integration* of processes crosscut the whole value peatland ecosystem from procurement of monitoring equipment and smart feeding regimes to optimised fish biomass to diversification of activities to include recycling of wastewater for irrigation of vertical farming and exploitation of the extensive "wilded" ecosystem for pollination/ecosystem service management. Essentially, all data generated from production processes is considered from the perspective of real-time efficiency, quality, risk mitigation and organization planning by digital technologies that will be optimised into an integrated system (Savastano et al., 2018).

Horizontal integration extends beyond the internal operations at Mount Lucas peatland site from suppliers to end-users including all key stakeholders. In recent times, seamlessly connecting this broad holistic ecosystem can be potentially met by the Quadruple Helix 'Empower Eco™' Concept that unites academia, industry, government and society regionally (Rowan and Casey, 2021). Digitization is the foundation for the development of new strategic plans and enable opportunities in real-time through specialist training across the water-energy and food nexus that will also stimulate new social enterprises, greater community engagement, and job creation in paliduculture. For example, exploiting the European Digital Innovation Hub (EIDH) concept at Mount Lucas will manage the digital transformation of bespoke peatland eco-innovations, along with connecting this site trans-regionally with complementary EIHDs to create and add value (Galanakis et al., 2021). Funding instruments such as Horizon Europe MSCA RISE, Erasmus Plus, European Just Transition and Interreg



**Fig. 1.** Aerial view of Mount Lucas peatland site in the Republic of Ireland for development of aquatic innovations powered by wind turbines. The wind turbine indicated (WT) provides all electrical requirements for the production site (red). The site consists for 4 D-end split culture ponds (blue), a 16-channel treatment lagoon (green) which contains algae and duckweed, and a water reservoir (yellow) for the system. Water is taken is from an adjacent bog river (- -). This occurs only to compensate for loss of water from the system via evaporation. Discharge from the system only occurs during times of excessive rainfall whereby filling the overflow tank (orange). The site is situated in the middle of a peatland which is undergoing natural restoration.

programmes can support and accelerate horizontal integration and at Mount Lucas will be matched with balanced rural enterprise and growth for economic, political and societal sustainability (Rowan and Casey, 2021; Rowan and Pogue, 2021).

Digitization provides access to an integrated network of unexploited big data with potential benefits to society and to the environment (Mondejar et al., 2021); moreover, it is the integration of digital technologies into everyday life. Appio et al. (2021) noted that our ability to make well-informed decisions underpinning how to efficiently use natural resources and services has a significant impact on sustainability and equal access. The Food Standards Agency UK (2021) highlighted the role of digital innovation as emerging technologies that will positively impact on the UK food system. Moreover, the Food Standards Agency UK (2021) stated "digital technologies are been adopted rapidly across the value chain that includes discrete applications at the field, farm and factory level that includes automation, robotics and performance monitoring - mostly aiming at process optimisation; at the consumer level with a multitude of innovative new internet-enabled food distribution platforms and services; and increasingly at an integrated system level, connecting actors at all stages of the value chain, including supply chain management, and secure and gap-less digital traceability of food items from farm to fork". It loosely grouped digital technologies (DT) across three main categories, (a) DTs applied directly to food production processes (such as sensor-based agriculture, traceability, monitoring of production and delivery) where resulting flow of information is based on input data from the food; (b) DTs that generate information relevant to food from input data, but not directly gathered from the actual food (such as social marketing for customer awareness, behaviour and choice); (c) DT platforms used to aggregate and transmit data securely (cyberphysical), record keeping and for making decisions either automomously or with human input. It is appreciated that digitization is a vast topic; consequently, this article will provide examples of digital technologies that are currently applied to, or will future enable peatland innovation with particular relevance to the development of the Mount Lucas site in the Republic of Ireland.

### 3.1. Robotics

No robotics are currently deployed at Mount Lucas; but, are being used and developed elsewhere for feeding, cleaning ponds (Osaka et al., 2010; Yue and Shen, 2022), oral or injecting vaccines (Lee et al., 2013), monitoring behaviours and removing diseased fish (Antononucci and Costa, 2020), all labour intensive and costly processes (Lucas et al., 2019). For the first time, consumption of farmed fish has exceeded that of wild-caught fish, and by 2030, digitization of aquaculture (such as by use of robotics) and aquaculture production will potentially contribute for two-thirds of fish that humans consume (Connolly, 2018). Robotics can facilitate risk mitigation and improve profit margins and sustainability (Table 1). For example, in the salmon industry automated underwater robots have been deployed for cleaning and inspecting nets and have reduced human operations (Paspalakisk et al., 2020) and have been used to evaluate fish health and monitor and prevent escape of farmed fish (Ohrem et al., 2020). Some of the advantages of robots is that they more profitable as they can work continuously and can be substituted to address none-ideal human workforce conditions, labour intensity and sophistication. The high density of fish during aquaculture increases the risk of disease, which reduces production and profits. Cermaq are overcoming this problem using robotics to sort diseased or harmed fish from good quality fish for processing (Connolly, 2018). SINTEF (an independent research organization in Norway) is developing underwater robots to examine and repair nets for marine aquaculture and contributing to the cost-effectiveness and sustainability of the salmon industry (Connolly, 2018). A survey of the literature revealed that several academic/industry partnerships, Innovasea (https://www.innovasea.com/), Cermaq (https://www.cermaq.com/), Robotfish (https://www.edlbf. com/), SeaVax (https://theexplorer.no/), and Sublue (https://www. sinfet.no/en/) are developing robotics for aquaculture (Yue and Shen, 2022). Moreover, with the development of Industry 5.0 human centric

technologies (Xu et al., 2021), collaborative robots (cobots) will play vital roles in agriculture, paludiculture and food production along the full value chain. These cobots will work closely with human operators to perform dangerous and less popular menial tasks along with more labor-intensive, monotonous and repetitive jobs (Wong et al., 2022). Cobots can also be used as transportation for in-field logis0ics applications (Ducket et al., 2018).

### 3.2. Drones and satellites

In terms of aquaculture, drones can function above and below the water (Sousa et al., 2019; Yue and Shen, 2022). Connolly (2018) noted that drones can be used for monitoring offshore fish farms such as for inspecting under water cages for damage and holes. Drones can be used to monitor duckweed coverage and production in the ponds of the Mount Lucas peatland site where they can contribute to increase operational efficacy of wastewater treatment, quality and recirculation (O'Neill et al., 2019). Yue and Shen (2022) noted research institutes and industries developing drones for deployment in aquaculture include Subblue (https://www.subblue. com/), Apium Swarm Robotics (http://apium.com), Blueye Pioneer (https://www.avetics.com/) and SeaDrone (https://seadronepro.com/). Connolly (2018) reported that drones produced by Apium Swam Robotics can be deployed en masse to survey the oceans and use sensors for analysis. Whereas Blueye Pioneer produced drones that provide live streaming of underwater exploration using a Blueye app on a smartphone, computer or headset/goggles. Drones can be used to collect data that would be challenging for humans to obtain and can contribute data that can be used to generate new algorithms for developing technologies (Yoo et al., 2020). Drones enable surveying of large areas and could be used to survey and monitor the status of the entire Irish peatlands (ca. 80,000 ha) and the collected data used to assist planning and monitoring of rewetting, rewilding and to support eco-innovation developments. Salidrone (https://salidrone. com/) is collecting data for evaluating fish stocks and environmental conditions and by integrating AI and cloud computing the cost of aquaculture can be reduced and the operational performance of aquaculture improved (Chen and Zhang, 2017). The potential of drones for aquaculture is enormous and currently the drone market is estimated to be worth \$5.1 Billion by 2025 (Yue and Shen, 2022).

Other adjacent studies have reported on the use of sensors for measuring water parameters can advance aquaculture (Xing et al., 2019; Su et al., 2020). Science Foundation Ireland (2021) reported on new 'Terrain AI' project between academic researchers and Microsoft Ireland focused on digital transformation that includes novel multimodal sensing technologies, IoT devices, Microsoft Azure-informed AI and cloud-based innovation to monitor carbon sequestration of peatlands and other land types to improve our understanding of the relationship between human activity and land use and how this may relate to climate change. Terrain-AI will build artificial intelligence (AI) models that can inform more effective and sustainable management practices, leading to significant carbon reduction. Data will be captured from satellites, airborne platforms, as well as infield instruments, from 14 test sites strategically located across Ireland. This digital project will help improve our understanding of the interactions between the land and human activities that lead to carbon emissions that will also support and enable comparison between peatlands and all land types including grasslands, croplands, forestry, wetlands, to urban areas. This will integrate the aforementioned-generated data into a modelling framework that will inform more effective policies to reduce carbon emissions. Terrain-AI will also help to inform future land use practices that will achieve reduced carbon outputs such as, precision aquaculture and farming, carbon sequestration of grassland, and new approaches to public transport, along with other adjacent needs such as tree planting in urban areas. Terrain-AI will design a cloud platform that can use the insights from these Irish findings that will be shared with other countries to enable everyone to explore land usage and carbon reduction in their respective geographical jurisdictions. Notable, Microsoft has been carbon neutral across the world since 2012 and is committed to achieving carbon negative

status by 2030; it seeks to promote sustainable development and lowcarbon business practices through the use of cloud-enabled technologies (Science Foundation Ireland, 2021). This builds upon delivery of Airband project in partnership with Teagasc (https://www.teagasc.ie/) tohelp the farming community to stay connected, where Terrain-AI will explore how we can leverage technology to reduce carbon emissions across different land types. This Triple HUB approach combines Earth Observation, Geocomputation and Climate Modelling with stakeholders that will collect high quality data, extract verifiable information and generate the facts to enable society make informed decisions about changing how we manage our climate and environment. It will deliver unique insights to help landowners and planners make better informed decisions to reduce carbon emissions. This platform leverages off the latest in AI and IoT technologies where open sharing of data and insights will collectively inform new solutions to reduce carbon emissions globally and will support the delivery of a net zero future.

### 3.3. Sensors

Many of the drones and robots use sensors to navigate underwater to collect data such as physicochemical parameters, or above water for geosurvey and referencing (Yue and Shen, 2022). Connolly (2018) noted the biosensors, such as those produced by Sense-T can improve efficiencies across he salmon to oyster industry by analysis of oxygen levels and water temperature; even extended to measuring heart rate and metabolism. Sensorex devices are used to monitor dissolved oxygen levels and pH to create appropriate environment for improve shrimp efficiency and yields (Connolly, 2018). The company YSI has also developed handheld sensing devices for automatic feeding technology and transportation tanks that maintains ideal environment for fish (Connolly et al., 2018). Recent studies at Mount Lucas has focused on the monitoring water physicochemical parameters along with matching use of handled algal-Torch for monitoring algal populations to that of using flow cytometry for real time determinations in living laboratories (O'Neill et al., 2022).

Biosensors have been exploited to advance the aquaculture industry for determining DO levels, water salinity and temperature (Antononucci and Costa, 2020). Svendsen et al. (2020) has used biosensors to monitor heart rate and other physiological conditions in salmon. While Zhou et al. (2019) described the potential of using underwater sensors linked to the internet to inform efficacy of feeding based upon hunger status of cultured fish in various aquatic environments including ponds and rivers. Yue and Shen (2022), reported on a European consortium project comprising academia and aquaculture companies that are developing an automated/integrated platform to detect and monitor chemical contaminants, harmful algal blooms, pathogens and toxins. Mount Lucas aquaculture can be advanced by developing and testing sensors combined with cloud management along with mobile phone apps to help inform the establishment of an optimised environment for fish and feeding; Yue and Shen (2022) also advocate need for sensors to monitor stress levels in individual fish species and emergence of pathogens in water, where devices could be inserted into live fish in such a manner to support detection on land, boats or satellites. Measurement of changes in microalgae species in the aquaculture pond (O'Neill et al., 2022; O'Neill and Rowan, 2022), and disruption brought on by flooding to IMTA system due to frequent storms were attributed to climate change (O'Neill et al., 2022).

### 3.4. Photonics

Optical and photonic technologies are currently adopted to measure crop health and agri-food quality using remote sensing data in the visible, near-infrared, and thermal-infrared wavebands (Massaro et al., 2020) (Table 1). Agri-photonics constitutes a new area of research that encompasses electronic and opto-electronic technological advances implemented on unmanned aerial vehicle (UAV), decision support systems (DSS), multispectral imaging, and precision agriculture sensing. Traditional methods in agriculture cultivation is labour-intensive and struggles to meet the increasing demands of our growing populations. Optics and photonic technologies have been proven as state-of-the-art-solutions for helping with crop production and harvesting technologies (Yeong et al., 2019); thus, have potential applicability for transforming peatland innovations.

### 3.5. Artificial intelligence and machine learning

Artificial intelligence (AI) can help inform reliable and appropriate decision making based upon large data sets measured using digital devices, such as drones, robots, and sensors (Yue and Shen, 2022) (Table 1). An Australian company, the Yield, provides a diverse suite of technologies for all types of agriculture which it uses Sensing + Aqua technology to enable predictive analytics for enhanced data-driven decision- making (Connolly, 2018). This author also noted that nearly 32% of wild-fish caught are procured unsustainability where AI through cameras and data collection can identify species, reduce overexploitation of fish species, and enable greater accountability of harvesting methods. It is envisaged that research from this peatland site will be linked to the Terrain-AI project (Science Foundation Ireland, 2021). Indeed, the need to rapidly develop aquaculture and adjacent innovations is reflected in many global partnerships between academia and aquaculture industry. Such efforts are deploying AI to inform decision making (Evensen, 2020) that can reduce reliance on labour intensive practices such as feeders, water quality, harvesting and processing. Josthiswaran et al. (2020) highlighted the potential of using AI in enhancing control systems in aquaculture, such as in the area of reducing waste streams and improving costs. However, there is commensurate opportunities to use machine learning and development of algorithms based on increased data sets through Open Innovation and knowledge exchange for advancing the industry linked to academia and digital companies. One research effort is to apply deep learning technologies in aquaculture, e.g. for fish classification, counting, behaviour monitoring, and fish fillet defect detection (Sun et al., 2020, Yang et al., 2021a, 2021b). Deep learning has outperformed the traditional machine learning algorithms in many application areas. However, one major drawback of deep learning is that it requires a large dataset to train the model, which is a significant challenge for applying deep learning in aquaculture.

Digital transformation can also advance bioprospecting of key bioactives from microalgae, duckweed and other peatlands resources for one health application. For example, AI can be used to potentially screen and inform the appropriate type of bioactives that elicit pro- and antiinflammatory responses for fish welfare with view to fortification of feed (Murphy et al., 2020a; Murphy e al., 2020b; Murphy et al., 2022; Pogue et al., 2021). Masterson et al. (2020) reported on the use of lentinan from Shiitake mushroom to ameliorate against clinical isolates of Klebsiella pneumoniae exhibiting antimicrobial resistance using novel lung infection models. Commensurately, Felix and Angnes (2018) also highlighted increased interest in electro-chemical immunsensors that may potentially disruptive the bioprospecting of such high value peatland products; such immunosensors explore measurements of an electrical signal produced from an electrochemical transducer. Moroever, this signal can be voltammetric, potentitiometric, conductiometric, or impedimetric that can be harnessed as tools since they are specific, simple, portable, generally disposable, and can carry out in situ or automated detection (Felix and Angnes, 2018). Use of bioinformatics linked to machine learning can be applied to understand diversity and richness of microbial and algal species in the aquaculture ponds (O'Neill et al., 2022). Use of drones and satellites can support and enable digital transformation of in situ living labs connected to environmental test beds at Mount Lucas (Rowan and Casey, 2021).

### 3.6. Immersive technologies, augmented reality (AR), virtual reality (VR)

Augmented reality is an interactive experience within which digital and context-based content is overlayed upon real-world objectives (Egan et al., 2016; Braga Rodrigues et al., 2020). As such, AR could be employed to inform aquaculture activities by their nature are highly variable and laborintensive that are frequently influenced by species, location and aquaculture process (FAO, 2020; Yue and Shen, 2022). Connolly (2018) reported that the U.S.Navy uses Divers Augmented Vision Display (DAVD) that superimposes high-resolution sonar imagery on a diver's visual experience. AR can be used to improve efficiency of aquaculture production, monitor mortalities and welfare status of fish under a plethora of environmental conditions. This was also noted by Yue and Shen (2022) that the use of AR, in combination VR, for training and education as it pertained to fish welfare, disease prevention, escaping fish and dangerous working conditions. Yue and Shen (2022) highlighted the capability of AR to inform efficacy and economics of deploying underwater drones and robots that encompassed monitoring fish behaviour and mortality. Yue and Shen (2022) have reported on the use of AR with a cloud system to advance aquaculture to increase fish biomass, and to monitor fish health linked to water parameter determinants. Augment and Virtualh Reality in may be possible to review and evaluate appropriate locations across ca. 80,000 ha of peatland for both sustainable carbon cycles and deployment of aquaculture systems that includes risk mitigation.

### 3.7. Georeferencing and mapping

ArcGIS mapping can be used to inform restoration and rehabilitation of peatlands linked to ecology/biodiversity to inform carbon cycles (Rowan and Casey, 2021). ArcGIS mapping of the 80,000 ha of peatlands under the management of Ireland's Bord Na Mona (State Body) would enable profiling of peatlands to match locations for development of aquatic innovations in terms of desirable production sites balanced with environmental protection and ecology. ArcGIS mapping of peatlands, linked to ecology/ biodiversity, ensures appropriate use of carbon cycles. Bord na Móna is rehabilitating and restoring bogs as part of its Peatlands Climate Action Scheme with the aim of reducing carbon emissions and the eventual creation of carbon sinks. This is complex as raised bog is very dependent on sphagnum moss activity that prefers acidic conditions; however, the ground water in cutaway bog from peat harvesting will have become alkaline. Thus, it is important to map the peatlands linked to physiochemical profiles that would be extremely challenging without using digital tools such as ArcGIS. For example, given differences in peatland acidity, solutions are more so focused not on bog restoration, but bog rehabilitation, which depends on the peatland providing a suite of habitat types, such as wetlands, fens, scrub and woodland that would benefit from end-to-end monitoring and use of cloud edge computing (Table 1). Bord na Móna, with Ireland's Economic and Social Research Institute (ESRI) has a trajectory to actively restore and rehabilitate 33,000 ha of peatlands. It uses ArcGIS to design the most appropriate rehabilitation measures and then implements a wide range of measures in real time; moreover, ArcGIS rapidly visualises the existing conditions across thousands of hectares of bogs using numerous datasets that informs design and implementation of the most appropriate rehabilitation measures to restore peatland function and deliver climate action benefits. For example, for each bog identified for rehabilitation, GIS specialists and ecologists use the desktop solution ArcGIS Pro and 3D spatial analysis tools to examine the ground level and create detailed, mapbased rehabilitation plans.

Tahar et al. (2018a, 2018b, 2018c) reported on the use of ArcGIS to enable monitoring and mapping of emerging contaminants of concern in aquatic environments for subsequent risk mitigation and management decision making (Tahar et al., 2017). Effective risk assessment and prediction for deployment of appropriate interventions to mitigate pollutants in waste water is at best 'semi-quantitative' given the enormous number of contributory factors and variables to inform management decision making (Tahar et al., 2017). Tahar et al. (2018a, 2018b, 2018c) highlighted that occurrence and geodatabase mapping of contaminations of emerging concern onto appropriate river basin catchment management tools will inform predictive and simulated risk determinations to inform strategic investment in necessary mitigation infrastructure to protect rivers and economic activities that rely on clean water. The medium to longer term ambition would be to utilize relevant European software and models for the development of spatially explicit Geography-Referenced Regional Exposure Assessment Tool for Peatlands to manage these resources, similar to what is been achieved for European River Basins.

Digital transformation of data sets would enable real-time and improve reliability of water quality determinants for risk management and policy decision making. There is help identify key constraints, such as knowledge underpinning sensitivity of existing sophisticated analytical equipment to measure low-level pollutants in real time; therefore, the commensurate development of appropriate risk management models will also inform future intensification and diversification of aquatic industries including peatland-based innovation. For example, Tahar et al. (2017) developed a semi-quantitative risk assessment model for evaluating the environmental threat posed by three EU watch list pharmaceutials namely, diclofenac, 17-beta-estradiol, and 17 alpha-ethinyestradiol, to aquatic ecosystems using Irish data; this model adopts EPA's Source-Pathway-Receptor concept to define relevant parameters including low, medium and high risk score for each agglomeration of waste water treatment plants, including catchment, treatment, operational and management. It is envisaged that a similar type semi-quantitative RA approach may aid development of peatlands globally in terms of screening for potential risks where there is a need to measure or predict environmental pollutant concentrations and where hydrological data are available. This approach is semi-quantitative, as other factors such as climate change will need to be considered for estimating and predicting risks with new aquatic innovations. Nair and Domnic (2022) noted the influence of machine leaning in advancing many aspects of this industry'; specifically, these authors described a combined strategy including non-learning enhanced method and deep CNN (convolution neural networks) for picture reduction and reconstruction in underwater imagine in aquaculture. These authors suggested that this model outperforms existing methods in terms of picture enhanced, compression, and reconstruction quality.

### 3.8. Edge-internet of things (IoT) systems

Globalization has radically informed the development of sustaining and disruptive technologies (Schuelke-Leech, 2018), where traditional industries such as agriculture and aquaculture employ vanguard technologies to expand upon opportunities that has enabled smart farming and the agri-food industry 4.0 (Klerkz et al., 2019). Perez-Pons et al. (2021) highlighted pressing need to make farms more profitable and sustainable via the analysis of data envelopment analysis and the application of the Internet of things and Edge computing; this approach allows monitoring environmental conditions with real-time data from the different sensors installed on the farm; thus, minimizing costs and achieving robustness by way of transitioning important data to the cloud after edge computing. Essentially, the edge devices process the data and then decide either to send this data to the cloud for further processing or make decision locally at the edge (Table 1). Edge computing can also be applied to sensor fault diagnosis and data repair (Wang et al., 2021). Technology requirements are increasingly important for agri-food industry with particular emphasis on meeting challenges faced by producers along value chain that also reflects a diversity of types of farms such as crop-cultivating or mixed farms that grow crops and produce livestock. These also rely upon fragile water resources; moreover, Eurostat noted that that total irrigable are in the EU-28 was ca. 15.5 Mha (8.9% of the total) whereas only 10.2 Mha (5.9% of the total) was irrigated thus highlighting the opportunities for implementing low-cost technological solutions (Fleming et al., 2016). The Industrial Internet of Things (IIoT) potentially enables technologies focused on implementation of monitoring and resource management solutions across many Industry 4.0 applications that embraces cloud computing, big data, AI or distributed ledger technologies (such as blockchain) that will improve traceability and productivity of commercial processes (Yu et al., 2017).

Perez-Pons et al. (2021) noted that when transmitting data to the cloud, there remains several challenges including data privacy, energy consumption, or costs associated with cloud services. Essentially, service providers charge users relative to the amount of data transferred, stored or processed in the cloud. However, by using Edge Computing technologies, one can decrease the amount of data transferred between the IoT layer and the cloud that will also allow for deployment of machine learning models at the edge of the network reducing response time and enabling service provision even if communication with the cloud is interrupted (such as rural areas associated with peatlands) (Alonso et al., 2020). Perez-Pons et al. (2021) had also reflected on the findings of Pedra-Munoz et al. (2016) who reviewed years of improvements of applying technologies at familyfarm level with sustainability, where the former reported that IoT and Edge computing can present a competitive advantage when measuring efficacy of decision making units. Perez-Pons et al. (2021) reported on different variables from the Environmental Performance Index with real-time sensors and the application of Edge-computing platforms that can reduce the data traffic to the cloud. In addition, 5G is a key enabler of edge computing, which provides low latency and high bandwidth communication services between sensors/edge devices and the cloud, and as well as direct device to device communication (Wang et al., 2021).

### 4. Pollination and ecosystem service management

Mount Lucas is an abundant source of heather and other peatland flowering plants, this provides a rich opportunity for honey production from nectar or from secretions from living part of plans by native bees; this can be addressed in on site living labs. The stressful environment created by the peatlands can influence emergence of novel bioactive properties produced in heather-honey. Shafiee et al. (2013) has previously reported on the use of machine learning to differentiate and classify polyfloral from monofloral honey where latter has a higher commercial value. There is a correlation between honey colour and its floral origin and some chemical parameters that can be discerned using image analysis and algorithms. Different monofloral honeys have distinctive flavour and colour due to variance in their main nectar sources (Escriche et al., 2011). Use of imagine analysis is an area of emerging importance that reflects a method that supports rapid, real-time, simple, selective and low-cost properties appropriate for honey characterization. Moreover, honey industry requires simple, non-invasive, fast and economic technologies for characterizations - other digital approaches include an electronic tongue and data fusion of electronic nose (Escriche et al., 2011) where there is good correlation evident by applying data fusion. The latter aids human panels in making decisions for application to honey quality evaluation that captures adulteration, classification of flora types along with their geographical sources. Image analysis can help discern honey colour that depends on a plethora of factors including phenolic and flavonoid contents, mineral contents and antioxidant activity (Dominiguez and Centurion, 2015). Machine learning as an innovation has been shown to enable objective assessment of visual attributes of food quality including colour. There are other multiple opportunities for machine learning for informing pollination; for example, Goblirsch et al. (2021) describe the potential of electron-beam for low-temperature treatment of pollen contaminated by complex bee parasites and viruses that may be used for commercial bumble bee purposes. The complexity of pollen contamination can be unravelled be potentially unravelled by use of flow cytometry using a suite of specific biomarkers in real time- where this non-invasive enumeration approach is potentially appropriate for machine learning and automation applications. The Irish company Apis Protect developed a digital platform using sensors and machine learning for monitoring hives; specifically, key parameters encompassing temperature, sound, humidity and temperature are recorded suing a wireless in-hive sensor device (Robb, 2021).

# 5. Waste water recirculation – nexus between monitoring, treatment and energy using a paludiculture framework

The Farm to Fork Strategy which is at the heart of the EU Green Deal is a key driver for efficiency within the nexus of food production, water and wastewater resources and treatment and broad sustainability issues. Indeed across the OECD Farm to Fork strategy reflects the requirements of food systems in terms of the "triple challenge" of food security and nutrition, livelihoods, and environmental sustainability (Rowan and Casey, 2021). While the concept of Industry 4.0 is now well established, when evaluating food production systems in terms of the use and reuse of resources such as water it is appropriate to consider the newer concept of Industry 5.0. Industry 5.0 sees digitisation as not just about productivity and growth; but is also part of a broader need for sustainable, human-centric and resilient industry. In the context of the positive transformation of peatland ecosystems, digitisation, applied to water resources can be seen in the context of Industry 4.0 and 5.0 concepts. The "Digital Water" programme within the International Water Association offers key lessons on how the water industry (or industry's where water is a key resource) can uptake and integrate next generation digital technologies. A series of white papers describe the various opportunities and challenges associated with the Digital Water Concept. Challenges with adoption include technical issues such as integrating smart actuators, sensors, and autonomous control systems in a sensible and transparent manner, cybersecurity issues, human resources issues and, crucially ensuring the need to have a clear value proposition.

Water management is key in paludiculture and should (i) maintain appropriate water levels for the activity in question and (ii) may be required to enable the supply of nutrients through the water inlet (Vroom et al., 2018). This may be required to address nitrogen losses after rewetting due to denitrification processes and anaerobic ammonium oxidation. Indeed this can provide an opportunity for local reuse of N rich agricultural wastewaters (Vroom et al., 2018). Recent work in the UK, analysed water management requirements through pumping of water to rivers when inputs to peatlands, due to rainfall, exceed evapotranspiration (Mulholland et al., 2020). Despite this there can be significant reductions in energy use, costs and associated greenhouse gas emissions for paludiculture when compared to arable activities on deep fen peat (Mulholland et al., 2020). Real time control of storm water control measures has been shown to have significant potential for urban water management (Xu et al., 2021) and reinforcement learning has also potential to mitigate flooding when compared to passive control and rule-based control systems (Bowes et al., 2021). Such innovation could be adapted for water management in paduciculture activities Furthermore, a digital transformation of how water and wastewater resources are managed, can impact both the use of water abstracted directly for paladiculture activities (e.g. for aquaculture) and the management from wastewaters resulting from such activities. Recent work in the water and wastewater treatment sectors can point the way forward in terms of digital adoption for efficient management of water resources in paludiculture. This can also help reduce concerns in terms of digital adoption through visibility of related case studies.

Artificial intelligence can be applied within the water sector under three headings namely (i) modelling, prediction and forecasting, (ii) decision support and operational management, and (iii) optimization. A broad digital transformation would also impact these areas but also a fourth, namely system and infrastructure design could be considered. There are lessons for paduliculture that can be gleaned from other sectors such as storm and flood management, wastewater treatment and the water treatment and distribution sectors. Table 2 summarises, in the context of paludiculture how a digital transformation could impact the sector. Developments in remote sensing could have significant implications for paludiculture. Weiss et al. (2020) presented a meta-review of remote sensing for agricultural applications. Technological improvements have meant that global, regional and local data on crop mapping, yield forecasting, biodiversity loss, water and soil impacts are, in many situations, readily available. Chawla et al. (2020) reviewed remote sensing products for analysis of water quality (e.g. surface water), water quantity (e.g. river or stream flow) and extremes (e.g. flooding and drought impacts). Remote sensing also provides an alternative for locations where in-situ sensing is problematic where satellite remote sensing can now monitor in near-real-time retrievals, most components of the terrestrial water cycle. Challenges remain relates to accuracy, data consistency, utility and also in retrieving data related to groundwater, water quality, surface water levels, and river flows and in relation to the products themselves (Chawla et al., 2020).

#### Table 2

Examples of impacts of digital transformation on water resources in paludiculture.

Modelling, prediction and forecasting	System and infrastructure design	Decision support and operational management	Optimization
<ul> <li>Scenario analysis to support design (e.g. robust design</li></ul>	<ul> <li>Optimise design to enable future expansion</li></ul>	<ul> <li>Management of water levels</li> <li>Real-time control of waste-</li></ul>	<ul> <li>Optimised pumping regimes for</li></ul>
under varying conditions) <li>Links to process optimisation and potential for real</li>	and digitisation <li>Reduce life cycle costs through development of</li>	water treatment processes <li>Fault detection and diagnosis</li>	water level management <li>Minimise energy consumption</li> <li>Optimised system maintenance (e.g.</li>
time process modelling using digital twins.	digital models and/or digital twins <li>Design to enable future expansion</li>	on water systems <li>Regulatory compliance</li>	preventative maintenance) <li>Optimise on-site productivity</li>

Data collection in the water sector has long been recognised as a key challenge. In many cases the may be collected from relatively harsh environments which results in added maintenance. Furthermore there can be concerns including (i) lack of trust in data veracity, (ii) poor data management systems and (iii) the use of systems that are over-complicated and unoperable for end-users. Therrien et al. (2020) also highlighted key steps and ways forward for ensuring the steps from adequate data collection to action can be completed. Corominas et al. (2018) conducted a comprehensive review of computer based techniques for data analysis to improve operation of wastewater treatment plants. The EU have led in terms of research in this area with the most cited techniques including artificial neural networks, principal component analysis, independent component analysis and partial least squares. However the review acknowledged a lack of objective comparison of techniques, the need for guidelines, the requirement for validation at full-scale, and the limited options for active optimization of data information content and quality. Clifford et al. (2017) proposed an approach balancing spatial resolution of data with the costs involved in collecting such data and It is clear the use of real-time monitoring and control systems has significant potential but further case-specific validation is necessary.

Newhart et al. (2019) comprehensively reviewed specific applications of various real-time control in wastewater treatment facilities and presented examples of on-site applications across various industrial and municipal sectors. While Newhart et al. (2019) pointed out the potential for such control to be more difficult to implement in decentralised wastewater treatment facilities, Fox et al. (2022) demonstrated how data driven real-time control of a decentralised wastewater treatment system (in this case a sequencing batch reactor) can be optimised to achieve regulatory compliance, reduce energy consumption and increase system throughput. Both standard statical approaches and more advanced approaches using neural networks and regression modelling (Fox et al., 2022) were implemented. Key to these approaches was their compatibility with standard (low-cost) programmable logic controllers and enabling the end-user optimise the control approach.

Fault detection and diagnosis, while common across various engineering sectors, has focused mainly on leak detection at a municipal water supply level rather than building or industrial settings (Seyoum et al., 2017; Hashim et al., 2020). Fault detection in these settings can reduce leaks but also enable efficient monitoring and preventative maintenance of equipment such as sensors, valves, pumps and motors used for water and wastewater management. There are various approaches that can be used to leverage data from water systems in industrial processes. Mulligan et al. (2021) developed a series of water distribution system performance assessment rules and demonstrated how these would result in significant energy and water savings and associated greenhouse gas emissions. Hashim et al. (2020) used principal component analysis and support vector machine techniques to enable accurate detection of various faults in a large public building and an industrial setting. The problem of false alarm moderation (false alarms can undermine user confidence) has also received recent attention using both modelled data (Chen, 2010) and using case-study data from industrial settings and large buildings (Hashim et al., 2020). Fault detection in wastewater treatment also requires further testing in full-scale facilities and a pathway forward of using hybrid models such as linking gaussian process regression, artificial neural networks with principal component analysis or reinforcement learning to challenges associated with issues around sensor accuracy and transient operational conditions (Sundui et al., 2021).

Nature based solutions (NBS) have been identified as key in addressing challenges in water management across urban, agricultural and ecological settings. In the agricultural landscape, NBS can be applied for soil health, carbon mitigation, downstream water quality protection, biodiversity benefits as well as assisting agricultural production and supply chains to achieve net-zero environmental emissions (Rowan and Casey, 2021). Evaluting the ecotoxicological safety of NBMS will also be importnat moving forward (Garvey et al., 2015). Examples of recent applications include wastewater treated using naturally occurring algae, bacterial and duckweed using IMTA process (O'Neill et al., 2020), use of earthworms based technology for composing and wastewater treatment applications (Cooney et al., 2021; Hylton et al., 2022; Arora and Saraswat, 2021), zooplankton for tertiary wastewater treatment to enable wastewater reuse (Pous et al., 2021). The INNOQUA (H2020) project presented reviewed and demonstrated various pathways for nature based treatment of wastewater and options for enabling wastewater reuse (Bumbac et al., 2021). The study reviewed constructed wetlands, waste stabilisation ponds, anaerobic treatment systems and vermifilters and presented details of performance, design and maintenance requirements and ability to meet various regulatory standards across a wide variety of applications. NBS technology can be underpinned by models and tools that support better land use, enable accurate nutrient flow modelling and support the development of sustainability metrics via life cycle assessment; moreover, while NBS are designed to minimise technological requirements, there are significant opportunities to further enhance their benefits and enable process flexibility through targeted real-time control (Basil et al., 2021).

### 6. Aquaculture and other aquatic systems

Peatlands presents an opportunity to establish new food production systems that can offset negative environmental consequences of resource constrained conventional terrestrial farming. Peatlands aquaculture can fulfil consumer needs by both intensifying and diversifying production of new freshwater areas and fish species. The emerging nature of the industry provides an opportunity to build in digital solutions that can enable high technological developments tailored to the specific needs of the production model. Technology interfaced with digital solutions can be used to address the challenges of low impact paludiculture (such as low-trophic IMTA, RAS, organic) and environmental services. Digital technologies can support efficient development of paludiculture products and can be used to develop climate-friendly, sustainable production systems generating high quality proteins and other products.

Essentially, Mount Lucas peatland demonstrator is ideal for establishing digital systems to allow digitalization and use of collected data across the full value chain to inform new eco-innovation balanced with environment protection. Digital tools such as Edge-cloud sensors, AI, machine learning and augmented reality can be used to build models from collected data and facilitate implementation of a circular strategy to produce biomass of high value fish (such as trout and perch). At the same time models that integrate biological constraints (microbiome, fish growth and health), system functioning, and effluent outputs can improve biomass productivity and represent a step towards precision aquaculture. Moreover, the integration of customized management systems throughout the supply chain will



Fig. 2. Imagery of microalgae from the genera Scenedesmus, Nitzschia, Monoraphidium, Chlorella, Chlamydomonas and Cyclotella identified in Mount Lucas IMTA.

enable a well-managed, responsive and crisis proof system with improved traceability and authentication (Fig. 2). Such a digitally transformed, intelligent management system will enable development, testing and validation of optimised integrated multi-trophic aquaculture and recirculating aquaculture systems (IMTA/RAS) on the peatlands by integrating multi-sensing (heterogenous sensors), multifunctional real-time modelling for decision making with provision for climate resilience through full system monitoring (microbiome, water physicochemical parameters).

There is a pressing need to design, implement and deploy ICT technologies to provide end-to-end monitoring systems for aquatic-based foods produced in the peatlands (O'Neill et al., 2022). Integrated software, and establishment of smart sensors for real-time monitoring and remote data logging of production, is being implemented at the Mount Lucas peatland site. LCA, PCA, MFA models can contribute to green business models and the standardization of emerging peatland aquaculture and associated businesses globally. The latter also includes use of AI, machine learning and Edge - cloud to develop and evolve strategies to reduce waste and to consolidate gains achieved in aquaculture supply chain management. On a related point, Ruiz-Salmón et al. (2020) noted that the water-energy-food nexus allows assessment of the life cycle of seafood products that enables clustering and knowledge transferring to add value in the European Atlantic region. LCA can be applied to help understanding the benefits of ecolabelling and eco-design in aquculture under a circular economy approach (Ruiz-Salmón et al., 2020). Open knowledge sharing can also help advance paludiculture, such as software that integrates sensors (cameras, remote sensing), and predictive analysis (biomass estimation and forecasting, water quality monitoring) to enable improved operational decisions for optimised production balanced with environmental protection. Combined use of augmented reality and virtual reality can be used for training to promote human resource development, so it accompanies 4IR and becomes part of the digital transformation and Just Transition. ICT can be used to create a quality of experience (QoE) that enables specialist training in paludiculture on site and remote by linking to living labs. Hatch blue accelerator is gaining in popularity as a novel sustainable aquaculture and innovation programme offered globally for innovators that includes those converging disciplines (https://www.hatch.blue/accelerator).

Traditionally, there has been worldwide reliance upon 'end-of-pipe' engineering solutions for discharge wastewater control to safeguard water resources (Barrett et al., 2016; Tahar et al., 2017). This IMTA system of Mount Lucas will provide solutions and data for wastewater management and be a model system for social marketing and studies of consumer awareness and acceptance of aquatic paludiculture processes (Domegan, 2021) that may lead to disruptive innovation (Schuelke-Leech, 2018; Schuelke-Leech, 2021a, 2021b). Mount Lucas will also enable emerging innovations such as oral vaccine testing and the potential use of other safety indicators (Taufek, 2020; Usuldin et al., 2021; Wan-Mohtar et al., 2021). The digital transformation "Terrain-AI" project uses drones across the peatlands for monitoring carbon sequestration as a means for sustaining carbon sinks (SFI, 2021). ICT systems coupled to drones for data collection can provide real-time ecosystem management to promote and preserve pollinators across the peatlands by preserving and enhancing habitats, improving food sources through rewilding and decrease bee-disease through use of innovative technologies (Goblirsch et al., 2021).

The development of fully recirculated systems, such as this aforementioned IMTA system, that relies upon natural processes for remediating water quality, and do not discharge to receiving water presents a step change or potential disruptive sustainable solution. O'Neill et al. (2020) described the first IMTA system developed in the Irish peatlands that uses a balanced ecosystem of naturally occurring microalgae, bacteria and potentially duckweed that regulates waste and maintains water quality. An onsite wind turbine provide a renewable source of energy to operate aeration systems in the circulatory aquaculture ponds. However, global warming has created greater opportunities for extreme weather events that can influence vital food production, by way of droughts and flooding. The ability to monitor and predict the impact of extreme weather events through digitalization of the pilot IMTA processes will help future proof and protect food supplies globally. Naughton et al. (2020) highlighted the potential benefit of digital technologies for connecting 'in field' monitoring devices with living lab sophisticated equipment for real-time decision making in freshwater aquaculture.

Previous researchers have also highlighted that microalgae species that constitute a major proportion of the peatland aquatic-biome are excellence candidates for CO<sub>2</sub> bio-capture (Lopez-Pacheco et al., 2021; Wang et al., 2021); thus, use of end-to-end monitoring linked to sensors that will enhance efficacy will be important going forward. The microalgae in the aquaculture ponds constitute potentially thousands of species (O'Neill et al., 2022) where there is a pressing use machine learning with bioinformatics to unlock their real-time monitoring and occurrence. O'Neill et al. (2022) has also reported on using the occurrence of key microalgae in aquaculture ponds as potential biosensors for assessing the impact of climate change on IMTA process, which can be informed by using Cloud-edge computing. It is notable that microalgae represent a superior option for carbon fixation than terrestrial plants for higher growth and faster biomass production, doubling their biomass in less than 24 h for most species (Farrelly et al., 2013; Guo et al., 2017; Lopez-Pacheco et al., 2021). Fig. 2 illustrates the different microalgae represented of the genera Chlorella, Raphidocelis, Scendesmus, Desmodesmus, Monaraphidium and Graesiella that were isolated from Mount Lucas aquatic environment and were previously noted to exhibit CO<sub>2</sub> capture abilities (Lopez-Pacheco et al., 2021). Tabatabaei et al. (2011) noted that microalgae use carbon dioxide for energy conversion while producing approximately half of the atmospheric oxygen. While Zhao and Su (2020) estimated that microalgae can capture a maximum of 2.35 GtCO<sub>2</sub> in 100,000 km<sup>2</sup> culture area, which shows the potential of using these organisms that naturally occur in the peatlands for CO<sub>2</sub> capture (Ramaraj et al., 2014). Use of microalgae in this IMTA closed system have many advantages, including easy control, insufficient space required, high CO2 sequestration rate, and no contamination risk, nearby all microalgae species may be cultivated, and high biomass density (Lopez-Pacheco et al., 2021). Lopez-Pacheco et al. (2021) also noted that through photosynthesis microalgae can fixate CO<sub>2</sub> by what is known as phyco-capture process. Thus, there is a pressing need to use digital tools to enable end-toend monitoring and to optimise ponds systems to cultivate microalgae for CO<sub>2</sub> bio-capture.

### 7. European Digital Innovation Hubs - Quo Vadis

Digital Innovation Hubs (DIHs) are one-stop shops that help companies to become more competitive with regard to their business/production processes, products or services using digital technologies, while remaining environmentally sustainable and reducing greenhouse gas emissions. DIHs are based on technology infrastructure (Competence Centre) and provide access to the latest knowledge, expertise and technology to support their customers with piloting, testing and experimenting with digital innovations. DIHs also provide business and financing support to implement these activities, if needed across the value chain. As proximity is considered essential, DIH act as first regional point of contact; consequently, a DIH is a regional multi-partner corporation (RTOs, universities, industry associations, chambers of commerce, incubator/accelerators, regional development agencies, and potentially governments), and can also provide strong nexus with other service providers outside their region supporting companies with access to their services. The rationale behind this DIH initiative is to help European industry, small or large, high-tech or not, to grasp the digital opportunities. The EC will focus 500 M€ over the next 5 years from Horizon Europe budget to support the development of DIHs as the level of digitalisation remains uneven, depending on the sector, country and size of company: only 20% of SMEs in the EU are highly digitised. This challenge is particularly pertinent for the digital transformation of peatland eco-products and services. Key assets aligned with DIHs include (a) information on infrastructures; (b) expertise; (c) network contacts of key players and communities at large; (d) expertise in initiating robust collaborations of key stakeholders to meet specialist USP offerings; (e) access to financial capital; and (f) effective digital marking to ensure trustworthy

brand that attracts stakeholders and ensures high quality delivery. Thus, digital transformation of a peatland-focused hub will enable real-time access to networks; upskill and satisfy R&D opportunities; track and communicate technical expertise and consultancy, enable seamless and managed access to facilities; and support the contribution to policy measures.

Currently, there are 706 DIHs registered on the Catalogue of Candidate European DIHs tool that are in following evolutionary stages; fully operational (413), in preparation (223), and potential new DIHs from H2020 (70) (https://europa.eu/!NX87WD). The purpose of this European catalogue is to support networking of DIHs and to provide an overview of the landscape of DIHs in Europe supported by regional, national and European initiatives for the digitization of the industry. European DIHs (EDIHs) will play a central role in the Digital Europe Programme to stimulate the broad uptake of AI, high performance computing (HPC) and cybersecurity, as well as other digital solutions/interoperability for the public sector. EDIHs will support companies, and the public sector organisations, in the use of digital technology to improve the sustainability of their processes and products, in particular with regard to energy consumption and reduction in carbon emissions. The 325 candidate EDIHs are the DIHs (existing or not) that are assigned by Member States to participate in the Call for Proposal of the EC to obtain funding to become European DIHs; a network of 200 EDIHs will be financed in Europe via the new Digital Programme 2021-2027. Currently, the EC is verifying all entries in the catalogue tool based on information provided by each DIH as to whether or not the comply with 4 criteria; (1) be part of a regional, national, or European policy initiative to digitise the industry; (2) be a non-profit organization; (3) have a physical presence in the region and present an updated website clearly explaining the DIHs' activities and services provided for the digital transformation of SMEs/Midcaps or industrial sectors currently insufficiently taking up digital technologies; and (4) have at least 3 examples of how the DIH has helped a company with their digital transformation, referring to publicly available information, identifying for: client profile, client need, and solution provided to meet the needs. Generally, the main functions of EDIHs include (1) Test before invest, (2) Skills and training, (3) Support to Find Investments, and (4) Innovation ecosystem and networking. EDIHs are embedded in a local economy; for example, if manufacturing is important, the hub will enable companies in adopting Industry 4.0 and circular economy methods. Traditional ICT method, such as simulation and supply chain integration, will take on an important role where these are becoming more AI and HPC orientated. In addition, by introducing digital manufacturing, cybersecurity becomes a prerequisite. .

It is likely that the applicability and digital maturity of entities supported by Digital Innovation Hubs alone or embedded in Quadruple Helix concepts will be profiled using the categories described in Table 3. Innovation Radar methodology is an approach adopted by the European Commission to assess the impact of "test before invest" and "support to find investments" services of EDIHs. Mount Lucas peatland HUB will support and enable 'train the trainer' developing ways to transfer knowledge generated in a HPC, AI and Cybersecurity through regular workshops. This will also enable community building for stakeholders that includes agriculture, horticulture, health, public administration and so forth. Other activities to be adopted, aligned with EDIH framework, digital matchmaking marketplace, InvestEU, short term training courses, engagement with regional and national policy makers, deploy effective media presence to highlight peatland activities across the network, and impact assessment of activities that includes analysis of indicators, and KPIs, developing targets, generating new knowledge including Open access provision to support benchmarking and policy recommendations.

# 8. Business modelling and sustainability for enabling accelerating Mount Lucas green innovation

Ziegler et al. (2021) highlighted that despite increasing sustaining or disruptive potential, it remains challenging for value propositions and value network for new paludiculture products and services remains to be hurdled, which presents a barrier for commercial applicability. Several authors with subject matter-expertise in peatland innovation have appropriately noted that digital transformation of paludiculture products and services will have limited utility in the absence of a robust and valid green business model that delivers scale. Ziegler et al. (2021) described paludiculture to be an emerging, science driven, and collaborative innovation that are rarely directly commercially viable, where the authors focused on fuel, fodder, horticulture substrate and construction material. These peatland products are currently not under development at Mount Lucas. Moreover, Ziegler et al. (2021) reported that paludiculture faces significant, adverse path-dependency due to subsidies and regulations that appear to preferentially support agriculture on drained peatlands. Paludiculture initiatives to date typically involve landowners and users where further economic models supporting experimentation and scaling up of paludiculture products and services are required (Ziegler et al., 2021). Zeigler (2020) also reported that paludiculture is the productive use of wet and rewetted peatlands and proposed a 3Ms-schema of mission, modes and making innovation as a device to create space for a wide and inclusive discussion of paludiculture. The reader is directed to the published work of Planko et al. (2016) for key processes for building up a technological innovation system (that includes paludiculture) and arguments supporting a system building model approach in strategic management.

Entrepreneurs and SMEs at Mount Lucas are supported and enabled through a quadruple helix hub approach under an Empower Eco sustainability framework, which connects stakeholders across the business ecosystem including academia, industry, government and society (end-users). This reflects that need to adopt far-reaching changes of the macro environment in order to implement innovative sustainable technologies (Planko et al., 2015; Planko et al., 2016). This Empower Eco holistic approach aligns with previous scholars who reported on the combined insights from the strategic management literature and the technological innovations system (TIS) literature in order to provide a strategy framework for entrepreneurs to collectively build a favourable environment for their sustainable technology (Planko et al., 2016). It is notable that insights from system-building literature originate mainly from the systems perspective (Musiolik et al., 2012) that had previously not considered specific subject-matter insights from the company perspective (Planko et al., 2016). The holistic system building approach, aligned strongly with Quadruple Helix Hub concept depicted by Empower Eco, connects actors across their ecosystem to collaborate strategically in order to shape their environment (Rowan and Casey, 2021). Planko et al. (2016) noted that term 'system building' originates from the TIS literature and is defined as the "deliberate creation or modification of broader institutional or organizational structures in a technological innovation system carried out by innovative actors" (Musiolik et al., 2012). While this system building approach harnesses the use of a diverse suite of assets, facilities and expert in collaborating universities for addressing broad ranging enterprise needs, this approach as supports entrepreneurs and start-ups in co-creation and testing of low cost green solutions (Table 4).

For example, Accelerate Green is the first Irish accelerator dedicated to scaling companies based in the peatlands with a particular focus on climate action and sustainability by developing eco-products and services based on green innovation with a base in Boora, which is adjacent to Mount Lucas (Rowan and Galanakis, 2020). Accelerate Green combines Bord Na Móna's commercial expertise and Resolve Partners expertise in building innovative companies, and is an equity-free scaling accelerator designed to create step change in green innovation (Bord Na Móna, 2022). It delivers eight 2-day deep progamme sessions that action planning, offsite innovation, strategic and business planning along with working with a peer group of successful entrepreneurs or 'green-change leaders'. Accelerate Green also provides an 'invest and enable' funding strategy with an initial focus on established scaling climate tech companies, high-growth SMEs pivoting to climate change economy, earlier stage innovative-driven enterprises, renewables/carbon reduction, waste/circular, and AgTech.

Use of this systems building approach has also supported UNIVIV OneHealth/Emerald SME partners to map a business canvas that includes the economic feasibility for commercially producing high value protein

from duckweed (Fig. 3) harvested from the aquaculture ponds at Mount Lucas that includes scalability. Irish demand for plan protein in animal feeds is approximately 900,000 tons, where current protein rich native crops (peas and beans) provide approximately 52,000 tons on 10,000 ha. The National Protein Stakeholders Group aspires to increase this production to 120,000 on 20,000 ha. Notably, 3000 ha of rewetted peatland ponds will produce 5000 tons of protein concentrate with zero artificial inputs including fertiliser and pesticide. Some 5000 tons of duckweed protein concentrate would displace 50% of imported protein concentrate Irish organic salmon industry. Thirty thousand hectares of duckweed on marginal lands would produce 50,000 tons of 65% protein concentrate that is approximately equivalent to 20,000 ha of pea and bean protein at approximately 30% to 35% protein. The direct commercial value is in excess of €50 m; thus, potentially displacing €50 m of imports from an Irish feed production and security perspective. Thus, rewetting peatland is commercially feasible from a farmers' perspective, if duckweed is cultivated as an alternative sustainable crop. Topics that require attention include processing duckweed to produce animal feed, optimizing duckweed growth conditions, and identifying lands most suitable for rewetting and change of use. The Mount Lucas site is licensed for fish farming where the fish waste stream supports the linked commercial production of duckweed that can be potentially used in the animal feed industry (O'Neill et al., 2022). O'Neill et al. (2019) also reported that duckweed is important for waste aquaculture stream quality and recirculation at Mount Lucas peatland site where duckweed works in concert with algae and microbes for this purpose. De Beukelaar et al. (2018) reported that duckweed is considered to be a promising source of protein for human food products due to its high protein content and environmentally friendly production properties. Albeit not peatland focused, Bonomo et al. (1997) had previously described the use of a pilot duckweed as biological approach to produce reliable, simple and cost-effective small wastewater treatment system. In spite of the profitable characteristics of duckweed (high productivity, high protein content, wide geographic distribution, control of negative impacts from conventional wastewater treatment ponds), the results intimate that extensive use in Italy seems difficult due to the high requirement of land area and the ceasing of growth in winter months (at least in Northern Italy). However, the peatlands represents a large expansive area in many countries internationally, including Ireland. In temperate climates, a reasonable use of duckweed looks to be the production of good quality secondary effluents (BOD and SS removal) from small communities, especially in seasonal (summer) wastewater treatment plants. Duckweed is the smallest and fastest-growing aquatic plant, and has advantages including simple processing and the ability to grow high biomass in smaller areas. Therefore, duckweed could also be used as a new potential bioreactor for biological products such as vaccines, antibodies, pharmaceutical proteins, and industrial enzymes. Moreover, Yang et al. (2021a, 2021b) recently reported on plant bioreactors have flourished into an exciting area of synthetic biology due to their product safety, inexpensive production cost, and easy scale-up.

Currently, the commercial production of high value 'organic' cultured perch and trout at Mount Lucas is no likely to be commercially viable given the low fish biomass or tonnage on a 4 ha site, but this also provides an excellent innovation system for education, training and research given it's state-of-the-art IMTA system focus. Fish health/welfare and the linked microbiome profile in cultured pond water will also strongly contribute as indicators or tools for studying impact of climate change on sustainable food production systems (O'Neill and Rowan, 2022; O'Neill et al., 2022), where these activities will also support social enterprises (Rowan and Casey, 2021), and the development of digital technologies.

The EIDH smart specialisation strategy for structured support and development of peatland innovation includes provision for Entrepreneurial Discovery process using bottom up approach. Typically, this embraces five categories– (1) launching strategic initiatives; (2) re-entering existing programmes; (3) updating stakeholder strategic agencies; (4) aligning infrastructure; and (5) setting up strategic fora. Digitising Peatland activities to manage projects, networks, skills and training, tech transfer, consultancy and strategy; transnational connectivity and communication; profile and

### Table 3

Categories used to describe common functions of European Digital Innovation Hubs.

Source: https://europa.eu/!NX87WD

Category	Description
INTELLIGENCE	are intelligent systems used for decision making that both understand and adjust to specific circumstances; these are systems than can predict or plan to improve quality and to optimise capacity
CONNECTIVITY	the ability to access data in a secure and real-time manner; appropriate systems and machine will exchange data that may also be an integrated part of the business process.
FLEXIBIITY	the ability to adapt and customise systems and business processes to specific needs so that personalized products can be produced at affordable, mass-production prices?
AUTOMATION	can repetitive task be automated in a reliable way.
SUSTAINABILITY	are natural resources used in a sustainable manner, whereby not wasting fragile resources, also ensuring that no harm is done to the environment nor quality of life of citizens. SERVICES – are new sustaining or disruptive business models used where products are offered as a service.
SOCIAL	are workers motivated, engaged and empowered to carry out their work in an autonomous manner when working within the new systems.

embed regional actors that can promote diffusion of best practices and knowledge transfer; expand project opportunities; facilitate learning, experimentation and capacity building; manage workshops, SAPs and supporting specialist modules for PhDs; promote and foster ad-hoc alliances to enable open innovation; broker/match make academia with industry; accelerate entrepreneurial activities; align interests to local industries and opportunities for converging these; connecting with complementary EIDH for information flow and sharing on EU topics; support trusted partners that will provide subject-matter expertise and complementary services; develop successful business models for collaborations; an efficient channel to other EIHs, regional and markets that includes access to capacities, best practices and skills.

Mount Lucas has focused on development innovations to support primary production of freshwater fish and to improve ecological performance with future provision for secondary process and diversification towards reduced waste and improving shelf-life. Development of customized ICT solutions for production and supply chain management to improve safety, quality and awareness towards enabling transition to eco-friendly aquaculture and adjacent innovations. This will be achieved by digitally connecting the living labs with in field aquatic and adjacent innovation in the peatlands (Fig. 1). Mount Lucas will support multi-stakeholder engagement to stimulate creation of innovative and disruptive solutions for eco-friendly aquatic food systems. Met through generation, validation and application of new knowledge at both pilot-scale and commercially in order to sustainably improve performance by enabling knowledge and technology exchanges that considers Open innovation. Means to consolidate sustainability include establishing and deploying best practice, quality assurance and risk management that can be met by specialist training and infrastructure with partnering companies. Essential is the development of effective and appropriate ICT management and related innovative tools throughout the supply chain that will embrace traceability and safety.

Mount Lucas will rely upon digital tools to develop, optimise, automate and validate land-based IMTA-RAS (IMRAS) models that include high protein foods that have a low environmental footprint or impact. Peatlandfocused models to satisfy a circularity approach to produce high-value fish (such as Perch and Trout) along with low trophic plant species, such as duckweed. Monitor and control physicochemical parameters of culture ponds to reflect optimised ratio of microalgae and bacteria with duckweed (*Lemna minor*) to produce biomass of fish along with using this natural ecosystem to regulate waste and maintain water quality through recirculation. Develop sensors that will respond to key performance indicators in the pond to match aquaculture effluent and biomass productivity, along with commercial production of duckweed that feeds of aquaculture waste stream.

Integrated digital (intelligent) management system for IMTA that includes provision for multi-sensing technologies aligned with multifunctional management platforms (advanced monitoring, modelling, data analytics, and decision making); this system should also measure and model variances in climate and other disruptive fluxes to system through real-time system monitoring (microbiome, water physicochemical parameters, toxic microalgae) and modelling.

Schuelke-Leech (2021a, 2021b) recently described disruptive innovation and how it applies to the changes proposed for the Green New Deal era, which is also highly relevant to digital transformation of the peatlands. This author noted that achieving the ambitions of the Green New Deal will require social, economic, and industrial revolutions, rarely experienced in our history. However, climate change is going to force monumental changes; the latter has already decoupled the end-to-end value chain at Mount Lucas where O'Neill et al. (2022) recently reported that IMTA production for trout and perch was affected by extreme weather events that lead to fish mortality. O'Neill and co-workers (2022) also reported also noted that the emergence of different species of microalgae in this IMTA peatland process can be used as an early warning tool for disruptive climate variance events. Schuelke-Leech (2021a, 2021b) reported that the Green New Deal offers a vision for controlling and directing society's transformation through the development of new green technologies and public policies that mitigate and support the transformation to a sustainable and just society. The reader is directed to the model of Schuelke-Leech (2018) for an understanding of magnitudes of disruptive technologies and applicability. Rowan (2019) also provided examples of potential disruptive technologies for disease mitigation that can be applied to safeguard peatlands aquatic products. A summary of potential key emerging innovative

### Table 4

Digital transformation of the multi-actor ecosystem (Quadruple Helix Hub) along with commensurate infrastructure, supports and enterprise activities.

Concentrated single-access supports for industry, entrepreneurs, disruptors	Linked acceleration activities	Digitisation
Step-Change Physical Infrastructure & Systems Supports	R&D Collaborative Facilitation	End-to-end Edge computing AI, ARVR, QOE, blockchain, cyber-physical systems
Pre-start Ups	Design Maturation Activities	ARVR, QoE, AI.
Ideation & Design Thinking	Technical Maturation Activities	AI, machine learning, augment reality, IoT
Market Research -Early Needs Analysis -Product Market Fit Analysis	Financial Planning Legal Assistance Social and Digital Marketing (including informing customer behavioural change)	Blockchain Blockchain ARVR, IoT
Early Technical Validation -Test the Tech	Networking Dedicated Grant Writing/Reporting	Edge – IoT systems End-to-end monitoring, Edge computing (sensors, robotics), photonics.
-Experimentation/Validation in Pre-Pilot -Scaled to Real-Life Setting	Connection to Academic Staff/Expertise and Equipment to support commercialisation	Edge computing, ARVR (QoE).
"Living lab" – specialist equipment usage for data generation (real-time analysis)	Test beds in the environment (traditional slow analysis and hand-held devices)	Big data, machine learning AI, ARVR.
Conduit to State Financial Supports/Agencies	Enable Social Enterprise - Outreach Functions	Blockchain, cyper-physical systems, AI.
Specialist training	Technical training - ecotourism	ARVR/QoE; digital

activities, models, products and services arising from the convergence of biological with digital domains over the next decade is illustrated in Fig. 4; it noticeable that some of these may lead to disruptive innovation.

### 9. Conclusions

There is a strong trajectory towards development of wet peatland innovation for new green job creation where this novel aquatic ecosystem must balance economic, environmental and social sustainability. Peatlands represents an important carbon sink where it is commensurately imperative to protect the environment, ecology aligned with regulatory policies. Digital transformation has radically changed adjacent Agriculture 4.0 that has been mainly driven new business models focusing on local productivity. Green innovation and social enterprises across the peatlands value chain will be accelerated through digital transformation, which will be harnessed through EDIH model structure that is complementary to that of the quadruple helix hub (academic-industry-government-society) concept. This new paludiculture enterprise ecosystem will test-the-tech and digitally connect green-minded companies, stakeholders, enables and end-users. It is advocated that future strategic orientation of digital policies and plans will be aimed at ensuring and achieving international consensus and agreement on harmonization of processes. For example, digital tools for enabling future strategic development of this Irish Mount Lucas peatland demonstrate will include focusing on protecting biodiversity; monitoring and ensure sustaining carbon sink with net zero gas emissions; developing new aquatic green-innovations that includes high quality protein sources, fully



Fig. 3. Duckweed harvested from the IMTA peatland site at Mount Lucas in the Republic of Ireland.



Fig. 4. Summary of potential key emerging innovative activities emerging from peatlands enabled by digital transformation over next decade.

recirculated waste water without need for end-of-pipe disease mitigation treatments, development of renewable energies, bioprospecting of bioactives from waste streams, in field testing of sensors and robotics, smart development of vertical farming and exotic mushrooms, agro-forestry, and pollination and ecosystem service management. Peatlands innovation will be advanced through digital solutions that address process automation, data analytics and processing, control and management systems; moreover, these activities reflect those applied to inform the 4th technological revolution under Industry 4.0 along with alignment with the main tenets of Industry 5.0 human-centric model, and with many of the UN's sustainable development goals. Digital transformation of paludiculture products and services as described herein will yield the next generation of multidisciplinary-trained researchers equipped with the necessary crosscutting knowledge and skills to advance sustainability and climate action agendas for society with a global orientation. Funding instruments such as the European Just Transition, Interreg, and Horizon Europe will advance these opportunities by fostering and supporting key partnerships with stakeholders for the betterment of society.

### 9.1. Implications and recommendations

- There is much to be learnt by generating a deeper appreciation of the central role of key digital technologies supporting Agriculture 4.0 for unlocking adjacent and cross-cutting opportunities for the peatlands that will be met by more collaborate projects aligned with EIDH and Quadruple Helix Hub concepts. For example, this holistic approach should also focus on gaining a better understanding of the impacts of climate change for the intensive sustaining of peatland innovations at Mount Lucas (Ireland), which will potentially provide a useful blueprint to replicate across peatlands globally using an Open access knowledge-sharing model.
- There remains a lack of clear understanding as to what the concept of sustainability means for industries supporting paludiculture; therefore, it is envisaged that digital technologies will help meet this gap by way of promoting and implementing virtual training, management and creating greater awareness of these emerging eco-enterprises, products and

services for end-users. Examples include using AI, machine learning, immersive technologies (AR/VR)

- Over the past 5 years there has been an accelerated transition from 'Brown to Green' that reflects what has become the paradigm shift away from harvesting peat as a fossil fuel to developing sustainable ecoinnovations on the same peatlands. Therefore, there is a pressing need to understanding the roles of local government, industry and society in the management development of paludiculture innovations, which will inform new policies and job creation.
- It is important to create an awareness surrounding Peatlands innovations and to understanding how the digital transformation of such enterprises can powerful contributors to new business models and regional economies, which includes meeting several of the United Nations' sustainable development goals.
- Big data created from digital transformation of peatlands using drones will inform geo-referencing of sites with view to optimised usage balanced with protecting environment and biodiversity including ecosystem service management.
- There is a trend to develop new regional digital innovation hubs to meet community transitions to low carbon economy; however, there is a pressing need to provide clear guidance on the roles of each of the separate 706 DIHs across Europe so as to optimise cross-cutting usage by stakeholders that includes provision for integration trans-regionally.
- There are growing opportunities to support the swarm of new green startups companies that will be enabled through digitised living labs connected to environment test beds that will include AI/ML informed process automation, blockchain and cyber-physical systems for transparency, safety and security, IoT Cloud based approach to optimise sensors, drones and satellites for carbon sequestration and development of aquatic innovations ranging from aquaculture to vertical farming.
- Digital technologies will be enable Peatland enterprises to become more efficient through effective business models in real time; moreover, lessons learnt from Agriculture 4.0 suggests that key activities will include process automation and robotics, information systems and applications, cyber-physical systems, data analytics and logistics.
- · Digital technologies will unlock societal challenges for rural and poor

regions as attested by the Irish peatland – midlands that will inform a fair and just transition across Europe.

- Digital technologies will help develop green innovation that will contribute to strategic management in order to mitigate against disruption in supply chains; this topic including mitigating against emerging risks including climate change, global conflict, and pandemics.
- Digital tools will help us appreciate key data outputs across several environmental test bed, such as use of LCA, PCA, MLF for generating data to help inform societal, political and technological readiness levels (or maturity) of green innovation across the aquatic peatlands ecosystem. Digital technologies will help define international standards for peatland innovation balanced with environmental protection including new policies, plans and licencing.
- Digital transformation can expedite the transition from 'brown to green'; for example, the trend away from peat harvesting as a fossil fuel to supporting and enabling the development of sustainable ecoinnovations such as e-aquaculture, aquaponics, vertical farming, agroforestry and social enterprises such as honey production, exotic mushroom cultivation, pollination and ecosystem service management, eco- and darksky tourism.
- There is a need to reach an agreed consensus from stakeholders on the implementation of appropriate digital strategies to develop the peatlands including use of georeferencing, drones and satellites to map the peatlands matched with physiochemical parameters, carbon sink and biodiversity.
- Digital transformation to wettable peatlands products and services will also inform a new generation of researcher and promote job creation for rural areas that were strongly relient-upon peat harvesting as a means of producing fossil fuels for their livelihoods that disturbed carbon sink. Notably, the adjacent Agriculture and food sector are responsible for 19.29% of global gas emissions.
- Use of blockchain will help local farmers to directly reach customers in a secure and efficient manner.
- AI, virtual reality and augmented reality will enable a quality of experience for specialist training in the peatlands living laboratories (wet-labs, ecotoxicology, biodiversity) and provide a connected experience to the suite of environmental test beds that will showcase new eco-innovation, products and services; such as through Erasmsus +, MSCA RISE programmes.
- Digital transformation of peatland research and enterprise will inform new specific polices met through Horizon Europe, European Just Transition, European Green Deal, UN Sustainable Development Goals and so forth
- Digital transformation of peatlands will inform a fair and just transition of communities that are pivoting to low carbon economies along with enabling and accelerating green innovation.

### CRediT authorship contribution statement

N. J. Rowan, Deborah Power: Conceptualization. E. A. O'Neill, Eoghan Clifford, Niall Murray, N.J. Rowan: Data Curation: N. J. Rowan, D. Barcelo, E. Clifford, Y. Qiao, D.M. Power, E. A. O'Neill: Data analysis; N. J. Rowan: Funding Acquisition; E.A. O'Neill, E. Clifford, N. J. Rowan; Investigation; N. Rowan; N. Murray, Y. Qiao, E. Clifford, D. Barcelo: methodology; E. A. O'Neill: Project Administration; N. Rowan, E. Clifford; D. Power: Resources; N. Murray, Y. Qiao; software; E.O'Neill, Y. Qiao, N. Rowan, N. Murray: validation; N. Rowan, N. Murray, Y. Qioa, D. Barcelo, E. Clifford, D. Power, E. O'Neill: roles/writing – original draft. N. Rowan, N. Murray, Y. Qioa, D. Barcelo, E. Clifford, D.M. Power, E.O'Neill: Writing – Review and Editing.

### Declaration of competing interest

The authors declare that there are no competing interests or conflicts of interest with respect to the publication of this article.

### Acknowledgments

The author acknowledges provision of aerial image of Mount Lucas peatland site furnished by Bord Iascaigh Mhara (BIM). The authors would like to thank Interreg Atlantic Area Neptunus (Project EAPA\_576/2018), Regional University Network European University (RUN-EU Project), and Bord Iascaigh Mhara (Project 2019 BIM-KGS-008) for funding support. We thank the SME, UNIVIV OneHealth Solutions, for providing business model information informing the commercial development of their their Green paludiculture innovation.

### References

- Ab Rahman, A., Hamid, U.Z.A., Chin, T.A., 2017. Emerging technologies with disruptive effects. A review. Perintis eJ, ournal 7, 111–128.
- Abualtaher, M., Bar, E.S., 2019. Review of applying material flow analysis-based studies for a sustainable Norwegian Salmon aquaculture industry. J. Appl. Aquac. 32 (1), 1–15.
- Aich, N., et al., 2020. A reveiw on recirculating aquaculture systems: Challenges and opportunities for sustainable aquaculture. Innovative Farming 5, 17–24.
- Alonso, R.S., Sitton-Candanedo, I., Garcia, O., Prieto, J., Rodrigueq-Gonsalez, S., 2020. An intelligent Edge-IoT platform for monitoring livestock and crops in a diary-farming scenario. Ad Hoc Netw. 98, 102047.
- Anderson, J.L., Asche, F., Garlock, T., 2019. Economics of aquaculture policy and regulation. Annu. Rev. Resour. Econ. 11, 101–123.
- Antononucci, F., Costa, C., 2020. Precision aquaculture: a short review on engineering innovations. Aquac. Int. 28, 41–57.
- Appio, F.P., Frattini, F., Petruzzelli, A.M., Nerotti, P., 2021. Digital transformation and innovation management: a synthesis of existing research and an agenda for future studies. J. Prod. Innov. Manag. 38, 4–20.
- Arora, S., Saraswat, S., 2021. Vermifiltration as a natural, sustainable and green technology for environmental remediation: a new paradigm for wastewater treatment process. Curr.Res.Green Sustain.Chem. 4. https://doi.org/10.1016/j.crgsc.2021.100061.
- Arvanitis, K.G., Symenonaki, E.G., 2020. Agriculture 4.0 the role of innovative smart technologies towards sustainable farm management. Open Agric. J. 14, 130–135.
- Barrett, M., et al., 2016. Detection, fate and inactivation of pathogenic norovirus employing settlement and UV treatment in wastewater treatment facilities. Sci. Total Environ. 568, 1026–1036. https://doi.org/10.1016/j.scitotenv.2016.06.067.
- Basil, J., Macedo, M., Lago, C., Oliveira, T., Junior, M., Oliveira, T., Mendiondo, E., 2021. Nature-based solutions and real-time control: challenges and opportunities. Water 13 (5), 651. https://doi.org/10.3390/w13050651.
- Bonomo, L., Pastorelli, G., Zambon, N., 1997. Advantages and limitations of duckweed-based water treatment systems. Water Sci. Technol. 35 (5), 239–246.
- Bord Na Móna, 2022. Accelerate Green –Irelands firsh accelerator dedicated to scaling companies leading the response to climate action and sustainability by developing products and services based on green innovation. https://accelerategreen.ie/. (Accessed 24 May 2022).
- Bowes, B.D., Tavakoli, A., Wang, C., Heydarian, A., Behl, M., Beling, P.A., Goodall, J.L., 2021. Flood mitigation in coastal urban catchments using real-time stormwater infrastructure control and reinforcement learning. J. Hydroinf. 23 (3), 529–547.
- Braga Rodrigues, T., O'Cathain, C., O'Connor, E., Murray, N., 2020. A quality of experience assessment of haptic and augmented reality feedback modalities in a gait analysis system. PLOS ONE https://doi.org/10.1371/journal.pone.0230570.
- Bumbac, Costel, Clifford, Eoghan, Dussaussois, Jean-Baptiste, Schaal, Alexandre, Tompkins, David (Eds.), 2021. Innovative Wastewater Treatment Technologies — The INNOQUA Project. now publishers, Boston-Delft https://doi.org/10.1561/9781680837810.
- Chawla, I., Karthikeyan, L., Mishra, A., 2020. A review of remote sensing applications for water security: quantity, quality, and extremes. J. Hydrol. 585. https://doi.org/10. 1016/j.jhydrol.2020.124826.
- Chen, T., 2010. On reducing false alarms in multivariate statistical process control. Chem. Eng. Res. Des. 88 (4), 430–436.
- Chen, Z.Y., Zhang, W.S., 2017. Construction and implementation of 3D aquaculture virtual simulation experiment systems. Mod.Comput. 16, 2017.
- Clifford, E., Mulligan, S., Comer, J., Hannon, L., 2017. Flow-signature analysis of water consumption in nonresidential building water networks using high-resolution and mediumresolution smart meter data: two case studies. Water Resour. Res. 54 (1), 88–106.
- Connolly, A., 2018. Eight digital technologies disrupting aquaculture. https://www. globalseafood.org/advocate/eight-digital-technologies-disrupting-aquaculture/ (accessed 06/04/2022).
- Cooney, R., Tahar, A., Kennedy, A., Clifford, E., 2021. The dilemma of opportunity in developing a life cycle assessment of emerging aquaculture systems – a case study of a Eurasian perch (Perca fluviatilis) hatchery recirculating aquaculture system. Aquaculture 536. https://doi.org/10.1016/j.aquaculture.2021.736403.
- Corominas, L.I., et al., 2018. Transforming data into knowledge for improved wastewater treatment operation: A critical review of techniques. Environmental Monitoring and Software 106, 89–103. https://doi.org/10.1016/j.envsoft.2017.11.023.
- De Beukelaar, M.F.A., Zeinstra, G.G., Mes, J.J., Fischer, A.R.H., 2018. Duckweed as human food. The influence of meal context and information on duckweed acceptability of Dutch consumers. Food Qual. Prefer. 71, 76–86.
- Domegan, C., 2021. Social marketing and behavioural change in a systems setting. Curr.Opin. Environ.Sci.Health https://doi.org/10.1016/j.coesh.2021.100275.
- Dominiguez, M.A., Centurion, M.E., 2015. Application of digital images to determine color in honey samples from Argentina. Microchem. J. 118, 110–114.

Ducket, T., Pearson, S., Blackmore, S., Grieve, B., Chen, W., Cielniak, G., Yan, G.V., 2018. Agriculatural Robotics: The Future of Robotic Agriculture arXiv preprint arViv: 1806.06762.

- Egan, D., Brennan, S., Barrett, J.Qiao, Timmerer, Y.C., Murray, N., 2016. An evaluation of Heart Rate and ElectroDermal Activity as an objective QoE evaluation method for immersive virtual reality environments. Eighth International Conference on Quality of Multimedia Experience (QoMEX), 2016, pp. 1–6 https://doi.org/10.1109/QoMEX. 2016.7498964.
- Escriche, I., Kadar, M., Domenech, E., Gil-Sanchez, L., 2011. A potentiometric electronic tongue for the discrimination of honey according to the botanical origin. Comparison with traditional methodologies: physicochemical parameters and volatile profile. J. Food Eng. 109, 449–456.
- Evensen, T., 2020. Fishy business: closing the gap between data-driven decision-making DDM) and aquaculture. An analysis of incumbents in the Norwegian aquaculture industry (NAI) and the use of big data for competitive advantage (accessed 06/04/2022).
- FAO, 2020. The State of Word Fisheries and Aquaculture 2020. Sustainability in Action, Italy. Farrelly, D.J., Everard, C.D., Fagan, C.C., McDonnell, K.P., 2013. Carbon sequestration and the role of biological carbon mitigation: a review. Renew. Sust. Energ. Rev. 21, 712–727. https://doi.org/10.1016/j.rser.2012.12.038.
- Felix, F.S., Angnes, L., 2018. Electrochemical immunsensors a powerful tool for analytical applications. Biosens. Bioelectron. 102, 470–478.
- Fleming, K., Waweru, P., Wambua, M., Ondula, E., Samuel, L., 2016. Toward quantified smallscale farms in Africa. IEEE Internet Comput. 20, 63–67.
- Food Standards Agency UK, 2021. Emerging technologies that will impact on the UK Food System – Rapid Evidence Assessment – May 2021. https://www.food.gov.uk/sites/ default/files/media/document/emerging-technologies-report.pdf (assessed 19 May, 2022).
- Fox, S., McDermott, J., Doherty, E., Cooney, R., Clifford, E., 2022. Application of neural networks and regression modelling to enable environmental regulatory compliance and energy optimisation in a sequencing batch reactor. Sustainability 14 (7), 4098. https://doi. org/10.3390/su14074098.
- Galanakis, C.M., Rizou, M., Aldawoud, T.M.S., Ucak, I., Rowan, N.J., 2021. Innovations and technology disruptions in the food sector within the COVID-19 pandemic and postlockdown era. Trends Food Sci. Technol. 110, 193–200.
- Garvey, M., et al., 2015. Ecotoxicological assessment of pulsed ultraviolet light-treated water containing microbial species and *Cryptosporidium parvum* using a microbiotest test battery. Water Environ. J. 29 (1), 27–35. https://doi.org/10.1111/wej.12073.
- Gjedrem, T., Robinson, N., 2014. Advances by selective breeding for aquaculture species a review. Agric. Sci. 5, 733–743.
- Goblirsch, M., Eakins, J., Rowan, N.J., 2021. Disease-mitigating innovations for the pollination service industry: challenges and opportunities. Curr.Opin.Environ.Sci.Health 22. https://doi.org/10.1016/j.coesh.2021.100265.
- Gratacap, R.L., et al., 2019. Potential of genoe ediingto improve aquaculure breeding and production. Trends Genet. 35, 672–684.
- Guo, Y., Yuan, Z., Xu, J., Wang, Z., Yuan, T., Zhou, W., Xu, J., Liang, C., Xu, H., Liu, S., 2017. Metabolic acclimation mechanism in microalgae developed for CO2 capture from industrial flue gas. Algal Res. 26, 225–233. https://doi.org/10.1016/j.algal.2017.07.029.
- Hashim, H., Ryan, P., Clifford, E., 2020. A statistically based fault detection and diagnosis approach for non-residential building water distribution systems. Adv. Eng. Inform. 46. https://doi.org/10.1016/j.aei.2020.101187.
- Hassan, S., De Filippi, P., 2021. Decentralized autonomous organization. InternetPolicy Review 10 (2), 1–10.
- Houston, R.D., et al., 2020. Harnessing genomics to fast-track genetic improvement in aquaculture. Nat. Rev. Genet. 21, 389–409.
- Hrustek, L., 2020. Sustainability driven by agriculture through digital transformation. Sustainability 12. https://doi.org/10.3390/su12208596.
- Hylton, E., Noad, L., Templeton, M.R., Sule, M.N., 2022. The rate of vermi-compost accumulation within 'Tiger Toilets' in India. Environ. Technol. 43 (3), 376–385.
- Intergovermental Panel on Climate Change (IPCC), 2022. Working Group III contribution, Climate Change 2022: Mitigation of Climate Change.
- Josthiswaran, V., Velumani, T., Jayaraman, R., 2020. Application of artificial intelligence in fisheries and aquaculture. Biotica Res.Today 2, 499–502.
- Kelly, A.M., Renekdas, N.N., 2020. Disease managemeth of aquatic animals, aquaculture, health management. Elsevier.
- Klerkz, L., Jakku, E., Labarthe, P., 2019. A review of social science on digital agriculture, smart farming and agriculture 4.0: new contributions and a future research agenda. NJAS Wangening. J. Life Sci. 90-91. https://doi.org/10.1016/j.njas.2019.100315.
- Laso, J, Ruiz-Salmon, I, et al., 2022. Achieving Sustainability of the Seafood Sector in the European Atlantic Area by Addressing Eco-Social Challenges: The NEPTUNUS Project. Sustainability 14 (5). https://doi.org/10.3390/su14053054.
- Lee, D.G., et al., 2013. Development of a vision-based automtic vaccine injecton system for flatfish. Aquac. Eng. 54, 78–84.
- Lopez-Pacheco, I.Y., Rodas-Zuluaga, L.I., Fuentes-Tristan, S., Castillo-Zacarias, C., Sosa-Hernandez, J.E., Barcelo, D., Iqbal, H.M.N., Parra-Saldivar, R., 2021. Phyocapture of CO2 as an option to reduce greenhouse gases in cities: carbon sinks in urban spaces. J. CO2 Utiliz. 53. https://doi.org/10.1016/j.jcou.2021.101704.
- Lucas, J.S., Southgate, P.C., Tucker, C.S., 2019. Aquaculture Farming Aquatic Animals and Plants. John Wiley & Sons.
- Massaro, A., Savino, N., Galiano, A., 2020. Agri-photonics in precision agriculture. 22nd International Conference on Transparent Optical Networks (ICTON), pp. 1–4 https://doi.org/ 10.1109/ICTON51198.2020.920377.
- Masterson, C.H., et al., 2020. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant *Klebsiella pneumoniae*-induced pulmonary sepsis. Lett. Appl. Microbiol. https://doi.org/10.1111/lam.13358.
- Mondejar, M.E., Avtar, R., Diaz, H.L.B., Dubey, R.K., Esteban, J., Gomez-Morales, A., Hallam, B., Mbungu, N.T., Okolo, C.C., Rrasad, K.A., She, W., Garcia-Segura, S., 2021.

Digitalization to achieve sustainable development goals: steps towards a smarter Green Plant. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2021.148539.

- Mulholland, B., Abdel-Aziz, I., Lindsay, R., McNamara, N., Keith, A., Page, S., Clough, J., Freeman, B., Evans, C., 2020. An Assessment of the Potential for Paludiculture in England and Wales. Report to Defra for Project SP1218 98 pp.
- Mulligan, S., Hannon, L., Ryan, P., Nair, S., Clifford, E., 2021. Development of a data driven FDD approach for building water networks: water distribution system performance assessment rules. J.Build.Eng. 34. https://doi.org/10.1016/j.jobe.2020.101773.
- Murphy, E.J., et al., 2020a. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects — Implications for coronavirus disease (COVID-19) immunotherapies. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.139330.
- Murphy, E.J., et al., 2020b. β-Glucan metabolic and immunomodulatory properties and potential for clinical application. J. Fungi https://doi.org/10.3390/jof6040356.
- Murphy, E.J., et al., 2022. Immunomodulatory activity of β-glucan polysaccharides isolated from different species of mushroom – A potential treatment for inflammatory lung conditions. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2021.152177.
- Musiolik, J., Markard, J., Hekkert, M.P., 2012. Networks and network resources in technological innovation systems: towards a conceptual framework for system building. Technol. Forecast. Soc. Chang. 79, 1032–1048.
- Nahavandi, S., 2019. Industry 5.0 a human-centric solution. Sustainability 1 (16), 4371. https://doi.org/10.3390/su11164371.
- Nair, R.S., Domnic, S., 2022. A combination of learning and non-learning based method for enhancement, compression and construction of underwater images. Aquac. Fish, 2022.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: a case study from the Republic of Ireland. Aquaculture 526. https://doi.org/10.1016/j.aquaculture.2020.735377.
- Newhart, K.B., Holloway, R.W., Hering, A.S., Cath, T.Y., 2019. Data-driven performance analyses of wastewater treatment plants : a review. Water Res. 157, 498–513.
- O'Neill, E.A., Rowan, N.J., 2022. Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wasterwater quality including provision for assessing impact of extremes in climate variance a comparative case study from the Republic of Ireland. Sci. Total Environ. 802. https://doi.org/10.1016/j.scitotenv.2021. 149800.
- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. Sci. Total Environ. 706. https://doi.org/10.1016/j. scitotenv.2019.136044.
- O'Neill, E.A., Fogarty, A., Rowan, N.J., 2019. Novel use of the alga Pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218. https://doi.org/10.1016/j.scitotenv.2019.07.243.
- O'Neill, E.A., Morse, A.P., Rowan, N.J., 2022. Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: implications and opportunities for advancing research and disruptive innovation post COVID-19 era. Sci. Total Environ. 819. https://doi.org/10.1016/j. scitotenv.2022.153073.
- O'Sullivan, K., 2022. Climate change report: it's now or never if we want to limit global warming to 1.5 degrees. https://www.irishtimes.com/news/environment/climatechange-report-it-s-now-or-never-if-we-want-to-limit-global-warming-to-1-5-degrees-1.4844226 (accessed 06/04/2022).
- Ohrem, S.J., Kelasidi, E., Bloecher, N., 2020. Analysis of a novel autonomous underwater robot for biofouling prevention and inspection in fish farms. 2020, 28th Mediterranean Conference on Control and Automation (MED). IEEE, pp. 1002–1008.
- Okolo, C.C., Gebresamuel, G., Retta, A.N., Zenebe, A., Haile, M., 2019. Advances in quantifying soil organic carbon under different land uses in Ethiopia: a review and synthesis. Bull. Natl. Res. Cent. 43, 99. https://doi.org/10.1186/s42269-019-0120-z.
- Osaka, T., Yakuskiji, H., Hirata, D., 2010. Underwater cleaning robot and auxiliary cleaning work machine. Google Patent https://nam11.safelinks.protection.outlook.com/? url = https%3A%2F%2Fpatents.google.com%2Fpatent%2FCN101695832B%2Fen& data = 05%7C01%7Cp.das%40elsevier.com%7Ccf8ceb5ef9074e4cb18b08da4551f560% 7C9274ee3f94254109a27f9fb15c10675d%7C0%7CC%7C637898515580904044%7C Unknown%7CTWFpbGZsb3d8eyJWIjoiMC4wLjAwMDAiLCJQIjoiV2luMzIiLCJBTiI6I k1haWwiLCJXVCI6Mn0%3D%7C3000%7C%7C%amp;sdata = RcFqwXCvikq6%2B omH6rBNtXY4NgVfF0Ij4A2PKDBFU2I%3D&reserved = 0. (Accessed 3 June 2022).
- Paspalakisk, S., Moirogiorou, K., Papandroulakis, N., Giakos, G., Zervakis, M., 2020. Automated fish cage net inspection using image processing techniques. IET Image Process. 14, 2028-2-34.
- Perez-Pons, M., Plaza-Hernandez, M., Alonso, R., Parra-Dominiquez, J., Prieto, J., 2021. Increasing profitability and monitoring environmental performance: a case study in the agri-food industry through an edge-IoT platform. Sustainability 13, 283. https://doi. org/10.3390/su13010283.
- Piedra-Munoz, L., Galdeano-Gomez, E., Perez-Mesa, J.C., 2016. Is sustainability compatible with profitability? An empirical analysis on family farming activity. Sustainability 8, 893.
- Planko, J., Cramer, J.M., Chappin, M.H.H., Hekkert, M.P., 2015. Strategic collective system building to commercialize sustainability innovations. J. Clean. Prod. 112. https://doi. org/10.1016/j.jclepro.2015.09.108.
- Planko, J., Cramer, J.M., Chappin, M.M.H., Hekkert, M.P., 2016. Strategic collective system building to commercialize sustainability. J. Clean. Prod. 112, 2328–2341.
- Pogue, R., Murphy, E.J., Fehrenbach, G.W., Rezoagli, E., Rowan, N.J., 2021. Exploiting immunomodulatory properties of β-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms: Concise review of existing knowledge, innovation and future opportunities. Curr. Opin. Environ. Sci. Health 21. https:// doi.org/10.1016/j.coesh.2021.100248.

- Pous, N., Barcelona, A., Sbardella, L., Higdalgo, M., Colomer, J., Serra, T., Salvado, V., 2021. Zooplankton-based reactors for tertiary wastewater treatment – a pilot scale case study. J. Environ. Manag. 278 (1). https://doi.org/10.1016/j.jenvman.2020.111538.
- Ramaraj, R., Tsai, D.D.-W., Chen, P.H., 2014. Freshwater microalgae niche of air carbon dioxide mitigation. Ecol. Eng. 68 (2014), 47–52. https://doi.org/10.1016/j.ecoleng.2014.03. 058.
- Rasmussen, L.V., Coosaet, B., Martin, A., Mertz, O., Pascual, U., Corbera, E., Dawson, N., Fisher, J.A., Franks, P., Ryan, C.M., 2018. Socio-ecological outcomes of agricultural intensification. Nat.Sustain. 1, 275–282.
- Robb, S., 2021. Irish company ApisProtect launches bee monitoring tech in US. https://www. farmersjournal.ie/irish-company-apisprotect-launches-bee-monitoring-tech-in-us-597351 (accessed 06/04/2022).
- Rowan, N.J., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries – Quo Vadis. Trends Food Sci. Technol. 88, 316–332.
- Rowan, N., Casey, O., 2021. Empower Eco multiactor HUB: A triple helix 'academia-industryauthority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. Curr. Opin. Environ. Sci. 21. https://doi.org/10.1016/j. coesh.2021.100254.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations – Quo Vadis? Sci. Total Environ. 748, 141362.
- Rowan, N.J., Laffey, J.G., 2020. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from coronavirus disease (COVID-19) pandemic –case study from the Republic of Ireland. Sci. Total Environ. 725, 138532.
- Rowan, N.J., Laffey, J.G., 2020b. Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from coronavirus disease (COVID-19) pandemic – implications for efficacy, re-use and sustainable waste management. Sci. Total Environ. 752. https://doi.org/10.1016/j.scitotenv.2020.142259.
- Rowan, N.J., Pogue, R., 2021. Editorial overview: green new deal era current challenges and emerging opportunities for developing sustaining and disruptive innovation. Curr. Opin. Environ.Sci.Health https://doi.org/10.1016/j.coesh.2021.100294.
- Ruiz-Salmón, I., Margallo, M., Laso, J., Villaneuva-Rey, P., Marino, D., Quinteiro, P., Diaz, A.C., Nunes, M.L., Marquez, A., Feijoo, G., Moreira, M.T., Lobet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N., Mendez-Paz, D., Aldaco, R., 2020. Eurpean seafood sector under a circular economy framework. Curr. Opin. Environ. Sci. Health 13, 101–106.
- Ruiz-Salmon, I., Laso, J., Margallo, M., Villanueva-Rey, P., Rodriguez, E., Quinteiro, P., Dias, A.C., Almeida, C., Nunes, M.L., Marquez, A., Cortes, A., Moreira, M.T., Feijoo, G., Loubet, P., Sonnemaan, G., Morse, A.P., Cooney, R., Clifford, E., Regueiro, L., Mendez, D., Anglada, C., Noirot, C., Rowan, N., Vazquez-Rowe, I., Aldaco, R., 2021. Life cycle assessment of fish and seafood processed products – a review of methodologies and new challenges. Sci. Total Environ. https://doi.org/10. 1016/j.scitotenv.2020.144094.
- Sargent, N., 2022. Peat pressure: industrial peat extraction in enforcement firing line. https:// www.thejournal.ie/peats-sake-pt9-unregulated-extraction-5731111-Apr2022/ accessed 7 April 2022.
- Savastano, M., Amendola, C., D'Ascenzo, F., 2018. How digital transformation is reshaping the manufacturing industry value chain: the new digital manufacturing ecosystem applied to a case study from the food industry. Netw. Smart Open 2. https://doi.org/10. 1007/978-3-319-62636-9\_9.
- Schuelke-Leech, B.A., 2018. A model for understanding the orders of magnitude of disruptive technologies. Technol. Forecast. Soc. Chang. 129, 261–274.
- Schuelke-Leech, B.A., 2021. Disruptive technologies for a Green New Deal. Curr.Opin.Environ.Sci.Health https://doi.org/10.1016/j.coesh.2021.100245.
- Schuelke-Leech, B.A., 2021. Disruptive technologies for a Green New Deal. Curr.Opin.Environ.Sci.Health https://doi.org/10.1016/j.coesh.2021.100245.
- Science Foundation Ireland, 2021. Microsoft Ireland and SFI Launch €5m Climate Change Research Project: Terrain-AI -Ambition to share insights and models with other countries to help reduce global carbon outputs. https://www.sfi.ie/research-news/news/microsoftsfi-terrain-ai/. (Accessed 15 December 2021).
- Seyoum, S., Alfonso, L., van Andel, S.J., Koole, W., Groenewegen, A., van de Giesen, N., 2017. A shazam-like household water leakage detection method. Procedia Eng. 186, 452–459.
- Shafiee, S., Minaei, S., Moghaddam-Charkari, N., Ghasemi-Vernamkhasti, M., Barzegar, M., 2013. Potential application of machine vision to honey characterization. Trends Food Sci. Technol. 30, 174–177.
- Shamin, A., Frolova, O., Makarychev, V., Yashkova, N., Kornilova, L., Akimov, A., 2019. Digital transformation of agricultural industry. IOP Conf. Series Earth Environ. Sci. 346, 012029.
- Shefat, S.H.T., 2018. Vaccines for use in finfish aquaculture. Aceta Scientific Pharmaceutical Sciences 19.
- Shen, Y., Ma, K., Yue, G.H., 2020. Status, challenges and trends of aquaculture in Singapore. Aquaculture.
- Skobelev, P.O., Borovik, S.Y., 2017. On the way from industry 4.0 to Industry 5.0: from digital manufacturing to digital society. Int. Sci. J. "Industry 4.0" 6, 307–311.
- Skobelev, P.O., Borovik, S.Y., 2017. On the way from industry 4.0 to Industry 5.0: from digital manufacturing to digital society. Int. Sci. J. "Industry 4.0" 6, 307–311.
- Sousa, D., Sargento, S., Perira, A., Luis, M., 2019. Self-adaptive team of aquatic drones with a communication network for aquaculture. EPIA Conference on Artificial Intelligence. Springer, pp. 569–580.
- Su, X, Sutarlie, I., Xie, Y., 2020. Sensors, biosensors, and anallyical echnologies for aquaculure ater quality. Research Ideas for Todays Investors https://doi.org/10.34133/2020/ 8272705.
- Sun, M., Yang, X., Xie, Y., 2020. Deep learning in aquaculture: a review. J. Comput. 31 (1), 294–319.

- Sundui, B., Calderon, O.A.R., Adbdeldayem, O.M., Lazaro-Gil, J., Rene, E.R., Sambuu, U., 2021. Applications of machine learning algorithms for biological wastewater treatment : updates and perspectives. Clean Techn. Environ. Policy 23, 127–143.
- Svendsen, E., Fore, M., Okland, F., Grans, A., Hedger, R.D., et al., 2020. Heart rate and swimming activity as stress indicators for Atlantic salmon. Aquaculture 531, 735804.
- Tabatabaei, M., Tohidfar, M., Jouzani, G.S., Safarnejad, M., Pazouki, M., 2011. Biodiesel production from genetically engineered microalgae: future of bioenergy in Iran. Renew. Sust. Energ. Rev. 15, 1918–1927. https://doi.org/10.1016/j.rser.2010.12.004.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. Peer J https://doi.org/10.7717/peerj.5281.
- Tahar, A., Kennedy, A., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3 (3), 28. https://doi.org/ 10.3390/fishes3030028 2018.
- Tahar, A., Tiedeken, E.J., Rowan, N.J., 2018c. Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – a first overview of the situation in the Republic of Ireland. Sci. Total Environ. 616–617, 187–197.
- Tacon, A.G., Metian, M., 2015. Feed matters: Satisfying the feed demand of aquaculture. Rev. Fish. Sci. Aquac. 1–10.
- Tan, Z.D., Lupascu, M., Mijedasa, L.S., 2021. Paludiculture as a sustainable land use alternative for tropical peatlands.
- Taufek, N.M., 2020. Performance of mycelial biomass and exopolysaccharide from Malaysian Ganoderma lucidum for the fungivore red hybrid Tilapia (Oreochromis sp.) in Zebrafish embryo. Aquac. Rep. https://doi.org/10.1016/j.aqrep.2020.100322.
- Therrien, J.-D., Nicolai, N., Vanrolleghem, P.A., 2020. A critical review of the data pipeline: how wastewater system operation flows from data to intelligence. Water Sci. Technol. 82 (12), 2613–2634.
- Tsolakis, N., Aivazidou, E., Srai, J., 2019. Sensor applications in agrifood systems: current trends and opportunities for water stewardship. Climate 7, 44.
- Uràk, I., Hartel, T., Gallé, R., Balog, A., 2017. Worldwide peatland degradation and the related carbon dioxide emissions: the importance of policy regulations. Environ. Sci. Pol. 69, 57–64.
- Usuldin, S.R.A., et al., 2021. In vivo toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications. Sci. Rep. 11. https://doi.org/10.1038/s41598-021-02486-7.
- Vindigni, G., Pulvirenti, A., Alaimo, S., Monaco, C., Spina, D., Peri, I., 2021. Bioinformatics approach to mitigate mislabelling in EU seafood market and protect consumer health. Int. J. Environ. Res. Public Health 18 (4). https://doi.org/10.3390/ijerph18147497.
- Vroom, R.J.E., Xie, F., Geurts, J.J.M., Chojnowska, A., Smolders, A.J.P., Lamers, L.P.M., Fritz, C., 2018. Typha latifolia paludiculture efectively improves water quality and reduces greenhouse gas emissions in rewetted peatlands. Ecol. Eng. 124, 88–98.
- Wang, C., Li, Z., Wang, T., Xu, X., Zhang, X., Li, D., 2021. Intelligent fish farm—the future of aquaculture. Aquac. Int. 29 (6), 2681–2711.
- Wan-Mohtar, W.A.A.Q.I., et al., 2021. Use of Zebrafish Embryo Assay to Evaluate Toxicity and Safety of Bioreactor-Grown Exopolysaccharides and Endopolysaccharides from European Ganoderma applanatum Mycelium for Future Aquaculture Applications. Int. J. Mol. Sci. 22 (4). https://doi.org/10.3390/ijms22041675.
- Weiskopt R, Sarah, Rubenstein A, Madeleine, Crozier C, Lisa, Gaichas, Sarah, Griffis, Roger, Halofsky E, Jessica, Hyde J.W, Kimberly, et al., 2020. Climate change effects on biodiversity, ecosystems, ecosystem services, and natural resource management in the United States. Sci. Total Environ. 733. https://doi.org/10.1016/j.scitotenv. 2020.137782.
- Weiss, M., Jacob, F., Duveiller, G., 2020. Remote sensing for agricultural applications: a metareview. Remote Sens. Environ. 236. https://doi.org/10.1016/j.rse.2019.111402.
- Wichmaan, W., Schroder, C., Joosten, H. (Eds.), 2016. Paludiculture Productive Use of Wet Peatlands: Climate Protection – Biodiversity – Regional Economic Benefits. Schweizbart Science Publishers, Stuttgart.
- Wong, K.H., Tan, I.S., Foo, H.C.Y., Chin, L.M., Cheah, J.R., Sia, J.K., Lam, M.K., 2022. Third generation bioethanol and L-lactic acid production from red macroalgae cellulosic residue: prospects of Industry 5.) algae. Energy Convers. Manag. 253, 115155.
- Xia, J., Gomes, G.B., Yue, G.H., 2022. Editorial emerging and disruptive technologies in aquaculture. Aquac. Fish. 7, 109–110.
- Xing, Q., et al., 2019. Monitoring seaweed aquaculure in the Yellow Sea with multiple sensors for managing the disese of macroalgal blooms. Remote Sens. Environ. 231. https://doi. org/10.1016/j.rse.2019.111279.
- Xu, X., Lu, Y., Vogel-Heuser, B., Wang, L., 2021. Industry 4.0 and Industry 5.0 inception, conception and perception. J. Manuf. Syst. 61, 530–535.
- Yang, G.-L., Feng, D., Liu, Y.-T., Lv, S.-M., Zheng, M.M., Tan, A.-J., 2021. Research progress of a potential bioreactor: duckweed. Biomolecules 11 (1), 93. https://doi.org/10.1007/978-3-319-62636-9\_9.
- Yang, X., Zhang, S., Liu, J., Gao, Q., Dong, S., Zhou, C., 2021. Deep learning for smart fish farming: applications, opportunities and challenges. Rev. Aquac. 13 (1), 66–90.
- Yeong, T.J., Pin Jern, K., Yao, L.K., Hannan, M.A., Hoon, S.T.G., 2019. Applications of photonics in agriculture sector: a review. Molecules 27 (10). https://doi.org/10.33890/molecules24102025.
- Yoo, S.-H., Ju, Y.T., Kim, J.S., Kim, E.K., 2020. Design and development of underwater drone for fish farm growth environment managment. The Journal of the Korea Institute of Electronic Xi, M., Adcock, M., Mcculloch, J. (2019). An end-to-end augmented reality solution to support aquaculture farmers with data collection, storage and analysis. The 17th International Conference on Virtual Reality Contimuum and Its Application in Industry, pp. 1–2.

- Yu, W., Liang, F., He, X., Hatcher, W.G., Lu, C., Lin, J., Yang, X., 2017. A survey on the edge computing for the Internet of Things. IEEE Access 6, 6900–6919. Yue, K., Shen, Y., 2022. An overview of disruptive technologies for aquaculture. Aquac.Fish.
- 7, 111–120.
- Zeigler, R., 2020. Paludiculture as a critical sustainability innovation mission. Res. Policy 49. https://doi.org/10.1016/j.respol.2020.103979.
- Zhao, B., Su, Y., 2020. Macro assessment of microalgae-based CO2 sequestration: environmental and energy effects. Algal Res. 51. https://doi.org/10.1016/j.algal.2020.102066, 102066.
- Zhou, C., et al., 2019. Evaluation of fish feeding intensity in aquaculture using a convolutional neural nework and machine vision. Aquaculture 507, 457–465.
- Ziegler, R., Wichtmaan, W., Abel, S., Kemp, R., Simard, M., Joosten, H., 2021. Wet peatland utilisation for climate protection an international survey of paludiculture innovation. Clean.Eng.Technol. https://doi.org/10.1016/j.clet.2021.100305.
- SFI, 2021. Terrain-AI Uncovering new insights to support effective climate change decision making (terrainai.com). (Accessed 31 May 2022).



Contents lists available at ScienceDirect

Trends in Food Science & Technology





# Innovations and technology disruptions in the food sector within the COVID-19 pandemic and post-lockdown era

Charis M. Galanakis<sup>a,b,c,\*</sup>, Myrto Rizou<sup>a</sup>, Turki M.S. Aldawoud<sup>b</sup>, Ilknur Ucak<sup>d</sup>, Neil J. Rowan<sup>e,f</sup>

<sup>a</sup> Research & Innovation Department, Galanakis Laboratories, Chania, Greece

<sup>b</sup> College of Science, King Saud University, Riyadh, Saudi Arabia

<sup>c</sup> Food Waste Recovery Group, ISEKI Food Association, Vienna, Austria

<sup>d</sup> Nigde Omer Halisdemir University, Faculty of Agricultural Sciences and Technologies, Nigde, Turkey

<sup>e</sup> Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Ireland

<sup>f</sup> Centre for Disinfection, Sterilization and Biosecurity, Athlone Institute of Technology, Athlone, Ireland

ARTICLE INFO

Keywords: SARS-CoV-2 COVID-19 Food innovation Lockdown transitioning Disruptions Sustainability

### ABSTRACT

*Background:* COVID-19 pandemic has caused a global lockdown that has abruptly shut down core businesses and caused a worldwide recession. The forecast for a smooth transition for the agri-food and drink industry is, at best, alarming. Given that COVID-19 shutdown multiple core services (such as aviation, food services, supply chains, and export and import markets), there is an enormous deficiency in critical information to inform priority decision making for companies where this uncertainly is likely to impact negatively upon recovery.

*Scope and approach:* The current article investigates potential innovations within the era of the COVID-19 crisis after framing them within the four issues of the food sector (food safety, bioactive food compounds, food security, and sustainability) that are directly affected by the pandemic. The prospect of foreseen innovations to disrupt the food sector during lockdown periods and the post-COVID-19 era is also discussed.

*Key findings and conclusions:* Internet and Communication Technologies, blockchain in the food supply chain and other Industry 4.0 applications, as well as approaches that redefine the way we consume food (e.g., lab-grown meat, plant-based alternatives of meat, and valorization of a vast range of bioresources), are the innovations with the highest potential in the new era. There is also an equally pressing need to exploit social marketing to understand attitudes, perceptions, and barriers that influence the behavior change of consumers and the agrifood industry. Subsequently, this change will contribute to adapting to new norms forged by the COVID-19 pandemic, where there is a significant gap in knowledge for decision making.

### 1. Introduction

The COVID-19 pandemic led to millions of infections and deaths worldwide, changing dramatically what we perceived as norms and impacting society, health systems, governmental policies, and businesses. The food sector is no exception, as the consequences of this *"Black Swan"* socio-economic event (Reid et al., 2020) has changed the way we think, buy and consume food by accelerating pre-existing innovation trends (Askew, 2020f), marking a "before" and "after" period. In the short term, the pandemic affected the sector by causing labor problems (e.g., lack of workers due to illness and quarantine measures), the shutdown of factories, food shortages on shelves, and some cases stress of cash flow for the active businesses (Reid et al.,

2020). We are also on the verge of a significant global recession (Guan et al., 2020), lacking critical information for recovery. The role of mapping trends and predicting consumer behavior towards new technologies, services, and products for transitioning beyond COVID-19 will be highly beneficial as per approaches described previously by Busse and Siebert (Busse & Siebert, 2018) and Suanda et al. (Suanda et al., 2018). On a long term basis, the pandemic affects the whole food sector into four main domains: food safety, bioactive food ingredients, food security, and sustainability (Galanakis, 2020).

Overcoming the pandemic's obstacles will not be achieved through austerity but by strong leadership, inspiration, and ambition. Even though vaccination to protect against COVID-19 has begun, the second pandemic wave has affected many countries. Thus it is impossible to

https://doi.org/10.1016/j.tifs.2021.02.002

Received 20 July 2020; Received in revised form 22 January 2021; Accepted 5 February 2021 Available online 10 February 2021 0924-2244/© 2021 Elsevier Ltd. All rights reserved.

<sup>\*</sup> Corresponding author. Research & Innovation Department, Galanakis Laboratories, Chania, Greece.

*E-mail addresses:* cgalanakis@chemlab.gr (C.M. Galanakis), myrtorizou@chemlab.gr (M. Rizou), tdawoud@ksu.edu.sa (T.M.S. Aldawoud), ilknurucak@ohu.edu. tr (I. Ucak), nrowan@ait.ie (N.J. Rowan).

exclude the possibility of repeated infection and lockdown waves, as the pandemic might continue up to the end of 2021, establishing further uncertainty to the food sector (Rowan & Laffey, 2020).

Subsequently, the pandemic's most significant impact may be from the changes induced during the following recovery period (Askew, 2020f). Today, it is uncertain with any degree of confidence what specific impact the global recession will have on the economy, as it relates to particular opportunities and needs encountered by emerging technologies in the food sector.

Through a long reflective lens, lessons have been learned from previous catastrophic global events, such as the Spanish Flu or Black Death, where inspiration has led to paradigm shifts in disruptive technologies. Subsequently, there is an increasing focus on innovative technologies to make the food sector sustainable to meet the opportunities arising from the COVID-19 pandemic (Munekata et al., 2020). The current perspective article explores relevant innovations within the era of the COVID-19 pandemic and post-lockdown era. The foreseen innovations are framed as two more dimensions within the boundaries of the four domains above (Fig. 1) before discussing their prospects of being implemented during expected lockdown periods and eventually to disrupt the food sector after the end of the pandemic crisis.

### 1.1. Implications of the pandemic in the food sector

As a 'Black Swan' socio-economic event occurring extremely rarely, the COVID-19 pandemic resulted in catastrophic consequences. In this kind of crisis, companies would not have predicted and planned accordingly (Reid et al., 2020). Food and food supply chain safety was the first emergent issue under consideration, requiring an increasing number of precautionary measures as long as we move from farm to fork (Rizou et al., 2020). Sustaining food production through COVID-19 brought challenges, including clustering of cases in agricultural food production, slaughterhouses, and food processing industries (Eccdc, 2020). This fact revealed operational issues of the food industry new era trying to maintain food supply balanced with social distancing. Likewise, the unified social solidarity to protect frontline workers from fighting COVID-19 has resulted in the abrupt mass closure of services and businesses globally, which has prompted governments to launch commensurate 'staggering' economic recovery packages to support a beleaguered industry (Govie, 2020; IBEC, 2020). COVID-19 pandemic has reset the new norm for society, with the frenzy of uncertainty fueling many companies to scramble to maintain functionality within their marketplace (Guan et al., 2020). Work practices have changed in terms of remote operation, including working from home and digital communication platforms and Internet and Communication Technologies (ICT) (Del Rio & Malani, 2020). What became evident during this crisis is that advanced and more digital traceability is a powerful tool in comprehending the implications of the supply chain in the case of a public health emerging event (Hahn, 2020).

Over the last year, boosting the immune system was a priority for consumers. This trend has accelerated within the COVID-19 era, and consumers' interest in sustainable, healthy, organic, and functional foods has been increasing rapidly (Askew, 2020a, 2020d; Galanakis, 2015, 2021; Zinoviadou et al., 2015). Moreover, products that are considered by consumers to boost their immune system (e.g., camomile, kombucha) have experienced a 3- to 4-fold increase in sales (Askew, 2020f). A recent survey of 23,000 European shoppers indicated that 72% of consumers would change their eating habits in the post-lockdown era to follow more healthy patterns (Fmcggurus, 2020). Following consumers' needs, food companies are commercializing products with bioactives, creating a trend towards seeking recognition of food bioactives as immune-boosting agents (Galanakis et al., 2020) (Daniells, 2020) that encompass forging more collaborations between governmental bodies and academic institutions to address this need (Koe, 2020a, 2020b).

On the other hand, the pandemic led to an instantaneous lack of critical information about consumers' preferences, attitudes, and bottlenecks in the post-lockdown period (Rowan & Laffey, 2020). The



Fig. 1. The food systems in the era of the COVID-19 pandemic and the prospect of implementing innovations during lockdowns and post COVID-19 era.

induced disruptions in the food supply chain have increased the risk for food fraud, whereas finding convincing evidence about the real health benefits of nutraceuticals and functional foods became critical. This gap also highlighted the emerging role of digitization using data analytics and Artificial Intelligence (AI) to support the real-time needs (e.g., remote monitoring and management decision tools) of the food industry, smart agriculture, supply chain, and food security (Bacco et al., 2019). Moreover, the proliferation of new technologies helped digitize the food supply chain and increased traceability systems' investments to mitigate risk, improve efficiency, and underpin sustainability initiatives (Kennedy et al., 2020).

Sectors benefiting from the current challenges of COVID-19 include companies producing non-perishable foods and processed food companies. For example, panic buying during the lockdown has driven consumers to develop new habits and taste products that they have not purchased before, e.g., freeze-dry ready meals. Since consumers have tasted rehydrated products and compared them with fresh food, it is expected that the demand for this kind of product will grow over the next years (Askew, 2020b). Companies are also experiencing a rapid increase in the need for plant-based ready meals (Askew, 2020e) and alternative protein sources to meat and dairy products (Foodnavigator, 2019). Direct-to-consumer services have rapidly increased, whereas many traditional grocery brands try to develop an easy route to market and become online first brands (Askew, 2020f). The online food trade has witnessed an explosive increase in demand, whereas there is a warning for sustainable packaging solutions due to the rapidly increased need for packaged food. Although the measures needed to ensure food safety, security, and sustainability converge more than ever before, it is essential to avoid hazardous and illegal food products reach the market due to shortages, false claims, or other reasons (e.g., economic). For example, food recalls due to authenticity concerns are likely to be reduced (Everstine, 2020). Since the online order of nutraceuticals has rapidly increased, attention should be given to these products' adulteration and safety (Bullimore, 2020). The second wave of infection that has already been witnessed worldwide will create additional opportunities for innovators and services such as online retail and deliveries (Rowan & Laffey, 2020).

### 2. Innovations and disruptions in the new era

COVID-19 pandemic has accelerated innovation all around the world. This change has already been seen in places where the virus hit first (e.g., in China), where large companies such as Huawei have increased their expenses for research and development activities. There is also reasoning behind this trend as companies that invested in the 2008 economic crisis (instead of just cutting costs) grew fast during the recession (Davey, 2020). On the other hand, defining and forecasting what constitutes a disruptive technology is complicated as the impact is more likely to be measured from a retrospective downstream perspective. Since the 1990s, researchers have referred to disruptive technologies as whirlwind, ground-breaking, game-changing, earth-quake, and emergent technologies that typically cause a substantial disturbance in established market structure and prominent companies. This disturbance is generated by producing highly efficient products and services that are more competitively priced, less complicated, and more accessible than established innovations (Christensen, 1997; Christensen & Bower, 1996; Schuelke-Leech, 2018). Disruptive technologies can substantially cause localized change within a market or industry (e.g., first-order disruption) or cause ground-breaking changes across many cross-cutting domains (e.g., second-order disruption) over a period that substantially influences societal norms. The challenges for technological forecasters and investors are those disruptive technologies are by their nature nascent - meaning that they can only be proven as disruptive in hindsight based upon demonstrating evidence-based impact. The uncertainty arising from COVID-19 will shape future disruptive technologies that may emerge from entrepreneurs, start-ups, small and

medium-sized enterprises (SMEs), and larger established companies. These companies are willing to integrate new solutions at providing smaller, lighter, more flexible, cheaper, and more convenient products (Rowan, 2019).

Fig. 2 illustrates the different innovations within the COVID-19 pandemic era and their application's field(s). The chord diagrams of Fig. 3 demonstrate the prospect of implementing the foreseen innovations in the food sector within the era of the COVID-19 pandemic and respective applications during lockdown periods and within the post-COVID-19 era. The chords start from the potential innovations during the lockdowns and end up in the post-COVID-19 era. The thicker the chords are, and the higher are the prospects of the new era's innovations. These innovations may be enhanced by adjacent industries and services dealing with digitization, using meteorological data linked with climate modeling (Ruiz-Salmón et al., 2020). Overall, it is challenging to predict the holistic consequences of this COVID-19 pandemic and at what time point we will emerge from it, and which innovations and technologies will disrupt the food sector.

### 2.1. Bioactive compounds

The pandemic generated opportunities and challenges for the commercialization of innovative functional foods and nutraceuticals containing target bioactive compounds (e.g., Vitamins and antioxidants) and highlighted the advances of personalized nutrition to boost consumers' immune system and improve their overall health (Galanakis et al., 2020). These prospects are expected to remain high within the post-lockdown and post-pandemic era due to the increased interest of health-conscious individuals. Subsequently, companies will seek new information and knowledge about consumer needs toward these products to address the pandemic's challenges. Nowadays, the development of these kinds of products are proposed with the simultaneous valorization of bioresources, e.g., the recovery of high added-value compounds from food waste (Galanakis, 2012, 2018; Galanakis et al., 2018; Nagarajan et al., 2019; Roselló-Soto, Barba, et al., 2015; Ruiz-Salmón et al., 2020; Wong et al., 2015), and the utilization of microalgae and plant foods (Ananey-Obiri et al., 2018; Bursać Kovačević et al., 2018;



**Fig. 2.** Foreseen innovations and disruptive technologies to tackle challenges of the four directions affected by the COVID-19 pandemic: food safety, bioactive compounds, food security, and sustainability.



Fig. 3. The prospect of implementing the foreseen innovations in the food sector within the era of COVID-19 pandemic and respective applications during lockdown and post-COVID-19 periods.

Galanakis, 2013; Roselló-Soto, Galanakis, et al., 2015). Adjacent advances will also be made in immune-boosting animal feed products such as recently reported by Taufek and co-workers (Taufek et al., 2020), who described the performance of mycelial biomass and exopolysaccharide from Malaysian edible mushroom *Ganoderma lucidum* for the fungivore red hybrid Tilapia (Oreochromis sp.) in Zebrafish embryo. Carballo and co-workers (Carballo et al., 2019) also reported on the use of  $\beta$ -glucans from yeast combined with microalgae extracted digest to improve the health of gut microbiome of high-value fish that prevented bacterial infection.

### 2.2. Food safety

On the other hand, innovations such as smart and active packaging, advanced traceability systems (e.g., using blockchain technology), new biosecurity arrangements (e.g., promoting a food safety culture in food processing facilities and farms), the application of biopesticides to agriculture and industry 4.0 (e.g., blockchain technology) are expected to grow substantially in the new era. The ultimate goal is to protect consumers by ensuring the food and food supply chain's safety and reduce food loss and the environmental impact of the food sector. These innovations may lead to new business models that could disrupt the food supply chain and the market of food products in a techno-socioeconomic way. By achieving international consensus on datasets, priority should be given to reliable processing and critical information for clinical trials (Rowan, 2019) by promoting open access to findings.

For instance, the FDA is planning to release a relevant blueprint targeting the development of traceable food systems and safer, more digital, and more secure food supply (FDA, 2020). Technologies such as artificial intelligence, blockchain, the Internet of Things (IoT), and sensor technology would allow the direct tracking of foods and commodities from farm to fork. The combination of advanced traceability systems with modern analytical and smart tools (e.g., remote or virtual inspections, root cause analyses) would reduce the response in foodborne outbreaks by using data streams. The latest could make the supply chain more visible, reducing the time between tracking the contamination origin of food and responding with mitigating actions.

These kinds of technologies would also assist in imbalances caused by panic buying and spot shortages due to extreme events and help comprehend the causes of food contamination and interpret predictive analytics. The latest use data to predict the contamination possibility and ultimately reduce food loss and waste, e.g., when lockdowns temporarily disrupt the chain of producers and customers in public places (e.g., schools and restaurants). New biosecurity arrangements will help to promote a food safety culture in food processing facilities and farms. There is also a pressing need for education and social enterprise to support the community transitioning to emerging changes and embrace new approaches for food sustainability and security that will accelerate consumer acceptance of green innovation.

### 2.3. Sustainability

With the changing lifestyle and rapid urbanization of the global population that has been accelerated by behavior changes arising from COVID-19, there is an increased generation of food waste from various industrial, agricultural, and household sources (Sharma et al., 2020).

From price spikes and panic buying to the acceleration of food waste, sustainability, and other economic implications, the COVID-19 crisis (as a real exercise) has reminded us that the current food systems are fragile. Thus, they should become more resilient, ensuring food security in future crises such as new pandemics and extreme events due to climate change. Two overarching priorities are to ensure that producers and processors can continue to operate effectively and keep supply lines open and that most services can work as effectively as possible during the pandemic. These are on top of the pressing need to develop innovative means to increase food production to meet growing populations internationally informed by digital technologies. Priorities also include the likely consolidation of significant industries with secure packaging and capacity for research and innovation that will potentially flourish during and post COVID-19 era, when socio-economic norms have been reset and countries quickly deploy economic recovery plans.

The dominant linear economy system, which is mainly based on increasing production to address increased consumption, has proved inefficient for the sustainable management of our resources (Hetemäki et al., 2017; Stegmann et al., 2020). On the other hand, the transition of the current development model to a circular bioeconomy approach could enhance resilience by converting biomass into various biobased products (Farcas et al., 2020; Mak et al., 2020). Thus, a more exceptional drive for innovations in this direction will balance the food supply chain's impact on the environment with the emergence of less-energy intensive, eco-friendly processes, products, and services (O'Neill et al., 2019). For instance, the traditional approaches for managing food waste include land-filling and incineration that generate toxic gases, causing severe environmental and human health hazards. The circular bioeconomy provides opportunities to valorize food waste utilizing biorefineries that produce biofuels, electrical energy, biosurfactants, biofertilizers and so forth (Mordorinelligence, 2020; O'Neill et al., 2019; Rahmanian et al., 2014). The use of biobased packaging materials and non-thermal disinfection technologies for packaging, such as pulsed light (Rowan, 2019), are two examples that could accelerate green innovation in the new era. Biobased materials have been developed to mitigate the complications instigated by conventional plastics. The bioplastic packaging market has been driven by the increasing awareness of traditional plastics' adverse effects, which has steered both consumers and regulatory bodies to opt for biobased materials in place of conventional plastic. The bioplastics packaging market was valued at USD 14.85 billion in 2019 and is expected to reach USD 39.37 billion by 2025 (Mordorinelligence, 2020).

Seafood and aquaculture sectors of Europe, are encounter also significant challenges concerning environmental threats (climate change, marine debris, resource depletion), social development (worker rights, consumer's awareness), or economic growth (market and nonmarket goods and services, global competitiveness). These issues are pressuring all stakeholders, from policy-makers to citizens and industries, to adopt more sustainable policies, practices, and processes. For example, O'Neill (O'Neill et al., 2020) reported Trout and Perch's organic fish farm production on the Irish peatlands using an aquaculture recirculation system (powered with wind energy) where water quality and waste remediation are controlled naturally by using indigenous microalgae and bacteria. Moreover, collaborations among different parties and beyond borders should be improved, aiming to create more efficient networks along the seafood and aquaculture sectors' supply chain. To achieve this, a "nexus thinking" approach (i.e., the analysis of actions in connected systems) combined with a life cycle thinking appears like an excellent opportunity to facilitate the transition to a circular bioeconomy. The emergence of centers of research excellence linked to enterprise and education with a global orientation that seeks to exploit added value to products derived from food waste will increasingly come to the fore. Future research would consider the impact of climate change on food supply chains, including circular bioeconomy, as recently demonstrated by O'Neill et al. (O'Neill et al., 2019), who showed the use of naturally occurring biological indicators to assess the environmental

impact of sustainable periods of drought on Irish freshwater aquaculture.

Besides, aquaculture is an emerging high-protein low carbon emission process that is of interest globally for intensive food sustainability (Tahar et al., 2018a, 2018b). The generated opportunities will be met in part by advances in the digitization of food technologies, innovation in manufacturing (e.g., emerging non-thermal technologies), and services for a diversity of markets and commensurate sustaining disruptive innovation in the adjacent manufacturing and materials sectors. Although most innovations (e.g., lab-grown aseptic meat, plant-based meat alternatives, biobased packaging, automation of food production, and robotics) targeting the sustainability of the food sector are today in a nascent development stage, they are expected to disrupt the food industry in the years to come (Galanakis, 2020). Also, exploiting advanced manufacturing (such as digitization, big data, ICT, blockchain, artificial intelligence, non-thermal technologies, robotics, augment and virtual reality, and 3D food printing) will address sustainability and security qualitative standards, and traceability along the entire food supply chain.

### 2.4. Food security

Disruptive innovation in digitization is transforming the pace and scale of the food and drinks industry globally, ensuring food safety and increasing food sustainability and food security applications. For instance, advanced digital solutions (e.g., IoT, blockchain in the supply chain) to ensure 24/7 order taking are expected to grow, whereas companies will have to promote their values and brand and their quality commitments (Askew, 2020c). The adaptation to new business models that allow the modernization of retail would also contribute to this direction. These models include, e.g., IoT and ICT (e.g., online delivery for supply chain), or e-commerce that utilizes mobile apps for shopping purposes, helping smallholders and producers to find different customers in small city centers (Askew, 2020f) (FDA, 2020; Galanakis, 2020). To this line, Naughton et al. (S. Naughton et al., 2020) demonstrated the need for real-time digitization and ICT in aquaculture to connect complex laboratory data analysis with in-field physicochemical measurements. These tools should determine optimum feeding rates at a fish farm or conditions' adjustments for disease mitigation.

The application value stream mapping is also expected to grow fast in the new era. This digital tool allows the proper management of the supply chain from agricultural production to processing, retail, and consumption. Likewise, it reduces food loss and mitigates environmental impacts (Wesana et al., 2019). Use of potential disruptive non-thermal technologies, such as pulsed light, for food packaging disinfection that also has potential for second-order disruption in areas of waste effluent treatment and critical ecosystem service management such as treatment of pollen frequently contaminated complex parasites and viruses for farmed bumblebees used in buzz pollination of soft fruits and crops across Europe (Garvey & Rowan, 2019; I. M. T. Murray et al., 2018; J. Naughton et al., 2017; Rowan, 2019). Disruptive technologies for promoting pollination security can be extended to commercial electron beam or x-ray treatments for large scale throughput, where the benefits of these emerging solutions have been comprehensively reviewed by (McEvoy & Rowan, 2019). MacFadden and co-workers (Mcfadden et al., 2016) also reported comparing the efficacy of using commercial electron beam and pulsed light for novel sterilization of Irish retailed infant milk formulae to further enhance safety and security from a non-thermal processing perspective. Traditional and emerging technologies were also comprehensively reviewed to help future-proof food systems contaminated with complex foodborne parasites (Franssen et al., 2019; Gérard et al., 2019; Herrero et al., 2020).

There is also a growing potential to exploit immersive Industry 4.0 advances (e.g., virtual and augmented reality) to advance remote workforce training that can embrace the need for social distance to prevent transmission of COVID-19 (N. Murray et al., 2019). Training and

competence development remains mostly grounded in traditional practice methods against a backdrop of highly innovative processes and technology advancements (N. Murray et al., 2019; Rowan, 2019). Likewise, the provision of specialist training in the agri-food workforce merits innovative approaches using technology as a tool in competence development. In the context of specialist training and educational programs, Immersive Multimedia (I.M.) technologies such as virtual reality are emerging as potential platforms based on their delivery of 360° visuals, spatial audio, and allowing the learner to move beyond the passive mode towards an active participant in their learning experience (Braga Rodrigues et al., 2020). In conjunction with wearable sensor technologies (capturing different user physiological metrics), task performance, and user interaction, these technologies facilitate a right "human-in-the-loop" system that supports adaptive, personalized while maintaining context-based learning. Combined educational (such as Cognitive Load Theory) and immersive training capture, at the individual level, critical abilities of the learner, which then informs how the presentation system challenges the learner, thus optimizing the learner experience (Braga Rodrigues et al., 2020; N.; Murray et al., 2019).

### 3. Conclusion

The COVID-19 pandemic has led to historical changes in our society's norms and the way people interact. It also showed direct and high impacts on the food sector, affecting mainly bioactive compounds, food safety, food security, and sustainability. The lockdown of billions of people during the last winter and spring and the lockdown waves that are expected to come in the next months/years led to different innovations in the food sector. Among them, Industry 4.0 applications (IoT, ICT, and blockchain technology) and innovations that disrupt the way we consider and consume food (e.g., lab-grown meat, plant-based alternatives of meat, and valorization of a vast range of bioresources) are the ones with the highest potential in the new era. Niche applications such as the development of nutritional and immune-boosting products to support the health and recovery of COVID-19 patients will become popular. There is also a trend towards intensive sustainable food production systems (such as digitization, AI, and automation in smart agriculture) with future-proofing for the potential impact of security risks and climate change through the supply chain to mitigate critical needs embrace opportunities. Education and training for the next generation of workforce in emerging technologies and accelerating initiatives that will foster behavior change of consumers to the merits of these new business services, technologies, and products will be becoming increasingly important. Globally, there will be a pressing focus on food security regionally and nationally to mitigate against challenges presented by the potential occurrence of future viral pandemics such as that caused by SARS-CoV-2 to protect vulnerable critical supply chains.

### Acknowledgenments

This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

### References

- Ananey-Obiri, D., Matthews, L., Azahrani, M. H., Ibrahim, S. A., Galanakis, C. M., & Tahergorabi, R. (2018). Application of protein-based edible coatings for fat uptake reduction in deep-fat fried foods with an emphasis on muscle food proteins. In *Trends* in food science and technology (Vol. 80, pp. 167–174). Elsevier Ltd. https://doi.org/ 10.1016/j.tifs.2018.08.012
- Askew, K. (2020a). COVID-19 crisis accelerates mainstreaming of immune boosting beta glucans. https://www.foodnavigator.com/Article/2020/04/20/COVID-19-crisis-a ccelerates-mainstreaming-of-immune-boosting-beta-glucans.
- Askew, K. (2020b). European freeze dry looks beyond COVID-19 boost: 'We will continue to see repeat purchase. https://www.foodnavigator.com/Article/2020/0 5/07/European-Freeze-Dry-looks-beyond-COVID-19-boost-We-will-continue-tosee-repeat-purchase.
- Askew, K. (2020c). Organic alliance CEO on COVID-19: 'This epidemic will mark a before and after for society. https://www.foodnavigator.com/Article/2020/05/13/

Organic-Alliance-CEO-on-COVID-19-This-epidemic-will-mark-a-before-and-after-forr-society.

- Askew, K. (2020d). Organic food gets coronavirus boost. https://www.foodnavigator.co m/Article/2020/05/06/Organic-food-gets-coronavirus-boost.
- Askew, K. (2020e). Plant-based ambient brand fiid's strong growth during coronavirus: 'There is a silver lining for our category. https://www.foodnavigator.com/Article/2 020/04/29/Plant-based-ambient-brand-Fiid-s-strong-growth-during-coronavirus -There-is-a-silver-lining-for-our-category.
- Askew, K. (2020f). The FoodNavigator Podcast: What does Coronavirus mean for the future of food?. https://www.foodnavigator.com/Article/2020/05/15/The-Foo dNavigator-Podcast-What-does-Coronavirus-mean-for-the-future-of-food.
- Bacco, M., Barsocchi, P., Ferro, E., Gotta, A., & Ruggeri, M. (2019). The digitisation of agriculture: A survey of research activities on smart farming. *Array*, 3–4, 100009. https://doi.org/10.1016/j.array.2019.100009
- Braga Rodrigues, T., Ó Catháin, C., O'Connor, N. E., & Murray, N. (2020). A Quality of Experience assessment of haptic and augmented reality feedback modalities in a gait analysis system. *PloS One, 15*(3), Article e0230570. https://doi.org/10.1371/ journal.pone.0230570
- Bullimore, S. (2020). COVID-19: Nutraceutical players respond. https://www.nutraceuticalbusinessreview.com/news/article\_page/COVID-19\_Nutraceutical\_players respond/163975.
- Bursać Kovačević, D., Barba, F. J., Granato, D., Galanakis, C. M., Herceg, Z., Dragović-Uzelac, V., & Putnik, P. (2018). Pressurized hot water extraction (PHWE) for the green recovery of bioactive compounds and steviol glycosides from Stevia rebaudiana Bertoni leaves. *Food Chemistry*, 254, 150–157. https://doi.org/10.1016/ j.foodchem.2018.01.192
- Busse, M., & Siebert, R. (2018). The role of consumers in food innovation processes. European Journal of Innovation Management, 21(1), 20–43. https://doi.org/10.1108/ EJIM-03-2017-0023. Emerald Group Publishing Ltd.
- Carballo, C., Pinto, P. I. S., Mateus, A. P., Berbel, C., Guerreiro, C. C., Martinez-Blanch, J. F., Codoñer, F. M., Mantecon, L., Power, D. M., & Manchado, M. (2019). Yeast β-glucans and microalgal extracts modulate the immune response and gut microbiome in Senegalese sole (Solea senegalensis). Fish & Shellfish Immunology, 92, 31–39. https://doi.org/10.1016/j.fsi.2019.05.044
- Christensen, C. M. (1997). The innovator's dilemma: When new technologies cause great firms to fail (M. H. B. S. P. Boston (ed.)).
- Christensen, C. M., & Bower, J. L. (1996). Customer power, strategic investment, and the failure of leading firms. *Strategic Management Journal*, 17, 197–218. https://doi.org/ 10.2307/2486845. Wiley.
- Daniells, S. (2020). NPA's fabricant: 'It's a global pandemic. We need to speak a global language when it comes to the science around vitamin D and zinc. https://www.nutr aingredients-usa.com/Article/2020/05/06/NPA-s-Fabricant-It-s-a-global-pandem ic.-We-need-to-speak-a-global-language-when-it-comes-to-the-science-around-vitam in-D-and-zinc.
- Davey, R. (2020). Now is the time to accelerate innovation Clarivate. https://clarivate. com/article/now-is-the-time-to-accelerate-innovation/.
- Del Rio, C., & Malani, P. N. (2020). COVID-19 new insights on a rapidly changing epidemic. JAMA - Journal of the American Medical Association, 323(14), 1339–1340. https://doi.org/10.1001/jama.2020.3072. American Medical Association.
- Ecdc. (2020). COVID-19 clusters and outbreaks in occupational settings in the EU/EEA and the U.K.
- Everstine, K. (2020). COVID-19 and food fraud risk | FoodSafetyTech. https://foodsa fetytech.com/column/covid-19-and-food-fraud-risk/.
- Farcas, A. C., Galanakis, C. M., Socaciu, C., Pop, O. L., Tibulca, D., Paucean, A., Jimborean, M. A., Fogarasi, M., Salanta, L. C., Tofana, M., & Socaci, S. A. (2020).
   Food security during the pandemic and the importance of the bioeconomy in the new era. Sustainability, 13(1), 150. https://doi.org/10.3390/su13010150
- FDA. (2020). new era of smarter food safety. FDA. https://www.fda.gov/food/new-erasmarter-food-safety.
- Fmcggurus. (2020). The impact of COVID-19 in 2020 & beyond. https://fmcggurus.com/ covid-19/.
- Foodnavigator. (2019). A matter of taste. Your protein proudly green!. https://www. foodnavigator.com/Product-innovations/A-matter-of-taste.-Your-protein-proudly-gr een.
- Franssen, F., Gerard, C., Cozma-Petruţ, A., Vieira-Pinto, M., Jambrak, A. R., Rowan, N., Paulsen, P., Rozycki, M., Tysnes, K., Rodriguez-Lazaro, D., & Robertson, L. (2019). Inactivation of parasite transmission stages: Efficacy of treatments on food of animal origin. Trends in Food Science & Technology, 83, 114–128. https://doi.org/10.1016/j. tifs.2018.11.009. Elsevier Ltd.
- Galanakis, C. M. (2012). Recovery of high added-value components from food wastes: Conventional, emerging technologies and commercialized applications. *Trends in Food Science & Technology*, 26(2), 68–87. https://doi.org/10.1016/j. tifs.2012.03.003, Elsevier.
- Galanakis, C. M. (2013). Emerging technologies for the production of nutraceuticals from agricultural by-products: A viewpoint of opportunities and challenges. *Food and Bioproducts Processing*, 91(4), 575–579. https://doi.org/10.1016/j.fbp.2013.01.004
- Galanakis, C. M. (2015). Separation of functional macromolecules and micromolecules: From ultrafiltration to the border of nanofiltration. *Trends in Food Science* &
- Technology, 42(1), 44–63. https://doi.org/10.1016/j.tifs.2014.11.005. Elsevier Ltd. Galanakis, C. M. (2018). Phenols recovered from olive mill wastewater as additives in meat products. Trends in Food Science & Technology, 79, 98–105. https://doi.org/10.1016/j.tifs.2018.07.010. Elsevier Ltd.
- Galanakis, C. M. (2020). The food systems in the era of the coronavirus (CoVID-19) pandemic crisis. *Foods*, 9(4). https://doi.org/10.3390/foods9040523. MDPI Multidisciplinary Digital Publishing Institute.

- Galanakis, C. M. (2021). Functionality of food components and emerging technologies. *Foods*, 10(1), 128. https://doi.org/10.3390/foods10010128
- Galanakis, C. M., Aldawoud, T. M. S., Rizou, M., Rowan, N. J., & Ibrahim, S. A. (2020). Food ingredients and active compounds against the coronavirus disease (COVID-19) pandemic: A comprehensive review. *Foods*, 9(11), 1701. https://doi.org/10.3390/ foods9111701
- Galanakis, C. M., Tsatalas, P., & Galanakis, I. M. (2018). Implementation of phenols recovered from olive mill wastewater as U.V. booster in cosmetics. *Industrial Crops* and Products, 111, 30–37. https://doi.org/10.1016/j.indcrop.2017.09.058
- Garvey, M., & Rowan, N. J. (2019). Pulsed U.V. as a potential surface sanitizer in food production processes to ensure consumer safety. *Current Opinion in Food Science*, 26, 65–70. https://doi.org/10.1016/j.cofs.2019.03.003. Elsevier Ltd.
- Gérard, C., Franssen, F., La Carbona, S., Monteiro, S., Cozma-Petrut, A., Utaaker, K. S., Režek Jambrak, A., Rowan, N., Rodríguez-Lazaro, D., Nasser, A., Tysnes, K., & Robertson, L. J. (2019). Inactivation of parasite transmission stages: Efficacy of treatments on foods of non-animal origin. *Trends in Food Science & Technology*, 91, 12–23. https://doi.org/10.1016/j.tifs.2019.06.015. Elsevier Ltd.
- Govie. (2020). Roadmap for reopening society and business. https://www.gov.ie/en/ne ws/58bc8b-taoiseach-announces-roadmap-for-reopening-society-and-business-andu/.
- Guan, D., Wang, D., Hallegatte, S., Davis, S. J., Huo, J., Li, S., Bai, Y., Lei, T., Xue, Q., Coffman, D. M., Cheng, D., Chen, P., Liang, X., Xu, B., Lu, X., Wang, S., Hubacek, K., & Gong, P. (2020). Global supply-chain effects of COVID-19 control measures. *Nature Human Behaviour*, 4(6). https://doi.org/10.1038/s41562-020-0896-8
- Hahn, S. (2020). Pandemic challenges highlight the importance of the new era of smarter food safety. FDA. https://www.fda.gov/news-events/fda-voices/pandemic-challengeshighlight-importance-new-era-smarter-food-safety.
- Herrero, M., Thornton, P. K., Mason-D'Croz, D., Palmer, J., Benton, T. G., Bodirsky, B. L., Bogard, J. R., Hall, A., Lee, B., Nyborg, K., Pradhan, P., Bonnett, G. D., Bryan, B. A., Campbell, B. M., Christensen, S., Clark, M., Cook, M. T., de Boer, I. J. M., Downs, C., ... West, P. C. (2020). Innovation can accelerate the transition towards a sustainable food system. *Nature Food*, 1(5), 266–272. https://doi.org/10.1038/s43016-020-0074-1
- Hetemäki, L., Hanewinkel, M., Muys, B., Ollikainen, M., Palahi, M., & Trasobares, A. (2017). Leading the way to a European circular bioeconomy strategy. European Forest Institute. https://efi.int/publications-bank/leading-way-european-circular-bioecon omy-strategy.
- IBEC. (2020). COVID19 Ibec publishes proposed solutions to liquidity crisis. https://www. ibec.ie/connect-and-learn/media/2020/04/03/covid19-ibec-publishes-proposed-so lutions-to-liquidity-crisis.
- Kennedy, A., Stitzinger, J., & Burke, T. (2020). Food traceability. In A. Demirci, H. Feng, & K. Krishnamurthy (Eds.), Food safety engineering, food engineering series. Springer Nature Switzerland.
- Koe, T. (2020a). COVID-19 and nutra research: Where does the industry need to focus its scientific endeavours?. https://www.nutraingredients-asia.com/Article/202 0/04/01/COVID-19-and-nutra-research-Where-does-the-industry-need-to-focusits-scientific-endeavours.
- Koe, T. (2020b). Immunity innovation: Gencor to study PEA's immune boosting effects via human trial. https://www.nutraingredients-asia.com/Article/2020/03/19/Immunit y-innovation-Gencor-to-study-PEA-s-immune-boosting-effects-via-human-trial.
- Mak, T. M. W., Xiong, X., Tsang, D. C. W., Yu, I. K. M., & Poon, C. S. (2020). Sustainable food waste management towards circular bioeconomy: Policy review, limitations and opportunities. *Bioresource Technology*, 297, 122497. https://doi.org/10.1016/j. biortech.2019.122497
- McEvoy, B., & Rowan, N. J. (2019). Terminal sterilization of medical devices using vaporized hydrogen peroxide: A review of current methods and emerging opportunities. *Journal of Applied Microbiology*, 127(5), 1403–1420. https://doi.org/ 10.1111/jam.14412. Blackwell Publishing Ltd.
- Mcfadden, E., Ramos, A.-L. C., Bradley, D., Vrain, O., Mcevoy, B., Rowan, N. J., & Student, P. (2016). Comparative studies on the novel sterilization of Irish retailed infant milk formulae using electron beam and pulsed light treatments. *International Journal of Science, Environment and Technology*, 5(6), 4375–4377.
- Mordorinelligence. (2020). Smart waste management market | growth, trends, and forecast (2020 2025). https://www.mordorintelligence.com/industry-reports/smart-waste -management-market.
- Munekata, P. E. S., Domínguez, R., Budaraju, S., Roselló-Soto, E., Barba, F. J., Mallikarjunan, K., Roohinejad, S., & Lorenzo, J. M. (2020). Effect of innovative food processing technologies on the physicochemical and nutritional properties and quality of non-dairy plant-based beverages. *Foods*, 9(3). https://doi.org/10.3390/ foods9030288. MDPI Multidisciplinary Digital Publishing Institute.
- Murray, N., Buckley, J., Seery, N., & Rowan, N. J. (2019). Blending immersive and educational technologies to inform sustainability and diversification of workforce training through machine interface learning using sterilization technologies as model - quo vadis?. https://research.thea.ie/handle/20.500.12065/3280.
- Murray, I. M. T., Rowan, N. J., McNamee, S., Campbell, K., & Fogarty, A. M. (2018). Pulsed light reduces the toxicity of the algal toxin okadaic acid to freshwater crustacean Daphnia pulex. *Environmental Science and Pollution Research*, 25(1), 607–614. https://doi.org/10.1007/s11356-017-0472-6
- Nagarajan, J., Krishnamurthy, N. P., Nagasundara Ramanan, R., Raghunandan, M. E., Galanakis, C. M., & Ooi, C. W. (2019). A facile water-induced complexation of lycopene and pectin from pink guava byproduct: Extraction, characterization and kinetic studies. *Food Chemistry, 296*, 47–55. https://doi.org/10.1016/j. foodchem.2019.05.135
- Naughton, S., Kavanagh, S., Lynch, M., & Rowan, N. J. (2020). Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing

efficacy of water quality in a freshwater aquaculture recirculation system: A case study from the republic of Ireland. *Aquaculture*, 526, 735377. https://doi.org/10.1016/j.aquaculture.2020.735377

- Naughton, J., Tiedeken, E. J., Garvey, M., Stout, J. C., & Rowan, N. J. (2017). Pulsed light inactivation of the bumble bee trypanosome parasite Crithidia bombi. *Journal of Apicultural Research*, 56(2), 144–154. https://doi.org/10.1080/ 00218839.2017.1289668
- O'Neill, E. A., Rowan, N. J., & Fogarty, A. M. (2019). Novel use of the alga Pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. *The Science of the Total Environment, 692*, 209–218. https://doi.org/10.1016/j. scitotenv.2019.07.243
- O'Neill, E. A., Stejskal, V., Clifford, E., & Rowan, N. J. (2020). Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. *The Science of the Total Environment*, 706, 136044. https://doi.org/10.1016/j.scitotenv.2019.136044
- Rahmanian, N., Jafari, S. M., & Galanakis, C. M. (2014). Recovery and removal of phenolic compounds from olive mill wastewater. JAOCS, Journal of the American Oil Chemists' Society, 91(1), 1–18. https://doi.org/10.1007/s11746-013-2350-9
- Reid, G., O'Beirne, N., & Gibson, N. (2020). How to reshape results during the COVID-19 crisis | E.Y. - Ireland. https://www.ey.com/en\_ie/transactions/companies-can-resha pe-results-and-plan-for-covid-19-recovery.
- Rizou, M., Galanakis, I. M., Aldawoud, T. M. S., & Galanakis, C. M. (2020 Aug). Safety of foods, food supply chain and environment within the COVID-19 pandemic. *Trends in Food Science & Technology*, 102, 292–299. https://doi.org/10.1016/j. ttifs.2020.06.008
- Roselló-Soto, E., Barba, F. J., Parniakov, O., Galanakis, C. M., Lebovka, N., Grimi, N., & Vorobiev, E. (2015). High voltage electrical discharges, pulsed electric field, and ultrasound assisted extraction of protein and phenolic compounds from olive kernel. *Food and Bioprocess Technology*, 8(4), 885–894. https://doi.org/10.1007/s11947-014-1456-x
- Roselló-Soto, E., Galanakis, C. M., Brnčić, M., Orlien, V., Trujillo, F. J., Mawson, R., Knoerzer, K., Tiwari, B. K., & Barba, F. J. (2015). Clean recovery of antioxidant compounds from plant foods, by-products and algae assisted by ultrasounds processing. Modeling approaches to optimize processing conditions. *Trends in Food Science & Technology*, 42(2), 134–149. https://doi.org/10.1016/j.tifs.2015.01.002. Elsevier Ltd.
- Rowan, N. J. (2019). Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo vadis? *Trends in Food Science & Technology, 88*, 316–332. https://doi.org/10.1016/j.tifs.2019.03.027. Elsevier Ltd.
- Rowan, N. J., & Laffey, J. G. (2020). Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic – case study from the Republic of Ireland. *The Science of the Total Environment*, 725, 138532. https://doi.org/10.1016/j. scitotenv.2020.138532
- Ruiz-Salmón, I., Margallo, M., Laso, J., Villanueva-Rey, P., Mariño, D., Quinteiro, P., Dias, A. C., Nunes, M. L., Marques, A., Feijoo, G., Moreira, M. T., Loubet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N., Méndez-Paz, D., Iglesias-Parga, X., ... Aldaco, R. (2020). Addressing challenges and opportunities of the European seafood sector under a circular economy framework. *Current Opinion in Environmental Science and Health*, 13, 101–106. https://doi.org/10.1016/j. coesh.2020.01.004
- Schuelke-Leech, B. A. (2018). A model for understanding the orders of magnitude of disruptive technologies. *Technological Forecasting and Social Change*, 129, 261–274. https://doi.org/10.1016/j.techfore.2017.09.033
- Sharma, P., Gaur, V. K., Kim, S. H., & Pandey, A. (2020). Microbial strategies for biotransforming food waste into resources. *Bioresource Technology*, 299, 122580. https://doi.org/10.1016/j.biortech.2019.122580
- Stegmann, P., Londo, M., & Junginger, M. (2020). The circular bioeconomy: Its elements and role in European bioeconomy clusters. *Resources, Conservation and Recycling X, 6.* https://doi.org/10.1016/j.rcrx.2019.100029. Elsevier B.V.
- Suanda, J., Cawley, D., Brenner, M., Domegan, C., & Rowan, N. (2018). Identification of behavioural change strategies to prevent cervical cancer among Malay women in Malaysia. *Social Marketing*, 555–566. https://doi.org/10.4324/9781315648590-34. Routledge.
- Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E., & Rowan, N. (2018a). Full water quality monitoring of a traditional flow-through rainbow trout farm. *Fishes*, 3(3), 28. https://doi.org/10.3390/fishes3030028
- Tahar, A., Kennedy, A. M., Fitzgerald, R. D., Clifford, E., & Rowan, N. (2018b). Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. *PeerJ*, (7), Article e5281. https://doi.org/ 10.7717/peerj.5281
- Taufek, N. M., Harith, H. H., Abd Rahim, M. H., Ilham, Z., Rowan, N., & Wan-Mohtar, W. A. A. Q. I. (2020). Performance of mycelial biomass and exopolysaccharide from Malaysian Ganoderma lucidum for the fungivore red hybrid Tilapia (Oreochromis sp.) in Zebrafish embryo. Aquaculture Reports, 17, 100322. https://doi.org/10.1016/j.aqrep.2020.100322
- Wesana, J., Gellynck, X., Dora, M. K., Pearce, D., & De Steur, H. (2019). Measuring food losses in the supply chain through value stream mapping: A case study in the dairy

### C.M. Galanakis et al.

sector. Saving Food, 249–277. https://doi.org/10.1016/b978-0-12-815357-4.00009-2. Elsevier.

- Wong, W. H., Lee, W. X., Ramanan, R. N., Tee, L. H., Kong, K. W., Galanakis, C. M., Sun, J., & Prasad, K. N. (2015). Two level half factorial design for the extraction of phenolics, flavonoids and antioxidants recovery from palm kernel by-product. *Industrial Crops and Products*, 63, 238–248. https://doi.org/10.1016/j. indcrop.2014.09.049
- Zinoviadou, K. G., Galanakis, C. M., Brnčić, M., Grimi, N., Boussetta, N., Mota, M. J., Saraiva, J. A., Patras, A., Tiwari, B., & Barba, F. J. (2015). Fruit juice sonication: Implications on food safety and physicochemical and nutritional properties. *Food Research International*, 77(77), 743–752. https://doi.org/10.1016/j. foodres.2015.05.032

**RESEARCH ARTICLE** 



### Pulsed ultraviolet (PUV) disinfection of artificially contaminated seawater seeded with high levels of pathogen disease indicators as an alternative for the shellfish industry depuration systems

Gustavo Waltzer Fehrenbach<sup>1</sup> · Emma Murphy<sup>1,2</sup> · Robert Pogue<sup>3,4</sup> · Frank Carter<sup>5</sup> · Eoghan Clifford<sup>6,7</sup> · Ian Major<sup>1</sup> · Neil Rowan<sup>4</sup>

Received: 11 January 2023 / Accepted: 24 April 2023 / Published online: 8 May 2023 © The Author(s) 2023

### Abstract

The increase in pathogen levels in seawater threatens the safety of entire aquatic ecosystems. Foodborne pathogens can potentially accumulate in shellfish, especially in filter feeders such as bivalves, requiring an efficient depuration process before consumption. Alternative approaches to promote a cost-efficient purge at depuration plants are urgently needed. A small prototype pulsed ultraviolet (PUV) light recirculation system was designed, and its depuration potential was tested in a seawater matrix artificially contaminated with high levels of microbial pathogens *Escherichia coli*, *Staphylococcus aureus*, *Salmonella typhimurium*, *Bacillus cereus* and *Candida albicans*. The analysis of treatment parameters including voltage, number of pulses and duration of treatment was performed to ensure the highest reduction in contaminant levels. Optimal PUV disinfection was attained at 60 pulses/min at 1 kV for 10 min (a UV output of 12.9 J/cm<sup>2</sup>). All reductions were statistically significant, and the greatest was observed for *S. aureus* (5.63 log<sub>10</sub>), followed by *C. albicans* (5.15 log<sub>10</sub>), *S. typhimurium* (5 log<sub>10</sub>), *B. cereus* (4.59 log<sub>10</sub>) and *E. coli* (4.55 log<sub>10</sub>). PUV treatment disrupted the pathogen DNA with the result that *S. aureus*, *C. albicans* and *S. typhimurium* were not detectable by PCR. Regulations were reviewed to address the applicability of PUV treatment as a promising alternative to assist in the reduction of microbial pathogens at depuration plants due to its high efficiency, short treatment period, high UV dose and recirculation system as currently employed in shellfish depuration plants.

Keywords Disinfection · Foodborne pathogens · Seafood · Depuration · Ultraviolet

Responsible Editor: Sami Rtimi

Gustavo Waltzer Fehrenbach gfehrenbach@ait.ie

- <sup>1</sup> Materials Research Institute, Technological University of the Shannon – Midlands Campus, N37 HD68, Athlone, Ireland
- <sup>2</sup> LIFE Health and Biosciences Research Institute, Technological University of the Shannon – Midwest Campus, V94 EC5T, Limerick, Ireland
- <sup>3</sup> Post-Graduate Program in Genomic Sciences and Biotechnology, Catholic University of Brasilia, 71966-700, Brasilia, Brazil

Introduction

Seafood is an important food source and includes finfish, crustaceans, echinoderms and molluscs obtained from fresh and saltwater. Seafood represents approximately 7% of the global food market and is projected to reach a value of \$336

- <sup>4</sup> Center for Sustainable Disinfection and Sterilization, Bioscience Research Institute, Technological University of the Shannon – Midlands Campus, N37 F6D7, Athlone, Ireland
- <sup>5</sup> Coney Island Shellfish Ltd., F91 YH56, Sligo, Ireland
- <sup>6</sup> School of Engineering, National University of Ireland Galway, H91 HX31, Galway, Ireland
- <sup>7</sup> Ryan Institute, National University of Ireland Galway, H91 HX31, Galway, Ireland

billion US dollars by 2025 (Shahbandeh, 2021). Global seafood consumption has increased at an average rate of 3% a year from 1961 to 2019, almost twice the annual world population growth for the same period (FAO, 2022). Consumption per capita grew from 9 kg in 1961 to 20.2 kg in 2020, and the seafood production chain evolved to meet the increased demand. One of the main changes was the rapid development of aquaculture systems with the use of technology such as automated feeding process and traceability of development to maximize yields and increase sustainability, expanding production from 19 million tonnes in 1950 to 179 million tonnes in 2018 (FAO, 2016). Shellfish is a seafood widely cultured in sea-based systems, relying on suitable water and environmental conditions. Bivalves are the most abundantly produced shellfish in Europe and will be referred to in this research. As a result of their filter-feeding mechanism, bivalves are susceptible to water impurities, which may lead to the accumulation of high levels of microbial pathogens (Martinez-Albores et al. 2020). Its filter physiology will also purge pathogens if placed in uncontaminated water and is often realized in depuration plants to purge it to levels within safe limits (EC, 2019).

After harvesting, the live bivalve is washed/cleaned in a processing plant to remove mud and debris, and graded by weight, a process whereby the underweight bivalve is returned to cages in water until achieving the commercial weight. The product is then transported to a depuration facility where it will remain for the depuration period. Seawater is mainly treated on recirculation and flow-through systems in large tanks of usually 1-3 m<sup>3</sup> and often pre-treated before usage due to the presence of contaminants. The type of depuration system deployed depends on factors such as availability and the average weight of the bivalve treated. Water quality is assessed by the presence of contaminants, phytoplankton concentration and parameters such as temperature, dissolved oxygen, turbidity and salinity. Re-circulation systems rely on water consistency since they can recirculate the same water for at least 24 h (Schneider et al. 2009). Such systems (Fig. 1a) usually consist of a pump that pumps water from the tank, recirculates it through a UV treatment and sprays it on the bivalve seated on trays. Chlorine, iodophors and ozone are also used in the depuration process to deactivate pathogens such as *E. coli* and NOV released from the bivalve and in the pre-treatment of seawater. In flow-through systems (Fig. 1b), seawater is continuously pumped through the depuration system and will discharge fractions of the water and replenish it during the depuration process. A clean and reliable source of seawater is necessary to provide water free of contaminants, and UV or membrane filtration can enhance seawater quality prior to pumping into the depuration system.

The disuse of chlorine treatment is due to several effects on bivalves such as shutting down the filter-feeding activity on oysters and carcinogenic potential of chlorinated metabolites, requiring a post-treatment with vigorous agitation and degassing with thiosulphate (Schneider et al. 2009; Martinez-Albores et al. 2020). Iodophors are substances whereby their bactericidal activity is based on elemental iodine, which penetrate the cell walls and membranes interfering with DNA synthesis. Their efficiency and safety for bivalve's depuration system have not been widely researched. Ozone is a strong oxidizing agent that inactivates contaminants by attacking the double bonds of organic compounds, supporting its use on lipophilic toxins. It has an increased operation cost and can generate cancerous co-products such as bromates. Of the available techniques, UV disinfection is distinct due to significant reductions in the viability of waterborne pathogens without leaving residuals and is the most widely used method in the UK, the USA and Australia (McLeod et al. 2017) and partially used in China where ozone is also applied (Lee et al. 2008).

The treatment efficacy is due to severe damage to cell structure from the activation of the photoreactive potential of purines and pyrimidines in DNA that triggers the



Fig. 1 Shellfish depuration system. a Recirculation system. b Flow-through. V: valves

70773





formation of mutations, oxidative stress and production of reactive oxygen species (ROS). These ROS attack nucleotide pools and stress-mediated mutagenesis (Ikehata and Ono, 2011). The current UV systems operate continuously and inactivate pathogens in recirculated water from depuration tanks. They are typically low-pressure mercury UV bulbs working mostly monochromatically from 100 to 400 nm and approximately 40 W. Treatment efficacy results from severe damage to a pathogen's cell structure. The efficacy of the method may also be reduced when enteric viruses (such as NOV) are present, because higher UV doses and extended retention times are required to reach safety levels. In this case, however, the use of high-power systems with wider wavelengths and UV doses can overcome these challenges, reducing the average time a certain amount of water stays in the depuration plant (hydraulic retention time) and ensuring food safety.

Pulsed ultraviolet (PUV) light treatment is based on high peak power pulses of  $> 1 \text{ kW/cm}^2$  generated by xenon lamps and delivered at short intervals (100 ns to 2 ms) within a broad wavelength spectrum ranging from 100 to 1100 nm. Its disinfection efficacy was reported for different pathogens and areas such as the packaging industry (Garvey and Rowan, 2015; Chen et al. 2015), surfaces in an animal laboratory (Li et al. 2020), artificially contaminated wastewater (Fitzhenry et al. 2021), different types of milk (Ansari et al. 2019) and cheese (Ricciardi et al. 2020). PUV is a promising alternative for the food industry as there is no by-product identified for PUV treatment (Rowan, 2019), and its application in food products is authorized by the United States Food and Drug Administration (FDA, 1999). Its high power and UV dose can potentially inactivate pathogens in the shellfish depuration system.

In Europe, the level of contaminants is constantly monitored in designated shellfish production areas which are classified as A, B and C. Bivalve molluscs from class A shall not exceed 230 *Escherichia coli* per 100 g of flesh and intravalvular liquid in 80% of samples collected, considered safe to consume if no other health risk is present. Class B is designated for areas where *E. coli* abundance does not exceed 4600 in 90% of the samples per 100 g of flesh and intravalvular liquid and 46,000 per 100 g of flesh and intravalvular liquid for class C. E. coli is still the main parameter for monitoring as most of the shellfish contamination arises from human and animal faeces, which suggests the presence of other contaminants also present in faeces but in lower concentrations. When harvested from class B and C areas, shellfish must be purged in depuration plants. Other contaminants such as paralytic shellfish poison (PSP) and norovirus (NOV) are also periodically monitored, and regulations follow the basic principles of food law to protect human health and consumer interests as set by Regulation (EC) No. 178/2002 (EC, 2002). The Centre for Environment, Fisheries and Aquaculture Science (CEFAS, 2022) recommends a minimum of 42 h to depurate the shellfish from class A and B harvesting areas. Then, it can be commercialized if it is within the limits established for the following parameters: organoleptic characteristics (1), biotoxins (2), E. coli (3), S. typhimurium (4), norovirus (5) and hepatitis A (6). The first parameter requires an absence of dirt on shells, a normal amount of intravalvular liquid and adequate response to percussion. Biotoxins must be within limits, such as paralytic shellfish poison (< 800  $\mu$ g/kg), amnesic shellfish poison (< 20 mg of domoic acid/kg), yessotoxins (1 mg/kg) and azaspiracods (160 µg/kg). Salmonella should not be detected in 25 g of bivalve tissue (EC, 2004). Hepatitis A and norovirus (NOV) are highly contagious, and there is no consensus

 Table 1
 PUV preliminary experiments to identify the best treatment conditions

Treatment	Voltage (V)	Number of pulses per min	Duration (min)	
1	500	6	1	
2	1000	6	1	
3	500	6	10	
4	1000	6	10	
5	500	60	10	
6	1000	60	10	

Table 2 Operating parameters of PUV lamp at 500 and 1000 V

Discharge voltage (V)	Energy per pulse (J)	Peak cur- rent (A)	Peak power (kW)	Peak admit- tance (S)	Current rise/fall time (µs)
500	5	443	175	1.10	12/36
1000	20	1173	985	1.46	7/27

or regulations for limits of RNA copies in shellfish which are monitored and controlled in a pro-active way to preserve consumer safety (FSAI et al. 2018).

In this paper, we describe and report the test results from a small prototype pulsed ultraviolet recirculation system for purification of seawater artificially contaminated with high levels of Escherichia coli, Staphylococcus aureus, Bacillus cereus, Candida albicans and Salmonella typhimurium. For example, *Escherichia coli* and *Salmonella* species were specifically chosen because they are used as index pathogens of faecal contamination for water quality assessment (Holcomb and Stewart, 2020). Candida albicans was chosen because it is representative of a clinical yeast occurring in wastewater (Babič et al. 2017). Staphylococcus aureus was chosen because it is associated with foodborne intoxications (FSAI, 2011). These organisms are also representative of Gram-positive and Gram-negative pathogens, which allows for verifying the effectiveness of PUV in different cellular structures. The proposed treatment described here was developed to be further tested in and ultimately employed by the shellfish industry at depuration plants to assist in the disinfection from common foodborne pathogens.

### **Material and methods**

### **Pulsed-ultraviolet treatment**

A small pulsed ultraviolet (PUV) light reactor was built to be further tested in the depuration process of the shellfish industry. The 60-kPa low-pressure xenon-filled flash lamp (Heraeus Noblelight XAP type NL4006) was fixed in stainless steel chamber of 100 mL volume internally walled with glass. The system was powered by a 1000-V scale power source (Samtech Pulsed UV 2018) equipped with an automatic trigger system. Purification was performed in a recirculation regime, and a schematic is presented in Fig. 2. A peristaltic pump (ColeParmer) was employed to recirculate 200 mL of contaminated seawater of salinity of 35 ppt and a temperature of 18 °C at a constant flowrate of 10 mL/s with hydraulic retention time (HRT) of 0.10 s. Using the same inoculum, no treatment (Fig. 2) was performed but not exposed to PUV, and each pathogen was recirculated through the system under the same conditions. The efficacy of PUV treatment was then compared to no treatment.

PUV treatment conditions were selected based on log reduction after treatment at different voltage intensities (V), number of pulses per min and time (min) as presented in Table 1.

Energy, peak current, peak power, peak admittance and current rise/fall time at 500 and 1000 V are presented in Table 2. Values are available in the PUV manufacturer's manual. Energy (J) corresponds to release per pulse in the flash lamp. Peak current (A) and power (kW) are the maximum current and power at the peak of pulsed energy, respectively, while peak admittance (S) is a measure of how easily the current will flow through the system and reach the lamp. Current rise/fall corresponds to time (µs) until it reaches the peak current and returns to 0.

UV output was determined by the product of UV irradiance (mJ/cm<sup>2</sup>), therefore a wavelength < 300 nm, and exposure time (s). The calculations are presented below.

- 1) Energy output (Eo) was determined by the product of energy per pulse (J or W/s) and pulses per second (0.1 and 1).
- Energy intensity (Ea) was determined by the division of Eo by the lamp surface area (170 cm<sup>2</sup> - height: 26 cm and radius: 1 cm).
- 3) Considering the ratio of energy generated at the UV range (< 300 nm) as 0.18 (Appendix 1), the product of Ea by 0.18 resulted in the energy intensity in the UV region (EaUV).
- Finally, UV output (J/cm<sup>2</sup>) after complete treatment was calculated by the product of exposure time (s) and EaUV (Table 3).

			11 5 0	1 1			
Discharge volt- age (V)	Energy per pulse (J)	Pulses per second	Eo (W)	Ea (mW/cm <sup>2</sup> )	EaUV (mW/cm <sup>2</sup> )	Exposure time (s)	UV output (J/cm <sup>2</sup> )
500	5	0.1	0.5	2.941	0.529	60	0.0324
		0.1	0.5	2.941	0.529	600	0.324
		1	5	29.41	5.29	600	3.24
1000	20	0.1	2	11.76	2.11	60	0.129
		0.1	2	11.76	2.11	600	1.29
		1	20	117.6	21.1	600	12.9

Table 3 UV output (J/cm<sup>2</sup>) at 500 and 1000 V when applying 0.1 and 1 pulses per second for 1 and 10 min


Fig. 3 Wavelength spectrum from emitted PUV flash (Samtech Pulsed UV 2018)

The spectrum emitted from the flash lamp is presented in Fig. 3. The voltages of 500 and 1000 V have similar spectra, with high output in the UVC region (220–80 nm) and visible light (430, 460 and 500 nm).

#### Foodborne pathogens and culture conditions

Bacterial species detected in shellfish and yeast associated with foodborne illness were selected to simulate a severe contamination of seawater. Contaminant levels in the seafood industry are much lower than the levels tested here; however, this approach was chosen to test the robustness of the PUV method. Escherichia coli (ATCC25922), Staphylococcus aureus (ATCC29213), Bacillus cereus (ATCC11778), Candida albicans (ATCC10231) and Salmonella typhimurium (IMD121) were selected and cultivated at under specific conditions and with specific media. E. coli and S. aureus were cultivated in Luria Bertani (10 g/L peptone, 5 g/L yeast extract and 5 g/L chloride), while B. cereus, C. albicans and S. typhimurium in Brain Heart Infusion (200 g/L calf brain, 250 g/L beef heart, 10 g/L proteose peptone, 5 g/L sodium chloride, 2.5 g/L sodium phosphate, 2 g/L dextrose), Potato Dextrose Broth (200 g/L potato infusion, 20 g/L dextrose) and Tryptic Soy Broth (17 g/L peptone from casein, 3 g/L peptone from soymeal, 2.5 g/L D(+) glucose monohydrate, 5 g/L sodium chloride, 2.5 g/L di-potassium hydrogen phosphate), respectively. Flasks containing media and respective strains were then cultivated in a shaker at 28 °C (C. albicans) and 37 °C (E. coli, S. aureus, B. cereus, S. typhimurium) to exponential phase and used as inoculum. Seawater was collected at Sligo Bay (54.3432° N, 8.5728° W) and autoclaved for 15 min at 121 °C to be used as a treatment matrix. A salinity of 35 ppt was observed, and pH was 7.6. The spiking was performed at room temperature (18 °C), and strains were treated in separate reactions at  $10^{6-8}$  CFU/mL. Experiments were run as presented in Fig. 2. The PUV system was washed and sanitized with isopropanol  $70^{\circ}$  several times to ensure complete cleaning. The average initial concentration of pathogens in preliminary assays was  $1.8 \times 10^8 \pm 1.68$  CFU/mL (Fig. 4a–e) and  $2 \times 10^9 \pm 2.31$  CFU/mL (Fig. 4f) when re-testing treatment 6 conditions.

#### **Cell viability**

A spread plate method was used to count the number of viable cells of no treatment (NT) and treated (T) with PUV. Samples were collected, serially diluted, spread onto agar plates and incubated at 28 °C (*C. albicans*) and 37 °C (other pathogens) for 36 and 24 h, respectively. Cell viability reduction (CVR) was calculated as presented in Eq. 1.

$$CVR = log_{10} \frac{NT}{T}$$

where NT is no treatment, and T is treated

#### **DNA extraction**

Bacterial genomic DNA was extracted as described on the GenElute kit (Sigma-Aldrich) with modifications. Samples of 2 mL were collected before and after PUV treatment and non-treated used as a positive control. Samples were centrifuged at 13,000 rpm for 2 min, and pellets of B. cereus, S. aureus and C. albicans were pre-treated with 200 µL of lysozyme for 30 min at 37 °C, while E. coli and S. typhimurium incubated with 20 µL of proteinase K for 30 min at 55 °C. To assist with DNA extraction, C. albicans was sonicated (35 W/40 °C) for 20 min prior to incubation with lysozyme. Proteinase K was then added to Gram-positive species, and 200 µL of lysis solution C was added to all strains and incubated for 10 min at 55 °C. The subsequent steps of column preparation, binding and washing steps were performed as described in the kit. DNA retained in the column was eluted with elution solution (10 mM Tris-HCl, 0.5 mM EDTA, pH 9.0) and stored at -20 °C.

#### Amplification conditions and gel electrophoresis

Polymerase chain reaction (PCR) was performed to verify the integrity and viability of pathogens after PUV treatment. The extracted DNA from *E. coli*, *S. aureus*, *B. cereus*, *C. albicans* and *S. typhimurium* were thawed, and 1 ng was mixed with the master mix according to Table 4. Bacterial 16S rDNA was amplified with the primers Pro341F and Pro805R (Table 5), with an expected product of 464 bp (Takahashi et al. 2014). For *C. albicans*, 18S rDNA primers SS5F and SS3R were used, and a product of 1800 bp was expected from amplification (Matsumoto et al. 2010).

PCR products were loaded with agarose 1X onto 1.5% agarose gel stained with SyBR green. Amplicons were



70776



Fig. 4 Cell viability (CFU/mL) of Escherichia coli (a), Staphylococcus aureus (b), Bacillus cereus (c), Candida albicans (d) and Salmonella typhimurium (e) not treated (NT) and after PUV treatment at different conditions (f). CVR of pathogens under optimal conditions identified on repeated treatment 6 (1000 V, 60 pulses/min and 10-min

treatment). Standard deviation bars are presented in ± CFU/mL. Statistical significance (p < 0.05) was observed in all pathogens treated with PUV at optimal conditions when comparing viabilities after treatment with no treatment levels

	Molarity	Cycling conditions	Molarity	Cycling conditions
5X Platinum Buffer 2 mM dNTP Mix Forward primer Reverse primer Taq polymerase Water Genomic DNA	1X 0.2 μM 0.2 μM 0.2 μM 0.04 U/μL - 1 ng	94 °C - 2' *94 °C - 15" *60 °C - 15" *68 °C - 15" *25 cycles	1X 0.2 μM 0.265 0.265 0.0275 - 1 ng	95 °C - 1' *95 °C - 15" *55 °C - 15" *72 °C - 30" 72 °C - 7' *25 cycles

Table 4 PCR reagents and cycling conditions

Table 5	Primer sec	juence and	expected	product in	base	pairs (	bp	)
---------	------------	------------	----------	------------	------	---------	----	---

Primers	Sequence	Product (bp)
Pro341F	AATGATACGGCGACCACCGAGATC TACACTCTTTCCCTACACGACGCT CTTCCGATCTCCTACGGGAGGCAG CAGCCTACGGGNBGCASCAG	464
Pro805R	CAAGCAGAAGACGGCATACGAGAT NNNNNGTGACTGGAGTTCAGACG TGTGCTCTTCCGATCTGACTACN- VGGGTATCTAATCC	
SS5F	GGTGATCCTGCCAGTAGTCATATG CTTG	1800
SS3R	GATCCTTCCGCAGGTTCACCTACG GAAACC	

visualized in a UV chamber, and band pixels were quantified on the software ImageJ (Rasband, 2018) to estimate the cDNA concentration.

#### **Statistical analysis**

Results were analysed on the *Statistica* software version 10 (Statsoft, USA) by Student's *t*-test comparing means, and p < 0.05 was considered significant. All experiments were done at least three times in triplicates.

#### **Results and discussion**

The objective of this paper was to construct and test a small prototype pulsed ultraviolet (PUV) light system to demonstrate how this method might assist the shellfish industry in the depuration processes. The analysis of treatment parameters such as voltage, number of pulses and duration of treatment was performed to ensure a high reduction in

**Fig. 5** Cell viability reduction of *E. coli, S. aureus, B. cereus, C. albicans* and *S. typhimurium* by UV output of 0.032 J/cm<sup>2</sup> for treatment 1, 0.129 J/cm<sup>2</sup> for treatment 2, 0.324 J/cm<sup>2</sup> for treatment 3, 1.29 J/cm<sup>2</sup> for treatment 4, 3.24 J/cm<sup>2</sup> for treatment 5 and 12.9 J/cm<sup>2</sup> for treatment 6

contaminant levels. The recirculation system was based on current depuration plants in the shellfish industry where animals like oysters, for example, are kept in recirculating water treated by stationary UV systems to reduce the level of the pathogens purged from the oysters. The initial concentration of pathogens in treatment tanks ranged from 10<sup>5</sup> to 10<sup>8</sup> CFU/ mL, simulating a highly contaminated environment. These levels are not usually found in seawater; however, the parameters were chosen to test the robustness and efficacy of PUV treatment in critical conditions, for example, a high level of bacteria can interfere with UV penetration depth. The preliminary tests to identify the voltage, number of pulses and duration of treatment were done to ensure high reduction and avoid excessive wear and tear of the PUV system.

The results from PUV treatment are presented in Fig. 4. Treatments 1 and 2, at 500 and 1000 V for 1 min, respectively, were the shortest and had fewer pulses (Table 1), resulting in microbe cell viability reductions ranging from  $0.08 \log_{10}$  for *E.* coli to 1.04  $\log_{10}$  for *B.* cereus (Fig. 4c) compared to the level in not treated (NT). The cell viability reduction (CVR) was not statistically significant between treatments 1 and 2 (Fig. 4a–e), showing the need for a prolonged treatment for improvement. Treatment 3 was prolonged to 10 min and the PUV operated at 500 V; however, reductions remained between 0 log<sub>10</sub> for *B. cereus* and 0.79 log<sub>10</sub> for *E. coli* compared to NT. A major increase in lethality was only observed when 1000 V pulses were applied for 10 min on treatment 4 (Fig. 4a), reducing the level of E. coli by  $2.62 \log_{10}$ . The relation between the number of pulses, duration and voltage was also observed for treatments 5 and 6 with a total of 60 pulses per minute applied in each treatment. Considering the results of all pathogens, lower CVR reductions from 2.97 log<sub>10</sub> to  $4.12 \log_{10}$  were obtained by applying 60 pulses per minute and 500 V when compared to the PUV unit operating at 1000 V and 60 pulses per minute, with a CVR from  $3.32 \log_{10}$  to



 Table 6
 Average peak area of amplicons bands from PCR of *E. coli*,

 S. aureus, B. cereus, C. albicans and S. typhimurium

	Pathogen	Average peak area		Ratio
		Non-treated	Treated	
1	E. coli	10,492.42	1692.26	6.2
2	S. aureus	10,111.01	0.00	-
3	B. cereus	11,770.74	796.48	14.78
4	C. albicans	12,607.45	0.00	-
5	S. typhimurium	8431.78	0.00	-

4.42  $\log_{10}$ . This fact is explained by the higher energy and UV output of 1000 V pulses delivering 20 J and 12.9 J/cm<sup>2</sup>, respectively, while 5 J and 3.2 J/cm<sup>2</sup> were delivered by 500 V pulses (Tables 1 and 2). CVR averages were statistically lower with 1000 V treatment than with 500 V for S. aureus (p = 0.0019), B. cereus (p = 0.0377) and S. typhimurium (p =0.0036). Therefore, the conditions of treatment 6 were chosen, and 1000 V, 60 pulses/min and 10 min employed in further assays. In terms of pathogen resistance to PUV treatment, the bacterium E. coli was the most affected. Reductions ranged from 0.08 (T1) to 4.42  $\log_{10}$  (T6) followed by *S. aureus* from 0.46 (T2) to 4.41 log<sub>10</sub> (T6), *S. typhimurium* from 0.15 (T1) to 4.19 (T6), *C. albicans* from 0.02 (T2) to 4.09 log<sub>10</sub> (T6) and B. cereus from 0 (T3) to 3.32 (T6). Therefore, the highest CVR reductions were observed for all pathogens when 1000 V, 60 pulses and 10 min treatment were applied.

The optimal conditions for the PUV treatment were verified, and CVR is presented in Fig. 4f. All CVR were statistically significant; the highest observed was 5.63  $\log_{10}$  for *S. aureus*, followed by 5.15  $\log_{10}$  for *C. albicans*, 5  $\log_{10}$  for *S. typhimurium*, 4.59  $\log_{10}$  for *B. cereus* and 4.55  $\log_{10}$  for *E. coli*. These results are in accordance with the same treatment

tested in preliminary assays as average CVR were not statistically different (p = 0.1044).

Log reduction of cell viability by UV output of treatments 1 to 6 is presented in Fig. 5. A UV output between 0.032 and 0.324 J/cm<sup>2</sup> resulted in the reduction of cell viability of approximately 1  $\log_{10}$  to all tested strains. *E. coli* and *S. aureus* were more sensitive to a UV output of 1.29 J/cm<sup>2</sup> and presented a 1.76 and 2.62  $\log_{10}$  reduction, respectively (Fig. 5). To achieve reductions from 3 to 4, a UV output from 3.24 to 12.9 J/cm<sup>2</sup> was required for all strains.

To ensure bivalve safety for consumers, alternative depuration methods and immune system stimulation approaches have been explored. A biological approach was reported by Jun et al. (2014) applying bacteriophage against V. parahaemolyticus in artificially contaminated oysters. Bacterial growth inhibition of  $7.4 \times 10^5$  CFU/mL after 12 h of phage application was reported. Probiotic bacteria can also be used to facilitate the depuration process and stimulate the shellfish immune system. Fajardo et al. (2014) obtained an isolate from a total of 365 bacteria from the shellfish digestive gland with in vitro antibacterial and antiviral activities. Ahmadi et al. (2015) reported a bio-physical method applying bacteriophages and high hydrostatic pressure for the inactivation of S. flexneri and V. cholerae in salmon and mussels. Complete inactivation of pathogens was achieved at 550 MPa for 5 min followed by the addition of a bacteriophage suspension at 10<sup>9</sup> PFU/mL.

The treatment system described in this study was developed to be easily installable at current depuration plants without the need for modifications. The true challenge for such complex alternatives such as those already mentioned (biological and bio-physical) emerges when upscaling. For a flow rate of 10 mL/s (36 L treated per hour) and 0.2 L of water, a real scale depuration tank of 500 L would require 14 h of operation. The usual levels of contaminants are



Fig. 6 Usual depuration system for bivalves

approximately 10<sup>1</sup>-10<sup>3</sup> CFU/mL, reaching 10<sup>4</sup> in class C areas, much lower than the 10<sup>8-9</sup> CFU/mL tested here, and would require a shorter period of treatment. Furthermore, additional PUV chambers can be set together in parallel to increase the depuration efficiency and also get benefited by operating with the current low-pressure UV system installed in most of the depuration plants. UV efficacy was reported by Garcia et al. (2015) with a 99% reduction of recombinant adenovirus and murine norovirus levels after 24 h of treatment by a low-pressure UV lamp and UV output of 6.4 J/cm<sup>2</sup>. Low-pressure efficacy in S. aureus and E. coli was also observed by Fitzhenry et al. (2021) with 5.3 and 6  $\log_{10}$  and 14 mJ/cm<sup>2</sup>. The economics of running costs and capital equipment compared with UV have yet to be calculated accurately; however, pilot PUV systems will cost more until full optimization is made. PUV can be considered as a bolt-on system for depuration for the shellfish in terms of sharing cooperative functionality. Shutdown of a processing plant and relocation of shellfish result in significant financial losses, affecting logistics and deliveries, requiring a re-validation of the depuration facility which takes weeks to complete, and massive damages to the reputation of the shellfish industry sector, causing foodborne infections in local and abroad consumers. PUV can be employed in terms of risk mitigation and food security.

The results of DNA integrity assessed qualitatively by gel electrophoresis and semi-quantitatively by analysing band pixels are presented in Table 6. No amplification was observed in S. aureus, C. albicans and S. typhimurium. We believe that PUV treatment has affected the availability and integrity of DNA while the same amount of template was used for PCR and quantified in highaccuracy fluorometer Qubit (Invitrogen), and it did not amplify. The disruptive and genotoxic effects of UV light are widely described. UV light induces the photoreactive potential of purines (adenine and guanine) and pyrimidines (cytosine, thymine and uracil) in DNA, triggering the formation of mutagenic DNA lesions and inactivating the replication, leading to a reduction in CVR. Oxidative stress is also observed in cells exposed to UV, leading to produce reactive oxygen species (ROS) that attack nucleotide pools and stress-mediated mutations (Ikehata and Ono, 2011).

Pullerits et al. (2020) investigated the effect of UV doses of 250, 400 and 600 J/m<sup>2</sup> on bacterial communities in water from a water treatment plant treated in a low-pressure UV system consisting of 10 rows of four UV lamps. The authors identified a long-term effect of UV irradiation that continued influencing the microbial dynamics after treatment and amplicons with greater guanine/cytosine contents were more resistant to UV treatment. In this study, *E. coli* and *B. cereus* had significantly lower amplification rates of 6.2 and 14.78fold reduction, respectively. It was not possible to correlate; however, the lower amplification with lower CVR shown in Fig. 4 as the same amount of DNA to PCR was used for all pathogens and resulted in amplifications similar to those observed in non-treated samples (Table 6).

B. cereus is Gram-positive spore-forming bacterium widely distributed in nature and has been frequently reported to be the causative agent of food poisoning (FSAI, 2016; Rowan, 2019). Bacillus endospores are tolerant of environmental stresses and are often used as bioindicator organisms for evaluating the efficacy of disinfection and sterilization modalities. Garvey and Rowan (2015) previously reported that pulsed UV can inactivate 1.5  $\log_{10} B$ . cereus and B. megaterium endospores in a flow-through system using a UV dose of 21.6  $\mu$ J/cm<sup>2</sup> and 6.46  $\mu$ J/cm<sup>2</sup>, respectively. A spore, B. cereus and its toxins can be heat resistant, requiring an extended pre-cooking procedure to reduce it to safe levels. Its resistance was observed by gel electrophoresis and detection of amplification bands post-PUV treatment and reduction in 4.59 log of B. cereus cells in a 10-min treatment. The PUV treatment method must be modified to better address the contaminant resistance. Taylor et al. (2020), for example, reported higher reductions of *B. cereus* spores when narrowing the UV to 222 nm, a peak also emitted in the present PUV protocol (Fig. 2). E. coli is a Gram-negative bacterium that is frequently associated with causing human and animal infections and can harbour multiple antibiotic resistance (AMR) genes. Zhang et al. (2017) reported its resistance when assessing the effect of a low-pressure UV treatment on antibiotic-resistant E. coli (AREC) isolated from a wastewater treatment plant. AREC required a higher UV dose of 20 mJ/cm<sup>2</sup> to cause reduction when compared to antibiotic-sensitive strains (8  $mJ/cm^2$ ).

The periodical monitoring of seawater and farm areas is important for safeguarding and for helping consumers helping the industry in decision-making (Fehrenbach et al. 2022). For shellfish producers, the water quality in these areas has been threatened by the expansion of cities and agriculture to shoreline. Irregular discharges of leachates and urban effluents represent a direct source of faecal contamination, mainly monitored and detected by E. coli levels, and this is a parameter considered when assessing the need for post-harvest manipulation. Other contaminants can also be present as chemicals and biotoxins that must also be periodically analysed. The location and boundaries of production areas are classified according to the level of faecal contamination. Regulation (EC) No. 854/2004 establishes the requirements for production areas in the European Union. Class A areas have the lowest risks and can be collected for direct human consumption if in accordance with health standards for live bivalves. Classes B and C cannot exceed 4600 and 46,000 E. coli per 100 g of flesh and intravalvular liquid, respectively, in the most probable number (MPN) test (EC, 2019). Classes B and C must be depurated at purification facilities to reduce the level of contaminants. For example, guidance for local action

on handling high *E. coli* results, pollution events and biotoxin results was released by Food Standards Agency (FSA) – England and Wales (2021). Levels of *E. coli* above 700, 18,000 and 46,000 per 100 g of shellfish for classes A, B and C, respectively, will require immediate action, from downgrading to temporary closure (FSA, 2021).

The usual system employed at purification facilities for bivalves is presented in Fig. 6. It consists of two main steps of washing and depuration. First, bivalves are transported by conveyor to a cleaning facility where they are graded according to size and serially cleaned in washing equipment. Then, the bivalves are placed into depuration tanks with clean water that recirculates in a purification system equipped with a UV system for disinfection. This system is usually effective for low contamination levels. The development of new alternatives is necessary to ensure efficient depuration, especially during winter when contaminants are higher and norovirus, for example, can reach 20,000 genome copies  $g^{-1}$  of digestive tissue (Rajko-Nenow et al. 2012). When this occurs, bivalves must be depurated for longer periods and/or also combined with higher water temperatures which can overload the facility period. Alternatively, they can be reintroduced into areas of cleaner water (EC, 2004; FSAI et al. 2018). These alternatives not only increase the production costs but also affect the production chain, possibly leading to delivery delays and affecting quality decrease to stressing of shellfish as well as increasing costs to the consumer. PUV treatment reached a bacterial removal of 5.63 log in highly contaminated artificial seawater after 10 min of treatment and could potentially support the removal of other contaminants such as NOV. It also supports the depuration of shellfish cultured in highly contaminated waters such as class C, where E. coli levels can reach 46,000 MPU per 100 g of flesh and intravalvular liquid (EC, 2019). Considering 46,000 MPU per 100 g as the E. coli load in shellfish, the PUV treatment can reduce it to safe consumption levels < 230 E. coli/100 g. However, it is necessary to consider the required incubation period for shellfish to depurate and shellfish load per batch. Figure 6 presents the usual depuration system for bivalves: (1) bivalves harvested from class B and C production areas or a purification centre or another dispatch centre. Production areas are classified according to the level of contamination, where A is the least and C the most contaminated; (2) bivalves must be kept at stressless conditions that support the filter-feeding act; (3) clean seawater collected and analysed before usage, it can be treated for safety measures; (4) water from depuration tanks is pumped through a UV chamber to inactivate pathogens; (5) treated water returns to depuration tank; (6) bivalves clean and readyto-eat (RTE) are immediately packed and kept at 4 °C; and (7) parameters that must be verified in the bivalves before commercialization.

In this research paper, we present a PUV system of broad efficacy reducing in  $4.5-5.6 \log_{10}$  the initial level of 5 common foodborne pathogens of different complexity and persistence. PUV treatment is a promising alternative to assist in the reduction of pathogen levels at depuration plants. Other pathogens, such as *Vibrio* sp. and norovirus, will also be tested in future studies to assess the PUV efficacy. The high disinfection efficiency and short treatment duration support the upscaling of PUV at the pilot-scale level, as well as the modelling of the treatment.

#### Conclusions

- The main parameters for PUV treatment such as voltage, number of pulses and duration successfully led to higher reductions in cell viability, as verified by posterior treatment.
- A UV output of 12.9 J/cm<sup>2</sup> at 1000 V, 60 pulses/min for 10 min of treatment reduced the levels of common foodborne pathogens to a maximum of 5.63 log<sub>10</sub> compared to no treatment.
- PUV treatment disrupted the pathogen DNA where *S. aureus, C. albicans* and *S. typhimurium* were not amplified by PCR; *E. coli* and *B. cereus* bands were detected by the software ImageJ, however, reduced by 6.2- to 14.8-fold compared to before treatment.
- Excellent depuration results were achieved with the same treatment conditions for all tested microorganisms without requiring specific UV conditions.
- The efficient depuration of different pathogens and simple set-up of the PUV system favours future testing at the pilot scale.

#### Appendix

See Table 7.

**Table 7** Spectrum output  $(\mu J/cm^2)$  at different distance from the flashlamp (cm) and bandwidth wavelengths

Dis- tance (cm)	< 300 nm	300–400 nm	400–500 nm	500–600 nm	600–700 nm	> 700 nm
10			630	370	444	1877
15	346	222	295	156	191	778
20	166	140	168	93	112	462
25	129	84	114	67	73	302
30	76	57	83	42	54	215

Dis-	< 300	300-400	400-500	500-600	600–700	> 700
tance (cm)	nm	nm	nm	nm	nm	nm
35	62	43	57	38	40	158
40	40	39	48	26	30	122
45	38	27	40	20	24	98
50	34	21	33	16	20	79

Authors' contributions GWF: conceptualization, methodology, investigation, writing—original draft preparation, and visualization. EM: methodology, writing—review and editing, supervision, and project administration. RP: conceptualization, methodology, writing—review and editing, supervision, and project administration. FC: writing review and editing. EC: conceptualization, writing—review and editing, supervision, and project administration. IM: conceptualization, methodology, writing—review and editing, supervision, and project administration. NR: conceptualization, writing—review and editing, supervision, and project administration.

**Funding** The research conducted in this publication was funded by the Irish Research Council under grant number GOIPG/2020/807 and Department of Agriculture, Food and the Marine under grant number PSS125.

**Data Availability** Data and materials for this study are available and will be provided to the journal upon request.

#### Declarations

Ethical approval Not applicable

Consent to participate Not applicable

Consent for publication Not applicable.

Competing interests The authors declare no competing interests.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

#### References

Ahmadi H, Anany H, Walkling-Ribeiro M, Griffiths MW (2015) Biocontrol of Shigella flexneri in ground beef and Vibrio cholerae in seafood with bacteriophage-assisted high hydrostatic pressure (HHP) treatment. Food Bioproc Tech 8:1160–1167. https://doi. org/10.1007/s11947-015-1471-6

- Ansari JA, Ismail M, Farid M (2019) Investigate the efficacy of UV pretreatment on thermal inactivation of Bacillus subtilis spores in different types of milk. Innov Food Sci Emerg Technol 52:387–393. https://doi.org/10.1016/j.ifset.2019.02.002
- Babič M, Gunde-Cimerman N, Vargha M et al (2017) Fungal contaminants in drinking water regulation? A tale of ecology, exposure, purification and clinical relevance. Int J Environ Res Public Health 14:636. https://doi.org/10.3390/ijerph14060636
- CEFAS (2022) Shellfish depuration advice. https://www.cefas.co. uk/services/laboratory-services-and-analysis/shellfish-depur ation-advice/. Accessed 23 Aug 2022
- Chen B-Y, Lung H-M, Yang BB, Wang C-Y (2015) Pulsed light sterilization of packaging materials. Food Package Shelf Life 5:1–9. https://doi.org/10.1016/j.fpsl.2015.04.002
- EC (2002) Regulation (EC) 178/2002. https://www.fsai.ie/uploa dedFiles/Legislation/Food\_Legisation\_Links/General\_Princ iples\_of\_Food\_Law/Consol\_Reg178\_2002.pdf. Accessed 5 Sep 2022
- EC (2004) Corrigendum to Regulation (EC) No 853/2004 of the European Parliament and of the Council of 29 April 2004 laying down specific hygiene rules for food of animal origin
- EC (2019) Regulation (EC) No 2019/627 of the European Parliament and the Council, 2019
- Fajardo P, Atanassova M, Garrido-Maestu A et al (2014) Bacteria isolated from shellfish digestive gland with antipathogenic activity as candidates to increase the efficiency of shellfish depuration process. Food Control 46:272–281. https://doi.org/ 10.1016/j.foodcont.2014.05.038
- FAO (2016) The State of World Fisheries and Aquaculture 2016. Contributing to food security ad nutrition for all. Rome. 200 pp. https://www.fao.org/3/i5555e/i5555e.pdf. Accessed 20 Dec 2022
- FAO (2022) The State of World Fisheries and Aquaculture 2022. Towards blue transformation. Rome, FAO. https://doi. org/10.4060/cc0461en
- FDA (1999) Pulsed light for the treatment of food. In: Code 21CFR179.41. https://www.accessdata.fda.gov/scripts/cdrh/ cfdocs/cfcfr/CFRSearch.cfm?fr=179.41 Accessed 5 Sep 2022
- Fehrenbach GW, Pogue R, Carter F et al (2022) Implications for the seafood industry, consumers and the environment arising from contamination of shellfish with pharmaceuticals, plastics and potentially toxic elements: a case study from Irish waters with a global orientation. Sci Total Environ 844:157067. https:// doi.org/10.1016/j.scitotenv.2022.157067
- Fitzhenry K, Clifford E, Rowan N, Val del Rio A (2021) Bacterial inactivation, photoreactivation and dark repair post flow-through pulsed UV disinfection. J Water Process Eng 41:102070. https://doi.org/10.1016/j.jwpe.2021.102070
- FSAI (2011) Staphylococcus aureus. https://www.fsai.ie/getmedia/ ae5bbe29-ecce-44a9-b75f-6ff61e14f37f/Staphylococcus-Facts heet-FINAL.pdf?ext=.pdf. Accessed 2 Jan 2023
- FSA (2021) Guidance for Local Action Groups (LAGs) on handling high E.coli results, biotoxin results and pollution events - classification and monitoring of live bivalve molluscs. https:// www.food.gov.uk/sites/default/files/media/document/finallocal-action-groups-guidance-lbm-sept2021.pdf. Accessed 25 Aug 2022
- FSAI (2016) Bacillus cereus. In: Microbial factsheet series. https://www.fsai.ie/WorkArea/DownloadAsset.aspxid= 10919#:~:text=Bacillus%20cereus%20is%20a%20Gram,capab le%20of%20facultative%20aerobic%20metabolism
- FSAI, SFPA, MI, BIM (2018) Interim guidance on the management of norovirus in oysters by shellfish producers. https://www.

fsai.ie/about\_us/industry\_fora/mssc/norovirus.html. Accessed 25 Aug 2022

- Garcia LAT, Nascimento MA, Barardi CRM (2015) Effect of UV light on the inactivation of recombinant human adenovirus and murine norovirus seeded in seawater in shellfish depuration tanks. Food Environ Virol 7:67–75. https://doi.org/10.1007/ s12560-014-9177-x
- Garvey M, Rowan N (2015) A pulsed light system for the disinfection of flow through water in the presence of inorganic contaminants. J Water Health 13:406–412. https://doi.org/10. 2166/wh.2014.176
- Holcomb DA, Stewart JR (2020) Microbial indicators of fecal pollution: recent progress and challenges in assessing water quality. Curr Environ Health Rep 7:311–324. https://doi.org/10. 1007/s40572-020-00278-1
- IKEHATA H, ONO T (2011) The mechanisms of UV mutagenesis. J Radiat Res 52:115–125. https://doi.org/10.1269/jrr.10175
- Jun JW, Kim HJ, Yun SK et al (2014) Eating oysters without risk of vibriosis: application of a bacteriophage against Vibrio parahaemolyticus in oysters. Int J Food Microbiol 188:31–35. https://doi.org/10.1016/j.ijfoodmicro.2014.07.007
- Lee R, Lovatelli A, Lahsen A (2008) Bivalve depuration: fundamental and practical aspects. FAO Fisheries Technical Paper. No 511. Rome, FAO. p 149
- Li J-J, Wang S-N, Qiao J-J et al (2020) Portable pulsed xenon ultraviolet light disinfection in a teaching hospital animal laboratory in China. J Photochem Photobiol B 207:111869. https:// doi.org/10.1016/j.jphotobiol.2020.111869
- Martinez-Albores A, Lopez-Santamarina A, Rodriguez JA et al (2020) Complementary methods to improve the depuration of bivalves: a review. Foods 9:129. https://doi.org/10.3390/foods9020129
- Matsumoto M, Sugiyama H, Maeda Y et al (2010) Marine diatom, Navicula sp. strain JPCC DA0580 and marine green alga, Chlorella sp. strain NKG400014 as potential sources for biodiesel production. Appl Biochem Biotechnol 161:483–490. https://doi.org/10.1007/s12010-009-8766-x
- McLeod C, Polo D, Saux J-C Le Guyader FS Le (2017) Final report: evaluating the effectiveness of depuration in removing norovirus from oysters. Seafood Safey Assessment Ltd and the French Research Institute for Exploitation of the Sea. https://www.food.gov.uk/sites/default/files/media/document/fs101068finrep.pdf Acessed 10 Oct 2022

- Pullerits K, Ahlinder J, Holmer L et al (2020) Impact of UV irradiation at full scale on bacterial communities in drinking water. NPJ Clean Water 3:11. https://doi.org/10.1038/s41545-020-0057-7
- Rajko-Nenow P, Keaveney S, Flannery J et al (2012) Characterisation of norovirus contamination in an Irish shellfishery using real-time RTqPCR and sequencing analysis. Int J Food Microbiol 160:105–112. https://doi.org/10.1016/j.ijfoodmicro.2012.10.001
- Rasband WS (2018) ImageJ. National Institutes of Health, Bethesda, Maryland, USA.http://imagej.nih.gov/ij
- Ricciardi EF, Pedros-Garrido S, Papoutsis K et al (2020) Novel technologies for preserving ricotta cheese: effects of ultraviolet and near-ultraviolet–visible light. Foods 9:580. https://doi. org/10.3390/foods9050580
- Rowan NJ (2019) Pulsed light as an emerging technology to cause disruption for food and adjacent industries – Quo vadis? Trends Food Sci Technol 88:316–332. https://doi.org/10.1016/j.tifs. 2019.03.027
- Samtech Pulsed UV (2018) System PUV-01 Operating Instructions, Glasgow
- Schneider KR, Cevallos J, Rodrick GE (2009) Molluscan shellfish depuration. Shellfish safety and quality. Elsevier, In, pp 509–541
- Shahbandeh M (2021) Global seafood market value 2020-2025. Statista https://www.statista.com/statistics/821023/globalseafood-market-value/#:~:text=Global%20seafood%20mar ket%20value%202020%2D2025&text=The%20global%20sea food%20market%20reached,336%20billion%20dollars%20by% 202025. Accessed 1 Sep 2022
- Takahashi S, Tomita J, Nishioka K et al (2014) Development of a prokaryotic universal primer for simultaneous analysis of bacteria and archaea using next-generation sequencing. PLoS One 9:e105592. https://doi.org/10.1371/journal.pone.0105592
- Taylor W, Camilleri E, Craft DL et al (2020) DNA damage kills bacterial spores and cells exposed to 222-nanometer UV radiation. Appl Environ Microbiol 86. https://doi.org/10.1128/AEM.03039-19
- Zhang CM, Xu LM, Wang XC et al (2017) Effects of ultraviolet disinfection on antibiotic-resistant *Escherichia coli* from wastewater: inactivation, antibiotic resistance profiles and antibiotic resistance genes. J Appl Microbiol 123:295–306. https://doi.org/10.1111/jam.13480

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Contents lists available at ScienceDirect

### Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

#### Review

# Implications for the seafood industry, consumers and the environment arising from contamination of shellfish with pharmaceuticals, plastics and potentially toxic elements: A case study from Irish waters with a global orientation



# Gustavo Waltzer Fehrenbach<sup>a,\*</sup>, Robert Pogue<sup>a,b</sup>, Frank Carter<sup>c</sup>, Eoghan Clifford<sup>d,e</sup>, Neil Rowan<sup>a,f</sup>

<sup>a</sup> Bioscience Research Institute, Technological University of the Shannon – Midlands Midwest, N37 F6D7, Ireland

<sup>b</sup> Post-Graduate Program in Genomic Sciences and Biotechnology, Catholic University of Brasilia, 71966-700, Brazil

<sup>d</sup> School of Engineering, National University of Ireland Galway, H91HX31, Ireland

<sup>e</sup> Ryan Institute, National University of Ireland Galway, Ireland

<sup>f</sup> Empower Eco™ Sustainability Hub, Technological University of the Shannon – Midlands Midwest, N37F6D7, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- Increased presence of pharmaceuticals, plastics and potentially toxic elements in shellfish waters
- Potential implications for industry and consumers arising from contaminated seafood
- Main contaminants and pressure points are reviewed offering potential solutions.
- Need for increased stakeholder awareness along with commensurate risk mitigation
- Risk assessment to safeguard shellfish industry and to protect food supply chain

#### ARTICLE INFO

Editor: Damia Barcelo

Keywords: Contamination Shelfish Plastics Pharmaceuticals Risk mitigation Potentially toxic elements



#### ABSTRACT

Shellfish are a rich source of minerals, B-vitamins and omega-3 to the human diet. The global population is expected to reach 9.6 billion people by 2050 where there will be increased demand for shellfish and for sustained improvements in harvesting. The production of most consumed species of shellfish is sea-based and are thus susceptible to *in situ* environmental conditions and water quality. Population growth has contributed to expansion of urbanization and the generation of effluent and waste that reaches aquatic environments, potentially contaminating seafood by exposure to non-treated effluents or inappropriately discarded waste. Environmental contaminants as microplastics (MP), pharmaceuticals (PHAR) and potentially toxic contaminants (PTE) are being identified in all trophic levels and are a current threat to both shellfish and consumer safety. Immunotoxicity, genotoxicity, fertility reduction, mortality and bioaccumulation of PTE are representative examples of the variety of effects already established in contaminated shellfish. In humans, the consumption of contaminated shellfish can lead to neurological and developmental effects, reproductive and gastrointestinal disorders and in extreme cases, death. This timely review provides insights into the presence of MP, PHAR and PTE in shellfish, and estimate the daily intake and hazard quotient for consumption behaviours. Alternatives approaches for seafood depuration that encompass risk reduction are addressed, to reflect state of the art knowledge from a Republic of Ireland perspective. Review of best-published literature revealed that

\* Corresponding author.

E-mail address: gfehrenbach@ait.ie (G.W. Fehrenbach).

http://dx.doi.org/10.1016/j.scitotenv.2022.157067

Received 30 March 2022; Received in revised form 9 June 2022; Accepted 26 June 2022 Available online 30 June 2022 0048-9697/© 2022 The Authors. Published by Elsevier B.V. This is an open access and a construction of the CC BY license (http://creativecommons.org/licenses/by/4.0/).

<sup>&</sup>lt;sup>c</sup> Coney Island Shellfish Ltd., Sligo F91 YH56, Ireland

MP, PHAR and PTE contaminants were detected in commercialised species of shellfish, such as *Crassostrea* and *Mytilus*. The ability to accumulate these contaminants by shellfish due to feeding characteristics is attested by extensive *in vitro* studies. However, there is lack of knowledge surrounding the distribution of these contaminants in the aquatic environment their interactions with humans. Preventive approaches including risk assessment are necessary to safeguard the shellfish industry and the consumer.

#### Contents

1.	Introduction	2
2.	Pharmaceuticals in shellfish	3
3.	Plastics and microplastics in shellfish	4
4.	Potentially toxic elements in shellfish	5
5.	Regulation and international guidelines	8
6.	Risk assessment and management for addressing contamination in shellfish	8
7.	Estimated daily intake and targeted hazard quotient of contaminants in shellfish	9
8.	Conclusions	2
CRe	diT authorship contribution statement	2
Decl	aration of competing interest	2
Ackı	nowledgements	2
Appe	endix A. Supplementary data	2
Refe	rences	2

#### 1. Introduction

Shellfish are aquatic invertebrates that have been used as a source of food by humans for over 160,000 years — and were one of the last additions to the human diet before the revolutionary culture of plants and domestication of animals (Marean et al., 2007). Shellfish can be classified as crustaceans or molluscs. There are approximately 40,000 known species of crustaceans, characterized by a segmented body protected by a hard shell, as observed in well-known species such as shrimp, crabs and lobsters (Hobbs, 2012; Prié, 2019). In the molluscs group, around 85,000 species such as slugs, snails, mussels, clams and octopuses, are grouped by distinctive characteristics including a muscular foot, and a fleshy mantle covering the viscera (Hobbs, 2012).

The importance of shellfish to the human diet and the rising demand for shellfish products has led to the development of production systems to increase availability and support harvesting from natural sources. In 2012, aquaculture (fish, shellfish, aquatic plants and molluscs) surpassed wild catches and production increased by 250 % between 2000 and 2015 (FAO, 2016; Hannah and Max, 2021). In 2018, global seafood production reached 179 million metric tons. Shellfish are a rich source of nutrients for the human diet. Benefits include the presence of a wide variety of essential amino acids, B-vitamins, and minerals in relatively high concentrations. They are an important source of omega-3,6 fatty acids, high levels of easily digestible protein, have low levels of fat and calories, and contain nutraceuticals with antioxidant, antiobesity, antidiabetes, and immunomodulatory activities (Venugopal and Gopakumar, 2017). However, the presence of contaminants in water and production plant systems can lead to harmful effects and illnesses, including disruptions in the human endocrine system, and immuno-genotoxicity in shellfish (Álvarez-Muñoz et al., 2015; Gadelha et al., 2019; Lacaze et al., 2015).

Most shellfish are cultivated in sea-based systems which rely on good water quality and environmental conditions. Geographical information, water quality reports from monitoring programs and risk modelling are important resources in the selection of appropriate sites for shellfish farms (Silva et al., 2011). Nonetheless, the discharge of industrial, domestic and agricultural effluents are unpredictable factors and result in chemical pollution, including pharmaceuticals, pesticides, metabolites, organic compounds and personal care products (Fairbairn et al., 2016). The growing population and agricultural demand for food has increased the use of pesticides and the discharge of domestic effluents, which also contaminate the seafood industry (Gadelha et al., 2019).

There is a constant requirement to remove persistent pharmaceuticals (PHAR) from the effluent matrix, including stimulants, anti-inflammatories, antibiotics and many other drugs (Hernando et al., 2006). If not removed, these PHAR will end up in the environment, reaching the shellfish and ultimately being consumed through the human diet. Potentially toxic elements (PTE) are naturally present in the environment at non-hazardous concentrations. However, these elements have been detected in toxic levels in industrialized areas such as ports, areas with mining activity, postenvironmental accidents, as well as in discharge of effluent into rivers and seas (Bayen et al., 2005). The persistence and high accumulative potential of PTE have been extensively detected in a wide variety of shellfish species (Wang et al., 2018). The intensive use of plastics and poor management of waste have also contributed to their dispersal in terrestrial and aquatic environments. Nowadays, plastic materials and microplastics (MP) can be found in the seabed and riverbed, suspended in water or deposited at coastal areas; studies have also demonstrated MP translocation to the circulatory system in mussels (Browne et al., 2008; Hantoro et al., 2019).

The interactions of the shellfish industry and environmental conditions are not only evident through contamination and pollution issues - but shellfish farms must also deal with the uncertainty arising from climate change. Ocean acidification and changing temperature are examples of processes that can impact plankton levels and shellfish metabolism, leading to complete decimation of cultures (Froehlich et al., 2018). The physiological effects of temperature and contamination on to the bivalve M. arenaria were reported by Greco et al. (2011). Increased temperature altered the response to a mixture of herbicides, enhancing the malondialdehyde content, cytochrome C oxidase and superoxide dismutase activity. Shellfish farms must also deal with natural predators, and the challenge of protecting the shellfish without affecting the ecological balance (Rheault, 2012). The survivability of shellfish farms is essential as a source of food and protection of wild stocks, and the dependence on a healthy environment for shellfish production stimulates the protection of cultured areas. Understanding the risks associated with shellfish production is the first step towards ensuring food safety and quality, therefore, a risk assessment (RA) strategy can represent an important resource to support the identification of risks, effects, and development of preventive strategies. The long coastline of Ireland, together with its importance as a marine habitat to many finfish and shellfish species, implies the need for new studies in preservation and protection. The Irish Marine Institute, an Irish state agency, has been providing essential monitoring of a wide range of contaminants in Irish waters. These studies were reviewed with available literature to diagnose the current situation in Ireland.

The objective of this review paper is to discuss the presence, in shellfish, of PHA, PTE and MP that represent a threat to the host and human safety, with insights into contaminants present in Irish waters, and state of the art. The regulations for maximum limits will be discussed in a way as to be used as a guide for researchers and for the shellfish farming industry. This paper will also address the concept and main steps involved in implementing a risk assessment to support the development of protective strategies, and also a guide to estimate the ingestion of contaminants and hazard quotient based on consumption profile and contaminant concentration.

#### 2. Pharmaceuticals in shellfish

The rate of increase of available medicines to treat and assist society is remarkable. Nonetheless, a wide variety of compounds produced for medical purposes have been identified as contaminants in different environments (Alygizakis et al., 2016; Mezzelani et al., 2018). The low degradation rate and high persistence of pharmaceutical (PHAR) compounds under environmental conditions represents a risk to aquatic environments, wildlife, and human safety (Almeida et al., 2020). Wastewater treatment plants are able to partially reduce the levels of PHAR from domestic effluent, however, the constant discharges and inadequate disposal methods increase their abundance in the environment, reaching aquatic organisms through the food chain, from phytoplankton, bivalves and crustaceans, to fish (Álvarez-Muñoz et al., 2015; Jjemba, 2018). The environmental concentration of PHAR in the marine environment can be found at ng  $L^{-1}$  to  $\mu g L^{-1}$ levels, and can lead to sub-lethal or chronic toxicity, while higher concentrations induce acute effects on reproduction and development (Capolupo et al., 2018; Fabbri and Franzellitti, 2016). When absorbed, PHAR compounds can lead to alterations in DNA, gene expression, antioxidant activity and immune responses, reflecting negatively on growth, behaviour and reproduction (Almeida et al., 2020; Lacaze et al., 2015).

Tiedeken et al. (2017) reported that pollution of European receiving waters with PHAR, such as with 17-beta-estradiol (a natural estrogenic hormone, E2), along with the pharmaceutically-active compounds diclofenac (an anti-inflammatory drug, DCL) and 17-alpha-ethynylestradiol (a synthetic estrogenic hormone, EE2) is a ubiquitous phenomenon. These authors noted that European surface water concentrations of DCL are typically reported below the proposed annual average environmental quality standard (AA EQS) of 100 ng  $L^{-1}$ , but that exceedances frequently occur. E2 and EE2 surface water concentrations are typically below 50 ng  $L^{-1}$  and 10 ng  $L^{-1}$ respectively, but these values greatly exceed the proposed AA EQS values for these compounds (0.04 and 0.035 ng  $L^{-1}$  respectively). Levels of these PHARs are frequently reported to be disproportionately high in EU receiving waters, particularly in effluents at control points that require urgent attention. Overall, it was found that DCL and EE2 enter European aquatic environments mainly following human consumption and excretion of therapeutic drugs, and by incomplete removal from influent at urban wastewater treatment plants (WWTPs) (Tiedeken et al., 2017). E2 is a natural hormone excreted by humans, which also experiences incomplete removal during WWTPs treatment (Tahar et al., 2018). Current conventional analytical chemistry methods are sufficiently sensitive for the detection and quantification of DCL, but not for E2 and EE2, thus alternative, ultra-trace, timeintegrated monitoring techniques such as passive sampling are needed to inform water quality for these estrogens (Tiedeken et al., 2017). DCL appears resistant to conventional wastewater treatment while E2 and EE2 have high removal efficiencies that occur through biodegradation or sorption to organic matter (Tahar et al., 2018).

The presence of PHAR in Irish surface waters has been reported by McEneff et al. (2014). Carbamazepine, diclofenac, gemfibrozil, mefenamic acid and trimetrophim were found in >85 % of the samples collected from two exposure sites in the West and East of Ireland, and two areas close to effluent treatment stations. These drugs were also detected at slightly lower concentrations in the marine surface water and can be accumulated in the shellfish, as observed for the marine mussels.

Beyond the importance highlighted by consumption, molluscs, especially bivalves, have also been used as bioindicators of pollution since they exhibit the required characteristics to evaluate an environment: they are filter feeders, show bioaccumulation, have a wide distribution, exhibit slow movement and have well-characterized life-cycles (Almeida et al., 2020; McEneff et al., 2014). Capolupo et al. (2018) address this characteristic by reporting the impact of contraceptive PHAR on molluscs in a study of in vitro inhibition of mussel (Mytilus galloprovincialis) gamete fertilization induced by 500 ng L<sup>-1</sup> of contraceptive 17- $\alpha$  ethinylestradiol (EE2), and also a reduction of fertility in sea urchins (*Paracentrotus lividus*) at 5000 ng  $L^{-1}$  of the lipid regulator gemfibrozil. Analyses of exposure during early phases are required to evaluate the effects of contaminants on animal development. EE2 and other active substances such as flame retardants, pesticides and fungicides are also considered emergent pollutants due to their endocrinedisruptive activity (Álvarez-Muñoz et al., 2015; Fehrenbach et al., 2021). Purdom et al. (1994) first registered the feminisation of male fish after exposure to treated effluent water, identifying the female hormone vitellogenin, naturally produced by the liver in females under use of estrogens. The occurrence of endocrine-disrupting compounds (EDC) is associated with human presence, and not restricted to contaminated areas or effluent discharges. As reported by Álvarez-Muñoz et al. (2015), in a study of 19 EDCs, 10 were present in bivalves and/or fish from the Tagus estuary (Portugal), Scheldt estuary (Netherlands), Po delta (Italy) and Ebro delta (Spain). The highest level of an EDC was observed in mullet, a fish found worldwide, containing 98.4 ng g<sup>-1</sup> dw of flame retardant Tris(2butoxyethyl) phosphate. Chiu et al. (2018) deployed the native mussels M. galloprovincialis, M. coruscus and P. viridis and semipermeable membrane devices in highly industrialized coastal areas to assess the presence of EDCs. In one month of exposure, the concentration of EDCs ranged from 99.4 to 326 ng  $g^{-1}$  dw and a correlation between the membranes and mussels was observed, confirming the potential of shellfish as indicators due to filter feeding ability.

In vitro studies with blue mussel (Mytilus edulis) using the psychotropic drugs paroxetine and venlafaxine at 1.5  $\mu g \ L^{-1}$  , generated DNA breaks in hemocytes, and caused immunomodulation, respectively; the antibiotic trimethoprim was genotoxic at 200  $\mu$ g L<sup>-1</sup> and immunotoxic at 20  $\mu$ g L<sup>-1</sup>, and the same effects were reported for erythromycin at concentrations higher than 20  $\mu$ g L<sup>-1</sup> (Lacaze et al., 2015). The ability of the organism to retain the tested compounds, together with high consumption of blue mussel worldwide, reveal the risk of consuming this filter feeder produced/ caught in contaminated environments. The presence of pharmaceuticals (PHAR) in the marine environment occurs uninterrupted (Capolupo et al., 2018), as demonstrated by Álvarez-Muñoz et al. (2015) with the identification of 15 out of 23 tested PHAR in Mytilus spp., Chamalea gallina and Crassostrea gigas collected in Portugal, Italy and Spain. The same authors identified 10 EDCs out of 20 compounds analysed in the mullet (L. aurata) and flounder (P. flesus) in Portugal and the Netherlands, respectively. On the Arabian Sea coast, the exposure to polycyclic aromatic hydrocarbons (PAH) and potentially toxic elements available in the water has induced DNA damage in the oyster Saccostrea cucullate (Sarker et al., 2018).

The wide variety of drugs present in different environments and shellfish species, and their effects, are described in Table 1.

PHARs can complex with organic compounds present in complex matrices as wastewater, resulting in higher persistence and accumulation in receiving waters. Wastewater treatment plants (WWTP) are responsible for reducing the impact of urban and industrial effluent through the removal of nutrients and contaminants. Mechanical, chemical, physical and biological processes are applied in the preliminary, primary, secondary and tertiary treatment steps (McEneff et al., 2015). PHAR are mostly removed in the tertiary treatment by biodegradation with adapted and capable bacteria in bioreactor and/or physical and mechanical as filtration and ozonation. Angeles et al. (2020) reported >95 % overall removal for 14 out of 11 PHAR compounds detected in WWTP, while biological treatment provided negative to <50 %. Higher removal efficiencies employing physical and mechanical as filtration are usually associated to increased treatment costs and maintenance.

Tahar et al. (2018) reported that of approximately 1000 WWTPs in the Republic of Ireland, only 16 have been monitored for PHARs. Diclofenac is

G.W. Fehrenbach et al.

#### Table 1

Pharmaceuticals and respective drugs detected in shellfish, marine water or tested in in vitro studies. The main effects observed are addressed with the duration of exposure.

	1 0				
Drug	Concentration and exposure time	Species	Environment or assay	Effect	Ref.
EE2	5 and 50 ng $L^{-1}$ ; 39 d	M. edulis	Marine	Reproductive system	(Blalock et al., 2018)
	5 to 425 ng L <sup>-1</sup> ; 20 h	C. gigas	Marine	Embryotoxicity	(Wessel et al., 2007)
	$1 \ \mu g \ L^{-1}$ ; 10 exp + 8 dep	C. virginica	Marine	Osmotic and oxidative stress, bioconcentration, metabolism	(Brew et al., 2020)
	5 ng L <sup>-1</sup> ; 12 d	E. complanata	Freshwater	Reproductive behaviour and biochemical parameters	(Leonard et al., 2017)
	5, 50 and 500 ng $L^{-1}$ ; 48 to 96 h	M. galloprovincialis, P. lividus and S. aurata	In vitro	Alteration in gamete fertilization and morphological abnormalities in <i>M. galloprovincialis</i> and <i>P. lividus</i> . Decreased survival of <i>S. aurata</i> .	(Capolupo et al., 2018)
IBP	1 and 100 $\mu g \ L^{-1};$ 7 d	C. gigas	In vitro	Increased transcription of certain genes and alterations in auxiliary enzymes and antioxidants	(Serrano et al., 2015)
	0.1 to 50 $\mu g$ $L^{-1};$ 14 d	R. philippinarum	In vitro	Decrease in lysosomal membrane stability; induced detoxification metabolism and oxidative stress; genotoxic effects	(Aguirre-Martínez et al., 2016)
	15 ng g <sup>-1</sup> ; 48 h	P. perna	Marine sediment	Development affected; decrease in lysosomal membrane stability	
DCF	100 and 600 $\mu$ g L <sup>-1</sup> ; 7 d	M. galloprovincialis	In vitro	Identification of 13 metabolites from DCF spiking	(Bonnefille et al., 2017)
	2.5 μg L <sup>-1</sup> ; 60 d	M. galloprovincialis	In vitro	IBU and KTP presented the same effects as DCF. Genotoxic, modulation of lipid metabolism, alterations of immunological parameters, and changes in cellular turn-over	(Mezzelani et al., 2018)
	$0.5 \ \mu g \ L^{-1}$ ; 96 h	R. philippinarum	<i>In vitro</i> flow through system	Mortality increases for larvae, shell alterations, increase in lactase activity	(Munari et al., 2016)
PAR	0.5, 5, 50 and 500 $\mu g  L^{-1};$ 96 $h$	Mytilus spp.	In vitro	Reduce of food intake, increase in glycogen levels, metabolic changes	(Piedade et al., 2020)
	40, 250 and 100 $\mu g$ $L^{-1};$ 7 d	M. edulis	In vitro	Alterations in gene expression affecting the reproductive system	(Koagouw and Ciocan, 2020)
	1 and 100 $\mu$ g L <sup>-1</sup> ; 1, 4 and 7 d	C. gigas	In vitro	Alterations at transcriptional level	(Bebianno et al., 2017)
PROP	500, 5000 and 50,000 ng L <sup>-1</sup> ; 48 to 96 h	M. galloprovincialis, and S. aurata	In vitro	Fertilization and survival reduction in <i>P. lividus</i> and morphological abnormalities in <i>M. galloprovincialis</i>	(Capolupo et al., 2018)
TCS	75 ng g <sup>-1</sup> ; 24 h	L. variegatus; M. charruana	Marine sediment	Development affected; decrease in lysosomal membrane stability	(Pusceddu et al., 2018)
	0.007, 0.014, and 0.036 mg L <sup>-1</sup> ; 20 d	C. catla	In vitro	Enhanced GOT, GPT, and GST enzyme activity; physiological alterations	(Hemalatha et al., 2019)

EE2: 17alpha-ethinylestradiol; IBP: ibuprofen; DCF: diclofenac; KTP: ketoprofen; PAR: paracetamol; PROP: propranolol; TCS: triclosan; GOT: glutamate oxaloacetate transaminase; GPT: glutamate pyruvate transaminase; GST: glutathione-S-transferase; dep: depuration; exp: exposure.

found in treated effluents from 5 WWTPs at levels at least as high as other European WWPTs, and sometimes higher. Measurements of E2 and EE2 in WWPT effluents were rare and effluents were more often evaluated for total estrogens; these PHARs were generally not detected using conventional analytical methods because of limits of detection being too high compared to environmental concentrations and WFD environmental quality standards (Tiedeken et al., 2017; Tahar et al., 2018). Tahar et al. (2018) reported a correlation between occurrence of these PHAR and regional drug dispensing data in Ireland. However, mapping the aforementioned data onto appropriate river basin catchment management tools will inform predictive and simulated risk determinations to inform investment in infrastructure that is necessary to protect rivers and beaches and economic activities that rely on clean water.

Adams et al. (2002) spiked sterile river water with common antibiotics to evaluate the effectiveness of conventional water treatment processes on antibiotic removal. They reported an effective removal when applying chlorine, ozone, powdered activated carbon and reverse osmosis, however, only powdered activated carbon (PAC) effectively removed the antibiotics at typical plant dosages of PAC and at a frequency that increases the treatment costs. However, matrices with higher concentrations of PHAR, such as urban wastewater, can make removal of PHAR more challenging. In shellfish, (McEneff et al., 2013) investigated the concentration of PHAR after a cooking process, spiking Mytilus spp. and artificial seawater with several PHAR at concentrations 1000 times greater than those observed in marine surface waters. The authors observed an increase of pharmaceutical residues in seawater and shellfish tissue after cooking by steaming. Currently, there is no standard method for PHAR removal in shellfish and seawater, as this depends on several parameters. The continuous water quality monitoring at shellfish farms and surroundings is essential to monitor the PHAR level. An early detection can reduce the bioconcentration and possibly the natural depuration of contaminants in controlled conditions.

#### 3. Plastics and microplastics in shellfish

Plastics are polymeric materials of highly diverse composition and uses, moulded under specific pressure and temperature to provide adequate resistance and performance (FAO, 2018). The demand for plastic materials has been constantly increasing over the past decades and reached a worldwide production of 368 million tonnes in 2019 (PlasticsEurope, 2020). The intensive usage of plastic materials and inadequate management to control its life cycle have led to dispersal in the terrestrial and aquatic environments, expanding to shorelines, open ocean and reaching the deep seas. It has been estimated that between 4.8 and 12.7 million tons of plastic entered the oceans in 2010, not including abandoned equipment and fishing nets (Jambeck et al., 2015). Contamination occurs on different levels, where low-density plastics are dominant and float at the top layers, and high-density plastics combined with biofouled low-density material sink and remain on the sea bottom (Andrady, 2011; Bellas et al., 2016). The physical properties of plastic materials, such as resistance, durability, light weight, and generally low cost make them appropriate for use in ropes, cages, tanks and general materials associated with the fish and shellfish industry. However, this can also threaten shellfish and finfish safety as these materials are not completely inert, where degradation under environmental conditions such as ultraviolet radiation and/or physical abrasion can release plastic fibres and particles in even higher concentrations than external contaminants (Hantoro et al., 2019).

The ingestion of plastic materials occurs involuntarily in finfish and shellfish; however, filter feeders are more susceptible to accumulating higher concentration of plastics than finfish due to non-selective feeding. Plastics can reach shellfish directly as primary plastic of specific size and composition, or as secondary materials of different sizes and composition, products of exposure to environmental conditions and ultraviolet radiation (Hantoro et al., 2019). The secondary plastics are mostly present as fibres,

powders, and pellets, classified by size as meso (>5 mm), micro (1000 nm) and nanoplastics (1-100 nm). In invertebrates, once the microplastic is ingested it can be transferred to body tissues though the gut epithelial lining or egested in faeces (Browne et al., 2008). To investigate the presence of plastic materials in the oyster C. gigas and mussel M. edulis, van Cauwenberghe et al. (2015) performed an overnight digestion of shellfish wet tissue using nitric acid (69 %) followed by a boiling step, where samples from a mussel farm and local supermarket resulted in 0.36 and 0.47 particles/g of tissue, respectively. Using the same digestion protocol, van Cauwenberghe and Janssen (2014) detected microplastics (MP) in all animal tissues and faeces collected from M. edulis and A. marina and an increase of energy consumption in the exposed M. edulis when compared to control, linked to increase of stress. Browne et al. (2008) used M. edulis as a model organism to investigate the route and biological consequences of ingesting microparticles of polystyrene, a polymer that reached a worldwide production of 15.61 million metric tons in 2021 (Statista, 2020). They identified a correlation between smaller particles and high accumulation in tissues. In 3 days, particles moved from the gut to the circulatory system and remained for 48 h, reaching their greatest level on the 12th day. Ingested microplastics can also work as a carrier of other contaminants to seafood and not only a direct contaminant. Polychlorinated biphenyls and bacteria were reported on polystyrene surface of eight of the 14 species analysed by Carpenter et al. (1972).

The accidental ingestion of plastics due to similarity to food can be summed up with the contamination of lower trophic levels and subsequent trophic transfer, as observed by the same authors in zooplankton samples. Crustaceans, for example, are non-selective and can have a wide variety of food sources. This fact might explain the presence of microplastics in the gastrointestinal tract of the crustacean *Nephrops norvegicus*, collected in Irish prawn grounds and presented by Hara et al. (2020). Fibres were predominant (98 % of plastic constituent) in the collected plastic and a positive correlation was observed between the prawn carapace condition and microplastic. Devriese et al. (2015) reported the natural ability for the shrimp *Crangon crangon* to ingest microplastics, reaching an uptake of 0.68 g w w<sup>-1</sup>. The role of shrimps in the trophic transfer of microplastics was also discussed as the variable diet of shrimps and the importance as food for a large range of predators contributes to transfer.

On the Irish continental shelf, an extension from the coastline of 200 nautical miles under the sea, microplastics were detected within superficial sediments and bottom water (Martin et al., 2017). The depth and environmental conditions at these regions slow the breakdown of plastic materials, contributing to their persistence and accumulation. Woodall et al. (2014) reports the high concentration of microplastics in samples collected from the North Atlantic Ocean, Mediterranean Sea and SW Indian Ocean. Microplastics were mostly in the form of fibres of 2–3 mm in length and <0.1 mm in diameter in deep-sea sediments and coral samples, regions widely habited by commercially explored finfish and shellfish. Lusher et al. (2018) reported the presence of macro-debris and micro-debris in Irish cetaceans, which shows trophic transfer as another route of contamination. The authors also observed a higher incidence of macrodebris ingestion in deep-diving species, however, it was not possible to investigate the relationship to habitat. Ireland is not located in accumulation zone of debris, however due to an extension of 7524 km, a combination of weather, season and geographical position can lead to high accumulation levels (Lusher et al., 2018). Concentration of microplastics such as fibres, fragments, films and filaments detected in shellfish species at different locations are presented in Table 2.

An alternative for reducing the levels of plastic in shellfish is a longer depuration process keeping the shellfish in tanks with recirculating water that pass through a treatment system, allowing the excretion of plastic materials from the gut. Depuration processes are designed to provide optimum conditions such as temperature, salinity and oxygenation, reducing the impact on organoleptic properties and increasing the filtering activity. van Cauwenberghe and Janssen (2014) reported a reduction in the levels of particles in M. edulis and C. gigas after depurating for three days. Alternatively, a change in the mode of consumption can be made by removing parts of the shellfish or cooking. However, this can impact on consumer experience and affect important characteristics to shellfish as freshness. Daniel et al. (2020) suggested the consumption of F. indicus in whole dried by cooking or peeled form to reduce the exposure to microplastics owing to the concentration in the foregut and midgut. However, more studies are needed to improve the depuration and cleaning processes while preserving the organoleptic properties.

#### 4. Potentially toxic elements in shellfish

Potentially toxic elements (PTE) are pollutants of high persistence and toxicity that contaminate a wide variety of ecosystems and organisms (Wang et al., 2018). The term PTE describes metals and metalloids that naturally present in soil, and some are considered essential nutrients, required at low doses and participating in biochemical and physiological functions (Tchounwou et al., 2012). Others, such as lead, mercury, and cadmium, are on a list of ten chemicals of major public concern (World Health Organization, 2020a). Anthropogenic activities such as general industrial processes, mining, agriculture, and domestic use of PTE have been associated with environmental contamination and human exposure (Tchounwou et al., 2012). Marine and freshwater ecosystems are highly affected by PTE, where shellfish can accumulate these compounds to toxic concentrations, releasing them back into the water after death, or transporting them to the next trophic level (Djedjibegovic et al., 2020). The ability of shellfish species to bioaccumulate chemicals, their long-life span, and high density, have supported their use as local bioindicators of contamination, where the determination of PTE is determined directly from the shellfish tissue (El-Shenawy et al., 2016).

The most common method to verify the presence of metals in shellfish is by total concentration, extracting all the contaminants from tissues and analysing by specialized techniques; they can also be determined in terms of bioacessibility, which represents the fraction of contaminants that is

#### Table 2

Concentration of microplastics in shellfish species and major findings.

Microplastic	Concentration	Specie	Comments	Location	Reference
Fibres, fragments and films	1.75 $\pm$ 2.01 items per shrimp	N. norvegicus	Most common range of 1 to 2 mm	Irish waters	(Hara et al., 2020)
Filaments	83 % contained plastic filaments	N. norvegicus	Fishing waste was probably the source for microfilaments	Clyde Sea	(Murray and Cowie, 2011)
Fibres	$0.68 \pm 0.55$ microplastics/g ww	C. crangon	The analysed specie was able to ingest plastic particles from natural habitat	Southern North Sea and Channel area	(Devriese et al., 2015)
Microplastics	$0.2 \pm 0.3$ microplastics/g; $1.2 \pm 2.8$ particles/g	M. edulis; A. marina (annelid)	Microplastics were present in all organisms collected in the field	French, Belgian and Dutch North Sea coast	(van Cauwenberghe et al., 2015)
Microplastics (LDP, HDP and POS)	$0.36 \pm 0.07$ particles/g (ww); plastic load 0.47 $\pm$ 0.16 particles/g ww	M. edulis; C. gigas	Annual dietary exposure of 11,000 microplastics	North Sea and Atlantic Ocean	(van Cauwenberghe and Janssen, 2014)
Microplastics	128 microplastics (83 % were fibres)	F. indicus	Contamination significantly higher in Monsoon season	Coastal waters off Cochin, India	(Daniel et al., 2020)

g: gram; ww: wet weight;

#### G.W. Fehrenbach et al.

#### Table 3

Potentially toxic elements quantified in shellfish and seawater. General comments were added to address the main findings of studies.

Metal	Concentration	Specie	Comments	Reference
Cd	2.0–12.4 $\mu$ g g <sup>-1</sup> dw	<i>C. angulate</i> oysters	Bio-accessibility from 13 to 58 %	(He et al., 2016)
	$1.7-5.1 \ \mu g \ g$ 0.644 mg kg <sup>-1</sup> ww	D. gahi	TWI would be reached in a 70 kg adult by a weekly consumption of 274 g of <i>M</i> .	(Djedjibegovic et al., 2020)
	0.015	P. monodon	edulis or 272 g of D. gahi	
	0.002	P. indicus		
	0.062 0.17/0.1/0.125 µg g <sup>-1</sup> www	M. edulis P. decussatus	Concentration in three different locations in the Lake Timesh: PTWI: 0.01. 0.06 up	(El Shonaway et al. 2016)
	0.23/0.23/0.23	P. undulate	$kg^{-1}/day$ ) - below the PTWI of 7 µg kg <sup>-1</sup> /week for both groups of consumers	(EF-SHellawy et al., 2010)
	$0.635 \text{ mg kg}^{-1} \text{ ww}$	Bivalve molluscs		(Wang et al., 2018)
	0.41 mg kg <sup><math>-1</math></sup> dw	O. glomerate	Concentration in seawater 0.08 mg kg <sup><math>-1</math></sup> dw	(Yuan et al., 2020)
	0.31	P. viriais C. scripta		
	0.06	M. edulis		
	1.14	G. divaricatum		
	0.66	B. virescens	Exceeded the 1981/2006/EC may limit in outer	(Irich Marine Institute 2018)
	0.02	Clams	Exceeded the 1001/2000/EC max mint in byster	(mish Marine filstitute, 2010)
	0.09-1.25	Oysters (Pacific and native)		
Cu	$173-2212 \ \mu g \ g^{-1} \ dw$	C. angulate oysters	Bio-accessibility from 42 to 95 %	(He et al., 2016)
	$1063-5314 \ \mu g \ g^{-1}$	C. hongkongensis Bivalve molluscs		(Wang et al., 2018)
	$12.04 \text{ mg kg}^{-1} \text{ dw}$	O. glomerate	Concentration in the seawater 1.48 mg kg $^{-1}$ dw	(Yuan et al., 2020)
	1.58	P. viridis		
	0.9	C. scripta M. edulis		
	1.1	G. divaricatum		
	1.32	B. virescens		
	2.53/3.79/3.16 μg g <sup>-1</sup> ww	R. decussatus	Concentration in three different locations in the Lake Timsah; EDI was lower than	(El-Shenawy et al., 2016)
	$0.38-1.32 \text{ mg kg}^{-1} \text{ ww}$	Blue mussels	the FAO/ WHO guidelines, meaning that is sale for consumption,	(Irish Marine Institute, 2018)
	0.77	Clams		
7-	1.81-22.2	Oysters (Pacific and native)	Die eenersikilike furm 12 to 50 0/	(IIe et al. 2016)
ZII	$2103-7021 \text{ µg g}^{-1}$	C. hongkongensis	BIO-accessibility from 13 to 58 %	(He et al., 2016)
	154.29 mg kg <sup><math>-1</math></sup> ww	Bivalve molluscs		(Wang et al., 2018)
	10.59 mg kg <sup>-1</sup> dw	O. glomerate	Concentration in the seawater 12.25 mg kg $^{-1}$ dw	(Yuan et al., 2020)
	11.95 9.79	P. viridis C. scripta		
	10.36	M. edulis		
	10.64	G. divaricatum		
	11.32 9.87-21 mg kg <sup>-1</sup> ww	B. virescens		(Irish Marine Institute 2018)
	11.4	Clams		(mon marine institute, 2010)
	91.8-448	Oysters (Pacific and native)		
Hg	$0.02 \text{ mg kg}^{-1} \text{ ww}$	D. gahi M. adulia		(Djedjibegovic et al., 2020)
	0.058	P. monodon		
	0.037	P. indicus		
	$0.01 \text{ mg kg}^{-1} \text{ ww}$	Bivalve molluscs		(Wang et al., 2018)
	0.02 mg kg <sup>-</sup> dw	O. glomerate P viridis	Concentration in seawater 0.007 mg kg <sup>+</sup> dw	(Yuan et al., 2020)
	0.02	C. scripta		
	0.01	M. edulis		
	0.02	G. divaricatum B. virescens		
	0.01–0.03 mg kg <sup>-1</sup> ww	Blue mussels		(Irish Marine Institute, 2018)
	0.01	Clams		
DP	0.01-0.04	Oysters (Pacific and native)	TWI would be reached in a 70 kg adult by a weakly consumption of $0.74 \times 10^{-1}$	(Diadiibagovic et al. 2020)
rU	0.014 WW	P. monodon	edulis or 272 g of D. gahi	(Djeujibegović et al., 2020)
	0.013	P. indicus		
	0.161	M. edulis	a	(m. 1. 0000)
	0.47 mg kg <sup>-</sup> dw 0.48	O. glomerate P viridis	Concentration in seawater 0.77 mg kg <sup>-</sup> dw	(Yuan et al., 2020)
	0.34	C. scripta		
	0.44	M. edulis		
	0.24	G. divaricatum B. virescens		
	0.23 mg kg <sup>-1</sup> ww	Bivalve molluscs		(Wang et al., 2018)
	0.05–0.37 mg kg <sup>-1</sup> ww	Blue mussels		(Irish Marine Institute, 2018)
	0.11	Clams		
Cr	0.02-0.13 $0.16 \text{ mg kg}^{-1} \text{ ww}$	Bivalve molluscs		(Wang et al., 2018)
	$0.76 \text{ mg kg}^{-1} \text{ dw}$	O. glomerate	Concentration in seawater 0.72 $\text{ mg kg}^{-1} \text{ dw}$	(Yuan et al., 2020)
	0.78	P. viridis		

Table 3 (continued)

Metal	Concentration	Specie	Comments	Reference
	1.79	C. scripta		
	1.08	M. edulis		
	0.78	G. divaricatum		
	0.23	B. virescens		
	0.08–0.35 mg kg <sup>-1</sup> ww	Blue mussels		(Irish Marine Institute, 2018)
	0.07	Clams		
	0.04-0.9	Oysters (Pacific and native)		
Ni	$0.26 \text{ mg kg}^{-1} \text{ ww}$	Bivalve molluscs		(Wang et al., 2018)
	$0.94/0.9/0.92 \ \mu g \ g^{-1} \ ww$	R. decussatus	Concentration in three different locations in the Lake Timsah; EDI between 0.01	(El-Shenawy et al., 2016)
	1.04/1.1/1.07	P. undulate	and 1.26 $\mu$ g kg <sup>-1</sup> /day for both bivalves	
	0.08–0.26 mg kg <sup>-1</sup> ww	Blue mussels		(Irish Marine Institute, 2018)
	0.05	Clams		
	0.03-0.1	Oysters (Pacific and native)		
As	1.27 mg kg <sup>-1</sup> ww	Bivalve molluscs		(Wang et al., 2018)
	0.48 mg kg <sup>-1</sup> dw	O. glomerate	Concentration in seawater 0.77 mg kg <sup>-1</sup> dw	(Yuan et al., 2020)
	0.50	P. viridis		
	0.29	C. scripta		
	0.22	M. edulis		
	0.50	G. divaricatum		
	0.39	B. virescens		
	$1.03-1.82 \text{ mg kg}^{-1} \text{ ww}$	Blue mussels		(Irish Marine Institute, 2018)
	1.65	Clams		
	1.12-2.69	Oysters (Pacific and native)		

ww: wet weight; dw: dry weight.

released from the food matrix and interacts with the human digestive tract (Brandon et al., 2006; He et al., 2016). Quantification by analytical techniques can be diverse. Wang et al. (2018) applied atomic absorption spectrophotometry to measure Cd, Pb, Cr, Ni, Cu, Zn, and atomic fluorescence spectrophotometry for Hg and As, while Djedjibegovic et al. (2020) analysed Cd and Pb by graphite furnace atomic absorption spectrometry and Hg by flow-injection cold vapour atomic absorption spectrometry. In vitro models can also be used to provide analysis of specific routes of consumption such as ingestion and sucking. As an example, He et al. (2016) investigated the bio-accessibility of Cd, Cu and Zn in Crassostrea angulate (green oyster) and Crassostrea hongkongensis (blue oyster). The oysters were collected from a station along the Jiulong River Estuary, China, and digested by an in vitro process that mimics the mouth, stomach and small intestine of humans. The total concentration was higher than maximum levels established by USEPA, and the bioacessibility range from 13 to 95 % highlighted the potential threat to human health and safety. The authors also found a correlation between the shellfish tissue colour and concentration of Cu, Cd and Zn.

The main exposure routes of humans to PTE occurs by ingestion of contaminated food and water. Due to their resistance to relatively polluted environments, shellfish species can bioaccumulate and biomagnify PTE through the food chain. PTE toxic effects are associated with gastrointestinal and kidney dysfunction, vascular damage, birth effects, bloody diarrhoea, and many other disorders (Balali-Mood et al., 2021). When exposed to low PTE doses, the effects can be more complicated to diagnose and correlate with source of intake after a long period of exposure. Mazumdar et al. (2011) found that lead exposure in childhood predicts intellectual function in young adulthood. Therefore, a long-term diet containing low doses can be a silent threat. In shellfish, the exposure to low doses of PTE can lead to incorporation of Pb and Zn instead of Ca, as they are incorporated by the same pathways. Stewart et al. (2021) recently reported the correlation between the weakening of shell strength in *Pecten maximus* and metal pollution in sediments, increasing the vulnerability of the bivalve in the environment.

In Irish Coastal Waters, the levels of PTE in waters and shellfish have been monitored mainly by state agencies such as the Irish Marine Institute (Ireland), and the Centre for Environment, Fisheries and Aquaculture (United Kingdom). A review of the contaminant status of the Irish Sea was reported by Kenny et al. (2005). This 100,000 km<sup>2</sup> area is bounded by Scotland, England, Wales and Ireland. In dredged samples collected from 1995 to 2005, zinc, lead and arsenic were present at higher concentrations than other PTE such as Cu, Ni and Cr. A variation of PTE levels was also observed, allowing the researchers to associate the presence of these chemicals in the environment as a consequence of society, industry and weather particularities. In mussel samples collected from inshore sites of the Irish Sea between 1999 and 2001, the authors also identified levels of Cd (208–465.69  $\mu$ g kg<sup>-1</sup> wet weight), Cu (923–1987  $\mu$ g kg<sup>-1</sup> wet weight), Pb (352–1668 µg kg<sup>-1</sup> wet weight), Zn (12,673–37,527 µg kg<sup>-1</sup> wet weight), Hg (18–70  $\mu$ g kg<sup>-1</sup> wet weight) and Ag (50–260  $\mu$ g kg<sup>-1</sup> wet weight). The concentrations were below the maximum acceptable limits. Another study by the Irish Marine Institute (2018) quantified several metals in bivalve molluscs collected in the Irish seawater in 2015 - from ovsters, blue mussels and clams; only Cd in oysters was above the maximum limit established by 1881/2006/EC. The determination was also carried in seawater and the levels of metals complied with the maximum limits established by the Statutory Instruments and Shellfish Waters Directive. However, the presence of these elements at non-natural levels alerts to an intoxication risk as the accumulation of PTE will depend on consumption habits and local body weight, discussed in the risk assessment.

In Table 3 are presented the main potentially toxic elements and their respective concentrations in shellfish tissues, seawater and after *in situ* studies, during bioacessibility and toxicology assays.

Han et al. (1993) reported one of the first depuration studies of metals in shellfish. *C. gigas* and *M. smarangdium* were collected from a region contaminated with copper and zinc and transferred to natural clean seawater for depuration. A 351  $\mu$ g g<sup>-1</sup> day<sup>-1</sup> depuration rate of cooper was reported for the first 6 days, approximately 67 % of the total in *C. gigas*, and only 36 % of accumulated zinc. Longer depuration periods are associate with increased costs of treatment and infrastructure. FAO (2008) describes the fundamentals and practical aspects of bivalve depuration. A minimum depuration period of 42 h is determined and extended according to contamination/removal levels.

Anacleto et al. (2015) evaluated the effect of depuration process on the levels of S, Cl, K, Ca, Fe, Zn, Br, Cu, Se, Rb and Sr in the bivalves *R. philippinarum, M. galloprovincialis*, and *S. plana*, collected at different sites of Tagus estuary. Depuration was performed in recirculating tanks set according to European Guidelines (European Commission, 2004). The level of toxic elements (Hg, Cd, Pb and As) after depuration were reduced in *R. philippinarum* (Hg, Cd, Pb and As), *M. galloprovincialis* (Pb) and *S. plana* (Pb).

Currently, there is no consensus about mitigation methods for removing/reducing PTE in shellfish. More studies are necessary to address effective technologies and factors associated with reduction and quality. The continuous monitoring of PTE levels in the full shellfish farming environment is still the main recommendation.

#### 5. Regulation and international guidelines

In order to protect consumer health and ensure product quality, international commissions have elaborated guidelines and limits for a wide diversity of contaminants. These limits were established to balance the potential benefit of consuming food with even a low concentration of contaminants that would not lead to serious effects, and also to keep the production/depuration costs affordable. European Commission Regulations and the U.S. Food and Drug Administration are the main references for maximum levels of contaminants (Table 4). The limits established by each regulatory agency also set the tolerance for legal actions such as recall of products from the market, shut down of production plants, and fines.

# 6. Risk assessment and management for addressing contamination in shellfish

The analysis of risks associated with an activity, or a hazard is a holistic approach involving all available knowledge and data that can support a prediction mode (Tahar et al., 2017). Besides the diversity of approaches, the risk assessment (RA) modelling is based on the main steps of hazard identification, exposure assessment, and hazard characterization, resulting in a broad risk characterization (Oscar, 2012; Tiedeken et al., 2017). RA can be applied to different environments, also delineating critical control points along the production and supply chain for industry — every stage of processing, production, storage, marketing and consumption is considered in an industrial environment.

Table 4

Environmental chemical contaminants and tolerable levels allowed in shellfish acc	cording to international ager	icies.
---	-------------------------------	--------

Substance	Group/species	Maximum level (concentration of contaminant per wet weight)
European Commission Regulation <sup>1</sup>		
Lead	Crustaceans	$0.5 \text{ mg kg}^{-1}$
	Bivalve molluscs	$1.5 \text{ mg kg}^{-1}$
	Cephalopods	$0.3 \text{ mg kg}^{-1}$
Cadmium	Bivalve molluscs	$1 \text{ mg kg}^{-1}$
	Crustaceans	$0.5 \text{ mg kg}^{-1}$
	Cephalopods	$1 \text{ mg kg}^{-1}$
Mercury	Crustaceans	$0.5 \text{ mg kg}^{-1}$
Dioxins and PCBS	Crustaceans	3.5  pg/g (sum of dioxins).
		6.6 (sum of dioxins anddioxin-like PCBs):
		75 ng g <sup>-1</sup> (PCB28, PCB52, PCB101, PCB138, PCB153 and PCB180)
Benzo(a) pyrene benzo(a) anthracene benzo(b)	Smoked crustaceans	$2.0 \text{ µg kg}^{-1}$
fluoranthene and chrysene		210 48 48
nuoranaione ana empoene	Smoked bivalve molluscs	$6 \text{ µg kg}^{-1} \text{Benzo}(a)$ pyrene.
		$35 \text{ µg kg}^{-1}$ Sum of benzo(a)-pyrene, benz(a)anthracene, benzo(b)
		fluoranthene and chrysene)
U. S. Food and Drug Administration <sup>2</sup>		
Aldrin/Dieldrin	Shellfish and finfish	0.3 ppm
Chlordane	Shellfish and finfish	0.3 ppm
Chlordecone	Shellfish and finfish	0.3 ppm
	Crabmeat	0.4 ppm
DDT, TDE, DDE	Shellfish and finfish	5 ppm
Heptachlor/Heptachlor Epoxide	Shellfish and finfish	0.3 ppm
Mirex	Shellfish and finfish	0.1 ppm
Arsenic	Crustacea	76 ppm
	Molluscan bivalves	86 ppm
Cadmium	Crustacea	3 ppm
	Molluscan bivalves	4 ppm
Chromium	Crustacea	12 ppm
	Molluscan bivalves	13 ppm
Lead	Crustacea	1.5 ppm
	Molluscan bivalves	1.7 ppm
Nickel	Crustacea	70 ppm
	Molluscan bivalves	80 ppm
Methyl Mercury	Shellfish and finfish	1 ppm
Diquat	Shellfish and finfish	0.1 ppm
Fluridone	Crayfish	0.5 ppm
Glyphosate	Shellfish	3 ppm
PCBs	Shellfish and finfish	2 ppm
2,4-D	Shellfish and finfish	1 ppm
Chinese National Food Safety Standard Maximum Levels	of Contaminants in Foods <sup>3</sup>	
Lead	Fish, crustacean	$0.5 \mathrm{mg  kg^{-1}}$
Luu	Bivalves	$1.5 \text{ mg kg}^{-1}$
Cadmium	Crustacean	0.5
	Bivalves, gastropods, cephalopods, echinoderms	2 (viscera removed)
Methyl mercury	Aquatic animals	$0.5 \text{ mg kg}^{-1}$
As	Aquatic animals and its products, not including fish	$0.5 \text{ mg kg}^{-1}$
Cr	Aquatic animal and its products	$2 \text{ mg kg}^{-1}$
Benzo( <i>a</i> )pyrene	Aquatic animals and its products	$5 \mu g kg^{-1}$
<i>N</i> -Nitrosodimethylamine	Aquatic animals and its products	$4 \mu g  kg^{-1}$
Polychlorinated hiphenyl	Aquatic animals and its products	$0.5 \text{ mg kg}^{-1a}$

<sup>1</sup> European Comission Regulation (ECR, 2006).

<sup>2</sup> U.S Food and Drug Administration (USDA, 1996a, 1996b, 1993).

<sup>3</sup> Chinese National Food Safety Standard Maximum Levels of Contaminants in Foods (NHFPC and CFDA, 2017).

<sup>a</sup> Polychlorinated biphenyl is calculated by total of PCB28, PCB52, PCB101, PCB118, PCB138, PCB153 and PCB180.

In the food safety area, RA is based on standard protocols established by governmental and international agencies, focused on standard methods and contaminants limits. However, Tahar et al. (2017) highlighted that developing and deploying an RA model for evaluating environmental threat posed by emerging contaminants of concern is challenging given the multiplicity of influencing and controlling factors to be addressed. The authors defined a semi-quantitative RA model that adopted the Irish EPA's Source-Pathway-Receptor concept to define relevant parameters for calculating low, medium or high-risk scores for each agglomeration of WWTP, which included catchment, treatments, operational and management factors. This RA model may be potentially applied on a transnational scale to (i) identify WWTPs that pose a particular risk as regards releasing disproportionally high levels of contaminants that may reach shellfish, and (ii) to identify priority locations for introducing or upgrading control measures (e.g., tertiary treatment, source reduction). This RA was deemed semi-quantitative as other influencing factors such as influence of climate change, hydrological data, pollutant usage or occurrence dose to be considered in a future point for estimating and predicting risk.

A general model and steps for conducting a risk assessment study is presented in Fig. 1. Correlating for the needs of the shellfish industry, the first step is the identification of hazards, where physicochemical contaminants, biotoxins, processing fails, inadequate handling and short depuration process are frequently observed (Fazil, 2005). Post identification, the contaminants and processing failures are measured and analysed to verify the severity of the potential risk; for example, if the contaminants are PTE that occur at high concentrations in shellfish waters, then represents an increased potential risk of exposure, that may also impact upon subsequent consumer safety. Next, the hazard characterization provides a better understanding of the problem and supports the development of mitigation alternatives. The fourth step in the proposed RA model reflects the development of an action plan and the main implementation of targeted measures. In the shellfish industry, the depuration process is one of the major mitigation strategies, and is mainly applied to reduce the shellfish bacterial and viral loads. Finally, the last step reflects the need for continuous monitoring of hazards identified and review of appropriate response(s); for example, Tahar et al. (2018) and Tiedeken et al. (2017) reported on the challenges limited data and lack of sufficiently sensitive analytical detection methods to inform robust RA modelling. Physical assets, such as depuration facilities, may also require upgrading to enable installation of more effective disinfection technologies for complex viral threats such as human norovirus. There is also potential to use the online graphic information system, ArcGIS, to inform the occurrence and geomapping for these targeted contaminants of concern (Tahar et al., 2018) that will inform location of concern with a view to upgrading mitigation infrastructure and assets.

In Europe, RA can be performed by permanent bodies or decentralised EU agencies and scientific communities (necessarily set up by the European Commission). A guidance to authorities is provided in the Article 20 of Regulation (EC) No 765/2008, and an implement to the risk assessment methodology described in 2015-IMP-MSG-15 (European Commission, 2016). In the United States of America, the Environmental Protection Agency (EPA) and National Research Council (NRC) provides the main guidance for risk assessment studies, separately according to the hazard and activity. International bodies such as the World Health Organization (WHO) and Food and Agriculture Organization (FAO) are also major references for risk assessment studies (Fazil, 2005; World Health Organization, 2020b). The International Organization for Standardization (ISO) is an important non-governmental organization responsible for standards, where the ISO 31000 is the international certification for risk management (ISO, 2018). In Ireland, with reference to the European and ISO guidelines, the Health Safety Authority (HSA) provides the guidance to protect employees and identify hazards (HSA, 2022).

A risk assessment is not only interesting to evaluate an ongoing activity, but essential to evaluate the environmental impact of new activities. In a recent report, the Irish Marine Institute (2020) provided a full analysis of a conservation area and its potential use for aquaculture and fishing projects. The description of *in site* activities, risks, effects and assessment through a score system were considered to endorse or reject the setting and usage of the natural habitat. The RA can be also focused on one contaminant/activity, or effect in shellfish/humans.

# 7. Estimated daily intake and targeted hazard quotient of contaminants in shellfish

The health risk due to contaminants in shellfish can be estimated by calculating the estimated daily intake (EDI) and targeted hazard quotient (THQ) (USEPA, 2000). First, EDI is calculated considering the concentration of contaminant in the shellfish (Eq. 1), usually expressed in  $\mu$ g g<sup>-1</sup>; the daily



Fig. 1. General steps of a risk assessment study.

mean ingestion (DMN) expresses the daily consumption of > shellfish *per capita* in the region/area where the shellfish is mainly consumed; and the average body weight of consumer (BW).

Then, THQ (Eq. 2) is obtained considering the oral reference dose (RFd) of the contaminant, an estimate of daily exposure to low risk of deleterious effects during a lifetime (USEDA, 1991)

For estimated daily intake (EDI), concentration of contaminant in shellfish is multiplied by the daily mean ingestion and divided by average body weight:

$$EDI = \frac{C.Contaminant \times daily mean ingestion}{Body weight}$$
(1)

For targeted hazard quotient (THQ), estimated daily intake is divided by oral reference dose, resulting in a constant value for THQ:

$$THQ = \frac{EDI}{RFd}$$
(2)

Concentration of PHAR, PTE and MP in shellfish (Tables 1, 2 and 3) were used to simulate the EDI for adults in Europe, Africa, Asia, America and Oceania based on average population weight and shellfish consumption per capita (Appendix A) (Rodriguez-Martinez et al., 2020). EDI and THQ were estimated based on two scenarios: continent-specific consumption, per capita, of crustaceans and molluscs reported by FAO (2020), where Europe, Africa, Asia, America and Oceania consume mean values of 10.16  $\pm$  0.23, 0.55  $\pm$  0.2, 16.24  $\pm$  1.77, 1.56  $\pm$  2 and 14.24  $\pm$  1.8 g day<sup>-1</sup>, respectively; and according to USDA and USDHHS (2020) in a guide for fish and shellfish consumption that classifies species that can be consumed in 2-3 servings (maximum 340 g) a week based on low mercury levels, and 1 serving (maximum 28 g) for species with higher levels. The consumption advice released by USDA and USDHHS (2020) intends to stimulate the healthy consumption of non-contaminated seafood due to benefits as part of a balanced eating pattern. An ingestion of 3 servings of 4 oz. (340 g) of shellfish (non-pregnant adults) was chosen in a hypothetical scenario to estimate the exposure to PHAR, PTE and MP in shellfish (Tables 1–3) when consuming the recommended amount, *per capita*, of shellfish. Therefore, 48.5 g was used as reference for daily consumption. The oral RFd for metals were based on USEPA (1991, 2000), a database platform part of the Integrated Risk Information System (IRIS) to provide the risk information and assessment of chemicals. For PHAR without a RFd, the parameter (Eq. 2) was replaced by the acceptable daily intake (ADI), corresponding to the lowest daily therapeutic dose without harmful effect. Currently, RFd availability is limited to PHAR applied on agriculture and/or animals. The daily intake of MP was estimated considering the number of units and particles/g and due to absence of reference values for MP to calculate the THQ.

EDI was calculated based on PHAR, PTE and MP concentrations reported in the literature. Then, THQ of PHAR and PTE were calculated and categorized according to levels and risk: no hazard (THQ  $\leq$  0.1), low hazard (THQ 0.1–1), moderate (THQ 1–10) and high hazard (THQ > 10) (Lemly, 1996). The total analysis for a specific contaminant, for example EDI and THQ of Cd in male and female according to FAO and USDA daily intakes, was considered the total (100 %), and each hazard level summed to calculate the specific percentage of the total. In Figs. 2 and 3 is presented the percentage of each hazard level per contaminant. The complete dataset is available in Appendix A.

EDI and THQ (Fig. 2) were higher for simulations with USDA and USDHHS (2020) consumption profile than FAO due to recommended ingestion of shellfish being higher than current mean. The lowest daily consumption of shellfish was identified in Africa (0.55 g day<sup>-1</sup>) and highest in Asia (16.24 g day<sup>-1</sup>).

THQ for potentially toxic elements (PTE) is presented in Fig. 2. Highest THQ were observed in Asia, and to Cu based on the PTE concentrations reported in shellfish. THQ was lower for As and Cr, with a moderate/low risk/ no hazard levels. EDI and THQ were expressively higher in studies where shellfish were sampled from severely contaminated areas as reported by He et al. (2016). The authors calculated the HQ for Cu, Cd, and Zn in *Crassostrea angulata* and *Crassostrea hongkongensis*, based on the reference doses (RfD) and estimated daily intake (EDI) as established by the USEPA (2005), considering the average body weight of a person from China and specific metal bioaccessibility (<100 %). THQ reported agreed with the



Fig. 2. Distribution of potentially toxic elements hazards (%) by continent.



Fig. 3. Distribution of pharmaceuticals hazards (%) by continent.

authors, signalling a high/moderate risk for Cu, Cd and Zn from consuming *C. angulate* and *C. hongkongensis* with contamination levels from reported waters. Cu concentration reported by the authors led to the highest THQ (Fig. 2), with Asia having 50 % of population in a potential scenario of high exposure. Yuan et al. (2020) considered the average weight of a person from China in their study and determined the HQ, bioconcentration factor (BF), cumulative risk (AR) and carcinogenic risk (CR) of Cu, Pb, Zn, Cd, Cr, Hg, and As in *P. viridis, M. edulis, O. glomerate, B. virescens, G. divaricatum*, and *C. scripta*. The authors identified high BF for Cu and Zn, calculated by dividing the concentration of PTE in bivalves by concentration of PTE in the sampled seawater.

Yuan et al. (2020) In another study in Bosnia and Herzegovina, Djedjibegovic et al. (2020) determined the health risk of exposure to Cd, Hg, and Pb in commercial seafood products by estimating the weekly intake, hazard index, target hazard quotients, and percentage of tolerable weekly intake by adults of average weight of 70 kg over a lifetime of 70 years. No risk was identified for shellfish based on the consumption profile of adults in Bosnia and Herzegovina. El-Shenawy et al. (2016) estimated the exposure to Al, Cd, Cr, Cu, Co, Fe, Mn, Mo, Ni, Pb, Sr, V, and Zn for local consumers of *Ruditapes decussatus* and *Paphia undulata* shellfish in Ismailia, Egypt, considering the consumption rate of a specific body weight group. Fe, Al, Zn, and Sr had the highest concentration in the bivalves, and Pb was nearly two times higher than the maximum limit; however, the HQ was lower than the limit for all metals.

THQ of PHARs values are presented in Fig. 3. Highest continental exposure was identified in Asia and a similar exposure between Europe, Africa, America and Oceania. Nonylphenol (NP) was the highest hazard PHAR, with moderate to no hazard THQs. NP is a synthetic organic compound widely used in the production of surfactants with extensively estrogenic activity reported. Its persistence and high accumulation in sediments was reported by Zhang et al. (2011) with a concentration of 1964.8 ng g<sup>-1</sup> in Yundang Lagoon of Xiamen, China. Asia would be more affected to pharmaceuticals in shellfish with a moderate exposure to ubiquitous contaminants such as EE2 and NP, although Europe, Africa, America and Oceania had a THQ < 1. BPA, OP, PAR, TRI, MA and CZ were not present in hazard levels (THQ  $\leq$  0.1) and no high hazard (THQ > 10) was observed for listed

PHAs. Interestingly, concentration of BPA in different shellfish species and locations reported by Chiu et al. (2018) and Zhang et al. (2011) resulted in the same exposure level in all continents. A similar THQ for EE2 was observed by Chiu et al. (2018), Zhang et al. (2011) and McEneff et al. (2014) at different locations and shellfish species. A variety of PHARs are usually present at same time in water and shellfish. If combined and calculated as hazard index, PHARs can potentially threat consumer safety and increase its hazard level.

Mello et al. (2022) reported the occurrence of PHAR in highly consumed bivalves and at Parnaiba River Delta (Brazil) and Sepetiba Bay (Brazil). Risk assessment and human exposure were also assessed. The presence of furosemide (FUR), carbamazepine (CBZ), ketoprofen (KET), bezafibrate (BZF), ibuprofen (IBU), gemfibrozil (GFB), diclofenac (DIC), simvastatin (SIM) was identified in the bivalve species A. brasiliana and M. edulis, with a consumer daily exposure to SIM, FUR and IBU up to  $10 \text{ ng kg}^{-1}$  body weight of bivalve. A. brasiliana had the highest human exposure of 20.3 ng kg $^{-1}$  body weight when considering all PHARs analysed. The estimated exposure was considered safe as target hazard quotient and hazard index were below 1 for all species and PHARs. However, only one metabolite was analysed in the study and the exposure could be underestimated. After ingestion, a drug can be metabolised by hydrolysis, oxidation, reduction and several reactions before reaching the shellfish. The constant monitoring of coastal areas and surroundings of shellfish farms is necessary to evaluate the level of contamination and register possible alterations from constant exposure to low concentrations.

The highest estimated daily exposure (EDI) to microplastics (MP) in shellfish was observed in Asia. Devriese et al. (2015) reported the presence of 0.68 microplastics/gram of shrimp. The consumption of 48.6 g of shell-fish would result in the ingestion of 0.53 g of microplastics a day — approximately 1 % of shrimp weight corresponding to microplastics. Levels of microplastics in the papers analysed were not distinct by species, van Cauwenberghe and Janssen (2014) and Devriese et al. (2015) reported 0.2 and 0.36 microplastics/g of *M. edulis*, respectively. As observed in Appendix A, EDI ranged from 0.005 to 0.53 microplastics/g, an annual maximum ingestion of 193 microplastics/g of shellfish. A RA of microplastics in bivalve molluscs was recently published by Ding et al.

#### Declaration of competing interest

Gustavo Waltzer Fehrenbach reports financial support was provided by Irish Research Council.

#### Acknowledgements

We would like to thank the Technological University of The Shannon – Midlands Midwest. The research conducted in this publication was funded by the Irish Research Council under grant number GOIPG/2020/807.

#### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi. org/10.1016/j.scitotenv.2022.157067.

#### References

- Adams, C., Wang, Y., Loftin, K., Meyer, M., 2002. Removal of antibiotics from surface and distilled water in conventional water treatment processes. J. Environ. Eng. 128, 253–260. https://doi.org/10.1061/(ASCE)0733-9372(2002)128:3(253).
- Aguirre-Martínez, G.V., DelValls, T.A., Martín-Díaz, M.L., 2016. General stress, detoxification pathways, neurotoxicity and genotoxicity evaluated in Ruditapes philippinarum exposed to human pharmaceuticals. Ecotoxicol. Environ. Saf. 124, 18–31. https://doi.org/10. 1016/j.ecoenv.2015.09.031.
- Almeida, Â., Solé, M., Soares, A.M.V.M., Freitas, R., 2020. Anti-inflammatory drugs in the marine environment: bioconcentration, metabolism and sub-lethal effects in marine bivalves. Environ. Pollut. 263, 114442. https://doi.org/10.1016/j.envpol.2020.114442.
- Álvarez-Muñoz, D., Rodríguez-Mozaz, S., Maulvault, A.L., Tediosi, A., Fernández-Tejedor, M., van den Heuvel, F., Kotterman, M., Marques, A., Barceló, D., 2015. Occurrence of pharmaceuticals and endocrine disrupting compounds in macroalgaes, bivalves, and fish from coastal areas in Europe. Environ. Res. 143, 56–64. https://doi.org/10.1016/j. envres.2015.09.018.
- Alygizakis, N.A., Gago-Ferrero, P., Borova, V.L., Pavlidou, A., Hatzianestis, I., Thomaidis, N.S., 2016. Occurrence and spatial distribution of 158 pharmaceuticals, drugs of abuse and related metabolites in offshore seawater. Sci. Total Environ. 541, 1097–1105. https://doi. org/10.1016/j.scitotenv.2015.09.145.
- Anacleto, P., Maulvault, A.L., Nunes, M.L., Carvalho, M.L., Rosa, R., Marques, A., 2015. Effects of depuration on metal levels and health status of bivalve molluscs. Food Control 47, 493–501. https://doi.org/10.1016/j.foodcont.2014.07.055.
- Andrady, A.L., 2011. Microplastics in the marine environment. Mar. Pollut. Bull. https://doi. org/10.1016/j.marpolbul.2011.05.030.
- Angeles, L.F., Mullen, R.A., Huang, I.J., Wilson, C., Khunjar, W., Sirotkin, H.I., McElroy, A.E., Aga, D.S., 2020. Assessing pharmaceutical removal and reduction in toxicity provided by advanced wastewater treatment systems. Environ. Sci. Water Res. Technol. 6, 62–77. https://doi.org/10.1039/C9EW00559E.
- Balali-Mood, M., Naseri, K., Tahergorabi, Z., Khazdair, M.R., Sadeghi, M., 2021. Toxic mechanisms of five heavy metals: mercury, lead, chromium, cadmium, and arsenic. Front. Pharmacol. 12. https://doi.org/10.3389/fphar.2021.643972.
- Bayen, S., Koroleva, E., Lee, H.K., Obbard, J.P., 2005. Persistent organic pollutants and heavy metals in typical seafoods consumed in Singapore. J. Toxic. Environ. Health A 68. https://doi.org/10.1080/15287390590890437.
- Bebianno, M.J., Mello, A.C.P., Serrano, M.A.S., Flores-Nunes, F., Mattos, J.J., Zacchi, F.L., Piazza, C.E., Siebert, M.N., Piazza, R.S., Gomes, C.H.A.M., Melo, C.M.R., Bainy, A.C.D., 2017. Transcriptional and cellular effects of paracetamol in the oyster Crassostrea gigas. Ecotoxicol. Environ. Saf. 144, 258–267. https://doi.org/10.1016/j.ecoenv.2017. 06.034.
- Bellas, J., Martínez-Armental, J., Martínez-Cámara, A., Besada, V., Martínez-Gómez, C., 2016. Ingestion of microplastics by demersal fish from the Spanish Atlantic and Mediterranean coasts. Mar. Pollut. Bull. 109, 55–60. https://doi.org/10.1016/j.marpolbul.2016.06.026.
- Bonnefille, B., Arpin-Pont, L., Gomez, E., Fenet, H., Courant, F., 2017. Metabolic profiling identification of metabolites formed in Mediterranean mussels (Mytilus galloprovincialis) after diclofenac exposure. Sci. Total Environ. 583, 257–268. https://doi.org/10.1016/j. scitotenv.2017.01.063.
- Blalock, B.J., Robinson, W.E., Loguinov, A., Vulpe, C.D., Krick, K.S., Poynton, H.C., 2018. Transcriptomic and network analyses reveal mechanistic-based biomarkers of endocrine disruption in the marine mussel, Mytilus edulis. Environ. Sci. Technol. 52, 9419–9430. https://doi.org/10.1021/acs.est.8b01604.
- Brandon, E.F.A., Oomen, A.G., Rompelberg, C.J.M., Versantvoort, C.H.M., van Engelen, J.G.M., Sips, A.J.A.M., 2006. Consumer product in vitro digestion model: bioaccessibility of contaminants and its application in risk assessment. Regul. Toxicol. Pharmacol. 44. https://doi.org/10.1016/j.yrtph.2005.10.002.
- Brew, D.W., Black, M.C., Santos, M., Rodgers, J., Henderson, W.M., 2020. Metabolomic investigations of the temporal effects of exposure to pharmaceuticals and personal care products and their mixture in the eastern oyster (Crassostrea virginica). Environ. Toxicol. Chem. 39, 419–436. https://doi.org/10.1002/etc.4627.
- Browne, M.A., Dissanayake, A., Galloway, T.S., Lowe, D.M., Thompson, R.C., 2008. Ingested microscopic plastic translocates to the circulatory system of the mussel, Mytilus edulis (L). Environ. Sci. Technol. 42, 5026–5031. https://doi.org/10.1021/es800249a.
- Capolupo, M., Díaz-Garduño, B., Martín-Díaz, M.L., 2018. The impact of propranolol, 17αethinylestradiol, and gemfibrozil on early life stages of marine organisms: effects and

(2022). The authors analysed the quantitative and qualitative data from 52 peer-reviewed papers and estimated the chemical composition risk, annual dietary intake and load index of microplastics by consumption of bivalve molluscs. Polyethylene terephthalate was the most abundant polymer in bivalves (20.4 % ± 21.2 %) followed by polyethylene (13 % ± 19.2 %), rayon (9 % ± 15.6 %), polypropylene (8.9 % ± 13.3 %), cellophane (8 % ± 17.7 %), polyester (7 % ± 12.4 %) and polyamide (5.3 % ± 14.8 %). The global abundance was in the range of 0.04–20 microplastics/g. Iran, Greece and China were the three countries with the highest levels. Median intake of microplastics *via* mollusc consumption, microplastics abundance, culture and geography were aspects associated with the wide intake range of 15–7333 microplastics/person.

The estimation of EDI and TDH for contaminants in shellfish based on levels available in publications, reports and papers allowed the construction of two consumption scenarios: first with the recommended portion of shellfish (USEPA), and the current mean in each continent (FAO). However, there is a significant variability in the shellfish consumption between countries of the same group (continent) and it would be necessary to calculate separately to provide accurately the status of each region. The objective of calculating EDI and THD showed step-by-step how risks can be easily estimated, allowing the small shellfish producers to estimate the local exposure and support decision-making. Many contaminants can be accumulated by shellfish when farmed in contaminated areas, a higher risk when compared to single analysis. The constant monitoring of water, soil and shellfish is essential to provide accurate assessment of shellfish risk and quality and must be associated to bioavailability studies to inform the %contaminant directly available. Most of the RA conducted for the shellfish industry are related to PTE (Hantoro et al., 2019). There remains a wide diversity of contaminants (e.g. plastics and pharmaceuticals) that must be comprehensively studied in order to ensure development of appropriate RA to inform management and where applicable, changes to policies.

#### 8. Conclusions

- In the present study, the environmental contaminants microplastic, pharmaceuticals and potentially toxic elements that threaten the shellfish industry were widely discussed.
- The effects and consequences in the most consumed shellfish species were highlighted and promising mitigation alternatives registered.
- Regulation and international guidelines were provided that can potentially assist decision making and contamination control.
- A panorama of Irish waters was provided. As noted in other countries, there is a gap in understanding the presence, extension of contamination and effects of the contaminants discussed.
- There is a need for the development of techniques for monitoring, standardization of risk assessments and detection of emerging contaminants in the environment.
- The reduction of contamination level in shellfish waters and employment of a risk assessment (RA) approach is essential to ensure consumer safety.
- International consensus must be reached on oral reference dose and exposure limits. There is a pressing need to simultaneously evaluate, model and predict the plethora of influencing factors governing the efficacy of RA models to inform decision making in real time.

#### CRediT authorship contribution statement

**Gustavo Waltzer Fehrenbach:** Conceptualization, Methodology, Investigation, Writing – original draft, Visualization. **Robert Pogue:** Conceptualization, Methodology, Writing – review & editing, Supervision, Project administration. **Frank Carter:** Methodology, Writing – review & editing. **Eoghan Clifford:** Conceptualization, Writing – review & editing, Supervision, Project administration. **Neil Rowan:** Conceptualization, Writing – review & editing, Supervision, Project administration.

#### G.W. Fehrenbach et al.

risk assessment. Environ. Sci. Pollut. Res. Int. 25, 32196-32209. https://doi.org/10. 1007/s11356-018-3185-6.

- Carpenter, E.J., Anderson, S.J., Harvey, G.R., Miklas, H.P., Peck, B.B., 1972. Polystyrene spherules in coastal waters. Science 178, 749–750. https://doi.org/10.1126/science. 178.4062.749.
- Chiu, J.M.Y., Po, B.H.K., Degger, N., Tse, A., Liu, W., Zheng, G., Zhao, D.-M., Xu, D., Richardson, B., Wu, R.S.S., 2018. Contamination and risk implications of endocrine disrupting chemicals along the coastline of China: a systematic study using mussels and semipermeable membrane devices. Sci. Total Environ. 624, 1298–1307. https://doi. org/10.1016/j.scitotenv.2017.12.214.
- Daniel, D.B., Ashraf, P.M., Thomas, S.N., 2020. Abundance, characteristics and seasonal variation of microplastics in indian white shrimps (Fenneropenaeus indicus) from coastal waters off Cochin, Kerala,India. Sci. Total Environ. 737. https://doi.org/10.1016/j. scitotenv.2020.139839.
- Devriese, L.I., van der Meulen, M.D., Maes, T., Bekaert, K., Paul-Pont, I., Frère, L., Robbens, J., Vethaak, A.D., 2015. Microplastic contamination in brown shrimp (Crangon crangon, Linnaeus 1758) from coastal waters of the Southern North Sea and Channel area. Mar. Pollut. Bull. 98. https://doi.org/10.1016/j.marpolbul.2015.06.051.
- Ding, J., Sun, Y., He, C., Li, J., Li, F., 2022. Towards risk assessments of microplastics in bivalve mollusks globally. J. Mar. Sci. Eng. 10, 288. https://doi.org/10.3390/ jmse10020288.
- Djedjibegovic, J., Marjanovic, A., Tahirovic, D., Caklovica, K., Turalic, A., Lugusic, A., Omeragic, E., Sober, M., Caklovica, F., 2020. Heavy metals in commercial fish and seafood products and risk assessment in adult population in Bosnia and Herzegovina. Sci. Rep. 10. https://doi.org/10.1038/s41598-020-70205-9.
- El-Shenawy, N.S., Loutfy, N., Soliman, M.F.M., Tadros, M.M., Abd El-Azeez, A.A., 2016. Metals bioaccumulation in two edible bivalves and health risk assessment. Environ. Monit. Assess. 188, 1–12. https://doi.org/10.1007/s10661-016-5145-2.
- European Commission Regulation, 2006. European Commission Regulation No. 1881/2006 of the European Parliament and of the Council Setting Maximum Levels for Certain Contaminants in Foodstuffs.
- European Commission, 2016. EU general risk assessment methodology (Action 5 of Multi-Annual Action Plan for the surveillance of products in the EU (COM(2013)76) [WWW Document]. URL https://ec.europa.eu/docsroom/documents/17107/attachments/1/ translations/en/renditions/pdf (accessed 3.3.22).
- European Commission, 2004. Regulation (EC) No 853/2004 of the European Parliament and of the Council of 29 April 2004 laying down specific rules for food of animal origin [WWW Document]. Official Journal of the European Union. URL https://www.fsai.ie/ uploadedFiles/Corrigendum-to-Regulation-EC-No-853-2004(1).pdf (accessed 3.28.22).
- Fabbri, E., Franzellitti, S., 2016. Human pharmaceuticals in the marine environment: focus on exposure and biological effects in animal species. Environ. Toxicol. Chem. 35, 799–812. https://doi.org/10.1002/etc.3131.
- Fairbaim, D.J., Arnold, W.A., Barber, B.L., Kaufenberg, E.F., Koskinen, W.C., Novak, P.J., Rice, P.J., Swackhamer, D.L., 2016. Contaminants of emerging concern: mass balance and comparison of wastewater effluent and upstream sources in a mixed-use watershed. Environ. Sci. Technol. 50, 36–45. https://doi.org/10.1021/acs.est.5b03109.
- FAO, 2018. Microplastics in Fisheries and Aquaculture: Status of Knowledge on Their Occurrence and Implications for Aquatic Organisms and Food Safety. Food & Agriculture Organization.
- FAO, 2016. The State of World Fisheries and Aquaculture: Contributing to Food Security and Nutrition for All.
- Fazil, A.M., 2005. A Primer on Risk Assessment Modelling: Focus on Seafood Products Rome. Fehrenbach, G., Arsand, D.R., Barbosa, S.C., Castagno, K., Filho, P., Primel, E., 2021. A fast, low-cost and eco-friendly method for routine determination of Bisphenol-A in landfill leachate employing vortex assisted liquid-liquid extraction. Anal. Methods Environ. Chem. J. 4, 59–67. https://doi.org/10.24200/amecj.v4.i03.146.
- Food and Agriculture Organization of the United Nations, 2008. Bivalve depuration: fundamental and practical aspects [WWW Document]. URL https://www.fao.org/3/i0201e/ i0201e.pdf (accessed 5.28.22).
- Food and Agriculture Organization of the United Nations (FAO), 2020. Shellfish consumption per capita [WWW Document]. URL https://www.fao.org/faostat/en/#data/FBS (accessed 5.21.22).
- Froehlich, H.E., Gentry, R.R., Halpern, B.S., 2018. Global change in marine aquaculture production potential under climate change. Nat. Ecol. Evol. 2, 1745–1750. https://doi.org/ 10.1038/s41559-018-0669-1.
- Gadelha, J.R., Rocha, A.C., Camacho, C., Eljarrat, E., Peris, A., Aminot, Y., Readman, J.W., Boti, V., Nannou, C., Kapsi, M., Albanis, T., Rocha, F., Machado, A., Bordalo, A., Valente, L.M.P., Nunes, M.L., Marques, A., Almeida, C.M.R., 2019. Persistent and emerging pollutants assessment on aquaculture oysters (Crassostrea gigas) from NW Portuguese coast (Ria De Aveiro). Sci. Total Environ. 666, 731–742. https://doi.org/10.1016/j. scitotenv.2019.02.280.
- Greco, L., Pellerin, J., Capri, E., Garnerot, F., Louis, S., Fournier, M., Sacchi, A., Fusi, M., Lapointe, D., Couture, P., 2011. Physiological effects of temperature and a herbicide mixture on the soft-shell clam Mya arenaria (Mollusca, Bivalvia). Environ. Toxicol. Chem. 30, 132–141. https://doi.org/10.1002/etc.359.
- Han, B.-C., Jeng, W.-L., Tsai, Y.-N., Jeng, M.-S., 1993. Depuration of copper and zinc by green oysters and blue mussels of Taiwan. Environ. Pollut. 82, 93–97. https://doi.org/10.1016/ 0269-7491(93)90166-L.
- Hannah, R., Max, R., 2021. Biodiversity. Published online at OurWorldInData.org.
- Hantoro, I., Löhr, A.J., van Belleghem, F.G.A.J., Widianarko, B., Ragas, A.M.J., 2019. Microplastics in coastal areas and seafood: implications for food safety. Food Addit. Contam. Part A Chem. Anal. Control Expo. Risk Assess. 36, 674–711. https://doi.org/ 10.1080/19440049.2019.1585581.
- Hara, J., Frias, J., Nash, R., 2020. Quantification of microplastic ingestion by the decapod crustacean Nephrops norvegicus from Irish waters. Mar. Pollut. Bull. 152. https://doi. org/10.1016/j.marpolbul.2020.110905.

- He, M., Ke, C.-H., Tian, L., Li, H.-B., 2016. Bioaccessibility and health risk assessment of Cu, Cd, and Zn in "colored" oysters. Arch. Environ. Contam. Toxicol. 70, 595–606. https:// doi.org/10.1007/s00244-015-0194-z.
- Hemalatha, D., Rangasamy, B., Nataraj, B., Ramesh, M., 2019. Assessment of triclosan impact on enzymatic biomarkers in an Indian major carp, Catla catla. J. Basic Appl. Zool. 80, 1–8. https://doi.org/10.1186/s41936-019-0094-2.
- Hernando, M., Mezcua, M., Fernandezalba, A., Barcelo, D., 2006. Environmental risk assessment of pharmaceutical residues in wastewater effluents, surface waters and sediments. Talanta 69. https://doi.org/10.1016/j.talanta.2005.09.037.
- Hobbs, H.H., 2012. Crustacea. Encyclopedia of Caves. Elsevier https://doi.org/10.1016/ B978-0-12-383832-2.00027-X.
- HSA, 2022. Your Steps to Chemical Safety A Guide for Small Business.
- Irish Marine Institute, 2020. Report Supporting Appropriate Assessment of Aquaculture Valentia Harbour/Portmagee Channel SAC (Site Code: 2262) Galway.
- Irish Marine Institute, 2018. Monitoring Results for Trace Metals and Organohalogens in Shellfish (2015) and Physicochemical Parameters and Trace Metals in Seawater (2016) in Accordance With Shellfish Varers Directive Galway.
- ISO, 2018. Risk Management: ISO 31000.
- Jambeck, J.R., Geyer, R., Wilcox, C., Siegler, T.R., Perryman, M., Andrady, A., Narayan, R., Law, K.L., 2015. Plastic waste inputs from land into the ocean. Science 1979 (347), 768–771. https://doi.org/10.1126/science.1260352.
- Jjemba, P.K., 2018. Pharma-Ecology: The Occurrence and Fate of Pharmaceuticals and Personal Care Products in the Environment. John Wiley & Sons.
- Kenny, A., Reynolds, W., Sheahan, D., McCubbin, D., Kershaw, P., Rycroft, R., Smith, A., Brooks, S., Kelly, C., Allchin, C., Lawton, E., 2005. A Review of the Contaminant Status of the Irish Sea.
- Koagouw, W., Ciocan, C., 2020. Effects of short-term exposure of paracetamol in the gonads of blue mussels Mytilus edulis. Environ. Sci. Pollut. Res. Int. 27, 30933–30944. https://doi. org/10.1007/s11356-019-06861-w.
- Lacaze, E., Pédelucq, J., Fortier, M., Brousseau, P., Auffret, M., Budzinski, H., Fournier, M., 2015. Genotoxic and immunotoxic potential effects of selected psychotropic drugs and antibiotics on blue mussel (Mytilus edulis) hemocytes. Environ. Pollut. 202, 177–186. https://doi.org/10.1016/j.envpol.2015.03.025.
- Leonard, J.A., Cope, W.G., Hammer, E.J., Barnhart, M.C., Bringolf, R.B., 2017. Extending the toxicity-testing paradigm for freshwater mussels: assessing chronic reproductive effects of the synthetic estrogen 17α-ethinylestradiol on the unionid mussel Elliptio complanata. Comp. Biochem. Physiol. C. Toxicol. Pharmacol. 191, 14–25. https://doi.org/10.1016/ j.cbpc.2016.09.002.
- Lemly, A.D., 1996. Evaluation of the hazard quotient method for risk assessment of selenium. Ecotoxicol. Environ. Saf. 35, 156–162. https://doi.org/10.1006/eesa.1996.0095.
- Lusher, A.L., Hernandez-Milian, G., Berrow, S., Rogan, E., O'Connor, I., 2018. Incidence of marine debris in cetaceans stranded and bycaught in Ireland: recent findings and a review of historical knowledge. Environ. Pollut. 232. https://doi.org/10.1016/j.envpol.2017.09. 070.
- Marean, C.W., Bar-Matthews, M., Bernatchez, J., Fisher, E., Goldberg, P., Herries, A.I.R., Jacobs, Z., Jerardino, A., Karkanas, P., Minichillo, T., Nilssen, P.J., Thompson, E., Watts, I., Williams, H.M., 2007. Early human use of marine resources and pigment in South Africa during the Middle Pleistocene. Nature 449. https://doi.org/10.1038/ nature06204.
- Martin, J., Lusher, A., Thompson, R.C., Morley, A., 2017. The deposition and accumulation of microplastics in marine sediments and bottom water from the Irish Continental Shelf. Sci. Rep. 7. https://doi.org/10.1038/s41598-017-11079-2.
- Mazumdar, M., Bellinger, D.C., Gregas, M., Abanilla, K., Bacic, J., Needleman, H.L., 2011. Low-level environmental lead exposure in childhood and adult intellectual function: a follow-up study. Environ. Health 10. https://doi.org/10.1186/1476-069X-10-24.
- McEneff, G., Barron, L., Kelleher, B., Paull, B., Quinn, B., 2014. A year-long study of the spatial occurrence and relative distribution of pharmaceutical residues in sewage effluent, receiving marine waters and marine bivalves. Sci. Total Environ. 476–477, 317–326. https://doi.org/10.1016/j.scitotenv.2013.12.123.
- McEneff, G., Barron, L., Kelleher, B., Paull, B., Quinn, B., 2013. The determination of pharmaceutical residues in cooked and uncooked marine bivalves using pressurised liquid extraction, solid-phase extraction and liquid chromatography-tandem mass spectrometry. Anal. Bioanal. Chem. 405, 9509–9521. https://doi.org/10.1007/s00216-013-7371-6.
- McEneff, G., Schmidt, W., Quinn, B., 2015. Pharmaceuticals in the Aquatic Environment: A Short Summary of Current Knowledge and the Potential Impacts on Aquatic Biota and Humans.
- Mello, F.V., Cunha, S.C., Fogaça, F.H.S., Alonso, M.B., Torres, J.P.M., Fernandes, J.O., 2022. Occurrence of pharmaceuticals in seafood from two Brazilian coastal areas: implication for human risk assessment. Sci. Total Environ. 803, 149744. https://doi.org/10.1016/j. scitotenv.2021.149744.
- Mezzelani, M., Gorbi, S., Regoli, F., 2018. Pharmaceuticals in the aquatic environments: evidence of emerged threat and future challenges for marine organisms. Mar. Environ. Res. 140, 41–60. https://doi.org/10.1016/j.marenvres.2018.05.001.
- Munari, M., Chemello, G., Finos, L., Ingrosso, G., Giani, M., Marin, M.G., 2016. Coping with seawater acidification and the emerging contaminant diclofenac at the larval stage: a tale from the clam Ruditapes philippinarum. Chemosphere 160, 293–302. https://doi. org/10.1016/j.chemosphere.2016.06.095.
- Murray, F., Cowie, P.R., 2011. Plastic contamination in the decapod crustacean Nephrops norvegicus (Linnaeus, 1758). Mar. Pollut. Bull. 62. https://doi.org/10.1016/j.marpolbul. 2011.03.032.
- NHFPC, CFDA, 2017. National Food Safety Standard Maximum Levels of Contaminants in Foods (GB 2762-2017) China.
- Oscar, T.P., 2012. Food Risk Analysis. https://doi.org/10.1007/978-1-4614-1177-2\_12.
- Piedade, F., Bio, S., Nunes, B., 2020. Effects of common pharmaceutical drugs (paracetamol and acetylsalicylic acid) short term exposure on biomarkers of the mussel Mytilus spp. Environ. Toxicol. Pharmacol. 73, 103276. https://doi.org/10.1016/j.etap.2019.103276.

#### G.W. Fehrenbach et al.

PlasticsEurope, 2020. Plastics - The Facts 2020.

Prié, V., 2019. Molluscs. Encyclopedia of Caves. Elsevier. https://doi.org/10.1016/B978-0-12-814124-3.00087-X.

- Purdom, C.E., Hardiman, P.A., Bye, V.J., Eno, N.C., Tyler, C.R., Sumpter, J.P., 1994. Estrogenic effects of effluents from sewage treatment works. Chem. Ecol. 8, 275–285. https://doi.org/10.1080/02757549408038554.
- Pusceddu, F.H., Choueri, R.B., Pereira, C.D.S., Cortez, F.S., Santos, D.R.A., Moreno, B.B., Santos, A.R., Rogero, J.R., Cesar, A., 2018. Environmental risk assessment of triclosan and ibuprofen in marine sediments using individual and sub-individual endpoints. Environ. Pollut. 232, 274–283. https://doi.org/10.1016/j.envpol.2017.09.046.
- Rheault, Robert, 2012. Aquaculture Production Systems. Wiley-Blackwell, Oxford, UK. https://doi.org/10.1002/9781118250105.
- Rodriguez-Martinez, A., Zhou, B., Sophiea, M.K., Bentham, J., et al., 2020. Height and bodymass index trajectories of school-aged children and adolescents from 1985 to 2019 in 200 countries and territories: a pooled analysis of 2181 population-based studies with 65 million participants. Lancet 396, 1511–1524. https://doi.org/10.1016/S0140-6736(20) 31859-6.
- Sarker, S., Vashistha, D., Saha Sarker, M., Sarkar, A., 2018. DNA damage in marine rock oyster (Saccostrea Cucullata) exposed to environmentally available PAHs and heavy metals along the Arabian Sea coast. Ecotoxicol. Environ. Saf. 151. https://doi.org/10.1016/j. ecoenv.2018.01.004.
- Serrano, M.A.S., Gonzalez-Rey, M., Mattos, J.J., Flores-Nunes, F., Mello, Á.C.P., Zacchi, F.L., Piazza, C.E., Siebert, M.N., Piazza, R.S., Alvarez-Muñoz, D., Rodriguez-Mozaz, S., Barceló, D., Bebianno, M.J., Gomes, C.H.A.M., Melo, C.M.R., Bainy, A.C.D., 2015. Differential gene transcription, biochemical responses, and cytotoxicity assessment in Pacific oyster Crassostrea gigas exposed to ibuprofen. Environ. Sci. Pollut. Res. Int. 22, 17375–17385. https://doi.org/10.1007/s11356-014-4023-0.
- Silva, C., Ferreira, J.G., Bricker, S.B., DelValls, T.A., Martín-Díaz, M.L., Yáñez, E., 2011. Site selection for shellfish aquaculture by means of GIS and farm-scale models, with an emphasis on data-poor environments. Aquaculture 318, 444–457. https://doi.org/10. 1016/j.aquaculture.2011.05.033.

Statista, 2020. Production Capacity of Polystyrene Worldwide From 2018 to 2024.

- Stewart, B.D., Jenkins, S.R., Boig, C., Sinfield, C., Kennington, K., Brand, A.R., Lart, W., Kröger, R., 2021. Metal pollution as a potential threat to shell strength and survival in marine bivalves. Sci. Total Environ. 755. https://doi.org/10.1016/j.scitotenv.2020.143019.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638. https://doi.org/10.1016/j.scitotenv. 2017.05.227.
- Tahar, A., Tiedeken, E.J., Rowan, N.J., 2018. Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – a first overview of the situation in the Republic of Ireland. Sci. Total Environ. 616–617, 187–197. https://doi.org/10.1016/j.scitotenv.2017.11.021.
- Tchounwou, P.B., Yedjou, C.G., Patlolla, A.K., Sutton, D.J., 2012. Heavy Metal Toxicity and theEnvironment. https://doi.org/10.1007/978-3-7643-8340-4\_6.
- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in

receiving waters - a 20 year systematic review. Sci. Total Environ. 574, 1140-1163. https://doi.org/10.1016/j.scitotenv.2016.09.084.

- USDA, 1996. Fish, Shellfish, Crustaceans and Other Aquatic Animals Fresh, Frozen, or Processed Methyl Mercury (CPG 7108.07).
- USDA, 1996. Pesticide Residues in Food and Feed Enforcement Criteria (CPG 7141.01).
- USDA, 1993. Guidance Document for Arsenic (or Cadmium/Chromium/Lead/Nickel) in Shellfish.

USDA and USDHHS, 2020. Dietary Guidelines for Americans, 2020-2025.

- USEPA, 2000. Guidance for assessing chemical contamination data for use in fish advisories. URL https://www.epa.gov/sites/default/files/2015-06/documents/volume2.pdf (accessed 6.7.22).
- U.S. Environmental Protection Agency, 1991. Nickel, soluble salts [WWW Document]. https://cfpub.epa.gov/ncea/iris/iris documents/documents/subst/0271 summary.pdf.
- USEPA, 2005. Risk-based Concentration Table. http://www.epa.gov/reg3hwmd/risk/ human/index.htm US.
- van Cauwenberghe, L., Claessens, M., Vandegehuchte, M.B., Janssen, C.R., 2015. Microplastics are taken up by mussels (Mytilus edulis) and lugworms (Arenicola marina) living in natural habitats. Environ. Pollut. 199. https://doi.org/10.1016/j.envpol.2015. 01.008.
- van Cauwenberghe, L., Janssen, C.R., 2014. Microplastics in bivalves cultured for human consumption. Environ. Pollut. 193, 65–70. https://doi.org/10.1016/j.envpol.2014.06.010.
- Venugopal, V., Gopakumar, K., 2017. Shellfish: nutritive value, health benefits, and consumer safety. Compr. Rev. Food Sci. Food Saf. 16. https://doi.org/10.1111/1541-4337.12312.
- Wang, X.N., Gu, Y.G., Wang, Z.H., Ke, C.L., Mo, M.S., 2018. Biological risk assessment of heavy metals in sediments and health risk assessment in bivalve mollusks from Kaozhouyang Bay, South China. Mar. Pollut. Bull. 133, 312–319. https://doi.org/10. 1016/j.marpolbul.2018.05.059.
- Wessel, N., Rousseau, S., Caisey, X., Quiniou, F., Akcha, F., 2007. Investigating the relationship between embryotoxic and genotoxic effects of benzo[a]pyrene, 17alphaethinylestradiol and endosulfan on Crassostrea gigas embryos. Aquat. Toxicol. 85, 133–142. https://doi.org/10.1016/j.aquatox.2007.08.007.
- Woodall, L.C., Sanchez-Vidal, A., Canals, M., Paterson, G.L.J., Coppock, R., Sleight, V., Calafat, A., Rogers, A.D., Narayanaswamy, B.E., Thompson, R.C., 2014. The deep sea is a major sink for microplastic debris. R. Soc. Open Sci. 1. https://doi.org/10.1098/rsos.140317.
- World Health Organization, 2020. 10 chemicals of public health concern [WWW Document]. URL https://www.who.int/news-room/photo-story/photo-story-detail/10-chemicals-ofpublic-health-concern (accessed 2.21.22).
- World Health Organization, 2020. Risk Assessment Geneva.
- Yuan, Y., Sun, T., Wang, H., Liu, Y., Pan, Y., Xie, Y., Huang, H., Fan, Z., 2020. Bioaccumulation and health risk assessment of heavy metals to bivalve species in Daya Bay (South China Sea): consumption advisory. Mar. Pollut. Bull. 150, 110717. https://doi.org/10.1016/j. marpolbul.2019.110717.
- Zhang, X., Gao, Y., Li, Q., Li, G., Guo, Q., Yan, C., 2011. Estrogenic compounds and estrogenicity in surface water, sediments, and organisms from Yundang Lagoon in Xiamen, China. Arch. Environ. Contam. Toxicol. 61, 93–100. https://doi.org/10.1007/ s00244-010-9588-0.

# DairyWater: striving for sustainability within the dairy processing industry in the Republic of Ireland

William Finnegan<sup>1,2</sup>, Eoghan Clifford<sup>1,2</sup>, Jamie Goggins<sup>1,2,3</sup>, Niall O'Leary<sup>4</sup>, Alan Dobson<sup>4</sup>, Neil Rowan<sup>5</sup>, Liwen Xiao<sup>6</sup>, Song Miao<sup>7</sup>, Kelly Fitzhenry<sup>1,2</sup>, Peter Leonard<sup>1,2</sup>, Emma Tarpey<sup>1</sup>, Beatriz Gil-Pulido<sup>4</sup>, Fei Gao<sup>6</sup> and Xinmin Zhan<sup>1,2</sup>\*

<sup>1</sup> College of Engineering and Informatics, National University of Ireland, Galway, Ireland

<sup>2</sup> Ryan Institute for Environmental, Marine and Energy Research, National University of Ireland, Galway, Ireland

<sup>3</sup> Centre for Marine and Renewable Energy (MaREI), Galway, Ireland

<sup>4</sup>School of Microbiology, University College Cork, Ireland

<sup>3</sup>Bioscience Research Institute, Athlone Institute of Technology, Ireland

<sup>b</sup> Department of Civil, Structural and Environmental Engineering, Trinity College Dublin, Ireland

<sup>7</sup> Teagasc Food Research Centre, Moorpark, Fermoy, Co. Cork, Ireland

Received 19 October 2017; accepted for publication 15 July 2018

This Review describes the objectives and methodology of the DairyWater project as it aims to aid the Irish dairy processing industry in achieving sustainability as it expands. With the abolition of European milk quotas in March 2015, the Republic of Ireland saw a surge in milk production. The DairyWater project was established in anticipation of this expansion of the Irish dairy sector in order to develop innovative solutions for the efficient management of water consumption, wastewater treatment and the resulting energy use within the country's dairy processing industry. Therefore, the project can be divided into three main thematic areas: dairy wastewater treatment technologies and microbial analysis, water re-use and rainwater harvesting and environmental assessment. In order to ensure the project remains as relevant as possible to the industry, a project advisory board containing key industry stakeholders has been established. To date, a number of large scale studies, using data obtained directly from the Irish dairy industry, have been performed. Additionally, pilot-scale wastewater treatment (intermittently aerated sequencing batch reactor) and tertiary treatment (flow-through pulsed ultraviolet system) technologies have been demonstrated within the project. Further details on selected aspects of the project are discussed in greater detail in the subsequent cluster of research communications.

Keywords: Dairy processing, Ireland, sustainability, tertiary treatment, wastewater treatment.

Ireland has a long tradition of dairy farming and the production of dairy products for international markets, primarily butter and cheese. Today, Ireland is one of Europe's largest producers of cow's milk and a globally recognised producer of dairy products and ingredients. Over the last two decades, Ireland has become one of the world's leading producers of infant nutritional products with the presence in Ireland of a number of leading infant nutrition companies and accounts for 15% of the global supply of infant formula (Barry, 2012).

In 2016, there were approximately 1.295 million dairy cows on over 17000 dairy farms in the Republic of Ireland, which produced 6654 million litres of milk with a

fat content of 4·1% and a protein content 3·45% (CSO, 2017). Unlike most of Ireland's competitors, the vast majority of cows in Irish dairy herds are fed on grass; for up to 300 d a year this is fresh pasture grazing, while, for the remainder of the year, the main source of fodder is grass silage. However, since milk production is essentially grass-based, it is very seasonal with a milk production ratio of 7 to 1, when comparing May to January (IFA, 2012). Irish milk is produced mainly by three cattle breeds; British Friesian, Holstein Friesian and Jersey. In 2016, an additional 813 million litres of milk was imported from Northern Ireland into Irish dairies, bringing the total amount of milk processed in the Republic of Ireland to 7467 million litres (CSO, 2017).

Currently, dairy ingredients and products comprise almost 30% of the Irish food and drink export market. In 2013, dairy ingredients and products surpassed  $\in$ 3 billion for the first time, making it Ireland's largest indigenous

<sup>\*</sup>For correspondence; e-mail: xinmin.zhan@nuigalway.ie



Fig. 1. Domestic milk production in the Republic of Ireland between 200 and 2016 and included the projected volume (<sup>a</sup>) for 2020; data obtained from CSO (2017) and NMA (2017).

industry that reached €3·4 billion in 2016 (NMA, 2017), which can be seen in Fig. 1. The increasing value of Irish dairy produce is also evident from this graphic, as Irish dairy processors move towards high value dairy products and ingredients. Traditionally individual dairy processing factories would have produced either cheese and whey products or butter and powder products (Geraghty, 2011). However, with the growth and development of the industry, certain dairy processing factories now specialise in the production of powders, cheeses or whey products. Approximately 7% of Ireland's milk is used for liquid milk consumption with the remainder (93%) manufactured into a variety of products, including butter, cheese, milk powder, whey and proteins (NMA, 2017).

At present, the Republic of Ireland is on the brink of a new era for the dairy industry as European milk quotas, which restricted milk production since 1984, were abolished in March 2015. As a result, milk production is expected to increase by 50% by 2020, based on the reference years 2007 to 2009 (Farrelly et al. 2014). Since their abolition, an immediate increase in milk production has been instigated, which is evident from Fig. 1.

This increase in the volume of milk being processed, together with stringent measures on emissions from the industry and growing commercial drive for operational efficiencies is driving the need for innovative technological and operational solutions within the dairy processing industry. In this context DairyWater, a multi-stakeholder research project, is developing innovative solutions for the efficient management of water consumption, wastewater treatment and the resulting energy use within the country's dairy processing industry. This project has the potential to position Ireland at the forefront of European, or indeed international, research in this sector as it strives to make the Irish dairy processing industry more efficient and environmentally sustainable by reducing carbon footprints, energy and water use. This will, in turn, lead to greater potential for exports, increased international competitiveness for Irish products and stimulate job creation.

This paper outlines the project aims and technologies that are being explored in the DairyWater project. Much of the data presented in this paper has been compiled through direct contact with the Irish dairy processing industry, due to the lack of published data available. However, although the presented study is based on the Irish dairy industry, it has relevance internationally as processing technologies are similar to those employed in other developed countries, in particular those European countries whose milk production was previously restricted by the milk quotas. The study would be particularly comparable to other countries with pasture based milk production, such as New Zealand, the United Kingdom, Galicia in Spain and Michigan in the United States of America. The overarching aim of the project is to aid the Irish dairy processing industry in becoming more sustainable. In order to achieve this the project intends to efficiently and effectively treat wastewater effluent from dairy processing factories using a range of innovative biological, nanomaterial-based and disinfection technologies. In parallel, the efficient use of water (and resulting energy costs) within the factories is also being explored. Additionally, a study on the environmental impact of the Irish dairy industry, along with an assessment of the contribution water and wastewater treatment at the dairy processing factory make to this, is performed. Selected aspects of the project have been discussed in greater detail as a cluster of research communications, which are entitled:

- Efficient treatment of dairy processing wastewater in a laboratory scale Intermittently Aerated Sequencing Batch Reactor (IASBR) (Leonard et al. 2018a)
- Efficient treatment of dairy processing wastewater in a pilot scale Intermittently Aerated Sequencing Batch Reactor (IASBR) (Leonard et al. 2018b)

- Dominance of the genus *Polaromonas* in the microbial ecology of an Intermittently Aerated Sequencing Batch Reactor (IASBR) treating dairy processing wastewater under varying aeration rates (Gil-Pulido et al. 2018)
- Potential of using synthesised nano-zeolite for ammonium and phosphate immobilisation in dairy wastewater (Gao et al. 2018)
- Microbiological characterisation and impact of suspended solids on pathogen removal from wastewaters in dairy processing factories (Fitzhenry et al. 2018)
- What is the environmental impact of the dairy processing industry in the Republic of Ireland? (Finnegan et al. 2018)

The main aim of the DairyWater project is to aid the Irish dairy processing industry in becoming more sustainable. In order to ensure that the research remains relevant to the Irish dairy processing industry throughout the project, the research team worked closely with leading industry stakeholders and a project advisory board is in place, which included leading members of the dairy processing industry and government funded bodies. Due to the fact that there is very little detailed information available in the published literature, it was essential for the industry partners to provide data, host site visits and facilitate pilot-scale activities, during the project; thus enabling potential commercial benefits of this research to be realised. The project can be divided into the following main research areas, which are discussed in the remainder of this section:

- water consumption and dairy wastewater characterisation;
- dairy wastewater treatment technologies and microbial analysis;
- water re-use and rainwater harvesting; and
- environmental assessment.

# Water consumption and dairy wastewater characterisation

Initially, in order to gain a greater insight into the Irish dairy processing industry, a series of site visits, which included site data surveys and sampling of water and wastewater streams, took place. The feedback from these visits and surveys further reinforced the need to a reduction in water consumption at the factories and for more efficient and environmentally friendly solutions for wastewater treatment. Grab samples were used to investigate the water and wastewater streams within the dairy processing factories, which were taken during site visits due to time and resource constraints. The results of part of the survey, relating to the onsite water consumption and a breakdown of water usage, is given in Table 1 for 5 Irish dairy processing factories. Overall, the largest consumer of water on-site is related to cleaning, where a summary of the main water users within a dairy processing factory is shown graphically in Fig. 3. The wastewater produced, as a result of its high nutrient and microbial quantities, needs to be treated to reduce its environmental impact (such as eutrophication) on receiving waters. Therefore, a nutrient and microbial analysis was

performed on samples of this wastewater to give an indication of the typical concentrations of key nutrients, where the results are given in Table 2. However, it can be seen in Table 2 that these nutrients vary greatly from site-to-site and, based on discussions with site personnel, can vary greatly throughout the day at a single site. For example, in this analysis, ammonium nitrogen, ortho-phosphate and coliforms vary between 0.9–184.2 mg/l, 5–102 mg/l and 3–450 000 CFU/ml, respectively. The details relating to the operation of the facility and the results of the site data surveys were used as direct inputs in order to perform the environmental assessment of the Irish dairy processing industry.

# Dairy wastewater treatment technologies and microbial analysis

One of the most central aspects of the project is to investigate the potential for the intermittently aerated sequencing batch reactor (IASBR) technology as solution for treating dairy wastewater. This technology has been selected as it previously showed potential for treating high strength wastewater and the resources of the project only allow for the complete investigation of one wastewater treatment technology, although the research team acknowledges that there may be other suitable solutions. Its viability within the Irish dairy industry, along with the development of an operating procedure for the IASBR technology to allow it to perform efficiently when treating dairy wastewaters, is investigated during the project. This technology has a distinct advantage over current systems as it is a biological wastewater treatment system in which wastewater is completely mixed with the microorganisms in the system for the duration of the react phase in order to remove nutrients and treat the wastewater. Previously, the IASBR technology has been applied in the treatment of municipal wastewater (Henry et al. 2013), high strength slaughterhouse wastewater (Henry, 2014; Pan et al. 2014) and, at laboratoryscale, dairy wastewater (Tarpey, 2016). Since wastewaters with high biodegradability are suitable for treatment by biological processes (Pan et al. 2014), the IASBR system has the potential to treat dairy wastewater efficiently. Additionally, the technology has been shown, if managed correctly, to be superior in certain industries in the removal of nitrogen and phosphorus than the more traditionally used sequencing batch reactor technology for the treatment of the wastewater (Pan et al. 2013; Henry, 2014).

In order to develop the technology, experimental procedures at both laboratory-scale and pilot-scale IASBR systems will be investigated. Nutrients which are commonly known to be responsible for eutrophication, primarily nitrogen and phosphorus, (Pan et al. 2013) will be removed and their removal efficiencies will be monitored, as well as the operating cost and energy demand. This will aid in determining the most efficient and effective solution when compared technologies currently employed in the Irish dairy processing industry. The initial step of the project was to

Table 1.	Results from	a survey c	of 5 Irish daiı	y processing	factories	relating to	water	consumption,	including	a breakdowr	in water	usage
within th	ne factory											

	Site 1 Fluid milk Cream		Site 2 Milk powder Butter		C:4- 4	Site 5 Milk powder Butter		Site 6 Infant formula
Products processed					Site 4 Milk powder			
Milk processed (million litres)		50	20	06	80	58	37	125
Total water consumption (m <sup>3</sup> )	58 000		155 000		82 000	500 000		438 000
Water consumption (litre per litre milk)	consumption (litre per litre milk) 1.16 0.75		75	1.03	0.82		3.5	
CIP	56%	1.7%	33%	10%	75%	53%	7%	15%
Boilers & flushing	3%	0.1%			25%	13%	2%	2%
Cooling						22%	3%	
Intake	14%	0.5%						
Separators				4%				
Butter factory				8%				
Deaerator				8%				
Dryers			14%					24%
Evaporators			14%					
Tanker wash			7%	2%				
WWTP	7%	0.2%						
Other	17%	0.5%						59%

Legend: clean-in-place system (CIP); wastewater treatment plant (WWTP).

Table 2. Results from a nutrient and microbial analysis of samples of wastewater (prior to on-site wastewater treatment) at 6 Irish dairy processing factories

	pH _	TSS (mg/l)	COD	BOD <sub>5</sub>	$NH_4$	Nitrite	PO <sub>4</sub>	Coliforms (CFU/ml)	E. coli
Site 1	N/A	2370	9770	4900	7.8	0.144	5.3	$1.00 \times 10^{2}$	$8.35 \times 10^{1}$
Site 2	N/A	750	1430	N/A	0.9	4.5	5.0	N/A	N/A
Site 5	9.8	915	3310	3300	184.2	0.1	8.1	$4.50 \times 10^{4}$	$0.00 \times 10^{0}$
Site 7	5	490	3270	1900	20.1	0.065	31.7	$1.35 \times 10^{5}$	$2.55 \times 10^{1}$
Site 8	6.3	311	1210	1200	6.1	0.0	6.6	$4.50 \times 10^{5}$	$5.63 \times 10^{2}$
Site 9	N/A	205	1171	375	1	0	102	$3.00 \times 10^{0}$	$0.00 \times 10^{0}$

Legend: ammonium nitrogen (NH<sub>4</sub>); 5-day biochemical oxygen demand (BOD<sub>5</sub>); chemical oxygen demand (COD); colony-forming unit (CFU); *Escherichia coli* (*E. coli*); ortho-phosphate (PO<sub>4</sub>); test not performed (N/A); total suspended solids (TSS).

replicate the technology in a laboratory setting (using a laboratory-scale system containing three 81 cylindrical units, which is shown in Fig. 2, to gain a greater understanding of the technology and to create parameters from which a pilot-scale system could be designed. This laboratory-scale system not only proved the validity of the technology for the treatment of dairy wastewater, but also determined initial operating parameters for the pilot-scale IASBR system. The pilot-scale IASBR system has been designed and installed on-site at the wastewater treatment plant of an Irish dairy processing factory and has a working volume of 3000 l, which is shown in Fig. 2. The system is supplied with wastewater from the processing factory and is monitored daily using an on-site automated refrigerated sampler. These samples are tested weekly in order to monitor the performance of the system. This pilot-scale system gives a greater insight into the performance of the IASBR technology at a near commercial-scale.

In order to gain a greater understanding about the performance of the IASBR technology, the bacterial communities underpinning nutrient removal efficiencies arising from the laboratory-scale and pilot-scale systems have been profiled. Microorganisms are they key contributors in biological wastewater treatment processes and the knowledge of bacterial community structures contributes to the development of strategies for process optimisation (Daims et al. 2006). Therefore, during this research, next generation sequencing (NGS) methods will be applied to IASBR biomass samples. Sample collection is intended to provide representation of varying operational parameters at laboratory-scale (e.g. aeration rates), pilot-scale (e.g. cycle length) and varying influent compositions (e.g. synthetic vs. industrial influent). NGS outputs were subjected to comprehensive in silico analyses to determine microbial community structures and to facilitate correlations between relative microbial abundance and reactor performance (Zhang et al. 2012; Weissbrodt



Fig. 2. Experimental systems used as part of the DairyWater project. Clockwise (from top left): laboratory-scale intermittently aerated sequencing batch reactor (IASBR) system; laboratory-scale pulsed ultra violet (PUV) system; low pressure ultra violet (LPUV) system; pilot-scale IASBR system.

et al. 2014). In silico metabolic profiling of the bacterial community was also performed to predict the abundance of functional gene families and to identify potential, key contributors to nutrient bioremediation (Ahmed et al. 2017). In an effort to corroborate the modelled outputs with respect to NGS determined dominant groups and metabolic functionalities, spatial distribution analyses was also performed. Fluorescence in situ Hybridisation (FISH) and quantitative polymerase chain reaction (gPCR) will be used for the enumeration of specific targeted microbial groups of interest and their spatial organisation within the representative biomass samples (Nielsen et al. 2009; van Loosdrecht et al. 2016). This study seeks to address the knowledge gap which exists in relation to the microbial community structures underpinning IASBR application. Such knowledge is critical to the optimisation of the IASBR technology, where a reliance on key microbiota can be demonstrated. While the target application of this project is the dairy processing sector, the findings have potential significance in the treatment of other agri-industrial wastewaters.

The use of nanomaterials to improve the efficiency of wastewater treatment processes in this sector has not previously been well explored. Therefore, a selection of nanomaterials have been tested in order to seek an appropriate one for simultaneous removal and recovery of nitrogen, phosphorus and other contaminants for the treatment of dairy wastewater. The nanomaterials tested in the study include nano-zeolite, surface modified nano-zeolite, carbon nanotubes and activated charcoal (Camblor et al. 1998). The physicochemical characteristics of nano-zeolite, such as high mechanical and chemical resistance and its high surface area, have formed the basis for its widespread use in catalysis, separation, and ion-exchange (Song et al. 2005). In this study, the feasibility of using coal fly ash to synthesise nano-zeolite was studied, along with the nitrogen and phosphorus adsorption efficiencies of the nano-zeolite. In addition, surface modifications were conducted by adhering Cu, Mg, Fe, Zn, Ca ions and Hexadecyltrimethylammonium (HDTMA) onto the surface of nano-zeolite in order to change the surface properties and enhance the contaminants removal efficiency.



Fig. 3. Schematic summarising processing within a dairy processing factory and the main water users within the manufacture of dairy products.

Comparative batch experiments were conducted to study the effect of sorption time, pH values and dosage of nanomaterials on the contaminant removal efficiency. Additionally, in order to minimise the operation cost, the regeneration of nanomaterials after use has been considered. A microbial fuel cell (MFC) has been chosen instead of typical acid/alkaline wash, as MFCs possess the ability of generating electricity from organic matter using exoelectrogenic bacteria during wastewater treatment (Logan, 2008). The possibility of regenerating the nanomaterials and energy recovered by using MFCs will be examined further in this research.

#### Water reuse and rainwater harvesting

Given the significant quantities of water consumed by the Irish dairy processing industry, the reuse of wastewater within the sector may be necessary for economic, regulatory and sustainability purposes (Geraghty, 2011). A key barrier to water reuse within this sector (and indeed other industries) is the implementation of pathogen removal. This study compares a flow-through pulsed ultraviolet (PUV) system (Barrett et al. 2016) and a continuous low-pressure UV (LPUV) system, which are both shown in Fig. 2, as potential technologies for (i) tertiary wastewater treatment plant effluent disinfection and (ii) disinfection systems to restore dairy wastewater to reusable levels for certain dairy processing factory practices. The two systems were compared in terms of energy usage and bacterial UV dose response and the impact of dairy wastewater parameters, primarily suspended solids (SS), on the system, along with the inactivation efficiency of *E. coli*.

Preliminary work included a nationwide dairy processing factory site survey followed by the microbiological characterisation of water/wastewater streams. The sample streams tested included cooling waters, condensate water and wastewater treatment plant (WWTP) influent and effluent. Feedback from this survey analysis indicated low levels of water reuse practices at dairy processing factories however the introduction of microbiological effluent discharge standards is likely in the near future. The key findings from the microbiological characterisation included the detection of faecal indicators in 11 out of the 12 samples collected including all three WWTP effluent samples. Pathogenic bacteria typically associated with the dairy industry were also detected; Listeria monocytogenes was present in all samples at two sites, while Campylobacter spp. was also present in cooling waters and wastewater treatment plant effluent at one of the sites. Bacillus pumilus and Bacillus subtilis endospores have been investigated as they are noted to exhibit increased resistance to LPUV in comparison to vegetative cells (WRF, 2010; Boczek et al. 2016).

From the results, inert suspended solids do not appear to inhibit PUV or LPUV efficiency at high SS concentrations (30 mg/l and above) while organic particles impacted on the PUV and to a lesser extent on LPUV efficiency. A PUV dose of approximately 2000 mJ/cm<sup>2</sup> was required for a 2 log inactivation of *B. pumilus* endospores while a LPUV dose of 30 mJ/cm<sup>2</sup> was required for a 5 log inactivation. The effect of agar supplement manganese sulphate (MnSO<sub>4</sub>) on the enhanced UV resistance of endospores has also been analysed. Cultivation agar supplemented with MnSO<sub>4</sub> appears to increase the UV resistance of Bacillus spp. endospores when exposed to both PUV and LPUV. Preliminary investigations into the energy efficiency of both systems indicated that the LPUV system exhibits a higher efficiency of converting electrical energy to UV energy in comparison to the PUV system. The photoreactivation potential of dairy pathogens post PUV and LPUV flow-through disinfection is to be investigated, along with 'on-site' dairy wastewater been incorporated into UV disinfection trials to evaluate the potential of UV treatment for low-level water reuse at dairy processing factories.

As dairy production increases, so too will the vast volumes of water required for dairy product output, which currently stands at  $2.5 \text{ m}^3/\text{m}^3$  of milk processed (Geraghty, 2011). Legislation regarding the discharge limitations of dairy wastewater effluent is becoming increasingly stringent along with initiatives to conserve and reuse water within the industry. Thus, alongside the potential for reusing treated wastewaters, there is significant potential to harvest and use rainwater. A key consideration for the industry in implementing rainwater harvesting is the potential for a positive and benefit to cost ratio and a relatively short pay-back time. Given the relatively low costs of water in Ireland (in some cases where water is withdrawn from private sources there may not be any charge associated with water use) this project focused on the development of design tools that would enable the industry determine the viability of implementing rainwater harvesting at a number of case-study sites. The toolkit also includes a life cycle assessment module which modelled expected life cycle emissions from the construction and operation/maintenance of the rainwater harvesting system. The tool was trialled, based on case-study data from a number of dairy facilities in Ireland. In general it was shown that given current water tariffs, the costs of deploying rainwater harvesting at each site and the potential end-uses for rainwater such systems might not at this time be economically viable.

#### **Environmental assessment**

Since the main aim of the DairyWater project is to help form a more sustainable Irish dairy processing industry, particularly from an environmental point-of-view, it is first necessary to establish a benchmark for the industry. This study will also aid in determining the true impact of water and wastewater treatment technologies on the overall environmental impact associated with the manufacture of dairy products. Therefore, life cycle assessment (LCA) is being used in order to estimate the current environmental impact of the Irish dairy industry. Additionally, LCA will be used to evaluate the environmental impact of the technologies developed during the DairyWater project in order to ensure that they will have a positive impact on the environment. In order to ensure the accuracy of this study, it is structured in accordance with the LCA guidelines of the International Organisation for Standardisation (ISO, 2006) and the LCA methodology for the dairy industry published by the International Dairy Federation (FIL-IDF, 2015).

Initially, a macro-scale LCA of the Irish dairy industry, which assessed the global warming potential associated with the main Irish dairy products, was performed (Finnegan et al. 2017a). The life cycle stages included in this study are raw milk production, raw milk transportation and dairy processing and publically available data, along with national statistics, were used. This study identified the critical processes, inputs and emissions that are necessary in delivering an accurate LCA. The main LCA performed during the study is an environmental LCA of selected dairy products in the Republic of Ireland, which included a number of environmental impact categories. This analysis was performed for a number of dairy products, including milk powder, butter, fluid milk, cream, infant formula, cheese and whey powder. So as to perform this analysis, data was collected from 11 dairy processing factories within the Republic of Ireland for 2013, which process approximately 49% of the total raw milk processed. Due to the availability of data, comprehensive studies relating to the production and manufacture of milk powder and butter have been carried out (Finnegan et al. 2017b, c). From these studies, the environmental impact of these two products have been estimated and the processes (within the dairy processing factories) that are the most significant contributor to the impact have been identified.

Along with performing an assessment of Irish dairy products, both an economic cost-benefit analysis and an environmental assessment (using LCA) of the potential positive impact of novel technologies investigated in this project will be carried out. The results of this analysis will be compared to existing dairy water and wastewater treatment technologies in order to demonstrate how sustainability may be improved.

The technologies being developed in this project have the potential to greatly increase the sustainability of the Irish dairy processing industry, while also returning a financial benefit. For example, the proposed IASBR system is a biological wastewater treatment system and, thus, does not require any chemicals, which is one of the largest expenses when treating wastewater. Additionally, as Ireland's agricultural and agri-food sector continues to grow, it has been suggested that a carbon tax on dairy processors, similar to the carbon tax paid by motorists for their cars and householders for their fuel bills, may be introduced as an incentive to reduce emissions (Melia, 2015).

At current milk processing levels, many dairy processors in Ireland are at the limits of their emissions to water, as set by the Irish environmental protection agency (EPA). As a result of the increase in raw milk production due to the European milk quotas being abolished, processors will be under increased pressures to remain within these limits. Additionally, in 2015, domestic water charges were introduced in the Republic of Ireland for the first time. Even though this didn't affect the Irish dairy processing industry, as groundwater and surface waters are the main source, it may in the future if extraction charges are introduced. Therefore, water reuse, particularly the reuse of wastewater as process water, may be necessary to reduce the water footprint of dairy processing factories and to reduce costs associated with production. The tertiary treatment systems proposed in this project, PUV and LPUV systems, would be vital in exploiting the potential of water reuse together with meeting the requirements of dairy processing factories to comply with strict regulations when emitting treated effluent to rivers and lakes, such as those implemented by the Water Framework Directive (WFD - 2000/60/EC), Surface Waters Regulations (S.I. No. 272 of 2009) and the European Community Shellfish Waters Directive (2006/113/EC).

In conclusion, this article outlines the project aims and technologies that are being explored in the DairyWater project, where the main aim is to aid the Irish dairy processing industry in becoming more sustainable. As Europe moves into a post-quota era, it will be essential for Ireland to improve its sustainability and maintain its 'green' image if it is to remain competitive internationally. Therefore, the DairyWater project has the potential to position Ireland at the forefront of European, or indeed international, research in this sector. The project strives to make the Irish dairy processing industry more efficient and environmentally sustainable, which will lead to greater potential for exports and increased international competitiveness for Irish products, along with stimulating job creation. In addition, the results of the study are suitable to be applied internationally as, although milk production conditions may not be pasture based, processing, and many of the associated environmental impacts and challenges that arise, are generally the same when using modern processing technologies.

The authors would like to acknowledge the funding provided by the Department of Agriculture, Food and the Marine for the DairyWater project (Ref.: 13-F-507); for additional details see http://www.dairywater.ie. Additionally, the DairyWater research team would like to the members of the Irish dairy sector for their cooperation and contributions throughout the research project.

#### References

- Ahmed W, Staley C, Sidhu J, Sadowsky M & Toze S 2017 Amplicon-based profiling of bacteria in raw and secondary treated wastewater from treatment plants across Australia. *Applied Microbiology and Biotechnology* **101** 1253–1266
- Barrett M, Fitzhenry K, O'Flaherty V, Dore W, Keaveney S, Cormican M & Clifford E 2016 Detection and inactivation of pathogenic Norovirus employing UV tertiary treatment in wastewater treatment facilities. *Science of the Total Environment* **568** 1026–1036
- Barry M 2012 Reducing Emissions in the World's Most Sustainable Argi-Food Industry. Dublin, Ireland: Irish Dairy Industries Association
- Boczek LA, Rhodes ER, Cashdollar JL, Ryu J, Popovici J, Hoelle JM, & Ryu H 2016 Applicability of UV resistant *Bacillus pumilus* endospores as a

human adenovirus surrogate for evaluating the effectiveness of virus inactivation in low-pressure UV treatment systems. *Journal of Microbiological Methods* **122** 43–49

- Camblor M, Corma A & Valencia S 1998 Characterization of nanocrystalline zeolite Beta. *Microporous and Mesoporous Materials* 25 59–74
- **CSO** 2017 *Statistical Product Milk Production*. Dublin, Ireland: Central Statistics Office
- Daims H, Taylor MW & Wagner M 2006 Wastewater treatment: a model system for microbial ecology. Trends in Biotechnology 24 483–489
- Farrelly P, Crosse S, O'Donoghue P, Whyte S, Farrelly PB, Burns T, Byrne D, Holmes O, Maklin R, McKearney JJ and Salley F 2014 Environmental Analysis of Scenarios Related to the Implementation of Recommendations in Food Harvest 2020. Dublin, Ireland: The Department of Agriculture, Food and the Marine
- FIL-IDF 2015 A Common Carbon Footprint Approach for the Dairy Sector: The IDF Guide to Standard Life Cycle Assessment Methodology. Brussels, Belgium: International Dairy Federation
- Finnegan W, Goggins J, Clifford E & Zhan X 2017a Global warming potential associated with dairy products in the Republic of Ireland. *Journal of Cleaner Production* 163 262–273
- Finnegan W, Goggins J, Chyzheuskaya A & Zhan X 2017b Global warming potential associated with Irish milk powder production. Frontiers of Environmental Science & Engineering 11 12–19
- Finnegan W, Goggins J, Clifford E & Zhan X 2017c Environmental impacts of milk powder and butter manufactured in the Republic of Ireland. *Science of the Total Environment* 579 159–168
- Finnegan W, Goggins J & Zhan X 2018 What is the environmental impact of the dairy processing industry in the Republic of Ireland? *Journal of Dairy Research* 85
- Fitzhenry K, Rowan N, Finnegan W, Zhan X & Clifford E 2018 Microbiological characterisation and impact of suspended solids on pathogen removal from wastewaters in dairy processing factories. *Journal of Dairy Research* 85
- Gao F, Xiao L & Zhang H 2018 The potential of nanomaterials in the treatment of dairy processing wastewater. *Journal of Dairy Research* 85
- Geraghty R 2011 Resource Efficiency in Ireland's Dairy Processing Sector. Dublin, Ireland: Enterprise Ireland
- Gil-Pulido B, Tarpey E, Finnegan W, Zhan X, Dobson A & O'Leary N 2018 Dominance of the genus Polaromonas in the microbial ecology of an Intermittently Aerated Sequencing Batch Reactor (IASBR) treating dairy processing wastewater under varying aeration rates. *Journal of Dairy Research* 85
- Henry LG 2014 Comparison of Intermittently Aerated Sequencing Batch Reactors (IASBRs) and Conventional Sequencing Batch Reactors (cSBRs) in Wastewater Treatment, PhD Thesis. Galway: National University of Ireland
- Henry LG, Xiao L, Zhan X 2013 Effect of aeration rate on domestic wastewater treatment using an intermittently aerated sequencing batch reactor (IASBR) technology. *Environmental Engineering and Management Journal* 12 1329–1334
- IFA 2012 Dairy Fact Sheet. Dublin, Ireland: Irish Farmers Association
- ISO 2006 ISO 14040: Environmental Management Life Cycle Assessment – Principles and Framework. Geneva, Switzerland: International Organization for Standardization
- Leonard P, Finnegan W, Tarpey E & Zhan X 2018a Efficient treatment of dairy processing wastewater in a laboratory scale Intermittently Aerated Sequencing Batch Reactor (IASBR). *Journal of Dairy Research* 85
- Leonard P, Finnegan W, Barrett M & Zhan X 2018b Efficient treatment of dairy processing wastewater in a laboratory scale Intermittently Aerated Sequencing Batch Reactor (IASBR). *Journal of Dairy Research* 85
- Logan BE 2008 Microbial Fuel Cells. New Jersey, USA: John Wiley & Sons Melia P 2015 Scientist backs tax on Ireland's food processors. Dublin,
- Ireland: Irish Independent 12/12/2015
- Nielsen PH, Daims H, Lemmer H, Arslan-Alaton I & Olmez-Hanci T 2009 FISH Handbook for Biological Wastewater Treatment. IWA Publishing, London, UK
- NMA 2017 Annual Report & Accounts 2016. Dublin, Ireland: National Milk Agency
- Pan M, Chen T, Hu Z & Zhan X 2013 Assessment of nitrogen and phosphorus removal in an intermittently aerated sequencing batch reactor

(IASBR) and a sequencing batch reactor (SBR). Water Science & Technology  ${\bf 68}$  400–405

- Pan M, Henry LG, Liu R, Huang X & Zhan X 2014 Nitrogen removal from slaughterhouse wastewater through partial nitrification followed by denitrification in intermittently aerated sequencing batch reactors at 11C. Environmental Technology 35 470–477
- Song W, Li G, Grassian VH & Larsen SC 2005 Development of improved materials for environmental applications: nanocrystalline NaY zeolites. *Environmental Science & Technology* **39** 1214–1220
- Tarpey E 2016 An Investigation into the Use of IASBRs for Treatment of Dairy Processing Wastewater, MEngSc Thesis. Galway: National University of Ireland
- van Loosdrecht MC, Nielsen PH, Lopez-Vazquez CM & Brdjanovic D 2016 Experimental Methods in Wastewater Treatment. London, UK: IWA Publishing
- Weissbrodt DG, Shani N & Holliger C 2014 Linking bacterial population dynamics and nutrient removal in the granular sludge biofilm ecosystem engineered for wastewater treatment. *FEMS Microbiology Ecology* 88 579–595
- WRF 2010 Challenge Organisms for Inactivation of Viruses by Ultraviolet Treatment. Denver, CO, USA: Water Research Foundation
- Zhang T, Shao MF & Ye L 2012 454 pyrosequencing reveals bacterial diversity of activated sludge from 14 sewage treatment plants. *The ISME Journal* 6 1137–1147

Contents lists available at ScienceDirect

# ELSEVIER





#### Review

# Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – A 20 year systematic review



CrossMark

#### Erin Jo Tiedeken<sup>a</sup>, Alexandre Tahar<sup>a</sup>, Brendan McHugh<sup>b</sup>, Neil J. Rowan<sup>a,\*</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Co. Westmeath, Ireland

<sup>b</sup> Marine Institute, Rinville, Oranmore, Co. Galway, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- Three EU watch list substances of emerging concern in receiving waters are reviewed
- Pharmaceuticals diclofenac and EE2 along with natural hormone E2 reported above environmental quality standards
- Under monitoring of these substances of emerging concern in many EU member countries
- Need for more sensitive estrogen detection methods to meet WFD limits
- Control measures frequently do not fully remove these harmful chemicals.

#### A R T I C L E I N F O

Article history: Received 12 July 2016 Received in revised form 25 August 2016 Accepted 11 September 2016 Available online xxxx

Editor: D. Barcelo

Keywords: Water framework directive Diclofenac 17-Beta-estradiol (E2) 17-Alpha-ethynylestradiol (EE2) Sources Receptors Control Watch list



#### ABSTRACT

Pollution of European receiving waters with contaminants of emerging concern (CECs), such as with 17-beta-estradiol (a natural estrogenic hormone, E2), along with pharmaceutically-active compounds diclofenac (an antiinflammatory drug, DCL) and 17-alpha-ethynylestradiol (a synthetic estrogenic hormone, EE2)) is a ubiquitous phenomenon. These three CECs were added to the EU watch list of emerging substances to be monitoring in 2013, which was updated in 2015 to comprise 10 substances/groups of substances in the field of water policy. A systematic literature review was conducted of 3952 potentially relevant articles over period 1995 to 2015 that produced a new EU-wide database consisting of 1268 publications on DCL, E2 and EE2. European surface water concentrations of DCL are typically reported below the proposed annual average environmental quality standard (AA EQS) of 100 ng/l, but that exceedances frequently occur. E2 and EE2 surface water concentrations are typically below 50 ng/l and 10 ng/l respectively, but these values greatly exceed the proposed AA EQS values for these compounds (0.04 and 0.035 ng/l respectively). However, levels of these CECs are frequently reported to be disproportionately high in EU receiving waters, particularly in effluents at control points that require urgent attention. Overall it was found that DCL and EE2 enter European aquatic environment mainly following human consumption and excretion of therapeutic drugs, and by incomplete removal from influent at urban wastewater treatment plants (WWTPs). E2 is a natural hormone excreted by humans which also experiences incomplete removal during WWTPs treatment. Current conventional analytical chemistry methods are sufficiently sensitive for the detection and quantification of DCL but not for E2 and EE2, thus alternative, ultra-trace, time-integrated monitoring techniques such as passive sampling are needed to inform water quality for these estrogens. DCL appears

\* Corresponding author at: Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co. Westmeath, Ireland. *E-mail address:* nrowan@ait.ie (N.J. Rowan). resistant to conventional wastewater treatment while E2 and EE2 have high removal efficiencies that occur through biodegradation or sorption to organic matter. There is a pressing need to determine fate and behaviour of these CECs in European receiving waters such as using GIS-modelling of river basins as this will identify pressure points for informing priority decision making and alleviation strategies for upgrade of WWTPs and for hospital effluents with advanced treatment technologies. More monitoring data for these CECs in receiving waters is urgently needed for EU legislation and effective risk management.

© 2016 Elsevier B.V. All rights reserved.

#### Contents

1.	1. Introduction										
2.	Materials and methods										
	2.1.	System	atic review protocol and defining search parameters								
	2.2.	Article	review and selection								
3.	Resul	ts and dis	cussion								
	3.1.	General	overview of the database								
	3.2.	Bibliogr	aphic analysis: State of European research on DCL E2 and EE2								
		3.2.1.	Pharmaceuticals studied								
		3.2.2.	Research theme studied								
		3.2.3.	Hormones: chemical vs biological monitoring method								
		3.2.4	Scale of the studies								
		325	Repartition of the studies by country 1146								
	33	Sources	and vectors of DCL F2 and FF2 1146								
	5.5.	331	General sources of CECs 1146								
		337	Sherific sources of DCL F2 and FF2 1147								
	34	Recento	res and occurrence of diclofenac F2 and FF2 in Furopean waters								
	5.4.	3/1	Occurrence of diclofenac								
		3.4.1	Occurrence of P2								
		2/2	Occurrence of EE								
	35	J.4.J. Monito	ordine for DL E 2 and EE2								
	J.J.	2 5 1	Monitoring of diclofonac								
		252	Monitoring of Parlotenae								
		J.J.Z. 2 E 2	Monitoring of L2 and L2								
	26	S.S.S.									
	5.0.	2.01	Medsules								
		3.0.1.									
		3.0.2.	Control measures of E2 and EE2.								
4	C	3.6.3.	Practicalities of implementing various tertiary technologies for Wastewater treatment								
4.	4. Conclusions and ruture research needs										
Ackr	nowled	igements									
Refe	rences										

#### 1. Introduction

Water is an essential resource, crucial to all living organisms and for a diversity of human activities (Barbosa et al., 2016). Drinking and food preparation, support of the natural environment and a growing economy all require a healthy and secure water supply (Rowan, 2011). Unfortunately, there are significant pressures on this fragile resource. Natural and along with other anthropogenic substances such as pharmaceuticals, pesticides, industrial compounds, personal care products, steroid hormones, drugs of abuse and others, end up in surface water, ground water and vital drinking water (Ribeiro et al., 2015) Specifically, this review focuses on monitoring, sources, receptors and control measures for addressing the pharmaceutically-active compounds diclofenac (DCL) and the synthetic hormone 17-alpha-ethinylestradiol (EE2) along with the natural hormone 17-beta-estradiol (E2), which hereafter will be referred to as contaminants of emerging concern (CECs). Pharmaceuticals and their pharmaceutically active metabolites/transformation products are a class of CECs that are widely used in human and veterinary medicine and are essential to modern healthcare (Fent et al., 2006; Nikolaou et al., 2007). Nevertheless, there are growing concerns about the negative impacts that may result from continuous contamination of the environment with pharmaceutically-active compounds (Barbosa et al., 2016; Verlicchi and Zambello, 2016). This research is important because of the potential toxic effects for aquatic biota and human health that may result from chronic exposure to such CECs (Fent et al., 2006; Kümmerer, 2009; Nikolaou et al., 2007). Characteristics specific to this class of environmental contaminants can however present significant challenges for research. For example, CECs exhibit wide variation in function, chemical structure and physiochemical properties, making it difficult to generalize about their behaviour, persistence or impact in the environment. CECs are also designed to be biologically active, have a specific mode of action and to be persistent in the body, meaning they can impact humans and wildlife at trace concentrations which are often hard to detect and quantify using traditional analytical methods (Fent et al., 2006).

Although there are no legal discharge limits for micropollutants into the environment, some regulations have been published in the last few years (Barbosa et al., 2016). CECs in the aquatic environment primarily originate from use in human medicines, however certain classes are also heavily used in veterinary practices (e.g. anti-inflammatory drugs, antibiotics) (Fent et al., 2006; Zhou et al., 2009). A large number of CECs have been detected in WWTPs influents and effluents and surface, ground and drinking water worldwide in recent years (Barreiros et al., 2016; Heberer, 2002; Nikolaou et al., 2007; Ternes, 1998; Zhou et al., 2009; Verlicchi and Zambello, 2016). In fact, it is now established that throughout the developed world, CECs are ubiquitous at µg to ng per litre levels in the aquatic environment (Nikolaou et al., 2007), although the concentrations of specific compounds depend on usage patterns in different countries and can vary temporally (Verlicchi et al., 2012). The impacts of chronic exposure to trace concentrations of many CECs on wildlife and human health may be severe (e.g. Verlicchi et al., 2012), thus it is critical to limit as much as possible the concentrations of this class of contaminants in our waterways. Certain CECs can specifically impact the endocrine system of humans or wildlife; such chemicals are part of emerging pollutants known as endocrine disrupting chemicals (EDCs). Much of the growing interest in this field of research stems from fears that chronic exposure to EDCs (in bathing or drinking water, for example) may be linked to adverse human health conditions such as declining male fertility, birth defects, and breast and testicular cancer (Nikolaou et al., 2007). Furthermore negative impacts of EDCs exposure on wildlife may include severe consequences such as feminisation in fish (Sumpter and Johnson, 2008). Similar to CECs as a whole, EDCs are mainly thought to be transported into the aquatic environment via incomplete removal at WWTPs (Nikolaou et al., 2007).

Until recently, environmental regulations worldwide had not required explicit testing for any CECs in water bodies. However given the growing concern about contamination of the aquatic environment with these compounds, legislation has recently begun to acknowledge this potential problem. The Water Framework Directive (WFD, 2000/ 60/EC) is an overarching piece of European environmental legislation aimed at protecting and improving water quality throughout the EU. The WFD committed EU Member States to achieve good qualitative and quantitative status of all water bodies by 2015. In order to reach this goal, certain chemicals identified by Annex X of the WFD have been deemed priority substances; these chemicals (e.g. some pesticides, metals such as lead or mercury, organic volatile compounds and other organics such as polycyclic aromatic hydrocarbon) must be monitored by all member states and cannot exceed specific concentration thresholds in surface waters (defined by the legislation as Environmental Quality Standards, or EQSs). Furthermore, article 16(4) of this legislation requires that the list of priority substances must be reviewed and adjusted as appropriate at regular intervals. As such, directive 2013/39/EU of 12 August 2013 added a further 12 substances to Annex X of the WFD. In addition, Article 8b of Directive 2013/39/EU states that "the Commission shall establish a watch list of substances for which EU-wide monitoring data are to be gathered for the purpose of supporting future prioritisation exercises." In response to growing EU concern about the release of untreated CECs into the aquatic environment, three compounds were included in the first watch list in 2013: diclofenac (CAS# 15307-79-6, hereafter referred as DCL), 17-beta-estradiol (CAS# 50-28-2, hereafter referred as E2) and 17-alpha-ethinylestradiol (CAS# 57-63-6, hereafter referred as EE2). It is relevant to note that the European Commission implemented decision 495 of 20 March 2015 that expanded substances or groups of substances on the watch list to 10 in the field of water policy. Besides the two pharmaceuticals (DCL and EE2) and the natural hormone (E2) that were previously recommended to be included by the Directive 39/2013/EU, the first watch list of 10 substances/groups of substances also refers to the three macrolide antibiotics (clarithromycin, azithromycin and erythromycin) other natural hormone (E1), some pesticides (oxadiazon, methiocarb, imidacloprid, thiacloprid, thiamethoxam, clothianidin, acetamiprid and triallate), a UVB filter (2-ethinylhexyl 4-methoxcinnamate) and an antioxidant (2,6-di-tert-butyl-4-methylphenol) commonly used as a food additive. This review focuses solely on the first three substances DCL, E2 and EE2 as there is a requirement to investigate policy implications for Ireland of these PhACs in receiving waters in the first instance. An overview of the EU policy in the water field as it relates to commonly used conventional and advanced treatment processes in the aqueous matrices of these 10 substances/groups can be found in Barbosa et al. (2016). The EU-wide monitoring data that will be produced in the next few years will help legislators determine whether or not these compounds are ultimately added to the list of priority substances from Annex X of the WFD. The WFD requires that all EU member states prepare river basin management plans (RBMPs) to address the many issues relating to water quality and protection in a holistic manner. These RBMPs identify the main pressures and activities affecting water status and propose environmental objectives that must be achieved during certain time periods. The recent European legislation on DCL, E2 and EE2 mentioned above has been identified as potentially significant water management issue that may need to be addressed in the next round of RBMPs (due for publication in 2017).

The overall aim of this literature review was to identify and evaluate all previous relevant EU-wide studies on contamination of the aquatic environment with the three watch list pharmaceuticals DCL, E2 and EE2 in order to anticipate their entrance in the WFD priority substances list and to identify gaps in knowledge aiming at guiding future research. This review is directed towards at-risk industries, companies, researchers, regulators and any sectors that would be affected by the addition of these compounds to future iterations of the WFD priority substance list (toxicology, water treatment, chemical analysis, biology, regulation). Risk assessment was not included in this literature review as it has been addressed by other authors (Camacho-Muñoz et al., 2012; Ferrari et al., 2004; Futran Fuhrman et al., 2015). However, this systematic literature review addresses four main research questions for each compound:

- 1) What are the likely sources/entry points of these CECs into European aquatic environment?
- 2) What are the likely receptors and loadings in European waters?
- 3) What monitoring methods are currently employed to measure aquatic concentrations of these CECs, and what are the current limits of detection/quantification?
- 4) What control measures (including both source control and treatment options) are effective (or potentially effective) and employed for lowering concentrations of these compounds in the aquatic environment?

#### 2. Materials and methods

#### 2.1. Systematic review protocol and defining search parameters

Even a cursory search of the literature reveals a vast amount of published material regarding the sources, receptors, monitoring and control measures of DCL, E2 and EE2 (Fatta-Kassinos et al., 2011; Johnson et al., 2013; Qian et al., 2015). Consequently this literature review was carried out using a defined systematic approach that answers research questions based on the published evidence, which is gathered using a predefined protocol that was adapted from the Centre for Evidence-Based Conservation's (CEBC) "Guidelines for Systematic Review in Conservation and Environmental Management" (Pautasso, 2013; Pullin and Stewart, 2006). The protocol comprised defining search parameters (databases to be searched, search times, types of publications), selecting search terms, developing eligibility (inclusion/exclusion) criteria, and conducting the literature search and carrying out the article review and selection process to produce publication database and bibliographic analysis. The article review was a two-step process including both a title and abstract filterStudies on the sources, receptors/monitoring and control measures of DCL, E2 and EE2 were identified using the Scopus database and from professional networks that included grey literature sources or sources that would not be returned by the database search (such as PhD theses or government reports) (Pullin and Stewart, 2006). The search was limited to literature published from 1995 to 2015 to ensure the publications included in the final database were up-to-date. The mid 1990s reflected time period when this field of research was in its infancy (Qian et al., 2015). Search terms were selected to ensure all potentially relevant articles were returned from the database searches. Two separate searches were run, one for DCL and one combined search for E2 and EE2. The E2 and EE2 searches were

combined due to a high percentage of overlap in these search results. For both final searches, results were limited to articles, articles in press or review papers. All 28 EU member states were included, as well as Switzerland, Norway and Turkey. Terms for each of the two searches included "water" and "wastewater" in order to focus on articles considering the CECs in aquatic matrices. In order to cover all relevant research, search terms included the class of CECs describing each drug of interest (i.e. "NSAID" or "estrogen") and all relevant synonyms for each specific compound. In order not to miss articles considering the veterinary usage of DCL, which can be a significant source of environmental pollution (Boxall, 2010; Hunt et al., 2015), the term "veterinary" was also included. A list of eligibility criteria was developed so that once all of the potentially relevant articles were located through the searches described above, articles for inclusion in the database could be distinguished (Table 1).

#### 2.2. Article review and selection

Once all potentially relevant articles were identified through the searches, a selection process was undertaken to find articles for inclusion in the final database framed upon meeting eligibility criteria (Table 1). Title and abstract review were undertaken by two researchers with 10% overlap in order to validate consistent choices. During the abstract review, additional fields were added to the spreadsheet by the reviewer (Fig. 1), which were organised into six domains namely topic of article, monitoring type, compounds studies, analytical methods used, study type, and country study was performed in. Articles with authors or fieldwork from multiple countries were counted as full publications for each country, rather than fractionally (Qian et al., 2015). These additional fields were filled in by reading the abstract, or if necessary, by downloading and reading the full-text of the article. The only exception was the analytical method employed for detection; this field was only filled out if the method was specified in the abstract. These additional fields, as well as the bibliographic information provided by Scopus, were utilized to conduct the bibliographic analysis (Section 3.2).

#### 3. Results and discussion

The aim of this systematic literature review was to evaluate current state of knowledge on contamination of the European aquatic environment with DCL, E2 and EE2, especially in regards to sources, receptors, monitoring and control measures. The following sections address the specific research questions this systematic review was concerned with: Sections 3.1 and 3.2 report the results from the bibliographic analysis, Section 3.3 details the key findings on the sources of these CECs in the aquatic environment; Section 3.4 discusses the receptors and

#### Table 1

Eligibility criteria for systematic literature review; used for title and abstract filter.

Eligibility criteria

- Must specifically discuss at least one of the three compounds of interest
   Cannot focus exclusively on impacts of compound for human/animal/plant health
- Exclude papers that focus only on ecological/environmental/toxicological impacts unless they also discuss relevant sources, receptors/monitoring or control measures
- Exclude clinical trial studies
- Must include some specific information on sources, receptors/monitoring or control measures
- Cannot focus on exposure routes other than water
- Study cannot be purely chemical, i.e. determining a chemical coefficient
- Exclude any papers on leaching of chemicals from bottled water/plastics
- Must be peer reviewed original article or review, or article in press
   Must be published between 1995-May 2015
- Must be published between 1995-May 2015
- Research must be conducted in Europe or by at least one author affiliated with a European country
- Article must be written in English
- Full text must be available

concentrations of these CECs in a European context; and Section 3.5 discusses the effectiveness and challenges associated with monitoring methods used to detect these compounds. Finally, Section 3.6 discusses DCL, E2 and EE2 current and potential control measures.

#### 3.1. General overview of the database

Even following strict exclusion criteria (see Section 2.1), the systematic review identified a large number of peer-reviewed publications on the sources, receptors/monitoring and control measures for DCL, E2 and EE2. Fig. 2 demonstrates the enormous number of articles returned by our searches, and the number of articles excluded (and reasons for exclusion) during the title and abstract filters. The database of publications and the summary information regarding this database (bibliographic analysis, Section 3.2) include 1268 publications deemed eligible by the systematic review protocol. Published review studies were analysed for data on monitoring, source, receptors and control measures for Sections 3.3 to 3.6 where summary data on three topics was extracted: (i) concentrations of DCL, E2 or EE2 in influent or effluent, and their removal efficiencies during various wastewater treatments; (ii) concentrations of these three CECs in surface, ground or drinking water; and (iii) methods of detection and limits of detection (LODs) for each of the three compounds.

#### 3.2. Bibliographic analysis: State of European research on DCL, E2 and EE2

Bibliographic analyses are particularly useful for fields with large bodies of research that are difficult or impossible to summarize via traditional, full-text review studies (Belter & Seidel, 2013). They are also important for defining gaps in the literature and directing future research (Qian et al., 2015). This bibliographic analysis originates from the database of publications created during the systematic review; it summarizes the state of European research on DCL, E2 and EE2 from 1995 to May 2015 (details of the methodology used to create the database are provided in Section 2).

#### 3.2.1. Pharmaceuticals studied

EU database constituted of 628, 697, and 665 EU studies reported on DCL, E2 and EE2 respectively as per alignment with eligibility criteria (Fig. 1). Many of the individual studies in the database reported on more than one of these CECs. In particular, studies that investigated hormones tended to include both the natural steroid estrogen E2 as well as the synthetic EE2. There are a large number of total studies (>600) focused on each of these three CECs, however slightly more research has been published on E2 and EE2 when compared with DCL; that may be due to particular concerns regarding environmental contamination with hormonal EDCs. Fig. 3 demonstrates the total annual number of published articles from the database that include information on each CEC of interest. It is clear that a large increase in research on the contamination of aquatic matrices with DCL, E2 and EE2 has occurred since the early 2000s. The annual counts of articles increased from 0 for all three CECs in 1995 to 76, 50 and 51 for DCL, E2 and EE2 respectively in 2014. The maximum number of annual publications on DCL sources, receptors or control measures occurred in 2012 (83), while E2 and EE2 reached a maximum in 2013 (68 and 67 respectively). This figure also demonstrates that most years, slightly more articles are published on E2 and EE2 when compared with DCL, although this trend reversed itself from 2011 onwards. Finally, the majority of publications (>84%) on these three CECs have occurred from 2005 onward. This trend likely relates to the recent increased concern regarding DCL, E2 and EE2 in regards to EU legislation (specifically via the WFD). The apparent sharp decrease in publications from 2014 to 2015 is an artefact as the search was conducted in May of 2015, thus presumably many more articles were published on these CECs in the second half of the year.



Data extracted from each included article

Fig. 1. Multi-step abstract review and data extraction approach used in the abstract filter step of the systematic literature review. Extracted data was used to carry out the bibliographic analysis.

#### 3.2.2. Research theme studied

This systematic review investigated three general themes regarding research on DCL, E2 and EE2 and found 595 studies for sources of contamination (Section 3.3); 775 studies for receptors or monitoring methods used to measure the levels of these compounds in the aquatic environment (Sections 3.4–3.5); and 651 for control measures for reducing contamination (Section 3.6). Studies often focus on more than one of these themes, for example, some monitoring studies also discuss removal of CECs via wastewater treatment. Furthermore, studies focused on receptors or monitoring methods outnumber source studies by nearly 200 articles and control measure studies by over a hundred articles. Many of these monitoring articles describe analytical methods and conditions used to detect low levels of the CECs of interest, but they often also report detected concentrations in wastewater influent and effluent; surface, ground and drinking water or other environmental matrices for validation of the developed analytical protocols (e.g. Ben



Fig. 2. Publications (articles) returned from the systematic review searches; the figure demonstrates the number of publications excluded plus reasons for exclusion during the title and abstract filter, as well as the total number of publications included in the final database.


Fig. 3. Total combined number of EU studies on sources, receptors or control measures for each DCL, E2 and EE2 from 1995 to May 2015, by year.

Fredj et al., 2015; Lacey et al., 2008; Ronan and McHugh, 2013). Studies on sources are the least common of the three research themes and often focus on consumption rates, the contribution of municipal vs industry wastewater to total CECs load, or contributions via agricultural or veterinary practices (e.g. Kümmerer, 2009; Rivera-Utrilla et al., 2013; Santos et al., 2010). Finally, studies on control measures occur frequently in the database, but these publications represent studies carried out on a variety of scales, from laboratory experiments, to pilot scale studies, to whole WWTP-level studies. They also include investigations of removal via primary, secondary and tertiary treatment technologies. Fig. 4 demonstrates the total number of studies from each research theme carried out each year, from 1995 to May 2015. While publications on all three themes of research have increased dramatically during this time period, the graph demonstrates that since 2010 studies on sources of contamination have become less popular and have begun to level out. Commensurately, the number of publications on monitoring methods has been slightly lower than the number of publications on control measures in recent years (2012 to 2015). This may indicate that while monitoring methods are still being developed and measurements of these CECs in water matrices are still taking place, the research community is increasingly concerned with investigating mitigation methods for CEC contamination. Given the potential for increased regulations regarding aquatic contamination with DCL, E2 and EE2, a further increase in control measure studies is expected.

In order to understand if source, monitoring and control measure studies are conducted equally for each CEC, Fig. 5 shows the number of each type of study conducted for each compound. The difference in monitoring studies compared with source or control measure studies is accentuated for the two hormones, while DCL studies are more evenly split between the three research themes. The number of source studies is approximately equal for each of the three CECs, however control studies are conducted more frequently for DCL. The inability of conventional WWTPs processes to remove this NSAID (see Section 3.6) has likely led







Fig. 5. Total number of EU studies on each pharmaceutical of interest investigating sources, of contamination, monitoring data or techniques, or control measures, from 1995 to May 2015.

to more investigations of alternative or advanced treatments that may improve removal efficiencies.

#### 3.2.3. Hormones: chemical vs biological monitoring method

In addition to traditional chemical monitoring, a variety of in vitro, effect-based monitoring assays can be used to identify the total estrogenic activity in environmental samples (Kunz et al., 2015). Fig. 6 compares the number of E2/EE2 monitoring studies that used traditional chemical (concentration) methods vs those that used biological effects monitoring. There is also a category for integrated or combined monitoring methods. Clearly chemical methods are much more common than biological effects monitoring. This trend is apparent both during the early years of research (1999-2001) and in more recent years (2007-2015). More information on these monitoring methods are presented in sections 3.5.2 and 3.5.3. The recent spike in concentration studies is likely related to an increase in the sensitivity of recent analytical approaches for measuring estrogens. Nevertheless, detecting environmentally relevant, low concentrations of estrogens remains a challenge, thus biological effect monitoring has become more popular as the field has developed.

#### 3.2.4. Scale of the studies

Studies on DCL, E2 and EE2 can be conducted on a variety of scales. Some studies take place at the field level, measuring CEC concentrations in various aquatic matrices such as surface or ground water (e.g. Camacho-Muñoz et al., 2013; McEneff et al., 2014). Some take place on a laboratory scale, e.g. measuring the removal or effectiveness of monitoring methodologies of spiked water samples in the lab (e.g. Rizzo et al., 2015; Zhou and Jiang, 2015). Others are conducted on a full WWTP level, where influent/effluent concentrations and removal efficiencies are measured at specific WWTPs (e.g. Clara et al., 2005b; Lacey et al., 2012). The concentrations of CECs in different matrices can also be modelled (Balaam et al., 2010; Johnson et al., 2007), and



Fig. 6. EU studies investigating E2 and EE2 using concentration measurements, biological effects measurements, or an integrated approach by year, 1995–May 2015.

many studies are reviews of recent literature (see Sections 3.3–3.6). The number of each of these study types published annually from the database results is presented in Fig. 7 below. This figure demonstrates that by far, laboratory scale studies are the most common type of investigations on DCL, E2 or EE2. Laboratory studies are manageable, have controlled conditions, and can be done relatively quickly, all factors that likely contribute to the high frequency of this study type. Field studies can be more time intensive and expensive as they involve travel to a variety of locations for the collection of samples; nevertheless these types of studies have occurred with increasing frequency in the past two decades as people become more concerned with the levels of these three CECs in the aquatic environment. WWTPs scale studies have increased slowly but steadily in frequency, and now >20 tend to be published each year on just these three CECs alone. Such studies contribute to valuable meta-analyses which can provide important information regarding removal efficiencies via various wastewater treatments (Miège et al., 2008; Verlicchi et al., 2012). Modelling studies have increased recently (from 2007 onward) as more data have become available in this field, and a further increase in this study type is likely. As total number of primary publications on these CECs increases, so does the number of reviews including data on these compounds.

#### 3.2.5. Repartition of the studies by country

This bibliographic analysis identified which European countries are producing the majority of research regarding contamination of the aquatic environment with DCL, E2 and EE2 (Table 2). As stated in Section 2.2, articles with authors from multiple countries were counted as full publications for each country, rather than fractionally (Qian et al., 2015). Spain and Germany effectively contribute 528 (35.5%) of total studies where review papers evaluating such national studies have been published (González et al., 2012; Jurado et al., 2012). The top 6 EU countries including Switzerland listed in Table 2 collectively published 971 (65%) studies where metadata on these CECs informs baseline and predictive modelling such as for river basins and catchments. However, the majority of EU countries have limited studies reported and will require substantial monitoring to effectively inform decision making and policy.

#### 3.3. Sources and vectors of DCL, E2 and EE2

As the bibliographic analysis above demonstrates, contamination of the environment with CECs is a relatively recent research field with the majority of studies conducted in the past 15 years (Qian et al., 2015; Rivera-Utrilla et al., 2013; Santos et al., 2010). Now that researchers have been able to identify and quantify a large number of potentially harmful CECs in the aquatic environment (Santos et al., 2010), there is increased interest in identifying sources and vectors of these compounds. Only when the sources and pathways of CEC contamination are understood can opportunities to reduce the input of these substances into the aquatic environment be identified (Jurado et al., 2012;



**Fig. 7.** Number of studies on three CECs (DCL, E2 and/or EE2) published in the EU from 1995 to May 2015 broken down by type of study: field, laboratory scale, WWTP, modelling and review.

#### Table 2

Number of articles produced by each EU country along with Switzerland, Norway and Turkey on sources, monitoring or control measures for DCL, E2 or EE2: 1995–May 2015.

Country	Total number (%) of studies
Spain	285 (19.2)
Germany	243 (16.3)
United Kingdom	179 (12.0)
France	93 (6.3)
Switzerland	87 (5.8)
Italy	84 (5.7)
The Netherlands	57 (3.8)
Sweden	51 (3.4)
Portugal	50 (3.4)
Greece	43 (2.9)
Belgium	42 (2.8)
Denmark	37 (2.5)
Poland	37 (2.5)
Czech Republic	26 (1.7)
Austria	24 (1.6)
Finland	23 (1.5)
Norway	21 (1.4)
Slovenia	21 (1.4)
Turkey	19 (1.3)
Ireland	17 (1.2)
Cyprus	14 (0.9)
Hungary	11 (0.7)
Romania	7 (0.5)
Luxembourg	6 (0.4)
Croatia	3 (0.2)
Slovakia	3 (0.2)
Bulgaria	2 (0.1)
Estonia	2 (0.1)
Northern Ireland	2 (0.1)
Lithuania	1 (0.06)
Latvia	0 (0)
Malta	0 (0)

Kümmerer, 2010). The main sources and vectors discussed by these articles are reviewed below in a general manner, because many of them are applicable to DCL, E2 and EE2, as well as other CECs (Section 3.3.1). However, sources and vectors specific to each of the three compounds of interest are also addressed below (Section 3.3.2).

#### 3.3.1. General sources of CECs

The largest source of environmental contamination with CECs comes from human use of therapeutic drugs (Kümmerer, 2009; Rivera-Utrilla et al., 2013; Santos et al., 2010). After consumption, unaltered CECs can enter the environment via excretion in urine and faeces (Santos et al., 2010). Medicines containing the CECs of interest in this study are almost exclusively prescription medications; this allows for relatively easy measurement of drug usage or consumption (Clouzot et al., 2008; Wise et al., 2011; Zhang et al., 2008), a critical factor for predicting the ultimate levels of environmental contamination in an area. Furthermore, review studies have noted that consumption of CECs varies temporally and spatially. For example, significant differences in consumption of individual compounds can occur from one country to another, often due to cultural or economic factors (Kümmerer, 2009). In addition to excretion of unaltered CECs, parent compounds can also be converted to metabolites or conjugates through various reactions in the body. These metabolites/conjugates are then excreted and can be harmful to aquatic organisms themselves, or can be transformed or deconjugated back into the parent compound in environmental matrices (Santos et al., 2010).

CECs that are excreted by humans will ultimately end up in wastewater, and will potentially receive treatment at a municipal WWTP or via a domestic treatment system (e.g. septic tank). However, WWTPs and domestic treatment systems are generally not designed to treat CECs (e.g. Verlicchi et al., 2012) (see Section 3.6). If incomplete removal of CECs during municipal or domestic wastewater treatment occurs, the compounds will enter the aquatic environment via WWTPs effluents discharged into receiving waters.

Another potential source of environmental contamination with CECs comes from household disposal of unused or out-of-date medications (Kümmerer, 2009; Santos et al., 2010). These medications are either discarded through the sink/toilet, in which case, they go directly to WWTPs via sewage influent, or they are disposed of via household waste. If household waste containing unused drugs is landfilled, CECs can enter the landfill effluent (Kümmerer, 2009; Santos et al., 2010) and consequently the aquatic environment. In addition to household waste, sludge from WWTPs can also be brought to landfills. In this case, leaching of CECs that were removed from wastewater via sorption to sludge could further increase the CEC content of landfill effluents (Santos et al., 2010). Treated sludge (biosolids) may also be applied to soil and recent studies have documented that CECs may also reach the environment by this entry route (Verlicchi et al., 2012).

Industrial effluent can be another significant source of pharmaceutically-active chemicals (PhACs) contamination (Kümmerer, 2009; Rivera-Utrilla et al., 2013; Santos et al., 2010). The effluents of pharmaceutical production facilities in particular can contain high levels of bioactive compounds (Santos et al., 2010). However, although very limited data exist, good manufacturing practices, regulatory requirements and the high value of the active ingredients in most pharmaceuticals have often led to the assumption that such emissions are negligible in a European context (Kümmerer, 2009). Another type of wastewater that could contain high levels of CECs is hospital effluent (Kümmerer, 2009; Rivera-Utrilla et al., 2013; Santos et al., 2010). Reviews studies indicate that while CEC concentrations in hospital wastewater tend to be much higher than those in municipal sewage, the total contribution of this source to environmental contamination with CECs is low because of the relatively lower volume of hospital effluent (Kümmerer, 2009). Verlicchi and Zambello (2016) studied chemical characterisation of hospital effluents in terms of predicted and measured concentrations in Italy. It was found that uncertainties in predictions are mainly attributed to the wastewater volume and extraction factor, whereas for measured concentrations, uncertainties are mainly due to the sampling mode. In the last fifteen years, Verlicchi and Zambello (2016) remarked that investigations and studies have focused on chemical characterisation of hospital effluents in terms of detection of a selection of CECs including diclofenac (Verlicchi et al., 2012) in order to inter alia; estimate the contribution of a hospital to the influent CEC load of a municipal WWTP (Herrmann et al., 2015); analyse the most appropriate hospital effluent management (Verlicchi et al., 2015); ascertain removal efficacy of conventional and advanced treatments with regards to selected CECs (Gautam et al., 2007); determine environmental risk evaluation posed by CEC in hospital effluent (Mendoza et al., 2015); establish ecotoxicology (Perrodin et al., 2015), define framework for proposing proper management and treatment (Al Aukidy et al., 2014; Daouk et al., 2016); prioritise compounds to monitor (Daouk et al., 2015). Verlicchi and Zambello (2016) noted that annual consumption data and weight percentage in the investigated hospital for DCL was 1.07 kg/year and 0.63% respectively. The concentration mean value for DCL in hospital effluent was found to be 223 ng/L and 510 ng/L for summer and winter monitoring periods respectively.

The use of CECs in agriculture and aquaculture can also be sources of environmental contamination, particularly in rural environments (Boxall, 2010; Rivera-Utrilla et al., 2013; Santos et al., 2010). First, CECs given to grazing or outdoor animals are excreted directly onto the ground or into surface waters without receiving any wastewater treatment. Furthermore, disposal of farmyard manure, slurry or litter containing unmetabolized CECs via application onto agricultural land can lead to leaching of compounds into groundwater, or runoff into surface water (Rivera-Utrilla et al., 2013). Municipal sewage sludge is also often spread on agricultural land as a fertilizer, and can contain CECs that were removed from wastewater during the treatment process (Santos et al., 2010). In the case of aquaculture, CECs can be used as veterinary medicines and may be applied through many routes, including via feed, topical application or injection; all of these uses have potential to lead to contamination of surface waters (Boxall, 2010).

#### 3.3.2. Specific sources of DCL, E2 and EE2

3.3.2.1. Sources of diclofenac. Sources of DCL were specifically addressed by two review articles in the database of publications (Vieno and Sillanpää, 2014; Zhang et al., 2008). DCL is an arylacetic acid NSAID. It is prescribed as oral tablets or a topical gel, and it is sold under many commercial names including Dicloabac, Diclofenbeta, Diclomex, Voltaren, among others (Vieno and Sillanpää, 2014). Vieno and Sillanpää (2014) comprehensively reviewed the human metabolism of this PhAC. They found that studies generally report that only 6-7% of the topical gel is absorbed, while the rest is washed off the skin or attaches to clothing. This is significant in regards to environmental contamination because a large percentage of topically applied DCL will end up washed down household drains, ultimately ending up in WWTP influent. Vieno and Sillanpää (2014) also summarized the metabolism of the tablet form; the studies they reviewed found that between 65 and 75% of the orally administered dose is excreted through urine and 20-30% is excreted in faeces as the parent drug or metabolites. This review also reports that both the topical and oral forms of DCL undergo almost complete biotransformation in the body, with <1% of the orally administered dose being excreted as unmetabolized DCL. The World Health Organization defined daily dose for DCL as 100 mg, of which less than 1 mg is eliminated as DCL and 11 mg as DCL conjugates. The rest is excreted as metabolites of DCL or their conjugates (Vieno and Sillanpää, 2014). This finding demonstrates the importance of analyzing environmental matrices for metabolites and conjugates, as well as the parent drug. Diclofenac is one of the most widely used NSAID, and consumption of this compound in a variety of regions is reviewed by Zhang et al. (2008) and Ziylan and Ince (2011). They summarized the annual consumed volumes of DCL for some European countries including Austria, France, Germany, and England. Consumption in the Zhang et al. (2008) study was compared using dose per capita, or the annual consumption of the drug in an area divided by that area's population. The authors reported that Germany had the highest dose per capita (915 mg), followed by Austria (750 mg), England (531 mg) and France (271 mg). The authors also calculated a simplified estimate of annual global DCL consumption of 940 tons. Such estimations of human consumption are critical for understanding the concentrations of this CEC expected in aquatic matrices.

Treated municipal wastewater effluent is considered to be the major vector of contamination of the aquatic environment with DCL (Vieno and Sillanpää, 2014). DCL is considered as a recalcitrant compound, meaning its removal efficiency during conventional wastewater treatment is poor (Miège et al., 2008; Verlicchi et al., 2012 and see Section 3.6). Thus, concentrations of this compound in effluent are generally high (Table 6), and DCL is commonly released via this pathway into surface waters. This compound is hydrophilic, meaning it dissolves in water and does not significantly sorb onto sludge during wastewater treatment to any significant extent (Vieno and Sillanpää, 2014 and Section 3.6). It is thus unlikely that DCL contamination will result from the spreading of sewage sludge on agricultural land. DCL may be found in landfill effluent, though only via disposal of the compound through household wastes, and not from sewage sludge deposited in landfills. To our knowledge, the removal of DCL in domestic treatment systems has not been investigated yet, but this could be another potential vector of environmental contamination. Veterinary use of DCL in Europe is a potential source of contamination with this CEC. Nevertheless, the European Medicines Agency (2014) reports that DCL is authorized for veterinary use in many member states. Increased regulations and risk assessments associated with veterinary use of DCL have been suggested, and may be implemented on a European level (European Medicines Agency, 2014).

3.3.2.2. Sources of E2. E2 is one of three naturally occurring steroid estrogens produced by the human body. Females excrete on average more E2 than males (males =  $1.6 \,\mu g/day$ ), and menstruating and pregnant women excrete particularly large amounts of this natural estrogenic compound (3.5 and 259 µg/day respectively) (reviewed in Wise et al., 2011). This natural CEC can also be used in prescribed drugs, including hormone replacement therapy and to treat infertility in women or advanced prostate and breast cancer (reviewed in Kunz et al., 2015). Compared with the other natural estrogenic hormones, E2 has the highest potency and levels of aquatic contamination of this CEC are therefore of great concern (Wise et al., 2011). Given that E2 is a naturally produced compound, humans represent one of the most important sources of contamination of the environment with this CEC. Similar to DCL, effluents from WWTPs are still one of the most important vector of aquatic contamination with E2 (Burkhardt-Holm, 2010; Hecker and Hollert, 2011; Verlicchi et al., 2012; Wise et al., 2011). This compound is easily eliminated during wastewater treatment (see Section 3.6), nevertheless removal of E2 is usually incomplete (Table 3). Trace concentrations of E2 are therefore released into surface waters via WWTPs effluents. E2 is also excreted by livestock, which in general excrete the same natural hormones as humans (Burkhardt-Holm, 2010; Wise et al., 2011). Research has demonstrated that surface waters downstream of agricultural land or farms often have relatively elevated levels of estrogens, including E2 (Wise et al., 2011). Sewage sludge is not thought to be a significant source of E2 contamination, again because the compound is readily biodegradable (see Section 3.6). Domestic treatment systems and landfill effluent can contribute to environmental contamination of E2 according to a review by Burkhardt-Holm (2010). Finally, E2 has been used as a veterinary medication for livestock, although determining the contribution of natural versus pharmaceutical estrogens to total livestock excretions is difficult (Wise et al., 2011).

3.3.2.3. Sources of EE2. The structure of the synthetic estrogen EE2 is more similar to E2 than any other natural estrogen (Clouzot et al., 2008). EE2 is the main estrogenic ingredient in oral contraceptive pills taken by women of reproductive age (Clouzot et al., 2008; Wise et al., 2011). It is also found in other prescription medications including hormone replacement therapies, palliative treatments for breast and prostate cancer, and lotions used to prevent androgen-dependent hair loss in women (reviewed in Kunz et al., 2015). Estimation of consumption of this CEC can be difficult because it is usually prescribed as a combination drug (usually in combination with a progestin). Wise et al. (2011) reviewed studies on the excretion of EE2, and they report that the average daily dose of this compound is 30–35 µg of EE2 per pill, and that women on oral contraceptives fully metabolize 20-48% of this dose. The rest is excreted in either its original form or as EE2 sulfate or glucuronide conjugates, but these conjugates are mostly deconjugated back to its original form in the environment (Clouzot et al., 2008; Wise et al., 2011). As with E2, effluent from municipal WWTPs is often considered to be the most important vector of environmental EE2 contamination (Burkhardt-Holm, 2010; Hecker and Hollert, 2011; Verlicchi et al., 2012). EE2 is prone to biodegradation during wastewater treatment (see Section 3.6), but it is significantly more recalcitrant (and therefore has lower removal rates, see Table 3) than E2 (Miège et al., 2008; Verlicchi et al., 2012). Because this CEC is not completely removed by conventional wastewater treatment, it enters surface waters via WWTP effluent discharge. Unlike E2, EE2 is not produced by livestock. Sewage sludge transferred to landfills or spread on agricultural land may contain traces of EE2, but this compound is thought to biodegrade readily and thus these practices also may not represent significant sources of EE2 contamination. As for E2, domestic treatment systems and landfill leachate may present pathways to groundwater contamination with EE2, again, related back to human usage of this compound (Burkhardt-Holm, 2010).

#### 3.4. Receptors and occurrence of diclofenac, E2 and EE2 in European waters

There is now evidence of contamination of the aquatic environment with hundreds of different CECs (Kümmerer, 2010) from a variety of therapeutic classes, including antibiotics, lipid regulators, psychiatric drugs, and of course, NSAIDs (e.g. DCL) and hormones (e.g. E2 and EE2) (Verlicchi et al., 2012). Levels of CECs in the aquatic environment can vary dramatically, but are usually present in low concentrations from the nanogram to microgram per litre range depending on the location and the aquatic matrix considered (Kümmerer, 2010; Verlicchi et al., 2012). In addition to global reviews (e.g. Ratola et al., 2012; Verlicchi et al., 2012; Vieno and Sillanpää, 2014) there are now also several studies summarizing the findings of CECs occurrence in particular European countries such as Spain (González et al., 2012; Vazquez-Roig et al., 2013) and Italy (Meffe and de Bustamante, 2014). Given the importance of WWTPs as point sources of CEC contamination (as mentioned in Section 3.3), it is essential to understand the levels of compounds entering the system via influent, as well as the final concentrations in treated effluent. Many of the published review studies are devoted specifically to evaluating the typical occurrence of CECs in WWTPs influents and effluents (e.g. Miège et al., 2008; Verlicchi et al., 2012; Vieno and Sillanpää, 2014). Here only inlet and outlet WWTPs concentrations for the three CECs of interest will be discussed; some more specific information on the removal efficiencies obtained with different treatment processes and potential interpretations of the encountered removal efficiencies for DCL, E2 and EE2 will be given in the next chapter (i.e. 3.6). Other reviews focus instead on the reported concentration of CECs in surface, ground and drinking water, as concentrations in these aquatic matrices ultimately have the most relevance for animal and human health (e.g Petrie et al., 2013; Lapworth et al., 2012; Martin and Voulvoulis, 2009). Generally, many more reviews summarize surface water concentrations than ground or drinking water concentrations, due to the low number of primary studies that consider the two later matrices. Tables 3 and 4 summarize the findings from recent review papers regarding the occurrence of DCL, E2 and EE2 in these aquatic matrices in Europe and internationally. Although concentrations of DCL, E2 and EE2 can vary a great deal in each of these aquatic matrices (even when considering each compound individually), typical concentrations (including averages and ranges) in each matrix are compared and contrasted below.

#### 3.4.1. Occurrence of diclofenac

Compared with E2 and EE2, DCL tends to be present in high concentrations in WWTP influents. This finding is common for compounds in this therapeutic class; in a recent meta-analysis Miège et al. (2008) found that NSAIDs had the highest WWTPs influent concentrations when compared with other drug classes (e.g. antibiotics, beta-blockers, lipid regulators, vasodilators). In the review papers evaluated, average DCL values varied from 80 to 2100 ng/l in this aquatic matrix (Table 3). The minimum DCL influent value reported by any of the reviews was 2 ng/l (Santos et al., 2010), while the maximum was 203,000 ng/l (Ratola et al., 2012). These large variations in reported influent concentrations may be partially explained by differences in consumption of DCL between and within countries (see Section 3.3.1), and also by the differences in analytical methods employed (see Section 3.5). Such differences can make describing or predicting DCL influent concentrations difficult (Zhang et al., 2008).

Meta-analyses that evaluate multiple PhACs repeatedly found that DCL is among the most frequently detected compound in WWTPs effluents (Miège et al., 2008; Verlicchi et al., 2012). DCL is rarely completely eliminated during wastewater treatment, especially using conventional treatment processes (Table 3). As a result, this recalcitrant compound rarely falls below the LODs of a few ng/l in WWTPs effluents (Zhang et al., 2008). In the reviews evaluated, mean DCL concentrations in effluents varied widely, from <2 to 2500 ng/l. These values do tend to be slightly lower than the average influent values reported in Table 3. Nevertheless, it is clear that high nanogram to microgram per litre levels of

#### Table 3

Summary of influent and effluent concentrations and removal efficiencies following various wastewater treatments throughout Europe. All values for influent and effluent concentrations reported in ng/l. Values reported as minimum, maximum, range or mean, depending on what was reviewed by the reference. Removal efficiencies could be determined using lab, pilot or whole plant scale studies. Removal efficiencies are also given for a variety of secondary or tertiary treatments. Note: removal efficiencies represent global removal, and are not based on direct comparisons between the listed influent and effluent concentrations. These values do not represent central tendencies of removal efficiencies unless specified; furthermore they may be influenced by factors such as artefacts of the analytical (detection) methods used. Data originate from summary information provided by review studies from published database: specific references cited for each PhAC.

Drug	Influent o (ng/l)	oncentratior	1		Effluent c (ng/l)	oncentratio	n		Removal efficiency (%)		Comment	Reference
	Mean	Max	Min	Range	Mean	Max	Min	Range	Mean	Range		
Diclofenac										90–100 91–99	Ozonation Bank filtration and soil aquifer treatment	Jekel et al., 2015
	80-2300	150-7100			<2–2500	120-4700			36 36		CAS Activated sludge with BNR	Vieno and Sillanpää, 2014
	250				215				48	28-46	MBR Activated carbon [Italy, Belgium, UK, Iraland, Carmany]	Rivera-Utrilla et
									100 >80		O3 based AOPs AOPs based on UV radiation [the Netherlands]	ai., 2015
										62.9–85	Gamma radiation, various parameters [Italy]	
				nd-203,000				nd-19,200		43–77 23–76 92–99	Generally <50% DCL removed (CAS at varying SRTs) [Austria] Biofiltration processes [Spain] Ozonation [Spain, Austria]	Ratola et al., 2012 Petrie et al., 2013
				105-4110				5-5450		69–98 9–60	Sorption processes [UK] Activated sludge plants, various treatment operations	Ziylan and Ince, 2011
				2, 2000				0.2. 2400	96 29		Ozonation UV radiation	Cambra at al. 2010
				2-3600				0.3-2400	<30		Conventional wastewater treatment, 66% of reviewed studies found removal rates of <30%	Oulton et al., 2010
										6–96	Removal efficiency of CW; WWTP removal for comparison was 24%	Matamoros and Bayona, 2008
	2400							670	45		15 WWTPs designed for up to 756,000 population [Portugal]	Pereira et al., 2015
										59–75	Full scale WWTPs, treatments not specified; high removal rates due to elimination of sludge during primary treatment and/or enhanced sorption to sludge during secondary treatment upon addition of inorganic salts for P precipitation [Spain]	Suárez et al., 2008
	882	4110	105		477	1720	35	<100-1750	35	0-80	WWTPs with activated sludge processes {France} Mainly 21 to 40% (European studies, various treatment processes including Austria, Denmark, France, Greece, Italy,	Miège et al., 2008 Zhang et al., 2008
	1000	1200			800	1100			29 60		Spain, Sweden, UK) International studies, CAS International studies, MBR	Verlicchi et al., 2012
E2				2.5-125				2.7-48 0.3-30			UK	Kralchevska et al., 2013 Ratola et al
				4-30				0.1-60		39-100	Removal of various estrogens	2012 Pereira et al.,
										94–100	via various oxidative treatments Removal of various estrogens via ozonation ISpainl	2011
				0.3–102 10–31				<0.3–85 3–8			Luxembourg Italy	Santos et al., 2010
									>90		CAS, 69% of reviewed international studies found removal rates of >90%	Oulton et al., 2010

Table 3 (continued)

Drug	Influent (ng/l)	concentrati	on		Effluent (ng/l)	concentrat	ion		Removal efficiency (%)		Comment	Reference
	Mean	Max	Min	Range	Mean	Max	Min	Range	Mean	Range		
								1–10			Germany, UK	Burkhardt-Holm, 2010
									36		Removal efficiency of constructed wetlands; WWTP removal for comparison was 85–99%) [Spain]	Matamoros and Bayona, 2008
								0–50			Median = $2 \text{ ng/L}[UK]$	Martin and Voulvoulis, 2009
										30-100	Full scale WWTPs, treatments not specified [Spain]	Suárez et al., 2008
	27.4	48.4	2.5		1.8	5.2	0.3		85		WWTPs with activated sludge processes [France]	Miège et al., 2008
	250	3000			10	80			80 99		International studies, CAS International studies, MBR	Verlicchi et al., 2012
EE2								0.1-8.9			Germany	Kralchevska et al., 2013
										7–82 >50–>66 >43	Biofiltration processes [Sweden] Ozonation [Sweden] Sorption processes [UK]	Petrie et al., 2013
				1.5–17.2				0.1-3.1			berphien processes [on]	Ratola et al., 2012
				7–50				2-60		39-100	Removal of various estrogens via various oxidative treatments	Pereira et al., 2011
										94-100	Removal of various estrogens via ozonation [Spain]	
				<1.6-24				<1.1-1.7	<i>/</i> 1		Removal efficiency of constructed	Santos et al., 2010 Matamoros and
									-11		wetlands; WWTP removal for comparison was 71–78%) [Spain]	Bayona, 2008
								nd-10			Germany, UK	Burkhardt-Holm, 2010
										(-18)-98	Full scale WWTPs, treatments not specified	Suárez et al., 2008
								0-25			Median = $1 \text{ ng/L}[\text{UK}]$	Martin and Voulyoulis 2009
	1.5	2.8	0.8		0.6	1.4	0.2		95		WWTPs with activated sludge	Miège et al., 2008
		13	<0.2	1.4-6.1		42	< 0.02	<0.2-9		<80	Activated sludge processes	Clouzot et al., 2008
	20	50			3	10			78 60		International studies, CAS International studies, MBR	Verlicchi et al., 2012

CAS = conventional activated sludge; BNR = biological nutrient removal; AOP = advanced oxidation process; SRT = solids retention time; MBR = membrane bioreactor; nd = not detected.

DCL in WWTPs effluents are common throughout Europe. Occasionally individual studies found that DCL showed negative removal rates during WWTPs treatment, i.e. concentrations are actually higher in effluent than influent (e.g. Clara et al., 2005b; Lacey et al., 2012; Lacey et al., 2008). Besides the impact of analytical uncertainty, two mechanisms have been proposed to explain this phenomenon, deconjugation of glucuronidated or sulfated DCL, or desorption of this compound from particles (Verlicchi et al., 2012; Vieno and Sillanpää, 2014). It should be noted that many review papers do not include these negative removal rates when calculating average removal via WWTPs processes (see removal efficiency, Table 3). Verlicchi et al. (2012) conducted the most comprehensive, recent meta-analysis of CEC concentrations in municipal WWTPs found in our database, and DCL was one of the compounds included. CEC concentrations of raw influent at >200 municipal WWTPs (all utilizing conventional activated sludge (CAS) systems) were compared with the concentrations in secondary effluents in order to calculate global removal efficiencies. The average concentration of DCL in influent was 1.0 µg/l, but even in this one review, the minimum and maximum reported values varied over an order of magnitude. The average concentration of DCL in effluent was 0.8 µg/l, but again the values ranged greatly; in one study, DCL was found in WWTP effluent at 11 µg/l, one of the highest absolute effluent concentrations found for all 118 CECs included in the study. In another study, Loos et al. (2012) analysed effluents from 90 WWTPs across Europe for 156 polar organic chemical contaminants and showed that DCL had a frequency of detection of 89%. The maximum concentration of DCL found was 174 ng/l and the median concentration was 43 ng/l. These levels are relatively low when compared with levels found in similar studies; the most recent review of DCL found that mean concentrations in wastewater effluents were usually above 100 ng/l, however mean values as low as 2 ng/l have been found also (Vieno and Sillanpää, 2014). Loos et al. (2012) hypothesize that the low levels could have been due to problems with different analytical standards.

DCL is frequently detected in surface waters throughout Europe (Table 4). This fact is not surprising given the high levels often found in WWTPs effluents. According to the most recent review, DCL concentrations in surface waters are generally reported below 100 ng/l (Vieno and Sillanpää, 2014). Other reviews however include maximum values as high as 1030 ng/l (Ziylan and Ince, 2011) or 1200 ng/l (Rivera-Utrilla et al., 2013). Still, such high levels are the exceptions rather than the rule in regards to concentrations of DCL in surface waters. Surface waters in the UK range in DCL concentrations from <0.5 to 261 ng/l while the same author reports a range of <12 to 154 ng/l for mainland Europe (Petrie et al., 2013). Similarly, an Italian review study found a maximum

#### Table 4

Concentrations of each CEC of interest in EU waters, including surface, ground and drinking water. Data originate from summary information provided in review studies from publication database; specific references listed for each CEC. All values reported in ng/l. Values reported as minimum, maximum, range or mean, depending on what was reviewed by the reference.

Drug	Concentrations in water matrices (ng/l) <sup>a</sup>		ntrations in water Comment		
	Surface water	Ground water	Drinking water		
Diclofenac	158	nd		Max in Italian studies	Meffe and de Bustamante, 2014
	<100	<lods< td=""><td><lods< td=""><td>Surface water: generally below 100 ng/l, almost always below 500 ng/l [UK, Spain, Italy,</td><td>Vieno and Sillanpää, 2014</td></lods<></td></lods<>	<lods< td=""><td>Surface water: generally below 100 ng/l, almost always below 500 ng/l [UK, Spain, Italy,</td><td>Vieno and Sillanpää, 2014</td></lods<>	Surface water: generally below 100 ng/l, almost always below 500 ng/l [UK, Spain, Italy,	Vieno and Sillanpää, 2014
				Germany, Sweden, Finland)	
				Ground water: generally low or below detection limits, max = $380 \text{ ng/l}$ [Spain, Italy, UK)	
				Drinking water: generally low or below detection limits, range = $1-7 \text{ ng/l}$ (Italy, Spain, France)	
	1200			Max of international studies between 1999 and 2004	Rivera-Utrilla et al., 2013
	<0.5-261			Range in UK	Petrie et al., 2013
	<12-154			Range in Austria	N
	1-90	477		Range in Spanish protected areas (wetlands, estuaries, watersheds)	Vazquez-Roig et al., 2013
		4//		Max in Spanish studies	Jurado et al., 2012
	1 1020	121		Mean value, the Netherlands ( $\max = 590 \text{ lig/l}, \min = 2.5 \text{ lig/l})$	Lapworth et al., 2012
	0.2 147	10 50	<0.25.7	Range in LIV. Cormany, Slovenia	Saptos et al. 2010
	15_135	10-50	<0.23=7	Range in Cermany	Diaz-Cruz and Barceló 2008
	< 50_290			Mean value range of international studies	Zhang et al. 2008
F2	129			Max in Italian studies	Meffe and de Bustamante 2014
LZ	0.11		02-21	Furonean studies [Italy and Germany]	Kralchevska et al. 2013
	nd		0.2 2.1	Studies in Llobregat River (Snain)	González et al 2012
		nd		Max in Spanish studies	Jurado et al., 2012
		31		Mean value, the Netherlands (max = 120 ng/l, min = $0.79$ ng/l)	Lapworth et al., 2012
	0.2-50	0.08-2	nd	Range of international studies, excluding outliers	Pereira et al., 2011
	<0.2-100	0.3-1.3	nd	in Germany France	Santos et al., 2010
	nd-200	nd-45	nd-2	Range in the Netherlands, France, Germany	Martin and Voulvoulis, 2009
	0.15-17		0.2-17	Germany, UK, the Netherlands	Wise et al., 2011
EE2	2.7	nd		Max in Italian studies	Meffe and de Bustamante, 2014
	4.3			Max of international studies between 1999 and 2004	Rivera-Utrilla et al., 2013
	0.04		0.15-2.4	Italy, Germany	Kralchevska et al., 2013
	nd			Studies in Llobregat River (Spain)	González et al., 2012
		nd		Max in Spanish studies	Jurado et al., 2012
	0.5-50	0.7-5	1-3	Range of international studies, excluding outliers	Pereira et al., 2011
	<0.2-73	0.5-3	<0.1	Germany, France	Santos et al., 2010
	nd-831		nd-0.5	the Netherlands, UK, France	Martin and Voulvoulis, 2009
	<0.1-5.1		0.15-1.4	Germany	Wise et al., 2011

<sup>a</sup> Indicates value published from cited references.

concentration of 158 ng/l of DCL in surface waters (Meffe and de Bustamante, 2014). Levels in protected areas may be lower, as was demonstrated by a review of DCL levels in Spanish wetlands, estuaries and watersheds where levels ranged from 1 to 90 ng/l (Vazquez-Roig et al., 2013). Given that the predicted no effect concentration (PNEC) for DCL is reported in the literature as approximately 14  $\mu$ g/l (Santos et al., 2007), the data in Table 4 suggest that typical surface water concentrations in Europe do not usually pose a significant environmental threat. However, point sources of pollution can lead to concerning levels of DCL contamination in European surface waters.

Levels of DCL in groundwater tend to be much lower than those in surface water (Table 4). The most recent review of DCL states that levels in groundwater are typically low or below LODs (of generally a few ng/l for this type of water, See Section 3.5) (Vieno and Sillanpää, 2014). According to a recent review, no Italian study has detected DCL in groundwater to date with LODs generally in the ng/l range (Meffe and de Bustamante, 2014, see Section 3.5.1), however Spanish studies have found a maximum concentration of 477 ng/l in groundwater (Jurado et al., 2012). In a review of international studies, Lapworth et al. (2012) found a mean groundwater concentration of 121 ng/l, while Santos et al. (2010) found values ranging from <10 to 50 ng/l. Finally, concentrational studies demonstrate levels between 1 and 7 ng/l (Vieno and Sillanpää, 2014) and <0.25 to 7 ng/l (Santos et al., 2010).

#### 3.4.2. Occurrence of E2

Levels of E2 in WWTPs influents tend to be in the nanogram per litre range (Miège et al., 2008; Pereira et al., 2011; Ratola et al., 2012;

Verlicchi et al., 2012). Of the reviews evaluated, mean E2 concentrations in influents ranged from 27.4 to 250 ng/l, considerably lower than those reported for DCL (Table 3) (Miège et al., 2008; Verlicchi et al., 2012). Similar to DCL, however, the range of E2 values the reviews report for influents are high; the lowest reported influent value in any review paper was 0.3 ng/l (Santos et al., 2010) and the highest was 3000 ng/l (Verlicchi et al., 2012). In the Verlicchi et al. (2012) meta-analysis, E2 in influent presented the highest absolute concentration and the highest average observed value among any of the hormones studied. In contrast, in a meta-analysis performed by Miège et al. (2008), the mean E2 value was lower, 27.4 ng/l, and the range was much smaller (min = 2.5 to 48.4 ng/l). The Verlicchi et al. review included three studies with extremely high E2 influent concentrations (> 1000 ng/l), which drove the overall reported mean value up considerably. In general however, European influent concentrations of E2 are much < 1000 ng/l.

A greater number of reviews provide summary information on E2 concentrations in WWTP effluents than in influents (Table 3). These reviews demonstrate that levels of E2 in WWTPs effluents are also usually found in the low nanogram per litre range. Furthermore, reported E2 concentrations in effluents are generally lower than average influent concentrations. For example, the Verlicchi et al. (2012) meta-analysis reported a mean E2 concentration of 10 ng/l in effluent, 25 times less than the mean concentration in influent. Similarly the Miège et al. (2008) meta-analysis reported a decrease in E2 effluent concentrations when compared with influent concentrations (1.8 ng/l vs 27.4 ng/l respectively). This decrease in E2 concentrations in effluent is likely due to the high removal rates of E2 during many wastewater treatment processes (often >90%, see Table 3 and Section 3.6.2). In contrast, Pereira et al. state in their 2011 review paper that estrogen concentrations in

effluent wastewaters are similar to those found in influent wastewaters; however, the values they report for each matrix do indicate a slight decrease in effluent levels for E2 specifically (Table 3).

The presence of estrogenic compounds (including E2) in surface water has been widely investigated (Table 4), supposedly largely due to concerns about the endocrine disrupting effects of these compounds. We found that the majority of recent review studies report surface water E2 concentrations of <50 ng/l (Meffe and de Bustamante, 2014; Pereira et al., 2011), although in some studies the maximum values extend as high as 200 ng/l (Martin and Voulvoulis, 2009; Santos et al., 2010). E2 surface water concentrations can reach these high levels of >100 ng/l when measurements are taken directly downstream from WWTPs effluent discharge (Pereira et al., 2011). However it is also not uncommon for studies to report that E2 is below the LOD in surface waters (generally a few ng/l or below in this type of water, see Section 3.5). For example, in a review of studies conducted in the Llobregat River (Spain), González et al. (2012) find no reports of E2 exceeding LODs (generally in the ng/l range, see Section 3.5.2). Similarly Santos et al. (2010) and Martin and Voulvoulis (2009) report that some of the studies they reviewed did not detect E2 in surface waters. Nevertheless, very low concentrations (i.e. sub ng/l range) of EDCs such as E2 can have a negative impact on aquatic organisms, especially via chronic exposure; thus even though on average, surface water concentrations of E2 are lower than many other CECs, the environmental impact of this compound should not be underestimated (Burkhardt-Holm, 2010; Abargues Llamas et al., 2012). Reviews examining the occurrence of E2 in the aquatic environment often consider levels in groundwater, but less frequently discuss levels in drinking water (Table 4). Measuring the low concentrations in drinking water can present a serious analytical challenge in terms of the sensitivity of the method (see Section 3.5), thus, there are not as many primary studies that are able to investigate this aquatic matrix. Out of all of the reviews evaluated, the highest E2 concentration reported for groundwater was 120 ng/l (Lapworth et al., 2012), however, most values were much lower than this (i.e. a few nanograms per litre), especially in reviews that excluded outliers (Pereira et al., 2011; Santos et al., 2010). Several reviews reported that E2 is often present in concentrations below detection levels in groundwater (Jurado et al., 2012; Martin and Voulvoulis, 2009; Pereira et al., 2011). Concentrations of E2 in drinking water usually are reported as even lower, reaching only a few ng/l according to most reviews (Table 4).

#### 3.4.3. Occurrence of EE2

Reviews that consider occurrence of E2 in aquatic matrices often also include figures for the synthetic estrogen EE2 (Pereira et al., 2011; Ratola et al., 2012; Verlicchi et al., 2012). According to reviews included in our study, EE2 concentrations in WWTPs influents range from <0.2 to 50 ng/l, with mean values ranging from 1.5 to 20 ng/l (Table 3). As with E2, two recent meta-analyses provide the best information on likely concentrations of these compounds in European WWTPs influents and effluents (Miège et al., 2008; Verlicchi et al., 2012). Verlicchi et al. (2012) include a small number of studies in their analyses with higher EE2 influent values (>10 ng/l), whereas Miège et al. (2008) report a maximum concentration of 5.2 ng/l EE2 in WWTPs influents. Both meta-analyses report lower concentrations of EE2 in influents compared with the natural hormone E2. The low concentrations of EE2 in WWTPs influents, as well as other aquatic matrices, makes it difficult to quantify or even detect this compound using standard analytical methods; this can limit the discussion about EE2 levels and removal during wastewater treatment (Clouzot et al., 2008 and see Section 3.5). Similar to influent concentrations, effluent concentrations are usually just a few nanograms EE2 per litre (Table 3). In the review studies evaluated, mean EE2 effluent concentrations ranged from 0.6 ng/l (Miège et al., 2008) to 3 ng/l (Verlicchi et al., 2012). As with influent concentrations, these values are lower than the corresponding mean E2 effluent concentrations. The minimum reported value in effluent is <0.02 ng/l (below LOD) (Clouzot et al., 2008), while the maximum value is 60 ng/l (Pereira et al., 2011). Generally average effluent concentrations are less than influent concentrations, however EE2 is known to be slightly more recalcitrant than E2, especially in regards to conventional WWT processes (Petrie et al., 2013).

Generally reviews of EE2 indicate that surface water concentrations are very low, often below LODs (González et al., 2012; Jurado et al., 2012; Martin and Voulvoulis, 2009). According to the reviews evaluated (Table 4), surface water concentrations of EE2 range from 0.04 ng/l (Kralchevska et al., 2013) to as high as 831 ng/l (Martin and Voulvoulis, 2009). The Martin and Voulvoulis (2009) review, which reported the highest EE2 surface water concentration of all the studies evaluated, is the only review to report a maximum value above 100 ng/l. In contrast, most reviews state that EE2 concentrations in surface waters do not exceed 10 ng/l (Clouzot et al., 2008; Meffe and de Bustamante, 2014; Rivera-Utrilla et al., 2013; Wise et al., 2011). Compared with other steroid estrogens such as E2 and estrone (E1), EE2 is detected in surface waters with the lowest frequency and at the lowest concentrations (Wise et al., 2011). Nevertheless, extremely low concentrations of EE2, even levels below most LODs, are known to cause endocrine disruptions such as intersex fish or vitellogenin induction (Clouzot et al., 2008). Thus similar to E2, the environmental risk of EE2 should not be underestimated just because surface water levels are low compared with other CECs.

As of 2011, only a small number of studies had measured EE2 in drinking water (Wise et al., 2011). Wise et al. (2011) reviewed these studies and found that in the UK, the EE2 levels were usually below reported LODs. Since then, a few more studies have reviewed concentrations of EE2 in drinking water and have found similarly low levels, ranging from 0.15 to 3 ng/l (Kralchevska et al., 2013; Pereira et al., 2011). Groundwater concentrations of EE2 have been reviewed by four studies; two reviews found no studies that detected EE2 in groundwater (Jurado et al., 2012; Meffe and de Bustamante, 2014), while two found values that ranged from 0.5 to 5 ng/l (Pereira et al., 2011; Santos et al., 2010).

#### 3.5. Monitoring for DCL, E2 and EE2

The majority of the review studies evaluated by this systematic literature review were summaries of various methods for monitoring CECs in different environmental matrices. Variation in monitoring techniques can greatly influence the results of studies that report levels of CECs in the aquatic environment (Vazquez-Roig et al., 2013). This variation is certainly one component responsible for the wide range of DCL, E2 and EE2 concentrations in different matrices reported in section 3.4 above. The following section is by no means a comprehensive review of CEC monitoring and analyse techniques; instead, it specifically focuses on some of the most common methods and problems for evaluating the presence, the concentrations and effects of DCL, E2 and EE2 in the context of WFD monitoring. It also addresses some major issues and concerns related to monitoring techniques for priority substances in general.

In order to evaluate and regulate the levels of priority substances in water, the WFD has defined environmental quality criteria (Environmental Quality Standards, EQSs) (European Parliament and Council of the EU, 2008). Two forms of EQSs are used, the annual average (AA) EQS and the maximum allowable concentration (MAC) EQS (units of both are µg/l or ng/l). The arithmetic mean of the concentrations of a given priority substance recorded during all representative monitoring points in a water body for a given year must not exceed the defined AA-EQS. In contrast, the measured concentration at any monitoring point within a water body may not exceed the WFD-defined MAC-EQS. EQS values can be proposed for inland surface waters (which encompass rivers and lakes and related artificial or heavily modified water bodies) as well as "other" surface waters (European Parliament and Council of the EU, 2008). To date, AA-EQS values for both inland

and other surface waters have been proposed by the WFD for DCL, E2 and EE2 (European Commission, 2011). Compliance with EQSs is necessary to achieve a good chemical status of surface waters with regards to the chemicals on the EU list of priority substances, which could soon include DCL, E2 and EE2. The EQS values set by the WFD legislation will therefore directly impact which monitoring techniques will be acceptable for reporting purposes for a given compound, and will dictate the required level of sensitivity of those monitoring methods (Kunz et al., 2015).

#### 3.5.1. Monitoring of diclofenac

The AA-EQS values proposed by the European Commission for DCL are 100 ng/l for inland surface waters and 10 ng/l for other surface waters (European Commission, 2011). The methods for detecting NSAIDs such as DCL were recently reviewed by Olives et al. (2012). These authors report that common identification and quantification methods include gas chromatography-mass spectrometry (GC-MS) and liquid chromatography (LC) coupled with a variety of detection methods, including ultraviolet (UV) detection, diode array detection, florescence detection and tandem MS. Because DCL is a polar compound, it is more suitable for analysis by LC as opposed to GC (Vazquez-Roig et al., 2013). Furthermore, in the review studies evaluated, LC was most often coupled with MS, a highly specific technique which can detect target compounds with high accuracy (Fischer et al., 2012; Hernández et al., 2014; Vazquez-Roig et al., 2013). Another recent review from the database of publications states that there is a clear trend towards the use of LC-MS over alternative detection methods for this class of emerging contaminants (Hernández et al., 2014). LC-MS/MS (liquid chromatography with tandem MS/MS detection) is preferred over LC-MS because the former method has greater analytical sensitivity and selectivity in the analysis of drug residues in complex samples (Olives et al., 2012). Table 5 shows that recent reviews from the database of publications indicate that when using these state-of-the-art analytical methods, the LOD for DCL are typically only a few nanograms per litre (Vieno and Sillanpää, 2014). It is therefore the case that current chemical analysis techniques can usually achieve the sensitivity required to detect DCL at the concentrations required for WFD reporting.

#### 3.5.2. Monitoring of E2 and EE2

The AA-EQS values for E2 are 0.4 ng/l in inland surface waters and 0.08 ng/l in other surface waters (European Commission, 2011). For EE2, the AA-EQS values are even lower, 0.035 ng/l and 0.007 ng/l in inland and other surface waters respectively (European Commission, 2011). The WFD-proposed AA-EQS values are derived based on species sensitivity distribution studies using the most sensitive taxonomic groups, which in this case are fish and amphibians (Kunz et al., 2015). Because even very low concentrations of E2 and EE2 can have endocrine disrupting effects for some aquatic organisms (reviewed in Burkhardt-Holm, 2010), the proposed AA-EQS values for these two compounds are low in order to provide adequate protection for the aquatic environment and human health (Kunz et al., 2015). The implications of these low standards for monitoring methods and reporting, however, are significant.

In comparison to DCL, many more review studies in our database of publications focused on monitoring methods for measuring the effects and concentrations of estrogens in aquatic matrices (Briciu et al., 2009; Kozlowska-Tylingo et al., 2010; Kunz et al., 2015; Simon et al., 2015; Sosa-Ferrera et al., 2013; Streck, 2009; Tomšíková et al., 2012). Similar to DCL, techniques for the separation of steroid estrogens are usually based on LC or GC (Briciu et al., 2009; Streck, 2009; Tomšíková et al., 2012). Detection of these compounds is also carried out using various techniques, including UV detection, florescence detection, diode detection, MS detection and tandem MS (MS/MS) (Streck, 2009; Tomšíková et al., 2012). It is difficult to achieve the required sensitivity with UV, diode or florescence detection, whereas GC–MS, GC–MS/MS, LC-MS and LC-MS/MS have much lower LODs (Briciu et al., 2009;

Streck, 2009; Tomšíková et al., 2012). The specificity and sensitivity of LC-MS/MS techniques are especially required for analysis of environmental samples with steroid estrogens because of the presence of endogenous steroids in biota; that LC-MS/MS can accurately identify endogenous and exogenous estrogens is a major advantage of this technique, and has led to it being the preferred method of choice for steroidhormone analysis (Briciu et al., 2009; Sosa-Ferrera et al., 2013; Tomšíková et al., 2012). Even LC-MS analyses typically fail to provide the required level of sensitivity to detect and quantify trace concentrations of these compounds in environmental samples (Streck, 2009), thus the most sensitive methodology for the identification and quantification of steroid estrogens is widely recognized as LC-MS/MS.

Table 5 contains summary information about the LODs for E2 and EE2 from the recent review articles in the database of publications. It is clear from this summary table that oftentimes, even when using the advanced analytical detection methods described above, current monitoring techniques are not sensitive enough to detect E2 and EE2 levels in the low ng/l or pg/l range. This can result in many studies reporting no detects for these two compounds, which makes discussions of their levels and removal rates in environmental matrices difficult. What is especially problematic is that the LODs for the most advanced analytical detection methods are usually higher than the proposed WFD EQS values. This results in a serious problem regarding monitoring and reporting of E2 and EE2 concentrations in surface waters for WFD compliance. In fact a recent review study demonstrated that only 35% of published methods are able to detect E2 at the AA-EQS value of 0.4 ng/l, and only one published method exists that can detect EE2 at the AA-EQS value of 0.035 ng/l (Kunz et al., 2015; Tomšíková et al., 2012).

#### 3.5.3. Possible alternatives in the monitoring of PhACs

Unlike the situation for DCL, current analytical detection methods are often insufficiently sensitive or robust for monitoring E2 and EE2 given the proposed WFD standards. Under this directive, methods of analysis must be able to achieve limits of quantitation (LOQ) equal to or below 30% of the associated EQS. For these emerging compounds extremely low EQS values, especially for marine waters, have been set which provide a great challenge to the analyst. One potential support technique for future monitoring lies in the application of passive sampling (PS) techniques in investigative and surveillance monitoring. Passive samplers are specifically designed to be deployed over a period of days to weeks, so that time-weighted average (TWA) concentrations of compounds in aquatic environments can be obtained (Wille et al., 2012). PS as a technique is based on the free flow of analyte molecules from a medium being sampled to a receiving medium due to a difference in chemical potentials (Miège et al., 2009).

PS is proving to be a valuable tool for the monitoring of a range of priority substances in water, sediment and biota, and can generally provide more representative profile information than infrequent spot sampling on the concentrations of pollutants in water bodies, particularly where concentrations fluctuate markedly in time. PS is rapidly gaining general acceptance as being applicable to monitoring the behaviour and (eco)toxicological effects and fate of polar compounds including, DCL, E2 and EE2 in the water column and generally can often enable much greater analytical sensitivity than can be achieved by "traditional" spot-sampling, potentially improving detection capabilities by orders of magnitude. While a variety of PS devices are now commercially available, several review studies describe the use of the Polar Organic Chemical Integrative Sampler (POCIS) (Buchberger, 2011; Vermeirssen et al., 2008; Wille et al., 2012). The POCIS has a polymer component sandwiched between two thin polythersulfone membranes. CECs with particular physiochemical properties will sorb onto this polymer while the device is deployed, and can then be extracted and analysed in the laboratory using analytical techniques (Buchberger, 2011; Vermeirssen et al., 2008). In addition to providing estimates of TWA concentration of compounds, passive samplers can be a potential solution for the

#### Table 5

Proposed WFD annual average environmental quality standard (AA EQS) values for each of the three CECs of interest vs current detection limits reported in literature reviews. Data originate from review studies from publication database, specific references are listed for each CEC (LOD/MDL values in ng/l). If no method of detection is provided with a LOD value, values represent summary (range or average) LODs for a variety of different methods.

Drug	Proposed AA-EQS (ng/l) <sup>a</sup>	LOD (ng/l)	Analytical method	Comment	Country	Review reference
DCL	100	A few 6 -0.3 20 7 30 29 100 6	LC-MS Immuno-assay LC-MS LC-MS HPLC-MS HPLC-MS GC-MS GC-MS GC-MS	WWT effluent Surface water; WWT effluent STP effluent STP effluent Hospital effluent Groundwater STP effluent STP effluent	Germany Luxemburg UK Spain Spain Germany Spain Switzerland	Vieno and Sillanpää, 2014 Buchberger, 2011 Santos et al., 2010
E2	0.4	1 0.008-40 0.6-3.5 0.04-2.01 0.3-2 0.6-1.6	GC-MS In vitro bioassays LC-MS LC-MS LC-MS LC-MS	STP effluent Surface water Sewage sludge Environmental water Surface water; STP effluent STP effluent	Greece Switzerland Spain Spain Luxembourg Germany	Kunz et al., 2015 Sosa-Ferrera et al., 2013 Tomšíková et al., 2012
		0.01-2 1 1.6 1.1 0.4 1.6 0.4 0.2	LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS	Groundwater STP effluent STP influent Surface water STP influent STP effluent STP effluent Surface water	France Luxembourg Italy Italy Italy Germany Germany Germany	Santos et al., 2010
EE2	0.035	0.003-15 2.3-10.6 0.1-0.2 0.01-50 - 0.04-2.01 0.3 0.6-1.6	LC-MS LC-MS HPLC-MS In vitro bioassays LC-MS LC-MS LC-MS	STP effluent, River Waters River water WWT effluent Surface water Environmental water STP effluent STP effluent	Italy Belgium UK Switzerland Spain Luxembourg Germany	Briciu et al., 2009 Kunz et al., 2015 Tomšíková et al., 2012
		0.0-1.0 0.01-0.2 2.0- 2.0 0.4 0.2 1.6 1.1 0.4 0.2 0.6-3.5 0.02-15 2.3-10.6 0.1-0.2	Immuno-assay LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS LM-MS LC-MS LC-MS LC-MS LC-MS LC-MS LC-MS	Surface water; WWT effluent STP effluent STP influent STP influent STP influent STP influent STP effluent Tibre river water Groundwater Sewage sludge STP effluent; river water River water WWT effluent	Germany Germany Germany Germany Italy Italy Italy Italy France Spain Italy Belgium UK	Buchberger, 2011 Santos et al., 2010 Sosa-Ferrera et al., 2013 Briciu et al., 2009

LOD = limit of detection. Sensitivity of various analytical techniques deployed is influenced by sample preparation method and volume used for extraction.

<sup>a</sup> AA EQS values are annual average environmental quality standards for inland surface waters, which according to WFD legislation encompass rivers and lakes and related artificial or heavily modified water bodies. (Carvalho et al., 2015)

problem presented by the low AA-EQS values for E2 and EE2. These compounds may accumulate in passive sampling devices over time, allowing for current analytical techniques to detect and quantify E2 and EE2 levels. Several recent review studies refer to the use of passive sampling to monitor various environmental matrices for DCL, E2 and EE2, especially as a potential useful screening method in regards to WFD monitoring (Buchberger, 2011; Vermeirssen et al., 2008; Wille et al., 2012).

While passive sampling shows potential in future monitoring of concentrations and fate of emerging contaminants, application of the technique (particularly in the case of polar compounds) does face some obstacles before passive sampling is considered as a viable sampling method for the WFD or other legislation. Although the risk of toxicity for aquatic organisms is based on the bioavailable, or dissolved pollutants in a water body, the EQS set out in the WFD for the priority substances, (with the exception of trace metals), are expressed as concentrations in 'whole water'. This means that current analysis must include both the dissolved fraction and any suspended matter when used in compliance monitoring. However, for samples in which the level of suspended solids are low, it is often very difficult to reach the required limits of detection (LODs) by conventional means, and in this situation passive sampling could provide a useful alternative since they will take up the freely dissolved analytes in the water and have been shown to reach generally lower LODs than conventional grab samples. PS is also affected by environmental variables (temperature, water flow rate, salinity) and on the development of biofilms on the surface of the device which as an external factor can impede the uptake rate. Ongoing research is required to further develop the area of performance reference compounds (PRCs) to generally account for such effects however currently in the case of polar compounds the reliability of PRC information is limited and thus use of polar passive samplers is primarily restricted to use as a screening tool. As noted throughout this review, generation of accurate concentration information on levels of pharmaceuticals and NSAIDs in aquatic environments is becoming much more relevant in respect of greater legislative monitoring requirements and/or in terms of the generation of accurate data to support consumer or ecosystem risk exposure assessments. Passive sampling exhibits great potential for application in future monitoring

programs for the screening of current priority and emerging compounds in water, identification of "new" pollutants of concern, source identification and its potential role in operational, investigative and surveillance monitoring under the WFD and for other legislation source attribution and fate studies of any other potential solution to the problem presented by the low EQS.

Another potential solution to the problem presented by the low EQS values of E2 and EE2 is the use of biological effects monitoring techniques (Kunz et al., 2015; Simon et al., 2015; Streck, 2009). In the case of the estrogens in particular, a variety of in vitro assays for effect monitoring can identify the total estrogenic activity in environmental samples, which is reported as E2 equivalent (or EEQ) concentrations (Kunz et al., 2015). Streck (2009) reviewed several in vitro bioassays to measure endocrine disruption and he categorizes them into three groups: ligand-binding assays; recombinant receptor-reporter assays; and assays based on the measurement of cell proliferation induced by endocrine active compounds. Effect-based monitoring techniques are particularly useful in the context of the WFD for two reasons: (i) they could be used in future elaborations of monitoring programs to provide a link between chemical and ecological assessments of water quality, and (ii) they are an excellent method for analyzing the overall impact of mixtures of xenoestrogens present in many water bodies (Kunz et al., 2015; Streck, 2009). Furthermore, for compounds with extremely low EQSs, effect-based techniques can provide increased sensitivity, and may be used as a screening tool in monitoring programs. Integrated monitoring is currently the recommended approach according to experts in the field, and future iterations of European and national CEC monitoring programs will thus likely incorporate both chemical and biological monitoring techniques (Hecker and Hollert, 2011; Kunz et al., 2015; Simon et al., 2015). Miège et al. (2015) reviewed outcomes of discussions at a NORMAN Network-supported workshop in Lyon (France) in November 2014 that aimed at providing a common position on PS community experts regarding concrete actions required to foster use of PS techniques in support of contamination risk assessment and management and for routine monitoring of contaminants in aquatic environment. Agreed list of recommendations focused on improving the acceptance of PS by policy makers; drafting of guidelines, quality assurance and control procedures; developing demonstration projects where biomonitoring and PS were undertaken alongside each other; organising proficiency testing schemes; and inter-laboratory comparison and establishing passive sampler-based assessment criteria in relation to existing EOS. This was in agreement with international findings reported by Booij et al. (2016) who reviewed compliance monitoring requirements in the European Union, the United States, and the Oslo-Paris Convention for the protection of marine environment of the North East Atlantic, and evaluated if these are met by PS methods for non-polar compounds. International experts concluded the existence of several knowledge gaps outlined above, PS presently is best available technology for chemical monitoring of non-polar compounds.

#### 3.6. Control Measures

This section reviews how the specific physiochemical properties of DCL, E2 and EE2 impact their removal from wastewater. It also discusses the control measures found to be effective for removal of these specific CECs. This section addresses three main issues: (i) how the chemical properties of DCL, E2 and EE2 impact their removal during wastewater treatment, (ii) how these three CECs respond during conventional secondary wastewater treatment (specifically in CAS plants) where we focused on the main elimination pathways (i.e. sorption and biodegradation), and (iii) which tertiary or advanced treatments are effective against each CEC of interest. For the later, we focused on the 4 main categories of advanced treatments, oxidation technologies, membrane technologies, activated carbon (AC) technologies and constructed wetlands (CWs).

#### 3.6.1. Control measures of diclofenac

3.6.1.1. Chemical properties of diclofenac impacting removal. Diclofenac is weakly soluble in water (water solubility = 2.37 mg/L at  $25 \degree$ C, (DrugBank, 2015a)), with an octanol-water coefficient (logK<sub>ow</sub>) of 4.51 (SRC, 2013). The pK<sub>a</sub> of DCL is 4.15 (DrugBank, 2015a; SRC, 2013). DCL has a carboxylic acid portion in its molecular structure, and this region becomes negatively ionized at a neutral pH. At acidic pH, DCL becomes electronically neutral, which increases its capacity for sorption (Vieno and Sillanpää, 2014). Thus, DCL is a compound for which D<sub>ow</sub> is a better predictor of hydrophobicity. Because D<sub>ow</sub> is pH dependent, the matrix in which it is measured must be specified. The LogDow value for DCL at a pH typical of wastewater treatment (approximately 8) is 2.51 (De Ridder et al., 2011). LogK<sub>d</sub> values for DCL are reviewed in Vieno and Sillanpää (2014), and listed in Table 6 As these values are typically less than three, very little removal of DCL due to sorption is predicted by this physiochemical property (Ternes et al., 2004). According to biodegradability studies, DCL biodegradation is slow or non-existent (Joss et al., 2005; Quintana et al., 2005). Studies investigating the biodegradation constant of DCL conclude that it is almost always  $< 0.1 \text{ lg}^{-1}$  ss d<sup>-1</sup>, indicating no substantial biodegradation (Joss et al., 2005; reviewed in Vieno and Sillanpää, 2014).

3.6.1.2. Removal of diclofenac during secondary treatment. In general, the physiochemical properties of DCL (summarized above in Section 3.6.1.1) lead to low removal via sorption (Joss et al., 2005; Martín et al., 2012; Radjenović et al., 2009; Suárez et al., 2012; Ternes et al., 2004). On average, DCL's sorption to secondary sludge is <5%, while its sorption to primary sludge is in the region of 5%-15% (Ternes et al., 2004). These removal percentages are actually often lower than would be predicted based on LogDow values (De Ridder et al., 2011). Furthermore, DCL is poorly biodegradable (Joss et al., 2005; Joss et al., 2006a, 2006b; Quintana et al., 2005). As a result of its low removal via sorption and biodegradation, incomplete elimination of DCL can be expected during conventional activated sludge treatment (Table 7, Luo et al., 2014; Vieno and Sillanpää, 2014). A study by Patrolecco et al. (2015) identified DCL as one of the PhACs that exhibited the most persistence to removal at four WWTPs in Rome. Mainly primary and CAS secondary treatments were performed at the plants investigated. DCL showed high concentrations at the four treatment plants tested in both the influent and effluent samples (range = 519-2230 ng/l in influent and 321-1424 ng/l in effluent), and had the lowest removal efficiency out of all of the PhACs studied. Mean removal efficiencies for DCL were 36% removal in spring and 39% removal in winter. These values are consistent with other CAS plants according to a recent review (Vieno and Sillanpää, 2014). Similarly to the Patrolecco et al. (2015) study, Martín et al. (2012), found that DCL had the poorest removal of any of the NSAIDs studied (mean removal efficiency = 14%). The authors hypothesized that the poor removal could be due to DCL's poor degradation in wastewater. They also hypothesized that low removal efficiencies could be a consequence of the release of further DCL molecules by de-conjugation of glucuronidated or sulfated DCL and/or desorption from particles. Furthermore, in this study the PhACs that were detected in the wastewater were also detected in the sludge, indicating partial removal from wastewater through sorption; DCL, however, was only detected in wastewater confirming its low potential for sorption onto sludge. Studies have shown that elimination of DCL can be enhanced during secondary treatment by changing process configuration (reviewed in Vieno and Sillanpää, 2014). There is limited evidence that membrane bioreactors (MBRs) can increase removal efficiencies compared with CAS (Radjenović et al., 2009). This may be due to the higher biomass content and longer sludge retention time (SRT) applied in MBR. However, some studies show no increase in removal of DCL from wastewater when comparing MBR to CAS (e.g. Clara et al., 2005a). CAS with biological nutrient removal (BNR) utilizes a combination of aerobic, anaerobic and anoxic treatment units in

#### Table 6

Chemical parameters of DCL, E2 and EE2 potentially impacting removal from wastewater.

Drug	Chemical formula	CAS no	Molecular mass	Water solubility (experimental)	рК <sub>а</sub>	Log K <sub>ow</sub>	Log K <sub>d</sub>	Proposed WFD AA EQS (inland surface waters)	Summary
Diclofenac	$C_{14}H_{10}Cl_2NO_2$	15307-86-5	296.15	2.37 mg/L (at 25 °C)	4.15	4.51	logK <sub>d,primary sludge</sub> 2.7 (Ternes et al.,	100 ng/l	Fairly soluble in water, moderately low
	(DrugBank,	15307-79-6	(DrugBank,		(DrugBank,	(SRC,	2004)	(European	octanol-water
	2015a)	(disodium salt)	2015a)	(DrugBank, 2015a)	2015a; SRC, 2013)	2013)	2.3 (Radjenović et al., 2009)	Commission, 2011)	coefficient; ionization at neutral pH, becomes electronically neutral
							logK <sub>d,secondary sludge</sub> 1.2 (Ternes et al., 2004) 2.1 (Radjenović		at acidic pH (reviewed in Vieno and Sillanpää, 2014)
							et al., 2009)		
E2	C <sub>18</sub> H <sub>24</sub> O <sub>2</sub> (DrugBank, 2015b)	50-28-2	272.38 (DrugBank, 2015b)	3.6 mg/L (at 27 °C) (DrugBank, 2015b)	10.4 (Joss et al., 2006b) (DrugBank, 2015b)	4.0 (Joss et al., 2006b)	logK <sub>d, sludge</sub> 2.3–2.8 (Carballa et al., 2008)	0.4 ng/l (European Commission, 2011)	E2 is weakly soluble in water, has high pK <sub>a</sub> , fairly hydrophobic (Ben Fredj et al., 2015)
EE2	C <sub>20</sub> H <sub>24</sub> O <sub>2</sub> (DrugBank,	57-63-6	296.40 (DrugBank,	11.3 mg/L (at 27 °C)	10.4–10.7 (DrugBank,	4.2 (Joss et al.,	logK <sub>d,primary sludge</sub> 2.28–2.67	0.035 ng/l	EE2 is weakly soluble in water, has high
	2015c)		2015c)	(DrugBank, 2015c)	2015c; Joss et al., 2006b)	2006b)	(Martín et al., 2012)	(European Commission, 2011)	pK <sub>a</sub> ; high logK <sub>d</sub> indicates it tends to be retained onto
							logK <sub>d,secondary</sub> sludge 2.77–3.54 (Martín et al., 2012)		sludge, consistent with high pK <sub>a</sub> value (Martín et al., 2012)

order to remove excess nutrients from wastewater. The use of BNR processes has been shown to sometimes increase removal of DCL from wastewater. However it should be noted that in a recent review of the impact process configuration has on DCL removal, MBR, BNR and CAS average removal efficiencies were very similar (48%, 36% and 36% average removal efficiencies respectively (Vieno and Sillanpää, 2014).

Elimination of DCL during conventional secondary treatment can also be enhanced by altering process parameters such as hydraulic retention time (HRT) and SRT. Increasing HRT to >2–3 days would increase the contact time of water with the biomass, leading to higher removal efficiencies (Suárez et al., 2012). However such an alteration would is likely to be unrealistic at an operational level due to the resulting need of increasing the volumetric capacity of the WWTP and high investment and operating costs associated. Moreover, enriching the bioreactor with DCL degrading microbes may also enhance elimination. This could be achieved by applying an SRT of >150 days; however, this may also not be a realistic option at full scale WWTPs (Fernandez-Fontaina et al., 2012). Bioaugmentation, which is the addition of cultured microbes possessing the ability to degrade DCL into the biological process, could be used, but this approach requires further research (Vieno and Sillanpää, 2014).

#### Table 7

Impact of conventional activated sludge on removal of CECs of interest during wastewater treatment.

PhAC	Sorption to sludge	Degradation potential	HRT	SRT	Removal efficiency (conventional activated sludge)
Diclofenac	Low potential Sorption to sludge observed to	Low potential Poorly biodegradable (Joss et al.,	Elimination of diclofenac could be enhanced by increasing HRT to >	Enriching the bioreactor with diclofenac degrading microbes may enhance	Variable but generally poorly removed; 0–81.4% (Luo et al., 2014)
	a low degree (Martín et al., 2012; Patrolecco et al., 2015; Radjenović et al., 2009; Suárez et al., 2012; Ternes et al., 2004)	2005; Joss et al., 2006a, 2006b; Quintana et al., 2005)	2–3 days; would increase the contact time of water with the biomass (Suárez et al., 2012)	elimination; could be achieved by applying a SRTs >150 days (Fernandez-Fontaina et al., 2012)	Mean concentrations in European municipal influents between 0.11 and 2.3 µg/l (110 and 2300 ng/l), effluents between 0.002 and 2.5 µg/l (2 and 2500 ng/l) (Vieno and Sillanpää, 2014)
E2 & EE2	Moderate potential Susceptible to removal by sorption (Ben Fredj et al., 2015; Carballa et al., 2008; Martín et al., 2012; Joss et al., 2006a; Zhang and Zhou, 2008)	High potential Generally biodegraded very effectively in WWTP processes under aerobic and anaerobic conditions (Abargues Llamas et al., 2012; Alvarino et al., 2014; Petrie et al., 2014)	Biodegradation was increased when the HRT was optimised by extending it from 8 to 24 h (Petrie et al., 2014)	Maximum achievable removal when at the maximum SRT studied (27 days) (Petrie et al., 2014) Critical SRT of 10 days for removal of natural estrogens and some micropollutants suggested (Clara et al., 2005a)	Highly removed: E2: 92.6–100% (Luo et al., 2014) EE2: 43.8–100% (Luo et al., 2014) Reduced by ~85%. Final effluents normally contain nanogram per litre concentrations (Griffith et al., 2014)
					EE2 typically more recalcitrant than E2 (Petrie et al., 2014)

3.6.1.3. Removal of diclofenac during tertiary treatment. The recalcitrant nature of DCL during conventional wastewater treatment has led to a large body of research investigating further removal of this compound from treated wastewater via tertiary treatments. Much of this research has focused on oxidation technologies, which have been found effective at mineralizing many NSAIDs (Malato, 2008; Oulton et al., 2010; Suárez et al., 2008; Ziylan and Ince, 2011). Ziylan and Ince (2011) compared the relative efficiencies of some basic advanced treatment processes and found that ozonation was among the most effective in terms of achieving the complete disappearance of NSAIDs, including DCL; they report that 95–100% of residues can be destroyed using this treatment. Some oxidation technologies that have been found to effectively degrade DCL in treated wastewater are gamma ray irradiation (Liu et al., 2011), ionizing radiation (Kimura et al., 2012) and UV or UV/H<sub>2</sub>O<sub>2</sub> (Lekkerkerker-Teunissen et al., 2012), among others (reviewed in Ziylan and Ince, 2011). Operating conditions, however, can impact DCL removal efficiencies when considering oxidation technologies (Malato, 2008; Ziylan and Ince, 2011). For example, initial DCL concentration (Liu et al., 2011) operation pH (Malato, 2008), TSS loading (Oulton et al., 2010) and oxidant dose and contact time (Oulton et al., 2010; Rivera-Utrilla et al., 2013) have all been shown to impact DCL removal efficiencies. Differences in such operating conditions can explain the range of removal efficiencies reported for a vast number of oxidation technologies. Combined homogenous advanced oxidation processes (AOPs) in particular (for example,  $UV/H_2O_2$ ,  $O_3/UV$ ,  $Fe^{2+}/H_2O_2$  (Fenton) and UV/Fenton oxidation) are thought to be very promising, and have shown great efficacy for DCL removal from treated wastewater (Ribeiro et al., 2015; reviewed in Ziylan and Ince, 2011). However, the major drawback of oxidation technologies for treating PhACs remains the potential formation of toxic or persistent by-products if a compound fails to be completely mineralized (Oulton et al., 2010). DCL is one of the compounds that has specifically been shown to produce by-products after treatment, especially if the oxidant dose or contact time are not adequate (Sein et al., 2008). The toxicity of these compounds must be evaluated in order to fully assess the potential of any oxidation technology for treating this particular CEC (Andreozzi et al., 2004).

The use of membrane filtration technologies has also been explored as a possibility for removing DCL from treated wastewater (Kimura et al., 2003; Snyder et al., 2007; reviewed in Suárez et al., 2008; Xu et al., 2005). In general, the effectiveness of membrane filtration for DCL removal greatly depends on the type of technology considered. For example, it has been shown that DCL is poorly eliminated by microfiltration or ultrafiltration membranes (Snyder et al., 2007), making these technologies poor choices for the removal of this PhAC from treated wastewater. However studies show that nanofiltration and reverse osmosis membranes can eliminate DCL very effectively (i.e. >90%) (Kimura et al., 2003; Snyder et al., 2007; Suárez et al., 2008), although lower elimination efficiencies (60%) have also been reported (Röhricht et al., 2009). Snyder et al. (2007) specify that charged compounds (including DCL), had high rejection efficiencies for the nano and reverse osmosis membranes utilized in their study due to electrostatic exclusion between the anionic compounds and the negatively charged membranes. Nevertheless, rejection efficiency via membrane filtration has been found to decrease as the concentration of DCL in treated wastewater decreases (Kimura et al., 2003). Biofouling of membranes can also impact the rejection efficiencies of some organic compounds; however, the physiochemical properties of contaminants can have an impact on their behaviour in regards to biofouling. In one study Botton et al. (2012) found that the rejection efficiencies of negatively charged compounds (including DCL) were no different when comparing virgin and biofouled nanofiltration membranes. Although the use of membrane filtration processes for removal of DCL is technically feasible and effective, some studies report that it may not be economical for wastewater treatment given high operational and investment costs (Röhricht et al., 2009; Suárez et al., 2008).

The use of both powdered AC (PAC) and granular AC (GAC) can also result in the removal of many PhACs - including DCL- from water (Delgado et al., 2012; Rivera-Utrilla et al., 2013; Snyder et al., 2007). Because removal via this technology type is based largely on sorption, the physiochemical properties of specific compounds influences their removal efficiencies (Baccar et al., 2012). For example, as sorption mechanisms are mostly hydrophobic when using AC materials (Delgado et al., 2012), logDow values can sometimes be good indicators of compound removal by AC. Although this type of tertiary treatment can partially remove DCL from water, in a recent review by Delgado et al. (2012), DCL was repeatedly cited as one of the most difficult compounds to remove using AC (e.g. below 85% at 35 mg PAC/L in Snyder et al., 2007). Removal efficiencies for DCL are also variable and can depend on factors such as contact time, pH, concentration of natural organic matter and AC dose (Baccar et al., 2012; Delgado et al., 2012; Snyder et al., 2007). Removal efficiencies of DCL can be enhanced when AC is used in combination with other technologies, such as AOPs. In this case, by-products or intermediates produced from the oxidation process can be removed via sorption onto the AC (Rivera-Utrilla et al., 2013).

The ability of CWs to remove CECs like DCL has been studied more extensively in the past decade (Hijosa-Valsero et al., 2010; Hijosa-Valsero et al., 2011; Matamoros and Bayona, 2008; Matamoros and Bayona, 2006). Although many CECs can be removed from wastewater extremely efficiently through the use of CWs, DCL is commonly cited by studies as a particularly recalcitrant compound in these systems (Hijosa-Valsero et al., 2010; Hijosa-Valsero et al., 2011; Matamoros and Bayona, 2008; Matamoros and Bayona, 2006; Oulton et al., 2010). Mean removal efficiencies for DCL in CW systems are very variable, ranging in just one study from 0 to 45%. This variability is similar to removal efficiencies for this compound in conventional wastewater treatment (Matamoros and Bayona, 2006 and see review in Section 3.6.1.2). Many factors can impact DCL removal in CWs, including process configuration (surface vs subsurface designs (Matamoros and Bayona, 2008; Oulton et al., 2010)); design parameters (water depths, presence of vegetation, plant species, etc. (Hijosa-Valsero et al., 2011; Matamoros and Bayona, 2006)); and environmental parameters (initial concentration of the compound, oxygen availability or the season (Hijosa-Valsero et al., 2011; Matamoros and Bayona, 2008)). These variables are obviously not independent as configuration and design parameters will impact many of the environmental conditions at a given treatment site. If these systems are going to be utilized with the aim of achieving significant DCL removal from wastewaters, specific parameters that have been shown in the literature to increase removal efficacy should be implemented. For example, recent research has demonstrated that high redox potential and the presence of plants appears to favour DCL removal (Hijosa-Valsero et al., 2011). It should also be kept in mind that removal efficiencies of PhACs at CWs can vary seasonally, with some evidence of lower removal in winter months due to lower bacterial activities at low temperatures (Hijosa-Valsero et al., 2011). Furthermore, the use of low-cost alternative sorbent materials (e.g. expanded clay, zeolite), as opposed to conventional inert materials such as sand and gravels or advanced materials such as AC, was shown to have a great potential for the removal of DCL in CW with removal efficiencies up to 90% (Dordio and Carvalho, 2013); however despite a great potential these studies were performed at pilot scale and the results need to be confirmed in real scale experiments.

#### 3.6.2. Control measures of E2 and EE2

3.6.2.1. Chemical properties of E2 and EE2 impacting removal. EDCs such as E2 and EE2 are mostly hydrophobic organic molecules, meaning they have a tendency to distribute in organic phases (Ben Fredj et al., 2015). E2 has a logK<sub>ow</sub> of 4.0 (Joss et al., 2006b) and EE2 has a logK<sub>ow</sub> of 4.2 (Joss et al., 2006b). The logK<sub>d</sub> values for E2 fall between 2.3 and 2.8 (Carballa et al., 2008) and as high as 3.54 for EE2 (Table 6) (Martín

et al., 2012). When logK<sub>d</sub> values are approximately 3–5, the compounds can be expected to have moderate potential for sorption to sludge (Ternes et al., 2004); even though values below three have been reported for these estrogens under specific conditions, they are close enough to this threshold that moderate sorption potential can be expected for E2 and EE2. Indeed, studies found that these two hormones tend to gather on underwater fauna, sediments or WWTP sludge when in aquatic matrices (Zhang and Zhou, 2008). Nevertheless, biodegradation is accepted as their foremost removal pathway (Petrie et al., 2014); the K<sub>biol</sub> constants for E2 and EE2 are much higher than that of DCL (300–800 and 7–9 respectively according to a review by Suárez et al. (2008)).

3.6.2.2. Removal of E2 and EE2 during secondary treatment. E2 and EE2 are both generally biodegraded very effectively in WWTPs under aerobic and anaerobic conditions (Table 7, Abargues Llamas et al., 2012). According to Alvarino et al. (2014), higher biodegradation of both compounds is achieved under aerobic conditions. Cometabolism (i.e. when an organic compound is transformed by microorganisms that cannot use the compound or its transformation products as a source energy (Grady et al., 1999)) has been shown to be the main mechanism in the removal of EE2 under nitrifying conditions, through the enzyme ammonium monooxygenase. Alvarino et al. (2014) found that in addition, other aerobic bacteria could be contributing to EE2 removal via biodegradation. As well as removal via biodegradation, the low polarity of these estrogenic compounds means sorption onto sludge may also be partially responsible for their removal from wastewater (Martín et al., 2012).

In a study by Petrie et al. (2014), the potential of CAS processes to simultaneously remove multiple micropollutants was evaluated. The study utilized a pilot-scale activated sludge plant in order to ensure process control and avoid variations in receiving sewage composition and flow; they then controlled SRT and HRT in order to evaluate the impact of these process parameters on micropollutant (including the estrogens E2 and EE2) removal. First, they evaluated whether an increase in SRT had an influence on removal at a fixed HRT of 8 h. The authors recorded maximum achievable micropollutant removal for all chemicals, including the estrogens, when at the maximum SRT studied (27 days). Furthermore, removal efficiencies were increased when the HRT was optimised by extending it from 8 h to 24 h. Most notably in the study was the improvement in the removal of the persistent EE2 (increased from 41% to  $65\% \pm 19\%$  at the 24 hour HRT). Improved removal of E2 (≥93%) was also demonstrated following this operational process change. Lengthening of the HRT saw a decrease in the food to microorganism ratio (F: M). A lower F: M ratio is indicative of a substrate limitation which in turn can lead to less-favoured carbon substrates like steroid estrogens being biodegraded as the primary food source (Cho et al., 2014). This together with an increased contact time for biodegradation might explain the improvement in the observed biodegradation at the longer HRTs (Petrie et al., 2014).

Similar to DCL, process configuration can have an impact on E2 and EE2 removal. In a 2011 study, the removal of E2 and EE2 was determined in four different WWTPs in the UK (Ifelebuegu, 2011). Removal ranges were 83–97% for E2 and 41–58% for EE2, demonstrating again that EE2 is often more persistent when compared with other estrogenic compounds. In this study, activated sludge plants that were configured for BNR showed better removal of the estrogens compared with other CAS plants. Again, both biodegradation and sorption to sludge were recognized as the primary pathways for removal (Ifelebuegu, 2011).

Jarošová et al. (2014) conducted a pan-European monitoring campaign of WWTPs effluents which included an effect-based assessment to determine estrogenicity. They found that one third of the tested municipal WWTPs effluents had EEQ values >0.5 ng/l, and that the values ranged from 0.53 to 17.9 ng/l EEQ. Overall this study shows that although removal efficiencies of E2 and EE2 (and other estrogenic compounds) are usually quite high, incomplete removal could still pose a threat to the environment; thus everything possible should be done at conventional wastewater treatment facilities to increase removal of these potent estrogenic compounds.

3.6.2.3. Removal of E2 and EE2 during tertiary treatment. Unlike DCL and as discussed above, E2 and EE2 are not particularly resistant to conventional wastewater treatment; however, tertiary treatment options for removing trace concentrations of these compounds are still being investigated because of their potential to negatively impact wildlife and humans even at the low levels found in conventional wastewater effluent (Burkhardt-Holm, 2010).

According to the literature, oxidative treatments (including ozonation and AOPs) are extremely efficient at eliminating estrogens from treated wastewater (Clouzot et al., 2008; Oulton et al., 2010; Pereira et al., 2011; Pereira et al., 2012; Ribeiro et al., 2015; Suárez et al., 2008). In a recent review, Pereira et al. (2011) found that estrogenic compound levels (including E2 and EE2) can be reduced between 94 and 99% using various AOP technologies. In fact, ozone is even more reactive with E2 and EE2 than with DCL, thus almost complete transformation of these compounds is expected following treatment (Suárez et al., 2008). However, removal of estrogens from treated wastewater through ozonation is pH dependent, and higher pH values reportedly lead to better reactivity of these compounds with ozone (Pereira et al., 2011). Furthermore the ozone reaction slows down at estrogen concentrations <100 ng/l, which is significant because such low levels are often found in water requiring treatment. In addition the presence of other compounds in treated wastewater can reduce reaction efficiencies between estrogenic compounds and oxidants (reviewed in Koh et al., 2008; Pereira et al., 2011). According to Pereira et al. (2011) some of the best oxidative technologies for the removal of estrogens are ozonation, ferrate oxidation, and disinfection with chlorine dioxide. In contrast, E2 and EE2 are poorly removed via direct phototransformation (Oulton et al., 2010). The presence of toxic by-products and intermediates produced through the treatment of estrogens with oxidation technologies is a growing concern. Although some studies suggest that the by-products produced are less estrogenic than the parent compounds (reviewed in Clouzot et al., 2008), more work is needed to identify and evaluate these by-products. If oxidation technologies are used to reduce the estrogenicity of wastewater, at the very least operating parameters (such as oxidant dose, contact time, and water pH) should be evaluated and adjusted in order to reduce by-product production (Pereira et al., 2011).

The ability of membrane filtration technologies to remove estrogenic compounds has also been investigated, and similar to DCL, removal efficiencies depend on the technology employed (reviewed by Koh et al., 2008; Oulton et al., 2010; Suárez et al., 2008). One in depth study found that microfiltration and ultrafiltration, while inefficient for the removal of most CECs, were very effective at removing steroid hormones (including E2 and EE2 (Snyder et al., 2007)). In general, however, microfiltration and ultrafiltration are not thought to perform as well as nanofiltration and reverse osmosis membranes (Koh et al., 2008), which are considered effective at removing estrogenic hormones from water (Braeken and Van der Bruggen, 2009; Dudziak and Bodzek, 2009; Koh et al., 2008; Snyder et al., 2007). In a review of the treatment and control strategies for removing estrogens from wastewater, Koh et al. (2008) state that nanofiltration and reverse osmosis can achieve up to 90% removal of estrogens. These figures however are variable, and lower estrogen removal via nanofiltration has been reported in other studies (between 60 and 85%) (Braeken and Van der Bruggen, 2009; Dudziak and Bodzek, 2009). This variability can be caused by differences in the properties of specific membranes (e.g. molecular weight cut-off, hydrophobicity, surface roughness or charge); the physiochemical properties of specific compounds (e.g. molecular size/weight, the acid dissociation constant, logKow values or polarity); or the characteristics of the wastewater (e.g. concentration of the compound, pH, presence of additional organic matter) (Dudziak and Bodzek, 2009; reviewed by Oulton et al., 2010). Finally, when considering E2 and EE2 in regards

to membrane filtration technologies, both hydrophobic adsorption and size exclusion should be considered as potential mechanisms of removal, and if ultra or microfiltration are used in MBR, biodegradation may also play an important role in the removal process (Koh et al., 2008; Oulton et al., 2010).

The removal of estrogenic compounds using AC has also been shown to be very effective (Clouzot et al., 2008; Delgado et al., 2012; Koh et al., 2008; Snyder et al., 2007; Suárez et al., 2008). In a comprehensive study by Snyder et al. (2007), both PAC and GAC were capable of removing E2 and EE2 to high levels (up to 100%). It is also specified in this study and elsewhere that the efficacy of AC for removing estrogens is reduced when natural organic matter is present, because it competes for binding sites thereby limiting removal (Koh et al., 2008; Snyder et al., 2007).

Finally, constructed wetlands (CWs) have been evaluated for their ability to remove estrogenic compounds from wastewater. A comprehensive review on this subject was carried out by Matamoros and Bayona (2008). They report that CWs can remove estrogens to similar extents as conventional wastewater treatment plants, and that certain configurations achieve >90% removal. The authors also suggest that the main mechanism of estrogen removal in CWs is sorption, and therefore subsurface flow configurations will be preferable to surface flow systems.

#### 3.6.3. Practicalities of implementing various tertiary technologies for wastewater treatment

All four of the tertiary treatment types discussed above (oxidation technologies, membrane filtration, AC and CWs) have shown some efficacy for the removal of DCL, E2 and/or EE2 (Table 8). However, whether or not tertiary treatments will ultimately be implemented at WWTPs depends on a number of factors besides their efficacy for removing micropollutants. Instillation and running costs, increases in consumer payments, overall environmental footprint, stage of development of different technologies, and general drawbacks associated with each type of treatment must be considered and weighed against potential benefits (Jones et al., 2007). Life cycle analysis (LCA) is currently a popular tool for evaluating the costs and benefits of products, services or processes in a number of sectors, and it recently has been applied to the wastewater treatment industry (reviewed in Corominas et al., 2013). LCA is unique in that it allows for a "cradle-to-grave" analysis of the technologies in question. LCA has been used to compare emerging technologies with conventional wastewater treatments (e.g. Igos et al., 2012; Kalbar et al., 2013; Machado et al., 2007). Researchers have also implemented it to evaluate the use of advanced treatment options for micropollutant (including CECs) removal (Hoibye et al., 2008; Rodríguez et al., 2012; Wenzel et al., 2008). This field of research is still relatively new (for example no LCA studies evaluating constructed wetlands in this context were found), but LCA studies for oxidation technologies (Hoibye et al., 2008; Rodríguez et al., 2012; Wenzel et al., 2008), AC (Igos et al., 2013) and membrane filtration technologies (Hoibye et al., 2008; Wenzel et al., 2008) regarding micropollutant removal do exist. A recent review of such studies states that overall, most findings indicate that there is little to no environmental benefit from the removal of CECs achieved by most advanced treatment technologies (Corominas et al., 2013). However, this is largely due to the uncertainty regarding the environmental impacts of CECs at very low concentrations; a better understanding of the implication of contamination of waters with trace concentrations of CECs is therefore necessary to improve LCA studies in this field. Additionally, economic analyses have also indicated that treating wastewater with advanced technologies for the purposes of micropollutant removal may not be feasible; Jones et al. (2007) suggest that the high costs of adopting tertiary treatments at wastewater facilities would most likely be passed onto industrial and domestic consumers. To avoid this phenomenon, they suggest that instead, parameters in conventional wastewater treatment plants should be adjusted to maximize CECs removal.

Barbosa et al. (2016) recently reviewed the occurrence and removal of organic micropollutants that comprise the complete watch list of EU Decision 2015/495. It was concluded that the efficiency of the treatment processes can decrease considerably when realistic water matices are used instead of simulated ones. As reported above, these reviewers also commented, since multiple factors can affect the efficiency of treatments, experiments should be performed as close as possible to the real conditions. Additionally, the formation of intermediates should be further investigated, considering that the produced by-products may be more toxic and/or persistent that the parent compounds. Barbosa et

Table 8	
Impact of various tertiary treatment types on remo	val of CECs of interest during wastewater treatment.

-				
Technology	Diclofenac removal	E2/EE2 removal	Costs	By-product danger
Membrane filtration technologies	Highly dependent on filtration technology; poor for micro and ultrafiltration, can be efficient for nano and reverse osmosis (Kimura et al., 2003; Snyder et al., 2007; Suárez et al., 2008)	Variable depending on technology; removal via nanofiltration ranges from > 50%–90% (Braeken and Van der Bruggen, 2009; Dudziak and Bodzek, 2009; Koh et al., 2008)	Capital costs include construction, engineering, materials costs, operational and management costs include replacing membranes and power to pump wastewater (US EPA, 1999a)	No (US EPA, 1999a)
Activated carbon	Can be efficient, depending on operational variables (Delgado et al., 2012)	Can be efficient, depending on operational parameters and wastewater characteristics (Delgado et al., 2012; Koh et al., 2008)	Dependent on different carbon contactor configurations and the cost of on-site vs off-site regeneration, as well as site and wastewater characteristics Capital costs include carbon contactors, storage tanks, regeneration systems (etc.) and operational costs include purchase of carbon, electrical power, flushing of carbon slurry piping, etc. (US EPA, 2000b)	No (Rivera-Utrilla et al., 2013)
Oxidation Technologies	Highly efficient processes for DCL removal (>90%) (Ribeiro et al., 2015; Ziylan and Ince, 2011)	Highly efficient process for estrogen removal (94–99%) (Pereira et al., 2011)	Dependent on technology type, capacity of the plant, wastewater characteristics, manufacturer and the site; e.g. ozonation costs generally high compared with other technologies, while UV can be competitive (US EPA, 1999b, 1999c)	High for DCL (Sein et al., 2008) and estrogenic compounds (Pereira et al., 2011)
Constructed wetlands	Very variable removal efficiencies, DCL considered recalcitrant compound (Matamoros and Bayona, 2008; Oulton et al., 2010)	Variable removal efficiencies but can be effective (>90%) depending on configuration and design parameters (Matamoros and Bayona, 2008)	Major capital costs include purchasing land, liner costs, engineering, etc., but both capital and operational and management costs tend to be much lower than conventional wastewater treatments (US EPA, 2000a, 2000c)	

al. (2016) also commented on the affordability of scale up of wastewater treatment options for tackling CECs, particularly, both in the implementation and management. Fate and behavioural geographical mapping and modelling of CEC hot spots can inform priority of decision making for upgrade of wastewater treatment plants and other pressure points of pollution, such as hospital effluent outputs, with advanced treatment technologies Aldekoa et al. (2013).

HRT: hydraulic residence time; SRT: solids retention time.

#### 4. Conclusions and future research needs

The overall aim of this study was to provide a baseline study for Europe exploring the implications of the addition of the three watch list compounds DCL, E2 and EE2 to the Water Framework Directive (WFD) priority substances list. This study utilized a systematic literature review to summarize the European state of knowledge in regards to the sources and prevalence of these CECs. Finally, a critical analysis of the effectiveness of potential control measures was carried out based on best-published information. Below, the main conclusions from each of these components of the study are summarized and future research needs are established.

The bibliographic analysis carried out by this study determined that the annual output of European research on DCL, E2 and EE2 has increased steadily from 1995 to 2015, with approximately 84% of all articles on aquatic contamination with these CECs published since 2005. More studies are performed on the estrogens than DCL annually, and studies focused on monitoring are more common than those on sources of contamination or control measures, though control measure studies are on the rise in recent years. Laboratory scale studies are the most common, while more realistic field and wastewater treatment plant (WWTP) level studies are rarer. This can likely be attributed to a lack of sensitive analytical techniques or accurate sensors. Spain and Germany are the European leaders in the field. The systematic literature review conducted by this study investigated the sources, receptors and monitoring methods for the three CECs of interest. Overall it was found that DCL and EE2 enter the European aquatic environment mainly following human consumption and excretion of therapeutic drugs, and incomplete removal from influent at urban WWTPs. E2 is a natural hormone excreted by humans which also experiences incomplete removal during WWTPs treatment, although livestock populations in Europe are also a significant non-point source of E2 contamination. In regards to receptors, throughout Europe DCL has on average higher concentrations (high ng/l or µg/l levels) in all aquatic matrices compared with the hormones E2 and EE2 (ng/l range); however, this does not necessarily translate to higher negative environmental impact/risk to aquatic organisms. Diclofenac concentrations found in European surface waters are generally below the annual average environmental quality standard (AA EQS) proposed by the WFD (100 ng/l), but several review studies report values exceeding this limit in the UK, Italy and other mainland European countries. E2 European surface water values are usually < 50 ng/l but nevertheless such values still greatly exceed the proposed AA EQS value (0.04 ng/l) of this bioactive compound. Similarly, EE2 is either not detected or found at low levels in European surface waters (usually below 10 ng/l), but reported values are often still higher than the proposed AA EQS value (0.035 ng/l). Finally, current standard, laboratory-based analytical chemistry methods are sufficiently sensitive for the detection and quantification of DCL, but limits of detection for E2 and EE2 are often higher than proposed AA EQS values, presenting serious analytical challenges in regards to chemical monitoring methods and reporting for these two CECs.

The systematic literature review results were expanded to analyse potential control measures that may be implemented at WWTPs to decrease levels of DCL, E2 and EE2 in final effluents. The review revealed that physiochemical properties or experimentally determined constants that can be used to predict the removal of CECs during wastewater treatment include the octanol–water partition coefficient ( $K_{ow}$ ),

n-octanol–water partition coefficient ( $D_{ow}$ ), solid-water distribution coefficient ( $K_d$ ), half-life and the biodegradation constant, ( $K_{bio}$ ). CECs with high water solubility and low biodegradability are the most recalcitrant during wastewater treatment. Studies showed that DCL is poorly removed during conventional wastewater treatment; removal percentages are variable but generally fall between 21 and 40%. Mean concentrations in European municipal influents are between 0.11 and 2.3 µg/l (110 and 2300 ng/l) and effluents between 0.002 and 2.5 µg/l (2 and 2500 ng/l). In contrast, E2 and EE2 are generally highly removed during conventional wastewater treatment: removal percentages generally are 85% or greater. Final European effluents normally contain nanogram per litre concentrations of these compounds, but EE2 is consistently more recalcitrant than E2.

Where secondary treatments are deemed insufficient, tertiary treatments may be used to further improve the quality of the final effluent. In particular, recent studies have mostly investigated four types of tertiary treatment technologies for removal of CECs from treated wastewater including oxidation technologies, membrane filtration, the use of activated carbon (AC) and constructed wetlands (CW). Oxidation technologies are considered highly efficient at DCL removal. Membrane filtration can also be efficient at removing DCL from treated wastewater, but it depends on the technology used; micro and ultra-filtration are typically ineffective while nano and reverse osmosis filtration are very efficient for this particular CEC. The application of AC can effectively reduce DCL concentrations in treated wastewater, but removal efficiencies largely depend on operational variables. Finally, CWs demonstrate variable removal of DCL, but this compound is generally considered recalcitrant in these systems. For E2 and EE2 removal, oxidation technologies are considered very effective treatments. Membrane filtration technologies, while exhibiting more variation than oxidation technologies, can also be extremely effective. The use of AC is also appropriate for estrogen removal, but similar to DCL removal efficiencies depend on operational variables and wastewater characteristics. Finally, CWs can perform estrogen removal, but the effectiveness depends upon the configuration and design of the system. Although more information is needed to accurately model the benefits of using tertiary treatments to reduce CEC concentrations in treated wastewaters, in general the literature suggests that the environmental benefits may not outweigh the costs. Some sources suggest that it may currently be more economically to adapt conventional wastewater treatment operational variables to decrease CEC emissions, rather than incur the costs/complications of adding tertiary treatments.

This study has highlighted areas for future research attention to include (a) development of more sensitive and validated analytical methods for different environmental samples (especially for the steroid estrogens) in order to be able to comply with WFD reporting requirements. Given the extremely low AA EQSs proposed for these compounds, it is also necessary to continue to investigate alternatives to chemical analyses, such as passive sampling, use of appropriate biological surrogates or effect-based monitoring techniques; (b) intensification of technology-focused studies for effective and efficient control measures for CECs removal particularly at areas showing disproportionally high levels of these CECs in terms of load; (c) evaluation of validated, bolt-on or mobile technologies effective for the removal of CECs in wastewater, (d) identify and quantify population level effects of wild biota from endocrine disrupting chemical exposure; and (e) investigate seasonal variations in CEC loading and removal efficiencies in future studies with particularly addressing influence of climate change.

#### Acknowledgements

The authors express gratitude to Ireland's Environmental Protection Agency STRIVE Programme for supporting this study (Project 2014-W-DS-18).

#### References

- Abargues Llamas, M.R., Robles Martínez, Á., Bouzas Blanco, A., Seco Torrecillas, A., 2012. Micropollutants removal in an anaerobic membrane bioreactor and in an aerobic conventional treatment plant. Water Sci. Technol. 65 (12), 2250–2422.
- Al Aukidy, M., Verlicchi, P., Voulvoulis, N., 2014. A framework for the assessment of the environmental risk posed by pharmaceuticals originating from hospital effluents. Sci. Total Environ. 493, 54–64.
- Aldekoa, J., Medici, C., Osorio, V., Pérez, S., Marcé, R., Barceló, D., Francés, F., 2013. Modelling the emerging pollutant diclofenac with the GREAT-ER model: application to the Llobregat River Basin. J. Hazard. Mater. 263 (Part 1), 207–213.
- Alvarino, T., Suarez, S., Lema, J.M., Omil, F., 2014. Understanding the removal mechanisms of PPCPs and the influence of main technological parameters in anaerobic UASB and aerobic CAS reactors. J. Hazard. Mater. 278, 506–513.
- Andreozzi, R., Campanella, L., Fraysse, B., Garric, J., Gonnella, A., Lo Giudice, R., Marotta, R., Pinto, G., Pollio, A., 2004. Effects of advanced oxidation processes (AOPs) on the toxicity of a mixture of pharmaceuticals. Water Sci. Technol. 50 (5), 23–28.
- Baccar, R., Sarrà, M., Bouzid, J., Feki, M., Blánquez, P., 2012. Removal of pharmaceutical compounds by activated carbon prepared from agricultural by-product. Chem. Eng. J. 211–212, 310–317.
- Balaam, J.L., Grover, D., Johnson, A.C., Jürgens, M., Readman, J., Smith, A.J., White, S., Williams, R., Zhou, J.L., 2010. The use of modelling to predict levels of estrogens in a river catchment: how does modelled data compare with chemical analysis and in vitro yeast assay results? Sci. Total Environ. 408 (20), 4826–4832.
- Barbosa, M.O., Moreira, N.F., Riberio, A.R., Pereira, M.F., Silva, A.M., 2016. Occurrence and removal of organic micropollutants: an overview of the watch list of EU Decision 2015/495. Water Res. 94, 257–279.
- Barreiros, L., Queiroz, J.F., Magalhaes, I.M., Silva, A.M.T., Segundo, M.A., 2016. Analysis of 17-β-estradiol nd 17-α-ethinylestradiol in biological and environmental matrices a review. Microchem. J. 126, 243–262.
- Belter, C.W., Seidel, D.J., 2013. A bibliometric analysis of climate engineering research. Wiley Interdisciplinary Reviews Climate Change 4 (5), 417–427.
- Ben Fredj, S., Nobbs, J., Tizaoui, C., Monser, L., 2015. Removal of estrone (E1), 17β-estradiol (E2), and 17α-ethinylestradiol (EE2) from wastewater by liquid–liquid extraction. Chem. Eng. J. 262, 417–426.
- Booij, K., Robinson, C.D., Burgess, R.M., Mayer, P., Roberts, C.A., Ahrens, L., Allan, I.J., Brant, J., Jones, L., Kraus, U.R., Larsen, M.M., Pepom, Petersen, J., Pröfrock, D., Roose, P., Schäfer, S., Smedes, F., Tixier, C., Vorhamp, K., Whitehouse, P., 2016. Passive sampling in regulatory chemical monitoring of nonpolar organic compounds in the aquatic environment. Environ. Sci. Technol. 50, 3–17.
- Botton, S., Verliefde, A.R.D., Quach, N.T., Cornelissen, E.R., 2012. Influence of biofouling on pharmaceuticals rejection in NF membrane filtration. Water Res. 46 (18), 5848–5860.
- Boxall, A.A., 2010. Veterinary medicines and the environment. In: Cunningham, F., Elliott, J., Lees, P. (Eds.), Comparative and Veterinary Pharmacology. Handbook of Experimental Pharmacology. Springer Berlin Heidelberg, pp. 291–314.
- Braeken, L., Van der Bruggen, B., 2009. Feasibility of nanofiltration for the removal of endocrine disrupting compounds. Desalination 240 (1–3), 127–131.
- Briciu, R.D., Kot-Wasik, A., Namiesnik, J., 2009. Analytical challenges and recent advances in the determination of estrogens in water environments. J. Chromatogr. Sci. 47 (2), 127–139.
- Buchberger, W.W., 2011. Current approaches to trace analysis of pharmaceuticals and personal care products in the environment. J. Chromatogr. A 1218 (4), 603–618.
- Burkhardt-Holm, P., 2010. Endocrine disruptors and water quality: a state-of-the-art review. International Journal of Water Quality 16 (3), 477–493.
- Camacho-Muñoz, D., Martín, J., Santos, J.L., Aparicio, I., Alonso, E., 2012. Effectiveness of conventional and low-cost wastewater treatments in the removal of pharmaceutically active compounds. Water Air Soil Pollut. 223 (5), 2611–2621.
- Camacho-Muñoz, D., Martín, J., Santos, J.L., Aparicio, I., Alonso, E., 2013. Distribution and risk assessment of pharmaceutical compounds in river sediments from Doñana Park (Spain). Water Air Soil Pollut. 224 (10), 1–15.
- Cho, S.H., Park, H.J., Lee, J.H., Kim, H.J., Cho, S., Yoon, C.Y., Kim, W.S., 2014. Monitoring of 35 illegally added steroid compounds in foods and dietary supplements. Food Addit Contam Part A Chem Anal Chem Control Risk Assess. 31 (9), 1470–1475.
- Carballa, M., Fink, G., Omil, F., Lema, J.M., Ternes, T., 2008. Determination of the solidwater distribution coefficient (K<sub>d</sub>) for pharmaceuticals, estrogens and musk fragrances in digested sludge. Water Res. 42 (1–2), 287–295.
- Carvalho, R.N., Ceriani, L., Ippolito, A., Lettieri, T., 2015. Development of the first Watch List under the Environmental Quality Standards Directive - Directive 2008/105/EC, as ammended by Directive 2013/39/EU in the field of water policy. Joint Technical Report EUR 27142 EN.
- Clara, M., Kreuzinger, N., Strenn, B., Gans, O., Kroiss, H., 2005a. The solids retention time—a suitable design parameter to evaluate the capacity of wastewater treatment plants to remove micropollutants. Water Res. 39 (1), 97–106.
- Clara, M., Strenn, B., Gans, O., Martinez, E., Kreuzinger, N., Kroiss, H., 2005b. Removal of selected pharmaceuticals, fragrances and endocrine disrupting compounds in a membrane bioreactor and conventional wastewater treatment plants. Water Res. 39 (19), 4797–4807.
- Clouzot, L., Marrot, B., Doumenq, P., Roche, N., 2008. 17α-Ethinylestradiol: an endocrine disrupter of great concern. Analytical methods and removal processes applied to water purification. A review. Environ. Prog. 27 (3), 383–396.
- Corominas, L., Foley, J., Guest, J.S., Hospido, A., Larsen, H.F., Morera, S., Shaw, A., 2013. Life cycle assessment applied to wastewater treatment: state of the art. Water Res. 47 (15), 5480–5492.
- Daouk, S., Chèvre, N., Vernaz, N., Bonnabry, P., Dayer, P., Daali, Y., Fleury-Souverain, S., 2015. Prioritization methodology for the monitoring of active pharmaceutical ingredients in hospital effluents. J. Environ. Manag. 160, 324–332.

- Daouk, S., Chèvre, N., Vernaz, N., Widmer, C., Daali, Y., Fleury-Souverain, S., 2016. Dynamics of active pharmaceutical ingredients loads in a Swiss university hospital wastewaters and prediction of the related environmental risk for the aquatic ecosystems. Sci. Total Environ. 547, 244–253.
- De Ridder, D.J., Verliefde, A.R.D., Heijman, S.G.J., Verberk, J.Q.J.C., Rietveld, L.C., Van Der Aa, L.T.J., Amy, G.L., Van Dijk, J.C., 2011. Influence of natural organic matter on equilibrium adsorption of neutral and charged pharmaceuticals onto activated carbon. Water Sci. Technol. 63 (3), 416–423.
- Delgado, L.F., Charles, P., Glucina, K., Morlay, C., 2012. The removal of endocrine disrupting compounds, pharmaceutically activated compounds and cyanobacterial toxins during drinking water preparation using activated carbon—a review. Sci. Total Environ. 435–436, 509–525.
- Díaz-Cruz, M.S., Barceló, D., 2008. Input of pharmaceuticals, pesticides and industrial chemicals as a consequence of using conventional and non-conventional sources of water for artificial groundwater recharge. In: Barceló, D., Petrovic, M. (Eds.), Emerging Contaminants from Industrial and Municipal Waste. The Handbook of Environmental Chemistry. Springer Berlin Heidelberg, pp. 219–238.
- Dordio, A., Carvalho, A., 2013. Constructed wetlands with light expanded clay aggregates for agricultural wastewater treatment. Sci Total Environ 463-464, 454–461.
- DrugBank (2015a) Diclofenac. http://www.drugbank.ca/drugs/DB00586.
- DrugBank (2015b) Estradiol. http://www.drugbank.ca/drugs/DB00783.
- DrugBank (2015c) Ethinyl Estradiol. http://www.drugbank.ca//drugs/DB00977.
- Dudziak, M., Bodzek, M., 2009. Selected factors affecting the elimination of hormones from water using nanofiltration. Desalination 240 (1–3), 236–243.
- European Commission, 2011. COM(2011) 876 Final, Proposal for a Directive of the European Parliament and of the Council Amending Directives 2000/60/EC and 2008/105/ EC as Regards Priority Substances in the Field of Water Policy. Brussels.
- European Medicines Agency, 2014. Opinion of the Committee for Medicinal Products for Veterinary Use Pursuant to Article 30(3) of Regulation (EC) No 726/2004.
- European Parliament and Council of the EU, 2008. Directive 2008/105/EC of the European Parliament and of the council of 16 December 2008. Off, J. Eur, Union.
- Fatta-Kassinos, D., Meric, S., Nikolaou, A., 2011b. Pharmaceutical residues in environmental waters and wastewater: current state of knowledge and future research. Anal. Bioanal. Chem. 299, 251–275.
- Fent, K., Weston, A.A., Caminada, D., 2006. Ecotoxicology of human pharmaceuticals. Aquat. Toxicol. 76 (2), 122–159.
- Fernandez-Fontaina, E., Omil, F., Lema, J.M., Carballa, M., 2012. Influence of nitrifying conditions on the biodegradation and sorption of emerging micropollutants. Water Res. 46 (16), 5434–5444.
- Ferrari, B., Mons, R., Vollat, B., Fraysse, B., Paxēaus, N., Lo Giudice, R., Rollio, A., Garric, J., 2004. Environmental risk assessment of six human pharmaceuticals: are the current risk assessment procedures sufficient for the protection of the aquatic environment. Environ. Toxicol. Chem. 23, 1344–1345.
- Fischer, K., Fries, E., Körner, W., Schmalz, C., Zwiener, C., 2012. New developments in the trace analysis of organic water pollutants. Appl. Microbiol. Biotechnol. 94 (1), 11–28.
- Futran Fuhrman, V., Tal, A., Arnon, S., 2015. Why endocrine disrupting chemicals (EDCs) challenge traditional risk assessment and how to respond. J. Hazard. Mater. 286, 589–611.
- Gautam, A.K., Kumar, S., Sabumon, P.C., 2007. Preliminary study of physcio-chemical treatment options for hospital wastewater. J. Environ. Manag. 83, 296–306.
- González, S., López-Roldán, R., Cortina, J.-L., 2012. Presence and biological effects of emerging contaminants in Llobregat River basin: a review. Environ. Pollut. 161, 83–92.
- Grady, L, Daigger, G., Lim, H., 1999. Biological Wastewater Treatment. 2nd edition. Marcel Dekker Inc., New York, New York.
- Griffith, D.R., Kido Soule, M.C., Matsufuji, H., Eglinton, T.I., Kujawinski, E.B., Gschwend, P.M., 2014. Measuring free, conjugated, and halogenated estrogens in secondary treated wastewater effluent. Environ. Sci. Technol. 48 (5), 2569–2578.
- Heberer, T., 2002. Tracking persistent pharmaceutical residues from municipal sewage to drinking water. J. Hydrol. 266 (3–4), 175–189.
- Hecker, M., Hollert, H., 2011. Endocrine disruptor screening: regulatory perspectives and needs. Environ. Sci. Eur. 23 (15), 1–14.
- Hernández, F., Ibáñez, M., Bade, R., Bijlsma, L., Sancho, J.V., 2014. Investigation of pharmaceuticals and illicit drugs in waters by liquid chromatography-high-resolution mass spectrometry. TrAC Trends Anal. Chem. 63, 140–157.
- Herrmann, M., Olsson, O., Fiehn, R., Herrel, M., Kümmerer, K., 2015. The significance of different health institutions and their respective contributions of active pharmaceutical ingredients in wastewater. Environ. Int. 85, 61–76.
- Hijosa-Valsero, M., Matamoros, V., Sidrach-Cardona, R., Martín-Villacorta, J., Bécares, E., Bayona, J.M., 2010. Comprehensive assessment of the design configuration of constructed wetlands for the removal of pharmaceuticals and personal care products from urban wastewaters. Water Res. 44 (12), 3669–3678.
- Hijosa-Valsero, M., Matamoros, V., Sidrach-Cardona, R., Pedescoll, A., Martín-Villacorta, J., García, J., Bayona, J.M., Bécares, E., 2011. Influence of design, physico-chemical and environmental parameters on pharmaceuticals and fragrances removal by constructed wetlands. Water Sci. Technol. 63 (11), 2527–2534.
- Hoibye, L., Clauson-Kaas, J., Wenzel, H., Larsen, H.F., Jacobsen, B.N., Dalgaard, O., 2008. Sustainability assessment of advanced wastewater treatment technologies. Water Sci. Technol. 58 (5), 963–968.
- Hunt, J.R., Knowles, T.G., Lascelles, B.D.X., Murrell, J.C., 2015. Prescription of perioperative analgesics by UK small animal veterinary surgeons in 2013. Vet. Rec. 176 (19), 493.
- Ifelebuegu, A.O., 2011. The fate and behavior of selected endocrine disrupting chemicals in full scale wastewater and sludge treatment unit processes. Int. J. Environ. Sci. Technol. 8 (2), 245–254.

- Igos, E., Benetto, E., Venditti, S., Kohler, C., Cornelissen, A., Moeller, R., Biwer, A., 2012. Is it better to remove pharmaceuticals in decentralized or conventional wastewater treatment plants? A life cycle assessment comparison. Sci. Total Environ. 438, 533–540.
- Igos, E., Benetto, E., Venditti, S., Köhler, C., Cornelissen, A., 2013. Comparative and integrative environmental assessment of advanced wastewater treatment processes based on an average removal of pharmaceuticals. Water Sci. Technol. 67 (2), 387–394.
- Jarošová, B., Erseková, A., Hilscherová, K., Loos, R., Gawlik, B., Giesy, J., Bláha, L., 2014. Europe-wide survey of estrogenicity in wastewater treatment plant effluents: the need for the effect-based monitoring. Environ. Sci. Pollut. Res. 21 (18), 10970–10982.
- Jekel, M., Dott, W., Bergmann, A., Dünnbier, U., Gnirß, R., Haist-Gulde, B., Hamscher, G., Letzel, M., Licha, T., Lyko, S., Miehe, U., Sacher, F., Scheurer, M., Schmidt, C.K., Reemtsma, T., Ruhl, A.S., 2015. Selection of organic process and source indicator substances for the anthropogenically influenced water cycle. Chemosphere 125, 155–167.
- Johnson, A.C., Keller, V., Williams, R.J., Young, A., 2007. A practical demonstration in modelling diclofenac and propranolol river water concentrations using a GIS hydrology model in a rural UK catchment. Environ. Pollut. 146 (1), 155–165.
- Johnson, A.C., Dumont, E., Williams, R.J., Oldenkamp, R., Cisowska, I., Sumpter, J.P., 2013. Do concentrations of ethinylestradiol, estradiol, and diclofenac in European rivers exceed proposed EU environmental quality standards? Environ. Sci. Technol. 47 (21), 12297–12304.
- Jones, O.A.H., Green, P.G., Voulvoulis, N., Lester, J.N., 2007. Questioning the excessive use of advanced treatment to remove organic micropollutants from wastewater. Environ. Sci. Technol. 41 (14), 5085–5089.
- Joss, A., Keller, E., Alder, A.C., Göbel, A., McArdell, C.S., Ternes, T., Siegrist, H., 2005. Removal of pharmaceuticals and fragrances in biological wastewater treatment. Water Res. 39 (14), 3139–3152.
- Joss, A., Carballa, M., Kreuzinger, N., Siegrist, H., Zabczynski, S., 2006a. Wastewater treatment. In: Termes, T.A.J., Joss, A. (Eds.), Human Pharmaceuticals, Hormones and Fragrances: The Challenge of Micropollutants in Urban Water Management. IWA Publishing, London, pp. 243–282.
- Joss, A., Zabczynski, S., Göbel, A., Hoffmann, B., Löffler, D., McArdell, C.S., Ternes, T.A., Thomsen, A., Siegrist, H., 2006b. Biological degradation of pharmaceuticals in municipal wastewater treatment: proposing a classification scheme. Water Res. 40 (8), 1686–1696.
- Jurado, A., Vàzquez-Suñé, E., Carrera, J., López de Alda, M., Pujades, E., Barceló, D., 2012. Emerging organic contaminants in groundwater in Spain: a review of sources, recent occurrence and fate in a European context. Sci. Total Environ. 440, 82–94.
- Kalbar, P.P., Karmakar, S., Asolekar, S.R., 2013. Assessment of wastewater treatment technologies: life cycle approach. Water Environ. J. 27 (2), 261–268.
- Kimura, K., Amy, G., Drewes, J.E., Heberer, T., Kim, T.-U., Watanabe, Y., 2003. Rejection of organic micropollutants (disinfection by-products, endocrine disrupting compounds, and pharmaceutically active compounds) by NF/RO membranes. J. Membr. Sci. 227 (1–2), 113–121.
- Kimura, A., Osawa, M., Taguchi, M., 2012. Decomposition of persistent pharmaceuticals in wastewater by ionizing radiation. Radiat. Phys. Chem. 81 (9), 1508–1512.
- Koh, Y.K.K., Chiu, T.Y., Boobis, A., Cartmell, E., Scrimshaw, M.D., Lester, J.N., 2008. Treatment and removal strategies for estrogens from wastewater. Environ. Technol. 29 (3), 245–267.
- Kozlowska-Tylingo, K., Namieśnik, J., Górecki, T., 2010. Determination of estrogenic endocrine disruptors in environmental samples—a review of chromatographic methods. Crit. Rev. Anal. Chem. 40 (3), 194–201.
- Kralchevska, R.P., Milanova, M.M., Hristova, I.L., Todorovsky, D.S., 2013. Some endocrine disrupting compounds in the environment and possibilities for their removal/degradation. Bulg. Chem. Commun. 45 (2), 131–143.
- Kümmerer, K., 2009. The presence of pharmaceuticals in the environment due to human use – present knowledge and future challenges. J. Environ. Manag. 90 (8), 2354–2366.
- Kümmerer, K., 2010. Pharmaceuticals in the environment. Annu. Rev. Environ. Resour. 35, 57–75.
- Kunz, P.Y., Kienle, C., Carere, M., Homazava, N., Kase, R., 2015. In vitro bioassays to screen for endocrine active pharmaceuticals in surface and waste waters. J. Pharm. Biomed. Anal. 106. 107–115.
- Lacey, C., McMahon, G., Bones, J., Barron, L., Morrissey, A., Tobin, J.M., 2008. An LC–MS method for the determination of pharmaceutical compounds in wastewater treatment plant influent and effluent samples. Talanta 75 (4), 1089–1097.
- Lacey, C., Basha, S., Morrissey, A., Tobin, J., 2012. Occurrence of pharmaceutical compounds in wastewater process streams in Dublin, Ireland. Environ. Monit. Assess. 184 (2), 1049–1062.
- Lapworth, D.J., Baran, N., Stuart, M.E., Ward, R.S., 2012. Emerging organic contaminants in groundwater: a review of sources, fate and occurrence. Environ. Pollut. 163, 287–303.
- Lekkerkerker-Teunissen, K., Benotti, M.J., Snyder, S.A., van Dijk, H.C., 2012. Transformation of atrazine, carbamazepine, diclofenac and sulfamethoxazole by low and medium pressure UV and UV/H2O2 treatment. Sep. Purif. Technol. 96, 33–43.
- Liu, Q., Luo, X., Zheng, Z., Zheng, B., Zhang, J., Zhao, Y., Yang, X., Wang, J., Wang, L., 2011. Factors that have an effect on degradation of diclofenac in aqueous solution by gamma ray irradiation. Environ. Sci. Pollut. Res. 18 (7), 1243–1252.
- Loos, R., Carvalho, R., Comero, S., António, D.C., Ghiani, M., Lettieri, T., Locoro, G., Paracchini, B., Tavazzi, S., Gawlik, B.M., 2012. EU Wide Monitoring Survey on Waste Water Treatment Plant Effluents. Publications Office of the European Union, Luxembourg.
- Luo, Y., Guo, W., Ngo, H.H., Nghiem, L.D., Hai, F.I., Zhang, J., Liang, S., Wang, X.C., 2014. A review on the occurrence of micropollutants in the aquatic environment and their fate and removal during wastewater treatment. Sci. Total Environ. 473–474, 619–641.
- Machado, A.P., Urbano, L, Brito, A.G., Janknecht, P., Salas, J.J., Nogueira, R., 2007. Life cycle assessment of wastewater treatment options for small and decentralized communities. Water Sci. Technol. 56 (3), 15–22.

- Malato, S., 2008. Removal of emerging contaminants in waste-water treatment: removal by photo-catalytic processes. In: Barceló, D., Petrovic, M. (Eds.), Emerging Contaminants From Industrial and Municipal Waste: Removal Technologies. Springer Berlin Heidelberg, Berlin, Heidelberg, pp. 177–197.
  Martin, O.V., Voulvoulis, N., 2009. Sustainable risk management of emerging contami-
- Martin, O.V., Voulvoulis, N., 2009. Sustainable risk management of emerging contaminants in municipal wastewaters. Philos. Trans. R. Soc. Lond. A: Math. Phys. Eng. Sci. 367 (1904), 3895–3922.
- Martín, J., Camacho-Muñoz, D., Santos, J.L., Aparicio, I., Alonso, E., 2012. Occurrence of pharmaceutical compounds in wastewater and sludge from wastewater treatment plants: removal and ecotoxicological impact of wastewater discharges and sludge disposal. J. Hazard. Mater. 239–240, 40–47.
- Matamoros, V., Bayona, J.M., 2006. Elimination of pharmaceuticals and personal care products in subsurface flow constructed wetlands. Environ. Sci. Technol. 40 (18), 5811–5816.
- Matamoros, V., Bayona, J., 2008. Behavior of emerging pollutants in constructed wetlands. In: Barceló, D., Petrovic, M. (Eds.), Emerging Contaminants from Industrial and Municipal Waste. The Handbook of Environmental Chemistry. Springer Berlin Heidelberg, pp. 199–217.
- McEneff, G., Barron, L., Kelleher, B., Paull, B., Quinn, B., 2014. A year-long study of the spatial occurrence and relative distribution of pharmaceutical residues in sewage effluent, receiving marine waters and marine bivalves. Sci. Total Environ. 476–477 (0), 317–326.
- Meffe, R., de Bustamante, I., 2014. Emerging organic contaminants in surface water and groundwater: a first overview of the situation in Italy. Sci. Total Environ. 481, 280–295.
- Mendoza, A., Aceňa, J., Pèrez, S., de Alda, M.L., Barceló, D., Gil, A., Valcárcel, Y., 2015. Pharmaceuticals and iodinated contrast media in a hospital wastewater: a case study to analyse their presence and characterise their environmental risk and hazard. Environ. Res. 140, 225–241.
- Miège, C., Choubert, J.M., Ribeiro, L., Eusebe, M., Coquery, M., 2008. Removal efficiency of pharmaceuticals and personal care products with varying wastewater treatment processes and operating conditions – conception of a database and first results. Water Sci. Technol. 57, 49–56.
- Miège, C., Choubert, J.M., Ribeiro, L., Eusèbe, M., Coquery, M., 2009. Fate of pharmaceuticals and personal care products in wastewater treatment plants – conception of a database and first results. Environ. Pollut. 157 (5), 1721–1726.
- Miège, C., Mazzella, N., Allan, I., Dulio, V., Smede, F., Tixier, C., Vermeirssen, E., Branti, J., O'Toole, S., Budzinski, H., Ghestem, J.-P., Stau, P.-F., Lardy-Fontan, S., Gonzalez, J.L., Coquery, M., Vrana, B., 2015. Position paper on passive sampling techniques for the monitoring of contaminants in the aquatic environment – achievements to date and perspectives. Trends in Environmental Analytical Chemistry 8, 20–26.
- Nikolaou, A., Meric, S., Fatta, D., 2007. Occurence patterns of pharmaceuticals in water and wastewater environments. Anal. Bioanal. Chem. 387, 1225–1234.
- Olives, A.I., Gonzalez-Ruiz, V., Martin, A., 2012. Isolation and quantitative methods for analysis of non-steroidal anti-inflammatory drugs. Anti-Inflammatory Anti-Allergy Agents Med. Chem. 11, 65–95.
- Oulton, R.L., Kohn, T., Cwiertny, D.M., 2010. Pharmaceuticals and personal care products in effluent matrices: a survey of transformation and removal during wastewater treatment and implications for wastewater management. J. Environ. Monit. 12 (11), 1956–1978.
- Patrolecco, L., Capri, S., Ademollo, N., 2015. Occurrence of selected pharmaceuticals in the principal sewage treatment plants in Rome (Italy) and in the receiving surface waters. Environ. Sci. Pollut. Res. 22 (8), 5864–5876.
- Pautasso, M., 2013. Ten simple rules for writing a literature review. PLoS Comput. Biol. 9 (7), e1003149.
- Pereira, R.O., Postigo, C., de Alda, M.L., Daniel, L.A., Barceló, D., 2011. Removal of estrogens through water disinfection processes and formation of by-products. Chemosphere 82 (6), 789–799.
- Pereira, V.J., Galinha, J., Barreto Crespo, M.T., Matos, C.T., Crespo, J.G., 2012. Integration of nanofiltration, UV photolysis, and advanced oxidation processes for the removal of hormones from surface water sources. Sep. Purif. Technol. 95, 89–96.
- Pereira, A.M., Silva, L.J., Meisel, L.M., Lino, C.M., Pena, A., 2015. Environmental impact of pharmaceuticals from Portuguese wastewaters: geographical and seasonal occurrence, removal, and risk assessment. Environ. Res. 136, 108–109.
- Perrodin, Y., Orias, F., Boillot, C., De Hugo, A., Brackers, J., Panouilleres, M., Emmanuel, E., 2015. Ecotoxicological risk assessment of hospital wastewater and management recommendations. Rev. Sci. Eau. 28 (1), 59–64.
- Petrie, B., McAdam, E.J., Scrimshaw, M.D., Lester, J.N., Cartmell, E., 2013. Fate of drugs during wastewater treatment. TrAC Trends Anal. Chem. 49, 145–159.
- Petrie, B., McAdam, E.J., Lester, J.N., Cartmell, E., 2014. Assessing potential modifications to the activated sludge process to improve simultaneous removal of a diverse range of micropollutants. Water Res. 62, 180–192.
- Pullin, A.S., Stewart, G.B., 2006. Guidelines for systematic review in conservation and environmental management. Conserv. Biol. 20 (6), 1647–1656.Qian, F., He, M., Song, Y., Tysklind, M., Wu, J., 2015. A bibliometric analysis of global re-
- Qian, F., He, M., Song, Y., Tysklind, M., Wu, J., 2015. A bibliometric analysis of global research progress on pharmaceutical wastewater treatment during 1994–2013. Environ. Earth Sci. 73 (9), 4995–5005.
- Quintana, J.B., Weiss, S., Reemtsma, T., 2005. Pathways and metabolites of microbial degradation of selected acidic pharmaceutical and their occurrence in municipal wastewater treated by a membrane bioreactor. Water Res. 39 (12), 2654–2664.
- Radjenović, J., Petrović, M., Barceló, D., 2009. Fate and distribution of pharmaceuticals in wastewater and sewage sludge of the conventional activated sludge (CAS) and advanced membrane bioreactor (MBR) treatment. Water Res. 43 (3), 831–841.
- Ratola, N., Cincinelli, A., Alves, A., Katsoyiannis, A., 2012. Occurrence of organic microcontaminants in the wastewater treatment process. A mini review. J. Hazard. Mater. 239–240, 1–18.

- Ribeiro, A.R., Nunes, O.C., Pereira, M.F.R., Silva, A.M.T., 2015. An overview on the advanced oxidation processes applied for the treatment of water pollutants defined in the recently launched Directive 2013/39/EU. Environ. Int. 75, 33–51.
- Rivera-Utrilla, J., Sánchez-Polo, M., Ferro-García, M.Á., Prados-Joya, G., Ocampo-Pérez, R., 2013. Pharmaceuticals as emerging contaminants and their removal from water. A review. Chemosphere 93 (7), 1268–1287.
- Rizzo, L., Fiorentino, A., Grassi, M., Attanasio, D., Guida, M., 2015. Advanced treatment of urban wastewater by sand filtration and graphene adsorption for wastewater reuse: effect on a mixture of pharmaceuticals and toxicity. J. Environ. Chem. Eng. 3 (1), 122–128.
- Rodríguez, A., Muñoz, I., Perdigón-Melón, J.A., Carbajo, J.B., Martínez, M.J., Fernández-Alba, A.R., García-Calvo, E., Rosal, R., 2012. Environmental optimization of continuous flow ozonation for urban wastewater reclamation. Sci. Total Environ. 437, 68–75.
- Röhricht, M., Krisam, J., Weise, U., Kraus, U.R., Düring, R.-A., 2009. Elimination of carbamazepine, diclofenac and naproxen from treated wastewater by nanofiltration. Clean: Soil, Air, Water 37 (8), 638–641.
- Ronan, J.M., McHugh, B., 2013. A sensitive liquid chromatography/tandem mass spectrometry method for the determination of natural and synthetic steroid estrogens in seawater and marine biota, with a focus on proposed Water Framework Directive Environmental Quality Standards. Rapid Commun. Mass Spectrom. 27, 738–746.
- Rowan, N.J., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, implications and outlooks. Int. J. Microbiol. 2011, 462832 (15 pages).
- Santos, J.L., Aparicio, I., Alonso, E., 2007. Occurrence and risk assessment of pharmaceutically active compounds in wastewater treatment plants. A case study: Seville City (Spain). Environ. Int. 33 (4), 596–601.
- Santos, L.H.M.L.M., Araújo, A.N., Fachini, A., Pena, A., Delerue-Matos, C., Montenegro, M.C.B.S.M., 2010. Ecotoxicological aspects related to the presence of pharmaceuticals in the aquatic environment. J. Hazard. Mater. 175 (1–3), 45–95.
- Sein, M.M., Zedda, M., Tuerk, J., Schmidt, T.C., Golloch, A., Sonntag, C.v., 2008. Oxidation of diclofenac with ozone in aqueous solution. Environ. Sci. Technol. 42 (17), 6656–6662.
- Simon, E., Lamoree, M.H., Hamers, T., de Boer, J., 2015. Challenges in effect-directed analysis with a focus on biological samples. TrAC Trends Anal. Chem. 67, 179–191.
- Snyder, S.A., Adham, S., Redding, A.M., Cannon, F.S., DeCarolis, J., Oppenheimer, J., Wert, E.C., Yoon, Y., 2007. Role of membranes and activated carbon in the removal of endocrine disruptors and pharmaceuticals. Desalination 202 (1–3), 156–181.
- Sosa-Ferrera, Z., Mahugo-Santana, C., Santana-Rodriguez, J.J., 2013. Analytical methodologies for the determination of endocrine disrupting compounds in biological and environmental samples. BioMed Research International, 2013.
- SRC (2013) PhysProp Database. http://www.syrres.com/esc/physdemo.htm.
- Streck, G., 2009. Chemical and biological analysis of estrogenic, progestagenic and androgenic steroids in the environment. TrAC Trends Anal. Chem. 28 (6), 635–652.
- Suárez, S., Carballa, M., Omil, F., Lema, J., 2008. How are pharmaceutical and personal care products (PPCPs) removed from urban wastewaters? Rev. Environ. Sci. Biotechnol. 7 (2), 125–138.
- Suárez, S., Reif, R., Lema, J.M., Omil, F., 2012. Mass balance of pharmaceutical and personal care products in a pilot-scale single-sludge system: influence of T, SRT and recirculation ratio. Chemosphere 89 (2), 164–171.
- Sumpter, J.P., Johnson, A.C., 2008. 10th anniversary perspective: reflections on endocrine disruption in the aquatic environment: from known knowns to unknown unknowns (and many things in between). J. Environ. Monit. 10 (12), 1476–1485.
- Ternes, T.A., 1998. Occurrence of drugs in German sewage treatment plants and rivers. Water Res. 32 (11), 3245–3260.
- Ternes, T.A., Herrmann, N., Bonerz, M., Knacker, T., Siegrist, H., Joss, A., 2004. A rapid method to measure the solid-water distribution coefficient (K<sub>d</sub>) for pharmaceuticals and musk fragrances in sewage sludge. Water Res. 38 (19), 4075–4084.
- Tomšíková, H., Aufartová, J., Solich, P., Nováková, L., Sosa-Ferrera, Z., Santana-Rodríguez, J.J., 2012. High-sensitivity analysis of female-steroid hormones in environmental samples. TrAC Trends Anal. Chem. 34, 35–58.

- US EPA (1999a) Wastewater Technology Fact Sheet: Intermittent Sand Filters, 1999a. Available online: http://www3.epa.gov/npdes/pubs/isf.pdf [Accessed.
- US EPA (1999b) Wastewater Technology Fact Sheet: Ozone Disinfection, 1999b. Available online: http://www.epa.gov/sites/production/files/2015-06/documents/ozon.pdf [Accessed.
- US EPA (1999c) Wastewater Technology Fact Sheet: Ultraviolet Disinfection, 1999c. Available online: http://www3.epa.gov/npdes/pubs/uv.pdf [Accessed.
- US EPA (2000a) Wastewater Technology Fact Sheet: Free Water Surface Wetlands, 2000a. Available online: http://www3.epa.gov/npdes/pubs/free\_water\_surface\_wetlands. pdf [Accessed.
- US EPA (2000b) Wastewater Technology Fact Sheet: Granular Activated Carbon, 2000b. Available online: http://www3.epa.gov/npdes/pubs/carbon\_absorption.pdf [Accessed.
- US EPA (2000c) Wastewater Technology Fact Sheet: Wetlands: Subsurface Flow, 2000c. Available online: http://www.epa.gov/sites/production/files/2015-06/documents/ wetlands-subsurface\_flow.pdf [Accessed.
- Vazquez-Roig, P., Blasco, C., Picó, Y., 2013. Advances in the analysis of legal and illegal drugs in the aquatic environment. TrAC Trends Anal. Chem. 50, 65–77.
- Verlicchi, P., Zambello, E., 2016. Predicted and measured concentrations of pharmaceuticals in hospital effluents. Examination of the strengths and weaknesses of the two approaches through the analysis of a case study. Sci. Total Environ. 565, 82–94.
- Verlicchi, P., Al Aukidy, M., Zambello, E., 2012. Occurrence of pharmaceutical compounds in urban wastewater: removal, mass load and environmental risk after a secondary treatment—a review. Sci. Total Environ. 429 (0), 123–155.
- Verlicchi, P., Al Aukidy, M., Zambello, E., 2015. What have we learned from worldwide experiences on the management and treatment of hospital effluent? – an overview and discussion on perspectives. Sci. Total Environ. 514, 467–491.
- Vermeirssen, E.L.M., Eggen, R.I.L., Escher, B.I., Suter, M.J.F., 2008. Estrogens in Swiss rivers and effluents- sampling matters. Chimia International Journal for Chemistry 62 (5), 389–394.
  Vieno, N., Sillanpää, M., 2014. Fate of diclofenac in municipal wastewater treatment plant —
- a review. Environ. Int. 69, 28–39. Wenzel, H., Larsen, H.F., Clauson-Kaas, J., Hoibye, L., Jacobsen, B.N., 2008. Weighing environmental advantages and disadvantages of advanced wastewater treatment of
- micro-pollutants using environmental life cycle assessment. Water Sci. Technol. 57 (1), 27–32. Wille, K., De Brabander, H.F., Vanhaecke, L., De Wulf, E., Van Caeter, P., Janssen, C.R., 2012.
- Coupled chromatographic and mass-spectrometric techniques for the analysis of emerging pollutants in the aquatic environment. TrAC Trends Anal. Chem. 35, 87–108.
- Wise, A., O'Brien, K., Woodruff, T., 2011. Are oral contraceptives a significant contributor to the estrogenicity of drinking water? Environ. Sci. Technol. 45 (1), 51–60.
- Xu, P., Drewes, J.E., Bellona, C., Amy, G., Kim, T.-U., Adam, M., Heberer, T., 2005. Rejection of emerging organic micropollutants in nanofiltration-reverse osmosis membrane applications. Water Environ. Res. 77 (1), 40–48.
- Zhang, Y., Zhou, J.L., 2008. Occurrence and removal of endocrine disrupting chemicals in wastewater. Chemosphere 73 (5), 848–853.
- Zhang, Y., Geißen, S.-U., Gal, C., 2008. Carbamazepine and diclofenac: removal in wastewater treatment plants and occurrence in water bodies. Chemosphere 73 (8), 1151–1161.
- Zhou, Z., Jiang, J.-Q., 2015. Treatment of selected pharmaceuticals by ferrate(VI): performance, kinetic studies and identification of oxidation products. J. Pharm. Biomed. Anal. 106, 37–45.
- Zhou, J.L., Zhang, Z.L., Banks, E., Grover, D., Jiang, J.Q., 2009. Pharmaceutical residues in wastewater treatment works effluents and their impact on receiving river water. J. Hazard. Mater. 166, 655–661.
- Ziylan, A., Ince, N.H., 2011. The occurrence and fate of anti-inflammatory and analgesic pharmaceuticals in sewage and fresh water: treatability by conventional and nonconventional processes. J. Hazard. Mater. 187 (1–3), 24–36.

Contents lists available at ScienceDirect

Science of the Total Environment

# ELSEVIER



#### journal homepage: www.elsevier.com/locate/scitotenv

### Development of a semi-quantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: An Irish case study



Alexandre Tahar<sup>a,\*</sup>, Erin Jo Tiedeken<sup>a,b</sup>, Eoghan Clifford<sup>c</sup>, Enda Cummins<sup>d</sup>, Neil Rowan<sup>a</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Ireland

<sup>b</sup> National Biodiversity Data Centre, Waterford, Ireland

<sup>c</sup> College of Engineering and informatics, Department of Civil Engineering, National University of Ireland Galway, Ireland

<sup>d</sup> School of Biosystems and Food Engineering, University College Dublin, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- Lack of data regarding the likelihood of one given pharmaceutical to be found in river
- Semi-quantitative risk assessment (RA) proposed for three watch-list PhACs
- The RA is based on influent source, PhACs properties, WWTP removal and receiving water
- A method to identify pollution hotspots and WWTPs responsible for PhACs release is provided



#### ARTICLE INFO

Article history: Received 23 February 2017 Received in revised form 23 May 2017 Accepted 24 May 2017 Available online 26 June 2017

Editor: D. Barcelo

Keywords: Pharmaceutical compounds EU-watch list Receiving water Risk assessment Wastewater treatment plants

#### ABSTRACT

Contamination of receiving waters with pharmaceutical compounds is of pressing concern. This constitutes the first study to report on the development of a semi-quantitative risk assessment (RA) model for evaluating the environmental threat posed by three EU watch list pharmaceutical compounds namely, diclofenac, 17-beta-estradiol and 17-alpha-ethinylestradiol, to aquatic ecosystems using Irish data as a case study. This RA model adopts the Irish Environmental Protection Agency Source-Pathway-Receptor concept to define relevant parameters for calculating low, medium or high risk score for each agglomeration of wastewater treatment plant (WWTP), which include catchment, treatments, operational and management factors. This RA model may potentially be used on a national scale to (i) identify WWTPs that pose a particular risk as regards releasing disproportionally high levels of these pharmaceutical compounds, and (ii) help identify priority locations for introducing or upgrading control measures (e.g. tertiary treatment, source reduction). To assess risks for these substances of emerging concern, the model was applied to 16 urban WWTPs located in different regions in Ireland that were scored for the three different compounds and ranked as low, medium or high risk. As a validation proxy, this case study used limited monitoring data recorded at some these plants receiving waters. It is envisaged that this semi-quantitative RA approach may aid other EU countries investigate and screen for potential

\* Corresponding author at: Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co. Westmeath, Ireland. *E-mail address:* atahar@ait.ie (A. Tahar). risks where limited measured or predicted environmental pollutant concentrations and/or hydrological data are available. This model is semi-quantitative, as other factors such as influence of climate change and drug usage or prescription data will need to be considered in a future point for estimating and predicting risks.

© 2017 Elsevier B.V. All rights reserved.

#### 1. Introduction

Pollution of European receiving waters containing pharmaceuticals is a ubiquitous phenomenon (Barbosa et al., 2016; Verlicchi and Zambello, 2016; Tiedeken et al., 2017). Pharmaceuticals are a class of emerging environmental contaminants that are widely used in human and veterinary medicine and are essential to modern healthcare (Streck, 2009; Kosma et al., 2014). From here on, these compounds will be referred to as pharmaceutically active chemicals (PhACs). Nevertheless, there are growing concerns about the negative impacts that may result from continuous contamination of the environment with PhACs. This research is important because of the potential toxic effects for aquatic biota and human health that may result from chronic exposure to PhACs (Miege et al., 2008; Streck, 2009; Kosma et al., 2014). PhACs exhibit wide variation in function, chemical structure and physiochemical properties, making it difficult to generalize about their behaviour, persistence or impact in the environment. PhACs are also designed to be biologically active, have a specific mode of action and to be persistent in the body, meaning they can impact humans and wildlife at trace concentrations that are often hard to detect and quantify using traditional analytical methods (Kosma et al., 2014). A large number of PhACs have been detected in wastewater treatment plants (WWTPs) influents and effluents and surface, ground and drinking water worldwide in recent years (Nikolaou et al., 2007; Cirja et al., 2008; Streck, 2009; Verlicchi and Zambello, 2016). The impacts of chronic exposure to trace concentrations of many PhACs on wildlife and human health may be severe (Verlicchi et al., 2012b), thus it is critical to limit as much as possible the concentrations of this class of contaminants in aquatic environments.

Until recently, environmental regulations worldwide had not required explicit testing for any PhACs in water bodies. However, given the growing concern about contamination of the aquatic environment with these compounds, legislation has recently begun to acknowledge this potential problem. The Water Framework Directive (WFD) requires that all EU member states prepare river basin management plans (RBMPs) to address the many issues relating to water quality and protection in a holistic manner. In response to growing EU concern about the release of untreated PhACs into the aquatic environment, three compounds were included on in the first EU watch list in 2013: diclofenac (CAS# 15307-79-6, hereafter referred as DCL), 17-betaestradiol (CAS# 50-28 - 2, hereafter referred as E2) and 17-alphaethinylestradiol (CAS# 57-63-6, hereafter referred as EE2). E2 and EE2 can impact the endocrine system of humans or wildlife where there is growing fears that chronic exposure to these endocrine disrupting chemicals or endocrine disrupting chemicals (EDCs) even at low concentrations (ng/L) (in bathing or drinking water, for example) may be linked to adverse human health conditions (Hernando et al., 2006; Streck, 2009). Similar to PhACs as a whole, EDCs are generally thought to be transported into the aquatic environment mostly via incomplete removal at wastewater treatment plants (WWTPs) (Streck, 2009). It is relevant to note that the European Commission implemented decision 495 of 20 March 2015 that expanded substances or groups of substances on the watch list to 10 in the field of water policy (Barbosa et al., 2016). This review focuses solely on the first three substances DCL, E2 and EE2 as there is a requirement to investigate policy implications for Ireland of these PhACs in receiving waters in the first instance.

Tiedeken et al. (2017) conducted a 20 year systematic review of 3945 potentially relevant articles published between 1995 and 2015

that produced a EU-wide database consisting of 1268 publications on DCL, E2 and EE2 in receiving waters. Overall, European surface water concentrations of DCL are typically reported below the proposed annual average environmental quality standard (AA EQS) of 100 ng/L with only a few extreme values exceeding this threshold (up to 1200 ng/L according to Rivera-Utrilla et al. (2013)). E2 and EE2 surface water concentrations are typically below 50 ng/L and 10 ng/L respectively, but these concentrations greatly exceed the proposed AA EQS values for these compounds (0.4 and 0.035 ng/L respectively). Furthermore, levels of these PhACs are frequently reported to be disproportionately high in EU receiving waters (up to 200 ng/L and 831 ng/L respectively in surface water according to Tiedeken et al. (2017)), particularly in effluents at control points that require urgent attention. Furthermore, the number of articles produced by each of the 28 EU countries along with Switzerland, Norway and Turkey varied greatly on sources, monitoring or control measures for DCL, E2 and EE2 over this 20-year systematic review period, where Spain, Germany and the United Kingdom contributed 707 (56%) of all reports. However, 24 and 16 EU countries produced under 50 and 20 reports respectively on these PhACs in their national receiving waters; consequently, very few countries have reported on use predicted or measured environmental concentrations to underpin modelling or to inform risks in their river basins (ter Laak et al., 2010; Guillén et al., 2012). Overall, it was found that DCL and EE2 enter European aquatic environment mainly following human consumption and excretion of therapeutic drugs, and by incomplete removal from influent at urban WWTPs. E2 is a natural hormone excreted by humans, which also experiences incomplete removal during WWTP treatments. Thus, WWTPs (initially not designed to remove PhACs) are considered as pressure point for control of PhACs in aquatic environment (Tiedeken et al., 2017). Current conventional analytical methods (i.e. LC-MS/MS) are sufficiently sensitive for the detection and guantification of DCL, but generally not for E2 and EE2 (at the very low EQS levels proposed for those compounds, i.e. 0.4 and 0.035 ng/L respectively, levels that cannot be reached for most of the standards chemical analysis labs) (Streck, 2009), thus alternative, ultra-trace, time-integrated monitoring techniques such as passive sampling are needed to inform water quality for these estrogens (Buchberger, 2011; Wille et al., 2012). Another emerging potential solution to the problem of low EQS values of E2 and EE2 is the use of biological effects monitoring techniques (Streck, 2009; Kunz et al., 2015; Simon et al., 2015). WWTPs are today widely considered as the main vector of PhACs through the aquatic environment (Hernando et al., 2006; Tiedeken et al., 2017). However, in Ireland as well in numerous other EU countries the number of reliable WWTPs and surface/ground waters monitoring data is still limited (Tiedeken et al., 2017) making difficult the direct and quantitative risk analysis associated to the selected PhACs. The aforementioned limitations of reliable data at WWTPs makes it difficult to directly quantify risks for aquatic environment. As an alternative, the application of a risk assessment (RA) method appeared to be the best way to obtain a full picture of the potential contamination of surface waters by the three selected PhACs and to identify PhACs emission hotspots. Other RA models related to PhACs in the aquatic environment aiming at: (i) prioritize certain compounds (Guillén et al., 2012), (ii) the evaluation of the ecological/environmental risk related to the presence of PhACs in rivers downstream from WWTP (e.g. Hernando et al., 2006; Ginebreda et al., 2010; Gros et al., 2010; Kosma et al., 2014; Pereira et al., 2015), (iii) the evaluation of PhACs concentrations in surface water based on their physico-chemical characteristics (Lindim et al., 2017), or on their

consumption (ter Laak et al., 2010). These models are generally "compound related" and focused on a certain number of compounds, never specifically on the WFD watch-list compounds. These studies generally aimed at the estimation of risk based the comparison of on predicted environmental concentration (PEC) (from WWTP monitoring) with predicted no effect concentrations (PNEC) (generally calculated from acute toxicity tests). To the best of our knowledge, no "WWTPs related" RA was developed so far.

This article aims to develop a semi-quantitative RA model for evaluating the environmental threats posed by Irish urban WWTPs in regards to PhACs contamination of aquatic ecosystems. As a case study, this risk screening model identified urban WWTPs in Ireland that are at risk of negatively impacting the aquatic environment via contamination with the three targeted WFD watch list PhACs of interest: namely DCL, E2 and EE2. While focused on Irish data, this RA model offers a promising tool to be used to identify WWTPs that pose a particular risk in regards to these PhACs, and to therefore prioritize the adoption of control measures at pollution 'hot-spots' (e.g. the application of efficient tertiary treatment such as advanced oxidation processes, activated carbon or membrane bioreactors or a combination of these). The model also identifies risk factors relevant to a variety of human pharmaceuticals, an emerging class of pollutants that is of ever-growing concern. Therefore, this study should be of interest to the international reader trying to protect the environment from pollution and to examine efficacy of WWTPs as critical pressure points for governing water quality in river catchment basins.

#### 2. Materials and methods

#### 2.1. Risk assessment model development

The model was designed following the risk screening guidelines of Ireland's Drinking Water Safety Plans (EPA, 2010) where the "Risk Screening Methodology for Cryptosporidium" was adapted to consider risk factors specific to the discharge of the PhACs of interest in this study. Information was also drawn from national and international peer-reviewed research in this field of study, such as Tiedeken et al. (2017) that constituted the foundation systematic bibliographic within this project. The general principles of this RA model align with this environmental protection agency (EPA)-sanctioned risk screening methodology, which uses the Source-Pathway-Receptor (SPR) concept to define relevant input parameters. A scoring system was employed that enables determination of each agglomeration as low, medium or high risk for each PhAC of interest with threshold values being as one third and two thirds of the maximum score respectively (Table 1). A risk score was calculated for each input based on specific factors; for this preliminary study and in order to build a semi-quantitative model, the risk scores associated to each input were defined as the sum of the score given to each factor. The final score was defined as the sum of the scores for each of the 4 inputs, allowing for WWTPs to be ranked relative to each other. The four input scores, considered being

#### Table 1

Maximum possible risk score for each PhAC of interest, and relevant scoring system enabling the classification of each WWTP as high risk, medium risk or low risk. Scores originate from four input parameters which are added to determine the final risk category.

	Diclofenac	E2	EE2	Risk classification
Input 1	5	13	6	
Input 2	0	0	0	
Input 3	11	11	11	
Input 4	8	8	8	
Total	24	32	25	Highest possible score
	≥17	≥21	≥17	High risk classification
	9 – 16	11 – 20	9 - 16	Medium risk classification
	≤8	≤10	≤8	Low risk classification

representative of the different sources of the targeted PhACs through the aquatic environment, were (i) source of influent, (ii) removal due to treatment, (iii) chemical properties of PhAC and (iv) fate of effluent. Where the level of uncertainty regarding an input parameter is high, a precautionary approach is adopted and the highest score is used; however, in such a case the source of uncertainty is noted and further review or monitoring for the factor is recommended. Due specifically to different substance characteristics, 3 separate RA are described within this same model.

#### 2.2. Input 1 - source of influent entering wastewater treatment plants

Agglomeration generated load (AGL) was included as a measure of the size of the population that is serviced by each WWTP. This factor assumes that the greater the AGL, the higher the potential for receiving influent contaminated with the PhAC of interest, i.e. higher population leads to higher levels of drug usage. According to the Terms and Definitions of the Urban Wastewater Treatment Directive 91/271/EEC, AGL is the organic biodegradable load of a WWTP expressed in population equivalents (PE), and consists of all urban wastewater requiring collection under Article 3(1) of the Directive. Furthermore, one PE is equal to the organic biodegradable load having a five-day biochemical oxygen demand (BOD<sub>5</sub>) of 60 g oxygen/day (Santos et al., 2010). AGL was grouped into 5 categories (Table 2), with risk scores increasing with higher PE values. AGL data were obtained from agglomeration specific 2015 annual environmental reports (AERs) submitted to the EPA. Some WWTPs are responsible for taking in sludge or effluent from domestic septic tanks and/or various industries. These data were obtained from agglomeration specific 2014 AERs submitted to the EPA. Drug utilization data was available in Ireland through Heath Service Executive (HSE) Primary Care Reimbursement Services to supplement AGL determinations, however as this data only covered 70% of Irish population it was not factored into this RA model. Agglomeration size can be indicatively used as a substitute to drug utilization data in RA models (EPA, 2010), particularly those that consider emissions of PhACs from specific WWTPs (Boxall et al., 2014). Domestic effluent and sludge could be a significant source of pharmaceuticals, especially the natural steroid estrogen E2 (UWWTD-REP Working Group, 2006; Gill et al., 2009). Industrial and healthcare inputs could potentially include effluent or sludge

Source of influent factors used in risk assessment model to calculate input one risk score. White indicates factor considered for all 3 compounds, light grey indicates factor considered only for E2 and EE2, dark grey indicates parameter considered only for E2. The colour of the risk score indicates whether there is increased risk (positive values, red), no impact on risk (zero values, blue) or decreased risk (negative values, green).

Factor	Source factor description	Risk
	-	score
1. Agglomeration generated load	PE served <500	1
(AGL)	PE served 500-50,000	2
	PE served >50,000	3
2. Domestic septic tank	No	0
sludge/effluent received?	Yes	1
3. Industrial sludge/	No	0
effluent received?	Yes	1
4. Gender ratio in county,	≤1	0
women/men	>1	1
5. Cattle score	No cattle/calves in region	0
	≤80 livestock unit per ha forage area in	2
	region	
	>80 livestock unit per ha forage area in	3
	region	
6. Sheep score	No sheep/lambs in region	0
	≤70 livestock unit per ha forage area in	1
	region	
	>70 livestock unit per ha forage area in	2
	region	
7. Pig score	No pigs in county	0
	≤20 livestock unit per ha forage area in	1
	region	
	>20 livestock unit per ha forage area in	2
	region	
Total for input 1		

from hospitals or pharmaceutical manufacturers. Previous studies have shown that effluent and sludge from hospitals and pharmaceutical manufacturing facilities can contain high levels of DCL, with concentrations up to 3000 ng/L (Fent et al., 2006; Kümmerer, 2009b) and estrogens, with concentrations up to 70 ng/L (Kümmerer, 2009b; Santos et al., 2010). As a result, if a WWTP is required to receive either one of these additional inputs, a point is awarded towards the final risk score for this factor (Table 2).

For the E2/EE2 models in terms of specific sources of hormones, additional factors are considered for input one. The natural steroid estrogen E2 is mainly excreted by humans and excretion efficiencies are especially high in pregnant women (Qian et al., 2015); this natural PhAC can also be used in prescribed drugs, including hormone replacement therapy and to treat infertility in women or advanced prostate and breast cancer (reviewed in Ben Fredj et al., 2015). EE2 is the main estrogenic ingredient in oral contraceptive pills taken by women of reproductive age (Verlicchi et al., 2012a; Tiedeken et al., 2016). Therefore, EE2 is mainly excreted by the portion of the population who ingest the contraceptive pill, i.e. women. It is also found in other prescription medications including hormone replacement therapies, palliative treatments for breast and prostate cancer, and lotions used to prevent androgen-dependent hair loss in women (reviewed in Ben Fredj et al., 2015). When considering E2 and EE2 (mainly excreted by women), the gender ratio of the county in which the agglomeration is located is considered. An additional point is therefore awarded towards the final risk score if the ratio of women: men is greater than one (Table 2). It is noticed that the score of locations where gender ratio is lower that one was kept to zero whereas such population will still emit some E2 and EE2; it was considered that it was sufficient accuracy for the development of such a semi-quantitative model as the one presented here. Population and gender information were obtained through the Central Statistics Office (CSO) 2011 census data. Agriculture is known to be another potential source of natural steroid estrogens (Vieno and Sillanpää, 2014). In particular dairy cows are known to be major contributors in comparison to sheep and pigs (Santos et al., 2010). For that reason cattle density in the regional authority in which the agglomeration and its primary discharge point are located is included as a factor in the RA model (as in EPA, 2010). Pig and sheep density were also considered, however a lower risk score is associated with these livestock compared with cattle (Table 2). Data on livestock numbers were obtained through the CSO StatBank online resource. The number of livestock in June for the year 2014 (file AAA07) was considered for total cattle, total sheep and total pigs. The spatial resolution of these data were obtained from the corresponding regional authority (NUTS3 classification), and the numbers were divided by the total land area of each region, as defined by the CSO. The excretions from livestock will not enter the WWTP for treatment; however high densities of livestock may lead to high background levels of estrogens in receiving waters. This factor is only considered when modelling for E2 as synthetic estradiol products (e.g. EE2) are banned in the EU as growth promoters, thus no products containing them are authorized by the Health Products Regulatory Authority for use in animals. To the best of our knowledge, despite authorized by the European Medicines Agency (2014), DCL is not used as a veterinary drug in Ireland and was therefore excluded from the livestock section of the model.

#### 2.3. Input factor 2 - removal of compounds due to wastewater treatment

Removal of PhACs from wastewater due to treatment can occur to varying extents (de Graaff et al., 2011); however, it is dependent upon a number of critical factors concerning the treatment, operation and management of WWTPs for each agglomeration (EPA, 2010). If not treated, a PhAC will end up in the environment making WWTPs the main vectors of PhACs for unwanted pollution. The first factor considered under input two is whether or not the WWTP effluent undergoes tertiary treatment. In Ireland, the majority of WWTPs do not include tertiary treatment (EPA, 2015). However, a limited number that discharge at/near a sensitive area employ tertiary treatments all year round, while others only treat for part of the season (i.e. the bathing season) in order to compensate the increased effluent load and to maintain a certain level of treatment during the tourism season. Advanced treatment technologies can increase removal efficiencies of PhACs, specifically E2 and EE2 (Boxall, 2010) and DCL (Joss et al., 2006). Therefore, if tertiary treatments is implemented for the whole year, the risk score associated with this factor is negative, because increased removal efficiencies of these three PhACs at such facilities would be expected (Table 3). The treatment technologies for the specific WWTPs of interest were identified through 2014 AERs and communication with the EPA.

The second factor under input two is the type of secondary treatment employed by the WWTP. Conventional WWTPs generally include both a primary physico-chemical and a secondary treatment characterized by biological treatments via activated sludge. Activated sludge plants have a limited capacity to remove many PhACs because most of these compounds cannot be metabolized by microorganisms (Joss et al., 2006). However, activated sludge plants including nutrient removal by extended aeration in the secondary treatment step have been shown to more effectively remove PhACs (including E2 and EE2 (Santos et al., 2010) and DCL (Kümmerer, 2009b; Lacey et al., 2012)). According to the Urban Wastewater report published by the EPA, 25.7% of Irish wastewater load (in PE) received secondary treatment plus biological nutrient reduction in the year 2014 (EPA, 2015). The risk score for this factor therefore decreases when the most recent AER indicates that a WWTP employs biological nutrient removal (Table 3). There is no additional or reduced risk when the plant employs conventional treatments without nutrient removal (no extended aeration) or sequence batch reactors (with or without phosphorous removal), as studies indicate these are not influencing the removal efficiencies of the PhACs of interest (Joss et al., 2006; Lacey et al., 2012). The type of secondary treatment technology employed by each plant was identified from the 2015 AERs for each agglomeration of interest.

The third factor taken into consideration for input two is a general measurement of the performances of WWTPs assuming that if standard

Treatment, operation and management factors used in risk assessment model to calculate input two risk score. The colour of the risk score indicates whether there is increased risk (positive values, red), no impact on risk (zero values, blue) or decreased risk (negative values, green).

Factor	Treatment factor description	Risk score
1- Tertiary treatment	Present year round	-2
	Implemented seasonally	0
	Absent year round	0
2- Type of secondary treatment (including nutrient removal)	Extended aeration (N removal)	-1
	Conventional activated sludge/sequence	0
	batch reactor (with or without P removal)	
3- WWTP quality measurement	Pass most recent UWWTD compliance criteria	-1
	Fail most recent UWWTD compliance criteria	0
Total for input 2		

parameters (e.g., BOD<sub>5</sub>, COD, TSS, nitrogen and phosphate compounds) are not efficiently eliminated then the PhACs won't be eliminated either. National compliance measures recorded by the EPA can have different requirements depending on the specifications in WWTP licenses. If a WWTP passed the most recent UWWTD compliance criteria, a negative score was awarded to the risk score for this factor (Table 3). Unpublished compliance data were provided by the EPA and originate from Irish Water reported values.

#### 2.4. Input factor 3 - chemical properties of compounds

The chemical properties of the specific PhACs of interest will play a contributing role in their persistence in water during the wastewater treatment process. For example, a characteristic specific to individual PhACs is metabolism in the human body. The rate of excretion, combined with drug usage data, will ultimately determine how much of the compound ends up in wastewater and enters a plant for treatment (Oaks et al., 2004; Santos et al., 2010) excluding some potential degradation/metabolization in the pipework leading to the WWTP (considered negligible in the present study due to low biomass content and short hydraulic residence time). Thus, the metabolism of the compound of interest was considered (Table 4). As the rate of excretion of the parent compound increases, so does the risk score. It is noticed that only excretion via urine was considered as the targeted PhACs residues or metabolites are demonstrated to be mostly excreted via urine compared to faeces (Lienert et al., 2007).

Efficiencies of excretion for DCL, E2 and EE2 were determined from the literature and with an EPA technical report (Tiedeken et al., 2016; Tiedeken et al., 2017). The main removal pathways for the PhACs of interest during wastewater treatment are sorption to sludge and biotransformation/biodegradation (Kümmerer, 2009b; Santos et al., 2010; Lacey et al., 2012). Sludge is a complex material comprised of live and dead microorganisms which provide a large surface area for PhACs to sorb onto; however, sorption is a complex process, which can depend on hydrophobicity, functional group polarity, ion exchange, chelation or other factors (Rivera-Utrilla et al., 2013). In this RA model, compounds that are recognized in the literature as having low water solubility/high hydrophobicity have a lower risk score because they will theoretically tend to sorb onto sludge and therefore not be discharged in the WWTP effluent (Table 4). It is noteworthy that some initially sorbed compounds could reach the aquatic environment through an overflow event that could accidentally cause the release of some untreated wastewater; this potential alternative route was considered negligible and therefore ignored in the design of the present model. The hydrophobicity and therefore the likelihood of a compound to sorb to sludge is often correlated with physico-chemical parameters such as a compound's octanol-water partition coefficient (Kow) (Joss et al., 2006) or its pH-dependent equivalent, Dow (Lapworth et al., 2012). It was suggested that Log (Kow) values could be related to organic compounds sorption potential on sludge and that this phenomenon could be considered as "not expected", moderate and high for values  $\leq 2.5, 2.5-4$  and  $\geq 4$ , respectively (Cirja et al., 2008). As a result, Log (K<sub>ow</sub>) (or Log (D<sub>ow</sub>) for DCL that is an ionisable compound, pKa = 4.15) values from the literature were considered when determining the likelihood of sorption for each PhAC of interest (Tiedeken et al., 2017) and were considered as threshold values for associated risk score assignments.

Similar to sorption, biodegradation and biotransformation are complex processes, and no single physico-chemical parameter can predict a compound's potential for elimination via this removal pathway (Kümmerer, 2009b). Again, information from the literature was used to determine whether the potential of each PhAC to experience biodegradation or biotransformation was high or low (Table 4). A property that helped make this determination was the compound's half-life in the environment; however, this value can vary widely depending on the environmental matrix being studied. The experimentally determined biological degradation constant (K<sub>biol</sub>) was therefore taken into consideration (Ziylan and Ince, 2011). Joss et al. (2006) suggested that the biological degradation constant  $K_{biol}$  (L/g<sub>ss</sub>/d) could be used to estimate the biodegradation potential of a compound, where values <0.1 translate to no substantial biodegradation, values between 0.1 and 10 indicate partial biodegradation and values above 10 indicate ready degradation (Table 4).

Some full scale WWTP studies report negative removal efficiencies of PhACs; in other words, the concentration of the PhAC of interest is higher in the effluent than in the influent (Thomas et al., 2007; Kümmerer, 2009b). Apart from an analytical artefact, it has been suggested that this phenomenon occurs because of deconjugation of conjugated metabolites of the PhACs of interest (Thomas et al., 2007; Lacey et al., 2012). To account for this, the final factor associated with this input was assigned a point towards the risk score if the literature

Chemical properties of compounds used in risk assessment model to calculate input three risk score. The colour of the risk score indicates whether there is increased risk (positive values, red), no impact on risk (zero values, blue) or decreased risk (negative values, green).

		1
Factor	Chemical properties factor description	Risk score
1– Metabolism	Rate of excretion 0–25%	1
	Rate of excretion 26–50%	2
	Rate of excretion 51–75%	3
	Rate of excretion 76–100%	4
2– Sorption potential	Low water solubility/high hydrophobicity $Log(D_{ow}) \ge 4$	1
	Medium water solubility/low hydrophobicity Log(D <sub>ow</sub> ) = 2.5-4	2
	High water solubility/low hydrophobicity $Log(D_{ow}) \le 2.5$	3
3- Degradation potential	High degradation through photolysis, hydrolysis or other mechanisms	1
	$K_{biol} > 10 L/g_{ss}/day$	
	Medium degradation through photolysis, hydrolysis or other mechanisms K <sub>biol</sub> = 1-10 L/g <sub>ss</sub> /day	2
	Low degradation through photolysis, hydrolysis or other mechanisms $K_{\rm biol}\!<\!1.10~{\rm L/g_{ss}}/{\rm day}$	3
4- Potential for deconjugation of conjugated metabolites during treatment	Not found to occur in the literature	0
	Deconjugation potential identified through literature	1
Total for input 3		·

indicated that deconjugation of conjugated metabolites during wastewater treatment was proven (Table 4).

Specific supporting information underpinning RA model for targeted PhACs is presented as follows.

Diclofenac: in regards to factor number one (metabolism), in this RA model only excretion via urine is considered (Table 4). Of the orally administered dose of DCL, between 65 and 70% is excreted in urine as the parent compound or metabolites (and Fent et al., 2006; reviewed in Lacey et al., 2012; Vieno and Sillanpää, 2014). According to the RA model methodology, DCL is thus assigned a risk score of three for this factor (Table 4). Removal efficiencies of DCL during wastewater treatment can vary widely depending on both the process configuration and parameters utilized by a treatment plant (reviewed in Lacey et al., 2012). In general however, DCL has only a weak ability for removal through sorption to sewage sludge (Zhang et al., 2008; Boxall, 2010; Wise et al., 2011; Vieno and Sillanpää, 2014). This is due in part to its physico-chemical properties; DCL is soluble in water and has a low  $Log(D_{ow})$  (2.50 at a pH value of 8, typical of wastewater treatment conditions) (Kümmerer, 2009a). As expected of compounds with low Log(Dow) values, sorption of DCL to sludge is unlikely. Due to these reports from the literature, in the RA model DCL is considered to have low sorption potential and is assigned a risk score of three (Table 4). Similar to the potential for sorption, biodegradation of DCL during wastewater treatment is generally accepted in the literature as being poor (Zhang et al., 2008; Vieno and Sillanpää, 2014). The K<sub>biol</sub> threshold values of 0.1 and 10 were used as suggested in Joss et al. (2006). The biodegradation values measured for DCL in the Joss et al. (2011) study were below 0.1, regardless of the biological process configuration. Additional evidence of the poor biodegradation of DCL is reviewed in Vieno and Sillanpää (2014), hence the risk score associated with this factor is three (Table 5). Finally, numerous studies (including one carried out in Ireland) have noted higher concentrations of DCL in WWTP effluent than influent (e.g. Thomas et al., 2007; Zhang et al., 2008; Santos et al., 2010). These studies have suggested that deconjugation of glucuronide or sulphate conjugates of DCL might occur during the wastewater treatment process, leading to negative removal efficiencies. Thus, DCL is assigned a risk score of two for the final factor in input three (Table 4).

*E2 and EE2* - in this RA model, only excretion of the parent metabolite or conjugates via urine is considered. Literature indicates that 22– 50% of the daily dose of EE2 is excreted in urine (Verlicchi et al., 2012a). Thus, EE2 is assigned a risk score of 2 for factor one, input three (Table 4). Metabolism for E2 is evaluated slightly differently; because this is a natural hormone, rate of excretion represented as percent of the daily dose ingested is inappropriate. Instead quantities (µg) of E2 excreted in the urine as conjugated or unconjugated forms must be considered. The literature reports that 14 µg and 2.4–3.0 µg of conjugated and unconjugated E2 respectively are excreted by women in urine per day (Verlicchi et al., 2012a; Petrie et al., 2013; Rivera-Utrilla et al., 2013). In order to assign a risk score for the model, these values were compared with the excretion efficiencies of other natural estrogenic hormones (estrone E1 and estradiol E3). In the studies examined, E2 was always the natural hormone with the lowest daily excretion rate (Petrie et al., 2013; Rivera-Utrilla et al., 2013), therefore it is assigned a risk score of two in our model (Table 4). The use of  $\mu$ g excreted per day (as opposed to percent of the daily dose excreted) prevents direct comparison of the E2 input 3 score with that of the other targeted PhACs.

Both E2 and EE2 have low polarity and like most other EDCs, are moderately hydrophobic organic molecules; this means they have a tendency to distribute into organic phases (Kunz et al., 2015). The compounds E2 and EE2 are not ionisable and  $log(K_{ow})$  values of 4.01 and 3.67 respectively are therefore taken into account. Sorption onto sludge is therefore considered partially responsible for removal of E2 and EE2 from wastewater, and they are assigned a score of 1 and 2 respectively in the RA model (Table 4). Compared with DCL, E2 and EE2 have a much higher potential for biodegradation (Burkhardt-Holm, 2010; Hecker and Hollert, 2011; Wise et al., 2011). However, studies comparing the removal efficiencies of estrogens indicate that EE2 is more recalcitrant than E2 (Wise et al., 2011; Kunz et al., 2015). K<sub>biol</sub> values for E2 range from 300 to 800  $L/g_{ss}/d$  in the literature, whereas values for EE2 are between 7 and 9 L/g<sub>ss</sub>/d (Verlicchi et al., 2012a). The risk score for this factor is thus classified as one for E2 and two for EE2 (Table 4). Finally, in regards to factor four (potential for deconjugation of conjugated metabolites during treatment), the same approach was taken as previous RA models in assuming that all the steroids excreted as conjugates are de-conjugated in the sewers before they reach the treatment plant (Wise et al., 2011). Therefore the risk score assigned for this factor is 0 for both E2 and EE2 (Table 4).

The total input three scores for DCL, E2 and EE2 are 12, 4 and 5 respectively (Table 5). These scores can be used to compare the risk associated with DCL, E2 and EE2 in regards to aquatic persistence during wastewater treatment. According to our model, of these three compounds, DCL presents approximately twice the risk of being found in receiving waters compared with E2 or EE2. This finding is consistent with reports from the literature as it demonstrates that DCL is among the most recalcitrant pharmaceuticals (Wise et al., 2011), and that its removal percentages during wastewater treatment, though variable, are on average much lower than those of E2 or EE2 (Lacey et al., 2012; Kunz et al., 2015; Tiedeken et al., 2017). The concentrations of DCL in effluent and receiving waters are generally in the high nanogrammicrogram per litre range, whereas E2 and EE2 are almost always reduced to nanogram per litre levels after wastewater treatment (Burkhardt-Holm, 2010; Lacey et al., 2012). In addition, the input three scores for E2 and EE2 indicate that the latter is slightly more persistent after wastewater treatment; this is also consistent with findings reported in the literature, which show that EE2 is considerably more recalcitrant than E2 and its removal efficiencies in activated sludge treatment plants are more variable (Wise et al., 2011). The congruence of these input three scores with the literature indicates the factors for this input are scored appropriately.

Table 5

Factor scoring for each PhAC of interest for risk assessment model, input three. Final score assigned in model for each factor is in colour and in bold.

	Factor 1 Metabolism	Factor 2 Sorption	Factor 3 Degradation	Factor 4 Conjugation	Total score
Diclofenac	3ª De	3 <sup>b</sup>	3 <sup>c</sup>	1 <sup>d</sup>	10
17-alpha-ethinylestradiol (EE2)	2 2 <sup>i</sup>	1 2 <sup>f</sup>	2 <sup>g</sup>	0 <sup>h</sup>	6

<sup>a</sup> (Vieno and Sillanpää, 2014) reviewed in (Fent et al., 2006; Lacey et al., 2012).

<sup>b</sup> (Zhang et al., 2008; Boxall, 2010; Wise et al., 2011; Vieno and Sillanpää, 2014) reviewed in Vieno and Sillanpää (2012).

<sup>c</sup> (Zhang et al., 2008; Ziylan and Ince, 2011; reviewed in Vieno and Sillanpää, (2014)

<sup>d</sup> (Thomas et al., 2007; Santos et al., 2010).

<sup>f</sup> (Wise et al., 2011; Kunz et al., 2015).

<sup>h</sup> (Wise et al., 2011).

<sup>i</sup> (Verlicchi et al., 2012a).

<sup>&</sup>lt;sup>e</sup> (Verlicchi et al., 2012a; Petrie et al., 2013; Rivera-Utrilla et al., 2013).

<sup>&</sup>lt;sup>g</sup> (Burkhardt-Holm, 2010; Hecker and Hollert, 2011; Wise et al., 2011; Verlicchi et al., 2012a).

#### 2.5. Input factor 4 - fate of treated effluent

Models of the risk and impact of PhACs in surface waters typically use information not only on drug usage, metabolism, and fate in WWTPs, but also on fate in receiving waters. Consequently, the final input used to calculate the overall risk score for this model considers factors specific to the receiving water the treated effluent from a WWTP is discharged to. For this RA study, only the primary discharge site of the WWTPs of interest and its potential PhACs dilution were considered. The self-purification potential of the receiving water was ignored for this preliminary study due to the pseudo-persistence concept and the fact that PhACs of interest are constantly introduced into receiving water via WWTPs (Hernando et al., 2006). The first factor associated with input four is the type of receiving water the treated effluent is discharged to. Receiving water is classified as one of four types: 1.) coastal; 2.) transitional, estuary, river, or lake; 3.) stream; or 4.) ground (Table 6). No WWTP evaluated released treated effluent directly to ground water, but theoretically this situation may occur, for example when considering domestic wastewater treatment systems (septic tanks) (UWWTD-REP Working Group, 2006; Gill et al., 2009; Ronan and McHugh, 2013). Dilution of the treated effluent is very high when released to coastal environments, hence, a low risk score is associated with this type of receiving water. Dilution will vary considerably in transitional/estuary/river/lake receiving waters (depending on their size, flow, etc.), but generally should be lower than in coastal receiving waters, hence, the risk score is increased. The risk score is the highest if the receiving water is classified as a stream, where dilution could be very low, especially during periods of low rainfall (Table 6). Data on the type of receiving water of each of the WWTP of interest were obtained from the 2015 AERs. Another factor is whether or not the primary discharge point is located at or near a sensitive area (Table 6). Sensitive areas are identified in the 2001 and 2010 Urban Wastewater Treatment Regulations. If the primary discharge point from a plant was reported in the 2015 AER as being directly into a sensitive area, an additional point is given to the risk score for this factor. The flow of the receiving water is considered in order to evaluate the fate of the treated effluent (Table 6). The hydrometric monitoring station closest to the primary discharge location of each WWTP was identified, and the 95 percentile flow  $(m^3/s)$  was obtained from the Water Data Unit of the EPA. The flow for each receiving water is classified as high (>10  $m^3/s$ ), medium (1–  $10 \text{ m}^3/\text{s}$ ) or low (<1 m<sup>3</sup>/s). This information is used to assign the risk score for this factor, which increases with decreasing flow. Several of the WWTPs used in the case study discharge into estuarine waters; where there is no estimate of 95 percentile flow available. These locations were thus assigned a risk score value of one, the lowest score available for this factor.

#### 2.6. Validation of the model

In order to validate the developed approach, the values obtained regarding the three RA models were compared to Irish monitoring data of the water downstream from each WWTP. Monitoring data for DCL, E2 and EE2 are not yet legally required in Ireland; thus existing data originate from independent research projects. All such monitoring data of DCL, E2, EE2 or EEQ at Irish WWTPs were compiled during the literature review (Tiedeken et al., 2016). For each publication identified as containing relevant Irish DCL, E2 or EE2 monitoring data, several parameters were extracted. First, the date water samples were taken was identified (including day, month and year if available, but year at minimum). The type of study (i.e. a study measuring the concentration of a compound or a study using effect-based methods) as well as the compound analysed (DCL, E2, EE2 or EEQ) was identified. The method of sampling (grab, passive, or the type of assay utilized) was identified, as was the matrix studied (marine water, lake water, ground water, effluent, etc.). The concentration of the compound recorded during each sampling event was also recorded if it was available. If multiple samples were taken at the same location during a study (i.e. repeat sampling over time), each sampling event was listed separately. However, if multiple analyses were run on the same samples (i.e. tests run in duplicate or triplicate) the individual results were not reported; instead the final value as presented by the authors of the study (usually an average) was utilized. For the concentration data, effect-based studies were ignored and only concentration studies were considered. The remaining concentration data were then sorted by compound, location and value, and for each unique location the highest concentration recorded for each compound was identified for mapping. These concentrations therefore represent the worst case scenario as recorded by monitoring studies to date at each location. It must be noted that this does not mean that these sites will always have such high concentrations of the compounds of interest; additional sampling events may have recorded lower values or non-detects. It should also be noted that just because a compound has not been detected at a site does not mean it will not exceed WFD proposed limits; it is possible (especially for the estrogens) that the LOD may exceed the WFD limit. Nevertheless, these monitoring data still provide an indication of which areas could be pollution "hotspots".

These data were used to test the effectiveness of the RA model developed. The distribution of those data and underpinning the scarcity of such data across Ireland (only a few data on 16 WWTPs) is presented in supplementary materials. This justifies the necessity and the relevance of the RA approach in order to get a full picture of the contamination by the three targeted PhACs at the country scale.

For the purposes of this RA model, monitoring data falls into one of two classes: 1.) PhAC concentrations are below WFD limits or below PNECs, and 2.) PhAC concentrations are above WFD proposed limits or equivalent to or higher than PNEC values. Although sparse, this pool of monitoring data was used to have an idea of the validity of the approach employed. Some of the WWTPs did not have some monitoring data for each PhAC of interest.

Fate of treated effluent factors used in risk assessment model to calculate input four risk score. The colour of the risk score indicates whether there is increased risk (positive values, red), no impact on risk (zero values, blue) or decreased risk (negative values, green).

Factor	Fate factor description	Risk score
1- Type of receiving water	Coastal	1
	Transitional/estuary/river/lake	2
	Stream	3
	Ground	4
2- Proximity to sensitive area	Primary discharge location not at/near sensitive area	0
	Primary discharge location at/near sensitive area	1
3- Flow of receiving water	High (>10 $m^3/s$ )	1
	Medium $(1 - 10 \text{ m}^3/\text{s})$	2
	Low $(<1 \text{ m}^3/\text{s})$	3
Total for input 4		

2.7. Details on the urban wastewater treatment plants used for this Irish case study

Sixteen urban WWTPs were identified for which there were some existing monitoring data on DCL, E2, EE2 or EEQ (see a map in Supplementary materials).

Details on the location and treatment characteristics for each WWTP is presented in Table 7. Coordinates for the 16 Irish urban WWTPs were recorded as the primary discharge point listed in the EPA wastewater license. These included WWTPs of a range of sizes and treatment technologies, which are distributed relatively evenly throughout the country. The evaluation of these 16 WWTPs was used as a case study to test the effectiveness of our semi-quantitative model.

#### 3. Results

#### 3.1. Case study to evaluate RA model using urban WWTPs in Ireland

The required data for each factor in all four inputs of our RA model was compiled for each of the 16 investigated Irish WWTPs, and the cumulative risk score for each was calculated (Table 1) and used to rank the agglomerations from highest to lowest risk of contamination of receiving waters with DCL, E2 and EE2 (Table 8). Seven agglomerations received the highest risk score for DCL, E2 and EE2 namely, Ringsend (East), Leixlip (East), Osberstown (East), Kilkenny (South), Killarney (South), Fermoy (South) and Longford (midlands) (Table 8). These seven plants vary in terms of their spatial distribution; Ringsend, Leixlip and Osberstown are eastern WWTPs, Kilkenny is located in the southeast of the country, Longford is in the midlands and Killarney and Fermoy are in the west. These high scores can be attributed to the following WWTP characteristics: they have no tertiary treatment and no nutrient removal (lack of extended aeration); they have a primary discharge point near/at a sensitive area; and/or they are discharging into receiving waters with relatively low 95 percentile flow. All of these aspects are relevant to the factors throughout input one, two and four, and as a result these WWTPs received high scores in all models corresponding to the three selected PhACs.

Compared with the high-risk agglomerations, the WWTPs with the lowest risk scores demonstrated more variation between the DCL and the E2/EE2 models (Table 8). Nevertheless, Ballincollig, Swords and Tralee consistently received some of the lowest scores in all models corresponding to the three selected PhACs. These three agglomerations received lower overall scores as the process configuration of the plants, i.e. Tralee plant utilizes tertiary treatment and Ballincollig/Swords plants employ extended aeration. These treatment processes lead to decreased levels of PhACs in receiving waters, and thus warrant a reduced risk score via the RA methodology. Tullamore WWTP, even if ranked high risk for DCL, was consistently ranked in the middle of the agglomerations in the E2 and EE2 models. This middle risk ranking is due to a relatively low input one score in all three models, contrasted by a high input four score. The low input one score was a result of the agglomeration not receiving any domestic or industrial inputs, while the high input four score was a result of discharging effluent into a low-flow receiving water (i.e., 95% percentile flow of 0.002  $m^3/s$ ) that is near a sensitive area. The lower cumulative risk score for Swords is due to a relatively low input one score as well as the use of extended aeration, which decreases the cumulative risk score of Swords WWTP further.

For DCL, the maximum cumulative risk score achieved by a WWTP was 19 and the minimum was 14 (Table 8). According to the risk classification scheme defined by our RA model (Table 1), this range spanned two risk classifications, high risk ( $\geq$ 17) and medium risk (9–16). None of the plants from the case study were categorized as low risk for DCL contamination. A similar range in WWTPs scores was observed for E2 (maximum of 20 and minimum of 14); however, all of the agglomerations are classified as medium risk (11–20) for E2 contamination (Table 8). EE2 scores were in general much lower than those for DCL and E2, with a maximum value of 16 and a minimum of 11. Again, all 16 Irish WWTPs were classified as medium risk (9–16) for contamination with EE2 (Table 1).

Even though the risk classifications assigned to WWTPs in this study were always high or medium, the overall distribution of risk scores was sufficient to allow for a meaningful risk ranking of these

Characteristics of the 16 WWTPs used to pilot the risk assessment model and corresponding data. Includes all WWTPs for which published monitoring data on diclofenac, E2, EE2 or estradiols equivalents is available.

WWTP	Size (PE)	Type of secondary	Domestic effluent	Industrial effluent	Gender ratio	Livestock density (ind/ha)		tock density Tertiary 'ha) treatment		WWTP Mo quality dat	Monitoring data	Receiving water	Flow of receiving	Sensitive area?
		treatment (final)				Cattle	Sheep	Pig		test			water (m <sup>3</sup> /s)	
Athlone	21,155	Extended aeration	No	Yes	>1	122	48	45	no	Pass	EEQ	River	18.2	Yes
Ballincollig	27,697	Extended aeration	Yes	No	>1	115	61	26	no	Pass	EEQ	River	2.65	No
Carlow	39,043	Extended aeration	No	Yes	≤1	141	73	34	no	Pass	EEQ	River	4.3	Yes
Clonmel	34,909	Extended aeration	Yes	Yes	≤1	141	73	34	no	Pass	EEQ	River	11.4	Yes
Fermoy	18,608	CAS	Yes	Yes	>1	115	61	26	no	Pass	EEQ	River	6.8	Yes
Galway	213,424	CAS	Yes	No	≤1	68	99	3	no	Pass	DCL	Coastal water	nd	No
Kilkenny	51,988	CAS	Yes	Yes	≤1	141	73	34	no	Pass	EEQ	River	3.72	Yes
Killarney	41,836	CAS	Yes	Yes	>1	115	61	26	no	Pass	EEQ	River	0.057	No
Leixlip	100,309	CAS	Yes	Yes	>1	85	112	12	no	Pass	EEQ, DCL	River	2	Yes
Longford	11,672	CAS	yes	Yes	≤1	122	48	46	no	Pass	EEQ	River	0.348	Yes
Osberstown	104,723	Sequence batch reactor	Yes	Yes	>1	85	112	12	no	Pass	EEQ, DCL, E2, EE2	River	2.6	Yes
Ringsend	2,124,000	Sequence batch reactor	No	Yes	>1	85	112	12	UV (part of the year)	Fail	EEQ, DCL, E2, EE2	Estuarine	nd	Yes
Roscommon	6989	CAS	No	No	≤1	68	99	3	no	Pass	EEQ	River	0.2	Yes
Swords	77,014	Extended aeration	No	No	>1	85	112	12	no	Pass	DCL	Estuarine	nd	Yes
Tralee	35,149	CAS	Yes	Yes	>1	115	61	26	UV (year-round)	Pass	EEQ	Estuarine	nd	Yes
Tullamore	24,055	CAS	No	No	≤1	122	48	46	no	Pass	EEQ	River	0.002	Yes

Table 8
Results of case study evaluating 16 Irish WWTP using the developed risk assessment model for diclofenac, E2 and EE2.

	Input 1			Input 2		Input 3			Input 4			Total			
WWIP	DCL	E2	EE2	DCL	E2	EE2	DCL	E2	EE2	DCL	E2	EE2	DCL	E2	EE2
Athlone	4	11	5	-2	-2	-2	10	4	6	4	4	4	16	17	13
Ballincollig	3	10	4	-2	-2	-2	10	4	6	4	4	4	15	16	12
Carlow	3	7	3	-2	-2	-2	10	4	6	5	5	5	16	14	12
Clonmel	4	11	4	-2	-2	-2	10	4	6	4	4	4	16	17	12
Fermoy	4	11	5	-1	-1	-1	10	4	6	5	5	5	18	19	15
Galway	4	9	4	-1	-1	-1	10	4	6	2	2	2	15	14	11
Kilkenny	5	12	5	-1	-1	-1	10	4	6	5	5	5	19	20	15
Killarney	4	11	5	-1	-1	-1	10	4	6	5	5	5	18	19	15
Leixlip	5	12	6	-1	-1	-1	10	4	6	5	5	5	19	20	16
Longford	4	11	4	-1	-1	-1	10	4	6	6	6	6	19	20	15
Osberstown	5	12	6	-1	-1	-1	10	4	6	5	5	5	19	20	16
Ringsend	4	11	5	0	0	0	10	4	6	3	3	3	17	18	14
Roscommon	2	7	2	-1	-1	-1	10	4	6	6	6	6	17	16	13
Swords	3	10	4	-2	-2	-2	10	4	6	3	3	3	14	15	11
Tralee	4	11	5	-3	-3	-3	10	4	6	3	3	3	14	15	11
Tullamore	2	8	2	-1	-1	-1	10	4	6	6	6	6	17	17	13

WWTPs (Tables 1 and 8). One reason that no agglomerations were classified as low risk was the size of the chosen WWTPs; the smallest plant had a PE of 6989, thus all plants received a relative high risk score for this factor (Tables 2, 7). This is also significant because PhACs contamination of effluents and receiving waters has yet to be considered for the many small WWTPs throughout the country. In addition, several of the plants classed as medium risk by this model could achieve low risk status if they were to employ process configurations known to reduce levels of pharmaceuticals in the effluent. For example, if a tertiary treatment and extended aeration (N removal) during secondary treatment were employed by some plants (i.e. Galway, Swords and Tralee for the EE2 model), they would have received low risk classification. Finally, obtaining higher quality or better resolved data for input into the model would increase the range of the final cumulative risk scores, and could lead to some WWTPs being classified as low risk. Overall, our model for DCL is likely to be a worse predictor of risk than the models for E2 and EE2. Input one, which considers the source of the contaminant being investigated by the RA model, has only three factors for DCL. In contrast, this input has four and seven factors associated with it for EE2 and E2 respectively. Although preliminary, this case study demonstrates the usefulness of this basic, semi-quantitative RA model for determining the relative risks posed by WWTPs in regards to environmental consequences of contamination with the PhACs of interest. It also identifies seven WWTPs with particularly high scores in regards to DCL, E2 and EE2 contamination (Table 8). Future monitoring studies should be considered including these WWTPs in order to better evaluate the levels of DCL, E2 and EE2 in their treated effluent/receiving waters. This kind of methodology allows the highest risk WWTPs to be prioritized for improvements such as the implementation of extended aeration and/or tertiary treatment in order to mitigate the impacts of PhACs pollution in the aquatic environment. It is essential, however, that this model be improved and additional sources of data be utilized in order to provide stakeholders with the most relevant and realistic RA possible. The following discussion section identifies areas for improvement which would increase the usefulness of this RA model or future models used for environmental risks regarding contamination with PhACs.

#### 3.2. Validation of the model

With such a semi-quantitative RA model, a quantitative validation would be a non-sense, especially with the low number of Irish monitoring data found (both WWTPs efficiency and corresponding downstream surface water). This is the reason why the following validation will only be assessed qualitatively.

Existing monitoring data from 5 of the 16 WWTPs included in this study (i.e. Ringsend, Swords, Galway, Leixlip and Osberstown), indicate that DCL is found in treated effluents as in other European WWTPs (Thomas et al., 2007; Kümmerer, 2009b; Wille et al., 2012). Furthermore, the level of accuracy of the analytical methods employed for the analysis of DCL was generally low, with LODs in the µg/L range (Lacey et al., 2008; Lacey et al., 2012). As a consequence, the monitoring data for Leixlip, Osberstown and Swords revealed only concentrations below LODs (Thomas et al., 2007; Kümmerer, 2009b; Wille et al., 2012) that limits the possible interpretation or correlation study with the scores from the present RA model. No data on concentrations of DCL in surface water downstream from the studied WWTPs were found. A comparison between the results from our RA model and existing DCL monitoring data is difficult as DCL levels in WWTPs effluent are unknown for the remaining eleven WWTPs included in this study (Table 7).

In regards to E2 and EE2, existing monitoring data are more complete (Table 7); measures of EEQ in treated effluent exist for all but two of the 16 WWTPs included in this case study (Swords and Galway) (Table 7). These monitoring data, however, were generated in different studies with different methods applied and are therefore difficult to compare; this is why we focused only on comparison of results from the same study where data were generated in the same condition (sampling, methodology, analytical methods, duration, etc). One study focused on the comparison of estrogenic responses upstream and downstream from two WWTPs considered in the present study, Osberstown and Ballincollig, by performing vitellogenin assays on wild brown trout (Tarrant et al., 2005). For Osberstown, it was demonstrated an impact of the WWTP and an evidence of estrogenic exposure of the brown trout was observed by a significant increase in plasma vitellogenin concentration from upstream to downstream from the plant. For Ballincollig, no impact of the WWTP on the fish vitellogenin concentration was observed and the authors conclude to an absence of estrogenic exposure that could potentially occur downstream from the plant (Tarrant et al., 2005). The respective E2/EE2 scores from the RA model were 20/16 and 16/12 for Osberstown and Ballincollig respectively and Osberstown plant was therefore given a higher score than Ballincollig one. In the same study, some EEQ values from downstream other WWTPs (e.g. Leixlip, Clonmel, Fermoy) were also given but the range obtained (1-3 ng/L) was deemed too narrow to be used in the present study as a qualitative validation tool. Another study quantified the EEQ for different locations situated downstream from some Irish WWTPs during a 2-years period using grab samples (Quinn-Hosey et al., 2012). The mean EEO values obtained ranged from 2.7 ng/L to 16.2 ng/L for Roscommon and Longford respectively; the corresponding E2/EE2 scores for these WWTPs were 16/13 and 20/15 respectively, showing a correlation between monitoring and RA model data. These observations, even qualitative, confirm the potential of such RA model to assess which WWTP could be considered as hotspot for the release of the targeted PhACs, especially in cases where monitoring data are scarce as it is the case for most of EU countries (Tiedeken et al., 2017).

#### 4. Discussion

This model allows for a preliminary RA evaluating the environmental threat posed by Irish WWTPs in regards to PhACs contamination of aquatic ecosystems. The case study results demonstrate that some improvements should be made before this model is fully implemented on a national/European scale, or used to make decisions regarding regulations or infrastructure and paving the way to the development of a fully quantitative model. Firstly, the risk classification system used in this model could be adapted. Currently, the three risk categories (high, medium and low) were split equally based on the highest possible score for each PhAC (Table 1). Redefining the risk categories so that the threshold for a "low risk" categorization is increased could be appropriate. Before taking this step, however, additional reliable studies evaluating the concentrations of the three selected PhACs and their main metabolites in WWTPs effluents and receiving waters should be carried out. How such monitoring data correlate with the cumulative risk scores from this model, and how they compare to PNEC and WFD proposed limits, will provide the best information on how to define the risk classifications. This would provide a reliable dataset that could be used as a real validation dataset for the future developments of the current RA model. To this aim, best practices in terms of analytical methods and sampling should be employed as it was noticed that most of Irish based studies found were still using non-representative grab samples and analytical methods not sensitive enough to quantify or even detect PhACs at environmental concentrations. Secondly, the same risk classification system could have been applied to all three drugs (i.e. DCL, E2 and EE2). For example, when scoring for DCL we chose to exclude factors in input one that were deemed irrelevant to this particular PhAC (i.e. livestock densities and gender ratios). Instead, these irrelevant factors could have received a risk score of zero. This alternative method assumes that if one compound has an additional source, it should be higher risk than compounds with fewer sources. Such a methodology would allow for a comparison between compounds at each WWTP, however it was deemed inappropriate due to lack of data. The best solution to this issue is to obtain drug usage data instead of relying on proxies (e.g. gender ratios) for information on compound consumption.

An additional change that could be applied to this RA model is the weighting of certain critical risk factors or inputs. For example, whether or not a plant utilizes a tertiary treatment will have a substantial impact on the removal efficiencies of DCL, E2 and EE2 during treatment (Joss et al., 2006; Boxall, 2010). It is thus possible that this factor should be considered to a greater extent than the rest of the risk factors, or that the input that this factor contributes to (input two) should be weighted. Because this was a preliminary study, we left all inputs and factors unweighted, but future iterations should consider the impact weighting could have on the model's predictive ability. In addition to structural changes to the existing RA model, the data sources for most of the factors could be improved, and additional data sources could be added as factors to relevant inputs. First, input one (source of the influent) is lacking drug utilization data, arguably one of the most important inputs to a RA model for any prescription drug (Oaks et al., 2004; Santos et al., 2010). Drug utilization data for DCL, E2 and EE2 are available for Ireland from the Health Service Executive (HSE) via the Primary Care Reimbursement Services (PCRS) sector, but is incomplete as approximately 80% and 60% are missing for DCL and estradiol respectively. Furthermore, these drug utilization data should be completed by data on legal or illegal internet purchase that could represent a substantial addition to the overall PhAC consumption. Thus, although drug utilization is important to include in future models, data quality for these three compounds in Ireland is imperfect and other relevant factors will need to be considered in order to ensure the predictive power of an RA model. In particular, the use of AGL as a factor in input one should not be abandoned, even if drug utilization is incorporated into future models (EPA, 2010; Boxall et al., 2014).

Factors two and three from input one (the addition of domestic or industrial septic tank sludge/effluent) would benefit from higher quality data. In regards to industrial effluent and sludge, the type of industry should be taken into account. Sludge that is likely to contain the PhACs of interest (e.g. from a pharmaceutical manufacturer or a hospital) should be assigned a higher risk score. Hospitals can be significant contributors to influent loads of PhACs (Fent et al., 2006; Kümmerer, 2009b; Boxall, 2010), thus an additional factor that could be added to input one of the RA model is data on the distribution of hospitals throughout the country. This factor should consider not just the location of hospitals, but also whether or not they have independent discharge licenses, whether they treat effluent before release, or whether any discharges are trucked to WWTPs. For the fifth, six and seventh factors in input one (livestock densities), data that were freely available from the CSO's online database StatBank were used. Additional factors could be considered regarding livestock, including whether sewage sludge/slurry spreading or slurry/dung stores occur in the catchment (EPA, 2010).

In regards to input two, removal of compounds due to treatment, there are many additional process parameters and configuration variables that could be considered in order to improve this RA model. At minimum, the solids retention time (SRT) and hydraulic residence time (HRT) utilized at each plant should be taken into consideration. SRT especially may impact PhACs removal because a higher value leads to increased microbial diversity, which could increase the metabolising and transforming capabilities of the sludge in general (Clara et al., 2005; Kümmerer, 2009b). An increased HRT increases the contact time of water with the biomass, potentially leading to higher removal efficiencies of PhACs (Gros et al., 2010; Vieno and Sillanpää, 2014). These variables are specific to each WWTP and can vary over time, thus information would need to be gathered through WWTPs staff and facilitated by the corresponding public authority.

In future RA models, input four (fate of the effluent) could consider secondary in addition to primary discharge points. In addition, only the dilution capacity of the receiving water was considered in the present model but the self-purification capacity of the receiving water should be taken into account considering that features such dissolved oxygen content or depth could impact the removal of PhACs once discharged through a river/lake/sea.

One general drawback of the current RA model is the scale of the data for many of the factors included in the various inputs. For example, gender ratio was calculated at the county level, and livestock densities were calculated for each regional authority. Ideally, all factors would consider data based on the population or land area served by the WWTP, instead of at the county or regional authority level. Gender ratio would thus consider the people living within the area serviced by a WWTP, and livestock density would be calculated based on the catchment in which the primary discharge point falls. These types of data manipulation are possible but time consuming, and often require permission from various national authorities. Future models should consider fully quantitative approach that is validated using standardized monitoring data (Oaks et al., 2004; Petrie et al., 2013; Rivera-Utrilla et al., 2013) and to consider potential ecotoxicological and biodiversity decline. In this perspective, potential mixture effect should be considered. This RA model should also be adapted to consider climate change such as large uncertainties associated with projections of precipitation variations at the river catchment scale reflecting the need to integrate hydrological data derived from application of catchment models (Bastola et al., 2011).

#### 5. Conclusion

A preliminary semi-quantitative RA model aiming at identifying WFD watch-list compounds pollution hotspots was developed. This model is based on a scoring system that encompasses different inputs that could potentially impact the presence of the three first watch list compounds in WWTPs receiving waters. The model focuses on (i) an estimation of the load of each PhAC in WWTPs influent, (ii) WWTPs and their potential for PhACs elimination from the water phase, (iii) on physico-chemical characteristics of the compounds that could impact this removal and (iv) on the dilution potential of the receiving water. Depending of the different scores obtained, a panel of 16 Irish WWTPs was ranked as low, medium or high risk for the three targeted compounds. The approach was tested using actual monitoring data found for the considered WWTPs; unfortunately, the low number of monitoring data found and their relative unreliability prevented from an actual validation of the RA model. However, the present work, although preliminary, is paving the way through the future development of a quantitative model. Indeed, some recommendations and potential improvements that could be used to strengthen the approach an being able to identify which WWTPs would necessitate some improvements as regards to the application of the WFD and specifically to reduce the presence of watch list PhACs in their receiving waters.

#### Acknowledgements

The authors express gratitude to Ireland's Environmental Protection Agency STRIVE Programme for supporting this study (Project 2014-W-DS-18).

#### Appendix A. Supplementary data

Supplementary data to this article can be found online at http://dx. doi.org/10.1016/j.scitotenv.2017.05.227.

#### References

- Barbosa, M.O., Moreira, N.F.F., Ribeiro, A.R., Pereira, M.F.R., Silva, A.M.T., 2016. Occurrence and removal of organic micropollutants: an overview of the watch list of EU decision 2015/495. Water Res. 94, 257–279.
- Bastola, S., Murphy, C., Sweeney, J., 2011. The sensitivity of fluvial flood risk in Irish catchments to the range of IPCC AR4 climate change scenarios. Sci. Total Environ. 409, 5403–5415.
- Ben Fredj, S., Nobbs, J., Tizaoui, C., Monser, L., 2015. Removal of estrone (E1),  $17\beta$ -estradiol (E2), and  $17\alpha$ -ethinylestradiol (EE2) from wastewater by liquid–liquid extraction. Chem. Eng. J. 262, 417–426.
- Boxall, A.A., 2010. Veterinary medicines and the environment. Comparative and Veterinary Pharmacology. Springer, Berlin Heidelberg, pp. 291–314.
- Boxall, A.B.A., Keller, V.D.J., Straub, J.O., Monteiro, S.C., Fussell, R., Williams, R.J., 2014. Exploiting monitoring data in environmental exposure modelling and risk assessment of pharmaceuticals. Environ. Int. 73, 176–185.

Buchberger, W.W., 2011. Current approaches to trace analysis of pharmaceuticals and personal care products in the environment. J. Chromatogr. A 1218, 603–618.

- Burkhardt-Holm, P., 2010. Endocrine disruptors and water quality: a state-of-the-art review. Int. J. Water Qual. 16, 477–493.
- Cirja, M., Ivashechkin, P., Schäffer, A., Corvini, P.F.X., 2008. Factors affecting the removal of organic micropollutants from wastewater in conventional treatment plants (CTP) and membrane bioreactors (MBR). Rev. Environ. Sci. Biotechnol. 7, 61–78.
- Clara, M., Strenn, B., Gans, O., Martinez, E., Kreuzinger, N., Kroiss, H., 2005. Removal of selected pharmaceuticals, fragrances and endocrine disrupting compounds in a membrane bioreactor and conventional wastewater treatment plants. Water Res. 39, 4797–4807.
- EPA, 2010. Environmental Protection Agency. A Handbook on the Implementation of the Regulations for Water Service Authorities for Public Water Supplies. Water Services Authorities, Ireland, pp. 1–62.
- EPA, 2015. Urban Waste Water Treatment in 2014. Environmental Protection Agency, Ireland, pp. 1–79.
- European Medicines Agency, 2014. Opinion of the Committee for Medicinal Products for Veterinary Use Pursuant to Article 30(3) of Regulation (EC) No 726/2004.
- Fent, K., Weston, A.A., Caminada, D., 2006. Ecotoxicology of human pharmaceuticals. Aquat. Toxicol. 76, 122–159.
- Gill, L., Misstear, B., Johnston, P., O' Luanaigh, N., 2009. Natural attenuation of endocrine disrupting chemicals in on-site domestic wastewater treatment systems. J. ASTM Int. 6, 1–13.
- Ginebreda, A., Muñoz, I., de Alda, M.L., Brix, R., López-Doval, J., Barceló, D., 2010. Environmental risk assessment of pharmaceuticals in rivers: relationships between hazard indexes and aquatic macroinvertebrate diversity indexes in the Llobregat River (NE Spain). Environ. Int. 36, 153–162.
- de Graaff, M.S., Vieno, N.M., Kujawa-Roeleveld, K., Zeeman, G., Temmink, H., Buisman, C.J.N., 2011. Fate of hormones and pharmaceuticals during combined anaerobic treatment and nitrogen removal by partial nitritation-anammox in vacuum collected black water. Water Res. 45, 375–383.
- Gros, M., Petrović, M., Ginebreda, A., Barceló, D., 2010. Removal of pharmaceuticals during wastewater treatment and environmental risk assessment using hazard indexes. Environ. Int. 36, 15–26.
- Guillén, D., Ginebreda, A., Farré, M., Darbra, R.M., Petrovic, M., Gros, M., Barceló, D., 2012. Prioritization of chemicals in the aquatic environment based on risk assessment: analytical, modeling and regulatory perspective. Sci. Total Environ. 440, 236–252.
- Hecker, M., Hollert, H., 2011. Endocrine disruptor screening: regulatory perspectives and needs. Europe]–>Environ. Sci. Eur. 23, 1–14.
- Hernando, M.D., Mezcua, M., Fernández-Alba, A.R., Barceló, D., 2006. Environmental risk assessment of pharmaceutical residues in wastewater effluents, surface waters and sediments. Talanta 69, 334–342.
- Joss, A., Zabczynski, S., Göbel, A., Hoffmann, B., Löffler, D., McArdell, C.S., Ternes, T.A., Thomsen, A., Siegrist, H., 2006. Biological degradation of pharmaceuticals in municipal wastewater treatment: proposing a classification scheme. Water Res. 40, 1686–1696.
- Kosma, C.I., Lambropoulou, D.A., Albanis, T.A., 2014. Investigation of PPCPs in wastewater treatment plants in Greece: occurrence, removal and environmental risk assessment. Sci. Total Environ. 466–467, 421–438.
- Kümmerer, K., 2009a. The presence of pharmaceuticals in the environment due to human use – present knowledge and future challenges. J. Environ. Manag. 90, 2354–2366.
- Kümmerer, K., 2009b. The presence of pharmaceuticals in the environment due to human use – present knowledge and future challenges. J. Environ. Manag. 90, 2354–2366.
- Kunz, P.Y., Kienle, C., Carere, M., Homazava, N., Kase, R., 2015. In vitro bioassays to screen for endocrine active pharmaceuticals in surface and waste waters. J. Pharm. Biomed. Anal. 106, 107–115.
- ter Laak, T.L., van der Aa, M., Houtman, C.J., Stoks, P.G., van Wezel, A.P., 2010. Relating environmental concentrations of pharmaceuticals to consumption: a mass balance approach for the river Rhine. Environ. Int. 36, 403–409.
- Lacey, C., McMahon, G., Bones, J., Barron, L., Morrissey, A., Tobin, J.M., 2008. An LC–MS method for the determination of pharmaceutical compounds in wastewater treatment plant influent and effluent samples. Talanta 75, 1089–1097.
- Lacey, C., Basha, S., Morrissey, A., Tobin, J., 2012. Occurrence of pharmaceutical compounds in wastewater process streams in Dublin, Ireland. Environ. Monit. Assess. 184, 1049–1062.
- Lapworth, D.J., Baran, N., Stuart, M.E., Ward, R.S., 2012. Emerging organic contaminants in groundwater: a review of sources, fate and occurrence. Environ. Pollut. 163, 287–303.
- Lienert, J., Bürki, T., Escher, B.I., 2007. Reducing micropollutants with source control: substance flow analysis of 212 pharmaceuticals in faeces and urine. Water Sci. Technol. 56, 87.
- Lindim, C., van Gils, J., Cousins, I.T., Kühne, R., Georgieva, D., Kutsarova, S., Mekenyan, O., 2017. Model-predicted occurrence of multiple pharmaceuticals in Swedish surface waters and their flushing to the Baltic Sea. Environ. Pollut. 223, 595–604.
- Miege, C., Choubert, J.M., Ribeiro, L., Eusebe, M., Coquery, M., 2008. Removal efficiency of pharmaceuticals and personal care products with varying wastewater treatment processes and operating conditions – conception of a database and first results. Water Sci. Technol. 57, 49–56.
- Nikolaou, A., Meric, S., Fatta, D., 2007. Occurence patterns of pharmaceuticals in water and wastewater environments. Anal. Bioanal. Chem. 387, 1225–1234.
- Oaks, J.L., Gilbert, M., Virani, M.Z., Watson, R.T., Meteyer, C.U., Rideout, B.A., Shivaprasad, H.L., Ahmed, S., Chaudhry, M.J.I., Arshad, M., Mahmood, S., Ali, A., Khan, A.A., 2004. Diclofenac residues as the cause of vulture population decline in Pakistan. Nature 427, 630–633.
- Pereira, A.M.P.T., Silva, L.J.G., Meisel, L.M., Lino, C.M., Pena, A., 2015. Environmental impact of pharmaceuticals from Portuguese wastewaters: geographical and seasonal occurrence, removal and risk assessment. Environ. Res. 136, 108–119.

- Petrie, B., McAdam, E.J., Scrimshaw, M.D., Lester, J.N., Cartmell, E., 2013. Fate of drugs during wastewater treatment. TrAC Trends Anal. Chem. 49, 145–159.
- Qian, F., He, M., Song, Y., Tysklind, M., Wu, J., 2015. A bibliometric analysis of global research progress on pharmaceutical wastewater treatment during 1994–2013. Environ. Earth Sci. 73, 4995–5005.
- Quinn-Hosey, K.M., Roche, J.J., Fogarty, A.M., Brougham, C.A., 2012. A toxicological assessment of endocrine disrupting chemicals found in BMW (border, midland and western) region of Ireland. J. Environ. Prot. 3, 304–315.
- Rivera-Utrilla, J., Sánchez-Polo, M., Ferro-García, M.Á., Prados-Joya, G., Ocampo-Pérez, R., 2013. Pharmaceuticals as emerging contaminants and their removal from water. A review. Chemosphere 93, 1268–1287.
- Ronan, J.M., McHugh, B., 2013. A sensitive liquid chromatography/tandem mass spectrometry method for the determination of natural and synthetic steroid estrogens in seawater and marine biota, with a focus on proposed water framework directive environmental quality standards. Rapid Commun. Mass Spectrom. 27, 738–746.
- Santos, L.H.M.L.M., Araújo, A.N., Fachini, A., Pena, A., Delerue-Matos, C., Montenegro, M.C.B.S.M., 2010. Ecotoxicological aspects related to the presence of pharmaceuticals in the aquatic environment. J. Hazard. Mater. 175, 45–95.
- Simon, E., Lamoree, M.H., Hamers, T., de Boer, J., 2015. Challenges in effect-directed analysis with a focus on biological samples. TrAC Trends Anal. Chem. 67, 179–191.
- Streck, G., 2009. Chemical and biological analysis of estrogenic, progestagenic and androgenic steroids in the environment. TrAC Trends Anal. Chem. 28, 635–652.
- Tarrant, H., Llewellyn, N., Lyons, A., Tattersall, N., Wylde, S., Mouzakitis, G., Maloney, M., McKenzie, C., 2005. Endocrine Dirsruptors in the Irish Aquatic Environment. Environmental Protection Agency, Dublin, Ireland, pp. 1–191.
- Thomas, K.V., Dye, C., Schlabach, M., Langford, K.H., 2007. Source to sink tracking of selected human pharmaceuticals from two Oslo city hospitals and a wastewater treatment works. J. Environ. Monit. 9, 1410–1418.
- Tiedeken, E.J., Clifford, E., Rowan, N., 2016. Investigation of the implications for Ireland of emerging standards on pharmaceuticals in receiving waters. EPA Research Report.

- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – a 20 year systematic review. Sci. Total Environ. 574, 1140–1163.
- UWWTD-REP Working Group, 2006. Terms and Definitions of the Urban Waste Water Treatment Directive 91/271/EEC. The Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, Brussels, Belgium.
- Verlicchi, P., Zambello, E., 2016. Predicted and measured concentrations of pharmaceuticals in hospital effluents. Examination of the strengths and weaknesses of the two approaches through the analysis of a case study. Sci. Total Environ. 565, 82–94.
- Verlicchi, P., Al Aukidy, M., Zambello, E., 2012a. Occurrence of pharmaceutical compounds in urban wastewater: removal, mass load and environmental risk after a secondary treatment—a review. Sci. Total Environ. 429, 123–155.
- Verlicchi, P., Al Aukidy, M., Zambello, E., 2012b. Occurrence of pharmaceutical compounds in urban wastewater: removal, mass load and environmental risk after a secondary treatment—a review. Sci. Total Environ. 429, 123–155.
- Vieno, N., Sillanpää, M., 2014. Fate of diclofenac in municipal wastewater treatment plant – a review. Environ. Int. 69, 28–39.
- Wille, K., De Brabander, H.F., Vanhaecke, L, De Wulf, E., Van Caeter, P., Janssen, C.R., 2012. Coupled chromatographic and mass-spectrometric techniques for the analysis of emerging pollutants in the aquatic environment. TrAC Trends Anal. Chem. 35, 87–108.
- Wise, A., O'Brien, K., Woodruff, T., 2011. Are oral contraceptives a significant contributor to the estrogenicity of drinking water? Environ. Sci. Technol. 45, 51–60.
- Zhang, Y., Geißen, S.-U., Gal, C., 2008. Carbamazepine and diclofenac: removal in wastewater treatment plants and occurrence in water bodies. Chemosphere 73, 1151–1161.
- Ziylan, A., Ince, N.H., 2011. The occurrence and fate of anti-inflammatory and analgesic pharmaceuticals in sewage and fresh water: treatability by conventional and nonconventional processes. J. Hazard. Mater. 187, 24–36.

Contents lists available at ScienceDirect



Science of the Total Environment



journal homepage: www.elsevier.com/locate/scitotenv

#### Review

## Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – A first overview of the situation in the Republic of Ireland



Alexandre Tahar<sup>a</sup>, Erin Jo Tiedeken<sup>a,b</sup>, Neil J. Rowan<sup>a,\*</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co. Westmeath, Ireland <sup>b</sup> School of Science, National College of New Jersey, Pennington Road Ewing, NJ 08628-0718, USA

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- Occurrence and mapping of 3 EU Watch list substances in Irish aquatic environment
- Lack of monitored data for CECs given number of river basin catchments and control points
- Need for new analytical techniques with low appropriate levels of detection to meet WFD limits
- Control measures frequently do not fully remove these harmful chemicals.
- Mapping of CECs will strategically inform future upgrades to important control points.

#### ARTICLE INFO

Article history: Received 2 August 2017 Received in revised form 30 October 2017 Accepted 2 November 2017 Available online xxxx

#### Editor: D. Barcelo

Keywords: EU watch-list substances Diclofenac Estrogens Occurrence Monitoring Receiving water Control points Republic of Ireland



#### ABSTRACT

This constitutes the first study to address occurrence and geodatabase mapping of the anti-inflammatory drug diclofenac (DCL) and the natural (17-beta-estradiol or E2) and synthetic (17-alpha-ethynylestradiol or EE2) estrogenic hormones in Republic of Ireland receiving waters over the period 1999 to 2015. Among these data, 317 samples came from concentration studies, while 205 were from effect-based studies. Monitoring data came from 16 waste water treatment plants (WWTPs), 23 water bodies (including rivers, lakes, marine and transitional waters) and 7 from domestic locations. Out of approximately 1000 WWPTs in the Republic of Ireland, only 16 have been monitored for at least one of these compounds of emerging concern (CECs). Diclofenac is found in treated effluents from 5 WWTPs at levels at least as high as other European WWPTs, and sometime higher. Measurements of E2 and EE2 in WWPT effluents were rare and effluents were more often evaluated for total estrogens; these CECs were generally not detected using conventional analytical methods because of limits of detection being too high compared to environmental concentrations and WFD environmental quality standards. There was good agreement between occurrence of these CEC and regional drug dispensing data in Ireland. Mapping the aforementioned data onto appropriate river basin catchment management tools will inform predictive and simulated risk determinations to inform investment in infrastructure that is necessary to protect rivers and beaches and economic activities that rely on clean water. There is a pressing commensurate need to refine/develop new analytical methods with low levels of detection for future CEC intervention.

© 2017 Elsevier B.V. All rights reserved.

\* Corresponding author.

E-mail address: nrowan@ait.ie (N.J. Rowan).

https://doi.org/10.1016/j.scitotenv.2017.11.021 0048-9697/© 2017 Elsevier B.V. All rights reserved. Contents

1.	Introd	uction .		188									
2.	Materials and methods												
3.	Results and discussion												
	3.1.	GIS ma	pping – an overview of Irish DCL, E2 and EE2 data (1999–2015)	189									
	3.2.	Monito	vring studies	190									
		3.2.1.	WWTPs monitoring and locations in Republic of Ireland	190									
		3.2.2.	Highest monitored concentrations in Republic of Ireland for the 3 targeted compounds	192									
		3.2.3.	Wastewater treatment plants and receiving waters monitoring studies	193									
		3.2.4.	Monitoring of on-site non collective treatment systems	195									
		3.2.5.	Studies on alternative control measures	195									
		3.2.6.	Comparison of Irish and EU monitoring data	195									
	3.3.	Relation	nship with drug utilising dispensing data	196									
	3.4.	Conclus	sion and perspectives.	196									
				197									
Apper	ndix A.	Supp	lementary data.	197									
Refe	rences.			197									

#### 1. Introduction

Pollution of European receiving waters containing pharmaceuticals is a ubiquitous phenomenon (Verlicchi and Zambello, 2016; Barbosa et al., 2016; Tiedeken et al., 2017). Until recently environmental regulations worldwide had not required explicit testing of these contaminants of emerging concern in water bodies. However, given concern about contamination of aquatic environment with these substances, legislation such as the Water Framework Directive (WFD) and the Environmental Quality Standards Directive (EQSD) at a European level and associated legislation at a local level has recently begun to acknowledge this problem (Tiedeken et al., 2017; Tahar et al., 2017) The identification of these contaminants and associated analytical methods may inform pressure points and efficacy of appropriate interventions for consideration in future WFD-monitoring programmes and regulations. Pharmaceuticals are a class of emerging environmental contaminants that are widely used in human and veterinary medicine (Tahar et al., 2017). From here on, these substances of emerging concern will be referred to as pharmaceutically active chemicals (PhACs) that includes not just pharmaceuticals but also their pharmaceutically active metabolites/ transformation products. This research is important because of the potential toxic effects for aquatic biota and human health that may result from chronic exposure to PhACs (Streck, 2009; Kümmerer, 2009; Kosma et al., 2014). PhACs exhibit wide variation in function, chemical structure and physiochemical properties, making it difficult to generalize about their behaviour, persistence or impact in the environment. PhACs are also designed to be biologically active, have a specific mode of action and to be persistent in the body, meaning they can impact humans and wildlife at trace concentrations which are often hard to detect and quantify using traditional analytical methods (Kosma et al., 2014). A large number of PhACs have been detected in wastewater treatment plants (WWTP) influents and effluents and surface, ground and drinking water worldwide in recent years (Cirja et al., 2008; Streck, 2009; Zhou et al., 2009; Verlicchi and Zambello, 2016). It is now established that throughout the developed world, PhACs are ubiquitous at µg to ng per litre levels in the aquatic environment (Streck, 2009). The impacts of chronic exposure to trace concentrations of many PhACs on wildlife and human health may be severe (Verlicchi et al., 2012); thus, it is critical to limit as much as possible the concentrations of this class of contaminants in our waterways.

Until recently, environmental regulations worldwide had not required explicit testing for any PhACs in water bodies. However given the growing concern about contamination of the aquatic environment with these compounds, legislation has recently begun to acknowledge this potential problem. The WFD requires that all EU member states prepare river basin management plans (RBMPs) to address the many issues relating to water quality and protection in a holistic manner. In response to growing EU concern about the release of untreated PhACs into the aquatic environment, three compounds were included on in the first EU watch list in 2013: diclofenac (CAS# 15307-79-6, hereafter referred as DCL), 17beta-estradiol (CAS# 50-28-2, hereafter referred as E2) and 17-alphaethinylestradiol (CAS# 57-63-6, hereafter referred as EE2) (Barbosa et al., 2016). Annual average environmental quality standard (AA-EQS) were defined for these 3 compounds as being the concentrations defining the boundary between good and moderate WFD status. The respective AA EQS in surface water for DCL, E2 and EE2 are 100 ng/L, 0.035 ng/L and 0.4 ng/L EE2 and E2 can impact the endocrine system of humans or wildlife (Verlicchi et al., 2012). There are growing fears that chronic exposure to these endocrine disrupting chemicals or EDCs (in bathing or drinking water, for example) may be linked to adverse human health conditions such as declining male fertility, birth defects, and breast and testicular cancer. Similar to PhACs as a whole, EDCs are mainly thought to be transported into the aquatic environment via incomplete removal at WWTPs (Streck, 2009). It is relevant to note that the European Commission implemented decision 495 of 20 March 2015 that expanded substances or groups of substances on the watch list to 10 in the field of water policy (Barbosa et al., 2016). Also, following the timetable for the common implementation of the WFD, the first management cycle ended in 2015, and the second river basin management plan combined with the first flood management plan is due to end in 2021.

A systematic review of these three first EU watch list PhACs in receiving waters was recently published, which reviewed 3945 potentially relevant articles over period 1995 to 2015 publications on uses, sources, monitoring and control measures to produce a EU-wide database (Tiedeken et al., 2017). Overall, European surface water concentrations of DCL are typically below reported annual proposed AA EQS of 100 ng/L, but exceedances frequently occur. E2 and EE2 surface water concentrations are typically below 50 ng/L and 10 ng/L respectively, but these values greatly exceed the proposed AA EQS values for these compounds (0.4 and 0.035 ng/L respectively). However, levels of these PhACs are frequently reported to be disproportionately high in EU receiving waters, particularly in effluents at control points that require urgent attention. In addition, the EPA reported in October 2017 that in 42 locations in the Republic of Ireland, sewage is discharged untreated, putting rivers and bathing areas at risk of pollution: 44 of 170 large urban areas did not comply with EU water quality standards (EPA, 2017). The review of Tiedeken et al. (2017) highlighted that there is a pressing need to conduct detailed mapping of the occurrences and control measures for CECs at a national scale that provides a platform for EU orientation so as to inform policy and decision-making on improving and protecting water quality. Thus, the aim of this case study was to evaluate best-published data on these three EU Watch list PhACs so as to geographically map their occurrence in Irish receiving waters and in effluent at WWTPs and to compare with regional drug dispensing data.

#### 2. Materials and methods

For each publication identified as containing relevant Irish DCL, E2 or EE2 monitoring data, several other parameters were extracted for use in mapping. First, the date water samples were taken was identified (including day, month and year if available, but year at minimum). The type of study (i.e. a study measuring the concentration of a compound or a study using effect-based methods) as well as the compound analysed (DCL, E2, EE2 or estradiols equivalents, EEQ) was identified. The method of sampling (grab, composite, passive, or any type of assay utilised) was identified, as was the matrix studied (marine water, lake water, ground water, WWTP influent/effluent). The name and GPS coordinated of the specific location where the sampling took place as well as the county was listed. If the sampling location was a WWTP, its coordinates were recorded as the primary discharge point listed in the EPA wastewater license (http://www.epa.ie/terminalfour/ wwda/index.jsp?disclaimer=yes&Submit=Continue#.VpPcJPmLTIX). The concentration (in ng/L) of the compound recorded during each sampling event was also recorded if it was available. If multiple samples were taken at the same location during a study (i.e. repeat sampling over time), each sampling event was listed separately. However, if multiple analyses were run on the same samples (i.e. tests run in duplicate or triplicate) the individual results were not reported; instead the final value as presented by the authors of the study (usually an average) was utilised. The publication (reference) associated with each data point was recorded. Frequently some of the data described above were not available from the publications, in which case corresponding authors were emailed or otherwise contacted and a data request was made.

Two aspects of these data were mapped for this report, (i) the distribution of the sampling events (or sampling effort) for each compound and (ii) the highest recorded concentrations of each compound at a sampling location. In order to map sampling events, the data were divided into three 5-6 year intervals: 1999-2004, 2005-2009, and 2010–2015. Results for samples that were taken at the same location, analysed for the same compound and sampled during the same interval were consolidated, and the number of sampling events at each location was specified. This allowed the sampling effort at each location to be mapped for each compound of interest. For the concentration data, effect-based studies were ignored. For some of these studies, the concentration values were not available. This was because either the study reported only presence/absence of the compound or reported average values instead of raw data. In the latter case, authors were contacted to attempt to obtain raw data values, however a small number of authors did not respond to data requests. These studies were thus excluded from concentration mapping. The remaining concentration data were then sorted by compound, location and value, and for each unique location the highest concentration recorded for each compound was identified for mapping. These concentration maps therefore represent the worst case scenario as recorded by monitoring studies to date at each location. It must be noted that this does not mean that these sites will always have such high concentrations of the compounds of interest; additional sampling events may have recorded lower values or non-detects. It should also be noted that just because a compound has not been detected at a site does not mean it will not exceed WFD proposed limits (EQS); it is possible (especially for the estrogens) that the limit of detection (LOD) may exceed the EQS limit. Nevertheless, these maps still provide an indication of which areas could be pollution "hotspots" based on currently available monitoring data.

Data were mapped using ArcGIS Desktop software (Arc Catalog and ArcMap 10.3.1, Environmental Systems Research Institute, via an ESRI single-user, one-year license). Additional data used to create the file geodatabase were downloaded from the Central Statistics Office database "StatBank Ireland," (http://www.cso.ie/px/pxeirestat/statire/ SelectTable/Omrade0.asp?Planguage=0, utilised for county bound-aries, city locations and population density) and the EPA's Geo Portal (http://gis.epa.ie/GetData/Download, utilised for river basin catchments, WFD river basin districts, WWTP locations and attribute data, and WFD protected areas). A map of the Republic of Ireland river basin catchment is provided in Supplementary materials (Fig. S1).

Reported drug utilisation or dispensing data is an important source of information on the quantities of PhACs that ultimately are released into the environment (Boxall, 2010; Rowan, 2011). Therefore, with the objective to correlate the concentration data gathered and to provide a proxy for the region where no environmental data were found, drug utilising dispensing data was provided by Ireland's Health Service Executive (HSE) for the 32 Local Health Offices (LHOs) for DCL and E2. The total volume (kg) of DCL and E2 dispensed to patients in all LHOs from 2009 to 2012 was provided, however data for years 2008 and 2013 were excluded as these periods were not full calendar years. EE2 is often prescribed as a combination drug, thus obtaining and summarizing dispensing data is more difficult than for DCL and E2 and was therefore not included in this study.

#### 3. Results and discussion

#### 3.1. GIS mapping – an overview of Irish DCL, E2 and EE2 data (1999–2015)

From the literature review performed, a total of 522 unique Irish monitoring data points were identified for DCL, E2, EE2 or EEQ concentrations. Of these samples, 151 were measurements of DCL concentrations, 83 each were measurements of E2 and EE2 concentrations and 205 were measurements of EEQ concentrations. These monitoring data include samples from 50 unique locations, comprised of influent or effluent from 16 Irish WWTPs, samples from 23 unique water bodies (including rivers, lakes, marine and transitional waters) and domestic effluent from 7 locations. Fig. 1 demonstrates the distribution and frequency of the national monitoring data for DCL, E2, EE2 and EEQ concentrations over the entire period reviewed by this report, from 1999 to 2014 (the different maps for the periods 1999-2004, 2005-2009 and 2010-2015 are presented in Supplementary material). Several patterns regarding the overall distribution of these monitoring data can be observed. First, the monitoring data are distributed relatively evenly throughout the country, though they do tend to be focused around population centres (Galway, Dublin and Cork). EEQ sampling is particularly evenly distributed, having been taken for many inland and coastal surface waters, including the east, south east, south west, and west coasts, as well as one location in the north of the country. Much of the concentration data for all three compounds, however, is carried out in coastal water bodies (marine or transitional waters), rather than in inland surface waters. This is particularly true of DCL, which has mainly been sampled in coastal waters surrounding Dublin and Galway. Concentration measurements of DCL, E2 and EE2 are particularly lacking in the midlands region. Finally, it is clear from this figure that all monitoring data on these watch list PhACs are lacking from the north, north east and northwest coasts of the republic of Ireland. A summary of all the monitoring studies across the country is provided in Supplementary materials (Figs. S3-S5).

In general, the estrogens/estrogenicity is better studied in the Republic of Ireland in terms of monitoring data when compared with DCL. Furthermore, in regards to the estrogens, effect data (measured by EEQ) are more common than concentration data. This is likely due to difficulties associated with obtaining sufficiently sensitive analytical methods of detecting trace amounts of estrogens in water samples. While data from effect-based studies reporting EEQ concentrations can help give an indication of the overall estrogenicity of Irish waters, the compounds responsible usually are not identified. Thus, in regards to quantifying contamination from these two specific estrogenic compounds, effect-based sampling is not as useful as concentration studies that report specific levels of E2 or EE2. Nevertheless, given the lack of sensitivity of current analytical methods, effect-based studies may have a place regarding reporting for WFD monitoring. Fig. 1 illustrates that in addition to surface water samples, monitoring data also came from measurements of WWTP influent or effluent.



Fig. 1. Summary of national monitoring distribution and frequency for diclofenac (blue triangles), E2 (red squares), EE2 (green diamonds), and estradiols equivalents (purple pentagons) in Ireland from 1999 to 2015. Symbol size increases with increasing number of samples taken at each location. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

In the period 1999–2004, studies of endocrine disrupting chemicals and pharmaceutical contamination of aquatic environments were just beginning to gain international popularity during this time. Irish monitoring data during these early years consist mostly of results from effect-based studies, reporting EEQ concentrations. They were carried out largely in the east and southwest coasts and midlands by a project focusing on the Shannon International River Basin District (Tarrant et al., 2005; Tarrant et al., 2008). Some E2 and EE2 concentration measurements were also taken for domestic effluent at sites in the south east of the country, but LODs for these compounds were still too high to compare to WFD proposed limits (Kelly et al., 2010; Quinn-Hosey et al., 2012). No DCL data were found for this period. In the second time period, from 2005 to 2009, the first DCL sampling took place on the east coast of the country, although all three sampling locations were samples from WWTP influent and effluent as opposed to surface waters (Lacey et al., 2008; Lacey et al., 2012). EEQ sampling for estrogenicity continued for the Shannon International River Basin District project during this time period (Quinn-Hosey et al., 2012) and EEQ sampling began on east, south east and west coast locations for the SeaChange project (Giltrap et al., 2013). The project with the most individual EEQ sampling events (89) also took place on a dairy farm in the north of Ireland during this time period (Cai et al., 2012). In the final and most recent time period, from 2010 to 2015, a shift in estrogen sampling is apparent; instead of sampling more inland freshwater sources using largely effect-based studies (EEQ measurements), coastal locations are more frequently monitored. This means that instead of freshwater samples there was a focus on marine and transitional waters. This is because the SeaChange project, which was meant to rectify the lack of estrogen concentration data in marine waters, took place largely during this time period. Less individual samples tend to be taken during concentration studies, because analyses are more expensive and time intensive than many effect-based studies; this is indicated by the smaller symbol sizes in Fig. 4, as opposed to 2 and 3. For DCL, more samples were taken in surface waters in both the east and west coasts of the country (McEneff et al., 2014), these data represent the best information to date on DCL levels in surface waters in Ireland. The GIS maps associated to these periods are presented in Supplementary material.

#### 3.2. Monitoring studies

#### 3.2.1. WWTPs monitoring and locations in Republic of Ireland

Fig. 2 highlights the monitoring data which originate from the 16 sampled Irish WWTPs, and demonstrates the distribution of these sampling locations in regards to all Irish WWTPs. Details of the 16 WWTPs is shown in Supplementary material (Table S1). In total, 178 (34%) of the national monitoring data points from the reviewed publications were



Fig. 2. Distribution of urban wastewater treatment plants (UWWTPs) with existing monitoring data on diclofenac, EE2 and/or estradiol equivalents (EEQ). Orange dots represent agglomerations with only EEQ measurements; yellow dots represent plants with only diclofenac monitoring data; green dots represent plants that have been monitored for all four compounds; yellow dots with orange outlines represent plants that have been monitored for diclofenac and EEQ only; and black dots represent the locations of agglomerations with no monitoring data for these compounds. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

measurements of one of the four investigated parameters (E2, EE2, DCL or EEQ concentrations) from WWTP influent or effluent. Most WWTPs were sampled for estrogenicity only. Fourteen WWTPs were sampled for estrogenicity in total, including Athlone, Ballincollig, Carlow,

Clonmel, Fermoy, Kilkenny, Killarney, Leixlip, Longford, Osberstown, Ringsend, Roscommon, Tralee and Tullamore. Only two WWTPs were sampled for E2 and EE2 specifically, Ringsend and Osberstown. Five plants were monitored for DCL levels including Galway, Leixlip,

#### Table 1

Sixteen WWTPs for which published monitoring data on diclofenac, E2, EE2 or estradiols equivalents is available in Republic of Ireland.

WWTP name	County	Catchment	WFD river basin district	Size (PE)	Type of secondary treatment
Carlow	Carlow	Barrow	South-eastern	39,043	Extended aeration
Ballincollig	Cork	Lee	South-western	27,697	Extended aeration
Fermoy	Cork	Blackwater	South-western	18,608	CAS
Ringsend	Dublin	Coastal	Eastern	2,124,000	Sequence batch reactor
Swords	Fingal	Broad Meadow Water	Eastern	77,014	Extended aeration
Galway	Galway	Coastal	Western	213,424	CAS
Killarney	Kerry	Laune	South-western	41,836	CAS
Tralee	Kerry	Coastal	Shannon	35,149	CAS
Leixlip	Kildare	Liffey	Eastern	100,309	CAS
Osberstown	Kildare	Liffey	Eastern	104,723	Sequence batch reactor
Kilkenny	Kilkenny	Nore	South-eastern	51,988	CAS
Longford	Longford	Shannon Upper	Shannon	11,672	CAS
Tullamore	Offaly	Shannon Lower	Shannon	24,055	CAS
Roscommon	Roscommon	Shannon Upper	Shannon	6989	CAS
Clonmel	Tipperary	Suir	South-eastern	34,909	Extended aeration
Athlone	Westmeath	Shannon Upper	Shannon	21,155	Extended aeration

CAS = conventional activated sludge; PE = population equivalents.
Ringsend Osberstown and Swords. The two WWTPs that were sampled for all 4 compounds were Ringsend and Osberstown. The 16 WWTPs for which monitoring data for at least one of these four compounds exists make up approximately 1.45% of all of Ireland's WWTPs, indicating that more data on the concentrations of these compounds in Irish agglomerations would be useful. The WWTPs with EEQ sampling data are distributed evenly throughout the country, though measurements are missing from the northwest. DCL again has only been sampled in WWTPs surrounding the major population centres of Dublin and Galway. Data on DCL levels in WWTPs influent and effluent in the midlands, south and north are severely lacking (Table 1).

192

# 3.2.2. Highest monitored concentrations in Republic of Ireland for the 3 targeted compounds

Figs. 3–4 and Supplementary material (Fig. S2) were created by mapping the highest concentration value recorded for each drug at each site where it was monitored. These figures thus allow for a comparison of the maximum recorded concentrations of each particular compound at the various sites for which monitoring data currently exist. These maps provide an indication of which areas could be pollution "hotspots" based on currently available monitoring data. The maximum E2 concentrations ranged from not being detected in any samples at a site to 15.2 ng/L. The AA EQS proposed for E2 is 0.4 ng/L (European Commission, 2011); it is clear from Fig. 3 that several locations had maximum values which exceed this limit, including locations near Cork, Galway and Dublin city. Given that E2 is a natural steroid estrogen, it is unsurprising that high concentrations are found near these population centres. Out of a total of 12 sites with reliable quantitative monitoring data, 7 had maximum values that exceeded the proposed AA EQS value. Of those 7, two were domestic wastewater (that are not to be compared with AA EQS that refer to surface water after mixing), but the rest were surface waters distributed throughout the country; future studies should consider these water bodies as potential monitoring sites as it is possible their E2 levels may exceed WFD limits.

EE2 could not be detected at 9 out of 11 sites where sampling occurred, and the maximum concentration recorded for EE2 at any site was 0.32 ng/L. This relatively high value was found in a sample of domestic effluent from a home in County Wicklow as opposed to in a surface water sample. The AA EQS value proposed for EE2 is 0.035 ng/L (European Commission, 2011). From the data presented in Fig. 4, it may seem like only two locations may exceed this AA EQS; however, this interpretation cannot be trusted because the sensitivity of many of the analytical methods used to detect and quantify EE2 was insufficient. Few methods exist that can reach the required LOD of 0.035 ng/L, hence it is



**Fig. 3.** Highest recorded concentrations (ng/L) of E2 at each sampling site where concentration monitoring data were collected. Relative concentration values are indicated by the symbol colour, where low concentrations are indicated by greens and high by reds. Yellow, orange and red indicate sites where the highest recorded concentration was greater than the proposed WFD AA-EQS value for E2 (0.4 ng/L). (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)



Fig. 4. Highest recorded concentrations (ng/L) of EE2 at each sampling site where concentration monitoring data were collected. Relative concentration values are indicated by the symbol colour, where low concentrations are indicated by greens and high by reds. Zero values represent no detects. There are two green dots near Cork and Dublin respectively, although only one is visible at this scale. (For interpretation of the references to colour in this figure legend, the reader is referred to the web version of this article.)

impossible to state with certainty that EE2 is below the WFD proposed limit if the compound is not detected at any of these locations. Additional sampling of surface waters for EE2 using sufficiently sensitive analytical methods is therefore of critical importance to our understanding of how concentrations of this compound in Irish waters compare to WFD limits.

For DCL there were clearly less quantitative available data in surface waters, even though analytical methods used to detect this compound are more easily available than for the estrogenic compounds. The AA EQS value proposed for DCL is 100 ng/L, and the maximum recorded concentration at every surface water sampling location except for one exceeded this value, including samples from Galway and Dublin Bay. Among all monitoring locations, the highest recorded values at two locations actually exceed 1000 ng/L, however these were effluent samples from WWTPs; however, these effluent waters are diluted in receiving waters and the expected concentrations in receiving water would be reduced. Nevertheless, it appears that certain locations in Ireland can have very high levels of DCL at times, indicating that additional monitoring for this compound is necessary.

3.2.3. Wastewater treatment plants and receiving waters monitoring studies

Table 2 gives a summary of the monitoring data found for the Republic of Ireland for the three compounds of interest. The first study investigated DCL in WWTPs for three agglomerations in the greater Dublin area (Lacey et al., 2008). Twenty-four hour composite samples were collected from the influent and effluent of each WWTP. The authors report that DCL was not detected in the influent samples (limit of detection or LOD =  $0.855 \,\mu g/L$ ), but was present in effluent samples  $(LOD = 0.743 \,\mu\text{g/L})$ , although not above the study's limit of quantification (LOQ) for this compound (LOQ in effluent =  $2.478 \,\mu\text{g/L}$ ). A second study by the same research group analysed monthly influent and effluent samples in three WWTPs for a full year (Lacey et al., 2012). Diclofenac was again detected in at least one effluent sample from each plant (detected in 5 effluent samples in total, LOD ranged from 0.5–2.95 µg/L). Similar to the previous study however, DCL was not detected in any influent samples (Lacey et al., 2012). The authors suggest this could be due to deconjugation of conjugated metabolites during the treatment process, which has been observed in other studies (Zhang et al., 2008), this could also be due to possible higher interferences in the analysis of the inlet water due to higher organic matter content that could have reduced the sensitivity of the chemical analysis; however, it is possible that DCL was present in the influent, but at levels below the LOD. In fact in comparison to other European studies, the LODs for both of the Lacey et al. (2012) studies were high. LOD in the ng/L range for DCL are currently standard in the international literature. These were the only studies found that investigated removal efficiencies of DCL, E2 or EE2 in Irish WWTPs.

#### Table 2

Summary of monitoring data for the three compounds of interest in the Republic of Ireland.

	Concentration (ng/L)	Matrix	Experiment	Reference
DCL	nd (LOD = $855 \text{ ng/L}$ )	WWTP influent	3 WWTPs sampled once	Lacey et al. (2008)
	nd (LOD = 743 ng/L)	WWTP effluent		
	nd (LOD = $855 \text{ ng/L}$ )	WWTPs influent	3 WWTPs sampled monthly during a year	Lacey et al. (2012)
	Detected in 5 samples (LOD $=$	WWTP effluent		
	500–2950 ng/L)			
	2630 (max value over a year long	WWTP effluent	2 WWTPs sampled (East and West coast) + receiving sea waters	Schmidt et al.
	study)		+ mussels	(2013)
	550 (max value over a year long study)	WWTP receiving sea water		
	nd (LOD = $29 \text{ ng/g}$ )	Mussels tissues		
	80.3	WWTP effluent	Pan EU campaign	Loos et al. (2013)
	144.3			
E2	nd (LOQ = $10 \text{ ng/L}$ )	WWTP effluent (2 plants)	Pan EU campaign	Jarošová et al.
				(2014)
EE2	nd (LOQ = $10 \text{ ng/L}$ )	WWTP effluent (2 plants)	Pan EU campaign	Jarošová et al.
				(2014)
EEQ	17.2	WWTP effluent (Co. Kildare)	YES assay	Tarrant et al.
	3.2	WWTP effluent (Co. Cork)		(2005)
	1.1–16.0	WWTPs effluents (8 plants)	YES assay	Tarrant et al.
	0.9–2.9	WWTPs receiving waters		(2008)
	0.53–2.67	WWTPs receiving waters + 2 control	YES assay	Tiedeken et al.
		locations		(2016)
	16.21 (max value over a 2-years	WWTPs receiving waters		
	survey)			
	<0.05	WWTP effluent (2 plants)	Pan EU campaign	Jarošová et al. (2014)

nd: not detected.

However, some studies of E2 and EE2 in Irish WWTPs used biomarkers or in vitro bioassays to quantify total estrogenic activity (reported as estradiols equivalents, EEQ) instead of direct chemical analysis. The first study to evaluate Irish WWTP effluent for estrogenicity was carried out in 2004 by Quinn et al. (2004). An effluent sample from Athlone WWTP was analysed for the presence of estrogenic chemicals using the yeast estrogen screen (YES) assay, and HPLC and GC-MS were used for chemical analyses. Lake water from a marina in Killenure Lough, Lough Ree (Co. Westmeath) was also sampled and analysed (Quinn et al., 2004). This study confirmed the presence of a complex mixture of compounds with estrogenic activity in the WWTP effluent. This estrogenicity was attributed mostly to E2, EE2 and bisphenol A. This was also the first study to demonstrate the presence of EDCs in lake water in Ireland. However, results in both effluent and lake water were limited; attempts to identify E2 and EE2 with GC-MS were unsuccessful due to relatively high LOOs associated to the chemical analysis, and the authors stated that further work was necessary to confirm their presence in these aquatic matrices. At approximately the same time that the Quinn study was performed (Quinn et al., 2004), the Tarrant et al. (2005) also began investigating Irish WWPT effluent for estrogenicity. They used the YES assay to screen undiluted effluent from Osberstown (Co. Kildare) and Ballincollig (Co. Cork) WWTPs, and found estrogen levels at 17.2  $\pm$  3.8 and 3.2  $\pm$  1.1 ng/L EEQ respectively. Ballincolig effluent levels were diluted upon entering the River Lee to a level that was below the threshold required to induce vitellogenesis in fish. The effluent form Osberstown WWTP, however, was not diluted enough by the River Liffey to reach levels below this threshold. Accordingly, the study also found raised vitellogenin in male brown trout from this study site, indicating the water was estrogenic enough to have an impact on wild fish (Tarrant et al., 2005). It should be noted however, that the compounds responsible for the estrogenicity of the water were not identified analytically; however, the authors hypothesized EE2 might be responsible because a component of Osberstown WWTP industrial effluent came from a manufacturer of the contraceptive pill. In the work described in Tarrant et al. (2008) estrogenicty using the YES assay was also measured for 8 additional Irish WWTPs and some of their receiving waters. Again, results are reported in EEQ; effluents ranged from 1.1-16.0 ng/L and receiving waters ranged from 0.9-2.9 ng/L. These results indicated that (besides a few notable exceptions in Co. Dublin), levels of estrogenic compounds were high in some Irish WWTP effluents, but that they were sufficiently diluted in most receiving waters so as not to cause an immediate threat to aquatic wildlife (Tarrant et al., 2008).

Given that previous studies mostly focused on surface waters near Dublin or Cork, in 2010 a study investigating estrogenicity in the Shannon International River Basin District (SIRBD) was carried out (Kelly et al., 2010). Five rivers were sampled and estrogenicity was assessed using the YES assay. Not only was estrogenicity evident in the rivers at all sites downstream of WWTPs, it was also present at two control locations. The EEQ values ranged from 0.53-2.67 ng/L, within the concentration range of E2 required to induce vitellogenesis in male rainbow trout (Kelly et al., 2010). This study also demonstrated that levels of EDCs in Irish rivers were elevated in comparison to EEQ values reported by previous studies (Tarrant et al., 2005). An additional study was carried out by the same research group further investigating EDCs in the Border, Midland and Western (BMW) region of Ireland (DoELG, 1999). In this work, influent and effluent samples were collected from inlet and outlet pipes at 4 different WWTPs, and corresponding upstream (control) and downstream water samples were taken from the Rivers Hind, Camlin, Shannon (at Athlone) and Silver (at Tullamore). Grab samples were taken over two year period to investigate seasonal changes. The results demonstrated that some WWTPs in this region were highly efficient in removing estrogen contamination; EEQ removal achieved was 98% and 92% respectively for Tullamore and Longford WWTPs. However, the EEQ levels in receiving water samples from downstream and upstream of the WWTPs demonstrated that this region of Ireland can obtain heavy loads of EDCs (maximum EEQ = 16.21 ng/L in a downstream sample from the River Camlin) (DoELG, 1999).

The SeaChange project provides the most recent monitoring data of E2 and EE2 in Irish waters (Giltrap et al., 2013). For example, in one study, water and mussel samples were collected from three locations on the Irish coast in June 2010: (i) in the estuary of the River Liffey (Dublin site), (ii) in Galway Bay (Galway site) and (iii) in Redbank hatchery, Aughinish Co. Clare (Clare site, mussels only analysed). Samples were analysed by LCMS/MS for estrone (E1), E2 and EE2. This study provided the first detection of E1 by LC/MS/MS in Irish marine waters (Dublin Bay, 0.76 ng/L) (Kunz et al., 2015). E2 and EE2 were not detected in the water samples from Galway Bay or in any mussel samples from

the Dublin, Galway or Clare sites (Kunz et al., 2015). E2 was detected in a sample from Dublin Bay, but levels were low (0.13 ng/L) and three additional samples were non-detects. Therefore the annual average value was below the proposed WFD AA EQS ·In general, the authors of the report concluded that the levels and associated endocrine disrupting effects were generally low at their study sites, and there was likely a low risk of estrogen-caused endocrine disruption to resident species; however areas with significant anthropogenic pressures are at higher risk and additional monitoring was suggested by the authors at such sites (e.g. Dublin Bay) (Giltrap et al., 2013). Further sampling for estrogens at additional sites along the Irish coastline was carried out during the SeaChange project using passive sampling and effect-based monitoring, however these results are not as easily compared with the proposed standards for the WFD.

The most recent data evaluating DCL levels in Irish wastewater effluent or surface waters come from Schmidt et al. (2013). In a year-long study, sewage effluent, receiving marine waters and marine bivalves were analysed for DCL and other pharmaceutical residues. Effluent was sampled (24-h composite samples) from one WWTP on both the east and west coast of Ireland. Mussels were deployed in a year-long cage experiment in a control site and two effluent exposure sites of the east and west coast, and grab samples of surrounding marine surface waters were also analysed. LOQ for DCL were 225 ng/L, 22 ng/L, and 29 ng/g respectively in effluent, marine water, and mussels. The highest DCL concentrations detected in effluent were 1.69 and 2.63 µg/L in the eastern and western WWTP respectively. In marine surface waters the highest values were 0.46 and 0.55  $\mu$ g/L in the east and west. Diclofenac was not detected in mussels (Wille et al., 2012). The concentrations of DCL in effluents and surface waters are comparable to those found in other EU monitoring programmes, but surface water values in particular are on average higher than DCL concentrations detected in other European marine waters (Wille et al., 2012). Furthermore, the highest marine surface water value detected in this study was approximately 5 times higher than the current proposed AA EQS for DCL. In a later experiment associated with the same project, wild mussels from a pristine site off the west of Ireland and a highly contaminated site on the east coast were analysed for several PhACs; again, DCL was not detected in any samples (McEneff et al., 2013). An additional study investigated PhACs residues (including DCL) in cooked and uncooked marine mussel tissue, and found that cooking increased PhACs residues in contaminated tissue. Although this does provide evidence for the potential exposure of humans to DCL via bioaccumulation, mussels in this experiment were artificially exposed in the lab and so this study did not provide monitoring data.

## 3.2.4. Monitoring of on-site non collective treatment systems

In Ireland the domestic wastewater of more than one-third of the population is treated by on-site systems (DoELG, 1999). Considering that human excretions represent a major source of PhACs contamination (Buchberger, 2011) on-site wastewater treatment systems could thus be an important source requiring consideration in Ireland. Work reported by Gill et al. (2009) which investigated the effectiveness of septic tank and secondary treatment on-site wastewater systems, produced two publications specific to EDC removal. The first study sampled domestic effluent at 4 sites in Ireland, two with effluent discharged following primary treatment (i.e. septic tank) and two with secondary treatment (i.e. peat filter) systems (Ó Súlleabháin et al., 2009). EDCs were found at all 4 sites, but E2 and EE2 were each found at two out the four sites with no straightforward relationship with the type of treatment. The sensitivity of the chemical analysis was poor in this study; EE2 was not determined quantitatively and E2 had a high LOD (2 µg/L). The second study aimed to answer similar questions using analytical methods with increased sensitivity (e.g. LOD for  $E2 = 0.05 \,\mu g/L$ ). Gill et al. (2009) again investigated the natural attenuation of EDCs in the most common on-site treatment system in Ireland, the septic tank and subsoil percolation area. The study focused on the transport of EDCs, including E2 and EE2, through the soil at three sites in the east of Ireland. Overall, the authors found that E2 and EE2 were significantly degraded with depth to sub ng/L levels at all sites investigated. These results are the only indication of how efficient on-site systems are at EDC removal in Ireland, and no equivalent study has investigated DCL removal.

### 3.2.5. Studies on alternative control measures

Studies investigating DCL, E2 and EE2 in wastewater effluent such as those reviewed above (Quinn et al., 2004; Tarrant et al., 2005; Tarrant et al., 2008; Lacey et al., 2008; Gill et al., 2009; Lacey et al., 2012; Wille et al., 2012; McEneff et al., 2013; Loos et al., 2013; Schmidt et al., 2013; Jarošová et al., 2014; Tiedeken et al., 2017) demonstrate that these compounds are often not removed completely during treatment at Irish WWTPs. Some Irish studies have therefore examined advanced treatment options which may remove DCL, E2 or EE2 from water more efficiently. For example, a study in 2010 examined the feasibility, kinetics and efficiency of using liquid-core microcapsules as a novel methodology for removal of DCL from water (Whelehan et al., 2010). The work demonstrated that liquid-core microcapsules are capable of rapid extraction of DCL (100% within 50 min of capsule addition to contaminated water) and other common PhACs. Another study investigated the temporal removal of estrogenic activity of several estrogens (including E2 and EE2) by UVA irradiation over an immobilised titanium dioxide (TiO<sub>2</sub>) catalyst (Coleman et al., 2004). UVA photolysis over the catalyst was equally effective at removing estrogenic activity of E2, EE2 and E1; the study demonstrated a 50% reduction in estrogenicity in samples treated for 10 min. Also, the work reviewed above by Cai et al. (2013) investigated removal of hormones from dairy farm wastewater with the aim of evaluating the efficiency of CWs to reduce estrogenic hormone concentrations in dairy wastewater as it was demonstrated in a study carried out in the UK that dairy cows were identified as the largest contributors to excreted estrogens (Johnson et al., 2006). Over the course of a year, monthly samples were taken at seven locations on the farm (i.e. the inflow pond, a plant covered area, close to the outlet of ponds, a lake on the farm, the receiving river, and a groundwater monitoring well) and analysed via a reporter gene assay (RGA). An average removal efficiency for estrogenic compounds of 95.2% was found, indicating that the CW was efficient at removal of such compounds; however, the lowest removal rate during the year was 83.7%, and the concentration at the final pond was 18.8 EEQ ng/L, which is above the proposed lowest observable effect concentration of 10 ng/L (Cai et al., 2012). These authors found that CWs currently employed in Ireland can eliminate hormones in dairy wastewater to low levels often acceptable for effluent. Cai et al. (2013) was also reported that advanced treatments, such as the employment of reactive and sorptive materials (granulated activated carbon, organoclay, etc.), can further improve CWS treated dairy farm wastewater quality, which may be particularly important in enhancing removal efficiency of peak hormone concentrations Additional studies on effective treatment options that can function in an Irish context are needed and could improve the effluent quality in regards to DCL, E2 and EE2 levels.

#### 3.2.6. Comparison of Irish and EU monitoring data

In 2010, two Irish WWTPs (one in Co. Kildare and one in Co. Dublin) were surveyed (via grab samples) for a wide range of PhACs, including DCL, E2 and EE2. This monitoring was part of a Joint Research Centre (JRC) pan-European campaign designed to provide the first concise overview of concentrations of many emerging pollutants in WWTP effluents across Europe (Jarošová et al., 2014). Along with conventional analytical techniques, effect-based monitoring was also carried out in order to determine total estrogenicity of the effluents, reported as EEQ. The study's results for Ireland found no steroid estrogens above their LOQ (i.e. 10 ng/L for steroidal estrogens) via chemical monitoring. In addition, the detected EEQ for both Irish WWTP swere <0.5 ng/L, even though one third of the municipal WWTP effluents from across

Europe that were included in the study had values >0.5 ng/L. This finding indicates that the two Irish WWTP effluents have relatively low estrogenicity in comparison to many European countries. One of the Irish WWTPs was one of only 4 municipal WWTPs in the study that utilised a tertiary treatment step (UV light); this advanced treatment could have contributed to the observed low levels of estrogenic contamination (Loos et al., 2013). The same year, another study based on the same Irish sampling events was published which analysed effluents for additional PhACs, including DCL (Loos et al., 2013). The study found that throughout Europe, DCL was among the compounds with the highest median concentration levels (it was found in 89% of all samples, max = 174 ng/L, median = 43.3 ng/L). In the Irish samples specifically, DCL was detected at concentrations of 80.3 and 144.3 ng/L for the two investigated WWTPs respectively (Loos et al., 2013).

Another European level study in 2013 used the GWAVA model to predict water concentrations of DCL, E2 and EE2 in rivers throughout Europe. The study recognized that the levels of these compounds found in receiving waters would vary considerably between European nations depending on available dilution of sewage effluent. Overall, the model predicted that 12%, 1% and 2% by length of Europe's rivers would exceed the EE2, E2 and DCL proposed annual average EQSs. For all three compounds however, <10% of Irish river length was predicted to exceed EQSs. There are significant sources of uncertainty in the model that should be noted; as far as we can tell, estimates are based on Northern Irish data only, and the parameters determining effluent concentrations, a critical component of estimating river concentrations, have additional uncertainty (Johnson et al., 2013).

## 3.3. Relationship with drug utilising dispensing data

Annual dispensing data for each LHO throughout the country is mapped for DCL and for E2 (heat maps and dispensing data presented in Supplementary material. Similar trend emerged where LHOs exhibiting higher dispensing data for DCL also matched locations reported previously for upper concentrations of the same compound using monitoring techniques (Supplementary material). However, this observation should be taken with care considering the low number of monitoring data referenced for DCL in surface waters. This trend was also evident for E2 where mapped baseline drug dispensing data from LHO's exhibiting upper concentrations of this estrogen were similar to locations identified for upper values using monitoring techniques (Fig. 3). While this proxy baseline dispensing data is useful for possibly intimating indicator areas that make disproportionately high contributions to overall pollution load, these "hotspots" may vary spatially and temporality and may occur due a variety of other contributing sources such as discharge points of high-risk WWTPs, high densities of livestock near water sources, effluent from hospitals and residential homes, and/ or pharmaceutical producers (Verlicchi et al., 2012; Tiedeken et al., 2017).

# 3.4. Conclusion and perspectives

The Water Framework Directive (WFD) and Irish River Basin Management Plans (RBMPs) establish both legal and operational frameworks to protect and restore clean water and to ensure its long-term, sustainable use. These goals require an integrated approach to the sustainable management and protection of water resources. Critical shortfalls in existing Irish RBMPs highlight the importance of affordability and prioritization considerations, particularly given the economic and social value of a clean and protected water supply. Therefore, the overall aim of this study was to provide a baseline study for Ireland exploring the implications of the addition of DCL, E2 and EE2 to the WFD priority substances list. This study mapped all national concentration data and concludes that DCL concentrations found in surface waters are generally below the limits proposed by the WFD, but that exceedances have occasionally been reported as it is the case in several European countries. In comparison, E2 and EE2 surface water concentrations are generally much lower, however reported values still commonly exceed the WFD proposed limits for these bioactive compounds. Perhaps most notably, while current standard, laboratory-based analytical chemistry methods are sufficiently sensitive for the detection and quantification of DCL, limits of detection for E2 and EE2 are often higher than proposed EQSs. This issue presents serious analytical challenges in regards to chemical monitoring methods and reporting for these two PhACs, and impacts Ireland's ability to meet European reporting requirements for these estrogenic compounds. The mapping work conducted during this study demonstrated that more monitoring data on DCL, E2 and EE2 in Irish waters is required. Nevertheless, based on the limited Irish data extracted from the literature and mapped in this study, it appears that the majority of Irish surface waters may not exceed WFD proposed limits for DCL, E2 and EE2, but that point sources of pollution could lead to occasional hotspots exceeding European limits. It must be noted that this prediction is based upon the use of very limited data, and is especially uncertain because of a lack of sufficiently sensitive analytical detection methods. This observation will resonate with the majority of EU countries in terms of current levels of CEC monitoring in respective receiving waters and at control points. Conventional analytical methods are sufficiently sensitive for the detection and quantification of DCL, but not for E2 and EE2, thus alternative, ultra-trace, time-integrated monitoring techniques such as passing sampling are needed to inform water quality for these estrogens. Another emerging potential solution to the problem of low EQS values of E2 and EE2 is the use of biological effects monitoring techniques (Streck, 2009; Kunz et al., 2015; Simon et al., 2015).

In comparison to other 28 EU countries along with Switzerland, Norway and Turkey, Ireland produced 21 studies on these three contaminants of emerging concern over 15-year systematic review period (Tiedeken et al., 2017) where Spain, Germany and the United Kingdom contributed 707 (56%) of all reports. However, 24 and 16 EU countries produced under 50 and 20 articles respectively on these PhACs in their national receiving waters; consequently, very few countries have reported on use predicted or measured environmental concentrations to underpin modeling or to inform risks in their river basins (ter Laak et al., 2010; Guillén et al., 2012; Murphy et al., 2017). Overall, this Irish study supported main tenets of Tiedeken et al. (2017) which found that DCL and EE2 enter European aquatic environment mainly following human consumption and excretion of therapeutic drugs, and by incomplete removal from influent at urban wastewater treatment plants. E2 is a natural hormone excreted by humans, which also experiences incomplete removal during WWTP treatments.

Future Irish-specific work in this research field is essential in order to ensure PhACs do not threaten our water supplies. Irish studies evaluating PhACs levels in WWTPs influents and effluents are also lacking; these are needed in order to develop effective and economic control measures for PhAC removal from wastewaters. The present study also found that on-site treatment systems could potentially be major sources of PhACs contamination in an Irish context, thus future research should address this issue. Given the positive results and outcomes from studies that utilise effect-based (biological) monitoring, passive sampling or an integrated monitoring approach (combined use of chemical and biological monitoring methods), thus Ireland along with other EU memberstates must consider broader acceptance of these types of methodologies for WFD priority substance reporting. Future projects evaluating concentrations in aquatic and other environmental matrices (sludge, sediment, biota) must be supported, particularly for compounds that are not yet considered priority substances (current and potential future watch list compounds). Furthermore, more data should be collected on prescriptions written and dispensed by public and private health agencies in Ireland, and such data should be made available to research projects. Another issue is the unavailability of commercially sensitive data such as PhAC sales/production information. This information would facilitate a more accurate determination of emission sources in different

401

catchments. Now that the limitations of these existing monitoring data are understood, additional types of data that can help inform regulators and policy makers about the levels of these substances in Irish waters should be explored. In particular, it would be useful to apply established European models for predicting fate, behaviour and concentrations of chemical pollutants in surface waters in an Irish context including considerations for influence of climate change given that 2015 was the wettest year recorded in Ireland over 250 years of annual measurements (Murphy et al., 2017). To this aim, longer term projects that utilise European software and models to predict fate of watch list compound concentrations in whole Irish watersheds should be carried out such development of spatially explicit Geography-Referenced Regional Exposure Assessment Tool for European River Basins (GREAT-ER).

# Acknowledgements

We also acknowledge Environmental Protection Agency STRIVE Programme for funding this research (Project 2014-W-DS-18).

# Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.scitotenv.2017.11.021.

# References

- Barbosa, M.O., Moreira, N.F.F., Ribeiro, A.R., Pereira, M.F.R., Silva, A.M.T., 2016. Occurrence and removal of organic micropollutants: an overview of the watch list of EU Decision 2015/495. Water Res. 94, 257–279.
- Boxall, A.A., 2010. Veterinary medicines and the environment. In: Cunningham, F., Elliott, J., Lees, P. (Eds.), Comparative and Veterinary Pharmacology, Handbook of Experimental Pharmacology. Springer, Berlin Heidelberg, pp. 291–314.
- Buchberger, W.W., 2011. Current approaches to trace analysis of pharmaceuticals and personal care products in the environment. J. Chromatogr. A 1218, 603–618.
- Cai, K., Elliott, C.T., Phillips, D.H., Scippo, M.-L., Muller, M., Connolly, L., 2012. Treatment of estrogens and androgens in dairy wastewater by a constructed wetland system. Water Res. 46, 2333–2343.
- Cai, K., Phillips, D.H., Elliott, C.T., Muller, M., Scippo, M.-L, Connolly, L., 2013. Removal of natural hormones in dairy farm wastewater using reactive and sorptive materials. Sci. Total Environ. 461–462, 1–9.
- Cirja, M., Ivashechkin, P., Schäffer, A., Corvini, P.F.X., 2008. Factors affecting the removal of organic micropollutants from wastewater in conventional treatment plants (CTP) and membrane bioreactors (MBR). Rev. Environ. Sci. Biotechnol. 7, 61–78.
- Coleman, H.M., Routledge, E.J., Sumpter, J.P., Eggins, B.R., Byrne, J.A., 2004. Rapid loss of estrogenicity of steroid estrogens by UVA photolysis and photocatalysis over an immobilised titanium dioxide catalyst. Water Res. 38, 3233–3240.
- DoELG, 1999. Department of the Environment and Local Government, Environmental Protection Agency, and Geological Survey of Ireland, Groundwater Protection Schemes. Department of the Environment and Local Government, Environmental Protection Agency and Geological Survey of Ireland, Ireland.
- Environmental Protection Agency, 2017. www.epa.ie/newsandevents/news/ name.53429.en.html#WfbhG1uoOUk, Accessed date: 30 October 2017.
- European Commission, 2011. COM(2011) 876 Final, Proposal for a Directive of the European Parliament and of the Council Amending Directives 2000/60/EC and 2008/105/EC as Regards Priority Substances in the Field of Water Policy. Brussels, Belgium.
- Gill, L., Misstear, B., Johnston, P., O' Luanaigh, N., 2009. Natural attenuation of endocrine disrupting chemicals in on-site domestic wastewater treatment systems. ASTM International]–>J. ASTM Int. 6, 1–13.
- Giltrap, M., McHugh, B., Ronan, J., Wilson, J., McGovern, E., 2013. Biological Effects and Chemical Measurements in Irish Marine Waters, Project Based Award, Final Report. Environmental Protection Agency, Ireland, pp. 1–110.
- Guillén, D., Ginebreda, A., Farré, M., Darbra, R.M., Petrovic, M., Gros, M., Barceló, D., 2012. Prioritization of chemicals in the aquatic environment based on risk assessment: analytical, modeling and regulatory perspective. Sci. Total Environ. 440, 236–252.
- Jarošová, B., Erseková, A., Hilscherová, K., Loos, R., Gawlik, B., Giesy, J., Bláha, L., 2014. Europe-wide survey of estrogenicity in wastewater treatment plant effluents: the need for the effect-based monitoring. Environ. Sci. Pollut. Res. 21, 10970–10982.
- Johnson, A.C., Williams, R.J., Matthiessen, P., 2006. The potential steroid hormone contribution of farm animals to freshwaters, the United Kingdom as a case study. Sci. Total Environ. 362, 166–178.
- Johnson, A.C., Dumont, E., Williams, R.J., Oldenkamp, R., Cisowska, I., Sumpter, J.P., 2013. Do concentrations of ethinylestradiol, estradiol, and diclofenac in European rivers exceed proposed EU environmental quality standards? Environ. Sci. Technol. 47, 12297–12304.
- Kelly, M.A., Reid, A.M., Quinn-Hosey, K.M., Fogarty, A.M., Roche, J.J., Brougham, C.A., 2010. Investigation of the estrogenic risk to feral male brown trout (*Salmo trutta*) in the Shannon International River Basin District of Ireland. Ecotoxicol. Environ. Saf. 73, 1658–1665.

- Kosma, C.I., Lambropoulou, D.A., Albanis, T.A., 2014. Investigation of PPCPs in wastewater treatment plants in Greece: occurrence, removal and environmental risk assessment. Sci. Total Environ. 466–467, 421–438.
- Kümmerer, K., 2009. The presence of pharmaceuticals in the environment due to human use – present knowledge and future challenges. J. Environ. Manag. 90, 2354–2366.
- Kunz, P.Y., Kienle, C., Carere, M., Homazava, N., Kase, R., 2015. In vitro bioassays to screen for endocrine active pharmaceuticals in surface and waste waters. J. Pharm. Biomed. Anal. 106, 107–115.
- ter Laak, T.L., van der Aa, M., Houtman, C.J., Stoks, P.G., van Wezel, A.P., 2010. Relating environmental concentrations of pharmaceuticals to consumption: a mass balance approach for the river Rhine. Environ. Int. 36, 403–409.
- Lacey, C., McMahon, G., Bones, J., Barron, L., Morrissey, A., Tobin, J.M., 2008. An LC–MS method for the determination of pharmaceutical compounds in wastewater treatment plant influent and effluent samples. Talanta 75, 1089–1097.
- Lacey, C., Basha, S., Morrissey, A., Tobin, J., 2012. Occurrence of pharmaceutical compounds in wastewater process streams in Dublin, Ireland. Environ. Monit. Assess. 184, 1049–1062.
- Loos, R., Carvalho, R., António, D.C., Comero, S., Locoro, G., Tavazzi, S., Paracchini, B., Ghiani, M., Lettieri, T., Blaha, L., Jarosova, B., Voorspoels, S., Servaes, K., Haglund, P., Fick, J., Lindberg, R.H., Schwesig, D., Gawlik, B.M., 2013. EU-wide monitoring survey on emerging polar organic contaminants in wastewater treatment plant effluents. Water Res. 47, 6475–6487.
- McEneff, G., Barron, L., Kelleher, B., Paull, B., Quinn, B., 2013. The determination of pharmaceutical residues in cooked and uncooked marine bivalves using pressurised liquid extraction, solid-phase extraction and liquid chromatography-tandem mass spectrometry. Anal. Bioanal. Chem. 405, 9509–9521.
- McEneff, G., Barron, L., Kelleher, B., Paull, B., Quinn, B., 2014. A year-long study of the spatial occurrence and relative distribution of pharmaceutical residues in sewage effluent, receiving marine waters and marine bivalves. Sci. Total Environ. 476–477, 317–326.
- Murphy, C., Noone, S., Duffy, C., Broderick, C., Matthews, T., Wilby, R.L., 2017. Irish droughts in newspaper archives: rediscovering forgotten hazards? Weather 72, 151–155.
- Ó Súlleabháin, C., Gill, L.W., Misstear, B.D.R., Johnston, P.M., 2009. Fate of endocrinedisrupting chemicals in percolating domestic wastewater effluent. Water Environ. J. 23, 110–118.
- Quinn, B., Gagné, F., Costello, M., McKenzie, C., Wilson, J., Mothersill, C., 2004. The endocrine disrupting effect of municipal effluent on the zebra mussel (*Dreissena polymorpha*). Aquat. Toxicol. 66, 279–292.
- Quinn-Hosey, K.M., Roche, J.J., Fogarty, A.M., Brougham, C.A., 2012. A toxicological assessment of endocrine disrupting chemicals found in BMW (Border, Midland and Western) region of Ireland. J. Environ. Prot. 3, 304–315.
- Rowan, N.J., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, implications and outlooks. Int. J. Microbiol. 5:206–214 (published online: http://www.hindawi.com/journals/ijmb/2011/462832.html (PMID: 209976256)).
- Schmidt, W., McEneff, G., Quinn, B., 2013. Pharmaceuticals in the Irish Aquatic Environment: The Assessment and Potential Human Impact of Exposure to Environmental Contaminants on Marine and Freshwater Bivalves. Environmental Protection Agency, Dublin, Ireland, pp. 1–332.
- Simon, E., Lamoree, M.H., Hamers, T., de Boer, J., 2015. Challenges in effect-directed analysis with a focus on biological samples. TrAC Trends Anal. Chem. 67, 179–191.
- Streck, G., 2009. Chemical and biological analysis of estrogenic, progestagenic and androgenic steroids in the environment. TrAC Trends Anal. Chem. 28, 635–652.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638.
- Tarrant, H., Llewellyn, N., Lyons, A., Tattersall, N., Wylde, S., Mouzakitis, G., Maloney, M., McKenzie, C., 2005. Endocrine Disruptors in the Irish Aquatic Environment. Environmental Protection Agency, Dublin, Ireland, pp. 1–191.
- Tarrant, H., Mousakitis, G., Wylde, S., Tattersall, N., Lyons, A., Maloney, M., Llewellyn, N., 2008. Raised plasma vitellogenin in male wide brown trout (*Salmo trutta*) near a wastewater treatment plant in Ireland. Environ. Toxicol. Chem. 27, 1773–1779.
- Tiedeken, E.J., Clifford, E., Rowan, N., 2016. Investigation of the implications for Ireland of emerging standards on pharmaceuticals in receiving waters. EPA Research Report No. 182.
- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – a 20 year systematic review. Sci. Total Environ. 574, 1140–1163.
- Verlicchi, P., Zambello, E., 2016. Predicted and measured concentrations of pharmaceuticals in hospital effluents. Examination of the strengths and weaknesses of the two approaches through the analysis of a case study. Sci. Total Environ. 565, 82–94.
- Verlicchi, P., Al Aukidy, M., Zambello, E., 2012. Occurrence of pharmaceutical compounds in urban wastewater: removal, mass load and environmental risk after a secondary treatment—a review. Sci. Total Environ. 429, 123–155.
- Whelehan, M., von Stockar, U., Marison, I.W., 2010. Removal of pharmaceuticals from water: using liquid-core microcapsules as a novel approach. Water Res. 44, 2314–2324.
- Wille, K., De Brabander, H.F., Vanhaecke, L., De Wulf, E., Van Caeter, P., Janssen, C.R., 2012. Coupled chromatographic and mass-spectrometric techniques for the analysis of emerging pollutants in the aquatic environment. TrAC Trends Anal. Chem. 35, 87–108.
- Zhang, Y., Geißen, S.-U., Gal, C., 2008. Carbamazepine and diclofenac: removal in wastewater treatment plants and occurrence in water bodies. Chemosphere 73, 1151–1161.
- Zhou, J.L., Zhang, Z.L., Banks, E., Grover, D., Jiang, J.Q., 2009. Pharmaceutical residues in wastewater treatment works effluents and their impact on receiving river water. J. Hazard. Mater. 166, 655–661.

Contents lists available at ScienceDirect





# Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

# Current decontamination challenges and potentially complementary solutions to safeguard the vulnerable seafood industry from recalcitrant human norovirus in live shellfish: Quo Vadis?



# Neil J. Rowan

Centre for Sustainable Disinfection and Sterilization, Bioscience Research Institute, Technological University of the Shannon Midlands Midwest, Athlone Campus, Ireland

# HIGHLIGHTS

# G R A P H I C A L A B S T R A C T

- Choice of appropriate non-thermal intervention(s) is limited as shellfish are consumed raw.
- Depuration using standard UV-irradiation effectively addresses bacterial contaminants in live shellfish.
- Additional non-thermal mitigation strategies are required to completely destroy norovirus in live oysters.
- New diagnostic method(s) are required to confirm non-viable norovirus post depuration.
- Global warming of coastal growing environments may increase risks for commercial shellfish producers.

#### ARTICLE INFO

Editor: Damià Barceló

Keywords: Norovirus Food security Sustainability Decontamination Shellfish Diagnostics



# ABSTRACT

Safeguarding the seafood industry is important given its contribution to supporting our growing global population. However, shellfish are filter feeders that bioaccumulate microbial contaminants in their tissue from wastewater discharged into the same coastal growing environments leading to significant human disease outbreaks unless appropriately mitigated. Removal or inactivation of enteric viruses is very challenging particularly as human norovirus (hNoV) binds to specific histo-blood ligands in live oyster tissue that are consumed raw or lightly cooked. The regulatory framework that sets out use of clean seawater and UV disinfection is appropriate for bacterial decontamination at the post-harvest land-based depuration (cleaning) stage. However, additional non-thermal technologies are required to eliminate hNoV in live shellfish (particularly oysters) where published genomic studies report that low-pressure UV has limited effectiveness in inactivating hNoV. The use of the standard genomic detection method (ISO 15, 216-1:2017) is not appropriate for assessing the loss of infectious hNoV in treated live shellfish. The use of surrogate viral infectivity methods appear to offer some insight into the loss of hNoV infectiousness in live shellfish during decontamination. This paper reviews the use of existing and potentially other combinational treatment approaches to enhance the removal or inactivation of enteric viruses in live shellfish. The use of alternative and complementary novel diagnostic approaches to discern viable hNoV are discussed. The effectiveness and virological safety of new affordable hNoV intervention(s) require testing and validating at commercial shellfish production in conjunction with laboratorybased research. Appropriate risk management planning should encompass key stakeholders including local government and the wastewater industry. Gaining a mechanistic understanding of the relationship between hNoV response at molecular and structural levels in individually treated ovsters as a unit will inform predictive modeling and appropriate treatment technologies. Global warming of coastal growing environments may introduce additional

E-mail address: nrowan@ait.ie.

http://dx.doi.org/10.1016/j.scitotenv.2023.162380

Received 11 January 2023; Received in revised form 16 February 2023; Accepted 17 February 2023

Available online 24 February 2023

0048-9697/© 2023 The Author. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

contaminant challenges (such as invasive species); thus, underscoring need to develop real-time ecosystem monitoring of growing environments to alert shellfish producers to appropriately mitigate these threats.

#### Contents

1.	Introduction
2.	Methods for human norovirus removal or destruction in contaminated live shellfish
	2.1. Using conventional low-pressure UV irradiation and filtration methods
	2.2. Pulsed light (PL) technology.
	2.3. Potential combined use of advanced oxidative processes (AOPs)
	2.4. Photonics
	2.5. Use of micro- and nano-bubble technology
3.	Use of safe enteric surrogates to monitor and assess disinfection of shellfish
4.	Advanced real-time diagnostics: future use of combined confocal microscopy and RAMAN spectroscopy (CRS)
5.	Advanced real-time diagnostics: future use of flow cytometry
6.	Modeling of enteric viral removal and inactivation and risk mitigation
7.	Climate change influences
8.	Quadruple Helix Hub Approach.    1
9.	Conclusions and implications for future research
CRec	rdiT authorship contribution statement
Fund	ding acknowledgement
Data	a availability
Decl	laration of competing interest
Refe	erences

# 1. Introduction

The seafood industry is an important sector of food production worldwide (Ruiz-Salmön et al., 2020). An estimated 18 million tons of marine molluscs are globally harvested each year with an estimated value of \$35 billion, comprising 9 % of the value of fisheries worldwide (Sharp et al., 2021). Shellfish consumption is an increasingly important part of human diet and is an emerging area for economic growth worldwide (Ruiz-Salmön et al., 2020; Ruiz-Salmön et al., 2021; Laso et al., 2022; Cooney et al., 2023). This is particularly relevant given that our growing global population recently reached eight billion people. Shellfish filter hundreds of litres of coastal water for nutrients and can bioaccumulate human pathogens including norovirus from their growing marine environment, if contaminated with faecal material (Compos and Lees, 2014). Human norovirus is found in high concentrations in faeces  $(10^{11} \text{ virus/g})$  (La Rosa et al., 2012); consequently, this virus is recognized as a high risk for environmental transmission (McLeod et al., 2017: Sharp et al., 2021). Pilotto et al. (2019) reported concentrations of 10<sup>11</sup> genome copies/per gram (cg/g) of murine norovirus (MNV1) that were achieved during 24 h bioaccumulation in the Pacific oyster Crassostrea gigas; whereas, Maalouf et al. (2011) noted that both human norovirus genogroups GI and GII were shown to simultaneously bioaccumulate in C. gigas oysters to  $10^6$  cg/g. Contamination of raw seafood with pathogenic microorganisms occurs through primary production and from infected food handlers (McLeod et al., 2017; Sharp et al., 2021).

In many parts of the world, microbiological pollution of coastal areas with human sewage, and agricultural run-off, can readily occur in commercial harvesting such that shellfish bioaccumulate large amounts of bacterial and/or viral pathogens (Bosch et al., 1995; Winterbourn et al., 2016); thus, requiring to be appropriately cleaned (depurated) (Rupnik et al., 2018; Razafimahefa et al., 2020; Rupnik et al., 2021). The faecal indicator organism *Escherichia coli* is frequently used as a general indicator of sewage contamination and for assessing the effectiveness of shellfish depuration processes (Sharp et al., 2021). These authors reported that non-pathogenic *E. coli*, pathogenic *E. coli* O157:H7 and hNoV GII RNA accumulate rapidly in mussels using simulated water contamination after a point-source release from a combined sewer overflow (CSO) and untreated

wastewater released directly into the coastal zone. "All three microbiological indicators reached close to maximum concentration within 3 h of exposure, demonstrating that short CSO discharges pose an immediate threat to shellfish harvesting areas" (Sharp et al., 2021). Depuration in clean seawater proved partially successful at removing pathogenic and non-pathogenic *E. coli* from shellfish tissue, but failed to eradicate hNoV GII RNA. The authors concluded that current EU standards for evaluating microbiological risk in shellfish are inadequate for protecting consumers against exposure to hNoV; thus, intimating a need to improve depuration efficiencies including developing new appropriate mitigation technologies (McLeod et al., 2017; McMenemy et al., 2018).

It is notable that hNoV is the leading viral cause of human gastroenteritis, where consuming contaminated shellfish contributes as a vector in this transmission (Yu et al., 2015). In developed countries, it accounts for 95 % of non-bacterial foodborne outbreaks, and over 50 % of all microbial outbreaks (Dewey-Mattia, 2018; Wikswo et al., 2021). In the US alone, noroviruses are responsible for ~20 million cases and >70,000 hospitalizations of infected children, annually (Smith and Smith, 2019). Oysters are frequently implicated as the source of this human gastroenteritis illness posing a particular risk to young, elderly and immunocompromised (Centers for Disease Control and Prevention (CDC), 2020). While most adults recover from viral diarrhoea, such illness in young children can lead to hospitalization and life-threatening dehydration (Centers for Disease Control and Prevention (CDC), 2020). Notably, contaminated shellfish are estimated to cause between 9 and 34 % of all foodborne norovirus cases in the US (Pouillot et al., 2021), and similar etiological ranges have been reported internationally (Havelaar et al., 2008; Davidson et al., 2011; Advisory Committee on the Microbiological Safety of Food, 2015). No vaccine currently exists that can prevent hNoV infection (Centers for Disease Control and Prevention (CDC), 2020). Best published evidence suggests that oysters are a main source of foodborne hNoV-transmission due to (a) the mode of transmission (as they mainly consumed raw or lightly cooked) (Rupnik et al., 2018); (b) production of oysters in same intertidal waters where human sewage is discharged as a source of hNoV (Rupnik et al., 2018; Sharp et al., 2021); and (c) specific retention of hNoV strains in oysters through binding to ligands enabling lengthy persistence (McLeod et al., 2017).

Oysters are premium food products of economic importance that are consumed globally (McLeod et al., 2017); however; failure to appropriately depurate contaminated shellfish can lead to reputational damage to the exporting industry and a commensurate loss of confidence for importing countries (Younger et al., 2020). Currently, it is technically challenging to confidently address the effective removal or destruction of recalcitrant enteric viruses, particularly from live contaminated shellfish (Leduc et al., 2020; Younger et al., 2020; Rupnik et al., 2021). The Pacific oyster (*C. gigas*) is the most commonly produced oyster globally; but other species are also commercially harvested including *Crassostrea viginica* (the Eastern oyster) in the US, *Saccostrea glomerata* in Australia, and flat oysters (*Ostrea edulis*) that are produced in many countries including Ireland, the United Kingdom, and Croatia (McLeod et al., 2017).

Norovirus is a small (approximately 30 nm in diameter) non-enveloped, single-stranded RNA virus that belongs to the family Caliciviridae (Sharp et al., 2021). Norovirus was named after the original Norwalk strain, which caused an outbreak of gastroenteritis in a school in Norwalk, Ohio in 1968. Noroviruses are now classified into ten genogroups (GI-GX) and 48 genotypes where typically GI and GII predominate (McLeod et al., 2017; Chhabra et al., 2019; Rupnik et al., 2021). While wastewater treatment plants (WWTPs) can contribute to enteric virus removal, it appears that improvements in the effectiveness of technologies or processes are required to fully remove or inactivate enteric viruses from human sewage (Barrett et al., 2016). Data published in a recent EFSA baseline survey highlights that peak hNoV concentration and prevalence in contaminated oysters was observed in the months of January and February when almost 65 % of samples tested were positive for hNoV [mean concentration of 661 genome copies/g] (EFSA, 2019).

Hepatitis A Virus (HAV) is also a small (approximately 30 nm in size), non-enveloped, icosahedron-shaped, RNA enteric virus that contaminates shellfish in polluted seafood leading to human illness (Woods and Burkhardt III, 2010). A vaccine exists for HAV (Centers for Disease Control and Prevention (CDC), 2020). Coincidently, viruses with icosahedral-shaped capsids efficiently package their RNA (Martin-Bravo et al., 2021), which will be discussed later in this review in the context of a focus for real-time detection and as a structural target for decontaminating enteric viruses of similar geometry (such as hNoV and HAV). Notably, researchers have reported on geometric defects and icosahedral viruses that may influence assembly, dissociation, or accessibility of cellular proteins to virion components (Wang et al., 2018). Many complex shellfish pathogens, such as these enteric viruses and waterborne protozoan enteroparasites (such as Cryptosporidium oocysts, and Giardia cysts), do not grow on standard laboratory based culture media, and require use of more sophisticated enumerations methods post-treatment, such as quantitative PCR (ISO standard) (Garvey et al., 2010; Garvey et al., 2013; Hayes et al., 2013; Gerard et al., 2019; Franssen et al., 2019; Sharp et al., 2021; Rupnik et al., 2021). There are in vitro cell culture methods to study the viability of treated surrogate enteric viruses (Barrett et al., 2016) and for waterborne parasites (Garvey et al., 2014a; Garvey et al., 2014b), which can be combined with the standard qPCR method to inform decontamination. Currently, there is no appropriate in vitro model for studying the infectivity of hNoV strains that has hindered development, testing and standardization of treatment approaches for the shellfish industry, particularly at commercial depuration phase (McLeod et al., 2017; Rupnik et al., 2018; Rupnik et al., 2021). Sophisticated diagnostic techniques are not routinely available in standard food-testing laboratories for enteric viruses; for example, hNoV and HAV are classified as belonging to Human Pathogen Hazard Group II viral pathogens necessitating use of more specialized Cat II facilities (Centers for Disease Control and Prevention (CDC), 2020).

These enteric viruses have a low infectious dose and may remain infectious for weeks in the environment or on food surfaces (Nasheri et al., 2021). However, limited physiological or mechanistic information is available regarding viral survival, persistence and transmission in contaminated shellfish. Interestingly, Kokkinos and co-workers (2021) noted that "the vast majority of viral agents, which are transmitted via the faecal-oral route are non-enveloped, highly stable under environmental conditions, characterized by extremely small size, and include emerging and reemerging Caliciviridae, Adenoviridae, Hepevirdae, Picornaviridae and Reoviridae. The enteric viruses of human stool and urine belong to more than 140 types (Kokkinos et al., 2011) where untreated wastewater has been identified as the most diverse viral metagenome examined thus far. Most sequence reads have little or no sequence relations to known problematic viruses, underscoring that most of the viruses have yet to be characterized", and are underestimated in prevalence (Cantalupo et al., 2011).

Previous researchers have reported that under EU law, sanitary classification of shellfish production areas is recognized based on the presence and concentration of the faecal bacterium Escherichia coli as designated EU Regulation 627/2019 (Rupnik et al., 2021; Hunt et al., 2023). Currently, the minimum time and water temperature used for commercial depuration are not stipulated in EU Regulation, however, Rupnik et al. (2021) had noted that such depuration should be performed for a minimum of 42 h with a water temperature of no <8 °C in Ireland. Specifically, shellfish harvesting waters are classified as A, B, or C, which is based on increasing E. coli concentrations measured in shellfish flesh and fluid (Hunt et al., 2023). Consequently, within each class, specific post-harvest decontamination methods such as depuration and relaying are mandated before any live product can be sold (Rupnik et al., 2018; Hunt et al., 2023). However, studies have indicated that monitoring faecal indicator bacteria in shellfish may be a poor indicator of water pollution and the risk of human exposure to pathogens from consuming shellfish (Romalde et al., 1994; Younger et al., 2018). Essentially, "bacterial species are traditionally used as indicators of faecal contamination of agricultural products, shellfish and shellfish waters" (Garcia et al., 2020). This led to the aforementioned formulation of legislation based on the measurement of faecal indicator bacteria (EU, 2020). Previous researchers have reported that there is a poor correlation between concentrations of E. coli and norovirus, making it less relevant as an indicator organism (Flannery et al., 2009; Lowther et al., 2019; Hunt et al., 2023). Although methods for detection and quantification of problematical pathogens (e.g., Vibrio spp.) in shellfish exist, these also have yet to be incorporated into EU legislation due to the lack of robust evidence and non-consensus agreements over what new standards should be incorporated (Hassard et al., 2017; EFSA, 2019; Sharp et al., 2021).

Post-harvest disease mitigation typically occurs in land-based tanks containing clean seawater. Duration of treatment is governed by several biological and environmental factors that are informed by the monitoring of enteric virus concentration (or viral load) (Rupnik et al., 2021). Thus, oysters and other bivalve molluscan shellfish harvested from class B category waters (accounting for the majority of overall oyster production from European countries) (McLeod et al., 2017), must undergo appropriate depuration before human consumption (EFSA, 2019). These regulations have informed the effective decontamination of bacterial-associated illness caused by contaminated oysters; however, despite same, there are numerous reports of enteric viral outbreaks caused by depurated oysters (Rajko-Nenow et al., 2013). For example, Doré et al. (2010) noted that illnesses have also been reported for the consumption of contaminated oysters that were harvested from category A waters where post-harvest decontamination is not mandatory. Consequently, in order to mitigate against the occurrence of enteric viral illness and to avoid the loss of consumer confidence in shellfish products, many commercial producers apply depuration treatment for oysters harvested from category A growing waters as part of their risk management procedures (Rupnik et al., 2018).

The most widely practised post-harvest treatments is depuration, whereby bivalve shellfish undergo self-purification in land-based tanks containing clean seawater (McLeod et al., 2017). However, the effect of depuration is reducing harmful norovirus in live shellfish is less well established compared with treating *E. coli*, particularly at the standard conditions of 48 h at 8 to 15 °C (McLeod et al., 2017; Hunt et al., 2023). The use of viral surrogates is a common approach to studying depuration efficacy such as using F + RNA bacteriophage type I or II (designated FRNA or FRNAPII) where there are established infectivity assays to offset this technical problem for assessing hNoV post treatments using the ISO standard detection method (Polo et al., 2014; Rupnik et al., 2018; Leduc et al., 2020; Rupnik

et al., 2021). "F-specific coliphages (F+ coliphages) are bacteriophages that infect *Escherichia coli* cells possessing F pili. F+ coliphages are classified into FDNA phages (FDNAPHs) or FRNA phages (FRNAPHs), depending on whether their genomes consist of single-stranded DNA or single-stranded RNA, respectively" (Hata et al., 2016). Oysters can bioaccumulate high concentrations of enteric viruses after a few hours (Flannery et al., 2012; Pilotto et al., 2019) which promotes rapid contaminated with potentially several hNoV strains. Optimal reduction of norovirus load was previously reported in the region of 1 log<sub>10</sub> (McLeod et al., 2017). Other experiments have reported marginal improvements in norovirus reductions, particularly using elevated seawater temperature (>11 °C) during depuration (Rupnik et al., 2021).

Oysters retain smaller particles, such as hNoV that bind to these particles depending on their isoelectric point (McLeod et al., 2017). However, it is also appreciated that the recognition of hNoV persists for longer periods than bacteria where oysters are depurated due to specific hNoV-ligand mechanisms (McLeod et al., 2017) as also reported in humans (Hutson et al., 2002). This is one of the main reasons for viral persistence in ovsters compared to bacteria where contaminated shellfish can retain virus copies for weeks or months after initial exposure; thus, acting as a reservoir for foodborne transmission (Maalouf et al., 2010; Mathijs et al., 2012). For example, hNoV GI.1 strain binds to the midgut and digestive diverticula of Pacific oysters, but not to other tissues (Le Guyader et al., 2012). While hNoV GII strain binds to various tissue types including digestive diverticula, midgut (intestine), gills, mantle, and labial palps, McLeod et al. (2017) has provided a comprehensive review of factors affecting hNoV binding in oysters including seasonal variations where these authors note that published findings support strain-specific variations in hNoV binding patterns. Oysters retain hNoV through specific ligand-binding in tissues that affects selective accumulation and persistence that may help explain their long retention in oysters (as observed in depuration, McLeod et al., 2017). However, there appears to be a lack of published studies on the ability of oysters to specifically bind surrogate viruses that would inform the comparative ability for infectious viral removal or deactivation at depuration. Hunt et al. (2023) advocated the need for new monitoring methods and regulatory regimens for the specific hazard of enteric viruses in oysters to better manage this risk. EFSA (2019) recommends better understanding of the actual risk associated with positive hNoV test results from contaminated production areas for all genogroups. However, Hunt et al. (2023) stated that "where to apply quantitative thresholds remains a core question in addition to what post-harvest or other interventions could be applied to further manage the risk of norovirus contamination in oysters. To determine the best methods for controlling shellfish virus risk is currently a matter of live discussion in the EU".

Consequently, this constitutes the first review to compare the effectiveness of different established and emerging technologies affecting the decontamination of enteric and surrogate viruses in live oysters. It also considers other complementary novel approaches for the real-time detection of viable and infectious hNoV strains to that of using a standard genomic method (RT-qPCR) that may potentially inform their effectiveness and advance studies underpinning the virological safety of shellfish, particularly addressing oysters.

# 2. Methods for human norovirus removal or destruction in contaminated live shellfish

A review of publications from PubMed and Scopus database over period 1981 to 2023 was used to address this research topic using PRISMA guiding framework. Of the combined keywords, norovirus (n = 7781) and "depuration" (n = 50), 43 publications were deemed eligible based on criteria: on the follow; (a) norovirus detection, persistence, and accumulation in shellfish, (b) decontamination technologies and operational factors affecting effectiveness for live shellfish applications; (c) use of surrogate microorganisms; (d) kinetic inactivation modeling and (e) foodborne transmission outbreaks. Seven publications were excluded for the reason that they addressed post processing of non-living shellfish; use of chlorine and heating in shellfish, food distribution chain and studies on norovirus aerosols in wastewater production plants. Combining "depuration"-based publications with "enteric viruses" (n = 10,305) over the period 1988 to 2023 revealed 22 matching publications where 6 were different to the aforementioned "norovirus" publication list. These addressed treatment of hepatitis A virus (HAV) in mussels using a closed circulatory system, treatment of the surrogates HAV, poliovirus type 1 and coliphage MS2 in shelled clambs, public health implications of viral-contaminated mussels, and development of diagnostic tests. One study was excluded for the reason that it addressed virological control of contaminated ground water that is not the subject of this review.

Decontamination of live shellfish is mainly carried out at the commercial depuration phase post-harvest using clean seawater where in-line or bolt-on solutions must be non-thermal in nature to mitigate against damage to the treated live ovsters (Rupnik et al., 2021). "Acceptable post-harvest treatments available to ensure oysters meet the E. coli standard include self-purification in land-based tanks containing clean seawater by a process called depuration or relaying bivalve shellfish to clean marine locations for an extended period (four weeks)". Norovirus-related gastroenteritis outbreaks have occurred even when oysters have been demonstrated to be fully compliant with regulatory end-product standards. Therefore, the combination of marine harvest area controls and post-harvest treatments as currently practised does not completely protect consumers from the virological safety risk associated with norovirus-contaminated shellfish (Rupnik et al., 2018). Findings from various researchers would support the idea that shorter, less-intense solar irradiation in the winter (such as in Northern hemisphere countries) may also contribute substantially to the environmental persistence of human noroviruses in shellfish (Younger et al., 2020; Rupnik et al., 2021).

Typically, solutions for live shellfish decontamination must consider inter alia: the volume of live seafood products to be treated; the cellular and molecular mechanism(s) of action of the applied decontamination approach(es) ensuring irreversible inactivation, or removal; the microbial load or initial starting population including composition, such as the type of pathogen(s) present as these frequently differ in behaviour and level of susceptibility to decontamination methods, particularly hNoV (Fig. 1); the amount of decontaminant applied or intensity of removal/disinfection methods; the exposure or treatment time (hours, days); the appropriate inactivation kinetic performance of applied decontamination approaches (log linear, or bi-/tri-phasic adaptive microbial responses); the environmental parameters (seawater depuration temperature, salinity, pH, presence of interfering suspended solids or turbidity), nutritional factors (oysters are consumed raw; thus nutritional and organoleptic characteristics must be maintained post non-thermal treatments); scalability; affordability; the availability of subject-matter technical assistance for equipment operation; operator safety (UV-irradiation), detection and infectivity enumeration methods; and biocompatibility post treatments including environmental (ecotoxicological) compliance. The sequence of susceptibility of treated microorganisms (and surrogates) to applied disinfection technologies, as shown in Fig. 1, is indicative as it is appreciated that the mode of action of biocidal action may differ depending upon the type of treatment methods. For example, specific hNoV strains (such as G1) may generally be more tolerant to a variety of technologies as specifically bind to ligands in oyster tissues that may confer greater protection (Leduc et al., 2020; Rupnik et al., 2021) (Table 1); thus, they are potentially more recalcitrant compared to treating similar hNoV genotypes under less complex planktonic situations as often replicated in laboratories. Waterborne parasites (such as Cryptosporidium parvum oocysts or Giardia lamblia cysts) appear to exhibit greater tolerance to reactive oxygen species (ROS) associated with advanced oxidative processes (AOP) yet appear to be more susceptible to UV-irradiation (Garvey et al., 2014a; Garvey et al., 2014b). However, combined use of UV/H<sub>2</sub>O<sub>2</sub> appears more effective for microbial disinfection compared to separately using the treatment technologies.

Effective post-harvest processing technologies that are recognized by regulators (such as the US FDA) have had limited acceptance in the domestic industry due to the high initial capital equipment costs and the

Science of the Total Environment 874 (2023) 162380



Fig. 1. Increasing microbial resistance to decontamination methods.

economics of transporting and storing shell oysters. While advances have been made to try to standardize or harmonize non-thermal disinfection technologies for food-based processes including packaging (Gómez-López et al., 2022), there is a marked knowledge gap in the development and reporting of appropriate treatment technologies (used alone or combined) for commercial-scale seafood applications, particularly, for technologies or approaches that can address complex challenges including enteric viruses, parasitic oocysts, algal toxins and other potential contaminants of concern (McLeod et al., 2017; Fehrenbach et al., 2022). However, the majority of physical and chemical disinfection technologies used in food and adjacent medical device processing industries would not be deemed appropriate for live shellfish decontamination. For example, researchers have highlighted similar challenges to effectively process and sterilize heatsensitive medical devices containing complex materials; however, such Medtech-approved chemical disinfectants (such as glutaraldehyde), and terminal sterilization modalities (gamma, x-ray, electron beam, ethylene oxide are not appropriate for use in seafood depuration due to killing of live shellfish, cost of equipment, and general non-practicalities. However, the use of vaporized hydrogen peroxide by the medical device industry may have a future role in commercial shellfish decontamination. In addition, there is a pressing commensurate need to establish the safety of new technologies including their environmental impact. For example, Hayes et al. (2013) reported that pulsed-plasma gas-discharge (PPGD) in oxygen-sparged water generated a range of short-lived highly-oxidative biocidal properties killing enteric bacterial pathogens and recalcitrant waterborne parasitic oocysts. This PPGD technology has been applied for contact food surface decontamination studies (Rowan et al., 2007, However, despite demonstrating biocidal efficacy, this PPGD system was reported to produce unwanted toxicological endpoints post treatments making it unsuitable for the intended use (Hayes et al., 2013). This is also relevant given scalability issues surrounding the transition from early discovery (labbased) to full commercial testing of technologies at depuration where there are limited published studies addressing shellfish decontamination. Such an innovative disease mitigation topic is underappreciated when considering the combined technological (TRL), societal (SRL) and policy readiness level (PRL) framework for evaluating new decontamination approaches ranging from discovery to full commercial deployment (as exemplified by Rowan and Casey, 2021), particularly for safely and affordably treating live shellfish. Given the amount of key governing factors, complexity of data generated and the need to achieve real time outcomes for effective decision-making (Naughton et al., 2020), potential practical

disease mitigation solutions will be informed, supported and enabled by digital technologies (Rowan, 2022; Rowan et al., 2022).

## 2.1. Using conventional low-pressure UV irradiation and filtration methods

Reference to the use of UV has appeared in 198,506 journals since 1946 to present day (combined PubMed and Scopus), where 117 journals combine UV with norovirus. A total of 48 papers were excluded from consideration in this review as they focused on the combined use of UV with other approaches (such as heating,  $H_2O_2$  generating wipes, peracetic acid, TiO2) for treating strawberries, onions, and lettuce or treating reclaimed water a golf course, simulating vomiting, SARS-CoV-2 (COVID-19) applications, therapeutic applications for intestinal microbiome, solar radiation, electrochemical paper-based devices, and chlorination

UV processes have been developed for food and water disinfection applications and have been divided into three main categories (a) low pressure/low intensity lamps (b) low pressure/high intensity (c) medium pressure/high intensity lamps (Fitzhenry et al., 2021). Barrett et al. (2016) noted that these UV systems vary with respect to operating pressure and output level; for example, it is low and medium pressure assets that are deployed for treating wastewater. Also, turbidity and suspended solids can affect UV disinfection efficiency by decreasing transmissivity (i.e., the transmission of UV light through the water body) (Barrett et al., 2016). Yet, these factors are not routinely considered when monitoring UV performance. UV light technologies have supported the effective treatment of drinking water for decades. Low pressure (LP) UV disinfection results in photochemical damage to viral RNA thus inhibiting viral reproduction (Fitzhenry et al., 2021). Advances have been recently made in the reduction of norovirus in contaminated oysters at the depuration phase using filtration and LP-UV (Rupnik et al., 2018; Leduc et al., 2020; Younger et al., 2020; Rupnik et al., 2021) (Table 1).

Previous researchers have reported that depuration time and seawater temperature are parameters that may influence human viral decontamination efficacy (Lees and Cen, 2010; Rupnik et al., 2018), but these parameters are not currently stipulated in EU regulation (Rupnik et al., 2021). Rupnik et al. (2021) reported that laboratory-based studies revealed up to 74 % of initial norovirus GII concentrations were achieved after three days of 17–21 °C [designated 'high' depuration temperature], and after 11 °C to 15 °C [designated 'medium' depuration temperature], compared to 44 % reductions at 7 °C to 9 °C [designated 'low' depuration temperature] where artificial seawater (mimicking water salinity of estuary) was treated

#### Table 1

Examples of the application of LP-UV or Pulsed UV irradiation for the decontamination of norovirus with relevance to shellfish.

Target	Source	Treatment	Method	Characteristics	Duration	Pre-depuration	Post-depuration Log10 r	eduction	Reference
NoV GI	WWTP effluent	LP-UV	RT-qPCR*	Flow through 60s, 120 S	24 h	10 <sup>6</sup> /100 mL	$0.09 \pm 0.04$ to 0.52	$0.15 \pm 0.12$ to 0.7 + 0.6 [120 c]	Barrett et al.
NoV GII	WWTP effluent	LP-UV	RT-qPCR*	Flow through 60s, I20 S	24 h	10 <sup>6</sup> /100 mL	$-0.11 \pm 0.55$ to 0.4	$\pm$ 0.0 [120 s] -0.11 $\pm$ 0.65 to 0.7	Barrett et al.
FRNAPH	WWTP effluent	LP-UV	RT-qPCR	Flow through 60s; 120 s	24 h	106/100 mL	$\pm$ 0.5 [60s] -0.43 to 1.07 $\pm$ 0.27	$\pm$ 0.1 [120 s] -1.27 $\pm$ 2.31 to 1.1	(2016) Barrett et al.
FRNAPH	WWTP effluent	LP-UV	Plaque	HRT (0.23 m <sup>3</sup> /h	24 h	10 <sup>5</sup> PFU/mL	Ca 2 log	± 0.2 [120 8]	Barrett et al.
FRNAPH	WWTP effluent	LP-UV	Plaque	HRT (0.52 m <sup>3</sup> /h	24 h	10 <sup>5</sup> PFU/mL	Ca 2 log		Barrett et al.
FRNAPH	WWTP effluent	LP-UV	Assay Plague	HRT (0.85 m <sup>3</sup> /h	24 h	105 PFU/mL	Ca 1.5 log		(2016) Barrett et al.
FRNAPH	WWTP effluent	PUV	Plague		24 h	10 <sup>7</sup> PFU/mL	1.24 to 2.21		(2016) Barrett et al.
FRNAPH	WWTP effluent	PUV	Plague		24 h	10 <sup>7</sup> PFU/mL	1.4 to 3.14 $\pm$ 0.03		Barrett et al.
FRNAPH	WWTP effluent	PUV	Plague	120 s [900 V, 1 pps]	24 h	10 <sup>7</sup> PFU/mL	1.3		Barrett et al.
FRNAPH	WWTP effluent	PUV $0.55 \text{ L/cm}^2$	Plague	120 s [600 V, 1 pps]	24 h	10 <sup>7</sup> PFU/mL	1.0		Barrett et al.
FRNAPH	WWTP effluent	PUV	Plague	HRT 120 s [300 V, 1 pps]	24 h	10 <sup>7</sup> PFU/mL	0.4		Barrett et al.
FRNAPH	WWTP effluent	PUV	Plague	60s, SS (57.5 mg/L); ToC	24 h	10 <sup>7</sup> PFU/mL	$2.2~\pm~0.2$		Barrett et al.
FRNAPH	WWTP effluent	PUV	Plague	120 s, SS (57.5 mg/l); ToC	24 h	$10^7  \text{PFU}/\text{mL}$	$3.1 \pm 0.2$		Barrett et al.
FRNAPH	Unfiltered WWTP offluent	PUV	Plague	60s	24 h	10 <sup>6</sup> PFU/mL	2.3 (Site 1)	Barrett et al. (2016)	Barrett et al.
FRNAPH	Unfiltered	PUV	Plague	90s	24 h	10 <sup>6</sup> PFU/mL	<1 PFU/ml (6 log) Site	Barrett et al. (2016)	Barrett et al.
NoV GI	G. gigas	Not stated	RT-PCR [LoQ 100	Commercial Production Site A 12–16 °C	7 d	$251~\mathrm{gcg}^{-1}$	121 (0.4 log or 67.5 % r	reduction)	Rupnik et al., 2021.
NoV GI	G. gigas	Not stated	RT-PCR [LoQ 100	Commercial Production Site B 18 °C	3 d	$241 \text{ gcg}^{-1}$	129 (0.4 log or 66.4 % r	reduction)	Rupnik et al., 2021.
NoV GII	C. gigas	Not Stated	RT-PCR [LoQ 100	Commercial Production Site A 12–16 °C	7 d	$281 \text{ gcg}^{-1}$	45 (0.79 log or 83.95 % reduction)		Rupnik et al., 2021.
NoV II	C. gigas	Not Stated	RT-PCR [LoQ 100	Commercial Production Site B 18 °C	3 d	$526 \text{ gcg}^{-1}$	66 (0.90. log or 87.47 re	eduction)	Rupnik et al., 2021.
NoV GII	C.gigas	LP-UV (3 W) No fluence	RT-PCR [LoQ 100	Lab; D.O. 80 % 8 °C, 2000 L/h	1–7 d	290 gcg <sup>-1</sup> [example]	310 gcg <sup>-1</sup> (d1); 284 (d2 249 (d5); 147 (d6); 225	2); 142 (d3); <loq (d4);<br="">(d7)</loq>	Rupnik et al., 2021
NoV GII	C.gigas	LP-UV (3 W) No fluence	RT-PCR [LoQ 100	Lab; D.O. 80 % 12 °C, 2000 L/h	1–7 d	452 gc/g <sup>-</sup> (example)	143 gcg <sup>-1</sup> (d1); 231 (d2 <219 (d5);181 (d6); <le< td=""><td>2); 202 (d3); &lt; LoQ (d4); oQ (d7)</td><td>Rupnik et al., 2021</td></le<>	2); 202 (d3); < LoQ (d4); oQ (d7)	Rupnik et al., 2021
NoV GII	C.gigas	LP-UV (3 W) No fluence	RT-PCR	Lab; D.O. 80 % 18 °C. 2000 L/h	1–7 d	$405~gc/g^-$	236 (d1); 147 (d2); <lo (d5) &lt; LoO (d6); <loo< td=""><td>Q (d3); <loq <="" d4)="" loq<br="">(d7)</loq></td><td>Rupnik et al., 2021</td></loo<></lo 	Q (d3); <loq <="" d4)="" loq<br="">(d7)</loq>	Rupnik et al., 2021
NoV GII	C.gigas	LP-UV (3 W) No fluence	RT-PCR [LoQ 100	Lab; D.O. 80 % 20 °C, 2000 L/h	1–7 d	$290 \text{ gcg}^{-1}$ [example	165 (d1); <loq (d2);="" <l<br="">(d5); <loq (<="" (d6);="" <loq="" td=""><td>.oQ (d3); <loq (d4);="" <loq<br="">(d7)</loq></td><td>Rupnik et al., 2021</td></loq></loq>	.oQ (d3); <loq (d4);="" <loq<br="">(d7)</loq>	Rupnik et al., 2021
NoV GII	C.gigas	LP-UV (3 W) No fluence	RT-PCR [LoQ 100 gcg <sup>-1</sup> ]	Lab; D.O. 80 % 20 °C, 2000 L/h	1–7 d	<1000  gc/g 5573 gcg <sup>-1</sup> [example >1000 gc/g]	4670 (d1); 5896 (d2); 2 (d5); 1513 (d6); 2054 (d	296 (d3); 2414 (d4); 1769 17)	Rupnik et al., 2021
NoV GI	Oyster digestive tissue	LP-UV (80 to 90 mJ/cm <sup>2</sup> )	RT-qPCR	10 °C	1 to 43 d	$293 \text{ gcg}^{-1}$	47.8 days (Time for 1 lo material)	g reduction in genomic	Leduc et al., 2020
NoV GI	Oyster digestive tissue	LP-UV (80 to 90 mJ/cm <sup>2</sup> )	RT-qPCR	10 °C	1 to 43 d	$802 \text{ gcg}^{-1}$	26.7 days (Time for 1 lo material)	g reduction in genomic	Leduc et al., 2020
FRNAPH	Oyster	LP-UV (80 to 90 mJ/cm <sup>2</sup> )	RT-qPCR	10 °C	1 to 43 d	463 gcg <sup>-1</sup>	43.9 days (Time for 1 lo material)	g reduction in genomic	Leduc et al., 2020
NoV GI	Oyster	LP-UV (80 to 90 mJ/cm <sup>2</sup> )	ICC- qRT-PCR	10 °C	1 to 43 d	$463 \text{ gcg}^{-1}$	26.7 days (Time for 1 lo particles)	g reduction in infectious	Leduc et al., 2020

Low Pressure UV (LPUV); Pulsed UV (PUV); Suspended Solids (SS); Total Organic Carbon (TOC), Total Inorganic Carbon (TIC), Waste Water Treatment Plant (WWTP).

with a 36 W UV-C lamp (fluence was not reported) (Table 1). Rupnik et al. (2021) determined norovirus GI and GII concentrations in contaminated oysters using standardized quantitative real-time reverse transcription PCR (RT-qPCR) in accordance with ISO, 2017. Seawater was circulated and maintained within 1 °C of the target depuration temperature and bar sprinklers were deployed for aeration where dissolved oxygen levels were kept in excess of 80 % in all trials. "Oysters were analysed for the presence of norovirus GI and GII before depuration where norovirus GI concentrations in environmentally contaminated oysters ranged from 178 to 16,426 norovirus gc/g (noting, the limit for quantification [LOQ] and

limit of detection [LOD] for Nov GII were  $\leq 100$  genome copies/g and 20 genome copies/g respectively)" (Rupnik et al., 2021). Results showed that heavily initial norovirus GII concentrations ( $\geq$  850 genome copies/g) remained above 300 genome copies/g in oysters irrespective of depuration water temperature after seven days of treatment (Rupnik et al., 2021). For example, samples were reduced from 1739 to 1248 norovirus GI concentration (genome copies/g) at 14 °C [medium temperature]; and from 5573 to 2054 norovirus GII concentration (genome copies/g) at 20 °C [high temperature], after seven days depuration. The authors noted that the ability to reduce hNoV concentration in oysters to <LOQ (100 genome copies/g, or 2

log hNoV g<sup>-1</sup> potentially remaining) using this LP-UV treatment method differed when contaminated below or above 1000 genome copies/g (or 4 log hNoV g<sup>-1</sup> present).

These important findings have potentially profound implications for determining the choice of disinfection technology for complementary or alternative use to LP-UV to improve human NoV removal or inactivation efficiencies in oysters that can be informed by viral kinetic inactivation modeling data (Rowan et al., 2015). In addition, the potential use of viral monitoring and screening approach for contaminated oysters (≤1000 NoV genome copies/g) may inform suitability for achieving appropriate levels of disinfection treatment based on current disinfection methods at depuration, or for relaying to clean seawater for four weeks (>1000 genome copies/g) from a risk-based assessment and virological safety perspective. However, the RT-qPCR detection method was solely used by Rupnik et al. (2021) where there is no infectivity assay for NoV; therefore, it is uncertain if the genomic copies of NoV per gram detected in oyster post these LP-UV treatments were viable where the use of ISO 15216-1 standard may underestimate the level of NoV lethality or removal achieved. It is appreciated that the combinational laboratory-commercial decontamination study of Rupnik et al. (2021) has made a significant contribution to advancing this technical NoV challenge in oysters. This is timely given that the EFSA has published a baseline survey on comprehensive scientific data for norovirus prevalence in European oyster harvesting areas that will inform future safety limits on norovirus concentration in oysters.

Rupnik et al. (2021) also noted that norovirus reductions were also assessed in two Irish commercial depuration systems that are routinely used to produce oysters. The authors reported up to 68 % reduction for hNoV GI and up to 90 % for hNoV GII reduction. This finding also supports the general observation by other researchers that intimates hNoV GI exhibits greater tolerance or persistence over hNoV GII strain possibly due to the specific behaviour of ligand binding in oyster tissue (Leduc et al., 2020; Younger et al., 2020). Additionally, other researchers have also deployed standard UV decontamination for removing norovirus and FRNAPII from oysters during depuration using genome and infectivity assays (Leduc et al., 2020; Younger et al., 2020). Specifically, these studies intimate that standard UV irradiation of seawater under aerated depuration circulatory conditions either inactivates human norovirus GI and GII genogroups (Leduc et al., 2020 [UV dose, 80 to 90 mJ/cm<sup>2</sup>], or, destroys and removes this enteric virus (Younger et al., 2020  $[2 \times 25 \text{ W lamps}]$ , no UV dose was reported]); thus, highlighting that greater studies are required to understand the mechanism of viral reduction and residual potential to cause infection at low concentrations (Table 1). Younger et al. (2020) found approximately 46 % removal of hNoV GII at 18 °C after two days and 60 % after five days compared with a maximum of 16 % hNoV GI removal. These researchers noted that "twice the rate of NoV GII removal was achieved at 18°C compared with 8°C after five days. Younger et al. (2020) also found that FRNAP-II was more readily removed than noroviruses. Notably, no significant difference was found between the rate of removal (as measured by RT-qPCR) and inactivation (as measured by bioassay) of FRNAPII". Younger et al. (2020) inferred from their results that the reduction in FRNAPII may be primarily due to physical removal (or destruction) rather than in situ inactivation of the virus. Also, the efficacy of RT-qPCR method to confirm the viability of the remaining norovirus post treatments remains questionable given that this molecular approach does not distinguish between live or dead viruses. Leduc et al. (2020) reported that FRNAPII infectivity bioassay (presence of viable phage) may be appropriate for informing disinfection effectiveness of viable hNoV given that a 1 log reduction in FRNAPII infectivity occurred after 20.6 days treatment using LP-UV (80 to 90 mJ/cm<sup>2</sup>) compared to its genome (43.9 days) and NoV GI genome (47.8 days). Leduc et al. (2020) reported that "FRNAPII and NoV genomes may display similar behaviours with low kinetic removal from the oysters under all purification conditions tested". In terms of surrogate representation of hNoV, Lowther et al. (2019) reported that "both viruses were in high concentrations in outbreak-related samples and that infectious FRNAPII were detected in all outbreak samples (n = 9)". Leduc et al. (2020) and Lowther et al. (2019) suggest combining RT-qPCR testing

with a test for infectious FRANPII detection in order to improve the hNoV risk assessment in shellfish. Leduc et al. (2020) also highlighted the importance of developing appropriate testing to identify new strategies for the effective elimination of hNoV, as addressed in this review. FRNAPII bacteriophages have been studied for shellfish decontamination due to their structural similarity to waterborne viruses and proof of faecal pollution coming from urban areas (Leduc et al., 2020).

Recent results underscore the importance of determining the typical range of norovirus prevalence and concentration in contaminated oysters to deploy appropriate decontamination methods. However, the best practice would be 'a multi-actor approach to strategically manage this problem by working with local government authorities and the water industry to reduce the occurrence of human enteric viruses in sewage effluent that may be discharged into the same intertidal growing areas where commercial oyster production occurs. Thus, placing greater emphasis on disease prevention, rather than relying on introducing complex removal strategies at the depuration phase for these enteric viruses in live oysters where there are currently limited appropriate solutions.

Jeong et al. (2021) also reported that after 60 h of depuration equipped with a standard UV light source *Vibrio vulnficus* cell numbers were reduced by <4.0 log MPN/g in Pacific oyster tissue from an initial population of ca. 8 log MPN/g. Lee (2020) reported that the use of standard UV for treating seawater in depuration tanks could extend the shelf life (two to three days) of raw oysters with minimal changes in food quality with faecal coliforms maintained at or below 2 log/g compared to non-depurated and generally packaged oysters. Depuration at temperatures between 7 °C and 15 °C using UV-treated seawater reduced *V. parahaemolyticus* populations in oysters by >3 log MPN/g after five days with no loss of live oysters (Phuvasate et al., 2012). The US National Shellfish Sanitation Program established time/temperature regulations that limit maximum hours of holding shellfish from harvest to refrigeration ( $\leq$ 10 °C) to reduce the risk of infections from *Vibro* spp. associated with shellfish consumption (Reid and Durance, 2000).

# 2.2. Pulsed light (PL) technology

Only 6 of 4239 "Pulsed UV" and "Pulsed Light" publications [2010 to 2023] have focused on treating norovirus or enteric viruses when assessing PubMed and Scopus databases. One study was excluded for the reason that it focused on PPE decontamination. Ten publications were also included that provided supporting context to background technology. Pulsed light (PL) technology is an exciting approach that delivers ultra-short bursts of broad spectrum (200 nm to 1100 nm light) (Rowan, 2019) and is commercially deployed by companies such as Claranor (France, https://www. claranor.com/en/) for food packaging sterilization with over 500 units installed in 52 countries. The benefits of using PUV reflect the potentially ultra-short disinfection against viruses, waterborne protozoan parasite oocysts (Cryptosporidium parvum) (Garvey et al., 2014a, 2014b) and cysts (Giardia lamblia) (Garvey et al., 2014b), and biological endospore indicators using laboratory-based static or limited flow-through treatment configurations (Garvey et al., 2010; Rowan, 2011; Hayes et al., 2012; Garvey et al., 2013; Garvey and Rowan, 2015). PL has been referred to as highintensity pulsed UV light (HIPL), pulsed UV (PUV), high-intensity broad spectrum UV light (BSPL), intense light pulsed (ILP) and pulsed white light (PWL) (Rowan et al., 1999; Rowan, 2019). PL has been approved by the US FDA in the production, processing and handling of foods since 1996 up to cumulative UV dose (or fluence) of 12 J cm<sup>-2</sup> where emission spectra are to be kept between 200 and 1100 nm and pulsed duration  $\leq$  2 ms (Rowan, 2019). The technological principle of PL disinfection is based upon the accumulation of high discharge voltage in a capacitor where the stored energy is delivered in ultra-short pulses through a light source filled with xenon gas. The xenon-light source emits a broad spectrum light flash typically in the range of ca. 200 to 1100 nm with approximately 25 % in the UV range (Gómez-López et al., 2022). PL disinfection efficiency is higher compared with continuous-wave low-pressure UV irradiation (CW-UV) due to its high peak power along with the ability to deliver stored

energy over short durations, typically 1 to 10 pulses per second (Garvey and Rowan, 2019). The main reporting parameters governing effective PL operation for disinfection are the fluence [J cm<sup>-2</sup>], exposure time [s], number of pulses applied [n], pulse width [ $\tau$ ], frequency [Hz], and the peak power [W] (Rowan et al., 2015; Gómez-López et al., 2022). Garvey et al. (2015) reported on a satisfactory ecological assessment of pulsed UV light treated water containing microbial species and *Cryptosporidium parvum* using a microbiotest test battery.

Very limited studies have been conducted on the inactivation of pulsed light for inactivating enteric viruses that have focused on bench-scale applications (Barrett et al., 2016). Barrett et al. (2016) compared the efficiency of pulsed light irradiation and low-pressure UV irradiation as a means of hNoV and FRNA bacteriophage using secondary treated wastewater effluent. While hNoV GI and GII inactivation could not be determined, it was found that a maximum UV dose of 6.9 J/cm<sup>2</sup> (at a hydraulic residence time of 120 s in a flow-through system) achieved 2.4 log <sub>10</sub> reduction of FRNA bacteriophage, which indicates the need for high pulsed UV doses to fully remove NoV (Table 1). Vimont et al. (2015) reduced murine NoV using pulsed UV by 3 log<sub>10</sub> in <3 s (fluence, 3.45 J/cm<sup>-2</sup>), while Jubinville et al. (2022) reported on the efficacy of pulsed UV against HAV on berries. Three studies have discussed the potential of using pulsed UV for enteric-virus decontamination on foods or for environmental applications (Jean et al., 2011; Pexara and Govaris, 2020).

All pulsed UV dosage rates related to wavelengths <300 nm and the average initial FRNAPII concentration was 10<sup>6</sup> PFU/mL (Barrett et al., 2016). The authors also found that increasing concentration of suspended solids impacted PUV disinfection efficiency. The use of LP-UV reduced FRNAPII phage by ca. 2  $\log_{10}$  using a significantly reduced UV dose of 31 mJ/cm<sup>2</sup>; however, the combined use of LP-UV and pulsed light for enteric virus reductions were not considered. Results also indicate that absorption of viral particles to solids in wastewater occurs; therefore, it would be prudent to also introduce a barrier process such as a tangential flow filtration system that filters the particulate matter and the flow through when using UVirradiation. Interestingly, PUV was capable of achieving up to 3 log10 reduction in FRNAPII infectivity within 24 h treatment under varying conditions (such as flow rates [120 s], total suspended solids [57.5 mg/L]) in unfiltered secondary wastewater effluent with enhanced destruction (up to 6 log10 PFU/ml) achieved in filtered secondary wastewater effluent (Table 1). This finding contrasts with other previously reported LP-UV studies that required up from 3 to 47 days to achieve a l log reduction in NoV and FRANPII in shellfish during depuration (Table 1). Thus, there is potential to augment enteric viral destruction in contaminated shellfish using a combinational LP-UV and PUV approach. PUV offers the benefits of delivering a broad high-intensity light spectrum that includes wavelengths in the blue light spectrum (such as 406 nm) that has been previously reported to generate hydroxyl radicals from oxygen in water via an advanced oxidation process (AOP) (Kingsley et al., 2018). Barrett et al. (2016) also noted that as RT-qPCR provided inconsistent results; therefore it was deemed not an appropriate method for assessing the inactivation of NoV and FRNAPII via pulsed light after treatments. Other researchers also observed this discrepancy (Pecson et al., 2011). Baert et al. (2009) found RT-qPCR results were unable to distinguish between infectious and non-infectious NoV using murine norovirus (MNV) as surrogate post-heat treatments. Uslu et al. (2016) employed PUV as a wastewater disinfection tool and indicated that in addition to pathogen removal/inactivation, it also reduces the organic load of municipal wastewater effluent by reducing chemical oxygen demand and total organic carbon.

However, Fitzhenry et al. (2021) recently reported that LP-UV is superior to that of using pulsed light for converting energy through the light source to UV dose for submerged wastewater-treatments. Thus, advances in PUV design is required (such as improving light source including the use of light emitting diodes, introducing more light sources such as in parallel; reflective surfaces for light scattering, and inclusion of smart materials that includes photocatalysis of TiO<sub>2</sub> for localized ROS generation) in order to realize the dual benefits of combining PUV (broad spectrum) and fixed-wavelength LP-UV for improved decontamination effectiveness. The

authors report that a pulsed UV system output of 2052  $mJ/m^{2-}$  (energy below 300 nm) was required for a 2 log inactivation of Bacillus pumilus, where low a lower LPUV system output of 12 mJ/cm<sup>2</sup> produced a similar level of inactivation in flow-through water systems. Complete inactivation of B. pumilus was achieved via LP-UV disinfection using a UV output of 30 mJ/cm<sup>2</sup>. Fitzhenry et al. (2021) reported that "while a typical xenon gas PUV system is comprised of light emissions within the broad spectrum range of UV, and infrared light, it may be an important consideration to prioritize UV dose/output calculations in terms of 'biocidal PUV dose/out' (i.e., the energy applied from wavelengths below 300 nm ahead of the 'total PUV dose/output', which infers the total energy applied across the whole spectrum output). This has been previously demonstrated in a number of studies with the aid of spectrometer/pyroelectric detectors and in some cases the UV dose/output from PUV systems in within the same order of magnitude as LPUV dose/outputs such as 1 – 100 mJ/m<sup>2"</sup>. However, measurements from Fitzhenry et al. (2021) indicated that only 26 % of the lamp energy reached the same sample (at 900 V and a distance of 10.75 cm), and of that, only 8 % was within the UV wavelength range.

Interestingly, the PUV and flow-through treatment configuration used by Fitzhenry et al. (2021) is the same system that was used by Barrett et al. (2016) to inactivate NoV GI and GII and FRNAPII phage with moderate success (such as 2.4 log reduction with 6.8 J/cm<sup>2</sup>) (Table 1). This outcome could be improved by replacing the xenon source with a specific wavelength light-emitting diode (LED). For example, Wen et al. (2022) report UV-LEDs are safe algicidal technologies for inactivating the marine microalgae *Tetraselmis* sp. These authors showed that the wavelength of 265 nm exhibited maximum inactivation efficiency, whereas 285 nm achieved optimal energy efficiency. UV irradiation is also affected by turbidity where viruses can be protected on particles, which must also be considered at the depuration stage, particularly in developing countries where water quality per se may not be appropriate for depuration.

Garvey and Rowan (2015) reported a reduction of 4.23  $\log_{10}$  in *Bacillus megaterium* vegetative cells using a UV dose of 6.48 µJ/cm<sup>2</sup> with no significant further microbial reduction after doubling the UV dose to 12.96um in a flow-through PUV system at a retention time of 60s and flow rate of 30 L/h. However, only a 1.48  $\log_{10}$  and 1.43  $\log_{10}$  a reduction occurred of *B. megaterium* and *B. cereus* endospores respectively at 12.98 µJ/L (RT 60s), indicating reduced efficacy of PUV in treating recalcitrant pathogens in submerged flow-through treatment configuration. The presence of inorganic contaminants did not significantly reduce PUV efficacy at the concentrations used.

## 2.3. Potential combined use of advanced oxidative processes (AOPs)

Of the 1764 published studies focusing on advanced oxidation processes or "AOPs" appearing in PubMed and Scopus databases over period 1981 to 2023, only 3 addressed norovirus. The reason for including AOPs in this review reflected the emergence of AOPs as non-thermal approaches to improve and enable decontamination performances for food treatments and for environmental applications. One was excluded for the reason that it was COVID-19 orientated. However, there is an increased sharing of knowledge on the potential for using advanced oxidation processes (or AOPs) for enteric viral destruction. This aligns with the need for new or more efficient methods that will destroy NoV in shellfish (Gerba et al., 2018; Kokkinos et al., 2021). AOPs relay on the in situ formation of chemical oxidants to disinfect liquids and degrade diverse harmful organic contaminants (Shabat-Hadas et al., 2017). Kokkinos et al. (2021) noted that "AOPs are, in practice, redox technologies that encompass different processes such as ozonation, ozonation coupled with hydrogen peroxide, and/or UV radiation, Fenton and alike reactions, photocatalysis activated by semiconductors, sonolysis, electrochemical oxidation, and various combinations of these." They are based on the generation of highly reactive oxygen species (ROS), characterized by the non-selectivity of the target and can be deployed before or after treatment of a biological process (Galeano et al., 2019).

The principle oxidizing agent is the hydroxyl radical; however, other ROS may be produced including hydroperoxyl radicals and superoxide radical anions (Shabat-Hadas et al., 2017). Kokkinos et al. (2021) noted that photo-Fenton AOP in which hydroxyl radicals are produced from light, iron and H<sub>2</sub>O<sub>2</sub> is a well-studied, environmentally-friendly, simple, lowcost process that inactivates complex or resistant microorganisms (Giannakis et al., 2017). Enteric viruses and a wide range of recalcitrant microorganisms are inactivated through the action of ROS such as singlet and triplet oxygen, anion-radical superoxide, hydroxyl and hydroperoxyl radical, and H<sub>2</sub>O<sub>2</sub>. ROS are recognized oxidants for inactivating a variety of molecules including proteins, lipids and nucleic acids. "When considering the lethal action of ROS on treated enteric viral nucleic acids, this AOP can change the nucleotides, break important phosphodiester bonds, enhance the formation of pyrimidine dimers, change the tri-dimensional structure, and affect RNA replication" (Kokkinos et al., 2021). Recently, photo-Fenton and alike processes have been developed as a 'green' alternative to chemical disinfection (such as the use of chlorine) for water and wastewater applications (Giannakis, 2018). Evaluation of the disinfection efficacy using light-mediated AOPs has been based on using varying methods such as computational fluid dynamics, chemical actinometry or biodosimetry (Shabat-Habas et al., 2017). An understanding of these assessment methods is critical to informing the efficacy of shellfish decontamination processes from a standardization, reliability and repeatability perspective (as per Rowan, 2019).

The utility of UV technologies has been enhanced through the combinational use of AOPs over the last few decades (Timchak and Gitis, 2012); particularly as an emerging high-efficiency technique for disinfecting enteric viruses leaving no unwanted disinfection by-products (Chu et al., 2012). Microbial inactivation is achieved through UV-induced photochemical reactions on genetic material (Rowan, 2019). Endogenous (direct) inactivation encompasses the absorbance of UVB light by the treated viral genome that causes its degradation. While the full antiviral mechanistic process has yet to be fully elucidated at molecular or structural levels, it is appreciated that UVC/UVB are strongly absorbed by enteric viral RNA with additional decontamination effects. "UVA cannot damage RNA and has no direct photochemical reactions; but it can produce reactive intermediates such as ROS (such as hydroxyl and superoxide radicals, hydrogen peroxide and so forth), which in turn can damage critical microbial targets (such as proteins, nucleic acids)" (Kokkinos et al., 2021). Thus, the role of using an optimised pulsed light technology that produces an intense broad spectrum (200 nm to 1100 nm) may potentially enhance AOP efficacy for enteric virus disinfection. Interestingly, compared to DNA, RNA is known to be more susceptible to the lethal action of UV irradiation (Galeano et al., 2019). The UVB/UVA and visible light wavelengths are absorbed by different water sensitizers through exogenous (or indirect) disinfection processes, such as organic matter, nitrate, and iron-containing complexes. Thus, the type of virus, the suspension menstruum (including suspended solids) and conditions (such as pH, temperature) are important governing factors affecting disinfection efficacy for MS-2 virus (Kosel et al., 2017).

Various AOP approaches have been studied including combined UV with  $H_2O_2$  for treating complex wastewater such as in the meat processing industry (Yapıcıoğlu, 2018). Another technique potentially applicable for the treatment of circulating water in the depuration tanks is pulsed-plasma gas-discharge. Such an AOP produces hydrodynamic cavitation that causes viral disinfection through photochemical (production of hydroxyl radicals), and physical mechanisms (pressure gradients, shock and acoustic waves, shear forces, and very high local temperatures) (Kosel et al., 2017). The mechanisms of hydrodynamic cavitation have yet to be elucidated, but it is theorized that it causes disruption to the viral capsid (icosahedral) and destabilizes recognition receptors.

Mycoystin\_LR (MC-LR) is produced by cyanobacteria that attract attention due to its high toxicity and high concentration in aquatic systems. Lu et al. (2018) showed that the combined use of UV/H<sub>2</sub>O<sub>2</sub> process and O<sub>3</sub>/ H<sub>2</sub>O<sub>2</sub> were effective methods to remove MC-LR from water and they performed better that UV-, O<sub>3</sub>-, H<sub>2</sub>O<sub>2</sub> alone processes under the same conditions. However, UV dosage of 1800 mJ/cm<sup>2</sup> was required to remove 90 % Of 100 mg/L MC-LR, where the amount significantly decreased to 500 mJ/cm<sup>2</sup> when 1.7 mg/L H<sub>2</sub>O<sub>2</sub> was added. Murray et al. (2017) reported that the use of pulsed light reduced the toxicity of the dinoflagellate algal toxin okadaic acid where ecotoxicological assessments were also performed using a miniaturised format of the conventional in vivo freshwater crustacean *Daphnia* sp. acute toxicity test. Findings revealed a 24-h EC50 of 25.87 µg/L for PL-treated okadaic acid at a UV dose of 12.98 µJ/cm<sup>2</sup> compared to a 24-h EC50 of 1.68 µg/L for the untreated okadaic acid control, suggesting a 15-fold reduction in toxicity to *Daphnia pulex*.

Despite positive observations as a clean decontamination technology, it remains uncertain as to the mechanism by which cavitation generated bubbles clean, disinfection and kill microbial organisms including viruses and enhance chemistry activity (Zupanc et al., 2019). Zupanc et al., 2019 also reported that "cavitation describes the formation of small vapour bubbles (cavities) inside an initially homogeneous liquid medium. It is a rapid physical phenomenon triggered by a sudden decrease in pressure. As the pressure recovers the bubble goes through a violent collapse and possible rebounds. By bubble growth, an energy from the surrounding liquid is collected and released by bubble collapsation, where extreme conditions can be formed locally. Bubble collapse can cause pressure shocks up to several 100 MPa and if the bubble collapses asymmetrically the so-called microjets with high velocities above 100 m/s can form". These observations are also aligned with the related studies of Chahine and Hsiao (2015). In addition, the so-called hot spots with extreme temperatures in order of several 1000 K can form at the centre of the bubble at its collapse, which can cause the formation of highly reactive radicals (Koda et al., 2003). Zupanc et al. (2019) also stated that the "exact manifestation of cavitation is influenced by liquid properties (temperature, density, viscosity and surface tension) and quality (number of solid particles and amount of dissolved gasses, which can both act as a nuclei). In general, two types of cavitation are recognized, hydrodynamic and acoustic cavitation. The difference is in the mechanism, which causes the local pressure to drop, while the principles which govern the hydrodynamic bubble and the acoustic bubble are basically the same".

The effects of cavitation including mechanical and thermal effects are: (a) microsteaming that can damage microorganism - together with shockwaves generated by bubble collapse; (b) chemical effects (implosion of bubbles and formation of hot spots for homolytic cleavage of H<sub>2</sub>O molecules and formation of highly oxidizing free radicals (\*OH and \*H) - \*OH readily oxidize particulate matter and also form H<sub>2</sub>O<sub>2</sub> - many other species can form (\*O<sub>2</sub>H, \*N, \*, 1O<sub>2</sub>) where different gases air/oxygen are dissolved in water; (c) oxidation of miroorganism's constituents (AC process produces a level of ROS beyond microbial antioxidant stress response capabilities) and kills a broad range of pathogens including viruses (McDonnell and Russell, 1999). Also different ROS affects polysaccharides, proteins, lipids and RNA/DNA); (c) oxidation of lipids, polysaccharides and nucleic acids by oxidative stress initiative by (ROS). High-frequency ultrasound inactives E. coli, Klebsiella pneumoniae, Pseudomonas aeruginosa, Legionella pneumophila, Enterobacter aerogenes, Vibrio cholera, Salmonella enterica, coliforms and Mycobacterium species albeit at different rates. However, different cellular and molecular mechanisms were postulated including cell wall breakage for Bacillus subtilis where this was the only study to report on simultaneous and sequential mechanisms using a pulsed plasma-gas-discharge approach (Hayes et al., 2013), where a similar approach was used to explain the lethal action of pulsed light (Farrell et al., 2011). Variations in effectiveness were also evident between using cavitation treating microalgae including Scendeesmus sp. The majority of researchers noted that the effect of cavitation on microalgae and microorganisms depends on the ultrasound frequency and that higher intensities and longer exposures correlate to more effective lethality (Wu et al., 2012). MS2 virus appears to be very susceptible to inactivation with all types of cavitation (Su et al., 2010) regardless of the initial concentration and medium tested only that in the case of higher concentrations more time for inactivation was needed. The exact mechanism of how cavitation causes virus inactivation is not yet elucidated. Su et al. (2010) suggested that the damages

inflicted during cavitation on the outer protein capsid itself or recognition sites on the capsid surface could be the reason for virus inactivation.

# 2.4. Photonics

There is also a potential role for developing photonics as a combined method for destroying enteric viruses, such as using visible blue light wavelengths (380–480 nm) that have been safely deployed for the inactivation of microbial pathogens in healthcare settings; thus, enabling the simultaneous activities of healthcare workers (Tomb et al., 2018). Maclean et al., 2009 have postulated that blue (405 nm) light inactivates bacteria via its interactions with porphyrins (five and six carbon multi-ring contain alternating single and double bonds) within the membrane of bacteria. The Irish SME, Atlantic Photonic Solutions Ltd., is developing innovative photonic solutions for the aquaculture Industry (APS, 2023). This company has identified specific wavelengths that can eliminate complex parasites without harming the hosts, such as high-throughput removal of sea lice from salmon in commercial aquaculture processes.

Kingsley et al. (2018) reported on oxygen-depended laser inactivation of murine norovirus using visible light lasers. Kingsley et al. (2018) used a tuneable mode-locked Sapphire laser where the frequency was doubled to generate femtosecond pulses at wavelengths of 400, 408, 450, 465 and 510 nm. >3 log murine NoV were achieved after 3 h exposure at 408, 425 or 450 femtosecond light pulses; thus, viral destruction was not wavelength specific. The use of photosensitizers, such as riboflavin, rose Bengal or methylene blue that generates singlet oxygen substantially improved the efficiency of inactivation. Findings indicate "a photochemical mechanism of the laser-induced inactivation where the action of relatively low-power blue laser light generates singlet oxygen" (Kingsley et al., 2018).

Kingsley et al. (2018) also noted that atmospheric sunlight at sea level contains visible light (400-700 nm), but also contains a UV component that is predominately 300-400 nm (Bird et al., 1983) which does not substantially penetrate water as UV is strongly absorbed. Recent research has indicated that sunlight can interact with dissolved organic molecules in aqueous settings to catalyse the formation of  $O_2$  ( $a^1\Delta_g$ ), commonly referred to as singlet oxygen, as well as other reactive oxygen species (Rosado-Lausell et al., 2013). These complex organic molecules act as sensitizers that become energetically excited by UV-Vis light and then transfer electronic energy to oxygen molecules producing singlet oxygen. Ultrashort pulse laser (USPL) light treatments were previously shown to be capable of inactivating murine norovirus (MNV) and other viruses (Tsen et al., 2007; Tsen et al., 2012); where impulsive stimulated Raman scattering (ISRS) was the postulated inactivation mechanism (Tsen et al., 2014; Tsen et al., 2014). Essentially the ISRS hypothesis was that high-frequency resonance vibrations are potentially induced by the 425 nm USPL, with a bandwidth of 420-430 nm, which may be capable of causing vibrations of sufficient strength such that the icosahedron capsid is destroyed after short nonthermal treatments (Kingsley et al., 2018). However, Kingsley et al. (2018) demonstrated that the inactivation of MNV by the USPL with variable wavelengths of blue light (400-450 nm) and by 408 nm CW indicated the inactivation mechanism was not specific to the 425 nm wavelength and, although more substantial inactivation was achieved with femtosecond pulse light, was not dependent on light pulses. This suggested that inactivation may have been via a non-ISRS mechanism. The addition of sodium bisulfate, an oxygen scavenger, substantially reduced CW laser inactivation, strongly implicating singlet oxygen as the cause of visible laser light-induced virus inactivation.

Kingsley et al. (2018) noted that as a priori, one would not expect vibrational resonance induced by ISRS to necessarily be wavelength-dependent. However, due to pulse-width dependence, it is well understood that ISRS cannot be induced using CW lasers since they do not generate light pulses. The authors demonstrated that CW lasers can inactivate MNV, which potentially conflicts with the hypothesis that ISRS is the mechanism of inactivation as proposed by Tsen et al. (2014) although it remains formally conceivable that both ISRS and singlet oxygen mechanisms could both contribute to laser inactivation observed by the USP laser. Also, it is difficult to envision a scenario in which the low concentrations of sodium bisulfate that reacts with, and sequesters, dissolved oxygen molecules within the MNV sample would inhibit a laser-induced vibrational mechanism. Indeed it was noted that purified MNV and MNV from a cell lysate had roughly equivalent inactivation rates (2014) suggesting that singlet oxygen enhancers are not substantially present in virus stocks, which are derived from virus-infected cell lysate. This suggests that either the virus capsid itself may function as an endogenous enhancer molecule, or that an enhancer may not be strictly required to produce singlet oxygen when interacting with intense blue light. This finding offers a potential explanation as to why norovirus illness was originally termed the 'winter vomiting disease.' The use of photonics has significant potential for nonthermal inactivation of microbial and parasitic pathogens, including, potentially, seafood.

# 2.5. Use of micro- and nano-bubble technology

Increased aeration and mixing can be potentially achieved in seawater depuration tanks using bulk micro-nanobubble (NB) water using hydrodynamic cavitation (Zhou et al., 2022). Nanobubbles (NBs) show technological potential in commercial applications including wastewater treatment, floatation, agriculture, nanoscope cleaning, and biological and medical applications (Ghadimkhani et al., 2016; Sun et al., 2022). The gas mix could be adjusted for delivery of hydrogen peroxide or ozone to accelerate disinfection of NoV deep in tissue. Such applications are attributed to the unique properties of NBs including low ascending velocity, long longevity retention time, massive interfacial surface area, high internal gas pressure, and negatively charged surface characteristics. Ultrafine NBs can form at the solid-liquid interface and in solutions and have a diameter of between 10 and 100 nm (Alheshibri et al., 2016). For commercial depuration of contaminated oysters, bubble concentrations and sizes in bulk NB water under different conditions should be evaluated along with the putative relationship between dissolved oxygen concentration and cavitation behaviours of bubbles used for this purpose. Thus, NBs have received attention for their unique physicochemical characteristics, including the previously mentioned large specific surface area, long residence time, high gas-liquid mass transfer efficiency, high zeta potential, and reactive oxygen species production (Zhang et al., 2022a, 2022b). Notably, NBs have exhibited outstanding performance, particularly in food safety and quality. Jaykus and Green (2015) describe a funded pilot-scale project addressing the efficacy of ozonated microbubbles for shellfish indicator organisms, Vibrio bacteria and NoV reduction in post-harvest depuration processing of Eastern oysters. The company ScanAqua now provides advanced nanobubble oxygenated technology with 'game-changing integrated engineering services and world-wide support for improved productivity, better water quality and fish welfare in related aquaculture.

The gas composition can be adjusted for NB generation to enhance antiviral activities, such as introducing reactive oxygen species (ROS) (Hayes et al., 2013). There is also potential to combine cold plasma with the nano-bubble biocide delivery concept. Despite positive observations as a clean decontamination technology, it remains uncertain as to the mechanism by which cavitation-generated bubbles clean, disinfect and kill microbial organisms including viruses and enhance chemistry activity (Zupanc et al., 2019) (Table 2). Cavitation describes the formation of small vapour bubbles (cavities) inside an initially homogeneous liquid medium. It is a rapid physical phenomenon triggered by a sudden decrease in pressure. As the pressure recovers the bubble goes through a violent collapse and possible rebounds. By bubble growth, energy from the surrounding liquid is collected and released by bubble collapses, where extreme conditions can be formed locally. Bubble collapse can cause pressure shocks up to several 100 MPa and if the bubble collapses asymmetrically the socalled microjets with high velocities above 100 m/s can form (Chahine and Hsiao, 2015). In addition, the so-called hot spots with extreme temperatures in order of several 1000 K can form at the centre of the bubble at its collapse, which can cause the formation of highly reactive radicals (Koda et al., 2003). Exact manifestation of cavitation is influenced by liquid properties (temperature, density, viscosity and surface tension) and quality

(number of solid particles and amount of dissolved gasses, which can both act as nuclei) (Zupanc et al., 2019). In general, two types of cavitation are recognized, hydrodynamic and acoustic cavitation. The difference is in the mechanism, which causes the local pressure to drop, while the principles which govern the hydrodynamic bubble and the acoustic bubble are the same.

The effects of cavitation including mechanical and thermal effects are: (a) microsteaming that can damage microorganism - together with shockwaves generated by bubble collapse; (b) chemical effects (implosion of bubbles and formation of hot spots for homolytic cleavage of H<sub>2</sub>O molecules and formation of highly oxidizing free radicals (\*OH and \*H) - \*OH readily oxidize particulate matter and also form H<sub>2</sub>O<sub>2</sub> - many other species can form (\*O<sub>2</sub>H, \*N, \*, 1O<sub>2</sub>) where different gases air/oxygen are dissolved in water; (c) oxidation of microorganism's constituents (AC process produces a level of ROS beyond microbial antioxidant stress response capabilities) and kills a broad range of pathogens including viruses (McDonnell and Russell, 1999). Also different ROS affects polysaccharides, proteins, lipids and RNA/DNA); (c) oxidation of lipids, polysaccharides and nucleic acids by oxidative stress initiative by ROS. High-frequency ultrasound inactives E. coli, Klebsiella pneumoniae, Pseudomonas aeruginosa, Legionella pneumophila, Enterobacter aerogenes, Vibrio cholera, Salmonella enterica, coliforms and Mycobacterium spp. albeit at different rates However, different cellular and molecular mechanisms were postulated including cell wall breakage for Bacillus subtilis where the study my laboratory was the only one to report on simultaneous and sequential mechanisms using a pulsed plasma-gas-discharge approach (Hayes et al., 2013), where similar approach was used to explain lethal action of pulsed light (Farrell et al., 2011).

Binding behaviour of noroviruses in oyster tissue may contribute to persistence and further challenge effectiveness of appropriate decontamination.

Careful considerations should be given to applying solutions that address the bioaccumulation of different NoV strains in oysters. This potentially contributes to the lengthy persistence of NoV in depurated oysters (McLeod et al., 2017). For example, immunochemistry studies carried on using Pacific oyster tissues, along with confirmatory tests using monoclonal antibodies and NoV particles with mutated capsids of similar tissues, established that GII binds specifically to the midgut and digestive tissue via an Alike carbohydrate, similar to the HBGAs used for NoV attachment to human epithelial cells (Le Guyader et al., 2006). In contrast, NoV GII binds to various tissue types in oysters, including the digestive tract, diverticula, midgut (intestine), gills, mantle, and labial palps (Wang et al., 2008; McLeod et al., 2009). Collectively, McLeod et al. (2017) noted that these findings intimate strain-specific variations in NoV binding patterns in contaminated oysters that can be seasonal due to higher bioaccumulation efficiency (GII strain), or no variations in seasonal patterns (GII strains). There is a need to understand typical viral load levels to deploy appropriate counter-measures; for example, there have been NoV outbreaks from depurated oysters

Examples of cavitation effects of treated microbial and algal species (adopted from Zupanc et al., 2019)

containing ca.  $10^3$  genome copies/g oyster tissue (far exceeding the median infectious dose [ID50]) (McLeod et al., 2017). Yet, the authors noted that "half of the published NoV reduction studies reported no decrease in NoV during depuration, and in the remaining studies it took between 9 and 45.5 d for a 1 – log reduction – significantly longer than commercial time frames".

# 3. Use of safe enteric surrogates to monitor and assess disinfection of shellfish

McLeod et al. (2017) noted that comparative elimination studies to date have shown that surrogate viruses (including phage, Feline calicivirus, Tulane virus, Mengovirus) are more rapidly depurated than NoV under a variety of depuration conditions, including temperatures of 8 to 25 °C, times varying between 23 h and 8 wk., and using both recirculating and flow-through systems that have UV and/or filtration disinfection. The rapid reductions noted for surrogate viruses may be partly attributable to the quantitation of infectious virions in some studies, whereas hNoV quantitation is based on genome detection; however, several studies (Ueki et al., 2007; Drouaz et al., 2015) have used genome detection for the analysis of both hNoV and surrogate viruses and large differences in reduction rates were still observed. McLeod et al. (2017) reported that "for a surrogate virus to provide useful information on hNoV infectivity, it is important that the characteristics of hNoV and the surrogate virus are similar within oyster tissues, including the way in which they interact with ligands, the stability of the virus capsid, and their persistence". Given that some researchers have reported more rapid depuration of certain surrogate viruses tested to date (such as FRNAPII), they may not be suitable for assessing the virological safety of depurated oysters when used alone without other confirmatory method. This is challenging given that there is currently no infectivity method for hNoV.

However, it is frequently challenging to use actual enteric viruses for developing and testing the efficacy of established or emerging disinfection technologies due to their infectious nature, the need to use Cat 2 facilities and the lack of a simple in vitro cell culture model. In addition, there is uncertainty surrounding the use of RT-qPCR for amplifying genomic material from treated enteric pathogens as this molecular approach does not distinguish between live or dead viruses. This is further complicated by the fact that using infectious hNoV at commercial plant level would not be appropriate for validating new technologies due to concerns over introducing pathogens into live shellfish. It is not uncommon to deploy different yet safe microorganism(s) as representative of the target pathogen of interest from a disinfection perspective; for example, *Geobacillus stearothermophilus* and *Bacillus atrophaeus* endospores are used as bioindicators for the terminal sterilization industry (McEvoy and Rowan, 2019), which exhibit a greater level of resistance to the applied lethal stress (such as UV) compared

Target organism	Characteristic size and shape	Treatment medium	Pre-treatment concentration	Ultrasound frequency (kHz)	Effectiveness (% reduction)	Reference
Enterobacter aerogenes	$0.7 \times 3.5 \mu m (rod)$	SS	$10^6$ CFU mL <sup>-1</sup>	20 [LFUS]	4.4 % after 20 min	Gao et al. (2014)
Haemophilus influenza	0.5 µm diameter (spherical)	BS	10 <sup>3</sup> CFU mL <sup>-1</sup>	20 [LFUS]	99 % after 10 min	Monsen et al. (2009)
Klebsiella pneumoniae	0.7 × 1.6 μm (rod)	BS	10 <sup>3</sup> CFU mL <sup>-1</sup>	20 [LFUS]	90 % after 15 min	Al Bsoul et al. (2010)
Legionella pneumophilia	0.5 × 2 μm (rod)	BS	$1.5~ imes~10^3~{ m CFU}~{ m mL}^{-1}$	33 [LFUS]	20 % after 60 min	Šarc et al., 2014
Vibrio cholerae	0.4 × 3.1 μm (rod)	ASW	$25~ imes~10^{6}~ ext{CFU}~ ext{mL}^{-1}$	19 [LFUS]	90 % after 0.9 min	Holm et al. (2008)
Coliforms		SUW	250 CFU mL <sup>-1</sup>	20 [LFUS]	70.8 % after 15 min	Al-Juboori et al. (2015)
Escherichia coli	0.5 × 1.5 μm (rod)	ASWS	$10^{6} \text{ CFU mL}^{-1}$	19 [LFUS]	90 % after 1.4 min	Holm et al. (2008)
Enterococcus faecalis	1 μm (oval)	DW	$10^{6} \text{ CFU mL}^{-1}$	-	100 % after 10 mins	Cerecedo et al. (2018)
Microcystis sp. (Algae)	4–5 μm (Oval)	GM	10 <sup>9</sup> cell mL <sup>-1</sup>	11 [LFUS]	5 % after 155 min	Zhang et al., 2006
Chlamydomonas sp. (Algae)	3–10 µm (spherical)	GM	$5.5 \times 10^7 \text{ cells ml}^1$	1100 [HFUS]	85 % after 7 min	Bigelow et al. (2014)
Scendesmus sp. (Algae)	6–8 µm (bean shaped)	GM	$5.2 \times 10^4$ cells ml <sup>-1</sup>		85 % after 60 min	Batista et al. (2017)
MS2 bacteriophage	24–27 nm	BS	$10^{6}  \text{PFU}  \text{mL}^{-1}$	20 [LFUS]	4.62 % after 30 min	Su et al. (2010)
Feline calicivirus	27–40 nm	BS	$10^{6}  \text{PFU}  \text{mL}^{-1}$	20 [LFUS]		Su et al. (2010)
Murine norovirus	28–35 nm	BS	$10^{6}  \text{PFU}  \text{mL}^{-1}$	20 [LFUS]	>3.8 % after 30 min	Su et al. (2010)

Artificial seawater (ASW); Surface Water (SUW); Salt Solution (SS); Buffer Solution (BS), Distilled Water (DS); Growth Medium (GM); Low Frequency ultrasound (LFUS); high frequency ultrasound (HFUS); Plague Forming Units (PFU); Colony Forming Units (CFU).

to similarly-treated microbial pathogens there is also a commensurate need to propagate high population levels of microorganisms (typically  $\geq 6 \log_{10}$  microbes/ml) in order to measure, standardize and to model the inactivation rate and kinetic shape to ensure sufficient disinfection performance. This is known as 'sterility assurance level' or SAL in the adjacent terminal sterilization industry that is used to inform appropriate modalities for the medical device industry. Several safe viral surrogates (or bioindicators [BIs]) have been used for assessment removal or disinfection performance of pathogenic hNoV in live shellfish, such as FRNAPII.

Norovirus is a fastidious virus as it does not grow on culture-based media where there is a reliance on molecular-based approaches to quantify the removal of norovirus and other enteric viruses post treatments; but, molecular approaches do not confirm if the material is alive or dead. Indeed, many existing molecular-based detection methods have great limitations and it is necessary to find alternative or complimentary methods for rapid multi-species and strain testing (Mi et al., 2021). Researchers have highlighted potential misalignment on the use of standard RT-qPCR that may be amplifying residual genetic determinants in the treatment process along with intact virion; which was informed by comparative studies that used FRNA surrogate virus (Rupnik et al., 2021). Plante et al. (2021) reported on improvements in efficiency to standard ISO 15216-1 genomic method using a Droplet-Digital PCR approach.

# 4. Advanced real-time diagnostics: future use of combined confocal microscopy and RAMAN spectroscopy (CRS)

Of the 53,612 publications that featured RAMAN in title, abstract or main body of the text in PubMed and Scopus during period 1995 to 2023, 12 focused on detecting norovirus and enteric viruses. The reason for including RAMAN spectroscopy reflected increasing evidence-based research that supported use of this technology for rapidly identifying and differentiating microorganisms including viruses. The development of rapid, cost-effective decontamination approaches for foodborne pathogens including viruses is of great importance for food safety, early detection of disease, and environmental monitoring (Yin et al., 2020).

There is a pressing opportunity to exploit the fact that hNoV strains and some surrogates (such as FRNAPII) have similar icosahedron-shaped capsids. Detection, and potentially differentiation between viable and inactivated (non-infectious) NoV and surrogates can be potentially achieved using Confocal Raman spectroscopy (CRS) combined with chemometrics as a novel assessment tool to inform disinfection efficacy. CRS is a fast, reliable and efficient method for detecting and identifying microorganisms without laborious pre-treatments. Raman spectroscopy is already accepted as a "powerful analytical technique for the rapid characterization of bacteria without external labels or tedious preparation. Raman spectrum, deriving from molecular vibrations, can be considered as a typical whole organism fingerprint of the biochemical composition of microorganisms where these vibrational spectra could show differences of molecular composition in various bacterial pathogens at the molecular level" (Lin et al., 2019). Thus, Raman spectra has been used to discern strain-specific physiological, metabolic and phenotypic states in microorganisms (Chisanga et al., 2020).

Huang et al. (2021) recently provided a comprehensive review addressing the use of Raman spectroscopy for fast and accurate detection of viruses; it also offers hand-held instruments for aiding practicality for point-of-use or care applicability. Mori et al. (2018) described the use of RAMAN spectroscopy for studying norovirus encapsulation. Other researchers have also exploiting use of this surface-enhanced scattering technology (SERS) for ultrasensitive and rapid identification of norovirus including; use of molybdenum trioxide nanocubes with graphene oxide (Achadu et al., 2020); molybdenum-trioxide quantum dots used with naonogels (Achadu et al., 2021), peroxidase-like graphene-gold particles (Ahmed et al., 2017), polyhedral Cu nanoshells (Kim et al., 2017) combined use of sulfur-doped carbon dots polydopamine-functionalized magnetic silver nanomaterials (Achadu et al., 2021), and plasmonic biosensing (Mauriz, 2020). Zhang et al. (2022a, 2022b) reported the use of CRS to identify and differentiate several types of bacterial pathogens (Vibrio cholera, Salmonella flexneri, L. monocytogenes, S. aureus, S. typhimurium, and Clostridium botulinum) based on characteristic peaks and peak intensity ratio. Principal component analysis (PCA), decision tree, artificial neural networks and Fisher's discrimination analysis were used to investigate differences in microbial detection. Combing Raman spectroscopy with confocal enables the identification of single bacterial cells in high spectral resolution, combining the power of 3D analysis with focused biological component in aqueous medium (Vlasov et al., 2020). The combined use of chemometrics to spectral data enables the analysis of nucleic acids, proteins, lipids, and carbohydrates for distinguishing different bacterial species (Zhang et al., 2022a, 2022b). The use of CRS may also be relevant for ensuring loss of all pathogens of concern including Vibrio species - it is particularly relevant to include a battery of test reference strains (E. coli, Vibrio sp., hNoV and surrogates) during the development of testing so as to ensure that bacterial and viral pathogens of concern are both addressed in live shellfish. For example, the use of pulsed light offers broader light spectrum targeting cellular constituents (cell membrane, lipid peroxidation and so forth) in Gram negatives bacteria; however, a focus on hNoV destruction from bestpublished data intimates delivering a short intensity of fixed wavelengths <300 nm that specifically targets viral RNA (and bacterial DNA) (Farrell et al., 2011; Hayes et al., 2012).

## 5. Advanced real-time diagnostics: future use of flow cytometry

Of the 253,560 publications cited in PubMed and Scopus on using "flow cytometry" of FCM, 26 papers described the combined use of FCM with noroviruses between period 1974 and 2023. Sixteen papers were excluded for the reason that they focused on non-enteric virus detection in shellfish or food systems, namely immune dysregulation associated with villous atrophy; NoV infection or Primary B Cell immune activation in vitro; interactions with Cryptosporidium parvum during infection of HCT-8 cells; identification of human NoV CD8 + T Cell restricted to HLA-A\*0201 allele; experimental inoculation of juvenile rhesus macaques; and bacterial-linked analysis of geothermal bathing pools. Detection or loss of important epitopic sites on NoV capsids by exposure to treatment technologies may affect the binding of these antibodies that can be rapidly determined in realtime using flow cytometry (FCM) - this is particularly relevant given that an intact capsid retaining surface receptors are critical for enabling viral attachment to specific human histo-blood group antigen (HGBA) sites to cause infection (Esseli et al., 2019). Razafimahefa et al. (2021a, 2021b) described the development of a specific anti-capsid antibody and magnet bead-based immunoassay to detect human NoV in stool samples and spiked mussels. Ponterio et al. (2013) reported on the phenotypic analysis use of FCM to detect pattern of human antigen presenting cells by genotype GII.4 norovirus. Hamza et al. (2011) noted that not all viruses are able to produce cytopathic effects where use of FCM may help with human NoV detection that have no available cell line for propagation. Annamalai et al. (2019) used FCM to evaluate infectivity of GII.4 human norovirus where the authors showed that this approach can inform differentiation between T-B-NK+ severe combined immunodeficiency (SCID) and non-SCID gnotobiotic pigs, implicating the role of NK cells in mediation of human NoV infection.

Thus, use of flow cytometry (FCM) is already an established immunological tool (Cossarizza et al., 2021) including for use in fisheries and shellfish research, where it has been applied to bivalve molluscan shellfish to analyse haemocytes (Van Nguyen and Alfaro, 2019). Cossarizza et al. (2021) noted that "FCM is a laser-based technique that is used to analyse physical characteristics of cells or particles in heterogenous fluid mixtures as they pass through the light source. When a sample of interest is injected into the FCM, targeted components are excited by the laser which emits light in a band of wavelengths. Here, the fluorescence intensity is measured for each particular target cell/particle (virus) at a rate of thousands per second. FCM provides a fast, accurate, convenient, simple to use and affordable tool that can achieve the desired simultaneous measurements of 'multiple' cellular or viral components in real time. This includes inter alia total cell or particle counts, cell viability including assessing sub-cell populations, quality of genomic content, phagocytosis, oxidative stress and apoptosis (Van Nguyen and Alfaro, 2019). Molecular targets of interest are typically labelled with fluorescent reagents that may include a broad range of dyes, stains, monoclonal antibodies or quantum dots (Van Nguyen and Alfaro, 2019). FCM also enables specific gating of plotted data for a more fine-grain focus on characteristic(s) of interest. For example, my research team (McEvoy et al., 2021) used flow cytometry as a real-time quantitative tool to enumerate biological indicators (*Bacillus atrophaeus* and *Geobacillus stearothermophilus*) in a commercial vaporized hydrogen peroxide (VHP) process used by the terminal sterilization industry.

FCM has been used for studying and quantifying viruses including clinical and aquatic applications for four decades (Hercher et al., 1979; McSharry, 1994; Marie et al., 1999; Kraus et al., 2007; Yang et al., 2008; Brussaard et al., 2010); for example, Marie et al. (1999) reported that FCM can successfully be used to enumerate viruses in seawater after staining and noted that the technique was first optimised by using the phaeocystis lytic virus PpV-01. Yang et al. (2008) advanced the use of microfluidic FCM for virus detection by introducing and integrating several functional micro-devices including antibody recognition and capturing of target viruses. Kraus et al. (2007) compared plaque and FCM-based methods for measuring dengue virus neutralization.

Given that this approach is based on the loss or reduction of a measured signal from fluorophore conjugated antibody that targets treated hNoV capsid - FCM can be deployed as a tool for determining the removal and inactivation modalities including assessing hNoV infectivity post treatments. FCM can be used to compare hNoV detection and quantitative efficiency of the established RT-qPCR genomic method at the laboratory bench (such as discovery phase, TRL 1-3) to compare and inform effectiveness of technologies at commercial plant level (TRL 6 to 9). Thus, and in theory, rapid detection (or loss of detection) of a cocktail of specific monoclonal antibodies matching epitopic sites on hNoV capsid may be used to directly inform quantification of inactivated hNoV by FCM (such as Van Nguyen and Alfaro, 2019). FCM can also be used to simultaneous assess the immunological status of treated live oysters in terms of vitality indicators (Van Nguyen and Alfaro, 2019). Commensurately, loss of specific spectral signals from combined use of CRS method will inform physical disruption of hNoV capsids post treatments, to be supplemented with use of scanning electron microscopy (SEM). For example, such approaches have also contributed to a generation of unique data in our laboratory that elucidates simultaneous and sequential changes at molecular and cellular level informing mechanistic of irreversible lethal action of pulsed light technology against clinical strains of Candida albicans (Hayes et al., 2013).

It is envisaged that this advanced suite of imaging techniques (FCM, CRS, SEM) will be centrally located in regional research and enterprise hubs linked to academic experts and industry (Rowan and Casey, 2021), where end-users can access and use these tools affordably that includes offering bespoke training to industry, as available in my laboratories. For example, this suite of advanced imaging tools will be initially used to compare effectiveness for rapid detection of hNoV methods including the provision of new real-time data to help understand the loss of hNoV infectivity during depuration treatments. This combined approach giving techniques that produce real-time spectral measurements would also be appropriate for future automation including the development of deep learning/machine learning algorithms. The Raman community needs to encourage the deposition of virus spectral findings to a central database so as to grow this technique for the food and healthcare sectors.

# 6. Modeling of enteric viral removal and inactivation and risk mitigation

Of the 1,973,267 papers published on mathematical "modeling" that appear in PubMed and Scopus databases [1933–2023], 26 addressed modeling for detecting and enumerating norovirus in shellfish [1996–2023]. Twelve papers were excluded for the reason that they focused on risk assessment and modeling of cooked or high-pressure-

processed shellfish, and development of a model virus or model system. The majority of modeling studies estimated distribution of norovirus or enteric viruses in shellfish, particularly oysters (Hunt et al., 2020) that included sewage impact (Winterbourn et al., 2016). For example, Campos et al. (2017) deployed modeling to determine the zone of impact of norovirus contamination in shellfish production areas through microbiological monitoring and hydrographic analysis. Whereas Razafimahefa et al. (2021b) used this approach to optimize a PMAxx<sup>TM</sup> -RT-qPCR assay and the preceding extraction method to selectively detect infectious murine norovirus particles in mussels.

Kinetic modeling of viral removal or inactivation performance is important in order to understand the behaviour of treated viruses to the applied treatment technology (Polo et al., 2014; Rowan et al., 2015; Polo et al., 2015; Oin et al., 2022). Oin et al. (2022) noted that a framework is needed to describe the complex nonlinear virus-oyster interactions that included a mathematical model addressing key processes for this viral dynamics, such as ovster filtration, viral replication, the antiviral immune response, apoptosis, autophagy, and selective accumulation. This has implications for informing the non-homogenous distribution of norovirus in ovsters and effectiveness of appropriate depuration or relaying. Ideally, a first-order log-linear inactivation plot is achieved over the treatment regime that eliminates or destroys the target pathogens of concern exponentially. However, depending on the efficacy of the treatment modalities, the resistance or tolerance potential of the microorganisms (Fig. 2), and other governing or interfering parameters (such as the presence of organic matter and particulates or biofilms, pH, salinity, dissolved oxygen and so forth), prior microbial stress adaption to applied lethal technologies, the shape of the inactivation plot may differ (Fig. 2). Microbial kinetic inactivation plots may exhibit an initial protective 'shoulder' effect followed by a linear kinetic plot, and/or a pronounced adaptive or resistant 'tailing' effect that may be evident after longer exposure times. These are referred to as bi-phasic if the shoulder or tail effect accompany a log linear kinetic plot, or triphasic if both the shoulder and tailing effects are present. For example, the absence of a shoulder and tailing effect in microbial inactivation plots for the adjacent sterilization industry is essential as medical devices must achieve a 12 log reduction in bacterial endospore numbers where the first 6 logs are enumerated, but the next 6 logs are predicted based on the probability of a linear plot from the first 6 log (half-cycle). If the tailing effect is evident in first 6 log half-cycle plot, then it cannot be assumed that linearity occurred; thus, introducing the possibility of microbial survivors in the latter treatment stages (McEvoy et al., 2023). For food treatments, it is desirable to have an initial starting microbial population of 5 to 6 log orders in order to standardize treatments between laboratories (Rowan, 2019; Gómez-López et al., 2022) and to achieved a discernible number of Dvalue reductions (a D value relates to the time taken for a 1 log reduction in microbial numbers achieved under a fixed treatment regime) (Rowan et al., 2000). US FDA also recommend using a 6 log starting population for PUV irradiation studies for food surface decontamination to include natural microbial contaminants during testing and validation (Rowan, 2019).

However, determining accurate viral kinetic modeling for a complete norovirus reduction in shellfish is challenging due to: (a) there is no infectivity assay that would allow the propagation of human NoV to artificially high levels (viral load) such as log 6 gcg<sup>-1</sup> for kinetic disinfection studies; (b) contamination of shellfish such as oysters appear to be typically contaminated at significantly less viral loads (such as  $\leq 5 \log$ ); (c) is its uncertain if the RT-qPCR alone will provide data on individual virus post treatments including potential for amplifying genomic artefacts in whole or disintegrated capsid; (d) the LOQ for RT-qPCR (100 gcg<sup>-1</sup>; Rupnik et al., 2021); (e) use of existing surrogates such as FRNAP II may be less resistant to actual hNoV strains in contaminated live virus due in part to differences in behaviour or non-ligand binding to oysters; and (f) as yet, there is no universally accepted model for determining kinetic destruction of hNoV in shellfish with particular relevance to commercial depuration deployment.

There is a marked gap in data on hNoV inactivation modeling for shellfish. However, Rupnik et al. (2021) reported that norovirus GII exhibited bi-phasic viral reduction kinetic performance using LP-UV irradiation in contaminated Pacific oysters. After the initial rapid linear reduction of GII, the rate of depuration decreased with a pronounced tailing effect; no further reduction was observed between days 3 and 7, or days 4 and 7 for



**Fig. 2.** Commonly observed types of inactivation curves during non-thermal UV processing expressed as  $\log_{10} N$  versus F (fluence, J/cm<sup>2</sup>). Plot A: sigmoidal-like, linear with a preceding shoulder, log-linear with a tailing. Plot B: biphasic, concave upwards or downwards. Plot C: Linear, Weibull incorporating a tailing effect, two mixed Weibullian distributions (adopted from Rowan et al., 2015).

high and medium depuration temperatures. Interestingly, the two phase viral reduction kinetics was not evident at low temperature trials, with no oyster mortalities observed during the first two weeks of depuration. This pronounced hNoV 'tailing' effect is a potential concern where there is a pressing need to test and develop complementary treatment technologies (such as possibly pulsed light, AOPs, photonics, micro- nano-bubbles) and improved hNoV detection techniques. Potential use of complementary approaches to assess effectiveness of hNoV removal or destruction below the existing LoQ or LoD for RT-qPCR method would be beneficial (as proposed and discussed in the next section). Also, for future modeling studies, it will be relevant to include a test reference hNoV strain, along other microbial reference strains (*E. coli, Vibrio* sp), in order to compare the broad-spectrum antimicrobial action of treatment technologies through collaborative inter-laboratory validation trials with commercial producers of oysters.

Thus, lessons can be learnt from the adjacent Medtech domain where complex disinfection and sterilization challenges to address disinfection and sterilization are met in tandem consultation with regulators (such as the USF FDA) for sterility assurance levels. Starting viral populations of 6 log<sub>10</sub> for NoV GI and GII (and surrogates such as FRNAPII) are required in order to model inactivation or removal profile to confirm efficiency of modalities where it is assumed that inactivation is exponential (or log-linear). McMenemy et al. (2018) proposed a mathematical model for estimating pathogen variability (including E. coli, hNoV) in shellfish and predicting minimum depuration times (including FRNAPII). This interesting model assumes a 'worst case scenario' for viability of pathogens and is then used to predict minimum depuration times to achieve hNoV levels which fall within possible risk management levels, which is a logical approach to mitigating this challenge as a tool to assist in future control strategies given its complexity. The authors state that this model is based on documented assumptions that hNoV is log-normally distributed throughout a population of oysters, and that the pathogen load decay during depuration is exponential. This model requires the "input of four parameters: i) the initial average hNoV load, ii) the depuration efficacy, iii) the desired assurance level (i.e., proportion of shellfish pathogen in depuration which must have pathogen loads less than pathogen threshold limit at the end of the depuration), and iv) the required hNoV threshold" (McMenemy et al., 2018).

However, observations from best-fit linear plots of hNoV and FRNAPII inactivation data in treated oysters appear to frequently show distinct initial shoulder and latter tailing effects that are associated with bi and triphasic viral survivor plot responses (Leduc et al., 2020; Rupnik et al., 2021); therefore, not all documented decontamination studies appear to exhibit viral data that are log-linear in performance. This is important as the pronounced tailing of the hNoV survivor plot would necessitate longer or alternative treatment approaches to ensure complete decontamination of enteric viruses in live shellfish (Rowan et al., 2015). The initial shoulder or increase in hNoV numbers after the onset of depuration (Rupnik et al., 2021) may be attributed to the release of viruses from more concentrated locations in oysters or possibly due to variance in treatment methods and conditions.

In the context of related modeling in adjacent domains, McEvoy et al. (2021) recently reported that vaporized hydrogen peroxide effectively kills recalcitrant bioindicators (Geobacillus stearothermophilus and Bacillus atrophaeus) in a commercial process where the inactivation produces linear death rate kinetics. This provides confidence for the industry where there were prior assumptions made through the half-cycle sterilization process (6-D log10 reductions) that linear inactivation occurs in a closed box endto-end monitoring treatment system. This is particularly important as the terminal sterilization makes inferences about the probability of linear inactivation in order to deliver treatments for 12 log10 reductions in Bacillus endospores. Modeling of kinetic data for norovirus destruction (such as longitudinal models for Covid-19, Rowan and Moral, 2021), along with gaining an understanding of the mechanistic destruction of this virus at a molecular level will also help inform reliable and repeatable destruction of this food-borne pathogen for the seafood industry. Interestingly, several publications refer to sterilization (Cong et al., 2021), when it is a 'high-level disinfection' process for human norovirus decontamination as one cannot

assume the destruction of all microbial life that is defined by 'sterilization'. There is a pressing need to develop an appropriate risk mitigation approach using a source-pathway-receptor model in order to determine the efficacy of hNoV removal or destruction from live shellfish. This is likely to be a quasi-quantitative approach given the complexity of the challenge as there is likely to be a residual risk to consumers and commercial seafood producers given their current understanding of existing treatment combinations. The question remains as to what is the acceptable risk given the uncertainly in diagnostic (RT-qPCR) for determining viable hNoV and the limited data available on the efficacy of established and emerging disinfection technologies for full removal or inactivation of enteric viruses deployed at commercial depuration for the shellfish industry. The quasi-quantitative, risk assessment-based approach of Tahar et al. (2017) may inform the aforementioned given that this study addressed the efficacy of WWTPs to remove complex contaminants of emerging concern including risks associated with discharge to receiving waters.

Modeling may also inform previous researcher observations that noted hNoV depuration to adopt a 'two-phase' response with elimination in the first few days being more rapid than subsequent days (Love et al., 2010; Polo et al., 2014, 2015). The first rapid phase of viral depuration is likely related to extracellular digestion and purging of the digestive tract; the speed of purging is governed by physiological traits of the shellfish species concerned, including filtration and clearance rates, digestion rate, and enzymatic activity. Thus, optimizing the physiological state of shellfish through adjusting different parameters such as temperature and salinity contributes to maximizing hNoV reductions in the first phase of depuration (that is, gut purging). However, the persistence of hNoV in shellfish during the second slower phase of elimination indicates that other properties are at play. Indeed, the binding of hNoV to HBGA-like ligands present on oyster gastrointestinal cells, gills, and mantle represents a major barrier to enhancing depuration and will be a key point to address in future studies (Maalouf et al., 2011; Polo et al., 2014). McLeod et al. (2017) also introduced the novel concept of enzymatic pre-treatment to help remove hNoV that could also be modelled for effectiveness.

Hunt et al. (2023) reported for the first time on a quantitative exposure assessment model to predict variations in per-serving norovirus consumption using results from the standard SIO 15216-1:2017 detection method. This model recognises that norovirus is infectious at low doses intimating that small variations in human exposure can potentially lead to significant differences in consumer health. Hunt et al. (2023) findings show "the boundaries for potential exposure following a given ISO detection result, and the relative importance of mean concentration, serving size, and oyster grade. This is directly relevant to potential regulatory thresholds being considered in the EU". However, improvements in the application of appropriate decontamination technologies would potentially impact positively on reduced risk in this model for consumers as these may remove or deactive noroviruses. Addressing norovirus reduction is a complex challenge for risk mitigation with reliance on the LoQ for the ISO method (i.e., 100 gc/ g), with a LoD of 20 gc/g. Hunt et al. (2023) noted that the EFSA baseline survey (EFSA, 2019) defined LoQ of 300 gc/g, as achievable for routine monitoring EU. Findings published by Hunt et al. (2023) show that a mean of 200 g/g predicts 95 % consumption range of 57 to 338 gc in a serving of a single oyster. However, the median infectious dose attributed to oysters must be below 2020 genomic copies of hNoV; thus, implying a high probability of infection even at levels lower than the LoQ. Hunt et al. (2023) noted that the median illness-causing does is higher, with a median predicted value of 260 copies needed, as calculated from outbreak events. Additional factors need to be considered to fully progress this QMRA model given the large difference between proportion of oyster batches testing positive for hNoV (such as 76.2 %, [Lowther et al., 2012]), and the proportion of positive samples associated with illness, which is estimated at <0.28 % (Lowther et al., 2010) in the UK. Future use of this QMRA model that incorporates additional factors such as disinfection potential of technologies can also potentially address challenges for hNoV risk assessment.

#### 7. Climate change influences

Climate change is increasing seawater temperatures that can cause disruption in aquatic microbiome enabling the emergence of toxigenic phytoplankton and invasive species, toxic algal blooms, and survival of Vibrio parahaemolyticus (Winder and Sommer, 2012; Duchenee-Moutiner and Netto, 2021) in marine environments (Noorian et al., 2023); thus, presenting additional shellfish depuration or relaying challenges. Moreover, 2022 was a year of climate extremes with record high temperatures and rising concentrations of greenhouse gas (Copernicus, 2023). This report highlighted that the "last eight years have been the eight warmest on record globally. La Niña conditions persisted during much of the year. The annual average temperature was 0.3°C above the reference period of 1991-2020, which equates to approximately 1.2°C higher than the period 1850-1900. Europe saw its second warmest year on record, exceeded by 2020 and only slightly warmer than 2019, 2015 and 2014. Europe experienced its hottest summer ever recorded in 2022, where autumn was the third warmest on record. The world's oceans reached temperatures in 2021 despite a La Niña event that typically has a cooling influence". In 2021, the upper 2000 m of the ocean, where most of the warming occurs, absorbed 14 more zettajoules (a unit of electrical energy equal to one sextillion joules) than it did in 2020. This amount of extra energy is 145 times greater than the world's entire electricity generation which, by comparison, is about half of a zettajoule (Milman, 2022). U.S. EPA (2021) reported that "sea surface temperature has been consistently higher during the past three decades than at any other time since reliable observations began in 1880". Seawater temperature rises have been attributed to changes in marine ecosystems such as the recent toxic algae bloom that caused significant losses at commercial salmon sites (Mowi) in Ireland (Intrafish, 2021).

Numerous reports have connected climate change impacts to influencing infectious diseases due to increased hazard, exposure and vulnerability, such as through modeling (Semenza and Paz, 2021; O'Neill et al., 2022a, 2022b). Extreme precipitation events have caused waterborne outbreaks and longer summer seasons (Semenza and Paz, 2021). Researchers have also noted that project risks from infectious disease can be reduced in the future by acting on hazard, exposure and vulnerability through appropriate mitigation and adaption, such as a described in this review (Semenza and Paz, 2021). Carlson et al. (2022) reported that climate change increases cross-species viral transmission risk where there is a mechanistic link between global environmental change and disease emergence. Notably, these authors stated that "this ecological transition may already be underway and holding warming under 2°C within the twenty-first century will not reduce future viral sharing. Our findings highlight an urgent need to pair viral surveillance and discovery efforts with biodiversity surveys tracking the range shifts of species". O'Neill et al. (2022b) reported on the role of using next-generation sequencing and bioinformatics to profile freshwater aquaculture-biome to understand the impact of climate variance of key algae and bacteria including the emergence of waterborne parasites (O'Neill and Rowan, 2023). Duchenne-Moutien and Neetoo (2021) also intimated a link between climate change and food safety issues.

Rindi (2014) reported that terrestrial algae (green algae and diatoms) are more directly affected by climate change and can therefore respond in a more immediate way. This is attributed in part to the fact that algae have short generations, fast turnovers and respond quickly to changes in environmental conditions. Sarmaja-Korjonen et al. (2006) demonstrated that algae appeared to be comparatively good indicators of environmental conditions by representing productivity disparities during changing climatic conditions. Hallegraeff (2010) has also indicated that changes in algal communities can putatively provide a sensitive early warning for climatedriven uncertainties in aquatic ecosystems. There has been increased interest in alternative uses of microalgae within aquaculture to assist with sustainability, in addition to enabling an ecotoxicological assessment and water quality control (O'Neill et al., 2019; O'Neill and Rowan, 2021). Eutrophication occurs when a water body is put under pressure with large levels of organic matter and nutrient waste that is taken in and biologically processed which in turn leads to algal blooms (Jegatheesan et al., 2011;

Martinez-Porchas et al., 2014; Sikder et al., 2016). Algal blooms in turn can lead to decreases in light and oxygen production, which can suffocate aquatic life (Jegatheesan et al., 2011; Chislock et al., 2013; O'Neill et al., 2019) and can be exacerbated by increased marine and freshwater temperatures.

Barry and Hoyne (2021) reported that changes in weather systems, such as increased precipitation, snow and ice events, heatwaves and storms, have led the European Commission to develop new policies and strategies to deal with extreme events. Moreover, Barry and Hoyne (2021) suggested seeking international agreement on indicators to inform ecological resilience, along with economic (such as a number of new SME creations, innovation, investment in training, and specialist upskilling), social enterprise and culture (such as diversity of youth initiatives to increase civil action, solidarity and engagement). Several researchers have also reported on the impact of climate change shifts on receiving waters including flooding (Blöschl et al., 2019).

# 8. Quadruple Helix Hub Approach

The role of Quadruple Helix HUB (academia-industry-regulators-society) to unlock challenges and to provide subject-matter inputs and shared use of complex or sophisticated equipment (such as novel treatment and diagnostic technologies) from an educational, training and innovation perspective (Rowan and Casey, 2021; Rowan, 2022) is vital. This approach can concentrate single-access supports for industry, entrepreneurs and disruptors; such as step-change in physical infrastructure and systems supports, pre-start-ups; ideation and design thinking to inform technology readiness level (TRLs); market research and enterprise support including early needs analysis and product market fit analysis; early technical validation (such as test the tech, experimentation and validation of pre-pilot and scale to commercial setting); digital transformation (including end-to-end monitoring, AI, remoting immersive training); and a conduit for grant or funding support (Rowan, 2022; Rowan et al., 2022). There is a pressing need for real-time monitoring and confirmation of disease mitigation efficacy - possibly through loss of essential housekeeping or virulence molecular determinant(s) in treated hNoV. This could be elucidated through combinational bioinformatics and next-generation sequencing (NGS). There is an enormity of potential parameters that require simplification that will inform decision making on technology adoption and deployment (Rowan and Galanakis, 2020). There is pressing need for concerted multiactor collaborations that address the open sharing of data and knowledge that involves hurdling IP discovery phase in TRLs from the screening pilot phase studies for realistic translation to commercial depuration (O'Neill and Rowan, 2022c; O'Neill et al., 2022c). This could all be accelerated through digital transformation that can also address the sustainability of the applied monitoring and disease mitigation technology from a commensurate energy, carbon footprint and risk evaluation perspective that will also help with investment in solutions. Studies conducted in the Hub linked to industry can inform toxicology (Usuldin et al., 2021; Wan-Mohtar et al., 2012

There is a need for a new generation of workers cross-trained in different converging topics to enable sustaining, and potentially, disruptive approaches to tackling industry and societal challenges. This can be met through the Quadruple Helix Hub concept that can also help streamline top-down policy with a bottom-up user understanding at a critical interface to inform decision-making. Testing across independent laboratories and pilot/commercial plants internationally for validation and to support investment in solutions. The greatest challenge is time, as effective solutions that includes confirmation of disease mitigation efficacy are required now.

# 9. Conclusions and implications for future research

Shellfish are an important source of nutritious food that will contribute to feeding our increasing global populations. However, bivalve molluscan shellfish (particularly oysters) are filter feeding animals and can bioaccumulate pathogenic microorganisms in their digestive tissue that lead to outbreaks of foodborne illness and damage the reputation of the seafood industry. Existing disease mitigation measures rely upon either relaying contaminated shellfish to intertidal locations of clean water for four weeks; or more commonly, depuration using clean seawater combined with a treatment technology (such as UV-irradiation) at the commercial shellfish production site. Growing intertidal environmental locations for shellfish production are graded A to C to dictate requirements for treatment based on microbiology quality; category B growing locations require mandatory depuration. However, many European shellfish producers operate depuration from category A sites that have resulted in contaminated oysters. While bacterial pathogens can be effectively removed at the depuration phase, there is a pressing need to improve existing technologies to address complex human NoV that can persist for lengthy periods due to specific ligand-binding of NoV strains to bivalve molluscan tissues (where the type of tissue for viral attached varies depending on the strain). Assessing the effectiveness of the removal or an inactivation approach for hNoV in live oysters is further complicated by the fact that there is no infectivity assay for this enteric pathogen with reliance upon the use of genomic RT-qPCR technique that does not distinguish between live or dead hNoV. Consequently, the use of surrogate viruses (seen as representative of hNoV to the applied treatment technologies) is adopted such as using the FRNA bacteriophage. However, best-published evidence suggests that existing depuration practices remove viral surrogates faster than targeted hNoV intimating differences in behaviour in viral binding to oyster tissue, and possibly the variance in enumeration methods. Therefore, there is a pressing need to improve both hNoV removal and inactivation methods at depuration along with implementing more appropriate detection and enumeration approaches for hNoV at the commercial depuration phase. There is also uncertainty regarding the appropriateness of using standard molecular-based method for assessing hNoV post treatments compared with surrogate infectious virus values, thus, it is potentially plausible that the level of viral lethality achieved is underestimated given current diagnostic challenges.

The following recommendations are proposed to help with this challenge:

- Further investigate and potentially apply combinational decontamination and treatment technologies for hNoV elimination in live shellfish such as PUV, AOPs, photonics, nano-bubbles and potentially enzymology.
- Develop appropriate laboratory-based diagnostic methods to focus on the destruction of icosahedron-shaped viral capsid, specifically addressing enteric viruses (noroviruses) and appropriate viral surrogates (such as FRNAPII).
- Compare the effectiveness of genomic RT-qPCR (ISO standard), and infectivity assays (such as FRNPH plaque method) against the aforementioned capsid-stability diagnostic method(s) (such as immuno-phenotypic profiling using FCM, confocal-RAMAN spectroscopy) pre and post treatment.
- Enhance support and engagement with the shellfish industry such as through the Quintuple Helix HUB framework (academia-industrygovernment-environment-society) that addresses access to specialist equipment, subject-matter experts and contract-service provision. This can inform testing and translation of new decontamination technologies to support commercial oyster producers in addition to implementing a practical strategy for unlocking affordable effective solutions.
- Determine actual representative hNoV viral load for testing and validating decontamination technologies.
- Determine the link between detection (ISO, 2017), disinfection efficacy and consumer exposure for norovirus contamination in oysters that is particularly relevant to commercial producers and for risk managers in EU.
- Develop appropriate inactivation kinetic models that embrace circumstances where there are deviations from exponential log-linear plots (such as bi-and tri-phasic performances associated with protective shouldering and tailing) that will inform decision-making for shellfish managers.
- Develop an appropriate holistic 'multi-actor' risk mitigation strategy for preventing hNoV contamination of shellfish including local government

#### N.J. Rowan

Science of the Total Environment 874 (2023) 162380

and wastewater industry. This can be combined with new ecological profiling of the same intertidal growing environments as commercial production shellfish production that will address early detection of established recalcitrant hNoV strains and emerging microbial or toxicological contaminations (such as algal biotoxins and invasive species that may be influenced by climate change/variance).

- Translate findings into supporting and facilitating new policies on hNoV in shellfish.
- Develop appropriate training that embraces new techniques including digital technologies.
- Given the complexity of multiple factors potentially governing reliable and repeatable hNoV decontamination of shellfish, introduce edgecomputing for end-to-end monitoring and AI/machine learning and blockchain to evaluate and analyse datasets for real-time decisionmaking.
- Routinely include standard microbial reference strains and for stakeholders to reach consensus on an agreed harmonized method for hNoV decontamination (for example, UV/PUV dose, viral load, depuration temperature, flow rate, duration and so forth).
- Enhance social marketing to help inform stakeholders in real-time and for behavioural change.

#### CRediT authorship contribution statement

Neil Rowan designed and wrote this paper solely.

#### Funding acknowledgement

The author acknowledges funded support from the MSCA RISE ICHTHYS Project [No. 872217] and Interreg Atlantic Area NEPTUNUS Project [No. EAPA\_576/2018].

## Data availability

Data will be made available on request.

#### Declaration of competing interest

The author declares no conflict of interest.

# References

- Achadu, O.J., Abe, F., Suzuki, T., Park, E.Y., 2020. Molybdenum trioxide nanotubes aligned on a graphene oxide substrate for the detection of norovirus by surface-enhanced Raman scattering. ACS Appl. Mater. Interfaces 12 (2020), 43522–43534.
- Achadu, O.J., et al., 2021b. Sulfur-doped carbon dots@polydopamine-functionalized magnetic silver nanocubes for dual modality detection of norovirus. Biosens. Bioelectron. 193. https://doi.org/10.1016/j.bios.2021.113540.
- Advisory Committee on the Microbiological Safety of Food, 2015. An Update on Viruses in the Food Chain. Zenodo https://doi.org/10.5281/zenodo.571415.
- Ahmed, S.R., et al., 2017. Size-controlled preparation of peroxidase-like graphene-gold nanoparticle hybrids for the visible detection of norovirus like particles. Biosens. Bioelectron. 87, 558–565.
- Al Bsoul, A., et al., 2010. Effectiveness of ultrasound for the destruction of Mycobacterium sp. strain (6PY1). Ultrason. Sonochem. 17, 106–110. https://doi.org/10.1016/j.ultsonch. 2009.04.005.
- Alheshibri, M., Qian, J., Jehannin, M., Craig, V.S.J., 2016. A history of nanobubbles. Langmuir 32 (43), 11086–11100.
- Al-Juboori, R.A., et al., 2015. Impact of pulsed ultrasound on bacteria reduction of natural waters. Ultrason. Sonochem. 27 (2015), 137–147. https://doi.org/10.1016/j.ultsonch.2015. 05.007.
- Annamalai, T., et al., 2019. Infectivity of GII.4 human norovirus does not differ between T-B-NK+ combined immunodeficiency (SCID) and non-SCID gnotobiotic pigs, implicating the role of NK cells in mediation of human norovirus infection. Virus Res. 267, 21–25.
- APS, 2023. Atlantic Photonic Solutions Ltd innovation solutions for the aquaculture industry. https://zaps.ie/research/ (accessed 10th February, 2023).
- Baert, L., et al., 2009. The efficacy of preservation methods to inactivate foodborne viruses. Inter. J. Food Microbiol. 131 (1–2), 83–94.
- Barrett, M., Fitzhenry, K., O'Flaherty, V.O., Dore, W., Keaveney, S., Cormican, M., Rowan, N., Clifford, E., 2016. Detection, fate and inactivation of pathogenic norovirus employing settlement and UV treatment in wastewater treatment facilities. Sci. Total Environ. 568, 1026–1036.

- Barry, D., Hoyne, S., 2021. Sustainable measurement indicators to assess impacts of climate change: Implications for the New Green Deal Era. Current Opinion in Environmental Science and Health 22.. https://doi.org/10.1016/j.coesh.2021.100259.
- Batista, M.D., et al., 2017. Use of hydrodynamic cavitation for algae removal: Effect on the inactivation of microalgae belonging to genus Scenedesmus. Water Air Soil Pollut 228. https://doi.org/10.1007/s11270-017-3624-x.
- Bigelow, T.A., et al., 2014. Lysis of Chlamydomonas reinhardtii by high-intensity focused ultrasound as a function of exposure time. Ultrason. Sonochem. 21 (2014), 1258–1264. https://doi.org/10.1016/j.ultsonch.2013.11.014.
- Bird, R.E., et al., 1983. Terrestrial solar spectral data sets. Sol. Energy 30 (6), 563.
- Blöschl, G., Hall, J., Viglione, A., et al., 2019. Changing climate both increases and decreases
- European river floods. Nature 573, 108–111. https://doi.org/10.1038/s41586-019-1495-6.
  Bosch, A., Pinto, R.M., Abad, F.X., 1995. Differential accumulation and depuration of human enteric viruses by mussels. Water Sci. Technol. 31, 447–451. https://doi.org/10.1016/ 0273-1223(95)00310-J.
- Brussaard, C.P.D., Payet, J.P., Winter, C., Weinbauer, M.G., 2010. Chapter 11: Quantification of aquatic viruses by flow cytometry.
- Campos, C., et al., 2017. Determining the zone of impact of norovirus contamination in shellfish production areas through microbiological monitoring and hydrographic analysis. Water Res. 124, 556–565.
- Cantalupo, P.G., et al., . Raw sewage harbors diverse viral populations. mBio https://doi.org/ 10.1128/mBio.00180-11.
- Carlson, C.J., Albery, G.F., Merow, C., et al., 2022. Climate change increases cross-species viral transmission risk. Nature 607, 555–562. https://doi.org/10.1038/s41586-022-04788-w.
- Centers for Disease Control and Prevention (CDC), 2020. Virus classification. https://www. cdc.gov/norovirus/lab/virus-classification.html Accessed 13 Dec, 2022.
- Cerecedo, L.M., Dopozo, C., Gomez-Lus, R., 2018. Water disinfection by hydrodynamic cavitation in a rotor-stator device. Ultrason. Sonochem. 48, 71–78.
- Chahine, G.L., Hsiao, C.T., 2015. Modelling cavitation erosion using fluid–material interaction simulations. Interface Focus 5 (5), 20150016. https://doi.org/10.1098/rsfs.2015.0016.
- Chhabra, P., de Graaf, M., Parra, G.I., Chan, M.C., Green, K., Martella, V., Wang, Q., White, P.A., Katayama, K., Vennema, H., Koopmans, M.P.G., Vinjé, J., 2019. Updated classification of norovirus genogroups and genotypes. J. Gen. Virol. 100 (10), 1393–1406. https:// doi.org/10.1099/jgv.0.001318.
- Chislock, M.F., Doster, E., Zitomer, R.A., Wilson, A.E., 2013. Eutrophication: causes, conse[1] quences, and controls in aquatic ecosystems. Nature Educ. Knowledge 4 (4), 10.. https:// doi.org/10.1007/978-90-481-9625-8.
- Chu, X., Hu, J., Xu, Y., 2012. Investigating the performance of a UV/H<sub>2</sub>O<sub>2</sub> integrated flowthrough system followed by free chlorine. Water Sci. Tech. Water Supply 12 (6), 715–719.. https://doi.org/10.2166/ws.2012.046.
- Compos, C.J.A., Lees, D.N., 2014. Environmental transmission of human noroviruses in shellfish waters. Appl. Environ. Microbiol. 80 3552-2561.
- Cong, X., et al., 2021. Temperature effects on the nutritional quality in Pacific oysters (Crassostrea gigas) during ultraviolet depuraton. J. Sci. Food Agric. 102 (4), 1651–1659.
- Cooney, R., Baptista de Sousa, D., Fernadez-Rios, A., Mellett, S., Rowan, N.J., Morse, A.P., Hayes, M., Laso, J., Regueiro, L., Wan, A.H.L., Clifford, E., 2023. A circular economy framework for seafood waste valorisation to meet challenges and opportunities for intensive production and sustainability. J. Clean. Prod. 392. https://doi.org/10.1016/j.jclepro. 2023.136283.
- Copernicus, 2023. Copernicus: 2022 was a year of climate extremes, with record high temperatures and rising concentrations of greenhouse gas. https://climate.copernicus.eu/ copernicus-2022-was-year-climate-extremes-record-high-temperatures-and-risingconcentrations (accessed 11 Jan, 2023).
- Cossarizza, A., et al., 2021. Guidelines for the use of flow cytometry and cell sorting in immunological studies. E. J. Immunol. 49 (10), 1457–1973.
- Davidson, V.J., Ravel, A., et al., 2011. Food-specific attribution of selected gastrointestinal illnesses: estimates from a Canadian expert elicitation survey. Foodboren Pathog. Dis. 8, 983–995.
- Dewey-Mattia, D., 2018. Surveillance for foodborne disease outbreaks United States, 2009–2015. MMWR Surveill. Summ. 67. https://doi.org/10.15585/mmwr.ss6710a1.
- Doré, W.J., Keaveney, S., Flannery, J., Rajko-Nenow, P., 2010. Management of health risk associated with oysters harvested from a norovirus contaminated area, Ireland, February-March 2010. EuroSurveillance 15 (19), 1–5.
- Drouaz, N., Schaeffer, J., Farkas, T., Le Pendu, J., Le Guyader, F.S., 2015. Tulane virus as a potential surrogate to mimic norovirus behaviour in oysters. Appl. Environ. Microbiol 81, 5249–5256.
- Duchenne-Moutien, R.A., Neetoo, H., 2021. Climate Change and Emerging Food Safety Issues: A Review. J. Food Protect. 84 (11), 1884–1897.
- EFSA, 2019. Analysis of the European baseline survey of norovirus in oysters. EFSA J. 17 (7), e05762. https://doi.org/10.2903/j.efsa.2019.5762.
- Esseili, M.A., Gao, X., Boley, P., Hou, Y., Saif, L.J., Brewer-Jensen, P., Lindesmith, L.C., Baric, R.S., Atmar, R.L., Wang, Q., 2019 Sep 8. Human norovirus histo-blood group antigen (HBGA) binding sites mediate the virus specific interactions with lettuce carbohydrates. Viruses 11 (9), 833. https://doi.org/10.3390/v11090833.
- EU, 2020. Regulation (EU) 2020/741 of the European Parliament And of the Council of 25 May 2020 on Minimum Requirements for Water Reuse. European Union, Brussels (Accessed 11th Jan, 2023).
- Farrell, H., Hayes, J., Laffey, J., Rowan, N., 2011. Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant Candida albicans. J. Microbiol. Methods 84 (2), 317–326.
- Fehrenbach, G.W., Pogue, R., Carter, F., Clifford, E., Rowan, N., 2022. Implications for the seafood industry, consumers and the environment arising from contamination of shellfish with pharmaceuticals, plastics and potentially toxic elements: a case study from Irish waters with a global orientation. Sci. Total Environ. 844. https://doi.org/10.1016/j. scitotenv.2022.157067.

- Fitzhenry, K., Clifford, E., Rowan, N., Val del Rio, A., 2021. Bacterial inactivation, photoreactivation and dark repair post flow-through pulsed UV disinfection. J.Water Process Eng. 41. https://doi.org/10.1016/j.jwpe.2021.102070.
- Flannery, J., Keaveny, S., Dore, W., 2009. Use of FRNA bacteriophages to indicate the risk of norovirus contamination in Irish oysters. J. Food Prot. 72, 2358–2362.
- Flannery, J., Keaveney, S., Rajko-Nenow, P., O'Flaherty, V., Doré, W.J., 2012. Concentration of norovirus during wastewater treatment and its impact on oyster contamination. Appl.Environ.Microbiol. 78 (9), 3400–3406.
- Franssen, F., et al., 2019. Inactivation of parasite transmission stages: efficacy of treatments on food of animal origin. Trends Food Sci. Technol. 83, 267–273.
- Galeano, L.A., Guerrero-Flórez, M., Sánchez, C.A., Gil, A., Vicente, M.Á., 2019. Disinfection by chemical oxidation methods. Applications of Advanced Oxidation Processes (AOPs) in Drinking Water Treatment. *The Handbook of Environmental Chemistry*. 67. Springer, Cham, pp. 257–295.
- Gao, S., et al., 2014. Inactivation of microorganisms by low-frequency high-power ultrasound:
  1. Effect of growth phase and capsule properties of the bacteria. Ultrason. Sonochem. 21 (1), 446–453. https://doi.org/10.1016/j.ultsonch.2013.06.006.
- Garcia, A.V., de Sa, S.H.G., Silva, G.D., Ballesteros, J.E.M., Barbieri, E., de Sousa, R.L.M., et al., 2020. Microbiological quality of shellfish and evaluation of compact dry EC for detecting total coliforms and Escherichia coli. Acta Aliment. 49, 32–39. https://doi.org/10.1556/ 066.2020.49.1.5.
- Garvey, M., Rowan, N., 2015. A pulsed light system for the disinfection of flow through water in the presence of inorganic contaminants. J. Water Health 13 (2), 406–412.
- Garvey, M., Rowan, N.J., 2019. Pulsed UV as a potential surface sanitizer in food production processes to ensure consumer safety. Curr. Opin. Food Sci. 26, 65–70.
- Garvey, M., Farrell, H., Cormican, M., Rowan, N., 2010. Investigations of the relationship between use of in vitro cell culture-quantitative PCR and a mouse-based bioassay for evaluating critical factors affecting the disinfection performance of pulsed UV light for treating Cryptosporidium parvum oocysts in saline. J. Microbiol. Methods 80, 267–273.
- Garvey, E., Clifford, E., O'Reilly, E., Rowan, N.J., 2013. Efficacy of using harmless Bacillus endospores to estimate the inactivation of Cryptosporidium parvum oocystes in water. J. Parasitol. 99 (3), 448–452.
- Garvey, M., Stocca, A., Rowan, N., 2014. Development of a combined in vitro cell culture quantitative PCR assay for evaluating the disinfection performance of pulsed light for treating the waterborne enteroparasite Cryptosporidium parvum. Exp. Parasitol. 144, 6–13.
- Garvey, M., Stocca, A., Rowan, N., 2014b. Development of a combined in vitro cell culture quantitative PCR assay for evaluating the disinfection performance of pulsed light for treating the waterborne enteroparasite Giardia lamblia. Exp. Parasitol. 144, 6–13.
- Garvey, M., Hayes, J., Clifford, E., Rowan, N.J., 2015. Ecological assessment of pulsed UV light treated water containing microbial species and Cryptosporidium parvum using a microbiotest test battery. WaterEnviron.J. 29 (1), 27–35.
- Gerard, C., et al., 2019. Inactivation of parasite transmission stages: efficacy of treatments on food of non-animal origin. Trends Food Sci. Technol. 91, 12–23.
- Gerba, C.P., Betancourt, W.Q., Katamima, M., Rock, C.M., 2018. Reducing uncertainty in estimating virus reduction by advanced water treatment processes. Water Res. 133, 282–288.
- Ghadimkhani, A., Zhang, W., Marhaba, T., 2016. Ceramic membrane defouling (cleaning) by air nano bubbles. Chemosphere 146, 379–384.
- Giannakis, S., Liu, S., et al., 2017. Effect of Fe(II)/Fe(III) species, pH, irradiance and bacterial presence on viral inactivation in wastewater by the photo-Fenton process. Kinetic modelling and mechanistic interpretation. Appl. Catal. B: Environ. 204, 159–166.
- Gómez-López, V.M., et al., 2022. Guidelines on reporting treatment conditions for emerging technologies in food processing. Critical reviews. Food Sci.Technol. 62 (21). https:// doi.org/10.1080/10408398.2021.1895058.
- Hallegraeff, G.M., 2010. Ocean climate change, phytoplankton community responses, and harmful algal bloom: a formidable predictive challenge. J. Phycology. 46 (2), 220–235.Hamza, I.A., et al., 2011. Methods to detection infectious human enteric viruses in environ-
- mental water samples. Int. J. Hyg. Environ. Health 214 (6), 424–436.
- Hassard, F., Sharp, J.H., et al., 2017. Critical review on the public health impact of norovirus contamination in shellfish and the environment: a UK perspective. Food Environ. Virol. 1–19. https://doi.org/10.1007/s12560-017-9279-3.
- Hata, A., Hanamoto, S., Shirasaka, Y., Yamashita, N., Tanaka, H., 2016. Quantitative distribution of infectious F-specific RNA phage genotypes in surface waters. Appl. Environ. Microbiol. 82 (14), 424–4252.
- Havelaar, A.H., Galindo, A.V., Kurowika, D., Cooke, R.M., 2008. Attribution of foodborne pathogens using structured expert elicitation. Foodboren Pathog. Dis. 5, 649–659.
- Hayes, J.C., Garvey, M., Fogarty, A.M., Clifford, E., Rowan, N., 2012. Inactivation of recalcitrant protozoan oocysts and bacterial endospores in drinking water using high-intensity pulsed UV light irradiation. Water Sci.Technol.Water Supply 12 (4), 513–522.
- Hayes, J., Kirf, D., Garvey, M., Rowan, N., 2013. Disinfection and toxicological assessments of pulsed UV and pulsed-plasma gas-discharge treated-water containing the waterborne protozoan enteroparasite Cryptosporidium parvum. J. Microbiol. Methods 94 (3), 325–337.
- Hercher, M., et al., 1979. Detection and discrimination of individual viruses by flow cytometry. J. Histochem. Cytochem. 27 (1). https://doi.org/10.1177/27.1.374599.
  Holm, E.R., et al., 2008. Sonication of bacteria, phytoplankton and zooplankton: application
- Holm, E.K., et al., 2008. Solication of bacteria, phytopiankton and zoopiankton application to treatment of ballast water. Mar. Pollut. Bull. 56 (2008), 1201–1208. https://doi.org/ 10.1016/j.marpolbul.2008.02.007.
- Huang, C.-C., Hsu, Z.-H., Lai, Y.S., 2021. Raman spectroscopy for virus detection and the implementation of unorthodox food safety. Trends Food Sci. Technol. 116, 525–532.
- Hunt, K., Doré, B., Keaveney, S., Rupnik, A., Butler, F., 2020. Estimating the distribution of norovirus in individual oysters. Int. J. Microbiol. 333. https://doi.org/10.1016/j. ijfoodmicro.2020.108785.
- Hunt, K., Doré, B., Keavney, S., Rupnik, A., Butler, F., 2023. A quantitative exposure assessment model for norovirus in oysters harvested from a classified production area. Microbial Risk Anal. 23. https://doi.org/10.1016/j.mran.2023.100247.

- Hutson, A.M., Atmar, R.I, Graham, D.Y., Estes, M.K., 2002. Norwalk virus infection and disease is associated with ABO histo-blood group type. J. Infect Dis. 185 (9), 1335–1337.
- Intrafish, 2021. Toxic algae bloom causes big losses at Mowi Ireland salmon farming sites. https://www.intrafish.com/salmon/toxic-algae-bloom-causes-big-losses-at-mowiireland-salmon-farming-sites/2-1-1094592 (Accessed 18th Dec, 2022).
- ISO, 2017. ISO 15216-1: 2017 Microbiology of the Food Chain Horizontal Method for Determination of Hepatitis A Virus And Norovirus Using Real-time RT-PCR–Part 1: Method for Quantification. Method 15216-1:2017. International Organization for Standardization, Geneva.
- Jaykus, L.A., Green, D., 2015. Efficacy of ozonated microbubbles for vibrio control in postharvest processing of shellfish. https://www.nal.usda.gov/research-tools/food-safetyresearch-projects/efficacy-ozonated-microbubbles-vibrio-control-post (Accessed 21 Nov, 2022).
- Jean, J., et al., 2011. Inactivation of hepatitis A virus and norovirus surrogate in suspension and on food-contact surfaces using pulsed UV light (pulsed light inactivation of foodborne viruses). Food Microbiol. 28 (3), 568–572.
- Jegatheesan, V., Shu, L., Visvanathan, C., 2011. Aquaculture Effluent: Impacts and Remedies for Protecting the Environment and Human Health. Elsevier B.V, New York, pp. 123–135. https://doi.org/10.1016/B978-0-444-52272-6.00340-8.
- Jeong, H., et al., 2021. Effectiveness of depuration of Pacific oyster (Crassostrea gigas): removal of bioaccumulated Vibrio vulnificus by UV-treatment. Food Sci. Biotechnol. 30 (5), 765–771.
- Jubinville, E., et al., 2022. Inactivation of hepatitis A virus and norovirus on berries by broadspectrum pulsed light. Inter. J. Food Microbiol. 364. https://doi.org/10.1016/j. ijfoodmicro.2021.109529.
- Kim, J.H., et al., 2017. Sensitive quantification naked-eye biodetection with polyhedral Cu nanoshells. Adv. Mater. 37. https://doi.org/10.1002/adma.201702945.
- Kingsley, D., Kuis, R., Perez, R., et al., 2018. Oxygen-dependent laser inactivation of murine norovirus using visible light lasers. Virol. J. 15, 117. https://doi.org/10.1186/s12985-018-1019-2.
- Koda, S., et al., 2003. A standard method to calibrate sonochemical efficiency of an individual reaction system. Ultrason. Sonochem. 10, 149–156. https://doi.org/10.1016/S1350-4177(03)00084-1.
- Kokkinos, P., Venieri, D., Mantzavions, D., 2021. Advanced oxidation processes for water and wastewater viral disinfection: A systematic review. Food Environ. Virol. 13, 283–302.
- Kosel, J., Gutirrez-Aguirre, I., Rački, N., Dreo, T., Ravnikar, M., Dular, M., 2017. Efficient inactivation of MS-2 virus in water by hydrodynamic cavitation. Water Research 124, 465–471.. https://doi.org/10.1016/j.watres.2017.07.077.
- Kraus, A.A., et al., 2007. Comparison of plaque- and flow cytometry-based methods for measuring dengue virus neutralization. J. Clin. Microbiol. https://doi.org/10.1128/JCM. 00827-07.
- La Rosa, et al., 2012. Emerging and potentially emerging viruses in water environments. Ann 1st Super. Santa 48 (4), 397–406.
- Laso, J., et al., 2022. Achieving sustainability of the seafood sector in the European Atlantic Area by addressing Eco-Societal challenges: the NEPTUNUS project. Sustainability 14 (5). https://doi.org/10.3390/su14053054.
- Le Guyader, F.S., Loisy, F., Atmar, R.L., Hutson, A.M., Estes, M.K., Ruveoen-Clouet, N., Pommepuy, M., Le Pendu, J., 2006. Norwalk virus - specific binding to oyster digestive tissues. Emerg. Infect. Dis. 12 (6), 931–936.
- Le Guyader, F.S., Atmar, R.L., Le Pendu, J., 2012. Transmission of viruses through shellfish: When specific ligands come into play. Current Opinion in Virology 2 (1), 103–110.
- Leduc, A., Leclerc, M., Challant, J., Loutreul, J., Robin, M., Maul, A., et al., 2020. F-specific RNA bacteriophages model the behavior of human noroviruses during purifcation of oysters: the main mechanism is probably inactivation rather than release. Appl.Environ. Microbiol. https://doi.org/10.1128/AEM.00526-20.
- Lee, D.H., 2020. Shelf-life extension of raw oyster Crassostrea gigas by depuration process. Korean J.Fish.Aquat.Sci. 53 (6), 842–850.
- Lees, D., Cen, W.T., 2010. International standardization of a method for detection of human pathogenic viruses in molluscan shellfish. Food Environ. Virol. 2, 146–155.
- Lin, Z., Zhao, X., Huang, J., Liu, W., Zheng, Y., Yang, X., et al., 2019. Rapid screening of colistin-resistant *Escherichia coli, Acinetobacter baumannii* and *Pseudomonas aeruginosa* by the use of Raman spectroscopy and hierarchical cluster analysis. Analyst 144, 2803–2810. https://doi.org/10.1039/c8an02220h.
- Love, D.C., Lovelace, G.L., Sobsey, M.D., 2010. Removal of Escherichia coli, Enterococcus fecalis, coliphages MS2, poliovirus and hepatitis A virus from oysters (Crassostrea virginica) and hardshell clam (Mercinaria mercinaria) by depuration. Int. J. Food Microbiol. 143, 211–217.
- Lowther, J.A., Avant, J.M., Giznski, K., Rangdale, R.E., Lees, D.N., 2010. Comparison between quantitative real-time reverse transcription PCR results for norovirus in oysters and self-reported gastroenteric illness in restaurant customers. J. Food Prot. 73, 305–311.
- Lowther, J.A., Gustar, N.E., Powell, A.L., Hartnell, R.E., Lees, D.N., 2012. Two-year systematic study to assess novovirus contamination in oysters from commercial harvesting areas in the United Kingdom. Appl. Environ. Microbiol. 78, 5812–5817.
- Lowther, J.A., Cross, L., Stapleton, T., Gustar, N.E., Walker, D.I., Sills, M., Treagus, S., Pollington, V., Lees, D.N., 2019. Use of F-specific RNA bacteriophage to estimated infectious norovirus levels in oysters. Food Environ. Virol. 11, 247–258.
- Lu, S., Wang, N., Wang, C., 2018. Oxidation and biotoxicity assessment of microcystin-LR using different AOPs based on UV, O3, H2O2. Front. Environ. Sci. Eng. 12 (12). https://doi.org/10.1007/s11783-018-1030-2.
- Maalouf, M., Pommepuy, M., Le Guyader, F.S., 2010. Environmental conditions leading to shellfish contamination and related outbreaks. Food Environ. Virol. 2, 136–145.
- Maalouf, H., Schaefer, J., Parnaudeau, S., Le Pendu, J., Atmar, R.L., Crawford, S.E., Le Guyader, F.S., 2011. Strain-dependent norovirus bioaccumulation in oysters. Appl. Environ. Microbiol. 77 (10), 3189–3196.

- Maclean, M., et al., 2009. Inactivation of bacterial pathogens following exposure to light from a 405-nanometer light-emitting diode array. Appl. Environ. Microbiol. 75 (7), 1932–1937.
- Marie, D., et al., 1999. Enumeration of marine viruses in culture and natural samples by flow cytometry. Appl. Environ. Microbiol. https://doi.org/10.1128/AEM.65.1.45-52.1999.
- Martin-Bravo, M., et al., 2021. Minimal design principles for icosahedral virus capsids. ACS Nano 15 (9), 14873–14884.
- Martinez-Porchas, M., Martinez-Cordova, L.R., Lopez-Elias, J.A., Porchas-Cornejo, M.A., (2014). Bioremediation of aquaculture effluents. Microbial Biodegradation and Bioremediation. 1st edn Elsevier, London, pp. 542–555. https://doi.org/10.1016/B978-0-12-800021-2.00024-8.
- Mathijs, E., Stals, L., Baert, N., et al., 2012. A review of known and hypothetical transmission routes for noroviruses. Food Environ. Virol. 4, 131–152. https://doi.org/10.1007/ s12560-012-9091-z.
- Mauriz, E., 2020. Recent progress in planonic biosensing schemes for virus detection. Sensors 17, 4745. https://doi.org/10.3390/s20174745.
- McDonnell, G., Russell, A.D., 1999. Antiseptics and disinfectants: activity, action, and resistance. Clin. Microbiol. Rev. 12, 147–179.
- McEvoy, B., Maksimovic, A., Rowan, N., 2023. Geobacillus stearothermophilus and Bacillus atrophaeus spores exhibit linear inactivation kinetic performance when treated with an industrial scale vaporized hydrogen peroxide (VH<sub>2</sub>O<sub>2</sub>) sterilization process. J. Appl. Microbiol. 134 (1).. https://doi.org/10.1093/jambio/lxac028.
- McEvoy, B., Rowan, N.J., 2019. Terminal sterilization of medical devices using vaporized hydrogen peroxide: a review of current methods and emerging opportunities. J. Appl. Microbiol. 127 (5), 1403–1420.
- McEvoy, B., Lynch, M., Rowan, N.J., 2021. Opportunities for the application of real-time bacterial cell analysis using flow cytometry for the advancement of sterilization microbiology. J. Appl. Microbiol. 130 (6), 1704–1812.
- McLeod, C., Polo, D., Le Saux, J.C., Le Guyader, F.S., 2017. Depuration and relaying: a review on potential removal of norovirus from oysters. Compr. Rev. Food Sci. Food Saf. https:// doi.org/10.1111/1541-4337.12271.
- McMenemy, P., Kleckowski, A., Lees, D.N., Lowther, J., Taylor, N., 2018. A model for estimating pathogen variability in shellfish and predicting minimum depuration times. PLoS One 13, e0193865-14 doi:1371/journal.pone.0193865.
- McSharry, J.J., 1994. Uses of flow cytometry in virology. Clinical Microbiology Reviews. https://doi.org/10.1128/CMR.7.4.576.
- Mi, F., Guan, M., Hu, C., Peng, F., Sun, S., Wang, X., 2021. Application of lectin-based biosensor technology in the detection of foodborne pathogenic bacteria: A review. Analyst 146, 429–443.
- Milman, O., 2022. Hottest ocean temperatures in history recorded last year. https://www. theguardian.com/environment/2022/jan/11/oceans-hottest-temperatures-researchclimate-crisis Accessed 11 Jan, 2023.
- Monsen, T., et al., 2009. In vitro effect of ultrasound on bacteria and suggested protocol for sonication and diagnosis of prosthetic infections. J. Clin. Microbiol. 47 (2009), 2496–2501. https://doi.org/10.1128/JCM.02316-08.
- Mori, H., et al., 2018. Raman spectroscopy insight into norovirus encapsulated in Bombyx mori cypovirus cubic microcrystals. Spectrochim. Acta A Mol. Biomol. Spectrosc. 203 (2018), 19–30.
- Murray, I.M.T., Rowan, N., McNamee, S., Campbell, K., Fogarty, A.M., 2017. Pulsed light reduces the toxicity of the algal toxin okadaic acid to freshwater crustacean Daphnia pulex. Environ.Sci.Pollut.Res. 25, 607–614. https://doi.org/10.1007/s11356-017-0472-6.
- Nasheri, N., Harlow, J., Chen, A., Corneau, N., Bidawid, S., 2021. Survival and Inactivation by Advanced Oxidative Process of Foodborne Viruses in Model Low-Moisture Foods. Food Environ. Virol. 13 (1), 107–116.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: a case study from the Republic of Ireland. Aquaculture 526. https://doi.org/10.1016/j.aquaculture.2020.735377.
- O'Neill, E.A., Rowan, N.J., 2021. Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance – a comparative case study from the Republic of Ireland. Sci. Total Environ. 802. https://doi.org/10.1016/j.scitotenv.2021.149800.
- O'Neill, E.A., Rowan, N.J., 2022c. Potential Disruptive Effects of Zoosporic Parasites on Peatland-based Organic Freshwater Aquaculture: Case study from the Republic of Ireland. Science of the Total Environment. Peatland-based innovation can potentially support and enable the sustainable development goals of the United Nations: Case study from the Republic of Ireland. Case Stud. Chem. Environ. Eng. 6. https://doi.org/10.1016/j. cscee.2022.100251.
- O'Neill, E.A., Fehrenbach, G., Murphy, E., Alencar, S.A., Pogue, R., Rowan, N., 2022a. Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: case study from the Republic of Ireland. Sci. Total Environ. 851. https://doi.org/10. 1016/j.scitotenv.2022.158392.
- O'Neill, E.A., Morse, A.P., Rowan, N.J., 2022b. Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: implications and opportunities for advancing research and disruptive innovation post COVID-19 era. Sci. Total Environ. 819. https://doi.org/10.1016/j. scitotenv.2022.153073.
- O'Neill, E.A., McKeon Bennett, M., Rowan, N.J., 2022c. Peatland-based innovation can potentially support and enable the sustainable development goals of the United Nations: case study from the Republic of Ireland. Case Stud. Chem. Environ. Eng. 6. https://doi.org/ 10.1016/j.cscee.2022.100251.
- Noorian, P., Hoque, M.M., Espinoza-Vergara, G., McDougald, D., 2023. In: Almagro-Moreno, S., Pukatzki, S. (Eds.), Vibrio spp. Infections. Advances in Experimental Medicine and Biology1404. Springer, Cham .. https://doi.org/10.1007/978-3-031-22997-8\_6.

- O'Neill, E., Rowan, N.J., 2023. Potential disruptive effects of zoosporic parasites on peatlandbased organic freshwater aquaculture: Case study from the Republic of Ireland. Sci. Total Environ. 868. https://doi.org/10.1016/j.scitotenv.2023.161495.
- O'Neill, E., Rowan, N.J., Fogarty, A., 2019. Novel use of the alga Pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692.. https://doi.org/10.1016/j.scitotenv.2019.07.243.
- Pecson, B.M., et al., 2011. Framework for Using Quantitative PCR as a Nonculture Based Method To Estimate Virus Infectivity. Environ. Sci. Technol. 45 (6), 2257–2263.
- Pexara, A., Govaris, A., 2020. Foodborne viruses and innovation non-thermal food-processing technologies. Foods 9 (11), 1520. https://doi.org/10.3390/foods9111520.
- Phuvasate, S., Chen, M.H., Su, Y.C., 2012. Reductions of Vibrio parahaemolyticus in Pacific oysters (Crassostrea gigas) by depuration at various temperatures. Food Microbiol. 31, 51–56.
- Pilotto, M.R., Souza, D.S.M., Barardi, C.R.M., 2019. Viral uptake and stability in Crassostrea gigas oysters during depuration, storage and steaming. Mar. Pollut. Bull. 149, 110524. https://doi.org/10.1016/j.marpolbul.2019.110524.
- Plant,e, D., Bran Barrera, J.A., Lord, M, Iugovaz, I, Nasheri, N., 2021. Development of an RNA Extraction Protocol for Norovirus from Raw Oysters and Detection by qRT-PCR and Droplet-Digital RT-PCR. Foods 10 (8), 1804.. https://doi.org/10.3390/foods10081804.
- Polo, D., et al., 2014. Depuration kinetics of murine norovirus in shellfish. Food Res. Int. 64, 182–187.
- Polo, D., Feal, X., Romalde, J.L., 2015. Mathematical model for viral depuration kinetics in shellfish: an unusual tool to estimate the risk for the consumers. Food Microbiol. 49, 220–225.
- Ponterio, E., et al., 2013. Pattern of activation of human antigen presenting cells by genotype GII.4. Norovirus virus-like particles. J. Tansl. Med. 24 (11), 127. https://doi.org/10. 1186/1479-5876-11-127.
- Pouillot, R., Smith, M., Van Doren, J.M., Catford, A., et al., 2021. Risk assessment of norovirus illness from consumption of raw oysters in the United States and in Canada. Risk Anal. 42 (2), 344–369.
- Qin, Q., Shen, J., Reece, K.S., 2022. A deterministic model for understanding nonlinear viral dynamics in oysters. Appl. Environ. Microbiol. 88 (8), e0236021.
- Rajko-Nenow, P., Keaveney, S., Flannery, J., McIntyre, A., Dore, W., 2013. Norovirus genotypes implicated in two oyster-related illness outbreaks in Ireland. Epidemiol. Infect. 142 (10), 2096–2104. https://doi.org/10.1017/S0950268813003014.
- Razafimahefa, R.M., Ludwig-Begall, L.F., Thiry, E., 2020. Cockles and mussels, alive, alive, oh: the role of bivalve molluscs as transmission vehicles for human norovirus infections. Transbound. Emerg. Dis. 67, 9–25. https://doi.org/10.1111/tbed.13165.
- Razafimahefa, R.M., et al., 2021. Development of a specific anti-capsid antibody and magnetic bead based immunoassay to detect human norovirus particles in stool samples and spiked mussels via flow cytometry. Food Environ. Virol. 13 (4\0), 493–506.
- Razafimahefa, R.M., et al., 2021b. Optimization of a PMAxx –RT-qPCR assay and the preceding extraction method to selectively detect infectious murine norovirus particles in mussels. Food Microbiol. Virol. 13 (1), 93–106.
- Reid, R.A., Timothy, T.D., 2000. The US National Shellfish Sanitation Program. Foodborne Disease Handbook. CRC Press, pp. 341–358.
- Rindi, F. (2014). Terrestrial green algae: systematics, biogeography and expected responses to climate change. T.R. Hodkinson, M.B. Jones, S. Waldren, J.A.N. Parnell (Eds.), Climate Change, Ecology and Systematics, Cambridge University Press, New York (2014), pp. 201-227.
- Romalde, J.L., Estes, M.K., et al., 1994. In situ detection of hepatitis A virus in cell cultures and shellfish tissues. Appl. Environ. Microbiol. 60 (6), 1921–1926.
- Rosado-Lausell, S.L., et al., 2013. Roles of singlet oxygen and triplet state of dissolved organic matter formed by different organic matters in bacteriophage MS2 inactivation. Water Res. 47 (14), 4869–4879.
- Rowan, N.J., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, implications, and outlooks. Int. J. Microbiol. 2011, 462832. https://doi.org/10.1155/2011/462832.
- Rowan, N.J., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo vadis? Trends Food Sci. Technol. 88, 316–332.
- Rowan, N.J., 2022. The role of digital technologies in supporting and improving fishery and aquaculture across the supply chain – Quo Vadis? Aquac.Fish. https://doi.org/10.1016/ j.aaf.2022.06.003.
- Rowan, N.J., Casey, O., 2021. Empower Eco multiactor HUB: a triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. Curr.Opin.Environ.Sci.Health 21. https://doi.org/10. 1016/j.coesh.2021.100254.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in Agri-food and green deal innovations. Quo Vadis? Sci. Total Environ. 746. https://doi.org/10.1016/j.scitotenv.2020.141362.
- Rowan, N.J., MacGregor, S.J., Anderson, J.G., Fouracre, R.A., McIIvaney, L., Farison, O., 1999. Pulsed light inactivation of food-related microorganisms. Appl. Environ. Microbiol. 65 (3), 1312–1315.
- Rowan, N.J., Espie, S., Harrower, J., Anderson, J.G., Marsili, L., MacGregor, S.J., 2007. Pulsed-plasma gas-discharge inactivation of microbial pathogens in chilled poultry wash water. J. Food Prot. 70 (12), 2805–2810.
- Rowan, N.J., Valdramidis, V.P., Gomez-Lopez, V.M., 2015. A review of quantitative methods to describe efficacy of pulsed light generated inactivation data that embraces the occurrence of viable but non culturable state microorganisms. Trends Food Sci. Technol. 44, 79–92.
- Rowan, N. J. et al. (2000). Pulsed electric field inactivation of diarrhoeagenic Bacillus cereus through irreversible electroporation, Letters in Applied Microbiology, Volume 31, Issue 2, 1 August 2000, Pages 110–114, https://doi.org/10.1046/j.1365-2672.2000.00772.x.
- Rowan, N.J., Moral, R.F., 2021. Disposable face masks and reusable face coverings as nonpharmaceutical interventions (NPIs) to prevent transmission of SARS-CoV-2 variants

N.J. Rowan

that cause coronavirus disease (COVID-19): Role of new sustainable NPI design innovations and predictive mathematical modelling. Sci. Total Environ. 772.. https://doi.org/ 10.1016/j.scitotenv.2021.145530.

- Rowan, N.J., Murray, N., Qiao, Y., O'Neill, E., Clifford, E., Barcelo, D., Power, D.M., 2022. Digital transformation of peatland eco-innovations ('Paludiculture'): enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products and services. Sci. Total Environ. 838 (3). https://doi.org/10.1016/j.scitotenv.2022.156328.
- Ruiz-Salmön, et al., 2020. Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Curr.Opin.Environ.Sci.Health 13, 101–106.
- Ruiz-Salmön, et al., 2021. Life cycle assessment of fish and seafood processed products: a review of methodologies and new challenges. Sci. Total Environ. 761. https://doi.org/10.1016/j.scitotenv.2020.144094.
- Rupnik, A., Keaveney, S., Devilly, L., et al., 2018. The impact of winter relocation and depuration on norovirus concentrations in Pacific oysters harvested from a commercial production site. Food Environ. Virol. 10, 288–296. https://doi.org/10.1007/s12560-018-9345-5.
- Rupnik, A., Doré, W., Devilly, L., et al., 2021. Evaluation of norovirus reduction in environmentally contaminated Pacific oysters during laboratory controlled and commercial depuration. Food Environ. Virol. 13, 229–240. https://doi.org/10.1007/s12560-021-09464-2.
- Šarc, A., et al., 2014. Can rapid pressure decrease induced by supercavitation efficiently eradicate legionella pneumophila bacteria? Desalin. Water Treat. 57 (2014), 2184–2194. https://doi.org/10.1080/19443994.2014.979240.
- Sarmaja-Korjonen, K., Nyman, M., Kultti, S., Valiranta, M., 2006. Palaeolimnological development of lake Njargajavir, Nothern Finnish Lapland, in a change holocence climate and environment. J. Paleolimnology 35, 65–81.
- Semenza, J.C., Paz, S., 2021. Climate change and infectious disease in Europe: impact, projection and adaptation. <sb:contribution><sb:title>Lancet Reg. Health</sb:title> </sb: contribution><sb:host><sb:issue><sb:series><sb:title>Eur.</sb:title></sb:series></ sb:issue></sb:host> 9. https://doi.org/10.1016/j.lanepe.2021.100230.
- Shabat-Hadas, E., Mamane, H., Gitis, V., 2017. Rhoaamine B in dissolved and non-bound forms: indicators for light-based advanced oxidation processes. Chemospher. 184, 1020–1027.
- Sharp, J.H., Clements, K., Diggens, M.N., McDonald, K.E., Malham, S.K., Jones, D.L., 2021. E. coli is a poor end-product criterion for assessing the general microbial risk posed from consuming norovirus contaminated shellfish. Front. Microbiol. https://doi.org/10. 3389/fmicb.2021.608888.
- Sikder, M.N.A., Min, W.W., Ziyad, A.O., Prem Kumar, P., Dinesh Kumar, R. (2016). Sustain[1] able treatment of aquaculture effluents in future-a review. International Research Journal of Advanced Engineering and Science 1 (4), 190–193 Available at: http://www.irjaes. com/pdf/IRJAES-V1N4Y16/IRJAES-V1N4P209Y16.pdf (Accessed 24 Feb, 2023).
- Smith, H.Q., Smith, T.J., 2019. The dynamic capsid structures of the Noroviruses. Viruses 11 (3), 235. https://doi.org/10.3390/v11030235 Mar 8.
- Su, X., et al., 2010. Inactivation of human enteric virus surrogates by high-intensity ultrasound. Foodborne Pathog. Dis. 7 (2010), 1055–1061. https://doi.org/10.1089/fpd. 2009.0515.
- Sun, L., Zhang, F., Guo, X., Qiao, Z., Zhu, Y., Jin, N., Cui, Y., Yang, W., 2022. Research progress on bulk nanobubbles. Particuology 60, 99–106.
- Tahar, A., Tiedeken, E., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. 603–604, 627–638. https://doi.org/10.1016/j.scitotenv. 2017.05.227.
- Timchak, E., Gitis, V., 2012. A combined degradation of dyes and inactivation of viruses by UV/H<sub>2</sub>O<sub>2</sub>. Chem. Eng. J. 192, 164–170.
- Tomb, R.M., et al., 2018. Review of the comparative susceptibility of microbial species to photoinactivation using 380–480 nm violet-blue light. Photochem. Photobiol. 94 (3), 445–458.
- Tsen, K.T., et al., 2007. Inactivation of viruses by coherent excitations with a low power visible femtosecond laser. Virol. J. 4, 50.
- Tsen, S.W., et al., 2012. Inactivation of enveloped virus by laser-driven protein aggregation. J. Biomed. Optics 17 (12), 128002.
- Tsen, S.W., et al., 2014. Studies of inactivation mechanism of non-enveloped icosahedral viruses by a visible ultrashort pulsed laser. Virol. J. 11, 20 2014a.
- Ueki, Y., Shoji, M., et al., 2007. Persistence of calciciviruses in artificially-contaminated oysters during depuration. Appl. Environ. Microbiol. 73, 5698–5701.

- U.S. EPA, 2021. Climate Change Indicators: sea surface temperature. https://www.epa.gov/ climate-indicators/climate-change-indicators-sea-surface-temperature (Accessed 11th Jan, 2023).
- Uslu, G. et al. (2016). Disinfection of synthetic and real municipal wastewater effluent by flow-through pulsed UV-light treatment system. J. Waste Processing Eng. 10, 89-97. https://doi.org/10.1016/j.jwpe.2016.02.004.
- Usuldin, S.R.A., et al., 2021. In vivo toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications. Sci. Rep. 11, 23079. https://doi.org/10.1038/s41598-021-02486-7.
- Van Nguyen, T., Alfaro, A.C., 2019. Application of flow cytometry in molluscan immunology: current status and trends. FishShellfish Immnol. 94, 239–248.
- Vimont, A., Fliss, I., Jean, J., 2015. Efficacy and mechanisms of murine norovirus inhibition by pulsed-light technology. Appl.Envrion. Microbiol. 81 (8), 2950–2957.
- Vlasov, A.V.; Maliar, N.L.; Bazhenov, S.V.; Nikelshparg, E.I.; Brazhe, N.A.; Vlasova, A.D.; Osipov, S.D.; Sudarev, V.V.; Ryzhykau, Y.L.; Bogorodskiy, A.O.; Zinovev, E.V.; Rogachev, A.V.; Manukhov, I.V.; Borshchevskiy, V.I.; Kuklin, A.I.; Pokorný, J.; Sosnovtseva, O.; Maksimov, G.V.; Gordeliy, V.I. Raman Scattering: From Structural Biology to Medical Applications. Crystals 2020, 10, 38. https://doi.org/10.3390/cryst10010038.
- Wang, J.C.-Y., Mukhopadhyay, S., Zlotnick, A., 2018. Geometric defects and icosahedral viruses. Viruses 10 (1), 25. https://doi.org/10.3390/v10010025.
- Wan-Mohtar, W.A.A.Q.I., Ilham, Z., Jamaludin, A.A., Rowan, N.J., 2012. Use of zebrafish embryo assay to evaluate toxicity and safety of bioreactor-grown exopolysaccharides and endopolysaccharides from European Ganoderma applanatum mycelium for future aquaculture applications. Int. J. Mol. Sci. 22 (4), 1675. https://doi.org/10.3390/ ijms22041675.
- Wen, D., et al., 2022. Evaluating disinfection performance of ultraviolet light-emitting diodes against the microalga Tetraselmis sp.: assay methods, inactivation efficiencies, and action spectrum. Chemosphere 308 (1). https://doi.org/10.1016/j.chemosphere.2022.136113.
- Wikswo, M.E., Karst, S.M., Hall, A.J., 2021. Norovius. In: Morris, J. (Ed.), Foodborne Infections And Intoxications. Elsevier, San Diego, pp. 287–306.
- Winder, M., Sommer, U., 2012. Phytoplankton response to a changing climate. Hydrobiologia 698, 5–16. https://doi.org/10.1007/s10750-012-1149-2.
- Winterbourn, J.B., et al., 2016. Use of Mytilus edulis biosentiels to investigate spatial patterns of norovirus faecal indicator organism contamination around coastal sewage discharges. Water Res. 15, 241–250.
- Wood, J.W., Burkhardt III, W., 2010. Occurrence of Norovirus and Hepatitis A virus in U.S. Oysters. Food Environ. Virol. 2, 176–182.
- Wu, Z., et al., 2012. Removal of blue-green algae using the hybrid method of hydrodynamic cavitation and ozonation. J. Hazard. Mater. 235–236 (2012), 152–158. https://doi.org/ 10.1016/j.
- Yang, S.Y., et al., 2008. Micro flow cytometry utilizing a magnetic bead-based immunoassay for rapid virus detection. Biosens. Bioelectron. 24 (4), 855–862.
- Yapıcıoğlu, P.S., 2018. Environmental impact assessment for a meat processing industry in Turkey: Wastewater treatment plant. Water Prac. Tech. 13 (3), 692–704.. https://doi. org/10.2166/wpt.2018.051.
- Yin, M., Jing, C., Li, H., Deng, Q., Wang, S., 2020. Surface chemistry modified upconversion nanoparticles as fluorescent sensor array for discrimination of foodborne pathogenic bacteria. J. Nanobiotechnol. 18, 41. https://doi.org/10.1186/s12951-020-00596-4.
- Younger, A.D., Neish, A., Walker, D.I., Jenkins, K.L., Lowther, J.A., Stapleton, T.A., Alves, M.T., 2020. Strategies to reduce norovirus (NoV) contamination from oysters under depuration conditions. Food Chem. Toxicol. https://doi.org/10.1016/j.fct.2020.111509.
- Zhang, G., et al., 2006. Ultrasonic damages on cyanobacterial photosynthesis. Ultrason. Sonochem. 13, 501–505.
- Zhang, J., Gao, P., Wu, Y., Yan, X., Ye, C., Liang, W., Yan, M., Xu, X., Jiang, H., 2022. Identification of foodborne pathogenic bacteria using confocal Raman microspectroscopy and chemometrics. Front. Microbiol. 13, 874658. https://doi.org/10.3389/fmicb.2022. 874658.
- Zhang, Z.-H., Wang, S., Cheng, L., Ma, H., Gao, X., Brennan, C.S., et al., 2022b. Micro-nanobubble technology and its applications in food industry: a critical review. Food Rev.Int. https://doi.org/10.1080/87559129.2021.2023172.
- Zhou, S., et al., 2022. The effect of preparation time and aeration rate on the properties of bulk micro-nanobubble water using hydrodynamic cavitation. Ultrason. Sonochem. 84. https://doi.org/10.1016/j.ultsonch.2022.105965.
- Zupanc, M., et al., 2019. Effects of cavitation on different microorganisms: the current understanding of the mechanisms taking place behind the phenomenon. A review and proposals for further research. Ultrason.Sonochem. 57, 147–165.



Contents lists available at ScienceDirect

# Science of the Total Environment



journal homepage: www.elsevier.com/locate/scitotenv

Review

# Life cycle assessment of fish and seafood processed products – A review of methodologies and new challenges



Israel Ruiz-Salmón <sup>a,\*</sup>, Jara Laso <sup>a</sup>, María Margallo <sup>a</sup>, Pedro Villanueva-Rey <sup>b</sup>, Eduardo Rodríguez <sup>b</sup>, Paula Quinteiro <sup>c</sup>, Ana Cláudia Dias <sup>c</sup>, Cheila Almeida <sup>d</sup>, Maria Leonor Nunes <sup>d,e</sup>, António Marques <sup>d,e</sup>, Antonio Cortés <sup>f</sup>, María Teresa Moreira <sup>f</sup>, Gumersindo Feijoo <sup>f</sup>, Philippe Loubet <sup>g</sup>, Guido Sonnemann <sup>g</sup>, Andrew P. Morse <sup>h</sup>, Ronan Cooney <sup>i,j</sup>, Eoghan Clifford <sup>i,j</sup>, Leticia Regueiro <sup>k</sup>, Diego Méndez <sup>k</sup>, Clémentine Anglada <sup>1</sup>, Christelle Noirot <sup>1</sup>, Neil Rowan <sup>m</sup>, Ian Vázquez-Rowe <sup>n</sup>, Rubén Aldaco <sup>a</sup>

- <sup>d</sup> IPMA Instituto Português do Mar e da Atmosfera (IPMA), Divisão de Aquacultura, Valorização e Bioprospeção, Avenida Doutor Alfredo Magalhães Ramalho 6, 1495-165 Lisboa, Portugal
- e CIIMAR Centro Interdisciplinar de Investigação Marinha e Ambiental, Terminal de Cruzeiros do Porto de Leixões, Avenida General Norton de Matos, S/N, 4450-208 Matosinhos, Portugal
- <sup>f</sup> Department of Chemical Engineering, Institute of Technology, Universidade de Santiago de Compostela, 15782 Santiago de Compostela, Galicia, Spain

<sup>g</sup> Univ. Bordeaux, CNRS, Bordeaux INP, ISM, UMR 5255, F-33400 Talence, France

<sup>h</sup> School of Environmental Sciences, University of Liverpool, Liverpool, UK

- <sup>i</sup> School of Engineering, NUI Galway, Galway H91 HX31 j Ryan Institute, NUI Galway, H91 TK33
- <sup>j</sup> Ryan Institute, NUI Galway, Ireland

k ANFACO-CECOPESCA, Campus University 16, 36310 Vigo PO, Spain

- <sup>1</sup> VertigoLab, Darwin Ecosystème, 87 Quai de Queyries, 33100 Bordeaux, France
- <sup>m</sup> Bioscience Research Institute, Athlone Institute of Technology, Ireland
- <sup>n</sup> Peruvian LCA Network (PELCAN), Department of Engineering, Pontificia Universidad Católica del Perú, Avenida Universitaria 1801, San Miguel, 15088 Lima, Peru

# HIGHLIGHTS

# GRAPHICAL ABSTRACT

- The review dissects 59 LCA studies about seafood, 90% of them over the last decade.
- LCA methodologies, the origin of the research centres and fish species are reviewed.
- LCA is key to climate change mitigation, energy and food sustainability and security.
- Challenges and potential opportunities for the seafood sector are addressed.
- Nexus water-energy-food to reduce environmental impact getting positive synergies.

# ARTICLE INFO

Article history: Received 23 September 2020 Received in revised form 17 November 2020 Accepted 22 November 2020 Available online 14 December 2020



# ABSTRACT

Life cycle assessment (LCA) has been widely applied in many different sectors, but the marine products and seafood segment have received relatively little attention in the past. In recent decades, global fish production experienced sustained growth and peaked at about 179 million tonnes in 2018. Consequently, increased interest in the environmental implications of fishery products along the supply chain, namely from capture to end of life, was recently experienced by society, industry and policy-makers.

\* Corresponding author.

E-mail address: Israel.ruizsalmon@unican.es (I. Ruiz-Salmón).

https://doi.org/10.1016/j.scitotenv.2020.144094 0048-9697/© 2020 Elsevier B.V. All rights reserved.

<sup>&</sup>lt;sup>a</sup> Departamento de Ingenierías Química y Biomolecular, Universidad de Cantabria, Avda. de Los Castros, s.n., 39005 Santander, Spain

<sup>&</sup>lt;sup>b</sup> EnergyLab, Fonte das Abelleiras s/n, Campus Universidad de Vigo, 36310 Vigo, Galicia, Spain

<sup>&</sup>lt;sup>c</sup> Centre for Environmental and Marine Studies (CESAM), Department of Environment and Planning, University of Aveiro, Campus Universitário de Santiago, 3810-193 Aveiro, Portugal

I. Ruiz-Salmón, J. Laso, M. Margallo et al.

Editor: Damia Barcelo

Keywords: Life cycle assessment Seafood Fisheries Nexus Environmental impacts Sustainability This timely review aims to describe the current framework of LCA and its application to the seafood sector that mainly focused on fish extraction and processing, but it also encompassed the remaining stages. An excess of 60 studies conducted over the last decade, along with some additional publications, were comprehensively reviewed; these focused on the main LCA methodological choices, including but not limited to, functional unit, system boundaries allocation methods and environmental indicators.

The review identifies key recommendations on the progression of LCA for this increasingly important sustaining seafood sector. Specifically, these recommendations include (i) the need for specific indicators for fish-related activities, (ii) the target species and their geographical origin, (iii) knowledge and technology transfer and, (iv) the application and implementation of key recommendations from LCA research that will improve the accuracy of LCA models in this sector. Furthermore, the review comprises a section addressing previous and current challenges of the seafood sector. Wastewater treatment, ghost fishing or climate change, are also the objects of discussion together with advocating support for the water-energy-food nexus as a valuable tool to minimize environmental negativities and to frame successful synergies.

© 2020 Elsevier B.V. All rights reserved.

## Contents

1	Introd	luction		2		
1. 2	Mator	ials and m		2		
2.	Poculi	tais and dice		5		
5.				Ј г		
	3.1.	Goal and		2		
		3.1.1.		5		
		3.1.2.	System boundaries	5		
		3.1.3.	Allocations	10		
		3.1.4.	Sensitivity analysis	12		
	3.2.	Life cycle	: impact assessment (LCIA)	12		
		3.2.1.	Impact categories in LCIA	12		
		3.2.2.	Fishery-specific impacts categories	13		
	3.3.	Seafood I	LCA practitioners: who and why	16		
	3.4.	Nexus w	ater-energy-food in the seafood sector.	17		
	35	Challeng	es and onnorthunities for LCA in the seafood sector	18		
	5.5.	351	Marina debria	18		
		J.J.I. 2 E D	Warte utofis	10		
		5.5.Z.		19		
		3.5.3.	Packaging	15		
		3.5.4.	Best available techniques	19		
		3.5.5.	Climate change	19		
4.	Concl	usions		20		
Decl	aration	of compe	ting interest	21		
Ackr	owled	gement .		21		
App	endix A	. Suppl	ementary data	21		
Refe	rences		-	21		

# 1. Introduction

The 17 Sustainable Development Goals (SDG), launched by the United Nations (UN) for the 2030 agenda, seek "*a shared blueprint for peace and prosperity for people and the planet, now and into the future*" (Sustainable Development Goals, 2020). Among these, SDG-14, "Conserve and sustainably use the oceans, seas and marine resources for sustainable development", specifically addresses the priceless environmental, cultural and social wealth of these water bodies that produce half of the oxygen we breath and provide 16% of the animal protein we eat (European Commission, 2019a). Therefore, addressing these goals are essential for human survival, not only from a biological, but also from a sustaining socioeconomic perspective. More specifically, goal SDG-14 focuses on the widely known 'blue economy'; that is, all activities related to oceans, seas and littoral environments including marine living resources, marine extraction of non-living resources, maritime transport, port activities, shipbuilding and repair, and coastal tourism.

In 2018, the production of global capture fisheries reached the highest level ever recorded at 96.4 million tonnes (live weight); whereas, the aquaculture sector attained another all-time record high of 82.1 million tonnes. The labour force involved in the primary sector

for fisheries and aquaculture is represented by a total of 59.5 million people (FAO, 2020). China leads the sector as attested to accounting for 35% of the global fish production that is followed by other Asian countries (34%), America (14%), Europe (10%), Africa (7%) and Oceania (1%). In general terms, total fish production followed an increasing trend but was geographically differenced: Asia and Africa have almost doubled their production in the past two decades and America fluctuated since the peak of the mid-1990s due to the strong influence of El Niño – Southern Oscillation on the abundance of anchoveta (*Engraulis ringens*). Whereas Europe suffered a gradual decline since the late 1980s that was buffered in recent years (FAO, 2020).

Beyond the macro data, the sustainability of the seafood sector involves maintaining a complex and dynamic equilibrium: namely, to: i) guarantee social protection for fishermen and fish farmers (Maritime Labour Convention, 2006) and subsidisation (Sumaila et al., 2019); ensure the periodical renewal of fishing grounds and the biodiversity preservation (Johnson et al., 2019), iii) keep the quality and security ruled by the standards of the food supply chain (Gephart et al., 2017); and, iv) face climate change consequences (Peck and Pinnegar, 2018) and other environmental negativities related to captures (e.g., ghost fishing, fuel leaks) or processing stages (e.g., waste streams). To deal with these complex issues, significant efforts have been made to establish an appropriate legal framework at different levels, particularly the need to progressively apply an environmental approach across many different sectors. Notably, the inclusion of environmental public criteria in decision-making is of paramount relevance, which address the provision of important information about the use of resources from raw materials to water and energy. Although the Blue Economy only represents 2.2% of employment in the European Union (EU)-28 along with a gross domestic product estimated at €15,900 billion (€13,500 without the UK) in 2018 (European Commission, 2020a), the continent has a long history of applying high standards related to environment, food production and consumption policies. The European Commission implemented a shared-management perspective of the oceans and seas through the Common Fisheries Policy (CFP) (European Commission, 2020b) in the 1970s and, more recently, a Green Deal to combat climate change and decouple economic growth from resource use. The implementation of the Circular Economy Action Plan reported in 2019 (European Commission, 2019b), the obligation of landing all discards in 2020 (European Commission, 2018a) or the ban of single-use plastics to minimize marine litter starting in 2021 (European Commission, 2018b) are some of the latest policy strategies in this regard (Ruiz-Salmón et al., 2020). In addition, EU members are also forced to promote policies under this framework and some of them are in place. For instance, Spain, with a coastline of almost 8000 km and the largest fishing industry in the EU, recently designated the agri-food and seafood sector as a priority intervention in the Spanish Circular Economy Strategy (SCES) (Spanish Government, 2020).

A quarter of global greenhouse gas emissions related to food production (Poore and Nemecek, 2018). Consequently, a grand challenge is to effectively promote the production of high-quality food at it' origin that have affordable prices along with commensurately reducing the impact derived from its production, both in terms of emissions, water use, nonvalued waste. Also, emphasis should be placed on reducing the impact derived from the generation of waste both in the field of production and arising from consumption. Notably, the SCES strongly recommends an increase in energy efficiency and a reduction of carbon dioxide (CO<sub>2</sub>) emissions from the seafood sector: thus, promoting efficient control of fisheries and data collection through improving knowledge in decision-making that underpin adaptation of products to consumer demand, reduce waste or improve recycling will add significant value. These recommendations will be enabled and advanced through applying life cycle assessment (LCA). LCA is the most established scientific tool for environmental analyses as it can fully address the quantification of footprints (e.g., energy, water, carbon) and environmental impact categories (e.g., global warming, eutrophication, human toxicity, etc.) along the life cycle of products and processes by studying the inputs (e.g., resources) and outputs (e.g., residues, by-products) of the system (ISO, 2006a, 2006b). Briefly, an LCA follows 4 inter-related stages: 1) statement of the goal and scope of the study, including the limits of the system and the functional unit; 2) development of inventory flows (inputs and outputs); 3) evaluation of the potential environmental impacts; and, 4) interpretation of the results drawing conclusions and recommendations.

Although LCA has been applied for decades, initially it focused on packaging, energy use and emission reductions in the 1960s. Following a gap in 1970s that coincided with methodological development, the first standardization occurred before the end of the millennium. Notable developments in the early 2000s included the release of the first datasets that emphasised increasing interest in the use of LCA as a potentially disruptive tool applied to national energy systems and waste management systems, among others. In the last two decades, LCA has experienced a methodological consolidation and a commensurate international collaboration across many sectors, such as business, research and innovation, product or process design, education, policy development, labelling, food and agriculture, consumer goods and energy industries (Hauschild et al., 2018). Nowadays, LCA is already part of the Single Market for Green Products Initiative launched by the European Commission (European Commission, 2013). Moreover, Product Environmental Footprint Category Rules (PEFCRs) are currently being updated with marine fish, including both fisheries and aquaculture production of live, fresh, chilled and frozen fish, as well as manufactured products, processed and preserved fish, crustaceans and molluscs (EC, 2020c). LCA has also been highlighted as an important tool to help companies pivot beyond COVID-19, including transitioning for uptake of new green deal innovations and services (Rowan and Galanakis, 2020).

In parallel, scientific LCA publication linked to the capture, farming and processing of fish started in mid-2000 where the LCA-related publications strongly increased from less than 20 cited in 2010 to more than 90 in 2019. Avadí and Fréon (2013) published one of the first reviews on LCA as applied to fisheries that was based on 16 studies issued in the first decade of 2000: these included the typical LCA phases: goal and scope definition, inventory analysis, impact assessment and interpretation. Other LCA reviews published by Vázguez-Rowe et al. (2012b) and Ziegler et al. (2016) also delved primarily into the fisheries sector and associated supply chain, whereas other groups focused on the aquaculture sector (Henriksson et al., 2012; Bohnes et al., 2019). More recently, Avadí et al. (2020) published another LCA review where the main objective was to present the first effort to aggregate and standardize seafoodrelated datasets in the Ecoinvent database. This LCA also explained the main data sources and commensurate methodological choices used in the building of the datasets. In this context, the current LCA review advances this topic by addressing the whole supply chain of the seafood sector with a particular focus on the fisheries sector. More specifically, the aim of the review is to assemble LCA-based studies published in the past decade, to analyse the evolution in research and innovation, and to describe the database and assessment models. Moreover, hotspots, challenges and opportunities arising from the current seafood sector are discussed from different perspectives including as climate change, economic market and environmental protection as well as the nexus between food, energy and water. No prior published literature review focused on this LCA-nexus integration has been reported. Overall, given the nature of bibliometric analysis performed, the expected audience of this review study are LCA practitioners. However, given the description of the utility and scientific challenges, this review will also appeal to broad range of stakeholders in the seafood sector, especially fish managers in the public and private sectors, and NGOs, which will help future proof the sector.

# 2. Materials and methods

The importance of the eco-perspective has been introduced in an increasing number of frameworks due to the historic push of the environmental movements and the multiple responses given by the rest of the actors. Occasionally, it is reflected in an eco-friendly commitment from policy-makers, companies and citizens. There are also instances of an eco-style approach to informing communities or green-washing campaigns from industries and big corporations. Nowadays, environmental awareness is a prominent cross-cutting key topic that features strongly in an increasing number of scientific publications along with permeating and influencing adjacent disciplines. In this context, this review addresses LCA studies applied to the fishing sector that focuses mainly on wild fish capture fisheries. It also commensurately addresses seafood processing and other stages of its life cycle based on their inclusion as topics appearing in system boundaries in publihed literature reviewed.

The bibliometric analysis was conducted using Scopus, which is the largest abstract and citation database of peer-reviewed literature; it addressed over 20,000 journals (Geng et al., 2017). This approach provides several options to make the search more accurate and reliable including use of keywords. Thus, three kinds of searches were applied using Scopus as a first step to appraise the breadth of LCA application in marine products and processes during the period 2010 to 2019: 1) using



Fig. 1. Number of publications per year in scientific journals that include the search terms "life cycle assessment" (grey circles), plus "food" (white circles) or plus "fish" and "seafood" (black circles) accessed in Scopus in June 2020.

the search term "life cycle assessment" and combining this with 2) "food" and 3) "fish" and "seafood". A total of 537 files were found in the literature search for the period analysed. Fig. 1 highlights an increasing use of LCA studies over the last decade where it appeared in almost 100 publications in both 2018 and 2019. Notwithstanding same, the seafood sector still represents a very low percentage of LCApublished studies that focused on food. Although the search included all kinds of publications, i.e., articles, reviews, conference papers, books, reports, etc., most of them (67-70%) were articles and about 76-83% involved articles or reviews. Thus, a refinement in the literature search was made combining "life cycle assessment" and "fish" or "seafood" terms only on titles, abstracts and keywords. The scope was reduced to 243 documents (198 English full-length articles). Finally, other papers were excluded because the subject area was out of scope -health, mathematics or engineering- (73) or addressed LCA related to diets (15), packaging and food waste (7), only aquaculture species (7), or did not include a case study.

In brief, 69 LCA-related publications have been extensively analysed for this review. Fig. 2 shows the 20 scientific journals where these articles were published. Two of these, the International Journal of Life Cycle Assessment and the Journal of Cleaner Production, represented 43% of the LCA publications that were included in this work. From all the publications assessed, 59 were case studies that included LCA impact categories along with occasional fisheries management indicator work. The other 10 studies encompassed two studies of Avadí and coauthors; namely, a general disposition to LCA (Avadí and Fréon, 2013), and a partial life cycle inventory review (Avadí et al., 2020). Other publications related to the best available techniques (BATs) in the fishing sector (Barros et al., 2009; Bello Bugallo et al., 2013), ecolabeling and certification (Thrane et al., 2009b; Vázquez-Rowe et al., 2016), best practices in LCA implementation and guidelines for managers and policy-makers (Vázquez-Rowe et al., 2012b; Vázquez-Rowe and Benetto, 2014), or regional context of fisheries in Peru (Fréon et al., 2014a) and seafood processing in Denmark (Thrane et al., 2009a). Among the 59 case studies, almost 90% were published after 2010 with only 7 appearing before that year; thus, making this present review representative of the previous decade.

This review does not include aquaculture despite its products representing approximately 15% of the worldwide fish production between 1986 and 1995: where the present day level of production is approximately 46% (FAO, 2020). Advances in science and technology have mirrored economic and labour growth in terms of rate of growth. A number of studies have comprehensively reviewed state-of-the-art in LCA for enhancing aquaculture over the past two years (Bohnes et al., 2019; Bohnes and Laurent, 2019; Philis et al., 2019). However, aquaculture was not a major focus of the current study, as mentioned in Section 1, although this particular sector is partially addressed in the discussion. Studies linked to the production of mussels in mussel rafts, along the coast of Galicia (Spain), were included within the scope of the current review given the fact that an auxiliary fishing fleet is needed to attend the production of this oceanic infrastructure.

The review follows the structure of the ISO standards 14,040 and 14,044 on LCA (ISO, 2006a, 2006b), also suggested by Avadí and Fréon (2013), yet advances this area much further by way of addressing new concepts linked to LCA in cross-cutting fields of study along with embracing current challenges facing the fishing sector. Firstly, a discussion focusing on the main LCA principles and requirements is addressed through use of several tables linked to text in which case studies were desegregated for ease of comparison: these included goal and scope, boundaries of the system, functional unit and life cycle impact assessment methods, impact categories and so forth. Thereafter, this



Fig. 2. Distribution of the 69 publications analysed according to the journal of publication.

constitutes the first study that comprehensively reviews use of LCA research for addressing the development of the fishing sector from a global perspective. Finally, the relationship or nexus between energy, water and food is discussed in addition to articulating previous and ongoing implications that hinder technical and environmental progress across the sector, such as marine debris, ghost shipping and climate change.

# 3. Results and discussion

#### 3.1. Goal and scope definition

The goal definition is framed on addressing key questions, such as, "why do we perform an LCA?", "who are the target audiences?" and "what is the product under study?" However, the scope definition is more complicated as it includes the definition of the system boundaries, the functional unit, the allocation strategy and other relevant hypotheses and assumptions (ISO, 2006a. 2006b). Table 1 presents some of the main characteristics of the reviewed LCA studies: namely, targeted species, functional unit, system boundaries, allocation method and sensitivity analysis.

## 3.1.1. Functional unit definition

The functional unit (FU) is the element that quantifies the function of the system, the calculation basis for which all inputs and outputs of the system must refer. The selection of a coherent FU is a fundamental step in order to perform a robust and comparable LCA. Choosing an appropriate FU allows direct comparison between alternative scenarios that perform the same function. However, FUs vary greatly, which highlight the difficulties of comparing results reported across many different articles. In this study, the choice of different FUs in the reviewed studies was assessed in detail. Around 30% of analysed papers focused their research on the extraction phase and use the amount of product landed in a port as the FU, typically 1 kg or 1 t. For instance, Abdou et al. (2018) defined their FU as 1 t of landed seafood (shrimp, demersal finfish, mullets, rays and sharks) by demersal trawlers in the Gulf of Gabes. Avadí et al. (2015) used 1 ton of landed fish by the Peruvian anchoveta purse seining fleet in the period 2005-2010 as the study FU. This FU was also used by Lourguioui et al. (2017) to assess mussel cultivation, Avadí et al. (2018) to address Peruvian hake capture, Fréon et al. (2014b) to analyse the Peruvian anchoveta caught in the north-centre fishing zone of Peru and Gonzalez-García et al. (2015) and Villanueva-Rey et al. (2018) to assess European pilchard capture. However, in some cases, such as Driscoll et al. (2015), this perspective includes other life cycle stages in the FU, such as transportation to the processing plant.

An alternative FU was used when studies focused on the processing stage. For example, Barros et al. (2009) analysed the operation of a mussel processing plant for one year, while Bello Bugallo et al. (2013) sought to characterise the operation of different seafood processing plants by the degree of implementation of BATs and an output-based FU was used. A similar case was analysed in Denham et al. (2016), where 1 t of processed fish sold at retail was used as the FU. Mass of packaged product ready for dispatch was a FU used by several authors. Findings of Ziegler et al. (2011), which was replicated in 2012 and 2016 but not included in the tables, defined the FU as 1 kg of shrimp and the accompanying packaging material ready to import to Europe. Almeida et al. (2015) used 1 kg of edible product (i.e., canned sardine with olive oil) as FU, whereas Avadí et al. (2014b) used 1 kg of fish (Peruvian anchoveta) in the final product. Other authors defined the FU as the commercial or serving fish product (Ziegler et al., 2003; Ziegler and Valentinsson, 2008). For example, Iribarren et al. (2010d) used one triple pack of round cans of canned mussels composed by 129 g of canned mussel flesh, 120 g of sauce, 81 g of tinplate can and 12.73 g of cardboard as the FU. Similarly, Laso et al. (2017a) selected one "octavillo" (i.e., a special can with the right amount of product, usually served as individual ration) of canned anchovy as the FU, which was composed by 30 g of canned anchovy, 20 g of extra virgin oil, one aluminium can and cardboard. Parker and Tyedmers (2012a) defined three FUs in accordance with the three Antarctic krill-derived products studied: 1 kg of krill meal, 1 L of krill oil and 1 consumer-ready bottle of 60 omega-3 krill oil capsules. On the other hand, Svanes et al. (2011) defined a specific FU for each of the four-derived cod products under study wetpack, burger, loin and processing residues to animal feed. Vázquez-Rowe et al. (2011a, 2013a) analysed two types of hake: production of 500 g of raw gutted fresh hake fillet reaching the household including packaging and 1 package of frozen fish sticks containing 10 fish sticks, which corresponds to approximately 320 g of edible product. Based on a nutritional point of view, some authors defined the FU as the amount of protein supplied to consumer. For instance, Vázquez-Rowe et al. (2014b) used the amount of protein (17.26 g) supplied by one can of sardines (85 g) in olive oil produced by a Galician canning industry as the FU, whereas Fréon et al. (2010) defined the FU as 100 g of animal protein of anchoveta or derivative product on the consumer plate.

However, another trend has been to analyse a feedstock-based FU. Hospido et al. (2006) analysed the production of canned tuna and the FU evaluated was 1 t of raw tuna entering the factory. Similarly, Laso et al. (2017b) assessed the production of canned anchovy and the selected FU was also 1 kg of raw anchovy entering the factory. Both options present different advantages that are worth emphasizing. Choosing an input-based FU allows for comparison of the environmental performance of different processes, which simultaneously determines the strengths and weaknesses of these processes. In contrast, selecting an output-based FU allows for detailed product analysis, including its use as a key step in eco-labelling. Fréon et al. (2017) evaluated two scenarios: firstly, the production of fishmeal and fish-oil as by-products of fishing for which an output-based FU (one tonne of byproduct produced) was used. However, Laso et al. (2016) evaluated the valorisation of anchovy residues by means of the production of fishmeal and fish oil defining an input-based FU (1 t of anchovy residues). Similarly, Iribarren et al. (2010b) addressed the valorisation of mussel shells and mussel organic remains using an input-based FU (100 t of each residue). Alternatively and secondly, Fréon et al. (2017) aimed at analysing the intrinsic characteristics of the processing process, for which a feedstock-based FU was chosen (1 ton of raw material). Even though they do not follow a conventional LCA approach, the studies by Hallstrom et al. (2019) and Hélias et al. (2018) are also noteworthy. Hallstrom et al. (2019) sought to establish a nexus between the carbon footprint and the nutritional impact of Swedish seafood consumption, for which the author makes a relative comparison between the results and their variation from the median value. Hélias et al. (2018) used fishery operations to develop characterization factors that allow determining biotic resource depletion.

#### 3.1.2. System boundaries

The reasons that influence an author to decide which processes should be included within the boundaries of the system must be duly justified, including what are the clearly defined criteria that govern this decision, and justify same. Studies included in this LCA review show some variability in the definition of the system boundaries. Thus, the scope of the analysis will depend on the approach of the system (Fig. 3): "cradle to grave", involving all stages of the life cycle; "cradle to gate" or "gate to grave", including limits from the beginning of the cycle to a specific "gate" (e.g., from capture to landing, from capture to get the product, etc.) or from a midpoint to the end of life stage, respectively; "gate to gate", for intermediate stages; or specific parts of the life cycle, such as the end of life, product components, etc. (ISO, 2006a, 2006b). Regarding fishery-related LCA studies, system boundaries usually include at least the use and maintenance of the vessel (Almeida et al., 2014; Driscoll and Tyedmers, 2010), while in other cases construction (Abdou et al., 2018; Avadí et al., 2014b; Avadí et al., 2018; Driscoll et al., 2015; Gonzalez-García et al., 2015; Hospido and Tyedmers,

## Table 1

Main characteristics of the reviewed LCA studies: Functional Unit, System Boundaries, Allocations and Sensitivity analysis.

Reference	Targeted species	Functional unit	System boundaries	Allocation method	Sensitivity analysis
Abdou et al.,	Shrimp and demersal finfish (Sparidae Diplodus annularis, Sparus aurata), mullete (Mullus besketus M	1 t of landed seafood	Cradle to gate	No	No
2018	munets (Multus barbatus, M. surmuletus), rays (e.g. Raja clavata) and sharks (e.g. Mustelus mustelus).		-		Voc. management constinct
Abdou et al., 2020	Shrimp and demersal finfish (SparidaeDiplodus annularis, Sparus aurata), mullets (Mullus barbatus, M. surmuletus), rays (e.g. Raja clavata) and sharks (e.g. Mustelus mustelus).	1 t of landed seafood	Cradle to gate	No	establishment of marine protected areas, extension of the biological rest period, and decrease in the number of demersal trawlers.
Almeida et al., 2014	European pilchard ( <i>Sardina</i> pilchardus)	1 kg of landed sardine,	Cradle to gate	Mass	Yes, gear and time lapse comparison.
Almeida et al., 2015	European pilchard ( <i>Sardina</i> pilchardus)	1 kg edible of canned sardine with olive oil	Cradle to gate	Mass	Yes, packaging analysis and comparison with other seafood products.
Avadí et al., 2014a	Peruvian anchoveta (Engraulis ringens)	1 kg of fish in the final product	Cradle to gate	Mass	Yes, electricity reduction, packaging material and reduction in-plant discards.
Avadí et al., 2014b	Peruvian anchoveta (Engraulis ringens)	1 t of landed fish	Cradle to gate	No	Analysis (DEA) to determine the relative efficiencies of multiple comparable units.
Avadí et al., 2015	Ecuadorian tuna yellowfin (Thunnus albacares), skipjack (Katsuwonus pelamis) and bigeye (Thunnus obesus)	1 t of tuna product	Cradle to gate	Mass	Yes, fuel use intensity and packaging material.
Avadí and Fréon, 2015	Peruvian anchoveta (Engraulis ringens), trout (Oncorhynchus mykiss), tilapia (Oreochromis spp.) and black pacu (Colossoma macropomum)	1 t of edible fish in a Direct Human Consumption product in the case of anchoveta and fresh fish edible portion for cultured species.	Cradle to gate	Mass	No
Avadí et al., 2018	Peruvian hake (Merluccius gayi peruanus)	1 t of whole hake landed	Cradle to gate	Mass	Yes, fuel use intensity and comparison with published results from other hake fisheries and with another Merlucciidae fish, the Patagonian grenadier ( <i>Macruronus magellanicus</i> ).
Avadí et al., 2020	South Pacific anchovies and hake (including Patagonian grenadier- Macruronus magellanicus) and Pacific tunas (Thunnus spp.), tilapia (Oreochromis spp.) and trout (Oncorhynchus mykiss)	N/A	Cradle to grave	Mass, economic	No
Denham et al., 2016	Different species of finfish: Crimson snapper ( <i>Lutjanus erythropterus</i> ), Bluespotted emperor ( <i>Lethrinus</i> <i>punctulatus</i> ) and Rosy threadfin bream ( <i>Nemipterus furcosus</i> ) among others.	1 t of processed fish sold at retail	Gate to gate	Mass	Yes, cleaner production strategies: solar electricity, biogas electricity, reduction of GHG emissions from refrigeration, and utilizing waste to develop by-products.
Driscoll and Tyedmers, 2010	Atlantic herring (Clupea harengus)	1 t of fish landed	Cradle to gate	N/A	Yes, variations of Total Allowable Catch and purse seine fishing effort. Yes, comparison of different
Driscoll et al., 2015	American lobster ( <i>Homarus</i> americanus)	1 t of live lobster	Cradle to gate	Mass	scenarios: no allocation between the main product and co-products, electricity use for storage, fuel use in vessel, different database for fuel combustion and post-capture
Farmery et al., 2015	White banana prawn (Fenneropenaeus merguiensis)	1 kg of frozen prawn	Cradle to gate	Mass	Yes, comparison of impact method (IPCC 100 years, CML 2 Baseline 2000 and ReCiPe). Also catch variation: 10% increase and decrease in catch with the same number of boat days.
Fréon et al., 2010	Peruvian anchoveta ( <i>Engraulis</i> ringens)	100 g of protein	Cradle to gate	No	No
Fréon et al., 2014b	Peruvian anchoveta (Engraulis ringens)	1 t of fresh fish	Cradle to gate	No	No
Fréon et al.,	Peruvian anchoveta (Engraulis	1 t of fresh fish	Cradle to gate	Mass and economic	Yes, simulations of fuel use

Reference	Targeted species	Functional unit	System boundaries	Allocation method	Sensitivity analysis
2014c	ringens)				variations of $\pm 20\%$ and recomputing single scores considering mass allocation.
Fréon et al., 2017	Peruvian anchoveta (Engraulis ringens)	(i) 1 t of fish oil or fishmeal at the gate of the plant and (ii) 1 t of raw material at the plant	(i) Cradle to gate and (ii) Gate to gate	Gross energy content, economic value and mass	Yes, cleaner production strategies: using natural gas instead of heavy fuel.
Gonzalez-García et al., 2015	European pilchard (Sardina pilchardus) Alaskan pollock (Theragra chalcogramma), Arctic char (Salvelinus alpinus), Cod (Gadus morhua), Atlantic halibut (Hippoglossus hippoglossus), Atlantic herring (Clupea harengus), Atlantic mackerel (Scomber scombrus), Atlantic salmon (Salmo salar), Cape hake (Merluccius capensis), Cephalopods (Cephalopoda spp.), European eel (Anguilla anguilla), European flounder (Platichtys flesus), Hake (Merluccius merluccius), Seabass (Dicentrarchus labrax), Sprat (Sprattus	1 t of landed pilchard	Cradle to gate	No	No
Hallstrom et al., 2019	sprattus), Gilt-head seabream (Sparus aurata), Haddock (Melanogrammus aeglefinus), Hoki (Macruronus novaezelandiae), Lobster (Homarus gammarus), Northern prawn (Pandalus borealis), Norway lobster (Nephrops norvegicus), Oyster (Ostreidae spp.), Pangasius (Pangasius hypophthalmus), Perch (Perca fluviatilis), Pike (Esox lucius), Pike perch (Sander lucioperca), Pink salmon (Oncorhynchus gorbuscha), Plaice (Pleuronectes platessa), Rain- bow trout (Oncorhynchus mykiss), Saithe (Pollachius virens), Scallop (Pecten maximus), Tilapia (Oreochromis niloticus), Trout (Salmo trutta), Turbot (Scophthalmus maxima), Whitefish (Coregonus spp.), Whiting (Merlangius merlangus) Black-bellied anglerfish (Lophius budegassa), White anglerfish (Lophius budegassa), White anglerfish (Lophius budegassa), White anglerfish, Jaddock (Melanogrammus aeglefinus), Hake (Merluccius merluccius), Greenland halibut (Reinhardtius hippoglossoides), Herring (Clupea harengus), Ling (Molva molva), Blue ling (Molva dypterygia), Mackerel	N/A. Results of CHG emissions and nutritional score are presented as a variation of the median of the entire analysed sample.	Cradle to gate	No	Yes, variation in nutritional results when using different methods.
Hélias et al., 2018	(Scomber scombrus), Horse mackerel (Trachurus trachurus), Megrim (Lepidorhombus whiffiagonis), Four-spot megrim (Lepidorhombus boscii), Plaice (Pleuronectes platessa), Beaked redfish (Sebastes mentella), Golden redfish (Sebastes norvegicus), Saithe (Pollachius virens), Sandeel (Ammodytes spp.), Seabass (Dicentrarchus labrax), Sole (Solea solea), Sprat (Sprattus sprattus), Tusk (Brosme brosme), Whiting (Merlangius merlangus) and Blue whiting (Micromesistius poutassou)	N/A	N/A	No	No
Hospido and Tyedmers, 2005	Skipjack tuna (Katsuwonus pelamis) and Yellowfin tuna ( <i>Thunnus</i> <i>albacares</i> )	1 t of frozen fish landed	Cradle to gate	No, they consider various target species within their global FU	Yes, increase and decrease fuel inputs by one standard deviation and the use of alternative emission factors from different courses
Hospido et al., 2006	Tuna (Thunnus albacares)	1 t of frozen fish entering the factory	Gate to grave	Economic (for transport from retailers to	Yes, improvement actions: recycled percentage of

(continued on next page)

Reference	Targeted species	Functional unit	System boundaries	Allocation method	Sensitivity analysis
				households)	packaging materials, substitution of packaging
lribarren et al., 2010a <sup>a</sup>	Mussel (Mytilus galloprovincialis)	(i) 1 kg of fresh mussels and (ii) 1 kg of canned mussels' flesh	Purification/transformation and consumption stages. Excluded mussel culture and valorization of mussel	Mass	No
Iribarren et al., 2010b <sup>a</sup>	Mussel (Mytilus galloprovincialis):	(i) 100 t of mussel shells and (ii) 100 t of mussel organic remains	Grave to grave: valorization of shells to calcium carbonate and organic waste to fish meal	System expansion for waste valorization	Yes, differences between current methodology of valorization (producing calcium carbonate) and others (landfilling, incineration) are considered. Also, similar differences between current organic waste valorization to fish meal and alternative production of mussel pate are analysed.
Iribarren et al., 2010c <sup>a</sup>	Mussel (Mytilus galloprovincialis)	100 kg of cultured mussel: 40 kg for fresh, 35 canning, 20 frozen in cooking-freezing plants and 5 from cooking plants for cannery. For comparative effects 1 kg of fresh, canned or frozen mussel	Cradle to grave named business-to-consumer (B2C)	System expansion for waste valorization (same procedure that Iribarren et al., 2010b)	Yes, regarding analysis based on 1 kg of protein supplied comparing mussels and chicken and canned tuna.
lribarren et al., 2010d <sup>a</sup>	Mussel (Mytilus galloprovincialis)	One triple pack or rounds cans format (129 g of canned mussels, 120 g of sauce, 81 g of primary packaging and 12.73 g of secondary packaging)	Cradle to grave (B2C)	System expansion for waste valorization (same procedure that Iribarren et al., 2010b)	No
Iribarren et al., 2010e	Species from coastal fishing (horse mackerel, Atlantic mackerel, European pilchard and blue whiting), offshore fishing (european hake, megrim and anglerfish), deep-sea fishing (skipjack and yellowfin tuna), extensive aquaculture (mussels) and intensive aquaculture (turbot)	1 t of fish	Cradle to gate	Economic and mass (sensitivity analysis)	Capital goods are relevant in carbon footprint results for extensive aquaculture species but not for the others.
lribarren et al., 2011	Species from coastal fishing, offshore fishing, deep-sea fishing, extensive aquaculture and intensive aquaculture (same species that Iribarren et al., 2010e)	1 t of fish	Cradle to gate	Economic and mass	Yes, based on the type of cooling agents.
Laso et al., 2016	Anchovy (Engraulis encrasicolus)	For heads and spines: 1 t of fish meat entering the plant. For the remaining and broken fish: 1 t paste processing	Gate to grave.	Economic and system expansion for anchovy waste valorisation	No
Laso et al., 2017a	Anchovy (Engraulis encrasicolus)	1 can of fish in extra virgin olive oil.	Cradle to gate (from fish to factory), gate to gate (factory process and canned products) and gate to grave (distribution and use and EoL)	System expansion for anchovy waste valorisation	Yes, different scenarios: packaging recycling is proposed -Application of BATs for canned anchovy industry such as recycle process water, recycle cardboard boxes, separate possible valorization streams, dry cleaning
Laso et al., 2017b	Anchovy (Engraulis encrasicolus)	1 kg of fish entering the	Cradle to grave	System expansion,	Yes, sensitive analysis based on mass or economic allocation
Laso et al., 2018a	Anchovy (Engraulis encrasicolus)	1 kg of processed fish	Cradle to grave	System expansion	Yes, Green protein footprint according the packaging type or no packaging
Laso et al.,	Anchovy (Engraulis encrasicolus)	1 kg of fish	Cradle to gate	System expansion	No
Laso et al., 2018c	Anchovy (Engraulis encrasicolus)	1 t fish food loss (FL)	2 scenarios: Food waste-to-energy-to-food"	System expansion	No
Lourguioui et al., 2017	Mussel (Mytilus galloprovincialis)	1 t of mussels	and "Food-waste-to-tood". Cradle to gate	No	Scenarios for mussel farms management and uncertainty
Lozano et al.,	Mussel	1 t of mussels for each	Cradle to gate	No	No

Reference	Targeted species	Functional unit	System boundaries	Allocation method	Sensitivity analysis
2010 Parker and Tyedmers, 2012a	Antarctic krill (Euphausia superba)	raft 1 kg of krill meal 1 L of krill oil 1 consumer-ready bottle of 60 omega-3 krill oil capsules	(i) Krill meal and oil: cradle to consumer (ii) Krill omega-3 capsules: cradle to retailer	<ul> <li>(i) Krill meal and oil:</li> <li>energy content in</li> <li>fishing and primary</li> <li>processing</li> <li>(ii) Krill meal and oil:</li> <li>mass in transport to</li> <li>port</li> </ul>	Application of three allocation scenarios to omega-3 capsules. Scenario analyses of different parameters for krill meal and omega-3 capsules.
Parker and Tyedmers, 2012b	Peruvian anchovy (Engraulis ringens), Atlantic herring (Clupea harengus), Gulf menhaden (Brevoortia patronus), blue whiting (Micromesistius poutassou) and Antarctic krill (Euphausia superba)	For each species: 100 GJ of combined meal and oil products, respecting the species-specific yields of meal and oil	N/A	(iii) Krill omega-3 capsules: system expansion in the secondary processing Output nutritional energy of meal and oil products	Uncertainty analysis/Monte Carlo Sensitivity analysis to the FU (basis of comparison between species):
Parker et al., 2015	4 tuna species: skipjack (Katsuwonus pelamis), yellowfin (Thunnus albacores), albacore (Thunnus	1 t of landed fish	Cradle to gate	Mass	<ul> <li>100 GJ of energy from meal and oil (baseline);</li> <li>1 t of protein from meal and oil</li> <li>1 t wet weight biomass No</li> </ul>
Ramos et al.,	alalunga), bigeye (Thunnus obesus) North East Atlantic Mackerel (NEAM)	1 t of landed round fish	Cradle to gate	Temporal allocation	No
2011 Svanes et al., 2011	(Scomber scombrus) Cod (Gadus morhua)	1 kg cod wetpack, frozen, in 400 g packages, delivered to retailer in Sweden 1 kg cod burger, frozen, in 5 kg packages, delivered to institutional buyer in Sweden 1 kg processed cod loin product in 2 kg package, delivered to distribution centre in the UK 1 kg processing residue, frozen, going to animal feed production in Norwav	Cod wetpack and cod burger: cradle to consumer Processed cod loin: cradle to distribution Cradle to gate (arrival at the processing plant)	Mass and economic	Sensitivity analysis based on either mass and economic allocation. Scenario analyses on different parameters.
Van Putten et al., 2016	Tropical rock lobster (TRL, Panulirus ornatus) and southern rock lobster (SRL, Jasus edwardsii)	1 kg of lobster	Cradle to consumer	a) mass, assuming heads are wasted; b) mass, assuming heads are used; c) nutritional value (total MJ of edible product); d) economic (ex-uscel price)	Scenario analyses, using base case mass allocation on different parameters.
Thrane, 2004	Codfish, flatfish, prawn, shrimp, Norway lobster, mussels, herring,	1 kg of fish	Cradle to gate	System expansion and mass and economic	No
Thrane, 2006	mackerel, industrial fish Flatfish	1 kg of frozen fish filet	Cradle to cradle	allocation System expansion and mass and economic	No
Vázquez-Rowe	Atlantic horse mackerel ( <i>Trachurus</i>	1 t of round fish	Cradle to gate	allocation Mass and economic	Yes
et al., 2010a Vázquez-Rowe et al., 2010b	European hake (Merluccius merluccius), horse mackerel (Trachurus trachurus), Atlantic mack- erel (Scomber scombrus), blue whiting (Micromesistius patassau)	1 kg of fish	Cradle to gate	No, global catch value was considered as FU	No
Vázquez-Rowe et al., 2011a	European hake (Merluccius merluccius)	500 g of gutted fish fillet	Cradle to grave	Mass and economic	No
Vázquez-Rowe et al., 2011b	Broad number of vessels within selected Galician fishing fleets (target species are quite varied, depending on the gear type and geographical zone where they fish)	1 t of landed fish	Cradle to gate	Mass	No
Vázquez-Rowe et al., 2012a	Common octopus (Octopus vulgaris)	24 kg of frozen octopus up to the point of import	Cradle to gate	Mass	No
Vázquez-Rowe	Hake (Macruronus magellanicus) fish	1 package of 10 frozen	Cradle to gate	Mass	No

(continued on next page)

Reference	Targeted species	Functional unit	System boundaries	Allocation method	Sensitivity analysis
et al., 2013a	sticks produced in a processing plant in Spain	fish sticks			
Vázquez-Rowe et al., 2013b	Hake ( <i>Macruronus magellanicus</i> ) fish sticks produced in a processing plant in Spain	1 package of 10 frozen fish sticks	Gate to grave	Mass	No
Vázquez-Rowe et al., 2013c	Goose barnacle (Pollicipes pollicipes)	1 kg of barnacles	Cradle to gate	No, due to the lack of co-products	No
Vázquez-Rowe et al., 2014a	Seafood species landed in Galician ports	1 of fish	Cradle to gate	Mass	No
Vázquez-Rowe et al., 2014b	European pilchard (Sardina pilchardus)	Amount of protein (17.26 g) supplied by one can of sardines (85.0 g) in olive oil	Cradle to gate	Mass, economic and energy	No
Villanueva-Rey et al., 2018	European pilchard (Sardina pilchardus)	1 t of fish	Cradle to gate	Mass	Yes, different assessment methods, allocation approach (economic), fishing gear life span, base port influence, engine type.
Winther et al., 2009	Norwegian seafood supply chain: a) aquaculture: Atlantic salmon (Salmo salar) and Blue mussels (Mytilus edulis); b) fishing: cod (Gadus morhua), saithe (Pollachius virens), haddock (Melanogrammus aeglefinus), herring (Clupea harengus) mackerel (Scomber scombrus)	1 kg of edible product	Cradle to gate	Mass	Yes, different scenarios were analysed: electricity mix, product waste, edible yield, allocation approach (economic), utilization of processing stage by-products, feed conversion ratio, refrigerant agent, etc.
Ziegler et al., 2003	Cod (Gadus morhua)	400 g of fish fillets	Cradle to gate	Economic	Yes, different scenarios were considered for fishing based on fishing gear.
Ziegler and Valentinsson, 2008	Norway lobster (Nephrops norvegicus)	300 g of lobster tails	Cradle to gate	Economic	Yes, fuel use, allocation choice, product yield, impact assessment method and background data.
Ziegler et al., 2011	Southern pink shrimp (Penaeus notialis)	1 kg of shrimp	Cradle to gate	Economic	No

<sup>a</sup> The production of mussels (*Mytilus galloprovincialis*) in Galicia, Spain, corresponds to extensive aquaculture. However, an important auxiliary fishing fleet (1267 vessels according to the 2020 regional census) supports the cultivation of mussels in mussel rafts along the Galician rias (Pesca de Galicia, 2020).

2005; Ramos et al., 2011; Vázquez-Rowe et al., 2010a; Vázquez-Rowe et al., 2010b; Vázquez-Rowe et al., 2011b) and end of life stages (Fréon et al., 2014b; Laso et al., 2018b) are also included. Only two studies assessed the consumption of fuel within the fishing activity (Parker et al., 2015; Thrane, 2004), distinguishing between fuel used during propulsion to reach fishing grounds and fuel used during the actual extraction of fish. Alternatively, Lozano et al. (2010) considered mussel cultivation farming, including the construction, operation and maintenance of mussel rafts. However, these authors also addressed the environmental impact of the auxiliary boats used by the sector to reach the rafts, which are usually located several miles away from the coast. Beyond vessel and farming-related activities, the systems commonly end at the harbour, when fish is landed. Within the sample analysed, a small number of studies included fish transport to the processing plant -e.g. Farmery et al. (2015), while others included transport to retail (Driscoll et al., 2015).

With regard to studies focusing on processing, the most recurrent feature is that the system boundaries include activities carried out directly in the processing plant, in addition to ancillary operations such as power or steam production, excluding the fishing or farming stage (Iribarren et al., 2010a). On the other hand, some authors considered the fishing stage together with processing (e.g. Almeida et al., 2015; Avadí et al., 2018). Fréon et al. (2017) also included the fishing stage within the boundaries of fishmeal and fish-oil production, which allowed performing an exhaustive study of the environmental impacts related to the production of these co-products. Svanes et al. (2011) and Vázquez-Rowe et al. (2012a) added one additional step along with including the distribution to retailer of cod and octopus' products, respectively. Winther et al. (2009) and Ziegler et al. (2012) also included the transportation of Norwegian seafood products to the whole-saler. Other authors advanced thee activities further; for instance,

Hospido et al. (2006) also analysed transportation stages to wholesale and consumption at the households that followed a gate-to-grave approach. Similarly, Fréon et al. (2010) and Laso et al. (2017a) assessed the whole life cycle of Peruvian and European anchovy, respectively, considering anchovy capture, production, transport, use and disposal. Vázquez-Rowe et al. (2011a) analysed the life cycle of fresh hake including the household consumption. Denham et al. (2016) included the transportation of all consumable items to city and regional retailer and waste disposal to landfill. In summary, the majority of studies were cradle-to-grave analysis, with some studies adopting either gate-togate or gate-to-cradle type approaches.

## 3.1.3. Allocations

Multifunctional processes are those economic systems that i) produce more than one valuable output (multi-output) (Huijungs and Guinée, 2007), ii) have more than one input (multi-input), such as waste treatment processes with a mixture of input waste flows (Iribarren et al., 2010c); or iii) transform a product into another product (open-loop recycling). In all these systems, the environmental burdens associated with a particular process must be partitioned over the various functional flows of that process (Margallo et al., 2014).

To handle this multifunctional problem, the ISO standard (ISO, 2006a, 2006b) establishes a preference order that consists on dividing processes into sub-processes, or expanding system boundaries to include the additional functions. If that is not possible, then the allocation problem must be solved by using physical causation or other relationships, including the economic value, mass or energy content of the functional outputs (Azapagic and Clift, 1999). In other circumstances, allocations are not needed as authors focus on the contribution of each production stage or fishing gear to the environmental impacts,


- - "gate to gate": intermediate stage of the life cycle - - - "gate to grave": from a midpoint to the end of the life cycle

Fig. 3. System boundaries applied in LCA for the seafood sector.

instead of the impact associated to the different species landed (Abdou et al., 2018; Lozano et al., 2010).

Most of the LCA studies analysed in this review can be classified as multi-output process. The most common need for allocation arises when fishing fleets land by-catch (organisms inadvertently captured while fishing for more valuable or legally permitted species) or target multiple species. In these cases, the use of system expansion is not usually implemented due to the lack of fisheries where only the by-catch species are caught (Ayer et al., 2007). Exceptionally, Thrane (2004, 2006) applied a combination of technical subdivision along with system expansion, mass and economic allocation to conduct an energy analysis and an LCA, respectively. In mass and economic allocation, the fuel consumption and impacts were distributed to each species based on the proportion to their weight and the catch value. Ideally, the system expansion should consider that a by-catch of a particular species affects the fishing vessels targeting that species, as their quotas are reduced proportionally to meet the overall quota -in those countries that fix it-(Thrane, 2004). Hence, the by-catch substitutes catch (or quota) in other Danish fisheries, which targets these species (Thrane, 2006).

Nonetheless, in most studies, multi-output fishery systems are solved using mass, energy or economic allocations. Despite that, some authors define mass allocation as arbitrary and unjustified; however, this procedure has been widely applied (e.g. Avadí et al., 2015; Avadí and Fréon, 2015; Avadí et al., 2018; Driscoll et al., 2015; Parker et al., 2015; Vázquez-Rowe et al., 2014b; Vázquez-Rowe et al., 2016; Villanueva-Rey et al., 2018; Winther et al., 2009) because there is no other way to particularize inputs to specific species during fishing operations. On the other hand, economic allocation has generally been defended as a reasonable and more socially relevant approach (Aver et al., 2007). However, several authors point out that the main problem of this approach is the highly volatile economic price of the product, which depends on the season, freshness of the product and many other market factors (Vázquez-Rowe et al., 2013a; Vázquez-Rowe et al., 2011a), making it difficult to establish a stable allocation over time (Winther et al., 2009). Pelletier and Tyedmers (2011) warn about the difficulties of applying economic criteria to partition natural systems.

Contrasting several allocation methods provides more robust and precise LCA results Avadí and Fréon, 2013). This type of approach, using economic and mass allocation was conducted in several studies (Fréon et al., 2017; Vázquez-Rowe et al., 2011a). Vázquez-Rowe et al. (2011a) compared the use of mass and economic allocation for fresh hake fillet captured by the Galician, founding similar results since the average sale price for European hake does not entail major differences. However, the use of economic allocation was shown to be preferred to mass allocation in mixed fisheries where the landed species have great differences in economic value (Ayer et al., 2007). Ramos et al. (2011) proposed a temporal allocation for the evaluation of fishing of North East Atlantic Mackerel (NEAM). This procedure was applied to construction and maintenance materials, as the Basque coastal purseseining fishing fleet presents three distinct fishing seasons. Fishing activity is focused on NEAM and on the anchovy during the first half of the year, while the albacore fishing season takes place in the second half of the year.

In the case of fish processing, the main allocation problem is obtaining data of secondary co-products. In these studies, the use of system expansion is more common, since it is easier to find an alternative by-product. Additionally, the substitution by another fishery or nonfishery protein source is always possible (Avadí and Fréon, 2013). The valorisation of shells and mussel meat by-product was allocated by substitution to alternative products. Mussel shells can be used as raw material for calcium carbonate production and it was assumed that 100 t of shells avoids the conventional production of 65 t of calcium carbonate (Iribarren et al., 2010b). Similarly, the valorisation of organic waste from anchovy canning plants results in anchovy paste (from remaining anchovies) and fishmeal and fish oil (from head and spines). The impact of these co-products was allocated using a system expansion based on substitution for alternative products (Laso et al., 2016; Laso et al., 2017a; Laso et al., 2017b; Laso et al., 2018a; Laso et al., 2018b; Laso et al., 2018c). Particularly, these authors assumed that 1 t of anchovy paste substituted the production of 1 t of tuna pâté given its similar protein content, whereas 1 t of heads and spines replaced the production of 212 kg of fishmeal and 108 kg of fish oil from fresh anchovy (Laso et al., 2016). This assumption was compared with mass and economic allocation (Laso et al., 2017b). Similarly to fisheries, if there is a lack of alternative production systems for the by-product under analysis, causality or non-causality allocation were applied (Vázquez-Rowe et al., 2013a; Winther et al., 2009).

Mass allocation was used for processed finfish (Denham et al., 2016), tuna (Avadí et al., 2014a), frozen prawn (Farmery et al., 2015), frozen common octopus (Vázquez-Rowe et al., 2012a), and frozen hake sticks (Vázquez-Rowe et al., 2013a; Vázquez-Rowe et al., 2013b). The use of mass or economic allocation was analysed for the Galician fishing activity (Iribarren et al., 2010e; Iribarren et al., 2011) providing changes in the carbon footprint from 0.3% to 57%, denoting the importance of the allocation method.

Svanes et al. (2011) evaluated the environmental performance of cod caught by the autoline fleet in Norwegian territorial waters and its processing in four cod-derived products: wetpack, burger, loin and processing residues to animal feed. Using mass allocation, the main differences were found in the choice of whether the head was considered a co-product or a waste, while low variations were observed with economic allocation.

Economic allocation was the preferred option for packaged cod fillets (Ziegler et al., 2003) as they were dominating both in quantity and in gross sales and for Norway lobster (Ziegler and Valentinsson, 2008) and Southern pink shrimp (Ziegler et al., 2011) due to its high economic value.

Allocation based on the energy content has been traditionally downplayed as an arbitrary method that does not generally reflect the relationship between the inputs and outputs of a studied system (Ayer et al., 2007). This is common for other non-causality allocations, such as mass or economic allocation. To analyse the impact of Peruvian fishmeal or fish oil, Fréon et al. (2017) applied gross energy content, as well as economic value and mass. In addition, Parker and Tyedmers (2012a) applied allocations based on energy content, mass allocation and system expansion. This study defined three functional units: 1 kg of krill meal, 1 L of krill oil and 1 consumer-ready bottle of 60 omega-3 krill oil capsules. The authors used energy allocation for fishing and primary processing, mass allocation for transport of meal to port, and system expansion to allocate between omega-3 capsules and lower grade meal. A sensitivity analysis evaluated three allocation scenarios to omega-3 capsules: i) energy allocation for fishing and primary processing; ii) mass allocation for transport of meal to port, and iii) system expansion to allocate between omega-3 capsules and lower grade meal.

LCA practitioners may avoid allocation, if the environmental impacts are assigned to the fishing gear or the production step, instead of the species. Nevertheless, when the species and by-products are considered, it is first recommended to expand the boundaries of the system prior to allocation and, if possible, use both mass and economic allocation for the same case study in order to compare approaches. Generally, mass allocation gives initial accuracy on data (widely known) if the energy content of all species is within the same range. If the variability in nutrient content of the species landed is high or will be transformed into sub-products (e.g., fish oil), energy allocation is suggested (Avadí and Fréon, 2013) together with the use of economic variables. However, as mentioned above, sometimes this information is not available and is strongly influenced by market fluctuations.

#### 3.1.4. Sensitivity analysis

When sensitivity analysis was conducted for the studies reviewed mainly discussed the fuel use of vessels (Avadí et al., 2015, 2018; Fréon et al., 2014c; Hospido and Tyedmers, 2005), packaging and cleaner production strategies. Although the manufacture of primary packaging for marine products is not the main contributor to the total environmental impact along it's life cycle (Molina-Besch, 2016), several authors evaluated the kind of material and the recycling percentage of the packaging material used (Almeida et al., 2015; Avadí et al., 2014a, 2015; Hospido et al., 2006).

Linked to cleaner production strategies, Laso et al. (2017a) proposed some BATs for canned anchovy industry, such as reuse of process water, recycling of cardboard boxes, separate possible valorisation streams or dry cleaning. Other authors also applied sensitivity analysis for BATs. For instance, Avadí et al. (2014a) and Driscoll et al. (2015) evaluated the use of electricity and, concretely, Denham et al. (2016), focused on solar electricity and biogas electricity, and other scenarios such as the reduction of greenhouse gas (GHG) emissions from refrigeration and utilizing waste to develop by-products. Meanwhile, Iribarren et al. (2010b) considered differences between current methodology of mussel valorisation (producing calcium carbonate) and others (landfilling, incineration) and concluded that the contribution of environmental impacts coming from valorisation processes are lower compared to mussel culture, fresh mussel purification or mussel transformation in cannery factories.

Finally, some studies collected the impacts of the fishing gear (Almeida et al., 2014; Driscoll and Tyedmers, 2010; Ziegler et al., 2003), catch variation (Driscoll and Tyedmers, 2010; Farmery et al., 2015), or multiple variables. For instance, Abdou et al. (2020) addressed management scenarios, varying the marine protected areas or the extension of the biological rest period, as well as evaluated a decrease in the number of demersal trawlers (Abdou et al., 2020). Driscoll et al.

(2015) studied the reduction of refrigerant leakage of the fishing fleet, the fuel consumption to wetpack and the use of energy, packaging and water, and the increment of gutted fish parts to human consumption. Winther et al. (2009) discussed electricity mix, product waste, edible yield, by-products and refrigerant agents.

#### 3.2. Life cycle impact assessment (LCIA)

Regarding life cycle impact assessment (LCIA) methods, CML-IA (Guinée, 2001) was the most widely used representing approximately 50% of the studies reviewed. There was a marked decline in the use of its versions (2000, 2001 and 2002), as evident in Table 3. The impact categories included in this method are those used in many LCA studies. The baseline indicators, which are the standard, are based on the best practice principle available and are category indicators at the outcome level (also referred to as the problem-oriented approach). LCA practitioners rarely applied the non-baseline, which addresses 11 impact categories in CML 2001, disaggregated into 50 subcategories, against the 8 general impact categories of the baseline (acidification, climate change, depletion of abiotic resources, ecotoxicity, eutrophication, human toxicity, ozone layer depletion and photochemical oxidation).

Likewise, more recent assessment methods were applied, such as ReCiPe (Huijbregts et al., 2017) that was implemented in approximately 25% of the publications analysed. This method can be described as an update and combination of 18 midpoint indicators of the CML 2001 and the three endpoint indicators (i.e., damage to human health, ecosystem quality, and resource availability) of the Ecoindicator 99 methodologies. In addition, utilization of other assessment methods addressing specific impacts, such as the carbon footprint that was updated through the 2001, 2007 and 2013 versions of the IPCC, the PAS 2050 British specification published in 2008 (Iribarren et al., 2010e, 2011), and the GHGs emission factors (Denham et al., 2016; Driscoll and Tyedmers, 2010; Hallstrom et al., 2019); or the energy consumption through the Cumulative Energy Demand (CED) (Vázquez-Rowe et al., 2014a), aiming at energy return on investment (EROI) calculation.

Some of the aforementioned methods were used for the same studies, or in combination with others. For instance, UseTox, developed under the United Nations Environment Program and the Society for Environmental Toxicology and Chemistry Life Cycle Initiative (UNEP-SETAC) (Rosenbaum et al., 2008), was applied together with ICLM and ReCiPe to evaluate the air, agricultural soil, natural soil, freshwater and seawater dimension of the impacts (Avadí et al., 2014a; Avadí and Fréon, 2015; Fréon et al., 2014b), while the ILCD recommendation (European Commission, 2011) was also implemented with ReCiPe and CED by Winther et al. (2009).

#### 3.2.1. Impact categories in LCIA

Almost 50 impact categories indicators were found in the 59 studies reviewed (see Supplementary Material). However, several indicators refer to the same environmental mechanism impact (midpoint) or damage (endpoint). In order to simplify the analysis, we were able to identify 17 types of impacts analysed by the papers, which can be computed by different methods: climate change, indicator of potential global warming due to GHG emissions; ozone depletion, for air emissions destroying the stratospheric ozone layer; photochemical oxidant formation, focused on the photochemical ozone created in the lower atmosphere (smog) catalysed by sunlight; particulate matter formation, for assessing damage to human health due to primary PM2.5 and PM2.5 precursor emissions; ionizing radiation, related to the damage that emissions of radionuclides produce in human health and ecosystems; acidification, of soils and water due to the release of gases such as nitrogen and sulphur oxides; eutrophication, measuring the abnormal nutrient enrichment of aquatic ecosystems due to nitrogen or phosphor containing compounds; ecotoxicity, which evaluates the toxic emissions on freshwater, sea water or land; human toxicity, based on the potential harm of emitted substances in people; land use, addressing the damage for occupation; abiotic depletion, referred to the consumption of non-biological resources; water use; and energy demand, indicator to quantify the primary energy usage (Acero et al., 2017).

Table 2 shows this classification and to which acronym the impact categories type refer (for: CML-IA, ReCiPe, ESA/IChemE). It also shows the number of studies (and the associated percentage) that consider each type of impact. The detail of the main impact categories studied in each study is shown in Table 3. These LCIA indicators also involved more than 20 indicators related to specific issues (specific to fisheries or socio-economic topics) that are discussed in the Section 3.2.2.

Focusing on emission related impacts, 54 studies (92%) computed climate change impacts as it constituted the most scrutinized impact in LCA. It should be noted that 6 studies measured carbon footprint exclusively and, therefore, only assessed climate change impacts. Ozone depletion (53%), photochemical oxidant formation (42%), acidification (64%), eutrophication (66%) and ecotoxicity (46%) were the 5 next most represented impact categories that were computed in most of the LCIA methods reviewed. Surprisingly, human toxicity impact category was only found in 32% of the studies reviewed, despite the fact that it is included in most of LCIA methods. The lack of consensus behind this impact category due to the uncertainty and variability related to both the health and ecotoxic effect data or the limited data on bioconcentration factors for fish or chemical degradation rates, among other parameters, could be one of the main reasons explaining its underrepresentation in the sample assessed (Rosenbaum et al., 2008). Finally, the two other emission-related impact categories, namely particulate matter formation and ionizing radiation, were found in only 14% and 10% of the studies, respectively. Such impact categories could be of interest for the fishery and seafood sector, more singularly particulate matter formation due to the influence of fishing boats to the ambient air and human health near the coast (Zhang et al., 2018).

Impact categories related to resource use and energy indicators were less represented in the studies reviewed. Abiotic depletion (including fossil and metal) and land use categories were found in 41% and 22% of the studies, respectively. Water use related impacts were computed in only 17% of the studies, which can be justified by the fact that methodology development in this field is recent. Indeed, the water foot printing was coined as "virtual water" in 1997, later renamed to the current terminology at the beginning of the millennium and finally adopted in the LCIA methodologies in 2010s (Pfister et al., 2017). Specific biotic resources use and sea use impact categories for fisheries have been applied in the reviewed papers and are more specifically analysed in the Section 3.3.2. Notwithstanding this, energy demand indicators are assessed in nearly half of the studies (46%). Even if such indicators are not part of LCIA methods as they are rather synthetic LCI indicators, it highlights that they are frequently used as complementary information.

Recent publications that describe the use of up to date and innovative methodologies, such as ReCiPe, suggests that it is worth including a large set of environmental impacts that may be of importance for the fisheries sector. This would also enable the computation of endpoint damage and would provide synthetized information. Modelling of endpoint damages was found in 8 studies (Laso et al., 2018c; Vázquez-Rowe et al., 2014b; Avadí and Fréon, 2015; Avadí et al., 2014a; Avadí et al., 2014b; Fréon et al., 2014b; Gonzalez-García et al., 2015) using ReCiPe. ReCiPe aggregates 18 midpoint impact categories in three endpoint damage categories or areas of protection, namely human health, ecosystem quality and resources. These 8 studies also compute single score that weights and normalizes the three areas of protection. Studies highlight the benefit of single score for communication, however, the large uncertainty associated with such metrics was evident in some studies (Laso et al., 2018a, 2018b, 2018c).

Another trend in LCIA is the regionalization of impacts in order to better represent site-specific environmental interventions (Patouillard et al., 2018). This may enhance the relevance of LCIA for seafood products because they usually generate direct impacts that are space dependant (e.g., impacts associated to toxic or eutrophying substances emitted in the marine environment or to water use in the supply chain). However, none of the reviewed studies used spatially differentiated LCIA methodologies (such as LC-Impact (Verones et al., 2020)), as they are too recent and not yet implemented in LCA software.

### 3.2.2. Fishery-specific impacts categories

Other impact categories related to the fish and seafood sector are assessed in the studies reviewed. Although this kind of indicators were rarely the main target of studies, 25 different indicators related to biotic (fish) resource, sea use, nutritional and socio-economic approaches were analysed (see Supplementary Material).

Impacts related to the removal of fish stocks have been estimated by several different indicators in 37% of the studies reviewed. A significant

#### Table 2

Frequency of LCIA impact categories and methods application in the reviewed studies.

Impact categories	CML-IA	ReCiPe	ESA IChemE	Others methods/ indicators	N studies	%
Climate change	GWP	СС	GWP	_	54	86%
Ozone depletion	ODP	OD	SOD	-	31	49%
Photochemical oxidant formation	POFP	POF	POF	-	25	40%
Particulate matter formation	-	PMF	-	-	8	13%
Ionizing radiation	-	IR	-	-	6	10%
Acidification	AP	TA	AA	-	38	60%
Eutrophication	EP	FE, ME	E, AOD	-	39	62%
Ecotoxicity	METP, TETP	TE, FE, MET	EAL, EAL2	-	27	43%
Human toxicity	HTP	HT	-	-	19	30%
Land use	LOP	ALO, ULO, NLT	-	-	13	21%
Abiotic depletion	ADP	MD, FD	-	-	24	38%
Water use	-	WD	-	-	10	16%
Energy demand	-	-	-	CED, TCED, EU	27	43%
Biotic (fish) resources use	-	-	-	see detailed analysis	22	37%
Nutritional impact	-	-	-	see detailed analysis	7	12%
Sea-bed damage	-	-	-	see detailed analysis	1	2%
Socio-economic	-	-	-	see detailed analysis	3	5%

GWP: Global Warming Potential; CC: Climate Change; ODP: Ozone Depletion Potential; OD: Ozone Depletion; SOD: Stratospheric Ozone Depletion; POFP: Photochemical Oxidant Formation Potential; POF: Photochemical Oxidant formation; PMF: Particulate matter formation; IR: Ionizing radiation; AP: Acidification Potential; TA: Terrestrial acidification; AA: Atmospheric acidification; EP: Eutrophication Potential; FE: Freshwater eutrophication; ME: Marine eutrophication; E: Eutrophication; AOD: Aquatic oxygen demand; METP: Marine Eco-Toxicity Potential; TETP: Terrestrial Eco-Toxicity Potential; TE: Terrestrial ecotoxicity; FE: Freshwater ecotoxicity; MET: Marine ecotoxicity; EL: Ecotoxicity to aquatic life (metals to seawater); EAL2: Ecotoxicity to aquatic life (other substances); HTP: Human Toxicity Potential; HT: Human toxicity; LOP: Land Occupation Potential; ALO: Agricultural land occupation; ULO: Urban land occupation; NLT: Natural land transformation; ADP: Abiotic Depletion Potential; MD: Metal depletion; FD: Fossil depletion; WD: Water depletion; CED: Cumulative energy demand; TCED: Total Cumulative Energy Demand; EU: Energy use.

	8
	1
	ŝ
	ŧ
ŝ	4
le	
ab	ŝ
<u> </u>	

Impact categories per sti	udy.																
Study	LCIA Method	Typical LCI	IA indicators										Ι	.CI F	ishery spec	ific	
		Climate change	Ozone depletion	Photochemical oxidant formation	Particlaute matter formation	lonizing radiation	Acidification	Eutrophication	Ecotoxicity	Human toxicity	Land use (	Abiotic V lepletion	Water   use o	Energyu demand r	Biotic (fish) esources use	Sea Nutritional Socio-ec use	conomic
Abdou et al., 2018	CML baseline 2000	×	×	×			×	×	×	×	×	×		×	×	×	
Abdou et al., 2020	Ecopath with Ecosim (EWE)	×	×	×			×	×	×	×	×	×		×	×		
Almeida et al., 2014	CML baseline	×	×				×	×						×	×		
Almeida et al., 2015	CML-IA baseline	×	×	×			×	×	×			×		×			
Avadí et al., 2014a	keure, cML baseline 2000 and	×					×	×		×	×	×	×	×	×		
Avadí et al., 2014b Avadí et al., 2015	usetox ReCIPE ReCiPe	××	×	××	××	×	×	××	×	××	×	××	×	×			
Avadí and Fréon, 2015	ReCIPE, CML-IA baseline 2000 and	×	×	×	×	×	×	×	×	×	×	×	×	×	×	~	×
Avadí et al., 2018	USEtox ReCiPe	×		×	×			×	×	×	×	×	×	×	×		
Denham et al., 2016	GHGs emission factors	×															
Driscoll and Tvedmers. 2010	GHGs emission factors	×													×		
Driscoll et al., 2015	CML baseline 22.000	×	×				×					×		×	×		
Farmery et al., 2015	Australian impact	×						×	×				×	×	×		
Fréon et al., 2010	N/A N/A															~	×
Fréon et al., 2014b	CML 2000 and	×	×	×	×	×	×	×	×	×	×	×	×				
Fréon et al., 2014c	2001 ReCiPe	×					×	×						×	×	~	×
Fréon et al., 2017	ReCiPe	×	×	×	×	×	×	×	×	×	×	×	×				
Gonzalez-Garcia et al., 2015	ReCiPe	×															
Hallstrom et al., 2019	GHGs emission factors	×															
Hélias et al., 2018	N/A														×		
Hospido and Tyedmers, 2005	CML baseline	×	×	×			×		×	×							
Hospido et al., 2006	CML baseline	×	×	×			× >	×	>	>	>	×					
Iribarren et al., 2010b	CML 2001	< ×	< ×	< ×			< ×	< ×	< ×	< ×	< ×	< ×					
Iribarren et al., 2010c	CML 2001	×	×	×			×	×	×	×	×	×					
Iribarren et al., 2010d	PAS 2050 and IPCC, 2007.	×															
Iribarren et al., 2010e	PAS 2050 and IPCC, 2007.	×															
Iribarren et al., 2011	PAS 2050 and IPCC 2007	×															
-	ESA with metrics	;	:				:	;	;								
Laso et al., 2016	trom IChemE 2002.	×	×				×	×	×								
Laso et al., 2017a	ESA with metrics from IChemE	×	×				×	×	×								
Laso et al., 2017b	2002. ESA with metrics	×	×				×	×	×								
	from Icheme																

																															3 5%
		×																													1 2%
													×							×					×			×	×	×	7 2% :
				;	×	×	×	×					×				×			×					×			×	×	×	22 17% 1
																															. >0
		×	×			×		×			×	×	××	<				>	~		×		×	×		×	×	×		×	27
		×																							×	×					10 17%
				;	×					×				>	<	×	×	>	~	×					×	×					24 41%
																									×	×					13 22%
				1	×								*	< >	<										×	×				×	19 32%
				;	×					×			*	< >	<		×	>	~	×					×	×			×	×	27 46%
×	×		×	: :	×	×				×	×	×	×	< >	<	×	×	>	×	×					×	×		×	×	×	39 66%
×	×		×	: :	×	×				×	×	×	*	< >	<	×	×	>	~	×					×	×		×	×	×	38 54%
																									×	×					6 10%
																									×	×					8 14%
				;	×							×	*	< >	<	×				×					×	×		×	×	×	25 42%
				:	×	×				×	×	×	*	< >	<	×	×			×					×	×			×	×	31 53%
																															4 %
×	× ť	×	×		×	×	L.	×		×	×	×	~		<	×	×	>	~	×	×	×	×		×	X D	×	×	×	×	es 5 <sup>2</sup>
JRC of EC. IPCC, 2013, ReCiPe,	LINIL-IA JRC of EC. ReCiPe CML-IA	IPCC, 2013 and specific	calculations CML baseline	2000 CML baseline	2000	CML 2 baseline 2000	Marine footprin	IPCC 2007 characterization	factors CMI haseline	2000	CML baseline 2000	CML 2 baseline	ISO 14040-43	CML baseline	2000	CML baseline 2000	CML baseline 2000	EMEP Corinair,	LML baseline 2000	CML baseline 2000	IPCC 2001	IPCC 2001	IPCC 2001	Cumulative Energy Demand	ReCiPe	ILCD, ReCiPe, CE	IPCC 2007; CED	LML baseline 2001	CML baseline 2001, Eccindicator 00	CML baseline	Number of studi %
Laso et al., 2018a	Laso et al., 2018b	Laso et al., 2018c	Lourguioui et al.,	2017	Lozano et al., 2010	Parker and Tyedmers, 2012a	Parker and Tyedmers, 2012b	Parker et al., 2015		Ramos et al., 2011	Van Putten et al., 2016	Svanes et al., 2011	Thrane, 2004 Thrane, 2006	Vázquez-Rowe et al.,	2010a	Vázquez-Rowe et al., 2010b	Vázquez-Rowe et al., 2011a	Vázquez-Rowe et al.,	2011b	Vázquez-Rowe et al., 2012a	Vázquez-Rowe et al., 2013a	Vázquez-Rowe et al., 2013b	Vázquez-Rowe et al., 2013c	Vázquez-Rowe et al., 2014a	Vázquez-Rowe et al., 2014b	Villanueva-Rey et al., 2018	Winther et al., 2009	Ziegler et al., 2003	Ziegler and Valentinsson, 2008	Ziegler et al., 2011	

number of studies assesses this impact given that overfishing ranks as one the most important threat to biodiversity loss in global marine ecosystems (Millennium Ecosystem Assessment, 2005; Emanuelsson et al., 2014). Such indicators include mean trophic index, lost potential yield, biotic natural resources index, discards and fish biomass extraction, among others. Avadí and Fréon (2013) already extensively discuss these indicators and their relevance, indicating that most of them are stand-alone indicators that are not part of LCIA methods. The diversity of indicators used shows the importance of setting up consensual or harmonized indicators to address the challenge of biotic resource use, a task that has been led by the Life Cycle Initiative through its Global Guidance on Environmental Life Cycle Impact Assessment Indicators (GLAM) project (LCI, 2020). It should be noted that there have been attempts to better integrate such indicators in LCIA framework, leading to natural resources area of protection (Hélias et al., 2018; Hélias and Heijungs, 2019).

Seabed damage is also a predominant driver for biodiversity loss in oceans worldwide (Millennium Ecosystem Assessment, 2005; Woods and Verones, 2019). Twelve percent of the studies consider this impact with various indicators that are discussed in Avadí and Fréon (2013) and Woods et al. (2016). Meanwhile, only one study considers nutritional indicator, through the use of protein content (Laso et al., 2018c). Nutritional impacts are more and more assessed in LCA of food and new methods are being developed in this area (Stylianou et al., 2016). Phase 3 of GLAM should focus on biotic resources and nutritional impact and therefore give guidance to the fishery and seafood sectors to address these important impact and damage categories (UNEP, 2020). Moreover, Woods et al. (2016) recommend the inclusion of (over)exploitation of fish and seabed damage in order to have a meaningful assessment of marine ecological impacts in LCA.

Other pathways recommended by Woods et al. (2016) include marine plastic debris related impacts. This was not found in the reviewed papers whereas fisheries can contribute to this impact category. This is because no operational methods exist yet to integrate such impacts in LCA. New methods are being developed in the frame of the Marine Impacts in Life Cycle Assessment initiative (MarilCA, 2020), following the call of the Medellin Declaration to develop new impact pathways to account for this environmental hazard (Sonneman and Valdivia, 2017).

LCA traditionally focuses on the evaluation of the environmental impacts of processes or products. Thus, discussion of socio-economic issues has been minimal in the context of fisheries LCA literature (Pelletier et al., 2007). All of the 59 reviewed studies focus on environmental impacts of production systems, and only three address socioeconomic indicators in fishery-specific impacts categories. Yet, in order to address sustainability objectives, assessments need to consider not only environmental aspects but also social and economic impacts (Kruse et al., 2008).

Avadí and Fréon (2015) proposed to complete LCA indicators with a set of other indicators to evaluate the sustainability performance of anchoveta fisheries and freshwater aquaculture industries. The set included nutritional profiling, energy and socio-economic assessments. The socio-economic indicators include production costs, added value, gross profit generation and employment. This approach allows accurate comparisons of different products by bringing an added value to LCA and gives a concrete perspective of sustainability by incorporating the social and economic perspectives together with the environmental one.

Other life cycle methods, namely Social LCA (SLCA) and Life Cycle Costing (LCC), have been developed as necessary complements for capturing trade-offs between environmental, social and economic aspects along the life cycle of production systems (Dreyer et al., 2006; Guinée et al., 2011). For instance, Soltanpour et al. (2020) used SLCA to analyse a case of fisheries management. However, SLCA approaches show that the perception of social impacts is highly variable, and the methodology is often debated, in particular regarding data frames (Jørgensen et al., 2008). Moreover, the Life Cycle Sustainability Assessment (LCSA) framework combines LCA, SLCA and LCC. LCSA evaluates environmental, social and economic negative impacts and benefits on decision-making processes towards more sustainable products throughout their life cycle (Valdivia et al., 2013). LCSA has also been applied to fisheries management research. For instance, Kruse et al. (2008) attempted to apply this approach in a seafood context.

Beyond life cycle methods, a great variety of system analysis tools have been developed, focusing on diverse types of impacts and dimensions of sustainability. Some of those methods could complement fisheries LCAs for wider, more holistic studies (Avadí and Fréon, 2013). By using an input-output model to test the socio-economic impacts on a few case studies in the Atlantic coast, the Interreg Neptunus project will also study economic implications of seafood circular economy, as well as economic benefits and drawbacks of implementing actions for proposed strategies under a circular and NEXUS eco-labelling approach (Neptunus, 2020). There is also scope to assign risk assessment categories for input-outputs model that will contribute to LCA knolwedge for seafood sector as described recently by Tahar et al. (2017) who focused on the waste water industry.

#### 3.3. Seafood LCA practitioners: who and why

From the revision of the LCA studies, the location of the research institutions involved and the species studied offers a good overview of the investigation on seafood. Fig. 4 qualitatively represents the geographic distribution of the authorship of the studies per country and fishing ground (area from which the studied species are caught). Most of the research was carried out by one or several institutions from only one territory, while 14 had international collaboration and involved researchers from two to four different countries. Geographically, Europe concentrates the largest number of research centres practicing LCA, constituting the FAO fishing area 27 and the most evaluated oceanid territory (FAO, 2015). Institutions from Spain participated in 33 studies collected in this review, followed by France and Peru (9 each), Sweden (6), Canada (5), Australia (4), Denmark and Portugal (3 each), Norway, Tunisia and United States of America (2 each) and, finally, Algeria, Ecuador, Italy, Luxembourg, Switzerland and United Kingdom (1 each). Regarding the fishing zones or waters, LCA studies in the 2000s focused on Atlantic and Pacific fisheries (Vázquez-Rowe et al., 2012b). However, to date, almost all major waters have been part of at least one article including -Mediterranean, Caribbean, Baltic, Tasman or North seas- but excluding, for instance, the Black or Caspian seas and Eastern Asia. This is notable as Asia is the largest worldwide fish producer: China remains a major fish producer, accounting for 35% of global fish production (FAO, 2020). Although numerous studies on aquaculture LCA have been also published in Asia in recent years (Henriksson et al., 2018; Järviö et al., 2018). It's appreciated that LCA typically remains a Western tool in this field of study, which is probably due to the higher environmental requirements for food production in European countries and others belonging to the Organization for Economic Co-operation and Development (EOCD, 2020). This unbalanced distribution can also be related with persistent differences in fish consumption levels. In developed countries the apparent fish consumption is 26.4 kg, 22% above global average (20.5 kg) while in developing countries it is considerably lower, 19.4 kg (FAO, 2020).

With respect to the analysed species, most studies focused on a single species or several similar species (Hospido and Tyedmers, 2005; Parker et al., 2015; Avadí et al., 2015). Typically, these species are emblematic of the regions analysed, such as anchovy in Peru (Fréon et al., 2010, 2014a, 2014b, 2014c, 2017) and Cantabria (Laso et al., 2016, 2017a, 2017b, 2018a, 2018b, 2018c), mussels in Galicia (Barros et al., 2009; Iribarren et al., 2010a, 2010b, 2010c, 2010d), pilchard in Portugal (Almeida et al., 2014, 2015; Gonzalez-García et al., 2015), white banana prawn in Australia (Farmery et al., 2015) or lobster in the United States (Driscoll et al., 2015). Few authors addressed the analysis of multiple species in the same study (Parker and Tyedmers, 2012b; Vázquez-Rowe et al., 2012b; Hélias et al., 2018; Avadí et al., 2020) and



Fig. 4. Geographic distribution of LCA publications per country (cap) and fishing ground (fish). The most intense colour in a territory indicates the largest number of studies.

some included species from coastal fishing, offshore fishing, deep-sea fishing, extensive aquaculture and intensive aquaculture (Iribarren et al., 2010e, 2011; Winther et al., 2009; Avadí and Fréon, 2015).

Almost all studies have been conducted by public bodies, such as universities, institutes and schools with the purpose of characterizing the seafood sector to address the potential environmental impact of the seafood supply chain, from fishing through processing up to consumers, as well as to identify hotspots and evaluate improvement opportunities to promote the sustainability of this part of the Blue economy. Research focused on evaluating the marine species stock, eventual increase of the wild-caught fishery, potential expansion of aquaculture, ecolabelling, effects of climate change in ocean ecosystems, and use of different fishing gears were issues that were the topic in these LCA studies. In contrast, very few LCA articles shared authorship with private companies, industries or consultancies (Laso et al., 2017a, 2017b, 2018a, 2018b, 2018c; Avadí et al., 2018). Only one research article was attributed to industry, which analysed and compared (with competing products from EU market) the carbon and energy footprint of Norwegian seafood supply chain (Winther et al., 2009). The remaining studies were developed by LCA practitioners in research centres from a theoretical perpsective based on direct enquiries to stakeholder and producers, official inventories, and so forth. This is probably due to scientific interest as the public and/or market relevance in their countries; or simply, ease of data collection.

The scientific and technical knowledge of some papers provides valuable inputs for fishery management and future regulations (Ziegler et al., 2016; Villanueva-Rey et al., 2018). For instance, fuel use and packaging in canning products were identified as important shortcomings to be addressed in the extraction and processing phases, respectively. These implications may guide policymakers and stakeholders about where best to make improvements that will lead to sustaining processes and design strategies, including behavioural change. Moreover, several papers promoted new approaches for the first time including 'geographically' focusing on LCA for aquaculture (Lourguioui et al., 2017) and fisheries (Abdou et al., 2018) in the Mediterranean or making a comprehensive LCA of the entire Peruvian anchovy fleet (Fréon et al., 2014b). Other studies report on the 'timely' development of the first fishery LCA study with inventory data for the period 2001 to 2008 (Ramos et al., 2011; Almeida et al., 2014). 'Technically', the first LCA study of fishmeal plants considered construction and maintenance phases (Fréon et al., 2017) or applying the 'water-energy-food-climate nexus" index to a case study for fisheries (Laso et al., 2018c). "Managerially', the first ecolabel in the Spanish fishing sector has been based on life cycle approaches for seafood products (Vázquez-Rowe et al., 2016).

#### 3.4. Nexus water-energy-food in the seafood sector

The assessment of individual environmental impact categories in LCA offers high added-value information, such as providing insights for further improvements in processes, products and services, as well as support in policy-making and environmental certification schemes. However, such an assessment gives a perspective that is circumscribed to the link between each impact category and the specific environmental problem that it represents. Although in some cases the environmental impacts of these impact categories are aggregated in endpoint damage indicators, which provide a more direct connection with the tecnosphere and the environment, in the case of seafood LCA studies, as mentioned above, these were only applied in 8 studies. Moreover, even the computation of endpoint damages with current LCIA methods lacks a strong interconnection between variables to reach robust results that allow a holistic analysis of production systems.

In this context, the nexus approach arises as a perspective that recognizes the fact that water, energy, food production systems (in this case, wild fish capture) and natural ecosystems show a series of robust and indivisible multi-dimensional interlinkages. We argue that when referring to life cycle studies in the seafood sector, nexus thinking is needed to appropriately transition to a circular economy (Ruiz-Salmón et al., 2020). This is essentially a transformative approach to governance, and also requires substantial changes in individual behavior. Therefore, the nexus implies how to govern such transformations, and the policy tools that will be required, including behavioural change interventions that go beyond mere education to influence how people make decisions regarding the purchase and consumption of marine products. In contrast, a traditional fragmented approach, that obviates this nexus perspective while attempting to achieve resource security independently, will not only generate sub-optimal outcomes, but may also endanger food security and sustainability (Staupe-Delgado, 2020).

Water, energy and food are basic requirements for everyday life and are key activities advancing the seafood sector. In this sense, the lack of a secure and economical provision of one of them might lead to disruption in the supply and accessibility of the two others (Machell et al., 2015). Thus, the application of a water-energy-food nexus under a holistic approach appears to cover the gap of the isolated impact categories aforementioned. LCA is particularly important for understanding the interconnections in the nexus, as it enables the consideration of entire supply chains. In fact, Hamiche et al. (2016) highlighted LCA as a valuable tool to shed light on the links between the water and energy sectors, since it is able to account for direct and indirect consumption. Similarly, De Laurentiis et al. (2016) and Mannan et al. (2018) also considered LCA as the best available tool to enable the developing of the nexus framework, driving the shift towards sustainable food systems thereby.

Having said this, a nexus approach using life cycle methods would have to redefine the selection of environmental impact categories that are traditionally used (see Table 3). For instance, the guantification of freshwater consumption by the seafood sector has received little attention (Vanham, 2016). This is remarkable considering that freshwater aquaculture, as well as fish processing in general, is a water-intensive industry and a large discharge of organic material. Therefore, we consider that new water consumption metrics, currently available in LCIA methods (e.g., AWARE), as well as a wider array of water degradation categories should be included in seafood LCA studies in order to enhance the meaningfulness of these under a nexus approach. Moreover, this would allow determining whether the planetary boundary of freshwater use is below its breaching risk in the studies conducted, ensuring, this way, clean, affordable and accessible energy generation and sustainable freshwater consumption to support food supply (Steffen et al., 2015).

In terms of energy, it must be noted that a high correlation exists between food and energy prices. This connection is even more pronounced for intensive capture fisheries, which are usually fully dependent on fossil fuels (Parker and Tyedmers, 2015). In contrast to water, energy has been repeatedly reported and included in seafood LCA studies, directly, through the use of the CED impact category or the EROI indicator, or indirectly, by monitoring environmental impacts that are heavily dependent on energy intensity (Weidema et al., 2008).

Finally, focusing on the food component of the nexus, seafood products are widely accepted to be an essential component of a balanced and healthy diet because they have a high "good fat" content and provide high quality proteins and many micronutrients, such as vitamins and minerals (Larsen et al., 2011; WHO, 2020). For instance, the NRF9.3 score, i.e., a nutritional index based on 9 nutrients to encourage and 3 nutrients to limit (Drewnowski, 2009) has been combined in the literature with LCA, mainly from a human diet perspective, in an effort to analysed the sustainability of diets across the world and determine, through linear programming or other methods (Vázquez-Rowe et al., 2019; Larrea-Gallegos and Vázquez-Rowe, 2020), future healthy diets. In this context, Castañé and Antón (2017) identified a considerable environmental impact due to a high global warming potential per kg of seafood produced due to energy consumption, but no details were given on the specific seafood products that were included in the assessment. Similarly, some attempts have been made to include a ranking for fish species based on nutrient density, climate impact and their combination (Hallstrom et al., 2019), but the nexus between the potential of the nutrients from seafood and environment effects is yet to be deeply researched.

Overall, the nexus approach can support the identification of synergies and trade-offs between water and energy systems and food systems aiming at resources efficiency and environmental impacts -production and consumption-reduction (Mannan et al., 2018). Advancements towards a greater linkage between terrestrial and marine systems, however, are necessary for fishing activities within a nexus framework. Freshwater consumption should be considered throughout the full supply chain of seafood products (Gephart et al., 2017; Vanham, 2016), as well as energy consumption produced by water resources (D'Odorico et al., 2018). The water dimension of the nexus should consider the water used during on-board activities (e.g. ice and water) and on-land activities where several processing activities consume freshwater and use energy produced by water resources (Gephart et al., 2017; Vanham, 2016). Some of the freshwater used in these processing activities can result in wastewater that, in turn, also consumes energy for its treatment (Vanham, 2016). Furthermore, there are also seafood processed-derived products that use crop ingredients (Vázquez-Rowe et al., 2013a), which, in turn, also consume freshwater that needs to be accounted for (Salmoral and Yan, 2018). Moreover, within a global context of increasing population and growth of water and energy use, the production, consumption and waste of food raises social questions, such as global food insecurity, food prices and international food shortages. Consequently, reducing food losses and waste is necessary to meet forecasted nutritional needs and to enhance the link with the energy and water implications.

All in all, a standardization for the nexus variable which defines, evaluates and modifies future strategies is lacking. The existence of this framework would allow the analysis of the interacting governing forces and balance the nutritional, economic and energetic value of the seafood sector to foster informed decisions, engaging industry stakeholders and consumers. Thus, the development of a unified water-energy-food nexus framework (Endo et al., 2017) would address the opportunity to strengthen the social, economic and environmental aspects of this specific but worldwide system. Furthermore, the nexus also helps international policies and scientific researchers to better adapt local and regional commerce for new perspectives. Finally, there are clear links between the SDGs of the Agenda 2030 of the UN and the water-energy-food nexus. Hence, this nexus can support the achievement of the SDGs when taking a transdisciplinary approach (Ghodsvali et al., 2019).

#### 3.5. Challenges and opportunities for LCA in the seafood sector

The application of LCA in the broader seafood sector presents a number of key challenges and opportunities, which in many cases are related to the circular economy and the management of process waste streams. Some of the most pressing challenges facing the sector are the: (i) proliferation of marine debris comprised of plastics and other artificial materials (Maximenko et al., 2019); (ii) generation of waste and its valorisation, and (iii) climate change (Ruiz-Salmón et al., 2020). These are challenges that LCA can help address in particular when evaluating the nature of the challenge, analysing how various parts of the seafood sector contribute to these issues and identifying where targeted improvements to the sector can yield most sustainability gains while transitioning to a CE.

#### 3.5.1. Marine debris

It is estimated that there are 120 million tonnes of plastic in the oceans and between 11.6 and 21.1 t in the Atlantic Ocean alone (Jambeck et al., 2015; Pabortsava and Lampitt, 2020). These plastics are non-biodegradable and breakdown into smaller pieces due to weakening by ultraviolet light and the motion of the ocean. They can impact on marine wildlife and trophic levels, where organisms mistake the plastics for food resulting in ingestion which may cause physical impairment or death (Maximenko et al., 2019; Provencher et al., 2019). The use of additional substances (e.g. colorants and stabilisers) in the production of these plastics, may make them toxic, further exacerbating their impacts. There remains, however, a significant gap of knowledge in the understanding of these toxicological and ecotoxicological impacts. Derelict and lost fishing gear, as well as shipwrecks, is the archetype of how seafood supply chains and processes can contribute to marine debris. It is estimated that lost gear makes up 46% of the Great Pacific Garbage Patch (Lebreton et al., 2018; Maximenko et al., 2019). As well as contributing to marine debris, lost fishing gear, commonly referred to as ghost nets, can continue to capture and kill fish and other organisms for many years after being lost. While switching to biodegradable nets is being evaluated, it has been reported that these nets may have a lower catch efficiency when compared to conventional nylon nets (Grimaldo et al., 2019). These losses in efficiency and a possible increase in fishing effort to offset this loss, is something which would benefit from an LCA perspective; particularly with regard to changes related to fuel use, the largest contributor to the impacts (Avadí et al., 2020), but also in other inventory items, such as refrigerating agents (Vázquez-Rowe et al., 2012a).

While the impact of ghost fishing (Vázquez-Rowe et al., 2012c) and seabed disturbance (Woods and Verones, 2019) has been described in some fishery LCA studies, current LCIA methods do not consider certain marine environmental impacts, such as biotic depletion or the degradation of the marine environment due to plastics accumulation in the ocean or damage to seafloor (Avadí et al., 2020), despite certain methodological advancement in recent years (Hélias et al., 2018). Furthermore, there is concern over the trophic transfer of these substances due to bioaccumulation and biomagnification (Maximenko et al., 2019; Provencher et al., 2019). Although Woods et al. (2019) have already developed effect factors due to entanglement from macroplastics, research into accounting for these impacts in LCA is ongoing through the MariLCA working group (Marilca, 2020). Finally, inventory flows linked to the presence of nano-, micro- and macroplastics, which can be potentially ingested, remain scarce, although certain initiatives, such as the study by Stefanini et al. (2020), or the PLP report recently published by Quantis (2020).

Similarly, the environmental consequences of the release of plastic debris to water bodies has arisen the interest of the LCA community, since this environmental dimension is not included in current metrics. In fact, Saling et al. (2020) have recently proposed a characterization model regarding the relationship between degradation and fragmentation of plastics in the marine environment. Moreover, Woods et al. (under review), have produced a detailed framework in which they describe the main cause-effect pathways that must be considered in LCIA in order to account for the different damages to human health (mainly through seafood), ecosystem quality (e.g., entanglement or invasive species) (Woods et al., 2019) and other endpoint damages.

#### 3.5.2. Waste valorisation

Another challenge the seafood sector is facing is the issue of seafood waste, which can indirectly increase pressure on fisheries through increased fishing effort to supplement this waste. It is estimated that around 40% of the total food supply is lost or wasted between harvesting, production and processing (Laso et al., 2016; Love et al., 2015). In Europe it estimated that seafood losses and wastage rates are greater than 30% (FAO, 2011). This loss and the knock-on effect of triggering increased fishing effort to meet market demand perpetuates a linear economy of use and waste in food production. The adaptation of a circular economy based on business models which replace the 'end-of-life' concept with reducing, alternative reuse, recycling and recovering materials in production/distribution and consumption processes, at the micro, meso and macro level can promote the minimisation of food loss and wastes through the development of more sustainable production loops (Kirchherr et al., 2017). One of the main avenues in closing the loop in seafood production is through the valorisation of classical waste streams and their utilization in other industries while eventually being fed back to the original industry (de la Caba et al., 2019). An example of this would be sludge from finfish aquaculture and its use as a substrate for anaerobic digestion and electricity generation. Examples of valorised waste which were mentioned in this review article were: trimmings, such as heads and spines, as fishmeal and fish oil, waste

meat or flesh as a paste (Laso et al., 2016) and mussel shells as a source for calcium carbonate (Iribarren et al., 2010a). A circular economy approach applied to these waste streams can highlight management and valorisation opportunities for sectors, processors and companies. Examples of waste or co-product valorisation were previously discussed in Section 3.1.3.

Emerging valorisation strategies include the use of by-products as bio-based materials (García-Santiago et al., 2020). Fish waste has been used for biodiesels and activated carbon production mainly via extractions from fish oil (Fadhil et al., 2017). Blood waters from pelagic processing plants have been shown to be a potential source of proteins, amino acids and vitamins which can be used as ingredients for biobased materials such as fuels, inks and feeds (Barr and Landis, 2018; Fadhil et al., 2017; Hayes and Gallagher, 2019). A similar approach of extracting glycogen from wastewater has been put forward for mussel processing sites (Barros et al., 2009), where it may be used for lactic acid production. In the instance of fish skin and bone, collagen from these wastes can be valorised as fish gelatine and in the manufacturing of active packaging (de la Caba et al., 2019).

#### 3.5.3. Packaging

A key area in which LCA was applied and valorisation of waste streams can play a role is that of seafood packaging. Several studies reviewed in this article have identified the issue of packaging as being a hotspot in the environmental impact of seafood products. The majority of these studies have focused on canned products, such as European anchovies (Laso et al., 2016, Laso et al., 2017b; Laso et al., 2018a, 2018b, 2018c), Peruvian anchovies (Avadí and Fréon, 2015; Avadí et al., 2014a), sardines (Almeida et al., 2015; Hospido et al., 2006; Vázquez-Rowe et al., 2014b; Laso et al., 2016) with several studies also considering fresh and frozen products, mussels (Iribarren et al., 2010; Svanes et al., 2011; Thrane, 2006; Van Putten et al., 2016). The studies have focused on the use of conventional packaging techniques and materials, but a benchmarking and deeper investigation of these materials from a life cycle perspective including their production, recyclability prospects, or substitution by innovative packaging materials (de la Caba et al., 2019) as well as on the packaging design optimization can help in reducing the environmental challenge associated with this stage of the product life cycle.

#### 3.5.4. Best available techniques

Other means of implementing LCA and circular economy philosophies in the seafood sector can be reached through BATs (Barros et al., 2009; Laso et al., 2016; Laso et al., 2017a; Morris et al., 2019). A number of studies (Barros et al., 2009; Laso et al., 2017a, 2017b) have demonstrated that BATs and the use of environmental management tools such as maintaining an accurate inventory and promoting recycling of byproducts and wastes could reduce the environmental burden on environmental aspects such as energy, water and raw material consumption.

As these processes and valorisation strategies exit the research phase and enter validation and trial stages, it will be important to consider their influence on the life cycle impacts of seafood products. There is an opportunity to implement circular economy and life cycle techniques in the seafood sector, which can help it to become an example of how a transition from a linear to a circular economy may be achieved.

#### 3.5.5. Climate change

Finally, climate change can interact with fisheries in many different ways, through the increase in water temperatures, extreme water flow events (floods and droughts), and warming of maritime environments (Ruiz-Salmón et al., 2020). Regarding the sea surface temperatures (SST), it was found that warming is not homogeneous, and the pattern is complex, as observed in the Atlantic Ocean. Garrett et al. (2018) noted that data were not uniform in 2017 at Malin Head (north of Ireland), and were the highest on record at 0.89 °C above the

1981-2010 average. Nevertheless, some areas of the North Atlantic show a slight cooling, such as the subpolar gyre close to 50°N. This location, sometimes referred to as the "cold blob" (Rahmstorf et al., 2015), has cooled by about 0.9 °C since 1900 (Allan and Allan, 2019). In the mid-Atlantic (longitude approx. of Reykjavik and latitude approx. of Liverpool/Galway) seasonal cycle ranges from less than 10 °C to almost 15 °C (NOAA, 2019). Additionally, in this location a general increase in annual SST has been observed, especially through the last 20 years with 2015 and 2018 being cooler than the long-term average (Rayner et al., 2003).

These changes may impact migration routes of fish, such as mackerel, and stock recovery. Fish migrate between different regions in part based on water temperature through the year as fish species have evolved narrow temperature tolerances (Cheung et al., 2016). This will also adapt their cellular machinery to tolerate a wider range of temperatures, which demands a lot of energy. Fish bodies start to fail when they find themselves in warmer water, so they have to use their energy to move to cooler waters instead of breeding or searching for food. This problem was studied by Pinnegar et al. (2017) for mackerel. The authors set out a series of impacts of climate on UK fisheries, noting the spread of this species into Icelandic and Faroese waters, impacting quota allocation between nations and fleets and governance. Although these aspects are yet to be included in LCA studies, it is possible that the impact of plastic litter working as a vector for invasive species will be included within the Marilca framework (Marilca, 2020).

Moreover, energy use in the capture stage of fishing is large especially when compared to the nutritional energy that is gained. Trawling for small pelagic fish, for instance, result in about 12.5 kg fish per kg of fuel, whereas when trawling for shrimp the fuel to fish ratio is can be between 3 and 4 kg fish per kg of fuel (Furuya et al., 2011; Parker et al., 2018). Despite the improved efficiency of midwater trawling as compared to bottom trawling, however, a wide range of LCA studies demonstrate the better energy performance of pelagic nets, such as purse seines (Driscoll and Tyedmers, 2010; Vázquez-Rowe et al., 2010a; Avadí and Fréon, 2013; Gonzalez-García et al., 2015). This excludes further energy costs in on shore processing, distribution or storage. In fact, for fisheries, EROI or edible protein EROI have shown to be highly relevant when examining the fishery stage, which as previously explained, is highly energy intensive (Vázquez-Rowe et al., 2014a). EROI considers the energy extracted from the edible content of the fish divided by the energy consumed in the production process (e.g., fuel use in the harvesting stage). Guillen et al. (2016) analysed the EROI of selected European fishing fleets in 2008. They estimated the total EROI average (here presented as a decimal) as being 0.11: the energy content of fuel burnt was 9 times greater than the edible energy content of the catch, with gear-specific EROI values ranging from 0.02 for beam trawl to 1.12 for pelagic trawlers and seiners capturing low value species (e.g. herring, mackerel, sand eels and sprats). In addition, they compared the EROI of fish species to those of other food systems. For instance, the EROI for soybeans was estimated at 4.92, corn 0.81, wheat 0.89, beef (pastured-based) 0.05, and broiler poultry 0.177. Strategies to improve the performance of the fishing industry should include behavioural, technological and managerial efforts since the potential to reduce fuel consumption varies substantially between fisheries (Parker et al., 2018).

### 4. Conclusions

LCA has emerged as a powerful methodology for the environmental evaluation of the seafood sector along its supply chain. In fact, the increasing number of studies published demonstrate the interest of researchers, decision makers and stakeholders in the seafood sector to use LCA for the decision-making process. In fact, the seafood sector has used LCA to address many environmental sustainability challenges and, in turn, the studies carried out by LCA practitioners have served to reinforce LCA as a valid tool to deal with sustainability challenges for seafood and other sectors. Both parties need to continue this path together in order to increasingly sustain the level of innovation necessary for the sector to grow in the current context. The common challenge is to continue enriching each other and move forward in attaining improved sustainability in the sector. To this end, there are a number of key areas to be addressed.

The main methodological challenges for the LCA tool applied to the seafood sector lie in: (i) identifying and defining the reference system so that different studies are more comparable and harmonized; (ii) establishing consensual rules for defining system boundaries, and (ii) clearly defining not only the functional unit, but also the function of the system. In terms of Life Cycle Inventory, the availability of inventory flows for nations and fisheries in the developing world are still substantially lower than for the developed world, especially Europe, where most seafood LCA studies have been conducted. In fact, most studies linked to developing countries have covered the extraction of fish resources that are ultimately exported to developed markets. Moreover, while certain databases have provided an upgrade of their inventory flows in recent years, some if which include fisheries and seafood products (e.g., Ecoinvent), there is still a lag between seafood-based products and other food commodities. From an LCIA perspective, the main challenges are linked to providing a more complete scope of environmental impacts and damages related to the marine environment. While other methodological challenges remain when assessing the impact of ocean-related activities, including fisheries, the addition of impact categories linked to plastic debris in the ocean and fisheries depletion would potentially enhance the utility and visibility of LCA within fisheries management authorities and scientists.

The main outlook of this review is that a life cycle approach is essential for understanding the nexus along the whole supply chains. The water-energy-food nexus approach appears to be crucial to monitoring of the SDGs, since it considers intersectoral synergies and complementarities that will be crucial to improve the sustainability of the fish and seafood processing sector within the CE framework and according to the 2030 EU agenda. Recommended future work should, therefore, include the development of guidelines adapted to the seafood sector, as well as additional empirical case studies that quantify in addition to the environmental impacts, social and economic impacts caused by the new challenges of the circular economy and the bio-economy. For the former, the release of the PEFCR for the fishing sector, which is expected in upcoming years, may shed light on a more harmonized way of conducting LCA studies for the sector. The development of nexus guidelines for the sector would also allow an enhanced interconnection of seafood LCA with other pillars of sustainability. For the latter, further integration of LCA with a wide array of economic, social, nutritional methods, or its integration with machine learning models may enhance the utility of the tool in the future. In this sense, we argue that although LCA must continue to expand its holistic perspective by updating and upgrading its inventory databases and impact assessment methods, it cannot cover the entire set of sustainability indicators, especially beyond those that are purely environmental, that fisheries managers seek to respond. Consequently, beyond a PEFCR for the sector, higher levels of harmonization between LCA and other management support tools must be undertaken to foster the utility of life cycle methodologies in the sector. The aforementioned may be also informed by carrying out risk assessment modelling, which together with LCA, will help future proof the seafood sector for sustainable development along with mitigating against uncertainties.

Accurate assessment by means of LCA based tools represents an excellent opportunity to contribute to the economic, social and environmental development of the seafood sector, but also implies a high responsibility that needs to be articulated through tangible mid and long-term actions. LCA can jointly address a global concern and interest in terms of policies and strategies aimed at climate change mitigation, energy and food security, marine debris or treatment of wastewater. To address the challenges posed by these objectives, sustainable and multilateral research cooperation is needed to define integrated methodologies and strategies, such as the water-energy-food nexus, a valuable tool to minimize environmental negativities and get successful synergies. Furthermore, the added methodological challenge is to integrate environmental, social and economic variables that meet national needs through transnational strategies. The establishment of synergies in knowledge and experiences and challenges at the local level will help overcome challenges at a global level.

#### **Declaration of competing interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgement

This work was supported by the EAPA\_576/2018 NEPTUNUS project. The authors would like to acknowledge the financial support of Interreg Atlantic Area.

Ana Cláudia Dias and Paula Quinteiro acknowledge FCT/MCTES for the financial support to CESAM (UIDB/50017/2020 + UIDP/50017/ 2020), through national funds, and to the research contracts CEECIND/ 02174/2017 and CEECIND/00143/2017, respectively.

#### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.scitotenv.2020.144094.

#### References

- Abdou, K., Gascuel, D., Aubin, J., Romdhane, M.S., Lasram, F.B.R., Le Loc'h, F., 2018. Environmental life cycle assessment of seafood production: a case study of trawler catches in Tunisia. Sci. Total Environ. 610, 298–307.
- Abdou, K., Le Loc'h, F., Gascuel, D., Romdhane, M.S., Aubin, J., Lasram, F.B.R., 2020. Combining ecosystem indicators and life cycle assessment for environmental assessment of demersal trawling in Tunisia. Int. J. Life Cycle Assess. 25 (1), 105–119.
- Acero, A. P., Rodriguez, C., Ciroth, A. Impact Assessment Methods in Life Cycle Assessment and Their Impact Categories, LCIA Methods, 2017.
- Allan, D., Allan, R.P., 2019. Seasonal changes in the North Atlantic cold anomaly: the influence of cold surface waters from coastal Greenland and warming trends associated with variations in subarctic sea ice cover. Journal of Geophysical Research: Oceans 124, 9040–9052.
- Almeida, C., Vaz, S., Cabral, H., Ziegler, F., 2014. Environmental assessment of sardine (Sardina pilchardus) purse seine fishery in Portugal with LCA methodology including biological impact categories. Int. J. Life Cycle Assess. 19, 297–306.
- Almeida, C., Vaz, S., Ziegler, F., 2015. Environmental life cycle assessment of a canned sardine product from Portugal. J. Ind. Ecol. 19 (4), 607–617.
- Avadí, A., Fréon, P., 2013. Life cycle assessment of fisheries: a review for fisheries scientist and managers. Fish. Res. 143, 21–38.
- Avadí, A., Fréon, P., 2015. A set of sustainability performance indicators for seafood: direct human consumption products from Peruvian anchoveta fisheries and freshwater aquaculture. Ecol. Indic. 48, 518–532.
- Avadí, A., Fréon, P., Quispe, I., 2014a. Environmental assessment of Peruvian anchoveta food products: is less refined better? Int. J. Life Cycle Assess. 19, 1276–1293.
- Avadí, A., Vázquez-Rowe, I., Fréon, P., 2014b. Eco-efficiency assessment of the Peruvian anchoveta steel and wooden fleets using the LCA+DEA framework. J. Clean. Prod. 70, 118–131.
- Avadí, A., Bolaños, C., Sandoval, I., Ycaza, C., 2015. Life cycle assessment of Ecuadorian processed tuna. Int. J. Life Cycle Assess. 20, 1415–1428.
- Avadí, A., Adrien, R., Aramayo, V., Fréon, P., 2018. Environmental assessment of the Peruvian industrial hake fishery with LCA. Int. J. Life Cycle Assess. 23 (5), 1126–1140. Avadí, A., Vázquez-Rowe, I., Symeonidis, A., Moreno-Ruiz, E., 2020. First series of seafood
- datasets in ecoinvent: setting the pace for future development. Int. J. Life Cycle Assess. 25 (7), 1333–1342.
- Ayer, N.W., Tyedmers, P.H., Pelletier, N.L., Sonesson, U., Scholz, A., 2007. Co-product allocation in life cycle assessments of seafood production systems: review of problems and strategies. Int. J. Life Cycle Assess. 12 (7), 480.
- Azapagic, A., Clift, R., 1999. Allocation of environmental burdens in co-product systems: product-related burdens (part 1). Int. J. Life Cycle Assess. 4 (6), 357.
- Barr, W.J., Landis, A.E., 2018. Comparative life cycle assessment of a commercial algal multiproduct biorefinery and wild caught fishery for small pelagic fish. Int. J. Life Cycle Assess. 23 (5), 1141–1150.
- Barros, M.C., Magán, A., Valiño, S., Bello, P.M., Casares, J.J., Blanco, J.M., 2009. Identification of best available techniques in the seafood industry: a case study. J. Clean. Prod. 17 (3), 391–399.

- Bello Bugallo, P.M., Cristóbal, Andrade, L., Magán Iglesias, A., Torres López, R., 2013. Integrated environmental permit through best available techniques: evaluation of the fish and seafood canning industry. J. Clean. Prod. 47, 253-264.
- Bohnes, F.A., Laurent, A., 2019. LCA of aquaculture systems: methodological issues and potential improvements. Int. J. Life Cycle Assess. 24 (2), 324–337.
- Bohnes, F.A., Hauschild, M.Z., Schlundt, J., Laurent, A., 2019. Life cycle assessments of aquaculture systems: a critical review of reported findings with recommendations for policy and system development. Rev. Aquac. 11 (4), 1061–1079.
- de la Caba, K., Guerrero, P., Trung, T.S., Cruz-Romero, M., Kerry, J.P., Fluhr, J., Maurer, M., Kruijssen, F., Albalat, A., Bunting, S., Burt, S., Little, D., Newton, R., 2019. From seafood waste to active seafood packaging: an emerging opportunity of the circular economy. J. Clean. Prod. 208, 86–98.
- Castañé, S., Antón, A., 2017. Assessment of the nutritional quality and environmental impact of two food diets: a Mediterranean and a vegan diet. J. Clean. Prod. 167, 929–937.
- Cheung, W.W.L, Reygondeau, G., Frölicher, T.L., 2016. Large benefits to marine fisheries of meeting the 1.5°C global warming target. Science 354, 1591–1594.
- De Laurentiis, V., Hunt, D., Rogers, C., 2016. Overcoming food security challenges within an energy/water/food Nexus (EWFN) approach. Sustainability 8, 95.
- Denham, F.C., Biswas, W.K., Solah, V.A., Howieson, J.R., 2016. Greenhouse gas emissions from a Western Australian finfish supply chain. J. Clean. Prod. 112, 2079–2087.
- D'Odorico, P., Davis, K.F., Rosa, L., Carr, J.A., Chiarelli, D., Dell'Angelo, J., Gephart, J., MacDonald, G.K., Seekell, D.A., Suweis, S., Cristina Rulli, M.C., 2018. The global foodenergy-water Nexus Paolo. AGU100. Rev. Geophys. 56, 456–531.
- Drewnowski, A., 2009. Defining nutrient density: development and validation of the nutrient rich foods index. J. Am. Coll. Nutr. 28 (4), 4215–426S.
- Dreyer, L, Hauschild, M., Schierbeck, J., 2006. A framework for social life cycle impact assessment. Int. J. Life Cycle Ass. 11 (2), 88–97.
- Driscoll, J., Tyedmers, P., 2010. Fuel use and greenhouse gas emission implications of fisheries management: the case of the new England Atlantic herring fishery. Mar. Policy 34, 353–359.
- Driscoll, J., Boyd, C., Tyedmers, P., 2015. Life cycle assessment of the Maine and Southwest Nova Scotia lobster industries. Fish. Res. 172, 385–400.
- Emanuelsson, A., Ziegler, F., Pihl, L., Sköld, M., Sonesson, U., 2014. Accounting for overfishing in life cycle assessment: new impact categories for biotic resource use. Int. J. Life Cycle Assess. 19 (5), 1156–1168.
- Endo, A., Tsurita, I., Burnett, K., Orencio, P.M., 2017. A review of the current state of research on the water, energy, and food nexus. Journal of Hydrology: Regional Studies 11, 20–30.
- EOCD, 2020. Organization for Economic co-operation and Development. Accessed on 30th August, 2020. https://www.oecd.org/environment/.
- European Commission, 2011. ILCD Handbook: Recommendations for Life Cycle Impact Assessment in the European Context.
- European Commission, 2013. Communication from the Commission to the European Parliament and the Council: building the single market for green products. Facilitating better information on the environmental performance of products and organisations, COM/2013/0196. Accessed on 20th June 2020. https://eur-lex.europa.eu/LexUriServ/ LexUriServ.do?uri=COM:2013:0196:FIN:EN:PDF.
- European Commission, 2018a. Communication from the Commission to the Council and the European Parliament: delegation of powers referred to in Article 11(2), Article 15(2), (3), (6), (7) and Article 45(4) of Regulation (EU) No 1380/2013 of the European Parliament and of the Council of 11 December 2013 on the Common Fisheries Policy. COM/2018/0079. Accessed on 20th June, 2020. https://eur-lex.europa.
- European Commission, 2018b. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: a European Strategy for Plastics in a Circular Economy. COM/ 2018/0016. Accessed on 20th June, 2020. https://ec.europa.eu/environment/circular-economy/pdf/plastics-strategy.pdf.
- European Commission, 2019a. The EU Blue Economy Report. Accessed on 20th June, 2020. https://prod5.assets-cdn.io/event/3769/assets/8442090163-fc038d4d6f.pdf.
- European Commission, 2019b. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: on the implementation of the circular economy action plan, COM/2019/0090. Accessed on 20th June, 2020. https://ec.europa.eu/commission/ sites/beta-political/files/report\_implementation\_circular\_economy\_action\_plan.pdf.
- European Commission, 2020a. The EU Blue Economy Report. Accessed on 30th June, 2020. https://ec.europa.eu/maritimeaffairs/sites/maritimeaffairs/files/2020\_06\_ blueeconomy-2020-ld\_final.pdf.
- European Commission, 2020b. Facts and figures on the common fisheries policy. Basic statistical data. Accessed on 30th June, 2020. https://ec.europa.eu/fisheries/sites/fisheries/files/docs/body/pcp\_en.pdf.
- European Commission, 2020c. Product environmental footprint category rules pilot fiche. Marine fish (discontinued). Accessed on 20th June 2020. https://ec.europa.eu/environment/eussd/smgp/pdf/Fiche\_fish.pdf.
- Fadhil, A.B., Ahmed, A.I., Salih, H.A., 2017. Production of liquid fuels and activated carbons from fish waste. Fuel 187, 435–445.
- FAO, 2011. Global Food Losses and Food Waste Extent, Causes and Prevention. Food and Agriculture Organisation of the United Nations, Rome.
- FAO, 2015. Major Fishing Areas for Statistical Purposes. Accessed on 20th June, 2020. http://www.fao.org/3/a-az126e.pdf
- FAO, 2020. Food and Agricultural Organization of the United Nations. The State of world fisheries and aquaculture. Accessed on 20th June, 2020. http://www.fao.org/3/ ca9229en/CA9229EN.pdf
- Farmery, A., Gardner, C., Green, B.S., Jennings, S., Watson, R., 2015. Life cycle assessment of wild capture prawns: expanding sustainability considerations in the Australian Northern Prawn Fishery. J. Clean. Prod. 87, 96–104.

- Fréon, P., Bouchon, M., Domalain, G., Estrella, C., Iriarte, F., Lazard, J., Legendre, M., Quispe, I., Mendo, T., Moreau, Y., Nuñez, J., Sueiro, J.C., Tam, J., Tyedmers, P., Voisin, S., 2010. Impacts of the Peruvian Anchoveta Supply Chains: From Wild Fish in the Water to Protein on the Plate (Globec International Newsletter).
- Fréon, P., Sueiro, J.C., Iriarte, F., Evar, O.F.M., Landa, Y., Mittaine, J.F., Bouchon, M., 2014a. Harvesting for food versus feed: a review of Peruvian fisheries in a global context. Rev. Fish Biol. Fish. 24 (1), 381–398.
- Fréon, P., Avadí, A., Vinatea-Chavez, A., Iriarte, F., 2014b. Life cycle assessment of the Peruvian industrial anchoveta fleet: boundary setting in life cycle inventory analyses of complex and plural means of production. Int. J. Life Cycle Assess. 19, 1068–1086.
- Fréon, P., Avadí, A., Marin-Soto, W., Negrón, R., 2014c. Environmentally extended comparison table of large-versus small- and medium- scale fisheries: the case of the Peruvian anchoveta fleet. Can. J. Fish. Aquat. Sci. 71 (10), 1459–1474.
- Fréon, P., Durand, H., Avadí, A., Huaranca, S., Orozco Moreyra, R., 2017. Life cycle assessment of three Peruvian fishmeal plants: toward a cleaner production. J. Clean. Prod. 145, 50–63.
- Furuya, A., Fukami, M., Ellingsen, H., Kagaya, S., 2011. A survey on energy consumption in fisheries, and measures to reduce CO2 emissions. 51st Congress of the European Regional Science Association: "New Challenges for European Regions and Urban Areas in a Globalised World", 30 August 3 September 2011. Spain, Barcelona.
- García-Santiago, X., Franco-Uría, A., Antelo, L.T., Vázquez, J.A., Pérez-Martín, R., Moreira, M.T., Feijoo, G., 2020. Eco-efficiency of a marine biorefinery for valorization of cartilaginous fish biomass. J. Ind. Ecol. https://doi.org/10.1111/jiec.13066.
- Garrett, A., Pinnegar, J., Dye, S., 2018. A Seafish/MCCIP Watching brief report 2017, SeaFish, pp15. Accessed on 30<sup>th</sup> June, 2020. https://www.seafish.org/media/1800222/seafish\_ climate\_change\_adaptation\_review\_digital\_version.pdf
- Geng, Y., Chen, W., Liu, Z., Chiu, A.S., Han, W., Liu, Z., Zhong, S., Qian, Y., You, W., Cui, X., 2017. A bibliometric review: energy consumption and greenhouse gas emissions in the residential sector. J. Clean. Prod. 159, 301–316.
- Gephart, J.A., Deutsch, L., Pace, M.L., Troell, M., Seekell, D.A., 2017. Shocks to fish production: identification, trends, and consequences. Glob. Environ. Chang. 42, 24–32.
- Ghodsvali, M., Krishnamurthy, S., de Vries, B., 2019. Review of transdisciplinary approaches to food-water-energy nexus: a guide towards sustainable development. Environ. Sci. Pol. 101, 266–278.
- Gonzalez-García, S., Villanueva-Rey, P., Belo, S., Vázquez-Rowe, I., Moreira, M.T., Feijoo, G., Arroja, L., 2015. Cros-vessel eco-efficiency analysis. A case study for purse seining fishing from North Portugal targeting European pilchard. International Journal of Life Cycle Assessment 20, 1019–1032.
- Grimaldo, E., Herrmann, B., Su, B., Føre, H.M., Vollstad, J., Olsen, L., Larsen, R.B., Tatone, I., 2019. Comparison of fishing efficiency between biodegradable gillnets and conventional nylon gillnets. Fish. Res. 213, 67–74.
- Guillen, J., Cheilari, A., Damalas, D., Barbas, T., 2016. Oil for fish: an energy return on in-
- vestment analysis of selected European Union fishing fleets. J. Ind. Ecol. 20, 145–153. Guinée, J., 2001. Handbook on life cycle assessment–operational guide to the ISO standards. Int. J. Life Cycle Assess. 6 (5), 255.
- Guinée, J., Heijungs, R., Huppes, G., Zamagni, A., Masoni, P., Buonamici, R., Ekvall, T., Rydberg, T., 2011. Life cycle assessment: past, present, and future. Environ. Sci. Technol. 45 (1), 90–96.
- Hallstrom, E., Bergman, K., Mifflin, K., Parker, R., Tyedmers, P., Troell, M., Ziegler, F., 2019. Combined climate and nutritional performance of seafood. J. Clean. Prod. 230, 402–411.
- Hamiche, A.M., Stambouli, A.B., Flazi, S., 2016. A review of the water-energy nexus. Renew. Sust. Energ. Rev. 65, 319–331.
- Hauschild, M.Z., Rosenbaum, R.K., Olsen, S., 2018. Life cycle assessment. Springer.
- Hayes, M., Gallagher, M., 2019. Processing and recovery of valuable components from pelagic blood-water waste streams: a review and recommendations. J. Clean. Prod. 215, 410–422.
- Hélias, A., Heijungs, R., 2019. Resource depletion potentials from bottom-up models: population dynamics and the Hubbert peak theory. Sci. Total Environ. 650, 1303–1308.
- Hélias, A., Langlois, J., Fréon, P., 2018. Fisheries in life cycle assessment: operational factors for biotic resource depletion. Fish Fish. 19, 951–963.
- Henriksson, P.J., Guinée, J.B., Kleijn, R., de Snoo, G.R., 2012. Life cycle assessment of aquaculture systems—a review of methodologies. Int. J. Life Cycle Assess. 17 (3), 304–313.
- Henriksson, P.J., Belton, B., Murshed-e-Jahan, K., Rico, A., 2018. Measuring the potential for sustainable intensification of aquaculture in Bangladesh using life cycle assessment. Proc. Natl. Acad. Sci. 115 (12), 2958–2963.
- Hospido, A., Tyedmers, P., 2005. Life cycle environmental impacts of Spanish tuna fisheries. Fish. Res. 76, 174–186.
- Hospido, A., Vazquez, M.E., Cuevas, A., Feijoo, G., Moreira, M.T., 2006. Environmental assessment of canned tuna manufacture with a life-cycle perspective. Resour. Conserv. Recycl. 47 (1), 56–72.
- Huijbregts, M.A., Steinmann, Z.J., Elshout, P.M., Stam, G., Verones, F., Vieira, M., Zijp, M., Hollander, A., van Zelm, R., 2017. ReCiPe2016: a harmonised life cycle impact assessment method at midpoint and endpoint level. Int. J. Life Cycle Assess. 22 (2), 138–147.
- Huijungs, R., Guinée, J.B., 2007. Allocation and 'what-if' scenarios in life cycle assessment of waste management systems. Waste Manag. 27, 997–1005.
- International Organization for Standardization: ISO 14040 Environmental management: life cycle assessment; principles and framework. 2006a. Accessed on 20th June 2020. http://www.cscses.com/uploads/2016328/20160328110518251825.pdf.
- International Organization for Standardization: ISO 14044 environmental management: life cycle assessment: requirements and guidelines. 2006b. Accessed on 20th June 2020. https://www.iso.org/obp/ui/#iso:std:iso:14044:ed-1:v1:en.
- Iribarren, D., Moreira, M.T., Feijoo, G., 2010a. Life Cycle Assessment of fresh and canned mussel processing and consumption in Galicia (NW Spain). Resour. Conserv. Recycl. 55 (2), 106–117.

- Iribarren, D., Moreira, M.T., Feijoo, G., 2010b. Implementing by-product management into the life cycle assessment of the mussel sector. Resour. Conserv. Recycl. 54 (12), 1219–1230.
- Iribarren, D., Moreira, M.T., Feijoo, G., 2010c. Revisiting the life cycle assessment of mussels from a sectorial perspective. J. Clean. Prod. 18, 101–111.
- Iribarren, D., Hospido, A., Moreira, M.T., Feijoo, G., 2010d. Carbon footprint of canned mussels from a business-to consumer approach. A starting point for mussel processors and policy makers. Environmental Science & Policy 13, 509–521.
- Iribarren, D., Vázquez-Rowe, I., Hospido, A., Moreira, M.T., Feijoo, G., 2010e. Estimation of the carbon footprint of the Galician fishing activity (NW Spain). Sci. Total Environ. 408, 5284–5294.
- Iribarren, D., Vázquez-Rowe, I., Hospido, A., Moreira, M.T., Feijoo, G., 2011. Updating the carbon footprint of the Galician fishing activity (NW Spain). Sci. Total Environ. 409, 1609–1611.
- Jambeck, J.R., Geyer, R., Wilcox, C., Siegler, T.R., Perryman, M., Andrady, A., Narayan, R., Law, K.L., 2015. Plastic waste inputs from land into the ocean. Science 347 (6223), 768–771.
- Järviö, N., Henriksson, P.J., Guinée, J.B., 2018. Including GHG emissions from mangrove forests LULUC in LCA: a case study on shrimp farming in the Mekong Delta, Vietnam. Int. J. Life Cycle Assess. 23 (5), 1078–1090.
- Johnson, D., Barrio Froján, C., Bax, N., Dunstan, P., Woolley, S., Halpin, P., Dunn, D., Hazin, C., Dias, M., Davies, T., Jiménez, J., Ross, E., Van Dover, C., Notarbartolo Di Sciara, G., Hoyt, E., Tetley, M.J., Gunn, V., Von Nordheim, H., 2019. The global ocean biodiversity initiative: promoting scientific support for global ocean governance. Aquat. Conserv. Mar. Freshwat. Ecosyst. 29, 162–169.
- Jørgensen, A., Le Bocq, A., Nazarkina, L., Hauschild, M., 2008. Methodologies for social life cycle assessment. Int. J. Life Cycle Ass. 13 (2), 96–103.
- Kirchherr, J., Reike, D., Hekkert, M., 2017. Conceptualizing the circular economy: an analysis of 114 definitions. Resour. Conserv. Recycl. 127, 221–232.
- Kruse, S., Flysjö, A., Kasperczyk, N., Scholz, A.J., 2008. Socioeconomic indicators as a complement to life cycle assessment – an application to salmon production systems. Int. J. Life Cycle Ass. 14 (1), 8–18.
- Larrea-Gallegos, G., Vázquez-Rowe, I., 2020. Optimization of the environmental performance of food diets in Peru combining linear programming and life cycle methods. Sci. Total Environ. 699, 134231.
- Larsen, R., Eilertsen, K.E., Elvevoll, E.O., 2011. Health benefits of marine foods and ingredients. Biotechnol. Adv. 29 (5), 508–518.
- Laso, J., Margallo, M., Celaya, J., Fullana, P., Bala, A., Gazulla, C., Irabien, A., Aldaco, R., 2016. Waste management under a life cycle approach as a tool for a circular economy in the canned anchovy industry. Waste Manag. Res. 34 (8), 724–733.
- Laso, J., Margallo, M., Fullana, P., Bala, A., Gazulla, C., Irabien, A., Aldaco, R., 2017a. Introducing life cycle thinking to define best available techniques for products: application to the anchovy canning industry. J. Clean. Prod. 155, 139–150.
- Laso, J., Margallo, M., Fullana, P., Bala, A., Gazulla, C., Irabien, Á., Aldaco, R., 2017b. When product diversification influences life cycle impact assessment: a case study of canned anchovy. Sci. Total Environ. 581, 629–639.
- Laso, J., Margallo, M., Serrano, M., Vázquez-Rowe, I., Avadí, A., Fullana, P., Bala, A., Gazulla, C., Irabien, A., Aldaco, R., 2018a. Introducing the Green protein footprint method as an understandable measure of the environmental cost of anchovy consumption. Sci. Total Environ. 621, 40–53.
- Laso, J., García-Herrero, I., Margallo, M., Vázquez-Rowe, I., Fullana, P., Bala, A., Gazulla, C., Irabien, Á., Aldaco, R., 2018b. Finding an economic and environmental balance in value chains based on circular economy thinking: an eco-efficiency methodology applied to the fish canning industry. Resour. Conserv. Recycl. 133, 428–437.
- Laso, J., Margallo, M., García-Herrero, I., Fullana, P., Bala, A., Gazulla, C., Polettini, A., Kahhat, R., Vázquez-Rowe, I., Irabien, A., Aldaco, R., 2018c. Combined application of life cycle assessment and linear programming to evaluate food waste-to-food strategies: seeking for answers in the nexus approach. Waste Manag. 80, 186–197.
- LCI, 2020. GLAM. Life Cycle Initiative. Retrieved from:. https://www.lifecycleinitiative.org/ category/glam/.
- Lebreton, L., Slat, B., Ferrari, F., Sainte-Rose, B., Aitken, J., Marthouse, R., Hajbane, S., Cunsolo, S., Schwarz, A., Levivier, A., Noble, K., Debeljak, P., Maral, H., Schoeneich-Argent, R., Brambini, R., Reisser, J., 2018. Evidence that the Great Pacific Garbage Patch is rapidly accumulating plastic. Sci. Rep. 8 (1), 1–15.
- Lourguioui, H., Brigolin, D., Boulahdid, M., Pastres, R., 2017. A perspective for reducing environmental impacts of mussel culture in Algeria. Int. J. Life Cycle Assess. 22 (8), 1266–1277.
- Love, D.C., Fry, J.P., Milli, M.C., Neff, R.A., 2015. Wasted seafood in the United States: quantifying loss from production to consumption and moving toward solutions. Glob. Environ. Chang. 35, 116–124.
- Lozano, S., Iribarren, D., Moreira, M.T., Feijoo, G., 2010. Environmental impact efficiency in mussel cultivation. Resour. Conserv. Recycl. 54, 1269–1277.
- Machell, J., Prior, K., Allan, R., Andresen, J.M., 2015. The water energy food nexus chalenges and emerging solutions. Environ. Sci.: Water Res. Technol. 1, 15–16.
- Mannan, M., Al-Ansari, T., Mackey, H.R., Al-Ghamdi, S.G., 2018. Quantifying the energy, water and food nexus: a review of the latest developments based on life-cycle assessment. J. Clean. Prod. 193, 300–314.
- Margallo, M., Aldaco, R., Irabien, A., Carrillo, V., Fischer, M., Bala, A., Fullana, P., 2014. Life cycle assessment modelling of waste-to-energy incineration in Spain and Portugal. Waste Manag. Res. 32 (6), 492–499.
- MarilCA, 2020. Marine Impacts in Life Cycle Assessment initiative. Accessed on 20th August, 2020. https://marilca.org/
- Maritime Labour Convention, 2006. Accessed on 30th June, 2020. https://www.ilo.org/ wcmsp5/groups/public/-ed\_norm/-normes/documents/normativeinstrument/ wcms\_090250.pdf.

Maximenko, N., Corradi, P., Law, K.L., Van Sebille, E., Garaba, S.P., Lampitt, R.S., Galgani, F., Martinez-Vicente, V., Goddijn-Murphy, L., Veiga, J.M., Thompson, R.C., Maes, C., Moller, D., Löscher, C.R., Addamo, A.M., Lamson, M.R., Centurioni, L.R., Posth, N.R., Lumpkin, R., Vinci, M., Martins, A.M., Pieper, C.D., Isobe, A., Hanke, G., Edwards, M., Chubarenko, I.P., Rodriguez, E., Aliani, S., Arias, M., Asner, G.P., Brosich, A., Carlton, J.T., Chao, Y., Cook, A.M., Cundy, A.B., Galloway, T.S., Giorgetti, A., Goni, G.J., Guichoux, Y., Haram, L.E., Hardesty, B.D., Holdsworth, N., Lebreton, L., Leslie, H.A., Macadam-Somer, I., Mace, T., Manuel, M., Marsch, R., Martinez, E., Mayor, D.J., Le Moigne, M., Molina Jack, M.E., Mowlem, M.C., Obbard, R.W., Pabortsava, K., Robberson, B., Rotaru, A.E., Ruiz, G.M., Spedicato, M.T., Thiel, M., Turra, A., Wilcox, C., 2019. Towards the integrated marine debris observing system. Front. Mar. Sci. 6, 447.

Millennium Ecosystem Assessment, 2005. Ecosystems and Human Well-Being: Health Synthesis. WHO. France.

- Molina-Besch, K., 2016. Prioritization guidelines for green food packaging development. Br. Food J., 118(10), 2512-253.
- Morris, J.P., Backeljau, T., Chapelle, G., 2019. Shells from aquaculture: a valuable biomaterial, not a nuisance waste product. Rev. Aquac. 11 (1), 42–57.
- Neptunus, 2020. Accessed on 30th June, 2020. https://neptunus-project.eu/
- NOAA, National Oceanic and Atmospheric Administration (2019). Global Climate report. Annual 2019. Accessed on 30<sup>th</sup> June, 2020. https://www.ncdc.noaa.gov/sotc/global/ 201913
- Pabortsava, K., Lampitt, R.S., 2020. High concentrations of plastic hidden beneath the surface of the Atlantic Ocean. Nat. Commun. 11 (1), 1–11.
- Parker, R.W.R., Tyedmers, P.H., 2012a. Life cycle environmental impacts of three products derived from wild-caught Antarctic krill (Euphausia superba). Environmental Science & Technology 46, 4958–4965.
- Parker, R.W.R., Tyedmers, P.H., 2012b. Uncertainty and natural variability in the ecological footprint of fisheries: a case study of reduction fisheries for meal and oil. Ecol. Indic. 16, 76–83.
- Parker, R.W., Tyedmers, P.H., 2015. Fuel consumption of global fishing fleets: current understanding and knowledge gaps. Fish Fish. 16 (4), 684–696.
- Parker, R.W., Vázquez-Rowe, I., Tyedmers, P.H., 2015. Fuel performance and carbon footprint of the global purse seine tuna fleet. J. Clean. Prod. 103, 517–524.
- Parker, R.W., Blanchard, J.L., Gardner, C., Green, B.S., Hartmann, K., Tyedmers, P.H., Watson, R.A., 2018. Fuel use and greenhouse gas emissions of world fisheries. Nat. Clim. Chang. 8 (4), 333–337.
- Patouillard, L., Bulle, C., Querleu, C., Maxime, D., Osset, P., Margni, M., 2018. Critical review and practical recommendations to integrate the spatial dimension into life cycle assessment. J. Clean. Prod. 177, 398–412.
- Peck, M., Pinnegar, J.K., 2018. In: Barange, M., Bahri, T., Beveridge, M.C.M., Cochrane, K.L., Funge-Smith, S., Poulan, F. (Eds.), Climate change impacts, vulnerabilities and adaptations: North Atlantic and Atlantic Arctic marine fisheries. Impacts of climate change on fisheries and aquaculture, FAO, pp. 5–87.
- Pelletier, N., Tyedmers, P., 2011. An ecological economic critique of the use of market information in life cycle assessment research. J. Ind. Ecol. 15 (3), 342–354.
- Pelletier, N.L., Ayer, N.W., Tyedmers, P.H., Kruse, S.A., Flysjö, A., Robillard, G., Ziegler, F., Scholz, A.J., Sonesson, U., 2007. Impact categories for life cycle assessment research of seafood production systems: review and prospectus. Int. J. Life Cycle Ass. 8 (1), 39–421.
- Pesca de Galicia, 2020. Rexistro de Buques Pesqueiros da Comunidade Autónoma de Galicia - Datos a 11/11/2020. Retrieved from: https://www.pescadegalicia.gal/ rexbuque/. Last accessed: November 11th 2020 (in Galician).
- Pfister, S., Boulay, A.M., Berger, M., Hadjikakou, M., Motoshita, M., Hess, T., Ridoutt, B., Weinzettel, J., Scherer, L., Döll, P., Manzardo, A., Núñez, M., Verones, F., Humbert, S., Buxmann, K., Hardling, K., Benini, L., Oki, T., Fikbeiner, M., Henderson, A., 2017. Understanding the LCA and ISO water footprint: a response to Hoekstra (2016) "a critique on the water-scarcity weighted water footprint in LCA". Ecol. Indic. 72, 352–359.
- Philis, G., Ziegler, F., Gansel, L.C., Jansen, M.D., Gracey, E.O., Stene, A., 2019. Comparing life cycle assessment (LCA) of salmonid aquaculture production systems: status and perspectives. Sustainability 11 (9), 2517.
- Pinnegar, J.K., Garrett, A., Simpson, S.D., Engelhard, G.H., van der Kooij, J., 2017. Fisheries. MCCIP Science Review 2017, 73–99.
- Poore, J., Nemecek, T., 2018. Reducing food's environmental impacts through producers and consumers. Science 360 (6392), 987–992.
- Provencher, J.F., Ammendolia, J., Rochman, C.M., Mallory, M.L., 2019. Assessing plastic debris in aquatic food webs: what we know and don't know about uptake and trophic transfer. Environ. Rev. 27 (3), 304–317.
- Quantis, 2020. Plastic Leak Project. Methodological guidelines. Accessed on 31<sup>st</sup> August, 2020. https://quantis-intl.com/report/the-plastic-leak-project-guidelines/
- Rahmstorf, S., Box, J.E., Feulner, G., Mann, M.E., Robinson, A., Rutherford, S., Schaffernicht, E.J., 2015. Exceptional twentieth-century slowdown in Atlantic Ocean overturning circulation. Nat. Clim. Chang. 5 (5), 475–480.
- Ramos, S., Vázquez-Rowe, I., Artetxe, I., Moreira, M.T., Feijoo, G., Zufía, J., 2011. Environmental assessment of the Atlantic mackerel (Scomber scombrus) season in the Basque Country. Increasing the timeline delimitation in fishery LCA studies. Int. J. Life Cycle Assess. 16 (7), 599–610.
- Rayner, N.A., Parker, D.E., Horton, E.B., Folland, C.K., Alexander, L.V., Rowell, D.P., Kent, E.C., Kaplan, A., 2003. Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century. J. Geophys. Res. 108, D14.
- Rosenbaum, R.K., Bachmann, T.M., Gold, L.S., Huijbregts, M.A., Jolliet, O., Juraske, R., Koehler, A., Lrsen, H.F., MacLeod, M., Margni, M., McKone, T.E., Payet, J., Schuhmacher, M., van de Meent, D., Hauschild, M.Z., 2008. USEtox—the UNEP-SETAC toxicity model: recommended characterisation factors for human toxicity and freshwater ecotoxicity in life cycle impact assessment. Int. J. Life Cycle Assess. 13 (7), 532.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.141362.

- Ruiz-Salmón, I., Margallo, M., Laso, J., Villanueva-Rey, P., Mariño, D., Quinteiro, P., Diaz, A.C., Nunes, M.L., Marques, A., Feijoo, G., Moreira, M.T., Loubet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N., Méndez-Paz, D., Iglesias-Parga, X., Anglada, C., Martin, J.C., Irabien, A., Aldaco, R., 2020. Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Current Opinion in Environmental Science & Health 13, 101–106.
- Saling, P., Gyuzeleva, L., Wittstock, K., Wessolowski, V., Griesshammer, R., 2020. Life cycle impact assessment of microplastics as one component of marine plastic debris. The International Journal of Life Cycle Assessment, 1-19.
- Salmoral, G., Yan, X., 2018. Food-energy-water nexus: a life cycle analysis on virtual water and embodied energy in food consumption in the Tamar catchment, UK. Resour. Conserv. Recycl. 133, 320–330.
- Soltanpour, Y., Peri, I., Temri, L., 2020. Co-management of fisheries through a negotiation lens: the case of prud'homies. Maritime Studies 19, 167–178.
- Sonneman, G., Valdivia, S., 2017. Medellin declaration on marine litter in life cycle assessment and management. Int. J. Life Cycle Assess. 22 (10), 1637–1639.
- Spanish Government. España Circular, 2020. Estrategia Española de Economía Circular. 2020. Accessed on 20th June, 2020. https://www.miteco.gob.es/es/calidad-yevaluacion-ambiental/temas/economia-circular/espanacircular2030\_def1\_tcm30-509532.PDF
- Staupe-Delgado, R., 2020. The water–energy–food–environmental security nexus: moving the debate forward. Environ. Dev. Sustain. 22, 6131–6147.
- Stefanini, R., Borghesi, G., Ronzano, A., Vignali, G., 2020. Plastic or glass: a new environmental assessment with a marine litter indicator for the comparison of pasteurized milk bottles. The International Journal of Life Cycle Assessment, 1-18.
- Steffen, W., Richardson, K., Rockström, J., Cornell, S.E., Fetzer, I., Bennett, E.M., Biggs, R., Carpenter, S.R., De Vries, W., De Wit, C.A., Folke, C., Gerten, D., Heinke, J., Mace, G.M., Persson, L.M., Ramanathan, V., Reyers, B., Sörlin, S., 2015. Planetary boundaries: guiding human development on a changing planet. Science 347, 1259855.
- Stylianou, K.S., Heller, M.C., Fulgoni, V.L., Ernstoff, A.S., Keoleian, G.A., Jolliet, O., 2016. A life cycle assessment framework combining nutritional and environmental health impacts of diet: a case study on milk. Int. J. Life Cycle Assess. 21 (5), 734–746.
- Sumaila, U.R., Ebrahim, N., Schuhbauer, A., Skerritt, D., Li, Y., Kim, H.S., Mallory, T.G., Lam, V.W.L., Pauly, D., 2019. Updated estimates and analysis of global fisheries subsidies. Mar. Policy 109, 103695.
- Sustainable Development Goals, 2020. Accessed on 20th June, 2020. https://sustainabledevelopment.un.org/?menu=1300
- Svanes, E., Vold, M., Hanssen, O.J., 2011. Environmental assessment of cod (Gadus morhua) from autoline fisheries. Int. J. Life Cycle Assess. 16, 611–624.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N., 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: an Irish case study. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2017.05.227.
- Thrane, M., 2004. Energy consumption in the Danish fishery. J. Ind. Ecol. 8, 223–239. Thrane, M., 2006, LCA of Danish Fish Products. New methods and insights (9 pp): The International Journal of Life Cycle Assessment, v. 11, p. 66-74.
- Thrane, M., Nielsen, E.H., Christensen, P., 2009a. Cleaner production in Danish fish processing–experiences, status and possible future strategies. J. Clean. Prod. 17 (3), 380–390.
- Thrane, M., Ziegler, F., Sonesson, U., 2009b. Eco-labelling of wild-caught seafood products. J. Clean. Prod. 17 (3), 416–423.
- UNEP, 2020. United Nations Environment Program. Accessed on 20th August, 2020. https://www.lifecycleinitiative.org/activities/key-programme-areas/life-cycleknowledge-consensus-and-platform/global-guidance-for-life-cycle-impact-assessment-indicators-and-methods-glam/.
- Valdivia, S., Ugaya, C.M., Hildenbrand, J., Traverso, M., Mazijn, B., Sonnemann, G., 2013. A UNEP/SETAC approach towards a life cycle sustainability assessment—our contribution to Rio+ 20. Int. J. Life Cycle Assess. 18 (9), 1673–1685.
- Van Putten, I.E., Farmery, A.K., Green, Bs, Hobday, A.J., Lim-Camacho, L., Norman-López, A., Parker, R.W., 2016. The environmental impact of two Australian rock lobster fishery supply chains under a changing climate. J. Ind. Ecol. 20 (6), 1384–1398.
- Vanham, D., 2016. Does the water footprint concept provide relevant information to address the water-food-energy-ecosystem nexus? Ecosystem Services. 17, 298–307.
- Vázquez-Rowe, I., Benetto, E., 2014. The use of a consequential perspective to upgrade the utility of Life Cycle Assessment for fishery managers and policy makers. Mar. Policy 48, 14–17.
- Vázquez-Rowe, I., Moreira, M.T., Feijoo, G., 2010a. Life cycle assessment of horse mackerel fisheries in Galicia (NW Spain): comparative analysis of two major fishing methods. Fish. Res. 106, 517–527.
- Vázquez-Rowe, I., Iribarren, D., Moreira, M.T., Feijoo, G., 2010b. Combined application of life cycle assessment and data envelopment analysis as a methodological approach for the assessment of fisheries. Int. J. Life Cycle Assess. 15, 272–283.
- Vázquez-Rowe, I., Moreira, M.T., Feijoo, G., 2011a. Life cycle assessment of fresh hake fillets captured by the Galician fleet in the northern stock. Fish. Res. 110, 128–135.
- Vázquez-Rowe, I., Iribarren, D., Hospido, A., Moreira, M.T., Feijoo, G., 2011b. Computation of operational end environmental benchmarks within selected Galician fishing fleets. J. Ind. Ecol. 15 (5), 776–795.
- Vázquez-Rowe, I., Moreira, M.T., Feijoo, G., 2012a. Environmental assessment of frozen common octopus (Octopus vulgaris) captured by Spanish fishing vessels in the Mauritanian EEZ. Mar. Policy 36, 180–188.
- Vázquez-Rowe, I., Hospido, A., Moreira, M.T., Feijoo, G., 2012b. Best practices in life cycle assessment implementation in fisheries. Improving and broadening environmental assessment for seafood production systems. Trends Food Sci. Technol. 28 (2), 116–131.
- Vázquez-Rowe, I., Moreira, M.T., Feijoo, G., 2012c. Inclusion of discard assessment indicators in fisheries life cycle assessment studies. Expanding the use of fishery-specific impact categories. Int. J. Life Cycle Assess. 17 (5), 535–549.

- Vázquez-Rowe, I., Villanueva-Rey, P., Mallo, J., De la Cerda, J.J., Moreira, M.T., Feijoo, G., 2013a. Carbon footprint of a multi-ingredient seafood product from a business-tobusiness perspective. J. Clean. Prod. 44, 200–210.
- Vázquez-Rowe, I., Villanueva-Rey, P., Moreira, M.T., Feijoo, G., 2013b. The role of consumer purchase and post-purchase decision-making in sustainable seafood consumption. A Spanish case study using carbon footprinting. Food Policy 41, 94–102.
- Vázquez-Rowe, I., Moreira, M.T., Feijoo, G., 2013c. Carbon footprint analysis of goose barnacle (Pollicipes pollicipes) collection on the Galician coast (NW Spain). Fish. Res. 143, 191–200.
- Vázquez-Rowe, I., Villanueva-Rey, P., Moreira, M.T., Feijoo, G., 2014a. Edible protein energy return on investment ratio (ep-EROI) for Spanish seafood products. Ambio 43 (3), 381–394.
- Vázquez-Rowe, I., Villanueva-Rey, P., Hospido, A., Moreira, M.T., Feijoo, G., 2014b. Life cycle assessment of European pilchard (Sardina pilchardus) consumption. A case study for Galicia (NW Spain). Sci. Total Environ. 475, 48–60.
- Vázquez-Rowe, I., Villanueva-Rey, P., Moreira, M.T., Feijoo, G., 2016. Opportunities and challenges of implementing life cycle assessment in seafood certification: a case study for Spain. Int. J. Life Cycle Assess. 21 (4), 451–464.
- Vázquez-Rowe, I., Laso, J., Margallo, M., Garcia-Herrero, I., Hoehn, D., Amo-Setién, F., Bala, A., Abajas, R., Sarabia, C., Durá, M.J., Fullana-i-Palmer, P., Aldaco, R., 2019. Food loss and waste metrics: a proposed nutritional cost footprint linking linear programming and life cycle assessment. The International Journal of Life Cycle Assessment, 1-13.
- Verones, F., Hellweg, S., Antón, A., Azevedo, L.B., Chaudhary, A., Cosme, N.M.D., Cucurachi, S., Baan, L., Dong, Y., Fantke, P., Golsteijn, L., Hauschild, M.Z., Heijungs, R., Jolliet, O., Juraske, R., Larsen, H., Laurent, A., Mutel, C.L., Margni, M., ... Huijbregts, M.A.J., 2020. LCIMPACT: a regionalized life cycle damage assessment method. J. Ind. Ecol. 24 (6), 1201–1219. https://doi.org/10.1111/jiec.13018.
- Villanueva-Rey, P., Vázquez-Rowe, I., Arias, A., Moreira, M.T., Feijoo, G., 2018. The importance of using life cycle assessment in policy support to determine the sustainability of fishing fleets: a case study for the small-scale xeito fishery in Galicia, Spain. Int. J. Life Cycle Assess. 23, 1091–1106.
- Weidema, B.P., Thrane, M., Christensen, P., Schmidt, J., Løkke, S., 2008. Carbon footprint: a catalyst for life cycle assessment? J. Ind. Ecol. 12 (1), 3–6.
- WHO. World Health Organization (2020). Accessed on 20th August, 2020. https://www. who.int/en/news-room/fact-sheets/detail/healthy-diet

- Winther, U., Ziegler, F., Hogner, E.S., Emanuelsson, A., Sund, V., Ellingsen, H., 2009. Carbon Footprint and Energy Use of Norwegian Seafood Products (Report SINTEF Fisheries and Aquaculture).
- Woods, J.S., Verones, F., 2019. Ecosystem damage from anthropogenic seabed disturbance: a life cycle impact assessment characterisation model. Sci. Total Environ. 649, 1481–1490.
- Woods, J.S., Veltman, K., Huijbregts, M.A., Verones, F., Hertwich, E.G., 2016. Towards a meaningful assessment of marine ecological impacts in life cycle assessment (LCA). Environ. Int. 89, 48–61.
- Woods, J.S., Rødder, G., Verones, F., 2019. An effect factor approach for quantifying the entanglement impact on marine species of macroplastic debris within life cycle impact assessment. Ecol. Indic. 99, 61–66.
- Woods, J.S., Verones, F., Jolliet, O., Boulay, A.M., Vázquez-Rowe, I. 2020 (under review). A framework for the assessment of marine litter impacts in life cycle impact assessment. Ecological Indicators, under review.
- Zhang, F., Chen, Y., Chen, Q., Feng, Y., Shang, Y., Yang, X., Gao, H., Tian, C., Li, J., Zhang, G., Matthias, V., Xie, Z., 2018. Real-world emission factors of gaseous and particulate pollutants from marine fishing boats and their total emissions in China. Environmental science & technology 52 (8), 4910–4919.
- Ziegler, F., Valentinsson, D., 2008. Environmental life cycle assessment of Norway lobster (Nephrops norvegicus) caught along the Swedish west coast by creels and conventional trawls – LCA methodology with case study. Int. J. Life Cycle Assess. 13, 487–497.
- Ziegler, F., Nilsson, P., Mattsson, B., Walther, Y., 2003. Life Cycle Assessment of Frozen Cod Fillets Including Fishery-Specific Environmental Impacts. International Journal of Life Cycle Assessment 8 (1), 39–47.
- Ziegler, F., Emanuelsson, A., Eichelsheim, J.L., Flysjö, A., Ndiaye, V., Thrane, M., 2011. Extended life cycle assessment of southern pink shrimp products originating in Senegalese artisanal and industrial fisheries for export to Europe. J. Ind. Ecol. 15 (4), 527–538.
- Ziegler, F., Winther, U., Hognes, E.S., Emanuelsson, A., Sund, V., Ellingsen, H., 2012. The carbon footprint of Norwegian seafood products on the global seafood market. Journal of the Industrial Ecology 17 (1), 103–116.
- Ziegler, F., Hornborg, S., Green, B.S., Eigaard, A.R., Farmery, A.K., Hammar, L., Smith A.D.M., Vázquez-Rowe, I., 2016. Expanding the concept of sustainable seafood using life cycle assessment. Fish and Fisheries 17(4), 1073-1093.

Contents lists available at ScienceDirect





# Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

# Seafood consumers engagement in reducing environmental impacts from packaging



Recycling rates for each packaging material

to be achieved by 2030

306 406

100

Cheila Almeida<sup>a,b,\*</sup>, Jara Laso<sup>c</sup>, David Baptista de Sousa<sup>d</sup>, Ronan Cooney<sup>e,f</sup>, Paula Quinteiro<sup>g</sup>, Neil Rowan<sup>h</sup>, Ana Cláudia Dias<sup>g</sup>, Eoghan Clifford<sup>e,f</sup>, Rodrigo G. Reboredo<sup>d</sup>, María Margallo<sup>c</sup>, Maria Leonor Nunes<sup>b</sup>, António Marques<sup>a,b</sup>

 <sup>a</sup> Instituto Português do Mar e da Atmosfera (IPMA), Divisão de Aquacultura, Valorização e Bioprospeção, Avenida Doutor Alfredo Magalhães Ramalho 6, 1495-165 Lisboa, Portugal
 <sup>b</sup> Centro Interdisciplinar de Investigação Marinha e Ambiental (CIIMAR), Terminal de Cruzeiros do Porto de Leixões, Avenida General Norton de Matos, S/N, 4450-208 Matosinhos, Portugal

<sup>c</sup> Departamento de Ingenierías Química y Biomolecular, Universidad de Cantabria, Avda. de Los Castros, s.n., 39005 Santander, Spain

<sup>d</sup> ANFACO-CECOPESCA, Campus University 16, 36310 Vigo, PO, Spain

<sup>e</sup> School of Engineering, National University of Ireland Galway, H91 HX31, Ireland

<sup>f</sup> Ryan Institute, National University of Ireland Galway, H91 HX31, Ireland

Kyun Insulute, National University of Ireland Galway, H91 HAS1, Ireland

<sup>8</sup> Centre for Environmental and Marine Studies (CESAM), Department of Environment and Planning, University of Aveiro, Campus Universitário de Santiago, 3810-193 Aveiro, Portugal <sup>h</sup> Bioscience Research Institute, Technological University of the Shannon Midlands Midwest, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

ap.ebi

Packaging

Waste

Packaging waste generated by packaging material

in 2019 in the European Unior

Plants: 19.4%

15.6

19.1%



- Consumers are not fully aware about what type of plastics can be recycled.
- Recycling information specific for packaging materials can help consumers' behaviour.
- Consumers were willing to pay more for products with more sustainable packaging.

#### ABSTRACT

A R T I C L E I N F O

Keywords: Packaging Fish Survey Recycling Plastic Consumer behaviour Packaging is essential to protect food, inform consumers, and avoid food waste, yet it can also contribute to the environmental footprint of products. Recycling waste treatment potentially provides more environmental benefits than other options (e.g., landfill), but only 66 % of packaging waste goes to recycling in the European Union. However, the prevention of packaging production with greater reuse, while extending the lifetime or improving packaging design should be firstly encouraged. This highlights the need to assess the willingness of consumers in reducing the environmental impact of seafood products from packaging. An online questionnaire was conducted in three countries (Portugal, Spain, and Ireland), composed of four sections: (i) seafood consumption, (ii) waste separation to be sent recycling, (iii) willingness to purchase seafood products with packaging designed to reduce environmental impact, and (iv) sociodemographic characteristics. Findings revealed that respondents from Spain and Portugal reported a slightly higher frequency of waste sent to recycle compared to Ireland. Irish waste management capabilities; whereas Spanish and Portuguese respondents were not fully aware that packaging does not need to be washed prior to recycling. The most popular alternatives to improve the sustainability of seafood packaging were the use of reusable packaging, compostable packaging material, glass jars for canned seafood instead of cans, and intelligent packaging.

\* Corresponding author at; Instituto Português do Mar e da Atmosfera (IPMA), Divisão de Aquacultura, Valorização e Bioprospeção, Avenida Doutor Alfredo Magalhães Ramalho 6, 1495-165 Lisboa, Portugal.

E-mail address: cheila.almeida@ipma.pt (C. Almeida).

http://dx.doi.org/10.1016/j.scitotenv.2022.160846

Received 29 July 2022; Received in revised form 27 September 2022; Accepted 7 December 2022

Available online 13 December 2022

0048-9697/© 2022 Elsevier B.V. All rights reserved.

Most respondents were willing to pay more for seafood products that use more sustainable packaging (62 % for Spain, 68 % for Ireland, 70 % for Portugal) and half of the respondents intimated that they avoid seafood products due to excessive packaging. With more detailed information on the waste management of packaging, seafood consumers could actively contribute with their attitudes where commensurate changes can improve environmental assessment of seafood.

#### 1. Introduction

While the primary function of packaging is to protect the product, less attention has been paid to packaging functions that reduce food waste. Such measures on packaging ability to reduce waste can be more important than the type of packaging material (Wikström et al., 2014; Williams and Wikström, 2011). Packaging effectively decreases the environmental burden of the product when considering the food loss related to injury during transportation and the environmental burden of additional production to compensate for the food loss (Sasaki et al., 2021). To fully assess the environmental impacts of food packaging, the trade-off between investment in packaging and potential reduced food waste should be quantified (Molina-Besch et al., 2019). Furthermore, the type of end-of-life treatment will compromise the contribution of the packaging to the total environmental assessment of the product. For example, packaging materials can be recycled into new products, produce energy if incinerated, or, in the worst-case scenario, end up in landfill (Wikström et al., 2019).

The current recycling rate of overall packaging waste in the European Union (EU) is around 66 % (European Environment Agency, 2021). In a production-based approach, the responsibility for packaging waste management is on the countries and producers, encouraging them to improve the recyclability of their packaging materials (Gao et al., 2022). However, for many years EU countries have shipped their plastic waste to other countries to meet their recycling targets, until 2017, when China refused to take any new waste (Brooks et al., 2018). This constraint reinforced the motivation to domestically recycle plastic waste and introduce more returnable or refillable containers in food packaging.

The transition towards a low-carbon footprint and circular economy, by extending the useful life of materials and encouraging recycling whilst reducing resource use, has become a priority in the EU (Tallentire and Steubing, 2020). The packaging sector is one of the major contributors to waste generation. In the EU, about 80 million tonnes of packaging waste was generated in 2019 (Eurostat, 2022). Paper and cardboard represented the main waste materials contributing 41 % of the total packaging waste generated, while plastic and glass represented both 19 %, wood 16 %, and metal 5 % (Eurostat, 2022). The Packaging Waste Directive contains updated measures to prevent the production of packaging waste, and promote its reuse, recycling, and other forms of recovery instead of final disposal (European Commission, 2018). It established the following recycling targets for each packaging material to be achieved by 2030: 85 % for paper/cardboard; 80 % for metal; 75 % for glass; 60 % for aluminium, 55 % for plastic; and 30 % for wood. These recycling targets should be calculated with the weight at the point where packaging waste enters the recycling operation (European Commission, 2018). Such targets are linked to the Circular Economy Action Plan which aims to make all packaging fully recyclable by 2030 (European Commission, 2020). In this sense, EU countries are required to adopt appropriate measures such as enhancing the circularity of waste systems, reduce the complexity of packaging, set up a minimum percentage of reusable packaging, design for bio-based materials application and re-use, and define deposit-return schemes. All these measures should be aligned with national programmes and economic instruments as incentives (e.g., taxes, subsidies, and charges) to achieve the established targets.

Studies on the environmental impacts of food production have shown that later stages in the supply system, such as packaging, retail, and transport combined contribute <14 % of greenhouse gas (GHG) emissions (Poore and Nemecek, 2018). However, in the case of specific types of products such as bottled drinks or canning products, packaging can contribute significantly to a product's GHG emissions since the production of glass

and metal requires a high level of energy (Poovarodom et al., 2012). Hence, packaging can contribute within a large range of values to the environmental burden of seafood products. The carbon footprint (CF) of packaging goes from almost zero up to 19 kg  $CO_2$  eq per kg of seafood, corresponding to <1 % to 89 % of the total CF of seafood products, differing substantially depending on the type of material used and its processing (Almeida et al., 2021). For example, the best case scenario is freezing lobster in a waxed cardboard box (van Putten et al., 2016) and the worst case is canned mussels in tinplate (Iribarren et al., 2010).

Most seafood studies focused on the production stage, whereas few have considered the complexity of seafood processing considering the supply chain from an integrated perspective. Packaging can have different levels, including primary packaging that is in direct contact with the product (e.g., aluminium can), secondary packaging corresponding to subsequent layers of material that contain one or more primary packaging (e.g., cardboard box), or tertiary packaging for transport, handling, and distribution (e.g., pallets) (ISO, 2016). The resources used along the seafood supply chain require consistent management to reduce waste and improve efficiency (Liu et al., 2020). However, seafood is highly prone to spoilage compared to other food (Love et al., 2015). As an example, canning allows for the preservation of perishable products such as small pelagic fish (e.g., sardines) caught in large volumes, thus increasing the proportion of fish available for human consumption (Almeida et al., 2015b). Viewed through recycling commitments, aluminium cans perform relatively well due to the 81 %<sup>1</sup> recycling rate for aluminium in the EU. However, plastic appears more favourable than aluminium if we prioritize climate impact since aluminium production relies on a more carbon-intensive production processes and is difficult to offset even with higher overall recycling rates. Therefore, given that frozen, chilled or cooked seafood packaging consists mainly of paper, plastic and wood, on average it contributes to <5 % of the CF, corresponding to <1 kg CO<sub>2</sub> eq per kg of seafood (Almeida et al., 2021). Plastics could also present an advantage in food distribution if their reuse is considered, being the most environmentally friendly option when compared to single-use cardboard boxes (Abejón et al., 2020). Expanded polystyrene (EPS) packaging represents one of the most widely used materials in seafood, due to its resistance to temperature changes during transport or storage and a significant environmental impact reduction is obtained when EPS is recycled and/or reused (Konstantinidis et al., 2021).

Plastic packaging waste has the lowest recycling rate compared with other materials such as glass, paper, and metals (Gao et al., 2022). Only 9 % of plastic waste is recycled worldwide, with the overwhelming majority of global plastic waste being landfilled or ending up contaminating the environment (Geyer et al., 2017). Plastic recycling represents a challenge due to the wide variety of additives and blends used in a multitude of products and the variation of material properties (e.g., shortening of the polymer chains) that limit the number of times products can be recycled (Geyer et al., 2017). A quality drop in the plastic recycled material reduces its application options, typically leading to down-cycling and delaying final disposal, but not avoiding it (van der Harst et al., 2016).

Measures to reduce the environmental impact of packaging, for example, switching from single-use to multi-use packaging (e.g., reusing a glass jar after the main product has been consumed) would further cut the carbon footprint, while also reducing waste. Reusable packaging systems are more often found in business-to-business markets for transport (e.g., boxes, pallets) while in business-to-consumer system addresses refillable-by-bulk-dispenser

<sup>&</sup>lt;sup>1</sup> https://international-aluminium.org/resource/aluminium-recycling-fact-sheet/

or parent packaging (e.g., refill packaging made with less material than parent packaging), and returnable packaging (Coelho et al., 2020). Other measures can be applied to reduce the environmental footprint of packaging such as the removal of excessive packaging, redesigning packaging to use less material, and plain packaging so that it can be easily recycled (Gao et al., 2022). Nevertheless, a shared responsibility approach is needed with incentives for producers and suppliers but also a consumer responsibility when the environmental impact of products includes consumption and end-of-life. As an example, consumers can pay a tax for the management of waste they generate, irrespective of where the packaging was produced, to guarantee that end-of-life of packaging waste follows the best interest of all (Gao et al., 2022).

Household waste sorting is an essential aspect of the waste management system; moreover, correct sorting of food packaging waste is cost-effective since it facilitates recycling and enhances the quality of the recycled materials (Nemat et al., 2019). However, environmental studies with data arising specifically on consumer behaviour from pre-purchase to postconsumption are scarce. There has been limited investigation into the relationship between sustainable packaging and consumer purchasing behaviour, and more studies should be conducted to cover the different products and consumer perspectives (Martinho et al., 2015).

The seafood industry is considered to be less innovative in new packaging development compared to other adjacent industries such as meat and poultry (Olsen et al., 2017). Furthermore, seafood packaging attributes appear to be underappreciated by researchers (Carlucci et al., 2015). Most of the research related to seafood consumption focus on consumers' preferences, motives and barriers (e.g., Cardoso et al., 2016; Pieniak et al., 2007; Verbeke et al., 2005). Attributes such as taste, appearance, freshness, and health benefits are often valued over production-related features (e.g., sustainability, production method) (Witter et al., 2021). Consumers value, above all, packaging attributes associated with quality, including the possibility to visualize the product (Olsen et al., 2017). To our knowledge, no research has been performed yet on the willingness of consumers in reducing the environmental impact of seafood products from packaging and if they proceed correctly at home when dealing with different seafood packaging waste materials. To collect survey data, consumer studies can be developed using web-based questionnaires (e.g., Altintzoglou et al., 2010a; Minnens et al., 2020), consumer panels (Altintzoglou et al., 2010b; Kole et al., 2009) or recruitment companies (Altintzoglou et al., 2021).

In this study, an online survey was developed to approach consumers in three countries, Spain (ES), Portugal (PT), and Ireland (IE). These countries were selected because they are European Atlantic countries, and the NEPTUNUS project, the framework where this research was developed, had data available. The main goals of this study were to establish among consumers from the three countries: 1) the reported rate of household waste sent to recycling for different seafood packaging types and associated materials; 2) if a relationship exists between seafood consumption frequency and the rate of seafood packaging separation for recycling; 3) knowledge regarding the recycling practices of different types of seafood packaging or materials; and 4) the willingness of consumers to purchase seafood products that use packaging with lower environmental impacts.

#### 2. Material and methods

The data collection was carried out online and it was preferred over other sampling methods (e.g., face-to-face questionnaires) based on convenience. Respondents can connect to the internet using various types of mobile devices and it allows instant access to a wide audience, irrespective of their geographical location, which makes it appropriate for cross-sectional studies and/or international comparisons (Evans and Mathur, 2018; Ilieva et al., 2002). A web-based questionnaire is easily accessible by a broader range of consumers and has the advantage to facilitate data entry, and overcoming time and budget constraints (Rogers, 2007). The anonymity possible by the internet is believed to help respondents in sharing their experiences and opinions (Van Selm and Jankowski, 2006). This sampling method has also disadvantages since only respondents with access to computer technology

and the internet will be able to participate and the answering instructions could be unclear (Callegaro et al., 2014). There is also a problem with sampling bias in non-probability panels (i.e., when people select themselves into the panel) with the impossibility to know in advance the probability of respondents' characteristics (Evans and Mathur, 2018). When results of online surveys cannot be defined as representative of the population, non-probability samples can be valuable as they may be representative of a sub-group of the total population (Van Selm and Jankowski, 2006). Therefore, a web-based questionnaire was a suitable sampling method for the explorative aim of this analysis. The survey questions were prepared in English, tested in an online form, and translated into other three languages: Portuguese, Spanish, and Gaelic. Adaptations were made in the different versions to better represent the specificities of each country (e.g., the recycling bin where plastic and metal should be discarded in Portugal and Spain is "yellow" and in Ireland is not associated exactly with a colour).

Data were collected through an online survey created on the LimeSurvey online tool. The links of four surveys were released corresponding to each language and shared via social media (e.g., Twitter, Facebook, LinkedIn) and email, with 640 validated and complete responses obtained: 228 from Portugal (PT), 234 from Spain (ES), and 154 from Ireland (IE). Additionally, seven answers were received from other countries and 17 more that did not present socio-economic data were eliminated. For those who agreed to participate, written informed consent was given at the beginning of the survey according to the General Data Protection Regulation (GDPR). Data collection was conducted from July to September 2021.

The questionnaire was structured in three parts (supplementary material): the first part collected information on seafood consumption patterns; the second section included questions about recycling and willingness to accept seafood products with packaging improvements to reduce environmental impacts; and the last section collected data on individuals' sociodemographic characteristics, such as gender, age, level of education completed, occupation, number of household members, living with children, living distance from the coast, and type of living place (e.g., rural versus city and inland versus coastal). Questions on seafood consumption habits referred explicitly to the period before the COVID-19 pandemic. Also, it was clarified in the consumption group of questions that "seafood" was meant as "fisheries and aquaculture products including fish, crustaceans and molluscs" and seafood consumption frequency should relate to an average weight per meal of a "portion of around 150 g or about the size and thickness of your hand".

Data were analysed by country; for the seafood consumption frequency, a cluster diagram was prepared to represent the respondents with high consumption which corresponded to more than twice a week, and low consumption which corresponded to equal or less than twice a week. Statistical differences were verified for socio-economic characteristics, seafood consumption frequencies, and the reported rate of household waste sent to recycling for different seafood packaging types with Pearson's chisquared test for contingency tables. For tests of associations between gender and answers related to seafood consumption and packaging, we used Chi-squared, Spearman's correlation, Kruskal-Wallis, Two-sided Student's *t*-test, and Wilcoxon signed-rank tests.



Fig. 1. Relative frequency of seafood consumption in the three countries.



Fig. 2. Relative frequency of seafood consumption in the three countries at home and out of home.

#### 3. Results

#### 3.1. Seafood consumption frequency of the respondents

Regarding the seafood consumption frequency (Fig. 1), most of the respondents reported consuming seafood between one to three times a week (ES - 68.8 %, PT - 57.0 %, IE - 61.0 %). However, significant differences were found between participating countries ( $p < 10^{-10}$ , Chi-squared test), which can be explained by a much higher percentage of Spanish and Portuguese who consumed seafood four to seven times a week (ES - 17.5 %, PT – 27.6 %) compared to Irish consumers (IE - 3.2 %), and a much higher percentage of Irish people who answered that they never consumed seafood (IE - 4.5 %). When analysing the frequency of seafood consumption in and outside the home (Fig. 2), similarities between Portugal and Spain were observed. For consumption at home, a higher percentage of respondents ate one to three times a week in both countries when compared to Ireland (ES - 72.6 %, PT - 61.4 %, IE - 58.8 %). For consumption of seafood outside the home, more respondents from Ireland eat seafood only once a month or less than compared to Portugal or Spain (ES - 53.0 %, PT - 48.7 %, IE -72.1 %).

When the analysis of the consumption frequency was by type of seafood product (Fig. 3), there was a higher consumption of fresh/chilled products in the three countries, with a higher number of respondents answering that they consumed these products one to three times a week (ES – 60.3 %, PT – 54.8 %, IE – 41.6 %). For frozen and canned seafood, the highest number of respondents occurred from Spain and Portugal who answered that they

consumed these products one to three times a week (ES – 35.9 % and 49.1 %; PT – 35.9 % and 34.2 %, respectively). However, it was once or less a month for Irish respondents (IE – 45.5 % and 41.6 %). Seafood products that were marketed as pre-cooked or ready-to-eat were less consumed in the three countries since the majority of respondents never consumed these products (ES – 58.1 %, PT – 47.4 %, IE – 46.8 %). Smoked products were consumed in general by most of the respondents from the three countries once or less a month (ES – 48.3 %, PT – 55.7 %, IE – 57.8 %) while dried and salted products were also eaten in Portugal once or less a month (PT – 48.2 %); but, in the case of Spain and Ireland most people never eat such products (ES - 43.6 %, IE – 76.0 %).

#### 3.2. Household waste sent to recycling and knowledge of recycling guidelines

In the case of Spain and Portugal, the largest number of respondents answered that they "always" send their non-organic waste (i.e., glass, paper, plastic waste) to specific bins to collect waste for recycling (Fig. 4), as reflected by 40.6 % and 42.3 % response rates respectively. In the case of respondents from Ireland, the largest number answered that they separate non-organic waste "often" (75 % of the times) to send it to recycling, corresponding to 29.5 % of the respondents. When the question was specific to seafood packaging types, in the case of Spain and Portugal the largest number of respondents answered that they "always" sent metal cans, and plastic package including expanded polystyrene, paper package, and glass jars to recycling. In the case of Ireland, a different pattern was observed where the largest number of Irish respondents answered that they "always" sent



Fig. 3. Realtive frequency of seafood consumption for different post-harvest processing products.

C. Almeida et al.



**Fig. 4.** Percentage of respondents that perform seperation of household waste of non-organic waste(i.e. glass, paper, plastic waste) to be sent to recycling in the three countries.

metal cans, paper packaging, and glass jars to recycling. However, for plastic packaging, the majority answered "often" and expanded polystyrene was "almost never" sent to recycling (Table 1).

When the respondents were divided into groups according to their seafood consumption frequency, a statistically significant difference related to the behaviour of sending household waste to recycling was found (p < 0.006, Chi-squared test). Respondents who self-reported higher consumption of seafood also self-reported higher intention of sending non-organic waste for recycling (Table 2).

By quantifying correct answers to the statements (Table 3) (question number 10 in the questionnaire available in the supplementary material), it was possible to characterize in part the knowledge of recycling practices when consumers deal with different types of packaging or materials. The answer that provided the greatest agreement by the three participating countries related to the statement that "metal cans from canned seafood packaging should be put in the yellow waste container or recycling bin" (TRUE was the correct answer in all cases). It received the highest number of correct answers from the three countries and significant differences were found between countries (p = 0.02, Chi-squared test), driven by the higher number of correct answers from PT. The answer that "empty plastic bags used for fresh seafood packaging should be put in yellow waste container

#### Table 2

Percentage of respondents that separate household packaging waste to recycling associated with their level of seafood consumption (%).

	Rate o	f househole	d waste ser	nt to recycli	ing
	0 %	25 %	50 %	75 %	100 %
High consumption (> $2 \times$ week)	1.3	7.7	11.1	34.2	45.7
Low consumption ( $\leq 2 \times$ week)	3.3	10.1	16.7	38.0	31.9

to recycle or recycling bin" (TRUE was the correct answer to ES and PT; FALSE was the correct answer to IE) received the second highest number of correct answers from the three countries and differences between countries were significative (p = 0.0028, Chi-squared test). In terms of whether it is necessary to wash metal cans from canned seafood packaging before they are sent to recycling (FALSE in the case of ES and PT, and TRUE in the case of IE), it received the lowest number of correct answers from Spain (47.9 %), less than half, and highest from Ireland (76.0 %). Differences between countries were significant (p = 0.00000008, Chi-squared test), driven by a higher number of correct answers from IE and lower from ES. The statement saying that "it is not necessary to wash plastic packaging from frozen seafood products before they are sent to waste" (TRUE was the correct answer in all cases) got overall the lowest number of correct answers and differences between countries were significant ( $p = 10^{-13}$ , Chi-squared test). Ireland is the country that overall had a lower average (63.9 %) when compared to Spain (68.1 %) and Portugal (78.5 %).

# 3.3. Willing to consume seafood products that use more sustainable packaging or avoid products with excess packaging

The most accepted alternatives for more sustainable seafood packaging according to responses from the three countries were (i) reusable packaging (e.g., return glass jar from canned seafood to the producer/retailer), (ii) a compostable type of packaging material (e.g., flexible film made from crustacean shells), (iii) "glass jar packaging for canning seafood instead of metal cans, and (iv) intelligent packaging (which extends shelf-life and maintains the quality of the food) (Table 4) (question number 11 in the questionnaire available in the supplementary material). Between 81.6 % and 96.5 % of the respondents answered that they are receptive to these practices. Only 37.0 % to 59.1 % of the respondents were willing to use a squeeze-tube type packaging instead of metal cans and plastic pouch packaging for

Table 1

Percentage of res	pondents that i	perform separation	n of the different sea	food packaging type	es/materials to be sen	t to recycling in the thr	ee countries (%)

		Spain ( <i>n</i> = 234)	Portugal ( $n = 228$ )	Ireland $(n = 154)$
Metal cans	Never (0 %)	12.0	4.4	9.7
	Almost never (25 %)	8.1	8.3	5.2
	Sometimes (50 %)	9.0	6.6	6.5
	Often (75 %)	14.1	11.0	11.0
	Always (100 %)	56.8	69.7	67.5
Plastic package	Never (0 %)	7.3	3.1	7.1
	Almost never (25 %)	6.8	4.8	10.4
	Sometimes (50 %)	8.1	9.6	16.9
	Often (75 %)	21.8	22.4	37.0
	Always (100 %)	56.0	60.1	28.6
Paper package	Never (0 %)	8.5	7.0	9.1
	Almost never (25 %)	12.4	9.2	9.7
	Sometimes (50 %)	11.1	12.3	7.8
	Often (75 %)	20.9	20.6	16.2
	Always (100 %)	47.0	50.9	57.1
Glass jar	Never (0 %)	9.4	4.8	13.0
	Almost never (25 %)	6.8	5.3	1.3
	Sometimes (50 %)	9.0	6.6	1.3
	Often (75 %)	13.7	9.6	9.1
	Always (100 %)	61.1	73.7	75.3
Expanded polystyrene	Never (0 %)	19.2	16.2	44.2
	Almost never (25 %)	12.4	10.5	14.9
	Sometimes (50 %)	14.5	8.8	13.6
	Often (75 %)	19.2	11.8	9.7
	Always (100 %)	34.6	52.6	17.5

Percentage of respondents from each country that gave the correct answer about recycling guidelines statements (%).

Countries version/statement [correct answers]	Spain (ES) (n = 234)	Portugal (PT) $(n = 228)$	Ireland (PT) (n = 154)
ES, PT / Metal cans from canned seafood packaging should be put in yellow waste container to recycle [TRUE] – IE / Metal cans from	88.0	94.7	87.7
canned seafood packaging should be put in the recycling bin [TRUE] *			
ES, PT / It is necessary to wash metal cans from canned seafood packaging before they are sent to waste [FALSE] - IE / It is not necessary	47.9	64.5	76.0
to wash metal cans from canned seafood packaging before they are sent to waste [FALSE] *			
ES, PT / The plastic packaging from frozen seafood products should be put in normal/undifferentiated waste container [FALSE] - IE / The	77.8	93.0	46.8
plastic packaging from frozen seafood products should be put in the general waste [FALSE] *			
ES, PT, IE / It is not necessary to wash plastic packaging from frozen seafood products before they are sent to waste [TRUE] *	57.7	72.4	32.5
ES, PT / Expanded polystyrene boxes used for fresh seafood should be put in yellow waste container to recycle [TRUE] - IE / Expanded	70.5	75.4	73.4
polystyrene boxes used for fresh seafood should be put in the recycling bin [FALSE]			
ES, PT, IE / It is necessary to wash expanded polystyrene boxes before they are sent to waste [FALSE] *	61.5	70.6	58.4
ES, PT / If empty, plastic bags used for fresh seafood packaging should be put in yellow waste container to recycle [TRUE] - IE / If empty,	77.8	88.2	72.1
plastic bags used for fresh seafood packaging should be put in the recycling bin [FALSE] *			
ES, PT / It is not necessary to wash plastic bags used for fresh seafood packaging before they are sent to waste [TRUE] - IE / It is not	63.7	68.9	64.3
necessary to wash plastic bags used for fresh seafood packaging before they are sent to waste [FALSE]			
Average	68.1	78.5	63.9

\* Statistically significant (*p* < 0.05, Chi-Squared test) differences were found between countries.

canning seafood instead of metal cans, which demonstrates that these should be considered less successful measures to improve seafood packaging from the consumers' point of view.

The avoidance of seafood products due to excess packaging (e.g., a package that has a secondary layer of carton) was confirmed by around half of the respondents, with Portuguese respondents having the highest rate (56.6 %) and Irish the lowest with 44.8 % of the respondents (Table 5). The dried and/or salted products (e.g., salted and dried cod) were considered to have the least excessive packaging for a post-harvest processed seafood product (ES - 5.1 %, PT - 4.8 %, IE - 3.2 %). The opposite trend was found for pre-cooked/ready-to-eat type of products, whereby significant numbers considered packaging excessive (ES - 37.2 %, PT - 39.5 %, IE - 30.5 %).

The majority of respondents, between 62.0 % for Spain, 68.8 % for Ireland, and 70.2 % for Portugal are willing to pay more for seafood products that use more sustainable packaging (Table 6). Of these respondents, the largest number answered that would be willing to pay 10 % more (ES - 43.6 %, PT - 53.1 %, IE - 46.1 %).

#### 3.4. Socio-economic characteristics

The socio-economic characteristics of the sample (Table 7) were statistically different between countries for all characteristics except gender. The percentage of female respondents was slightly greater in the three countries (between 64.5 % to 67.1 %). The sample from Spain presented a younger population (more in 21–40 years old), fewer people lived with parents (25.2 %), a greater percentage of households without children (46.2 %), and more people that lived in a small town (51.7 %) compared to the other two countries. The majority of respondents from Portugal were between 41 and 60 years old (64.9 %) and, compared to the other two countries, had a higher level of education (30.7 % of respondents had a PhD), the majority lived in <50 km distance from the coast (93.9 %) and in a large town (60.5 %) and also did not live with parents (89.9 %). The Irish sample had the highest percentage of people with three or more children (13.6 %), the highest household size comprising five to six members (16.2 and 5.2 % respectively), and the highest proportion of respondents that lived rurally (40.3 %). Despite being an island, 46.1 % of the Irish respondents lived further than 50 km from the sea, as opposed to 6.1 % of Portuguese and 16.7 % of Spanish respondents.

When the results from the questions were tested against the sociodemographic characteristics of the samples (Table 8), it was possible to find some statistical differences. Older respondents (Rho = 0.17; p = 0.00001, Spearman's correlation test), not living with parents (p = 0.0045, Kruskal-Wallis test), and living on the seaside reported a significantly higher seafood consumption frequency (p = 0.000001, Kruskal-Wallis test). A higher frequency of seafood consumption at home was reported by older respondents (Rho = 0.19, p = 0.0000014, Spearman's correlation test), not living with parents (p = 0.03, Kruskal-Wallis test), but living with children (Rho = 0.11, p = 0.007, Spearman's correlation test), and living on the seaside (p = 0.000001, Kruskal-Wallis test).

Males appear to have a significantly higher frequency of seafood consumption outside the home (p = 0.0005, Chi-squared test). This trend seems essentially driven by Portugal, the only sample for which the difference is significant (p = 0.002, Chi-squared test). Age follows the same pattern which might be explained by the fact that older people have usually better jobs and a higher economic level (Rho = 0.15, p = 0.0002, Spearman's correlation test). Students and the unemployed have lower seafood consumption outside the home (p = 0.002, Kruskal-Wallis test) which is normal due to budget limitations, together with respondents with a higher number of people living at the household (Rho = -0.098; p =0.02 Spearman's correlation test). Respondents not living with parents (p = 0.00002, Kruskal-Wallis test) and living in cities presented higher seafood consumption outside the home (p = 0.005, Kruskal-Wallis test).

Females appear to have a higher willingness to reuse packaging (88.5 %) than males (81.3 %) (p = 0.02, Chi-squared test). There is also a slight tendency that respondents from larger households are not willing to use reusable packaging (p = 0.01, Wilcoxon signed-rank test). The older respondents are more willing to use a compostable type of packaging (p = 0.001, Two-sided Student's *t*-test), the result is mostly explained by the differences found among the Spanish cohort where those that responded positively were on average 11 years older than those who respondents that are

#### Table 4

Percentage of respondents from each country that answered "yes" about their willingness to use the seafood packaging alternatives (%).

Seafood packaging alternative	Spain (n = 234)	Portugal ( $n = 228$ )	Ireland ( $n = 154$ )
Reuse packaging (e.g., return glass jar from canning seafood to the producer/retailer)	81.6	88.6	88.3
Use a compostable type of packaging material (e.g., flexible film made from crustacean shells) *	90.2	96.5	94.8
Use a squeeze tube type packaging instead of metal cans	44.0	48.7	37.0
Use plastic pouch packaging for canning seafood instead of metal cans	54.7	52.2	59.1
Use glass jar packaging for canning seafood instead of metal cans	93.2	95.6	90.9
Use of intelligent packaging (which extends shelf-life and maintain the quality of the food)	90.6	92.1	90.9

\* Statistically significant (p < 0.05, Chi-Squared test) differences were found between countries.

Percentage of respondents from each country that avoid seafood products with excess packaging and consider that post-harvest processing products have excess packaging (%).

		Spain ( <i>n</i> = 234)	Portugal $(n = 228)$	Ireland $(n = 154)$
Avoid seafood products with Type of post-harvesting processing products	excess packaging Fresh / Chilled Frozen Canned Pre-cooked / Ready-to-eat	49.1 7.3 18.8 15.0 37.2	56.6 12.3 33.8 23.2 39.5	44.8 23.4 18.2 4.5 30.5
	Smoked Dried and/or salted	12.0 5.1	20.2 4.8	13.6 3.2

students (p = 0.0003, Chi squared test), living in larger households (p = 0.046, Wilcoxon signed-rank test), living with parents (p = 0.0000002, Chi squared test), or have a higher number of children in the household (p = 0.02, Wilcoxon signed-rank test) answered that they are not willing to use a compostable type of packaging, and again the result is mostly explained by the differences found among the Spanish cohort. Finally, it was found that there is a slight trend for more educated people to accept the use of a squeeze-tube type packaging instead of metal cans (p = 0.004, Wilcoxon signed-rank test).

#### 4. Discussion

Addressing consumer concerns by way of understanding behavioural aspects and preferences is important as this can also inform changes towards sustainable practices and promote innovation through social marketing (Domegan, 2021; Rowan and Pogue, 2021). The present study focused on the role of consumers across Portugal, Spain and Ireland in terms of understanding practices towards waste of seafood packaging that can affect environmental sustainability. Results revealed that seafood consumption frequency is relatively similar between Spain and Portugal, with the majority of respondents reporting the consumption of these products between one to three times a week, with a slightly higher frequency occurring for the Portuguese cohort. Ireland showed a different pattern where there was a general lower frequency of seafood consumption, including a small percentage of respondents (4.5 %) that never ate seafood. This result is not surprising as apparent per capita consumption of fishery and aquaculture products is highest in Portugal and Spain, with 59.9 kg and 46.0 kg per year respectively, and slightly above the EU average (24.0 kg) in the case of Ireland with 25.5 kg (EUMOFA, 2021). Data obtained for Portugal in previous studies provided a seafood consumption of four to five times a week, which is extremely high (Almeida et al., 2015a; Cardoso et al., 2013). Furthermore, Spanish citizens were associated with regular fish consumers, with 46 % of respondents saying that they consumed fish more than once a week, while Ireland was at 18 % (Zander and Feucht, 2018).

Regarding the different processing products, there is a higher consumption in the three countries for fresh/chilled seafood products. Frozen and canned seafood has the highest consumption frequency in Spain and Portugal, with the largest number of respondents answering one to three

#### Table 6

Percentage of respondents from each country that are willing to pay more for seafood products that use more sustainable packaging (e.g., package that allows to send all the packaging materials to recycling) and how much more (%).

		Spain (n = 234)	Portugal ( $n = 228$ )	Ireland ( $n = 154$ )
Pay more for seafor	od			
products that use	e more	62.0	70.2	68.8
sustainable packa	aging			
How much more	10 %	43.6	53.1	46.1
2	25 %	14.1	14.9	18.2
	50 %	3.4	0.4	2.6
	> 50 %	0.9	1.8	1.9

times a week for both types of products; but, in Ireland it was one or fewer times a month. Pre-cooked or ready-to-eat seafood was the least consumed in the three countries with the largest number of respondents never eating these products. Respondents that reported higher seafood consumption related to responses with higher rates of household separation of nonorganic waste to recycling. This could be of relevance since most of the seafood consumption occurs at home (one to three times a week had the majority of answers from the three countries: 72.6 % in ES, 61.4 % in PT, 57.8 % in IE) and the seafood packaging waste could thus be significant in this setting. Furthermore, a higher frequency of seafood consumption at home was reported by older respondents, not living with parents, but living with children, and living on the seaside.

Respondents appear to be aligned with recycling behaviour as the largest number of respondents from Spain and Portugal said they separate all household waste to be sent to recycling whereas Ireland sends almost all. Differences in responses were noted when the question posed was specific to the type of packaging materials. There were lower rates of waste separation to recycling for Ireland for plastic waste, especially with packaging made by EPS. The reason behind these results could be related to the lack of information provided to consumers on how to deal with different waste materials or a lack of waste management capacity. Providing

#### Table 7

Summary statistics of the socio-economic characteristics of the samples (%).

		Spain sample	Portugal sample	Ireland sample
Gender	Female	64.5	67.1	66.2
	Male	34.2	32.5	29.9
	Other	0.0	0.0	1.3
	Prefer not to say	1.3	0.4	2.6
Age (years) *	≤ 20	14.1	0.0	0.0
	21-40	41.9	25.0	44.2
	41-60	38.9	64.9	46.1
	≥ 61	4.7	10.1	9.7
Education level *	Primary or below	2.1	0.0	0.0
	Lower secondary	13.7	0.9	0.6
	Higher secondary	3.4	6.1	5.2
	Additional technical	11.1	1.8	8.4
	training			
	BSc (or similar)	23.5	27.6	30.5
	MSc (or similar)	32.5	32.9	36.4
	Bellow upper education			
	Tertiary education	-	-	-
	PhD (or similar)	13.7	30.7	18.2
Current occupation *	Student	16.7	7.9	16.2
	Employed	72.6	79.4	68.2
	Self employed	3.4	6.6	4.5
	Unemployed	4.3	2.2	2.6
	Responsible for the	0.9	1.8	2.6
	household			
	Retired	2.1	2.2	5.8
Number of people living in	1	9.4	12.3	9.1
the household *	2	25.2	28.5	32.5
	3	31.2	27.6	16.9
	4	27.8	25.0	19.5
	5	4.3	5.3	16.2
	≥ 6	1.7	0.9	5.2
Live with parents *	No	74.8	89.9	81.8
	Yes	25.2	10.1	18.2
Number of children in the	0	46.2	55.7	56.5
household *	1	25.6	21.1	14.9
	2	23.9	18.4	14.9
	≥ 3	3.8	4.8	13.6
Type of living area *	Large town	21.8	60.5	37.7
	Rural area or village	26.5	13.6	40.3
	Small or middle-sized	51.7	25.9	22.1
Living distance from the sea *	Inland (>50 km from the coast)	16.7	6.1	46.1
	On the seaside (<50 km from the coast)	83.3	93.9	53.9

 $^*$  Statistically significant (p < 0.05, Chi-Squared test) differences were found between countries.

Statistical test results (p values) between socio-demographic variables and answers related with seafood consumption and packaging. \* Statistically significant (p < 0.05) differences were found; and the following statistical test was used: <sup>1</sup> Chi-squared test; <sup>2</sup> Spearman's correlation test; <sup>3</sup> Kruskal-Wallis test; <sup>4</sup> Two-sided Student's *t*-test, <sup>5</sup> Wilcoxon signed-rank test.

	Gender	Age	Education level	Occupation	N° people in the household	Live with parents	N° children in the household	Type of living area	Living distance from the sea
Seafood consumption frequency	_1	Rho = 0.17; $p = 0.00001^{2} *$	- 2	- 3	- 2	0.0045 * <sup>3</sup>	- 2	_ 3	0.000001 * 3
Seafood consumption at home	_1	Rho = 0.19; $p = 0.0000014^{2} *$	- 2	- 3	_ 2	0.029 * <sup>3</sup>	Rho = 0.11; $p = 0.007 * ^{2}$	- 3	0.00002 * <sup>3</sup>
Seafood consumption outside the home	0.0005* 1	Rho = 0.15; $p = 0.00018^{2} *$	- 2	0.0017 * <sup>3</sup>	Rho = $-0.098$ ; p = 0.017 * <sup>2</sup>	0.00002 * <sup>3</sup>	2	0.0047 * <sup>3</sup>	0.0076 * <sup>3</sup>
Willingness to reuse packaging	0.021 * 1	4	_ 5	- 1	0.046 5	- 1	- 5	- 1	- 1
Willingness to use a compostable type of packaging material	- 1	0.001 * 4	_ 5	0.0003 1	0.01 5	0.0000002	0.02 5	- 1	- 1
Willingness to use a squeeze tube type packaging instead of metal cans	- 1	- 4	0.004 * <sup>5</sup>	- 1	- 5	- 1	- 5	- 1	- 1
Willingness to use plastic pouch packaging for canned seafood	- 1	- 4	- 5	- 1	- 5	- 1	_ 5	- 1	- 1
Willingness to use glass jar packaging for canned seafood	- 1	- 4	_ 5	- 1	_ 5	- 1	_ 5	- 1	- 1
Willingness to use of intelligent	- 1	- 4	_ 5	- 1	_ 5	- 1	_ 5	- 1	- 1
Avoid seafood products with excess	- 1	- 4	- 5	- 1	_ 5	- 1	_ 5	- 1	- 1
Pay more for seafood products that use more sustainable packaging	- 1	_ 4	_ 5	_ 1	_ 5	- 1	_ 5	- 1	_ 1

recycling information is particularly important to enhance consumer knowledge of the value of packaging and to eliminate uncertainty about sorting different packaging materials (Nemat et al., 2019). From open questions, it was possible to understand that some respondents seem to be confused about what type of plastics can be recycled. Since many types of plastic exist and only a few cannot be recycled, a more efficient message could be to request consumers to separate all plastic to be sent to recycling and, in this way, avoid the risk of losing plastics from packaging that would go to undifferentiated waste treatment, for example, landfilling. Furthermore, in Ireland non-rigid plastic (e.g., plastic film, grocery bags) could only be sent to recycling waste bins since September 2021, which might explain the plastic recycling responses in Ireland to some extent were contrary to Spain and Portugal.

When we analysed the four statements relating to knowledge about recycling including where metal cans, plastic packaging from frozen, EPS boxes, and plastic bags from fresh seafood should be put or not in the recycling bin, on average, Ireland received the lowest percentage of correct answers to these statements (63.9 %) compared to Spain (68.1 %) and Portugal (78.5 %). The fact that in Ireland plastic packaging, metals, and all paper and cardboard materials should be sent to the same recycling bin (whereas in Spain and Portugal there is a yellow container for plastic and metal packaging and a blue container for paper and cardboard) can have a counter-productive effect on an individual's capacity to send packaging to recycling. There is also another difference between countries as in Ireland, in contrast to Portugal and Spain, consumers are asked to wash packaging before sending them to the recycling bin. This happens because in Ireland, since recycling materials are mixed, dirty or unsuitable items could contaminate the entire contents. This behaviour seems to be evident among Irish consumers since Irish participants reported a higher percentage of correct answers compared to the other two countries when asked if it is necessary to wash metal cans from canned seafood packaging before they are discarded (76.0 % for IE against 47.9 % for ES and 64.5 % for PT). One reason that might explain these results is that metal cans might be perceived as dirtier packaging compared to plastic due to the associated oil used to preserve canned seafood. It seems that a higher effort exists in Ireland in communicating this specific behaviour of washing dirty packaging when it is needed. In Spain and Portugal, this seems to be confusing since results for washing metal had the lowest number of correct answers from all the statements in both countries' samples. Washing packaging does not add any benefit to the recycling process since grease, traces of food, or even liquid residues do not interfere with the process, as there is a washing phase in the recycling treatment that eliminates these traces. This means that, when washing the packages at home before sending them to the recycling bin, consumers in Portugal and Spain are doubling the work and wasting water. The other three statements related to washing plastic packaging from frozen seafood, EPS boxes, and plastic bags from fresh seafood all got a higher level of correct answers from respondents from Portugal, with 72.4 %, 70.6 % and 68.9 %, and lower from Spain (57.7 %, 61.5 %, and 63.7 % respectively) and Ireland (32.5 %, 58.4 %, and 64.3 % respectively). The results suggest that improved communication using more appropriate guiding and policy information could help consumers in Spain and Portugal adjust their behaviour. This is supported by observations made by Klaiman et al. (2017) on the testing of plastic or boxboard sandwich containers, where having to clean packaging can alter preferences for packaging material and affect consumer recycling behaviour at the household.

The statement that had the highest consensus and highest number of correct answers, with 90.4 % on average, from the three countries, was if metal cans from canned seafood should be put in the recycling bin. This result is in accordance with the high level of aluminium recycling rate in Europe, which in Spain and Portugal, it can also be recovered from undifferentiated waste. Nevertheless, it demonstrates that more effort should be done for other types of packaging materials such as plastics. In general, it seems that there is a lack of information about plastic packaging recycling since some respondents intimated in open answers that they do not fully know the guidelines and believe that seafood plastic packaging cannot be sent to the recycling bin and, for example, that EPS package cannot be recycled. Also, some respondents believe that they need to wash the plastic and EPS to be recycled and, since they do not want to do it, they avoid contaminating "clean" packaging in the recycling of packages (in the case of Spain and Portugal). To overcome these misunderstandings, it would be important to communicate detailed procedures applicable to each type of material, including the different types of plastics (e.g., EPS, High-Density Polyethylene (HDPE), Low-Density Polyethylene (LDPE)), and also explain the destination of these types of materials after recycling. Specific communication for certain types of seafood products could also support the willingness of consumers to send more household waste to recycling, especially in Spain and Portugal, where seafood consumption at home is so frequent. Nevertheless, the different recycling guidelines implemented in the EU countries could impair the consistency of recycling behaviours in households at the European level.

The most consensual alternatives to using more sustainable packaging for seafood from the three countries were the use of reusable packaging (e.g., return glass jars from canning seafood to the producer/ retailer), glass jar packaging for canning seafood instead of metal cans, compostable types of packaging material (e.g., flexible film made from crustacean shells), and intelligent packaging (which extends shelf-life and maintains the quality of the food). By reusing packaging, a significant reduction in the environmental impact of packaging could be feasibly achieved as a second use will allocate part of the environmental cost of the packaging. Therefore, this alternative could be promoted in case companies find it feasible (e.g. implement a return deposit scheme of glass jars for some beverage products such as water bottles). The main barrier identified by producers is a major logistical complexity, which requires the reorganisation of supply chains to ensure that packaging is available and returned through distribution and that consumers maintain their loyalty (Coelho et al., 2020). Another possibility could be the use of reusable containers from consumers when purchasing seafood at the fishmongers to avoid the use of plastic bags/heat-sealed bags. Females appear to have a higher willingness to reuse packaging than males, but not respondents from larger households that may be due to an idea that this possibility could imply extra work for the household routine which might be already complicated in households with more people.

Glass jar packaging for canning seafood could represent environmental benefits when compared to aluminium (Laso et al., 2017). In the same way, through packages with intelligent packaging consumers may also reduce seafood waste and prevent unnecessary transport and logistics from an early stage (Cammarelle et al., 2021; Salgado et al., 2021). The use of a compostable type of packaging material could improve the end-of-life of seafood packaging, which is mainly an alternative to plastic; but, it raises the point if municipalities are prepared to collect waste compost and if it is an appropriate disposal alternative. Older respondents seem to be more willing to use a compostable type of packaging, and on the contrary, respondents that are students, living in larger households, living with parents, or having a higher number of children in the household answered that they are not so willing to use compostable packaging. These results are mostly explained by the differences found among the Spanish cohort which is characterized also by younger respondents when compared with the other two samples, meaning older people could be more aware of the possibilities of composting their waste.

New design features that make packaging more sustainable could also increase the likelihood that sustainable packaging is chosen which could potentially increase the level of positive-environmental behaviour from consumers (Martinho et al., 2015). However, the use of a "squeeze-tube type packaging instead of metal cans", commonly marketed in northern European countries, and "plastic pouch packaging for canning seafood instead of metal cans" had half or less than half of the answers so these should be considered less successful measures to improve seafood packaging from the consumers' point of view. These options imply a redesign of the packaging which may make the package less recognizable to consumers or less convenient, and the use of different types of material may reduce the appeal and attractiveness (Prendergast and Pitt, 1996). Even though, a slight trend was found for more educated people to accept the use of a squeeze-tube type packaging instead of metal cans.

Respondents appear to be concerned with packaging because around half of them intimated that they avoid seafood products due to excessive packaging (e.g., packaging that has a secondary layer of carton). This is especially the case for pre-cooked/ready-to-eat types of products since it was the type of post-harvest processed product that had a higher number of respondents who considered it to have excessive packaging. These products are less consumed in the three participating countries since the majority of respondents answered that they never eat them. The second level of products that respondents associate with excessive packaging is frozen products. In contrast, dried and/or salted products (e.g., *bacalhau* or mackerel) were the seafood products that fewer respondents from the three countries considered had excessive packaging. The majority of respondents from Portugal answered that they consumed dried and salted seafood products once or fewer times a month; but, in the case of Spain and Ireland the highest number of respondents never tried them. Fresh or chilled seafood products, the most common way of consuming seafood in these countries, were also not considered as having excessive packaging, especially by respondents from Spain and Portugal.

The majority of respondents were willing to pay more for seafood products that use more sustainable packaging (for example a 10 % increase in price was acceptable for 62.0 % of respondents in Spain, 68.8 % in Ireland 70.2 % in Portugal). This result shows the commitment of consumers in contributing to an effort that may need to be carried out in partnership with industries, presenting opportunities for innovation and more environmentally-friendly solutions (Steenson and Creedon, 2022). This result is aligned with consumer motivation to pay for more sustainable seafood products which is positively correlated to respondents' income and their environmental concerns, with a maximum premium of 10 % of the product price that consumers were happy to pay (Salladarré et al., 2016). Nevertheless, many people find difficulties in defining terms such as 'sustainability' in relation to something as specific as packaging and the principles in improving environmental impacts need to be translated into more clear requirements (Lewis, 2005).

Overall, the results highlight a positive attitude of consumers towards recycling and care for packaging with lower environmental impact. However, the results could also be biased by the highly-educated cohort sample which may affect the consumers' knowledge as well as their reported attitudes. This also highlights an area to improve upon for future research to reach consumers from a broader socio-demographic range and consequently obtain a more holistic perspective of environmental concerns. Other limitations of this present research should be noted. Firstly, the sample is not representative of countries' populations and results should be discussed as representing only the group of people, with their respective sociodemographic characteristics, that answered the survey. Additional research is recommended to ensure a full characterisation of the entire seafood consumer market in European countries. A second limitation is related to the data collection method adopted as the survey was processed online, representing a strong barrier for people with low motivations and/or unfamiliar with new technologies or social networks and as a result, higher educated people were more likely to answer to the survey (e.g., Cammarelle et al., 2021). Nevertheless, results can give an outlook on how much respondents are willing to contribute to reducing the environmental impact of seafood products through packaging and start a discussion on what is needed to improve the end-of-life of seafood packaging. A third issue that one should bear in mind is that answers with self-declared intentions related to more responsible behaviours or willingness to pay can be overestimated by people responding too positively when compared to reality.

Food packaging and its attributes can potentially hinder or motivate consumers to sort packaging waste correctly. The design of packaging influences recycling behaviour and, therefore, the recyclability of packaging should be considered an inherent value of the packaging, similar to other attributes such as beauty or durability (Nemat et al., 2019). Addressing consumer behavioural and attitude changes towards seafood packaging is important. In this instance the transition towards increased circularity where communication can be done by deploying appropriate marketing practices to educate and empower consumers (Domegan, 2021; Rowan and Casey, 2021). Consumers are only one of the key players in the seafood value chain and efforts are also needed from other groups of stakeholders (e.g., policymakers, dieticians, consumer organisations, retailers, fishermen, seafood processing industry) to ensure that future seafood production, processing, distribution and consumption become more sustainable. In this sense, transparent and trustful information with the help of certification labelling is recommended (Sacchettini et al., 2021). As an example, labelling based on Product Environmental Footprint (PEF) guide could be developed to deliveri information to consumers integrating topics that are still poorly described, such as the end-of-life of packaging materials (Hélias et al., 2022).

#### 5. Conclusions

The reported rates of household waste sent to recycling of different seafood packaging types and associated materials were relatively high in the three countries analysed: Spain, Portugal, and Ireland. Respondents also appear to be willing to pay more for seafood products that use more sustainable packaging and avoid seafood products with excess packaging. A general relationship was observed between seafood consumption frequency, which mainly occurs at home, and seafood packaging separation for recycling. These results need to be linked with a sample of respondents that in general is characterized by highly-educated people and mostly employed. However, these findings can provide insights as to where to invest resources to inform on sustaining behaviour in seafood packaging waste disposal. Recycling practices at the household level could be improved upon with more appropriate information provided about specific types of seafood packaging and materials, especially for plastics that present a huge diversity of types. This may be significantly important in countries where there is high seafood consumption, especially at home, as is the case for Spain and Portugal. An increase in the quantity of packaging sent to recycling and efficiency of the waste management system can hence improve the circularity and reduce the environmental cost of seafood related to packaging and its materials.

#### CRediT authorship contribution statement

Cheila Almeida: Conceptualization, Methodology, Investigation, Data curation, Project administration, Writing – original draft. Jara Laso: Methodology, Investigation, Visualization, Writing – original draft. David Baptista de Sousa: Methodology, Investigation, Writing – original draft. Ronan Cooney: Methodology, Investigation, Writing – original draft. Paula Quinteiro: Methodology, Writing – original draft. Neil Rowan: Methodology, Writing – review & editing. Ana Cláudia Dias: Methodology, Writing – review & editing. Eoghan Clifford: Methodology, Writing – review & editing. Rodrigo G. Reboredo: Writing – review & editing. María Margallo: Writing – review & editing. Maria Leonor Nunes: Methodology, Writing – review & editing. António Marques: Methodology, Writing – review & editing, Supervision.

#### Data availability

Data will be made available on request.

#### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgements

We are most grateful to Filipe Ferreira from IPMA Informatic Systems Division for help in preparing the online survey using LimeSurvey software and Nuno Barbosa Morais from iMM for help with data analysis. The authors would like to acknowledge the promotion of the survey by the various agencies, stakeholders and groups. This work was supported by the NEPTUNUS project (EAPA\_576/2018). The authors would like to acknowledge the financial support of Interreg Atlantic Area. Ana Cláudia Dias and Paula Quinteiro acknowledge FCT/MCTES for the financial support to CESAM (UIDB/50017/2020 + UIDP/50017/2020 + LA/P/0094/2020), through national funds, and to the research contracts CEECIND/02174/ 2017 and CEECIND/00143/2017, respectively.

#### Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi. org/10.1016/j.scitotenv.2022.160846.

#### References

- Abejón, R., Bala, A., Vázquez-Rowe, I., Aldaco, R., Fullana-i-Palmer, P., 2020. When plastic packaging should be preferred: life cycle analysis of packages for fruit and vegetable distribution in the Spanish peninsular market. Resour. Conserv. Recycl. 155 (November 2019), 104666. https://doi.org/10.1016/j.resconrec.2019.104666.
- Almeida, C., Altintzoglou, T., Cabral, H., Vaz, S., 2015a. Does seafood knowledge relate to more sustainable consumption? Br. Food J. 117 (2), 894–914. https://doi.org/10. 1108/BFJ-04-2014-0156.
- Almeida, C., Loubet, P., da Costa, T.P., Quinteiro, P., Laso, J., Baptista de Sousa, D., Cooney, R., Mellett, S., Sonnemann, G., Rodríguez, C.J., Rowan, N., Clifford, E., Ruiz-Salmón, I., Margallo, M., Aldaco, R., Nunes, M.L., Dias, A.C., Marques, A., 2021. Packaging environmental impact on seafood supply chains: a review of life cycle assessment studies. J. Ind. Ecol. 1–18. https://doi.org/10.1111/jiec.13189.
- Almeida, C., Vaz, S., Ziegler, F., 2015b. Environmental life cycle assessment of a canned sardine product from Portugal. J. Ind. Ecol. 19 (4), 607–617. https://doi.org/10.1111/jiec. 12219.
- Altintzoglou, T., Einarsdottir, G., Valsdottir, T., Schelvis, R., Skåra, T., Luten, J., 2010. A voiceof-consumer approach in development of new seafood product concepts. J. Aquat. Food Prod. Technol. 19 (2), 130–145. https://doi.org/10.1080/10498850.2010.493636.
- Altintzoglou, T., Verbeke, W., Vanhonacker, F., Luten, J., 2010. The image of fish from aquaculture among europeans: impact of exposure to balanced information. J. Aquat. Food Prod. Technol. 19 (2), 103–119. https://doi.org/10.1080/10498850.2010.492093.
- Altintzoglou, T., Honkanen, P., Whitaker, R.D., 2021. Influence of the involvement in food waste reduction on attitudes towards sustainable products containing seafood byproducts. J. Clean. Prod. 285. https://doi.org/10.1016/j.jclepro.2020.125487.
- Brooks, A.L., Wang, S., Jambeck, J.R., 2018. The Chinese import ban and its impact on global plastic waste trade. Sci. Adv. 4 (6), 1–8. https://doi.org/10.1126/sciadv. aat0131.
- Callegaro, M., Baker, R., Bethlehem, J., Göritz, A.S., Krosnick, J.A., Lavrakas, P.J., 2014. Online panel research: history, concepts, applications and a look at the future. Online Panel Research: A Data Quality Perspective. 2012, pp. 1–22. https://doi.org/10.1002/ 9781118763520.ch1.
- Cammarelle, A., Viscecchia, R., Bimbo, F., 2021. Intention to purchase active and intelligent packaging to reduce household food waste: evidence from Italian consumers. Sustainability (Switzerland) 13 (8). https://doi.org/10.3390/su13084486.
- Cardoso, C., Lourenço, H., Costa, S., Gonçalves, S., Nunes, M.L., 2013. Survey into the seafood consumption preferences and patterns in the portuguese populationGender and regional variability. Appetite 64, 20–31. https://doi.org/10.1016/j.appet.2012.12.022.
- Cardoso, C., Lourenço, H., Costa, S., Gonçalves, S., Leonor Nunes, M., 2016. Survey into the seafood consumption preferences and patterns in the portuguese population: education, age, and health variability. J. Food Prod. Mark. 22 (4), 421–435. https://doi.org/10. 1080/10454446.2014.949982.
- Carlucci, D., Nocella, G., De Devitiis, B., Viscecchia, R., Bimbo, F., Nardone, G., 2015. Consumer purchasing behaviour towards fish and seafood products. Patterns and insights from a sample of international studies. Appetite 84, 212–227. https://doi.org/10.1016/ j.appet.2014.10.008.
- Coelho, P.M., Corona, B., ten Klooster, R., Worrell, E., 2020. Sustainability of reusable packaging–current situation and trends. Resources, Conservation & Recycling: X 6 (November 2019), 100037. https://doi.org/10.1016/j.rcrx.2020.100037.
- Domegan, C., 2021. Social marketing and behavioural change in a systems setting. Curr. Opin. Environ. Sci. Health 23, 100275. https://doi.org/10.1016/j.coesh.2021.100275.
- EUMOFA, 2021. The EU fish market 2021 edition. European Market Observatory for Fisheries and Aquaculture Products https://doi.org/10.2771/563899.
- European Commission, 2018. DIRECTIVE (EU) 2018/852 OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL of 30 May 2018, Amending Directive 94/62/EC on Packaging and Packaging Waste. http://data.europa.eu/eli/dir/2018/852/oj.
- European Commission, 2020. Circular economy action plan. European Commission (Issue March) https://doi.org/10.2775/855540.
- European Environment Agency, 2021. Waste Recycling in Europe. . https://www.eea.europa. eu/ims/waste-recycling-in-europe.
- Eurostat, 2022. Packaging Waste Statistics. https://ec.europa.eu/eurostat/statistics-explained. Evans, J.R., Mathur, A., 2018. The value of online surveys: a look back and a look ahead. In-
- ternet Res. 28 (4), 854–887. https://doi.org/10.1108/IntR-03-2018-0089.
   Gao, X., Dall'erba, S., Ellison, B., Avelino, A.F.T., Yang, C., 2022. When one cannot bypass the byproducts: plastic packaging waste embedded in production and export. Journal of Industrial Ecology, 1–15 https://doi.org/10.1111/jiec.13282.
- Geyer, R., Jambeck, J.R., Law, K.L., 2017. Production, use, and fate of all plastics ever made. Sci. Adv. 3 (7), 3–8. https://doi.org/10.1126/sciadv.1700782.
- van der Harst, E., Potting, J., Kroeze, C., 2016. Comparison of different methods to include recycling in LCAs of aluminium cans and disposable polystyrene cups. Waste Manag. 48, 565–583. https://doi.org/10.1016/j.wasman.2015.09.027.
- Hélias, A., van der Werf, H.M.G., Soler, L.-G., Aggeri, F., Dourmad, J.-Y., Julia, C., Nabec, L., Pellerin, S., Ruffieux, B., Trystram, G., 2022. Implementing environmental labelling of food products in France. Int. J. Life Cycle Assess. 0123456789. https://doi.org/10. 1007/s11367-022-02071-8.
- Ilieva, J., Baron, S., Healey, N.M., 2002. Online surveys in marketing research: pros and cons. Int. J. Mark. Res. 44 (3).
- Iribarren, D., Hospido, A., Moreira, M.T., Feijoo, G., 2010. Carbon footprint of canned mussels from a business-to-consumer approach. A starting point for mussel processors and policy makers. Environ. Sci. Policy 13 (6), 509–521. https://doi.org/10.1016/j.envsci.2010.05.003.
- ISO, 2016. Packaging Vocabulary Part 1: General Terms. ISO21067-1. International Organization for Standardization.
- Klaiman, K., Ortega, D.L., Garnache, C., 2017. Perceived barriers to food packaging recycling: evidence from a choice experiment of US consumers. Food Control 73, 291–299. https:// doi.org/10.1016/j.foodcont.2016.08.017.

#### C. Almeida et al.

- Kole, A.P.W., Altintzoglou, T., Schelvis-Smit, R.A.A.M., Luten, J.B., 2009. The effects of different types of product information on the consumer product evaluation for fresh cod in real life settings. Food Qual. Prefer. 20 (3), 187–194. https://doi.org/10.1016/j.foodqual. 2008.09.003.
- Konstantinidis, E., Perdikaris, C., Ganias, K., 2021. Life cycle assessment during packaging of market-sized seabass and meagre: necessary adaptations toward GHG neutrality. Int. J. Life Cycle Assess. 26 (7), 1456–1470. https://doi.org/10.1007/s11367-021-01943-9.
- Laso, J., Margallo, M., Fullana, P., Bala, A., Gazulla, C., Irabien, A., Aldaco, R., 2017. Introducing life cycle thinking to define best available techniques for products: application to the anchovy canning industry. J. Clean. Prod. 155, 139–150. https://doi.org/10.1016/j. jclepro.2016.08.040.
- Lewis, H., 2005. Defining product stewardship and sustainability in the Australian packaging industry. Environ. Sci. Policy 8 (1), 45–55. https://doi.org/10.1016/j.envsci. 2004.09.002.
- Liu, G., Arthur, M., Viglia, S., Xue, J., Meng, F., Lombardi, G.V., 2020. Seafood-energy-water nexus: a study on resource use efficiency and the environmental impact of seafood consumption in China. J. Clean. Prod. 277, 124088. https://doi.org/10.1016/j.jclepro. 2020.124088.
- Love, D.C., Fry, J.P., Milli, M.C., Neff, R.A., 2015. Wasted seafood in the United States: quantifying loss from production to consumption and moving toward solutions. Glob. Environ. Chang. 35, 116–124. https://doi.org/10.1016/j.gloenvcha.2015.08.013.
- Martinho, G., Pires, A., Portela, G., Fonseca, M., 2015. Factors affecting consumers' choices concerning sustainable packaging during product purchase and recycling. Resour. Conserv. Recycl. 103, 58–68. https://doi.org/10.1016/j.resconrec.2015.07.012.
- Minnens, F., Marques, A., Domingo, J.L., Verbeke, W., 2020. Consumers' acceptance of an online tool with personalized health risk-benefit communication about seafood consumption. Food Chem. Toxicol. 144 (311820), 311820. https://doi.org/10.1016/j.fct.2020. 111573.
- Molina-Besch, K., Wikström, F., Williams, H., 2019. The environmental impact of packaging in food supply chains—does life cycle assessment of food provide the full picture? Int. J. Life Cycle Assess. 24 (1), 37–50. https://doi.org/10.1007/s11367-018-1500-6.
- Nemat, B., Razzaghi, M., Bolton, K., Rousta, K., 2019. The role of food packaging design in consumer recycling behavior—a literature review. Sustainability 11 (16), 4350. https://doi.org/10.3390/su11164350.
- Olsen, S.O., Tuu, H.H., Grunert, K.G., 2017. Attribute importance segmentation of Norwegian seafood consumers: the inclusion of salient packaging attributes. Appetite 117, 214–223. https://doi.org/10.1016/j.appet.2017.06.028.
- Pieniak, Z., Verbeke, W., Scholderer, J., Brunsø, K., Olsen, S.O., 2007. European consumers' use of and trust in information sources about fish. Food Qual. Prefer. 18 (8), 1050–1063. https://doi.org/10.1016/j.foodqual.2007.05.001.
- Poore, J., Nemecek, T., 2018. Reducing food's environmental impacts through producers and consumers. Science 360 (6392), 987–992. https://doi.org/10.1126/science.aaq0216.
- Poovarodom, N., Ponnak, C., Manatphrom, N., 2012. Comparative carbon footprint of packaging systems for tuna products. Packag. Technol. Sci. 25 (5), 249–257. https://doi.org/10. 1002/pts.975.
- Prendergast, G., Pitt, L., 1996. Packaging, marketing, logistics and the environment: are there trade-offs? Int. J. Phys. Distrib. Logist. Manag. 26 (6), 60–72. https://doi.org/10.1108/ 09600039610125206.
- van Putten, I.E., Farmery, A.K., Green, B.S., Hobday, A.J., Lim-Camacho, L., Norman-López, A., Parker, R.W., 2016. The environmental impact of two australian rock lobster fishery supply chains under a changing climate. J. Ind. Ecol. 20 (6), 1384–1398. https://doi. org/10.1111/jiec.12382.
- Rogers, P.J., 2007. Theory-Based Evaluation: Reflections Ten Years On. 114, pp. 63–81. https://doi.org/10.1002/ev.

- Rowan, N.J., Casey, O., 2021. Empower eco multiactor HUB: a triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new green Deal opportunities under a United Nations sustainable development goals fram. Curr. Opin. Environ. Sci. Health 21, 100254. https://doi.org/10. 1016/j.coesh.2021.100254.
- Rowan, N.J., Pogue, R., 2021. Editorial overview: green new deal era current challenges and emerging opportunities for developing sustaining and disruptive innovation. Curr. Opin. Environ. Sci. Health 22, 100294. https://doi.org/10.1016/j.coesh.2021.100294.
- Sacchettini, G., Castellini, G., Graffigna, G., Hung, C.Y., Lambri, M., Marques, A., Perrella, F., Savarese, M., Verbeke, W., Capri, E., 2021. Assessing consumers' attitudes, expectations and intentions towards health and sustainability regarding seafood consumption in Italy. Sci. Total Environ. 789, 148049. https://doi.org/10.1016/j.scitotenv.2021.148049
- Salgado, P.R., Di Giorgio, L., Musso, Y.S., Mauri, A.N., 2021. Recent developments in smart food packaging focused on biobased and biodegradable polymers. Front. Sustain. Food Syst. 5 (April), 1–30. https://doi.org/10.3389/fsufs.2021.630393.
- Salladarré, F., Brécard, D., Lucas, S., Ollivier, P., 2016. Are french consumers ready to pay a premium for eco-labeled seafood products? A contingent valuation estimation with heterogeneous anchoring. Agric. Econ. (U. K.) 47 (2), 247–258. https://doi.org/10.1111/ agec.12226.
- Sasaki, Y., Orikasa, T., Nakamura, N., Hayashi, K., Yasaka, Y., Makino, N., Shobatake, K., Koide, S., Shina, T., 2021. Life cycle assessment of peach transportation considering trade-off between food loss and environmental impact. Int. J. Life Cycle Assess. 26 (4), 822–837. https://doi.org/10.1007/s11367-020-01832-7.
- Steenson, S., Creedon, A., 2022. Plenty more fish in the sea ? is there a place for seafood within a healthier and more sustainable diet ? February, 261–273. https://doi.org/10. 1111/nbu.12553.
- Tallentire, C.W., Steubing, B., 2020. The environmental benefits of improving packaging waste collection in Europe. Waste Manag. 103, 426–436. https://doi.org/10.1016/j. wasman.2019.12.045.
- Van Selm, M., Jankowski, N.W., 2006. Conducting online surveys. Qual. Quant. 40 (3), 435–456. https://doi.org/10.1007/s11135-005-8081-8.
- Verbeke, W., Sioen, I., Pieniak, Z., Van Camp, J., De Henauw, S., 2005. Consumer perception versus scientific evidence about health benefits and safety risks from fish consumption. Public Health Nutr. 8 (4), 422–429. https://doi.org/10.1079/phn2004697.
- Wikström, F., Williams, H., Verghese, K., Clune, S., 2014. The influence of packaging attributes on consumer behaviour in food-packaging life cycle assessment studies - a neglected topic. J. Clean. Prod. 73, 100–108. https://doi.org/10.1016/j.jclepro.2013. 10.042.
- Wikström, F., Verghese, K., Auras, R., Olsson, A., Williams, H., Wever, R., Grönman, K., Kvalvåg Pettersen, M., Møller, H., Soukka, R., 2019. Packaging strategies that save food: a research agenda for 2030. J. Ind. Ecol. 23 (3), 532–540. https://doi.org/10. 1111/jiec.12769.
- Williams, H., Wikström, F., 2011. Environmental impact of packaging and food losses in a life cycle perspective: a comparative analysis of five food items. J. Clean. Prod. 19 (1), 43–48. https://doi.org/10.1016/j.jclepro.2010.08.008.
- Witter, A., Murray, G., Sumaila, U.R., 2021. Consumer seafood preferences related to alternative food networks and their value chains. Mar. Policy 131 (February). https://doi.org/ 10.1016/j.marpol.2021.104694.
- Zander, K., Feucht, Y., 2018. Consumers' willingness to pay for sustainable seafood made in Europe. J. Int. Food Agribusiness Mark. 30 (3), 251–275. https://doi.org/10.1080/ 08974438.2017.1413611.





# Empower Eco multiactor HUB: A triple helix 'academia-industryauthority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework Neil J. Rowan<sup>1</sup> and Orla Casey<sup>2</sup>

#### Abstract

There is a pressing drive to address climate change and environmental degradation that are global existential threats. Europe has strategically responded by unifying efforts to transform its connected communities into a modern, resource-efficient and competitive economy with a trajectory to enable net nil greenhouse gas emissions by 2050; thus, ensuring economic growth is decoupled from resource utilisation, and that no person or place is left behind. The European Green Deal is an ambitious plan to make the European economy sustainable; however, there is no reference blue-print for the safe and just transitioning to a low carbon economy. This constitutes the first description of a triple helix (academic-industry-authority) concept underpinning operation of multiactor innovation hub that can be strategically applied to enable this transition that develops green innovation and enterprises. Innovative tools for meeting the United Nations' Sustainable Development Goals are informed by appropriate technology, policy and society readiness levels from idea to final market/wider society deployment. "Empower-Eco Sustainability HUB," is a digitised "living lab" established in the Irish peatlands that converges academia, communities, social enterprises, industries, policy and decision-makers. It develops green innovation in intended environments at demo/test-beds, such as for digital, agri-food, bioeconomy and bio-based sectors, and embraces climate-proofing and COVID-19 recovery.

#### Addresses

<sup>1</sup> Bioscience Research Institute, Athlone Institute of Technology, Dublin Road, Athlone, Co. Westmeath, Ireland

<sup>2</sup> Momentum Educate + Innovate, Orchard Court, Leitrim Village, Co. Leitrim, Ireland

Corresponding author: Rowan, Neil J (nrowan@ait.ie)

Current Opinion in Environmental Science & Health 2021, 21:100254

This reviews comes from a themed issue on Environmental Toxicology 2021: Disruptive Green Deal Innovations

Edited by Neil J. Rowan and Robert Pogue

For a complete overview see the Article Collection https://doi.org/10.1016/j.coesh.2021.100254

2468-5844/© 2021 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

### Keywords

Just transition, New Green Deal, Sustainability, Open research, Multiactor hub, Circularity, UN Sustainable Development Goals.

# Introduction

Climate change and environmental degradation are recognised as existential threats that require urgent transnational action [1-3]. Ebi et al. [2] also stated that human health and wellbeing and the health of the biosphere are inextricably linked. Ebi et al. identified four main themes: (1) risk identification and management (including related to water, hygiene, sanitation and waste management); food production and consumption; oceans; and extreme weather events and climate change. (2) strengthening climate-resilient health systems; (3) monitoring, surveillance and evaluation; and (4) risk communication. They reported that research needs to be "transdisciplinary, multi-scalar, inclusive, equitable and broadly communicated; thus, promoting resilient and sustainable development are critical for achieving human and planetary health". The European Green Deal is an ambitious plan to make the European economy sustainable, which will inform a panoply of new policies, programmes and legislation that will propel Europe's Green Deal, Biodiversity, Farm to Fork and Circular Economy plans [4,5]; this is set against a backdrop of economic recovery, including changes to trade agreements due to BREXIT and ongoing COVID-19 pandemic [6].

Addressing pressing climate and environmental challenges will potentially create opportunities leading to sustaining or potentially new disruptive solutions [7]. For example, there is a pressing need to develop breakthrough 'circularity' solutions to replace the current 'take-makedispose' economy that will positively have an impact on food waste regeneration, resource utilisation and climate change [8]. Adopting nonconventional approaches to

Given his role as Guest Editor Neil J. Rowan had no involvement in the peer-review of this article and has no access to information regarding its peer-review. Full responsibility for the editorial process for this article was delegated to Damia Barceló Culleres.

research will avoid untapped outcomes that typically can occur in silos associated with traditional research, which limit their use and benefit to stakeholders. Unlocking the value of impact from research innovation in these areas remains a significant challenge due to inter alia barriers, lack of trust between different actors, perceptions of the social consequences to change, economic barriers to investment, and unclear returns [9]. Gaining an appreciation of the effective use, re-use and upcycle loops will inform next-generation eco-solutions that that will meet research gaps of enterprises along with tailoring education and training [1,10]. Recovering from the ongoing COVID-19 pandemic will create opportunities directly influencing our environment, our health and our society [11]. There is also a growing interest in supporting "One Health" as a global concept that recognises that humans, animals and their shared environments are interconnected. It is increasingly apparent that it is necessary to take a One Health approach to tackle many of the health challenges we face in today's world with nexus to environmental protection, resource utilisation and sustainablity.

# Developing and exploiting multiactor HUB approaches to unlock eco-innovation for new Green Deal era – lessons learnt from transnational modelling

One approach is to developing multiactor eco-innovation hubs that will drive open responsible research and innovation, such as in the areas of smart agri-food, forestry, marine, and bio-based systems [8,12]. Such a surge in interest will be informed by the outputs of transnational models and clusters that focus on drivers for change [1,13]. Multiactor HUBs will enable future solutions to improve waste mitigation and improve food security are likely to embrace open sharing of knowledge intra-regionally that can stimulate regeneration; this will be informed by life cycle assessment such as unlocking the nexus between food, energy and water to inform enterprise [14,15]. Eco-hubs will enable a structured 'holistic systems' approach that cuts across sectors and disciplines and engages multiple actors to deliver cobenefits for health, sustainability, climate and inclusion [1,12]. This is similar to the "triple-helix" concept of converging academia, industry and authorities to address complex challenges and to unlock green opportunities. For example, a circular food and resource supply can potentially become a reality where wasteful practices can stop, and circularity can be designed into all new products and services from the start [8]; however, for this to occur, a disruptive approach to channel effort will be required, including step changes in education that informs environmental literacy and behavioural change. Transnational modelling of technology core facilities and clusters for regional development, such as that described by European Interreg projects [1,13], have highlighted the need to develop multiactor hubs that link academia with industry and stakeholders. These create

opportunities for sharing infrastructure and skills that underpin the needs of SMEs linked to academia that potentially lead to advances in sustaining and new disruptive technologies [7,16,17]. Loorbach et al. [18] also conceptualised that the development of such transformative initiatives can occur through growing, replicating, partnering, instrumentalising and embedding; this is supported through translocal networks that connect initiatives by sharing ideas, objects and activities across local contexts. This need also extends to econometric models for assessing government policies to support SMEs in the context of meeting ongoing and future pandemics [19]. However, there has been a significant interest in advancing supply chain disruption and risk mitigation to manage COVID-19 and to support recovery, including improving the networking of multiactors and beneficiaries [20]. Development of Green innovation from idea to market introduction, along with wider societal deployment, will also be informed technology, policy and societal readiness levels that are increasingly being used by research funding bodies [9] (Table 1).

# Emerging opportunities using the peatlands

The peatlands account for ca. 3% of the earth's surface and as an important  $CO_2$  sink where it stores approximately 1.4 trillion tonnes of carbon that is equivalent to 75% of carbon in our atmosphere [1]. Several European-funded Interreg projects, such as NWE Carbon Connects [21], North Sea CANAPE [ 22], NWE Care Peat [23]), have focused on restoring peatlands and biodiversity, sequestering carbon and supporting wet-peatland innovation. 'Paludiculture' crops described in the aforementioned include sphagnum moss, moor-grasses, wild rice, typha and so forth. These transnational projects, along with EC's "Platform for Coal Regions in Transition" initiative, support transitioning away from exploiting fossil fuels to develop more climate-resilient solutions [1,24]. Paludiculture focused-activities could be extended to supporting enterprise and research in designated 'green development zones', such as for eco-design, testing and eco-labelling of new environmental-friendly products and services [25]. For example, Bord Na Mona, a State Body that manages 80,000 ha of 'organic status' peatland in the Republic of Ireland, has aggressively adopted such a new 'Brown to Green Strategy' that also includes controlled re-wetting of the peatlands for conservation along with forging climate solutions based on sustainable resource utilisation and green innovation. Consequently, many workers require retraining/upskilling in these areas. It is noteworthy that the global Green Technology and Sustainability market size is to potentially grow from USD 11.2 billion in 2020 to USD 36.6 billion by 2025, at a Compound Annual Growth Rate (CAGR) of 26.6% during the forecast period [26]; these include targeted technologies (such as IoT, AI and analytics, digital twin, cloud computing), and applications (such green building, carbon footprint management, weather monitoring and forecasting).

www.sciencedirect.com

Current Opinion in Environmental Science & Health

2021,

21:100254

Technology Readiness Levels <sup>a</sup> (TRL) <sup>a</sup>		Policy Readiness Levels <sup>b</sup> (PRL	)	Society Readiness Levels <sup>c</sup> (SFI	_)
<ul> <li>TRL 1 – Basic Research – Principles postulated and observed –no experimental proof (Discovery)</li> <li>TRL 2 – Technology Concept Formulated – concept and application defined (Concept Definition)</li> </ul>	Knowledge development Academia	<ul> <li>PRL 1 – Basic Research – identifying issue/problem and identifying policy readiness (Discovery)</li> <li>PRL 2 – Formulation of issue/problem, proposed solutions and potential impact; expected policy readiness; concept identification relevant to stakeholders (Concept Definition)</li> </ul>	Knowledge development Academia	<ul> <li>SRL 1 – Basic Research – identifying a problem and identifying societal readiness (Discovery)</li> <li>SRL 2 – Formulation of problem, proposed solution(s) and potential impact, expected societal readiness; identifying intended stakeholders for project (Concept Definition)</li> </ul>	ł
TRL 3 – Experimental Applied Research Concept – first laboratory tests completed (Proof of Concept)		PRL 3 – First testing of proposed solution(s) with relevant stakeholders; modelling, consultations, feedback, development complete (Proof of Concept).		SRL 3 – Applied Research - initial testing of proposed solution(s) with intended stakeholders (Proof of Concept)	
TRL 4 – Technology Validated in Lab – Small scale prototype – built and tested in lab (lab validation)		PRL 4 – Problem validated 'in lab' through pilot testing in intended environment to substantiate proposed impact, policy readiness, feedback development (lab validation)		SRL 4 – Pilot-Test Scale - concept validated through pilot testing in relevant environment to substantiate proposed impact and societal readiness (Concept Validation)	
TRL 5 Large-Scale Prototype tested in intended environment (test facility validation	Technology development Collaboration	PRL 5 – Proposed solution(s) validated; now by intended stakeholders in the area for application ('open water' validation)	Policy Development Collaboration	SRL 5 – Large Scale Test/system - proposed solution(s) validated; with intended stakeholders ('open water' validation)	S
TRL 6 – Technology demonstrated in intended environment – close to expected performance ('open water' validation)		PRL 6 – Demonstration system in intended environment and with intended stakeholders at pre-role out scale for feedback on impact (system demo)		SRL 6 – Demonstrated system - solution(s) demonstrated in relevant environment and with intended stakeholders for feedback on policy	
TRL 7 – System prototype demonstration in operational environment – at pre-commercial scale (system demo)		PRL 7 – System refinement of scheme and/or solution(s), and possibly, retesting in intended environment with intended stakeholders to gain feedback (refinement)		SRL 7 – System Refinement – refinement of product, and/or solution(s), and if needed, retesting in intended environment with stakeholders (refinement)	
TRL 8 – First system complete, qualified, verified – First commercial system – manufactured issues solved (verification)	Business development Industry	PTL 8 - First System - proposed solution(s), as well as plan for policy adaptation complete, and qualified (verification)	Scheme development Government	SRL 8 – First System – issues solved, proposed solution(s), as well as plan for societal adaption complete, and qualified (verification)	ę
TRL 9 – Actual Full commercial system proven in operational environment – technology available for beneficiaries (deployment)		PRL 9 – Policy Implementation – actual project solution(s) proven in relevant environment. Issues solved, continued monitoring, evaluation, and review of scheme/solution (deployment)		SRL 9 – Full Social System – actual project solution(s) in intended or relevant environment (deployment)	

a process, a management practice, an

<sup>a</sup> TRL – are indicators of status or maturity level of particular technology been researched and commonly used for European Commission in the context of Horizon Europe. Adapted from [9].

<sup>b</sup> PRL – Used to assess the level of societal adaptation to project, technology, product, process or management practice, or innovation to be integrated into policy. Adapted from [9].

<sup>c</sup> SRL – Used to assess the level of societal adaptation to project, technology, product, process or management practice, or innovation to be integrated into society. Adapted from [9].

Knowledge

Solution

Stakeholder

Society

development/

Development Collaboration

development Academia





Empower Eco Sustainability HUB" supporting a dynamic converging community and enterprise ecosystem for transitioning to the new Green Deal era; this fosters a triple-helix concept (academia-industry-authority) to enabling and accelerating change to green innovation in the Irish midlands.

# Empower Eco – Irish midlands multiactor platform to support and accelerate a just transition to low carbon economy

Empower Eco constitutes the first new Green Deal "multiactor hub" accelerator that supports and connects research, innovation, social enterprises along with enabling community transitioning to low carbon economy through engagement with broad stakeholder from a bottom-up perspective (Figure 1). It constitutes a signature multifunctional digitised building at Boora, in the heart of the Irish peatland that connects research, entrepreneurship, enterprise and the community (Figure 2). While the focus of Empower Eco is geared to

support and enable other green-innovative projects funded under Just Transition [24], it is specifically developing environmental testbed activities that include the use of 6 ha of freshwater aquaculture and aeroponics, 34 ha of horticulture comprising 12 different medicinal plants and herbs, 5 ha of agroforestry. Social enterprises include deploying 200 new hives across the peatlands to promote pollination and ecosystem service management, where 'Bell Heather' honey is produced at the regional Ferbane Food Hub (Figure 1). Activities conducted across Empower Eco ecosystem meet EC Green Deal ambitions [27], including (a) supporting, investing, testing environmentally-friendly technologies





Empower Eco Sustainability HUB has four inter-related foundation pillars connecting entrepreneurship, research, education and training, community and social activities that will guide just transition to low carbon economy in Irish midlands with a national and international orientation.

across agri-food, horticulture, agroforestry, verticalfarming, biofuels, circular economy and digitisation; (b) supporting industry to innovate, (c) decarbonising the energy sector - specifically expanding biofuel and wind energy initiatives, (d) ensuring that buildings are more energy-efficient and (e) working with international partners to provide global environmental standards and international mobility for SME training. Herrero et al. [16] also recently highlighted the potential for such innovation to accelerate the transition to sustainable food systems. Empower Eco can also escalate this approach by way of addressing eco-design and eco-labelling that focuses on green products ranging from concept to testing at pilot scale linked to LCA [25]; this also embraces carbon emissions and energy measurements using these peatland testbeds [21-23]. In addition, this provides useful testbed data for impact climate monitoring and forecasting.

The 'living lab' concept connects the need for real-time sophisticated analysis HUB infrastructure, such as using molecular profiling (such as MinION genomic sequencing) and bioreactors for mass-balance simulated studies to inform testing and validation of in-field monitoring devices at the testbeds [28]. Early needs analysis for market research, product-market fit analysis and design for scale—up processes are addressed that includes (1) early pilot and technical validation, including 'test the tech' (beta testing/field testing); (2) experimentation and validation at prepilot scale in reallife environments, (3) providing a commercial platform, and commercialisation vehicle for research for SMEs, and emanating from connected 3rd level institutions; and (4) embracing ICT, digitisation and exploit immersive technologies for linked research, innovation and training. Empower Eco activities strongly align across several policy areas, including biodiversity (measures to protect our fragile ecosystem); from farm to fork (ways to ensure more sustainable food systems); sustainable agriculture (sustainability in agriculture and rural needs informed by common agricultural policy (CAP)); clean energy (wind and biofuels); sustainable industry (ways to ensure more sustainable, more environmentally respectful production cycles); green building and renovating; eliminating pollution (measures to cut and monitor pollution rapidly and efficiently); climate action (making Ireland and the EU climate neutral by 2050, including creating monitoring of innovation and testbeds) [9].

Empower Eco will evaluate fundamental data trends, scenario modelling, mitigation measures, risk analysis in order to implement future policies that will strengthen the environmental sustainability and competitiveness of Irish and European partnering food industries. Empower Eco will address the development of datadriven and natural capital approaches to inform bioresources, businesses, services and value chains, which includes promoting new partnerships and innovation ecosystems between producers, processors, retailers and society. Examples of how this triple helix approach can be used as sustaining or potential disruptive green innovation for addressing United Nation's Sustainable Development Goals of this approach was also applied to

UN Sustainable Development Goal <sup>a</sup>	Indicative sustaining or potentially disruptive activity under linked Empower Eco triple-helix management platform
No poverty	Food security, food systems, and sustainability along with digitisation via open knowledge and technology sharing with developing countries. Researcher mobility and exchanges creating opportunities in education.
Zero hunger	Development and future proofing sustainable agri-food processes and
Good health and wellbeing	crops along with alternative sources for protein that includes training. Adopting One Health approach to informing green innovation – there is a strong focus on community transition and social enterprise for health – this includes food for good approach that includes provision for research
Quality Education	and innovation to pivot beyond COVID-19 [33]. Sharing knowledge, discoveries and growing collaborations in academia, ex-Delta Africa and Irish Research Council Coalesce Programme. Appointing Visiting Research Fellows from Developing Countries to Empower Eco projects for the dual translation of know-how that includes food for good, health and wellbeing that cross-cuts STEM with Social Science and humanities to inform behavioural change
Gender Equality	Empower Eco projects are strong gender equality focused with equal representation in research, innovation and entrepreneurship across projects, tasks, management that includes Project Advisory Team.
Clean Water and Sanitation	Innovative green research and enterprise to promote natural resources for water quality and mitigate waste that moves beyond end-of-pipe solutions [1,38].
Affordable and Clean Energy	Empower Eco enables and accelerates green innovations by developing affordable and clean energy that includes "Hub and spoke model" to testing and verifying technologies in intended environments using LCA, modelling and by applying technology, policy and society readiness levels [Table 1]. Carbon sink measurements on peatlands with nexus to specialist training [50,52]. Recirculating aquaculture system with 'organic status' in his postlands is driven by wind turbines [1, 38]
Decent work and Economic Growth	Empower Eco adopts a triple helix approach (academia-industry-authority) to inform techno-social-economic feasibility and impact that is framed upon pivoting activities to hurdle regional, national and international challenges (including COVID-19) informed by econometric and transnational modelling of key clusters for sustainable change
Industry, Innovation, Infrastructure	Empower Eco framework, or connected ecosystem, has assembled appropriate mix of academia, industries, and enterprises to accelerate green innovation that includes sharing and access to specialist infrastructure, equipment and resources where focus is on regional transition to low carbon economy and impact. With nexus to Education
Reduced Inequalities	Empower Eco triple helix management team approach that ensures education and research for all as its' core underpinning tenet.
Sustainable Cities and Communities	Empower Eco is focused on community transitioning to low carbon economy where there was a strong reliance on fossil fuels – community groups are strongly represented in management group and Empower Eco has registered as a 'Company with Limited Guarantee' to support not-for-profit and charitable activities to support its communities
Responsible Consumption and Production	Responsible consumption and production are core activities of Empower Eco in terms of research and enterprise – Empower Eco also develops digital twin, factory of the future, industry 4.0 concepts
Climate Action	Empower Eco addresses climate action by developing more sustainable green innovation that are climate proofed – this includes biosensor technology for monitoring more resilient innovation [36] – this is informed by networking with international partners such as EC Horizon 2020 and Interreg programmes that includes data from transnational models. Demo and test-beds in relevant and intended environments will render big data for climate impact modelling.
Life Below Water	Empower Eco is supporting freshwater aquaculture and studies on biodiversity related to natural aquatic operators in postlands
Life On Land	Empower Eco is supporting studies on biodiversity, pollination and ecosystem service management that includes studies on key drivers affecting decline of animal pollinators (bees) [3,35]. This is co-funded by the Environmental Protection Agency.

Empower Eco activities and tools to support, enable and accelerate potential green innovation disruption under United Nations' Sus-

Table 2 (continued)	
UN Sustainable Development Goal <sup>a</sup>	Indicative sustaining or potentially disruptive activity under linked Empower Eco triple-helix management platform
Peace, Justice and Strong Institutions	Peace, justice and strong institutions are key founding tenets of Empower Eco that blends academia, industry, authority with communities.
Partnerships for the goals	Empower Eco supports and enable national and international partnerships aligned with UN SDGs that includes mobility and training.

<sup>a</sup> Alignment with https://www.un.org/sustainabledevelopment/sustainable-development-goals/.

describing potential sustaining or disruptive tools for meeting needs of the United Nations Sustainable Development Goals Framework are provided in (Table 2).

# Bio-based and bio-inspired smart functional products from the peatlands

Due to its harsh and stressed environment [29], the peatlands of Europe provide a rich resource for refining new bio-inspired materials with improved functionality and sustainability for potential health and wellbeing applications, such as high value nutraceuticals, cosmetics and personal care products. This includes the development of novel bioactive compounds or ingredients for promoting health and wellbeing in humans and animals [30-33], along with testing and developing potentially disruptive green technologies [16,28,34,35]. Opportunities for harvesting immune-priming bioactives from food waste streams for potentially addressing the surge in antimicrobial resistance (AMR) that contributes to pulmonary sepsis and pneumonia should be pursued through a One Health approach [31,32]. Data-driven development solutions to optimise biomass availability and to inform the development of robust supply chains, business cases, including potential social enterprise models for sustainable and circular biobased value chains, will be advanced [9].

# Example of sustainable innovation using "Empower Eco" HUB: freshwater aquaculture

Aquaculture is the fastest growing food production section in the world [36]. It has an average annual growth rate of 6-8% since 2000 [36] that has been positively influenced by smart fishing and the increase in the seafood and aquaculture trade. Negative connotations associated with farmed seafood, such as pollution, poor animal welfare, nonsustainable depletion of resources, disease spreading and drug use, erodes the social license to produce [37]. Freshwater aquaculture in the Irish peatlands is framed upon using naturally occurring microalgae, bacteria and duckweed in the ponds as a means of water quality and waste recycling, where trout and perch are farmed based upon organic principles and use wind turbines [38]. The process is defined to optimise 'circularity' knowledge and to mitigate waste. This sustainable system moves away from end-of-pipe effluent treatment to embrace a full recirculation system; however, unexpected operational challenges have been experienced due to large rainfalls associated with multiple storms in 2020 that caused a 'flux' in the system, causing fish fatalities. This testbed also has exploited innovative environmentally friendly diagnostic and monitoring technologies, along with connected real-time determination potential of living labs (using real-time 'in lab' evaluation) with onsite handheld microalgae torch to monitor production processes [Figure 3]. This system will be digitised to inform the full process, validation and management decision tools that include the use of machine learning along with the potential for virtual Quality of Experience training [39]. Table 3 highlights the potential for future-proofing the aquaculture industry using green innovation developed through this triple-helix management approach.

# COVID-19 response and sustainability

There are increasing waves of SARS-CoV-2 infection globally that include more transmissible and pathogenic variants that exert enhanced pressure on personal and protective equipment (PPE) supply chains globally [40]. PPE shortages, such as filtering facepiece respirators or medical face masks, have been met by increased manufacture, by emergency use authorisation of nonthermal reprocessing technologies (such as VH2O2, UV) for PPE reuse and digitisation of the supply chain [41]. PPE is an important nonpharmaceutical intervention strategy that reduces the relative risk of SARS-CoV-2 infection and will continue to accompany vaccination for mitigating this COVID-19 disease and future pandemics. PPE that has single use-plastic items has generated a new waste management challenge where there are opportunities for using sustainable bio-inspired alternatives to that of using plastic constituents. These alternative bio-based solutions are to be potentially met from natural material such as food waste stream, bioplastic upcycling, and so forth [8]. For example, these biomaterials could be introduced by electrospinning to improve filtration efficacy below 300 nm cut pore size in face masks (where SARS-CoV-2 is typically 60–140 nm in size) [42,43]; however, other functional parameters must be met by original equipment manufacturers, namely pressure

#### Figure 3



Pressing need for end-to-end digital monitoring of big data and machine learning that connects (a) manual physiochemical parameters and (b) hand-help on-farm monitoring with (c) with real-time living lab analysis for (d) sustainable freshwater aquaculture using IMTA approach.

drop and comfort fit post nonthermal reprocessing to appropriate standards. The peatlands potentially provide a resource of alternative bioplastics to inform new PPE design and functionality for ongoing and future pandemics.

# Community-focused transitioning – pressure points and challenges to overcome

Transnational modelling, and regional observational studies have highlighted key drivers to improve the transition of communities and enterprises to low carbon economies [44, 45]. These include:

### Building resilient enterprises through

(a) extended supports, such as EI supports, learning from BREXIT/COVID-19 pandemic, transition vouchers; (b) SME digitisation, such as boosting awareness and take-up, simplify appreciation for innovation supports, (c) Enterprise-led networks, such as funding, lessons learned from existing clustering projects, (d) Early-movers, such as if and how to support initiatives.

# Continuous pre-emptive workforce development through

(a) increased ambition, such as improved resources, higher targets, incentivise employers; (b) improve

information, such as more skills audits, recognised informed skills and information sharking, (c) life-long learning, such as greater effort linked to academic institutions, and (d) better advice, such as preunemployment, more one-to-one coaching.

## Delivering high-impact targeted funded through

(a) targeted support, including social enterprises and new business/community level engagements and mechanisms, (b) improve support, such as seed funding applications, fund local plan development, and more place-based schemes, and (c) better engagement, such as working with multiactor institutions and monitoring EU developments.

#### Invest and enable green enterprises and innovation

There is a need to develop and invest in enterprises informed by expert training and research, including pilot testbeds that will accelerate new products and services and dissolve technical barriers [44,46–49]. Emerging business models to roadmap initiatives supporting this include (a) transnational business modelling tailored for circularity – such as developing further successful transnational econometric and cluster models [19,47]; (b) "Invest and enable for accelerating green innovations"; (c) Enterprise HPSU Green SPRINT Programme; (d) support from local enterprise office, local government, community groups, a policy such as Just

Current Opinion in Environmental Science & Health 2021, 21:100254

#### Table 3

Climate proofing sustainable freshwater aquaculture using Irish peatlands.

Challenge Area	Detail	Technical, Commercial, Social Resolution
Environment: Climate Change	Internationally, climate change can interact with fisheries in many different ways; a decline in the industry has been attributed in part to increased water temperatures, extreme water flow events (floods and droughts), and warming of maritime and freshwater environments [14,15,47].	Harvesting the potential for peatlands to help with new innovation to transition economy beyond same [38] and designing systems for recirculating avoiding end-of-pipe treatment options [1,28]. Using big data and digitisation from aquaculture testbeds to inform climate impact modelling along with biosensors that detect real-time variations in the aquatic environment due to extreme weather events [28,36].
Environment: Waste Streams and food waste redcuction	<ul> <li>Negative connotations associated with farmed seafood, such as pollution, poor animal welfare, nonsustainable depletion of resources, disease spreading and drug use have affected the development of aquaculture. Landbased fish farms require monitoring and treatment to ensure effluent containing nutrients and other physicochemical properties meet discharge to receiving water licence. Failure to achieve this can contribute to problematic algae blooms and fish death.</li> <li>Chemicals, such as antibiotics, used in the aquaculture industry, can be released into waterways. Future opportunities for Aquaculture systems to be closed or wastewater treated prior to discharge.</li> </ul>	<ul> <li>There has been an enhanced focus on valorising wastes from food production from farm to fork. For example, in the aquaculture industry, solid waste from finfish production has been identified as a potential substrate for anaerobic digestion with a secondary use as a fertiliser. However, there is a pressing need to leverage emerging 'natural 'processes to reduce operational cost and the environmental burden of food production for sustainability of the aquaculture sector. This presents opportunities for green innovation that will be informed by monitoring [37], Life Cycle Assessment, Risk modelling and cosbenefit analysis [15] and novel processes [16,34]</li> <li>Extraction of bioactives, or bioinspired materials with increased functionality from aquaculture waste streams (including from microalgae) that can be used in feed ingredients to promote immune health of farmed fish, thus limiting or avoiding the use of antibiotics</li> </ul>
Disease	Aquaculture operations can spread parasites and disease into the wild. Farmed fish have an increased chance of getting parasites such as sea lice, as opposed to fish that live and breed in their natural environments. Farmed fish are exposed to diseases through the use of unprocessed fish as a food source.	<ul> <li>[32,33,51].</li> <li>Introduce precision farming techniques that monitor and control disease and prevent infection. Utilise nutritional technologies to maintain the optimum health of the farm. Investigate symbiotic feeding systems that are biodiverse and sustainable. Opportunities for reviewing green innovations from a holistic technology, policy and societal readiness from initial idea to market or system deployment</li> </ul>
Cost Future need for food security, and disruptive innovation surrounding supply chain and resource utilisation	Treatments lead to higher costs and lower profitability The sustained expansion of aquaculture has arisen due to the increasing global population and commensurate demand for more food. Aquaculture now accounts for ca 50% of fishery products, which is estimated to reach ca. 62% by 2030 [37]. However, issues associated with the aquaculture licensing process and the potential environmental impact caused by aquaculture effluent have hampered the expansion of the Irish Aquaculture	[1able 2]. Build upon the potential demonstrated in the first radically-new concept in integrated-multitrophic aquaculture (IMTA) that uses cutaway peatland (bogs) to farm fish with an associated organic status that is powered by wind energy and utilises bacteria, microalgae, duckweed to treat rearing water. Developing IMTA systems, along with eco- innovation and monitoring/management processes, is critical for transforming the sector [38]. Use of triple-helix (academic-industry- authority) to inform new innovation linked to stakeholders and beneficiaries that also includes social change.
Feed	The feed-protein crisis has exposed the feed industry to shortages in quality protein and the need to import nonsustainably produced soy from places such as the amazon, and/or use wild-caught fish to grow fish.	Plant proteins constitute a novel avenue that has not been adequately explored in Ireland. The development of vertical farming to promote and develop high protein foods and feeds will also advance this area. It is envisaged that Irish aquaculture-farmed perch and trout benefit (continued on next page)

<sup>r</sup> able 3. ( <i>continued</i> )		
Challenge Area	Detail	Technical, Commercial, Social Resolution
Land Impact	Maintaining biodiversity, along with ensuring compliance with policies and statutory regulations, is important when	from consuming duckweed, which along with algae and bacteria, naturally control water quality and waste [1]. Utilising bogs that support natural ecosystems. This also enables the transition from burning peat to lower carbon emissions [1]. Approx.
	considering developing and deploying land-based facilities. Aquaculture businesses locate near coastlines for easy access to clean and natural water.	5% of Ireland is comprised of peat bogs, comprising potential aquaculture and other green enterprises. This will also support the community transitioning to a low carbon Bioeconomy.
COVID-19 crisis and recession	Presents uncertainly and considerable challenges and opportunities for food sustainability and security	Potential for peatlands to help with new innovation to transition economy beyond same, including agri-food and bio-based sectors informing by training [32,50,52].

Transition [24]; (e) entrepreneurship – such as Irelands New Frontiers delivered by Enterprise Ireland; mobility and specialist training for SMEs such as offered through the Hatch Blue Aquaculture Accelerator programme [50] and by National Digital Research Centre (NDRC) for accelerating start-up companies [52]. The aforementioned will also be informed by enabling innovation management through ISO 56000 series that will advance enterprises and grow capability in the green technology and service space.

# Conclusions

There is a pressing need to support communities in transitioning to low carbon economies where there is an underpinning drive to empower behavioural change to meet ambitious climate and environmental challenges through creative opportunities. Empower Eco is a multitiered ecosystem approach established to support and accelerate this just transition in the Irish midland peatlands that paradoxically supported rural communities through the burning of peat or supply to products to horticulture. The controlled wetting of the peatlands presents an essential carbon sink for Ireland to meet its net zero GHS emission targets for 2050; however, this also presents considerable innovative opportunities to create social enterprises and to support businesses in forging new green innovations to help with this community transitioning and to increase employment. While lessons have been learnt from outputs of transnational sustainability modelling, life cycle assessment and costbenefit analysis; Empower Eco constitutes the first multiactor connected ecosystem of stakeholders, spanning academia, industry, authority and communities, for informing a unified transition to low carbon economy using the Irish midland peatlands as a targeted region for regeneration. Unknowns yet to be discerned include elucidating the key drivers that inform sustainable behavioural change of citizens to low carbon economy. Also, the financial stimulus to support for enabling and accelerating green industry and businesses need to be ambitious and flexible to fast tract needs; these should also be linked to academia that can share technological core facilities and provide specialist training. For the impact of climate variance on emerging new sustainable activities, such as farm to fork, an example is the fully recirculated freshwater aquaculture process. Future studies will evaluate critically large-scale meta-data generated from these ambitious multiactor projects that will exploit advances in transnational and risk modelling, life cycle assessment and social marketing framed upon open sharing and access of knowledge. Therefore, Empower Eco describes a tool by which key green innovation and knowledge can be openly and directly transferred to decision makers and shared with stakeholders and beneficiaries. This triple-helix management approach also describes potential disruptive solutions that will inform green innovations such as required

under the UN Sustainable Development Goals (SDG) framework.

## **Declaration of competing interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

### Acknowledgement

The authors wish to acknowledge funding from Just Transition Fund (Project No. MRTT 112\_2021); Regional University Network - European Universities (Project No. 101004068) and Interreg Atlantic Area Neptunus (Project No. EAPA\_576/2018).

#### References

- Rowan NJ, Galanakis CM: Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: quo Vadis? Sci Total Environ 2020, https://doi.org/10.1016/ j.scitotenv.2020.141362.
- Ebi K, Harris F, Sioen GB, Wannous C, Anyamba A, Bi P, Boeckmann M, Bowen K, Cisse G, Dasgupta P, Dida GO, Gasparatos A, Gatzweiler F, Javadi F, Kanbara S, Kone B, Maycock B, Morse A, Murakami T, Mustapha A, Pongsiri M, Suzan G, Wantabe C, Capon A: Transdisciplinary research priorities for human and planetary health in the context of 2030 Agenda for sustainable development. Int J Environ Res Publ Health 2020, 17. 10.3390/ijerph17238890.
- All Ireland Pollinator Plan. 2020. https://pollinators.ie/. [Accessed 26 January, 2021].
- A European Green Deal. 2020. https://ec.europa.eu/info/strategy/ priorities-2019-2024/european-green-deal\_en. [Accessed 26 January, 202. [Accessed 26 January, 2021].
- European Green Deal striving to be the first climate neutral continent. https://ec.europa.eu/info/strategy/priorities-2019-2024/ european-green-deal\_en. [Accessed 1 January, 2021].
- Guan D, Wang D, Hallegatte S, Davis SJ, Huo S, Li L, Bai Y, Lei T, Xue Q, Coffman D, Cheng D, Chen P, Liang X, Xu B, Lu X, Wang S, Hubacek K, Gong P: Global supply-chain effects of COVID-19 control measures. Nature Human Behaviour 2020, https://doi.org/10.1038/s41562-020-0896-8.
- Schuelke-Leech B-A: A model for understanding the orders of magnitude of disruptive technologies. Technol Forecast Soc Change 2018, 129:261–274.
- Silva ALP, Prata JC, Walker TR, Campos TRW, Duarte AC, Soares AMVM, Barcelo D, Rocha-Santos T: Rethininkg and optimising plastic waste management under COVID-19 pandemic: policy solutions based on redesign and reduction of single-use plastics and person protective equipment. *Sci Total Environ* 2020. 10.1016/j.scitotenv.2020.140565; 2020.
- Department of Agriculture, Food and the Marine 2021 call for research proposals. https://www.gov.ie/en/publication/3d715dafm-announces-2021-call-for-research-proposals/. [Accessed 10 February, 2021].
- Sousa MJ, Rocha A: Skills for the disruptive digital business. J Bus Res 2018, https://doi.org/10.1016/j.busres.2017.12.051.
- Sarkis J, Cohen MJ, Dewick P, Schröder P: A brave new world: lessons from the COVID-19 pandemic for transitioning to sustainable supply and production resources. *Resour Conserv Recycl* 2020, 159, https://doi.org/10.1016/ j.resources.2020.104894.
- 12. Michelini L, Principato L, Iseavoli G: Understanding food sharing models to tackle sustainability challenges. *Ecol Econ* 2018, **145**:205–217.
- SHAREBIOTECH: Sharing life science infrastructure and skills to benefit the Atlantic area biotechnology sector. 2013. https://keep.

eu/projects/440/Sharing-life-science-infrastr-EN/. Accessed 10 February 2021.

- Ruiz-Salmón I, Laso J, Margallo M, Morse AP, Cooney R, Clifford E, Regueiro L, Medez D, Anglada C, Noirot C, Rowan N, Vasquez-Rowe I, Aldaco R: Life cycle assessment of fish and seafood processed products – a review of methodologies and new challenges. *Sci Total Environ* 2020, **761**, https://doi.org/ 10.1016/j.scitotenv.2020.144094.
- Ruiz-Salmon I, Margalo M, Laso J, Villanueva-Rey P, Marino D: Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Current Opinion Environmental Science Health 2020, 13:101–106.
- Herrero M, Thornton PK, D'Croz DM, Palmer J, Benton G, et al.: Innovation can accelerate the transition towards a sustainable food system. Nature Food 2020, 1:266–272.
- Loorbach D, Wittmayer J, Avelino F, von Wirth T, Frantzeskaki N: Transformative innovation and translocation diffusion. Environmental Innovations Societal Transitions 2020, 35:251–260.
- Yongfu S, Liheng W, Zontan S, Kunsheng W, Liangyan JH, Jian C, *et al.*: Connotation and selection of disruptive technologies that lead industrial change. *Strategic Study CAR* 2017, https://doi.org/10.153302/J-SSCAE-2017.05.002.
- Razumovskaia E, Yuzvovich L, Kniazeva E, Klimenko M, Shelyakin V: The effectiveness of Russian government policy to support SMEs in the COVID-19 pandemic. J. Open Innov Technol Mark Complex 2020, 6:160, https://doi.org/10.3390/ joitmc6040160.
- Shahed KS, Azeem A, Ali S,M, Moktadir MA: A supply chain disruption risk mitigation model to manage COVID-19 pandemic risk. *Environ Sci Pollut Control Ser* 2021, 275, https:// doi.org/10.1007/s11356-020-12289-4.
- Carbon Connects. 2020. https://www.nweurope.eu/projects/ project-search/cconnects-carbon-connects/. [Accessed 26 July, 2020].
- 22. CANAPE. 2020. https://northsearegion.eu/canape/. [Accessed 26th July 2020].
- Care-Peat. 2020. Restoring the carbon capacity of the peatlands https://www.nweurope.eu/projects/project-search/care-peatcarbon-loss-reduction-from-peatlands-an-integrated-approach/. [Accessed 26th July 2020].
- Department of the Environment, Climate and communications just transition fund. https://www.gov.ie/en/publication/ed10d-justtransition-fund/. [Accessed 10 Febraury 2020].
- Interreg Atlantic area NEPTUNUS project. Energy-Water-Seafood Nexus: eco-Innovation and ciruclar economy strategies in the Atlantic area. https://neptunus-project.eu/. [Accessed 10 Febraury 2020].
- MarketsandMarkets. 2020. https://www.marketsandmarkets.com/ top-market-reports.asp . [Accessed 10 Febraury 2020].
- European Commision: First "Green Deal" funding from European Innovation Council awards over €307 million to 64 startups set to support the recovery plan for Europe. https:// ec.europa.eu/info/news/first-green-deal-funding-european-innovation-council-awards-over-eu307-million-64-startups-set-support-recovery-plan-europe-2020-jul-23\_en. Accessed 26 July 2020.
- Naughton S, Kavanagh S, Lynch M,, Rowan NJ: Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: a case study from the Republic of Ireland. *Aquaculture* 2020, 526, https://doi.org/10.1016/ i.aquaculture.2020.735377.
- Robroek BJM, Jassey VE, Payne RJ, Martí M, Bragazza L, Bleeker A, Buttler A, Caporn SJM, Dise NB, Kattge J, Zajac K, Svensson BH, Ruijven JV, Verhoven: Taxonomic and functional turnover are decoupled in European peat bogs. Nat Commun 2017, 8:1161, https://doi.org/10.1038/s41467-017-01350-5.
- Galanakis CM, Aldawoud T, Rizou M, Rowan NJ, Ibrahim SA: Food ingredients and active compounds against coronavirus disease (COVID-19) pandemic – a comprehensive review. Foods 2020, 9:1701.
- Masterson CH, Murphy EJ, Gonzalez H, Major I, McCarthy SD, O'Toole D, Laffey JG, Rowan NJ: Purified beta-glucans from Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumonia-induced pulmonary sepsis. Lett Appl Microbiol 2020, 71:405–412.
- 32. Murphy EJ, Masterson C, Rezoagli E, O'Toole D, Major I, Stack GD, Lynch M, Laffey JG, Rowan NJ: β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects — implications for coronavirus disease (COVID-19) immunotherapies. Sci Total Environ 2020, 732:139330.
- Galanakis CM, Rizou M, Aldawoud TMS, Ucak I, Rowan NJ: Innovations and technology disruptions in the food sector within the COVID-19 pandemic and post-lockdown era. *Trends Food Sci Technol* 2021, 110:193–200.
- Rowan NJ: Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo Vadis? *Trends Food Sci Technol* 2019, 88:316–332.
- Naughton J, Tiedeken EJ, Garvey M, Stout JC, Rowan NJ: Pulsed light inactivation of the bumble bee trypanosome parasite Crithidia bombi. J Apicult Res 2017, 56:144–154.
- O'Neill EA, Rowan NJ, Fogarty AM: Novel use of alga Pseudokirchneriella subcapitatia, as an early-warning indicator of identify climate change ambiguity in aquatic environments using freshwater finfish. Sci Total Environ 2019, 692:209–218.
- Tahar A, Kennedy A, Fitzgerald RD, Clifford E, Rowan N: Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 2018, 3:28.
- O'Neill EA, Stejskal V, Clifford E, Rowan NJ: Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. *Sci Total Environ* 2020, 706, https://doi.org/10.1016/j.scitotenv.2019.136044.
- Braga Rodriguez T, O' Cathain C, O'Connor NE, Murray N: A Quality of Experience assessment of haptic and augmented reality feedback modalities in a gait analysis system. *PloS One* 2020, 15, e0230570, https://doi.org/10.1371/ journal.pone.0230570.
- 40. Rowan NJ, Laffey JG: Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from the Coronavirus disease (COVID19) pandemic- case study from the Republic of Ireland. Sci Total Environ 2020, 725:138532.
- 41. Rowan NJ, Laffey JG: Unlocking the surge in demand for personal and protective equipment (PPE) and improvised

face coverings arising from Coronavirus disease (COVID-19) pandemic – implications for efficacy, re-use and sustainable waste management. *Sci Total Environ* 2020, **752**:142259.

- Rowan NJ, Moral AM: Disposable face masks and reusable face coverings as non-pharmaceutical interventions (NPIs) to prevent transmission of SARS-Cov-2 variants that cause coronavirus disease (COVID-19): role of new sustainable NPI design innovations and predictive mathematical modelling. *Sci Total Environ* 2021, **772**, https://doi.org/10.1016/ i.scitotenv.2021.145530.
- Rubio-Romero JC, del Carmen M, Ferreira P, Torrecilla JA, Santiago G, Castro C: Disposable masks: disinfection and sterilization for reuse, and non-certified manufacturing, in the face of shortages during the COVID-19 pandemic. *Sci Total Environ* 2020, https://doi.org/10.1016/j.ssci.2020.104830.
- National Economic and Social Fund.: Four case studies on just transition – lessons from Ireland. https://www.nesc.ie/ publications/four-case-studies-on-just-transition-lessons-forireland/. Accessed 10 February 2021.
- 45. Geels FW: Disruption and low-carbon system transformation; progress and new challenges in socio-technical transitions research and the multi-level perspective. Energy Res Social Science 2018, 86:2317–2324.
- Saguy IS, Roos YH, Cohen E: Food engineering and food science and technology: forward looking journey to future horizons. Innovat Food Sci Emerg Technol 2018, 47:326–334.
- Tavares AO, Areia NP, Mellett S, James J, Intrigliolo DS, Gouldrick LB, Berthoumieu JF: The European media portrayal of climate change: implications for social mobilization towards climate action. Sustainability 2020, 12:8300, https:// doi.org/10.3390/su12208300.
- Lajoie EW, Bridges L: Innovative decisions using the gartner hype cycle. Library Leadership Management 2014, 28, https:// doi.org/10.5860/llm.v28i4.7083.
- Li M, Porter AL, Suominen A: Insights into relationship between disruptive technology/innovation and emerging technology – a bibliometric perspective. J Food Eng 2018, 220: 20–27.
- 50. Blue Hatch Aquaculture Accelerator Program. https://www.hatch. blue/program. Accessed 10 February 2021.
- 51. Wan-Mohtar WAAQI, Taufek NM, Fhiran JP, Rahman JFP, Yerima G, Subramamiam K, Rowan NJ: Investigations on the use of exopolysaccharides derived from mycelial extract of *Ganoderma lucidum* as functional feed ingredient for aquaculture-farmed red hybrid Tilapia (Oreochromis sp.). *Future Foods* 2021, https://doi.org/10.1016/ j.fufo.2021.100018.
- Darmody J: Dogpatch labs and regional hubs win €17m NDRC contract. 2020. https://www.siliconrepublic.com/start-ups/ dogpatch-labs-ndrc-contract. Accessed 14 March 2021.



Contents lists available at ScienceDirect

### Journal of Cleaner Production



journal homepage: www.elsevier.com/locate/jclepro

#### Review

## A circular economy framework for seafood waste valorisation to meet challenges and opportunities for intensive production and sustainability

Ronan Cooney <sup>a,b</sup>, David Baptista de Sousa <sup>c</sup>, Ana Fernández-Ríos <sup>d</sup>, Sinead Mellett <sup>e</sup>, Neil Rowan <sup>e</sup>, Andrew P. Morse <sup>f</sup>, Maria Hayes <sup>g</sup>, Jara Laso <sup>d</sup>, Leticia Regueiro <sup>c</sup>, Alex HL. Wan <sup>a,b,h</sup>, Eoghan Clifford <sup>a,b,\*</sup>

<sup>a</sup> School of Engineering, University of Galway, Galway City, H91 HX31, Ireland

<sup>b</sup> Ryan Institute, University of Galway, Galway City, H91 TK33, Ireland

<sup>c</sup> ANFACO-CECOPESCA, Department of Circular Economy, Campus University 16, 36310, Vigo, Spain

<sup>d</sup> Department of Chemical and Biomolecular Engineering, University of Cantabria, Av. de Los Castros, s.n., 39005, Santander, Spain

<sup>e</sup> Bioscience Research Institute, Technological University of the Shannon, Midlands Midwest Athlone Campus, N37 HD68, Ireland

f School of Environmental Sciences, University of Liverpool, Liverpool, UK

<sup>g</sup> Department of Food BioSciences, Teagasc Food Research Centre, Ashtown, Dublin, D15 DY05, Ireland

h Aquaculture Nutrition Research Unit (ANRU), Carna Research Station, Ryan Institute and School of Natural Sciences, University of Galway, Carna, Connemara, Co.

Galway, H91 V8Y1, Ireland

ARTICLE INFO

Handling Editor: Cecilia Maria Villas Bôas de Almeida

Keywords: Seafood loss Aquaculture Nutrient recovery Seafood supply chain Processing by-products

#### ABSTRACT

There is a growing concern among societies and consumers over food security and the sustainability of food production systems. For seafood, it has been highly advocated as a healthy food source and its sustainability credentials. However, the increasing global demand for seafood and the need to supply the quantities are creating sustainability issues, e.g., the importation of plant and marine proteins for aquafeed production. Consequently, there is a necessary need to analyse the supply chain and life cycle of these systems to determine their sustainability merits and how to enhance them. The circular economy (CE) aims to reduce processing byproduct underutilisation, increase the rate of reuse, and reduce pressure on natural resources and systems. For seafood, there are large quantities of biomass that are being lost through bycatch/discards, waste from aquaculture (e.g., sludge and wastewater), and by-products generated through processing (e.g., trimmings and offal). These can all be valorised for the generation of feeds, value-added products, or further food production. This review will focus on seafood by-products generated during the processing into consumer products, and the current methods that could be used to manage or treat these waste streams. The review presents a stepwise framework that outlines valorisation opportunities for seafood by-products. This framework can enable producers, operators, regulators, and investors to integrate with the principles of the CE with the consideration of achieving economic viability. The challenges of seafood loss due to climate change and emerging recycling strategies will also need to be considered and integrated into the valorisation pathways. Communication, education, and engagement with stakeholders are key to transitioning to a circular economy. Where increase awareness and acceptance will create drivers and demand for seafood by-product valorisation. Overall, the impact of such a circular production system will potentially lead to higher production efficiency, reduce demand for natural resources, and greater seafood production. All of which addresses many of the United Nation's Sustainable Development Goals by contributing towards future food security and sustainability.

#### 1. Introduction

Human activities contribute to the significant decline in environmental quality and biodiversity. To the present date, interactions between humanity and the environment have been based on a model of extraction, processing, production, and discarding the unused products and the by-products produced along the core product as waste back into the environment (Tan and Lamers, 2021). This linear economy model is

\* Corresponding author. School of Engineering, University of Galway, Galway City, H91 HX31, Ireland. *E-mail addresses:* rcooney@nuigalway.ie (R. Cooney), eoghan.clifford@universityofgalway.ie (E. Clifford).

https://doi.org/10.1016/j.jclepro.2023.136283

Received 23 May 2022; Received in revised form 27 January 2023; Accepted 30 January 2023 Available online 2 February 2023 0959-6526/© 2023 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/). no longer compatible with the current capacities of this planet (Borrello et al., 2020; European Commission, 2020b; Laso et al., 2018; Regueiro et al., 2021). With a growing population compounded by a growing middle class with increased spending power, the global demand for food is increasing (Belton et al., 2020; Béné et al., 2015; Fernández-Ríos et al., 2021; Rohm et al., 2017). In particular, the demand for seafood and seafood products is rising also due to the advocacy as part of a healthy diet (Bohnes et al., 2020; Regueiro et al., 2021).

Seafood is a colloquial and highly broad food category. For example, in much of Europe, this often encompasses both freshwater and marine finfish species (e.g., farmed salmon, trout, carp, seabass, seabream), bivalves, (e.g., mussels, clams and oysters), decapods, (e.g., crabs, shrimp, and lobsters), cephalopods (e.g., squid and octopus), but also algae (macro and micro) and cyanobacteria. These diverse groups of seafood are considered to have a better environmental performance than other protein-rich foods, such as terrestrially farmed animals (European Commission, 2019, 2020a). Their production is either derived through farming (aquaculture) or wild caught (fisheries). Together these food production systems produced 214 million tonnes of global seafood in 2020 and over half of this production was from aquaculture (FAO, 2022). The Food Agricultural Organisation of the United Nation has estimated that the seafood trade is worth USD 151 billion and is projected that this will further grow by over 13% in production value in 2030. Much of this growth will be driven by aquaculture due to its increasing efficiency, farmed species diversification, and new production opportunities.

Like many other food production systems, seafood must employ the concepts of life cycle thinking and the circular economy (CE) to increase production efficiency and mitigate its environmental impact (Cortés et al., 2021b; Ruiz-Salmón et al., 2021). This is either through valorisation strategies or nutrient recovery technologies (de la Caba et al., 2019; Venugopal, 2021). Within the EU, policies for increased seafood consumption are being introduced to increase food and seafood circularity (European Commission, 2020b). The Circular Economy Action Plan (European Commission, 2020b) was launched by the European Union (EU) as part of the European Green Deal. The plan aims to transition the European economic bloc from a linear to a circular economy. This transition has been advocated to open new avenues for resource efficiency and places the concept of life cycle thinking as its core action (Gheewala and Silalertruksa, 2021; Ruiz-Salmón et al., 2020). This drive for circularity in value chains and processes will help to support (Raimondo et al., 2021; Regueiro et al., 2021; Zilia et al., 2021).

- (i) Greater use of renewable energy,
- (ii) More responsible use of resources and, crucially,
- (iii) Reuse and valorisation of by-products and residue streams generated from seafood processing.

To combat the seafood nutrients, energy, and elemental loss along the supply chain, there is a need to identify value in this lost material. Thereafter, there is a need for novel designs for the valorisation and exploitation of seafood by-products, as well as the promotion of more environmentally, socially, and economically sustainable business models. Furthermore, any proposed solutions should contribute to increasing seafood circularity by maximising the potential value that can be derived. The underutilisation of by-products is regarded as a wasted opportunity. Therefore, the present review aims to identify the opportunities for CE within seafood production chains. This will be achieved by using a stepwise valorisation framework, with a particular focus on.

- (i) The current protocols for seafood by-product treatment,
- (ii) Opportunities for CE,
- (iii) Emerging seafood loss, and
- (iv) Emerging strategies for CE and opportunities.

Furthermore, this review will evaluate the barriers that prevent or limit seafood circularity and potential solutions. Overall, this review will form a catalyse in allowing researchers, industry, and policymakers in focusing key innovations to drive a CE in seafood.

#### 2. Research methodology

The present review was carried out by searching for peer-reviewed studies relevant to the area of seafood waste, by-product valorisation, and the circular economy. Relevant literature was reviewed using academic databases (inc. Scopus, Google Scholar, Web of Science and Science Direct). The terms used for the literature search were "seafood loss", "seafood waste", "seafood nutrient recovery", "aquaculture waste/ loss", "fisheries waste/loss", "seafood circular economy" and "seafood circularity". The inclusion criteria for the results were that the articles had to be peer-reviewed and published in the English language and must have been published in the last 10 years (2010-2022). The exclusion criteria that were applied focused on the thematic relevance of the article and that it not be an opinion article, conference article, or from the grey literature. Further refinement of the results was reached by implementing the stepwise approach to key areas of seafood waste that the article aimed to review. These areas were (i) seafood waste streams and (ii) current waste treatment protocols. The next thematic area focused on approaches which incorporate the circular economy into seafood waste management, (iii) nutrient recovery technologies, (iv) nutrient recovery strategies and management practices, (v) seafood loss, (vi) recycling strategies and (vii) bio-based resources.

Using these search terms and criteria, 142 articles derived from these searches were then broken down into the relevant thematic areas. This critical analysis of this article concludes with a discussion of the challenges with recommendations for implementation and actualisation, which will transform societies from a linear into a circular economy. This analysis will be European-centric due to the over-exploitation of aquatic environments and the economic bloc's 88 million tonnes of food waste per annum (European Commission, 2022).

#### 3. Current practices for seafood by-products

Within seafood production systems, large quantities of by-products are generated from both wild capture fisheries and farmed aquatic species-aquaculture. Very frequently these by-products are disposed of as waste or discharged into the aquatic environment. This brings the need to implement effective and novel treatment and utilisation protocols to reduce any environmental impact on aquatic environments or land and promote more efficient production practices that reduce biomass, energy, or nutrient losses (Caruso, 2016).

Within the seafood category, there are significant volumes of byproducts being generated from the production, processing, distribution, consumption, and disposal stages (Hayes and Gallagher, 2019; Venugopal, 2021). It has been estimated that as much as 36% of seafood can be lost or wasted (FAO, 2018; Gustafsson et al., 2013). These losses are often complex and each by-product streams are unique from different seafood production systems which leads to varying composition, quantities, and quality (Fig. 1). Consequently, this results in the need for different technological requirements in its management after its produced.

In terms of by-product generation, there are a variety of sources from the seafood sector, particularly depending on the level being studied. Byproduct volumes, value and quality can vary from species to species, between regions, availability, and at different stages of the supply chain. For example, the use of pond culturing systems (flow through aquaculture) are extensively used for freshwater aquaculture across the world (Bohnes et al., 2019; Bohnes and Laurent, 2019; FAO, 2018). Its attractiveness to farmers is the low technology required to set up and maintain but more importantly can be easily built with limited cost in relation to aquatic farm systems. In many instances, pond systems may



Fig. 1. An overview of the several types of seafood, their production/capture systems, and the types of by-products that are generated.



Fig. 2. A proposed framework for top-down CE strategies for seafood by-product valorisation. Animal by-products categories are presented as Cat. The triangle indicates the added value of the products and volume needed.

not allow for efficient process control, this can result in wasted and uneaten feed that can settle and accumulate over time on the pond floor. This uneaten feed can become a nutrient-rich layer that provides a substrate for microbes to convert and breakdown down biochemically (Dauda et al., 2019). This layer can also reduce process efficiencies by consuming oxygen, thus requiring larger amounts of supplementary aeration (Tahar et al., 2018). The low technology requirements of pond-based aquaculture are in contrast to the high technology requirements of recirculating aquaculture systems. These closed or semi closed aquaculture systems re-use a large proportion of the water by undergoing treatment processes. Commercial recirculating aquaculture systems as mechanical solids removal, bioreactors, heating, and cooling, and ozonation and/or ultraviolet sterilisation (Martins et al., 2010). From the solid's removal, sludge which comprises uneaten feed and biogenic wastes is captured and stored on the site for further treatment. Generally, this sludge is treated in centralised facilities such as publicly owned treatment works used for the treatment of other livestock waste as well as domestic and industrial waste (van Rijn, 2013). Other routes of food loss and waste in open aquaculture systems can be due to disease outbreaks and environmental events such as jellyfish blooms, and algal blooms which can result in mortality events (Brooks et al., 2022; Clinton et al., 2021).

Within fisheries, the main by-products are by-catches and discards. The former is classified as non-targeted caught species which can impact marine food webs, e.g., cetaceans, echinoderms, and molluscs (Bielli et al., 2020). While the latter are species that are captured which may not be of suitable grade (e.g., below harvest size), or economic value, or the fishers may not have a quota for the species. It has been estimated by

the EU that between 7 and 10 million tonnes of fish are discarded annually across the world (EC, 2022). Like bycatches, discards are returned to the sea dead which has led to a significant negative public image of fisheries (FAO, 2020). In recent years there has been a growing trend to utilise these fish for the production of fishmeal, fish oil, fertilisers, biostimulants, and even food ingredients for human consumption (Dineshbabu et al., 2013; Madende and Hayes, 2020). This trend is supported by several mitigation strategies, e.g., the EU's Common Fisheries Policy on discards ban and landing obligation. They aim to reduce the levels of bycatch and discards through modifications that include the mesh size, use of fisheye devices (FAO, 2020), implementation of circle hooks, alternate baits in longline fisheries, improvements in remote sensing, and animal tracking technologies on vessels (Komoroske and Lewison, 2015).

While the volumes of by-products from aquaculture and fisheries seem high, the major source of lost or unused material can be found in

the processing stage. By-products generated during processing typically consist of biomass produced during filleting and preparation of processed seafood products direct to the end consumer. For fish byproducts, there is the skin, bones, trimmings, heads, offal, shells and byssal threads from bivalve and mollusc species (Table 1). By-products should be treated on-site as food grade if required for food ingredient generation or feed grade if processed further by approved animal byproduct operators. In some instances, these may be discharged to the marine environment, sent to sewage treatment plants, or disposed of in landfills (Cadavid-Rodríguez et al., 2019). For example, "stick-water" produced from the processing of fish (e.g., blood, mucus, and residue muscle proteins) can be an issue due to their quantities being generated and the requirement for sanitary disposal. However, in a number of countries, actions have been taken to reduce and valorise these by-products into food-grade and value-added food ingredients, such as collagen, chitosan, and proteins (Erasmus et al., 2021; Mathew et al.,

#### Table 1

0 1 1	1 . 1		c 1 · .·	1	1. 1
Citreront contood r	w nroducte and	TATOCTO CONTROOC	tor walorication	biorofining	or dicpocol
CULLETH SEALOOU L	//- // // // // // // // // // // // //		IOI VAIOLISALIOIL.	DIOLCHIIIII 2.	UL ULSDUSAL

Stage	Treatment/Process	Current waste hierarchy option	Seafood by-products	Current management strategy	Reference
Aquaculture	Use of pond systems	Disposal	Uneaten feed and sludges	Accumulation of organic by-products at the bottom of the system, where microbes act converting it to less toxic material	Dauda et al. (2019)
	Use of recirculating aquaculture systems	Disposal	Uneaten feed and sludges	Partial removal of organic by-products from aquaculture water through sedimentation and filters. Subsequent treatment in publicly owned wastewater treatment plants.	Martins et al. (2010)
Fishing	No treatment	Disposal	Bycatch	Return of the dead catch to the sea	FAO (2020)
Ū	Fishmeal production	Valorisation	Fish bycatch	Production of fishmeal in processing plants destined for aquafeed products	Dineshbabu et al. (2013)
	Fertiliser production	Valorisation	Fish bycatch	Production of liquid or solid fertilizer, silage, or compost through fermentation or composting	
	Mitigation	Prevention	Fish bycatch	Modifications in mesh size Use of fisheye devices Implementation of circle hooks and alternate baits in longline fisheries Improvements in remote sensing and animal tracking technologies	FAO (2020) Komoroske and Lewison (2015)
Processing	No treatment	Disposal	Heads, bones, offal, and skin	Waste is eliminated through its discharge into the ocean or its disposal in landfills	Cadavid-Rodríguez et al. (2019)
	Bait	Valorisation/ biorefining	Heads, bones, offal, and skin	Valorisation of by-products for fish bait	Masilan et al. (2021)
	Silage production	Valorisation	All fish and selfish by- products	Production of silage (protein-rich liquid) through enzymatic hydrolysis of fish by-products for aquatic and terrestrial animals feeding	Islam and Peñarubia (2021)
	Fishmeal production (human consumption)	Valorisation	Heads	Production of meal in fishmeal processing plants for human consumption, such as fish mince, fish smoked sausages, or fish patties	Erasmus et al. (2021)
	Fishmeal production (animal feeds)	Valorisation	All fish by-products	Production of aquafeed for a low trophic level or warm water fish (like carp or tilapia)	Saleh et al. (2020)
	Functional food	Valorisation/ Biorefining	All fish by-products	Production of functional foods or products such as collagen, peptides, chitin, enzymes, and gelatin for human consumption	Coppola et al. (2021)
	Fertilizers production	Valorisation	Bones, heads, viscera, and wastewater (blood water)	Production of biostimulants and liquid fertilisers through degradation of viscera, biological treatment of wastewater for agricultural application	Kim et al. (2010) Kim (2011) Ching and Redzwan (2017)
	Biofuels production	Valorisation/ Biorefining	Offal and bones	Extraction of oils from fish waste and subsequent biogas or biodiesel production and by-production of glycerine for pharmaceutical, food and cosmetic applications	Karkal and Kudre (2020) Jayathilakan et al. (2012) El-Gendy et al. (2014)
	Omega-3 fatty acids extraction	Valorisation/ Biorefining	All fish by-products	Extraction of omega-3 fatty acids through anaerobic digestion for food applications (food supplements)	Mbatia et al. (2010) Nges et al. (2012)
	Collagen extraction	Valorisation/ Biorefining	Skin, scales, fins, and bones	Extraction of collagen for food, cosmetic, pharmaceutical, tissue engineering, and biomedical industries	Bhuimbar et al. (2019) Araujo et al. (2021)
	Chitin and chitosan extraction	Valorisation/ Biorefining	Shellfish shells	Extraction and purification of chitin and production of chitosan for chemical and food applications	Mathew et al. (2020)
	Protein hydrolysates extraction	Valorisation/ Biorefining	All fish and selfish by- products (especially skin and bones)	Isolation and purification of proteins for human nutrition, cosmetics, and pharmaceutical purposes	Anal et al. (2013)
Consumption	No treatment	Disposal	Bones, viscera, heads, and skin	Waste is eliminated through its disposal in organic recycling bins and later deposited in landfills	Cadavid-Rodríguez et al. (2019)

2020). These products can directly contribute to reducing the environmental impact and production costs of the primary fish product. This may provide an impact positively human health. Whereby, the production of novel, functional food ingredients potentially can reduce the risk of diseases or enhance health (Hayes, 2021).

Markets for these by-products can include low-value animal feeds or agricultural products. A common way to produce low-value feeds is to use seafood by-product-based on ensilage processes. This is used as it is a relatively simple and cheap process, taking advantage of all aspects of the by-product, i.e., protein, lipids, and bone materials (Islam and Peñarubia, 2021; Mousavi et al., 2013). Examples of higher-value alternatives for by-product use include the production of hydrolysates for human consumption, fish mince, fish smoked sausages, fish patties (Erasmus et al., 2021), and aquafeeds (Saleh et al., 2020). The use of by-products for the production of fertilisers and biostimulants has also received attention over the years as a means to increase the economic and ecological sustainability of the fishing industry (Ahuja et al., 2020), by finding avenues to derive value from bones, heads (Kim et al., 2010), viscera (Kim, 2011), and even the wastewater from processing and effluent from aquaculture farms (Ching and Redzwan, 2017; Hayes and Gallagher, 2019).

An area which has been widely studied is the valorisation of fish byproducts into high-value products. Some of these studies have assessed the extraction of omega-3 fatty acids, for food supplements because of their preventive role in cardiovascular diseases (Mbatia et al., 2010; Nges et al., 2012), collagen for food, cosmetic, pharmaceutical, tissue engineering, and biomedical industries (Araujo et al., 2021; Bhuimbar et al., 2019), chitin or chitosan from shells for chemical applications (Mathew et al., 2020), and protein hydrolysates (Anal et al., 2013). Other opportunities and strategies for the extraction of oils and the subsequent production of biofuels, like biogas or biodiesel, have also been assessed (Karkal and Kudre, 2020). These valorisation processes make use of by-products generated from seafood production activities such as fishing, aquaculture, processing (e.g., offal and trimmings, Jayathilakan et al., 2012), and waste from consumers (e.g., bones) (El-Gendy et al., 2014). However, research and the uptake of seafood valorisation into commercial practices must be underpinned by the CE model. This would ensure the seafood by-products valorised for further use are sustainable through evidence-based metrics, such as life cycle assessment (LCA).

#### 4. Incorporating the circular economy in seafood value chains

To incorporate CE principles into seafood value chains, there is a need for measurable value(s) from the utilisation of the by-products, e. g., economic, consumer perception, de-risk production portfolio, and legislation compliance. A way in which this can be facilitated is through the use and promotion of eco-design and eco-efficiency (de la Caba et al., 2019; Regueiro et al., 2021; Ruiz-Salmón et al., 2020). The former is integrating environmental attributes into the design of the value chain, while the latter is producing more from fewer resources. Opportunities exist based on a value and volume hierarchy, which can be used to valorise potential products from seafood by-products. Within the seafood industry, there are two main types of by-products that are considered waste. These by-products can be effluents from process water (i.e., sludge, aquaculture wastewater, and processing and cooking effluents) or the biological by-products resulting from the processing (e.g., crustacean, and bivalve shells, offal, fish heads, frame, and trimmings). Some whole fish material could be sourced as discarded from fisheries due to changes in the EU's Common Fisheries policy. Although the quantities are reduced in recent years due to policy changes when compared to a more static and inherent loss from seafood processing for human consumption.

Research activities have identified the potential for the valorisation of liquid effluents from seafood and processing activities, however, these are not yet as extensively applied as they could be (Alkaya and Demirer, 2016; Zilia et al., 2021). For example, it is possible to extract pigments, proteins, or flavour compounds (Tremblay et al., 2020) from processing by-products. Other avenues for the recovery of valuable products include the blood waters from the processing of fish. These waters contain substances which could be valorised into products such as antioxidant peptides, renin, and dipeptidyl peptidase (Hayes and Gallagher, 2019). Other recoverable materials from effluents include sludge or biosolids. Alternative management for sludge could generate novel resources, generating valuable elements such as carbon and different nutrients. Furthermore, as an energy resource in the form of biogas or biodiesel, sludge and recovered solids can be integrated into sustainable solutions that can help mitigate energy consumption in the sector (Gherghel et al., 2019).

With regards to organic processing by-products such as heads, skin, fins, bones, viscera, and scales are often derived into low-value commercial products such as feed, fish meal and oils (Al Khawli et al., 2019; Bruno et al., 2019a). Adhering to the principles of sustainability, these products could be important sources of new high-quality and high-value commercial compounds such as proteins, peptides, vitamins, amino acids, collagen, chitin, enzymes, gelatine, glycosaminoglycans, polyunsaturated fatty acids, minerals, etc. Furthermore, they can provide important functional and bioactive properties for food, agriculture, cosmetics, pharmaceuticals, and/or nutraceutical industries (Al Khawli et al., 2019; Ghalamara et al., 2020; Wang et al., 2019). In addition, different processes have been developed to exploit these by-products efficiently in the form of food packaging, silage, fertiliser enrichers, biofuels, etc. (Nawaz et al., 2020).

In a CE context and to achieve a zero-waste goal, a top-down classification strategy could be established according to the quality or value of the resulting products from liquid or solid by-products. This is with a view to minimising the generation of waste material, which does not result in the recovery of energy or products of interest. It is essential to identify appropriate extraction technologies that will minimise energy consumption, maximise quality and yield, guarantee the safety of the resultant product, and ensure the objectives of sustainable development. However, the valorisation of these by-products may be impacted by certain regulations. Within the EU, the European Commission Regulation (EC) 1774/2002 established the sanitary standards applicable to animal by-products not intended for human consumption (currently repealed by regulation (EC) 1069/2009). For these purposes, byproducts are classified into 3 categories based on their risk to human and animal health and specify the conditions under which they can be managed (Table 2). These regulations place a limit on some of the material which can be valorised but does not unduly impact most seafood. Only designated category 3 can re-enter the food chain, typically in the form of farmed animal feed or aquafeeds.

In Fig. 2, the blue arrows represent the flows of by-products or lowvalue discards that are used in industries with "lower added value". These levels utilise a higher volume of material for a lower value-added product. As the material moves towards the base of the pyramid it is placed in the category below. This is because each time a lower level is used, the less value the resultant product can obtain. In the case of the green arrow, this indicates the co-product that results from energy production that could be reused in the fertilizer category, promoting recirculation of the system (i.e., digestate). The colours for wastewater/ sludge (orange) and by-products (red) have no value per se. In the figure, they symbolize the types of by-products and effluents obtained during industrial processes for each of the animal by-products categories.

#### 4.1. Pharma-industry, cosmetics, and biotechnology opportunities

There are a number of processing/biorefinery technologies that can be used to extract, concentrate, refine, and transform compounds from seafood by-products into high-value market bioactive/functional products for nutraceutical, pharma-industry, cosmetics, and biotechnology. These technologies can often include the use of supercritical extraction,

#### Table 2

Animal by-products (ABP) categories in seafood by-products and waste (European Union Regulation EC No. 1069/2009).

ABP Category	Risk level	By-product management	ABP material	Permitted uses
Category 1	High	Disposal only	Diseased fish, fish with notifiable diseases etc.	Incineration or as fuel in approved plants.
Category 2	Med	Not intended for animal consumption	Livestock carcasses and non-disease mortalities	Fertiliser, landfill (after sterilisation), and safe technical use.
Category 3	Low	'fit for human consumption' and derived from processing plants	Fish and shellfish processor by- products, fish trimmings.	Fertiliser, biofuel, petfood, farmed animal feeds, and aquafeeds.

A stepwise seafood by-product valorisation framework is outlined in Fig. 2. This framework is based on five levels of value for seafood by-products using a volume-value relationship (i.e., low volume, high value products at level 1 and high volume, low value products at level 5). High value applications for by-products include (Level 1) pharma-industry, cosmetics, and biotechnology, (Level 2) food. Medium value applications are valorisation as (Level 3) animal feed. Lower value applications include (Level 4) fertilisers (nitrogen, N; phosphorus, P; potassium, K) and (Level 5) energy. The nutrient recovery technologies, the value-added opportunities, and products for each of the levels of seafood by-product valorisation are presented below from high to low-value products.

membrane separation, and/or hydrolysis, inc. chemical and enzymatic. Such use allows compounds such as vitamins, flavours, essential oils, carotenoids, enzymes, amino acids, lecithin, and polyunsaturated fatty acids to be obtained from seafood by-products. The advantages of using technologies such as supercritical fluid extraction eliminates the need for extractive organic solvents that is traditionally used for extracting bioactive compounds. These solvents might not be food-safe, or environmentally friendly, and at elevated temperatures could compromise recovery rates of the compound(s) (Al Khawli et al., 2019; Haque et al., 2014; Kuvendziev et al., 2018; Uddin et al., 2011). Most studies evaluating the potential of supercritical fluids have focused on recovering lipid-soluble and antioxidative compounds. These products can often have the greatest economic value because they can improve certain technological properties in food matrices in novel foods (Al Khawli et al., 2019).

In contrast, the long-chain polysaccharides chitin and chitosan recovered from shrimp and crab shells are typically through the use of microbial proteolytic enzymes to deproteinise crustacean by-products (Wang et al., 2019). The other methodology is through the use of inorganic acids to demineralise the shells, followed by strong alkalis for deproteinisation. While the latter method produces a purer final product than the biological technique, it does however produce waste by-products, and if not removed, it can contaminate the final product.

The use of pressure membrane technology can purify recovered protein/peptide from seafood. The usefulness of this technology can avoid the need for solvents or adsorbents, as Through this technology, it is possible to obtain permeate fractions enriched in small bioactive peptides with bioactive potential as antioxidative, antimicrobial, and angiotensin-converting enzyme inhibitory activity (Chi et al., 2015; Karnjanapratum et al., 2017; Ngo et al., 2014). For example, it has been observed that cod blood can be a potential source of peptides with antioxidative properties and could be exploited as a functional food ingredient (Ghalamara et al., 2020). In addition, peptides purified from fish by-products also showed interesting antimicrobial activity (Ennaas et al., 2015; Song et al., 2012), i.e., peptides purified from cod blood and sardine cooking wastewater against *Escherichia coli* (Ghalamara et al., 2020).

The use of sludge derived from seafood was recently reviewed by Gherghel et al. (2019). The authors reported the obtention of different compounds of interest such as adsorbents obtained using microwave and pyrolysis treatments, high-quality enzymes, and proteins comparable to commercial versions using ultrasound, or bioplastics using activated sludge as raw material during bacterial fermentation. The bycatch small-spotted catshark (*Scyliorhinus canicula*) viscera have been used as a substrate to produce hyaluronic acid by *Streptococcus zooepidemicus* fermentation (Vázquez et al., 2015). Scales have also been used as a substrate to generate collagenase-like enzymes by microbial fermentation (Wang et al., 2019). Cephalopod by-products such as squid skin, it is a source of collagen for the manufacture of cosmetic products. In addition, this by-product has been investigated as a potential plasticiser in the preparation of biofilms in combination with chitosan (Wang et al., 2019).

#### 4.2. Food opportunities

The use of seafood by-products continues to be a challenge due to food safety, their interactions with other ingredients used in the final food product, and public perception and consumer acceptance. Several products of interest can be obtained from fish by-products such as protein hydrolysates and polyunsaturated fatty acids from trimmings, heads, and frames. However, any by-products used for human consumption must be treated as food grade standards during their collection and processing, e.g., HACCP, which meets below limits of foodborne pathogens. Neglecting these standards can result in hygiene issues, spoilage, and food-borne illness due to seafood's inherent highly perishable nature.

Protein hydrolysates are perhaps the most common use of fish byproducts, e.g., fish heads, frames, and offal. The generation of functional food ingredients containing bioactive peptides that can provide the consumer with a health benefit that goes beyond basic, human nutrition is a growing area of both research and commercial venture. Protein hydrolysates, concentrates and isolates are distinguished by the level and quality of protein contained in each and can command different market values based on protein content but also technofunctional and sensory attributes as well as health benefits for the consumer (Hayes et al., 2016). Hydrolysates also have applications as functional feed ingredients to improve the health of farmed aquatic species, ruminants, and companion animals (Naik et al., 2021).

Similarly, omega-3 fatty acid-rich oils from fish livers for the food/ health supplement market (Al Khawli et al., 2019; Anal et al., 2013; Bhuimbar et al., 2019). Marine-derived oils (e.g., fish oil) are valuable by-products but were once treated as a low-value commodity until the recognition for their high nutritional value and now it is exploited as an omega-3 fatty acid-rich supplement. Besides this, other applications have been tested including the omega-3 fatty acid enrichment of bakery and pasta products (Nawaz et al., 2020). The traditional method of oil extraction is through cooking and separation. Although, there are other technologies such as ultrasound combined with assisted enzymatic extraction that can improve the oil extraction efficiency. Recent studies have shown that waste parts from fish (e.g., heads and frames) that were pre-treated with assisted enzymatic extraction before enzymatic hydrolysis led to a higher level of oil recovery. This included a higher percentage of polyunsaturated fatty acids level and greater oxidative stability, lower apparent viscosity, and an overall sensitivity to temperature-dependent degradation. All these attributes would lead to wider applications in food products (Al Khawli et al., 2019; Bruno et al., 2019b). While marine oils could also be co-produced along with the production of protein hydrolysates, where centrifugation or filtration technologies are typically used for recovery.

Calcium from fish bones has received attention as a natural calcium supplement for individuals that has calcium deficiency or as a health supplement (Nawaz et al., 2020; Venugopal, 2021). Studies have previously reported that calcium bioavailability is higher in tuna bones in

comparison to calcium from other sources such as milk, vegetable, and salts.

However, all previous studies suggested pre-treatment, including heating, boiling, tempering, or chemical treatment before adding it to the food matrix (Gupta et al., 2016; Nawaz et al., 2020). The reason is that the bone matrix is composed of a complex inorganic part and an organic part of collagen fibres. These fibres are difficult to break down in simple enzymatic digestion without prior softening of the bones. Likewise, by incorporating boiled fish bones from Nile tilapia (*Oreochromis niloticus*) into biscuits, it was reported that fish bone fortification may be a rich source of calcium and other minerals, along with improved fatty acids (Nawaz et al., 2020).

#### 4.3. Animal feeds opportunities

The majority of animal by-products from fisheries and processing plants have long been used in fish meal production that is destined for animal feeds (Cho and Kim, 2011). Although from a circular business economic model perspective, the aim would be to use these by-products for greater value outputs, specifically Level 1 (pharma-industry, cosmetics and biotechnology and human food) and 2 (human food) categories (Fig. 2). However, not all by-products are suitable for CE implementation. For example, the low production yield of the seafood by-product to be economically viable, loss of quality, sporadic times of production, long distances between the by-product producer and the valorisation plant, or insufficient logistical resources to be valorised under Levels 1 and 2, the by-product could therefore be more suitable for animal feed rendering use that is lower grade. One example can be the red and vascularised fish flesh that is typically produced as waste from the fish filleting plants. It is a high-quality protein source that is often used for animal feed production or discarded without revenue being generated (Herpandi et al., 2011; Nawaz et al., 2020). This is often due to the low quantities being generated and logistical difficulty in attaining Level 1 or 2 use.

However, if it is commercially, technically, and/or logistically viable then seafood waste could undergo a series of biorefinery processes to produce functional ingredients for feed use. For instance, the use of supercritical fluid extraction makes it possible to reduce the fat content of fishmeal without affecting the quality of the protein. When operated under certain extraction conditions (10–40 MPa, 25–80 °C and with CO<sub>2</sub> flows of 9.5 g min<sup>-1</sup>), it is possible to achieve a 90% reduction in lipid content (Al Khawli et al., 2019). Fish oil is extracted from fish viscera by pressing, microwave-assisted extraction, supercritical fluid extraction, solvent extraction, autolysis, and enzymatic hydrolysis (Wang et al., 2019). The application of supercritical fluid extraction has increased in recent years as CE and resource-efficient practices have been incorporated into commercial production practices (Al Khawli et al., 2019; Venugopal, 2021).

The high cost of fishmeal used in fish feed has prompted alternative ways of obtaining protein for feeds. A low-cost method for processing seafood by-products for feed is fish silage (Islam and Peñarubia, 2021; Mousavi et al., 2013). This process results in excellent protein products and is a valuable source of amino acids for protein biosynthesis, with high amounts of polyunsaturated fatty acids. The resulting biorefined products are widely used as feed ingredients in aquaculture for different aquaculture species. During silage processing, endogenous enzymes hydrolyse proteins and transform them into more soluble forms of nitrogen, this helps contribute to their widespread use (Ahuja et al., 2020; Herpandi et al., 2011; Mousavi et al., 2013).

As a potential solution to liquid by-product streams for aquaculture and fish processing, the cultivation of high nutritional value macroalgae (seaweeds) in integrated multitrophic aquaculture systems presents great potential. Ammonia/ammonium from protein metabolism and uneaten aquafeeds is the main aquaculture effluent. It is also often one of the most difficult nutrients to limit in flow through and open aquaculture systems, e.g., sea cages (Badiola et al., 2012; Liu et al., 2016; Song et al., 2019). In general, intensive aquaculture systems, solids are removed by sedimentation or screening, and the nitrogenous nutrient is converted to nitrate (NO<sub>3</sub>), through nitrification in bacterial filters (Milhazes-Cunha and Otero, 2017). Therefore, current effluent treatment technology depends on bacterial systems and does not add value to the process beyond converting the toxic nutrient to a lesser form in the effluent. This is except for some aquaculture facilities possessing additional units where that host anaerobic denitrification bacteria where that convert nitrates into nitrogen gas. Although this process is technically prohibitive, e.g., the conversion process is relatively slow. Microalgae can be used for the efficient collection and recycling of nutrients in aquaculture effluents and can reduce chemical and biochemical oxygen demand concentrations and potentially toxic metals. To further enhance the economic strength of integrated multitrophic aquaculture systems, can be achieved by extracting high-value added compounds from algal biomass (fatty acids, pigments, polysaccharides, etc.) that can be valorised in premium animal feeds or as a feedstock for biobased fuels or plastics (Laurens et al., 2017; Maiolo et al., 2020; Milhazes-Cunha and Otero, 2017; Shah et al., 2018).

#### 4.4. Fertiliser and plant biostimulant opportunities

The use of seafood by-products to produce plant fertilisers and biostimulants (i.e., enhances plant health, crop quality and stress tolerance) can potentially reduce large biomasses into commodity. This is particularly relevant if the biomass has deteriorated in shelf-life quality (i.e., rancid), or not fit for use in human or animal use. Furthermore, the recycling of elements back into food in both fisheries and aquaculture production system can mitigate the need for artificial fertilisers. For example, synthetically produced nitrogenous based fertilisers (e.g., ammonium nitrate), which uses the energy-intensive process of converting nitrogen gas in the air to ammonia using natural gas. More importantly, the generation of potassium and phosphate fertilizer from seafood by-products can also displace the need for finite mined sources, e.g., phosphorite and potash-rich rock. This could be strategically important as there is an increasing concern over the growing need for phosphorus fertilizers as a result of food demand. In addition, there is a rising scarcity of phosphate-rich mines, and more recently, global conflicts and climate impact, all of which threatens future food security (Nedelciu et al., 2020).

The benefits of using seafood by-products to produce fertiliser and biostimulant products include being a natural soil conditioner, improving soil texture, and enhancing growth performance (Radziemska et al., 2019). Different fertilisers based on fish by-products are commercially available (e.g., fish blood and bone) and some are even authorised for ecological or organic agriculture. For example, composted fish by-products with pine bark have been evaluated as organic fertilisers. The results showed that there was a positive enhancement in the nitrogen, phosphorus, potassium, sodium, calcium, and magnesium content in arable plant leaves, i.e., ice lettuce (Lactuca sativa) and white mustard (Sinapis alba) (Radziemska et al., 2019). Although the calcium: phosphorous ratio simultaneously had worsened as a result of the fertiliser regime. Similarly, the fermentation of squid feather bones inoculated with the Lactobacillus paracasei bacteria has been reported as a means of producing biobased fertilisers (Wang et al., 2019). In addition, as a fertiliser, it can increase the carbon storage capacity of the soil, potentially minimising greenhouse gas emissions to the atmosphere (Radziemska et al., 2019).

The recovery of phosphorus (usually as struvite) from waste sludge has been reported after anaerobic digestion, where the recovered phosphorus can be used as fertilisers. In some studies, there has been reported that aquaculture-derived sludge derived from anaerobic digestion has a higher bioavailability of nitrogen than undigested sludge (Aas and Åsgård, 2017; Del Campo et al., 2010). Although these processes involve low investment costs and can remove potentially toxic metals simultaneously, they usually require specialised equipment (e.g., digester vessels), high operating costs, and chemical and energy consumption (Gherghel et al., 2019). While the use of microwave technology at a laboratory scale had been shown to obtain cadmium, chromium, copper, nickel, lead and zinc after anaerobic digestion of sludge, with a total recovery of 95.3–100%. After anaerobic digestion and sludge dewatering, the production of adsorbents for metal ions ( $Cu^{2+}$  and  $Pb^{2+}$ ) has been reported, with improved control of the heating process, energy savings and reduction of equipment and wastes (Gherghel et al., 2019; Madende and Hayes, 2020).

#### 4.5. Energy generation opportunities

Different methods to obtain energy from seafood-derived sludge have been reported. For instance, biogas generation has been investigated through microwaves, ozonation, ultrasound, enzymatic treatment, and treatments with alkalis or acids. Consolidated technologies have been shown to enhance biodegradability and the capacity to obtain methane from sludge (thermal pre-treatments and high pressures), or the co-digestion of food waste with sewage sludge. Although the complexity of the reactors and the process requires a high level of technical expertise for the operation and production optimisation.

In general, biogas is produced through anaerobic digestion, which obtains a varying degree of methane (50-70%) and  $CO_2$  (30-50%)depending on the quality of the substrate and with a minor impurity concentration, e.g., nitrogen and hydrogen sulphide. Although the process can be slow, carried out at higher than ambient temperatures, and can require large bioreactors to produce viable quantities of biogas. Anaerobic processes can remove organic matter (80-90%) (Del Campo et al., 2010; Parvathy et al., 2017). Beyond anaerobic digestion, gasification, pyrolysis, and sludge can also produce useful products such as syngas (carbon monoxide and hydrogen from gasification/pyrolysis), biochar (pyrolysis), and bio-oils. For the latter, calcined flakes have been used as a catalyst for biodiesel synthesis (Wang et al., 2019). While pyrolysis of sludge after the standard anaerobic digestion for energy production can produce biochar, which can be used for soil remediation or as fuel (Gherghel et al., 2019). The combination of anaerobic digestion and aerobic processes can also be an effective approach to reducing negative characteristics in fish processing wastewater, e.g., high biological and chemical oxygen demand, volatile solids, and typically a low pH. Other technologies such as supercritical fluid extraction have also been explored after sludge dewatering. However, they require high capital investment and maintenance costs (Gherghel et al., 2019; Parvathy et al., 2017). In the review by Pan et al. (2015), a more critical analysis of waste-to-energy supply chains was undertaken. The authors identified a series of barriers that could be categorised as technological, financial, institutional, and regulatory for the uptake of waste into energy production systems. Furthermore, evaluated successful and sustainable waste-to-energy businesses.

#### 5. Discussion

Outlined in the previous sections of this article is the current state of the art with regards to seafood by-product materials, the processes, and technologies available and a volume to value valorisation framework. However, to increase the levels of circularity in seafood value chains, a multidisciplinary and holistic approach to its implementation is required. In order to realise this, a number of stages and steps will need to be included, expanded on and developed (Fig. 3).

- Valorisation levels in a circular economy
- Emerging waste recycling strategies
- · Beyond waste to bio-based resources
- Climate impact on seafood loss in fisheries and aquaculture as well as along the value chain
- Education and outreach
- · Measuring environmental performance for intensive sustainability



Fig. 3. The thematic areas that require implementation for increased uptake of circularity in seafood value chains.

These thematic areas address some of the key environmental, economic, and social aspects, which will impact on the successful implementation of CE in seafood value chains (Fig. 3). These areas can allow consumers, producers, and waste managers to tackle areas such as seafood loss, prepare climate adaptation measures and find value in almost all aspects of the production chain.

#### 5.1. Valorisation levels in a circular economy

The levels for valorisation and nutrient recovery from seafood waste demonstrate the role of reuse and recovery within the CE, which subsequently adds to the sustainable seafood value chain credentials. These levels offer a decision tree framework which can allow operators, regulators, and investors to comply with the principles of CE while following options that make economic and business sense in their respective cases. For each site, the key considerations will include the volume of available material for valorisation, the distance that the material must travel and indeed, and financial costs that can be considered within this framework and allow economic and environmental considerations to develop viable valorisation strategies.

Feedback loops for by-product material in the various stages of the value chain can contribute to raw material production, in terms of feed, energy, and fertiliser (Fig. 4). By-product material from seafood processing can be valorised as feed ingredients or in higher applications such as food and biopharma products and shifted to the consumption



**Fig. 4.** Seafood circularity using the 5 levels of valorisation potential. Large arrows indicate the bulk material transfer from one step of the value chain to another, while smaller arrows demonstrate potential material transfer for regeneration along the value chain. Omitted from the diagram is the value component for each of the levels.

stage of the value chain. Opportunities for increased valorisation and recovery of lost food and material can be implemented at numerous stages in the seafood supply chain.

The CE model aims to maximise the efficiencies of resource use to reach a high level of return on the energy, time and space invested into the activity and product. The perception that the by-products from these processes are value-less is changing. However, to engage all value chain actors, there are significant knowledge gaps which need to be closed to implement sustainable CE models, which balance the needs of the business with the needs of the environment, i.e., profitability versus limiting environmental impact and sustainability. Therefore, CE practices must be implemented across the product life cycle. Extending beyond the current state of the art, where much of the focus has been on seafood production and processing.

#### 5.2. Climate impact on seafood loss

One of the greatest threats to food and seafood security is climate change. Changes in environmental conditions can greatly impact wild fish stocks, and aquaculture productivity.

For instance, there are increasing concerns over emerging frequency and prevalence of infectious diseases, especially amongst farmed aquatic species. Brooks et al. (2022) considered the impacts of emerging infectious diseases in aquatic systems, but the case studies are mostly on tropical and crustacean-farmed species. This suggests aquatic systems elsewhere need to be evaluated against potential future climate-related impacts.

Furthermore, the increasing frequency in extreme weather events caused by climate change may have negative impacts on processing, packaging, and distribution channels, which further contribute to seafood loss. For example, Collins et al. (2020) postulated that climate change may cause an increased flood risk that could impact both shore-based facilities and their access points and routes. While increasing storm and extreme weather events (e.g., heatwaves) may also increase the time required to transport or shorten the life of seafood within and between markets. This consequently brings about a reduced useable life, quality, and value of the seafood product (Maulu et al., 2021).

#### 5.2.1. Fisheries loss

Fisheries is one of the food production sectors that face the greatest threat of food loss from climate change. Sainsbury et al. (2021) found that fishers in the southwest region of the UK examined the trade-off of the economic rewards of continued fishing compared with the physical risk at sea when adverse weather conditions impact fishing. They looked at the socio-economic risks and potential benefits across a range of wind speeds and wave heights at sea, and how these influence the decision-making of the ship's skipper on whether to set out to fish or not. There was variability across the types of ships, gear type and other factors but in general, the utility values were seen to reduce with windspeeds above about 40 kph and wave heights >3 m. Other factors which contribute to seafood loss in the fisheries segment of seafood production can also include; discarded catch, poor chilling facilities on board the fishing vessel and damage to stock while being removed from nets (Kruijssen et al., 2020). These factors can broadly be categorised as physical, quality, nutritional and market loss (Kruijssen et al., 2020; Kumolu-Johnson and Ndimele, 2011; Love et al., 2015).

#### 5.2.2. Aquaculture loss

The same factors for seafood loss in fisheries apply to aquaculture. Where aquaculture differs from fisheries is that the artificial conditions in which aquatic species are cultured offer little protection from natural events for example, in marine environments, storms and extreme weather events can cause cage structures to fail and in the instance of freshwater flow through systems flood events can cause considerable damage to facilities and lead to the escape of farmed stock. Food loss in aquaculture systems is something which should be mitigated given the artificial nature of the practice, which is similar to the culture of cattle and sheep, where it relies on inputs from the technosphere for production and success. Naylor et al. (2021) recognised in a comprehensive review of global aquaculture and its increasing importance over the last 20 years the impacts and thus potential losses that climate change may have on farmed seafood. They noted that losses (i.e., mortalities) from aquaculture occur mainly due to suboptimal growing temperatures, saltwater intrusion due to sea-level rise, damage to infrastructure, freshwater shortages, and droughts. They also noted climate impacts on rising costs of feed as well as climate-driven risks due to pathogens, parasites, and pests as well as harmful algal blooms.

#### 5.2.3. Processing and distribution loss

The processing and packaging sections of seafood value chains are likely to offer the greatest opportunity and impact in recovering lost waste in the short and medium term (Cortés et al., 2021a; de la Caba et al., 2019; Laso et al., 2016; Tan and Lamers, 2021; Venugopal and Sasidharan, 2021). Seafood loss can occur in this stage of the value chain through a number of different challenges. Some of these overlap with losses that have been highlighted in other parts of the value chain. One of the ways in which loss can occur in the processing stage can be through spoilage due to inadequate equipment or equipment failure (i.e. refrigeration and conveyor systems), low levels of processing control (i. e., staff removing too little from the carcass), low use of packaging material, or low processing capacities (Kruijssen et al., 2020; Love et al., 2015; Spang et al., 2019). Another contributing factor to food loss can be the implementation of high-quality food standards. These quality standards, while they present appealing-looking products to consumers can also contribute to food and nutrient loss, by diverting damaged, but safe and healthy seafood products from the food supply (Spang et al., 2019).

From a distribution perspective, losses can occur through poor handling or stocking of the products. Damaged packaging can shorten the shelf life of the products and in cases where there is a sizeable distance to market lead to unsaleable products. Poor road infrastructure and remote landing or production sites can also contribute to seafood loss. Geopolitical events, trade barriers, and disease outbreaks can also shorten the shelf-life and availability of seafood. This can be evident by delays in the delivery of seafood products to continental Europe from Ireland and air freight of seafood from Europe to Far-East Asia and vice versa (Ahearne and Hynes, 2020; Barnes, 2020; Mahfouz et al., 2019).

Improved process control and supply chain management practices can help to reduce food loss and waste in seafood supply chains (Bruno et al., 2019a; de la Caba et al., 2019; Yan and Chen, 2015). With industry and industrial processes moving into the 4th industrial revolution (Industry 4.0), a number of advanced and smart technologies are coming online (Hassoun et al., 2022). These technologies in conjunction with wider smart systems such as energy monitoring can help to reduce the costs of cooling and refrigeration; cooling and freezing being one of the main drivers of cost, energy use, and environmental burden of seafood processing (Avadí and Acosta-Alba, 2021; Cortés et al., 2021a; Hassoun et al., 2022; Vazquez-Rowe et al., 2012). Industry 4.0 is still an emerging vision of how industrial processes and economies can increase their efficiencies. However, the EU has already begun to build capacity for Industry 5.0. It aims to complement Industry 4.0 practices by also transitioning to sustainable, human-centric, and resilient industries (European Commission, 2022).

#### 5.2.4. Consumption loss

Some of the greatest losses and waste of seafood occur at the consumption stage. Statistics on seafood loss at the consumer level can range from 10 to 11% (Gustafsson et al., 2013) to 41–56% (Love et al., 2015) and by some estimates can be as high as 70% (Stenmarck et al., 2016). These values can vary from region to region and can reflect the cultural value that is placed on seafood within a nation's diet. Contributing factors to food loss at this stage of the supply chain can include spoilage of the products, excess preparation (the loss of edible parts due to poor preparation), and diet culture, i.e., only eating certain parts of the seafood (Birney et al., 2017; Kruijssen et al., 2020).

Many of the recommendations which have been put forward to reduce the levels of loss and waste generated by consumers have focused on behavioural changes through educational efforts (Kruijssen et al., 2020; Love et al., 2015). These strategies aim to encourage consumers to plan their meals, control the portion size of the meals, and minimise the levels of leftovers. Other suggestions include a switch from fresh to frozen seafood products, which could lead to a decrease in levels of waste through longer shelf-life spans. Action plans are being developed by national and interregional policymakers to raise awareness on the impacts and to highlight the unsustainability of current food consumption practices. This includes the United Nations' Sustainable Development Goals, policies, and actions such as those by the EU, which have incorporated food loss and waste minimisation within many of its plans (Circular Economy, European Green Deal) and strategies (Farm to Fork, European Commission, 2018; 2020a, 200b). These actions will lead to national and interregional initiatives by member states to combat food loss and waste. However, there will also be a need to match these efforts in waste management through novel and emerging waste recycling techniques and strategies throughout seafood supply chains.

#### 5.3. Emerging waste recycling strategies

Emerging waste recycling strategies in the seafood supply chain are essential to tackle the growing seafood waste problem. Within the EU trading bloc, the Waste Framework Directive (Commission Decision, 2011/753/EU, EU 2019) outlines some basic waste management principles with the preferred option of preventing waste. However, according to Sharma et al. (2021) and Ghosh et al. (2016), an effective waste management strategy would need to include waste minimisation, characterising the waste, and waste recycling. From a supply chain perspective, some emerging strategies are being introduced to meet sustainability goals. For example, location-based tracking technology is being introduced by the EU's commercial fisheries. This initiative will be a step towards increasing transparency and discernibility in seafood supply chains. Identifying the waste levels and the location of waste within the supply chain allows the development of an effective waste management plan, e.g., the stage, location, type, and volume of seafood waste being generated (Ghosh et al., 2016). Emerging waste recycling plans aim to develop value-added supply chains. For instance, the valorisation of abundant and available bio-wastes (de la Caba et al., 2019; Sorg et al., 2019). These plans can be facilitated through the separation of the waste into liquid/solid streams and using the best available extraction method to retain the nutritional component(s) of interest (Coppola et al., 2021; Lam et al., 2020; Rejula and Mohanty, 2018). There is also a great interest in developing sustainable and biorefinery practices to produce disruptive and high-value bioactive compounds from food waste streams. These range from the use of novel antimicrobials for eco-packaging applications, immune-stimulatory ingredients for health and wellbeing (Murphy et al., 2020; Rowan and Galanakis, 2020), and novel ingredients for fish feed from biorefining fisheries and seaweed by-products into the aquafeed ingredients (Pogue et al., 2021; Colombo et al., 2022).

Some researchers have outlined the importance of reframing what is food and what is waste as a key step in the cradle-to-cradle considering waste as part of the cycle (Martínez-Córdova et al., 2017; Soma, 2020). For example, in aquaculture, the conventional waste that is produced can take a place as a by-product alongside traditional seafood products. This approach can decrease the release of contaminants into the environment and add new revenue streams to the seafood sector (Regueiro et al., 2021).

One of the key challenges in emerging waste management strategies is consumer acceptance of recycling seafood waste and turning it into a new product. These strategies will need to balance several factors to be effective and commercially viable approaches. For example, new market opportunities, market trends, current market developments, and an end product that is competitive in the marketplace (Borrello et al., 2017). The role of the consumer in the success of the product is particularly important, as the consumer has the final say in the economic success of a product (Borrello et al., 2017; Ghosh et al., 2016). Finally, food marketing, consumer education and food labelling by retailers and producers can assist with changing consumer behaviour as a catalyst for waste management plans (Altintzoglou et al., 2021; Aschemann-Witzel et al., 2016, 2016de la Caba et al., 2019).

#### 5.4. Beyond waste to bio-based resources

Within CE an emerging field is the circular bioeconomy (CBE). The CBE aims to utilise biomass or biological resources from all industries and economic sectors (Tan and Lamers, 2021). There are a number of definitions in use for the CBE, but it can be accepted as complementary to the CE and can help in the transition to cleaner and sustainable production (Raimondo et al., 2021; Tan and Lamers, 2021; Venugopal and Sasidharan, 2021). One of the core aims of the CBE is the production of biobased materials and products. Bio-based products are often considered to be more sustainable than their conventional counterparts. Although these claims need to be validated using life cycle thinking (Adom et al., 2014; Brunklaus and Riise, 2018; Spierling et al., 2020; Wojnowska-Baryla et al., 2020). Under a CBE framework, there are efforts to develop functional foods, feeds, and additives from seafood by-products.

Recent research and frameworks for the use of machine learning in the processing and production of feeds and food products have identified opportunities for the application of such technologies across seafood supply chains (Cooney et al., 2021; Hassoun et al., 2022; Kang et al., 2017; López-Cortés et al., 2017; Manoharan et al., 2020; Suryawanshi and Eswari, 2022). There are strong roles for machine learning, the internet of things and digitisation in the development and production of efficient, precise food products from raw or waste sources.

#### 5.5. Measuring sustainability and environmental performance

One of the most important concepts in the use and implementation of the technologies and approaches for seafood by-product valorisation is sustainability. The sustainability of these interventions and actions needs to balance the economic and environmental needs of society and business operators. One of the approaches that are widely used to assess the environmental performance and sustainability credentials of a system, technology, process, or product is life cycle thinking, primarily in the form of life cycle assessment (LCA, Bohnes and Laurent, 2019; Ruiz-Salmón et al., 2021). Life cycle thinking uses a multimeric approach that can be used to assess the performance of a product across the three pillars of sustainability or triple bottom line: 1) environmental, 2) economic, and 3) social. These assessments can take the form of environmental LCA, social LCA (SLCA) or life cycle costing (LCC). When combined, these distinct aspects of life cycle thinking become a life cycle sustainability assessment (LCSA). LCSA has received growing attention since the early 2010s but to date has not been widely applied (Guinée, 2016). This is in part due to the ongoing development of frameworks and methodological approaches which can be used in LCSA (Lam et al., 2020). While the use of LCSA is the end goal, in the interim the use of key performance indicators (KPI) derived from LCA, LCC and SLCA can help to better inform and support decisions for the valorisation and increased utilisation of seafood by and co-products. For example, under the environmental pillar of sustainability, the use of KPIs such as carbon dioxide (kg CO2 eq. kg-1 of product), water use (m3 kg-1 of product) or energy use (MJ/kg of product) can provide very useful information on the associated environmental burden of operational decisions. Economic KPIs under an LCC approach could take the form of a payback period of the investment, internal rate of return or benefit-cost ratio (Valenti et al.,

2018). Social KPIs for example that could be applied in this context could include investment to create direct employment or indices on the seafood circularity (amount of seafood prevented from being wasted) and its contribution to nutrition (using recommended daily allowances for human nutrition or energy content if for feed) (Hallström et al., 2019; Valenti et al., 2018). The use of simple KPIs like those listed above can help to promote the sustainable use of seafood co-products under a circular economy.

Previous studies have demonstrated LCA as a suitable tool to guide and make decisions on selecting and handling strategies in food waste recycling (Lam et al., 2020; Vilariño et al., 2017; Zilia et al., 2021). The use of the water-energy-food nexus can be an effective tool in developing a waste recycling plan applied for seafood by-products. A study undertaken by Chang et al. (2016) found it was particularly useful for identifying synergies, challenges, and opportunities. In particular, the authors noted that some of the outputs which would lead to increased social justice enhanced resource efficiency and reduced environmental impact. Gephart et al. (2017) argued that there was a gap in the 'seafood' water-energy-food nexus literature. They highlighted the importance of looking at water usage in seafood production. In addition, Slorach et al. (2020) conducted a study incorporating 'health' into the water-energy-food nexus, using LCA for each sector to give a true perspective on environmental impacts. The study looked at the impact on the nexus of four treatment options: anaerobic digestion, in-vessel composting, incineration, and landfilling. The results found that anaerobic digestion was the most sustainable and in-vessel composting the least. Similarly, a study carried out by Laso et al. (2016) analysed the environmental impact of anchovy waste valorisation in contrast to incineration and landfilling. The results showed that the valorisation waste management options resulted in the least impact on the environment.

Numerous studies have outlined the benefits of using social, economic, and environmental variables including economic performance assessment tools to assist decision-makers in the waste management of seafood supply chain strategy (Alkaya and Demirer, 2016; Jacob et al., 2021; Zilia et al., 2021). Zilia et al. (2021) demonstrated that by incorporating the three pillars of economic, social, and environmental in the example of reusing sea urchin waste. They did this by presenting the frameworks under a business model canvas approach and identified social opportunities through additional jobs being created as part of CE implementation and reduced environmental burden by valorising the waste material. Vilariño et al. (2017) followed a similar approach in reviewing LCA approaches for food by-product management. A number of recent studies have suggested that the use of technological, social, and political readiness levels to supplement the use of LCA as means of developing new green innovations, including sustainable waste management (Ruiz-Salmón et al., 2020; Stead, 2019; Villanueva-Rey et al., 2018; Voyer and van Leeuwen, 2019).

According to Ghosh et al. (2016) 'undervaluing and 'underreporting' are referred to as the 'hidden costs' in food waste management. Economic benefits are key enablers of green innovation (Kelliher et al., 2020); exploring the true value of these, 'hidden costs' can enable a more accurate assessment of the economic impact of waste valorisation and can act as a facilitator for change (Ghosh et al., 2016). A good example is the use of recirculating aquaculture systems and how they can reduce water usage, increase nutrient recycling, and improve waste management (Martins et al., 2010).

#### 5.6. Driving a seafood circular agenda through education and outreach

The linear economy has increased the distance between producers and consumers. As a result, it has affected society across all strands and social classes. In particular, how food is produced, this gap has resulted in a loss of awareness and value for the food being consume and the associated production systems. Given the increasing global demand for seafood, there is an opportunity for many seafood production systems to become "model" CE systems. Applying a multi-stakeholder approach to inform future sustainable waste management will enable an effective and inclusive strategy that will affect behavioural change. The power of the consumer as a catalyst in driving waste management in the seafood supply chain is vital. For example, the raising in public awareness on the environmental impact of single use plastics has led to a drive to prohibit their use, which resulted in the EU passing the Directive 2019/904 and bans certain types of plastics used and sold within the EU. There is a scarcity of literature on gauging socioeconomic impact, and this limits the ability of decision-makers to develop successful and long-term sustainable waste management strategies into existing business models.

Education and communication (vertical and horizontal) using the concept of a Triple Helix (academia-industry-government) ecosystem approach is required to change the behaviour to value waste as a resource in the value-added supply chain. This Triple helix approach will also support and accelerate the societal transition to a low-carbon economy using a bottom-up approach, which will inform future research and enterprise-performing activities regionally and nationally, with a global orientation. A number of emerging waste recycling strategies are reviewed from the seafood supply chain to the consumer. A holistic Triple-Helix framework approach to supporting and enabling sustainable waste recycling and management will provide strategies for by-product resource efficiency incorporating all stakeholders comprising producers, policymakers and consumers in the decisionmaking process and contribute to sustainable waste recycling goals. There should also be equally public-academia-industry-government driven initiatives for better seafood valorisation innovation and strategic impact. For instance, changing the misconception of fishmeal being used in aquafeeds where it is derived from the fish processing (e. g., trimming, offal, heads, and frames, Colombo et al., 2022) or hydrolysates waste by-products (partially fish protein hydrolysate and cook water, Egerton et al., 2020).

#### 6. Conclusion

This review offers insight into the diversity of seafood by-products being generated as part of current EU and global food production practices. It also presented a stepwise framework which can allow operators to identify valorisation strategies based on a volume-value-based approach. Opportunities exist through the valorisation or reuse, which can reduce environmental impact or loss of resources, i.e., nutrients, energy, and biomass. Some of these opportunities, for example, can include nutrient recovery at the level of production in aquatic farms using polyculture, aquaponics, integrated multitrophic aquaculture, or through mechanical means such as solid waste recovery from the produced sludge. Similarly, there are many opportunities from the fisheries sector, where non-targeted species, over quotas, or poor quality and unsuitable targeted species could be exploited as a valued resource, rather than discarded back into the aquatic environment. There is now a greater need for resource efficiency use as fisheries production levels are plateauing due to over-exploitation and climate impact. However, there are significant barriers that prevent the large-scale regeneration of seafood waste into valued products. A conjunction of incentives, capital investments, policy changes, commercial, and social and consumer acceptance is needed to realise the true potential of seafood waste.

Another key area to address is that many of the technologies which had been discussed remain at the research level. To increase the technological readiness level, there needs to be greater engagement with the industry in discerning the viability and readiness with which these technologies and approaches will be received. The markets for valorised seafood by-products as they move into the higher value levels of the valorisation framework (i.e., food and biopharma). The amounts and quality of by-products that are available for valorisation will also need to be established. Eco-efficiency and eco-design will need to play strong roles in the sustainable development of seafood by-product valorisation. For example, would centralising or decentralising to valorising a particular seafood waste be the most appropriate for commercial exploitation? To measure this environmental impact LCA would need to be employed and allow the decision-making of which valorisation option is the most sustainable. Human development has now reached a critical juncture, where the realisation of many natural resources is finite and incompatible with society's linear production chains. As the global human population increases, the regeneration of nutrients from seafood waste into new foods and products is paramount to creating a future of sustainable and responsible consumption.

#### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Data availability

No data was used for the research described in the article.

#### Acknowledgements

This work was supported by the EAPA\_576/2018 NEPTUNUS project. The authors would like to acknowledge the financial support of Interreg Atlantic Area. A.H.L Wan was co-funded under the HYDROfish project (2019–2022) which was funded under the Disruptive Technologies Innovation Fund (DTIF), established under Project Ireland 2040, run by the Department of Enterprise Trade and Employment with administrative support from Enterprise Ireland. His opinions expressed are his own. The authors would also like to thank Matt Bell for his editorial assistance.

#### References

- Aas, T.S., Åsgård, T.E., 2017. Estimated content of nutrients and energy in feed spill and faeces in Norwegian salmon culture. Nofima rapportserie. Available from: http://hdl .handle.net/11250/2452167, 16/January0/2022.
- Adom, F., Dunn, J.B., Han, J., Sather, N., 2014. Life-cycle fossil energy consumption and greenhouse gas emissions of bioderived chemicals and their conventional counterparts. Environ. Sci. Technol. 48 (24), 14624–14631. https://doi.org/ 10.1021/es503766e.
- Ahearne, A., Hynes, S., 2020. Challenges and opportunities for Ireland's major ocean economy industries. SEMRU, Whitaker Institute, University of Galway, Ireland. Available from: http://whitakerinstitute.ie/wp-content/uploads/2021/01/Cha llenges-facing-major-ocean-economy-industries.pdf, 16/January0/2022.
- Ahuja, I., Dauksas, E., Remme, J.F., Richardsen, R., Løes, A.K., 2020. Fish and fish wastebased fertilizers in organic farming-with status in Norway: a review. Waste Manag. 115, 95–112. https://doi.org/10.1016/j.wasman.2020.07.025.
- Al Khawli, F., Pateiro, M., Domínguez, R., Lorenzo, J.M., Gullón, P., Kousoulaki, K., Ferrer, E., Berrada, H., Barba, F.J., 2019. Innovative green technologies of intensification for valorization of seafood and their by-products. Mar. Drugs 17 (12). https://doi.org/10.3390/md17120689.
- Alkaya, E., Demirer, G.N., 2016. Minimizing and adding value to seafood processing wastes. Food Bioprod. Process. 100, 195–202. https://doi.org/10.1016/j. fbp.2016.07.003.
- Altintzoglou, T., Honkanen, P., Whitaker, R.D., 2021. Influence of the involvement in food waste reduction on attitudes towards sustainable products containing seafood by-products. J. Clean. Prod. 285, 125487 https://doi.org/10.1016/j. jclepro.2020.125487.
- Anal, A.K., Noomhorm, A., Vongsawasdi, P., 2013. Protein hydrolysates and bioactive peptides from seafood and crustacean waste: their extraction, bioactive properties and industrial perspectives. Marine Proteins. Peptides: Biological Activities. Appl. 709–735. https://doi.org/10.1002/9781118375082.ch36.
- Araujo, J., Sica, P., Costa, C., Márquez, M.C., 2021. Enzymatic hydrolysis of fish waste as an alternative to produce high value-added products. Waste. Biomass. Valorization. 12 (2), 847–855. https://doi.org/10.1007/s12649-020-01029-x.
- Aschemann-Witzel, J., De Hooge, I., Normann, A., 2016. Consumer-related food waste: role of food marketing and retailers and potential for action. J. Int. Food & Agribus. Mark. 28 (3), 271–285. https://doi.org/10.1080/08974438.2015.1110549.
- Avadí, A., Acosta-Alba, I., 2021. Eco-efficiency of the fisheries value chains in the Gambia and Mali. Foods 10 (7). https://doi.org/10.3390/foods10071620.
- Badiola, M., Mendiola, D., Bostock, J., 2012. Recirculating Aquaculture Systems (RAS) analysis: main issues on management and future challenges. Aquacult. Eng. 51, 26–35. https://doi.org/10.1016/j.aquaeng.2012.07.004.

- Barnes, M.D., 2020. Marine foresight study. Marine Institute. Available from: https: //oar.marine.ie/bitstream/handle/10793/1666/Marine%20Foresight%20Study.pd f?sequence=1&isAllowed=y, 16/January0/2022.
- Belton, B., Little, D.C., Zhang, W., Edwards, P., Skladany, M., Thilsted, S.H., 2020. Farming fish in the sea will not nourish the world. Nat. Commun. 11 (1), 1–8. https://doi.org/10.1038/s41467-020-19679-9.
- Bhuimbar, M.V., Bhagwat, P.K., Dandge, P.B., 2019. Extraction and characterization of acid soluble collagen from fish waste: development of collagen-chitosan blend as food packaging film. J. Environ. Chem. Eng. 7 (2) https://doi.org/10.1016/j. jece.2019.102983.
- Bielli, A., Alfaro-Shigueto, J., Doherty, P.D., Godley, B.J., Ortiz, C., Pasara, A., Wang, J. H., Mangel, J.C., 2020. An illuminating idea to reduce bycatch in the Peruvian smallscale gillnet fishery. Biol. Conserv. 241 https://doi.org/10.1016/j. biocon.2019.108277.
- Birney, C.I., Franklin, K.F., Davidson, F.T., Webber, M.E., 2017. An assessment of individual foodprints attributed to diets and food waste in the United States. Environ. Res. Lett. 12 (10) https://doi.org/10.1088/1748-9326/aa8494.
- Bohnes, F.A., Hauschild, M.Z., Schlundt, J., Laurent, A., 2019. Life cycle assessments of aquaculture systems: a critical review of reported findings with recommendations for policy and system development. Rev. Aquacult. 11 (4), 1061–1079. https://doi.org/ 10.1111/raq.12280.
- Bohnes, F.A., Laurent, A., 2019. LCA of aquaculture systems: methodological issues and potential improvements. Int. J. Life Cycle Assess. 24 (2), 324–337. https://doi.org/ 10.1007/s11367-018-1517-x.
- Bohnes, F.A., Rodriguez, U.P., Nielsen, M., Laurent, A., 2020. Are aquaculture growth policies in high-income countries due diligence or illusionary dreams? Foreseeing policy implications on seafood production in Singapore. Food Pol. 93 https://doi. org/10.1016/j.foodpol.2020.101885.
- Borrello, M., Caracciolo, F., Lombardi, A., Pascucci, S., Cembalo, L., 2017. Consumers' perspective on circular economy strategy for reducing food waste. Sustainability 9 (1). https://doi.org/10.3390/su9010141.
- Borrello, M., Pascucci, S., Cembalo, L., 2020. Three propositions to unify circular economy research: a review. Sustainability 12 (10). https://doi.org/10.3390/ su12104069.
- Brooks, D.R., Hoberg, E.P., Boeger, W.A., Trivellone, V., 2022. Emerging infectious disease: an underappreciated area of strategic concern for food security. Transbound. Emerg. Dis. 69 (2), 254–267. https://doi.org/10.1111/tbed.14009.
- Brunklaus, B., Riise, E., 2018. Bio-based materials within the circular economy: opportunities and challenges. Designing Sustainable Technologies. Product. Pol. 43–47. https://doi.org/10.1007/978-3-319-66981-6.
- Bruno, S.F., Ekorong, F.J.A.A., Karkal, S.S., Cathrine, M.S.B., Kudre, T.G., 2019a. Green and innovative techniques for recovery of valuable compounds from seafood byproducts and discards: a review. Trends Food Sci. Technol. 85, 10–22. https://doi. org/10.1016/j.tifs.2018.12.004.
- Bruno, S.F., Kudre, T.G., Bhaskar, N., 2019b. Impact of pretreatment-assisted enzymatic extraction on recovery, physicochemical and rheological properties of oil from Labeo rohita head. J. Food Process. Eng. 42 (3) https://doi.org/10.1111/jfpe.12990.
- Bruno, S.F., Kudre, T.G., Bhaskar, N., 2019c. Impact of pretreatment-assisted enzymatic extraction on recovery, physicochemical and rheological properties of oil from *Labeo rohita* head. J. Food Process. Eng. 42 (3), e12990 https://doi.org/10.1111/ ifpe.12990.
- Béné, C., Barange, M., Subasinghe, R., Pinstrup-Andersen, P., Merino, G., Hemre, G.I., Williams, M., 2015. Feeding 9 billion by 2050–Putting fish back on the menu. Food Secur. 7 (2), 261–274. https://doi.org/10.1007/s12571-015-0427-z.
- Cadavid-Rodríguez, L.S., Vargas-Muñoz, M.A., Plácido, J., 2019. Biomethane from fish waste as a source of renewable energy for artisanal fishing communities. Sustain. Energy Technol. Assessments 34, 110–115. https://doi.org/10.1016/j. seta 2019.05.006
- Caruso, G., 2016. Fishery wastes and by-products: a resource to be valorised. J. FisheriesSciences.com 10 (1), 80–83. URL. http://www.fisheriessciences.com/fish
- Chang, Y., Li, G., Yao, Y., Zhang, L., Yu, C., 2016. Quantifying the water-energy-food nexus: current status and trends. Energies 9 (2), 1–17. https://doi.org/10.3390/ en9020065.
- Chi, C.F., Wang, B., Hu, F.Y., Wang, Y.M., Zhang, B., Deng, S.G., Wu, C.W., 2015. Purification and identification of three novel antioxidant peptides from protein hydrolysate of bluefin leatherjacket (Navodon septentrionalis) skin. Food Res. Int. 73, 124–129. https://doi.org/10.1016/j.foodres.2014.08.038.
- Ching, Y.C., Redzwan, G., 2017. Biological treatment of fish processing saline wastewater for reuse as liquid fertilizer. Sustainability 9 (7). https://doi.org/10.3390/ su9071062.
- Cho, J.H., Kim, I.H., 2011. Fish meal–nutritive value. J. Anim. Physiol. Anim. Nutr. 95 (6), 685–692. https://doi.org/10.1111/j.1439-0396.2010.01109.x.
- Clinton, M., Ferrier, D.E., Martin, S.A., Brierley, A.S., 2021. Impacts of jellyfish on marine cage aquaculture: an overview of existing knowledge and the challenges to finfish health. ICES (Int. Counc. Explor. Sea) J. Mar. Sci. 78 (5), 1557–1573. https:// doi.org/10.1093/icesjms/fsaa254.
- Collins, C., Bresnan, E., Brown, L., Falconer, L., Guilder, J., Jones, L., Kennerley, A., Malham, S., Murray, A., Stanley, M., 2020. Impacts of Climate Change on Aquaculture, pp. 482–520. https://doi.org/10.14465/2020.arc21.aqu.
- Colombo, S.M., Roy, K., Mraz, J., Wan, A.H., Davies, S.J., Tibbetts, S.M., Øverland, M., Francis, D.S., Rocker, M.M., Gasco, L., Spencer, E., 2022. Towards achieving circularity and sustainability in feeds for farmed blue foods. Revi. Aquacult. Early view. https://doi.org/10.1111/raq.12766.

Cooney, R., Wan, A.H., O'Donncha, F., Clifford, E., 2021. Designing environmentally efficient aquafeeds through the use of multicriteria decision support tools. Curr. Opinion. Environ. Sci. Health. 23 https://doi.org/10.1016/j.coesh.2021.100276.

Coppola, D., Lauritano, C., Palma Esposito, F., Riccio, G., Rizzo, C., de Pascale, D., 2021. Fish waste: from problem to valuable resource. Mar. Drugs 19 (2). https://doi.org/ 10.3390/md19020116.

Cortés, A., Esteve-Llorens, X., González-García, S., Moreira, M.T., Feijoo, G., 2021a. Multi-product strategy to enhance the environmental profile of the canning industry towards circular economy. Sci. Total Environ. 791, 148249 https://doi.org/ 10.1016/j.scitotenv.2021.148249.

Cortés, A., González-García, S., Franco-Uría, A., Moreira, M.T., Feijoo, G., 2021b. Evaluation of the environmental sustainability of the inshore great scallop (Pecten maximus) fishery in Galicia. J. Indus. Ecol. Early view. https://doi.org/10.1111/ jiec.13153.

Dauda, A.B., Ajadi, A., Tola-Fabunmi, A.S., Akinwole, A.O., 2019. Waste production in aquaculture: sources, components and managements in different culture systems. Aquacult. Fisheries. 4 (3), 81–88. https://doi.org/10.1016/j.aaf.2018.10.002.

de la Caba, K., Guerrero, P., Trung, T.S., Cruz-Romero, M., Kerry, J.P., Fluhr, J., Maurer, M., Kruijssen, F., Albalat, A., Bunting, S., Burt, S., 2019. From seafood waste to active seafood packaging: an emerging opportunity of the circular economy. J. Clean. Prod. 208, 86–98. https://doi.org/10.1016/j.jclepro.2018.09.164.

Del Campo, L.M., Ibarra, P., Gutiérrez, X., Takle, H.R., 2010. Utilization of sludge from recirculation aquaculture systems. Nofima rapportserie. Available from: http://hdl. handle.net/11250/2559653, 16/January0/2022.

Dineshbabu, A.P., Radhakrishnan, E.V., Thomas, S., Maheswarudu, G., Manojkumar, P. P., Kizhakudan, S.J., Pillai, S.L., Chakraborty, R.D., Josileen, J., Sarada, P.T., Sawant, P.B., 2013. Appraisal of trawl fisheries of India with special reference on the changing trends in bycatch utilization. J. Mar. Biol. Assoc. India 55 (2), 69–78. Available from: http://eprints.cmfri.org.in/id/eprint/10057, 16/January0/2022.

EC, 2022. Discarding in fisheries. European commission, ocean and fisheries. Available from: https://oceans-and-fisheries.ec.europa.eu/fisheries/rules/discarding-fisheries \_en, 15/January0/2022.

Egerton, S., Wan, A., Murphy, K., Collins, F., Ahern, G., Sugrue, I., Busca, K., Egan, F., Muller, N., Whooley, J., McGinnity, P., 2020. Replacing fishmeal with plant protein in Atlantic salmon (*Salmo salar*) diets by supplementation with fish protein hydrolysate. Sci. Rep. 10 (1), 1–16.

El-Gendy, N.S., Hamdy, A., Abu Amr, S.S., 2014. An investigation of biodiesel production from wastes of seafood restaurants. Int. J. Biomater. 2014, 609624 https://doi.org/ 10.1155/2014/609624.

Ennaas, N., Hammami, R., Beaulieu, L., Fliss, I., 2015. Purification and characterization of four antibacterial peptides from protamex hydrolysate of Atlantic mackerel (Scomber scombrus) by-products. Biochem. Biophys. Res. Commun. 462 (3), 195–200. https://doi.org/10.1016/j.bbrc.2015.04.091.

Erasmus, V.N., Kadhila, T., Gabriel, N.N., Thyberg, K.L., Ilungu, S., Machado, T., 2021. Assessment and quantification of Namibian seafood waste production. Ocean Coast Manag. 199, 105402 https://doi.org/10.1016/j.ocecoaman.2020.105402.

European Commission, 2018. EU food loss and waste prevention hub. Available from: https://ec.europa.eu/food/safety/food\_waste/eu-food-loss-waste-prevention-hub/ about, 16/January0/2022.

European Commission, 2019. The European green deal. European Commission Brussels. COM(2019) 630 final. pp.1–24. Available from. https://ec.europa.eu/info/sites/de fault/files/european-green-deal-communication\_en.pdf, 16/January0/2022.

European Commission, 2020a. Communication from the commission to the European parliament, the council, the European economic and social committee and the committee of the regions: a farm to Fork strategy for a fair, healthy and environmentally-friendly food system. European Commission Brussels. COM/2020/ 381 final Available from. https://eur-lex.europa.eu/legal-content/EN/TXT/? uri=CELEX%3A52020DC0381, 16/January/2020.

European Commission, 2020b. Communication from the commission to the European parliament, the European council, the council, the European economic and social committee and the committee of the regions. A new Circular Economy Action Plan. For a cleaner and more competitive Europe 1–19. European Commission Brussels. COM(2020) 98 final. Available from: https://eur-lex.europa.eu/legal-content/EN /TXT/?uri=COM%3A2020%3A98%3AFIN, 11/March/2020.

European Commission, 2022. Discarding in fisheries. European Commission. Available from: https://oceans-and-fisheries.ec.europa.eu/fisheries/rules/discarding-fisheries \_en, 6/January0/2022.

FAO, 2018. The state of world fisheries and aquaculture 2018 - meeting the sustainable development goals. Rome. FAO. pp. 1–227. Available from: https://www.fao.org/3/ i9540en/i9540en.pdf.

FAO, 2020. The state of world fisheries and aquaculture 2020. Sustain. Action: Food. Agricult. Organization. United Nations 1–244. https://doi.org/10.4060/ca9229en. FAO.

FAO, 2022. The state of world fisheries and aquaculture 2022. https://doi.org/10.4060/ cc0461en [10/10/22. FAO. pp. 1–266. Available from:

Fernández-Ríos, A., Laso, J., Campos, C., Ruiz-Salmón, I., Hoehn, D., Cristóbal, J., Batlle-Bayer, L., Bala, A., Fullana-i-Palmer, P., Puig, R., Aldaco, R., 2021. Towards a Water-Energy-Food (WEF) nexus index: a review of nutrient profile models as a fundamental pillar of food and nutrition security. Sci. Total Environ. 789, 147936 https://doi.org/10.1016/j.scitotenv.2021.147936.

Gephart, J.A., Troell, M., Henriksson, P.J., Beveridge, M.C., Verdegem, M., Metian, M., Mateos, L.D., Deutsch, L., 2017. The 'seafood gap' in the food-water nexus literature—issues surrounding freshwater use in seafood production chains. Adv. Water Resour. 110, 505–514. https://doi.org/10.1016/j.advwatres.2017.03.025.

Ghalamara, S., Silva, S., Brazinha, C., Pintado, M., 2020. Valorization of fish by-products: purification of bioactive peptides from codfish blood and sardine cooking wastewaters by membrane processing. Membranes 10 (3), 44. https://doi.org/ 10.3390/membranes10030044.

Gheewala, S.H., Silalertruksa, T., 2021. Life cycle thinking in a circular economy. In: Liu, L., Ramakrishna, S. (Eds.), An Introduction to Circular Economy. Springer, Singapore, pp. 35–53. https://doi.org/10.1007/978-981-15-8510-4\_3.

Gherghel, A., Teodosiu, C., De Gisi, S., 2019. A review on wastewater sludge valorisation and its challenges in the context of circular economy. J. Clean. Prod. 228, 244–263. https://doi.org/10.1016/j.jclepro.2019.04.240.

Ghosh, P.R., Faweett, D., Sharma, S.B., Poinern, G.E.J., 2016. Progress towards sustainable utilisation and management of food wastes in the global economy. Int. J. Food Sci. 2016, 3563478 https://doi.org/10.1155/2016/3563478.

Gupta, A., Gangwar, K., Parihar, R., 2016. In vitro bioavailability of calcium in products incorporated with green leafy vegetables. Int. J.Food Sci. 2 (1).

- Gustafsson, J., Cederberg, C., Sonesson, U., Emanuelsson, A., 2013. The methodology of the FAO study: global food losses and food waste-extent, causes and prevention". FAO, 2011. SIK Institutet för livsmedel och bioteknik 1–70. Available from: http: //www.diva-portal.org/smash/get/diva2:944159/FULLTEXT01.pdf, 11/January0/ 2022.
- Haque, A.S.M., Asaduzzaman, A.K.M., Chun, B.S., 2014. Fatty acid composition and stability of extracted mackerel muscle oil and oil-polyethylene glycol particles formed by gas saturated solution process. Fisheries. Aquatic Sci. 17 (1), 67–73. https://doi.org/10.5657/FAS.2014.0067.
- Hassoun, A., Siddiqui, S.A., Smaoui, S., Ucak, İ., Arshad, R.N., Garcia-Oliveira, P., Prieto, M.A., Aït-Kaddour, A., Perestrelo, R., Câmara, J.S., Bono, G., 2022. Seafood processing, preservation, and analytical techniques in the age of industry 4.0. Appl. Sci. 12 (3), 1703. https://doi.org/10.3390/app12031703.
- Hayes, M., 2021. Food waste reduction for peptide production: food by-products and gut endogenous proteins as sources of bioactive peptides for use in health maintenance and disease prevention. Curr. Pharmaceut. Des. 27 (11) https://doi.org/10.2174/ 138161282711210402151234, 1331-1331.
- Hayes, M., Gallagher, M., 2019. Processing and recovery of valuable components from pelagic blood-water waste streams: a review and recommendations. J. Clean. Prod. 215, 410–422. https://doi.org/10.1016/j.jclepro.2019.01.028.
- Herpandi, N.H., Rosma, A., Wan Nadiah, W.A., 2011. The tuna fishing industry: a new outlook on fish protein hydrolysates. Compr. Rev. Food Sci. Food Saf. 10 (4), 195–207. https://doi.org/10.1111/j.1541-4337.2011.00155.x.

Islam, M.J., Peñarubia, O.R., 2021. Seafood waste management status in Bangladesh and potential for silage production. Sustainability 13 (4). https://doi.org/10.3390/ su13042372.

Jacob, C., Noirot, C., Anglada, C., Binet, T., 2021. The benefits of integrating socioeconomic dimensions of circular economy practices in the seafood sector. Curr. Opinion. Environ. Sci. Health. 22 (100255), 1–7. https://doi.org/10.1016/j. coesh.2021.100255.

- Jayathilakan, K., Sultana, K., Radhakrishna, K., Bawa, A.S., 2012. Utilization of byproducts and waste materials from meat, poultry and fish processing industries: a review. J. Food Sci. Technol. 49 (3), 278–293. https://doi.org/10.1007/s13197-011-0290-7.
- Kang, M.S., Jung, Y.G., Jang, D.H., 2017. A study on the search of optimal aquaculture farm condition based on machine learning. J. Inst.Internet, Broadcasting. Commun. 17 (2), 135–140. https://doi.org/10.7236/JIIBC.2017.17.2.135.

Karkal, S.S., Kudre, T.G., 2020. Valorization of fish discards for the sustainable production of renewable fuels. J. Clean. Prod. 275, 122985 https://doi.org/ 10.1016/j.jclepro.2020.122985.

Karnjanapratum, S., O'CALLAGHAN, Y.C., Benjakul, S., O'Keeffe, M.B., FitzGerald, R.J., O'Brien, N.M., 2017. Purification and identification of antioxidant peptides from gelatin hydrolysates of unicorn leatherjacket skin. Ital. J. Food Sci. 29 (1), 158–170. https://doi.org/10.14674/1120-1770/ijfs.v588.

Kelliher, F., Mellett, S., Harrington, D., 2020. Enablers of green innovation in the microfirm—perspectives from Ireland and Canada. J. Small Bus. Enterpren. 34 (1), 74–93. https://doi.org/10.1080/08276331.2020.1789826.

Kim, J.K., 2011. Cost-effectiveness of converting fish waste into liquid fertilizer. Fisheries. Aquatic Sci. 14 (3), 230–233. https://doi.org/10.5657/FAS.2011.0230.

- Kim, J.K., Kong, I.S., Lee, H.H., 2010. Identification and characterization of microorganisms from earthworm viscera for the conversion of fish wastes into liquid fertilizer. Bioresour. Technol. 101 (14), 5131–5136. https://doi.org/10.1016/j. biortech.2010.02.001.
- Komoroske, L.M., Lewison, R.L., 2015. Addressing fisheries bycatch in a changing world. Front. Mar. Sci. 2 (83), 1–11. https://doi.org/10.3389/fmars.2015.00083.
- Kruijssen, F., Tedesco, I., Ward, A., Pincus, L., Love, D., Thorne-Lyman, A.L., 2020. Loss and waste in fish value chains: a review of the evidence from low and middle-income countries. Global Food Secur. 26 (100434), 1–13. https://doi.org/10.1016/j. gfs.2020.100434.
- Kumolu-Johnson, C.A., Ndimele, P.E., 2011. A review on post-harvest losses in artisanal fisheries of some African countries. J. Fish. Aquat. Sci. 6 (4), 365–378. https://doi. org/10.3923/jfas.2011.365.378.

Kuvendziev, S., Lisichkov, K., Zeković, Z., Marinkovski, M., Musliu, Z.H., 2018. Supercritical fluid extraction of fish oil from common carp (Cyprinus carpio L.) tissues. J. Supercrit. Fluids 133, 528–534. https://doi.org/10.1016/j. supflu.2017.11.027.

Lam, C.M., Iris, K.M., Hsu, S.C., Tsang, D.C., 2020. Life-cycle assessment of food waste recycling. In: Waste Biorefinery, pp. 481–513. https://doi.org/10.1016/B978-0-12-818228-4.00018-6.

Laso, J., Margallo, M., Celaya, J., Fullana, P., Bala, A., Gazulla, C., Irabien, A., Aldaco, R., 2016. Waste management under a life cycle approach as a tool for a circular economy in the canned anchovy industry. Waste Manag. Res. 34 (8), 724–733. https://doi.org/10.1177/0734242X16652957.

- Laso, J., Margallo, M., Serrano, M., Vázquez-Rowe, I., Avadí, A., Fullana, P., Bala, A., Gazulla, C., Irabien, Á., Aldaco, R., 2018. Introducing the Green Protein Footprint method as an understandable measure of the environmental cost of anchovy consumption. Sci. Total Environ. 621, 40–53. https://doi.org/10.1016/j. scitotenv.2017.11.148.
- Laurens, L.M., Markham, J., Templeton, D.W., Christensen, E.D., Van Wychen, S., Vadelius, E.W., Chen-Glasser, M., Dong, T., Davis, R., Pienkos, P.T., 2017. Development of algae biorefinery concepts for biofuels and bioproducts; a perspective on process-compatible products and their impact on cost-reduction. Energy Environ. Sci. 10 (8), 1716–1738. https://doi.org/10.1039/C7EE01306J.
- Liu, Y., Rosten, T.W., Henriksen, K., Hognes, E.S., Summerfelt, S., Vinci, B., 2016. Comparative economic performance and carbon footprint of two farming models for producing Atlantic salmon (*Salmo salar*): land-based closed containment system in freshwater and open net pen in seawater. Aquacult. Eng. 71, 1–12. https://doi.org/ 10.1016/j.aquaeng.2016.01.001.
- Love, D.C., Fry, J.P., Milli, M.C., Neff, R.A., 2015. Wasted seafood in the United States: quantifying loss from production to consumption and moving toward solutions. Global Environ. Change 35, 116–124. https://doi.org/10.1016/j. gloenvcha.2015.08.013.
- López-Cortés, X.A., Nachtigall, F.M., Olate, V.R., Araya, M., Oyanedel, S., Diaz, V., Jakob, E., Ríos-Momberg, M., Santos, L.S., 2017. Fast detection of pathogens in salmon farming industry. Aquaculture 470, 17–24. https://doi.org/10.1016/j. aquaculture.2016.12.008.
- Madende, M., Hayes, M., 2020. Fish by-product use as biostimulants: an overview of the current state of the art, including relevant legislation and regulations within the EU and USA. Molecules 25 (5). https://doi.org/10.3390/molecules25051122.
- Mahfouz, A., Allen, D., Arisha, A., Elbert, R., Gleser, M., 2019. A Post-Brexit transportation scenario analysis for an agri-fresh produce supply chain. In: 2019 Winter Simulation Conference (WSC) Pp. 1789-1800. IEEE. https://doi.org/ 10.1109/WSC40007.2019.9004790.
- Maiolo, S., Parisi, G., Biondi, N., Lunelli, F., Tibaldi, E., Pastres, R., 2020. Fishmeal partial substitution within aquafeed formulations: life cycle assessment of four alternative protein sources. Int. J. Life Cycle Assess. 25 (8), 1455–1471. https://doi. org/10.1007/s11367-020-01759-z.
- Manoharan, H., Teekaraman, Y., Kshirsagar, P.R., Sundaramurthy, S., Manoharan, A., 2020. Examining the effect of aquaculture using sensor-based technology with machine learning algorithm. Aquacult. Res. 51 (11) https://doi.org/10.1111/ are.14821.
- Martins, C.I.M., Eding, E.H., Verdegem, M.C., Heinsbroek, L.T., Schneider, O., Blancheton, J.P., d'Orbcastel, E.R., Verreth, J.A.J., 2010. New developments in recirculating aquaculture systems in Europe: a perspective on environmental sustainability. Aquacult. Eng. 43 (3), 83–93. https://doi.org/10.1016/j. aquaeng.2010.09.002.
- Martínez-Córdova, L.R., Martínez-Porchas, M., Emerenciano, M.G.C., Miranda-Baeza, A., Gollas-Galván, T., 2017. From microbes to fish the next revolution in food production. Crit. Rev. Biotechnol. 37 (3), 287–295. https://doi.org/10.3109/ 07388551.2016.1144043.
- Masilan, K., Neethiselvan, N., Sivaraman, B., Ravikumar, T., Nagarajan, M., Karthy, A., Sukumar, D., 2021. Valorization of discarded industrial fish processing wastes for the extraction of gelatin to use as biodegradable fish bait matrix using RSM. PeerJ Mater. Sci. 3, e14. https://doi.org/10.7717/peerj-matsci.14.
- Mathew, G.M., Mathew, D.C., Sukumaran, R.K., Sindhu, R., Huang, C.C., Binod, P., Sirohi, R., Kim, S.H., Pandey, A., 2020. Sustainable and eco-friendly strategies for shrimp shell valorization. Environ. Pollut. 267 https://doi.org/10.1016/j. envpol.2020.115656.
- Maulu, S., Hasimuna, O.J., Haambiya, L.H., Monde, C., Musuka, C.G., Makorwa, T.H., Munganga, B.P., Phiri, K.J., Nsekanabo, J.D., 2021. Climate change effects on aquaculture production: sustainability implications, mitigation, and adaptations. Front. Sustain. Food Syst. 5 https://doi.org/10.3389/fsufs.2021.609097.
- Mbatia, B., Adlercreutz, D., Adlercreutz, P., Mahadhy, A., Mulaa, F., Mattiasson, B., 2010. Enzymatic oil extraction and positional analysis of  $\omega$ -3 fatty acids in Nile perch and salmon heads. Process Biochem. 45 (5), 815–819. https://doi.org/10.1016/j. procbio.2010.02.010.
- Milhazes-Cunha, H., Otero, A., 2017. Valorisation of aquaculture effluents with microalgae: the integrated multi-trophic aquaculture concept. Algal Res. 24, 416–424. https://doi.org/10.1016/j.algal.2016.12.011.
- Mousavi, S.L., Mohammadi, G., Khodadadi, M., Keysami, M.A., 2013. Silage production from fish waste in cannery factories of Bushehr city using mineral acid, organic acid, and biological method. Intl. J. Agric. Crop Sci. 6 (10), 610–616.
- Murphy, E.J., Masterson, C., Rezoagli, E., O'Toole, D., Major, I., Stack, G.D., Lynch, M., Laffey, J.G., Rowan, N.J., 2020. β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—implications for coronavirus disease (COVID-19) immunotherapies. Sci. Total Environ. 732 https://doi.org/10.1016/j. scitotenv.2020.139330.
- Naik, A.S., Whitaker, R.D., Albrektsen, S., Solstad, R.G., Thoresen, L., Hayes, M., 2021. Mesopelagic fish protein hydrolysates and extracts: a source of novel antihypertensive and anti-diabetic peptides. Front. Mar. Sci. 8 https://doi.org/10.3389/ fmars.2021.719608.
- Nawaz, A., Li, E., Irshad, S., Xiong, Z., Xiong, H., Shahbaz, H.M., Siddique, F., 2020. Valorization of fisheries by-products: challenges and technical concerns to food industry. Trends Food Sci. Technol. 99, 34–43. https://doi.org/10.1016/j. tifs.2020.02.022.
- Naylor, R.L., Hardy, R.W., Buschmann, A.H., Bush, S.R., Cao, L., Klinger, D.H., Little, D. C., Lubchenco, J., Shumway, S.E., Troell, M., 2021. A 20-year retrospective review of

global aquaculture. Nature 591 (7851), 551–563. https://doi.org/10.1038/s41586-021-03308-6.

- Nedelciu, C.E., Ragnarsdottir, K.V., Schlyter, P., Stjernquist, I., 2020. Global phosphorus supply chain dynamics: assessing regional impact to 2050. Global Food Secur. 26, 100426.
- Nges, I.A., Mbatia, B., Björnsson, L., 2012. Improved utilization of fish waste by anaerobic digestion following omega-3 fatty acids extraction. J. Environ. Manag. 110, 159–165. https://doi.org/10.1016/j.jenvman.2012.06.011.
- Ngo, D.H., Ryu, B., Kim, S.K., 2014. Active peptides from skate (Okamejei kenojei) skin gelatin diminish angiotensin-I converting enzyme activity and intracellular free radical-mediated oxidation. Food Chem. 143, 246–255. https://doi.org/10.1016/j. foodchem.2013.07.067.
- Pan, S.Y., Du, M.A., Huang, I.T., Liu, I.H., Chang, E.E., Chiang, P.C., 2015. Strategies on implementation of waste-to-energy (WTE) supply chain for circular economy system: a review. J. Clean. Prod. 108, 409–421. https://doi.org/10.1016/j. iclepro.2015.06.124.
- Parvathy, U., Jeyakumari, K.R.A., Zynudheen, A.A., 2017. Biological treatment systems for fish processing wastewater-A review. Bioresour. Technol. 101 (2), 439–449. https://doi.org/10.1016/j.biortech.2009.08.065.
- Pogue, R., Murphy, E.J., Fehrenbach, G.W., Rezoagli, E., Rowan, N.J., 2021. Exploiting immunomodulatory properties of β-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms: concise review of existing knowledge, innovation and future opportunities. Curr. Opinion. Environ. Sci. Health. 21, 100248 https://doi.org/10.1016/j.coesh.2021.100248.
- Radziemska, M., Vaverková, M.D., Adamcová, D., Brtnický, M., Mazur, Z., 2019. Valorization of fish waste compost as a fertilizer for agricultural use. Waste. Biomass. Valorization. 10 (9), 2537–2545. https://doi.org/10.1007/s12649-018-0288-8.
- Raimondo, M., Caracciolo, F., Cembalo, L., Chinnici, G., Pappalardo, G., D'Amico, M., 2021. Moving towards circular bioeconomy: managing olive cake supply chain through contracts. Sustain. Prod. Consum. 28, 180–191. https://doi.org/10.1016/j. spc.2021.03.039.
- Regueiro, L., Newton, R., Soula, M., Mendez, D., Kok, B., Little, D.C., Pastres, R., Johansen, J., Ferreira, M., 2021. Opportunities and limitations for the introduction of circular economy principles in EU aquaculture based on the regulatory framework. J. Ind. Ecol. https://doi.org/10.1111/jiec.13188. Early view.
- Rejula, K., Mohanty, A.K., 2018. Innovative extension approaches for fish waste management. ICAR-Central Institute of Fisheries Technology, Cochin. pp. 236–242. Available from: https://krishi.icar.gov.in/jspui/handle/123456789/20435. (Accessed 10 January 2022).
- Rohm, H., Oostindjer, M., Aschemann-Witzel, J., Symmank, C., Almli, V.L., De Hooge, I. E., Normann, A., Karantininis, K., 2017. Consumers in a sustainable food supply chain (COSUS): understanding consumer behavior to encourage food waste reduction. Foods 6 (12), 104. https://doi.org/10.3390/foods6120104.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: quo Vadis? Sci. Total Environ. 748, 141362 https://doi.org/10.1016/j. scitotenv.2020.141362.
- Ruiz-Salmón, I., Laso, J., Margallo, M., Villanueva-Rey, P., Rodríguez, E., Quinteiro, P., Dias, A.C., Almeida, C., Nunes, M.L., Marques, A., Cortés, A., 2021. Life cycle assessment of fish and seafood processed products–a review of methodologies and new challenges. Sci. Total Environ. 761, 144094 https://doi.org/10.1016/j. scitotenv.2020.144094.
- Ruiz-Salmón, I., Margallo, M., Laso, J., Villanueva-Rey, P., Mariño, D., Quinteiro, P., Dias, A.C., Nunes, M.L., Marques, A., Feijoo, G., Moreira, M.T., 2020. Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Curr. Opinion. Environ. Sci. Health. 13, 101–106. https://doi. org/10.1016/j.coesh.2020.01.004.
- Sainsbury, N.C., Schuhmann, P.W., Turner, R.A., Grilli, G., Pinnegar, J.K., Genner, M.J., Simpson, S.D., 2021. Trade-offs between physical risk and economic reward affect Fishers' vulnerability to changing storminess. Global Environ. Change 69, 102228. https://doi.org/10.1016/j.gloenvcha.2021.102228.
- Saleh, N., Wassef, E., Barakat, K., Helmy, H., 2020. Bioconversion of fish-wastes biomass into a fish meal alternate for European seabass (*Dicentrarchus labrax*) diets. Int. Aquat. Res. 12 (1), 40–52. https://doi.org/10.22034/IAR(20).2020.671145.
- Shah, M.R., Lutzu, G.A., Alam, A., Sarker, P., Chowdhury, K., Parsaeimehr, A., Liang, Y., Daroch, M., 2018. Microalgae in aquafeeds for a sustainable aquaculture industry. J. Appl. Phycol. 30 (1), 197–213. https://doi.org/10.1007/s10811-017-1234-z.
- Sharma, H.B., Vanapalli, K.R., Samal, B., Cheela, V.S., Dubey, B.K., Bhattacharya, J., 2021. Circular economy approach in solid waste management system to achieve UN-SDGs: solutions for post-COVID recovery. Sci. Total Environ. 800, 149605 https:// doi.org/10.1016/j.scitotenv.2021.149605.
- Slorach, P.C., Jeswani, H.K., Cuéllar-Franca, R., Azapagic, A., 2020. Environmental sustainability in the food-energy-water-health nexus: a new methodology and an application to food waste in a circular economy. Waste Manag. 113, 359–368. https://doi.org/10.1016/j.wasman.2020.06.012.
- Soma, T., 2020. Cradle to cradle: the role of food waste in a regenerative food system. In: Duncan, J., Carolan, M., Wiskerke, J.S.C. (Eds.), Routledge Handbook of Sustainable and Regenerative Food Systems. Routledge, Taylor and Francis Group, London, UK and New York, USA, pp. 406–419.
- Song, R., Wei, R.B., Luo, H.Y., Wang, D.F., 2012. Isolation and characterization of an antibacterial peptide fraction from the pepsin hydrolysate of half-fin anchovy (*Setipinna taty*). Molecules 17 (3), 2980–2991. https://doi.org/10.3390/ molecules17032980.
- Song, X., Liu, Y., Pettersen, J.B., Brandão, M., Ma, X., Røberg, S., Frostell, B., 2019. Life cycle assessment of recirculating aquaculture systems: a case of Atlantic salmon

#### R. Cooney et al.

farming in China. J. Ind. Ecol. 23 (5), 1077–1086. https://doi.org/10.1111/jiec.12845.

- Sorg, F., Kahle, J., Wehner, N., Mangold, M., Peters, S., 2019. Clarity in diversity: how the sustainability standards comparison tool and the global sustainable seafood initiative provide orientation. In: Sustainable Global Value Chains, pp. 239–264. https://doi.org/10.1007/978-3-319-14877-9\_14.
- Spang, E.S., Moreno, L.C., Pace, S.A., Achmon, Y., Donis-Gonzalez, I., Gosliner, W.A., Jablonski-Sheffield, M.P., Momin, M.A., Quested, T.E., Winans, K.S., Tomich, T.P., 2019. Food loss and waste: measurement, drivers, and solutions. Annu. Rev. Environ. Resour. 44, 117–156. https://doi.org/10.1146/annurev-environ-101718-033228.
- Spierling, S., Venkatachalam, V., Mudersbach, M., Becker, N., Herrmann, C., Endres, H. J., 2020. End-of-life options for bio-based plastics in a circular economy—status quo and potential from a life cycle assessment perspective. Resources 9 (7), 90. https:// doi.org/10.3390/resources9070090.
- Stead, S.M., 2019. Using systems thinking and open innovation to strengthen aquaculture policy for the United Nations Sustainable Development Goals. J. Fish. Biol. 94 (6), 837–844. https://doi.org/10.1111/jfb.13970.
- Stenmarck, Â., Jensen, C., Quested, T., Moates, G., Buksti, M., Cseh, B., Juul, S., Parry, A., Politano, A., Redlingshofer, B., Scherhaufer, S., 2016. Estimates of European food waste levels. IVL Swedish Environmental Research Institute, Sweden. Available from: https://www.eu-fusions.org/phocadownload/Publications/Estimates%200f% 20European%20food%20waste%20levels.pdf, 16/January0/2022.
- Suryawanshi, N., Eswari, J.S., 2022. Chitin from seafood waste: particle swarm optimization and neural network study for the improved chitinase production. J. Chem. Technol. Biotechnol. 97 (2), 509–519. https://doi.org/10.1002/jctb.6656.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, e5281. https://doi.org/10.7717/peerj.5281.
- Tan, E.C., Lamers, P., 2021. Circular bioeconomy concepts—a perspective. Frontier. Sustain. 2, 701509 https://doi.org/10.3389/frsus.2021.701509.
- Tremblay, A., Corcuff, R., Goulet, C., Godefroy, S.B., Doyen, A., Beaulieu, L., 2020. Valorization of snow crab (*Chionoecetes opilio*) cooking effluents for food applications. J. Sci. Food Agric. 100 (1), 384–393. https://doi.org/10.1002/ jsfa.10066.
- Uddin, M.S., Kishimura, H., Chun, B.S., 2011. Isolation and characterization of lecithin from squid (*Todarodes pacificus*) viscera deoiled by supercritical carbon dioxide extraction. J. Food Sci. 76 (2), C350–C354. https://doi.org/10.1111/j.1750-3841.2010.02039.x.

- Valenti, W.C., Kimpara, J.M., Preto, B.d.L., Moraes-Valenti, P., 2018. Indicators of sustainability to assess aquaculture systems. Ecol. Indicat. 88, 402–413.
- Van Rijn, J., 2013. Waste treatment in recirculating aquaculture systems. Aquacult. Eng. 53, 49–56. https://doi.org/10.1016/j.aquaeng.2012.11.010.
- Venugopal, V., 2021. Valorization of seafood processing discards: bioconversion and biorefinery approaches. Front. Sustain. Food Syst. 5, 611835 https://doi.org/10.3389/ fsufs.2021.611835.
- Venugopal, V., Sasidharan, A., 2021. Seafood industry effluents: environmental hazards, treatment and resource recovery. J. Environ. Chem. Eng. 9 (2) https://doi.org/ 10.1016/j.jece.2020.104758.
- Vilariño, M.V., Franco, C., Quarrington, C., 2017. Food loss and waste reduction as an integral part of a circular economy. Front. Environ. Sci. 5, 21. https://doi.org/ 10.3389/fenvs.2017.00021.
- Villanueva-Rey, P., Vázquez-Rowe, I., Arias, A., Moreira, M.T., Feijoo, G., 2018. The importance of using life cycle assessment in policy support to determine the sustainability of fishing fleets: a case study for the small-scale xeito fishery in Galicia, Spain. Int. J. Life Cycle Assess. 23 (5), 1091–1106. https://doi.org/10.1007/s11367-017-1402-z.
- Voyer, M., van Leeuwen, J., 2019. 'Social license to operate' in the blue economy. Resour. Pol. 62, 102–113. https://doi.org/10.1016/j.resourpol.2019.02.020.
- Vázquez, J.A., Pastrana, L., Piñeiro, C., Teixeira, J.A., Pérez-Martín, R.I., Amado, I.R., 2015. Production of hyaluronic acid by *Streptococcus zooepidemicus* on protein substrates obtained from *Scyliorhinus anicular* discards. Mar. Drugs 13 (10), 6537–6549. https://doi.org/10.3390/md13106537.
- Vazquez-Rowe, I., Moreira, M.T., Feijoo, G., 2012. Environmental assessment of frozen common octopus (Octopus vulgaris) captured by Spanish fishing vessels in the Mauritanian EEZ. Mar. Pol. 36 (1), 180–188. https://doi.org/10.1016/j. marpol.2011.05.002.
- Wang, C.H., Doan, C.T., Nguyen, V.B., Nguyen, A.D., Wang, S.L., 2019. Reclamation of fishery processing waste: a mini-review. Molecules 24 (12), 2234. https://doi.org/ 10.3390/molecules24122234.
- Wojnowska-Baryła, I., Kulikowska, D., Bernat, K., 2020. Effect of bio-based products on waste management. Sustainability 12 (5), https://doi.org/10.3390/su12052088.
- Yan, N., Chen, X., 2015. Sustainability: don't waste seafood waste. Nature 524 (7564), 155–157. https://doi.org/10.1038/524155a.
- Zilia, F., Bacenetti, J., Sugni, M., Matarazzo, A., Orsi, L., 2021. From waste to product: circular economy applications from sea urchin. Sustainability 13 (10), 5427. https:// doi.org/10.3390/su13105427.

Contents lists available at ScienceDirect

## Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

#### Review

# Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis?

## Neil J. Rowan<sup>a,b,c,\*</sup>, Charis M. Galanakis<sup>d,e,f</sup>

<sup>a</sup> Bioscience Research Institute, Athlone Institute of Technology, Ireland

- <sup>b</sup> Centre for Disinfection, Sterilization, and Biosecurity, Athlone Institute of Technology, Ireland
- <sup>c</sup> Empower Eco Sustainability Hub, Lough Boora, Co. Offaly, Ireland
- $^{\rm d}$  Research & Innovation Department, Galanakis Laboratories, Chania, Greece
- <sup>e</sup> College of Science, King Saud University, Riyadh, Saudi Arabia
- f Food Waste Recovery Group, ISEKI Food Association, Vienna, Austria

#### HIGHLIGHTS

GRAPHICAL ABSTRACT

AGRI-FOOD &

MANUFACTURING

SERVICES AND

USINESS PROCESSES

÷

DIGITIZATION

3

CLIMATE ACTION & ENVIRONMENT

٠

ASTE MANAGEMEN

SECURITY & SUSTAINABILITY

- COVID-19 pandemic presents opportunities for sustainable agri-food production and to accelerate green innovation.
   Consider under arbitrage technical
- Sensible yet ambitious technicaleconomical recovery plans are urgently needed when countries reopen.
- COVID-19 may create disruptive technologies that cross-cuts agri-food, ICT, health, and environment.
- Multi-agency converging innovation hubs have the potential to accelerate socio-economic recovery.

#### ARTICLE INFO

Article history: Received 8 June 2020 Received in revised form 27 July 2020 Accepted 28 July 2020 Available online 31 July 2020

Editor: Damia Barcelo

Keywords: COVID-19 Disruptive technologies Agri-food Climate change Green innovation Sustainability



COVID-19 pandemic is on a trajectory to cause catastrophic global upheaval with the potential to alter geopolitical and socio-economic norms. Many countries are frantically responding with staggering financial stimulus recovery initiatives. This opinion-paper reviews challenges, opportunities, and potential solutions for the post-COVID-19 era that focuses on intensive sustaining of agri-food supply chain in tandem with meeting the high demand for new green deal innovation. For example, the development of wet peatland innovation, known as Paludiculture, can intensively sustain and blend agri-food and green innovations that will help support COVID-19 pandemic transitioning. The future looks bright for the creation of new sustainability multi-actor innovation hubs that will support, connect, and enable businesses to recover and pivot beyond the COVID-19 pandemic. The nexus between first 'Green Deal' initiative supporting 64 selected European Startups and SMEs (European Innovation Council) and 43 Irish Disruptive Technology projects are addressed in the context of cross-cutting developments and relevance to COVID-19. Candidate areas for future consideration will focus on climate action, digitization, manufacturing, and sustainable food production, security, and waste mitigation. Recommendations are also provided to facilitate community transitioning, training, enterprise, and employment to low carbon economy.

MULTI-AGENCY

INNOVATION

HUBS

COVID-19

DISRUPTIVE

CHALLENGES

**OPPORTUNITIES** 

SOLUTIONS

ASSESSING & PREDICTING CONSUMER BEHAVIOURAL CHANGE & TRENDS IN REAL TIME

TECHNOLOGIES

© 2020 Elsevier B.V. All rights reserved.

\* Corresponding author at: Centre for Disinfection, Sterilization, and Biosecurity, Athlone Institute of Technology, Ireland. *E-mail address*: nrowan@ait.ie (N.J. Rowan).







#### Contents

1.	Introduction	2
2.	International financial recovery stimulus support initiatives for business during and post COVID-19	2
3.	Agri-food industry – an example of challenges ahead post COVID-19	3
4.	Transitioning COVID-19 crisis by exploiting green peatland innovations (Paludiculture) - a useful nexus between agri-food, climate action, and circ	cu-
	lar economy	3
5.	First "Green Deal" funding from European Innovation Council to support the recovery plan for Europe – strong alignment with Paludiculture platform am	bi-
	tions	6
6.	Solutions to accelerate agri-food recovery for post-COVID-19 potential technology disruption.	7
7.	Definitions of disruptive technologies	7
8.	How might DTs accelerate recovery post-COVID-19 and transition towards sustainable food systems?	8
9.	Strategic funding initiatives to identify and accelerate DTs and relevance to COVID-19 – a case study from the Republic of Ireland	11
10.	Potential technology, product and business service disruptors in food beyond COVID-19 crisis	13
11.	Conclusion	14
Auth	nors' contributions	14
Cons	sent for publication	14
Ackr	nowledgments	14
Refe	rences	14

#### 1. Introduction

Coronavirus (COVID-19) has caused dramatic and unprecedented upheaval to socio-economic norms since first reported in Wuhan, China, earlier this year (Rowan and Laffey, 2020; Guan et al., 2020). COVID-19 pandemic is impossible to predict, with many countries experiencing second or successive waves of infection. Akin to a 'Black Swan' event, there is a significant gap in published research to inform companies where most have neither prepared for it nor have they planned to transition beyond it (Taleb, 2010; Reid et al., 2020). COVID-19 impacts upon health systems, governments, and businesses alike with unprecedented implications for companies worldwide. Companies are feeling the market and financial shock of the COVID-19 outbreak by factories' shutdown, labor shortages to cash flow stress, and disruptions in the supply chain (Reid et al., 2020). The global mitigation response has focused on public health strategies to curtail and reduce the spread of viral transmission through hand hygiene, social distancing, lockdown (staying-at-home, cocooning), and, community wearing of protective face masks that now occurs in 50 countries (Rowan and Laffey, 2020; Silva et al., 2020). This fact has resulted in an unprecedented technical-socioeconomic earthquake that has left many sectors seeking emergency COVID-19-related unemployment benefits (Rowan and Laffey, 2020; Guan et al., 2020). The food supply chain sector does not comprise an exception (Galanakis, 2020). Sarkis et al. (2020) noted that there is a window now for transitioning to sustainable supply chains in the aftermath of COVID-19 that includes rethinking vulnerabilities created by over-reliance on 'just-in-time' or 'business-as-usual' practices. Barcèló (2020) highlighted challenges affecting the environment and health caused by COVID-19 pandemic along with recommendations for monitoring and mitigation.

Similarly to other countries, the Republic of Ireland has been in lockdown for 3 months to flatten the curve in COVID-19 cases. On the other hand, it is in a relatively privileged position where it is adopting a phased reopening of the country that commenced 18th May 2020, with phase five intimated for 10th August 2020 (Fig. 1). While Ireland is emerging from the first phase of reopening, likely, the vast majority of foodservice and businesses will not return to regular operation for several weeks. Covid-19 adjusted employment in the Republic of Ireland has reached 28.2%, compared with traditional 5.4% for the same month last year, which is estimated to cost the Irish exchequer  $\in$ 6.4bn until the country reopens. COVID-19 pandemic has introduced dramatic uncertainty and unpredictability where this volatility has the potential to cause socio-economic chaos across many sectors when Ireland and other countries reopen for business (Guan et al., 2020).

While there are financial stimulus support packages (Government of Ireland, 2020), there is an unprecedented dearth in critical information (such as marketing of new consumer trends and behavioural change to offset sheltering-in-place and social distancing) in order to enabling businesses to make priority decisions on core needs for recovery post-COVID-19 (Sarkis et al., 2020). As there was almost simultaneous closure of countries due to COVID-19 mediated lockdowns (Guan et al., 2020), the Republic of Ireland will re-emerge from the COVID-19 bubble ahead of many other trading countries that will provide a useful lens to gaze through to inform recovery. This scenario will help inform realtime situational-analysis and will reveal the main challenges where different geopolitical and socio-economic landscape now potentially exists internationally. The lack of food security means that countries are dependent upon imports but may not earning sufficient foreign currency to purchase vital imported goods. The post-COVID-19 era also presents equal opportunities to address climate change and to advance green innovation matched with driving sustainable food production and security.

## 2. International financial recovery stimulus support initiatives for business during and post COVID-19

Europe needs to be reopened as quickly as sensible, but this will involve significant risk. In response, there has been major national, EU, and international stimulus financial aid to offset this potential socioeconomic crisis potentially. On 27th May 2020, the European Commission presented an economic stimulus plan of €750 billion to help mitigate the shock from this COVID-19 pandemic and pave the way for a sustainable future. While there is a commitment to support recovery financially, there is currently a lack in specific detail underpinning a techno-socio-economic ecosystem unroadmap to guide F&D companies through and beyond COVID-19 when countries reopen. There is a recognition that the EU will still need to prioritize climate actions and a digital strategy, but also to update the new EU health program. This action is important to ensure continuity of supply chain for food and medical products, services across the EU to offset critical shortages arising for existing and future pandemics. In the Republic of Ireland, the Department of Health and the Department of Agriculture, Food and the Marine (DAFM) are collaborating as part of a holistic response to COVID-19, e.g., although remaining open for business sector (DAFM, 2020). At the time of writing, two very stark documents published in Ireland the Irish Fiscal Advisory Council (2020) and the ESRI UK and Ireland (2020) both warned of ballooning unemployment, huge deficits, massive accumulations of public debt and painful budgetary adjustments

Phase 1	Phase 2	Phase 3	Phase 4	Phase 5
May 18 <sup>th</sup> , 2020	8 <sup>th</sup> June, 2020	29 <sup>th</sup> June, 2020	20 <sup>th</sup> July, 2020	10 <sup>th</sup> August, 2020
Phased return of outdoor	Limited return to onsite	Return to low-interaction	Return to work where	Phased return to work
workers	working subject	work. Remote working	employees	across all sectors.
Remote working	to compliance capability	continues for all that can	cannot remote work.	Remote working continues
continues	Remote working continues	do so.	Staggered hours.	for all that can do so.
for all that can do so.	for all that can do so.		Remote working continues	
			for all that can do so.	
Retail that is mainly	Small retail outlets with	Open non-essential retail	Gradual easing of restrictions	Further easing of restrictions
outdoor and homeware,	control of numbers open.	outlets with street level	on higher-risk services. e.g.	higher-risk
opticians, motor, bicycle	Marts open. All subject to	access.	Barbers and hairdressers	services. e.g. shopping centres,
& repair, office products,	social distancing			tattoo, piercing.
electrical, IT, phone sales				
& repair open. All subject				
to social distancing.				
Farmers markets,	Small retail outlets can	The opening of all other	Restrictions will gradually be	Enclosed shopping centres can
gardners and other	reopen with a small	non-essential retail outlets	decreased on the numbers	reopen, with social distancing.
outdoor workers return	number of staff on the	will be phased in on basis	travelling in major urban	A further loosening of
to work – social	basis that the retailer can	of a restriction on number	centres on public transport	restrictions for services
distancing requirements	control the number of	of staff and customers per	and in private cars.	involving direct physical contact
continue to apply .	individuals that staff and	square metre so that social		for periods of time between
	customers interact with at	distancing can be	Specific measures will be	people for which there is not a
Remote working	any one time.	maintained.	introduced at ports and	population-wide demand for
continues for all others		This is to be limited to	airports.	later phases due to risks .
that can do so .		retail outlets with a street-		Non-resident tourist travel to
Return to Work Safety		level entrance and exit and		offshore islands can resume.
Protocol is the operative		does not include those in		Social distancing and hygiene
guide for employers and		enclosed shopping centres		measures are to continue for
employess		due to higher risk.		public and private transport.
Data source - https://www	.gov.ie/en/news/58bc8b-taois	each-announces-roadmap-for-	-reopening-society-and-business-	and-u/ (accessed June 4 <sup>th</sup> , 2020)

Fig. 1. Reopening of economic, retail and commercial activities in the Republic of Ireland using a phased approach during 2020.

to the tune of up to €14bn by 2025. IBEC (2020), the group that represents Irish business, published proposed solutions to COVID-19 imposed liquidity crisis, e.g., supporting vulnerable businesses using emergency cashflow and liquidity measures from the Irish government. IBEC also denoted the importance of maintaining the operation of food and other processing facilities. This action includes specific liquidity and financing needs of farmers, fishers, and agri-food businesses and for the banks to offer flexibility to their customers at an early stage to discuss emerging cashflow issues.

#### 3. Agri-food industry – an example of challenges ahead post COVID-19

The agri-food sector comprises a dynamic societal-technical innovation ecosystem and is one of the largest manufacturing industries (Saguy et al., 2018; Rowan, 2019). In the EU, this increasingly important sector accounts for €1098 billion turnovers and employs 4.24 million (Saguy et al., 2018). The food and drink (F&D) industry has doubled in size in the United States over the past decade, where it was estimated to be worth £6 trillion in 2015 with packaging comprising almost £1.9 trillion (Statista, 2020a). Revenue in the food market amounts to US \$924,389m in 2020; the market is expected to grow annually by 1.8% (Statista, 2020a). The rate of global population growth is staggering, and there is a commensurate need to match this demand that will put increasing pressure on food production and security (Richie et al., 2018; Michelini et al., 2018). There is a pressing need to diversify the food supply chain and identify sustainable technologies across food systems that will meet changing diets, increasingly aging, ethnic and cultural population, diet-related diseases, more personalized products (Galanakis, 2020). DBEI (2018) projects that there will be a ca. 70% rise in demand for more food products and services over the next 40 years. In the years ahead, it is envisaged that organic, unprocessed, and healthy food will drive growth in domestic markets (Statista, 2020b). For example, the estimated value of shipments of the industry was US\$795.4 billion in 2019, where. 15.1% of the cost of shipments generated from dairy product manufacturing.

The agri-food and beverage sector is Ireland's most important indigenous industry where the sector produced €13bn of exports to overseas markets in 2019 on foot of decades of accelerated growth where the value of exports in this sector increased by 67% (Duffy, 2020). However, at this time of writing, the impact of COVID-19 pandemic with associated international lockdown has pretty much instantaneously shut down world economies. Duffy (2020) reported that COVID-19 would also bring about a dramatic collapse in merger and acquisition activity with deals stalled, put on hold, or canceled outright. With such uncertainty and unpredictability - post-crisis, the likely immediate focus of business owners will be on managing core business and recovery. It is very challenging to forecast at this stage, for the F&D sector, the outcome and when new dawn may emerge. Duffy (2020) postulated that deals likely to be created from COVID-19 circumstances include "(a) forced mergers and acquisitions transactions as those companies unable to recover from this crisis will seek buyers, (b) corporate entitles with competitive cost structures, strong balance sheets, and case resources may seek cheaper deals including lower valuations for businesses post COVID-19, (c) consolidation of some companies of similar size to withstand market uncertainties and strengthen their financial standing by securing new customers/markets; (d) companies may seek to remedy supply chain problems, such as cash-rich retailers transitioning to procure certainty of supply for certain food products; (e) companies to see out value by adding traceability solutions to improve food security - where they will find themselves facing enhanced requirements to satisfy consumer needs for accurate and transparent information on the food they are producing". COVID-19 will be trying to understand if changes to consumer purchasing practices are permanent and how/where to adjust post-COVID-19.

## 4. Transitioning COVID-19 crisis by exploiting green peatland innovations (Paludiculture) - a useful nexus between agri-food, climate action, and circular economy

Europe has set out its ambitious vision to become the world's first climate-neutral continent by 2050 (A European Green Deal, 2020), while in 2019, the Irish Government published of the All of Government Climate Action Plan - to tackle Climate Breakdown. The food and environmental sectors are tightly bound. The environmental industry is nowadays facing societal changes that force companies to pay more

#### Table 1 European wet-peatland technologies (Paludiculture) funded projects and relevance to green deal innovation and COVID-19 pandemic recovery.

Project	Aims and approaches	Partners	Activities	Impacts
Interreg NW Europe Carbon Connects 2018–2021 €4.5 M funding https://www.nweurope.eu /projects/project-search/ cconnects-carbon-connects/	To reduce the high carbon footprint of peatland soils in Northwest Europe by introducing new bio-based business models developed for sus- tainable land management practices. Estimated stats offered are 50% reduction in CO <sub>2</sub> emissions caused by traditional practices Promote alternative approaches to wet agriculture to protect environment Low lying peatlands – can be done by raising water levels and introducing alternative crops Facilitating transformation in land use to wet agriculture – develop new business models	Research partners Netherlands Belgium, France UK Ireland Local authorities Farmers Consumers Research Institutes	'Paludiculture or wet peatland innovation' Wet crops New business models Cattail [food/construction material/biomass] Sphagnum moss Alder carr Wild rice Blue services [water retention, cooling] Carbon credits. Online toolbox for processes leading to Carbon sink modelling Carbon and blue credit scheme for peatlands Pilot sites for testing of new business models	To reduce emissions by ca. 3200 t per year on pilot sites across ca. 80 ha Protection against environmental issues Partnerships create innovation ecosystem (living labs) with local authorities, Research Institutes, farmers, landowners Educational Programme Share best processes and knowledge
Interreg North Sea Region CANAPE	Enhancing wetland ecosystems To measure greenhouse gases in wetlands.	UK Denmark Cermany	Paludiculture – typha, mosses and grasses Retrain Carbon sink and water store Subaryum mossee	Clear understanding of production methods for each product
€5.5 M	and support the restoration of 95 that peatland, and support the restoration of 3 peatland lakes. The project Creating a New Approach to Peatland Ecosystems responds to these issues	Belgium The Netheralnds	Reed (compost, biocher] Waste wood (cooking charcoal and biocher)	Identify market for products Produce Carbon Pocket Guide
/canape/measuring-greenhouse-gases-in-wetlands/	by restoring wetland areas to reduce their CO <sub>2</sub> emissions and improve their capacity to store water, and by aiming to develop the markets for products produced from wetland ecosystems - a type of farming known as Paludiculture. Physical restoration of the landscape, raising water levels in drained bogs and fens, and restoring lake edges to improve their water quality. This will balt the relaxe of CO <sub>2</sub> that	Pan Euroean (Nordic Baltic Wetland Group)	Greenhouse Gas Emission Site Types (GEST) Methodology. This uses the water classification and the plant types growing on the site to estimate emissions. These are based on a review of scientific literature, and uses previous studies to establish an average for each site which meets certain criteria.	as alternative form of land use
	occurs on drained peatlands, and restore the capacity of the land to act as a buffer to floods and droughts. Improving water quality will improve the recreational value of the		Deliverables Create over 60 ha of new bog habitat Create over 20 ha of new reedland habitat Trial 10 ha of new agricultural production	
	improve the recreational value of the	488	That to ha of new agricultural production,	

400

	waterways and support tourism by reducing incidents of toxic algal blooms.		including reed, sphagnum moss, and purple moorgrass.	
	Demonstrating sustainable use of the land, through piloting agricultural products that can be grown on wet land (known as Paludiculture), and showing that there is a viable economic alternative to draining land for agriculture.			
Interreg NW Europe Care-Peat	Reduce $CO_2$ emission and restore Carbon storage capacity in peatlands to set up and	Scientific Research National Centre		By 2022 – expected ca 7800 t of Carbon emissions to be retained in 5 pilot sites –
€6.24 M	demonstrate innovative technologies for new restoration and carbon measurement techniques and involve local and regional	French Geological Survey Lancashire Wildlife		equivalent greenhouse gas emission of 6072 passenger cars driven per year
2019-2022	stakeholders. Therefore the nature organisations, together	Trust Manchester		
https://www.nweurope.	with local landowners, restore peatlands of five	Metropolitan		
eu/projects/project-search/care-peat-carbon-loss-reduction- from-peatlands-an-integrated-approach/	different pilot sites ranging from 10 to 250 ha and demonstrate the (potential) carbon savings of the restoration. For each pilot site different restoration techniques are used - from manual labour to growing additional peat moss.	University National University of Ireland Galway Eurosite Natuurmonumenten University of Orleans		
Eurpean Commission – LIFE PEAT RESTOTE Project –	To reduce greenhouse gas emissions by 40% until 2030 compared to 1990. Therefore	Poland, Latvia, Estonia, Lithuania,	The LIFE Project Peat Restore aims to rewet degraded peatlands in the partner countries	During the drainage of peatlands, oxygen gets into the peat, which leads to aeration and
https://life-peat-restore.eu/en/	conservation of peatlands must be integrated in the climate and energy policy. Especially the Baltic States as well as Poland and Germany have huge areas of peatlands, which are partly heavily degraded and which need conservation and restoration. In the project the emissions and storage of greenhouse gases, the water level as well as the wildlife (flora and fauna) will be documented, analysed and compared. This helps to prove rewetting measures and gives the chance to regulate aberrations quickly. Also the potential climate effects of the rewetting can be calculated.	Germany	Poland, Latvia, Estonia, Lithuania and Germany, covering an area of 5300 ha to restore the func- tion as carbon sinks. National events, information materials, a photo exhibition and a documentary film will contribute to raise public awareness and to inform about the project's progress.	decomposition of organic matter into smallest parts. In this aerobic zone microorganisms are extremely active, so the rate of decomposition is especially high. As a result the greenhouse gases (CO <sub>2</sub> , N <sub>2</sub> O) are emitted into the atmosphere, what reverses the function of peatlands as a store into a source of greenhouse gases.

сī

considerable attention to develop services, processes, and products that meet the strict legislative requirements as well as a result in a greener production, cleaner environment, and a more sustainable world. Simultaneously, companies deal with increasingly competitive markets in which innovation is regarded as a survival requirement (Galanakis, 2019). Subsequently, there is a pressing need to consider smart new solutions for ensuring the intensive sustainability of agriculture and food production processes that respond to the challenges of pandemics and climate change.

Globally, peatlands account at present for ca. 3% of the earth's surface and play a significant role in offsetting CO2 emissions through sequestration. Peatlands store 1.4 trillion tonnes of carbon-equivalent to 75% of all carbon in the atmosphere, which is twice that stored in forests. Bord Na Mona, the Irish state body that owns 80,000 ha of peatlands, are transitioning from burning peat to transitioning to a green footprint. For example, there is a strong trajectory for Bord Na Mona to develop wet peatland innovation (known as paludiculture) for new business models with academic partners and other industries under a new Empower Eco Sustainability HUB platform that is also supported by the government (Just Transition Programme, 2020). Bord Na Mona is exploiting the 'Empower Eco' platform to accelerate its transition from 'Brown to Green' during COVID-19 that will also stimulate regional development, community transitioning to low carbon economy, training, enterprise, and employment. These transitioning activities will be informed by necessary knowledge transfer from several related European-funded Interreg projects that seek to restore peatlands, including measuring greenhouse gases (GHGs), to improve biodiversity and carbon sequestration, along with developing green paludiculture innovations and blue services such as water retention (Table 1). Wet peatland crops such as cattail, sphagnum mosses, moor-grasses, Alder carr, wild rice, typha, reed, and waste wood for the food and construction industry (Carbon Connects, 2020; Restore, 2020; Care-Peat, 2020; Canape, 2020). Those mentioned above will enable carbon and blue credit scheme for the peatlands. Paludiculture will also facilitate the ambitious of advancing EU collective agriculture policy beyond 2020 (Peatlands in the EU, 2020).

This approach will also facilitate the urgent need for sustainable alternatives and innovative business models for farmers and land managers on rewetted peatlands, which will also enable significant carbon sinking. O'Neill et al. (2020) have recently reported on the development of freshwater aquaculture for high-value perch and trout in the Irish peatlands using cut-away flooded peatlands using organic principles, which is powered by a wind turbine (Fig. 2). Water quality and waste remediation are maintained in this paludiculture-based aquaculture process by using natural microalgae and duckweed. Naughton et al. (2020) also reported on the use of this freshwater aquaculture process for testing new innovations, such as the real-time relationship between the use of hand-held AlgalTorch™ on the farm and relationship with flow-cytometry to measure vital physicochemical parameters. O'Neill et al. (2019) also used this freshwater aquaculture process to demonstrate the efficacy of using microalgae as a smart green innovative tool to evaluate the ecotoxicological quality of effluent that included robustness for used during climate change, such as drought. Advances in bioinformatics and next-generation sequencing will also help with improvements in different microalgae used for this purpose as well as the determination of microbial populations in the system, including the emergence of pathogens or problematic microorganisms (Naughton et al., 2020). This is particularly relevant as less than 5% of aquatic microorganisms are culturable on conventional agar media (Rowan, 2011; Rowan et al., 2015; Fitzhenry et al., 2019). It is envisaged that there will be continued advances and potential for technology disruption in forestry and horticulture for future environmental-proofing such as the delivery of cocktail of helper microorganisms and bioactives through hydrogels from adjacent manufacturing and materials industry to respond variances in climate change and resilience. Peatlands also present an opportunity for improving pollination and ecosystem service management (Naughton et al., 2017; All Ireland Pollinator Plan, 2020). These actions include supporting new medical plant and herbs (such as Calendula, Meadowsweet, Plantain, Valerian, Marsh Mellow, Peppermint), extraction of medicinal sap from birch trees, and production of honey as a social enterprise for rural communities. These timely activities align strongly with the finding of Robroek et al. (2017), who reported that plants that makeup peat bogs adapt exceptionally well to climate change compared to diverse plant communities (such as dunes and grasslands). Peatlands lose fewer plant species as a result of climate and environmental change as they are more resilient. Robroek et al. (2017) also noted that peatlands protect ca. 10% of global CO<sub>2</sub> emissions and act as a sponge in our landscape for holding vast amounts of drinking water, which can be used to offset uncertainty for climate change, such as drought.

The development of peatlands presents opportunities to accelerate leading-edge knowledge and competency through innovation hubs, which would tap into abundant, diverse natural resources for enterprise development and to facilitate transition beyond COVID-19. This will inform the next generation of entrepreneurs to stimulate regional creation and environmental knowledge base with a low carbon orientation. Exploiting wet peatland innovation, known as "paludiculture"), will contribute to the new EU Farm to Fork Strategy, which is at the heart of the European Green Deal, aiming to make food systems fair, healthy, and environmentally-friendly. It will also support Europe's ambitious vision to become the world's first climate-neutral continent by 2050 (A European Green Deal, 2020). In so doing, accelerating paludiculture innovation will advance the economic diversification of territories most affected by the climate transition and the reskilling and active inclusion of their workers and jobseekers. Companies are coming under pressure to engage in sustainability and eco-friendly innovation, but lack a clear progression pathway. This new paludiculture multi-actor ecosystem platform will accelerate entrepreneurship and commercialization that will contribute to supporting the recovery from COVID-19, and possibly other future pandemics. Such a paludiculture innovation platform is being actively pursued in the Irish midlands, known as Empower Eco, which will provide sustainable transitioning-solutions for those unemployed and underemployed as a result of reliance upon burning peat as fossil fuel. It is envisaged that exploiting digitization will also advance the rate of paludiculture innovation along with creating opportunities for promoting the recovery of tourism with the support of Citizen Science. Multi-actor sustainable Hubs will also build enterprise programs by bundling added value through R & D collaboration, facilitating design and technological maturation activities, and commercial planning that should encompass financial, legal, supply chain, and market entry activities. This platform should support startup and SME funding instruments to be leveraged through elevant regional, national and European funding models, such as under Green deal and sustainable agriculture that will potentially inform technology disruption. The core tenets of multi-actor transnational innovation hubs, that connects complimentary technological core facilitities, have been successfully delineated through Interreg Atlantic Area Sharebiotech Project (2012). Such activities will also be supported and accelerated thourugh transnational modelling (Interreg Atlantic Area Neptunus Project, 2018).

## 5. First "Green Deal" funding from European Innovation Council to support the recovery plan for Europe – strong alignment with Paludiculture platform ambitions

On 23rd July 2020, the European Innovation Council (EIC) announced the awarding of over €307 million in funding to 64 gamechanging 'green deal' startups and SMEs that will contribute to the objectives of the European Green Deal Strategy (2020), and the Recovery Plan for Europe (2020). Successful proposals ranged from innovative solutions for the automotive, aerospace, and maritime sectors to advanced materials or Internet of Things technologies (full listing of selected projets are highlighted at European Innovation Council, 2020). Many of these EIC-funded 'Green Deal' projects align with the core tenets of Science Foundation Ireland Disruptive Technology Innovation Funded projects of 2018 and 2019 (Figs. 3 and 4). Examples of EIC 'Green Deal' selected projects that focus on pollination, bioplastics, renewable energy and GHG emission reduction, waste remediation along with specific COVID-19 interventions are noted in Table 2. Over one-third of EIC-funded companies are led by women CEOs, which is a significant increase (tripling) of the number in previous EIC funding rounds.

This paludiculture research platform aligns with The European Innovation Council (EIC), which supports visionary entrepreneurs who create transformative solutions to pressing societal and environmental challenges, supporting the Green Deal and the recovery plan for Europe (European Commision, 2020; European Innovation Council, 2020; Green Deal for Start-Ups and SMEs, 2020). These startups and SMEs are set to scale up, creating jobs and growth, and giving Europe the global leader in green technologies and solutions. Akin to the EIC's Green Deal, the paludiculture-based hub also supports companies to benefit from exclusive business acceleration services to enable rapid growth and scale (Table 3). It is noteworthy that the EIC is currently in a €3.3 billion pilot phase and is due to be fully-fledged in 2021 as part of the new Horizon Europe program (European Innovation Council, 2020). In March 2020, the Commission amended the pilot EIC 2020 Work Programme to include a €300 m budget through the EIC pilot Accelerator for funding game-changing, market-creating innovations that contribute to the goals of The European Green Deal and the United Nation's 2030 Agenda for Sustainable Development. The scheme offers startups and SMEs the option to apply for either grants or blended finance (grant or equity support). Since the start of the EIC Accelerator Pilot in autumn 2019, a total of 140 startups and SMEs active in all technology-intensive sectors (health, digital, energy, etc.) have been pre-selected for equity financing for a total of over €500 m. The amended pilot EIC Work Programme also introduced special provisions to support applications by SMEs and startups with female CEOs.

Climate change and environmental degradation are an existential threat to Europe and the world. To overcome these challenges, Europe needs a new growth strategy that transforms the Union into a modern, resource-efficient, and competitive economy where there are no net emissions of greenhouse gases by 2050 (European Innovation Council, 2020). The European Green Deal is an ambitious roadmap for making the EU's economy sustainable, which will happen by turning climate and environmental challenges into opportunities across all policy areas and making the transition just and inclusive for all. The European Green Deal provides a timely roadmap with actions that will (1) boost the efficient use of resources by moving to a clean, circular economy, (2) to restore biodiversity and cut pollution. It outlines the investments needed and financing tools available and explains how to ensure a just and inclusive transition. Those mentioned above paluciculturefocused transitioning initiative strongly aligns with A European Green Deal Action Plan (2020) including (1) climate ambition, (2) clean, affordable and secure energy, (3) industrial strategy for a clean and circular economy, (4) sustainable and smart mobility; (5) greening the ordinary agricultural policy / 'Farm to Fork Strategy, (6) preserving and protecting biodiversity, (7) towards a zeropollution ambition for a toxic-free environment, (8) mainstreaming sustainability in EU policies, and (10) supporting the EU as a global leader in the green innovation.

#### 6. Solutions to accelerate agri-food recovery for post-COVID-19 potential technology disruption

Several economies across the globe are braced for a short-term economic slowdown (Guan et al., 2020), where they have implemented economic stimulus measures. Reid et al. (2020) suggested that corporate leaders should consider five key steps they can take to reshape their business and plan for recovery: (i) prioritize people safety, (ii) reshape strategy for business continuity through stress-testing to evaluate short-term liquidy and risks, (iii) communicate with relevant stakeholders in a clear, transparent and timely manner, (iv) maximize the use of government support policies and packages, and finally (v) build resilience in preparation of the new reality with bold strategies based on stress tests. Companies will then have to define internal guidelines based on experience learned from the first COVID-19 wave, as well as to develop contingency to respond better to a second wave, and faster to future pandemic crises.

Virtual accelerator hubs for connecting Micro with SMEs that exploit advances in ICT through immersive technologies will become more popular for informing disruptive innovation in situ and for remote end-user applications. This tendency will enable hurdling restrictions that may come with networking and training innovators or employees in meeting rooms that may persist as a barrier to innovate for post-COVID-19 disruptors. It can be actualized through ICT and Quality of Experience (QoE) such as virtual reality (VR) and augmented reality (AR) (Braga Rodrigues et al., 2020). Rowan (2019) reported on the potential benefits of combining immersive with educational technologies for remote workforce training that has significant implications for the future provision of remote workforce training for new skills linked to education. EIC 'Green Deal' startups and SME selected projects also focus on virtual reality and training that includes provision for COVID-19 (Table 2). There is a pressing need to exploit existing or to create new multi-agency enterprise-hubs related to academia that will support and accelerate innovators and businesses (such as in agri-food) as well as full-span commercialization of products and services using the 9-stage technology readiness assessment (Mankins, 2009). This fact should include off-site pilot-data generation where there is an increasing trajectory towards sustainable innovation, the green agenda, and digitization (Table 3). This multi-agency innovation HUB concept is under strategic development in the Irish midlands region to facilitate transitioning to green innovation and for recovery post-COVID-19 era, which will facilitate resource utilization, help recovery of businesses, and accelerate technology disruption post-COVID-19 (Just Transition, 2020).

#### 7. Definitions of disruptive technologies

Disruptive technologies (DTs) or disruptive innovations were initially defined to address market disruption in established markets, where a new product or service (a technology) is introduced (Bower and Christensen, 1995; Christensen et al., 2004). DTs arise from a global drive to discover innovations that will lead to greater competitiveness, impact and value to businesses and society (Christensen and Bower, 1996; Lauer and D'Agostin, 2013; Yongfu et al., 2017; Geels, 2018; Li et al., 2018; Sousa and Rocha, 2018). Innovations may be viewed as disruptive when they take the place of established or broadly accepted ideas arising from scientific inquiry, or in methodologies or in paradigms that disrupts knowledge (Kuhn, 1962).

In recent times, definitions of DT focus on broad factors affecting the industry and address the nexus between learning experience arising from substitutable innovations that relate explicitly to competitive pricing and performance (Rowan, 2019). The recent review by Schuelke-Leech (2018) provides an excellent insight into a diversity of DTs - where products are reduced in size (such as exploiting leading developments in nanotechnology); more lightweight and efficient (such as utilizing additive manufacturing and material science); more competitively and affordably priced (such as exploiting resource management and production including advances in innovative service and business processes); more excellent dexterity and convenience in design and functionality (using researcher creativity blended with artificial intelligence, augmented and virtual reality and so forth that includes future-proofing for needs across various platforms); and more significant performing products and services (such as exploiting physico-chemical developments combined with the use of robotics

and AI for design linked to advances in education and workforce training). For example, this concept may be applied to the food industry for the introduction and training of new technologies across the supply chain from production, distribution, and storage. Developing DTs in the agri-food domain is core to supporting and driving national strategic development plans as these generate jobs, add-value, troubleshoot, and enhance quality in changing marketplaces. These important deliverables are also strongly evident in Green Deal selected projects awarded by European Innovation Council (2020).

In the interesting work of Schuelke-Leech (2018), Beth Ann described a conceptual model to understand the orders of magnitude of DTs that may disrupt markets, businesses, institutions, and the societal norm, which constitute 'the innovation ecosystem.' Specifically, such disruptions occur at two different levels. First order-level disruption is the focus of much the business literature where it considers and addresses disrupters in innovation. Second-order includes technological disruptions that permeate through society, influencing substantial change. Second-order disruptions are more extensive than first-order disruptions. There is great desire to understand the process whereby one identify candidate technology disruptors as that is connected to innovation ecosystem (Nagji and Tuff, 2012) Schuelke-Leech (2018) noted that factors leading to the creation of DTs arise for the localized opportunity (Christensen, 2003), creativity and problem solving (Rowan, 2019), financial investments (such as from self-financing to Venture Capital and Angel investors, Rowan, 2019), networks (Rowan, 2019), broad applicability for an innovative technology (Schuelke-Leech, 2018), and supporting infrastructure and institutions (such as clustering of human capital, networking and so forth to enable a creative process to occur, Drucker, 1985).

## 8. How might DTs accelerate recovery post-COVID-19 and transition towards sustainable food systems?

Herrero et al. (2020) recently reviewed and identified technologies and approaches that have the potential to accelerate the transition towards a sustainable food system. While this was a pre-COVID-19 study, it is evident that many of the innovations identified as future technologies may also be led to disruption and inform recovery post-COVID-19 pandemic in terms of sustainable food production and socio-economic recovery plans. For instance, preference for



Fig. 3. Number of funded projects awarded per topic in 2018 and 2019 by the Irish Government under Disruptive Technology Innovation Fund (DTIF).

functional foods or nutraceuticals that boost immune-system and wellbeing for COVID-19 are likely to become popular (Masterson et al., 2019; Murphy et al., 2020; Galanakis, 2020). Given the gap in knowledge, there has been an increasing number of preprint publications that specifically focused on the potential benefits of nutraceuticals for used to improve the health and mental wellbeing of citizens, including a focus on those recovering post-COVID-19 pandemic. Target compounds under evaluation include vitamins, polysaccharides, natural polyphenols, bioactive lipids, and peptides (Gonzalez, 2020). Companies claim more recognition of immune-boosting ingredients (Daniells, 2020). They have seen an opportunity to develop relevant products (e.g., chocolate balls rich in  $\beta$ -glucan) targeting immunity during the post-lockdown era (Koe, 2020). These trends will most likely continue to drive the market within the next years (Galanakis, 2020). Subsequently, businesses will be seeking to fill the demand for nutraceuticals and functional foods to address the challenges and opportunities created by COVID-19 disease (Galanakis, 2020). These products may emerge from food processing by-products (Galanakis, 2012, 2013), seaweeds, yeast, algae, plants, and fungi or mushrooms that reduce inflammatory responses that are typically associated with cytokine storm in severe COVID-19 patients (Murphy et al., 2020). Masterson et al. (2020) also



Fig. 2. Peatlands-based Freshwater Aquaculture process (adapted from O'Neill et al., 2020).



**Fig. 4.** Amount in Euro of funded projects awarded per topic in 2018 and 2019 by the Irish Government under Disruptive Technology Innovation Fund (DTIF). Proportional representation for each topic in overall DTIF funding is shown as percentage.

reported on the novel use of bioactives from medicinal fungi ameliorates antibiotic-resistant *Klebsiella pneumoniae*-induced pulmonary sepsis.

Future intensive sustainability of the food sector will also be influenced by pressures applied to supply chain, including uncertainties associated with the impact of global warming on crops that will include more flooding and droughts (O'Neill et al., 2019). Fisheries and seafood are viewed as desirable high protein, low carbon-intensive products with the emergence of smart aquaculture processes to meet growing consumer demands (Tahar et al., 2018a, 2018b, O'Neill et al., 2020). However, Ruiz-Salmón et al. (2020) also reported that the seafood and aquaculture sectors across European countries are embracing opportunities to mitigate key environmental pressure points (depletion of resources and climate change), social needs (changing customer attitudes and preferences) or growth in markets (services and business processes along with enhanced competition and worldwide competitiveness). Maintaining satisfactory water quality through developing sustainable innovations to meet growing populations globally will also present significant challenges and opportunities (Tahar et al., 2018c; Tiedeken et al., 2017). De-risking and policy decision making will be increasing influenced by predicitve modelling (Tahar et al., 2017). These pressing challenges are influencing the innovation ecosystem from citizens to policymakers to adopt and foster more sustainable practices. Sharing of new knowledge across European seafood and aquaculture sector, including innovation in ecolabelling and ecodesign, will have far-reaching and crosscutting influences to the circular economy (Ruiz-Salmón et al., 2020). Smart changes in these areas may lead to disruptive products and businesses.

Blockchain offers a security-proof approach to recording every digital transaction that can inform a broad spectrum of smart innovations from business processes to 5G networks. In the food disruption context, it has the potential to radically transform and disrupt safety and quality, waste remediation and recycling, security, and authenticity and traceability (Medical Expo, 2020). The robotics industry is estimated to be worth ca. €2.2billion by 2022 and has to potential to transform the food industry through automation, such as personalized food processing (StartUS-Insight, 2020). There has been a global push to re-address dependency on single-use plastics with a greater focus on smart packaging, including the emergence of potential for bioplastics led by large companies such as

#### Table 2

Examples of EIC Accelerator 'Green Dea'l Selected Projects 2020 including COVID-19.

Startup/SME	Country	Selected Project
B4plastics	Belgium	Complete biodegradable and resistent polymers that allow for circularity and a
		reduction in plastic consumption and microplastics pollution.
UniSieve	Switzerland	Advanced molecular separation solutions for a wide range of applications of
		different scales, ranging from propylene or hydrogen purification to biogas
		upgrading.
Vatorex AG	Switzerland	An eco-friendly solution to kill the Varroa mites and preserve the beekeeping
		industry.
Nanogence SA	Switerland	Innovative techniques for increasing the efficiency of cement manufacturing and
		the robustness of materials while achieving energy savings and reducing CO2
		emissions in the process
Electrohaea GmbH	Germany	Using biological methanation technology for cost-effective large-scale energy
		storage and the production of multi-purpose e-fuel for use in transportation, power generation or industrial heating.
Celllugy	Denmark	Using bacteria and yeast starting from sugar or agro-industrial waste for obtaining a biomaterial able to replace plastics in several packaging
		applications
Brite Hellas SA	Greece	A unique transparent (80%) solar glass panel that generates clean energy. The solar glass combines a nanostructured coating material
		with silicon solar cell technology to de liver a product ideally suited for greenhouse applications.
Cascade	France	Game-changing disruptive solution for photosynthesis and plant development that makes growing food more sustainable by
		significantly reducing water, fertilizer, and pesticide use.
Outsight Sas	France	Key enabling laser technology for autonomous vehicles and drones, opening the
		future of emissions reduction and ecosystem monitoring
Altar	France	A disruptive platform harnessing the power of natural selection for the development of novel microorganisms fulfilling specific industrial
		needs.
NVP Energy	Ireland	Technology to treat municipal sewage wastewater at ambient temperatures and
		convert sewage pollutants into renewable biogas
N2 Applied AS	Norway	SmartNitroFarm is a ground-breaking alternative to chemical fertilisers, which
**		reduces greenhouse gas emissions, reduces odour, stops ammonia loss and air pollution.
GlasPort Bio Ltd	Ireland	An innovative platform technology to eliminate greenhouse gas emissions from
		organic residues and manures that specifically inhibits methane-producing microbes and traps nitrogen.
LightSpace Technologies	Latvia	Next Generation Enhanced Augmented Reality 3D Glasses for medical education,
		pre-procedural planning, intra-procedural visualization, and patient rehabilitation
Hpnow APS	Denmark	Secure, safe, sustainable and affordable on-site generation of Hydrogen Peroxide
Aquila Bioscience Ltd	Ireland	COVID-19 pandemic: An innovative, safe and effective bio-decontamination
		technology for non-toxic removal of biological agents, including coronavirus
Reisistell AG	Switerland	Rapid diagnostic for bacterial SEPSIS and AMR urgently needed for ICU patients in
		COVID-19-like epidemic
Nanoscent Ltd	Isreal	Novel COVID-19 POC Screening Tool Based on Proprietary Nano-Sensors and ML
		Techniques

#### Table 3

Multi-agency ecosystem HUB activities to support and accelerate innovators, businesses post COVID-19.

Concentrated Single-Access Supports for Industry, Entrepreneurs, Disruptors	Linked Acceleration Activities
Step-Change Physical Infrastructure & Systems Supports	R&D Collaborative Facilitation
Pre-start Ups	Design Maturation Activities
Ideation & Design Thinking to inform Technology Readiness Level (TRL)	Technical Maturation Activities
Market Research and Enterprise Support	Financial Planning
<ul> <li>Early Needs Analysis</li> </ul>	Legal Assistance
<ul> <li>Product Market Fit Analysis</li> </ul>	Social and Digital Marketing (including informing customer behavioural change)
Early Technical Validation	Networking
<ul> <li>Test the Technologies</li> </ul>	Dedicated Grant Writing/Reporting
<ul> <li>Experimentation/Validation in Pre-Pilot</li> </ul>	Connection to Academic Staff/Expertise and Equipment to support commercialisation
<ul> <li>Scaled to Real-Life Setting</li> </ul>	
Digitization (including AI, in situ and remote training via AR/VR)	Workforce training - placements
Conduit to State Financial Supports/Agencies	Enable Social Enterprise - Outreach Functions

Diago and Nestlé (Medical Expo, 2020). Food waste management is also the subject of many transnational research and innovation projects such as the European Commission Interreg Neptunus project that combines academic expertise with industry across the Atlantic area to address waste recycling in the fisheries and seafood area, including life cycle assessment, valorization, and ecolabeling (https://neptunus-project.eu/). There is also increasing interest in the development of 3D printers, also known as additive manufacturing, as a sustaining and potentially disruptive technology for a wide range of possibilities for the food industry. For example, 3D food bioprinters permit personalized and repeatable nutrition where it is considered to provide the correct amount of nutrients to match different lifestyles, gender, and health requirements (Brunner et al., 2018). Besides, experimental 3D Bioprinters are designed to prints living cells that have the potential to advance food supply chain needs (StartUS-Insight, 2020). However, the role of social marketing and communication to inform behavior changes and to seek feedback on attitudes, perceptions, and barriers for the uptake of this technology will be necessary (Brunner et al., 2018).

Artificial intelligence (AI) is increasingly used to develop new foods and flavors, such as Coca Cola's research into the Cherry Stripe in 2017 (Medical Expo. 2020). AI will play a prominent role in the personalization of foods and nutrition, exploiting the vast potential of digitalization. Besides, food delivery companies are beginning to concentrate on using the position of artificial intelligence (AI) for problem-solving matched with automation, such as automated guided vehicles. As an example, slow-moving pavement droids to deliver food have been tested by Just Eat, who has partnered with Starship Technologies for this exciting opportunity (StartUS-Insight, 2020). The Internet of Things (IoT) is increasingly becoming relevant for the next-generation food industry, which includes forging innovation in services and business processes. For example, Innet introduced a change suitable for all kitchen devices such as analysis of items for food refrigeration, including taking note of expiration dates with provision for suggesting recipes along with meal preparation. Food security is also an essential factor to have to the fore post-COVID-19, where the monitoring of food from the field to fork using IoT technologies presents a logical solution to this challenge. Given the necessity for food globally, disruption in products and services is likely to emerge from innovations in the delivery and online retail sector as most people remain at home to prevent infection. This infers a focus on food security (such as blockchain and the internet of things in the food supply chain), safety including smart packaging, and alternative, disruptive approaches to food sources such as protein sources. Traceability of infected workers is also very important across the entire farm to fork continuum where there have been clusters of COVID-19 cases reported in meat packaging plants, such as in Ireland (Power, 2020) and Germany (Scally, 2020). The Republic of Ireland has introduced a new COVID-19 Tracker mobile phone app for it's citizens that has ca. 200,000 daily visits.

In terms of potential global economic recovery plans post-COVID-19, and the emergence of Food DTIFs, the significant value may be placed on such things as a review of antibody testing data. For this application, it is hoped that epidemiology will show that many people were infected where this may inform a v-shaped recession with a short sharp recovery. A desirable v-shape economic recovery trend may be more likely due to a wave of online shopping and people working from home. Approximately \$6.2 trillion (12.5% of retail total) is spent on food and beverage in the USA: 2015 was the year that more food was brought in than prepared in the home. COVID-19 has shocked that trend, in the US in Q2, \$100bn dollars shifted from restaurants to retail space. Monopolies in food grocery services may arise, where smaller independent stores may struggle. In the UK, 7% of the population shop for groceries online, with 4% in the US. Yet, one-third of the US population bought online during the 2nd week of March, and half of them was their first time. Confidence must be provided to ensure continuity in the food supply chain to avoid friction in the food system. It is uncertain as to the state with any degree of confidence what would be the specific impact caused by the global downturn (or potential recession) in the economy as it relates to particular needs and opportunities met by emergent technologies in agri-food (including ingress from adjacent industries). Reuters (2020) described that an alternative U-shaped economic recovery might occur, that takes more than a couple of quarters as economies have suffered a faster and deeper, which Reuters feel may be the likeliest outcome. This reflects thinking that lockdown impact may last for a while after their lifting with a gradual easing of the lockdown where social distancing will continue that will continue to influence the tourist industry and so forth negatively. The occurrences of these combined risks may affect the appearance of a COVID-19-induced recession. The flexibility and adaptability of companies to meet change and adjust business models, including provision for ICT, including online delivery for supply chain, will be better placed for sustain and for potentially cause food disruption practices.

In response to COVID-19, countries may consider nationalizing their supply chains for greater control as to avoid reliance on another country that will significantly affect exports. Yet global trade feeds one-third of the world, and producing locally means buying less and the need for more land. COVID-19 may cause a contraction in the extension of the supply chain, and countries will trade with who they can trust. Question of relative advantage, will countries afford to produce things they aren't familiar with or can do, or will this be a necessity arising from potential supply chain shortage issues – for example, reprocessing Personal and Protection Equipment (Rowan and Laffey, 2020)? There is likely to be an increased demand for ICT, including areas such as robotics, blockchain, algorithms to improve processes, efficiency, and sustain or create more jobs. Industries will need to adapt in real-time – which is challenging, given very little market data available to underpin critical decisions.

#### 9. Strategic funding initiatives to identify and accelerate DTs and relevance to COVID-19 – a case study from the Republic of Ireland.

COVID-19 has been hailed as the "big equalizer," but the reality is that we are not equally resilient as a society. It will fuel the next wave of innovators, both for economic and social impact. Given the importance of transitioning beyond COVID-19, there is likely to be increased interest internationally on providing strategic funding initiatives that merge academia and industry to identify the next disruptive technology. There is a shortage of strategic initiatives that specifically focus on DTs. Still, Enterprise-Ireland has been quick to embrace this need with its Disruptive Technology Innovation Fund (DTIF) that launched in 2018. This DTIF initiative has committed €500 m to identify and supporting DFs Project Ireland 2040 (https://dbei.gov.ie/DTIF) that will run over the period 2018 to 2027 aligned with enterprise co-funding. Relevant questions and responses are available to view on this host website. The nature of the 43 DTIF awards will also have relevance to tackling COVID-19 pressure points (Table 4). DTIF Fund is aligned with the Irish Government's Future Jobs Ireland framework with a focus on 'Embracing Innovation and Technological Change,' where there is an emphasis on creating and advancing technology disruption on a commercial footing. It is envisaged that pursuit of these strategic domains, and harnessing the potential of DTIFS emerging from these cross-cutting areas will also support national economic recovery plan for COVID-19 pandemic.

The DTIF funding initiative in Ireland is resourced to  $\in$ 65 m up to 2022 for projects across many thematic domains encompassing emergent preferences for advancing medical devices, ICT, artificial intelligence, blockchain, robotics, nutraceuticals, therapeutics, manufacturing and environmental and so forth. A more in-depth analysis of all funded project awards over the 2018 and 2019 period highlights the potential benefit of addressing COVID-19 often through cross-cutting contributions from these previously considered separate domains of innovation (Table 3).

Analysis of the data provided shows that increased funding for this vital initiative in the priority areas Innovations in Services and Business Processing (1 project,  $\in 3.9$  m (2.7%)); Food (3 projects from DTIF 2, €5.2 m (3.6%)); Energy, Climate Action and Sustainability (6 projects, €8.3 m (5.6%); Manufacturing and Materials – Advanced and Additive Manufacturing (3 projects, €8.7 m (6.0%); ICT (11 projects,€31.1 m (21.6%)), and Health and Wellbeing including Medical Devices, Diagnostics, and Therapeutics (21 projects at €86.8 m (60.3%) (Figs. 3 and 4). These domains reflect the Republic of Ireland's priority strategic areas of research and innovation to 2023. Fig. 3 describes the number of DTIF project awards in the various fields, including food for this Republic of Ireland government initiative since its' launch in 2018. However, it is appreciated that distribution of funding award to date reflects in part the presence of global Medtech and ICT industries in Ireland, in addition to the crucial partnership with leading academic institutions, Science Foundation Ireland-funded Research Centres and Enterprise-Ireland Technology Gateways that all support MNC, SMEs, startups, and entrepreneurs in a closely-knit innovation ecosystem.

The strong performance of medical technologies and ICT (Table 3) reflects in part the significant presence of global multinationals in this space connected to benchmarking universities and Institutes of Technologies across Ireland. Primary production in Ireland is represented by agriculture, fisheries, and forestry (includes food, drink, and horticulture), which pre-COVID-19 accounted for 10% of total exports worth €13bn reaching 180 markets worldwide. Approximately 80% of Ireland's agricultural land is devoted to grasslands, which makes it highly suitable for food production (Bord Bia, 2020). The innovation ecosystem is such that many of the leading food companies such as Diageo, Kerry Group, Glanbia, and so forth also are key collaborating partners in national research and innovation centers, enterprise-technology gateways, and benchmarking academic institutions that are focused sustaining and food disruption. For example, the Enterprise-Ireland funded Technology Gateways in Athlone

#### Table 4

COVID-19 (Coronavirus) - list of successful projects and case studies funded in the Republic of Ireland.

Topic of Research	Funding (€)
Irish Coronavirus Sequencing Consortium	378 716
NSPIRE: Eacilitation Non-Invasive Ventilation (NIV) during the COVID-19 Crisis	205 667
Critical Research Production Addressing Sum() Chain Risk for COVID-19 Diagnostics	540 263
Data Plate Plate Tradecont Response Analyzement (DPFRM)	402 323
Dram nation in the Energy Response Management (DEEM)	192,525
SABS-TILDA: SARS-CoV-2 specific AntiBodieS in The Irish LongituDinal Study on Aging (TILDA): an opportunity to assess COVID-19 rates and phenotypes in older adults in Ireland	199,875
Improving long-term patient recovery and reducing disability after COVID-19 critical illness using microRNA-based approaches	199,621
Defining the disease course and immune profile of COVID-19 in the immunosuppressed patient (DeCOmPRESS study)	199,908
Irish COVID-19 Vasculopathy Study (iCVS)	199,036
Screening for antiviral compounds active against SARS-CoV-2	190,036
COVID19: The North Dublin Cohort Study	190,237
Using a back-calculation model to estimate the scale of asymptomatic Covid-19 prevalence by age and determine the critical threshold of available susceptible persons within the community	199,098
Using telehealth to enhance management of vulnerable groups during the Covid-19 pandemic	96,340
Impact of Covid 19 on Individuals with Intellectual and Developmental Disabilities and Caregivers	199,965
Cupid Covid-19: paediatric emergency department attendance during covid-19	62,857
A Rapid Resource Repository for Health Professionals (RRR-HP): An online and social media individualised support intervention for return to practice, reassigned and new to practice, nursing and allied	107,015
Rapid response and learning for later: establishing high quality information networks and evaluation frameworks for the National Ambulance Service response to COVID-19	109,188
Expediting the diagnosis of COVID-19 in a clinical setting using AI enabled analysis of CT scans	62,858
Creating an evidence-based toolbox for targeted public health interventions during COVID-19: a cross-border analysis to disentangle psychological, behavioural, media and governmental responses	107,015
An investigation of psychological responses to COVID-19 in health care workers during the delay and mitigation phase of disease management: longitudinal and nested qualitative study	83,690
Altered lives in a time of crisis: Preparing for recovery from the impact of the COVID-19 pandemic on the lives of older adults	175.013
Autism specific Transition RESources: a response to the COVID-19 related restrictions (TRes)	123.370
Ontimising Covid-19 social distancing communications: Identifying and addressing nychosocial determinants of social distancing during the Covid-19 pandemic	148 745
Covid-19: Estimating the burden of symptomatic disease in the community and the impact of public health measures on physical mental and social wellbeing	199 945
dentifying mental health needs and best practice for psychological support in frontline healthcare workers during and after the COVID-19 outbreak and in future	187 254
pandemics	107,204
Modelling Real Time Population Wide Impacts of COVID-19	38,310

\*\* For further details see https://www.gov.ie/en/publication/1b4099-covid-19-list-of-successful-projects-and-case-studies/

#### Table 5

DTIF Projects funded by Irish Government between 2018 and 2019 - putative relationship with 2nd Order disruption for Food and COVID-19.

Title of DTIF Project and Value of Award	Priority Area of Award	Potential cross-link to Food (potential 2nd Order Disruptive Technology)	Potential COVID-19 Use
Disruptive Gene Therapy Platform, Replacing Viruses in the	Health & Wellbeing	Not obvious, as yet	Not obvious, as yet
HOLISTICS - Holistic Human Sensing for Health, Aging and Wellness [€7.4 m]	Health & Wellbeing	Training, as Smart Wearables Industry Value - human-centric intelligent sensors and their wireless communications for products.	Yes, remote training and monitor cocooning
AuriGen Solution for Persistent Atrial Fibrillation [€5.9 m] 'Future of Colorectal Cancer Diagnosis and Treatment: Combining Tissue Responsive Probes, AI and Machine Learning for Medical Care [€5.7 m]	Health & Wellbeing Health & Wellbeing	Not obvious, as yet Not obvious, as yet	Not obvious, as yet Yes, use of AI and Machine Learning
Therapeutic enzymes as a treatment for sepsis and other immune disorder diseases [€5 m]	Health & Wellbeing	Cytoflow5, had potential for informing new innovation in Food – such as nutraceuticals	Yes, potentially
Towards safe and effective off the shelf cellular therapy for cancer [€4.3 m]	Health and wellbeing	Not obvious.	Not obvious as yet
Photonics Manufacturing Pilot Line€4.1 m	ICT- Manufacturing	Pilot Line Hub will develop packaging designs tailored to fast cost-effective packaging processes and equipment and develop and next gen packaging equipment (including test) with reduced cycle-times.	Yes, smart packaging relevant for COVID-19
Microfluidic Gene Transfection Cell Analysis and Sorting Platform (€3.4 m]	Health & Wellbeing	Not obvious	Not obvious, as yet
Cooperative Energy Trading System (CENTS)	ICT	Consumers and communities will be empowered with the necessary infrastructure to generate their own electricity for artisan food production with lower carbon footprint	Yes, by stimulus to COVID-19 recovery for Micro and Small Enterprises
Nex ARDENT II [€2.8 m] Medical Imaging Ireland €2 m]	ICT – Internet of Things Health & Wellbeing Health & Wellbeing	Not obvious Not obvious. Assess impact on new nutraceuticals for lung health (Masterson et al. 2019)	Potentially Potentially Yes, lung health
ArtEngine 2.0 bridging automated, AI-Driven 3D World Creation to Market [€2 m]	ICT – AI/AR/VR	Food 3D printing using AI as tool creation of 3D models – cost of 3D content creation is prohibitive for small studios and enables co-development and adoption of AR/VR.	Yes, in Vit D supplementing products and virtual modulities
BioHealx [€1.9]	Health and Wellbeing	Not obvious	Not obvious
Sustainable Bio-Renewable Energy from Wastewater (S-BREW) for the Food & Drink wastewater sector that will reduce land-spread waste and produce high-quality renewable energy [€1.8 m]	Energy, Climate Action and Sustainability	Food waste reprocessing	Yes – economic recovery
E-BAMBI - Enhanced Biocompatibility of Additively Manufactured Biomedical Implants for Improved Clinical Outcomes [€1.9 m]	Health & Wellbeing – Medical Devices	Not obvious – but 3D printing focused	Yes, potentially
High throughput microfluidic drug screening platform [€1.9 m]	Health & Wellbeing – Diagnostics/Therapeutics	Response models for drug testing – may have cross-link to nutraceuticals (GRAS)	Yes, potentially
Future Software Systems Architectures [€1.6 m]	ICT – IoT, AI,	Future use to rapidly operationalise new software systems that are slow – with Al	Possibly, via socio-economic recovery
Irish Lasers for the Internet of the Future (iLife) [€1.6 m]	ICI – Future Networks	Not obvious, as yet	Possibly, in socio – economic recovery
€1.5 m]	ICI – AI, Data Analytics - Blockchain	platform for 2-way communication of safety-critical security information (vulnerability) across food chain	such as PPE distribution
Creating Bionic Man – neural training suit for semimotor impairments[€1.5 m]	Health & Wellbeing	Not obvious	Not obvious
Advanced Environmental Decision Support System for Coastal Areas [€1.1 m]	Energy, Climate Action & Sustainability	Not obvious	Not obvious
Smart-Cardio – a paradigm shift in Cardiac Arrhythmia treatment [€1.1 m]	Health & Wellbeing – Medical Devices	Not obvious	Possibly
DEFINE- AM – Disruptive finishing using electrochemical machining for additive manufacturing €1 m]	Manufacturing & Materials	Future link to food for challenges of post- processing 3D-printed metallic parts	Not obvious
Blockchain in Technology Product Supply Chain [€1 m] Developing inhaled bioengineered exosome therapeutics	ICT - Blockchain Health and Wellbeing	Food technology product supply chain Delivery of smart nutraceuticals via tailored aerosol delivery	Yes, in supply chain Yes, in lung delivery of
$[\in 9.4 \text{ m}]$ Qunatum computing – a software platform for multiple	ICT	Possible role in financial services and logistics supporting	Yes, in stimulus support
Qubit technologies [€7.3 m] Point-of-care iron stores/Ferritin testing for at risk blood donors [€7 m]	Health and Wellbeing	Not obvious	Not obvious
Data-center audio/visual intelligence on-device [€6.9 m]	ICT	Possible role between in lab and field work for audio and vision-data on devices	Yes
Pharam Latch – [€4.4]	Health & Wellbeing	Not obvious	Yes, possibly
Stroke-CIS [€4 m] Blockchain and AI-Enabled Stratified Trial System [€3.9 m]	Health & Wellbeing Innovations in Service.	Not obvious Food security – ensuring complete (GDPR) trustworthv.	Yes, possibly Yes, in supply chain and
FreeSpace Project [€3.6 m]	Business Processes ICT	control and ownership of data Wireless connectivity with ultra-high capacity wireless laser communication technology for broad food industry – delivers combination of bandwidth, availability and distance	security Yes, in communications, distribution networks and economic recovery

Table 5 (continued)

Title of DTIF Project and Value of Award	Priority Area of Award	Potential cross-link to Food (potential 2nd Order Disruptive Technology)	Potential COVID-19 Use
Transfer print technology for heterogeneous integration of components [€3.6 m]	Manufacturing and Materials	Possibly in food packaging	Yes
EyeVU	Health & Wellbeing	Not obvious	Not obvious
Next generation heat pump for affordable decarbonisation	Energy, Climate Action,	Possible role in food distribution and storage as zero	Yes, in efficiency of
of heating [€2.4 m]	Sustainability	carbon-emission, refrigerant-free, heat pump	food supply and security
Haemodialysis Outcomes & Patient empowerment [€2.1 m]	ICT	Possible role of AI enable software and wearable device for chronic diseases	Yes, via AI informed disease mitigation
Connected Enteral feeding healthcare system [€2 m]	Health & Wellbeing	New innovative feed delivery device design, connective and	Yes
		apps	
TRANSPIRE – a trained AI platform for regulation [ $\in 2 \text{ m}$ ]	ICT	Combines human and AI to demystify laws and regulations making it easier to do business while protecting consumers	Yes, in innovations in business services as post disrupter
Video Intelligent Search Platform (VISP) [ $\in$ 1.5 m]	ICT	Not obvious	Yes, in socio-economic recovery
Optimised commercial-scale cultivation of protein-rich biomass from <i>Palmaria palmata</i> for the generation of health enhancing plant based proteinaceous ingredients.	Food	Plant-based proteinaceous ingredients as source of high quality protein and contribute to meeting the growing global demand for plant-based proteinaceous ingredients for animal and human food	Yes, economic recovery and health promoting
Beyond Food Labelling	Food	Using massively multiplexed Next Generation Sequencing to provide a crypto-anchor for food authentication and as a substitute for costly, error prone labelling and certification systems	Yes, economic recovery
HYDRO-fish: Combining targeted nutraceuticals and traceability technology for a smarter and sustainable Irish fish aquaculture industry	Food	The project entails reinforcing the supply chain of Irish salmon production	Yes, economic recovery

Institute of Technology supports ca. 300 projects per year with different startups, SMEs, MNCs nationally with cross-cutting links to the agri-food sector, including smart packaging, 3D printing, and nutraceuticals in terms of intelligent delivery systems. Drivers for informing future technology disruption in the agri-food domain will be influenced by needs arising from COVID-19 pandemic along with balancing environmental concerns for more eco-sustainable, climate-friendly products and services.

If one conducts a more in-depth review of the Irelands 43 DTIF projects, it becomes apparent that potentially 20 (46%) at a combined award value of €86.8 m (60.2%) have cross-cutting abilities to cause 2nd order disruption in the food domain (Fig. 3). This would include disruptive training using wearables via wireless communication, use of cytoflow5 for exploiting benefits of nutraceuticals, 3D printing of food, disruptive feed delivery, and future use of nutraceuticals for lung health using aerosol delivery under Health and Wellbeing domain. Disruption in the area of food security could be potentially achieved through the sole Innovative Services and Business Processes project (Table 3). Besides, disruption may be likely made in food distribution and storage through the Energy, Climate Action, and Sustainability project for reveals the use of a new zero-carbon emission, refrigerator-free, heat pump. Science Foundation Ireland has supported a Rapid COVID-19 Response Initiative that funded 26 of 350 applications to address health service readiness, medical countermeasures along with social and policy countermeasures with potential for disruption (Table 5). There is also good thematic agreement with recent Green Deal selected startup and SMEs for COVID-19 interventions (European Innovation Council, 2020).

## 10. Potential technology, product and business service disruptors in food beyond COVID-19 crisis

In recent times, there has been increasing interest in the augmented use of microorganisms, such as yeast, microalgae, and bacteria, in the form of protein sources. Microorganisms are commonly used in fermented products that we are very familiar with, such as yogurts and sauerkraut. This practice offers a more efficient, innovative approaches to producing the same proteins that we are already familiar with (Medical Expo, 2020). The demand for such alternative food ingredients is pushed by Millennials and Generation X, with changes in eating habits along with corresponding changes in personalized nutrition. From the disruption of introducing Greek yogurt to the emergence of new functional foods such as seaweeds (Mohamed et al., 2012), there has been increasing interest in food ingredients. Such things have informed a trend towards personalized nutrition. Considerable development in this space has been the recent partnership of Nestlé with Corbion. This combines exciting expertise of Corbion's microalgae innovation with Nestlé fermentation abilities that is renowned for its smart plant-based products. Other examples include Impossible Foods, who are making soy heme (typically found in soy plants) for plant-based burgers through microbial fermentation. Impossible food burger is made with soy leghemoglobin that mimics the taste of meat. Such innovations in food ingredients may also complement growing consumer demand for eco-sustainable food sources, which also reflects changing eating habits, diets, and the new role of personalized nutrition. Startup-Insight (2020) noted laboratory-cultured meat might provide an alternative or complementary source to actual meat where the latter requires approximately seven tons of water to produce 450 g of beef. Interestingly, the price for producing around 140 g of artificial meat dropped to €9,59 Euro in 2017 from a non-affordable initial costing base of €274.366. A useful trend to follow for disruption in food production and services is to monitor activities in the United States as more than a third of the world's top food and drink processing companies are headquartered there including Unilever, Danone, Diageo, Kirin, SABMiller, Cadbury Schweppes, Heineken and Asahi Breweries.

High protein feed for animal and human usage will prove relevant, which has been exemplified by an intensive focus on this product for intensive aquaculture production globally. Aquaculture is rapidly developing worldwide and highlights one of the fastest growth areas for the food industry (Fečkaninová et al., 2017; Liu et al., 2017; Tahar et al., 2018a; Tahar et al., 2018b; O'Neill et al., 2019). Aquacultures' pace and scale of expansion reflect a substantial increase in our worldwide population, and the commensurate demand for more safe, nutritious food (Seoane et al., 2014). Fish stocks are depleting on the oceans, and there is a countermeasure push to develop sustainable aquaculture processes to enhance disruption. Hatch-Blue is an example of accelerator SME focused on investing and progressing entrepreneurs to fast-track potential disruptive technologies for the fisheries, seafood, and aquaculture sector

(https://www.hatch.blue/) globally. Precisely, hatch-blue constitutes the first accelerator program for sustainable aquaculture that seeks out, develops, and nurtures Start-ups for disruptive innovation.

#### 11. Conclusion

If one takes a long lens and examines how societies responded to previous catastrophic Black Swan events, (such as the Black Death the devastated the Byzantine Empire, the Cholera outbreak of the 1830s, the Spanish Flu of 1918-1920), World War II (1939-1945), such traumas unleased tremendous changes in thinking, literature, and culture, inspiration, creativity, and ambition will prevail over austerity and fuel the discovery of next-generation disruptive innovations beyond this COVID-19 pandemic. Enhanced innovation leading to the creation of new disruptive technologies in the agri-food domain will inform new exciting new products and services that will address challenges and opportunities for the intensive sustainability of the industry. Modern-day and future disruptive technologies for the agri-food sector will be influenced by the growing demand to produce more safe, nutritious foods to meet growing populations that reflects dynamic changes in eating habits such as personalized nutrition, alternative protein sources, and attitudes towards climate change and digitization. The Black Swan and anti-fragile influences, such as shock to the economy by the occurrence of COVID-19 pandemic, will create both challenges and new opportunities that include emergent or disruptive innovations in service and business processes such as home delivery. A review of the recent 43 projects funded by the Irish government under Science Foundation Ireland's Disruptive Technology Initiatives was used to highlight trends in the innovation ecosystem and the potential for both crosscutting and future ground-breaking disruption in the agri-food, health, ICT, manufacturing and circular economy sectors with a global orientation. This trend is also reflected in the first 64 'Green Deal' selected startups and SME projects recently funded by the European Innovation Projects, which will help support transitioning beyond this COVID-19 pandemic and for future pandemics. Understanding where potential food technology disruptions are likely to occur will be aided by having a holistic perspective and appreciation of the complex socio-technological innovation ecosystem.

#### Authors' contributions

NJR and CMG conceptualized the manuscript. NJR drafted the manuscript. Both authors read, edited, and approved the final manuscript.

#### **Consent for publication**

Not applicable.

#### **Declaration of competing interest**

The authors declare that they have no competing or conflict of interests.

#### Acknowledgments

The authors acknowledge Pixabay for the use of a free black swan image in the graphic abstract. The authors acknowledge funding support from the Irish Health Research Board [HRB-2014-644-POR] and from H2020 MSCA RISE [ICHTHYS].

#### References

- A European Green Deal (2020). https://ec.europa.eu/info/strategy/priorities-2019-2024/ european-green-deal\_en (accessed 26th July, 2020).
- A European Green Deal Action Plan (2020) https://ec.europa.eu/info/sites/info/files/european-green-deal-communication-annex-roadmap\_en.pdf (accessed 26th July, 2020). All Ireland Pollinator Plan (2020). https://pollinators.ie/ (accessed 26th July, 2020).

- Barcèló, D. (2020). An environmental and health perspective for COVID-19 outbreak: meteorology and air quality influence, sewage epidemiology indicator, hospital disinfection, drug therapies and recommendations. Journal of Environmental Chemical Engineering. 8 (4) https://doi.org/10.1016/j.ece.2020.104006.
- Bower, J.L, Christensen, C.M., 1995. Disruptive technologies catching the wave. Harv. Bus. Rev. 73 (1), 506–520.
- Braga Rodrigues, T., Ó Catháin, C., O'Connor, N.E., Murray, N., 2020. A Quality of Experience assessment of haptic and augmented reality feedback modalities in a gait analysis system. PLoS ONE 15 (3), e0230570. https://doi.org/10.1371/journal.pone.0230570.
- Brunner, T.A., Delley, M., Denkel, C., 2018. Consumers' attitudes and change of attitude toward 3D-printed food. Food Qual. Prefer. 68, 389–396.
- CANAPE (2020) https://northsearegion.eu/canape/ (accessed 26<sup>th</sup> July 2020).
- Carbon Connects (2020). https://www.nweurope.eu/projects/project-search/cconnectscarbon-connects/ (accessed 26<sup>th</sup> July, 2020).
- Care-Peat (2020) Restoring the carbon capacity of the peatlands https://www.nweurope. eu/projects/project-search/care-peat-carbon-loss-reduction-from-peatlands-an-integrated-approach/ (accessed 26<sup>th</sup> July, 2020).
- Christensen, C.M., 2003. The Innovator's Solutions:- Creating and Sustaining Successful Growth. Harvard Business School Press, Boston, MA.
- Christensen, C.M., Bower, J.L., 1996. Customer power, strategic investment, and the failure of leading firms. Strateg. Manag. J. 17 (3), 197–218.
- Christensen, C.M., Anthony, S.D., Roth, E.A., 2004. Seeing What's Next: Using the Theories of Innovation to Predict Industry Change. Harvard Business Press, Cambridge, MA.
- DAFM, 2020 https://www.agriculture.gov.ie/customerservice/coronaviruscovid-19/ (accessed 5th June, 2020).
- Daniells, S. NPA's fabricant: 'It's a global pandemic. We need to speak a global language when it comes to the science around vitamin D and zinc.' (2020). https://www. nutraingredients. com/Article/2020/05/06/NPA-s-Fabricant-It-s-a-global-pandemic.-We-need-to-speak-a-global-language-when-it-comes-to-the-science-around-vitamin-D-and-zinc. Accessed 11 May 2020.
- DBEI, 2018. Ireland's research priority areas 2018-2023. https://dbei.gov.ie/en/Publications/Publication-files/Research-Priority-Areas-2018-to-2023.pdf.
- Drucker, P.F., 1985. Innovation and Entrepreneneurship: Practice and Principles. Harper & Row, New York, NY.
- Duffy, R., 2020. Ireland's agri-food and drinks sector M & A outlook during and post-COVID-10. Horizongs – BDOS global viw of mid-market deal activity. https://www. europarl.europa.eu/news/en/headlines/economy/20200513STO79012/covid-19-theeu-plan-for-the-economic-recovery.
- ESRI UK & Ireland. 2020. Business recovery a strategic approach to recovery –https:// www.esriuk.com/en-gb/covid-19/business/recovery (accessed 5th June, 2020).
- European Commision, (2020). First "Green Deal" funding from European Innovation Council awards over €307 million to 64 startups set to support the recovery plan for Europe. https://ec.europa.eu/info/news/first-green-deal-funding-european-innovation-council-awards-over-eu307-million-64-startups-set-support-recovery-planeurope-2020-jul-23\_en (Accessed 26th July, 2020).
- European Green Deal Strategy, 2020. https://ec.europa.eu/info/strategy/priorities-2019-2024/europeangreen-deal\_en (Accessed 31 July, 2020).
- European Innovation Council (2020). List of successful funded startups and SMEs under first European Green Deal Strategy. https://ec.europa.eu/info/funding-tenders/opportunities/docs/cap/h2020/eic-smeinst-2018-2020/1900113-eic\_list\_for\_public\_accelerator\_19\_may\_2020\_en.pdf (Accessed 26th July, 2020).
- Fečkaninová, A., Koščová, J., Mudroňová, D., Popelka, P., Toropilová, J., 2017. The use of probiotic bacteria against *Aeromonas* infections in salmonid aquaculture. Aquaculture 469, 1–8.
- Fitzhenry, K., Rowan, N., del Rio, A.V., Cremillieux, A., Clifford, E., 2019. Inactivation efficiency of Bacillus endospores via modified flow-through PUV treatment with comparison to conventional LPUV treatment. Journal of water process engineering 27, 67–76.
- Galanakis, C.M., 2012. Recovery of high added-value components from food wastes: conventional, emerging technologies and commercialized applications. Trends Food Sci. Technol. 26 (2), 68–87.
- Galanakis, C.M., 2013. Emerging technologies for the production of nutraceuticals from agricultural by-products: a viewpoint of opportunities and challenges. Food Bioprod. Process. 91, 575–579.
- Galanakis, C.M., 2019. Preface. In: Galanakis, C.M. (Ed.), Innovations Strategies in Environmental Science. Elsevier Inc, Waltham.
- Galanakis, C.M., 2020. The food systems in the era of the coronavirus (COVID-19) pandemic crisis. Foods 9, 523.
- Geels, F.W., 2018. Disruption and low-carbon system transformation; progress and new challenges in socio-technical transitions research and the multi-level perspective. Energy Res. Soc. Sci. 86, 2317–2324.
- Gonzalez, S., 2020. Dietary bioactive compounds and human health and disease. Nutrients 12, 348–351.
- Government of Ireland, 2020. https://www.gov.ie/en/news/58bc8b-taoiseach-announces-roadmap-for-reopening-society-and-business-and-u/ (accessed 5th June, 2020).
- Green Deal for Start-Ups and SMEs (2020). https://ec.europa.eu/info/funding-tenders/opportunities/docs/cap/h2020/eic-smeinst-2018-2020/1900113-eic\_list\_for\_public\_accelerator\_19\_may\_2020\_en.pdf (accessed 26th July, 2020).
- Guan, D., Wang, D., Hallegatte, S., Davis, S.J., Huo, S., Li, L., Bai, Y., Lei, T., Xue, Q., Coffman, D., Cheng, D., Chen, P., Liang, X., Xu, B., Lu, X., Wang, S., Hubacek, K., Gong, P., 2020. Global supply-chain effects of COVID-19 control measures. Nat. Hum. Behav. https://doi.org/10.1038/s41562-020-0896-8.
- Herrero, M., Thornton, P.K., D'Croz, D.M., Palmer, J., Benton, G., Bodirsky, B. L., Bogard, J.R., Hall, A., Lee, B., Nyborg, K., Pradhan, P.4, Graham D. Bonnett1, Brett A. Bryan 6, Bruce M. Campbell7,8, Svend Christensen 7, Michael Clark 9, Mathew T. Cook1, Imke J. M.

de Boer10, Chris Downs1, Kanar Dizyee1, Christian Folberth 11, Cecile M. Godde1, James S. Gerber 12, Michael Grundy1, Petr Havlik11, Andrew Jarvis8, Richard King 3, Ana Maria Loboguerrero 8, Mauricio A. Lopes 11, C. Lynne McIntyre1, Rosamond Naylor13, Javier Navarro1, Michael Obersteiner 11, Alejandro Parodi 10, Mark B. Peoples1, Ilje Pikaar 14,15, Alexander Popp4, Johan Rockström4,16, Michael J. Robertson1, Pete Smith 17, Elke Stehfest 18, Steve M. Swain 1, Hugo Valin 11, Mark van Wijk19, Hannah H. E. van Zanten 10, Sonja Vermeulen3,20, Joost Vervoort21 and Paul C. West . 2020. Innovation can accelerate the transition towards a sustainable food system, Nature Food. Vol 1, 266–272.

- IBEC, 2020 https://www.ibec.ie/connect-and-learn/media/2020/04/03/covid19-ibecpublishes-proposed-solutions-to-liquidity-crisis (accessed 5th June, 2020).
- Interreg Atlantic Area Sharebiotech Project (2012). https://keep.eu/projects/440/ (Accessed 31 July, 2020).
- Interreg Atlantic Area Neptunus Project (2018). Water-Energy-Seafood Nexus:Eco-innovation and Circular EconomyStrategies in the Atlantic Area. https://neptunus-project.eu/ (Accessed 31 July, 2020.
- Irish Fiscal Advisory Council (2020) Fiscal assessment report, May 2020. https:// fiscalcouncil.ie (accessed 7th June, 2020).
- Just Transition Fund, 2020. https://www.gov.ie/en/publication/ed10d-just-transitionfund/ accessed 24th June, 2020.
- Koe, T. Chocolate balls for immune health: Singapore firm's new product for kids amid COVID-19. 2020. https://www. nutraingredients-asia. com/Article/2020/04/30/ Chocolate-balls-for-immune-health-Singapore-firm-s-new-product-for-kids-amid-COVID-19 Accessed 11 May 2020.
- Kuhn, T.S., 1962. The Structure of Scientific Revolutions. The University of Chicago Press, Chicago, IL.
- Lauer, M.S., D'Agostin, R.B., 2013. The randomized registry trial the next disruptive technology in clinical research. N. Engl. J. Med. 369 (17), 1579–1581.
- Li, M., Porter, A.L., Suominen, A., 2018. Insights into relationship between disruptive technology/innovation and emerging technology – a bibliometric perspective. J. Food Eng. 220, 20–27.
- Liu, X., Steele, J.C., Meng, X.Z., 2017. Usage, residue, and human health risk of antibiotics in Chinese aquaculture: a review. Environ. Pollut. 223, 161–169.
- Mankins, J.C., 2009. Technology readiness assessments: a retrospective. Acta Astronauticia 65, 1216–1223.
- Masterson, C.H, Murphy, E., Major, I., González, H., O'Toole, D., McCarthy, S., Laffey, J.G., Rowan, NJ. 2019. Purified Beta-Glucan from the Lentinus Edodes mushroom attenuates antibiotic resistant *Klebsiella pneumoniae*-induced pulmonary sepsis. Am. J. Respir. Crit. Care Med. 2019;199:A1222. doi:https://doi.org/10.1164/ajrccm-conference.2019.199.1\_MeetingAbstracts.A1222.
- Masterson, C.H., Murphy, E.J., Gonzalez, H., Major, I., McCarthy, S.D., O'Toole, D., Laffey, J.G., Rowan, N.J., 2020. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumoniae-induced pulmonary sepsis. Letters in Applied Microbiology https://doi.org/10.1111/lam.13358.
- Medical Expo, 2020. Five Major Trends That Will Impact the Food Industry in 2020. http:// emag.medicalexpo.com/five-major-trends-will-impact-the-food-industry-in-2020/ (Accessed 31 July, 2020).
- Michelini, L., Principato, L., Iseavoli, G., 2018. Understanding food sharing models to tackle sustainability challenges. Ecol. Econ. 145, 205–217.
- Mohamed, S., Nadia, S., Hafeedza, H., Rahman, A., 2012. Seaweeds: a sustainable functional food for complementary and alternative therapy. Trends Food Sci. Technol. 23 (3), 83–96.
- Murphy, E.J., Masterson, C., Rezoagli, E., O'Toole, D., Major, I, Stack, G.D., Lynch, M., Laffey, J.G., Rowan, N.J. (2020). β-Glucan extracts from the same edible shiitake mushroom Lentinus edodes produce immunomodulatory and pulmonary cytoprotective effects implications for coronavirus disease (COVID-19). Sci. Total Environ., 732, 25th August 2020, 139330. doi:https://doi.org/10.1016/j.scitotenv.2020.139330.

Nagji, B., Tuff, G., 2012. Managing your innovation portfolio. Harv. Bus. Rev. 90 (5), 66-74.

- Naughton, J., Tiedeken, E.J., Garvey, M., Stout, J., Rowan, N., 2017. Pulsed light inactivation of the bumble bee trypanosome parasite Crithidia bombi. Journal of Apiculture Research 56, 144–154.
- Naughton, S., Kavanagh, S., Lynch, M., Rowan, N.J., 2020. Synchronizing use of sophisticated wetlaboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: A case study from the Republic of Ireland. Aquaculture https://doi.org/10.1016/j.aquaculture.2020.735377.
- O'Neill, E.A., Rowan, N.J., Fogarty, A.M., 2019. Novel use of the alga *Pseudokirchneriella* subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish farming as a case study. Sci. Total Environ. 692, 209–218.
- O'Neill, E.A., Stejskal, V., Clifford, E., Rowan, N.J., 2020. Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. Sci. Total Environ. 706. https://doi.org/ 10.1016/j.scitotenv.2019.136044.
- Peat Restore (2020). Climate change mitigation by means of rewetting the peatlands. https://life-peat-restore.eu/en/ (Accessed 26<sup>th</sup> July, 2020).
- Peatlands in the EU (2020). Common Agriculture Policy (CAP) beyond 2020. https:// www.eurosite.org/wp-content/uploads/CAP-Policy-Brief-Peatlands-in-the-new-European-Union-Version-4.8.pdf (accessed 26th July, 2020).
- Power, J. (2020). Coronavirus sharp rise of cases in meat plants concering. https://www. irishtimes.com/news/ireland/irish-news/coronavirus-sharp-rise-in-meat-plantcases-concerning-health-officials-1.4257698 (accessed 27th July, 2020).
- Recovery Plan for Europe. https://ec.europa.eu/commission/presscorner/detail/en/ip\_20\_ 940 (Accessed 31 July, 2020).

- Reid, G., O'Beirne, N., Gibson, N. (2020). How companies can reshape results and plan for a COVID-19 recovery https://www.ey.com/en\_ie/transactions/companies-can-reshape-results-and-plan-for-covid-19-recovery (accessed 6th June, 2020).
- Reuters, 2020. Alphabet soup: how will post-virus economic recovery shape up? https:// uk.reuters.com/article/uk-health-coronavirus-economy-graphic/alphabet-souphow-will-post-virus-economic-recovery-shape-up-idUKKCN21R25J?il=0
- Richie, S.C., Hacker, J.C., Yearkey, K.L., Shi, Z., Pak, S.H., Rainwater, C.E., 2018. Unravelling of food production microbiome: concepts and future directions. Food and Feed Safety Systems and Analysis. Academic Press.
- Robroek, B.J.M, Jassey, V.E., Payne, R.J., Martí, M., Bragazza, L., Bleeker, A., Buttler, A., Caporn, S.J.M., Dise, N.B., Kattge, J., Zajac, K., Svensson, B.H., Ruijven, J.V., Verhoven (2017). Taxonomic and functional turnover are decoupled in European peat bogs. Nat. Commun. 8, 1161. doi:https://doi.org/10.1038/s41467-017-01350-5.
- Rowan, N., 2011. Defining established and emerging microbial risks in the aquatic environment: current knowledge, implications, and outlooks. International Journal of Microbiology 2011, 462832 15 pages. https://doi.org/10.1155/2011/462832.
- Rowan, N., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo vadis? Trends Food Sci. Technol. 88, 316–332. https://doi. org/10.1016/j.tifs.2019.03.027.
- Rowan, N., Laffey, J.G., 2020. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic – case study from the Republic of Ireland. Sci. Total Environ. 725. https://doi.org/10.1016/j.scitotenv.2020.138532.
- Rowan, N., Valdramidis, V.P., Gómez-López, V.M., 2015. A review of quantitative methods to describe efficacy of pulsed light generated inactivation data that embraces the occurrence of viable but non culturable state microorganisms. Trends Food Sci. Technol. 44 (1), 79–92.
- Ruiz-Salmón, I., Margallo, M., Laso, J., Villanueva-Rey, P., Mariño, D., Quinteiro, P. Dias, A.C., Nunes, M.L., Marques, A., Feijoo, G., Moreira, M.T., Loubet, P., Sonnemann, G., Morse, A., Cooney, R., Clifford, E., Rowan, N. J., Méndez-Paz, D., Parga, X, Anglada, C., Martin, J.C., Rubén-Aldaco, A., 2020. Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Current Opinion in Environmental Science & Health, Volume 13, 101–106.
- Saguy, I.S., Roos, Y.H., Cohen, E., 2018. Food engineering and food science and technology: forward looking journey to future horizons. Innovative Food Sci. Emerg. Technol. 47, 326–334.
- Sarkis, J., Cohen, M.J., Dewick, P., Schröder, P., 2020. A brave new world: lessons from the COVID-19 pandemic for transitioning to sustainable supply and production resources. Resources, Conservations, Recycling. 159. https://doi.org/ 10.1016/j.resources.2020.104894.
- Scally, D. (2020). Germany imposes local lockdown after meat plant outbreak. https:// www.irishtimes.com/news/world/europe/coronavirus-germany-imposes-local-lockdown-after-meat-plant-outbreak-1.4286213 (Accessed 27th July, 2020).
- Schuelke-Leech, B.-A., 2018. A model for understanding the orders of magnitude of disruptive technologies. Technol. Forecast. Soc. Chang. 129, 261–274.
- Seoane, M., Rioboo, C., Herrero, C., Cid, Á., 2014. Toxicity induced by three antibiotics commonly used in aquaculture on the marine microalga *Tetraselmis suecica* (Kylin) Butch. Mar. Environ. Res. 101, 1–7.
- Silva, A.L.P., Prata, J.C., Walker, T.R., Campos, T.R.W., Duarte, A.C., Soares, A.M.V.M., Barceló, D., Rocha-Santos, T., 2020. Rethinking and optimising plastic waste management under COVID-19 pandemic: policy solutions based on redesign and reduction of single-use plastics and personal protective equipment. https://10.1016/j.scitotenv.2020.140565.
- StartUS-Insight, 2020. Disrupting The Food Industry: A Breakdown On Startup Driven Innovation. https://www.startus-insights.com/innovators-guide/disrupting-food-industry-breakdown-startup-driven-innovation/ (Accessed 31 July, 2020).
- Sousa, M.J. & Rocha, A. (2018). Skills for the disruptive digital business. J. Bus. Res. http:// dx.doi.org/https://doi.org/10.1016/j.busres.2017.12.051.
- Statista, 2020a. Food United States. https://www.statista.com/outlook/40060000/109/ food/united-states (Accessed 31 July, 2020).
- Statista, 2020b. Industry report on food manufacturing. https://www.statista.com/study/ 15804/industry-report-food-manufacturing/. Accessed date: 23rd April, 2020.
- Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E., Rowan, N.J., 2017. Development of a semi-quantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants – a Irish case study. Sci. Total Environ. 603, 627–638.
- Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E., Rowan, N., 2018a. Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3, 28.
- Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E., Rowan, N., 2018b. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, 1–22.
- Tahar, A., Tiedeken, E.J., Rowan, N.J., 2018c. Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants a first overview of the situation in the Republic of Ireland. Sci. Total Environ. 616, 187–197.
- Taleb, N.N., 2018. The Black Swan: Second Edition: The Impact of the Highly Improbable Fragility. Random House Publishing Group ISBN 0679604189, 9780679604181.
- Tiedeken, E.J., Tahar, A., McHugh, B., Rowan, N.J., 2017. Monitoring, sources, receptors and control measures for three European Union watch list substances of emerging concern in receiving waters – a 20 year systematic review. Sci. Total Environ. 574, 1140–1163.
- Yongfu, S., Liheng, W., Zontan, S., Kunsheng, W., Liangyan, J.H., Jian, C., et al. (2017). Connotation and selection of disruptive technologies that lead industrial change. Strategic Study of CAR. https://doi.org/10.153302/J-SSCAE-2017.05.002.

Contents lists available at ScienceDirect

### Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv

#### Review

Challenges and future opportunities to unlock the critical supply chain of personal and protective equipment (PPE) encompassing decontamination and reuse under emergency use authorization (EUA) conditions during the COVID-19 pandemic: Through a reflective circularity and sustainability lens



#### Neil J. Rowan\*

Department of Nursing and Healthcare, Technological University of the Shannon Midlands Midwest, Ireland Centre for Disinfection and Sterilization, Technological University of the Shannon Midlands Midwest, Ireland School of Medicine, Nursing and Health Sciences, University of Galway, Ireland CURAM SFI Research Centre for Medical Devices, University of Galway, Ireland

#### HIGHLIGHTS

#### GRAPHICAL ABSTRACT

- COVID-19 pandemic caused disruption in supply chains to vital PPE
- Lack of knowledge to inform key policies and appropriate disease mitigation strategies globally
- Decontamination technologies may enable safe treatment of PPE waste for recycling and long term sustainability
- Co-circulating COVID-19 variants of concern will continue to challenge our healthcare system
- Emerging opportunities to meet harmonized pandemic responses, circular bioeconomy and green innovations

#### ARTICLE INFO

Editor: Damia Barcelo

Keywords: Sustainability PPE reuse Medical device supply chain COVID-19 SARS-CoV-2 Sterilization decontamination Circularity



#### ABSTRACT

Severe acute respiratory syndrome Coronavirus-2 (SARS-CoV-2), and the resulting coronavirus disease (COVID-19), was declared a public health emergency of global concern by the World Health Organization (WHO) in the early months of 2020. There was a marked lack of knowledge to inform national pandemic response plans encompassing appropriate disease mitigation and preparation strategies to constrain and manage COVID-19. For example, the top 16 "most cited" papers published at the start of the pandemic on core knowledge gaps collectively constitute a staggering 29,393 citations. Albeit complex, appropriate decontamination modalities have been reported and developed for safe reuse of personal and protective equipment (PPE) under emergency use authorization (EUA) where critical supply chain shortages occur for healthcare workers (HCWs) caused by the COVID-19 pandemic. Commensurately, these similar methods may provide solutions for the safe decontamination of enormous volumes of PPE waste promoting opportunities in the circular bioeconomy that will also protect our environment, habitats and natural capital. The co-circulation of the highly transmissive mix of COVID-19 variants of concern (VoC) will continue to challenge our embattled healthcare systems globally for many years to come with an emphasis placed on maintaining effective disease mitigation strategies. This viewpoint article addresses the rationale and key developments in this important area since the onset of the COVID-19 pandemic and provides an insight into a variety of potential opportunities to unlock the long-term sustainability of single-use medical devices, including waste management.

\* Department of Nursing and Healthcare, Technological University of the Shannon Midlands Midwest, Ireland. *E-mail address*: nrowan@ait.ie.

http://dx.doi.org/10.1016/j.scitotenv.2023.161455

Received 23 October 2022; Received in revised form 2 January 2023; Accepted 3 January 2023 Available online 6 January 2023

0048-9697/© 2023 The Author. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### Contents

1.	Introduction	2							
2.	Observations from the early phase of COVID-19 pandemic	2							
3.	Developments during the COVID-19 pandemic	б							
4.	Impact of SARS-CoV-2 'variants of concern' of PPE usage – bracing for a winter COVID-19 surge	5							
5.	Surge of PPE use and impact on waste management and our environment	7							
6.	Addressing future circularity and sustainability of single-use PPE	7							
7.	Emerging opportunities and challenges for PPE supply chain	Э							
8.	Future recommendations, research and enterprise opportunities.	C							
Consent for publication									
CRediT authorship contribution statement									
Data	availability	0							
Decl	aration of competing interest	C							
Ackr	nowledgement	0							
Refe	rences	0							

#### 1. Introduction

Severe acute respiratory syndrome Coronavirus-2 (SARS-CoV-2), and the resulting coronavirus disease (COVID-19), was declared a public health emergency of global concern by the World Health Organization (WHO) in the early months of 2020 (Sohrabi et al., 2020). The occurrence of COVID-19 took the world by surprise (Ahmed et al., 2020; WHO, 2020). In addition to significant loss of life and morbidity, COVID-19 disrupted social daily norms that also impacted world trade (White et al., 2021). COVID-19 is also seen as a rare "black swan" event where there is no playbook to advise on best-practice solutions from prior lessons learnt, including the availability of appropriate countermeasures (Rowan and Galanakis, 2020). The aim of this viewpoint article is to critique the main developments affecting the personal and protective equipment (PPE) supply chain during the COVID-19 pandemic along with embracing emerging trends and opportunities. For example, deficiencies in our ability to maintain a supply chain of single-use PPE for frontline healthcare workers (HCWs) was underappreciated; yet, emphasis is shifting away from the potential reuse of critical PPE under conditions of extreme shortages (emergency use authorization, [EUA]) (US FDA, 2020; US FDA, 2020b) towards potentially co-creating safe decontamination and bioeconomy opportunities for recycling enormous volumes of PPE waste to protect our environment (Rubio-Romero et al., 2020; Alt et al., 2022).

A hallmark of the COVID-19 onset, in early months of 2020, was a tremendous lack of knowledge and understanding surrounding SARS-CoV-2, where the emphasis was placed on constraining and managing the spread of infection as there was no available vaccine to safeguard HCWs and citizens alike (Rowan and Laffey, 2020a; Rowan and Laffey, 2020b). Moreover, there was a marked gap in published information that addressed key needs to inform interventions and policies including viral pathogenesis, persistence, transmissiveness and efficacy of available disease prevention, control and waste management methods (Wu et al., 2020). There was a vast chasm in knowledge where the world scrambled to obtain and openly share new information in real-time, which would unlock appropriate solutions (Table 1). For example, at the time of writing, the top 16 "most cited" papers that were published in leading journals on core gaps and potential COVID-19 solutions collectively constitute a staggering 29,393 citations (Google Scholar) (Table 1). Having conducted a systematic literature review using Scopus and PubMed databases, the appearance of the key word "PPE" in journal publications increased by 71.4 % and 73.9 % respectively during the COVID-19 pandemic (Table 2). Over 7000 journal papers were published using key words "COVID-19" and "PPE". The trend showed substantial research and enterprise activities in the areas of "decontamination", "reuse" and "sterilization" generating 526 notable publications since the start of the COVID-19 pandemic (2020 to 2022 reporting period), where similar related studies had not been reported previously (Table 2). Of 516 papers of interest from the combined databases, 89 were included

in this review to highlight key developments and emerging trends. The final set of complete information for this paper, represented in both Tables 1 and 2, was assembled and cross-referenced on the same day (19 October 2022). The purpose of this review is not to exhaustively digest and present copious data on the efficacy or shortcomings influencing appropriate intervention strategies (as there are many narrative and systematic reviews published on this topic cited here); but, it is to reflectively consider and articulate key findings that underpinned cues to action informing policies along with discussing commensurate opportunities that will potentially shape future needs for society, economy and our environment.

#### 2. Observations from the early phase of COVID-19 pandemic

The World Health Organization (WHO) recommended that countries worldwide draw up a "Pandemic Plan" that would achieve clear results in managing pandemics from the early stages (Fusco et al., 2020). Each emergency in healthcare is characterized by different phases: mitigation, preparation, response and recovery (Public Health Emergency, 2012). National overarching plans provided scenarios that benefit from communication and cooperation between different multi-actors and sectors, such as healthcare management, HCWs, logistics, communication, finance and so forth (Public Health Management, 2012). In the absence of a vaccine, emphasis was placed on containing and managing the spread of SARS-CoV-2 outbreaks in the mitigation phase through the deliberate deployment of non-pharmaceutical interventions (NPIs) that addressed various modes of transmission, including contaminated contact surface and infectious aerosols (Charkaborty and Maity, 2020; Rowan and Moral, 2021). NPIs included exercising social distancing, where the WHO recommended wearing single-use N95s or FFP2 standard (or equivalent) facepiece respirators (FFRs) (WHO, 2020). A simplified holistic diagram highlighting the role of NPIs (including PPE) in breaking the chain of SARS-CoV-2 infection is illustrated in Fig. 1. Preventative and disease-controlling factors are typically focused on disrupting linkages between the infectious agent, susceptible host, mode of transmission, reservoir, mode of entry and mode of exit. Face shields and N95s or FFP2s and medical-grade face masks were also used by the general public; however, these were typically replaced by the use of improvised fabric face masks for public transport and in community settings as the months progressed and when COVID-19 vaccines were introduced (de Man et al., 2020). To protect frontline HCWs and to limit transmission, the WHO and US officials recommended wearing 100 to 300 million FFRS per month with the US having only 1 % of this in its supply chain (Lovelace, 2020).

When in sufficient supply, FFRs provide safe work environments for HCWs (Rowan and Moral, 2021). However, most countries experienced critical shortages of FFRs; therefore, it was essential to expedite alternative approaches to meet supply chain deficiencies where the US Food and Drug Administration (FDA) provided guidance for obtaining EUA to market

#### Table 1

Examples of "most cited" influential publications addressing critical new challenges and solutions for meeting PPE supply chain, disease mitigation and waste management during early stages of COVID-19 pandemic in 2020.

Citations	Торіс	Paper title	Author(s)
10,553	Viral persistence	Aerosol and Surface Stability of SARS-CoV-2 as Compared with SARS-CoV-1	Van Doremalen et al. (2020)
5691	COVID-19 review	World Health Organization declares global emergency: A review of the 2019 novel coronavirus (COVID-19)	Sohrabi et al. (2020).
4477	Biocide efficacy	Persistence of coronaviruses on inanimate surfaces and their inactivation with biocidal agents	Kampf et al. (2020)
2284	Genome sequencing	Genome Composition and Divergence of the Novel Coronavirus (2019-nCoV) Originating in China	Wu et al. (2020)
1542	Environment, Prevention	COVID-19 outbreak: Migration, effects on society, global environment and prevention	Charkaborty and Maity (2020)
1276	Wastewater surveillance	First confirmed detection of SARS-CoV-2 in untreated wastewater in Australia: A proof of concept for the wastewater surveillance of COVID-19 in the community	Ahmed et al. (2020)
946	Air pollution & COVID-19	Association between short-term exposure to air pollution + COVID-19 infection: Evidence from China	Zhu et al. (2020)
924	Lockdown, air quality	Effect of lockdown amid COVID-19 pandemic on air quality of the megacity Delhi, India	Mahato et al., 2020
384	Disinfection + Supply Chain + solutions	Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from COVID19 pandemic – Case study from the Republic of Ireland	Rowan and Laffey (2020a)
375	Hospital waste & wastewater	Disinfection technology of hospital wastes and wastewater: Suggestions for disinfection strategy during COVID-19 pandemic in China	Wang et al. (2020)
221	Disinfecting, Sterilizing face masks	Disposable masks: Disinfection and sterilization for reuse, and non-certified manufacturing, in the face of shortages during the COVID-19 pandemic	Rubio-Romero et al. (2020).
206	PPE reuse extended use	PPE and intensive care unit healthcare worker safety in the COVID-19 era (PPE-SAFE): an international survey. Journal of Critical Care.	Tabah et al. (2020)
175	COVID-19 Interventions	An environmental and health perspective for COVID-19outbreak: Meteorology and air quality influence, sewage epidemiology indicator, hospitals disinfection, drug therapies and recommendations.	Barcelo (2022)
148	PPP waste management	Environmentally Sustainable Management of Used Personal Protective Equipment	Singh et al. (2020)
107	Face coverings, reuse, waste management	Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from COVID-19 – Implications for efficacy, re-use and sustainable waste	Rowan and Laffey (2020b)
84	Sterilizing face masks	Sterilization of disposable face masks via standardized dry/steam sterilization processes; an alternative in fight against mask shortages	de Man et al. (2020).

devices (equipment) for the decontamination of N95s during this period of critical use (US Food and Drug Administration, 2020). For example, Tabah et al. (2020) received 2711 responses from a worldwide survey of 1797 (67 %) physicians, 744 (27 %) nurses, and 170 (6 %) allied HCWs on the availability and use of PPE caring for COVID-19 patients in intensive care units (ICU) where at least one piece of standard PPE was unavailable for 1402 (52 %), and 817 (30 %) reported reusing single use PPE, PPE was worn for a median of 4 h. Adverse effects of PPE were associated with longer shift durations and included heat (1265, 51 %), thirst (1174, 47 %), pressure areas (1088, 44 %), headaches (686, 28 %), inability to use the bathroom (661, 27 %), and extreme exhaustion (4992, 20 %). Battista et al. (2021) also reported adverse reactions in HCWs to PPE wearing during COVID-19. Alt et al. (2022) noted that sessional use, re-use, or use of alternative PPE to that of FFRs are the limited options that face HCWs during these periods of critical shortages. The US Centres for Disease Control and Prevention also published recommendations for extended use or limited re-use, (including decontamination), as potential mitigation approaches to offset known shortages of N95s. Mtetwa et al. (2021) reported on the effective decontamination of gowns during COVID-19.

Most countries experienced several "lockdowns" in order to contain and manage COVID-19 infection where the virus evolved into different variants of concern (VoC) through the process of viral mutational change enabled by incubation in large populations of infected people including asymptomatic carriers, over the pandemic period (Rowan et al., 2021). In the early waves of COVID-19, SARS-CoV-2 was reported to exhibit a higher transmission rate (Jiang and Shi, 2020), but a lower mortality rate (2–3 %) (Shi et al., 2020) to that of outbreaks caused previously by other coronavirus (such as 40 % for Middle East respiratory syndrome) (Shi et al., 2020). Due to its enveloped morphology, the SARS-CoV-2 virus has been shown to exhibit relatively low resistance to front-line disinfection, which became its "Achilles heel". Initial emphasis was placed on trying to protect the supply chain of PPE, particularly for HCWs, where there was a focus on keeping the R<sub>0</sub>

#### Table 2

Systematic Review of Scopus and PubMed databases for journal publications containing key words on personal protective equipment (PPE) use, reuse and decontamination under EUA conditions during COVID-19 pandemic over period 2010 to 2022.

Scopus			PubMed			
Key word(s) used	Encompassing COVID-19 pandemic [2010 to 2022]	Prior to COVID-19 pandemic [2010 to 2019]	Key word(s) used	Encompassing COVID-19 pandemic [2010 to 2022]	Prior to COVID-19 pandemic [2010 to 2019]	
PPE	20,831	5943	PPE	46,554	12,164	
PPE + COVID-19	7594	0	PPE + COVID-19	7081	0	
PPE + COVID-19 +	211	0	PPE + COVID-19 +	388	0	
Sterilization			Sterilization			
PPE+ COVID-19 +	102	0	PPE + COVID-19 +	290	0	
Decontamination			Decontamination			
PPE + SARS-COV-2	1500	0	PPE + SARS-COV-2	6125	0	
PPE + Disinfection +	128	0	PPE + Disinfection +	498	0	
SARS-CoV-2			SARS-CoV-2			
Decontamination + N95	232	3	Decontamination + N95	243	17	
PPE + Reuse	150	5	PPE + Reuse	169	8	



Fig. 1. Role of key interventions strategies in breaking the chain of COVID-19 infection.

value below 1 in order to limit the number of SARS-COV-2 infected cases requiring hospital care (particularly for vulnerable patients that may need ventilator assistance in intensive care unit ICU) (Rowan and Moral, 2021).

A hallmark of the initial months of COVID-19 was the introduction of creative means of ensuring the continuity of a vital PPE supply chain including 3-D printing of face shields, bespoke production of Starmed Hoods for point-of-use in ICU (Rowan and Laffey, 2020b), use of blockchain to manage and distribute PPE to essential HCWs (Fusco et al., 2020; Nandi et al., 2021), and reuse and decontamination of single use PPE where severe shortages occurred due to this pandemic situation (Rowan and Moral, 2021). For example, blockchain has been increasingly applied to inform risk in healthcare management, as a strategic tool to strengthen efficient and effective evidence-based decision making. Fusco et al. (2020) reported on the benefits of using blockchain combined with artificial intelligence systems to affect a trace route for COVID-19 safe clinical practice. Smart use of blockchain allows for the creation of a generalizable predictive system that could contribute to the containment and management of pandemic risk nationally (Fusco et al., 2020). At the outset of COVID-19, Rowan and Laffey (2020b) reported on the trial use of bleach (sodium hypochlorite at  $\leq$  4000 ppm), along with a counter water immersion phase to remove residuals, for testing the disinfection performance of Starmed Hoods for ICU. This point-of-use chemical disinfection strategy was not put into practice due to operator concerns over the lingering unpleasant chlorine odor; also, Viscusi et al. (2009) measured the filtration performance of two FFR models submerged into a range of sodium hypochlorite solutions (0.525 %–5.25 % sodium hypochlorite) and noted some degradation in filtration performance, but not below acceptable levels.

In Ireland, as well as many countries internationally, responses and communication to the general public was coordinated by the government through the Department of Health where there was reliance on contributions from subject matter experts across various fields ranging from medical device manufacturing to clinicians/epidemiologists that were connected via a dedicated mobile WhatsApp group. This holistic approach helped to inform a rapid and flexible response to shaping new effective policies, including appropriate disease mitigation strategies. This unified approach was a stand-out abiding "response" feature from this shared COVID-19

pandemic experience as it hurdled conventional silo-based scenarios through holistic engagement where key individuals were joined by a common purpose to solve a societal challenge. This essentially unlocked the collective intellectual and creative wealth of multi-actor stakeholders to inform the government including generating and openly sharing vital information surrounding COVID-19 which informed effective guidelines and policies to constrain and manage the spread of the SARS-CoV-2 virus. Most countries had a limited number of critical-use ventilators and emphasis was also placed on the bespoke production of key equipment and safe reuse. Logistically, this was challenging given that all countries were also similarly affected by the same supply chain shortage issues for PPE; this was supported through the fast-tracking research funding nationally, and internationally from a bottom-up perspective. Addressing the safe reprocessing of vital single-use PPE for reuse applications was a complex and underappreciated task. PPE is manufactured, tested, verified and validated under very strict conditions that include an appropriate terminal sterilization step, which informs authorization by regulators, such as the FDA. As PPE items consist of varying materials where many are heat sensitive, there is a commensurate need to match this need with the use of appropriate and effective decontamination and sterilization modalities. For example, the choice of a decontamination method for the reuse of PPE must also negate any dysfunctionality post-treatment, such as retaining filtration efficacy (for FFRs), comfort fit, material compatibility and safety. Original equipment manufacturers (OEMs) of medical devices (including PPE) had not envisaged, possibly ever, that there would be a requirement to reuse PPE where vital supply was disrupted due to the pandemic. OEMs willingness to share knowledge and to provide expert advice contributed to unlocking this challenge where EUA was granted by regulators for PPE reuse using limited treatment technologies (example, VH2O2 used alone or with Ozone). At the outset of the COVID-19 pandemic, there was a gap in evidence-based publications on the persistence and survival of Coronavirus on contact surfaces (example, Kampf et al., 2020 that has reached 4477 citations) (Table 1). Based on tacit and subject matter expert knowledge of working closely with disinfection and sterilization modalities over many decades, a paper was framed on candidate appropriate technologies that could potentially address safe decontamination of PPE under EUA conditions (Rowan and Laffey, 2020a [386 citations]). The timing of this published paper also coincided with EUA US FDA on alternative strategies for PPE reuse (Rowan and Laffey, 2020b; Alt et al., 2022). Interestingly, the duration to write, review, accept and appear online in the journal in open access format was five days, which reflected the urgency in responding to critical COVID-19 knowledge gaps. At the then time of writing (3 April 2020) this paper, the global number of confirmed COVID-19 cases had reached 1,000,249 with 51,515 deaths which markedly contrasts with current epidemiology data recording 625,359,756 cases and 6,569,533 deaths (19 October 2022). Candidate technologies proposed in April 2020 for safe PPE decontamination, such as the use of vaporized hydrogen peroxide (VH2O2), mild heating and UV irradiation, remain the most popular methods used today. The potential use of VH2O2 for PPE decontamination was informed by the timely systematic review paper of McEvoy and Rowan (2019), which comprised a SWOT analysis on the use of VH2O2 compared with other technologies for terminal sterilization applications, such as physical (gamma-, x-ray, electron-beam) and gaseous (ethylene oxide) methods. This McEvoy and Rowan (2019) paper was written to provide valuable information in order to advance the field of terminal sterilization for established medical device manufacture and safe use, and not for PPE reuse.

It was also appreciated that Coronavirus was a large enveloped virus; therefore, it was likely to be susceptible to the lethal action of frontline disinfectants. Thus, moderate to high-level disinfection processes ( $6 \log_{10}$  reduction in mycobacterial cells, but not bacterial endospores) could be theoretically applied for the safe reprocessing of PPE during COVID-19 without the need for full 12 D reduction in bacterial spores that is allowed for in a sterilization processes. A D value is the time taken to reduce a population of microorganisms by 1  $\log_{10}$  order using a concentration of disinfectant or steriliant (Rowan, 2019). During the early phase of COVID-19, the US FDA and US officials authorized similar decontamination modalities to address pressing PPE supply chain deficiencies under EUA for the reprocessing of PPE (encompassing Umbrella EUA for surgical masks, N95 and other respirator EUA and face shields and other barrier EUAs) during this pandemic that was restricted to a limited number of modalities, such as the use of VH2O2 (US FDA, 2022). At no time was it deemed appropriate to use physical radiation modalities (x-ray, gamma and electronbeam) or ethylene oxide gas for the reprocessing of contaminated PPE during COVID-19. Recent publications in this area corroborated this approach where Haedi et al. (2021) reported the decreased functionality of FFRs after gamma irradiation treatments. US FDA (2020) advised that "once availability of PPE (including facemasks) returns to normal (non-EUA conditions), healthcare facilities should promptly resume conventional practices".

There was a commensurate surge in interest in the use of different types of decontamination methods focused on addressing infectious aerosols and contaminated surfaces including PPE applications with some success. Viana Martins et al. (2022) reviewed 1229 studies from two databases of which only 16 studies reported on methods to "recondition PPE". However, there was frequent variance in the type and level of challenge microorganism(s) used where some researchers used SARS-VoC-2 for PPE decontamination. However, the majority of these in vitro studies did not encompass PPE functionality and material compatibility testing as part of the reported decontamination methods. Various other sophisticated decontamination methods were applied for PPE decontamination with reported successes, including the use of VH2O2 plasma vapor, ionized VHO2 (Cramer et al., 2021), plasma-generated ozone and reactive oxygen species (ROS) (Huang et al., 2022), supercritical CO<sub>2</sub> (Bennet et al., 2021), photoactivated methylene blue (Floreine et al., 2022). For example, plasma-ROS treated PPE (N95s) maintained acceptable mechanical and functionality properties for point-of-use decontamination applications (Huang et al., 2022). Lendvay et al. (2022) reported on the successful decontamination of N95s artificially inoculated with 2 SARS-CoV-2 viruses using novel methylene blue photochemical treatments where FFR integrity was maintained after five processing cycles. The same authors noted that one FFR model failed after five cycles using FDA-authorized VH2O2 plus ozone decontamination method that was included as a comparative control. McAvoy et al. (2021) reported on the successful use of 3D-printed frames to enable reuse, extended-use and improved fit of N95 and KN95 respirators. However, Chen et al. (2019) reported that 3D-printed polymers are less stable than injection molding fabrication when exposure to terminal sterilization by VH2O2 and electron beam technologies.

There was a concerted focus on testing less complicated, more available and affordable disinfection approaches for PPE reuse. For example, Côrtes et al. (2021) evaluated the re-use of 45 surgical masks and 69 respirators by analyzing their performance and safety before and after decontamination using oven, thermal drying, autoclave, and hydrogen peroxide plasma vapor. In addition, 14 used respirators were analyzed after work shifts before and after decontamination using reverse transcription polymerase chain reaction (RT-qPCR) and viral culturing. Oven decontamination (75 °C for 45 min) exhibited the simplest decontamination method that maintained acceptable physical and filtrations performance of treated masks and respirators for at least five processing cycles. Reprocessed respirators used in work shifts (hospital settings) were evaluated positively by users in terms of functionality and comfort, even after three decontamination cycles. Alt et al. (2022) noted that "appropriate decontamination technologies for PPE reuse under EUA FDA conditions must meet the following criteria: (1) include screening and replacement processes for these FFRs not suitable for reuse; (2) effectively inactivate SARS-CoV-2 or representative virus and other relevant bioburden on the FFR; (3) be compatible with the FFRs to avoid rendering it dysfunctional; (4) be available and practical for healthcare settings; and (5) minimize risk to operators of the decontamination equipment and end-users alike".

Clustering of decontamination and sterilization activities and services for PPE reuse under EUA pandemic conditions was centered around the use of regional hospitals, such as in Ireland that was coordinated by the Health Service Executive (HSE) in partnership with Science Foundation Ireland (SFI), academic institutions, industry and with established terminal
sterilization companies. A limited number of published studies had used actual Coronavirus strains as a bioindicator for establishing decontamination efficacy and sterility assurance levels of treated PPE during the COVID-19 pandemic. It is appreciated that the reprocessing and sterilization industries are a bastion of rigor that communicates closely with regulators to ensure compliance with standards in order to safely decontaminate and sterilize products. Thus, several non-thermal candidate laboratories pilot or commercially-available technologies used for food processing (such as gas plasma, high pressure, pulsed electric fields, photonics and so forth (Gómez-López et al., 2021)) that would not be deemed appropriate for PPE reprocessing for point-of-use healthcare applications until due to the nature of their biocidal action, material compatibility and lack of data to support regulator approval. Despite the establishment of PPE reuse methods in healthcare facilities in Ireland, there was a lack of interest in their use by HCWs. Emphasis was always placed on ensuring a continuous supply chain of new PPE items during the COVID-19 pandemic. Absent was the establishment of independent testing laboratories nationally for verification and validation of reprocessed PPE for in-house healthcare applications during COVID-19. However, a new European pandemic response hub, managed by the University of Galway, Ireland, now coordinates and supports research and enterprise in this area including provision for mitigating against future pandemics.

#### 3. Developments during the COVID-19 pandemic

Through the passage of time, new combinational approaches were adopted to inform disease prevention and control including the introduction of effective COVID-19 vaccines (Table 3). Examples of interesting activities and innovations include the modeling of COVID-19 infection data and positive case occurrences (Zeroual et al., 2020; IHME COVID-19 Forecasting team, 2021; Rowan and Moral, 2021), and the use of machine learning and blockchain to review data from a supply chain perspective. For example, Haug et al. (2020) combined statistical modeling and machine learning techniques and estimated that wearing a mask yields a reduction of R<sub>0</sub> between 1.8 % and 12 %, while social distancing contributed to a reduction of approximately 20 %. Genomic, next-generation sequencing and bioinformatic breakthroughs also provided key information on countermeasures. Longitudinal modeling studies yielded new information on the efficacy of COVID-19 mitigation strategies and surveillance of transmission rates in various susceptible populations internationally, where evaluating related data over short timeframes provides limited value (Rowan and Moral, 2021). Efficacy of front-line biocides must also consider the appropriateness for meeting the emerging SARS-CoV-2 variants of concern that differed in transmissiveness and pathogenesis. There was a holistic sharing of key findings by multi-stakeholders for solutions that included the use of the Quintuple Helix Hub approach (combination of academia-industryhealthcare-government-society) to support research, development and innovation linked to education. Interestingly, there was a commensurate push for the digital transformation of adjacent industries including additive manufacturing and food systems (Rowan et al., 2022), which can potentially improve efficiencies in terms of monitoring, surveillance, sustainability, and automation. Currently, there are 706 established or new European Digital Innovation Hubs (EDIHs) where there are opportunities to align some of these appropriate EDIHs for meeting pandemic disease response, innovation, education and training (Table 3) (Rowan et al., 2022). These could also support new SMEs, start-ups, and entrepreneurs, particularly with a focus on transitioning to sustainable solutions, which may also inform new businesses and possibly, future technology disruption (Rowan and Casey, 2021). Opportunities to advance pandemic disease preparedness and responses could also be further supported through bespoke specialist training using immersive and educational technologies.

## 4. Impact of SARS-CoV-2 'variants of concern' of PPE usage – bracing for a winter COVID-19 surge

PPE encompassing face masks (medical grade, FFP2, N95s) remain an important intervention strategy to constrain and manage SARS-CoV-2. Despite the availability of effective vaccines and antiviral drugs such as

Table 3

Examples of new issues, activities and solutions arising for meeting PPE supply chain shortage, disease mitigation and waste management during COVID-19 pandemic.

Citations	Topic	Paper title	Author(s)
752	Symptoms (smell, taste)	Anosmia and Ageusia: Common Findings in COVID-19 Patients	Vaira et al. (2020)
326	Infection modeling	Modeling COVID-19 scenarios for the United States	IHME COVID-19 Forecasting Team
301	Diagnostics	COVID-19 diagnostics and context	Wesssleder et al. (2020)
226	Wastewater surveillance	Wastewater surveillance for population-wide Covid-19: The present and future	Daughton (2020)
212	Blockchain in supply chain bioeconomy	Redesigning Supply Chains using Blockchain-Enabled Circular Economy and COVID-19 Experiences	Nandi et al. (2021)
181	Energy & Environ footprint	The energy and environmental footprints of COVID-19 fighting measures – PPE, disinfection, supply chains	Klemeš et al. (2020)
177	Quadruple interactive HUBs	Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis?	Rowan and Galanakis (2020)
171	Multi- actor approach	COVID-19: A Call for Physical Scientists and Engineers	Huang et al. (2020)
116	Environmental pollution	Occurrence of personal protective equipment (PPE) associated with the COVID-19 pandemic along the coast of Lima, Peru	De la Torre et al. (2021)
112	Open research datasets	COVID-19 Data Hub	Guidotti and Adria (2020)
87	3 D printed PPE	3-D Printed Protective Equipment during COVID-19 Pandemic.	Wesemann et al. (2020)
79	Blockchain in Healthcare	Blockchain in Healthcare: Insights on COVID-19	Fusco et al. (2020)
70	Sustainable PPE	COVID-19 Creating another problem? Sustainable solution for PPE disposal through LCA approach	Kumar et al., 2021
62	PPE pyrolysis waste to energy	Current plastics pollution threats due to COVID-19 and its possible mitigation techniques: a waste-to-energy conversion via Pyrolysis.	Aragaw and Mekonnen (2021)
53	Modeling of Non- pharmaceutical interventions (NPIs)	Disposable face masks and reusable face coverings as non-pharmaceutical interventions (NPIs) to prevent transmission of SARS-CoV-2 variants: Role of new sustainable NPI design innovations and predictive mathematical modeling	Rowan and Moral (2021)
33	Alternative sanitizing for Masks	Photocatalytic Rejuvenation Enabled Self-Sanitizing, Reusable, and Biodegradable Masks against COVID-19	Li et al. (2021)
30	Biodegradable	Biodegradable and multifunctional surgical face masks: A brief review on	Badaahmadi et al. (2021)
	Multifunctional	demands during COVID-19 pandemic, recent developments, and future	
	Surgical Face masks	perspectives	
21	Sustainable PPE – materials and recycling	Key ingredients and recycling strategy of personal protective equipment (PPE): Towards sustainable solution for the COVID-19 like pandemics.	Singh Siwal et al. (2021)
15	Machine learning – case mortality	Identifying mortality factors from Machine Learning using Shapley values – a case of COVID19	Smith and Alvarez, 2021

Paxlovid, COVID-19 cases caused by a mix of new VoC remain a worry given that these collectively represent one in three new SARS-CoV-2 infections in the US (Wang et al., 2022). COVID-19 cases are rising in Europe and the UK, where these VoC have taken hold (Goodman, 2022). The Omicron subvariant BA.5 causes most COVID-19 infections in the US (Smith-Schoenwalder, 2022), accounting for 88 % of new infections. Those with weakened immunity will be particularly vulnerable (Rhee et al., 2022), where recent research intimates that the last laboratory-created antibodies do not provide adequate protection against these VoC (Cao et al., 2022). These VoC descend from slightly different branches of the Omicron lineage, and this "convergent evolution" infers that several VoC can be cocirculating in our population at the same time as we enter winter. With a reduced emphasis on wearing face masks in many countries, there will be a co-challenge to meet potential surges in both COVID-19, influenza and possibly Norovirus (winter vomiting) that will place added pressure on our healthcare system and our HCWs. Interestingly, adherence to NPIs practices generally reduced the incidence of influenza and Norovirus cases over the past two winters. Real-time assessment of the large mix in VoC is becoming challenging as countries reduce surveillance (WHO, 2022). However, <10 % of the US population (14.8 million people) have received an updated COVID-19 bivalent booster vaccine, which is of concern. Thus, the use of PPE, particularly for frontline HCWs remains important; and therefore, the commensurate need to address effective sustainable clinical waste management.

Published work intimates that VoC would be similarly sensitive to disinfection and PPE as preventive measures (Rowan et al., 2021). Meister et al. (2021) reported disinfection effectiveness against SARS-CoV-2 VoC B.1.1.7 and B.1.351 using heat, soap and ethanol where treatment was carried out using a variety of artificially-seeded surfaces including face masks. While society now has effective disease mitigation strategies, we cannot become complacent which includes continued investments in vaccines and antibody therapies. Of concern is societal fatigue towards the use of NPIs (Michie et al., 2020), particularly when one considers that we will be living with COVID-19 for many years ahead. Rhee et al. (2022) surveyed COVID-19 infection control policies at 30 leading US academic hospitals in the context of the initial pandemic surge of the SARS-CoV-2 omicron variant and found that infection control practices vary substantially. The authors recommended clearer public health guidance and transparency around hospital policies that also aligns with meeting VoC responses, in order to facilitate consistent and harmonized standards.

## 5. Surge of PPE use and impact on waste management and our environment

The US healthcare facilities generate an estimated 1 million tons of noninfectious plastic waste every year where there are opportunities to improve recycling (Healthcare Plastics Recycling Council, 2019). Creative means of monitoring and surveillance of COVID-19 pandemic have been applied including reporting the occurrence of this virus in municipal waste that would reflect community-level transmission rates (Daughton, 2020; Barceló, 2020; Ahmed et al., 2020) (Table 1). There is also a pressing need to address appropriate clinical waste management for used PPE (Rubio-Romero et al., 2020; Rowan and Laffey, 2020b). Given the sure in single-use PPE that contains plastic globally (De la Torre et al., 2021) there is commensurate interest in effective and alternative approaches to clinical waste management. For example, Zhao et al. (2022) describe a novel pyrolysis process for used PPE that enables energy recovery through a detailed life cycle assessment approach and includes sustainability. The authors point to the environmental advantages of reducing 35.42 % of total greenhouse gas emissions from the conventional incineration and 43.50 % of total fossil fuel use from landfill processing, the optimal number, sizes, and locations of established facilities within PPE processing system in New York State where one integrated fast pyrolysis facility is used in Rockland County.

There is increasing evidence of PPE accumulation in our environment including the marine that has accelerated interest in identifying workable solutions. Waste PPE was have been reported globally that contaminates our natural capital and habitats (De la Torre et al., 2021). Consequently, there is a surge in interest in the development of alternative biodegradable materials to replace plastics in PPE including considering waste-to-energy conversion (Badaahmadi et al., 2021). It is estimated that some 44 million non-woven PPE items are used by frontline HCWs every day, resulting in some 15,000 tons of waste that are destined for landfills or incineration. Alt et al. (2022) noted that while many materials in PPE are recyclable, SARS-CoV-2 contamination significantly influences such disposal strategies, and this coupled with the natural resource consumption during manufacture may impact our natural environment. Alt et al. (2022) also evaluated and validated technologies suitable for the decontamination and re-use of contaminated N95 FFRs in response to the COVID-19 pandemic. Multiple low-temperature steam (65 to 71 °C) and vaporized hydrogen peroxide technologies were shown to be successful that inactivated feline calcivirus (FCV,  $>3 \log_{10}$ ) and Mycobacterium species ( $\ge 6 \log_{10}$ ) (employed as representatives on the contamination challenge) without affecting the performance of the treated PPE. This work was reported to be suitable for 10 to 20 decontamination cycles of the same PPE. This is potentially a "game-changer" as the deployment of commercial VH2O2 processes could decontaminate large amounts of PPE making it safe for circular bioeconomy applications and long-term sustainability, including reducing PPE waste destined for landfill or incineration (Aragaw and Mekonnen, 2021). Several studies have corroborated the effective use of VH2O2 technologies to decontaminate N95 face masks (Jatta et al., 2021; Deer et al., 2022). Alt et al. (2022) also reported that there are additional challenges to be considered for the reuse of PPE that includes user acceptance, traceability and stock management. Doos et al. (2022) and Kea et al. (2021) reported that the reuse of non-decontaminated PPE worn by HCWs is not an appropriate practice.

Industrial VH2O2 is a strong candidate technology for sustainable PPE waste management as it meets scalability, penetration, and compatibility with materials including innovations in sterilization chamber design and process development (McEvoy and Rowan, 2019). Not all sterilization technologies would be deemed appropriate for large-scale decontamination of PPE waste. For example, the challenges of using ethylene oxide (EO) relate to the hazardous and carcinogenic nature of the gas combined with prolonged treatment times (McEvoy and Rowan, 2019; McEvoy et al., 2021). While radiation is a relatively rapid process leaving no unwanted toxic residues, it is limited by the availability of the radiation source (cobalt), in the case of gamma and not applicable (McEvoy and Rowan, 2019). Emerging non-thermal technologies, such as cold gas plasma (Hayes et al., 2013; Qin et al., 2022), have the potential for novel point-of-use decontamination of clinical waste before transport to landfill or incineration; thus, offering new opportunities for reducing, reusing and recycling. Qin et al. (2022) reported that cold gas plasma (CAP) destroyed six major epidemic strains of SARS-CoV-2 variants of concern within 300 s, where this CAP method affects the SARS-CoV-2 spike protein rather than damaging viral RNA through an oxidative reaction. The presence of soiling on surfaces must always be considered from a cleaning perspective as the presence of organic matter can affect disinfection efficacy on contaminated surfaces including medical devices (Rowan et al., 2021).

#### 6. Addressing future circularity and sustainability of single-use PPE

There has been a marked surge in research interest focusing on the future sustainability of medical devices, particularly, the role of using reprocessing versus single-use plastic items. The sustainability of medical devices (particularly, single-use PPE) has intensified by meeting supply chain shortage issues arising from the COVID-19 pandemic that has placed increased emphasis on reuse (where appropriate), and clinical waste management. Popular circularity and sustainability topics and activities have been captured in Table 4. In addition, use of life cycle assessment (LCA) tools to address all end-to-end stages for medical devices from design thinking to authorized commercial use will help inform resource consumption, carbon footprint, supply chain and transport energy efficiency,

#### Table 4

Examples of new and related research and innovation informing the future circularity and sustainability of medical devices (including PPE) arising a consequence of COVID-19 occurrence.

Topic(s)	Description of research and innovation activities	Reference
Environmental Impact Assessment LCA assessment Circularity assessment	<ul> <li>&gt; 16 different environmental impact categories highlighting superior use of reprocessed devices over single use in 13 categories with focus on electrophysiology catheter. Also informed by LCA.</li> <li>&gt; Healthcare could cut emissions by half for some devices if opting for regulated, represent fiscale user; increased.</li> </ul>	Schulte et al. (2021)
Resource Consumption & Emissions; New job creation	<ul> <li>Avoiding use of virgin materials, remanufacturing alternators can environmental impacts of resource consumption, emissions [reducing abiotic resource use and the Global Warming impact (GWD) – Uses LCA.</li> </ul>	Peters (2016) Zhang et al. (2020) D'Adamo and Rosa (2016)
Use of VH2O2 for PPE waste management	<ul> <li>Development and validation of technologies for decontamination and reuse of con- taminated N95 filtering facepiece respirators with provision for future sustainable waste management</li> </ul>	Alt et al. (2022).
Rapid manufacturing; 3D printing; Biocompatability Becycling challenges	<ul> <li>Consider material composition, toxicological end-points and improvements in com- patibility testing given future sustainability and reuse opportunities for medtech</li> <li>Recycling of complex medical device product needs actionize material flow analy.</li> </ul>	Antonini et al. (2021) Hayes et al. (2013) Conjunth et al. 2020
Recycling chanenges	<ul> <li>recycling of complex interfal device products needs extensive material now analysis is to make it economically and ecologically reasonable</li> <li>The more complex a device, the more process steps, energy, and resources are</li> </ul>	D'Adamo and Rosa (2016)
	<ul> <li>needed for recycling</li> <li>Complex devices may have specific requirements for collection and disposal after use from an infection-prevention perspective and may contain complex materials not suited to municipal waste management &amp; recycling infrastructure (including Green technology)</li> </ul>	Eze et al. (2020) Lee et al. (2017)
CE concept using R-strategies	<ul> <li>Design smarter [Refuse, Rethink, Reduce]</li> <li>Extend lifetime [Reuse, Repair, Refurbish, Remanufacture, Repurpose]</li> <li>Circularly end-point activities: Recycle and Recover</li> </ul>	Potting et al. (2017)
Reuse of Polyvinyl Chloride (PVC)	<ul> <li>Recycle PVC several times (most widely used plastic in medical devices) without loss of critical properties</li> </ul>	I'Ons (2021)
Design thinking	<ul> <li>Design medical devices smarter for circularity that includes appropriateness for reprocessing and sterilization</li> <li>Design from day one with disassembly and cleaning in mind, including reducing number of device components</li> </ul>	I'Ons (2021)
Improve energy efficiency	<ul> <li>Improving energy efficiency and opting for clean energy source could reduce overall costs</li> <li>Additive manufacturing can reduce material waste by as much as 90 % compared to conventional manufacturing and can speed up device prototyping and testing</li> </ul>	I'Ons (2021)
Logistical and design challenge	<ul> <li>Consider device materials in sustainable design</li> <li>For example, disposable surgical drapes contain polypropylene and polyethylene, each can be recycled, but used together they cannot be recycled.</li> </ul>	La Plante (2022)
Reduce carbon footprint	<ul> <li>Recycled materials can be used without loss of technical properties or need for addition of virgin materials</li> <li>For example, recycled polyethylene terephthalate (PET) has 79 % lower GHG emis-</li> </ul>	La Plante (2022)
Multi-actor collaborations	<ul> <li>stons that virgin PET.</li> <li>Work with Medical Device component and OEM manufacturers to advance circular medical device production, and to move away from traditional "take-make-waste" linear systems.</li> </ul>	La Plante (2022) Rowan and Laffey (2020b)
AI, automation and use of novel diagnostic technologies	<ul> <li>Consider automating processes including use of artificial processes to potentially reduce resources and disinfection along with future sustainability production and waste management</li> <li>Role of novel diagnostic techniques in monitoring reprocessing and sterilization of devices</li> </ul>	McEvoy et al. (2021)
Role of Supply Chain and Transport	<ul> <li>What steps are used to reduce energy, water and chemical use</li> <li>Reduce transport of waste from used medical devices by recycling in same region to reduce carbon footprint</li> </ul>	La Plante (2022)
Research translation, Education, Advocacy	<ul> <li>Surveyed 'state-of-the-art' environmental sustainability research in anaesthesia and critical care where avoid, reduce, reuse, recycle and reprocess addressed</li> <li>Moving beyond clinical care, energy (renewables vs fossil fuel) and energy efficiency are important influencers in healthcare's ecological footprint.</li> </ul>	McGain et al. (2020).

profitability, automation for future circularity and sustainability. Lessons can also be learnt from adjacent industries on the combined use of other sustainability tools that has influenced innovation in packaging (Ruiz-Salmón et al., 2020; Almeida et al., 2021;). There is enhanced interest among researchers in the incorporation of smart bioactive materials into medical devices that exhibit virocidal and bacteriocidal properties, which could potentially be used in the design and development of more sustainable innovations (Masterson et al., 2021); however, the appropriateness of these new materials to tolerate established high-level disinfection or sterilization processes would need to be incorporated into new product evaluation and validation including maintaining the functionality of new design innovations during testing or reuse. Addressing such challenges is likely to have potential broader disease preventive benefits including addressing antimicrobial resistant bacterial and fungal pathogens (AMRs) that are at a crisis point globally (Garvey et al., 2022). Masterson et al. (2021) reported, for the first time, the development of a low-temperature extrusion process for the production of GRAS bioactive-polymer loaded compounds for targeting such AMR bacteria that may also be suitable for decontamination processes. There is significant scope to develop appropriate sustainability tools for evaluating the potential impact of applied decontamination technologies on the environment (Hayes et al., 2013) including elucidating cellular mechanisms of inactivation (Farrell et al., 2013). The use of life cycle assessment (LCA) confirmed that more sustainable strategies for the disposal of PPE waste are needed (Kumar et al., 2021). However, published studies have also highlighted that data generated by medical-device-focused researchers would benefit from having an understanding of decontamination and sterilization technologies in their method (Rowan and Moral, 2021). This knowledge gap can be addressed by delivering specialist training on medical device reprocessing and terminal sterilization to academia and other stakeholders including SMEs, start-ups and entrepreneurs; thus, co-creating solutions in the adjacent circular bioeconomy or digital technologies domains (Table 3). The latter can be enabled and accelerated through sustainable business models that will de-risk investments in innovations (Rowan and Galanakis, 2020).

#### 7. Emerging opportunities and challenges for PPE supply chain

Table 5 describes 24 observations from tackling critical PPE supply chain challenges and other adjacent COVID-19 activities and will inform the ongoing and future pandemic response plans with opportunities for promoting innovation and sustainability that will potentially benefit society, the economy and the environment. There is an emergence of opportunities to consider such as automation using robotics in medicine and healthcare including transforming reprocessing of complex medical devices (such as endoscopes) in healthcare (Allescher et al., 2022). This includes keeping

#### Table 5

New observations and lessons learnt from tackling critical PPE supply chain challenges during COVID-19 pandemic.

No.	Topic	Reference
1.	Efficacy in forging <b>multi-actor contributions to techno-, socio-economic</b> and environmental challenges for <b>real time solutions</b> in innovation and policies underpinning pandemic plan led by Government with subject-matter experts	Rowan and Laffey (2020a)
2.	Global terminal sterilization technologies have potential to scale and treat large volumes of PPE waste during pandemic for future recyclable options	Alt et al., 2022
3.	Importance of open <b>real-time communication</b> (selecting appropriate channels, messaging, timing) and <b>Open-Access dissemination</b> for sharing and reflecting upon break-through solutions (ex. COVID-19 tracker phone apps)	Zhang et al, 2022
4.	Greater need for education, training and awareness of the role of PPE supply chain, infection microbiology, disinfection/sterilization	
-	modalities, sustainability and circular bioeconomy in universities with industry	Rowan and Moral, (2021)
5. 6	Co-creation of regional clusters and convergence of multi-actor innovation nubs nationally for pandemic response plans Interact in sustainable management of DPE waste	University of Gaiway (2022)
0.	nicrest in sustainable management of 11 E waste	Singh et al. (2020)
7.	Reliance on front-line HCWs to deliver on critical disease prevention and control service that can be prolonged and highly stressful where	Tabah et al. (2020)
	there is limited understanding on mental health and wellbeing including long-term wearing of medical face masks and possible use of	White et al. (2021)
0	reprocessed PPE	Kea et al. (2021)
о.	repeat waves of COVID-19 interctions introduces societal ratigue in terms of anterence to mandatory lockdowns and restricted social intovenient; imposing future lockdowns will challenge society further with present-day economic pressure caused by increased cost of living,	Rowan and Moral (2021)
9	energy/commodity crisis, initiation and potential energence of a global economic recession. Reliance on single-use-plastic based PPE in healthcare globally can have a dramatic and underestimated impact on environmental pollution	Kumar et al. (2021)
	during a pandemic with urgent need to identify alternative biodegradable polymers in PPE design and appropriate clinical waste management	Rowan and Galanakis (2020)
10.	SARS-CoV-2 virus mutated and changed to variants of concern that is likely to persist for many years akin to influenza, where there is requirement	Rowan et al. (2021)
	for regular monitoring to confirm efficacy of vaccines and boosters. Also, despite genetic adaptive changes to virion genome, VoC are similarly	Rhee et al. (2022)
	sensitive to iron-ine non-pharmaceutical interventions (masks) and to non-specific chemical disintectants and physical treatments. Serendinitously, society was fortunate that this is a complex enveloped virus rendering it less likely to presist for long time on surfaces and readily.	Meister et al. (2021)
	disinfected	
11.	Creativity and ingenuity of society in improvising and problem solving ranging from improvised wearing of fabric-based face mask to	Rowan and Laffey (2020b).
	determining indicative levels of virus in communities by monitoring of municipal wastewater, to tracking, modeling and reporting on COVID-19 globally in real-time dashboard managed by Johns Hopkins University	Rowan and Moral (2021)
12.	Unlike deadly Spanish flue of 1918 where rate of international transmission was comparatively slow due to absence of commercial air travel; COVID-19 <b>rapidly spread globally</b> , but society had means of disease counters measures.	Sohrabi et al. (2020).
13.	OEMs of medical devices (PPE) and contract sterilization pivoted to supporting and providing solutions for what are typically single use devices	Alt et al., 2022
14.	Regulators and healthcare authorities acted rapidly to develop appropriate pandemic plan including EUA for PPE reuse such as point-of-use in bushlesses.	US FDA 2020
15	nearmcare Despite suite of physical and gaseous sterilization modalities and greater appreciation of efficacy of emerging non-thermal technologies (such as	WHO, 2020 Gómez-López et al. (2021)
10.	use in food industry, reprocessing of PPE for point-of-use in healthcare under EUA is likely is restricted to use of a limited number of validated	Floreine et al. (2022)
	modalities until such time that other modalities are appropriately developed, tested, commercially scaled and meet regulatory need with	Lendvay et al. (2022)
1.0	matching ISO standards	1 (0001)
16.	Fast tracking funding in research and enterprise to develop solutions, which reduces red-tape and barriers to innovation and open knowledge barring has been seen as impactful and henceficial. Stude as 2 D printing of medical durings, and hencele production of real-comment Starmagh bands	McAvoy et al. (2021)
	sharing has been seen as impactual and beneficial, such as 5 D printing of medical devices, and bespoke production of replacement starmed noods for ICU.	Rowan and Laffey (2020b)
17.	Many developed small countries including Ireland could afford maintaining single use PPE supply chains through procurement, and benefited	United Nations (2021)
	from early freely available vaccines – however, there is a greater need in society to share and expedite knowledge, innovation and solutions	
10	with <b>developing countries</b> that are equally addressing co-morbidities in terms of HIV and TB	Power and Marel (2021)
18.	immersive technologies, blockchain) along within in situ 3D printable sterilized devices show promise as <b>disruptive innovation</b> in pandemic	Huang et al. (2020)
	plan	
19.	Due to adopting effective face mask wearing and use of other non-pharmaceutical interventions (NPIs), there was a marked reduction or	Speare-Cole (2021)
	absence in the occurrence of influenza and winter vomiting Norovirus that challenge healthcare during winter seasons – their reemergence	Agha and Avner (2021)
	(including KSV) will place added pressure on embattied nearthcare system that are a laready playing catching from a resource and $H_{cW}$	
20.	Wearing of face masks in society (out with typically healthcare) became the social norm and will be used in healthcare for the foreseeable	Rowan and Moral (2021)
	future	
21.	Only in the passage of time, such as through <b>reflection on longitudinal mixed-methods studies</b> including modeling will true effectiveness of	Rowan and Moral, 2021
21	Interventions be understood for <b>narmonizing global pandemic response plan</b> <b>Strong community engagement</b> including bespoke 3 D printing of face shields in local schools	Wesemann et al. (2020)
22.	Emergence of opportunities for digital technologies and use of machine learning during COVID-19 pandemic	Allescher et al. (2022)
		Taylor (2022)
23.	Surveillence of COVID-19 by use of sewage as epidemiological indicator	Barceló (2020)
24.	VoCs adopt and co-circulate in society, but use of disinfectants are effective.	Meister et al. (2021)
		MICC CL dl. (2022)

Abbreviations: Variants of Concern (VoC); Non-pharmaceutical Interventions (NPIs); vaporized hydrogen peroxide (VH202); Emergency Use Authorization (EUA); Healthcare workers (HCWs).

pace with sustaining and disruptive practices in surgery for healthcare applications. Leading sterilization companies in partnership with universities are also developing and deploying state-of-the-art biotechnology tools such as flow-cytometry to unlock real-time microbial inactivation where such approaches may supplement conventional plate counts (McEvoy et al., 2021). Immersive (digital) technologies are also partnering with medtech companies for complex virtual training on specific technical operations. For example, Mersus Technology (Immersive) has partnered with Boston Scientific to test and apply an "Avatar Academy Program that uses computer gaming to recreate virtual laboratories and cleanrooms, allowing medtech employees to familiarize themselves remotely with a complex work environment and processes. This approach will potentially automate training where one could theoretically run six bespoke training sessions in one day that previously would have taken a month, and can be extrapolated to address the full production chain delivered in a virtual environment" (Westmeath Independent, 2020). There is a pressing interest in also ensuring the safe reprocessing of reusable medical devices in healthcare applications that are challenged by the presence of recalcitrant biofilms harboring pathogenic microorganisms where failure to effectively clean, reprocess and sterilize may lead to significant patient risk.

HCWs have been in the constant face of COVID-19 for over two years where anxiety and stress can lead to mental health issues (Peteet, 2020); thus, highlighting the importance of maintaining effective disease prevention and control measures (Vaira et al., 2020). Peteet (2020) reported growing concerns about anxiety with COVID-19 that have led to recommendations for effective staff care, and greater availability of mental health treatment. This researcher has also noted existential concerns raised by the pandemic suggesting the importance of religious resources, as seen in research into patients dealing with advanced cancer. Many Asian countries introduced tougher COVID-19 restrictions compared to other countries which caused elevated levels of stress, anxiety and isolation. A-Singapore-based "Intellect" company recently raised \$20 m to develop and manage a mobile phone app that regularly checks on users' moods with connectivity to exercise and recovery sessions and to therapists for real-time interventions (Cheung and Ripley, 2022).

#### 8. Future recommendations, research and enterprise opportunities

- Identify and invest in appropriate scalable decontamination technologies that can effectively treat large volumes of used PPE during the COVID-19 pandemic with a view to recovery, reusing and recycling that will drive a long-term sustainable waste management system.
- Harmonize pandemic response plans (disease mitigation, preparation, response and recovery phases) including infrastructure, living labs etc. that facilitate opportunities to test, verify and validate point-of-use decontamination technologies for medical devices (including 3D printing) in healthcare, which also makes provision for ongoing and future pandemics where PPE (FFRs) may require reuse under EUA conditions.
- Promoter greater engagement of OEMS, sterilization companies with academic institutions, new businesses, policymakers and civil society, such as through Quintuple Helix Hub framework to inform new green solutions, such as the expansion of Rowan and Casey (2021). An example is the efficacy of new medical devices comprising biodegradable materials from a life cycle assessment and 360 degree holistic thinking perspective that can be decontaminated without loss of functionality or biocompatibility.
- Develop and apply risk models and pathways for new businesses to help investment in entrepreneurial activities that include increased funding support for new sustainable green innovations.
- Develop further, more rapid methods in microbiology to meet the need for real-time monitoring, sustainability and diversification including "Industry 5.0" human-centric models (Rowan et al., 2022).
- Greater interface between broad users (including academia, SMEs, MNCs) and policymakers/regulators to understand policies and standards for developing innovations and processes. MNCs in medtech and connected sterilization already very effectively communicate regulators at important policy interface.

- Increased education and training for greater engagement with multi actors including social sciences, humanities and communities for feedback on pandemic response and innovations.
- Promotion of open access publications and pandemic data hubs (such as Guidotti and Adria, 2020).
- Development of regional spatial and economic strategies to support the co-creation and development of innovations for the medtech sector in partnership with academia that also addresses circular bioeconomy opportunities, and digital transformation.
- Increased multi-disciplinary role of blending Science, Technology, Engineering and Mathematics (STEM) with Arts, Humanities and Social Sciences (AHSS), and industry for solutions.
- Greater research and provision of appropriate tailored support and intervention services for HCWs suffering from stress, anxiety and mental health as a result of meeting this pandemic response for society.
- Reflect to ensure effective communication and dissemination channels, messaging and timing to inform all stakeholders in society.
- Develop appropriate sustainability tools to assist SMEs, start-ups and entrepreneurs in the creation of a new medical device that addresses the life cycle from discovery to validation and addresses technology, society and policy readiness levels. This will de-risk investment and create local employment and regional development.
- Nurture and train a talent pool of high-caliber researchers for these and future applications.
- Resource regional pandemic response hubs that address a critical supply chain for PPE linked to education and outreach, which also includes digital transformation and sustainability along with disease prevention and control.
- Effective clinical waste management balanced with environmental protection and natural capital.
- Develop and introduce immersive technologies (digital) to address important practices in disease prevention and control from a training and education perspective.
- Digital transformation of PPE supply chain from EUA perspective such as using blockchain and AI.

#### Consent for publication

Not applicable.

#### CRediT authorship contribution statement

The author solely designed and wrote this paper.

#### Data availability

Data will be made available on request.

#### Declaration of competing interest

The author declares no conflict of interest.

#### Acknowledgement

The author acknowledges the assistance of Gillian Kean (Momentum Consulting, Ireland) in the preparation of graphics used in this review article. The author acknowledges funding support under Phase 2 Development of CÚRAM SFI-Research Centre for Medical Devices (University of Galway) in Ireland.

#### References

Agha, R., Avner, J.D., 2021. Delayed seasonal RSV surge observed during the COVID-19 pandemic. Pediatrics 148 (3), e2021052089. https://doi.org/10.1542/peds.2021-052089.

- Ahmed, W., et al., 2020. First confirmed detection of SARS-CoV-2 in untreated wastewater in Australia: a proof of concept for the wastewater surveillance of COVID-19 in the community. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020.138764.
- Allescher, H.D., Voigt, F., Mangold, M., Haddadin, S., 2022. Integration of robotic in the reprocessing and transfer of endoscopes. Endosc. Int. Open 10 (8), E1022–E1028. https://doi.org/10.1055/a-1789-0532 PMID: 35979027; PMCID: PMC9377825.
- Almeida, C., et al., 2021. Packaging environmental impact on seafood supply chains: a review of life cycle assessment studies. J. Ind. Ecol. https://doi.org/10.1111/jiec.13189.
- Alt, J., Eveland, R., Fiorello, A., Hass, B., Mezzaros, J.J., McEvoy, B., Ridenour, C., Shaffer, D., Yirava, W., Ward, L., 2022. Development and validation of technologies suitable for the decontamination and re-use of contaminated N95 filtering facepiece respirators in response to the COVID-19 pandemic. J. Hosp. Infect. 119, 141–148. https://doi.org/10. 1016/j.jhin.2021.10.004.
- Antonini, M.J., Plana, D., Srinivasan, S., Atta, L., Achanta, A., Yang, H., Cramer, A.K., Freake, J., Sinha, M.S., Yu, S.H., NR, LeBoeuf, Linville-Engler, B., Sorger, P.K., 2021. A Crisis-responsive framework for medical device development applied to the COVID-19 pandemic. Front. Digit. Health 3, 617106. https://doi.org/10.3389/fdgth.2021.617106 2021 Mar.
- Aragaw, T.A., Mekonnen, B.A., 2021. Current plastics pollution threats due to COVID-19 and its possible mitigation techniques: a waste-to-energy conversion via pyrolysis. Environ. Syst.Res. https://doi.org/10.1186/s40068-020-00217-x.
- Badaahmadi, V., et al., 2021. Biodegradable and multifunctional surgical face masks: a brief review on demands during COVID-19 pandemic, recent developments, and future perspectives. Sci. Total Environ. 798. https://doi.org/10.1016/j.scitotenv.2021.149233.
- Barcelo, D., 2020. An environmental and health perspective for COVID-19 outbreak: Meteorology and air quality influence, sewage epidemiology indicator, hospitals disinfection, drug therapies and recommendations. J. Environ. Chem. Eng. 8 (4), 104006. https:// doi.org/10.1016/j.jece.2020.104006.
- Battista, R.A., et al., 2021. Personal protective equipment (PPE) in COVID 19 pandemic -related symptoms and adverse reactions in healthcare workers and general population. J. Occup. Environ. Med. 63 (2), e80–e85. https://doi.org/10.1097/JOM.00000000002100.
- Bennet, D., et al., 2021. Evaluation of supercritical CO2 sterilization efficacy for sanitizing personal protective equipment from the coronavirus SARS-CoV-2. Sci. Total Environ. 780. https://doi.org/10.1016/j.scitotenv.2021.146519.
- Cao, Y., et al., 2022. Imprinted SARS-CoV-2 Humoral Immunity Induces Convergent Omicron RBD Evolution.
- Charkaborty, I., Maity, P., 2020. COVID-19 outbreak: migration, effects on society, global environment and prevention. Sci. Total Environ. https://doi.org/10.1016/j.scitotenv.2020. 138882.
- Chen, Y., et al., 2019. 3D printed polymers are less stable than injection moulded counterparts when exposed to terminal sterilization processes using novel vaporized hydrogen peroxide and electron beam processes. Polymer 183. https://doi.org/10.1016/j.polymer.2019. 121870.
- Cheung, E., Ripley, W., 2022. How Covid prompted Asian startups to use tech in revolutionizing mental health support. https://edition.cnn.com/2022/10/19/tech/covidsingapore-startup-mental-health-revolution/index.html (Accessed 21st October, 2022).
- Côrtes, M.F., et al., 2021. Decontamination and re-use of surgical masks and respirators during the COVID-19 pandemic. Int. J. Infect. Dis. 104, 320–328.
- Cramer, A.K., Plana, D., Yang, H., et al., 2021. Analysis of SteraMist ionized hydrogen peroxide technology in the sterilization of N95 respirators and other PPE. Sci. Rep. 11, 2051. https://doi.org/10.1038/s41598-021-81365-7.
- D'Adamo, I., Rosa, P., 2016. Remanufacturing in industry; advices from the field. Int. J. Adv. Manuf. Technol. 86, 2573–2584.
- Daughton, C.G., 2020. Wastewater surveillance for population-wide Covid-19: the present and future. Sci. Total Environ. 736. https://doi.org/10.1016/j.scitotenv.2020.139631.
- De la Torre, G.E., et al., 2021. Occurrence of personal protective equipment (PPE) associated with the COVID-19 pandemic along the coast of Lima, Peru. Sci. Total Environ. 774. https://doi.org/10.1016/j.scitotenv.2021.145774.
- de Man, P., van Straten, B., van den Dobbelsteen, V.van der Eijk, 2020. Sterilization of disposable face masks by means of standardized dry and steam sterilization processes; an alternative in the fight against mask shortages due to COVID-19. J. Hosp. Infect. 105, 356–357.
- Deer, T.H., et al., 2022. Aerosolized hydrogen peroxide decontamination of N95 respirators, with fit-testing and viral inactivation, demonstrates feasibility for reuse during the COVID-19 pandemic. mSphere https://doi.org/10.1128/msphere.00303-22.
- Doos, D., et al., 2022. The dangers of reused personal protective equipment: healthcare workers and workstation contamination. J. Hosp. Infect. 127, 59–68.
- Eze, S., Ijomah, W., Wong, T.C., 2020. Remanufacturing: a potential sustainable solution for increasing medical device equipment availability. J. Manuf. 10, 141–159.
- Farrell, H., Hayes, J., Laffey, J.G., Rowan, N.J., 2013. Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant Candida albicans. J. Microbiol. Methods 84 (2), 317–326. https://doi.org/10.1016/j.mimet.2010.12.021.
- Floreine, E.M., et al., 2022. Exploring inactivation of SARS-CoV-2, MERS-CoV, Ebola, Lassa, and Nipah viruses on N95 and KN95 respirator material using photoactivated methylene blue to enable reuse. Am. J. Infect. Control 50, 863–870. https://doi.org/10.1016/j.ajic. 2022.02.016.
- Fusco, A., Dicuonzo, G., Dell'Atti, V., Tatullo, M., 2020. Blockchain in healthcare: insights on COVID-19. Int. J. Environ. Res. Public Health 17, 7167. https://doi.org/10.3390/ ijerph17197/167.
- Garvey, M., Meade, E., Rowan, N.J., 2022. Effectiveness of frontline and emerging fungal disease prevention and control interventions and opportunities to address appropriate ecosustainable solutions. Sci. Total Environ. 851 (2).. https://doi.org/10.1016/j. scitotenv.2022.158284.
- Gómez-López, V.M., et al., 2021. Guidelines on reporting treatment conditions for emerging technologies in food processing. Crit.Rev.Food Sci.Nutr. 62 (21), 5925–5949. https:// doi.org/10.1080/10408398.2021.18.95058.

- Goodman, B., 2022. Quicky and stealthy 'scrabble variants' are poised to drive a winder COVID-19 surge. https://edition.cnn.com/2022/10/20/health/variants-covid-wintersurge/index.html accessed 21 Oct, 2022.
- Gopinath, K.P., et al., 2020. A crucial review of the influence of energy, environmental and economic factors on various processes used to handle and recycle plastic wastes: development of a comprehensive index. J. Clean Prod. 274, 123031.
- Guidotti, E., Adria, D., 2020. COVID-19 data hub. J.Open Source Softw. https://doi.org/10. 21105/joss.02376.
- Haedi, E., et al., 2021. Gamma radiation sterilization of N95 respirators leads to decreased respirator performance. PLOS One 16 (4), e0248859. https://doi.org/10.1371/journal. pone.0248859.
- Haug, N., et al., 2020. Ranking the effectiveness of worldwide COVID-19 government interventions. Nat. Hum. Behav. 4 (12) 1303-1212.
- Hayes, J., Kirf, D., Garvey, M., Rowan, N.J., 2013. Disinfection and toxicological assessments of pulsed UV and pulsed-plasma gas-discharge treated-water containing the waterborne protozoan enteroparasite Cryptosporidium parvum. J. Microbiol. Methods 94 (3), 325–337. https://doi.org/10.1016/j.mimet.2013.07.012.
- Healthcare Plastics Recycling Council, 2019. Why Recylcers Should Consider Healthcare Plastics as a Valuable Feedstock. (Accessed 29 November, 2022) https://www.hprc.org/singpost/2019/01/22/recyclers-advantage-healthcare-plastics.
- Huang, H., et al., 2020. COVID-19: a call for physical scientists and engineers. ACS Nano 14, 3747–3754.
- Huang, M., et al., 2022. Plasma generated ozone and reactive oxygen species for point of use PPE decontamination system. PLOS One 17 (2), e0262818. https://doi.org/10.1371/ journal.pone.0262818.
- IHME COVID-19 Forecasting Team, 2021. Modelling COVID-19 scenarios for the United States. Nat. Med. 27, 94–105.
- I'Ons, G., 2021. Optimising end-to-end sustainability for single-use devices. Med-Tech Innovation News (accessed 29 November, 2022) https://www.med-technews.com/medtechinsights/medtech-materials-and-assembly-insights/green-goals-optimising-end-to-endsustainability-for-single-/.
- Jatta, M., et al., 2021. N95 reprocessing by low temperature sterilization with 59% vaporized hydrogen peroxide during the 2020 COVID-19 pandemic. Am. J. Infect. Control 49 (1), 8–14.
- Jiang, S., Shi, Z.L., 2020. The first disease X caused by a highly transmissible acute respiratory syndrome coronavirus. Viorl. Sin. 2020. https://doi.org/10.1007/s12250-020-00206-5.
- Kampf, G., Todt, D., Pfaender, S., Steinmaan, E., 2020. Persistence of coronaviruses on inanimate surfaces and their inactivation with biological agents. J. Hosp. Infect. 104 (3), 246–251. https://doi.org/10.1016/j.jhin.2020.01.022.
- Kea, B., et al., 2021. An international survey of healthcare workers use of personal protective equipment during the early stages of the COVID-19 pandemic. JACEP Open https://doi. org/10.1002/emp2.12392.
- Klemeš, J.J., Fan, Y.V., Jiang, P., 2020. The energy and environmental footprints of COVID-19 fighting measures - PPE, disinfection, supply chains. Energy (Oxf) 15 (211), 118701. https://doi.org/10.1016/j.energy.2020.118701.
- Kumar, H., Azad, A., Gupta, A., et al., 2021. COVID-19 creating another problem? Sustainable solution for PPE disposal through LCA approach. Environ. Dev. Sustain. 23, 9418–9432. https://doi.org/10.1007/s10668-020-01033-0.
- La Plante, B., 2022. How to design medical devices with sustainability in mind. Med Device Online (accessed 29 Nov, 2022) https://www.meddeviceonline.com/doc/how-todesign-medical-devices-with-sustainability-in-mind-0001.
- Lee, C.-M., Woo, W.-S., Roh, Y.-H., 2017. Remanufacturing: trends and issues. Int. J. Precis. Manuf. Green Technol. 4, 113–125.
- Lendvay, et al., 2022. Addressing personal protective equipment (PPE) decontamination: methylene blue and light inactivates severe acute respiratory coronavirus virus 2 (SARS-CoV-2) on N95 respirators and medical masks with maintenance of integrity and fit. Infect.Control Hosp.Epidemiol. 43, 876–885.
- Li, Q., et al., 2021. Photocatalytic rejuvenation enabled self-sanitizing, reusable, and biodegradable masks against COVID-19. ACS Nano https://doi.org/10.1021/acsnano. 1c03249.
- Lovelace Jr., B., 2020. HHS clarified US has about 1% of face masks needed for 'full blown' coronavirus pandemic. CNBC, Englewood Cliffs, NJ. https://www.cnbc.com/2022/03/ 04/hhs-clarifies-us-has-about-1percent-of-face-masks-needed-for-full-blown-pandemic. html. (Accessed 20 October 2022).
- Mahato, S., Pal, S., Ghost, K.G., 2020. Effect of lockdown amid COVID-19 pandemic on air quality of the megacity Delhi, India. Sci. Total Environ. https://doi.org/10.1016/j. scitotenv.2020.139086.
- Masterson, K., Meade, E., Garvey, M., Lynch, M., Major, I., Rowan, N.J., 2021. Development of a low-temperature extrusion process for production of GRAS bioactive-polymer loaded compounds for targeting antimicrobial-resistant (AMR) bacteria. Sci. Total Environ. 800. https://doi.org/10.1016/j.scitotenv.2021.149545.
- McAvoy, M., et al., 2021. 3 D printed frames to enable reuse and improved fit of N95 and KN95 respirators. BMC Biomed.Eng. https://doi.org/10.1186/s42490-021-00055-7.
- McEvoy, B., Rowan, N.J., 2019. Terminal sterilization of medical devices using vaporized hydrogen peroxide: a review of current methods and emerging opportunities. J. Appl. Microbiol. 127 (5), 1403–1420. https://doi.org/10.1111/jam.14412.
- McEvoy, B., Lynch, M., Rowan, N.J., 2021. Opportunities for the application of real-time bacterial cell analysis using flow cytometry for the advancement of sterilization microbiology. 130 (6), pp. 1794–1812. https://doi.org/10.1111/jam.14876.
- McGain, F., Murety, J., Lawson, C., Sherman, J.D., 2020. Environmental sustainability in anaesthesia and critical care. Br. J. Anaesth. 125 (5), 680–692.
- Meister, T.L., et al., 2021. Comparable environmental stability and disinfection profiles of the currently circulating SARS-Cov-2 variants of concern B.1.1.7 and B.1.3.5. J.Infect.Dis. 224 (3), 420–424. https://doi.org/10.1093/infdis/jiab260.
- Michie, S., West, R., Harvey, N., 2020. The concept of 'fatigue' in tracking COVID-19. MBJ https://doi.org/10.1136/bmj.m4171.

- Mtetwa, G., et al., 2021. Sterilization of gowns: making the most of a scarce commodity during the COVID-19 pandemic. Public Health Action 11 (3), 112–113. https://doi.org/10. 5588/pha.21.0024.
- Nandi, S., et al., 2021. Redesigning supply chains using blockchain-enabled circular economy and COVID-19 experiences. Sustain. Prod. Consum. 27, 10–22.
- Peteet, J.R., 2020. COVID-19 anxiety. J. Relig. Health 59, 2203-2204.
- Peters, K., 2016. Methodological issues in life cycle assessment for manufactured products: a critical review of existing studies and an illustrative case study. J. Clean. Prod. 126, 21–37.
- Potting, J., et al., 2017. Circular Economy: Measuring Innovation in the Product Chain. PBL Publishers: Netherlands Environmental Assessment Agency, The Hauge, The Netherlands.
- Public Health Emergency, 2012. Emergency Management and the Incident Command System. US Department of Health and Human Services, Washington, DC, USA.
- Qin, H., et al., 2022. Efficient disinfection of SARS-CoV-2-like coronavirus, pseudotyped SARS-CoV-2 and other coronaviruses using cold plasma induces spike protein damage. J. Hazard. Mater. 430. https://doi.org/10.1016/j.jhazmat.2022.128414.
- Rhee, C., et al., 2022. Survey of coronavirus disease 2019 (COVID-19) infection control policies at leading US academic hospitals in the context of the initial pandemic surge of the severe acute respiratory coronavirus virus 2 (SARS-CoV-2) omicron variant. Infect. Control Hosp. Epidemiol. https://doi.org/10.1017/ice.2022.155.
- Rowan, N.J., 2019. Pulsed light as an emerging technology to cause disruption for food and adjacent industries – quo vadis? Trends Food Sci. Technol. 88, 316–332. https://doi. org/10.1016/j.tifs.2019.03.027.
- Rowan, N.J., Casey, O., 2021. Empower Eco multiactor HUB: a triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. Curr.Opin.Environ.Sci.Health 21. https://doi.org/10. 1016/j.cosh.2021.100254.
- Rowan, N.J., Galanakis, C.M., 2020. Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: quo vadis? Sci. Total Environ. 748. https://doi.org/10.1016/j.scitotenv.2020.141362.
- Rowan, N.J., Laffey, J.G., 2020a. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from coronavirus disease (COVID19) pandemic – case study from the Republic of Ireland. Sci. Total Environ. 725. https://doi.org/10.1016/j.scitotenv.2020.138532.
- Rowan, N.J., Laffey, J.G., 2020b. Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from coronavirus disease (COVID-19) pandemic – Implications for efficacy, re-use and sustainable waste management. 752. https://doi.org/10.1016/j.scitotenv.2020.142259.
- Rowan, N.J., Moral, R.A., 2021. Disposable face masks and reusable face coverings as nonpharmaceutical interventions (NPIs) to prevent transmission of SARS-CoV-2 variants that cause coronavirus disease (COVID-19): role of new sustainable NPI design innovations and predictive mathematical modeling. Sci. Total Environ. 772. https://doi.org/ 10.1016/j.scitotenv.2021.145530.
- Rowan, N.J., Meade, E., Garvey, M., 2021. Efficacy of frontline chemical biocides and disinfection approaches for inactivating SARS-CoV-2 variants of concern that cause coronavirus disease with the emergence of opportunities for green eco-solutions. Curr.Opin. Environ.Sci.Health 23. https://doi.org/10.1016/j.coesh.2021.100290.
- Rowan, N.J., Murray, N., Qiao, Y., O'Neill, E., Clifford, E., Barcelo, D., Power, D.M., 2022. Digital transformation of peatland eco-innovations ('Paludiculture'): enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products and services. Sci. Total Environ. 838 (3). https://doi.org/10.1016/j.scitotenv.2022.156328.
- Rubio-Romero, J.C., del Carmen Pardo-Ferreira, M., Torrecilla-Garcia, J.A., Calero-Castro, S., 2020. Disposable masks: disinfection and sterilization for reuse, and non-certified manufacturing in the face of shortages during the COVID-19 pandemic. Sci. Total Environ. https://doi.org/10.1016/j.ssci.2020.104830.
- Ruiz-Salmón, I., et al., 2020. Addressing challenges and opportunities of the European seafood sector under a circular economy framework. Curr.Opin.Environ.Sci.Health 13, 101–106.
- Schulte, A., Maga, D., Thonemaan, N., 2021. Combining life cycle assessment and circularity assessment to analyse environmental impacts of the medical remanufacturing of electrophysiology catheters. Sustainability 13, 898. https://doi.org/10.2290/st13020898.

Shi, Y., et al., 2020. An overview of COVID-19. J. Zhejiang Univ. Sci. B 21, 343–360.

Singh Siwal, S., et al., 2021. Key ingredients and recycling strategy of personal protective equipment (PPE): towards sustainable solution for the COVID-19 like pandemics. J.Environ.Chem.Eng. https://doi.org/10.1016/j.jece.2021.106284.

- Singh, N., Tang, Y., Ogunseitan, O.A., 2020. Environmentally sustainable management of used personal protective equipment. Environ. Sci. Technol. 54 (14), 8500–8502.
- Smith-Schoenwalder, C., 2022. CDC data: BA.5 plateaus at 88% of new COVID-19 cases. USNews.com https://www.usnews.com/news/health-news/articles/2022-08-26/cdcdata-omicron-subvariant-ba-5-plateaus-at-88-of-new-covid-19-cases. (Accessed 11 November 2022).
- Smith, M., Alvarez, F., 2021. Identifying mortality factors from machine learning using Shapley values – a case of COVID19. Expert Syst. Appl. 176. https://doi.org/10.1016/j. eswa.2021.114832.
- Sohrabi, C., et al., 2020. World Health Organization declares global emergency: a review of the 2019 novel coronavirus (COVID-19). Int. J. Surg., 76–77 https://doi.org/10.1016/j. ijsu.2020.02.034.
- Speare-Cole, R., 2021. Vomiting bug norovirus surging as COVID-19 restrictions life, experts warn. https://news.sky.com/story/vomiting-bug-norovirus-surging-as-covid-19restrictions-lift-experts-warn-12357264 accessed 23 Oct, 2022.

- Tabah, A., et al., 2020. Personal protective equipment and intensive care unit healthcare worker safety in the COVID-19 era (PPE-SAFE): an international survey. J. Crit. Care 59, 70–75. https://doi.org/10.1016/j.jcrc.2020.06.005.
- Taylor, D., 2022. As Robot Technology Surges, Sterilization Lags Behind. OR Management News – Perspective. (accessed 18th Oct, 2022) https://www.ormanagement.net/ Perspective/Article/06-22/As-Robot-Technology-Surges-Sterilization-Lags-Behind/ 66972.
- Told, R., et al., 2022. A state-of-the-art guide to the sterilization of thermoplastic polymers and resin materials used in the additive manufacturing of medical devices. Mater. Des. 223. https://doi.org/10.1016/j.matdes.2022.111119.
- United Nations, 2021. UN rolls out new technology initiative to help developing countries fight COVID-19. https://www.un.org/technologybank/news/un-rolls-out-newtechnology-initiative-help-developing-countries-fight-covid-19 Accessed 23rd Oct, 2022.
- University of Galway, 2022. Strengthening EU preparedness and response to future pandemics. https://stories.nuigalway.ie/-strengthening-eu-preparedness-and-response-tofuture-pandemics-/index.html accessed 23 Oct, 2022.
- US FDA, 2020. Emergency Use Authorization (EUA) information, and list of all current EUAs. https://www.fda.gov/emergency-preparedness-and-response/mcm-legal-regulatoryand-policy-framework/emergency-use-authorization#covidppe (Accessed 8th July, 2020).
- US Food and Drug Administration, 2020. Enforcement policy for face masks and respirators during the coronavirus disease (COVID-19) public health emergency, guidance for industry and Food and Drug Administration staff. Revisions dated March 2020, April 2020 and May 2020. 2020US FDA, Washington, DC (Accessed 20 Oct, 2022) https://www.fda. gov/regulatory-information/search-fda-guidance-documents/enforcement-policy-freemasks-and-respirators-during-coronavirus-disease-covid-19-public-health.
- US Food and Drug Administration, 2022. Personal protective equipment EUAs. ttps://www. fda.gov/medical-devices/coronavirus-disease-2019-covid-19-emergency-useauthorizations-medical-devices/personal-protective-equipment-euas Accessed 1st Dec,
- 2022. Vaira, L.A., Salzano, G., Deiana, G., de Riu, G., 2020. Ansomia and ageusia: common findings
- in COVID-19 patients. Laryngoscope https://doi.org/10.1002/lary.28692. Van Doremalen, N., et al., 2020. Aerosol and surface stability of SARS-CoV-2 as compared
- with SARS-CoV-1. N. Engl. J. Med. 382, 1564–1567.Viana Martins, C.P., Xavier, C.S.F., Cobrado, L., 2022. Disinfection methods against SARS-CoV-2: a systematic review. J. Hosp. Infect. 119, 84–117.
- Viscusi, et al., 2009. Evaluation of the five decontamination methods for filtering facepice respirators. Ann. Occup. Hyg. 53, 815–827.
- Wang, J., et al., 2020. Disinfection technology of hospital wastes and wastewater: Suggestions for disinfection strategy during coronavirus Disease 2019 (COVID-19) pandemic in China. Environ. Pollut. 262. https://doi.org/10.1016/j.envpol.2020.114665.
- Wang, L., Berger, N.A., Davis, P.B., Kaelber, D.C., Volkow, N.D., Xu, R., 2022. COVID-19 rebound after Paxlovid and Molnupiravir during January-June 2022. medRxiv [Preprint] https://doi.org/10.1101/2022.06.21.22276724.
- Wesemann, C., et al., 2020. 3-D printed protective equipment during COVID-19 pandemic. Materials https://doi.org/10.3390/ma13081997.
- Wesssleder, R., Lee, H., Ko, J., Pittet, M.J., 2020. Sci. Transl. Med. 12 (546). https://doi.org/ 10.1126/scitranslmed.abc1931.
- Westmeath Independent, 2020. Second National Award for Mersus. (accessed 19th Oct, 2022) https://www.westmeathindependent.ie/2022/09/23/second-national-award-forathlone-based-mersus/.
- White, E.M., et al., 2021. Front-line nursing home staff experiences during the COVID-19 pandemic. J. Am. Med. Dir. Assoc. 22 (1), 199–203. https://doi.org/10.1016/j.jamda.2020. 11.022.
- WHO, 2020. Transmission of SARS-CoV2: implications for infection control prevention. https://www.who.int/news-room/commentaries/detail/transmission-of-sars-cov-2implications-for-infection-prevention-precautions (Accessed 20th Oct, 2020).
- WHO, 2022. Tracking SARS-CoV-2 variants. . https://www.who.int/activities/tracking-SARS-CoV-2-variants. (Accessed 20 December 2022).
- Wu, A., Peng, Y., Huang, B., Ding, X., Wang, X., Niu, P., Meng, J., Zhu, Z., Zhang, Z., Wang, J., Sheng, J., Quan, L., Xia, Z., Tan, W., Cheng, G., Jiang, T., 2020. Genome composition and divergence of the novel coronavirus (2019-nCoV) originating in China. Cell Host Microbe 27 (3), 325–328. https://doi.org/10.1016/j.chom.2020.02.001.
- Zeroual, A., Harrou, F., Dairi, A., Sun, Y., 2020. Deep learning methods for forecasting COVID-19 time-series data: a comparative study. Chaos, Solitons Fractals 140, 110121.
- Zhang, X., et al., 2020. A review on energy, environment and economic assessment in remanufacturing based on life cycle assessment method. J. Clean. Prod. 255, 120160.
- Zhang, L., Malviya, S., Tandoc, Ho., S.S., 2022. Exploring channels and gaps in information dissemination and acquisition among energy scientists and the public: The southeast Asian context. PLoS One https://doi.org/10.1371/journal.pone.0273626.
- Zhao, X., Klemes, J.J., You, F., 2022. Energy and environmental sustainability of waste personal protective equipment (PPE) treatment under COVID-19. Renew.Sustain.Energy Prev. 153. https://doi.org/10.1016/j.rser.2021.111786.
- Zhu, Y., Xie, J., Huang, F., Cao, L., 2020. Association between short-term exposure to air pollution and COVID-19 infection: evidence from China. Sci. Total Environ. https://doi.org/ 10.1016/j.scitotenv.2020.138704.

### Appendix 1.

Full list of publications for Neil J Rowan that includes papers that were not used in this higher DSc submission (period covers from 1996). Red indicates publications cited below that were <u>not included</u> as full-length papers in thesis [44 of 142 papers (**31% unused**)].

Noting, papers included in this DSc are <u>representative</u> of my leading independent research contributions to field of study. It is not practical to include full length versions for all my published work in this thesis.

Sectio	on 1: Food Safety Microbiology
1	Rowan, N.J., Anderson, J.G. and Anderton, A. (1997). Bacteriological Quality of Infant Milk Formulae Examined under a Variety of Preparation and Storage Conditions. <i>Journal of Food</i> <i>Protection</i> , <b>60</b> (1), 1089-1094.
2	Rowan, N.J., Anderson, J.G. and Anderton, A. (1997). The bacteriological quality of hospital-prepared infant feeds. <i>Journal of Hospital Infection</i> , <b>35</b> (4), <u>https://doi.org/10.1016/S0195-6701(97)90219-X</u>
3	Rowan, N.J. and Anderson, J.G. (1998). Growth and enterotoxin production by <i>diarrhoeagenic Bacillus cereus</i> in dietary supplements prepared for hospitalized HIV patients. <i>Journal of Hospital Infection</i> , <b>38</b> (2), 139-146.
4	Rowan, N.J. and Anderson, J.G. (1997). Maltodextrin stimulates growth of <i>Bacillus cereus</i> and synthesis of diarrheal enterotoxin in infant milk formulae. Applied and Environmental Microbiology, 63 (3), <u>https://doi.org/10.1128/aem.63.3.1182-1184.1997</u>
5	Rowan, N.J. and Anderson, J.G. (1998). Diarrhoeal enterotoxin production by psychrotrophic <i>Bacillus cereus</i> present in reconstituted milk-based infant formulae (MIF). <i>Letters in Applied Microbiology</i> , <b>26</b> (2), 161-165).
6	Rowan, N.J. and Anderson, J.G. (1998). Effectiveness of Cleaning and Disinfection Procedures on the Removal of Enterotoxigenic <i>Bacillus cereus</i> From Infant Feeding Bottles. <i>Journal of Food Protection</i> , <b>61</b> (2), 196-200.
7	Rowan, N.J., Caldow, G., Gemmell, C.G. and Hunter, I.S. (2003). Production of Diarrheal Enterotoxins and Other Potential Virulence Factors by Veterinary Isolates of <i>Bacillus</i> Species Associated with Nongastrointestinal Infections. <i>Applied and Environmental Microbiology</i> , <b>69</b> (4), <u>https://doi.org/10.1128/AEM.69.4.2372-2376.2003</u>
8	Rowan, N.J. and Anderson, J.G. (1998). Effects of Above-Optimum Growth Temperature and Cell Morphology on Thermotolerance of <i>Listeria monocytogenes</i> Cells Suspended in Bovine Milk. <i>Applied and Environmental Microbiology</i> , <b>64</b> (6), <u>https://doi.org/10.1128/AEM.64.6.2065-2071.1998</u>
9	Bradley, D., McNeil, B., Laffey, J.G. and Rowan, N.J (2012). Studies on the pathogenesis and survival of different culture forms of <i>Listeria monocytogenes</i> to pulsed UV-light irradiation after exposure to mild-food processing stresses. <i>Food Microbiology</i> , <b>30</b> (2), 330-339.

10	Rowan, N.J, Anderson, J.G., and Candlish, A.A.G. (2000). Cellular morphology of rough forms of <i>Listeria monocytogenes</i> isolated from clinical and food samples. <i>Letters in Applied Microbiology</i> , <b>31</b> (4), 319-322.
11	Rowan, N.J., Candlish, A.A.,G., Bubert, A., Andreson, J.G., Kramer, K. and McLauchlin, J. (2000). Virulent Rough Filaments of <i>Listeria monocytogenes</i> from Clinical and Food Samples Secreting Wild-Type Levels of Cell-Free p60 Protein. <i>Journal of Clinical Microbiology</i> , <b>38</b> (7), <u>https://doi.org/10.1128/jcm.38.7.2643-2648.2000</u>
12	Rowan, N.J., Kirf, D. and Tomkins, P. (2009). Studies on the susceptibility of different culture morphotypes of <i>Listeria monocytogenes</i> to uptake and survival in human polymorphonuclear leukocytes. <i>FEMS Immunology and Medical Microbiology</i> , <b>57</b> (2), 183-192.
13	Rowan, N.J. (1999). Evidence that inimical food-preservation barriers alter microbial resistance, cell morphology and virulence. <i>Trends in Food Science and Technology</i> , <b>10</b> (8), 261-270.
14	Geraghty, L., Booth, M., Rowan, N.J. and Fogarty, A. (2013). Investigations on the efficacy of routinely used phenotypic methods compared to genotypic approaches for the identification of staphylococcal species isolated from companion animals in Irish veterinary hospitals. Ir Vet J 66, 7 (2013). <u>https://doi.org/10.1186/2046-0481-66-7</u>
15	Rowan, N. (2004). Viable but non-culturable forms of food and waterborne bacteria: quo vadis? <i>Trends in Food Science and Technology</i> , <b>15</b> (9), 462-467.

Sectio	on 2: Non-thermal Decontamination Technology Elucidation and Development
16	Rowan, N.J., MacGregor, S.J., Anderson, J.G., Fouracre, R.A., McIIvaney, L. and Farish, O. (1999). Pulsed light inaction of food related microorganisms. <i>Applied and Environmental Microbiology</i> , <b>65</b> (3), <u>https://doi.org/10.1128/AEM.65.3.1312-1315.1999</u>
17	Garvey, M., Fernandes, J.P.A. and Rowan, N.J. (2015). Pulsed light for the inactivation of fungal biofilms of clinically-relevant pathogenic <i>Candida</i> species. <i>Yeast</i> , <b>32</b> , 533-542.
18	Farrell, H.P., Garvey, M., Cormican, M., Laffey, J.G., Rowan, N.J. (2010). Investigation of critical inter-related factors affecting the efficacy of pulsed light for inactivating clinically relevant bacterial pathogens. <i>Journal of Applied Microbiology</i> , <b>108</b> (5), 1494-1508.
19	Farrell, H., Hayes, J., Laffey, J and Rowan, N. J. (2011). Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant <i>Candida albicans</i> . <i>Journal of Microbiological Methods</i> , <b>84</b> (2), 317-326.
20	Garvey, M., Farrell, H., Cormican, M. and Rowan, N.J. (2010). Investigations of the relationship between use of in vitro cell culture-quantitative PCR and a mouse-based bioassay for evaluating critical factors affecting the disinfection performance of pulsed UV light for treating <i>Cryptosporidium parvum</i> oocysts in saline. <i>Journal of Microbiological Methods</i> , <b>80</b> (3), 267-273.
21	Garvey, M., Stocca, A and Rowan, N.J. (2014). Development of a combined <i>in vitro</i> cell culture – Quantitative PCR assay for evaluating the disinfection performance of pulsed light

	for treating the waterborne enteroparasite <i>Giardia lamblia</i> . <i>Experimental Parasitology</i> , <b>144</b> , 6-13.
22	Garvey, M., Clifford, E., O'Reilly, E., Rowan, N.J. (2013). Efficacy of Using Harmless <i>Bacillus</i> Endospores to Estimate the Inactivation <i>of Cryptosporidium parvum</i> Oocysts in Water. <i>The Journal of Parasitology</i> , <b>99</b> (3), 448-452.
23	Garvey, M., Rabbitt, D., Stocca, A. and Rowan, N.J. (2014). Pulsed ultraviolet light inactivation of <i>Pseudomonas aeruginosa</i> and <i>Staphylococcus aureus</i> biofilms. <i>Water and the Environmental Journal</i> , <b>29</b> (1), 36-42.
24	Hayes, J.C., Laffey, J.G., McNeil, B. and Rowan, N.J. (2012). Relationship between growth of food-spoilage yeast in high-sugar environments and sensitivity to high-intensity pulsed UV light irradiation. <i>International Journal of Food Science and Technology</i> , <b>47</b> (9) 1925-1934.
25	Garvey, M., Stocca, A. and Rowan, N.J. (2016). Use of a real-time PCR assay to assess the effect of pulsed light inactivation on bacterial cell membranes and associated cell viability. <i>Water Environment Research</i> , <b>88</b> (2) 168-174.
26	Rowan, N.J., Valdramidis, V.P. and Gomez-Lopez, V.M (2015). A review of quantitative methods to describe efficacy of pulsed light generated inactivation data that embraces the occurrence of viable but non culturable state microorganisms. <i>Trends in Food Science and Technology</i> , <b>44</b> (1), 79-92.
27	Fitzhenry, K., Clifford, E., Rowan, N.J. and Val de Rio, A. (2021). Bacterial inactivation, photoreactivation and dark repair post flow-through pulsed UV disinfection. <i>Journal of Water Processing Engineering</i> , <b>41</b> , <u>https://doi.org/10.1016/j.jwpe.2021.102070</u>
28	Garvey, M., Coughlan, G., Murphy, N. and Rowan, N.J. (2016). The pulsed light inactivation of veterinary relevant microbial biofilms and the use of a RTPCR assay to detect parasite species within biofilm structures. <i>Open Veterinary Journal</i> , <b>6</b> (1), <u>https://doi.org/10.4314/ovj.v6i1.3</u>
29	Garvey, M., Hayes, J., Clifford, E., Kirf, D. and Rowan, N.J. (2013). Efficacy of measuring cellular ATP levels to determine the inactivation of pulsed UV treated <i>Cryptosporidium parvum</i> oocysts suspended in water. <i>Water Supply</i> , <b>13</b> (2), 202-213.
30	Rowan, N.J. (2019). Pulsed light as an emerging technology to cause disruption for food and adjacent industries – Quo vadis? Trends in Food Science and Technology, 88, 316-332.
31	Rowan, N.J., MacGregor, S.J., Anderson, J.G., Fouracre, R.A. and Farish, O. (2000). Pulsed electric field inactivation of diarrhoeagenic <i>Bacillus cereus</i> through irreversible electroporation. <i>Letters in Applied Microbiology</i> , <b>31</b> (2), 110-114.
32	Yaqub, S., Anderson, J.G., MacGregor, S.J. and Rowan, N.J. (2004). Use of a fluorescent viability stain to assess lethal and sublethal injury in food-borne bacteria exposed to high-intensity pulsed electric fields. <i>Letters in Applied Microbiology</i> , <b>39</b> (30, 246-251.
33	Beveridge, J.R., Wall, K., MacGregor, S.J., Anderson, J.G. and Rowan, N.J. (2003). Pulsed electric field inactivation of spoilage microorganisms in alcoholic beverages and the influence of pulse profile. Digest of Technical Papers. PPC-2003. 14th IEEE International

	Pulsed Power Conference (IEEE Cat. No.03CH37472), Dallas, TX, USA, 2003, pp. 1132- 1135 Vol.2, doi: 10.1109/PPC.2003.1278011.
34	Rowan, N.J. MacGregor, S.J., Anderson, J.G., Cameron, D. and Farish, O. (2001). Inactivation of <i>Mycobacterium paratuberculosis</i> by Pulsed Electric Fields. <i>Applied and</i> <i>Environmental Microbiology</i> , <b>67</b> (6), <u>https://doi.org/10.1128/AEM.67.6.2833-2836.2001</u>
35	Rowan, N.J., Espie, S., Harrower, J., Anderson, J.G., Marsili, L., MacGregor, S.J. (2007). Pulsed-Plasma Gas-Discharge Inactivation of Microbial Pathogens in Chilled Poultry Wash Water. <i>Journal of Food Protection</i> , <b>70</b> (12), 2805-2810.
36	Hayes, J., Kirf, D., Garvey, M. and Rowan, N. (2013). Disinfection and toxicological assessments of pulsed UV and pulsed-plasma gas-discharge treated-water containing the waterborne protozoan enteroparasite <i>Cryptosporidium parvum</i> . Journal of Microbiological Methods, <b>94</b> (3), 325-337.
37	Rowan, N.J., Espie, S., Harrower, J., Farrell, H., Marsili, L., Anderson, J.G. and MacGregor, S. J. (2008). Evidence of lethal and sublethal injury in food-borne bacterial pathogens exposed to high-intensity pulsed-plasma gas discharges. <i>Letters in Applied Microbiology</i> , <b>46</b> (1), 80-86.
38	McEvoy, B. and Rowan, N.J. (2019). Terminal sterilization of medical devices using vaporized hydrogen peroxide: a review of current methods and emerging opportunities. <i>Journal of Applied Microbiology</i> , <b>127</b> (5), 1403-1420.
39	Rowan, N.J., Kremer, T., McDonnell, G. (2023). A review of Spaulding's classification system for effective cleaning, disinfection and sterilization of reusable medical devices: Viewed through a modern-day lens that will inform and enable future sustainability. <i>Science of the Total Environment</i> , <b>878</b> , <u>https://doi.org/10.1016/j.scitotenv.2023.162976</u>
40	Chen, Y., Neff, M., McEvoy, B., Cao, Z., Pessoli, R., Murphy, A., Gately, N., Hopkins Jnr, M., Rowan, N.J. and Devine, M. (2019). 3D printed polymers are less stable than injection moulded counterparts when exposed to terminal sterilization processes using novel vaporized hydrogen peroxide and electron beam processes. <i>Polymer</i> , <b>183</b> , <u>https://doi.org/10.1016/j.polymer.2019.121870</u>
41	Garvey, M. and Rowan, N.J. (2023). Pathogenic Drug Resistant Fungi: A Review of Mitigation Strategies. <i>International Journal of Molecular Sciences</i> , <b>24</b> (2), <u>https://doi.org/10.3390/ijms24021584</u>
42	Garvey, M., Meade, E. and Rowan, N.J. (2022). Effectiveness of front line and emerging fungal disease prevention and control interventions and opportunities to address appropriate eco-sustainable solutions. <i>Science of the Total Environment</i> , <b>85</b> 1 (2), <u>https://doi.org/10.1016/j.scitotenv.2022.158284</u>
43	Rowan, N.J., and Laffey, J.G. (2020). Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic – Case study from the Republic of Ireland. <i>Science of the Total Environment</i> . <b>725</b> , <u>https://doi.org/10.1016/j.scitotenv.2020.138532</u>
44	Rowan, N., J. and Laffey, J.G. (2020). Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from coronavirus disease (COVID-19) pandemic – Implications for efficacy, re-use and sustainable waste

	management. Science of the Total Environment, <b>752</b> , <u>https://doi.org/10.1016/j.scitotenv.2020.142259</u>
45	Rowan, N.J. and Moral, R.A. (2016). Disposable face masks and reusable face coverings as non-pharmaceutical interventions (NPIs) to prevent transmission of SARS-CoV-2 variants that cause coronavirus disease (COVID-19): Role of new sustainable NPI design innovations and predictive mathematical modelling. <i>Science of the Total Environment</i> , <b>772</b> , <u>https://doi.org/10.1016/j.scitotenv.2021.145530</u>
46	Kremer, T.A., Rowan, N., McDonnell (2024). A new quantitative method for determining patient risk for reusable medical device categorization based on using and interpreting Kremer's cleaning classification system. Journal of Hospital Infection, https://doi.org/10.1016/j.jhin.2024.09.024
47	Murray I.M.T., Rowan, N.J., McNamee, S., Campbell, K. and Fogarty, A.M. (2018). Pulsed light reduces the toxicity of the algal toxin okadaic acid to freshwater crustacean Daphnia pulex. Environmental Science and Pollution Research, 25, 607-614.
48	MacGregor, S.J., Farish, O., Fouracre, R., Rowan, N.J. and Anderson, J.G. (2000). Investigations of pathogenic and spoilage microorganisms in a test liquid using pulsed electric fields. <i>IEEE transactions on Plasma Science</i> , <b>28</b> (1), 144-149.
49	Garvey, M. and Rowan, N.J. (2019). Pulsed UV as a potential surface sanitizer in food production processes to ensure consumer safety. <i>Current Opinion in Food Science</i> , <b>26</b> , 65-70.
50	Farrell, H., Garvey, M. and Rowan, N.J. (2009). Studies on the inactivation of medically important <i>Candida</i> species on agar surfaces using pulsed light. FEMS Yeast Research, 9 (6), 956-966.
51	MacGregor, S.J., Rowan, N.J. McIlvaney, Anderson, J.G. and Fouracre, R.A. (1998). Light inactivation of food-related pathogenic bacteria using pulsed power source. <i>Letters in Applied Microbiology</i> , 27 (2), 67-70
52	Rowan, N.J., Anderson, J.G. and Smith, J.E. (1998). Potential infective and toxic microbiological hazards associated with the consumption of fermented foods. <i>Microbiology</i> of <i>Fermented Foods</i> , 812-837.
53	Rowan, N.J., Meade, E. and Garvey, M. (2021). Efficacy of frontline chemical biocides and disinfection approaches for inactivating SARS CoV-2 variants of concern that cause coronavirus disease with the emergence of opportunities for green eco-solutions. <i>Current Opinion in Environmental Science and Health</i> , <b>23</b> , https://doi.org/10.1016/j.coesh.2021.100290
54	Broderick, M., Fogarty, A. and Rowan, N.J. (2017). Development of a high-intensity, pulsed -plasma, gas-discharge technology for destruction of hazardous aqueous environmental micropollutants. <i>International Journal of Science, Environment and Technology</i> , <b>6</b> (1), 7-19.

Section	Section 3: Large-Scale Technology Development and Demonstration		
55	Eakins, J., Lynch, M., Carolan, J.C. and Rowan, N.J. (2023). Studies on the novel effects		
	of electron beam treated pollen on colony reproductive output in commercially-reared		
	bumblebees (Bombus terrestris) for mass pollination applications. Science of the Total		
	Environment, 899, https://doi.org/10.1016/j.scitotenv.2023.165614		
56	Naughton, J., Tiedeken, E.J., Garvey, M., Stout, J.C. and Rowan, N.J. (2017). Pulsed light		
	inactivation of the bumble bee trypanosome parasite Crithidia bombi. Pathology and		
	Parasitology, 144-154.		
57	Goblirsch, M., Eakins, J. and Rowan, N.J. (2021). Disease-mitigating innovations for the		
	pollination service industry: Challenges and opportunities. Current Opinion in		
	Environmental Science and Health, 22, https://doi.org/10.1016/j.coesh.2021.100265		

58	McEvoy, B., Maksimovic, A. and Rowan, N.J. (2022). <i>Geobacillus stearothermophilus</i> and <i>Bacillus atrophaeus</i> spores exhibit linear inactivation kinetic performance when treated with an industrial scale vaporized hydrogen peroxide (VH2O2) sterilization process. <i>Journal of Applied Microbiology</i> , <b>134</b> (1), <u>https://doi.org/10.1093/jambio/lxac028</u>
59	McEvoy, B., Maksimovic, A., Howell, D., Reppert, P., Ryan, D., Rowan, N.J. and Herve, M. (2023). Studies on the comparative effectiveness of X-rays, gamma rays and electron beams to inactivate microorganisms at different dose rates in industrial sterilization of medical devices. <i>Radiation Physics and Chemistry</i> , <b>208</b> , <u>https://doi.org/10.1016/j.radphyschem.2023.110915</u>
60	McEvoy, B., Lynch, M. and Rowan, N.J. (2021). Opportunities for the application of real- time bacterial cell analysis using flow cytometry for the advancement of sterilization microbiology. <i>Journal of Applied Microbiology</i> , <b>130</b> (6), 1794-1812.
61	McFadden, E., Costa Ramos, A., Bradley, D., Vrain, O., McEvoy, B. and Rowan, N.J. (2016). Comparative studies on the novel sterilisation of Irish retailed infant milk formula using electron beam and pulsed light treatments. <i>International Journal of Science, Environment and Technology</i> , <b>5</b> (6), 4375-4377.
62	Kremer, T., Rowan, N.J. and McDonnell, G. (2024). A proposed cleaning classification system for reusable medical devices to complement the Spaulding classification. <i>Journal of Hospital Infection</i> , <b>145</b> , 88-98.
63	Kremer, T.A., Felgar, J., Rowan, N.J. and McDonnell, G. (2023). Validation of the Device Feature Approach for Reusable Medical Device Cleaning Evaluations. <i>Biomedical</i> <i>Instrumentation and Technology</i> , <u>https://doi.org/10.2345/0899-8205-57.4.143</u>
64	Rowan, N.J., Johnstone, C.M., McLean, R.C., Anderson, J.G. and Clarke, J.A.(1999).Prediction of Toxigenic Fungal Growth in Buildings by Using a Novel Modelling System.AppliedandEnvironmentalMicrobiology,65(11),https://doi.org/10.1128/AEM.65.11.4814-4821.1999
65	Rowan, N.J (2024). Digital technologies to unlock safe and sustainable opportunities for medical device and healthcare sectors with a focus on the combined use of digital twin and extended reality applications: A review. <i>Science of the Total Environment</i> , <b>926</b> , <u>https://doi.org/10.1016/j.scitotenv.2024.171672</u>
66	Franssen, F., Gerard, C., Cozma-Petrut, A., Vieira-Pinto, M., Rezek Jambrak, A, Rowan, N.J., Paulsen, P., Rozcki, M., Tysnes, K., Rodrigues-Lasaro, D and Roberston, L. (2019). Inactivation of parasite transmission stages: Efficacy of treatments on food of animal origin. <i>Trends in Food Science and Technology</i> , <b>83</b> , 114-128.
67	Gerard, C., Franssen, F., La Barbona, S., Monteiro, S., Cozma-Petrut, A., Uttaker, K.S., Rezek Jambrak, A., Rowan, R., Rodriguez-Lasaro, D., Nasser, A., Tysnes, K and Robertson, LJ. (2019). Inactivation of parasite transmission stages: Efficacy of treatments on foods of non-animal origin. <i>Trends in Food Science and Technology</i> , <b>91</b> , 12-23.
68	Kremer, T., Murray, N., Buckley, J. and Rowan, N.J. (2023). Use of real-time immersive digital training and educational technologies to improve patient safety during the processing of reusable medical devices: Quo Vadis? <i>Science of the Total Environment</i> , <b>900</b> , <u>https://doi.org/10.1016/j.scitotenv.2023.165673</u>

(0)	
69	Byrne, T., Murray, N., McDonnell-Naughton, M. and Rowan, N.J. (2023). Perceived factors informing the pre-acceptability of digital health innovation by aging respiratory patients: a case study from the Republic of Ireland. <i>Frontiers in Public Health</i> , <b>11</b> , <u>https://doi.org/10.3389/fpubh.2023.1203937</u>
70	Rowan, N.J. and Laffey, J.G. (2020). Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic – Case study from the Republic of Ireland. <i>Science of the Total Environment</i> , <b>725</b> , <u>https://doi.org/10.1016/j.scitotenv.2020.138532</u>
71	Anderson, J.G., Rowan, N.J., MacGregor, S.J., Fouracre, R.A., and Farish, O. (2000). Inactivation of food-borne enteropathogenic bacteria and spoilage fungi using pulsed light. <i>IEEE transactions on Plasma Science</i> , <b>28</b> (1), 83-88.
72	Clarke, J.A., Johnstone, C.M., Kelly, N.J., McLean, R.C., Rowan, N.J. and Smith, J.E. (1999). A technique for the prediction of the conditions leading to mould growth in buildings. <i>Building and the Environment</i> , <b>34</b> (4), 515-521.
73	Julinawati, S., Cawley, D., Domegan, C., Brenner, M. and Rowan, N.J. (2013). A review of the perceived barriers within the health belief model on pap smear screening as a cervical cancer prevention measure. <i>Journal of Asian Scientific Research</i> , <b>3</b> (6), 677-692
74	Curley, C., Hayes, J.C., Rowan, N.J. and Kennedy, J.E. (2014). An evaluation of the thermal and mechanical properties of a salt-modified polyvinyl alcohol hydrogen for a knee meniscus application. <i>Journal of mechanical behaviour of biomedical materials</i> , <b>40</b> , 13-22.
75	Kirf, D., Higginbotham, C.L., Rowan, N.J. and Devery, S.M. (2010). Cyto- and genotoxicological assessment and functional characterization of N-vinyl-2-pyrrolidone– acrylic acid-based copolymeric hydrogels with potential for future use in wound healing applications. <i>Biomedical Materials</i> , <b>5</b> , DOI 10.1088/1748-6041/5/3/035002
76	Rowan, N.J., Anderson, J.G., Smith, J.E., Clarke, J.A., McLean, R.C., Kelly, N.J. and Johnstone, C.M. (1997). <i>Indoor and Built Environment</i> , <b>6</b> (1), 4-11.
77	Meade, E., Rowan, N.J. and Garvey, M. (2023). Bioprocessing and the Production of Antiviral Biologics in the Prevention and Treatment of Viral Infectious Disease. <i>Vaccines</i> , <b>11</b> (5), <u>https://doi.org/10.3390/vaccines11050992</u>
78	Tunku Salha, T.A., O'Neill, C. and Rowan, N.J. (2013). The use of cointegration and error correction modelling to investigate the influence of diabetes and associated medical services expenditure on economic growth in Malaysia. <i>Journal of Asian Scientific Research</i> , <b>3</b> (6), 644-653.
79	MacGregor, S.J., Farish, O., Fouracre, R., Rowan, N.J. and Anderson, J.G. (2000). Nonthermal medical/biological treatments using electromagnetic waves and ionized gases – inactivation of pathogenic and spoilage microorganisms. <i>IEEE Transactions in Plasma</i> <i>Science</i> , <b>8</b> (1), 144-149.

Section 4: Complementary and Alternative Therapies and Approaches	
80	Masterson, C.H., Murphy, E.J., Gonzalez, H., Major, I., McCarthy, S.D., O'Toole, D.,
	Laffey, J.393G. and Rowan, N.J (2020). Purified β-glucans from the Shiitake mushroom

	ameliorates a405ntibiotic-resistant <i>Klebsiella pneumoniae</i> -induced pulmonary sepsis. <i>Letters in Applie424d Microbiology</i> <b>71</b> (4), 405-412.
81	Murphy, E.J., Masterons, C., Rezoagli, E., O'Toole, D., Major, I., Stack, D., Lynch, M., Laffey, J.G. and Rowan, N.J (2020). β-Glucan extracts from the same edible shiitake mushroom <i>Lentinus edodes</i> produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects — Implications for coronavirus disease (COVID-19) immunotherapies. <i>Science of the Total Environment</i> , <b>732</b> , <u>https://doi.org/10.1016/j.scitotenv.2020.139330</u>
82	Murphy, E.J., Rezoagli, E, Pogue, R., Simonassi-Paiva, B., Aldin, I., Fehrenbach, G.W., O'Niell, E., Major, I., Laffey, J.G. and Rowan, N. (2022). Immunomodulatory activity of $\beta$ -glucan polysaccharides isolated from different species of mushroom – A potential treatment for inflammatory lung conditions. <i>Science of the Total Environment</i> , <b>809</b> , <u>https://doi.org/10.1016/j.scitotenv.2021.152177</u>
83	Murphy, E.J., Rezoagli, E., Major, I., Rowan, N.J., Laffey, J.G (2020). β-Glucan Metabolic and Immunomodulatory Properties and Potential for Clinical Application. <i>Journal of Fungi</i> , <b>6</b> (4), <u>https://doi.org/10.3390/jof6040356</u>
84	Galanakis, C.M., Aldawoud, T.M., Rizou, M., Rowan, N.J. and Ibrahim, S.A. (2020). Food Ingredients and Active Compounds against the Coronavirus Disease (COVID-19) Pandemic: A Comprehensive Review. <i>Foods</i> , <b>9</b> (11), <u>https://doi.org/10.3390/foods9111701</u>
85	Wan Mohtar, W.A.A.Q.I, IIham, Z., Jamaludin, A.A and Rowan, N.J (2021). Use of Zebrafish Embryo Assay to Evaluate Toxicity and Safety of Bioreactor-Grown Exopolysaccharides and Endopolysaccharides from European <i>Ganoderma applanatum</i> Mycelium for Future Aquaculture Applications. <i>International Journal of Molecular Sciences</i> , <b>22</b> (4), <u>https://doi.org/10.3390/ijms22041675</u>
86	Wan Mohtar, W.A.A.Q.I., Fafek, N.A., Thiran, J.P., Rahman, J.F.P., Yerima, G., Subramaniam, K. and Rowan, N.J. (2021). Investigations on the use of exopolysaccharide derived from mycelial extract of <i>Ganoderma lucidum</i> as functional feed ingredient for aquaculture-farmed red hybrid Tilapia (Oreochromis sp.). <i>Future Foods</i> , <b>3</b> , <u>https://doi.org/10.1016/j.fufo.2021.100018</u>
87	Usuldin, S.R.A., Wan-Mohtar, W.A.A.Q.I., IIham, Z., Jamaludin, A.A., Abdullah, N.R. and Rowan, N. (2021). <i>In vivo</i> toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications. <i>Nature Scientific Reports</i> , <b>11</b> , <u>https://doi.org/10.1038/s41598-021-02486-7</u>
88	Masterson, K., Meade, E., Garvey, M., Lynch, M, Major, I and Rowan, N.J (2021). Development of a low-temperature extrusion process for production of GRAS bioactive-polymer loaded compounds for targeting antimicrobial-resistant (AMR) bacteria. <i>Science of the Total Environment</i> , <b>80</b> 0, <u>https://doi.org/10.1016/j.scitotenv.2021.149545</u>
89	Masterson, K., Major, I., Lynch, M. and Rowan, N.J. (2023). Synergy Assessment of Four Antimicrobial Bioactive Compounds for the Combinational Treatment of Bacterial Pathogens. <i>Biomedicines</i> , <b>1</b> 1 (8), <u>https://doi.org/10.3390/biomedicines11082216</u>

90	Rowan, N.J., Meade, E., Garvey, M. (2021). Efficacy of frontline chemical biocides and disinfection approaches for inactivating SARS-CoV-2 variants of concern that cause coronavirus disease with the emergence of opportunities for green eco-solutions. <i>Current Opinion in Environmental Science and Health</i> , <b>23</b> , <u>https://doi.org/10.1016/j.coesh.2021.100290</u>
91	Meade, E., Hehir, S., Rowan, N.J. and Garvey, M. (2022). Mycotherapy: Potential of Fungal Bioactives for the Treatment of Mental Health Disorders and Morbidities of Chronic Pain. <i>Journal of Fungi</i> , <b>8</b> (3), <u>https://doi.org/10.3390/jof8030290</u>
92	Sullivan, R., Smith, J.E., and Rowan, N.J. (2006). Medical mushrooms and cancer therapy: translating a traditional practice into Western medicine. <i>Perspectives in Biology and Medicine</i> . <b>49</b> (2), 159-170.
93	Smith, J.E., Rowan, N.J. and Sullivan, R. (2002). Medical mushrooms: a rapidly developing area of biotechnology for cancer therapy and other bioactives. <i>Biotechnology Letters</i> , <b>24</b> , 1839-1835
94	Rowan, N.J., Smith, J.E., and Sullivan, R. (2003). Immunomodulatory activities on mushroom glucans and polysaccharide-protein complexes in animals and humans. International Journal of Medicinal Mushrooms, 5 (2), 1- 16. DOI: 10.1615/InterJMedicMush.v5.i2.10
95	Smith, J.E., Sullivan, R. and Rowan, N.J. (2003). The role of polysaccharides derived from medicinal in cancer treatment programs – current perspectives. <i>International Journal of Medicinal Mushrooms</i> . <b>5</b> (3), 20-38. DOI: 10.1615/InterJMedicMush.v5.i3.10
96	Pogue, R., Murphy, EJ., Fehrenbach, G.W., Rezoagli, E. and Rowan, N.J. (2021). Exploiting immunomodulatory properties of beta-glucans derived from natural products for improving health and sustainability in aquaculture-farmed organisms. <i>Current Opinion in Environmental Science and Health</i> , <b>21</b> , <u>https://doi.org/10.1016/j.coesh.2021.100248</u>
97	Taufek, N.M., Harith, H.H., Rahim, M.H.A., Ilham, Z, Rowan, N.J. and Wan-Mohtar W. (2020). Performance of mycelial biomass and exopolysaccharide from Malaysian <i>Ganoderma lucidum</i> for the fungivore red hybrid Tilapia ( <i>Oreochromis</i> sp.) in Zebrafish embryo. <i>Aquaculture Reports</i> , <b>17</b> , <u>https://doi.org/10.1016/j.aqrep.2020.100322</u>
98	Murphy, E.J., Rezoagli, E., Major, I., Rowan, N.J. and Laffey, J.G. (2021). Beta-glucans. <i>Encyclopedia</i> , <b>1</b> (3), 831-847.
99	Abidin, I.Z., Rezoagli, E., Simonassi-Paiva, B., Fehrenbach, G.W., Masterson, K., Rowan, N.J., and Major, I. (2020). A bilayer vaginal table for the localized delivery of disulfiram and 5-fluorouracil to the cervix. Pharmaceutics, 12 (12), https://doi.org/10.3390/pharmaceutics12121185
100	Crowley, E., Rowan, N.J., Faller, D., and Friel, A.M. (2019). Natural and synthetic isothiocyanates possess anticancer potential against liver and prostate cancer in vitro. <i>Anticancer Research</i> , <b>39</b> (7), 3469-3485.
101	Smith, J.E., Sullivan, R., and Rowan, N.J. (2005). Mushrooms and cancer therapy. <i>Biologist</i> <b>52</b> (6), 328.
102	McNeil, B., Harvey, L.M., Rowan, N.J. and Giavasis, I. (2013). Fermentation monitoring and control of microbial cultures for food ingredient manufacture. <i>Food Science, Technology and Nutrition</i> , 125-143. <u>https://doi.org/10.1533/9780857093547.1.125</u>

103	Hughes, P., Kealy, C., Rowan, N.J., and Brady, D.B. (2013). Evaluation of vegetable oils and their respective fatty acids on the viability of <i>Streptococcus mutans</i> , a persistent oral pathogen. <i>Journal of Asian Scientific Research</i> , <b>3</b> (6), 670-578.

Section 5: Sustainable innovation and frameworks	
104	Naughton, S., Kavanagh, S., Lynch, M. and Rowan, N.J. (2020). Synchronizing use of sophisticated wet-laboratory and in-field handheld technologies for real-time monitoring of key microalgae, bacteria and physicochemical parameters influencing efficacy of water quality in a freshwater aquaculture recirculation system: A case study from the Republic of Ireland. <i>Aquaculture</i> , <b>526</b> , https://doi.org/10.1016/j.aquaculture.2020.735377
105	Tahar, A., Kennedy, A.M., Fitzgerald, R.D., Clifford, E. and Rowan, N.J. (2018) Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. <i>PeerJ 6 Aquatic Biology</i> , <b>e5281</b> , <u>https://doi.org/10.7717/peerj.5281</u>
106	Tahar, A., Kennedy, A., Fitzgerald, R., Clifford, E. and Rowan, N.J. (2018). Full water quality monitoring of a traditional flow-through rainbow trout farm. <i>Fishes</i> <b>3</b> (3),https://doi.org/10.3390/fishes3030028
107	Rowan, N.J., Fort, A., O'Neill, E.A., Clifford, E., Jansen, M., Helfert, M., Toner, D., Maguire, J. and Tiwari, B. (2024). Development of a novel recirculatory multitrophic peatland system for the production of high-value bio-based products at scale embracing zero waste and pollution principles to unlock sustainable development goals. <i>Case Studies in Chemical and Environmental Engineering</i> , <b>9</b> , <u>https://doi.org/10.1016/j.cscee.2024.100763</u>
108	O'Neill, E.A. and Rowan, N.J. (2022). Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance – A comparative case study from the Republic of Ireland. <i>Science of the Total Environment</i> , <b>802</b> , <u>https://doi.org/10.1016/j.scitotenv.2021.149800</u>
109	O'Neill, E.A., Morse, A.P. and Rowan, N.J. (2022). Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: Implications and opportunities for advancing research and disruptive innovation post COVID-19 era. <i>Science of the Total Environment</i> , <b>819</b> , <u>https://doi.org/10.1016/j.scitotenv.2022.153073</u>
110	O'Neill, E.A., Rowan, N.J. and Fogarty, A.M. (2019). Novel use of the alga <i>Pseudokirchneriella subcapitata</i> , as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish as a case study. <i>Science of the Total Environment</i> , <b>692</b> , 209-218.
111	O'Neill, E.A., Fehrenbach, G., Murphy, E., Alencar, S.A., Pogue, R. and Rowan, N.J. (2022). Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: Case study from the Republic of Ireland. <i>Science of the Total Environment</i> , 851 (2), <u>https://doi.org/10.1016/j.scitotenv.2022.158392</u>
112	O'Neill, E.A., Stejskal, V., Paolacci, S., Jansen, M.A.K. and Rowan, N.J. (2024). Quo vadis - Development of a novel peatland-based recirculating aquaculture multi-trophic pond

	system (RAMPS) in the Irish midlands with a global orientation. <i>Case Studies in Chemical and Environmental Engineering</i> , <b>9</b> , <u>https://doi.org/10.1016/j.cscee.2024.100748</u>
113	O'Neill, E.A. and Rowan, N.J. (2023). Potential disruptive effects of zoosporic parasites on peatland-based organic freshwater aquaculture: Case study from the Republic of Ireland. <i>Science of the Total Environment</i> , <b>868</b> , <u>https://doi.org/10.1016/j.scitotenv.2023.161495</u>
114	Rowan, N.J., Murray, N., Qiao, Y., O'Neill, E., Clifford, E., Barcelo, D. and Power, D.M. (2022). Digital transformation of peatland eco-innovations ('Paludiculture'): Enabling a paradigm shift towards the real-time sustainable production of 'green-friendly' products and services. <i>Science of the Total Environment</i> , <b>838</b> (3), https://doi.org/10.1016/j.scitotenv.2022.156328
115	Galanakis, C.M., Rizou, M., Aldawoud, T.M.S., Ucak. I. and Rowan, N.J. (2021). Innovations and technology disruptions in the food sector within the COVID-19 pandemic and post-lockdown era. <i>Trends in Food Science and Technology</i> , <b>110</b> , 193-200.
116	Fehrenbach, G.W., Murphy, E., Pogue, R., Carter, F., Clifford, E., Major, I. and Rowan, N. (2023). Pulsed ultraviolet (PUV) disinfection of artificially contaminated seawater seeded with high levels of pathogen disease indicators as an alternative for the shellfish industry depuration systems. <i>Environmental Science and Pollution Research</i> , <b>30</b> , 70771-70782.
117	Fehrenbach, G.W., Pogue, R., Carter, F., Clifford, E. and Rowan, N.J. (2022). Implications for the seafood industry, consumers and the environment arising from contamination of shellfish with pharmaceuticals, plastics and potentially toxic elements: A case study from Irish waters with a global orientation. <i>Science of the Total Environment</i> , <b>844</b> , <u>https://doi.org/10.1016/j.scitotenv.2022.157067</u>
118	Finnegan, W., Clifford, E., Goggins, J., O'Leary N., Dobson, A., Rowan, N., Xiao, L., Miao, S., Fitzhenry, K., Leonard, P., Tarpey, E., Gil-Pulido, B., Gao, F. and Zhan, X. (2018). Dairywater – striving for sustainability within the diary processing industry in the Republic of Ireland. <i>Journal of Diary Research</i> , <b>85</b> (3), 366-374.
119	Tiedeken, E.J., Tahar, A., McHugh, B. and Rowan, N.J. (2017). Monitoring, sources, receptors, and control measures for three European Union watch list substances of emerging concern in receiving waters – A 20 year systematic review. <i>Science of the Total Environment</i> , <b>574</b> , 1140-1163.
120	Tahar, A., Tiedeken, E.J., Clifford, E., Cummins, E. and Rowan N.J. (2017). Development of a semi-quantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater treatment plants: An Irish case study. <i>Science of The Total Environment</i> , <b>603-604</b> -, 627-638.
121	Tahar, A., Tiedeken, E.J. and Rowan, N.J. (2018). Occurrence and geodatabase mapping of three contaminants of emerging concern in receiving water and at effluent from waste water treatment plants – A first overview of the situation in the Republic of Ireland. <i>Science of the Total Environment</i> , <b>616-617</b> -, 187-197.
122	Rowan, N.J. (2023). Current decontamination challenges and potentially complementary solutions to safeguard the vulnerable seafood industry from recalcitrant human norovirus in live shellfish: Quo Vadis? <i>Science of the Total Environment</i> , <b>874</b> , <u>https://doi.org/10.1016/j.scitotenv.2023.162380</u>

123	Ruiz-Salmon, I., Laso, J., Margallo, M., Villanaueva-Rey, P., Rodriguez, E., Quinteiro, P., Claudia Dias, A., Almeida, C., Clifford, E., Reguiero, L., Mendez, D., Anglada, C., Noirot, C., Rowan, N.J., Vazquez-Rowe, I. and Aldaco, R. (2021). Life cycle assessment of fish and seafood processed products – A review of methodologies and new challenges. <i>Science of the Total Environment</i> , <b>761</b> , <u>https://doi.org/10.1016/j.scitotenv.2020.144094</u>
124	Almeida, C., Laso, J., Baptista de Sousa, Cooney, R., Quinterio, P., Rowan, N.J., Dias, A.C., Clifford E., Reboredo, R.G., Margallo, M., Nunes, M.L. and Marques, A. (2023). Seafood consumers engagement in reducing environmental impacts from packaging. <i>Science of the Total Environment</i> , <b>863</b> , <u>https://doi.org/10.1016/j.scitotenv.2022.160846</u>
125	Rowan, N.,J. and Casey, O. (2021). 'Empower Eco' multi-actor HUB: A triple helix 'academia-industry-authority' approach to creating and sharing potentially disruptive tools for addressing novel and emerging new Green Deal opportunities under a United Nations Sustainable Development Goals framework. <i>Current Opinion in Environmental Science and Health</i> , <b>21</b> , <u>https://doi.org/10.1016/j.coesh.2021.100254</u>
126	Cooney, R., Baptista de Sousa, D., Fernandez Rios, Mellett, S., Rowan, N., Morse, A.P., Hayes, M., Laso, J., Regueiro, L., Wan, A., Clifford, E. (2023). A circular economy framework for seafood waste valorisation to meet challenges and opportunities for intensive production and sustainability. <i>Journal of Cleaner Production</i> , <b>392</b> , <u>https://doi.org/10.1016/j.jclepro.2023.136283</u>
127	Rowan, N.J. and Galanakis, C.M. (2020). Unlocking challenges and opportunities presented by COVID-19 pandemic for cross-cutting disruption in agri-food and green deal innovations: Quo Vadis? <i>Science of the Total Environment</i> . 748, <u>https://doi.org/10.1016/j.scitotenv.2020.141362</u>
128	Rowan, N.J. (2023). Challenges and future opportunities to unlock the critical supply chain of personal and protective equipment (PPE) encompassing decontamination and reuse under emergency use authorization (EUA) conditions during the COVID-19 pandemic: Through a reflective circularity and sustainability lens. <i>Science of the Total Environment</i> , <b>866</b> , <u>https://doi.org/10.1016/j.scitotenv.2023.16145</u>
129	Barrett. ,M., Fitzhenry, K., O'Flaherty, Dore, W., Keaveney, S., Rowan, N.J. and Clifford, E. (2016). Detection, fate and inactivation of pathogenic norovirus employing settlement and UV treatment in wastewater treatment facilities. <i>Science of the Total Environment</i> , <b>568</b> , 1026-1036
130	McGillicuddy, E., Murray, I., Kavanagh, S., Morrison, L., Fogarty, A., Cormican, M., Dockery, P., Prendergast, M., Rowan, N. and Morris, D. (2017). Silver nanoparticles in the environment: Sources, detection and ecotoxicology. <i>Science of the Total Environment</i> , <b>575</b> , <u>https://doi.org/10.1016/j.scitotenv.2016.10.041</u>
131	O'Neill, E.A., Stejskal, V., Paolacci, S., Jansen, M.A.K. and Rowan, N.J. (2024). Quo vadis - Development of a novel peatland-based recirculating aquaculture multi-trophic pond system (RAMPS) in the Irish midlands with a global orientation. <i>Case Studies in Chemical and Environmental Engineering</i> , <b>9</b> , <u>https://doi.org/10.1016/j.cscee.2024.100748</u>
132	Wan-Mohtar, W.H., Wan-Mohtar, W.A.A.Q.I., Zahuri, A.A., Ibrahim, M.F., Show, P.W., Ilham, Z. and Rowan, N.J. (2022). Role of ascomycete and basidiomycete fungi in meeting established and emerging sustainability opportunities: a review. <i>Bioengineered</i> , <b>13</b> (7), 14903-14935.

133	Rowan, N.J. (2011). Defining established and emerging microbial risks in the aquatic environment – current knowledge, implications and outlooks. <i>International Journal of Microbiology</i> , (1), 462832.
134	Rowan, N.J. (2003). The role of digital technologies in supporting and improving fishery and aquaculture across the supply chain – Quo Vadis? <i>Aquaculture and Fisheries</i> , <b>8</b> (4), 365-374.
135	Ruiz-Salmon, I., Margallo, M., Laso, J., Villanueva-Rey, P., Marno, D., Rowan, N. J. (2020). Addressing challenges and opportunities of the European seafood sector under a circular economy framework. <i>Current Opinion in Environmental Science and Health</i> , <b>13</b> , 101-106.
136	O'Neill, E.A., Stejskal, V., Clifford, E. and Rowan, N.J. (2020). Novel use of peatlands as future locations for the sustainable intensification of freshwater aquaculture production – a case study from the Republic of Ireland. 706, <u>https://doi.org/10.1016/j.scitotenv.2019.136044</u>
137	Almeida, C., Loubet, Pl, da Costa, T.P., Quinteiro, P., Laso, de Sousa, D.P., Cooney, R., Mellett, S., Sonnemann, G., Rodriguez, C.J., Rowan, N.J., Clifford, E., Ruiz-Salmon, I., Margallo, M., Nunez, M.L., Dias, A.C., and Marqus, A. (2021). Packaging environmental impact on seafood supply chains: A review of life cycle assessment studies. <i>Journal of Industrial Ecology</i> , <b>26</b> (6), 1961-1978.
138	O'Neill, E.A., Bennett, M.M.K., and Rowan, N.J. (2022). Peatland-based innovation can potentially support and enable the sustainable development goals of the United Nations: Case study from the Republic of Ireland. <i>Case studies in Chemical and Environmental Engineering</i> , <b>6</b> , <u>https://doi.org/10.1016/j.cscee.2022.100251</u>
139	Laso, J., Ruiz-Salmon, I., Margallo, M., Villanueva-Rey, P., Poceiro, L., Quinteiro, R., Dias, A., Almeida, C., Marquez, A., Entrena-Barbero, E., Cooney, R., Rowan, N.J., Noirot, C. (2022). Achieving Sustainability of the Seafood Sector in the European Atlantic Area by Addressing Eco-Social Challenges: The NEPTUNUS Project. Sustainability, 13 (5), <u>https://doi.org/10.3390/su14053054</u>
140	Finnegan, W., Clifford, E., Goggins, J., O'Leary N., Dobson, A., Rowan, N., Xiao, L., Miao, S., Fitzhenry, K., Leonard, P., Tarpey, E., Gil-Pulido, B., Gao, F. and Zhan, X. (2018). Dairywater – striving for sustainability within the diary processing industry in the Republic of Ireland. <i>Journal of Diary Research</i> , <b>85</b> (3), 366-374.
141	Garvey, M., Hayes, J., Clifford E., Rowan, N.J (2015). Ecotoxicological assessment of pulsed light-treated water containing microbial species and <i>Cryptosporidium parvum</i> using a microbiotest test battery. <i>Water and Environment Journal</i> , <b>29</b> (1), 27-35.
142	Jahis, B.M., Ilham, Z., Supramani, S., Sohedein, M.N.A., Ibrahim, M.F., Abd-Azis, S., Rowan, N., and Wan-Mohtar, W. (2022). Ganodiesel: A new biodiesel feedback from biomass of the mushroom <i>Ganoderma lucidum</i> . <i>Sustainability</i> , <b>14</b> (17), <u>https://doi.org/10.3390/su141710764</u>
143	O'Neill, E., Jansen, M., Tiwari, B., Clifford, E., and Rowan, N.J. (2024). Development of IMTA-based bioeconomy sites in peatlands for change of land use to green innovation that promotes zero-waste, zero-pollution and climate-action principles. In "Degrowth and Green Growth, Sustainable Innovation, IntechOpen Publishers
144	Wan-Mohtar, W., Ilham, Z. and Rowan, N.J (2023). The value of microbial bioreactors to meet challenges in the circular bioeconomy. <i>Frontiers in Bioengineering and Biotechnology</i> , 11, <u>https://doi.org/10.3389/fbioe.2023.1181822</u>







University Hospital Galway Ospidéal na h-Ollscoile, Gaillimh GALWAY UNIVERSITY HOSPITALS

16<sup>th</sup> April, 2021

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am a Professor of Anaesthesia and Intensive Care Medicine at NUI Galway, Ireland. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, data analysis and in manuscript preparation, submission and proof-editing.

Examples of these collaborative efforts are listed below.

- NJ Rowan, JG Laffey. 2020. Challenges and solutions for addressing critical shortage of supply chain for personal and protective equipment (PPE) arising from Coronavirus disease (COVID19) pandemic–Case study from the Republic of Ireland. Science of The Total Environment 725, 138532. [IF 6.551].
- HP Farrell, M Garvey, M Cormican, JG Laffey, NJ Rowan. 2010. Investigation of critical inter-related factors affecting the efficacy of pulsed light for inactivating clinically relevant bacterial pathogens. Journal of Applied Microbiology 108 (5), 1494-1508 79 [IF 3.066]
- 3. H Farrell, J Hayes, J Laffey, N Rowan. 2011. Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant *Candida albicans*. Journal of Microbiological Methods 84 (2), 317-326. [IF 1.81]
- EJ Murphy, C Masterson, E Rezoagli, D O'Toole, I Major, GD Stack, J. Laffey, Rowan, N.J. 2020. β-Glucan extracts from the same edible shiitake mushroom *Lentinus edodes* produce differential in-vitro immunomodulatory and pulmonary cytoprotective effects—Implications for COVID-19. Science of the Total Environment 732, 13933 [IF 6.551]
- 5. D Bradley, B McNeil, JG Laffey, NJ Rowan. 2012. Studies on the pathogenesis and survival of different culture forms of *Listeria monocytogenes* to pulsed UV-light irradiation after exposure to mild-food processing stresses. Food Microbiology 30 (2), 330-339 [IF 4.155]
- NJ Rowan, JG Laffey. 2020. Unlocking the surge in demand for personal and protective equipment (PPE) and improvised face coverings arising from coronavirus disease (COVID-19) pandemic–Implications for efficacy, re-use and sustainable waste management ...Science of the Total Environment, 142259. [IF 6.551]
- JC Hayes, JG Laffey, B McNeil, NJ Rowan. 2012. Relationship between growth of food-spoilage yeast in high-sugar environments and sensitivity to high-intensity pulsed UV light irradiation. International Journal of Food Science & Technology 47 (9), 1925-1934. [IF 2.773]

OE Gaillimh, Bóthar na hOllscoile, Gaillimh, Eire NUI Galway, University Road, Galway, Ireland

T +353 91 492566 F +353 91 494561

anaesthesia@nuigalway.ie www.nuigalway.ie/anaesthesia/







University Hospital Galway Ospidéal na h-Ollscoile, Gaillimh GALWAY UNIVERSITY HOSPITALS

- CH Masterson, EJ Murphy, H Gonzalez, I Major, SD McCarthy, D O'Toole, Laffey, J, Rowan, N.J. 2020. Purified β-glucans from the Shiitake mushroom ameliorates antibiotic-resistant Klebsiella pneumoniae-induced pulmonary sepsis. Letters in Applied Microbiology 71 (4), 405-412. [IF 2.173]
- EJ Murphy, E Rezoagli, I Major, NJ Rowan, JG Laffey. 2020. β-glucan metabolic and immunomodulatory properties and potential for clinical application. Journal of Fungi 6 (4), 356 [IF 4.621]

If you require any further details, please do not hesitate to contact me.

**Yours Sincerely** 

John Laffey, Established Professor, Anaesthesia and Intensive Care Medicine, School of Medicine, NUI Galway. Consultant Anaesthesiologist and Intensivist, Galway University Hospitals Vice Dean for Research, College of Medicine, Nursing and Health Sciences, NUI Galway. Director of Clinical Research, Clinical Research and Development Office, Saolta Hospitals and NUI Galway.

OE Gaillimh, Bóthar na hOllscoile, Gaillimh, Eire NUI Galway, University Road, Galway, Ireland

T +353 91 492566 F +353 91 494561 anaesthesia@nuigalway.ie www.nuigalway.ie/anaesthesia/



Abbott Diagnostics, Lisnamuck, Longford. Tel: 00 353 43 3334904 Fax: 00 353 43 3331065

18<sup>th</sup> April, 2021

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am Principal Scientist (NPD) at Abbott Diagnostics Business in Ireland. Professor Rowan and I have co-authored several articles on which he was a lead or a senior author. By this I mean that he contributed to the conception, design and writing of the work, data analysis, in manuscript preparation and submission.

Examples of these collaborative efforts are listed below.

- H Farrell, M Garvey, M Cormican, JG Laffey, NJ Rowan. 2010. Investigation of critical inter-related factors affecting the efficacy of pulsed light for inactivating clinically relevant bacterial pathogens. Journal of Applied Microbiology 108 (5), 1494-1508[IF 3.066]
- H Farrell, J Hayes, J Laffey, N Rowan. 2011. Studies on the relationship between pulsed UV light irradiation and the simultaneous occurrence of molecular and cellular damage in clinically-relevant Candida albicans. Journal of Microbiological Methods 84 (2), 317-326. [IF 1.81]
- M Garvey, H Farrell, M Cormican, N Rowan. 2010. Investigations of the relationship between use of in vitro cell culture-quantitative PCR and a mouse-based bioassay for evaluating critical factors affecting the disinfection ...Journal of Microbiological Methods 80 (3), 267-273. [IF 1.81]
- 4. NJ Rowan, S Espie, J Harrower, H Farrell, L Marsili, JG Anderson. 2008. Evidence of lethal



and sub-lethal injury in food-borne bacterial pathogens exposed to high-intensity pulsed-plasma gas discharges. Letters in Applied Microbiology 46 (1), 80-86. [IF2.173]

 H Farrell, M Garvey, N Rowan. 2009. Studies on the inactivation of medically important Candida species on agar surfaces using pulsed light. FEMS Yeast Research 9 (6), 956-966.
 [IF 3.13]

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,

Dr Hugh Farrell

Principal Scientist New Product development Abbott Diagnostics Division





Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am a Senior Lecturer in the School of Engineering and Informatics at NUI Galway, Ireland. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, data analysis and in manuscript preparation, submission and proof-editing.

Examples of these collaborative efforts are listed below.

- M Barrett, K Fitzhenry, V O'Flaherty, W Dore, S Keaveney, M Cormican, Rowan, N., Clifford, E. 2016. Detection, fate and inactivation of pathogenic norovirus employing settlement and UV treatment in wastewater treatment facilities. Science of the Total Environment 568, 1026-1036 [IF 6.551]
- 2. A Tahar, EJ Tiedeken, E Clifford, E Cummins, N Rowan. 2017. Development of a semiquantitative risk assessment model for evaluating environmental threat posed by the three first EU watch-list pharmaceuticals to urban wastewater .Science of The Total Environment 603, 627-638 [6.551]
- JC Hayes, M Garvey, AM Fogarty, E Clifford, NJ Rowan. 2012. Inactivation of recalcitrant protozoan oocysts and bacterial endospores in drinking water using high-intensity pulsed UV light irradiation. Water Science and Technology: Water Supply 12 (4), 513-522
- 4. A Tahar, AM Kennedy, RD Fitzgerald, E Clifford, N Rowan. 2018. Longitudinal evaluation of the impact of traditional rainbow trout farming on receiving water quality in Ireland. PeerJ 6, e528111 [IF 2.38]
- M Garvey, J Hayes, E Clifford, N Rowan. 2015. Ecotoxicological assessment of pulsed ultraviolet light-treated water containing microbial species and Cryptosporidium parvum using a microbiotest test battery. Water and Environment Journal 29 (1), 27-35 [IF 1.369]
- A Tahar, A Kennedy, RD Fitzgerald, E Clifford, N Rowan. 2018, Full water quality monitoring of a traditional flow-through rainbow trout farm. Fishes 3 (3), 28 [IF 1.3]

National University of Ireland Galway, University Road, Galway, Ireland H91 TK33 T. +353 91 524411 www.nuigalway.ie



- M Garvey, E Clifford, E O'Reilly, NJ Rowan. 2013. Efficacy of Using Harmless Bacillus Endospores to Estimate the Inactivation of Cryptosporidium parvum Oocysts in Water. The Journal of Parasitology 99 (3), 448-452 [IF 1.236]
- K Fitzhenry, N Rowan, AV del Rio, A Cremillieux, E Clifford. 2019. Inactivation efficiency of Bacillus endospores via modified flow-through PUV treatment with comparison to conventional LPUV treatment. Journal of Water Process Engineering 27, 67-76 [IF 3.465]
- K Fitzhenry, N Rowan, W Finnegan, X Zhan, E Clifford. 2018. Microbiological characterisation and impact of suspended solids on pathogen removal from wastewaters in dairy processing factories. Journal of Dairy Research 85 (3), 391-395 [IF 1.628]
- M Garvey, J Hayes, E Clifford, D Kirf, N Rowan. 2013. Efficacy of measuring cellular ATP levels to determine the inactivation of pulsed UV treated Cryptosporidium parvum oocysts suspended in water. Water Science and Technology: Water Supply 13 (2), 202-213 4 citations. [IF 1.050]
- W Finnegan, E Clifford, J Goggins, N O'Leary, A Dobson, N Rowan, 2018. DairyWater: striving for sustainability within the dairy processing industry in the Republic of Ireland. Journal of Dairy Research 85 (3), 366-374. 2 citations. [IF 1.628]
- I Ruiz-Salmón, J Laso, M Margallo, P Villanueva-Rey, E Rodríguez, Cooney, R., Rowan, N., Noiret, C., Aldaco, Clifford, E. et al. 2020. Life cycle assessment of fish and seafood processed products—a review of methodologies and new challenges. Science of The Total Environment, 144094 1 citations. [IF 6.551]
- 13. JC Hayes, AM Fogarty, E Clifford, NJ Rowan. 2011. A comparison of the efficacy of pulsed UV light and pulsed plasma gas-discharge systems for the novel inactivation of Cryptosporidium spp. and other clinically relevant microorganisms. 2011 IEEE Pulsed Power Conference, 1574-1581
- 14. Cooney, R., Baptista de Sousa, D., Fernandez Rios, Mellett, S., Rowan, N., Morse, A.P., Hayes, M., Laso, J., Regueiro, L., Wan, A., Clifford, E. (2023). A circular economy framework for seafood waste valorisation to meet challenges and opportunities for intensive production and sustainability. Journal of Cleaner Production, 392, https://doi.org/10.1016/j.jclepro.2023.136283

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,

Dr Eoghan Clifford Eoghan Clifford BE PhD CEng MIEI Senior Lecturer School of Engineering, Academic Director CÉIM Engineering programme. NUI Galway Community and University Sustainability Partnership Executive Board. Visiting research Fellow Athlone Institute of Technology National University of Ireland Galway, University Road, Galway, Ireland H91 TK33 T. +353 91 524411

www.nuigalway.ie



Room 1035: Alice Perry Engineering Building NUI Galway, Galway. H91HX31, Ireland.

Eoghan.clifford@nuigalway.ie

National University of Ireland Galway, University Road, Galway, Ireland H91 TK33 T. +353 91 524411 www.nuigalway.ie



Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am a Senior Lecturer in the Department of Life and Physical Science at Athlone Institute of Technology, Ireland. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, data analysis and in manuscript preparation, submission and proof-editing.

Examples of these collaborative efforts are listed below.

- JC Hayes, M Garvey, AM Fogarty, E Clifford, NJ Rowan. 2012. Inactivation of recalcitrant protozoan oocysts and bacterial endospores in drinking water using high-intensity pulsed UV light irradiation. Water Science and Technology: Water Supply 12 (4), 513-522
- 2. L Geraghty, M Booth, N Rowan, A Fogarty. 2013. Investigations on the efficacy of routinely used phenotypic methods compared to genotypic approaches for the identification of staphylococcal species isolated from companion animals. rish Veterinary Journal 66 (1), 1-9
- 3. E McGillicuddy, I Murray, S Kavanagh, L Morrison, A Fogarty, M Cormican, Rowan, N., Morris, D. (2017). Silver nanoparticles in the environment: Sources, detection and ecotoxicology. Science of the Total Environment 575, 231-246.
- 4. EA O'Neill, NJ Rowan, AM Fogarty. 2019. Novel use of the alga Pseudokirchneriella subcapitata, as an early-warning indicator to identify climate change ambiguity in aquatic environments using freshwater finfish. Science of the Total Environment 692, 209-218.
- 5. M Broderick, A Fogarty, NJ Rowan. 2107. Development of a high-intensity, pulsedplasma, gas-discharge technology for destruction of hazardous aqueous environment micropollutants. International Journal of Science, Environment and Technology. 1
- 6. IMT Murray, NJ Rowan, S McNamee, K Campbell, AM Fogarty. 2018. Pulsed light reduces the toxicity of the algal toxin okadaic acid to freshwater crustacean Daphnia pulex. Environmental Science and Pollution Research 25 (1), 607-614
- 7. JC Hayes, AM Fogarty, E Clifford, NJ Rowan. 2011. A comparison of the efficacy of pulsed UV light and pulsed plasma gas-discharge systems for the novel inactivation of

Cryptosporidium spp. and other clinically relevant microorganisms. 2011 IEEE Pulsed Power Conference, 1574-1581

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,



Andy Fogarty PhD, Senuir Lecturer and Researcher, Faculty of Science and Health, Department of Life and Physical Science, Phone: +353 906471861 Email: afogarty@ait.ie





Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am a Senior Director Global Technologies at STERIS Advanced Sterilization Technologies. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design, data analysis and in manuscript preparation.

Examples of these collaborative efforts are listed below.

- 1. B McEvoy, NJ Rowan. 2019. Terminal sterilization of medical devices using vaporized hydrogen peroxide: a review of current methods and emerging opportunities. Journal of Applied Microbiology 127 (5), 1403-1420.
- E McFadden, AL Cost Ramos, D Bradley, O Vrain, B McEvoy, N Rowan. 2016. Comparative studies on the novel sterilisation of Irish retailed infant milk formula using electron beam and pulsed light. International Journal of Science, Environment and Technology 5 (6), 4375-4377.
- 3. Y Chen, M Neff, B McEvoy, Z Cao, R Pezzoli, A Murphy, N Gately, Rowan, N. 2019. 3D printed polymers are less stable than injection moulded counterparts when exposed to terminal sterilization processes using novel vaporized hydrogen peroxide and electron beam. Polymer 183, 121870
- B McEvoy, M Lynch, NJ Rowan. 2020. Opportunities for the application of real-time bacterial cell analysis using flow cytometry for the advancement of sterilization microbiology. Journal of Applied Microbiology. [IF 3.066] PMID: 33155740 DOI: 10.1111/jam.14876

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,

Brian McEvoy

21<sup>th</sup> Sept, 2024

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

# Re: Neil J Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am a Senior Research Fellow at the Technological University of the Shannon here in Ireland. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, along with playing a key role in data analysis and in manuscript preparation, submission and proofediting.

Examples of these collaborative efforts are listed below.

- O'Neill, E.A. and Rowan, N.J. (2022). Microalgae as a natural ecological bioindicator for the simple real-time monitoring of aquaculture wastewater quality including provision for assessing impact of extremes in climate variance – A comparative case study from the Republic of Ireland. Science of the Total Environment, 802, https://doi.org/10.1016/j.scitotenv.2021.149800
- O'Neill, E.A., Morse, A.P. and Rowan, N.J. (2022). Effects of climate and environmental variance on the performance of a novel peatland-based integrated multi-trophic aquaculture (IMTA) system: Implications and opportunities for advancing research and disruptive innovation post COVID-19 era. Science of the Total Environment, 819, https://doi.org/10.1016/j.scitotenv.2022.153073
- c. O'Neill, E.A., Fehrenbach, G., Murphy, E., Alencar, S.A., Pogue, R. and Rowan, N.J. (2022). Use of next generation sequencing and bioinformatics for profiling freshwater eukaryotic microalgae in a novel peatland integrated multi-trophic aquaculture (IMTA) system: Case study from the Republic of Ireland. Science of the Total Environment, 851 (2), https://doi.org/10.1016/j.scitotenv.2022.158392
- O'Neill, E.A., Stejskal, V., Paolacci, S., Jansen, M.A.K. and Rowan, N.J. (2024). Quo vadis -Development of a novel peatland-based recirculating aquaculture multi-trophic pond system (RAMPS) in the Irish midlands with a global orientation. Case Studies in Chemical and Environmental Engineering, 9, https://doi.org/10.1016/j.cscee.2024.100748
- e. O'Neill, E.A. and Rowan, N.J. (2023). Potential disruptive effects of zoosporic parasites on peatlandbased organic freshwater aquaculture: Case study from the Republic of Ireland. Science of the Total Environment, 868, https://doi.org/10.1016/j.scitotenv.2023.161495

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,

Emer Meill

Dr Emer O'Neill Senior Research Fellow, TUS (emer.oneill@tus.ie)

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am an Emeritus Professor (granted on retiral from the former Department of Bioscience and SIPBS) and now currently a Research Fellow at the Department of Electronic and Electrical Engineering at the University of Strathclyde. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, along with playing a key role in data analysis and in manuscript preparation, submission and proof-editing.

Examples of these collaborative efforts are listed below.

- NJ Rowan, SJ MacGregor, JG Anderson, RA Fouracre, L McIlvaney, O. Farish. (1999). Pulsed-light inactivation of food-related microorganisms. Applied and environmental microbiology 65 (3), 1312-1315 [IF 4.016].
- 2. JG Anderson, NJ Rowan, SJ MacGregor, RA Fouracre, O Farish. 2000. Inactivation of food-borne enteropathogenic bacteria and spoilage fungi using pulsed-light. IEEE Transactions on Plasma Science 28 (1), 83-88. [IF 1.309]
- 3. SJ MacGregor, NJ Rowan, L McIlvaney, JG Anderson. 1998. Light inactivation of food-related pathogenic bacteria using a pulsed power source. Letters in Applied Microbiology 27 (2), 67-70. 149 [IF 2.713]
- NJ Rowan, SJ Macgregor, JG Anderson, RA Fouracre, O Farish.2000. Pulsed electric field inactivation of diarrhoeagenic Bacillus cereus through irreversible electroporation. Letters in Applied Microbiology 31 (2), 110-114. [IF 2.173]
- SJ MacGregor, O Farish, R Fouracre, NJ Rowan, JG Anderson. 2000. Inactivation of pathogenic and spoilage microorganisms in a test liquid using pulsed electric fields. IEEE Transactions on Plasma Science 28 (1), 144-149 [IF 1.309]
- 6. NJ Rowan, JG Anderson. 1998. Effects of Above-Optimum Growth Temperature and Cell Morphology on Thermotolerance of Listeria monocytogenes Cells Suspended in Bovine Milk. Applied and Environmental Microbiology 64 (6), 2065-2071. [IF 4.016]
- NJ Rowan, SJ MacGregor, JG Anderson, D Cameron, O Farish. 2001. Inactivation of *Mycobacterium paratuberculosis* by pulsed electric fields. Applied and Environmental Microbiology 67 (6), 2833-2836 [4.016]
- NJ Rowan, S Espie, J Harrower, JG Anderson, L Marsili, SJ MacGregor. 2007. Pulsedplasma gas-discharge inactivation of microbial pathogens in chilled poultry wash water. Journal of Food Protection 70 (12), 2805-2810. [IF 1.60]
- 9. JR Beveridge, SJ MacGregor, L Marsili, JG Anderson, NJ Rowan. 2002. Comparison of the effectiveness of biphase and monophase rectangular pulses for microbial inactivation using pulsed electric fields. IEEE Trans Plasma Sci 30 (4), 1525-1531.

- NJ Rowan, AAG Candlish, A Bubert, JG Anderson, K Kramer, .2000. Virulent Rough Filaments of Listeria monocytogenesfrom Clinical and Food Samples Secreting Wild-Type Levels of Cell-Free p60 Protein. Journal of Clinical Microbiology 38 (7), 2643-2648. [IF 5.897]
- NJ Rowan, JG Anderson. 1997. Maltodextrin stimulates growth of Bacillus cereus and synthesis of diarrheal enterotoxin in infant milk formulae. Applied and Environmental Microbiology 63 (3), 1182-1184. [IF 4.016]
- NJ Rowan, JG Anderson. 1998. Diarrhoeal enterotoxin production by psychrotrophic Baccillus cereus present in reconstituted milk-based infant formulae (MIF). Letters in Applied Microbiology 26 (2), 161-165 [2173]
- 13. S Yaqub, JG Anderson, SJ MacGregor, NJ Rowan. 2004; Use of a fluorescent viability stain to assess lethal and sublethal injury in food-borne bacteria exposed to high-intensity pulsed electric fields. Letters in Applied Microbiology 39 (3), 246-25
- 14. JR Beveridge, K Wall, SJ MacGregor, JG Anderson, NJ Rowan. 2004. Pulsed electric field inactivation of spoilage microorganisms in alcoholic beverages. Proceedings of the IEEE 92 (7), 1138-1143 . [IF 10.694].
- 15. NJ Rowan, JG Anderson. 1998. Growth and enterotoxin production by diarrhoeagenic Bacillus cereus in dietary supplements prepared for hospitalized HIV patients. Journal of Hospital Infection 38 (2), 139-146. [IF 3.441]
- 16. NJ Rowan, JG Anderson, A Anderton. 1997. Bacteriological quality of infant milk formulae examined under a variety of preparation and storage conditions. Journal of Food Protection 60 (9), 1089-1094.
- NJ Rowan, JG Anderson, AAG Candlish. Cellular morphology of rough forms of Listeria monocytogenes isolated from clinical and food samples. Letters in Applied Microbiology 31 (4), 319-322 [IF 2.173]
- NJ Rowan, JG Anderson. 1998. Effectiveness of cleaning and disinfection procedures on the removal of enterotoxigenic Bacillus cereus from infant feeding bottles. Journal of food Protection 61 (2), 196-200 [IF 1.60]
- 19. NJ Rowan, JG Anderson, A Anderton. 1997. The bacteriological quality of hospitalprepared infant feeds. Journal of Hospital Infection 35 (4), 259-267 [IF 3.441]
- 20. NJ Rowan, JG Anderson, JE Smith. 1998. Potential infective and toxic microbiological hazards associated with the consumption of fermented foods. Microbiology of Fermented foods, 812-837. 13 citations.
- NJ Rowan, JG Anderson, JE Smith, JA Clarke, RC McLean, NJ Kelly. 1997. Development of a computer programme for the prediction and control of mould growth in buildings using the ESP-r modelling system. Indoor and Built Environment 6 (1), 4-11. 7 citations. [IF 1.9]

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,

56 Autor

Prof John Anderson



Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Neil Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

To whom it may concern,

I am a Lecturer in the Department of Life Science at Sligo Institute of Technology, Ireland. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, along with playing a key role in data analysis and in manuscript preparation, submission and proof-editing.

Examples of these collaborative efforts are listed below.

- 1. HP Farrell, M Garvey, M Cormican, JG Laffey, NJ Rowan. 2010. Investigation of critical inter-related factors affecting the efficacy of pulsed light for inactivating clinically relevant bacterial pathogens. Journal of Applied Microbiology 108 (5), 1494-1508 79 [IF 3.066]
- 2. M Garvey, H Farrell, M Cormican, N Rowan. 2010. Investigations of the relationship between use of in vitro cell culture-quantitative PCR and a mouse-based bioassay for evaluating critical factors affecting the disinfection ...Journal of Microbiological Methods 80 (3), 267-273. [IF 1.81]
- 3. J Hayes, D Kirf, M Garvey, N Rowan. 2013. Disinfection and toxicological assessments of pulsed UV and pulsed-plasma gas-discharge treated-water containing the waterborne protozoan enteroparasite Cryptosporidium parvum. Journal of Microbiological Methods. 94 (3), 325-337. [1.81]
- M Garvey, D Rabbitt, A Stocca, N Rowan. 2015. Pulsed ultraviolet light inactivation of Pseudomonas aeruginosa and Staphylococcus aureus biofilms. Water and Environment Journal 29 (1), 36-42 [IF 1.369]
- 5. P Haughton, M Garvey, NJ Rowan. 2010. Emergence of Bacillus cereus as a dominant organism in irish retailed powdered infant formulae (pif). Journal of Food Safety 30 (4), 814-831 [IF 1.133]
- JC Hayes, M Garvey, AM Fogarty, E Clifford, NJ Rowan. 2012. Inactivation of recalcitrant protozoan oocysts and bacterial endospores in drinking water using high-intensity pulsed UV light irradiation. Water Science and Technology: Water Supply 12 (4), 513-522
- 7. M Garvey, NJ Rowan. 2019. Pulsed UV as a potential surface sanitizer in food production processes to ensure consumer safety. Current Opinion in Food Science 26, 65-70 [IF 4.577]
- 8. M Garvey, N Rowan. 2015. A pulsed light system for the disinfection of flow through water in the presence of inorganic contaminants. Journal of Water and Health 13 (2), 406-412. [IF 1.349]



- 9. M Garvey, J Hayes, E Clifford, N Rowan. 2015. Ecotoxicological assessment of pulsed ultraviolet lighttreated water containing microbial species and Cryptosporidium parvum using a microbiotest test battery. Water and Environment Journal 29 (1), 27-35 [IF 1.369]
- M Garvey, A Stocca, N Rowan. 2014. Development of a combined in vitro cell culture–Quantitative PCR assay for evaluating the disinfection performance of pulsed light for treating the waterborne enteroparasite ... Experimental parasitology 144, 6-13. [IF 1.690]
- M Garvey, E Clifford, E O'Reilly, NJ Rowan. 2013. Efficacy of Using Harmless Bacillus Endospores to Estimate the Inactivation of Cryptosporidium parvum Oocysts in Water. The Journal of Parasitology 99 (3), 448-452 [IF 1.236]
- 12. J Naughton, EJ Tiedeken, M Garvey, JC Stout, NJ Rowan. 2017. Pulsed light inactivation of the bumble bee trypanosome parasite Crithidia bombi. Journal of Apicultural Research 56 (2), 144-154 [IF 1 .872]
- 13. M Garvey, N Thokala, N Rowan. 2014. A comparative study on the pulsed UV and the low-pressure UV inactivation of a range of microbial species in water. Water Environment Research 86 (12), 2317-2324 [IF 1.269]
- 14. M Garvey, JP Andrade Fernandes, N Rowan. 2015. Pulsed light for the inactivation of fungal biofilms of clinically important pathogenic Candida species. Yeast 32 (7), 533-540. [IF 3.143]
- 15. M Garvey, E Clifford, E O'Reilly, NJ Rowan. 2012. Efficacy of using harmless Bacillus endospores as novel surrogate organisms to indicate the inactivation performance of recalcitrant cryptosporidium parvus oocysts suspended in ...American Society of Parasitologists [IF 1.238]
- 16. M Garvey, A Stocca, N Rowan. 2016. Use of a real time PCR assay to assess the effect of pulsed light inactivation on bacterial cell membranes and associated cell viability. Water Environment Research 88 (2), 168-174 [IF 1.369]
- M Garvey, J Hayes, E Clifford, D Kirf, N Rowan. 2013. Efficacy of measuring cellular ATP levels to determine the inactivation of pulsed UV treated Cryptosporidium parvum oocysts suspended in water. Water Science and Technology: Water Supply 13 (2), 202-213 [IF 1.050]
- M Garvey, G Coughlan, N Murphy, N Rowan. 2016. The pulsed light inactivation of veterinary relevant microbial biofilms and the use of a RTPCR assay to detect parasite species within biofilm structures. Open Veterinary Journal 6 (1), 15-22 [IF 2.1]

If you require any further details, please do not hesitate to contact me at Garvey.mary@itsligo.ie.

Yours Sincerely,

De rlay Same

Dr Mary Garvey

Dr Mary Garvey MSc BSc Dept of Life Science Institute of Technology Sligo Ash Lane Sligo Ireland F91 YW50 www.itsligo.ie 27<sup>th</sup> September, 2024

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

**Re**: Professor Neil J Rowan, applicant for higher Doctorate of Science (D.Sc.) Degree on published work to the University of Strathclyde

To whom it may concern,

I am the Director for Microbiological Quality at Johnson and Johnson. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, along with playing a key role in data analysis and in manuscript preparation, submission and proof-editing.

Examples of these collaborative efforts are listed below.

Rowan, N.J., Kremer, T., McDonnell, G. (2023). A review of Spaulding's classification system for effective cleaning, disinfection and sterilization of reusable medical devices: Viewed through a modern-day lens that will inform and enable future sustainability. Science of the Total Environment, 878, https://doi.org/10.1016/j.scitotenv.2023.162976

Kremer, T., Rowan, N.J. and McDonnell, G. (2024). A proposed cleaning classification system for reusable medical devices to complement the Spaulding classification. *Journal of Hospital Infection*, **145**, 88-98.

Kremer, T.A., Felgar, J., Rowan, N.J. and McDonnell, G. (2023). Validation of the Device Feature Approach for Reusable Medical Device Cleaning Evaluations. *Biomedical Instrumentation and Technology*, https://doi.org/10.2345/0899-8205-57.4.143

Kremer, T., Murray, N., Buckley, J. and Rowan, N.J. (2023). Use of real-time immersive digital training and educational technologies to improve patient safety during the processing of reusable medical devices: Quo Vadis? *Science of the Total Environment*, **900**, <u>https://doi.org/10.1016/j.scitotenv.2023.165673</u>

If you require any further details, please do not hesitate to contact me.

Yours Sincerely,

1

Terra Kremer


## **To Whom It May Concern**

24<sup>th</sup> September 2024

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

## Re: Neil J. Rowan, applicant for Doctorate of Science (D.Sc.) on published work to the University of Strathclyde

I am an Associate Professor in the Institute of Biological Sciences at the Universiti Malaya, Malaysia. Prof Rowan and I have co-authored several articles on which he was a lead author. By this I mean that he contributed to the conception, design and writing of the work, data analysis and in manuscript preparation, submission and proof-editing.

- Wan Mohtar, W.A.A.Q.I, IIham, Z., Jamaludin, A.A and Rowan, N.J (2021). Use of Zebrafish Embryo Assay to Evaluate Toxicity and Safety of Bioreactor-Grown Exopolysaccharides and Endopolysaccharides from European *Ganoderma applanatum* Mycelium for Future Aquaculture Applications. International Journal of Molecular Sciences, 22 (4), <u>https://doi.org/10.3390/ijms22041675</u>
- Wan Mohtar, W.A.A.Q.I., Fafek, N.A., Thiran, J.P., Rahman, J.F.P., Yerima, G., Subramaniam, K. and **Rowan, N.J.** (2021). Investigations on the use of exopolysaccharide derived from mycelial extract of *Ganoderma lucidum* as functional feed ingredient for aquaculture-farmed red hybrid Tilapia (Oreochromis sp.). Future Foods, 3, <u>https://doi.org/10.1016/j.fufo.2021.100018</u>
- Usuldin, S.R.A., Wan-Mohtar, W.A.A.Q.I., IIham, Z., Abdullah, N.R. and Rowan, N. (2021). In vivo toxicity of bioreactor-grown biomass and exopolysaccharides from Malaysian tiger milk mushroom mycelium for potential future health applications. Nature Scientific Reports, 11, <u>https://doi.org/10.1038/s41598-021-02486-7</u>

Your sincerely,

Official stamp:



**Associate Professor Ts. Dr. Wan Abd Al Qadr Imad Wan Mohtar** Biotechnology Program, Institute of Biological Sciences, Faculty of Science Universiti Malaya, 50603, Kuala Lumpur, Malaysia Email: qadyr@um.edu.my

INSTITUT SAINS BIOLOGI Fakulti Sains, Universiti Malaya, 50603 Kuala Lumpur, MALAYSIA

Tel: (603) 7967 4208 / 4389 (Pentadbiran), (603) 7967 4209 (HEP+ Ijazah Dasar), (603) 7967 4190 (HEP + Ijazah Tinggi) Tel: (603) 79674118 (Pejabat Ketua) • Faks: (603) 79674178 • Laman Web: http://biology.um.edu.my



Energy Systems Research Unit Mechanical & Aerospace Engineering James Weir Building 75 Montrose Street Glasgow G1 1XJ Scotland, UK



21 April 2021

Chair, Higher Doctorate Committee On behalf of Senate, University of Strathclyde

To Whom it May Concern

## Neil Rowan: DSc application, University of Strathclyde

I write in connection with Professor Rowan's DSc application. I collaborated with Neil on several research projects over the period 1996-2000 and co-authored, with him and others, publications in which he was a lead or senior author:

- Rowan N J, Johnstone C M, McLean R C, Anderson J G and Clarke J A (1999) 'Prediction of toxigenic fungal growth in buildings by using a novel modelling system', Applied and Environmental Microbiology, 65 (11), pp. 4814-4821. [IF 4.016]
- Clarke J A, Johnstone C M, Kelly N J, McLean R C, Rowan N J and Smith J E (1999) 'A technique for the prediction of the conditions leading to mould growth in buildings', Building and Environment, 34 (4), pp. 515-521. [IF 4.971]
- Rowan N J, Anderson J G, Smith J E, Clarke J A, McLean R C and Kelly N J (1997) 'Development of a computer program for the prediction and control of mould growth in buildings using the ESP-r modelling system', Indoor and Built Environment, 6 (1), pp. 4-11. [IF 1.9]

Throughout our collaboration, Neil took the lead in the bioscience aspects of the research and delivered world leading outcomes: his contributions to the conception and execution of the underlying research was substantial.

Please do not hesitate to contact me should you require any further details.

Yours sincerely,

Professor Emeritus J A Clarke